

DIVISION OF ENVIRONMENTAL QUALITY

SECRETARY

November 18, 2024

Via email to: philipantici@ffcmail.com & First Class Mail

Philip Antici HSES Manager FutureFuel Chemical Company PO Box 2357 Batesville, AR 72503

Re: Notice of Final Permitting Decision; Permit No. 1085-AOP-R17

Dear Mr. Antici,

After considering the application and other applicable materials as required by APC&EC Rule 8.211 and Ark. Code Ann. § 8-4-101 *et seq.*, this notice of final permitting decision is provided for:

FutureFuel Chemical Company 2800 Gap Road Batesville, AR 72501

Permit Number: 1085-AOP-R17

Permitting Decision: approval with permit conditions as set forth in final Permit No. 1085-AOP-R17

Accessing the Permitting Decision: https://www.adeq.state.ar.us/downloads/WebDatabases/PermitsOnline/Air/1085-AOP-R17.pdf.

Accessing the Statement of Basis: https://www.adeq.state.ar.us/downloads/WebDatabases/PermitsOnline/Air/1085-AOP-R17-SOB.pdf. Rule 26.903 of the Rules of the Arkansas Operating Air Permit Program do not require a public notice or public comment period for Administrative Amendments.

Sincerely,

Demetchbrough

Demetria Kimbrough Associate Director, Office of Air Quality, Division of Environmental Quality 5301 Northshore Drive, North Little Rock, AR 72118-5317

Enclosure: Certificate of Service cc: lynncornelius@ffcmail.com

CERTIFICATE OF SERVICE

I, Natasha Oates, hereby certify that the final permit decision notice has been mailed by first class mail to FutureFuel Chemical Company, PO Box 2357, Batesville, AR, 72503, on this 18th day of November, 2024.

Natasha Oatis

Natasha Oates, AA, Office of Air Quality

RESPONSE TO COMMENTS

FUTUREFUEL CHEMICAL COMPANY PERMIT #1085-AOP-R17 AFIN: 32-00036

On September 15, 2024, the Director of the Arkansas Department of Energy and Environment, Division of Environmental Quality ("Division") gave notice of a draft permitting decision for the above referenced facility. On October 21, 2024, written comments on the draft permitting decision were submitted by Lynn Cornelius of FutureFuel. The Division's response to these issues follows.

Note: The following page numbers and condition numbers refer to the draft permit. These references may have changed in the final permit based on changes made during the comment period.

Comment #1: Page 30- FFCC suggests to remove condition OCI 9. It is redundant with conditions OCI 7, 8, and MON 10. Plantwide condition #33 only applies to CAM sources, of which, there are none in the OCI Section.

Response to Comment #1: The Condition OCI 9 is necessary and not redundant. Based on information in the permit application, SN-5N09-01, 5N09-02, and 5N09-03 are subject to CAM. The condition contains the CAM requirements for those sources. The supplemental CAM information provided by the facility states that sources 5N09-01, 5N09-02, and 5N09-03 are exempt due to being subject to a post 1990 federal regulation, 40 C.F.R. Part 63, Subpart FFFF. However, in order for that exemption to apply, the federal regulation must contain emission standards for the specific pollutant that would otherwise be subject to CAM. Subpart FFFF contains standards for HAP, not for VOC. These sources have the potential to emit VOC greater than 100 percent of the amount required for a source to be classified as a major source; therefore, the sources are subject to CAM. However, the compliance mechanisms already established by the permit are adequate to satisfy CAM requirements. OCI 9 was added with a regulatory citation to the CAM rule to specify the CAM requirements, which were already required by other conditions, and to require records for CAM. No changes to the permit were made.

Comment #2: Page 91- FFCC suggests to remove condition US 18. It is redundant with conditions US 15, 16 and MON 10. Plantwide condition #33 only applies to CAM sources, of which, there are none in the US Section.

Response to Comment #2: No changes to the permit were made. The reasoning being the same as in Response to Comment #1 above for condition US 18 as it was for OCI 9. While SN-6M01-01 is subject to requirements in 40 C.F.R. Part 63, Subpart FFFF, this regulation does not contain standards specific to PM_{10} and the potential to emit is greater than 100 percent of the amount required for a source to be classified as a major source; therefore, the sources are subject to CAM.

Comment #3: Page 164- FFCC suggests to remove condition IB 6. It is redundant with conditions IB 4, 5 and MON 10. Plantwide condition #33 only applies to CAM sources, of which, there are none in the IB Section.

Response to Comment #3: No changes to the permit were made. The reasoning being the same as in Response to Comment #1 above for condition IB 6 as it was for OCI 9. While SN-5N03-54 is subject to requirements in 40 C.F.R. Part 63, Subpart FFFF, this regulation does not contain standards specific to VOC and the potential to emit is greater than 100 percent of the amount required for a source to be classified as a major source; therefore, the sources are subject to CAM.

Comment #4: Page 205, EG 1, in the table the tpy for SO₂ and CO for 6N02-EG did not get changed. It should be SO₂ 0.6 tpy and CO 0.4 tpy in the table.

Response to Comment #4: The requested changes were made.



DIVISION OF ENVIRONMENTAL QUALITY

OPERATING AIR PERMIT

PERMIT NUMBER: 1085-AOP-R17

IS ISSUED TO: FutureFuel Chemical Company 2800 Gap Road Batesville, AR 72501 Independence County AFIN: 32-00036

PURSUANT TO THE RULES OF THE ARKANSAS OPERATING AIR PERMIT PROGRAM, RULE 26: THIS PERMIT AUTHORIZES THE ABOVE REFERENCED PERMITTEE TO INSTALL, OPERATE, AND MAINTAIN THE EQUIPMENT AND EMISSION UNITS DESCRIBED IN THE PERMIT APPLICATION AND ON THE FOLLOWING PAGES. THIS PERMIT IS VALID BETWEEN:

November 18, 2024 AND November 17, 2029

THE PERMITTEE IS SUBJECT TO ALL LIMITS AND CONDITIONS CONTAINED HEREIN.

Signed:

etchbrough-

Demetria Kimbrough Associate Director, Office of Air Quality Division of Environmental Quality

November 18, 2024

Date

Table of Contents

SECTION I: FACILITY INFORMATION5SECTION II: INTRODUCTION6Summary of Permit Activity6Process Description6Rules and Regulations9Emission Summary10SECTION III: PERMIT HISTORY18SECTION IV: SPECIFIC CONDITIONS28Organic Chemical Intermediates2840 CFR 63 Subpart GGG - National Emission Standards for Pharmaceutical Production30
Ingredient Production
Utilities
from Off-Site Waste and Recovery Operations
 40 CFR Part 60 Subpart Y - Standards of Performance for Coal Preparation Plants
40 CFR Part 63 Subpart DDDDD—National Emission Standards for Hazardous Air Pollutants for Major Sources: Industrial, Commercial, and Institutional Boilers and Process Heaters
Organic Sulfonation Process
Chemical Waste Destructor
Solvent Recovery157Wastewater Treatment Facility159Isopropyl Benzene Production (DIPB)16140 CFR Part 61 Subpart FF - NESHAP for Benzene Waste Operations162
40 CFR Part 61 Subpart V - National Emission Standard for Equipment Leaks (Fugitive Emission Sources)
40 CFR Part 61 Subpart J – National Emission Standard for Equipment Leaks (Fugitive Emission Sources of Benzene
40 CFR Part 61 Subpart Y - National Emission Standard for Benzene Storage Vessels 188

Storage Tanks and Miscellaneous Sources 5N07 Production Facility	
Aldehyde Processing Section	
40 CFR Part 63 Subpart DDDDD—National Emission Standards for Hazardous	Air
Pollutants for Major Sources: Industrial, Commercial, and Institutional Boilers and	nd Process
Heaters	196
Anode Material Process	
Emergency Generators	
Plantwide Sources Subject to MON	
SECTION V: COMPLIANCE PLAN AND SCHEDULE	
SECTION VI: PLANTWIDE CONDITIONS	
Acid Rain (Title IV)	
Title VI Provisions	
SECTION VII: INSIGNIFICANT ACTIVITIES	
SECTION VIII: GENERAL PROVISIONS	
Appendix A – CEMS Requirements	
Appendix B - 40 CFR Part 63 Subpart MMM	
Appendix C – 40 CFR Part 60 Subpart Kb	
Appendix D – 40 CFR Part 60 Subpart Y	
Appendix E – 40 CFR Part 60 Subpart VV	
Appendix F – 40 CFR Part 60 Subpart NNN	
Appendix G – 40 CFR Part 61 Subpart J	
Appendix H – 40 CFR Part 61 Subpart V	
Appendix I – 40 CFR Part 61 Subpart Y	
Appendix J – 40 CFR Part 61 Subpart FF	
Appendix K – 40 CFR Part 63 Subpart DD	
Appendix L – 40 CFR Part 63 Subpart EEE (Phase I and II)	
Appendix M – 40 CFR Part 63 Subpart FFFF	
Appendix N – Notification of Compliance (NOC) – Boilers and Incinerator	
Appendix O – 40 CFR 63 Subpart ZZZZ	
Appendix P – 40 CFR 63 Subpart DDDDD	
Appendix Q – 40 CFR 60 Subpart JJJJ	
Appendix R – 40 CFR Part 63 Subpart GGG	
Appendix S – 40 CFR Part 60 Subpart Db	
Appendix T – 40 CFR Part 61 Subpart E	

List of Acronyms and Abbreviations

Ark. Code Ann.	Arkansas Code Annotated
AFIN	Arkansas DEQ Facility Identification Number
C.F.R.	Code of Federal Regulations
СО	Carbon Monoxide
COMS	Continuous Opacity Monitoring System
НАР	Hazardous Air Pollutant
Нр	Horsepower
lb/hr	Pound Per Hour
NESHAP	National Emission Standards (for) Hazardous Air Pollutants
MVAC	Motor Vehicle Air Conditioner
No.	Number
NO _x	Nitrogen Oxide
NSPS	New Source Performance Standards
PM	Particulate Matter
PM ₁₀	Particulate Matter Equal To Or Smaller Than Ten Microns
PM _{2.5}	Particulate Matter Equal To Or Smaller Than 2.5 Microns
SNAP	Significant New Alternatives Program (SNAP)
SO_2	Sulfur Dioxide
SSM	Startup, Shutdown, and Malfunction Plan
Тру	Tons Per Year
UTM	Universal Transverse Mercator
VOC	Volatile Organic Compound

SECTION I: FACILITY INFORMATION

PERMITTEE:	FutureFuel Chemical Company
AFIN:	32-00036
PERMIT NUMBER:	1085-AOP-R17
FACILITY ADDRESS:	2800 Gap Road Batesville, AR 72501
MAILING ADDRESS:	PO Box 2357 Batesville, AR 72503
COUNTY:	Independence County
CONTACT NAME:	Philip Antici
CONTACT POSITION:	HSES Manager
TELEPHONE NUMBER:	(870) 698-5358
REVIEWING ENGINEER:	Shawn Hutchings
UTM North South (Y):	Zone 15: 3954180.53 m
UTM East West (X):	Zone 15: 633409.13 m

SECTION II: INTRODUCTION

Summary of Permit Activity

FutureFuel Chemical Company, located in Batesville, Arkansas, is a supplier of specialty organic chemical intermediates used in the manufacture of color film and photographic paper, paints and coatings, plastics and bottle polymers, medical supplies, prescription medicines, food supplements, household detergents, agricultural products, and biofuel. This permit is the Title V renewal for the facility and updates the permit conditions for the MON requirements due to regulation changes. The method of compliance using the CEMS for the CO lb/hr limit for SN-6M03-05 was specified to allow for exclusion of startup and shutdown emissions, compliance during startup and shutdown will be shown by other means. VOC and Organic Pollutant emission rates dropped by 0.3 tpy. There was no change in emissions only corrections to the total. All other pollutants remain unchanged.

Process Description

Organic Chemical Intermediates Section (OCI)

FutureFuel's batch organic chemical intermediates facilities are located in Buildings 5N01, 5N03, and 5N07. These production buildings contain multi-purpose/product equipment which may produce a variety of chemicals. The contained or captured vapors from the equipment in the three batch production buildings are vented through a collection system to the RTO units via a common duct. Volatile organic compounds (VOCs) are destroyed by combustion. Fugitive emissions from organic chemical intermediates are designated as source number OCI-FUG. Two other TOs (SN-5N09-02 and SN-5N09-03) are also present, but in conjunction with a caustic scrubber. Fugitive emissions are designated as source OCI-FUG.

Utilities (US)

There are three coal fired (6M01-01) and two natural gas fired boilers (6M06-01 and 6M07-01) at the facility. The coal fired boilers are balanced draft, coal-fired steam generation boilers that have been fitted with atomizing nozzles to facilitate burning of liquid chemical wastes. Each coal fired boiler system is designed as a 70 million Btu/hr unit and is equipped with its own electrostatic precipitator (ESP) to control particulate emissions. The spent solvent from the 2,000 gallon liquid process tank to be used for the purpose of flushing the chemical distribution piping is routed to either the coal-fired boiler auxiliary waste chemical burners or to the burner of the chemical waste destructor. Emissions from tank venting are collected and routed to the coal-fired boilers (6M01-01). There are also two natural gas fired boilers at the facility. The #4 boiler (6M06-01) burns natural gas at 78 million BTU/hr. The #5 boiler (6M07-01) burns natural gas at 221 million BTU/hr. These natural gas burning sources do not have control devices. Utility fugitive emissions are accounted for in the BLR-FUG source.

Organic Sulfonation (OSP)

The organic sulfonate facility produces a solid material for use as a household consumer product. The two organic sulfonation facilities include reactors, centrifuges, scrubbers, distillation equipment, raw materials and process tanks. Scrubbers are the primary means for controlling emissions from the production facilities. The low vapor pressures of the contents of the storage tanks minimize the potential for VOC emissions from these emission points, but there are fugitive emissions taken into account from the processes. Equipment from this section is typically notated with 5M in the source name. The 5MNOBS-TNK source is a bubble consisting of four tanks. VOC fugitive emissions are accounted for in NOBS-FUG, while particulate fugitives are accounted for within 5M01-TSP.

Chemical Waste Destructor (CWD)

The chemical waste destructor at FutureFuel Chemical Company is designed to burn a mixture of waste streams resulting from various fine chemical manufacturing facilities at the plant. Some of the waste is mainly organic solvents, but the majority is comprised of aqueous solutions containing some organic and salt compounds. The equipment used to burn the waste includes a burner assembly, oxidizer chamber, weir tank, quench separator tank, high-energy scrubber, vane separator, and a stack. The chemical destructor is a vertically down fired unit. Emissions were calculated for the incinerator (6M03-05) and for fugitive equipment leaks (DEST-FUG). The chemical waste destructor is subject to 40 CFR Part 63, Subpart EEE, *National Emission Standards for Hazardous Air Pollutants from Hazardous Waste Combustors*.

Solvent Recovery (SR)

Solvent Recovery equipment is designed to recover solvents that become contaminated during the manufacturing processes. Individual streams from the chemical manufacturing processes are transferred to storage tanks in the solvent recovery area. These streams are pumped to a pH adjustment system and then to a series of distillation columns. After distillation, the solvents are reused in the manufacturing processes or are sold for other uses. Process emissions from the Solvent Recovery Facility are controlled with Regenerative Thermal Oxidizers (RTOs), 5N09-01 and designated as 4PSR-00. Fugitives from this section are SR-FUG.

Wastewater Treatment Facility (WWT)

The Wastewater Treatment Plant (WWT) treats process wastewater from various areas of the plant, sanitary sewage, and some storm water. The wastewater treatment system is an extended aeration activated sludge design consisting of equalization and neutralization, aeration, and clarification. The excess biomass is aerobically digested and either land applied on-site via a spray irrigation system, dewatered and burned in the coal-fired boilers, or landfilled. A diversion tank is used to reduce organic or hydraulic peaks in the untreated wastewater. Sources in this section include WWT Facilities (7K01-01), a Decant Tank (7M01-02), two pH control tanks (7M01-03 and 7M01-03B), and a WWT chemical tank (7M01-04).

Isopropyl Benzene Section (IB)

Future Fuel Chemical Companies (FFCC) Isopropyl Benzene Production facilities are located in building 5N03 and in a large distillation complex located east of 5N03. The isopropyl benzene process consists of alkylation of benzene with propylene. A catalyst is used to promote the reaction. 5N03-48 and 5N03-55 are scrubbers associated with the DIPB process and vent to the atmosphere. 5N03-48, is a water scrubber that is located on the vent discharge of an Aluminum Chloride baghouse. 5N03-55 is a water scrubber that receives vapor off of a tank containing an Aluminum Chloride solution. Additional sources in this section are a large tank bubble (10 tanks, 5NDIPB-TNK), a Flare (5N03-54), and Fugitive Emissions from Isopropyl Benzene Process (DIPB-FUG).

5N07 Production Facility (BD)

The 5N07 production facility contains multi-purpose production equipment which may produce a variety of chemicals including biofuel. Biodiesel is the primary product from this facility. The contained or captured vapors from the equipment in this facility are vented through a collection system to the RTO (SN 5N09-01) units via a common duct. Two methanol tanks (T-242 and T-243) are in this area. Emissions from these tanks are accounted for in tank bubble 5N03TK-01. All emissions from Biodiesel Refining and Methanol Recovery, Material Storage, Glycerol Refining, and Fugitive Emissions are part of the 5N07 emission source.

Storage Tanks and Miscellaneous Sources (STMS)

Many different chemicals can be manufactured; therefore many tanks are needed for raw material, intermediate, and product storage. Emissions from storage tanks that vent to the atmosphere and other miscellaneous sources are identified in the STMS section of the permit. With the exception of a fume scrubber TF-5FS on TF-5, there are no specific controls on these tanks other than conservation vents.

Sources in this section include a Diesel Tank (6N01-02), a Gasoline Tank (6N01-03), and several process tanks, 5N03TK-01.

Aldehyde Processing Section (AP)

Raw materials are unloaded into storage tanks T-271, T-272, and TFS-75. TFS-75 vents to the RTO. Raw materials are transferred to the process as needed. Both columns and tanks vent to the RTOs (5N09-01).

The aldehyde process reactors are periodically cleaned and the vent gas is routed to a water scrubber, SV-03. The vent stream from this source discharges to the RTO. The water stream is routed to the wastewater treatment facilities. The reactors are heated with two 5 MMBTU/hr hot oil system, SN 4P05-01 and SN 4P05-03. The hot oil system SN 4P05-01 is designed to burn natural gas, fuel oil, biodiesel, and process vent streams. Hot oil system SN 4P05-03 is designed to burn natural gas and process vent streams.

Products are refined in distillation column SB-02 and extraction column SX-03. The distillation column and the extraction column both vent back to the hot oil system or the RTOs. The refined product is transferred to three lot tanks, VC-PT-01, VC-PT-02, and VC-PT-03. These tanks are equipped with vapor balancing and do not vent to the atmosphere. VC-ST-01 shutdown tank vents to the hot oil system for thermal recovery and is not an emission source itself. Tank WDT-03, a waste tank, is located in the Utilities section and is connected to the coal-fired boiler closed vent system and control device. VC-PT-03, VC-PT-01, and VC-PT-02 are all subject to NSPS Kb.

The emission points for the Aldehyde Processing Section are the hot oil system (4P05-01), hot oil system (4P05-03), equipment fugitives (4PSR-FUG), and the RTO (5N09-01).

Anode Material Process (CP2)

The Anode Material Process (CP2), permitted as CP-2, consists of solids handling equipment, continuous stirred tank reactors, and dryers. Pelletized and granular material will be fed through a metering system into vessels where the material is heated and mixed. The formulated material is then dried and repackaged. All of this equipment is located inside the 5M11 building. Dust from all of the solids handling equipment is vented to a continuous dust control system (CDCS), 5M11-08. The CDCS consists of a baghouse for solid particle separation, a collection hopper, and an induced draft fan. A central vacuum cleaning system (CVC), 5M11-09 will be used to clean spills. All VOC vents will be routed to the RTOs, 5N09-01. Emissions from tanks and equipment containing VOC vent to the thermal oxidizer SN-5N09-01.

Rules and Regulations

The following table contains the rules and regulations applicable to this permit.

Rules and Regulations
Arkansas Air Pollution Control Code, Rule 18, effective March 14, 2016
Rules of the Arkansas Plan of Implementation for Air Pollution Control, Rule 19, effective May 6, 2022
Rules of the Arkansas Operating Air Permit Program, Rule 26, effective March 14, 2016
40 CFR 52.21- Prevention of Significant Deterioration of Air Quality (PSD).
40 CFR Part 63 Subpart GGG - National Emission Standards Pharmaceuticals Production
40 CFR Part 63 Subpart MMM - National Emission Standards for Hazardous Air Pollutants for Pesticide Active Ingredient Production
40 CFR Part 60 Subpart Kb - Standards of Performance for Volatile Organic Liquid Storage Vessels (Including Petroleum Liquid Storage Vessels) for Which Construction, Reconstruction, or Modification Commenced after July 23, 1984
40 CFR Part 60 Subpart Y- Standards of Performance for Coal Preparation Plants

Rules	and	Regulations	
ivaics	anu	Regulations	

40 CFR Part 60 Subpart VV - Standards of Performance for Equipment Leaks of VOC in the Synthetic Organic Chemicals Manufacturing Industry

40 CFR Part 60 Subpart NNN - Standards of Performance for Volatile Organic Compound (VOC) Emissions From Synthetic Organic Chemical Manufacturing Industry (SOCMI) Distillation Operations

40 CFR Part 61 Subpart E - National Emission Standards for Equipment Mercury

40 CFR Part 61 Subpart J - National Emission Standards for Equipment Leaks (Fugitive Emission Sources) of Benzene

40 CFR Part 61 Subpart V - National Emission Standards for Equipment Leaks (Fugitive Emission Sources)

40 CFR Part 61 Subpart Y - National Emission Standards for Benzene Emissions from Benzene Storage Vessels

40 CFR Part 61 Subpart FF - National Emission Standard for Benzene Waste Operations

40 CFR Part 63 Subpart DD - National Emission Standards for Hazardous Air Pollutants from Off-Site Waste and Recovery Operations

40 CFR Part 63 Subpart EEE (Phase I and II) - National Emission Standard for Hazardous Air Pollutants from Hazardous Waste Combustors

40 CFR Part 63 Subpart FFFF - National Emission Standard for Hazardous Air Pollutants: Miscellaneous Organic Chemical Manufacturing

40 CFR Part 60 Subpart Db - Standards of Performance for Industrial-Commercial-Institutional Steam Generating Units

40 CFR Part 63 Subpart ZZZZ - National Emissions Standards for Hazardous Air Pollutants for Stationary Reciprocating Internal Combustion Engines

40 CFR Part 63 Subpart DDDDD - National Emission Standards for Hazardous Air Pollutants for Major Sources: Industrial, Commercial, and Institutional Boilers and Process Heaters

40 CFR Part 60 Subpart JJJJ – Standards of Performance for Stationary Spark Ignition Internal Combustion Engines

Emission Summary

The following table is a summary of emissions from the facility. This table, in itself, is not an enforceable condition of the permit.

EMISSION SUMMARY				
Source	Decorintian	Pollutant	Emission Rates	
Number	Description		lb/hr	tpy
		PM	56.7	178.7
Total Allowable Emissions		PM_{10}	56.7	178.7
		PM _{2.5}	See Note*	

EMISSION SUMMARY				
Source			Emission Rates	
Number	Description	Pollutant	lb/hr	tpy
	·	SO_2	1447.0	6144.4
		VOC	318.9	491.0
		СО	309.8	1224.6
		NO _X	246.7	875.4
		Pb	0.9	3.5
	LLA D _c	Inorganics*	273.2	1092.9
	HAPS	Organic Pollutants**	318.8	491.3
		PM	9.9	17.5
		PM_{10}	9.9	17.5
	Decompositive Themas	SO_2	8.4	36.8
5NI00 01	Regenerative Thermal	VOC	43.6	191.0
JIN09-01	(2 Unita)	СО	11.0	25.2
	(2 Offics)	NO _x	8.7	38.1
		Inorganics*	10.0	43.8
		Organic Pollutants**	43.6	191.0
	Thermal Oxidizer and Caustic Scrubber	PM	0.1	0.1
		PM_{10}	0.1	0.1
		SO_2	3.0	13.1
5NI00 02		VOC	2.9	12.7
51109-02		СО	0.5	2.2
		NO _x	3.0	13.1
		Inorganics*	1.0	4.4
		Organic Pollutants**	2.9	12.7
		PM	0.1	0.2
		PM_{10}	0.1	0.2
		SO_2	3.0	13.1
5NI00 03	Thermal Oxidizer and	VOC	2.9	12.7
51109-05	Caustic Scrubber	СО	0.5	2.2
		NO _x	3.0	13.1
		Inorganics*	1.0	4.4
		Organic Pollutants**	2.9	12.7
OCI-FUG	Fugitives	VOC	3.7	15.9
001-100	i ugitives	Organic Pollutants**	3.7	15.9
		Utilities		
6M01	Coal Dila	PM	0.1	0.1
01/101		PM10	0.1	0.1
	2 Coal Fired Deilars (70	PM	26.1	112.8
6M01-01	MMRtu/br anch)	PM_{10}	26.1	112.8
-	MMBtu/hr each)	SO_2	1391.0	5982.9

EMISSION SUMMARY				
Source	Description	2.11	Emission Rates	
Number		Pollutant	lb/hr	tpy
		VOC	0.7	2.9
		СО	236.3	1034.0
		NO _x	106.0	488.2
		Pb	0.8	3.4
		Inorganics*	259.0	1031.6
		Organic Pollutants**	0.7	2.9
	Coal Bunker Fabric	PM	0.2	0.7
6M01-01A	Filter	PM_{10}	0.2	0.7
		PM	0.6	2.6
		PM_{10}	0.6	2.6
	#4 Boiler (78	SO_2	0.1	0.3
6M06-01	MMBtu/hr) Natural	VOC	0.5	1.9
	Gas	СО	6.6	28.7
		NO _x	13.3	58.3
		Organic Pollutants**	0.5	1.9
	#5 Boiler (221 MMBtu/hr) Natural Gas	PM	1.7	7.4
		PM_{10}	1.7	7.4
		SO_2	0.2	0.6
6M07-01		VOC	1.3	5.4
		СО	18.6	81.4
		NO _x	22.0	96.4
		Organic Pollutants**	1.3	5.4
	Utilities Area Fugitive	VOC	0.5	1.8
BLK-FUG	Emissions	Organic Pollutants**	0.5	1.8
	Organ	ic Sulfonation		
51401 01	S a mala la an	VOC	0.1	0.4
31VI01-01	Scrubber	Organic Pollutants**	0.1	0.4
51401 02	S a mala la an	VOC	0.1	0.4
51101-02	Scrubber	Organic Pollutants**	0.1	0.4
5M01.05	Samuhhan	VOC	0.1	0.4
51101-05	Scrubber	Organic Pollutants**	0.1	0.4
5M01.06	Scrubber	VOC	0.5	1.8
5101-00	Scrubber	Organic Pollutants**	0.5	1.8
5M01-07	Scrubber	VOC	0.1	0.4
5101-07	Scrubber	Organic Pollutants**	0.1	0.4
5101.09	Sombhor	VOC	0.1	0.4
5101-00	Scrubber	Organic Pollutants**	0.1	0.4
5M01.00	Completer	VOC	0.2	0.9
51101-07	Scrubber	Organic Pollutants**	0.2	0.9

EMISSION SUMMARY				
Source			Emission Rates	
Number	Description	Pollutant	lb/hr	tpy
5M03-01	Scrubber	VOC	0.1	0.4
510105 01	Seruoser	Organic Pollutants**	0.1	0.4
5M03-02	Scrubber	VOC	0.2	0.8
		Organic Pollutants**	0.2	0.8
5M04-01	Scrubber	VOC	0.6	2.3
		Organic Pollutants**	0.6	2.3
5M04-02	Scrubber	VOC	0.2	0.7
		Organic Pollutants**	0.2	0.7
5M04-10	Scrubber	SO_2	0.1	0.4
51405-01	Samuhhan	VOC	0.1	0.4
51/103-01	Scrubber	Organic Pollutants**	0.1	0.4
51405 02	Sombhor	VOC	0.1	0.4
51105-02	Schubber	Organic Pollutants**	0.1	0.4
5M11.01	Samuhhan	VOC	0.1	0.4
51111-01	Schubber	Organic Pollutants**	0.1	0.4
5M11.04	Samuhhan	VOC	0.1	0.4
JIVI11-04	Scrubber	Organic Pollutants**	0.1	0.4
51/11 05	Samila	VOC	0.1	0.4
51/11-05 501	Schubber	Organic Pollutants**	0.1	0.4
5M 11 06	Scrubber	VOC	0.1	0.4
51/111-00	Scrubber	Organic Pollutants**	0.1	0.4
5M11.07	Scrubber	VOC	0.1	0.4
510111-07	Scrubber	Organic Pollutants**	0.1	0.4
	SPS Supersack Load	РМ	0.1	03
5M11-15	Hopper Dust Control	PM ₁₀	0.1	0.3
	System	1 10110	0.1	0.5
5M13-01	Scrubber	VOC	0.1	0.4
		Organic Pollutants**	0.1	0.4
5M16-01	Supersack Loadout	PM	0.1	0.4
	Dust Control System	PM ₁₀	0.1	0.4
5M18-01	Continuous Dust	PM	3.9	17.1
	Control System	PM ₁₀	3.9	17.1
5M18-02	Central Vacuum	PM	3.4	3.7
	Cleaning System	PM ₁₀	3.4	3.7
5M18-03	Bin Vacuum Cleaning	PM	0.3	0.9
	System	PM ₁₀	0.3	0.9
5MNOBS-	Aggregate Tank (5	VOC	0.6	2.6
TNK	tanks)	Organic Pollutants**	0.6	2.6

EMISSION SUMMARY				
Source	D			Rates
Number	Description	Pollutant	lb/hr	tpy
NOBS-FUG	Fugitive Emissions from Organic Sulfonation Process	VOC Organic Pollutants**	1.2 1.2	5.3 5.3
5M01-TSP	Dust Control Maintenance Eugitives	PM PM10	3.1	0.2
	Chemi	cal Destructor	5.1	0.2
			1.5	6.6
	Chemical Waste	PM_{10} SO_{2} VOC	1.5 20.4 1.0	6.6 89.0 4.0
6M03-05	Destructor (50 MMBtu/hr)	CO NO _x Pb Inorganics*	11.2 33.2 0.1 1.4	27.0 146.0 0.1 6.1
		Organic Pollutants**	1.0	4.0
DEST-FUG	Destructor Fugitives	VOC Organic Pollutants**	1.0 1.0	3.0 3.0
	Solve	ent Recovery		I
4PSR-00	Solvent Recovery Facility	VOC Organic Pollutants**	4.0 4.0	17.0 17.0
SR-FUG	Solvent Recovery Fugitive Emissions	VOC Organic Pollutants**	5.0 5.0	21.9 21.9
	Waste W	ater Treatment		
7K01-01	WWT Facilities	VOC Organic Pollutants**	28.6 28.6	125.3 125.3
7M01-02	Decant Tank	VOC Organic Pollutants**	0.8 0.8	3.5 3.5
7M01-03	pH Control Tank	Inorganics*	0.1	0.4
7М01-03-В	pH Control Tank	Inorganics*	0.1	0.1
7M01-04	WWT Container	VOC Organic Pollutants**	0.1 0.1	0.4 0.4
	Isopr	opyl Benzene		
5NDIPB-TNK	Tank Bubble (10 tanks)	VOC Organic Pollutants** Inorganic*	0.2 0.2 0.1	0.6 0.6 0.1
5N03-48	Scrubber	Inorganics*	0.1	0.4

EMISSION SUMMARY				
Source	D		Emission Rates	
Number	Description	Pollutant	lb/hr	tpy
		PM	0.1	0.4
		PM_{10}	0.1	0.4
		SO_2	0.5	1.9
5N03-54	Flare	VOC	0.9	3.9
		СО	2.4	9.8
		NO _x	0.5	1.9
		Organic Pollutants**	0.9	3.9
5N03-55	Scrubber	Inorganics*	0.1	0.4
	Fugitive Emissions	VOC	0.2	1.2
DIPB-FUG	from Isopropyl Benzene	VOC Organia Ballutanta**	0.3	1.5
	Process	Organic Pollutants	0.3	1.3
	Storage Tanks an	d Miscellaneous Source	28	
6NI01 02	Diago1 Tomly	VOC	0.1	0.4
01001-02	Diesei Talik	Organic Pollutants**	0.1	0.4
CNI01 02	Cos Torit	VOC	200.0	0.8
0101-05	Gas Tank	Organic Pollutants**	200.0	0.8
	Process Tanks	VOC	4.0	17.5
5N03TK-01		Organic Pollutants**	4.0	17.5
	(30 Taliks)	Inorganics*	0.3	1.2
	5N07 Pro	duction Facility		
51107	Diadiagal Draduation	VOC	2.9	12.4
51NU7	Biodiesel Production	Organic Pollutants**	2.9	12.4
	Aldehyde l	Processing Section		
		РМ	0.2	0.9
		PM_{10}	0.2	0.9
		SO_2	1.1	4.9
4P05-01	Hot Oil System	VOC	4.0	6.8
		СО	1.5	6.4
		NO _x	3.2	13.8
		Organic Pollutants**	4.0	6.8
		PM	0.2	0.6
		PM_{10}	0.2	0.6
		SO_2	0.1	0.1
4P05-03	Hot Oil System #2	VOC	0.8	3.4
		СО	1.5	6.4
		NO _x	0.9	3.8
		Organic Pollutants**	0.8	3.4

EMISSION SUMMARY					
Source Number	Description	Pollutant	Emission Rates		
			lb/hr	tpy	
ADSD FUG	Aldehyde Processing	VOC	0.6	2.5	
4151-100	Fugitives	Organic Pollutants**	0.6	2.5	
	Anode Material Process				
5M11-08	Continuous Dust	РМ	1 1	47	
	Collection System	PM_{10}	1.1	ч.7 Д7	
	Baghouse	1 10110	1.1	7.7	
5M11-09	Central Vacuum	PM	1.1	0.9	
	System Baghouse	PM ₁₀	1.1	0.9	
CP2-FUG	Anode Material Process	VOC	0.4	1.4	
012-100	VOC/HAP Fugitives	Organic Pollutants**	0.4	1.4	
	Emerge	ncy Generators			
		PM	0.3	0.1	
		PM_{10}	0.3	0.1	
		SO_2	0.2	0.1	
5N01-WA	Diesel Glycol Pump	VOC	0.3	0.1	
		СО	0.8	0.1	
		NO _x	3.8	0.2	
		Organic Pollutants**	0.3	0.1	
		PM	0.6	0.1	
		PM_{10}	0.6	0.1	
71404 117	Discul Wester Discussed	SO_2	0.5	0.1	
/M04-H1-	Diesel Waste Disposal Pump	VOC	0.6	0.1	
G01		СО	1.7	0.1	
		NO _x	8.0	0.4	
		Organic Pollutants**	0.6	0.1	
	Diesel Waste Disposal Pump	PM	0.6	0.1	
		PM_{10}	0.6	0.1	
71404 117		SO_2	0.5	0.1	
/M04-H1-		VOC	0.6	0.1	
G04		СО	1.7	0.1	
		NO _x	8.0	0.4	
		Organic Pollutants**	0.6	0.1	
	Emergency Diesel Generator	PM	0.7	0.1	
		PM10	0.7	0.1	
		SO_2	12.1	0.6	
6N02-EG		VOC	1.1	0.1	
		СО	8.5	0.4	
		NO _x	15.5	0.8	
		Organic Pollutants**	1.1	0.1	

.

EMISSION SUMMARY				
Source	Description	Pollutant	Emission Rates	
Number	Description		lb/hr	tpy
		PM	0.5	0.1
8M01	Diesel Fire Water Pump	PM_{10}	0.5	0.1
		SO_2	5.7	0.3
		VOC	0.5	0.1
		СО	3.8	0.2
		NO _x	16.8	0.8
		Organic Pollutants**	0.5	0.1
		PM	0.1	0.1
4P-EG-01	Emergency Natural Gas Generator	PM_{10}	0.1	0.1
		SO_2	0.1	0.1
		VOC	0.1	0.1
		СО	3.2	0.4
		NO _x	0.8	0.1
		Organic Pollutants**	0.1	0.1

*Inorganics are considered to be non-organic Hazardous Air Pollutants ** Organic Pollutants are considered to be VOC, all organic HAPs, and organic Air Contaminates. ***PM_{2.5} limits are source specific, if required. Not all sources have PM_{2.5} limits.

SECTION III: PERMIT HISTORY

- 262-A Permit was issued to Arkansas Eastman in December of 1974 for the installation of a facility to manufacture various specialty and organic intermediate chemicals through batch operations. Three 70 MM Btu/hr coal-fired boilers were installed to provide steam for the processes.
- 262-AR-1 Issued in 1976, this permit recognized suspension of construction plans for the hydroquinone plant, authorized a higher number of reactors for the chemical intermediates plant, acknowledged the use of ESPs for control of boiler emissions, and permitted the chemical destructor at 9 pounds of particulate per hour.
- 487-A Permit was issued in 1978. This permit allowed the facility to add 8 batch reactors and 10 storage tanks. Each of the reactors were vented through a caustic scrubber. The particulate emissions were routed through fabric filters.
- 262-AR-2 This permit, issued in 1978, authorized an expansion of the chemical products and intermediates. Emission control was provided by caustic and water scrubbers. The permit required the facility to develop an ambient air monitoring program in order to evaluate emission concentrations beyond the property line.
- Issued on July 25, 1980. This permit approved an expansion in production to allow a greater variety and larger quantity of chemicals. New process equipment included reactors, filters, dryers, distillation columns, and storage tanks. Emission control equipment included scrubbers using sodium hydroxide or water. The permit also allowed the installation of a new coal fired boiler (193 MM Btu/hr). The coal boiler utilized an ESP for particulate control, and the boiler was limited to coal at or below 1 percent sulfur, and a heat content of 12,500 Btu per pound. This permitting action required PSD review.
- PSD-AR-311 Issued by the U.S. Environmental Protection Agency on March 27, 1981. This was a PSD permit which addressed the installation and operation of (coal-fired) Boiler #4 and the associated coal handling system. The permit imposed a coal sulfur limit of 1 percent by weight and an ash content of 20 percent by weight. The permit also specified limits on throughput, opacity, emissions, monitoring, and stack testing for the new boiler.
- 262-AR-4 Permit was issued on September 25, 1981. This permit allowed the installation of additional process equipment and a coal-fired boiler. The permit also authorized cessation of certain continuous monitoring equipment, subsequent to the demonstration that criteria pollutant concentrations were well below the NAAQS.
- 262-AR-5 Permit revision was issued on July 23, 1982. This permit authorized an increase in sulfur content of the coal fueling the coal boilers. The sulfur limit was raised

from 1 to 4 percent. Upon evaluation of emission increases and dispersion modeling, this permitting action did not require PSD review.

- 262-AR-6 Issued on March 21, 1986. This permit authorized the installation and operation of an oxidized cellulose facility. Emission control was provided by a packed column scrubber using sodium hydroxide.
- 744-A Issued on November 5, 1984. This permit was issued to allow the operation of a new isopropylbenzene production process. Emission control included a fabric filter and a water scrubber for the catalyst storage and transfer system. Reaction and refining emissions were routed to a flare.
- 829-A Issued on July 14, 1987. This permit authorized the installation and operation of one 78 MMBtu/hr steam boiler. Nitrogen oxides emissions from this boiler were estimated at above the 40 ton/yr Prevention of Significant Deterioration (PSD) threshold, and the permit application was therefore required to undergo PSD review. The BACT analysis found that emissions controlled by either staged combustion/low excess air burners or flue gas recirculation would not substantially improve ambient air quality and were not economically feasible. No additional controls were therefore required, and standard-register burners were approved for use.
- 981-A Issued on February 20, 1990. This permit was issued to allow the operation of a new polymer production facility. Emissions were controlled by conservation vents on the tanks and 2-stage scrubbers on the centrifuges, reactors, and distillation columns.
- 268-I Permit issued on March 25, 1976 to permit the facility's incinerator.
- 1085-A Issued on January 11, 1991. This permit was issued to modernize some of the older permits and to put all of the company's permits into one package. This permit also required Eastman to install and operate a Regenerative Thermal Oxidizer (RTO) on the batch organic chemicals production facilities in buildings 5N01 and 5N03 for the control of VOC emissions by July, 1992.
- 1085-AR-1 Issued on May 14, 1992. This permit involved the installation of a 221 MMBtu/hr natural gas fired boiler (6M-07-01), which required a PSD permit due to significant nitrogen oxide emissions (98 tons per year).
- 1085-AR-2 Issued on February 9, 1994. This permit was issued to document the burning of wastewater sludge in all three of the coal fired boilers at the facility. Eastman proposed to dewater wastewater treatment plant sludge before atomizing it using compressed air, into the high temperature combustion zone of the boilers.

- 1085-AR-3 Issued on April 18, 1994. The modification involved the addition of a packed-bed water scrubber to source 5N01-45, a 24,000 gallon aboveground storage tank which stores crotonaldehyde. This was an uncontrolled source prior to this minor permit modification. Potential emissions from this source were calculated to be 5.7 tons per year after the controls.
- 1085-AR-4 Issued on October 20, 1994. This permit involved venting several temporary storage tanks to the RTOs. The main purpose for this modification was to control the odor generated from the use of ethyl mercaptan, which is mainly used to odorize natural gas. The following tanks were vented to the RTO: 5N01-11, 5N01-12, 5N01-13, 5N01-14, 5N01-16, 5N01-19, 5N01-20, 5N01-21, 5N01-29, 5N01-30, 5N01-34, 5N01-35, 5N01-36, 5N01-37, 5N01-50, 5N01-51, 5N01-52, 5N01-53, 5N01-60, 5N01-62, 5N03-09, 5N03-10, and 5N03-61.
- 1085-AR-5 Issued on October 18, 1994. This was a minor modification for producing a new polymer in the Polymer Production Facility. Emissions from this modification were controlled by the RTOs, scrubbers, and conservation vents on tanks.
- 1085-AR-6 Issued on June 6, 1995. This modification involved modifying existing solvent recovery equipment used to recover additional solvent and to remove potential odor producing compounds by destroying them in the existing RTOs. The main purpose of this modification was to control the odor generated from the use of ethyl mercaptan. Ethyl mercaptan is mainly used to odorize natural gas. The odor threshold of ethyl mercaptan is 0.4 ppb. To eliminate this odor, the facility proposed that the scrubber atmospheric vents be connected to the RTOs. Additionally, the permittee proposed to modify the existing wastewater treatment system by closing the existing equalization basin, discontinuing the use of the existing diversion basin for processing wastewater, and constructing aboveground tanks for equalization/neutralization and diversion of the wastewater. The system modification included the addition of two 30,000 gallon pump station clearwells, two 750,000 gallon equalization tanks, and one 1,000,000 gallon diversion tank. Also a new lift station, neutralization system, and a floating organic skimmer and decant system was to be provided. The existing diversion basis was to be used to capture noncontact cooling water and storm water runoff should it become contaminated.
- 1085-AR-7 Permit was issued on November 27, 1995. This permit was issued to raise the particulate emission limit on the RTOs.
- 1085-AR-8 Permit was issued on May 8, 1996. This permit covered routing emissions from eleven waste storage tanks to the coal-fired boilers to abate odors within the utilities area of the plant, to burn waste solvent fuel in the boilers at the rates certified under the Boiler and Industrial Furnace regulation (BIF), to increase the rate of rubber and paper pellet fuel burning to 100% of the total heat input of the

coal-fired boilers, and to construct one 20,000 gallon storage tank containing a final polymer product.

- 1085-AR-9 Permit was issued on November 12, 1996. This permit involved increasing potential VOC emission from the Waste Chemical Destructor from 0.5 tpy to 8.8 tpy due to an anticipated future increase in business and a corresponding increase in the amount of wastes that could potentially be generated; and to increase potential inorganic emissions from 16.3 tpy to 43.8 tpy from the two RTOs due to an anticipated increase in chlorinated compounds production.
- 1085-AR-10 Permit was issued on March 11, 1997. This permit involved the construction and operation of a continuous dust collection system and central vacuum cleaning system. Five additional emission points discharging from venturi scrubbers and fabric filters, and an emission point designating fugitive emission from maintenance activities, were created with the startup of this dust collection and vacuum cleaning system. This permit also allowed the organic sulfonation facility to produce alternative products, which required minor changes in the process chemistry to meet new markets. Eight new emission points were created with this modification.
- 1085-AOP-R0 Permit was issued on June 24, 2002. This permit (1085-AOP-R0) was issued in order to satisfy the requirements of Title V of the Clean Air Act. This permit also incorporated the requirements of 40 CFR Part 60, Subpart EEE, National Emission Standards for Hazardous Air Pollutants from Hazardous Waste Combustors, promulgated on September 30, 1999. In addition, the facility was authorized to: burn wood chips in the three coal-fired boilers; install a system of tanks, strippers, dryers, and distillation columns necessary to recover dimethyl sulfoxide from wastewater; incorporate a project to collect and reduce the accumulation of process dust within the organic sulfonate manufacturing area; install a small-scale laboratory for research and development activities; re-route emissions from 23 tanks to the Regenerative Thermal Oxidizer (RTO); replace five waste storage tanks; and to re-route three distillation column vents to the Regenerative Thermal Oxidizer (RTO) control system for the purpose of odor abatement.
- 1085-AOP-R1 Permit was issued on January 20, 2004. This permit was issued in response to a Permit Appeal Resolution (PAR, Docket No. 02-006-P) concerning Air Permit 1085-AOP-R0. Changes based upon the PAR include: the deletion of individual unit pound-per-hour emission limits for Hazardous Air Pollutants (HAPs); the addition of a plantwide condition to clarify types of permit deviations and reporting schedules; the removal of the carbon monoxide (CO) stack testing requirement for the Chemical Waste Destructor (SN-6M03-05); the addition of a mechanism by which the facility may use a correlation study to petition the Department for less frequent (non-MACT) stack testing of NO_X, SO₂, and/or PM at SN-6M03-05; the revision of conditions related to 40 CFR Part 63, Subpart

EEE to reflect the most recent version of the interim rule; a modification of former Plantwide Condition 23 to clarify that a compliance report is required for state-only enforceable terms and conditions; and the incorporation of the requirements of 40 CFR Part 63, Subpart GGG, National Emission Standards for Pharmaceuticals Production.

1085-AOP-R2 Permit was issued on June 18, 2004. The permit was modified in order to connect three general-purpose bulk storage tanks to the Regenerative Thermal Oxidizers (RTOs) to meet the deadline of December 23, 2003 for 40 CFR Part 63, Subpart MMM – National Emission Standards for Hazardous Air Pollutants for Pesticide Active Ingredient Production. VOC and HAP emissions were reduced by 2.8 tons/yr as a result of the tank controls. In addition, the requirements of the MACT Subpart MMM were incorporated into the permit as well as changes to the Insignificant Activities list.

1085-AOP-R3 Permit was issued on May 20, 2005. Four changes were incorporated into the permit. First, the HCl (inorganic HAP) dispersion modeling demonstration to meet PAER requirements was changed as to allow the submitted modeling in combination with facility records of inorganic HAP emissions to verify that the off-site concentration is protective of public health. Second, a 2,000 gallon liquid process tank was installed. The tank was used for the purpose of flushing the chemical distribution piping at the Chemical Waste Destructor (6M03-05). The spent solvent is then routed to either the coal-fired boiler auxiliary waste chemical burners or to the burner of the chemical waste destructor. Emissions from tank venting will be collected and routed to the coal-fired boilers (6M01-01). Estimated emissions from the tank were less than 0.1 ton/yr VOC or HAP. Third, a bleach scrubber (D75-02) in the Organic Chemical Intermediates section was removed from service. The scrubber, while not actually an atmospheric emission source, removes ethyl mercaptan, an odorous compound generated by an existing batch process. The exiting gases are then routed to the regenerative thermal oxidizers (SN-5N09-01). There is no permitted change to emission estimates at SN-5N09-01. Finally, the facility also requested changes to final Specific Condition CDW 9 to include a compliance option, as CDW 9b, which was not included when the language was originally added to the permit. This option was already provided by 40 CFR Part 63 Subpart EEE. There were no permitted emission changes with this modification.

1085-AOP-R4 Issued on August 14, 2006. The facility modified their permit to: use tanks and scrubbers in the Solvent Recovery and Storage Tanks and Miscellaneous Sources areas to produce Biodiesel; to install upgrade equipment to the Regenerative Thermal Oxidizers (RTOs) to increase the destruction removal efficiency (DRE) from 95% to 98%; and to install a replacement air seal inlet and outlet main valves and to add a chamber purge system to prevent leakage and air infiltration around the valves that will increase the DRE, thus reducing emissions of VOC and HAP with no increased usage of natural gas.

- 1085-AOP-R5 Issued on February 15, 2007. The facility requested a minor permit modification to construct a new production line to manufacture wood fuel pellets.
- 1085-AOP-R6 Issued on May 8, 2007. The facility increased biodiesel production capacity and added several new tanks (accounted for within an existing tank bubble, 5N03TK-01) and loading racks which vent to the atmosphere. Controlled emissions from process equipment and storage tanks are routed to scrubbers SV-01 and SV-03 (in the 4PSR-00 emission bubble of the Solvent Recovery Section), and two regenerative thermal oxidizers (SN-5N09-01 of the OCI Section). The loading racks and BD-01 Biodiesel Sales Tank qualify as A-13 Insignificant Activities. The Building 5N07 acrylic resins (5N07-06 and 5N07-FUG) and polymer production (5NPOLY-TNK and POLY-FUG) facility will be retrofitted for biodiesel production. Acrylic resins and polymers will no longer be manufactured. Instead, a new source, the 5N07 Production Facility, which will produce primarily biodiesel, was added.
- 1085-AOP-R7 Issued on December 17, 2007. The permit was modified to allow for the production of Aldehyde products. The emission points for the new Aldehyde Processing Section were the hot oil system (4P05-01), the water scrubber (4P05-02), equipment fugitives (4PSR-FUG), and the RTO (5N09-01).
- 1085-AOP-R8 This revision as issued on June 6, 2008. This permit incorporated the following changes: added 40 CFR 63, Subpart FFFF requirements for several sources at the facility, routed the process equipment going to control devices 4P02-01 and 4P94-02 to the RTOs, route storage tank TF-2, PES # 5N01-44, to the RTOs; Install three new storage tanks, T-271 (30,000 gallons), T-272 (30,000 gallons), and T-273 (40,000 gallons), use an existing storage tank, PT-50, for storage of off-site waste; rename T-212A to VC-PT-03; remove TFB-30 from NSPS Subpart Kb applicability, and redirect the vent from the SB-01 (Source # 4P05-02), located in the Aldehyde Section, to SV-03 (Source # 4P94-02), a water scrubber located in Solvent Recovery.
- 1085-AOP-R9 This permit was issued on September 19, 2013. In this permit the following changes were made:
 - Renew the Title V Air Permit

As part of the renewal permit, the Wood Pellet process was removed from the facility. In addition, rubber is no longer used at the Coal Boilers and the relevant specific conditions have been removed.

Emergency engines and NESHAP ZZZZ - *National Emission Standards for Hazardous Air Pollutants for Reciprocating Internal Combustion Engines*conditions were also been added to the permit. These five engines were existing sources. The natural gas fired boilers, 6M06-01 and 6M07-01 emissions were

recalculated using current AP-42 factors for natural gas combustion. Emissions for the coal boilers, 6M01-01, were re-evaluated based upon the maximum coal firing rate on an hourly basis, and an average rate for the annual emissions. Inorganics from this source were revised in accordance with the most recent test event for EEE. Small amounts of lead emissions were also added to the permit. The facility increased production in the Aldehyde processing section from 45 million to 58.5 million pounds per year of vinyl compound products.

• Added tanks T-271 (30,000 gallons), T-272 (30,000 gallons), and T-273 (40,000 gallons)

These tanks are operated at a pressure below 29.7 psia, thus triggering NSPS Kb applicability. Each tank is vapor balanced with incoming railcars and tank trucks at the Aldehyde Processing Section.

• Produce a multi-use anode material within the Organic Sulfonation section

Emissions from this process are 2.3 tpy PM/PM₁₀, 1.4 tpy VOC, and 1.4 tpy Organic HAP. A dormant part of the Organic Sulfonation plant will be retrofitted to process anode material. Existing and new process equipment both are used in the process. The Anode Material Process (CP2), which will be permitted as a separate section, CP-2, consists of solids handling equipment, continuous stirred tank reactors, and dryers. Pelletized and granular materials are fed through a metering system into vessels where the material is heated and mixed. The formulated material is then dried and repackaged. All of this equipment is located inside the 5M11 building. Dust from all of the solids handling equipment is vented to a continuous dust control system (CDCS), 5M11-08. The CDCS consists of a baghouse for solid particle separation, a collection hopper, and an induced draft fan. A central vacuum cleaning system (CVC), 5M11-09 will be used to clean spills. All VOC vents are routed to the RTOs, 5N09-01. A 17,500 gallon solvent tank (CP2-T-004) and a 36,000 gallon residue tank (CP2-T-003) was added. Emissions from tanks and equipment containing VOC's will vent to existing thermal oxidizers SN-5N09-01. Previously, SN-5M11-08 and SN-5M11-09 are permitted as scrubbers (SRE-VE-501 and SER-VE-502) in the OSP section. As a result of this change, the source will be permitted as a baghouse (CP2-C-501 and SER-C-503).

Neither CP2-T-003 or CP2-T-004 are regulated by 40 CFR Subpart Kb (Standards of Performance for Volatile Organic Liquid Storage Vessels) per 40 CFR 60.11b(b). Both tanks contain liquids with vapor pressures less than 15.0 kPa. Both tanks are vented to the RTOs, 5N09-01. As a miscellaneous organic chemical manufacturing process unit (MCPU), this section is subject to 40 CFR Part 63 Subpart FFFF, and was required to comply with these provisions upon startup.

• Install a 2.5 MMBTU thermal oxidizer and caustic scrubber, SN-5N09-2, in the OCI section.

Thermal oxidizer exhaust gases vents to a caustic scrubber suitable for halogen halide and chlorinated hydrocarbon control. The source is subject to 40 CFR 63, Subpart FFFF. Permitted emissions increased by 0.1 tpy particulates, 13.1 tpy SO₂, 12.7 tpy VOC, 2.2 tpy CO, 13.1 tpy NO_x, 4.4 tpy Inorganic HAP, and 12.7 tpy Organic HAP. A small increase of 0.3 tpy of fugitive VOC and Organic HAP emissions from the OCI section was added from the installation of the source.

• To change the scrubbing fluid on SPS-VE-02 (SN-5M04-01 of the Organic Sulfonation Section) from 2,2,4-trimethyl-1-3-pentanediol diisobutyrate (TXIB) to water.

There were no changes to emissions as a result of the scrubber liquid change.

• Add a new storage tank, PSA-TF-01 to the Organic Sulfonation Section

The new tank was added to the 5MNOBS-TNK tank bubble. Emissions are estimated to be less than 40 lb/yr of VOC from this source. Neither tank is subject to NSPS Kb.

- This permit was issued May 14, 2015. With this permit modification, 1085-AOP-R10 FutureFuel added three odor control scrubbers to reduce odors of trimethylamine and triethylamine emitted by process equipment and added a third scrubber on tank TF-14. Tank TF-14 is a pressurized tank. The scrubber is located on the pressure relief valve of the tank and is not permitted as a source. The other odor scrubbers are before Thermal Oxidizer SN 5N09-02 and SN-5N09-01. No increases in permitted emissions are made due to these scrubbers. The emissions for the gasoline tank SN 6N01-03 were updated. Hourly rates were increased. The "Organic HAPs" in the permit were changed to "Organic Pollutants" to include organic air contaminates and HAPs which are not VOC. Two insignificant diesel tanks were removed. Opacity observation requirements for 5N09-01 and 5N09-02 were updated. A requirement to submit an application to incorporate the boiler MACT into the permit was added to Plantwide conditions.
- 1085-AOP-R11 This permit was issued January 1, 2016. The facility submitted an application to add 40 C.F.R.§ 63 Subpart DDDDD to the permit. The facility also requested the approval to install a second 500 cfm natural gas fired thermal oxidizer with quench and scrubber (SN-5N09-03). Total facility wide emission changes were as follows: +0.2 tpy PM/PM₁₀, +13.1 tpy SO₂, +13.0 tpy VOC, +2.2 tpy CO, +13.1 tpy NO_x, +4.4 tpy Inorganics, and +12.7 tpy Organic Pollutants.

1085-AOP-R12 This permit was issued on August 31, 2016. The facility submitted a minor modification application to:

- Remove tanks TF-2, TFV-1, and TFV-3 from the 5N03-TK01 emission bubble.
- Remove tanks TF-6, TF-7, TF-10, and TF-11 from the Insignificant Activities list as they are part of SN 5N03TK-01.
- Correct TFB-2 to TFB-22 in SN 5N03TK-01 emission bubble list.
- Add tanks T-273, TFS-6, and TF-8 to the 5N03TK-01 emission bubble.
- Move tank TF-5FS from the Insignificant Activities List to SN 5N03TK-01, as it will now be subject to NESHAP FFFF (MON MACT).
- There were no permitted changes in emissions with these actions.

1085-AOP-R13 This permit was issued on November 29, 2018. The facility submitted a Title V permit renewal, with modification, to update several emission factors due to new test data. Permitted emission increases were 0.8 tpy of PM and PM_{10} , 7.6 tpy SO₂, 9.07 tpy VOC and organic pollutants, 89.5 tpy CO, 38.2 tpy NO_X, and 1.2 tpy of inorganic pollutants.

1085-AOP-R14 This Permit was issued on February 24, 2020. The facility submitted a Minor Modification to add a second Hot Oil System (SN 4P05-03) to the Aldehyde Processing section of the permit. This will increase the production limits of the Aldehyde section from 58.5 million pounds per year of vinyl compound products to 75 million pounds per year.

Permitted emissions increases are 0.6 tpy of PM and PM_{10} , 0.1 tpy SO₂, 3.4 tpy VOC and Organic Pollutants, 6.4 tpy CO, and 3.8 tpy NO_X

Based on calculations submitted by the facility, this debottlenecking does not constitute a significant increase in emissions for any NSR pollutant.

1085-AOP-R15 – This permit was issued on June 27, 2022. This application was submitted as a minor modification to Permit No. 1085-AOP-R14 to do the following:

- Update the process description for the Aldehyde Processing Section (AP);
- Removing tank 5N01-39 from bubble source 5N03TK-01, this tank is reported under SN-5N09-01;
- Replacing 6N02 Diesel Generator with 6N02-EG (emergency generator)
- Update Plantwide Limits due to the change in generators.

Permitted emissions decreased 0.1 tpy of both PM and PM_{10} , 0.3 tpy VOC, 0.3 tpy Organic Pollutants and 0.6 tpy NO_X. Permitted emissions increased 0.1 tpy SO₂ and CO.

1085-AOP-R16 - This permit was issued on June 21, 2023. This permit was a minor modification to Permit No. 1085-AOP-R15:

• Add a natural gas emergency engine (SN-4P-EG-01) to the permit. SN-4P-EG-01 is subject to 40 C.F.R. Part 63, Subpart ZZZZ and 40 C.F.R. Part 60, Subpart JJJJ.

Permitted emission increases are 0.1 tpy SO₂, PM, PM_{10} , VOC, NO_X and Organic Pollutants as well as 0.4 tpy of CO.

SECTION IV: SPECIFIC CONDITIONS

Organic Chemical Intermediates 5N09-01, 5N09-02, 5N09-03, and OCI-FUG

Source Description

FutureFuel's batch organic chemical intermediates facilities are located in Buildings 5N01, 5N03, and 5N07. These production buildings contain multi-purpose/product equipment which may produce a variety of chemicals. The contained or captured vapors from the equipment in the three batch production buildings are vented through a collection system to the RTO units via a common duct. Volatile organic compounds (VOCs) are destroyed by combustion. Fugitive emissions from organic chemical intermediates are designated as source number OCI-FUG. Two other TOs (SN-5N09-02 and SN-5N09-03) are also present, but in conjunction with a caustic scrubber.

Some portion of the organic chemical intermediate facility is subject to NSPS Subpart Kb, NESHAP Subpart GGG, MMM, and FFFF.

Specific Conditions

OCI 1 The permittee shall not exceed the emission rates set forth in the following table. Emission rates are based upon the maximum capacity of equipment. [Reg.19.501 *et seq.* and 40 C.F.R. § 52 Subpart E]

SN	Description	Pollutant	lb/hr	tpy
5N09-01		PM ₁₀	9.90	17.5
	Regenerative	SO_2	8.40	36.8
	Thermal Oxidizer	VOC	43.6	191.0
	(2 Units)	CO	11.0	25.2
		NO _x	8.7	38.1
5N09-02		PM10	0.1	0.1
	Thermal Oxidizer	SO_2	3.0	13.1
	and Caustic	VOC	2.9	12.7
	Scrubber	CO	0.5	2.2
		NO _x	3.0	13.1
		PM_{10}	0.1	0.2
5N09-03	Thermal Oxidizer	SO_2	3.0	13.1
	and Caustic	VOC	2.9	12.7
	Scrubber	CO	0.5	2.2
		NO _x	3.0	13.1
OCI-FUG	Fugitives	VOC	3.7	15.9

OCI 2 The permittee shall not exceed the emission rates set forth in the following table. The facility shall show compliance with the facility total ton/yr limits using the procedures outlined in Plantwide Conditions 12 through 16. Hourly Plantwide Hazardous Air Pollutant emissions are limited by Plantwide Conditions 16. Additional HAP limitations are included in Plantwide Condition 13. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]

SN	Description	Pollutant	lb/hr	tpy
5N09-01	Regenerative	PM	9.9	17.5
	Thermal Oxidizer	Inorganics*	10.0	43.8
	(2 Units)	Organic Pollutants**	43.6	191.0
5N09-02	Thermal Oxidizer	PM	0.1	0.1
	and Caustic	Inorganics*	1.0	4.4
	Scrubber	Organic Pollutants**	2.9	12.7
	Thermal Oxidizer	PM	0.1	0.2
5N09-03	and Caustic	Inorganics*	1.0	4.4
	Scrubber	Organic Pollutants**	2.9	12.7
OCI-FUG	Fugitives	Organic Pollutants**	3.7	15.9

*Inorganics are considered to be non-organic Hazardous Air Pollutants. **Organic Pollutants are considered to be VOC, all organic HAPs, and organic Air Contaminates.

- OCI 3 The permittee shall perform periodic testing of 5N09-01 (RTO) for SO₂, VOC, CO, and NO_x, using Methods 6C, 25A, 10, and 7E, respectively. The VOC destruction efficiency shall be determined during the Method 25A testing. Testing shall commence no later than 61 months from the date of the previous test. Testing at 5N09-01 shall conform with the requirements of Plantwide Conditions 3 and 4. [Reg.19.702 and 40 C.F.R. § 52 Subpart E]
- OCI 4 The permittee shall commence testing of SN-5N09-02 (TO) and SN-5N09-03 (TO) no later than 61 months from the date of the previous test. The halogen emission rate shall be determined using Method 26 or 26A, or 320 and shall be limited to no more than 0.45 kg/hr, contained within Specific Condition MON 10. The VOC destruction efficiency shall be determined using Method 25. Testing at SN-5N09-02 and SN-5N09-03 shall conform with the requirements of Plantwide Conditions 3 and 4. [Reg.19.702 and 40 C.F.R. § 52 Subpart E]
- OCI 5 The permittee shall not exceed 20% opacity as measured by Method 9 at SN-5N09-01, SN-5N09-02 and SN-5N09-03 during normal operations. Method 22 observations shall be performed on a weekly basis if any visible emissions are detected the permittee shall immediately conduct a Method 9 observation to determine if visible emissions are above permitted limits. [Reg.19.503 and 40 C.F.R. § 52 Subpart E]
- OCI 6 If visible emissions in excess of 20% are detected from SN-5N09-01, SN-5N09-02 or SN-5N09-03, then the permittee will conduct corrective action. The results of these observations and corrective action shall be kept on site and made available for

inspection upon request. Opacity observations at the RTOs shall not be required during times when the RTOs are being "baked out." [Reg.19.702 and 40 C.F.R. § 52 Subpart E]

- OCI 7 The permittee shall continuously monitor and record the temperature in the combustion chamber of the RTOs and TOs during normal operations. These records shall be kept in accordance with General Provision 7, kept on site, and made available to Division personnel upon request. [Reg.19.703, 40 C.F.R. § 52 Subpart E, and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]
- OCI 8 The permittee shall maintain the temperature in the combustion chamber of the RTOs and TOs during normal operations as outlined in the most current version of the Facility Operating Plan. [Reg.19.303 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]
- OCI 9 The permittee shall monitor SN-5N09-01, 5N09-02, and 5N09-03 as outlined in OCI 7 and OCI 8 above and the monitoring requirements of the MON MACT outlined in MON 10. The permittee shall submit reports as required by Plantwide Condition 33. [Rule 19.304 and 40 C.F.R. § 64]

40 CFR 63 Subpart GGG - National Emission Standards for Pharmaceutical Production

APPLICABILITY

OCI 10 A portion of this facility is subject to 40 CFR § 63 Subpart GGG, National Standards for Pharmaceuticals Production. Applicable requirements include the following conditions. [Reg.19.304 and 40 C.F.R. § 63.1250]:

Affected Source

a. The permittee is an affected source subject to 40 CFR Part 63, Subpart GGG as defined in 40 CFR §63.1250(a). The source is an existing source with a compliance date of October 21, 2002. [Reg.19.304 and 40 C.F.R. § 63.1250(a)]

General Provisions Requirements

 b. The provisions of Subpart A, specified in Table I of Subpart GGG are the only general provisions that apply to an affected source subject to this subpart. [Reg.19.304 and 40 C.F.R. § 63.1250(c)]

Storage Tank Ownership

c. The requirements of §63.1250(e), storage tank ownership determination, do not apply until such a time the permittee either installs or activates a tank for use in an applicable Pharmaceutical Manufacturing Process (PMPU). The permittee does not currently have storage tanks subject to this requirement. [Reg.19.304 and 40 C.F.R. § 63.1250(e)]

Compliance Date

d. The compliance date for the existing affected source is October 21, 2002. [Reg.19.304 and 40 C.F.R. § 63.1250(f)(1)]

Applicability except during periods of startup, shutdown, and malfunction

e. The permittee shall comply with all applicable requirements of 40 CFR 63, Subpart GGG except that emission limitations shall not apply during periods of startup, shutdown, and malfunction. [Reg.19.304 and 40 C.F.R. § 63.1250(g)]

Consistency with other Regulations

- f. The permittee shall identify in the Notice of Compliance Status report [the report was submitted on March 20, 2003] required by §63.1260(f) the compliance options cited in §63.1250(h)(1) through (6) for those regulations identified that may overlap Subpart GGG. [Reg.19.304 and 40 C.F.R. § 63.1250(h)]
 - i. The permittee will be subject to MACT standards with upcoming compliance dates including the Pesticide Active Ingredient (PAI) MACT, and upon promulgation, the Miscellaneous Organic NESHAP (MON rule). These regulations are not specifically mentioned in the overlap section of the MACT (Subpart GGG). [Reg.19.304 and 40 C.F.R. § 63.1250(h)(1)]
 - ii. The permittee may elect to comply with the monitoring recordkeeping and reporting requirements of either 40 CFR Part 63, Subpart GGG or RCRA Subparts AA, BB, CC for process vents, equipment leaks, and containers/storage tanks covered under both regulations. [Reg.19.304 and 40 C.F.R. § 63.1250(h)(2)]
 - iii. A storage tank with a fixed roof, closed-vent system and control device in accordance with NSPS Kb, must comply with Subpart GGG monitoring, recordkeeping, and reporting requirements for that vessel. Currently the permittee has no tanks in Subpart GGG applicable service. [Reg.19.304 and 40 C.F.R. § 63.1250(h)(3)
 - iv. Equipment subject to Subpart I of this part may elect to comply with either the provisions of §63.1255 or the provisions of Subpart H of this part for all such equipment. The permittee does not have equipment in Subpart I or Subpart H applicable service. [Reg.19.304 and 40 C.F.R. § 63.1250(h)(4)]
 - v. The permittee does not operate any process subject to the Polyether Polyols MACT. [Reg.19.304 and 40 C.F.R. § 63.1250(h)(6)]
STANDARDS: GENERAL

OCI 11 The permittee shall control HAP emissions to levels specified in this section on and after the compliance dates specified in §63.1250(f) [the compliance date for an existing source is specified as October 21, 2002]. Initial compliance with the emission limits is demonstrated in accordance with the provisions of §63.1257 [Test Methods and Compliance Procedures], and compliance is demonstrated in accordance with the provisions of §63.1258 [Monitoring Requirements]. [Reg.19.304 and 40 C.F.R. § 63.1252]

Opening of a safety device

a. The opening of a safety device, as defined in §63.1251, definitions, is allowed at any time conditions require it to do so to avoid unsafe conditions. [Reg.19.304 and 40 C.F.R. § 63.1252(a)]

Closed-vent systems

b. If the permittee installs a by-pass line that could divert a vent stream away from a control device used to comply with the requirements of §63.1253 [storage tanks], §63.1254 [process vents], and §63.1256 [wastewater provisions], the permittee shall comply with the requirements of §63.1252(b)(1) and (2).

The permittee operates regenerative thermal oxidizers (RTOs), which have emergency dampers meeting the definition of a safety device of §63.1251. Bypass lines do not exist on this closed-vent system and control device. [Reg.19.304 and 40 C.F.R. § 63.1252(b)(1) and (2)]

Heat exchange systems

c. The permittee shall comply with the requirements in §63.1252(c)(1) of this section for heat exchange system that cool process equipment or materials used in pharmaceutical manufacturing operations except as provided by §63.1252(c)(2). [Reg.19.304 and 40 C.F.R. § 63.1252(c)(1)]

Heat exchangers (HON) requirements

d. Applicable heat exchange systems shall be treated according to the provisions of §63.104 [HON Heat Exchangers] except that monitoring shall be no less than quarterly. [Reg.19.304 and 40 C.F.R. § 63.1252(c)(1)]

Heat exchangers (cGMP) option

e. For identifying leaking heat exchange systems of equipment, which meet current good manufacturing practice (cGMP) requirements of 21 CFR Part 211. The permittee may elect to use the physical integrity of the reactor as a surrogate of the heat exchange system leaks around the reactor.

Unit D1-01 meets the criteria of this subpart, cGMP, so the physical integrity of the equipment (pressure vessel) is used as the surrogate indicator of heat exchange system leaks. [Reg.19.304 and 40 C.F.R. § 63.1252(c) and (c)(2)]

Emissions averaging

f. The permittee may choose to comply with the provisions of §63.1253 [storage tanks] and §63.1254 [process vents] by using emissions averaging requirements specified in §63.1257(g) and (h) except as provided in §63.1252(d)(1). [Reg.19.304 and 40 C.F.R. § 63.1252(d)]

At this time, the permittee does not choose to opt for an emissions averaging compliance method.

Pollution prevention (P2) alternative

g. The permittee may choose, except as provided in §63.1252(e)(1) of this section, to meet the pollution prevention alternative requirement specified in either §63.1252(e)(2) or (3) of this section, in lieu of the requirements specified in §63.1253 [tanks], §63.1254 [process vents], §63.1255 [LDAR], and §63.1256 [wastewaters]. Compliance shall be demonstrated through the procedures in §63.1257(f). Any Pharmaceutical Manufacturing Process Unit (PMPU) for which the permittee seeks to comply by using the pollution prevention alternative shall begin with the same starting material(s) and end with the same product(s). The permittee shall not comply with the pollution prevention alternative by eliminating any steps of a process by transferring the step offsite and to another manufacturing location. [Reg.19.304 and 40 C.F.R. § 63.1252(e)]

The permittee presently does not choose to opt for the P2 alternative.

Control requirements for certain liquid streams in open systems within a PMPU

h. The permittee does not operate any liquid streams in open systems as described in §63.1252(f). Therefore, this requirement is not applicable. [Reg.19.304 and 40 C.F.R. § 63.1252(f)]

Control requirements for halogenated vent streams that are controlled by combustion devices

i. If a combustion device is used to comply with the provisions of §63.1253 [storage tanks], §63.1254 [process vents], or §63.1256(h) [wastewater vent streams] for a

halogenated vent stream, then the vent stream shall be ducted to a halogen reduction device such as, but not limited to, a scrubber, before it is discharged to the atmosphere. The halogen reduction device must reduce emissions by amounts specified in either $\S63.1252(g)(1)$ or (2) of Subpart GGG.

The permittee does not manage any halogenated vent streams in its PMPU. Therefore, this requirement is not applicable. If halogenated compounds are to be vented from the PMPU, the permittee shall comply with the requirements of this subpart. [Reg.19.304 and 40 C.F.R. § 63.1252(g)]

Planned routine maintenance for centralized combustion control devices

j. The permittee does not operate any non-dedicated PMPUs during periods of planned routine maintenance for centralized combustion control devices (CCCD) and is not subject to this citation. [Reg.19.304 and 40 C.F.R. § 63.1252(h)]

STANDARDS: Storage Tanks

OCI 12 The requirements of §63.1253 do not apply until such a time the permittee either installs or activates a storage tank for use in an applicable Pharmaceuticals Manufacturing Process Unit. [Reg.19.304 and 40 C.F.R. § 63.1253]

STANDARDS: Process Vents - Existing Sources

OCI 13 The permittee shall comply with the requirements in either §63.1254(a)(1) [process-based emission reduction] and (3) [individual vent emission reduction], or §63.1254(a)(2) [process-based annual mass limit] and (3) [individual vent emission reduction]. Initial compliance with the required emission limits or reductions in §63.1254(a)(1) through (3) are demonstrated in accordance with the initial compliance procedures described in §63.1257(d) [Initial Compliance with Process Vents], and continuous compliance is demonstrated in accordance with the monitoring requirements in [Monitoring]. [Reg.19.304 and 40 C.F.R. § 63.1254(a)]

Process-based emission reduction requirement

a. If the permittee chooses the compliance option in §63.1254(a)(1), uncontrolled HAP emissions from the sum of all process vents with a process that are not subject to the requirements of §63.1254(a)(3)[individual vent emission reduction requirement] shall be reduced by 93% or greater by weight, as specified in §63.1254(a)(1)(ii) [process-based emission reduction requirement]. Notification of changes in the compliance method shall be reported according to the procedures in §63.1260(h) [notification of process change]. [Reg.19.304 and 40 C.F.R. § 63.1254(a)(1)]

Process-based annual mass limit

b. If the permittee chooses the compliance option in §63.1254(a)(2), the permittee shall not allow actual HAP emissions from the sum of all process vents within a process (individual PMPU) not to exceed 900 kg (1894 lbs) in any 365-day period. Actual HAP emissions from the sum of all process vents within processes (all PMPUs) complying with §63.1254(a)(2)(i) are limited to a maximum of 1,800 kg (3,968 lbs) in any 365-day period.

Initial compliance is demonstrated by determining controlled HAP emissions by:

- (1) Computing the uncontrolled emissions from the PMPU and,
- (2) By applying a demonstrated control efficiency to obtain "controlled HAP emissions"

The process is described in the Test Methods and Compliance Procedures section Subpart GGG 63.1257(d)(1)(ii)(A). The permittee has chosen the process-based annual mass limit option for initial compliance. [Reg.19.304 and 40 C.F.R. § 63.1254(a)(2)]

c. Emissions from vents that are subject to the requirements of §63.1254(a)(3) [individual vent emission reduction] and emissions from vents that are controlled in accordance with the procedures in §63.1254(c)[alternative standards] may be excluded from the sums calculated in §63.1254(a)(2)(i) and (ii).

Emissions from vents subject to 98% HAP control or to less than 20 ppmv and that are meeting the alternative standard requirements do not have to be included in the 900 kg or 1,800 kg actual HAP emissions sums in 63.1254(a)(2)(i) and (ii). [Reg.19.304 and 40 C.F.R. 63.1254(a)(2)(ii)]

d. The permittee may switch from compliance with §63.1254(a)(2) [process-based annual mass limit] to compliance with §63.1254(a)(1) [process-based reduction] after at least one year of operation in compliance with the §63.1254(a)(2) [process-based annual mass limit]. Notification of such a change in the compliance method shall be reported according to the procedures in §63.1260(h) [notification of process change]. [Reg.19.304 and 40 C.F.R. § 63.1254(a)(2)(iv)]

Individual vent emission reduction requirements

e. If uncontrolled HAP emissions from a process vent exceeds 25 tons per year and the flow weighted average flowrate (FRA) is less than or equal to the flowrate index (FRI), the uncontrolled HAP emissions from the vent must be controlled to 98%, unless the vent is "grandfathered", installed on or before April 2, 1997. The permittee's RTOs were installed in 1992 and are "grandfathered" under the language of §63.1254(a)(3)(ii) and (A)(1), which requires a HAP emissions

reduction greater than or equal to 93% by weight but less than 98% by weight. [Reg.19.304 and 40 C.F.R. § 63.1254(a)(3)]

STANDARDS: EQUIPMENT LEAKS

OCI 14 Equipment means each pump, compressor, agitator, pressure relief device, sampling connection system, open-ended valve or line, valve, connector and instrumentation system in OHAP service. In OHAP service means that the equipment either contains or contacts a fluid, liquid or gas that is at least 5% by weight total OHAP. [Reg.19.304 and 40 C.F.R. § 63.1255]

General equipment leak requirements

a. The provisions of §63.1255(a) apply to pumps, compressors, agitators, pressure relief devices, sampling connection systems, open-ended valves or lines, valves, connectors, instrumentation systems, control devices, and closed-vent systems that are intended to operate in OHAP service 300 hours or more during a calendar year within a source subject to this subpart. [Reg.19.304 and 40 C.F.R. § 63.1255(a)]

LDAR (Leak Detection and Repair) Provision summary

b. An attached table provides a summary of the equipment leak requirements of Subpart GGG. Because of the complexities of the LDAR requirements, this table should be considered a reference tool only and the regulations should be referenced when developing a detailed plan of compliance. The permittee shall develop a comprehensive LDAR program to fully meet the Subpart GGG equipment leak requirements including developing a list of equipment and identification numbers subject to the requirements and a monitoring schedule. Connectors, except those determined to be unsafe-to-monitor, difficult to monitor, or inaccessible, do not have to be individually identified, but the lines must be identified. Physical tagging of components is not required by 40 CFR Part §63.1255(a)(7) and §63.1255(g)(2)(i)(C). [Reg.19.304 and 40 C.F.R. § 63.1255(a)(1)]

Summary of Equipment Leak Requirements for Subpart GGG ¹						
Equipment Pharma MACT/HON	Design Requirements/ Exemptions	Monitoring Frequency	Method	Leak Limit	Calculations	Recordkeeping Requirements (40 CFR §63.1255(g))
Pumps in Light Liquid Service (63.1255(c))		Quarterly with Instrument (If 10% of pumps or three of the pumps in the group of the process, leak, then monitor monthly) Weekly visual inspection	Method 21 (40 CFR Part 60 Appendix A) Method 21 (40 CFR Part 60 Appendix A) Visual	2,000 ppm 2,000 ppm	Calculate Leakers per 40 CFR Part 63.1255(c)(4) Calculate Leakers	Keep records/statistics on leakers. Develop a list of identification numbers of equipment subject to the requirements of this section. List is to be updated within 15 calendar days of the completion of each monitoring survey.
Pressure Relief Devices in Gas/Vapor Service (§63.165)	OHAP Service Exempt if routed to vent header	Monitor after every pressure relief episode		Operated with instrument reading less than 500 ppm above background		(Connectors need not be identified if all connectors or length of a pipe is designated as a group). Develop and keep
Sampling Connection Systems (63.166)	Must be equipped with closed purge, closed loop, or closed vent system. Shall return fluid to process line	Initially				a schedule for monitoring connectors and valves subject to the standards for connectors in gas/vapor and light liquid service Develop a list of compressors designated as operating at less than 500 ppm above the background. Develop a list of identification numbers of pressure relief devices in HAP

Summary of Equipment Leak Requirements for Subpart GGG ¹						
Equipment Pharma MACT/HON	Design Requirements/ Exemptions	Monitoring Frequency	Method	Leak Limit	Calculations	Recordkeeping Requirements (40 CFR §63.1255(g))
						equipped with rupture discs.

¹ Does not summarize the requirements of 40 CFR Part 63.169, standards for pumps, valves, connectors, and agitators in heavy liquid service, instrumentation systems; and pressure relieve devices <u>in liquid</u> service because these requirement do not apply to FutureFuel

Open-Ended Valves or Lines (§63.1255(d))	Must be equipped with flanges, plugs, or another valve If poses a safety hazard, is designed to open automatically, or if equipped with double block and bleed exempt by 40 CFR Part 63.1255(d)(4)- (6)	Initially				Develop a list of instrumentation systems used to comply with PAI regulations. For dual mechanical seal systems, record design criteria and changes. Keep a list of equipment designated as unsafe, difficult, or inaccessible to monitor, and a copy of plan to monitor these devices. Keep a list of any connectors removed or added to the process and documentation of the integrity of the weld for any removed connectors. Keep dates of visual inspections Keep records of initial pressure tests of compressors and
Valves in Gas/Vapor and Light Liquid Service (§63.1255(e))		Initial survey within 1 year of compliance date	Method 21 of 40 CFR Part 60 Appendix A	500 ppm	Calculate Leakers per 40 CFR Part 63.1255(e) (5)	
		>2% of leakers - monthly		500 ppm	Calculate Leakers	
		<2% of leakers - quarterly		500 ppm	Calculate Leakers	
		<1%- once/2 quarters		500 ppm	Calculate Leakers	pressure relief valves.

	<0.5%- once/4 quarters <0.25%- every 2 years		500 ppm 500 ppm	Calculate Leakers Calculate Leakers	Keep a record background and initial reading. Keep design data for closed vent systems
Connectors in Gas/Vapor and in Light Liquid Service (§63.174)	Once within a year of compliance date <0.5%- once/4 quarters <0.25% - every 2 years	Method 21 of 40 CFR Part 60 Appendix A	500 ppm	Calculate Leakers Per 40 CFR Part 63.174(h)(3)(i) Calculate Leakers Calculate Leakers	Keep records of components in heavy liquid service, including analysis used to determine heavy liquid status. Maintain records of
Agitators in Gas/Vapor and Light Liquid Service (§63.1255(c)	Quarterly with instrument Weekly visual inspection	Method 21 of 40 CFR Part 60 Appendix A	10,000 ppm		exempt components

The following are key exemptions provided for the Subpart GGG standards for equipment leaks:

- i. Equipment that is intended to operate in OHAP service for less than 300 hours for a calendar year. [Reg.19.304 and 40 C.F.R. § 63.1255(d)(4)(viii)]
- ii. Equipment that is in vacuum service, which is operated at an internal pressure at least 5 kPa (0.725 psia) below ambient pressure, is excluded from the equipment leaks provisions of Subpart GGG. [Reg.19.304 and 40 C.F.R. § 63.1255(a)(8)]
- iii. Lines and equipment not containing process fluids are not subject to the LDAR requirements. Utilities and other non-process lines, such as heating and cooling systems which do not combine their materials with those in the processes they serve, are not considered part of a process and are not subject. [Reg.19.304 and 40 C.F.R. § 63.1255(a)(5)]

Consistency with other regulation

c. After the compliance date for a process, equipment subject to both §63.1255(a)(2) and either 40 CFR Part 60 and Part 61 will be required to only comply with the provisions of Subpart GGG. [Reg.19.304 and 40 C.F.R. § 63.1255(a)(2)]

- d. The provisions of §63.1(a)(3) of Subpart A do not alter the provisions in §63.1255(a)(2). [Reg.19.304 and 40 C.F.R. § 63.1255(a)(4)]
- e. The permittee shall comply with all applicable portions of §63.1255(b) though (h), including all recordkeeping, reporting, and monitoring requirements necessary for submitting information required in the Notification of Compliance Status report under §63.1260(f). [Reg.19.304 and 40 C.F.R. § 63.1255(b)]

STANDARDS: WASTEWATER

OCI 15 The permittee shall comply with the general wastewater requirements §63.1256(a)(1) through (3), and the maintenance wastewater provisions of §63.1256(a)(4). The permittee may transfer wastewater to a treatment operation not owned by the permittee in accordance with §63.1256(a)(5). [Reg.19.304 and 40 C.F.R. § 63.1256]

Identification of wastewater that requires control

a. The permittee shall comply with the requirements in §63.1256(a)(1) (i) [determine characteristics of a wastewater stream] or (ii) [designate wastewater as affect wastewater] to determine whether a wastewater stream is an affected wastewater stream that requires control for soluble and/or partially soluble HAP compounds or to designate the wastewater stream as an affected wastewater stream, respectively. The permittee may use a combination of approaches in §63.1256(a)(1)(i) and (ii) for different affected wastewater generated at the source. [Reg.19.304 and 40 C.F.R. § 63.1256(a)(1)]

Requirements for affected wastewater

- b. The permittee shall comply with the applicable requirements for wastewater tanks, surface impoundments, containers, individual drains, systems, and oil/water separators as specified in §63.1256(b) through (f), except as provided in §63.1256(g)(3) [biological treatment process]. [Reg.19.304 and 40 C.F.R. § 63.1256(a)(2)(i)]
- c. The permittee shall comply with the applicable requirements for control of soluble and partially soluble compounds as specified in §63.1256(g) [performance standard for processes managing wastewater and/or residuals removed from wastewater]. Alternatively, the permittee may elect to comply with the treatment provisions specified in §63.1256(a)(5) [offsite treatment or onsite treatment not owned/operated by the source]. [Reg.19.304 and 40 C.F.R. § 63.1256(a)(2)(ii)]

- d. The permittee shall comply with the applicable monitoring and inspection requirements in §63.1258 [monitoring requirements]. [Reg.19.304 and 40 C.F.R. § 63.1256(a)(2)(iii)]
- e. The permittee shall comply with the applicable recordkeeping and reporting requirements in §63.1259 [recordkeeping] and §63.1260 [reporting]. [Reg.19.304 and 40 C.F.R. § 63.1256(a)(2)(iv)]

Requirements for multi-phase discharge

f. The permittee shall not discharge a separate phase that can be isolated through gravity separation from the aqueous phase to a waste management or treatment unit, unless the stream is discharged to a treatment unit in compliance with §63.1256(g)(13) [treatment in RCRA unit option]. [Reg.19.304 and 40 C.F.R. § 63.1256(a)(3)]

Maintenance wastewater requirements

g. The permittee shall comply with the requirements of §63.1256(a)(4)(i) through (iv) for maintenance wastewater containing partially soluble or soluble HAP listed in Tables 2 and 3 of Subpart GGG. Maintenance wastewater is exempt from all other provisions of Subpart GGG. [Reg.19.304 and 40 C.F.R. § 63.1256(a)(4)]

Offsite treatment or onsite treatment not owned or operated by the source

h. The permittee may elect to transfer affected wastewater streams or a residual removed from such affected wastewater to an onsite treatment operation not owned or operated by the owner or operator of the source generating the wastewater or residual, or to an offsite treatment operation. [Reg.19.304 and 40 C.F.R. § 63.1256(a)(5)]

Wastewater tanks

i. The permittee shall comply with the requirements of either §63.1256(b)(1) or (2) of Subpart GGG as specified in Table 6 of this subpart for each wastewater tank that receives, manages, or treats affected wastewater or a residual removed from affected wastewater.

The permittee does not have wastewater tanks associated with the present pharmaceutical processes. This condition does not apply until the permittee places tanks into service as wastewater tanks. [Reg.19.304 and 40 C.F.R. § 63.1250(b)]

Surface impoundments

j. The permittee shall comply with §63.1256(c)(1),(2), and (3) of Subpart GGG for each surface impoundment that receives, manages, or treats affected wastewater or a residual removed from affected wastewater.

The permittee does not treat affected wastewaters or residuals in surface impoundments. This provision does not apply until such a time as the permittee chooses this treatment option. [Reg.19.304 and 40 C.F.R. § 63.1256(c)]

Containers

k. The permittee shall comply with the requirements of §63.1256(d)(1) through (5) of Subpart GGG for each container that receives, manages, or treats affected wastewater or a residual removed from affected wastewater. [Reg.19.304 and 40 C.F.R. § 63.1256(d)]

Individual drain systems

1. The permittee shall comply with the requirements of §63.1256(e)(1), (2), and (3), or with §63.1256(e)(4), (5), and (6) of Subpart GGG for each individual drain system that receives or manages affected wastewater or a residual from affected wastewater.

The permittee does not have individual drain systems associated with the present pharmaceutical process. This condition does not apply unless the permittee installs individual drain systems meeting the applicability criteria. [Reg.19.304 and 40 C.F.R. § 63.1256(e)]

Oil/water separators

m. The permittee shall comply with the requirements for oil/water separators that receives, manages, or treats affected wastewater or a residual removed from affected wastewater.

The permittee does not have oil/water separators associated with the pharmaceutical processes. This condition does not apply until such a time as the permittee implements this equipment. [Reg.19.304 and 40 C.F.R. § 63.1256(f)]

Performance standards for treatment processes managing wastewater and/or residuals removed from wastewater

n. The permittee shall comply with the requirements in §63.1256(g)(1) through (6) of Subpart GGG. Where multiple compliance options are provided, the options may be used in combination for different wastewater and/or for different compounds (e.g. soluble versus partially soluble compounds) in the same

wastewater, except where otherwise provided in Subpart GGG. Once affected wastewater or a residual removed from affected wastewater has been treated in accordance with Subpart GGG, it is no longer subject to the requirements of Subpart GGG. [Reg.19.304 and 40 C.F.R. § 63.1256(g)]

Existing source

i. For a wastewater stream at an existing source that exceeds or is designated to exceed the concentration and load criteria in §63.1256(a)(1)(i)(A), the permittee shall comply with a control option in §63.1256(g)(8) [wastewater containing partially soluble HAP compounds]. For a wastewater stream at an existing source that exceeds the concentration and load criteria in either §63.1256(a)(1)(i)(B) or (C), the permittee shall comply with the control option in §63.1256(g)(8) and a control option in §63.1256(g)(9) [wastewater containing soluble HAP].

As an alternative to the control options in 63.1256(g)(8) and (9), the permittee may comply with a control option in either 63.1256(g)(10) [enhanced bio-treatment], (11) [95% mass reduction for biological treatment processes], or (13) [treatment in a RCRA unit], as applicable.

The permittee has initially chosen 63.1256(g)(13) [RCRA unit option] as the control option. [Reg.19.304 and 40 C.F.R. 63.1256(g)(1)]

Biological treatment process

ii. Biological treatment processes in compliance may be either open or closed biological treatment processes as defined in §63.1251. [Reg.19.304 and 40 C.F.R. § 63.1256(g)(3)]

Performance tests and Design evaluation

iii. If the RCRA option specified in §63.1256(g)(13) or the enhanced biological treatment process for soluble HAP compounds option in §63.1256(g)(10) is selected, neither a design evaluation nor a performance test is required. For any other nonbiological treatment process, and for closed biological treatment processes as defined in §63.1251, the permittee shall conduct either a design evaluation as specified in §63.1257(e)(2)(ii) or performance test as specified in §63.1257(e)(2)(iii). For each open biological treatment process as defined in §63.1257(e)(2)(iii). For each open biological treatment process as defined in §63.1257(e)(2)(iii)(E) or (F). [Reg.19.304 and 40 C.F.R. § 63.1256(g)(4)]

Control device requirements

iv. When gases are vented from the treatment process, the permittee shall comply with the applicable control device requirements in §63.1256(h) [control device requirements] and §63.1257(e)(3) [test methods and compliance procedures – control device requirements], and the applicable leak inspection provisions specified in §63.1258(h) [leak inspection provisions for vapor suppression equipment]. This requirement is additional to the requirements for treatment systems specified in §63.1256(g)(8) [wastewater containing partially soluble HAP] and (14) [residuals]. This requirement does not apply to any open biological treatment process that meets the mass removal requirement. [Reg.19.304 and 40 C.F.R. § 63.1256(g)(5)]

Residuals: general

v. When residuals result from treating affected wastewater, the permittee shall comply with the requirements for residuals specified in §63.1256(g)(14).

The permittee's current selected wastewater treatment process does not generate residuals. This condition does not apply until such time that the permittee selects an applicable treatment option that produces a residual. [Reg.19.304 and 40 C.F.R. § 63.1256(g)(6)]

Treatment using a series of treatment processes

vi. In all cases where the wastewater provisions of Subpart GGG allow or require the use of a treatment process or control device to comply with emissions limitations, the permittee may use multiple treatment processes or control devices, respectively. For combinations of treatment processes where the wastewater stream is conveyed by hard-piping, the permittee shall comply with either §63.1256(g)(7)(i) [compliance across the combination of all treatment units or control devices in series], or (ii) [compliance across individual units]. For combinations of treatment processes where the wastewater stream is not conveyed by hard-piping, the permittee shall comply with the requirements in §63.1256(g)(7)(ii). For combinations of control devices, the permittee shall comply with the requirements of §63.1256)(g)(7)(i) of Subpart GGG.

The permittee shall identify, and keep a record of, the combination of treatment processes, including identification of the first and last treatment process. The permittee shall include this information as part of the treatment process description reported in the Notification of Compliance status report. [Reg.19.304 and 40 C.F.R. § 63.1256(g)(7)]

Treatment in RCRA unit option

> vii. The permittee shall treat the affected wastewater or residual in a unit identified in, and complying with, §63.1256(g)(13)(i), (ii), or (iii) of Subpart GGG. These units are exempt from the design evaluation or performance tests requirements specified in §63.1256(g)(4) [performance tests and design evaluations] and §63.1257(e)(2) [compliance with treatment unit control provisions], and from the monitoring requirements specified in §63.1256(a)(2)(iii) [requirements for affected wastewater], as well as the recordkeeping and reporting requirements associated with monitoring and performance tests.

This is the initial compliance option performance standard the permittee has chosen for the management of affected wastewaters. [Reg.19.304 and 40 C.F.R. \S 63.1256(g)(13)]

Residuals

viii. When residuals are generated, the permittee shall control for air emissions by complying with §63.1256(b) through (f) of Subpart GGG, and by complying with one of the provisions in §63.1256(g)(14)(i) through (iv).

The permittee's current selected wastewater treatment option process does not generate residuals. This condition does not apply until the permittee selects a wastewater treatment option that produces a residual. [Reg.19.304 and 40 C.F.R. § 63.1256(g)(14)]

Wastewater control devices

o. For each control device or combination of control devices used to comply with the provisions of §63.1256(b) through (f) and §63.1256(g)(5) [control device requirements], the permittee shall operate and maintain the control device or combination of control devices in accordance with the requirements of §63.1256(h)(1) through (5) of Subpart GGG. [Reg.19.304 and 40 C.F.R. § 63.1256(h)]

TEST METHODS AND COMPLIANCE PROCEDURES: GENERAL

OCI 16 The permittee is subject to the following requirements of 40 CFR §63.1257:

a. Except as provided in §63.1257(a)(5), the procedures specified in §63.1257(c) [storage tanks], (d) [process vents], (e) [wastewater], and (f) [pollution prevention] of Subpart GGG, are required to demonstrate initial compliance with §63.1253 [tanks], §63.1254 [process vents], §63.1256 [wastewater] and §63.1252(3) [heat exchangers], respectively. The provision in §63.1257(a)(2)

through (3) apply to performance tests that are specified in §63.1257(c) [tanks], (d) [process vents], and (e) [wastewater]. The provisions in §63.1257(a)(5) of this section are used to demonstrate initial compliance with the alternative standards specified in §63.1253(d) [tanks] and §63.1254(c) [new source alternative standards]. The provisions in §63.1257(a)(6) [initial compliance with the 20 ppmv limit] are used to comply with the outlet concentration requirements specified in §63.1253(c) [tanks], §63.1254(a)(2)(i) [process vent process-based annual mass limit] and §63.1254(a)(3)(ii)(B) [individual vent emission reduction], §63.1254(b)(i) [new sources], and §63.1256(h)(2) [control devices]. [Reg.19.304 and 40 C.F.R. § 63.1257(a)]

Test methods

b. When testing is conducted to measure emissions from an affected source, the test methods specified in §63.1257(b)(1) through (10) shall be used. [Reg.19.304 and 40 C.F.R. § 63.1257(b)]

Initial compliance with storage tanks

c. Initial compliance with the outlet concentration requirement of §63.1253(d) is demonstrated by fulfilling the requirements of §63.1257(a)(5).

The permittee does not currently operate any storage tank meeting the definition of PMPU storage tank. Therefore, the permittee is not currently subject to the storage tank standards of this subpart. The requirements of §63.1253 [storage tanks] do not apply until such time the permittee either installs or activates a tank for use in an applicable Pharmaceuticals Manufacturing Process Unit. Upon installing or activating a storage tank, which would be subject to this subpart, the permittee must at that time comply with the provisions of §63.1253, as well as the initial compliance provisions in §63.1257(c). [Reg.19.304 and 40 C.F.R. § 63.1257(c)]

Initial compliance with process vent provisions

d. The permittee shall demonstrate compliance using the procedures described in §63.1257(d)(1) through (4) for the process vent standards in §63.1254 [process vents]. [Reg.19.304 and 40 C.F.R. § 63.1257(d)]

Compliance with wastewater provisions

e. The wastewaters being treated in a RCRA unit are exempt from the design evaluation or performance tests requirements specified in §63.1256(g)(4) [performance testing and design evaluations] and §63.1257(e)(2), and from the monitoring requirements in §63.1256(a)(2)(iii) [requirements for affected wastewater], as well as the recordkeeping and reporting requirements associated

with performance tests. [Reg.19.304 and 40 C.F.R. § 63.1256(g)(13) and §63.1257(e)(2)]

The permittee has chosen the RCRA treatment option specified in (63.1256(g))(13). If the permittee opts for wastewater treatment controls other than allowed by (63.1256(g))(13), the permittee will be subject to the applicable requirements of (63.1257(e)) [compliance with wastewater provisions].

MONITORING REQUIREMENTS

OCI 17 The permittee is subject to the following requirements of 40 CFR §63.1258:

a. The permittee shall provide evidence of continued compliance with the standard as specified. During the initial compliance demonstration, maximum or minimum operating parameter levels, as appropriate, shall be established for emission sources that will indicate the source is in compliance. Test data, calculations, or information from the evaluation of the control device design shall be used to establish the operating parameter level. [Reg.19.304 and 40 C.F.R. § 63.1258(a)]

Monitoring of control devices

b. Except as provided by §63.1258(b)(1)(i), for each control device, the permittee shall install and operate monitoring devices and operate within the established parameter levels to ensure continued compliance with the standard. Monitoring parameters are specified for control scenarios in Table 4, and in §63.1258(b)(1)(ii) through (ix), of Subpart GGG. [Reg.19.304 and 40 C.F.R. § 63.1258(b)]

Averaging periods

i. Averaging periods for parametric monitoring levels shall be established according to §63.1258(b)(2)(i) through (iii). [Reg. 19.304 and 40 CFR §63.[Reg.19.304 and 40 C.F.R. § 63.1258(b)(2)]

Procedures for setting parameter levels for control devices used to control emissions – Large control devices

ii. For devices controlling greater than 10 tpy of HAP for which a performance test is required the parameter level must be established according to §63.1258(b)(3)(ii)(A) through (C). [Reg.19.304 and 40 C.F.R. § 63.1258(b)(3)(ii)]

Request approval to monitor alternative parameters

iii. The permittee may request approval to monitor parameters other than those required by §63.1258(b)(1)(ii) through (ix). The request shall be submitted according to the procedures in §63.8(f) [use of an alternative monitoring method] or included in the Precompliance report. [Reg.19.304 and 40 C.F.R. § 63.1258(b)(4)]

Exceedances of operating parameters

- iv. Exceedance of an operating parameter is defined as one of the following: [Reg.19.304 and 40 C.F.R. § 63.1258(b)(6)]
 - 1. If the parameter, averaged over the operating day or block, is below the minimum value established during the initial compliance determination;
 - 2. If the parameter, average over the operating day or block, is above the maximum value established during the initial compliance test; or
 - 3. Each loss of pilot flame for flares.

Excursions

- v. Excursions are defined as either of the two cases listed in §63.1258(b)(7)(i) or (ii) as follows: [Reg.19.304 and 40 C.F.R. § 63.1258(b)(7)]
 - 1. When the period of control devices operation is 4 hours or greater in an operating day and monitoring data are insufficient to constitute a valid hour of data as defined in §63.1258(b)(7)(iii), for at least 75 percent of the operating day.
 - 2. When the period of control device operation is less than 4 hours in an operating day and more than one of the hours during the period of operation does not constitute a valid hour of data due to insufficient monitoring data, or
 - 3. Monitoring data are insufficient to constitute a valid hour of data, as used in §63.1268(b)(7)(i) and (ii). If measured values are unavailable for any of the required 15-minute periods within the hour.

Violations

vi. Exceedances of parameters monitored according to §63.1258(b)(1)(ii), (iv) through (ix), and §63.1258(b)(5)(ii)(A) and (B), or excursions as defined by §63.1258(b)(7)(i) through (iii) constitute violations of the operating

limit according to (3.1258(b)(8)(i), (ii)), and (iv). Exceedances of the temperature limit monitored according to (3.1258(b)(1)(iii)) or exceedances of the outlet concentrations monitored according to the provisions of (3.1258(b)(1)(x)) constitute violations of the emission limit according to (3.1258(b)(1)(x)) constitute violations of the outlet concentration monitored according to (3.1258(b)(3)(i)), (ii), and (iv). Exceedances of the outlet concentration monitored according to (3.1258(b)(5)) constitute violations of the emission limit according to the provisions of (3.1258(b)(8)(ii)) and (iv) of Subpart GGG. [Reg.19.304 and 40 C.F.R. (3.1258(b)(8))]

Monitoring for emission limits

c. Compliance with §63.1254(a)(2) [process-based annual mass limit] shall demonstrate continuous compliance with the 900 and 1,800 kg/yr emission limits by calculating daily 365-day rolling summations of emissions. During periods of planned routine maintenance when emissions are controlled as specified in §63.1252(h), the permittee must calculate controlled emissions assuming the HAP emissions are reduced by 93 percent. If the permittee opts to switch compliance strategy from the 93 percent control requirement to the annual mass emission limit method, as described in §63.1254(a)(1)(i), the rolling summations beginning with the first day after the switch must include emissions from the past 365 days. [Reg.19.304 and 40 C.F.R. § 63.1258(c)]

Monitoring for equipment leaks

d. If the permittee is complying with the requirements of §63.1255 [LDAR], the monitoring requirements of §63.1255 shall be met. [Reg.19.304 and 40 C.F.R. § 63.1258(d)]

Inspection and monitoring of waste management units and treatment processes

e. The permittee shall comply with the inspection requirements specified in Table 7 of Subpart GGG for each wastewater tank, surface impoundment, container, individual drain system, and oil-water separator that receives, manages, or treats wastewater, a residual removed from wastewater, a recycled wastewater, or a recycled residual removed from wastewater. [Reg.19.304 and 40 C.F.R. § 63.1258(g)(1)]

Leak inspection provisions for vapor suppression equipment

f. The permittee shall comply with the requirements of §63.1258(h)(2) through (8), except as provided in §63.1258(h)(9) and (10), for each vapor collection system, closed-vent system, fixed roof, cover, or enclosure required to comply with this section. [Reg.19.304 and 40 C.F.R. § 63.1258(h)]

g. The permittee shall comply with the requirements of §63.1258(h)(10) in lieu of complying with the requirements of §63.1258(h)(2) through (8). The permittee shall maintain the closed-vent system below atmospheric pressure during normal RTO operation. The system shall be equipped with at least one pressure gauge or other pressure measurement device that can be read from a readily accessible location to verify that negative pressure is being maintained in the closed-vent system when the control devices are operating. [Reg.19.304 and 40 C.F.R. § 63.1258(h)(10)]

RECORDKEEPING REQUIREMENTS

- OCI 18 The permittee is subject to the following requirements of 40 CFR §63.1259:
 - a. The permittee shall comply with the recordkeeping requirements in Subpart A of Part 63, as specified in Table 1 of Subpart GGG and in §63.1259(a)(1) through (5). [Reg.19.304 and 40 C.F.R. § 63.1259(a)]

Records of equipment operation

b. The permittee shall keep up-to-date and readily accessible records of equipment operation as specified in §63.1259(b)(1) through (13), which conform to the sources applicability determination and operations. [Reg.19.304 and 40 C.F.R. § 63.1259(b)]

Records of operating scenarios

c. The permittee shall keep records of each operating scenario, which demonstrates compliance with Subpart GGG. [Reg.19.304 and 40 C.F.R. § 63.1259(c)]

Records of LDAR programs

d. A requirement to implement a leak detection and repair (LDAR) program under §63.1255, shall require the permittee to implement the recordkeeping requirements of §63.1255 of Subpart GGG. [Reg.19.304 and 40 C.F.R. § 63.1259(d)]

Records of emission averaging

e. If the permittee elects to comply with the requirements of §63.1252(d), the permittee shall maintain up-to-date records of the information specified in 63.1259(e)(1) through (4). [Reg.19.304 and 40 C.F.R. § 63.1259(e)]

Records of delay of repair

f. Documentation of a decision to use a delay of repair due to unavailability of parts, as specified in §63.1256(i) [delay of repair – wastewater], shall include a description of the failure, the reason additional time was necessary (including a statement of why replacement parts were not kept onsite and when delivery from the manufacturer is scheduled), and the date when the repair was completed. [Reg.19.304 and 40 C.F.R. § 63.1259(f)]

Record of wastewater stream and residual transfer

g. If the permittee transfers an affected wastewater stream or residual removed from an affected wastewater stream in accordance with §63.1256(a)(5) [offsite treatment or onsite treatment not owned/operated by the source] shall keep a record of the notice sent to the treatment operator stating that the wastewater stream or residual contains organic HAP, which are required to be managed and treated in accordance with the provisions of Subpart GGG. [Reg.19.304 and 40 C.F.R. § 63.1259(g)]

Records of extension

h. The permittee shall keep documentation of a decision to use an extension, as specified in §63.1256(b)(6)(ii) [wastewater tanks-floating roof] or (b)(9) [wastewater tanks – delay of repair], in a readily accessible location. The documentation shall include a description of the failure, documentation that alternate storage capacity is unavailable, and specification of a schedule of actions that will ensure that the control equipment will be repaired and the tank will be emptied as soon as possible. [Reg.19.304 and 40 C.F.R. § 63.1259(h)]

Currently, the permittee does not have wastewater tanks associated with the present pharmaceutical processes. This condition does not apply until the permittee places tanks into service as wastewater tank, upon which action this condition becomes effective.

Records of inspection

- i. The permittee shall keep records of all applicable inspection requirements as specified in §63.1259(i)(1) through (9). [Reg.19.304 and 40 C.F.R. § 63.1259(i)]
- OCI 19 The permittee is subject to the following requirements of 40 CFR §63.1260:
 - a. The permittee shall comply with the reporting requirements in §63.1260(b) through (l) of Subpart GGG. Applicable reporting requirements of §63.9 [notification requirements] and 63.10 [recordkeeping requirements] are also summarized in Table 1 of Subpart GGG. [Reg.19.304 and 40 C.F.R. § 63.1260(a)]

The Initial Notification report specified in §63.1260(b) was submitted to ADEQ on January 8, 1999. The Precompliance Report specified in §63.1260(e) was submitted to ADEQ on April 19, 2002.

Application for approval of construction or reconstruction

b. Any application for approval of construction of a new major affected source, the reconstruction of a major affected source, or the reconstruction of a major source such that the source becomes a major affected source subject to the standards shall be prepared in accordance with §63.5(d) [application for approval of construction or reconstruction]. [Reg.19.304 and 40 C.F.R. § 63.1260(c)]

Notification of CMS performance evaluation

c. Any owner/operator who is required by the Administrator to conduct a performance evaluation for a continuous monitoring system shall notify the Administrator of the date of the performance evaluation as specified in §63.8(e)(2). [Reg.19.304 and 40 C.F.R. § 63.1260(d)]

Notification of Compliance Status Report

d. The Notification of Compliance Status report required under §63.9 shall be submitted no later than 150 days after the compliance date of October 21, 2002 and shall include information specified in §63.1260(f)(1) through (7). [Reg.19.304 and 40 C.F.R. § 63.1260(f)]

Periodic reports

e. The permittee shall prepare Periodic Reports in accordance with §63.1260(g)(1) and (2) of Subpart GGG. [Reg.19.304 and 40 C.F.R. § 63.1260(g)]

Notification of process change

f. Except as specified in §63.1260(h)(2), whenever a process change is made, or a change in any of the information in the Notification of Compliance Status Report, the permittee shall submit the information specified in §63.1260(h)(1)(i) through (iv) with the next Periodic Report required under §63.1260(g). [Reg.19.304 and 40 CFR 63.1260(h)(1)]

Reports of LDAR programs

g. The permittee implementing the LDAR program specified in §63.1255 shall implement the reporting requirements in §63.1255 of Subpart GGG. Copies of all reports shall be retained as records for a period of 5 years, in accordance with

> the requirements of §63.10(b)(1) [record keeping and reporting]. [Reg.19.304 and 40 C.F.R. § 63.1260(j)]

Reports of emission averaging

h. If the permittee chooses to comply with the requirements of §63.1252(d) [emission averaging provisions], the implementation plan required by §63.1259(e) [records of emission averaging] must be submitted 6-months prior to the compliance date of the standard and the following information in §63.1260(k)(1) through (6) [reporting of emission averaging]. [Reg.19.304 and 40 C.F.R. § 63.1260(k)]

Notification of performance test and test plan

i. The permittee shall notify the Administrator of the planned date of a performance test at least 60-days before the test in accordance with §63.7(b) [notification of performance tests]. The permittee shall also submit the test plan required by §63.7(c) [quality assurance program] and the emission profile required by §63.1257(b)(8)(ii) with the notification of the performance test. [Reg.19.304 and 40 C.F.R. § 63.1260(l)]

40 CFR Part 63, Subpart MMM, National Emission Standards for Pesticide Active Ingredient Production

APPLICABILITY

OCI 20 A portion of this facility is subject to 40 CFR Part 63, Subpart MMM, National Emission Standards for Pesticide Active Ingredient Production. Applicable requirements include the following conditions. [Reg.19.304 and 40 C.F.R. § 63.1360]

Affected Source

a. The permittee is an affected source subject to 40 CFR Part 63, Subpart MMM as defined in 40 CFR §63.1360(a) [Applicability]. The source is an existing source with a compliance date of December 23, 2003. [Reg.19.304 and 40 C.F.R. § 63.1360(a)]

General Provisions Requirements

b. The provisions of Subpart A, specified in Table 1 of 40 CFR 63, Subpart MMM are the only general provisions that apply to an affected source subject to this subpart. [Reg.19.304 and 40 C.F.R. § 63.1360(c)]

Applicability of this Subpart except During Periods of Startup, Shutdown, and Malfunction

c. The permittee shall comply with all applicable requirements of 40 CFR 63, Subpart MMM except that emission limitations shall not apply during periods of startup, shutdown, and malfunction as defined in 40 CFR §63.1361, provided the conditions in 40 CFR §63.1360(e)(1) through (4) are met. [Reg.19.304 and 40 C.F.R. § 63.1360(e)]

Storage Vessel Applicability Determination

d. The permittee shall follow the procedures in 40 CFR §63.1360(f)(1) through (5) to determine whether a storage vessel is part of the affected PAI source.
[Reg.19.304 and 40 C.F.R. § 63.1360(f)]

Designating Production of an Intermediate as a PAI Process Unit

e. With the exception of 40 CFR §63.1360(d) [Exemptions]: The permittee may elect to designate production of any intermediate that does not meet the definition of integral intermediate as a PAI process unit. Storage vessels containing the intermediate is assigned to the PAI process unit according to the procedures in 40 CFR §63.1360(f) [storage vessel applicability determination]. Any process tank containing the intermediate is part of the process unit used to produce the intermediate. [Reg.19.304 and 40 C.F.R. § 63.1360(g)]

Applicability of Process Units Included in a Process Unit Group

f. The permittee may elect to develop process unit groups in accordance with 40 CFR §63.1360(h)(1). For PAI process units in these process unit groups, the permittee may comply with the provisions in the overlapping MACT standards as specified in 40 CFR §63.1360(h)(2) through (4), as an alternative means of demonstrating compliance with this subpart. [Reg.19.304 and 40 C.F.R. § 63.1360(h)]

Overlap with other MACT Standards

g. If the permittee is subject to the provisions of Subpart MMM and also subject to the provisions of any other subpart under 40 CFR Part 63, the permittee may elect, to the extent the subparts are consistent, under which subpart to maintain records and report to EPA. The permittee shall identify in the Notice of Compliance Status (NOCS) report required by 40 CFR §63.1368(f) under which subpart such records shall be maintained. [Reg.19.304 and 40 C.F.R. § 63.1360(i)(1)]

Overlap with RCRA Subparts AA, BB, and/or CC

OCI 21 The permittee may elect to comply with the monitoring, reporting, and recordkeeping requirements of 40 CFR 63, Subpart MMM or RCRA Subparts AA, BB, and/or CC for devices covered under both regulations. Compliance with the recordkeeping, monitoring, and reporting requirements in 40 CFR Parts 264 and/or 265 shall constitute compliance with the monitoring, reporting and recordkeeping of Subpart MMM. The permittee shall identify in the NOCS report required in §63.1368(f) the authority under which compliance is demonstrated. [Reg.19.304 and 40 C.F.R. § 63.1360(i)(2)]

Overlap with NSPS Kb

a. The permittee is only required to comply with the provisions of Subpart MMM for Group 1 and Group 2 storage vessels that are also subject to the requirements of 40 CFR 60, Subpart Kb. [Reg.19.304 and 40 C.F.R. § 63.1360(i)(3)]

Overlap with Subpart I

b. If the permittee has equipment in a process unit subject to 40 CFR 63, Subpart I; the permittee may elect to comply with either Subpart MMM, or 40 CFR 63, Subpart H. The permittee shall identify in the NOCS report required in §63.1368(f) the elected option of compliance. [Reg.19.304 and 40 C.F.R. § 63.1360(i)(4)]

Overlap with RCRA Regulations for Wastewater

c. If the permittee has affected wastewater streams subject to 40 CFR 260 through 272, compliance shall be based on the more stringent control requirements and the more stringent testing, monitoring, recordkeeping and reporting requirements that overlap between the requirements of Subpart MMM and Parts 260 through 272. The permittee shall keep a record of the information used to determine which requirements are the most stringent and shall submit this information if requested by the Administrator. [Reg.19.304 and 40 C.F.R. § 63.1360(i)(5)]

Overlap with NSPS Subparts III, NNN, and RRR

OCI 22 If the permittee has any process vent subject to Subpart MMM that is also subject to 40 CFR 60, Subparts III, NNN, or RRR and elects to reduce organic HAP emissions from the process vent by 98% as specified in §63.1362(b)(2)(iii)(A), then the permittee is only required to comply with Subpart MMM. Otherwise the permittee shall comply with Subpart MMM and the provisions of 40 CFR 60 Subparts III, NNN, and RRR as applicable. [Reg.19.304 and 40 C.F.R. § 63.1360(i)(6)]

Meanings of Periods of Time

a. All terms of Subpart MMM that define a period of time for completion of required tasks (e.g., weekly, monthly, quarterly, annual), unless specified otherwise in §63.1360 [Applicability], or subsection that imposes the requirement, refer to standard calendar periods of time. [Reg.19.304 and 40 C.F.R. § 63.1360(j)]

DEFINITIONS

OCI 23 Terms used in Subpart MMM are defined in the CAA, in Subpart A of 40 CFR 63, or in §63.1361. If the same term is defined in Subpart A and in §63.1361, it shall have the same meaning given in §63.1361 for the purposes of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.1361]

STANDARDS

- OCI 24 The permittee is subject to the following requirements of 40 CFR §63.1362:
 - a. Affected sources subject to Subpart MMM shall control HAP emission to the levels specified in §63.1362 [Standards: General] and in §63.1363 [Standards: Equipment Leaks], as summarized in Table 2 of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.1362(a)]

Process Vents

b. Subpart MMM existing sources shall comply with the requirements of §63.1362(b)(2) and (3). New sources shall comply with the requirements of §63.1362(b)(4) and (5). Compliance with §63.1362(b)(2) through (5) shall be demonstrated through the applicable test methods and initial compliance procedures in §63.1365 and the monitoring requirements in §63.1366. [Reg.19.304 and 40 C.F.R. § 63.1362(b)(1)]

Organic HAP Emissions from Existing Sources

- c. Existing effected sources must comply with the requirements in either §63.1362(b)(2)(i), or with §63.1362(b)(2)(ii) through (iv). [Reg.19.304 and 40 C.F.R. § 63.1363(b)(2)]
 - i. The uncontrolled organic HAP (OHAP) emission rate shall not exceed 0.15 Mg/yr from the sum of all process vents within a process. [Reg.19.304 and 40 C.F.R. § 63.1362(b)(2)(i)]

The permittee has chosen to not comply with the option under (63.1362(b)(2)(i)) at the present time, but reserves the ability to switch to this option at a later date providing proper notification under (63.1368(f)).

ii. Except as provided in §63.1361(b)(2)(ii)(B): Uncontrolled OHAP emissions from a process vent shall be reduced by 98% by weight or greater if the flow-weighted average flow rate for the vent, as calculated using Equation 1 is less than or equal to the flow rate using Equation 2 as specified in §63.1362(b)(2)(ii). [Reg.19.304 and 40 C.F.R. § 63.1362(b)(2)(ii)(A)]

The requirement under 63.1362(b)(2)(ii)(A) is not applicable at the present time, but the permittee may switch to this option at a later date providing proper notification under §63.1368(f)

iii. Control devices installed on or before November 10, 1997 on a process vent subject to §63.1362(b)(2)(ii)(A), and reducing inlet emissions of the total organic HAP by greater than 90% by weight, but less than 98% by weight, must be operated to reduce inlet emissions of total organic HAP by 90% weight or greater. [Reg.19.304 and 40 C.F.R. § 63.1362(b)(2)(ii)(B)]

This option does not apply at present to the permittee's operations. The permittee may comply with this option at a later date providing proper notification under §63.1368(f).

- iv. The permittee shall reduce, uncontrolled organic HAP emissions from the sum of all process vents within a process shall be reduced by 90% or greater by weight, excluding process vents that are subject to §63.1362(b)(2)(ii). [Reg.19.304 and 40 C.F.R. § 63.1362(b)(2)(iii)]
- v. As an alternative to §63.1362(b)(2)(ii) and (iii), uncontrolled OHAP emissions from any process vent may be reduced in accordance with and of the provisions in §63.1362(b)(2)(iv)(A) through (D) as listed below. All remaining process vents must be controlled in accordance with §63.1362(b)(2)(ii) and (iii). [Reg.19.304 and 40 C.F.R. § 63.1362(b)(2)(iv)]
 - 1. To outlet concentrations less than or equal to 20 ppmv; or
 - 2. By a flare that meets the requirements of §63.11(b); or
 - 3. By a control device specified in §63.1365(a)(4); or
 - 4. In accordance with the alternative standard specified in §63.1362(b)(6).

HCL and CL Emissions from Existing Sources

- d. The permittee shall comply with either of the following emission reduction requirements for HCL and CL from existing process vents:
 - i. The uncontrolled HCL and CL emissions, including HCL generated from the combustion of halogenated process vent emissions, from the sum of all process vents within a process shall not exceed 6.8 Mg/yr; or
 - ii. HCL and CL emissions, including HCL generated from combustion of halogenated process vent emissions, from the sum of all process vents within a process shall be reduced by 94% or greater or to outlet concentrations less than or equal to 20 ppmv.

[Reg.19.304 and 40 C.F.R. § 63.1362(b)(3)]

Alternative Standard - Process Vents

e. As an alternative to the standards for existing and new process vent emission control requirements, the permittee may route emissions from a process vent to a combustion control device achieving an outlet TOC concentration (calibrated on methane or the predominate HAP) of 20 ppmv or less, and an outlet concentration of HCL and CL of 20 ppmv or less. If routing to a non-combustion control device or series of control devices, the control devices(s) must achieve an outlet TOC concentration of 50 ppmv or less, and an outlet concentration of HCL and CL of 50 ppmv or less. Process vents not routed to a control device must be controlled according to §63.1362(b)(2)(ii) through (iv), §63.1362(b)(3)(ii), §63.1362(b)(4)(ii), §63.1362(b)(5)(ii) or (iii) of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.1362(b)(6)]

Storage Vessels

f. The permittee shall either determine the group status of a storage vessel or designate it as a Group 1 storage vessel. Storage vessels designated as Group 1 are not required to have the maximum true vapor pressure of the material stored to be determined. [Reg.19.304 and 40 C.F.R. § 63.1362(c)]

Storage Vessel Standard for Existing Sources

- i. Except as specified in §63.1362(c)(4), (5), and (6), the permittee shall equip each Group 1 storage vessel at an existing affected source with one of the following:
 - 1. A fixed roof and internal floating roof; or
 - 2. An external floating roof; or
 - 3. An external floating roof converted to an internal floating roof; or

- 4. A closed-vent system meeting the requirements in §63.1363(j) [closed-vent systems] and a control device that meets any of the following conditions:
- 5. Reduces organic HAP emissions by 95% weight or greater; or
- 6. Reduces organic HAP emissions to outlet concentrations of 20 ppmv or less; or
- 7. Is a flare meeting the requirements of §63.11(b); or
- 8. Is a control device specified in §63.1365(a)(4) [boiler/process heater].

[Reg.19.304 and 40 C.F.R. § 63.1362(c)(2)(i) through (iv)]

Storage Vessel Standard for New Sources

ii. Presently the permittee is not subject to the new source requirements for storage vessels. The new source provisions do not apply until such time that the permittee installs or modifies a PAI storage vessel to meet the definition of new affected source as defined in §63.1361. Group 1 storage vessels at a new source shall equip the affected storage vessel with any of the controls specified in §63.1362(c)(2)(i) through (iv) listed above. [Reg.19.304 and 40 C.F.R. § 63.1362(c)(3)]

Storage Vessels – Alternative Standard

iii. As an alternative to the standards for existing and new storage vessel emission control requirements, the permittee may route emissions from a storage vessel to a combustion control device achieving an outlet TOC concentration (calibrated on methane or the predominate HAP) of 20 ppmv or less, and an outlet concentration of HCL and CL of 20 ppmv or less. If routing to a non-combustion control device or series of control devices, the control devices(s) must achieve an outlet TOC concentration of 50 ppmv or less, and an outlet concentration of HCL and CL of 50 ppmv or less. [Reg.19.304 and 40 C.F.R. § 63.1362(c)(4)]

Storage Vessel Planned Routine Maintenance

iv. The permittee is exempt from the storage vessel existing and new source standards and the alternative standard during periods of planned routine maintenance of the control device that does not exceed 240 hours/yr. The permittee may submit an extension to the Administrator requesting an extension of this time limit to 360 hours/yr. The request must explain why the extension is needed and it must indicate that no material will be added to the storage vessel between the time the 240-hour limit is exceeded and this control device is again operational. The request must

> be submitted at least 60-days before the 240 hour limit will be exceeded. [Reg.19.304 and 40 C.F.R. § 63.1362(c)(5)]

Storage Vessel - Vapor Balancing Alternative

v. As an alternative to the storage vessel existing and new source standards, the permittee may implement vapor balancing as specified in §63.1362(c)(6)(i) through (vii). [Reg.19.304 and 40 C.F.R. § 63.1362(c)(6)]

Storage Vessel Compliance Provisions

vi. Compliance with storage vessel existing and new source standards is demonstrated using the initial compliance procedures in §63.1365(d) [Initial compliance with storage vessel provisions] and the monitoring requirements in §63.1366 [Monitoring and inspection requirements]. Compliance with outlet concentrations in the alternative standard shall be determined by the initial compliance provisions in §63.1365(a)(5) and the continuous emission monitoring requirements in §63.1366(b)(5). [Reg.19.304 and 40 C.F.R. § 63.1362(c)(7)]

WASTEWATER

OCI 25 The permittee shall comply with the requirements of 40 CFR §63.132 through §63.147, with the differences noted in §63.1362(d)(1) through (16) for the purpose of compliance with Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.1362(d)]

Definitions

- a. When the term "process wastewater is referred to in §63.132 through §63.147 of Subpart G, the term "wastewater" as defined in §63.1361 shall apply for the purposes of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.1362(d)(7)]
- b. When the term "Group 1 wastewater stream" is used in §63.132 through §63.147 of Subpart G, the definition of "Group 1 wastewater stream" in §63.1361 shall apply for both new and existing sources for the purposes of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.1362(d)(8)]
- c. When the term "Storage vessel" is used in §63.119 through §63.123 of Subpart G, the definition of "storage vessel" in §63.1361 shall apply for the purposes of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.1362(d)(2)(i)]

Statement of Table 8 Non-Applicability

 d. The requirements in §63.132 through §63.147 for compounds listed on Table 8 of Subpart G shall not apply for the purposes of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.1362(d)(9)]

EXISTING SOURCE WASTEWATER PROVISIONS – GENERAL

OCI 26 The permittee shall comply with the requirements of §63.132(a)(1) through (3) no later than the applicable date of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.132(a)]

Determination of Group 1 or Group 2 Status for Table 9 Compounds

a. The permittee shall determine the Group 1 or Group 2 status for Table 9 compounds according to the requirements of §63.132(c). [Reg.19.304 and 40 C.F.R. § 63.132(c)]

Designation of a Group 1 Wastewater Stream

- b. The permittee may elect to designate a wastewater stream a Group 1 wastewater stream in order to comply with the requirements of §63.132(a)(1) or (b)(1) by following the procedures in §63.132(e). [Reg.19.304 and 40 C.F.R. § 63.132(e)]
- c. The permittee shall not discard liquid or solid organic materials with a concentration of greater than 10,000 ppm of Table 9 compounds (as determined by analysis of the stream composition, engineering calculations, or process knowledge, according to the provisions of §63.144(b) of this subpart) from a chemical manufacturing process unit to water or wastewater, unless the receiving stream is managed and treated as a Group 1 wastewater stream. This prohibition does not apply to materials from the activities listed in §63.132(f)(1) through (f)(4) below:
 - i. Equipment leaks;
 - ii. Activities included in maintenance or SSM plans;
 - iii. Spills; or
 - iv. Samples of a size not greater than reasonably necessary for the method of analysis that is used.

[Reg.19.304 and 40 C.F.R. § 63.132(f)]

Off-site Treatment not Owned or Operated by the Source

d. The permittee may elect to transfer Group 1 wastewater or residuals removed from Group 1 wastewater streams to an off-site treatment operation by complying with the requirements of §63.132(g). [Reg.19.304 and 40 C.F.R. § 63.132(g)]

PROCESS WASTEWATER PROVISIONS – WASTEWATER TANKS

- OCI 27 The permittee is subject to the following requirements of 40 CFR §63.133:
 - a. The permittee shall comply with either §63.133(a)(1) or (2), as specified in Table 10, for each wastewater tank that receives, manages, or treats a Group 1 wastewater stream or wastewater residual removed from a wastewater stream. [Reg.19.304 and 40 C.F.R. § 63.133(a)]
 - b. The maximum true vapor pressures in Table 10 shall be limited to the HAP listed in Table 9 to 40 CFR 63, Subpart G. [Reg.19.304 and 40 C.F.R. § 63.1362(d)(15)]
 - c. The permittee shall comply with the requirements of §63.133(a)(2)(i) for fixed-roof tanks. The fixed roof shall meet the requirements of §63.133(b)(1), the control device shall meet the requirements of §63.133(b)(2), and the control device shall meet the requirements of §63.133(b)(3). [Reg.19.304 and 40 C.F.R. § 63.133(b)]
 - d. The permittee shall inspect each wastewater tank initially, and semi-annually for improper work practices in accordance with 63.143, with the exception provided in §63.133(e)(2). [Reg.19.304 and 40 C.F.R. § 63.133(f)]
 - e. The permittee shall inspect each wastewater tank for control equipment failures as specified in §63.133(g)(1) through (3). [Reg.19.304 and 40 C.F.R. § 63.133(g)]
 - f. The permittee shall initiate first efforts to repair control equipment failures or improper work practices within 5 calendar days, and complete the repairs within 45 calendar days. Two extensions of up to 30 additional calendar days each may be utilized provided documentation supporting the decision as identified in §63.133(h) is maintained. [Reg.19.304 and 40 C.F.R. § 63.133(h)]

PROCESS WASTEWATER PROVISIONS – SURFACE IMPOUNDMENTS

OCI 28 The permittee is subject to the following requirements of 40 CFR §63.134:

The permittee shall comply with §63.134(a) through (d) for each surface impoundment that receives, manages, or treats a Group 1 wastewater stream or residual removed from a Group 1 wastewater stream. [Reg.19.304 and 40 C.F.R. § 63.134(a)]

Presently, the permittee does not treat Group 1 wastewaters or residuals in surface impoundments. This specific condition does not apply until such time as the permittee chooses this option.

PROCESS WASTEWATER PROVISIONS – CONTAINERS

OCI 29 The permittee is subject to the following requirements of 40 CFR §63.135:

The permittee shall comply with §63.135(b) through (f) for each container that receives, manages or treats a Group 1 wastewater stream or a residual from a Group 1 wastewater stream. [Reg.19.304 and 40 C.F.R. § 63.135(a)]

PROCESS WASTEWATER – INDIVIDUAL DRAIN SYSTEMS

OCI 30 The permittee is subject to the following requirements of 40 CFR §63.136:

The permittee shall comply with the requirements in §63.136(b) through (d), or with §63.136(e) through (g) for each individual drain system that receives or manages a Group 1 wastewater stream or residual from a Group 1 wastewater stream. [Reg.19.304 and 40 C.F.R. § 63.136(a)]

PROCESS WASTEWATER – OIL/WATER SEPARATORS

- OCI 31 The permittee is subject to the following requirements of 40 CFR §63.137:
 - a. The permittee shall comply with §63.137(c) and (d) for each oil-water separator that receives, manages, or treats a Group 1 wastewater stream or residual from a Group 1 wastewater stream, and shall maintain and operate a fixed roof and closed-vent system and control device as specified in §63.137(a)(1), and which meets the requirements of §63.137(b). [Reg.19.304 and 40 C.F.R. § 63.137(a)]
 - b. As an alternative to §63.137(a)(1), the permittee may elect to comply with the floating roof requirements in §63.137(a)(2), or an equivalent means of emission limitation as specified in §63.137(a)(3). [Reg.19.304 and 40 C.F.R. § 63.137(a)(2) and (3)]
 - c. The permittee shall inspect each oil-water separator initially, and semi-annually for improper work practices in accordance with §63.143. [Reg.19.304 and 40 C.F.R. § 63.137(d)]
 - d. The permittee shall inspect each wastewater tank for control equipment failures as specified in §63.137(e). [Reg.19.304 and 40 C.F.R. § 63.137(e)]
 - e. Except as provided in §63.140, when an improper work practice or control equipment failure is identified, the first attempt at repair shall be made no later

than 5 calendar days after identification and repair shall be completed within 45 calendar days. [Reg.19.304 and 40 C.F.R. § 63.137(f)]

PROCESS WASTEWATER PROVISIONS – PERFORMANCE STANDARDS FOR TREATMENT PROCESSES MANAGING GROUP 1 WASTEWATER STREAMS

OCI 32 The permittee is subject to the following requirements of 40 CFR §63.138:

General Requirements

a. The permittee shall comply with the requirements specified in §63.138(a)(1) through (6). Where multiple compliance options are provided, the options may be used in combination for different wastewater streams and/or different compounds in the same wastewater streams, except where otherwise provided in §63.138. Once a Group 1 stream or residual removed from a Group 1 stream has been treated in accordance with Subpart MMM, it is no longer subject to the requirements of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.138(a)]

Existing Source

i. The permittee is an existing source for the purpose of these requirements. If the wastewater stream is Group 1 for Table 9 compounds, the permittee shall comply with §63.138(b). [Reg.19.304 and 40 C.F.R. § 63.138(a)(1)]

New Source

ii. If the permittee becomes subject to the new source wastewater standard, the permittee shall comply with §63.138(a)(2) for Group 1 compounds in a Table 9, as determined by the procedures in §63.132. [Reg.19.304 and 40 C.F.R. § 63.138(a)(2)]

Control Options: Group 1 Wastewater Streams for Table 9 Compounds

b. The permittee shall comply with §63.138(b)(2) [other compliance options] for the control of Table 9 compounds. [Reg.19.304 and 40 C.F.R. § 63.138(b)]

Options:

- i. Operate a design steam stripper meeting the requirements of §63.138(d);
- ii. Percent mass removal option meeting the requirements of §63.138(e);
- iii. Required mass removal option meeting the requirements of §63.138(f);
- iv. 95% required mass removal option for biological treatment processes meeting the requirements of §63.138(g);

- v. Treatment in a RCRA unit option meeting the requirements of §63.138(h); or
- vi. One megagram total source mass flow rate option meeting the requirements of §63.138(i).

[Reg.19.304 and 40 C.F.R. § 63.138(b)(1) and (b)(2)]

Design Evaluations or Performance Tests for Treatment Processes

c. The permittee shall demonstrate compliance with the elected treatment option by following the requirements of §63.138(j), as applicable to the treatment option specified, with the exceptions provided in §63.138(j)(3) or §63.138(h). [Reg.19.304 and 40 C.F.R. § 63.138(j)]

Exemptions from Performance Testing and Design Analysis

- d. The provisions of §63.138(j)(1) [design evaluation mass balance] and (j)(2) [performance tests] do not apply to steam strippers which meet the requirements of §63.138(d). [Reg.19.304 and 40 C.F.R. § 63.138(j)(3)]
- e. The provisions of §63.138(h) [RCRA unit treatment option] exempt the permittee from the design evaluation and performance test requirements specified in §63.138(a)(3) and §63.138(j), and from the monitoring requirements specified in §63.132(a)(2)(iii) and §63.132(b)(3)(iii), as well as the recordkeeping and reporting requirements associated with monitoring and performance tests. [Reg.19.304 and 40 C.F.R. § 63.138(j) and §63.138(j)

Residuals

- f. The permittee shall control residuals from Group 1 wastewater streams by complying with §63.133 through §63.137 and by complying with one of the following options:
 - i. Recycle the residual to a production process or sell the residual for the purpose of recycling. Once a residual is returned to the production process, the residual is no longer subject to Subpart MMM;
 - ii. Return the residual to the treatment process;
 - iii. Treat the residual to destroy the total combined mass flow rate of Table 9 compounds by more than 99% or more; or
 - iv. Comply with the requirements for RCRA treatment options specified in §63.138(h).

[Reg.19.304 and 40 C.F.R. § 63.138(k)]

PROCESS WASTEWATER PROVISIONS - CONTROL DEVICES

OCI 33 The permittee is subject to the following requirements of 40 CFR §63.139:

The permittee shall operate and maintain control device or combination of control devices in accordance with §63.138(b) through (f) for control devices used to comply with the requirements of §63.133 through §63.138. [Reg.19.304 and 40 C.F.R. § 63.139(a)]

PROCESS WASTEWATER PROVISIONS – DELAY OF REPAIR

OCI 34 The permittee is subject to the following requirements of 40 CFR §63.140:

The permittee is allowed delays in repair of equipment for which a control equipment failure or a gap, crack, tear, or hole has been identified, provided the permittee complies with the exceptions specified in §63.140(a) through (c). [Reg.19.304 and 40 C.F.R. § 63.140(a) through (c)]

PROCESS WASTEWATER PROVISIONS – INSPECTIONS AND MONITORING OF OPERATIONS

OCI 35 The permittee is subject to the following requirements of 40 CFR §63.143:

- a. The permittee shall comply with the inspection requirements in Table 11 for each wastewater tank, surface impoundment, container, individual drain system, and oil-water separator that receives, manages, or treats a Group 1 wastewater stream, a residual removed from a Group 1 wastewater stream, a recycled Group 1 wastewater stream, or recycled residual removed from a Group 1 wastewater stream. [Reg.19.304 and 40 C.F.R. § 63.143(a)]
- b. The permittee shall comply with the monitoring requirements in Table 12 for each design steam stripper and biological treatment unit used to comply with §63.138. [Reg.19.304 and 40 C.F.R. § 63.143(b)]
- c. If the permittee elects to comply with Item 1 of Table 12, the permittee shall request approval to monitor appropriate parameters that demonstrate proper operation of the biological treatment unit. The request shall be submitted according to the procedures in §63.151(f) and shall include a description of the planned reporting and recordkeeping procedures. The basis for the selected monitoring frequencies and the methods used shall be included in the submittal. [Reg.19.304 and 40 C.F.R. § 63.143(c)]

- d. If the permittee elects to comply with Item 3 of Table 12, the permittee shall request approval to monitor appropriate parameters that demonstrate proper operation of the selected treatment process. The request shall be submitted according to the procedures in §63.151(f) and shall include a description of the planned reporting and recordkeeping procedures. [Reg.19.304 and 40 C.F.R. § 63.143(d)]
- e. For each control device used to comply with the requirements of §63.133 through §63.139, the permittee shall comply with the requirements of §63.139(d), and with the requirements of §63.143(e)(1), (e)(2), or (e)(3), except as provided in §63.143(e)(4) and (5). [Reg.19.304 and 40 C.F.R. § 63.143(e)]
- f. The permittee shall establish a range that indicates proper operation of the treatment process or control device for each parameter monitored in accordance with §63.143(c), (d), or (e). In order to establish the range, the permittee shall comply with the requirements of §63.146(b)(7)(ii)(A) and (b)(8)(ii). [Reg.19.304 and 40 C.F.R. § 63.143(f)]
- g. Monitoring equipment shall be installed, calibrated, and maintained according to the manufacturer's specifications or other written procedures that provide adequate assurance that the equipment would be reasonably expected to monitor accurately. [Reg.19.304 and 40 C.F.R. § 63.143(g)]

PROCESS WASTEWATER PROVISIONS – TEST METHODS AND PROCEDURES FOR DETERMINING APPLICABILITY AND GROUP 1/GROUP 2 DETERMINATION (Determining which Wastewater Streams Require Control)

OCI 36 The permittee is subject to the following requirements of 40 CFR §63.144:

The permittee shall comply with 63.144(a)(1) [determine Group 1 or Group 2 status] or (a)(2) [designate as Group 1] for each wastewater stream to determine which wastewater streams require control for Table 9 compounds. The permittee may use a combination of the approaches in 63.144(a)(1) or (a)(2) for different wastewater streams generated at the source. [Reg.19.304 and 40 C.F.R. § 63.144(a)]

- a. The permittee may determine the Group 1 and Group 2 status under §63.144(a)(1) and use the procedures in 63.144(b) to establish concentration limits, and 63.144(c) to determine flow rates. [Reg.19.304 and 40 C.F.R. § 63.144(a)(1)]
- b. The permittee may designate as a Group 1 wastewater stream a single wastewater stream or mixture of wastewater streams under §63.144(a)(2). The permittee is not required to determine the concentration of flow rate of each designated
Group 1 wastewater stream for the purposes of §63.144. [Reg.19.304 and 40 C.F.R. § 63.144(a)(2)]

PROCESS WASTEWATER PROVISIONS – TEST METHODS AND PROCEDURES TO DETERMINE COMPLIANCE

OCI 37 The permittee is subject to the following requirements of 40 CFR §63.145:

General

63.145 specifies the procedures for performance tests that are conducted to demonstrate compliance of a treatment process or a control device with the control requirements specified in 63.138. If conducting a design evaluation, the permittee shall comply with the requirements of 63.145(a)(1) and (a)(2). If conducting a performance test, the permittee shall comply with the requirements in 63.145(a) through (i). [Reg.19.304 and 40 C.F.R. 63.145(a)]

Performance Tests and Design Evaluations for treatment Processes

- a. If the permittee has chosen the design steam stripper option in §63.138(d), or RCRA option in §63.138(h) to comply with §63.138, neither a design evaluation nor a performance test is required. [Reg.19.304 and 40 C.F.R. § 63.145(a)(1)]
- b. If the permittee chooses to use any other non-biological treatment process, the permittee shall conduct either a design evaluation as specified in §63.138(j), or a performance test as required in §63.145. [Reg.19.304 and 40 C.F.R. § 63.145(a)(1)]
- c. If the permittee chooses to use a closed biological treatment process, the permittee shall conduct either a design evaluation according to §63.138(j), or a performance test according to §63.145. If using an open biological treatment system, the permittee shall conduct a performance test according to §63.145. [Reg.19.304 and 40 C.F.R. § 63.145(a)(1)]

PROCESS WASTEWATER PROVISIONS - REPORTING

OCI 38 The permittee is subject to the following requirements of 40 CFR §63.146:

a. For each waste management unit, treatment process, or control device used to comply with §63.138(b)(1), (c)(1), (d), (e), (f), or (g) for which the permittee seeks to monitor a parameter other than those specified in Tables 11, 12, or 13, the permittee shall submit a request for approval to monitor alternative

parameters according to the procedures in §63.8(f) of Subpart A, as referenced in §63.1366(b)(4). [Reg.19.304 and 40 C.F.R. § 63.146(a), 63.8(f), 63.1362(d)(3), and §63.1366(b)(4)]

- b. The permittee shall submit the information specified in §63.146(b)(1) through (b)(9) as part of the Notification of Compliance Status report required by §63.1368(f) of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.146(b), §63.1368(f), and §63.1362(d)(4)]
- c. The permittee shall submit as part of the Periodic Report required by §63.1368(g) the results of each inspection required by §63.143(a). Each Periodic Report shall include the date of the inspection, identification of each waste management unit in which a control equipment failure was detected, description of the failure, and description of the nature of and date the repair was made for each waste management unit that receives, manages, or treats a Group 1 wastewater stream or residual removed from a Group 1 wastewater stream. [Reg.19.304 and 40 C.F.R. § 63.146(c), §63.1362(d)(6) and §63.1368(g)]
- d. The permittee shall submit as part of the Periodic Report required by §63.1368(g) the information specified in §63.146(d)(1) through (3) for the monitoring required by §63.143(b), (c), and (d). [Reg.19.304 and 40 C.F.R. § 63.146(d) and §63.1368(g)]
- e. The permittee shall submit as part of the Periodic Report the information specified in §63.146(e)(1) or (e)(2) for each control device used to comply with §63.133 through §63.139. [Reg.19.304 and 40 C.F.R. § 63.146(e)]
- f. If the permittee utilizes and extension for delay or repair in accordance with §63.133(e)(2) or §63.133(h) the information shall be included in the next Periodic Report. [Reg.19.304 and 40 C.F.R. § 63.146(g)]

PROCESS WASTEWATER PROVISIONS - RECORDKEEPING

- OCI 39 The permittee is subject to the following requirements of 40 CFR §63.147:
 - a. If the permittee transfers a Group 1 wastewater stream or residual removed from a Group 1 stream in accordance with §63.132(g), the permittee shall keep a record of the notice sent to the treatment operating stating that the wastewater stream or residual contains organic HAP which are required to be managed and treated in accordance with the provisions of this subpart. [Reg.19.304 and 40 C.F.R. § 63.147(a)]
 - b. The permittee shall keep in a readily accessible location the records specified in §63.147(b)(1) through (8). [Reg.19.304 and 40 C.F.R. § 63.147(b)]

- c. The permittee shall keep records of the daily average value of each continuously monitored parameter for each operating day as specified in §63.1367, except as provided in §63.147(d)(1) through (3). [Reg.19.304 and 40 C.F.R. § 63.146(d) and §63.1362(d)(5)]
- d. If the permittee obtains approval to use a control device other than the one for which monitoring requirements are specified in §63.143, or to monitor parameters other than those specified in Tables 12 or 13, the Administrator will specify the appropriate recordkeeping requirements. [Reg.19.304 and 40 C.F.R. § 63.147(e)]
- e. If the permittee uses process knowledge to determine the annual average concentration of a wastewater stream as specified in §63.144(b)(3) and/or uses process knowledge to determine the annual average flow rate as specified in §63.144(c), and determines that the wastewater stream is not a Group 1 wastewater stream, the permittee shall keep in a readily accessible location the documentation of how process knowledge was used to determine the annual average concentration and/ or the annual average flow rate of the wastewater stream. [Reg.19.304 and 40 C.F.R. § 63.147(f)]

BAG DUMPS AND PRODUCT DRYERS

OCI 40 The permittee is subject to the following requirements of 40 CFR §63.1362:

- a. The following standards apply to bag dumps and product dryers:
 - i. The permittee shall reduce particulate matter emissions to a concentration not to exceed 0.01 gr/dscf from product dryers that dry PAI or integral intermediate that is a HAP. [Reg.19.304 and 40 C.F.R. § 63.1362(e)(1)]
 - ii. The permittee shall reduce particulate matter emissions to a concentration not to exceed 0.01 gr/dscf from bag dumps that introduce to a PAI process unit a feedstock that is a solid material and a HAP, excluding bag dumps where the feedstock contains HAP only as an impurity. [Reg.19.304 and 40 C.F.R. § 63.1362(e)(2)]
 - iii. The permittee shall control gaseous HAP emissions from product dryers and bag dumps in according to the provisions of the process vent requirements in §63.1362(b). [Reg.19.304 and 40 C.F.R. § 63.1362(e)(3)]

Heat exchange systems

b. With the exception of the conditions specified in 40 CFR 63, Subpart F, §63.104(a)(1) through (6), the permittee shall monitor each heat exchange system that is used to cool PAI process units that are part of an affected source as defined in §63.1360(a) according to one of the provisions in 40 CFR §63.104(b) or (c) of Subpart F. Whenever a leak is detected, the permittee shall comply with the requirements in 40 CFR 63, Subpart F, §63.104(d). Delay of repair of heat exchange systems for which leaks have been detected is allowed according to §63.104(e) of Subpart F. [Reg.19.304 and 40 C.F.R. § 63.1362(f)]

Pollution Prevention Alternative

c. Except as provided in §63.1362(g)(1) of Subpart MMM, for a process that has an initial startup prior to November 10, 1997, the permittee may choose to meet the pollution prevention alternative requirement specified in §63.1362(g)(2) or (3) for any PAI process unit, in lieu of the requirements specified in §63.1362(b) [Process vents], (c) [Storage vessels], (d) [Wastewater], and (e) [Bag dumps] and in §63.1363 [Equipment Leaks]. Compliance with the requirements of §63.1363(g)(2) and (3) shall be demonstrated through the procedures in §63.1365(g) and §63.1366(f). [Reg.19.304 and 40 C.F.R. § 63.1362(g)].

Emissions Averaging Provisions

d. Except as provided in §63.1362(h)(1) through (7), the permittee may choose to comply with the emission standards in §63.1362(b), (c), and (d) of this section by using emissions averaging procedures specified in §63.1365(h) for organic HAP emissions from any storage vessel, process or waste management unit this is part of an affected source subject to Subpart MMM. [§19.304 of Regulation 19 and §40 CFR 63.1362(h)]

Presently, the permittee does not choose to opt for the emissions averaging compliance method.

Opening of a Safety Device

e. Opening of a safety device, as defined in §63.1361 is allowed at any time conditions require it to avoid unsafe conditions. [Reg.19.304 and 40 C.F.R. § 63.1362(i)]

Closed-vent Systems

f. Presently, the permittee does not have a closed-vent system containing a bypass line that could divert a vent stream away from a control device used to comply with the requirements of §63.1362(b) [Process vents], (c) [Storage vessels], or (d) [Wastewater]. If any bypass line is installed in the future, the permittee shall

comply with the requirements of §63.1362(j)(1) and (2) [Closed-vent systems], and Table 2 of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.1362(j)]

The permittee operates regenerative thermal oxidizers (RTOs), which have emergency vent dampers meeting the definition of a safety device as specified in §63.1361. Bypass lines do not exist on the closed-vent system and control device configuration.

Exception for RCRA Treatment Units

- g. The permittee shall be exempt from the initial compliance demonstrations and monitoring provisions in §63.1365 [Test methods and initial compliance] and §63.1366 [Monitoring and inspections] and the recordkeeping and reporting requirements in §63.1367 [Recordkeeping] and §63.1368 [Reporting] for emissions from process vents, storage vessels, and waste management units that are discharging to the following devices:
 - i. A boiler or process heater burning hazardous waste for which the permittee has been issued a final permit under 40 CFR Part 270 and complies with the requirements of 40 CFR Part 266, Subpart H; or
 - ii. Has certified compliance with the interim status requirements of 40 CFR Part 266, Subpart H.
 - iii. A hazardous waste incinerator for which the permittee has been issued a final permit under 40 CFR Part 270 and complies with the requirements of 40 CFR Part 264, Subpart O, or has certified compliance with the interim status requirements of 40 CFR Part 265, Subpart O. [Reg.19.304 and 40 C.F.R. § 63.1362(1)]

STANDARDS – EQUIPMENT LEAKS

OCI 41 The permittee is subject to the following requirements of 40 CFR §63.1363:

General Equipment Leak Requirement

- a. The following General Equipment Leak requirements apply:
 - i. For the purpose of §63.1363 [Standards: Equipment Leaks], equipment means each pump, compressor, agitator, pressure relief device, sampling connection system, open-ended valve or line, valve, connector, and instrument system in organic HAP service. In organic HAP service means that a piece of equipment contains or contacts a fluid (liquid or gas) that is at least 5 percent by weight of total organic HAP. These

provisions also apply to any closed-vent systems and control devices required under §63.1363. [Reg.19.304 and 40 C.F.R. § 63.1363(a)(1) and §63.1361]

Consistency with Other Regulations

- ii. After the compliance date for a process, equipment subject to both §63.1363 [Standards: Equipment Leaks] and either of the following (40 CFR Part 60 and Part 61) will be required to only comply with the provisions of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.1363(a)(2)]
- iii. The provisions in §63.1(a)(3) of subpart A of Part 63 do not alter the provisions in §63.1363(a)(2). [Reg.19.304 and 40 C.F.R. § 63.1363(a)4]

Exemptions

- iv. The following are key exemptions regarding Subpart MMM standards for equipment leaks.
 - 1. Equipment that is in vacuum service is excluded from the requirements of §63.1363. [Reg.19.304 and 40 C.F.R. § 63.1363(a)(8)]
 - Equipment that operates in organic HAP service for less than 300 hours per calendar year, if it is identified as required in §63.1363(g)(9). [Reg.19.304 and 40 C.F.R. § 63.1363(a)(9)]
 - 3. Lines and equipment not containing process fluids are not subject to §63.1363. Utilities and other nonprocess lines, such as heating and cooling systems which do not combine their materials with those in the processes they serve, are not part of a process. [Reg.19.304 and 40 C.F.R. § 63.1363(a)(5)]

LDAR Provision Summary

v. An attached table provides a summary of the Subpart MMM equipment leak requirements. Because of the complexity of the LDAR requirements, this table should be considered a reference tool only and the regulation should be referenced when developing a detailed compliance plan. Moreover, the permittee shall develop a comprehensive LDAR to fully meet the Subpart MMM requirements, including developing a list of equipment and identification numbers subject to the requirements and monitoring schedule. Connectors, except those determined to be unsafeto-monitor, difficult-to-monitor, or inaccessible, do not have to be individually identified but the associated lines must be identified. Physical tagging of components is not required per 40 CFR §63.1363(a)(7) and §63.1363(g)(2).

PAI MACT Equipment Leak Requirement Summary – Part 1					
Equipment (PAI MACT / HON Cite- 40 CFR Part)	Design Requirements (Exemptions in parentheses)	Monitoring Frequency	Method	Leak Limit	Calculations
	OHAP HAP Service	Quarterly with Instrument			Calculate
Pumps in Light Liquid Service (63.1363(c))	(Dual mechanical seal systems that include a barrier fluid system are exempt per 40 CFR Part 63.1363(c)(5))	If 10% of pumps or 3 pumps in the process group leak, then monitor monthly	Method 21 ¹	2,000 ppm	Leakers per 40 CFR Part 63.1363(c)(4)
		Weekly visual inspection	Visual		
Pressure Relief Devices in Gas/Vapor Service (63.165)	OHAP Service (Exempt if routed to vent header)	Monitor after every pressure relief episode		Operated with instrument reading less than 500 ppm above background	
Sampling Connection Systems (63.166)	Must be equipped with closed purge, closed loop, or closed vent system Shall return fluid to process line	Initially			
Open-Ended Valves or Lines (63.1363(d))	Must be equipped with flanges, plugs, or another valve (If poses a safety hazard, is designed to open automatically, or if equipped with double block and bleed exempt by 40 CFR Part 63.1363(d)(4)-(6))	Initially			

¹ Method 21 citation is 40 CFR Part 60 Appendix A.

PAI MACT Equipment Leak Requirement Summary – Part 1					
Equipment (PAI MACT / HON Cite- 40 CFR Part)	Design Requirements (Exemptions in parentheses)	Monitoring Frequency	Method	Leak Limit	Calculations
Valves in Gas/Vapor and Light Liquid Service (63.1363(e))		Once within year of compliance date	Method 21 ²	500 ppm	Calculate Leakers per 40 CFR Part 63.1363(e)(5)
		>2% leakers- monthly		500 ppm	
		<2% leakers- quarterly		500 ppm	
		<1%-once/2 quarters		500 ppm	
		<0.5%-once/4 quarters		500 ppm	
		<0.25%-every 2 years		500 ppm	
Connectors in Gas/Vapor and in Light Liquid Service (63.174)		Once within year of compliance date	Method 21 ²	500 ppm	Calculate Leakers per 40 CFR Part 63.174(h)(3)(i)
		<0.5% - once/4 quarters		500 ppm	
		<0.25-every 2 years		500 ppm	
Agitators in Gas/Vapor and Light Liquid Service (63.1363(c))		Quarterly with instrument	Method 21 ²	10,000 ppm	
		Weekly visual Inspection	Visual		

References to 40 CFR Subpart H

b. The permittee shall comply with the provisions of 40 CFR 63, Subpart H as specified in §63.1363(b)(1) through (3) of Subpart MMM. When the term "process unit" is used in Subpart H, it shall mean any group of processes for the purpose of Subpart MMM. Groups of processes, as used in Subpart MMM, may be any individual process or combination of processes. [Reg.19.304 and 40 C.F.R. § 63.1363(b)]

Standards for Designated Equipment

- c. The permittee shall comply with all specific equipment leak standards §63.1363(c) through (f), including all documentation and calculations necessary for submitting information required in the Notification of Compliance Status Report (NOCS) under §63.1368(f). [Reg.19.304 and 40 C.F.R. § 63.1363(c) through (f)]
- d. The permittee may comply with recordkeeping requirements of more than one group of processes in a one recordkeeping system if the system identifies with each record the program being implemented. (e.g., quarterly monitoring) for each equipment type. All records shall be maintained in a manner that can be readily accessed at the plant site. This includes accessing the records from a central location by computer at the plant site. [Reg.19.304 and 40 C.F.R. § 63.1363(g)(1)]
- e. The permittee shall record all information required under §63.1363(g)(2) through (10), except as allowed under §63.1363(g)(5). [Reg.19.304 and 40 C.F.R. § 63.1363(g)(2) through (10)]

LDAR Reporting

f. The permittee shall submit a Notification of Compliance Status report as specified in §63.1363(h)(2) and periodic reports identified in §63.1363(h)(3). [Reg.19.304 and 40 C.F.R. § 63.1363(h)(1) through (3)]

COMPLIANCE DATE

OCI 42 The permittee is subject to the following requirements of 40 CFR §63.1364: The existing source compliance date for Subpart MMM is December 23, 2003. [Reg.19.304 and 40 C.F.R. § 63.1364(a)(1)]

TEST METHODS AND INITIAL COMPLIANCE PROCEDURES

OCI 43 The permittee is subject to the following requirements of 40 CFR §63.1365:

General

a. Except as specified in §63.1365(a)(5) [Alternative standard], the procedures specified in (c) [Process vents], (d) [Storage vessels], (e) [Wastewater], (f) [Bag dump/product dryer], and (g) [Pollution prevention alternative] are required to demonstrate initial compliance with 63.1362(b) [Process vents], (c) [Storage vessels], (d) [Wastewater], (f) [Bag dumps], and (g) [Pollution prevention alternative] respectively. Design evaluations that are used to demonstrate compliance with the standards for process vents and storage tanks are subject to

the provisions of 63.1365(a)(1). Performance tests that are specified in 63.1365(c), (d), and (e) are subject to the requirements in 63.1365(a)(2). Initial compliance procedures for flares are subject to 63.1365(a)(3). Alternative standards specified in 63.1362(b)(6) and (c)(4) are subject to the requirements in 63.1365(a)(5). The outlet concentration requirements of 63.1362(b)(2)(iv)(A), 63.1362(b)(3)(ii), 63.1363(b)(4)(ii)(A), 63.1362(b)(5)(ii), and 63.1632(b)(5)(iii) are subject to the requirements of 63.1365(a)(6). [Reg.19.304 and 40 C.F.R. 63.1365(a)]

Initial Compliance Procedures

b. The permittee shall demonstrate initial compliance by following the applicable procedures in §63.1365(a)(1) through (7). [Reg.19.304 and 40 C.F.R. § 63.1365(a)]

Test Methods and Conditions

c. The permittee shall use the appropriate test methods in §63.1365(b)(1) through (9) when testing is used to measure emissions. Compliance tests shall be performed under the conditions specified in §63.1365(b)(10) and (11). [Reg.19.304 and 40 C.F.R. § 63.1365(a)]

Initial Compliance with Process Vents Provisions

d. The permittee shall demonstrate compliance with the process vent standards in §63.1362(b) using the procedures described in §63.1365(c)(1) through (3). [Reg.19.304 and 40 C.F.R. § 63.1365(c)]

Initial Compliance with Storage Vessel Provisions

e. The permittee shall demonstrate initial compliance with the storage vessel standards in §63.1362(c)(2) through (4) by meeting the requirements in either §63.1365(d)(1), (2), (3), (4), (5), or (6). The demonstration of compliance for the planned routine maintenance provision in §63.1362(c)(5) is fulfilled by meeting the requirements in §63.1365(d)(7). [Reg.19.304 and 40 C.F.R. § 63.1365(d)]

Initial Compliance with the Wastewater Provisions

f. The permittee shall demonstrate initial compliance with the wastewater requirements by complying with the applicable provisions of §63.145, except the permittee need not comply with the requirement to determine visible emissions that are specified in §63.145(j)(1), and the references to compounds in Table 8 of Subpart G are not applicable for the purposes of Subpart MMM. When §63.145(i) refers to Method 18 of 40 CFR Part 60, Appendix A-6, the permittee may use any method specified in §63.1362(d)(12) to demonstrate initial compliance with Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.1365(e)]

Initial Compliance with the Bag Dump and Product Dryer Provisions

g. Compliance with the particulate matter concentration limits specified in §63.1362(e) [Bag dumps] is demonstrated when the concentration of particulate matter is less than 0.01 gr/dscf, as measured using the method described in §63.1365(b)(7). [Reg.19.304 and 40 C.F.R. § 63.1365(f)]

Initial Compliance with the Pollution Prevention Alternative Standard

h. If the permittee chooses to comply with the pollution prevention alternative standard, the permittee shall demonstrate initial compliance with 63.1362(g)(2) and (3) for a PAI process unit by preparing the demonstration summary in accordance with 63.1365(g)(1) and by calculating baseline and target annual HAP and VOC factors in accordance with 63.1365(g)(2) and (3). To demonstrate compliance with 63.1362(g)(3), the permittee must also comply with the procedures for add-on control devices that are specified in 63.1365(g)(4). [Reg.19.304 and 40 C.F.R. 63.1365(g)]

Compliance with Emissions Averaging Provisions

i. If the permittee chooses to comply with the emissions averaging provisions, the permittee must meet the requirements of §63.1365(h)(1). [Reg.19.304 and 40 C.F.R. § 63.1365(h)]

MONITORING AND INSPECTION REQUIREMENTS

- OCI 44 The permittee is subject to the following requirements of 40 CFR §63.1366:
 - a. The permittee shall provide evidence of continued compliance with the standard as specified in §63.1366. During the initial compliance demonstration, maximum or minimum operating parameter levels, as appropriate, shall be established for emission sources that will indicate the source is in compliance. Test data, calculations, or information from the evaluation of the control device design shall be used to establish operating parameter level. [Reg.19.304 and 40 C.F.R. § 63.1366(a)]

Monitoring for Control Devices

b. Except as provided by §63.1366(b)(1)(i), for each control device, the permittee shall install and operate monitoring devices and operate within the established parameter levels to ensure continued compliance with the standard. Monitoring parameters are specified for control scenarios in Table 3, and in §63.1366(b)(1)(ii) through (xii), of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.1366(b)(1)]

Averaging Periods

c. The permittee shall establish averaging periods for parametric monitoring levels according to §63.1366(b)(2)(i) through (iii). [Reg.19.304 and 40 C.F.R. § 63.1366(b)(2)]

Procedures for Setting Parameter Levels for Control Devices used to Control Emissions from Process Vents Process Vents - Small Control Devices

d. The permittee shall set the parametric monitoring levels for control devices (controlling less than 10 tons/yr of HAP), for which a performance test is not required, by conducting a design evaluation. If a performance test is conducted it shall be established according to the procedures in §63.1366(b)(3)(i). [Reg.19.304 and 40 C.F.R. § 63.1366(b)(3)]

Process Vents – Large Control Devices

i. The permittee shall establish the parameter monitoring level for large control devices (controlling greater than or equal to 10 tons/yr), for which a performance test is required by Subpart MMM, by following the procedures in §63.1366(b)(3)(ii). [Reg.19.304 and 40 C.F.R. § 63.1366(b)(3)(ii)]

Process Vents - Parameter Levels for Control Devices Controlling Batch Process Vent

ii. The permittee shall establish parameter monitoring levels for devices controlling batch process vents, alone or in combination with other streams according to §63.1366(b)(3)(iii). [Reg.19.304 and 40 C.F.R. § 63.1366(b)(3)(iii)]

Request for Approval to Monitor Alternative Parameters

e. The permittee may request approval to monitor parameters other than those required by §63.1366(b)(1)(ii) through (xiii). The request shall be submitted according to the procedures in §63.8(f) of Subpart A or in the Precompliance Report as specified in §63.1368(e). [Reg.19.304 and 40 C.F.R. § 63.1366(b)(4)]

Monitoring for the Alternative Standard

f. The permittee may monitor for the alternative standards identified in §63.1362(b)(6) and (c)(4) by following the requirements in §63.1366(b)(5). [Reg.19.304 and 40 C.F.R. § 63.1366(b)(5)]

Exceedances of Operating Parameters

g. An exceedance of an operating parameter is defined as one of the following:

- i. If the parameter level, averaged over the operating day block, is below a minimum value established during the initial compliance demonstration;
- ii. If the parameter level, averaged over the operating day block, is above the maximum value established during the initial compliance demonstration;
- iii. A loss of all pilot flame for a flare during an operating day or block. Multiple losses of pilot flame during an operating day constitutes an exceedance;
- iv. Each operating day or block for which the time interval between replacement of a nonregenerative carbon absorber exceeds the interval established in §63.1366(b)(1)(v); or
- v. Each instance in which procedures to initiate the response to a bag lead detection alarm within 1-hour of the alarm as specified in the corrective action plan.

[Reg.19.304 and 40 C.F.R. § 63.1366(b)(6)]

h. Monitoring data are insufficient to constitute a valid hour of data, as used in §63.1366(b)(7)(i) and (ii) if measured values are unavailable for any of the required 15-minute periods within the hour. [Reg.19.304 and 40 C.F.R. § 63.1366(b)(7)]

Excursions are defined by either of the two cases listed in §63.1366(b)(7)(i) or (ii) as follows:

- i. When the period of control device operation is 4-hours or greater in an operating day or block and monitoring data are insufficient to constitute a valid hour of data as defined in §63.1366(b)(7)(iii) for at least 75% of the operating hours. [Reg.19.304 and 40 C.F.R. § 63.1266(b)(7)(i)]
- ii. When the period of control device operation is less than 4-hours in an operating day or block and more than 1 of the hours during the period of operation does not constitute a valid hour of data due to insufficient monitoring data. [Reg.19.304 and 40 C.F.R. § 63.1366(b)(7)(ii)]

Violations

i. Exceedances of parameters monitored according to §63.1366(b)(1)(ii), (iv) through (ix), and §63.1366(b)(5)(i)(A) and (B), or excursions as defined by §63.1258(b)(7)(i) and (ii) constitute violations of the operating limit according to §63.1366(b)(8)(i), (ii), and (iv). Exceedances of the temperature limit monitored according to §63.1366(b)(1)(iii) or exceedances of the outlet concentrations monitored according to the provisions of §63.1366(b)(1)(x) constitute violations of the emission limit according to §63.1366(b)(8)(i), (ii), and (iv). Exceedances of the outlet concentrations monitored according to the provisions of §63.1366(b)(1)(x) constitute violations of the emission limit according to §63.1366(b)(8)(i), (ii), and (iv). Exceedances of the outlet concentration

monitored according to §63.1366 (b)(5) constitute violations of the emission limit according to the provisions of §63.1366(b)(8)(iii) and (iv) of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.1366(b)(8)]

Exceptions to Exceedances, Excursions, and Violations

- i. Except as provided in §63.1366(b)(8)(iv), for episodes occurring more than once per day, exceedances of established parameter limits or excursions will result in no more than one violation per operating day for each monitored item or equipment utilized in the process. [Reg.19.304 and 40 C.F.R. § 63.1366(b)(8)(i)]
- Except as provided in §63.1366(b)(8)(iv), for control devices used for more than one process in the course of an operating day, exceedances or excursions will result in no more than one violation per operating day, per control device, for each process for which the control device is in service. [Reg.19.304 and 40 C.F.R. § 63.1366(b)(8)(ii)]
- Except as provided in §63.1366(b)(8)(iv), exceedances of the 20 ppmv TOC outlet emission limit, or the HCL and chlorine emission limit, averaged over the operating day, will result in no more than one violation per operating day per control device. [Reg.19.304 and 40 C.F.R. § 63.1366(b)(8)(iii)]
- iv. Periods of time when the monitoring measurements exceed the parameter values as well as periods of inadequate monitoring data do not constitute a violation if they occur during a startup, shutdown, or malfunction, and the facility follows its startup, shutdown, and malfunction plan. [Reg.19.304 and 40 C.F.R. § 63.1366(b)(8)(iv)]

Monitoring for Equipment Leaks

j. The permittee shall comply with the equipment leak monitoring requirements in §63.1363. [Reg.19.304 and 40 C.F.R. § 63.1366(d)]

Emission Monitoring for Heat Exchanger Systems

k. The permittee shall comply with the heat exchanger monitoring requirements in §63.1362(f) for those heat exchangers subject to Subpart MMM. [[Reg.19.304 and 40 C.F.R. § 63.1366(e)]

Monitoring for the Pollution Prevention Alternative Standard

1. If the permittee chooses to comply with the pollution prevention alternative standards in §63.1362(g)(2) or (3) the requirements of §63.1366(f) shall be followed. [§19.304 of Regulation 19 and §40 CFR §63.1366(f)]

Monitoring of Emissions Averaging

m. If the permittee chooses to comply with the emissions averaging requirements in §63.1362(h), the monitoring requirements of §63.1366(b) [Monitoring for control devices] must be followed for all processes, storage tanks, and waste management units included in the emissions average. [Reg.19.304 and 40 C.F.R. § 63.1366(g)]

Leak Inspection Provisions of Vapor Suppression Equipment

- n. The following General Equipment Leak requirements apply:
 - i. The permittee shall comply with the requirements of §63.1366(h)(2) through (8), except as provided in §63.1366(h)(9) and (10), for each vapor collection system, closed-vent system, fixed roof, cover, or enclosure. [Reg.19.304 and 40 C.F.R. § 63.1366(h)(1)]
 - ii. If a closed-vent system subject to §63.1366 [Monitoring and inspection requirements] is also subject to the equipment leak provisions of §63.1363, the permittee shall comply with the provisions of §63.1363 and is exempt from the requirements in §63.1366. [Reg.19.304 and 40 C.F.R. § 63.1366(h)(9)]
 - iii. The permittee is not required to comply with the requirements specified in §63.1366(h)(2) though (8) for any closed-vent system that is operated and maintained under negative pressure. [Reg.19.304 and 40 C.F.R. § 63.1366(h)(10)]

RECORDKEEPING REQUIRMENTS

- OCI 45 The permittee is subject to the following requirements of 40 CFR §63.1367:
 - a. The permittee shall comply with the recordkeeping requirements in 40 CFR 63, Subpart A as specified in Table 1 of Subpart MMM and in §63.1367(a)(1) through (5). [Reg.19.304 and 40 C.F.R. § 63.1367(a)]

Records of Equipment Operations

b. The permittee shall keep the records specified in §63.1367(b)(1) through (11) up-todate and readily accessible, that conforms to the sources applicability determination and operations. [Reg.19.304 and 40 C.F.R. § 63.1367(b)]

Records of Equipment Leak Detection and Repair

c. The permittee shall implement recordkeeping requirements specified in §63.1363(g) [Recordkeeping] for equipment subject to the equipment leak standards in §63.1363. All records shall be kept for a period of 5-years, in

accordance with the requirements in 63.10(b)(1) of 40 CFR 63, Subpart A. [Reg.19.304 and 40 C.F.R. § 63.1367(c)]

Records of Emissions Averaging

d. If the permittee chooses to comply with the emissions averaging requirements of §63.1362(h), up-to-date records of the information in §63.1367(d)(1) through (4) must be kept. [Reg.19.304 and 40 C.F.R. § 63.1367(d)]

Heat Exchanger Records

e. If the permittee is subject to the heat exchanger system requirements of §63.1362(g), records as specified in §63.104(f)(1)(i) through (iv) of 40 CFR 63, Subpart G must be retained. [Reg.19.304 and 40 C.F.R. § 63.1367(e)]

Records of Inspections

f. The permittee shall keep records of inspections specified in §63.1367(f)(1) through (6). [Reg.19.304 and 40 C.F.R. § 63.1367(f)]

Records of Primary Use

g. The permittee shall keep records of each PAI process unit that is used to produce a given material for use as a PAI as well as for other purposes. The permittee shall keep records of total production and the production for use as a PAI on a semiannual or more frequent basis if the use as a PAI is not the primary use. [Reg.19.304 and 40 C.F.R. § 63.1367(g)]

REPORTING REQUIREMENTS

OCI 46 The permittee is subject to the following requirements of 40 CFR §63.1368:

a. The permittee shall comply with the reporting requirements in §63.1368(b) through (l) of Subpart MMM. Applicable reporting requirements of §63.9 [notification requirements] and §63.10 [recordkeeping requirements] are also summarized in Table 1 of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.1368(a)]

Initial Notification

b. The permittee shall submit the applicable initial notification in accordance with §63.9(b) or (d) of 40 CFR 63, Subpart A, as specified in Table 1 of Subpart MMM. [Reg.19.304 and 40 C.F.R. § 63.1368(b)]

Application for Approval of Construction or Reconstruction

c. Any application for approval of construction of a new major affected source, the reconstruction of a major affected source, or the reconstruction of a major source such that the source becomes major affected source subject to the standards shall be prepared in accordance with §63.5(d) [Application for approval of construction or reconstruction]. [Reg.19.304 and 40 C.F.R. § 63.13268(c)]

Notification of Continuous Monitoring System Performance Evaluation

d. If the permittee is required by the Administrator to conduct a performance evaluation for a continuous monitoring system that is used to comply with the alternate standard in §63.1362(b)(6) or (c)(4), the permittee shall notify the Administrator of the date of the performance evaluation as specified in §63.8(e)(2). [Reg.19.304 and 40 C.F.R. § 63.1368(d)]

Precompliance Plan

e. The permittee shall submit the Precompliance Plan at least 3-months prior to the compliance date of the standard. The Precompliance Plan shall include the information specified in §63.1368(e)(1) through (5). [Reg.19.304 and 40 C.F.R. § 63.1368(e)]

Notice of Compliance Status Report

f. The permittee shall submit the Notification of Compliance Status report required under §63.9 no later than 150 days after the compliance date and shall include information specified in §63.1360(f)(1) through (9). [Reg.19.304 and 40 C.F.R. § 63.1368(f)]

Periodic Reports

g. The permittee shall prepare and submit periodic reports specified in §63.1368(g)(1) and (2) to the Administrator. [Reg.19.304 and 40 C.F.R. § 63.1368(g)]

Notification of Process Change

h. Except as specified in §63.1368(h)(2), whenever a process change is made, or a change in any of the information in the Notification of Compliance Status Report, the permittee shall submit the information specified in §63.1268(h)(1)(i) through (iv) with the next Periodic Report required under §63.1368(g) [Periodic reports]. [Reg.19.304 and 40 C.F.R. § 63.1368(h)]

Reports of Startup, Shutdown, and Malfunction

i. The permittee shall prepare startup, shutdown, and malfunction (SSM) reports as specified in §63.1368(i). [Reg.19.304 and 40 C.F.R. § 63.1368(i)]

Reports of Equipment Leaks

j. The permittee shall implement the reporting requirements specified in §63.1363(h) [LDAR Reporting] for sources subject to the equipment leak standards in §63.1363. Copies of all reports shall be retained for a period of 5 years in accordance with the requirements of §63.10(b)(1) of Subpart A. [Reg.19.304 and 40 C.F.R. § 63.1368(j)]

Reports of Emissions Averaging

k. If the permittee chooses to comply with the emissions averaging requirements in §63.1362(h), all information specified in §63.1367(d) shall be submitted for all emission points included in the emissions average. Additionally, the report shall include all information specified in §63.1368(g) [Periodic reports] for each emission point included in the emissions average, and all information listed in §63.1368(k)(1)(i) through (iv). [Reg.19.304 and 40 C.F.R. § 63.1368(k)]

Reports of Heat Exchange Systems

1. The permittee shall submit reports of applicable heat exchange systems as specified in §63.1368(l). [Reg.19.304 and 40 C.F.R. § 63.1368(l)]

Notification of Performance Test and Test Plan

m. The permittee shall notify the Administrator of the planned date of a performance test at least 60-days before the test in accordance with §63.7(b) [notification of performance tests]. The permittee shall also submit the test plan required by §63.7(c) [quality assurance program] and the emission profile required by §63.1365(b)(8)(ii) with the notification of the performance test. [Reg.19.304 and 40 C.F.R. § 63.1368(m)]

Utilities

6M01, 6M01-01, 6M01-01A, 6M06-01 and 6M07-01

Source Description

There are three coal fired (6M01-01) and two natural gas fired boilers (6M06-01 and 6M07-01) at the facility.

The coal fired boilers are balanced draft, coal-fired steam generation boilers that have been fitted with atomizing nozzles to facilitate burning of liquid chemical wastes. Each coal fired boiler system is designed as a 70 million Btu/hr unit and is equipped with its own electrostatic precipitator (ESP) to control particulate emissions. The three coal fired boilers share a common primary fuel conveying system, a common ash handling system, and a common 200 foot tall stack. The boilers are independently controlled by a Distributed Control System (DCS). All interactions from the operator to the burners are made through this computer system.

The three coal fired boilers were installed in 1975, and are rated for 70 million Btu/hr per unit. Due to size and installation date, these boilers are not subject to any of the NSPS requirements.

The three coal fired boilers are subject to 40 CFR Part 63, Subpart EEE, National Emission Standards for Hazardous Air Pollutants from Hazardous Waste Combustors.

The solvent from the 2,000 gallon liquid process tank to be used for the purpose of flushing the chemical distribution piping is routed to either the coal-fired boiler auxiliary waste chemical burners or to the burner of the chemical waste destructor. Emissions from tank venting will be collected and routed to the coal-fired boilers (6M01-01).

There are two natural gas fired boilers at the facility. The #4 boiler (6M06-01) burns natural gas at 78 million BTU/hr. The #5 boiler (6M07-01) burns natural gas at 221 million BTU/hr. Each boiler system consists of a water tube boiler, economizer, superheater and a stack.

The #4 boiler was installed in 1986 and is rated for 78 million Btu/hr. The #5 boiler was installed in 1993 and is rated for 221 million Btu/hr. Due to size and installation date, the #4 boiler is not subject to NSPS requirements. However, the #5 boiler is subject to NSPS Subpart Db, with requirements pertaining to NO_x are applicable. Both the #4 (6M06-01) and #5 (6M07-01) natural gas fired boilers are subject to PSD emissions limitations. Initial testing to confirm PSD emission limits for NOx were performed on August 2, 1988 for the #4 Boiler and August 6-7, 1992 for the #5 Boiler.

NSPS Kb requirements are identified and addressed in the Plantwide Conditions of this permit for all facility storage vessels, including those used in waste chemical service in the Utilities section. Emissions from utilities waste chemical storage tanks are routed through a closed-vent system to three coal-fired boilers as control devices.

BACT Analysis for Boilers #4 and #5

<u>Boiler #4</u>. This boiler is subject to a PSD emission rate limitation for NO_x which is simply 13.3 lb/hr. BACT for NO_x at the time of permit issuance was considered to be a standard register burner. BACT analysis for this source was performed in Permit No. 829-A.

<u>Boiler #5</u>. This boiler is subject to both PSD and NSPS Subpart Db requirements. The PSD BACT limit for NO_x is 22 lb/hr (0.1 lb/million Btu), which is more stringent than the NSPS emissions standard for NO_x (0.2 lb/million Btu). The BACT analysis was performed in Permit No. 1085-AR-1.

Specific Conditions

US 1. The permittee shall not exceed the emission rates set forth in the following table. [Reg.19.501 *et seq.* and 40 C.F.R. § 52 Subpart E]

SN	Description	Pollutant	lb/hr	tpy
6M01	Coal Pile	PM_{10}	0.1	0.1
6M01-01	3 Coal Fired Boilers (70 MMBtu/hr each)	PM ₁₀ SO ₂ VOC CO NO _x Pb	26.1 1391.0 0.7 236.3 106.0 0.8	112.8 5982.9 2.9 1034.0 488.2 3.4
6M01- 01A	Coal Bunker Fabric Filter	PM ₁₀	0.2	0.7
6M06-01	#4 Boiler (78 MMBtu/hr) Natural Gas	PM ₁₀ SO ₂ VOC CO NO _x	0.6 0.1 0.5 6.6 13.3	2.6 0.3 1.9 28.7 58.3
6M07-01	#5 Boiler (221 MMBtu/hr) Natural Gas	PM ₁₀ SO ₂ VOC CO NO _x	1.7 0.2 1.3 18.6 22.0	7.4 0.6 5.4 81.4 96.4
BLR-FUG	Utilities Area Fugitive Emissions	VOC	0.5	1.8

US 2. The facility shall show compliance with the facility total ton/yr limits using the procedures outlined in Plantwide Conditions 12 through 16. Hourly Plantwide Hazardous Air Pollutant emissions are limited by Plantwide Conditions 16. Additional HAP limitations are included in Plantwide Condition 13. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

SN	Description	Pollutant	lb/hr	tpy
6M01	Coal Pile	PM	0.1	0.1
6M01-01	3 Coal Fired Boilers (70 MMBtu/hr each)	PM Inorganics* Organic Pollutants**	26.1 259.0 0.7	112.8 1031.6 2.9
6M01-01A	Coal Bunker Fabric Filter	РМ	0.2	0.7
6M06-01	#4 Boiler 78 MMBtu/hr) Natural Gas	PM Organic Pollutants**	0.6 0.5	2.6 1.9
6M07-01	#5 Boiler (221 MMBtu/hr) Natural Gas	PM Organic Pollutants**	1.7 1.3	7.4 5.4
BLR-FUG	Utilities Area Fugitive Emissions	Organic Pollutants**	0.5	1.8

*Inorganics are considered to be non-organic Hazardous Air Pollutants

**Organic Pollutants are considered to be VOC, all organic HAPs, and organic Air Contaminates.

US 3. Visible emissions may not exceed the limits specified in the following table of this permit as measured by EPA Reference Method 9, except during periods of startup, shutdown, and malfunction. Compliance with this condition shall be demonstrated through operating the ESP as specified by the manufacturer, and in accordance with the most current version of the Facility Operating Plan. [Reg.18.501 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

SN	Limit	Regulatory Citation
6M01	5%	§18.501
6M01-01	20%	§19.503
6M01-01A	5%	§18.501
6M06-01	5%	§18.501
6M07-01	20%	§19.304 and NSPS Db

US 4. The permittee shall conduct weekly observations of the opacity from all sources in the Utilities section and keep a record of these observations. If the permittee detects visible emissions, the permittee must immediately take action to identify and correct the cause of the visible emissions. After implementing the corrective action, the permittee must document that the source complies with the visible emissions requirements. The permittee shall maintain records of the cause of any visible emissions and the corrective action taken. The permittee must keep these records onsite and make them available to Department personnel upon request. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

- US 5. Boiler #4 and Boiler #5 shall be limited to NO_x emission rates of 13.3 and 22.0 lb/hr, respectively. [Reg.19.901 et seq. and 40 C.F.R. § 52 Subpart E]
- US 6. The permittee shall maintain the power input to the ESP (6M01-01 Coal Fired Boilers) as outlined in the most current version of the Facility Operating Plan. [Reg.19.303 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]
- US 7. The permittee shall maintain daily records of the power input at the ESP (6M01-01 Coal Fired Boilers). [Reg.19.705 and 40 C.F.R. § 52 Subpart E]
- US 8. The permittee shall maintain compliance with the VOC, SO₂, NO_x, CO, and inorganic emission limits of 6M01-01 (Coal Fired Boilers) per the methodology outlined in most current version of the Facility Operating Plan. [Reg.19.303 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]
- US 9. The permittee shall not combust coal with a sulfur content greater than 3.8% by weight. [Reg.19.705, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 70.6]
- US 10. The permittee shall perform testing of 6M01-01 (Coal Fired Boilers) every 61 months for NOx, using EPA Reference Method 7E. This test shall be conducted in accordance with Plantwide Condition 3. [§19.702 of Regulation 19 and 40 CFR Part 52 Subpart E]
- US 11. The permittee shall record the amount and type of coal, biosludge and liquids fed to the coal fired boilers (6M01-01 Coal Fired Boilers) during a 30-day period. These records shall be kept on site and made available upon request. [Reg.19.705 and 40 C.F.R. § 52 Subpart E]
- US 12. The permittee shall track natural gas usage in the #4 Boiler (6M06-01) as outlined in the most current version of the Facility Operating Plan. [Reg.19.705, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 70.6]
- US 13. The permittee shall use a predictive emission monitoring system (PEMS) to monitor NOx emissions from the #5 Boiler (6M07-01) as outlined in the most current version of the Facility Operating Plan. [Reg.19.705, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 70.6 and 40 C.F.R. § 60.48b(g)(2)]
- US 14. The permittee shall combust only pipeline quality natural gas in 6M06-01 and 6M07-01. [Reg.19.705, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 70.6]
- US 15. The permittee shall maintain the pressure drop across the fabric filter at 6M01-01A as outlined in the most current version of the Facility Operating Plan. [Reg.19.705, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311, and 40 C.F.R. § 70.6]
- US 16. The permittee shall keep records on site of the pressure drop across 6M01-01A. [Reg.19.705 and 40 C.F.R. § 52 Subpart E]
- US 17. The permittee shall operate all CEMS at this source in accordance with all applicable conditions of Section III Notification and Recordkeeping of the Department's Continuous Emission Monitoring Systems Conditions as found in Appendix A of this permit.

 $[Reg.19.703, 40\ C.F.R. \$ 52 Subpart E, and Ark. Code Ann. $\$ 8-4-203 as referenced by Ark. Code Ann. $\$ 8-4-304 and 8-4-311]

US 18. The permittee shall monitor SN-6M01-01 as outlined in US 15 and US 16 above and the monitoring requirements of the MON MACT outlined in MON 10. The permittee shall submit reports as required by Plantwide Condition 33. [Rule 19.304 and 40 C.F.R. § 64]

40 CFR Part 60 Subpart Db - Standards of Performance for Industrial-Commercial-Institutional Steam Generating Units

- US 19. The permittee is to comply with the following nitrogen oxides emission limitation (expressed as NO₂) at SN-6M07-01: The NO₂ limitation is 0.20 lb/MMBtu based on a high heat release rate. [Reg.19.304 and 40 C.F.R. § 60.44b(a)(1)(ii)]
- US 20. The nitrogen oxide standard at SN-6M07-01 applies at all times including periods of startup, shutdown, or malfunction. [Reg.19.304 and 40 C.F.R. § 60.44b(h)]
- US 21. Compliance with the emission limitations at SN-6M07-01 is determined on a 30-day rolling average basis. [Reg.19.304 and 40 C.F.R. § 60.44b(i)-(j)]
- US 22. The permittee is limited to opacity at SN-6M07-01 of 20%. This limit shall apply at all times except periods of startup, shutdown, or malfunction. [Reg.19.304 and 40 C.F.R. § 60.46b(a)]
- US 23. The permittee shall use a continuous parametric monitoring system (PEMS) at SN-6M07-01 to determine compliance with monitoring nitrogen oxides under §60.48b. [Reg.19.304 and 40 C.F.R. § 60.46b(e)]
- US 24. The permittee shall monitor steam generating unit operating conditions at SN-6M07-01 and predict nitrogen oxides emission rates as specified in a plan submitted pursuant to §60.49(c). [Reg.19.304 and 40 C.F.R. § 60.48b(g)(2)]
- US 25. The permittee shall comply with all provisions of this citation for monitoring steam generating unit operating conditions at SN-6M07-01 under §60.48b(g)(2). [Reg.19.304 and 40 C.F.R. § 60.49b(c)]
- US 26. The permittee shall record and maintain records of amounts of natural gas combusted at SN-6M07-01 each day and calculate the annual capacity factor for the reporting period. The annual capacity factor is determined on a 12-month rolling average basis with a new annual capacity factor calculated at the end of each calendar month. [Reg.19.304 and 40 C.F.R. § 60.49b(d)]
- US 27. The permittee shall maintain and record at SN-6M07-01, for each steam generating unit operating day, the information required by §60.49b(g). [Reg.19.304 and 40 C.F.R. § 60.49b(g)]
- US 28. The permittee shall submit excess emission reports for any excess emission which occur at SN-6M07-01 during the reporting period. [Reg.19.304 and 40 C.F.R. § 60.49b(h)]
- US 29. The reporting period for the reports required at SN-6M07-01 under this subpart is each 6month period. All reports shall be submitted to the Administrator and shall be postmarked

by the 30th day following the end of the reporting period. [Reg.19.304 and 40 C.F.R. 60.49b(w)]

40 CFR Part 63 Subpart DD - National Emission Standards for Hazardous Air Pollutants from Off-Site Waste and Recovery Operations

- US 30. The permittee shall comply with any of the requirements specified in 40 CFR §63.683(b)(1) for Off-site Material Management Units within an affected source designation under 40 CFR §60.680(c). [Reg.19.304 and 40 C.F.R. § 63.683(a)]
- US 31. Specific units identified as applicable under this regulation include the following liquid waste storage tanks: WB-01, WB-02, WB-03, WB-04, WB-05, WB-06, WB-07, WB-08, WB-09, WDT-01, WDT-02, and PT-50. These tanks, with the exception of PT-50, are routed through the coal-fired boilers via a closed-vent vapor recovery system. PT-50 is routed to the RTO. [Reg.19.304 and 40 C.F.R. § 63.683(b)(1)]
- US 32. The permittee is exempt from those off-site material management units identified in 40 CFR §63.683(b)(2). [Reg.19.304 and 40 C.F.R. § 63.683(b)(2)]
- US 33. The permittee controls air emissions from off-site material management units in accordance with the applicable standards specified in 40 CFR §63.685 through §63.689. [Reg.19.304 and 40 C.F.R. § 6363.683(b)(1)(i)]
- US 34. The permittee shall comply with the requirements of 40 CFR §63.684(a) and any of the treatment processes under 40 CFR §63.684 (b), as applicable, for the treatment of off-site material to remove or destroy HAP for which §63.683(b)(1)(i) references such treatment. [Reg.19.304 and 40 C.F.R. § 63.684(a) and (b)]
- US 35. The permittee shall maintain records of each treatment process in accordance with the requirements in 40 CFR §63.696. [Reg.19.304 and 40 C.F.R. § 63.684(f)]
- US 36. The permittee shall submit and prepare reports for each treatment process in accordance with 40 CFR §63.697(a). [Reg.19.304 and 40 C.F.R. § 63.684(g)]
- US 37. The permittee shall comply with the requirements of §63.685(a) and (b), and control air emissions from tanks for which §63.683(b)(1)(i) references such air emission control. [Reg.19.304 and 40 C.F.R. § 63.685(a) and (b)]
- US 38. The permittee shall comply with the requirements of §63.685(c) when controlling air emissions from tanks using Tank Level 1 controls, unless the permittee has implemented Tank Level 2 controls. [Reg.19.304 and 40 C.F.R. § 63.685(c)]
- US 39. The permittee shall comply with §63.685(d) for controlling air emissions from a tank, which requires the use of Tank Level 2 controls. [Reg.19.304 and 40 C.F.R. § 63.685(d)]
- US 40. The permittee shall comply with the requirements of §63.685(g)(1) through (3) for the control of tank air emissions if venting to a control device. [Reg.19.304 and 40 C.F.R. § 63.685(g)]
- US 41. The permittee shall comply with the requirements of either §63.689(b) or (c), as applicable, for the control of air emissions from transfer systems for which §63.683(b)(1)(i) references such air emission control. [Reg.19.304 and 40 C.F.R. § 63.689(a)]

- US 42. The permittee shall comply with the requirements of §63.691(a) and (b) for the control of equipment leaks for which §63.680(c)(3) references such air emission control. [Reg.19.304 and 40 C.F.R. § 63.691(a)]
- US 43. The permittee shall meet the requirements of 40 CFR §63.693(b)(1) for each closed-vent system. [Reg.19.304 and 40 C.F.R. § 63.693(b)(1)]
- US 44. The permittee shall meet the requirements of 40 CFR §63.693(b)(2) for each control device. [Reg.19.304 and 40 C.F.R. § 63.693(b)(2)]
- US 45. The permittee shall perform testing as specified in 40 CFR §63.694 for all applicable treatment processes and/or control devices used for compliance with applicable standards under this subpart. [Reg.19.304 and 40 C.F.R. § 63.694]
- US 46. The permittee shall comply with the inspection and monitoring requirements of 40 CFR §63.695 for all affected tanks, closed-vent systems, transfer systems, and control devices as applicable. [Reg.19.304 and 40 C.F.R. § 63.695]
- US 47. The permittee shall comply with all applicable recordkeeping requirements in 40 CFR §63.696, including requirements in 40 CFR §63.10, General Provisions that applies as specified in Table 2 of 40 CFR §63, Subpart DD. [Reg.19.304 and 40 C.F.R. § 6363.696]

40 CFR Part 60 Subpart Y - Standards of Performance for Coal Preparation Plants

- US 48. The coal processing and conveying equipment, coal storage, and coal transfer equipment shall be limited to 20% opacity. This condition applies to 6M01, storage pile and coal unloading area. [Reg.19.304 and 40 C.F.R. § 60.252(c)]
- US 49. In conducting the initial performance tests required in §60.8, the permittee shall use Method 9 to determine opacity. [Reg.19.304 and 40 C.F.R. § 60.254(b)(2)]

40 CFR Part 63 Subpart EEE - National Emission Standards for Hazardous Air Pollutants from Hazardous Waste Combustors

US 50. The permittee shall maintain the operating limits as outlined in the Notification of Compliance (NOC) for the coal boilers when combusting hazardous waste. The NOC is required by 40 CFR Part 63, Subpart EEE. [Reg.19.705 and 40 C.F.R. § 52 Subpart E]

Operating Parameters Limitations (OPLs)

Maximum Waste Feed Limitations

Waste Feed Rate (aqueous HAP + organic HAP), lb/hr

Chlorine Feed Rate, lb/hr

Ash Feed Rate, lb/hr

Total Mercury Feed Rate, lb/hr

Total Semi-Volatile Metals Feed Rate, lb/hr

Total Low-Volatile Metals Feed Rate, lb/hr

Operating Parameters Limitations (OPLs)

Combustion Chamber Limitations

Minimum Combustion Chamber Temperature, $^{\rm o}{\rm F}$

Minimum Atomization Pressure, psi

Maximum Furnace Pressure, psi

Maximum Combustion Air Flow Rate, scfm

ESP Limitations

Minimum ESP Power, kW

Maximum Inlet Temperature to the ESP, °F

Operating Parameters Limitations (OPLs)

Stack Gas Limitations

Maximum Total Hydrocarbon, ppmv

Maximum Percent Oxygen, %

- US 51. The permittee shall maintain records operating limits as specified in the NOC. These records shall be maintained on site and available for inspection upon request. [Reg.19.705 and 40 C.F.R. § 52 Subpart E]
- US 52. This facility is subject to 40 CFR Part 63, Subpart EEE, National Emission Standards for Hazardous Air Pollutants from Hazardous Waste Combustors when burning hazardous waste. Applicable requirements include, but are not limited to, the following conditions [Reg.19.304 and 40 C.F.R. § 63.1200 of EEE]:
 - a. The permittee shall meet the dioxin and furan standards of 40 CFR Part 63, Subpart EEE by complying with the hydrocarbon emission standards of condition US.51(e) of this permit. [Reg.19.304 and 40 C.F.R. § 63.1216(a)(1)]
 - b. The permittee shall not discharge or cause combustion gases to be emitted to the atmosphere that contain mercury in excess of $11 \mu g/dscm$, corrected to 7 percent oxygen. [Reg.19.304 and 40 C.F.R. § 63.1216(a)(2)]
 - c. The permittee shall not discharge or cause combustion gases to be emitted to the atmosphere that contain lead and cadmium in excess of 180 μ g/dscm, combined emissions, corrected to 7 percent oxygen. [Reg.19.304 and 40 C.F.R. § 63.1216(a)(3)]
 - d. The permittee shall not discharge or cause combustion gases to be emitted to the atmosphere that contain arsenic, beryllium, and chromium in excess of 380µg/dscm, combined emissions, corrected to 7 percent oxygen. [Reg.19.304 and 40 C.F.R. § 63.1216(a)(4)]

- e. The permittee shall not discharge or cause combustion gases to be emitted to the atmosphere that contain hydrocarbons in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, and corrected to 7 percent oxygen, and reported as propane. [Reg.19.304 and 40 C.F.R. § 63.1216 (a)(5)(ii)]
- f. The permittee shall not discharge or cause combustion gases to be emitted to the atmosphere that contain hydrochloric acid and chlorine gas in excess of 440 parts per million by volume, combined emissions, expressed as hydrochloric acid equivalents, dry basis and corrected to 7 percent oxygen or the permittee may comply with the health-based compliance alternatives in lieu of the emission standards for total chlorine provided under §§ 63.1216(a)(6) by complying with the health-based compliance alternatives for total chlorine under the procedures prescribed in 40 CFR 63.1215. [Reg.19.304 and 40 C.F.R. § 63.1216(a)(6)]
- g. The permittee shall not discharge or cause combustion gases to be emitted to the atmosphere that contain particulate matter in excess of 68 mg/dscm corrected to 7 percent oxygen. [Reg.19.304 and 40 C.F.R. § 63.1216(a)(7)]

Destruction and Removal Efficiency (DRE) Standard

h. The permittee shall maintain a 99.99% destruction and removal efficiency (DRE) for each principal organic hazardous constituent (POHC) designated under paragraph (c)(3) of this section. The DRE shall be calculated using the following equation:

DRE = [1-(Win / Wout)] X 100% Where:

Win = mass feedrate of one principal organic hazardous constituent (POHC) in a waste feed stream; and

Wout = mass emission rate of the same POHC present in exhaust emissions prior to release to the atmosphere. [Reg.19.304 and 40 C.F.R. § 63.1216(c)]

- The permittee must treat the POHCs in the waste feed that are specified under paragraph (c)(3)(ii) of this section to the extent required by paragraphs §63.1216(c)(1) and (c)(2) (i.e. 99.99% as stated in the previous Specific Condition). [Reg.19.304 and 40 C.F.R. § 63.1216(c)(3)(i)]
- j. The permittee shall specify one or more POHCs from the list of hazardous air pollutants established by 42 U.S.C. 7412(b)(1), excluding caprolactum as provided by §63.60, for each waste to be burned. The permittee must base this specification on the degree of difficulty of incineration of the organic constituents in the waste and on their concentration or mass in the waste feed, considering the results of waste analyses or other data and information. [Reg.19.304 and 40 C.F.R. § 6363.1216(c)(3)(ii)]
- k. The emission limits provided by paragraphs §63.1203(a) and §63.1203(b) are presented with two significant figures. Although the permittee must perform intermediate calculations using at least three significant figures, the resultant

emission levels may be rounded to two significant figures to document compliance. [Reg.19.304 and 40 C.F.R. § 63.1216(d)]

Compliance Provisions

- 1. The permittee shall comply with the emission standards of §63.1219 no later than October 14, 2008, unless the Administrator grants an extension under §63.6(i) or §63.1213. [Reg.19.304 and 40 C.F.R. § 6363.1206(a)(1)(ii)]
- m. The permittee shall comply with the emission standards and operating requirements set forth in 40 CFR Part 63, Subpart EEE at all times when hazardous wastes are in the combustion chamber, except as specified in §63.1206(b)(1)(i) and (ii). [Reg.19.304 and 40 C.F.R. § 63.1206(b)(1)]
- n. The permittee shall demonstrate compliance based on performance testing under operating conditions representative of the extreme range of normal conditions. This performance test shall be conducted as required by 40 CFR §63.1206(b)(12). Prior to the completion of the performance test, the permittee shall document compliance with 40 CFR Part 63, Subpart EEE no later than October 14, 2008. This documentation of compliance (DOC) will ensure that operating parameters are established to ensure compliance with this subpart. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(2)]
- The permittee may petition the Administrator to grant an extension of compliance with the emission standards of this subpart as provided by §63.6(i) and §63.1213.
 [Reg.19.304 and 40 C.F.R. § 63.1206(b)(4)]
- p. The permittee shall comply with the requirements of notification, performance testing, and waste-burning restrictions as outlined in §63.1206(b)(5)(i)(A) through (C) if the facility plans to make a change in design, operation, or maintenance that could adversely affect compliance. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(5)(i)]
- q. The permittee shall document any changes not affecting compliance in the facility operating record. Revisions reflecting such changes shall also be made, as necessary, to the performance test plan, Documentation of Compliance, Notification of Compliance, and the start-up, shutdown, and malfunction plan. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(5)(ii)]
- r. The permittee shall demonstrate destruction removal efficiency (DRE) of at least 99.99% during the comprehensive performance test conducted in compliance with the conditions of §63.1207(b)(1) of this subpart. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(7)]
- s. Any particulate matter and opacity standards or any permit or other emissions operating parameter limits or conditions, including any limitation on workplace practices, that are applicable to hazardous waste combustors to ensure compliance with any particulate matter or opacity standard of parts 60, 61, 63, 264, 265, and 266 of this chapter (i.e., any title 40 particulate or opacity standards) do not apply while the permittee conducts particulate matter continuous emissions monitoring system (CEMS) correlation tests. However, compliance with this condition is not

required until such time that the Agency promulgates all performance specifications and operational requirements applicable to PM CEMS. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(8)(i) and (ii)]

- t. For provisions of this section to apply, the permittee must develop a particulate matter CEMS correlation test plan that includes the following information. This test plan may be included as part of the comprehensive performance test plan required under §63.1207(e) and (f):
 - i. Number of test conditions and number of runs for each test condition;
 - ii. Target particulate matter emission level for each test condition;
- iii. How you plan to modify operations to attain the desired particulate matter emission levels; and
- iv. Anticipated normal emission levels.

The permittee shall submit the particulate CEMS correlation test plan to the Administrator for approval at least 90 calendar days before the correlation test is scheduled to be conducted. However, compliance with this condition is not required until such time that the Agency promulgates all performance specifications and operational requirements applicable to PM CEMS. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(8)(iii)(A) and (B)]

- u. If the Administrator fails to approve or disapprove the correlation test plan with the time period specified by §63.7(c)(3)(i), the plan is considered approved, unless the Administrator has requested additional information. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(8)(iv)]
- v. The particulate matter and associated operating limits and conditions will not be waived for more than 96 hours, in the aggregate, for a correlation test, including all runs of all test conditions, unless more time is approved by the Administrator. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(8)(v)]
- w. The permittee must return to operating conditions indicative of compliance with the applicable particulate matter and opacity standards as soon as possible after correlation testing is completed. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(8)(vii)]
- x. The permittee must calculate the hazardous waste residence time and include the calculation in the performance test plan under §63.1207(f) and the operating record. The permittee must also provide the hazardous waste residence time in the Documentation of Compliance under §63.1211(c) and the Notification of Compliance under §63.1207(j) and §63.1210(b). [Reg.19.304 and 40 C.F.R. § 63.1206(b)(11)]
- y. The permittee must conduct a minimum of three runs of a performance test required under §63.1207 to document compliance with the emission standards of this subpart. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(12)(i)]
- z. The permittee must document compliance with the emission standards based on the arithmetic average of the emission results of each run, except that the

permittee must document compliance with the destruction and removal efficiency standard for each run of the comprehensive performance test individually. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(12)(ii)]

General Operating Requirements

- aa. The permittee must operate only under the operating requirements specified in the Notification of Compliance under §63.1207(j) and §63.1210(b), except: [Reg.19.304 and 40 C.F.R. § 63.1206(c)(1)(i)]
 - i. During performance tests under approved test plans according to §63.1207(e), (f), and (g), [40 CFR §63.1206(c)(1)(i)(A)]
 - ii. Under the conditions of paragraph (b)(1)(i) or (ii) of this section [Reg.19.304 and 40 C.F.R. § 63.1206(c)(1)(i)(B)]
 - 1. The Notification of Compliance must contain operating requirements including, but not limited to, the operating requirements of this section and §63.1209. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(1)(ii)]
 - 2. Failure to comply with the operating requirements is failure to ensure compliance with the emissions standards of this subpart [Reg.19.304 and 40 C.F.R. § 63.1206(c)(1)(iii)]
 - 3. Operating requirements in the Notification of Compliance are applicable requirements for purposes of parts 70 and 71 of this chapter [Reg.19.304 and 40 C.F.R. § 63.1206(c)(1)(iv)]
 - 4. The operating requirements specified in the Notification of Compliance will be incorporated in the Title V permit. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(1)(v)]
 - bb. Except as provided in by paragraph (c)(2)(ii) of this section, the permittee is subject to the startup, shutdown, and malfunction plan requirements of §63.6(e)(3). [Reg.19.304 and 40 C.F.R. § 63.1206(c)(2)(i)]
 - i. If the permittee elects to comply with §270.235(a)(1)(iii), §270.235(a)(2)(iii), or §270.235(b)(1)(ii) of this chapter to address RCRA concerns, the permittee must comply with the provisions of §63.1206(c)(2)(ii)(A) and (B). [Reg.19.304 and 40 C.F.R. § 63.1206(c)(2)(ii)]
 - ii. The permittee must identify in the plan the projected oxygen correction factor based on normal operations to use during periods of startup and shutdown. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(2)(iii)]
 - iii. The permittee must record the plan in the operating record. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(2)(iv)]

- iv. The permittee must comply with this requirement for operation under the startup, shutdown, and malfunction plan. [Reg.19.304 and 40 C.F.R. 63.1206(c)(2)(v)]
- cc. Upon the compliance date, the permittee must operate the combustor with a functioning system that immediately and automatically cuts off the hazardous waste feed, except as provided by paragraph (c)(3)(viii) of this section, when the following conditions apply: [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(i)]
 - i. When operating parameter limits specified under §63.1209; an emission standard monitored by CEMS; and the allowable combustion chamber pressure; [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(i)(A)]
 - ii. When the span value of any CMS detector, except a CEMS, is met or exceeded; [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(i)(B)]
- Upon malfunction of a CMS monitoring an operating parameter limit specified under §63.1209 or an emission level; or [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(i)(C)]
- iv. When any component of the automatic waste feed cutoff system fails. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(i)(D)]
- dd. During an automatic waste feed cutoff (AWFCO) the permittee must continue to duct combustion gases to the air pollution control system while hazardous waste remains in the combustion chamber. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(ii)]
- ee. The permittee must continue to monitor during the cutoff the operating parameters for which limits are established under §63.1209 and the emissions required under that section to be monitored by a CEMS, and the permittee shall not restart the hazardous waste feed until the operating parameters and emission levels are within specified limits. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(iii)]
- ff. If the AWFCO system fails to automatically and immediately cutoff the flow of hazardous waste upon exceedance of a parameter required to be interlocked with the AWFCO system under paragraph (c)(3)(i) of this section, the permittee has failed to comply with the AWFCO requirements of paragraph (c)(3) of this section. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(iv)]
- gg. If, after any AWFCO, there is an exceedance of any emission standard or operating requirement, irrespective of whether the exceedance occurred while hazardous waste remained in the combustion chamber, the permittee shall investigate the cause of the AWFCO, take appropriate corrective measures to minimize future AWFCOs and record the findings and corrective measures in the operating record. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(v)]
- hh. For each set of 10 exceedances of an emissions standard or operating requirement while hazardous waste remains in the combustion chamber during a 60-day block period, the permittee must submit to the Administrator a written report within 5 calendar days of the 10th exceedance documenting the exceedances and the

results of the investigation and corrective measures taken. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(vi)(A)]

- ii. On a case-by-case basis, the Administrator may require excessive exceedance reporting when fewer than 10 exceedances occur during a 60-day block period. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(vi)(B)]
- jj. The AWFCO system and associated alarms must be tested at least weekly to verify operability, unless the permittee documents in the operating record that weekly inspections will unduly restrict or upset operations and that less frequent inspection will be adequate. At a minimum, the permittee must conduct operability testing at least monthly. The permittee must document and record in the operating record AWFCO operability test procedures and results. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(vii)]
- kk. The permittee may ramp down waste feed according to the requirements of §63.1206(c)(3)(viii), except as provided in §63.1206(c)(3)(B). The permittee must document ramp down procedures in the operating and maintenance plan. If the AWFCO is triggered by an exceedance of any of the following operating limits, the permittee may not ramp down the waste feed cutoff: Minimum combustion chamber temperature, maximum hazardous waste feedrate, or any hazardous waste firing system operating limits that may have been established. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(viii)]
- 11. The permittee is subject to the emergency safety vent (ESV) operating and reporting requirements set forth in this section. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(4)(i through iv)]
- mm. The permittee is subject to the combustion system leak control system operating and reporting requirements set forth in these sections. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(5)(i)(A) and (ii)]
- nn. The permittee is subject to the operator training and certification standards set forth in this section. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(6)(i through vii)]
- oo. The permittee must prepare and at all times operate according to an operation and maintenance plan which complies with the requirements set forth in these sections. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(7)(i)(A-D)]

Performance Testing Requirements

- pp. The permittee must conduct performance testing in accordance with the applicable requirements contained in this section. [Reg.19.304 and 40 C.F.R. § 63.1207(a-m)]
- qq. The permittee must commence the initial comprehensive performance test not later than six months after the compliance date. [Reg.19.304 and 40 C.F.R. § 63.1207(c)(1)]
- rr. The permittee must conduct testing periodically as described in paragraphs (d)(1) through (3) of this section. The date of commencement of the initial

comprehensive performance test is the basis for establishing the deadline to commence the initial confirmatory performance test and the next comprehensive performance test. The permittee may conduct performance testing at any time prior to the required date. The deadline for commencing subsequent confirmatory and comprehensive performance testing is based on the date of commencement of the previous comprehensive performance test. [Reg.19.304 and 40 C.F.R. § 63.1207(d)(1) through (3)]

- i. The permittee must commence comprehensive testing no later than 61 months after the date of commencing the previous comprehensive performance test.
- ii. The permittee must commence confirmatory performance testing no later than 31 months after the date of commencing the previous comprehensive performance test. To ensure that the confirmatory test is conducted approximately midway between comprehensive performance tests, the Administrator will not approve a test plan that schedules testing within 18 months of commencing the previous comprehensive performance test.
- iii. The permittee must complete performance testing within 60 days after the date of commencement, unless the Administrator determines that a time extension is warranted based on documentation in writing of factors beyond the permittee's control that prevent testing from being completed within 60 days.

Applicable Testing Requirements under the Interim Standard

- ss. Waiver of periodic comprehensive performance tests. Except as provided by §63.1207(c)(2), the permittee must conduct only an initial comprehensive performance test under the interim standards (i.e., the standards published in the Federal Register on February 13, 2002). All subsequent comprehensive performance testing requirements are waived under the interim standards. The provisions in the introductory test to paragraph (d) and in paragraph (d)(1) of this section do not apply until EPA promulgates permanent replacement standards pursuant to the Settlement Agreement noticed in the Federal Register on November 16, 2001. [Reg.19.304 and 40 C.F.R. § 63.1207(d)(4)(i)].
- tt. Waiver of periodic confirmatory performance tests. The permittee is not required to conduct a confirmatory test under the interim standards (i.e., the standards published in the Federal Register on February 13, 2002). The confirmatory testing requirements in the introductory text to paragraph (d) and in (d)(2) of §63.1207 are waived until EPA promulgates permanent replacement standards pursuant to the Settlement Agreement noticed in the Federal Register on November 16, 2001. [Reg.19.304 and 40 C.F.R. § 63.1207(d)(4)(ii)].
- uu. The permittee must submit to the Administrator a notification of intent to conduct a comprehensive performance test and CMS performance evaluation and a site specific test plan and CMS performance evaluation plan at least one year before

the performance test and performance evaluation are scheduled to begin. [Reg.19.304 and 40 C.F.R. § 63.1207(e)(1)(i)]

- vv. The permittee must submit to the Administrator a notification of intent to conduct the comprehensive performance test at least 60 calendar days before the test is scheduled to begin. [Reg.19.304 and 40 C.F.R. § 63.1207(e)(1)(i)(B)]
- ww. The permittee must submit to the Administrator a notification of intent to conduct a confirmatory performance test and CMS performance evaluation and a test plan and CMS performance evaluation plan at least 60 calendar days before the performance test is scheduled to begin. [Reg.19.304 and 40 C.F.R. § 63.1207(e)(1)(ii)]

Test Methods

xx. The permittee shall use the test methods contained in this section when determining compliance with the emissions standards of this subpart. [Reg.19.304 and 40 C.F.R. § 63.1208(a-b)]

Monitoring Requirements

- yy. The permittee is subject to the applicable monitoring requirements contained in these sections. [Reg.19.304 and 40 C.F.R. § 63.1209 (a-q)]
- zz. The permittee must either use a carbon monoxide or hydrocarbon CEMS to demonstrate compliance with either the carbon monoxide and hydrocarbon standards under this subpart. The permittee must also use an oxygen CEMS to continuously correct the carbon monoxide and hydrocarbon levels to 7 percent oxygen. [Reg.19.304 and 40 C.F.R. § 63.1209(a)(1)(i)]
- aaa. The permittee must install, calibrate, maintain, and operate a particulate matter CEMS to demonstrate and monitor compliance with the particulate matter standards under this subpart. However, compliance with this condition is not required until such time that the Agency promulgates all performance specifications and operational requirements applicable to PM CEMS. [Reg.19.304 and 40 C.F.R. § 63.1209(a)(1)(iii)]
- bbb. The permittee must install, calibrate, maintain, and continuously operate the CEMS in compliance with the quality assurance procedures provided in the appendix to this subpart and Performance Specifications 1 (opacity), 4B (carbon monoxide and oxygen), and 8A (hydrocarbons) in Appendix B, Part 60 of this chapter. [Reg.19.304 and 40 C.F.R. § 63.1209(a)(2)]
- ccc. The permittee must comply with the span requirements of § 63.1209(a)(4). [Reg.19.304 and 40 C.F.R. § 63.1209(a)(4)]
- ddd. The permittee may petition the Administrator to use CEMS for compliance monitoring for other standards in lieu of compliance with the corresponding operating parameter limits under this section. [Reg.19.304 and 40 C.F.R. § 63.1209(a)(5)]

- eee. The permittee will begin recording one-minute and hourly rolling average values as necessary to ensure that 60 one-minute values will be available for calculating the initial hourly rolling average before the compliance date. The permittee will continue to use the CEMS to monitor parameters as required in § 63.1209(a)(6). [Reg.19.304 and 40 C.F.R. § 63.1209(a)(6)]
- fff. The permittee will use Continuous Monitoring Systems where necessary to ensure compliance with operating parameters established in the Documentation of Compliance or the Notification of Compliance. [Reg.19.304 and 40 C.F.R. § 63.1209(b)]
- ggg. Prior to feeding the material, the permittee must obtain an analysis of each feedstream that is sufficient to document compliance with the applicable feedrate limits provided in this section. [Reg.19.304 and 40 C.F.R. § 63.1209(c)(1)]
- hhh. The permittee must develop and implement a feedstream analysis plan and record it in the operating record. [Reg.19.304 and 40 C.F.R. § 63.1209(c)(2)]
 - iii. The permittee must submit the feedstream analysis plan to the Administrator for review and approval, if requested. [Reg.19.304 and 40 C.F.R. § 63.1209(c)(3)]
 - jjj. To comply with the applicable feedrate limits of this section, the permittee must monitor and record the feedrates as follows: [Reg.19.304 and 40 C.F.R. § 63.1209(c)(4)]
 - i. Determine and record the value of the parameter for each feedstream by sampling and analysis or other method;
 - ii. Determine and record the mass or volume flowrate of each stream by a CMS. If the permittee determines flowrate of a feedstream by volume, the permittee must determine and record the density of the feedstream by sampling and analysis (unless the permittee reports the constituent concentration in units of weight per volume); and
 - iii. Calculate and record the mass feedrate of the parameter per unit time.
- kkk. The requirements of §63.8(d) (Quality control program) and (e) (Performance evaluation of continuous monitoring systems) apply, except that the permittee must conduct performance evaluations components of the CMS under the frequency and procedures (for example, submittal of performance evaluation test plan for review and approval) applicable to performance tests as provided by §63.1207. [Reg.19.304 and 40 C.F.R. § 63.1209(d)(1)]
- III. The permittee shall maintain and operate each CMS as specified in §63.8(c), except for § 3.8(c)(3) and §63.8(c)(4)(ii). The permittee shall have the CMS installed, calibrated, and operational on the compliance date. The permittee must sample the regulated parameter without interruption, and evaluate the detector response at least once each 15 seconds, and compute and record the average values at least every 60 seconds. [Reg.19.304 and 40 C.F.R. § 63.1209(f)]

- mmm. The permittee shall follow the requirements for the reduction of monitoring data as specified in 40 CFR §63.8(g). [Reg.19.304 and 40 C.F.R. § 63.1209(h)]
 - nnn. When one operating parameter is used to ensure compliance with one or more standards, the permittee must use the most stringent limit, determined during the comprehensive performance test, as the limit for that operating parameter. [Reg.19.304 and 40 C.F.R. § 63.1209(i)]
- ooo. To remain in compliance with the destruction and removal efficiency (DRE) standards, the permittee must establish operating limits during the comprehensive performance test (or during a previous DRE test under provisions of §63.1206(b)(7)) for the following parameters, unless the limits are based on manufacturer specifications and comply with those limits at all times that hazardous waste remains in the combustion chamber. [Reg.19.304 and 40 C.F.R. § 63.1209(j)]
- ppp. The permittee must measure the temperature of each combustion chamber at locations that best represents, as practicable, the bulk gas temperature in the combustion zone. The permittee must document the temperature measurement location in the test plan submitted under §63.1207(e), and establish a minimum rolling average limit as the average of the test run values. [Reg.19.304 and 40 C.F.R. § 63.1209(j)(1)(i) and (ii)]
- qqq. As an indicator of gas residence time in the control device, the permittee must establish and comply with a limit on the maximum flue gas flowrate, the maximum production rate, or another parameter that is documented in the site-specific test plan as an appropriate surrogate for gas residence time, as the average of the maximum hourly rolling averages for each run. [Reg.19.304 and 40 C.F.R. § 63.1209(j)(2)(i)]
 - rrr. The permittee must establish limits on the maximum pumpable and total (i.e., pumpable and nonpumpable) hazardous waste feedrate for each location where hazardous waste is fed. [Reg.19.304 and 40 C.F.R. § 63.1209(j)(3)(i)]
 - sss. The permittee must specify operating parameters and limits to ensure that good operation of each hazardous waste firing system is maintained. [Reg.19.304 and 40 C.F.R. § 63.1209(j)(4)]
 - ttt. The permittee must comply with the dioxin and furans emission standard by establishing and complying with the following operating parameter limits. You must base the limits on operations during the comprehensive performance test, unless the limits are based on manufacturer specifications. [Reg.19.304 and 40 C.F.R. § 63.1209(k)]
- uuu. The permittee must establish and comply with a limit on the maximum temperature of the gas at the inlet to the electrostatic precipitator (ESP) on an hourly rolling average. The permittee must establish the hourly rolling average limit as the average of the test run averages. [Reg.19.304 and 40 C.F.R. § 63.1209(k)(1)(i)]
- vvv. The permittee must measure the temperature of each combustion chamber at a location that best represents, as practicable, the bulk gas temperature in the combustion zone. The permittee must document the temperature measurement location in the test plan and establish a minimum hourly rolling average limit as the average of the test runs. [Reg.19.304 and 40 C.F.R. § 63.1209(k)(2)(i) and (ii)]
- www. As an indicator of gas residence time in the control device, the permittee must establish and comply with a limit on the maximum flue gas flowrate, the maximum production rate , or another parameter which is an appropriate surrogate for residence time, as the hourly rolling averages for each run. Compliance with this limit is on an hourly rolling average basis. [Reg.19.304 and 40 C.F.R. § 63.1209(k)(3)(i) and (ii),]
- xxx. The permittee must establish limits on the maximum pumpable and total (pumpable and nonpumpable) waste feedrate for each location where waste is fed and establish limits as the average of the maximum hourly rolling averages for each run. Compliance shall be based on an hourly rolling average basis. [Reg.19.304 and 40 C.F.R. § 63.1209(k)(4)(i B iii)]
- yyy. The permittee shall ensure compliance with the mercury emission standard by establishing a maximum mercury feed rate limit. The limit is established as a 12-hour rolling average limit for the total feedrate of mercury in all feedstreams as the average of the test run values, unless mercury feedrate limits are extrapolated from performance test feedrate levels as described under §63.1209(l)(v). [Reg.19.304 and 40 C.F.R. § 63.1209(l)]
- zzz. The permittee must comply with the particulate matter emission standard by establishing and complying with the applicable operating parameter limits found in §63.1209(m) of this subpart. [Reg.19.304 and 40 C.F.R. § 63.1209(m)]
- aaaa. As an indicator of gas residence time in the control device, the permittee must establish and comply with a limit on the maximum flue gas flowrate, the maximum production rate, or another parameter that is documented in the sitespecific test plan as an appropriate surrogate for gas residence time, as the average of the maximum hourly rolling averages for each run. [Reg.19.304 and 40 C.F.R. § 63.1209(m)(2)]
- bbbb. The permittee must establish a maximum ash feedrate limit as a 12-hour rolling average based on the average of the test run averages. [Reg.19.304 and 40 C.F.R. § 63.1209(m)(3)]
- cccc. The permittee must comply with the semivolatile metal (cadmium and lead) and low volatile metal (arsenic, beryllium, and chromium) emission standards by establishing and complying with the following operating parameter limits: [Reg.19.304 and 40 C.F.R. § 63.1209(n)]
 - i. The permittee must establish a limit on the maximum inlet temperature to the ESP on an hourly rolling average basis as the average of the test run averages. [Reg.19.304 and 40 C.F.R. § 63.1209(n)(1)]

- ii. The permittee must establish a total feed rate limit for semivolatile metals and low volatile metals, with compliance based on 12-hour rolling average limits as the average of the test run averages unless the metal feedrate limits are extrapolated from performance test feedrate levels as described under §63.1209(n)(2)(vii). [Reg.19.304 and 40 C.F.R. § 63.1209(n)(2)(ii)]
- iii. The permittee must establish a separate feedrate limit for low volatile metals in pumpable feedstreams, with compliance based on 12-hour rolling average limits as the average of the test run averages, unless the metal feedrate limits are extrapolated from performance test feedrate levels as described under §63.1209(n)(2)(vii). [Reg.19.304 and 40 C.F.R. § 63.1209(n)(2)(vi)]
- dddd. The permittee must establish a 12-hour rolling average limit for the total feedrate of chlorine in all feedstreams as the average of the test run averages. [Reg.19.304 and 40 C.F.R. § 63.1209(o)(1)(i)]
- eeee. As an indicator of gas residence time in the control device, the permittee must establish a limit on the maximum flue gas flowrate, the maximum production rate, or another parameter documented in the site-specific test plan as an appropriate surrogate for gas residence time, as the average of the maximum hourly rolling averages for each run. This limit must be maintained on an hourly rolling average basis. [Reg.19.304 and 40 C.F.R. § 63.1209(o)(2)(i)]
- ffff. The permittee must perform instantaneous monitoring of pressure and the automatic waste feed cutoff system must be engaged when negative pressure is not adequately maintained. [Reg.19.304 and 40 C.F.R. § 63.1209(p)]
- gggg. The permittee may choose to operate under different modes of operation as described in this section. The permittee must establish operating parameter limits for each mode and must document in the operating record when the permittee changes modes of operation and begins complying with the operating limits for an alternative mode of operation. [Reg.19.304 and 40 C.F.R. § 63.1209(q)]
- hhhh.The permittee may elect to use shorter averaging periods than those specified in this section. [Reg.19.304 and 40 C.F.R. § 63.1209(r)]

Notification Requirements

- iiii. The permittee shall submit all of the applicable notifications prior to the deadlines established in this subpart. [Reg.19.304 and 40 C.F.R. § 63.1210(a)(1)]
- jjjj. The permittee must submit the required notifications outlined in this section to the Administrator in order to request or elect to comply with the alternative requirements contained in this subpart. [Reg.19.304 and 40 C.F.R. § 63.1210(a)(2)]
- kkkk. Upon postmark of the Notification of Compliance, the operating parameter limits identified in the Notification of Compliance, as applicable, shall be complied with, the limits identified in the Document of Compliance or a previous

Notification of Compliance are no longer applicable. [Reg.19.304 and 40 C.F.R. § 63.1210(b)(2)]

Recordkeeping and Reporting Requirements

IIII. The permittee shall submit the reports required by this subpart to the Administrator prior to the deadlines set forth in this subpart. [Reg.19.304 and 40 C.F.R. § 63.1211]

Procedure for Extending the Compliance Date

mmmm. The permittee may request an extension of the compliance date to install pollution prevention or waste minimization controls provided that the conditions outlined in this section are met. [Reg.19.304 and 40 C.F.R. § 63.1213]

40 CFR Part 63 Subpart DDDDD—National Emission Standards for Hazardous Air Pollutants for Major Sources: Industrial, Commercial, and Institutional Boilers and Process Heaters

- US 53. Sources SN-6M06-01 and SN-6M07-01 are subject to the provisions of 40 CFR Part 63, Subpart DDDDD – National Emission Standards for Hazardous Air Pollutants for Major Sources: Industrial, Commercial, and Institutional Boilers and Process Heaters. The applicable provisions of this subpart include, but are not limited to, the items found in Specific Conditions US 53 through US 76. [Reg.19.304 and 40 C.F.R. §§ 63.6585, 63.6590(c), 63.6595(a) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- US 54. The permittee must meet the requirements in paragraphs (a)(1) through (3) of §63.7500, except as provided in paragraphs (b), through (e) of §63.7500. The permittee must meet these requirements at all times the affected unit is operating, except as provided in paragraph (f) of §63.7500. [Reg.19.304 and 40 C.F.R. § 63.7500(a)]
- US 55. The permittee must meet each emission limit and work practice standard in Tables 1 through 3, and 11 through 13 to 40 CFR Part 63, Subpart DDDDD that applies to the permittee's boiler or process heater, for each boiler or process heater at the permittee's source, except as provided under §63.7522. The output-based emission limits, in units of pounds per million Btu of steam output, in Tables 1 or 2 to Subpart DDDDD are an alternative applicable only to boilers and process heaters that generate steam. The output-based emission limits, in units of pounds per megawatt-hour, in Tables 1 or 2 to Subpart DDDDD are an alternative applicable only to boilers that generate electricity. If the permittee operates a new boiler or process heater, the permittee can choose to comply with alternative limits as discussed in paragraphs (a)(1)(i) through (a)(1)(iii) of §63.7500, but on or after January 31, 2016, the permittee must comply with the emission limits in Table 1 to Subpart DDDDD.
 - a. If the permittee's boiler or process heater commenced construction or reconstruction after June 4, 2010 and before May 20, 2011, the permittee may comply with the emission limits in Table 1 or 11 to Subpart DDDDD until January 31, 2016.

- b. If the permittee's boiler or process heater commenced construction or reconstruction after May 20, 2011 and before December 23, 2011, the permittee may comply with the emission limits in Table 1 or 12 to Subpart DDDDD until January 31, 2016.
- c. If the permittee's boiler or process heater commenced construction or reconstruction after December 23, 2011 and before January 31, 2013, the permittee may comply with the emission limits in Table 1 or 13 to Subpart DDDDD until January 31, 2016.

[Reg.19.304 and 40 C.F.R. § 63.7500(a)(1)(i-iii)]

- US 56. At all times, the permittee must operate and maintain any affected source (as defined in §63.7490), including associated air pollution control equipment and monitoring equipment, in a manner consistent with safety and good air pollution control practices for minimizing emissions. Determination of whether such operation and maintenance procedures are being used will be based on information available to the Administrator that may include, but is not limited to, monitoring results, review of operation and maintenance procedures, review of operation and maintenance records, and inspection of the source. [Reg.19.304 and 40 C.F.R. § 63.7500(a)(3)]
- US 57. Boilers and process heaters in the units designed to burn gas 1 fuels subcategory are not subject to the emission limits in Tables 1 and 2 or 11 through 13 to Subpart DDDDD, or the operating limits in Table 4 to Subpart DDDDD. [Reg.19.304 and 40 C.F.R. § 63.7500(e)]
- US 58. These standards apply at all times the affected unit is operating, except during periods of startup and shutdown during which time the permittee must comply only with Table 3 to Subpart DDDDD. [Reg.19.304 and 40 C.F.R. § 63.7500(f)]
- US 59. In response to an action to enforce the standards set forth in §63.7500 the permittee may assert an affirmative defense to a claim for civil penalties for violations of such standards that are caused by malfunction, as defined at §63.2. Appropriate penalties may be assessed if the permittee fail to meet the permittee's burden of proving all of the requirements in the affirmative defense. The affirmative defense shall not be available for claims for injunctive relief. [Reg.19.304 and 40 C.F.R. § 63.7501]
- US 60. The permittee must be in compliance with the emission limits, work practice standards, and operating limits in this subpart. These limits apply to the permittee at all times the affected unit is operating except for the periods noted in §63.7500(f). [Reg.19.304 and 40 C.F.R. § 63.7505(a)]
- US 61. For existing affected sources (as defined in §63.7490), the permittee must complete the initial compliance demonstration, as specified in paragraphs (a) through (d) of §63.7510, no later than 180 days after the compliance date that is specified for the permittee's source in §63.7495 and according to the applicable provisions in §63.7(a)(2) as cited in Table 10 to Subpart DDDDD, except as specified in paragraph (j) of §63.7510. The permittee must

complete an initial tune-up by following the procedures described in 63.7540(a)(10)(i) through (vi) no later than the compliance date specified in 63.7495, except as specified in paragraph (j) of 63.7510. The permittee must complete the one-time energy assessment specified in Table 3 to Subpart DDDDD no later than the compliance date specified in 63.7495, except as specified in paragraph (j) of 63.7510. [Reg.19.304 and 40 C.F.R. 63.7510(e)]

- US 62. If the permittee is required to meet an applicable tune-up work practice standard, the permittee must conduct an annual, biennial, or 5-year performance tune-up according to §63.7540(a)(10), (11), or (12), respectively. Each annual tune-up specified in §63.7540(a)(10) must be no more than 13 months after the previous tune-up. Each biennial tune-up specified in §63.7540(a)(11) must be conducted no more than 25 months after the previous tune-up. Each 5-year tune-up specified in §63.7540(a)(12) must be conducted no more than 61 months after the previous tune-up. For a new or reconstructed affected source (as defined in §63.7490), the first annual, biennial, or 5-year tune-up must be no later than 13 months, 25 months, or 61 months, respectively, after the initial startup of the new or reconstructed affected source. [Reg.19.304 and 40 C.F.R. § 63.7515(d)]
- US 63. For affected sources (as defined in §63.7490) that have not operated since the previous compliance demonstration and more than one year has passed since the previous compliance demonstration, the permittee must complete the subsequent compliance demonstration, if subject to the emission limits in Tables 1, 2, or 11 through 13 to Subpart DDDDD, no later than 180 days after the re-start of the affected source and according to the applicable provisions in §63.7(a)(2) as cited in Table 10 to Subpart DDDDD. The permittee must complete a subsequent tune-up by following the procedures described in §63.7540(a)(10)(i) through (vi) and the schedule described in §63.7540(a)(13) for units that are not operating at the time of their scheduled tune-up. [Reg.19.304 and 40 C.F.R. § 63.7515(g)]
- US 64. The permittee must include with the Notification of Compliance Status a signed certification that the energy assessment was completed according to Table 3 to Subpart DDDDD and is an accurate depiction of the permittee's facility at the time of the assessment. [Reg.19.304 and 40 C.F.R. § 63.7530(e)]
- US 65. The permittee must submit the Notification of Compliance Status containing the results of the initial compliance demonstration according to the requirements in §63.7545(e). [Reg.19.304 and 40 C.F.R. § 63.7530(f)]
- US 66. The permittee must demonstrate continuous compliance with each emission limit in Tables 1 and 2 or 11 through 13 to Subpart DDDDD, the work practice standards in Table 3 to Subpart DDDDD, and the operating limits in Table 4 to Subpart DDDDD that applies to the permittee according to the methods specified in Table 8 to Subpart DDDDD and paragraphs (a)(1) through (19) of §63.7540. [Reg.19.304 and 40 C.F.R. § 63.7540(a)]

- US 67. If the unit is not operating on the required date for a tune-up, the tune-up must be conducted within 30 calendar days of startup. [Reg.19.304 and 40 C.F.R. § 63.7540(a)(13)]
- US 68. The permittee must submit to the Administrator all of the notifications in §§63.7(b) and (c), 63.8(e), (f)(4) and (6), and 63.9(b) through (h) that apply to the permittee by the dates specified. [Reg.19.304 and 40 C.F.R. § 63.7545(a)]
- US 69. If the permittee is required to conduct an initial compliance demonstration as specified in §63.7530, the permittee must submit a Notification of Compliance Status according to §63.9(h)(2)(ii). For the initial compliance demonstration for each boiler or process heater, the permittee must submit the Notification of Compliance Status, including all performance test results and fuel analyses, before the close of business on the 60th day following the completion of all performance test and/or other initial compliance demonstrations for all boiler or process heaters at the facility according to §63.10(d)(2). The Notification of Compliance Status report must contain all the information specified in paragraphs (e)(1) through (8), as applicable. If the permittee is not required to conduct an initial compliance demonstration as specified in §63.7530(a), the Notification of Compliance Status must only contain the information specified in paragraphs (e)(1) and (8).
 - a. A signed certification that the permittee have met all applicable emission limits and work practice standards.
 - b. If the permittee had a deviation from any emission limit, work practice standard, or operating limit, the permittee must also submit a description of the deviation, the duration of the deviation, and the corrective action taken in the Notification of Compliance Status report.
 - c. In addition to the information required in §63.9(h)(2), the permittee's notification of compliance status must include the following certification(s) of compliance, as applicable, and signed by a responsible official:
 - i. "This facility complies with the required initial tune-up according to the procedures in §63.7540(a)(10)(i) through (vi)."
 - ii. "This facility has had an energy assessment performed according to §63.7530(e)."

[Reg.19.304 and 40 C.F.R. § 63.7545(e)(6-8)]

- US 70. The permittee must submit each report in Table 9 to Subpart DDDDD that applies to the permittee. [Reg.19.304 and 40 C.F.R. § 63.7550(a)]
- US 71. Unless the EPA Administrator has approved a different schedule for submission of reports under §63.10(a), the permittee must submit each report, according to paragraph (h) of §63.7550, by the date in Table 9 to Subpart DDDDD and according to the requirements in paragraphs (b)(1) through (4) of §63.7550. For units that are subject only to a requirement to conduct an annual, biennial, or 5-year tune-up according to §63.7540(a)(10), (11), or (12), respectively, and not subject to emission limits or operating limits, the permittee may submit only an annual, biennial, or 5-year compliance report, as applicable, as specified in paragraphs (b)(1) through (4) of §63.7550, instead of a semi-annual compliance report.

- a. The first compliance report must cover the period beginning on the compliance date that is specified for each boiler or process heater in §63.7495 and ending on July 31 or January 31, whichever date is the first date that occurs at least 180 days (or 1, 2, or 5 years, as applicable, if submitting an annual, biennial, or 5-year compliance report) after the compliance date that is specified for the permittee's source in §63.7495.
- b. The first compliance report must be postmarked or submitted no later than July 31 or January 31, whichever date is the first date following the end of the first calendar half after the compliance date that is specified for each boiler or process heater in §63.7495. The first annual, biennial, or 5-year compliance report must be postmarked or submitted no later than January 31.
- c. Each subsequent compliance report must cover the semiannual reporting period from January 1 through June 30 or the semiannual reporting period from July 1 through December 31. Annual, biennial, and 5-year compliance reports must cover the applicable 1-, 2-, or 5-year periods from January 1 to December 31.
- d. Each subsequent compliance report must be postmarked or submitted no later than July 31 or January 31, whichever date is the first date following the end of the semiannual reporting period. Annual, biennial, and 5-year compliance reports must be postmarked or submitted no later than January 31.

[Reg.19.304 and 40 C.F.R. § 63.7550(b)(1-4)]

- US 72. A compliance report must contain the following information depending on how the facility chooses to comply with the limits set in this rule.
 - a. If the facility is subject to the requirements of a tune up they must submit a compliance report with the information in paragraphs (c)(5)(i) through (iv) and (xiv) of §63.7550.

[Reg.19.304 and 40 C.F.R. § 63.7550(c)(1)]

- US 73. Company and Facility name and address.
 - a. Process unit information, emissions limitations, and operating parameter limitations.
 - b. Date of report and beginning and ending dates of the reporting period.
 - c. The total operating time during the reporting period.
 - Include the date of the most recent tune-up for each unit subject to only the requirement to conduct an annual, biennial, or 5-year tune-up according to §63.7540(a)(10), (11), or (12) respectively. Include the date of the most recent burner inspection if it was not done annually, biennially, or on a 5-year period and was delayed until the next scheduled or unscheduled unit shutdown. [Reg.19.304 and 40 C.F.R. § 63.7550(c)(5)(i-xiv)]
- US 74. The permittee must keep records according to paragraphs (a)(1) and (2) of §63.7555.
 - a. A copy of each notification and report that the permittee submitted to comply with this subpart, including all documentation supporting any Initial Notification

or Notification of Compliance Status or semiannual compliance report that the permittee submitted, according to the requirements in (63.10(b)(2)(xiv)).

- Records of performance tests, fuel analyses, or other compliance demonstrations and performance evaluations as required in §63.10(b)(2)(viii).
 [Reg.19.304 and 40 C.F.R. § 63.7555(a)(1-2)]
- US 75. The permittee's records must be in a form suitable and readily available for expeditious review, according to §63.10(b)(1). [Reg.19.304 and 40 C.F.R. § 63.7560(a)]
- US 76. As specified in §63.10(b)(1), the permittee must keep each record for 5 years following the date of each occurrence, measurement, maintenance, corrective action, report, or record. [Reg.19.304 and 40 C.F.R. § 63.7560(b)]
- US 77. The permittee must keep each record on site, or they must be accessible from onsite (for example, through a computer network), for at least 2 years after the date of each occurrence, measurement, maintenance, corrective action, report, or record, according to §63.10(b)(1). The permittee can keep the records off site for the remaining 3 years. [Reg.19.304 and 40 C.F.R. § 63.7560(c)]

40 CFR Part 61 Subpart E—National Emission Standard for Mercury

- US 78. The emissions from 6M01-01 shall not exceed 3.2 kg (7.1 lb) of mercury per 24-hour period. [Reg.19.304 and 40 C.F.R. § 61.51]
- US 79. The permittee shall sample and test the wastewater treatment plant sludge using EPA Reference Method 105 and in accordance with § 61.54. [Reg.19.304 and 40 C.F.R. § 61.52]
- US 80. If mercury emissions are in excess 1.6 kg (3.5 lb) per 24-hour period, demonstrated either by stack sampling according to § 61.53 or sludge sampling according to § 61.54, shall monitor mercury emissions at intervals of at least once per year by use of Method 105 of appendix B or the procedures specified in § 61.53 (d) (2) and (4). The results of monitoring shall be reported and retained according to § 61.53(d) (5) and (6) or § 61.54 (f) and (g). [Reg.19.304 and 40 C.F.R. § 61.54]

Organic Sulfonation Process

5M01-01, 5M01-02, 5M01-05, 5M01-06, 5M01-07, 5M01-08 5M01-09, 5M03-01, 5M03-02, 5M04-01, 5M04-02, 5M04-10, 5M05-01, 5M05-02, 5M11-01, 5M11-04, 5M11-05, 5M11-06, 5M11-07, 5M11-15, 5M13-01, 5M16-01, 5M18-01, 5M18-02, 5M18-03, 5MNOBS-TNK, NOBS-FUG, 5M01-TSP

Source Description

The organic sulfonate facility produces a solid material for use as a household consumer product. The two organic sulfonation facilities include reactors, centrifuges, scrubbers, distillation equipment, raw materials and process tanks. Scrubbers are the primary means for controlling emissions from the production facilities. The phenol and solvent storage tanks vent to a scrubber. The low vapor pressures of the contents of the storage tanks minimize the potential for VOC emissions from these emission points.

NSPS subpart NNN (SOCMI Distillation Operations) applies to a scrubber associated with an acetic acid distillation column (5M01-02).

NSPS Subpart VV (SOCMI VOC Equipment Leaks) applies to certain equipment in this process such as pumps, compressors, pressure relief devices, sampling connection systems, and valves.

NSPS Subpart Kb (VOC Storage Vessels) applies to several tanks in the organic sulfonate production area. These requirements are located within the Plantwide Conditions.

Specific Conditions

OSP 1. The permittee shall not exceed the emission rates set forth in the following table. Emission rates are based upon the maximum capacity of equipment. [Reg.19.501 *et seq.* and 40 C.F.R. § 52 Subpart E]

SN	Description	Pollutant	lb/hr	tpy
5M01-01	Scrubber	VOC	0.1	0.4
5M01-02	Scrubber	VOC	0.1	0.4
5M01-05	Scrubber	VOC	0.1	0.4
5M01-06	Scrubber	VOC	0.5	1.8
5M01-07	Scrubber	VOC	0.1	0.4
5M01-08	Scrubber	VOC	0.1	0.4
5M01-09	Scrubber	VOC	0.2	0.9
5M03-01	Scrubber	VOC	0.1	0.4
5M03-02	Scrubber	VOC	0.2	0.8

SN	Description	Pollutant	lb/hr	tpy
5M04-01	Scrubber	VOC	0.6	2.3
5M04-02	Scrubber	VOC	0.2	0.7
5M04-10	Scrubber	SO_2	0.1	0.4
5M05-01	Scrubber	VOC	0.1	0.4
5M05-02	Filter	PM ₁₀	0.1	0.4
5M11-01	Scrubber	VOC	0.1	0.4
5M11-04	Scrubber	VOC	0.1	0.4
5M11-05	Scrubber	VOC	0.1	0.4
5M11-06	Scrubber	VOC	0.1	0.4
5M11-07	Scrubber	VOC	0.1	0.4
5M11-15	SPS Supersack Load Hopper Dust Control System	PM ₁₀	0.1	0.3
5M13-01	Scrubber	VOC	0.1	0.4
5M16-01	Supersack Loadout Dust Control System	PM ₁₀	0.1	0.4
5M18-01	Continuous Dust Control System	PM ₁₀	3.9	17.1
5M18-02	Central Vacuum Cleaning System	PM_{10}	3.4	3.7
5M18-03	Bin Vacuum Cleaning System	PM_{10}	0.3	0.9
5MNOBS- TNK	Aggregate Tank (4 tanks)	VOC	0.6	2.6
NOBS- FUG	Fugitive Emissions from Organic Sulfonation Process	VOC	1.2	5.3
5M01-TSP	Dust Control Maintenance Fugitives	PM10	3.1	0.2

OSP 2. The permittee shall not exceed the emission rates set forth in the following table. Emission rates are based upon the maximum capacity of equipment. The facility shall show compliance with the facility total ton/yr limits using the procedures outlined in Plantwide Conditions 12 through 16. Hourly Plantwide Hazardous Air Pollutant emissions are limited by Plantwide Conditions 16. Additional HAP limitations are included in Plantwide Condition 13. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

SN	Description	Pollutant	lb/hr	tpy
5M01-01	Scrubber	Organic Pollutants**	0.1	0.4
5M01-02	Scrubber	Organic Pollutants**	0.1	0.4
5M01-05	Scrubber	Organic Pollutants**	0.1	0.4
5M01-06	Scrubber	Organic Pollutants**	0.5	1.8
5M01-07	Scrubber	Organic Pollutants**	0.1	0.4
5M01-08	Scrubber	Organic Pollutants**	0.1	0.4
5M01-09	Scrubber	Organic Pollutants**	0.2	0.9
5M03-01	Scrubber	Organic Pollutants**	0.1	0.4
5M03-02	Scrubber	Organic Pollutants**	0.2	0.8
5M04-01	Scrubber	Organic Pollutants**	0.6	2.3
5M04-02	Scrubber	Organic Pollutants**	0.2	0.7
5M05-01	Scrubber	Organic Pollutants**	0.1	0.4
5M05-02	Filter	PM	0.1	0.4
5M11-01	Scrubber	Organic Pollutants**	0.1	0.4
5M11-04	Scrubber	Organic Pollutants**	0.1	0.4
5M11-05	Scrubber	Organic Pollutants**	0.1	0.4
5M11-06	Scrubber	Organic Pollutants**	0.1	0.4
5M11-07	Scrubber	Organic Pollutants**	0.1	0.4
5M11-15	SPS Supersack Load Hopper Dust Control System	РМ	0.1	0.3
5M13-01	Scrubber	Organic Pollutants**	0.1	0.4
5M16-01	Supersack Loadout Dust Control System	РМ	0.1	0.4
5M18-01	Continuous Dust Control System	РМ	3.9	17.1
5M18-02	Central Vacuum Cleaning System	РМ	3.4	3.7

SN	Description	Pollutant	lb/hr	tpy
5M18-03	Bin Vacuum Cleaning System	PM	0.3	0.9
5MNOBS- TNK	Aggregate Tank (5 tanks)	Organic Pollutants**	0.6	2.6
NOBS- FUG	Fugitive Emissions from Organic Sulfonation Process	Organic Pollutants**	1.2	5.3
5M01-TSP	Dust Control Maintenance Fugitives	РМ	3.1	0.2

*Inorganics are considered to be non-organic Hazardous Air Pollutants

**Organic Pollutants are considered to be VOC, all organic HAPs, and organic Air Contaminates.

- OSP 3. The permittee shall not exceed 5% opacity over a three (3) hour period at sources 5M05-02, 5M11-15, 5M16-01, 5M18-01, 5M18-02, and 5M18-03. Compliance with this limit shall be demonstrated as outlined in the most current version of the Facility Operating Plan. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]
- OSP 4. The permittee shall conduct weekly visual inspections at 5M05-02, 5M11-15, 5M16-01, 5M18-01, 5M18-02, and 5M18-03 for possible emissions using EPA Method 22 and monthly observations of the system using EPA Method 9. The permittee shall record the presence of any excessive visible emissions and the subsequent actions taken to correct the exceedance. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]
- OSP 5. The permittee shall measure the pressure drop at least daily at 5M05-02, 5M11-15, 5M16-01, 5M18-01, 5M18-02, and 5M18-03 as outlined in the most current version of the Facility Operating Plan. [Reg.19.705, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 70.6]
- OSP 6. The permittee shall keep records on site of the pressure drop at 5M05-02, 5M11-15, 5M16-01, 5M18-01, 5M18-02, and 5M18-03. [Reg.19.705 and 40 C.F.R. § 52 Subpart E]
- OSP 7. The permittee shall maintain a daily flowrate on scrubbers in this section in accordance with most current version of the Facility Operating Plan. [Reg.19.303 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]
- OSP 8. The permittee shall continuously monitor the liquid flow rate of the scrubber SN-5M04-02. The permittee shall submit reports as required by Plantwide Condition 33. [Rule 19.304 and 40 C.F.R. § 64]

40 CFR Part 60 Subpart NNN - Manufacturing Industry (SOCMI) Distillation Operations

- OSP 9. The permittee shall maintain a TRE index value of greater than 1.0 without the use of VOC emission control device for 5M01-02. The permittee shall document and record all calculations performed to determine the TRE index value of the vent stream per §60.664(d), (e) and (f). [Reg.19.304 and 40 C.F.R. § 60.662(c)]
- OSP 10. The permittee shall keep up-to-date, readily accessible records of:
 - a. Any changes in production capacity, feedstock type, or catalyst type, or of any replacement, removal or addition of recovery equipment or a distillation unit,
 - b. Any recalculation of the TRE index value performed pursuant to §60.664(f), and,
 - c. The results of any performance test performed pursuant to the methods and procedures required by §60.664(d). [Reg.19.304 and 40 C.F.R. § 60.665(h]
- OSP 11. The provisions of this subpart apply to affected sources as defined in paragraph (b) of this section, and is part of process or production unit that produces any of the chemicals listed in §60.667 as a product, co-product, by-product, or intermediate, except as provided in paragraph (c). [Reg.19.304 and 40 C.F.R. § 60.660]
- OSP 12. This source is operated under the exemption allowed by this citation; being, an affected facility with a TRE index value greater than 8.0. This source is exempt from all provisions of this subpart except for §60.662; §60.664(d), (e), and (f); and §60.665(h) and (l). [Reg.19.304 and 40 C.F.R. § 60.660(c)(4)]
- OSP 13. The permittee shall use any of the options listed in §60.662(a), (b), or (c) for an applicable treatment standard, providing proper notification is provided to the Department to document the change in treatment standard. The permittee shall then comply with the requirements of §60.663, §60.664, and §60.665 as applicable to the emission standard chosen. [Reg.19.304 and 40 C.F.R. § 60.662]
- OSP 14. The permittee shall comply with all recordkeeping and reporting requirements in §60.665 as applicable to the treatment standard and control devices used to meet compliance with this subpart. [Reg.19.304 and 40 C.F.R. § 60.665]
- OSP 15. The permittee is exempt from the quarterly reporting requirements contained in §60.7(c) of the General Provisions. [Reg.19.304 and 40 C.F.R. § 60.665(k)]
- OSP 16. The permittee shall submit semiannual reports of the following information: Any recalculation of the TRE index value, as recorded under §60.665(h). [Reg.19.304 and 40 C.F.R. § 60.665]

40 CFR Part 60, Subpart VV - Standards of Performance for Equipment Leaks of VOC in SOCMI

OSP 17. The equipment, including each valve, pump, pressure relief device, sampling connection system, open-ended valve or line, and flange or other connector in VOC service at the acetic acid recovery area, are affected facilities under the terms of 40 CFR Part 60 Subpart VV – Standards of Performance for Equipment Leaks of VOC in SOCMI. [Reg.19.304 and 40 C.F.R. § 60.480]

Standards: General

- OSP 18. For equipment subject to these standards, the following General Requirements apply:
 - a. Each owner or operator subject to the provisions of this subpart shall demonstrate compliance with the requirements of §§ 60.482-1 through 60.482-10 or § 60.480(e) for all equipment within 180 days of initial startup. [Reg.19.304 and 40 C.F.R. § 60.482-1(a)]
 - b. Compliance with §§ 60.482-1 to 60.482-10 will be determined by review of records and reports, review of performance test results, and inspection using the methods and procedures specified in § 60.485. [Reg.19.304 and 40 C.F.R. § 60.482-1(b)]
 - c. An owner or operator may request a determination of equivalence of a means of emission limitation to the requirements of §§ 60.482-2, 60.482-3, 60.482-5, 60.482-6, 60.482-7, 60.482-8, and 60.482-10 as provided in § 60.484. [Reg.19.304 and 40 C.F.R. § 60.482-1(c)(1)]
 - d. If the Administrator makes a determination that a means of emission limitation is at least equivalent to the requirements of §§ 60.482-2, 60.482-3, 60.482-5, 60.482-6, 60.482-7, 60.482-8, or 60.482-10, an owner or operator shall comply with the requirements of that determination. [Reg.19.304 and 40 C.F.R. § 60.482-1(c)(2)]
 - e. Equipment that is in vacuum service is excluded from the requirements of §§ 60.482-2 to 60.482-10 if it is identified as required in § 60.486(e)(5). [Reg.19.304 and 40 C.F.R. § 60.482-1(d)]
 - f. Equipment that an owner or operator designates as being in VOC service less than 300 hours (hr)/yr is excluded from the requirements of §§ 60.482-2 through 60.482-10 if it is identified as required in § 60.486(e)(6) and it meets any of the conditions specified in paragraphs (e)(1) through (3) of this section. [Reg.19.304 and 40 C.F.R. § 60.482-1(e)(i-iii)]
 - i. The equipment is in VOC service only during startup and shutdown, excluding startup and shutdown between batches of the same campaign for a batch process.
 - ii. The equipment is in VOC service only during process malfunctions or other emergencies.
 - iii. The equipment is backup equipment that is in VOC service only when the primary equipment is out of service.
 - g. If a dedicated batch process unit operates less than 365 days during a year, an owner or operator may monitor to detect leaks from pumps and valves at the frequency specified in the following table instead of monitoring as specified in §§ 60.482-2, 60.482-7, and 60.483-2:

Operating time (percent of hours during	Equivalent monitoring frequency time in use		
year)	Monthly	Quarterly	Semiannually
0 to <25	Quarterly	Annually	Annually.
25 to <50	Quarterly	Semiannually	Annually.
50 to <75	Bimonthly	Three quarters	Semiannually.
75 to 100	Monthly	Quarterly	Semiannually.

[Reg.19.304 and 40 C.F.R. § 60.482-1(f)(1)]

- h. Pumps and valves that are shared among two or more batch process units that are subject to this subpart may be monitored at the frequencies specified in paragraph (f)(1) of this section, provided the operating time of all such process units is considered. [Reg.19.304 and 40 C.F.R. § 60.482-1(f)(2)]
- i. The monitoring frequencies specified in paragraph (f)(1) of this section are not requirements for monitoring at specific intervals and can be adjusted to accommodate process operations. An owner or operator may monitor at any time during the specified monitoring period (e.g., month, quarter, year), provided the monitoring is conducted at a reasonable interval after completion of the last monitoring campaign. Reasonable intervals are defined in paragraphs (f)(3)(i) through (iv) of this section. [Reg.19.304 and 40 C.F.R. § 60.482-1(f)(3)(i-iv)]
 - i. When monitoring is conducted quarterly, monitoring events must be separated by at least 30 calendar days.
 - ii. When monitoring is conducted semiannually (*i.e.*, once every 2 quarters), monitoring events must be separated by at least 60 calendar days.
- iii. When monitoring is conducted in 3 quarters per year, monitoring events must be separated by at least 90 calendar days.
- iv. When monitoring is conducted annually, monitoring events must be separated by at least 120 calendar days.
- j. If the storage vessel is shared with multiple process units, the process unit with the greatest annual amount of stored materials (predominant use) is the process unit the storage vessel is assigned to. If the storage vessel is shared equally among process units, and one of the process units has equipment subject to subpart VVa of this part, the storage vessel is assigned to that process unit. If the storage vessel is shared equally among process units, none of which have equipment subject to subpart VVa of this part, the storage vessel is assigned to any process unit subject to this subpart. If the predominant use of the storage vessel varies from year to year, then the owner or operator must estimate the predominant use initially and

reassess every 3 years. The owner or operator must keep records of the information and supporting calculations that show how predominant use is determined. All equipment on the storage vessel must be monitored when in VOC service. [Reg.19.304 and 40 C.F.R. § 60.482-1(g)]

Standards: Pumps in light liquid service

- OSP 19. For pumps in light liquid service, the permittee shall:
 - a. Monitor monthly to detect leaks by the methods specified in § 60.485(b), except as provided in § 60.482-1(c) and (f) and paragraphs (d), (e), and (f) of this section. A pump that begins operation in light liquid service after the initial startup date for the process unit must be monitored for the first time within 30 days after the end of its startup period, except for a pump that replaces a leaking pump and except as provided in § 60.482-1(c) and (f) and paragraphs (d), (e), and (f) of this section. [Reg.19.304 and 40 C.F.R. § 60.482-2 (a)(1)]
 - b. Check by visual inspection each calendar week for indications of liquids dripping from the pump seal, except as provided in § 60.482-1(f). [Reg.19.304 and 40 C.F.R. § 60.482-2 (a)(2)]
 - c. If an instrument reading of 10,000 ppm or greater is measured, a leak is detected. [Reg.19.304 and 40 C.F.R. § 60.482-2(b)(1)]
 - d. If there are indications of liquids dripping from the pump seal, the owner or operator shall follow the procedure specified in either paragraph (b)(2)(i) or (ii) of this section. This requirement does not apply to a pump that was monitored after a previous weekly inspection if the instrument reading for that monitoring event was less than 10,000 ppm and the pump was not repaired since that monitoring event. [Reg.19.304 and 40 C.F.R. § 60.482-2(b)(2)(i and ii)]
 - i. Monitor the pump within 5 days as specified in § 60.485(b). If an instrument reading of 10,000 ppm or greater is measured, a leak is detected. The leak shall be repaired using the procedures in paragraph (c) of this section.
 - ii. Designate the visual indications of liquids dripping as a leak, and repair the leak within 15 days of detection by eliminating the visual indications of liquids dripping.
 - e. When a leak is detected, it shall be repaired as soon as practicable, but not later than 15 calendar days after it is detected, except as provided in § 60.482-9. [Reg.19.304 and 40 C.F.R. § 60.482-2(c)(1)]
 - f. A first attempt at repair shall be made no later than 5 calendar days after each leak is detected. First attempts at repair include, but are not limited to, the practices described in paragraphs (c)(2)(i) and (ii) of this section, where practicable. [Reg.19.304 and 40 C.F.R. § 60.482-2(c)(2)(i and ii)]
 - i. Tightening the packing gland nuts;

- ii. Ensuring that the seal flush is operating at design pressure and temperature.
- g. Each pump equipped with a dual mechanical seal system that includes a barrier fluid system is exempt from the requirements of paragraph (a) of this section, provided the requirements specified in paragraphs (d)(1) through (6) of this section are met. [Reg.19.304 and 40 C.F.R. § 60.482-2(d)(1)-(6)]
 - i. Each dual mechanical seal system is—
 - 1. Operated with the barrier fluid at a pressure that is at all times greater than the pump stuffing box pressure; or
 - 2. Equipped with a barrier fluid degassing reservoir that is routed to a process or fuel gas system or connected by a closed vent system to a control device that complies with the requirements of § 60.482-10; or
 - 3. Equipped with a system that purges the barrier fluid into a process stream with zero VOC emissions to the atmosphere.

[Reg.19.304 and 40 C.F.R. § 60.482-2(d)(1)(i-iii)]

- ii. The barrier fluid system is in heavy liquid service or is not in VOC service. [Reg.19.304 and 40 C.F.R. § 60.482-2(d)(2)]
- iii. Each barrier fluid system is equipped with a sensor that will detect failure of the seal system, the barrier fluid system, or both. [Reg.19.304 and 40 C.F.R. § 60.482-2(d)(3)]
- iv. Each pump is checked by visual inspection, each calendar week, for indications of liquids dripping from the pump seals. [Reg.19.304 and 40 C.F.R. § 60.482-2(d)(4)(i)]
- v. If there are indications of liquids dripping from the pump seal at the time of the weekly inspection, the owner or operator shall follow the procedure specified in either paragraph (d)(4)(ii)(A) or (B) of this section. [Reg.19.304 and 40 C.F.R. § 60.482-2(d)(4)(ii)(A&B)]
 - 4. Monitor the pump within 5 days as specified in § 60.485(b) to determine if there is a leak of VOC in the barrier fluid. If an instrument reading of 10,000 ppm or greater is measured, a leak is detected.
 - 5. Designate the visual indications of liquids dripping as a leak.
 - vi. Each sensor as described in paragraph (d)(3) of this section is checked daily or is equipped with an audible alarm. [Reg.19.304 and 40 C.F.R. § 60.482-2(d)(5)(i)]
 - vii. The owner or operator determines, based on design considerations and operating experience, a criterion that indicates failure of the seal system,

the barrier fluid system, or both. [Reg.19.304 and 40 C.F.R. § 60.482-2(d)(5)(ii)]

- viii. If the sensor indicates failure of the seal system, the barrier fluid system, or both, based on the criterion established in paragraph (d)(5)(ii) of this section, a leak is detected. [Reg.19.304 and 40 C.F.R. § 60.482-2(d)(5)(iii)]
 - ix. When a leak is detected pursuant to paragraph (d)(4)(ii)(A) of this section, it shall be repaired as specified in paragraph (c) of this section. [Reg.19.304 and 40 C.F.R. § 60.482-2(d)(6)(i)]
 - x. A leak detected pursuant to paragraph (d)(5)(iii) of this section shall be repaired within 15 days of detection by eliminating the conditions that activated the sensor. [Reg.19.304 and 40 C.F.R. § 60.482-2(d)(6)(ii)]
 - xi. A designated leak pursuant to paragraph (d)(4)(ii)(B) of this section shall be repaired within 15 days of detection by eliminating visual indications of liquids dripping. [Reg.19.304 and 40 C.F.R. § 60.482-2(d)(6)(iii)]
- h. Any pump that is designated, as described in § 60.486(e)(1) and (2), for no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, is exempt from the requirements of paragraphs (a), (c), and (d) of this section if the pump:
 - i. Has no externally actuated shaft penetrating the pump housing,
 - ii. Is demonstrated to be operating with no detectable emissions as indicated by an instrument reading of less than 500 ppm above background as measured by the methods specified in § 60.485(c), and
- iii. Is tested for compliance with paragraph (e)(2) of this section initially upon designation, annually, and at other times requested by the Administrator.

[Reg.19.304 and 40 C.F.R. § 60.482-2(e)(1-3)]

- i. If any pump is equipped with a closed vent system capable of capturing and transporting any leakage from the seal or seals to a process or to a fuel gas system or to a control device that complies with the requirements of § 60.482-10, it is exempt from paragraphs (a) through (e) of this section. [Reg.19.304 and 40 C.F.R. § 60.482-2(f)]
- j. Any pump that is designated, as described in § 60.486(f)(1), as an unsafe-tomonitor pump is exempt from the monitoring and inspection requirements of paragraphs (a) and (d)(4) through (6) of this section if:
 - i. The owner or operator of the pump demonstrates that the pump is unsafeto-monitor because monitoring personnel would be exposed to an immediate danger as a consequence of complying with paragraph (a) of this section; and

> ii. The owner or operator of the pump has a written plan that requires monitoring of the pump as frequently as practicable during safe-tomonitor times but not more frequently than the periodic monitoring schedule otherwise applicable, and repair of the equipment according to the procedures in paragraph (c) of this section if a leak is detected.

[Reg.19.304 and 40 C.F.R. § 60.482-2(g)(1&2)]

k. Any pump that is located within the boundary of an unmanned plant site is exempt from the weekly visual inspection requirement of paragraphs (a)(2) and (d)(4) of this section, and the daily requirements of paragraph (d)(5) of this section, provided that each pump is visually inspected as often as practicable and at least monthly. [Reg.19.304 and 40 C.F.R. § 60.482-2(h)]

Standards: Sampling connection systems

- OSP 20. For Sampling connection systems, the following requirements apply:
- Each sampling connection system shall be equipped with a closed-purge, closed-loop, or closed-vent system, except as provided in § 60.482-1(c) and paragraph (c) of this section. [Reg.19.304 and 40 C.F.R. § 60.482-5(a)]
- b. Each closed-purge, closed-loop, or closed-vent system as required in paragraph (a) of this section shall comply with the requirements specified in paragraphs (b)(1) through (4) of this section. [Reg.19.304 and 40 C.F.R. § 60.482-5(b)(1-4)]
 - i. Gases displaced during filling of the sample container are not required to be collected or captured.
 - ii. Containers that are part of a closed-purge system must be covered or closed when not being filled or emptied.
 - iii. Gases remaining in the tubing or piping between the closed-purge system valve(s) and sample container valve(s) after the valves are closed and the sample container is disconnected are not required to be collected or captured.
 - iv. Each closed-purge, closed-loop, or closed-vent system shall be designed and operated to meet requirements in either paragraph (b)(4)(i), (ii), (iii), or (iv) of this section.
 - 1. Return the purged process fluid directly to the process line.
 - 2. Collect and recycle the purged process fluid to a process.
 - 3. Capture and transport all the purged process fluid to a control device that complies with the requirements of § 60.482-10.
 - 4. Collect, store, and transport the purged process fluid to any of the following systems or facilities:
 - a. A waste management unit as defined in § 63.111, if the waste management unit is subject to and operated in

compliance with the provisions of 40 CFR part 63, subpart G, applicable to Group 1 wastewater streams;

- b. A treatment, storage, or disposal facility subject to regulation under 40 CFR part 262, 264, 265, or 266;
- c. A facility permitted, licensed, or registered by a state to manage municipal or industrial solid waste, if the process fluids are not hazardous waste as defined in 40 CFR part 261;
- d. A waste management unit subject to and operated in compliance with the treatment requirements of § 61.348(a), provided all waste management units that collect, store, or transport the purged process fluid to the treatment unit are subject to and operated in compliance with the management requirements of §§ 61.343 through 61.347; or
- e. A device used to burn off-specification used oil for energy recovery in accordance with 40 CFR part 279, subpart G, provided the purged process fluid is not hazardous waste as defined in 40 CFR part 261.
- c. In situ sampling systems and sampling systems without purges are exempt from the requirements of paragraphs (a) and (b) of this section. [Reg.19.304 and 40 C.F.R. § 60.482-5(c)]

Standards: Valves in gas/vapor service and in light liquid service

- OSP 21. For Valves in gas/vapor service and in light liquid service, the following apply:
 - a. Each valve shall be monitored monthly to detect leaks by the methods specified in § 60.485(b) and shall comply with paragraphs (b) through (e) of this section, except as provided in paragraphs (f), (g), and (h) of this section, § 60.482-1(c) and (f), and §§ 60.483-1 and 60.483-2. [Reg.19.304 and 40 C.F.R. § 60.482-7(a)(1)]
 - b. A valve that begins operation in gas/vapor service or light liquid service after the initial startup date for the process unit must be monitored according to paragraphs (a)(2)(i) or (ii), except for a valve that replaces a leaking valve and except as provided in paragraphs (f), (g), and (h) of this section, § 60.482-1(c), and §§ 60.483-1 and 60.483-2. [Reg.19.304 and 40 C.F.R. § 6060.482-7(a)(2)(i and ii)]
 - i. Monitor the valve as in paragraph (a)(1) of this section. The valve must be monitored for the first time within 30 days after the end of its startup period to ensure proper installation.
 - ii. If the valves on the process unit are monitored in accordance with § 60.483-1 or § 60.483-2, count the new valve as leaking when calculating the percentage of valves leaking as described in § 60.483-

2(b)(5). If less than 2.0 percent of the valves are leaking for that process unit, the valve must be monitored for the first time during the next scheduled monitoring event for existing valves in the process unit or within 90 days, whichever comes first.

- c. If an instrument reading of 10,000 ppm or greater is measured, a leak is detected. [Reg.19.304 and 40 C.F.R. § 60.482-7(b)]
- d. Any valve for which a leak is not detected for 2 successive months may be monitored the first month of every quarter, beginning with the next quarter, until a leak is detected. As an alternative to monitoring all of the valves in the first month of a quarter, an owner or operator may elect to subdivide the process unit into 2 or 3 subgroups of valves and monitor each subgroup in a different month during the quarter, provided each subgroup is monitored every 3 months. The owner or operator must keep records of the valves assigned to each subgroup. [Reg.19.304 and 40 C.F.R. § 60.482-7(c)(1)(i and ii)]
- e. If a leak is detected, the valve shall be monitored monthly until a leak is not detected for 2 successive months. [Reg.19.304 and 40 C.F.R. § 60.482-7(c)(2)]
- f. When a leak is detected, it shall be repaired as soon as practicable, but no later than 15 calendar days after the leak is detected, except as provided in § 60.482-9. [Reg.19.304 and 40 C.F.R. § 60.482-7(d)(1)]
- g. A first attempt at repair shall be made no later than 5 calendar days after each leak is detected. [Reg.19.304 and 40 C.F.R. § 60.482-7(d)(2)]
- h. First attempts at repair include, but are not limited to, the following best practices where practicable:
 - i. Tightening of bonnet bolts;
 - ii. Replacement of bonnet bolts;
- iii. Tightening of packing gland nuts;
- iv. Injection of lubricant into lubricated packing.

[Reg.19.304 and 40 C.F.R. § 60.482-7(e)(1-4)]

- i. Any valve that is designated, as described in § 60.486(e)(2), for no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, is exempt from the requirements of paragraph (a) if the valve:
 - i. Has no external actuating mechanism in contact with the process fluid,
 - ii. Is operated with emissions less than 500 ppm above background as determined by the method specified in § 60.485(c), and
- iii. Is tested for compliance with paragraph (f)(2) of this section initially upon designation, annually, and at other times requested by the Administrator.

[Reg.19.304 and 40 C.F.R. § 60.482-7(f)]

- j. Any valve that is designated, as described in § 60.486(f)(1), as an unsafe-tomonitor valve is exempt from the requirements of paragraph (a) if:
 - i. The owner or operator of the valve demonstrates that the valve is unsafe to monitor because monitoring personnel would be exposed to an immediate danger as a consequence of complying with paragraph (a), and
 - ii. The owner or operator of the valve adheres to a written plan that requires monitoring of the valve as frequently as practicable during safe-to-monitor times.

[Reg.19.304 and 40 C.F.R. § 60.482-7(g)(1 and 2)]

- k. Any valve that is designated, as described in § 60.486(f)(2), as a difficult-tomonitor valve is exempt from the requirements of paragraph (a) if:
 - i. The owner or operator of the valve demonstrates that the valve cannot be monitored without elevating the monitoring personnel more than 2 meters above a support surface.
 - ii. The process unit within which the valve is located either becomes an affected facility through § 60.14 or § 60.15 or the owner or operator designates less than 3.0 percent of the total number of valves as difficult-to-monitor, and
 - iii. The owner or operator of the valve follows a written plan that requires monitoring of the valve at least once per calendar year.

[Reg.19.304 and 40 C.F.R. § 60.482-7(g)(1-3)]

Standards: Pumps and valves in heavy liquid service, pressure relief devices in light liquid or heavy liquid service, and connectors

- OSP 22. For Pumps and valves in heavy liquid service, pressure relief devices in light liquid or heavy liquid service, and connectors, the following apply:
- a. If evidence of a potential leak is found by visual, audible, olfactory, or any other detection method at pumps and valves in heavy liquid service, pressure relief devices in light liquid or heavy liquid service, and connectors, the owner or operator shall follow either one of the following procedures:
 - i. The owner or operator shall monitor the equipment within 5 days by the method specified in § 60.485(b) and shall comply with the requirements of paragraphs (b) through (d) of this section.
 - ii. The owner or operator shall eliminate the visual, audible, olfactory, or other indication of a potential leak within 5 calendar days of detection. [Reg.19.304 and 40 C.F.R. § 60.482-8(a)(1&2)]
- b. If an instrument reading of 10,000 ppm or greater is measured, a leak is detected. [Reg.19.304 and 40 C.F.R. § 60.482-8(b)]
- c. When a leak is detected, it shall be repaired as soon as practicable, but not later than 15 calendar days after it is detected, except as provided in § 60.482-9. [Reg.19.304 and 40

C.F.R. § 60.482-8(c)(1)]

- d. The first attempt at repair shall be made no later than 5 calendar days after each leak is detected. [Reg.19.304 and 40 C.F.R. § 60.482-8(c)(2)]
- e. First attempts at repair include, but are not limited to, the best practices described under §§ 60.482-2(c)(2) and 60.482-7(e). [Reg.19.304 and 40 C.F.R. § 60.482-8(d)]

Standards: Delay of repair

OSP 23. For leaks detected, the following are requirements for delay of repair:

- a. Delay of repair of equipment for which leaks have been detected will be allowed if repair within 15 days is technically infeasible without a process unit shutdown. Repair of this equipment shall occur before the end of the next process unit shutdown. Monitoring to verify repair must occur within 15 days after startup of the process unit. [Reg.19.304 and 40 C.F.R. § 60.482-9(a)]
 - b. Delay of repair of equipment will be allowed for equipment which is isolated from the process and which does not remain in VOC service. [Reg.19.304 and 40 C.F.R. § 60.482-9(b)]
- c. Delay of repair for valves will be allowed if:
 - i. The owner or operator demonstrates that emissions of purged material resulting from immediate repair are greater than the fugitive emissions likely to result from delay of repair, and
 - ii. When repair procedures are effected, the purged material is collected and destroyed or recovered in a control device complying with § 60.482-10.

[Reg.19.304 and 40 C.F.R. § 60.482-9(c)(1 and 2)]

- d. Delay of repair for pumps will be allowed if:
 - i. Repair requires the use of a dual mechanical seal system that includes a barrier fluid system, and
 - ii. Repair is completed as soon as practicable, but not later than 6 months after the leak was detected.

[Reg.19.304 and 40 C.F.R. § 60.482-9(d)(1 and 2)]

- e. Delay of repair beyond a process unit shutdown will be allowed for a valve, if valve assembly replacement is necessary during the process unit shutdown, valve assembly supplies have been depleted, and valve assembly supplies had been sufficiently stocked before the supplies were depleted. Delay of repair beyond the next process unit shutdown will not be allowed unless the next process unit shutdown occurs sooner than 6 months after the first process unit shutdown. [Reg.19.304 and 40 C.F.R. § 60.482-9(e)]
- f. When delay of repair is allowed for a leaking pump or valve that remains in service, the pump or valve may be considered to be repaired and no longer subject to delay of repair requirements if two consecutive monthly monitoring instrument readings are below the leak definition. [Reg.19.304 and 40 C.F.R. § 60.482-9(f)]

Standards: Closed vent systems and control devices

- OSP 24. The permittee shall comply with the following standards for closed vent systems and control devices subject to this subpart:
- a. Owners or operators of closed vent systems and control devices used to comply with provisions of this subpart shall comply with the provisions of this section. [Reg.19.304 and 40 C.F.R. § 60.482-10(a)]
- b. Vapor recovery systems (for example, condensers and absorbers) shall be designed and operated to recover the VOC emissions vented to them with an efficiency of 95 percent or greater, or to an exit concentration of 20 parts per million by volume, whichever is less stringent. [Reg.19.304 and 40 C.F.R. § 60.482-10(b)]
- c. Enclosed combustion devices shall be designed and operated to reduce the VOC emissions vented to them with an efficiency of 95 percent or greater, or to an exit concentration of 20 parts per million by volume, on a dry basis, corrected to 3 percent oxygen, whichever is less stringent or to provide a minimum residence time of 0.75 seconds at a minimum temperature of 816 °C. [Reg.19.304 and 40 C.F.R. § 60.482-10(c)]
- d. Flares used to comply with this subpart shall comply with the requirements of § 60.18. [Reg.19.304 and 40 C.F.R. § 60.482-10(d)]
- e. Owners or operators of control devices used to comply with the provisions of this subpart shall monitor these control devices to ensure that they are operated and maintained in conformance with their designs. [Reg.19.304 and 40 C.F.R. § 60.482-10(e)]
- f. Except as provided in paragraphs (i) through (k) of this section, each closed vent system shall be inspected according to the procedures and schedule specified in paragraphs (f)(1) and (f)(2) of this section.
 - i. If the vapor collection system or closed vent system is constructed of hard-piping, the owner or operator shall comply with the requirements specified in paragraphs (f)(1)(i) and (f)(1)(i) of this section:
 - 6. Conduct an initial inspection according to the procedures in § 60.485(b); and
 - 7. Conduct annual visual inspections for visible, audible, or olfactory indications of leaks.
 - ii. If the vapor collection system or closed vent system is constructed of ductwork, the owner or operator shall:
 - Conduct an initial inspection according to the procedures in § 60.485(b); and
 - Conduct annual inspections according to the procedures in § 60.485(b).

[Reg.19.304 and 40 C.F.R. § 60.482-10(f)(1&2)]

g. Leaks, as indicated by an instrument reading greater than 500 parts per million by volume above background or by visual inspections, shall be repaired as soon as practicable except

as provided in paragraph (h) of this section.

- i. A first attempt at repair shall be made no later than 5 calendar days after the leak is detected.
- ii. Repair shall be completed no later than 15 calendar days after the leak is detected.

[Reg.19.304 and 40 C.F.R. § 60.482-10(g)(1&2)]

- h. Delay of repair of a closed vent system for which leaks have been detected is allowed if the repair is technically infeasible without a process unit shutdown or if the owner or operator determines that emissions resulting from immediate repair would be greater than the fugitive emissions likely to result from delay of repair. Repair of such equipment shall be complete by the end of the next process unit shutdown. [Reg.19.304 and 40 C.F.R. § 60.482-10(h)]
- If a vapor collection system or closed vent system is operated under a vacuum, it is exempt from the inspection requirements of paragraphs (f)(1)(i) and (f)(2) of this section. [Reg.19.304 and 40 C.F.R. § 60.482-10(i)]
- j. Any parts of the closed vent system that are designated, as described in paragraph (l)(1) of this section, as unsafe to inspect are exempt from the inspection requirements of paragraphs (f)(1)(i) and (f)(2) of this section if they comply with the requirements specified in paragraphs (j)(1) and (j)(2) of this section:
 - i. The owner or operator determines that the equipment is unsafe to inspect because inspecting personnel would be exposed to an imminent or potential danger as a consequence of complying with paragraphs (f)(1)(i)or (f)(2) of this section; and
 - ii. The owner or operator has a written plan that requires inspection of the equipment as frequently as practicable during safe-to-inspect times.

[Reg.19.304 and 40 C.F.R. § 60.482-10(j)(1&2)]

- k. Any parts of the closed vent system that are designated, as described in paragraph (l)(2) of this section, as difficult to inspect are exempt from the inspection requirements of paragraphs (f)(1)(i) and (f)(2) of this section if they comply with the requirements specified in paragraphs (k)(1) through (k)(3) of this section:
 - i. The owner or operator determines that the equipment cannot be inspected without elevating the inspecting personnel more than 2 meters above a support surface; and
 - ii. The process unit within which the closed vent system is located becomes an affected facility through §§ 60.14 or 60.15, or the owner or operator designates less than 3.0 percent of the total number of closed vent system equipment as difficult to inspect; and

iii. The owner or operator has a written plan that requires inspection of the equipment at least once every 5 years. A closed vent system is exempt from inspection if it is operated under a vacuum.

[Reg.19.304 and 40 C.F.R. § 60.482-10(k)(1-3)]

- 1. The owner or operator shall record the information specified in paragraphs (l)(1) through (l)(5) of this section.
 - i. Identification of all parts of the closed vent system that are designated as unsafe to inspect, an explanation of why the equipment is unsafe to inspect, and the plan for inspecting the equipment.
 - ii. Identification of all parts of the closed vent system that are designated as difficult to inspect, an explanation of why the equipment is difficult to inspect, and the plan for inspecting the equipment.
 - iii. For each inspection during which a leak is detected, a record of the information specified in § 60.486(c).
 - iv. For each inspection conducted in accordance with § 60.485(b) during which no leaks are detected, a record that the inspection was performed, the date of the inspection, and a statement that no leaks were detected.
 - v. For each visual inspection conducted in accordance with paragraph (f)(1)(ii) of this section during which no leaks are detected, a record that the inspection was performed, the date of the inspection, and a statement that no leaks were detected.

[Reg.19.304 and 40 C.F.R. § 60.482-10(1)(1-5)]

m. Closed vent systems and control devices used to comply with provisions of this subpart shall be operated at all times when emissions may be vented to them. [Reg.19.304 and 40 C.F.R. § 60.482-10(m)]

Alternative standards for valves—allowable percentage of valves leaking

- OSP 25. The permittee may comply with the following alternative standards for leaking valves:
- a. An owner or operator may elect to comply with an allowable percentage of valves leaking of equal to or less than 2.0 percent. [Reg.19.304 and 40 C.F.R. § 60.483-1(a)]
- b. The following requirements shall be met if an owner or operator wishes to comply with an allowable percentage of valves leaking:
 - i. An owner or operator must notify the Administrator that the owner or operator has elected to comply with the allowable percentage of valves leaking before implementing this alternative standard, as specified in § 60.487(d).
 - ii. A performance test as specified in paragraph (c) of this section shall be conducted initially upon designation, annually, and at other times requested by the Administrator.

iii. If a valve leak is detected, it shall be repaired in accordance with § 60.482-7(d) and (e).

[Reg.19.304 and 40 C.F.R. § 60.483-1(b)(1-3)]

- c. Performance tests shall be conducted in the following manner:
 - i. All valves in gas/vapor and light liquid service within the affected facility shall be monitored within 1 week by the methods specified in § 60.485(b).
 - ii. If an instrument reading of 10,000 ppm or greater is measured, a leak is detected.
 - iii. The leak percentage shall be determined by dividing the number of valves for which leaks are detected by the number of valves in gas/vapor and light liquid service within the affected facility.

[Reg.19.304 and 40 C.F.R. § 60.483-1(c)(1-3)]

d. Owners and operators who elect to comply with this alternative standard shall not have an affected facility with a leak percentage greater than 2.0 percent, determined as described in § 60.485(h).

[Reg.19.304 and 40 C.F.R. § 60.483-1(d)]

Alternative standards for valves—skip period leak detection and repair

- OSP 26. The following requirements apply as alternative standards for valves—skip period leak detection and repair:
- a. An owner or operator may elect to comply with one of the alternative work practices specified in paragraphs (b)(2) and (3) of this section. [Reg.19.304 and 40 C.F.R. § 60.483-2(a)(1)]
- b. An owner or operator must notify the Administrator before implementing one of the alternative work practices, as specified in § 60.487(d). [Reg.19.304 and 40 C.F.R. § 60.483-2(a)(2)]
- c. An owner or operator shall:
 - i. Comply initially with the requirements for valves in gas/vapor service and valves in light liquid service, as described in § 60.482-7.
 - ii. After 2 consecutive quarterly leak detection periods with the percent of valves leaking equal to or less than 2.0, an owner or operator may begin to skip 1 of the quarterly leak detection periods for the valves in gas/vapor and light liquid service.
 - After 5 consecutive quarterly leak detection periods with the percent of valves leaking equal to or less than 2.0, an owner or operator may begin to skip 3 of the quarterly leak detection periods for the valves in gas/vapor and light liquid service.

- iv. If the percent of valves leaking is greater than 2.0, the owner or operator shall comply with the requirements as described in § 60.482-7 but can again elect to use this section.
- v. The percent of valves leaking shall be determined as described in § 60.485(h).
- vi. An owner or operator must keep a record of the percent of valves found leaking during each leak detection period.
- vii. A value that begins operation in gas/vapor service or light liquid service after the initial startup date for a process unit following one of the alternative standards in this section must be monitored in accordance with 60.482-7(a)(2)(i) or (ii) before the provisions of this section can be applied to that value.

[Reg.19.304 and 40 C.F.R. § 60.483-2(b)(1-7)]

Equivalence of means of emission limitation

OSP 27. Each owner or operator subject to the provisions of this subpart may apply to the Administrator for determination of equivalence for any means of emission limitation that achieves a reduction in emissions of VOC at least equivalent to the reduction in emissions of VOC achieved by the controls required in this subpart. The requirements for such a request and determination of equivalence are detailed within (b) through (f) of this section. [Reg.19.304 and 40 C.F.R. § 60.484(a-f)]

Test methods and procedures

- OSP 28. The permittee shall conduct performance testing in accordance with the provisions of this section.
- a. In conducting the performance tests required in § 60.8, the owner or operator shall use as reference methods and procedures the test methods in appendix A of this part or other methods and procedures as specified in this section, except as provided in § 60.8(b). [Reg.19.304 and 40 C.F.R. § 60.485(a)]
- b. The owner or operator shall determine compliance with the standards in §§ 60.482-1 through 60.482-10, 60.483, and 60.484 as follows:
 - i. Method 21 shall be used to determine the presence of leaking sources. The instrument shall be calibrated before use each day of its use by the procedures specified in Method 21. The following calibration gases shall be used:
 - 1. Zero air (less than 10 ppm of hydrocarbon in air); and
 - 2. A mixture of methane or n-hexane and air at a concentration of about, but less than, 10,000 ppm methane or n-hexane.

[Reg.19.304 and 40 C.F.R. § 60.485(b) (1)(i&ii)]

c. The owner or operator shall determine compliance with the no detectable emission standards in §§ 60.482-2(e), 60.482-3(i), 60.482-4, 60.482-7(f), and 60.482-10(e) as

follows:

- i. The requirements of paragraph (b) shall apply.
- Method 21 shall be used to determine the background level. All potential leak interfaces shall be traversed as close to the interface as possible. The arithmetic difference between the maximum concentration indicated by the instrument and the background level is compared with 500 ppm for determining compliance.

[Reg.19.304 and 40 C.F.R. § 60.485(c)]

- d. The owner or operator shall test each piece of equipment unless he demonstrates that a process unit is not in VOC service, i.e., that the VOC content would never be reasonably expected to exceed 10 percent by weight. For purposes of this demonstration, the following methods and procedures shall be used:
 - i. Procedures that conform to the general methods in ASTM E260-73, 91, or 96, E168-67, 77, or 92, E169-63, 77, or 93 (incorporated by reference—see § 60.17) shall be used to determine the percent VOC content in the process fluid that is contained in or contacts a piece of equipment.
 - ii. Organic compounds that are considered by the Administrator to have negligible photochemical reactivity may be excluded from the total quantity of organic compounds in determining the VOC content of the process fluid.
 - iii. Engineering judgment may be used to estimate the VOC content, if a piece of equipment had not been shown previously to be in service. If the Administrator disagrees with the judgment, paragraphs (d) (1) and (2) of this section shall be used to resolve the disagreement.

[Reg.19.304 and 40 C.F.R. § 60.485(d)]

- e. The owner or operator shall demonstrate that a piece of equipment is in light liquid service by showing that all the following conditions apply:
 - i. The vapor pressure of one or more of the organic components is greater than 0.3 kPa at 20 °C (1.2 in. H₂ O at 68 °F). Standard reference texts or ASTM D2879-83, 96, or 97 (incorporated by reference—see § 60.17) shall be used to determine the vapor pressures.
 - ii. The total concentration of the pure organic components having a vapor pressure greater than 0.3 kPa at 20 °C (1.2 in. H₂ O at 68 °F) is equal to or greater than 20 percent by weight.
 - iii. The fluid is a liquid at operating conditions.

[Reg.19.304 and 40 C.F.R. § 60.485(e)]

f. Samples used in conjunction with paragraphs (d), (e), and (g) of this section shall be representative of the process fluid that is contained in or contacts the equipment or the gas

being combusted in the flare. [Reg.19.304 and 40 C.F.R. § 60.485(f)]

- g. The owner or operator shall determine compliance with the standards of flares as follows:
 - i. Method 22 shall be used to determine visible emissions.
 - ii. A thermocouple or any other equivalent device shall be used to monitor the presence of a pilot flame in the flare.
 - iii. The maximum permitted velocity for air assisted flares shall be computed using the following equation:

 $V_{\max} = K_1 + K_2 H_T$

Where:

V_{max} = Maximum permitted velocity, m/sec (ft/sec)

 H_T = Net heating value of the gas being combusted, MJ/scm (Btu/scf).

 $K_1 = 8.706$ m/sec (metric units)

= 28.56 ft/sec (English units)

 $K_2 = 0.7084 \text{ m}^4 / (\text{MJ-sec}) \text{ (metric units)}$

= 0.087 ft⁴ /(Btu-sec) (English units)

iv. (4) The net heating value (H_T) of the gas being combusted in a flare shall be computed using the following equation:

$$H_{I} = K \sum_{i=1}^{n} C_{i} H_{i}$$

Where:

K = Conversion constant, 1.740×10^{-7} (g-mole)(MJ)/(ppm-scm-kcal) (metric units) = 4.674×10^{-6} [(g-mole)(Btu)/(ppm-scf-kcal)] (English units)

C_i = Concentration of sample component "i," ppm

 H_i = Net heat of combustion of sample component "i" at 25 °C and 760 mm Hg (77 °F and 14.7 psi), kcal/g-mole

- v. Method 18 or ASTM D6420-99 (2004) (where the target compound(s) are those listed in Section 1.1 of ASTM D6420-99, and the target concentration is between 150 parts per billion by volume and 100 parts per million by volume) and ASTM D2504-67, 77 or 88 (Reapproved 1993) (incorporated by reference—see § 60.17) shall be used to determine the concentration of sample component "i."
- vi. ASTM D2382-76 or 88 or D4809-95 (incorporated by reference—see § 60.17) shall be used to determine the net heat of combustion of component "i" if published values are not available or cannot be calculated.

vii. Method 2, 2A, 2C, or 2D, as appropriate, shall be used to determine the actual exit velocity of a flare. If needed, the unobstructed (free) cross-sectional area of the flare tip shall be used.

[Reg.19.304 and 40 C.F.R. § 60.485(g)(1-7)]

- h. The owner or operator shall determine compliance with § 60.483-1 or § 60.483-2 as follows:
 - i. The percent of valves leaking shall be determined using the following equation:

 $%V_{L} = (V_{L} / V_{T}) * 100$

Where:

 $%V_L$ = Percent leaking values

 V_L = Number of valves found leaking

 V_T = The sum of the total number of valves monitored

- ii. The total number of valves monitored shall include difficult-to-monitor and unsafe-to-monitor valves only during the monitoring period in which those valves are monitored.
- iii. The number of valves leaking shall include valves for which repair has been delayed.
- iv. Any new valve that is not monitored within 30 days of being placed in service shall be included in the number of valves leaking and the total number of valves monitored for the monitoring period in which the valve is placed in service.
- v. If the process unit has been subdivided in accordance with § 60.482-7(c)(1)(ii), the sum of valves found leaking during a monitoring period includes all subgroups.
- vi. The total number of valves monitored does not include a valve monitored to verify repair.

[Reg.19.304 and 40 C.F.R. § 60.485(h)(1-6)]

Recordkeeping requirements

- OSP 29. The permittee shall show compliance with the recordkeeping requirements of this subpart. [Reg.19.304 and 40 C.F.R. § 60.486]
- a. An owner or operator of more than one affected facility subject to the provisions of this subpart may comply with the recordkeeping requirements for these facilities in one recordkeeping system if the system identifies each record by each facility. [Reg.19.304 and 40 C.F.R. § 60.486(a)(2)]
- b. When each leak is detected as specified in §§ 60.482-2, 60.482-3, 60.482-7, 60.482-8, and 60.483-2, the following requirements apply:

- i. A weatherproof and readily visible identification, marked with the equipment identification number, shall be attached to the leaking equipment.
- ii. The identification on a valve may be removed after it has been monitored for 2 successive months as specified in § 60.482-7(c) and no leak has been detected during those 2 months.
- iii. The identification on equipment except on a valve, may be removed after it has been repaired.

[Reg.19.304 and 40 C.F.R. § 60.486(b)(1-3)]

- c. When each leak is detected as specified in §§ 60.482-2, 60.482-3, 60.482-7, 60.482-8, and 60.483-2, the following information shall be recorded in a log and shall be kept for 2 years in a readily accessible location:
 - i. The instrument and operator identification numbers and the equipment identification number.
 - ii. The date the leak was detected and the dates of each attempt to repair the leak.
 - iii. Repair methods applied in each attempt to repair the leak.
 - iv. "Above 10,000" if the maximum instrument reading measured by the methods specified in § 60.485(a) after each repair attempt is equal to or greater than 10,000 ppm.
 - v. "Repair delayed" and the reason for the delay if a leak is not repaired within 15 calendar days after discovery of the leak.
 - vi. The signature of the owner or operator (or designate) whose decision it was that repair could not be effected without a process shutdown.
 - vii. The expected date of successful repair of the leak if a leak is not repaired within 15 days.
 - viii. Dates of process unit shutdowns that occur while the equipment is unrepaired.
 - ix. The date of successful repair of the leak.

[Reg.19.304 and 40 C.F.R. § 60.486(c)(1-9)]

- d. The following information pertaining to the design requirements for closed vent systems and control devices described in § 60.482-10 shall be recorded and kept in a readily accessible location:
 - i. Detailed schematics, design specifications, and piping and instrumentation diagrams.
 - ii. The dates and descriptions of any changes in the design specifications.

- A description of the parameter or parameters monitored, as required in § 60.482-10(e), to ensure that control devices are operated and maintained in conformance with their design and an explanation of why that parameter (or parameters) was selected for the monitoring.
- iv. Periods when the closed vent systems and control devices required in §§ 60.482-2, 60.482-3, 60.482-4, and 60.482-5 are not operated as designed, including periods when a flare pilot light does not have a flame.
- v. Dates of startups and shutdowns of the closed vent systems and control devices required in §§ 60.482-2, 60.482-3, 60.482-4, and 60.482-5.

[Reg.19.304 and 40 C.F.R. § 60.486(d)(1-5)]

- e. The following information pertaining to all equipment subject to the requirements in §§ 60.482-1 to 60.482-10 shall be recorded in a log that is kept in a readily accessible location:
 - i. A list of identification numbers for equipment subject to the requirements of this subpart.
 - ii. A list of identification numbers for equipment that are designated for no detectable emissions under the provisions of §§ 60.482-2(e), 60.482-3(i) and 60.482-7(f).
 - iii. The designation of equipment as subject to the requirements of § 60.482-2(e), § 60.482-3(i), or § 60.482-7(f) shall be signed by the owner or operator. Alternatively, the owner or operator may establish a mechanism with their permitting authority that satisfies this requirement.
 - iv. A list of equipment identification numbers for pressure relief devices required to comply with § 60.482-4.
 - v. The dates of each compliance test as required in §§ 60.482-2(e), 60.482-3(i), 60.482-4, and 60.482-7(f).
 - 1. The background level measured during each compliance test.
 - 2. The maximum instrument reading measured at the equipment during each compliance test.
 - vi. A list of identification numbers for equipment in vacuum service.
 - vii. A list of identification numbers for equipment that the owner or operator designates as operating in VOC service less than 300 hr/yr in accordance with § 60.482-1(e), a description of the conditions under which the equipment is in VOC service, and rationale supporting the designation that it is in VOC service less than 300 hr/yr.

[Reg.19.304 and 40 C.F.R. § 60.486(d)(1-7)]

f. The following information pertaining to all valves subject to the requirements of § 60.482-7(g) and (h) and to all pumps subject to the requirements of § 60.482-2(g) shall be recorded

in a log that is kept in a readily accessible location:

- i. A list of identification numbers for valves and pumps that are designated as unsafe-to-monitor, an explanation for each valve or pump stating why the valve or pump is unsafe-to-monitor, and the plan for monitoring each valve or pump.
- ii. A list of identification numbers for valves that are designated as difficult-to-monitor, an explanation for each valve stating why the valve is difficult-to-monitor, and the schedule for monitoring each valve.

[Reg.19.304 and 40 C.F.R. § 60.486(f)(1&2)]

- g. The following information shall be recorded for valves complying with § 60.483-2:
 - i. A schedule of monitoring.
 - ii. The percent of valves found leaking during each monitoring period.

[Reg.19.304 and 40 C.F.R. § 60.486(g)(1&2)]

- h. The following information shall be recorded in a log that is kept in a readily accessible location:
 - i. Design criterion required in §§ 60.482-2(d)(5) and 60.482-3(e)(2) and explanation of the design criterion; and
 - ii. Any changes to this criterion and the reasons for the changes.

[Reg.19.304 and 40 C.F.R. § 60.486(h)(1&2)]

- i. The following information shall be recorded in a log that is kept in a readily accessible location for use in determining exemptions as provided in § 60.480(d):
 - i. An analysis demonstrating the design capacity of the affected facility,
 - ii. A statement listing the feed or raw materials and products from the affected facilities and an analysis demonstrating whether these chemicals are heavy liquids or beverage alcohol, and
 - iii. An analysis demonstrating that equipment is not in VOC service.

[Reg.19.304 and 40 C.F.R. § 60.486(i)(1-3)]

- j. Information and data used to demonstrate that a piece of equipment is not in VOC service shall be recorded in a log that is kept in a readily accessible location. [[Reg.19.304 and 40 C.F.R. § 60.486(j)]
- k. The provisions of § 60.7 (b) and (d) do not apply to affected facilities subject to this subpart. [Reg.19.304 and 40 C.F.R. § 60.486(k)]

Reporting requirements

- OSP 30. The permittee shall comply with the reporting requirements of this section. [Reg.19.304 and 40 C.F.R. § 60.487]
- a. Each owner or operator subject to the provisions of this subpart shall submit semiannual

reports to the Administrator beginning six months after the initial startup date. [Reg.19.304 and 40 C.F.R. § 60.487(a)]

- b. The initial semiannual report to the Administrator shall include the following information:
 - i. Process unit identification.
 - ii. Number of valves subject to the requirements of 60.482-7, excluding those valves designated for no detectable emissions under the provisions of § 60.482-7(f).
 - iii. Number of pumps subject to the requirements of § 60.482-2, excluding those pumps designated for no detectable emissions under the provisions of § 60.482-2(e) and those pumps complying with § 60.482-2(f).
 - Number of compressors subject to the requirements of § 60.482-3, excluding those compressors designated for no detectable emissions under the provisions of § 60.482-3(i) and those compressors complying with § 60.482-3(h).

[Reg.19.304 and 40 C.F.R. § 60.487(b)(1-4)]

- c. All semiannual reports to the Administrator shall include the following information, summarized from the information in § 60.486:
 - i. Process unit identification.
 - ii. For each month during the semiannual reporting period,
 - 1. Number of valves for which leaks were detected as described in § 60.482-7(b) or § 60.483-2,
 - 2. Number of valves for which leaks were not repaired as required in § 60.482-7(d)(1),
 - 3. Number of pumps for which leaks were detected as described in § 60.482-2(b), (d)(4)(ii)(A) or (B), or (d)(5)(iii),
 - 4. Number of pumps for which leaks were not repaired as required in § 60.482-2(c)(1) and (d)(6),
 - 5. Number of compressors for which leaks were detected as described in § 60.482-3(f),
 - 6. Number of compressors for which leaks were not repaired as required in § 60.482-3(g)(1), and
 - 7. The facts that explain each delay of repair and, where appropriate, why a process unit shutdown was technically infeasible.
 - iii. Dates of process unit shutdowns which occurred within the semiannual reporting period.

iv. Revisions to items reported according to paragraph (b) if changes have occurred since the initial report or subsequent revisions to the initial report.

[Reg.19.304 and 40 C.F.R. § 60.487(c)(1-4)]

- d. An owner or operator electing to comply with the provisions of §§ 60.483-1 or 60.483-2 shall notify the Administrator of the alternative standard selected 90 days before implementing either of the provisions. [Reg.19.304 and 40 C.F.R. § 60.487(d)]
- e. An owner or operator shall report the results of all performance tests in accordance with § 60.8 of the General Provisions. The provisions of § 60.8(d) do not apply to affected facilities subject to the provisions of this subpart except that an owner or operator must notify the Administrator of the schedule for the initial performance tests at least 30 days before the initial performance tests. [Reg.19.304 and 40 C.F.R. § 60.487(e)]
- f. The requirements of paragraphs (a) through (c) of this section remain in force until and unless EPA, in delegating enforcement authority to a State under section 111(c) of the Act, approves reporting requirements or an alternative means of compliance surveillance adopted by such State. In that event, affected sources within the State will be relieved of the obligation to comply with the requirements of paragraphs (a) through (c) of this section, provided that they comply with the requirements established by the State. [Reg.19.304 and 40 C.F.R. § 60.487(f)]
Chemical Waste Destructor 6M03-05 and DEST-FUG

Source Description

The chemical waste destructor at FutureFuel Chemical Company is designed to burn a mixture of waste streams resulting from various fine chemical manufacturing facilities at the plant. Some of the waste is mainly organic solvents, but the majority is comprised of aqueous solutions containing some organic and salt compounds. The equipment used to burn the waste includes a burner assembly, oxidizer chamber, weir tank, quench separator tank, high-energy scrubber, vane separator, and a stack. The chemical destructor is a vertically downfired unit. Emissions were calculated for the incinerator (6M03-05) and for fugitive equipment leaks (DEST-FUG).

The chemical waste destructor is subject to 40 CFR Part 63, Subpart EEE, *National Emission Standards for Hazardous Air Pollutants from Hazardous Waste Combustors*.

Specific Conditions

CWD 1. The permittee shall not exceed the emission rates set forth in the following table. Compliance with the CO lb/hr rate will be shown by CEMs data on a 3-hour average excluding periods of start-up and shut down. CO tpy numbers will be shown by the CEMs data including all operations. [Reg.19.501 *et seq.* and 40 C.F.R. § 52 Subpart E]

SN	Description	Pollutant	lb/hr	tpy
6M03-05	Chemical Waste Destructor (50 MMBtu/hr)	PM ₁₀ SO ₂ VOC CO NO _x Lead	1.5 20.4 1.0 11.2 33.2 0.1	6.6 89.0 4.0 27.0 146.0 0.1
DEST-FUG	Destructor Fugitives	VOC	1.0	3.0

CWD 2. The permittee shall not exceed the emission rates set forth in the following table. Emission rates are based upon the maximum capacity of equipment. The facility shall show compliance with the facility total ton/yr limits using the procedures outlined in Plantwide Conditions 12 through 16. Hourly Plantwide Hazardous Air Pollutant emissions are limited by Plantwide Conditions 16. Additional HAP limitations are included in Plantwide Condition 13. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

SN	Description	Pollutant	lb/hr	tpy
	Chemical Waste	PM	1.5	6.6
6M03-05	Destructor (50	Inorganics*	1.4	6.1
	MMBtu/hr)	Organic Pollutants**	1.0	4.0

				ιpy
DEST-FUG	Destructor Fugitives	Organic Pollutants**	1.0	3.0
*Inorgan **Organ	ics are considered to be no ic Pollutants are considered	n-organic Hazardous Air Polluta d to be VOC, all organic HAPs, a	ints and organic Air C	Contaminates.
The per destruc reading Part 52	rmittee shall not excee tor, except during peri s will be conducted or , Subpart E]	d 20% opacity as measured ods of startup, shutdown, a 1 a weekly basis. [§19.503	l by Method 9 nd malfunction of Regulation	at the chemical n. Opacity 19 and 40 CFR
The per months efficien Based o higher.	mittee shall measure t and using Method 25. by measuring the in on maximum rates, the [Reg.19.705 and 40 0	he VOC emissions at the cl A. The permittee shall also nlet and outlet concentratio destruction efficiency duri C.F.R. § 52 Subpart E]	hemical destru determine the ns of VOC du ng testing sha	ctor every 61 e destruction ring this test. Il be 99.99% or
The permittee shall measure the particulate emissions from the chemical destructor using Method 5, with the PM_{10} portion using Methods 201 or 201A. The permittee shall measure the NO _x emissions using Method 7E. The permittee shall measure the SO ₂ emissions using Method 6C. Testing shall commence no later than 61 months from the date of the previous test. [Reg 19,705 and 40 C.F.R. § 52 Subpart E]				
The per with me Ann. §	rmittee shall maintain ost current version of t 8-4-203 as referenced	a daily flowrate on scrubbe he Facility Operating Plan. by Ark. Code Ann. §§ 8-4-	rs in this section [Reg.19.303 a -304 and 8-4-3	on in accordance and Ark. Code 11]
The permittee shall minimize the time SN-6M03-05 is spent in start-up and shut down. [Reg.19.705, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 52 Subpart E]				
The per condition Continu permit. reference	rmittee shall operate all ons of Section III – <u>No</u> lous Emission Monito [Reg.19.703, 40 C.F. ced by Ark. Code Ann	Il CEMS at this source in ac <u>otification and Recordkeepi</u> oring Systems Conditions as R. § 52 Subpart E, and Ark a. §§ 8-4-304 and 8-4-311]	ccordance with <u>ng</u> of the Depa s found in App c. Code Ann. §	a all applicable artment's bendix A of this 8-4-203 as
	DEST-FUG *Inorgan *Norgan The per destruc reading Part 52 The per months efficient Based of higher. The per Method measur emission date of The per with me Ann. § The per [Reg.19 8-4-311 The per condition Continue permit. reference	DEST-FUGDestructor Fugitives*Inorganics are considered to be not **Organic Pollutants are considered The permittee shall not exceet destructor, except during peri readings will be conducted or Part 52, Subpart E]The permittee shall measure t months and using Method 25. efficiency by measuring the in Based on maximum rates, the higher. [Reg.19.705 and 40 C The permittee shall measure the Method 5, with the PM10 port measure the NOx emissions u emissions using Method 6C. date of the previous test. [Reg The permittee shall maintain with most current version of t Ann. § 8-4-203 as referenced The permittee shall minimize [Reg.19.705, Ark. Code Ann. 8-4-311, and 40 C.F.R. § 52 S The permittee shall operate al conditions of Section III – No Continuous Emission Monito permit. [Reg.19.703, 40 C.F. referenced by Ark. Code Ann	DEST-FUGDestructor FugitivesOrganic Pollutants***Inorganics are considered to be non-organic Hazardous Air Polluta **Organic Pollutants are considered to be VOC, all organic HAPs, a The permittee shall not exceed 20% opacity as measured destructor, except during periods of startup, shutdown, a readings will be conducted on a weekly basis. [§19.503 Part 52, Subpart E]The permittee shall measure the VOC emissions at the cl months and using Method 25A. The permittee shall also efficiency by measuring the inlet and outlet concentratio Based on maximum rates, the destruction efficiency duri higher. [Reg.19.705 and 40 C.F.R. § 52 Subpart E]The permittee shall measure the particulate emissions from Method 5, with the PM10 portion using Methods 201 or 27 measure the NOx emissions using Method 7E. The permitee shall commence not date of the previous test. [Reg.19.705 and 40 C.F.R. § 52 Subpart B]The permittee shall maintain a daily flowrate on scrubbe with most current version of the Facility Operating Plan. Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-311, and 40 C.F.R. § 52 Subpart E]The permittee shall operate all CEMS at this source in ac conditions of Section III – Notification and Recordkeepi Continuous Emission Monitoring Systems Conditions at permit. [Reg.19.703, 40 C.F.R. § 52 Subpart E, and Ark referenced by Ark. Code Ann. §§ 8-4-311]	DEST-FUGDestructor FugitivesOrganic Pollutants**1.0*Inorganics are considered to be non-organic Hazardous Air Pollutants**Organic Pollutants are considered to be VOC, all organic HAPs, and organic Air OThe permittee shall not exceed 20% opacity as measured by Method 9destructor, except during periods of startup, shutdown, and malfunctionreadings will be conducted on a weekly basis. [§19.503 of RegulationPart 52, Subpart E]The permittee shall measure the VOC emissions at the chemical destrumonths and using Method 25A. The permittee shall also determine the efficiency by measuring the inlet and outlet concentrations of VOC duBased on maximum rates, the destruction efficiency during testing shal higher. [Reg.19.705 and 40 C.F.R. § 52 Subpart E]The permittee shall measure the particulate emissions from the chemicMethod 5, with the PM ₁₀ portion using Method 7E. The permittee shall measureemissions using Method 6C. Testing shall commence no later than 61 date of the previous test. [Reg.19.705 and 40 C.F.R. § 52 Subpart E]The permittee shall maintain a daily flowrate on scrubbers in this sectio with most current version of the Facility Operating Plan. [Reg.19.303 a Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-33The permittee shall minimize the time SN-6M03-05 is spent in start-up [Reg.19.705, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code A 8-4-311, and 40 C.F.R. § 52 Subpart E]The permittee shall operate all CEMS at this source in accordance with conditions of Section III – <u>Notification and Recordkeeping</u> of the Depa Continuous Emission Monitoring Systems Conditions as found in App permit. [Reg.19.703, 40 C.F.R. § 52 Subpart E, and Ark. Co

40 CFR Part 63 Subpart EEE - National Emission Standards for Hazardous Air Pollutants from Hazardous Waste Combustors

CWD 9. The permittee shall maintain the operating limits as outlined in the Notification of Compliance (NOC) for the chemical destructor. The NOC is required by 40 CFR Part 63, Subpart EEE. [Reg.19.705 and 40 C.F.R. § 52 Subpart E]

Operating Parameters Limitations (OPLs)

Maximum Waste Feed Limitations

Operating Parameters Limitations (OPLs)
Waste Feed Rate (aqueous HAP + organic HAP), lb/hr
Chlorine Feed Rate, lb/hr
Ash Feed Rate, lb/hr
Total Mercury Feed Rate, lb/hr
Total Semi-Volatile Metals Feed Rate, lb/hr
Total Low-Volatile Metals Feed Rate, lb/hr
Combustion Chamber Limitations
Minimum Combustion Chamber Temperature, °F
Minimum Atomization Pressure, psi
Maximum Combustion Air Flow Rate, scfm
High Energy Scrubber Limitations
Minimum Scrubber Differential Pressure, inches water column
Minimum Scrubber Water pH
Minimum Scrubber Water Flow Rate (1st and 2nd Stages), gpm
Minimum Scrubber Water Tank Level, %
Minimum Scrubber Blowdown Rate, gpm
Stack Gas Limitations
Maximum Carbon Monoxide Concentration, ppmv @ 7% O2

Maximum Total Hydrocarbon, ppmv

- CWD 10. The permittee shall maintain records of the chemical destructor operating limits as specified in the NOC. These records shall be maintained on site and available for inspection upon request. [Reg.19.705 and 40 C.F.R. § 52 Subpart E]
- CWD 11. This facility is subject to 40 CFR Part 63, Subpart EEE, National Emission Standards for Hazardous Air Pollutants from Hazardous Waste Combustors. Applicable requirements include, but are not limited to, the following conditions. [Reg.19.304 and 40 C.F.R. § 63.1200 of EEE]:

Emission Limits

a. The permittee shall not discharge or cause combustion gases to be emitted to the atmosphere that contain dioxin and furans in excess of 0.4 ng TEQ/dscm, corrected to 7 percent oxygen. [Reg.19.304 and 40 C.F.R. § 63.1219(a)(1)(A)]

- b. Emissions in excess of 0.40 ng TEQ/dscm corrected to 7 percent oxygen provided that the combustion gas temperature at the inlet to the initial particulate matter control device is 400 °F or lower based on the average of the test run average temperatures. For purposes of compliance, operation of a wet particulate control device is presumed to meet the 400 °F or lower requirement. [Reg.19.304 and 40 C.F.R. § 63.1219(a)(1)(B)]
- c. The permittee shall not discharge or cause combustion gases to be emitted to the atmosphere that contain mercury in excess of 130 μ g/dscm, corrected to 7 percent oxygen. [Reg.19.304 and 40 C.F.R. § 63.1219(a)(2)]
- d. The permittee shall not discharge or cause combustion gases to be emitted to the atmosphere that contain lead and cadmium in excess of 230 μg/dscm, combined emissions, corrected to 7 percent oxygen. [Reg.19.304 and 40 C.F.R. § 63.1219(a)(3)]
- e. The permittee shall not discharge or cause combustion gases to be emitted to the atmosphere that contain arsenic, beryllium, and chromium in excess of 92µg/dscm, combined emissions, corrected to 7 percent oxygen. [Reg.19.304 and 40 C.F.R. § 63.1219(a)(4)]
- f. For carbon monoxide and hydrocarbons, either:
 - i. Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (a)(5)(ii) of this section, you must also document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by § 63.1206(b)(7), hydrocarbons do not exceed 10 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or
 - ii. Hydrocarbons in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; [Reg.19.304 and 40 C.F.R. § 63.1219(a)(5)(i and ii)]
- g. Hydrogen chloride and chlorine gas (total chlorine) in excess of 32 parts per million by volume, combined emissions, expressed as a chloride (Cl⁽⁻⁾) equivalent, dry basis and corrected to 7 percent oxygen. [Reg.19.304 and 40 C.F.R. § 63.1219(a)(6)]
- h. Except as provided by paragraph (e) of this section, particulate matter emissions in excess of 0.013 gr/dscf corrected to 7 percent oxygen. [Reg.19.304 and 40 C.F.R. § 63.1219(a)(7)]
- i. In lieu of complying with the particulate matter standards of this section, you may elect to comply with the following alternative metal emission control requirement.

- i. You must not discharge or cause combustion gases to be emitted into the atmosphere that contain cadmium, lead, and selenium in excess of 230 µgm/dscm, combined emissions, corrected to 7 percent oxygen; and,
- ii. You must not discharge or cause combustion gases to be emitted into the atmosphere that contain antimony, arsenic, beryllium, chromium, cobalt, manganese, and nickel in excess of 92 μgm/dscm, combined emissions, corrected to 7 percent oxygen.

[Reg.19.304 and 40 C.F.R. § 63.1219(e)(1&2)]

Destruction and Removal Efficiency (DRE) Standard

j. The permittee shall maintain a 99.99% destruction and removal efficiency (DRE) for each principal organic hazardous constituent (POHC) designated under paragraph (c)(3) of this section. The DRE shall be calculated using the following equation:

DRE = [1-(Win / Wout)] X 100%

Where:

Win = mass feedrate of one principal organic hazardous constituent (POHC) in a waste feed stream; and

Wout = mass emission rate of the same POHC present in exhaust emissions prior to release to the atmosphere. [Reg.19.304 and 40 C.F.R. \S 63.1219(c)]

- k. The permittee must treat the POHCs in the waste feed that are specified under paragraph (c)(3)(ii) of this section to the extent required by paragraphs §63.1203(c)(1) and (c)(2) (i.e. 99.99% as stated in the previous Specific Condition). [Reg.19.304 and 40 C.F.R. § 63.1219(c)(3)(i)]
- The permittee shall specify one or more POHCs from the list of hazardous air pollutants established by 42 U.S.C. 7412(b)(1), excluding caprolactum as provided by §63.60, for each waste to be burned. The permittee must base this specification on the degree of difficulty of incineration of the organic constituents in the waste and on their concentration or mass in the waste feed, considering the results of waste analyses or other data and information. [Reg.19.304 and 40 C.F.R. § 63.1219(c)(3)(ii)]
- m. The emission limits provided by paragraphs §63.1203(a) and §63.1203(b) are presented with two significant figures. Although the permittee must perform intermediate calculations using at least three significant figures, the resultant emission levels may be rounded to two significant figures to document compliance. [Reg.19.304 and 40 C.F.R. § 63.1219(d)]

Compliance Provisions

n. The permittee shall comply with the standards of 40 CFR Part 63, Subpart EEE no later than September 30, 2003 unless the Administrator grants an extension under §63.6(i) or §63.1213. [Reg.19.304 and 40 C.F.R. § 63.1206(a)(1)]

- o. The permittee shall comply with the emission standards and operating requirements set forth in 40 CFR Part 63, Subpart EEE at all times when hazardous wastes are in the combustion chamber, except as specified in §63.1206(b)(1)(i) and (ii). [Reg.19.304 and 40 C.F.R. § 63.1206(b)(1)]
- p. The permittee shall demonstrate compliance based on performance testing under operating conditions representative of the extreme range of normal conditions. This performance test shall be conducted as required by 40 CFR §63.1206(b)(12). Prior to the completion of the performance test, the permittee shall document compliance with 40 CFR Part 63, Subpart EEE no later than September 30, 2003. This documentation of compliance (DOC) will ensure that operating parameters are established to ensure compliance with this subpart. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(2)]
- q. The permittee may petition the Administrator to grant an extension of compliance with the emission standards of this subpart as provided by §63.6(i) and §63.1213.
 [Reg.19.304 and 40 C.F.R. § 63.1206(b)(4)]
- r. The permittee shall comply with the requirements of notification, performance testing, and waste-burning restrictions as outlined in §63.1206(b)(5)(i)(A) through (C) if the facility plans to make a change in design, operation, or maintenance that could adversely affect compliance. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(5)(i)]
- s. The permittee shall document any changes not affecting compliance in the facility operating record. Revisions reflecting such changes shall also be made, as necessary, to the performance test plan, Documentation of Compliance, Notification of Compliance, and the start-up, shutdown, and malfunction plan. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(5)(ii)]
- t. The permittee shall ensure and document compliance with the CO emission standard using a continuous emission monitoring system (CEMS). The permittee shall ensure and document compliance with the hydrocarbon emission standard by complying with the CO emission standard, and by demonstrating that the highest hourly rolling average hydrocarbon level emitted during the comprehensive performance test does not exceed the hydrocarbon emission limit. [Reg.19.304 and 40 C.F.R. § 631206(b)(6)]
- u. The permittee shall demonstrate destruction removal efficiency (DRE) of at least 99.99% during the comprehensive performance test conducted in compliance with the conditions of §63.1207(b)(1) of this subpart. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(7)]

- v. Any particulate matter and opacity standards or any permit or other emissions operating parameter limits or conditions, including any limitation on workplace practices, that are applicable to hazardous waste combustors to ensure compliance with any particulate matter or opacity standard of parts 60, 61, 63, 264, 265, and 266 of this chapter (i.e., any title 40 particulate or opacity standards) do not apply while the permittee conducts particulate matter continuous emissions monitoring system (CEMS) correlation tests. However, compliance with this condition is not required until such time that the Agency promulgates all performance specifications and operational requirements applicable to PM CEMS. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(8)(i) and (ii)]
- w. For provisions of this section to apply, the permittee must develop a particulate matter CEMS correlation test plan that includes the following information. This test plan may be included as part of the comprehensive performance test plan required under §63.1207(e) and (f):
 - i. Number of test conditions and number of runs for each test condition;
 - ii. Target particulate matter emission level for each test condition;
 - iii. How you plan to modify operations to attain the desired particulate matter emission levels; and
 - iv. Anticipated normal emission levels.

The permittee shall submit the particulate CEMS correlation test plan to the Administrator for approval at least 90 calendar days before the correlation test is scheduled to be conducted. However, compliance with this condition is not required until such time that the Agency promulgates all performance specifications and operational requirements applicable to PM CEMS. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(8)(iii)(A) and (B)]

- x. If the Administrator fails to approve or disapprove the correlation test plan with the time period specified by §63.7(c)(3)(i), the plan is considered approved, unless the Administrator has requested additional information. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(8)(iv)]
- y. The particulate matter and associated operating limits and conditions will not be waived for more than 96 hours, in the aggregate, for a correlation test, including all runs of all test conditions, unless more time is approved by the Administrator. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(8)(v)]
- z. The permittee must return to operating conditions indicative of compliance with the applicable particulate matter and opacity standards as soon as possible after correlation testing is completed. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(8)(vii)]

- aa. The permittee must calculate the hazardous waste residence time and include the calculation in the performance test plan under §63.1207(f) and the operating record. The permittee must also provide the hazardous waste residence time in the Documentation of Compliance under §63.1211(c) and the Notification of Compliance under §63.1210(b). [Reg.19.304 and 40 C.F.R. § 63.1206(b)(11)]
- bb. The permittee must conduct a minimum of three runs of a performance test required under §63.1207 to document compliance with the emission standards of this subpart. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(12)(i)]
- cc. The permittee must document compliance with the emission standards based on the arithmetic average of the emission results of each run, except that the permittee must document compliance with the destruction and removal efficiency standard for each run of the comprehensive performance test individually. [Reg.19.304 and 40 C.F.R. § 63.1206(b)(12)(ii)]

General Operating Requirements

- dd. The permittee must operate only under the operating requirements specified in the Notification of Compliance under §63.1207(j) and §63.1210(b), except: [Reg.19.304 and 40 C.F.R. § 63.1206(c)(1)(i)]
 - i. During performance tests under approved test plans according to §63.1207(e), (f), and (g), [Reg.19.304 and 40 C.F.R. § 63.1206(c)(1)(i)(A)]
 - ii. Under the conditions of paragraph (b)(1)(i) or (ii) of this section [Reg.19.304 and 40 C.F.R. § 63.1206(c)(1)(i)(B)]
 - The Notification of Compliance must contain operating requirements including, but not limited to, the operating requirements of this section and §63.1209. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(1)(ii)]
 - 2. Failure to comply with the operating requirements is failure to ensure compliance with the emissions standards of this subpart [Reg.19.304 and 40 C.F.R. § 63.1206(c)(1)(iii)]
 - 3. Operating requirements in the Notification of Compliance are applicable requirements for purposes of parts 70 and 71 of this chapter [Reg.19.304 and 40 C.F.R. § 63.1206(c)(1)(iv)]
 - The operating requirements specified in the Notification of Compliance will be incorporated in the Title V permit. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(1)(v)]
- ee. Except as provided in by paragraph (c)(2)(ii) of this section, the permittee is subject to the startup, shutdown, and malfunction plan requirements of §63.6(e)(3). [Reg.19.304 and 40 C.F.R. § 63.1206(c)(2)(i)]

- i. If the permittee elects to comply with §270.235(a)(1)(iii), §270.235(a)(2)(iii), or §270.235(b)(1)(ii) of this chapter to address RCRA concerns, the permittee must comply with the provisions of §63.1206(c)(2)(ii)(A) and (B). [Reg.19.304 and 40 C.F.R. § 63.1206(c)(2)(ii)]
- ii. The permittee must identify in the plan the projected oxygen correction factor based on normal operations to use during periods of startup and shutdown. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(2)(iii)]
- iii. The permittee must record the plan in the operating record. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(2)(iv)]
- iv. The permittee must comply with this requirement for operation under the startup, shutdown, and malfunction plan. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(2)(v)]
- ff. Upon the compliance date, the permittee must operate the combustor with a functioning system that immediately and automatically cuts off the hazardous waste feed, except as provided by paragraph (c)(3)(viii) of this section, when the following conditions apply: [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(i)]
 - i. When operating parameter limits specified under §63.1209; an emission standard monitored by CEMS; and the allowable combustion chamber pressure; [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(i)(A)]
 - ii. When the span value of any CMS detector, except a CEMS, is met or exceeded; [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(i)(B)]
 - iii. Upon malfunction of a CMS monitoring an operating parameter limit specified under §63.1209 or an emission level; or [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(i)(C)]
 - iv. When any component of the automatic waste feed cutoff system fails. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(i)(D)]
- gg. During an automatic waste feed cutoff (AWFCO) the permittee must continue to duct combustion gases to the air pollution control system while hazardous waste remains in the combustion chamber. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(ii)]
- hh. The permittee must continue to monitor during the cutoff the operating parameters for which limits are established under §63.1209 and the emissions required under that section to be monitored by a CEMS, and the permittee shall not restart the hazardous waste feed until the operating parameters and emission levels are within specified limits. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(iii)]
- ii. If the AWFCO system fails to automatically and immediately cutoff the flow of hazardous waste upon exceedance of a parameter required to be interlocked with the AWFCO system under paragraph (c)(3)(i) of this section, the permittee has failed to comply with the AWFCO requirements of paragraph (c)(3) of this section. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(iv)]

- jj. If, after any AWFCO, there is an exceedance of any emission standard or operating requirement, irrespective of whether the exceedance occurred while hazardous waste remained in the combustion chamber, the permittee shall investigate the cause of the AWFCO, take appropriate corrective measures to minimize future AWFCOs and record the findings and corrective measures in the operating record. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(v)]
- kk. For each set of 10 exceedances of an emissions standard or operating requirement while hazardous waste remains in the combustion chamber during a 60-day block period, the permittee must submit to the Administrator a written report within 5 calendar days of the 10th exceedance documenting the exceedances and the results of the investigation and corrective measures taken. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(vi)(A)]
- On a case-by-case basis, the Administrator may require excessive exceedance reporting when fewer than 10 exceedances occur during a 60-day block period. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(vi)(B)]
- mm. The AWFCO system and associated alarms must be tested at least weekly to verify operability, unless the permittee documents in the operating record that weekly inspections will unduly restrict or upset operations and that less frequent inspection will be adequate. At a minimum, the permittee must conduct operability testing at least monthly. The permittee must document and record in the operating record AWFCO operability test procedures and results. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(vii)]
- nn. The permittee may ramp down waste feed according to the requirements of §63.1206(c)(3)(viii), except as provided in §63.1206(c)(3)(B). The permittee must document ramp down procedures in the operating and maintenance plan. If the AWFCO is triggered by an exceedance of any of the following operating limits, the permittee may not ramp down the waste feed cutoff: Minimum combustion chamber temperature, maximum hazardous waste feedrate, or any hazardous waste firing system operating limits that may have been established. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(3)(viii)]
- oo. The permittee is subject to the emergency safety vent (ESV) operating and reporting requirements set forth in this section. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(4)(i through iv)]
- pp. The permittee is subject to the combustion system leak control system operating and reporting requirements set forth in these sections. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(5)(i)(A) and (ii)]
- qq. The permittee is subject to the operator training and certification standards set forth in this section. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(6)(i through vii)]
- rr. The permittee must prepare and at all times operate according to an operation and maintenance plan which complies with the requirements set forth in these sections. [Reg.19.304 and 40 C.F.R. § 63.1206(c)(7)(i)(A-D)]

Performance Testing Requirements

- ss. The permittee must conduct performance testing in accordance with the applicable requirements contained in this section. [Reg.19.304 and 40 C.F.R. § 63.1207(a-m)]
- tt. The permittee must commence the initial comprehensive performance test not later than six months after the compliance date. [Reg.19.304 and 40 C.F.R. § 63.1207(c)(1)]
- uu. The permittee must conduct testing periodically as described in paragraphs (d)(1) through (3) of this section. The date of commencement of the initial comprehensive performance test is the basis for establishing the deadline to commence the initial confirmatory performance test and the next comprehensive performance test. The permittee may conduct performance testing at any time prior to the required date. The deadline for commencing subsequent confirmatory and comprehensive performance testing is based on the date of commencement of the previous comprehensive performance test. [Reg.19.304 and 40 C.F.R. § 63.1207(d)(1) through (3)]
 - i. The permittee must commence comprehensive testing no later than 61 months after the date of commencing the previous comprehensive performance test.
 - ii. The permittee must commence confirmatory performance testing no later than 31 months after the date of commencing the previous comprehensive performance test. To ensure that the confirmatory test is conducted approximately midway between comprehensive performance tests, the Administrator will not approve a test plan that schedules testing within 18 months of commencing the previous comprehensive performance test.
 - iii. The permittee must complete performance testing within 60 days after the date of commencement, unless the Administrator determines that a time extension is warranted based on documentation in writing of factors beyond the permittee's control that prevent testing from being completed within 60 days.

Applicable Testing Requirements under the Interim Standard

vv. *Waiver of periodic comprehensive performance tests*. Except as provided by §63.1207(c)(2), the permittee must conduct only an initial comprehensive performance test under the interim standards (i.e., the standards published in the Federal Register on February 13, 2002). All subsequent comprehensive performance testing requirements are waived under the interim standards. The provisions in the introductory test to paragraph (d) and in paragraph (d)(1) of this section do not apply until EPA promulgates permanent replacement standards pursuant to the Settlement Agreement noticed in the Federal Register on November 16, 2001. [Reg.19.304 and 40 C.F.R. § 63.1207(d)(4)(i)].

- ww. Waiver of periodic confirmatory performance tests. The permittee is not required to conduct a confirmatory test under the interim standards (i.e., the standards published in the Federal Register on February 13, 2002). The confirmatory testing requirements in the introductory text to paragraph (d) and in (d)(2) of §63.1207 are waived until EPA promulgates permanent replacement standards pursuant to the Settlement Agreement noticed in the Federal Register on November 16, 2001. [Reg.19.304 and 40 C.F.R. § 63.1207(d)(4)(ii)].
- xx. The permittee must submit to the Administrator a notification of intent to conduct a comprehensive performance test and CMS performance evaluation and a site specific test plan and CMS performance evaluation plan at least one year before the performance test and performance evaluation are scheduled to begin. [Reg.19.304 and 40 C.F.R. § 63.1207(e)(1)(i)]
- yy. The permittee must submit to the Administrator a notification of intent to conduct the comprehensive performance test at least 60 calendar days before the test is scheduled to begin. [Reg.19.304 and 40 C.F.R. § 63.1207(e)(1)(i)(B)]
- zz. The permittee must submit to the Administrator a notification of intent to conduct a confirmatory performance test and CMS performance evaluation and a test plan and CMS performance evaluation plan at least 60 calendar days before the performance test is scheduled to begin. [Reg.19.304 and 40 C.F.R. § 63.1207(e)(1)(ii)]

Test Methods

 aaa. The permittee shall use the test methods contained in this section when determining compliance with the emissions standards of this subpart. [Reg.19.304 and 40 C.F.R. § 63.1208(a-b)]

Monitoring Requirements

- bbb. The permittee is subject to the applicable monitoring requirements contained in these sections. [Reg.19.304 and 40 C.F.R. § 63.1209 (a-q)]
- ccc. The permittee must either use a carbon monoxide or hydrocarbon CEMS to demonstrate compliance with either the carbon monoxide and hydrocarbon standards under this subpart. The permittee must also use an oxygen CEMS to continuously correct the carbon monoxide and hydrocarbon levels to 7 percent oxygen. [[Reg.19.304 and 40 C.F.R. § 63.1209(a)(1)(i)]
- ddd. The permittee must install, calibrate, maintain, and operate a particulate matter CEMS to demonstrate and monitor compliance with the particulate matter standards under this subpart. However, compliance with this condition is not required until such time that the Agency promulgates all performance specifications and operational requirements applicable to PM CEMS. [Reg.19.304 and 40 C.F.R. § 63.1209(a)(1)(iii)]

- eee. The permittee must install, calibrate, maintain, and continuously operate the CEMS in compliance with the quality assurance procedures provided in the appendix to this subpart and Performance Specifications 1 (opacity), 4B (carbon monoxide and oxygen), and 8A (hydrocarbons) in Appendix B, Part 60 of this chapter. [Reg.19.304 and 40 C.F.R. § 63.1209(a)(2)]
- fff. The permittee must comply with the span requirements of §63.1209(a)(3). [Reg.19.304 and 40 C.F.R. § 63.1209(a)(3)]
- ggg. The permittee may petition the Administrator to use CEMS for compliance monitoring for other standards in lieu of compliance with the corresponding operating parameter limits under this section. [Reg.19.304 and 40 C.F.R. § 63.1209(a)(5)]
- hhh. The permittee will begin recording one-minute and hourly rolling average values as necessary to ensure that 60 one-minute values will be available for calculating the initial hourly rolling average before the compliance date. The permittee will continue to use the CEMS to monitor parameters as required in §63.1209(a)(6). [Reg.19.304 and 40 C.F.R. § 63.1209(a)(6)]
- iii. The permittee will use the Comprehensive Performance Test to demonstrate that the THC standard is met to establish operating parameters for DRE. [Reg.19.304 and 40 C.F.R. § 63.1209(a)(7)]
- jjj. The permittee will use Continuous Monitoring Systems where necessary to ensure compliance with operating parameters established in the Documentation of Compliance or the Notification of Compliance. [Reg.19.304 and 40 C.F.R. § 63.1209(b)]
- kkk. Prior to feeding the material, the permittee must obtain an analysis of each feedstream that is sufficient to document compliance with the applicable feedrate limits provided in this section. [Reg.19.304 and 40 C.F.R. § 63.1209(c)(1)]
- lll. The permittee must develop and implement a feedstream analysis plan and record it in the operating record. [Reg.19.304 and 40 C.F.R. § 63.1209(c)(2)]
- mmm. The permittee must submit the feedstream analysis plan to the Administrator for review and approval, if requested. [Reg.19.304 and 40 C.F.R. § 63.1209(c)(3)]
- nnn. To comply with the applicable feedrate limits of this section, the permittee must monitor and record the feedrates as follows: [Reg.19.304 and 40 C.F.R. § 63.1209(c)(4)]
 - i. Determine and record the value of the parameter for each feedstream by sampling and analysis or other method;
 - Determine and record the mass or volume flowrate of each stream by a CMS. If the permittee determines flowrate of a feedstream by volume, the permittee must determine and record the density of the feedstream by sampling and analysis (unless the permittee reports the constituent concentration in units of weight per volume); and

- iii. Calculate and record the mass feedrate of the parameter per unit time.
- ooo. The requirements of §63.8(d) (Quality control program) and (e) (Performance evaluation of continuous monitoring systems) apply, except that the permittee must conduct performance evaluations components of the CMS under the frequency and procedures (for example, submittal of performance evaluation test plan for review and approval) applicable to performance tests as provided by §63.1207. [Reg.19.304 and 40 C.F.R. § 63.1209(d)(1)]
- ppp. The permittee shall maintain and operate each CMS as specified in §63.8(c), except for §63.8(c)(3) and §63.8(c)(4)(ii). The permittee shall have the CMS installed, calibrated, and operational on the compliance date. The permittee must sample the regulated parameter without interruption, and evaluate the detector response at least once each 15 seconds, and compute and record the average values at least every 60 seconds. [Reg.19.304 and 40 C.F.R. § 63.1209(f)]
- qqq. The permittee shall follow the requirements for the reduction of monitoring data as specified in 40 CFR §63.8(g). [Reg.19.304 and 40 C.F.R. § 63.1209(h)]
- rrr. When one operating parameter is used to ensure compliance with one or more standards, the permittee must use the most stringent limit, determined during the comprehensive performance test, as the limit for that operating parameter. [Reg.19.304 and 40 C.F.R. § 63.1209(i)]
- sss. To remain in compliance with the destruction and removal efficiency (DRE) standards, the permittee must establish operating limits during the comprehensive performance test (or during a previous DRE test under provisions of §63.1206(b)(7)) for the following parameters, unless the limits are based on manufacturer specifications and comply with those limits at all times that hazardous waste remains in the combustion chamber. [Reg.19.304 and 40 C.F.R. § 63.1209(j)]
- ttt. The permittee must measure the temperature of each combustion chamber at locations that best represents, as practicable, the bulk gas temperature in the combustion zone. The permittee must document the temperature measurement location in the test plan submitted under §63.1207(e), and establish a minimum rolling average limit as the average of the test run values. [Reg.19.304 and 40 C.F.R. § 63.1209(j)(1)(i) and (ii)]
- uuu. As an indicator of gas residence time in the control device, the permittee must establish and comply with a limit on the maximum flue gas flowrate, the maximum production rate, or another parameter that is documented in the site-specific test plan as an appropriate surrogate for gas residence time, as the average of the maximum hourly rolling averages for each run. [Reg.19.304 and 40 C.F.R. § 63.1209(j)(2)(i)]
- vvv. The permittee must establish limits on the maximum pumpable and total (i.e., pumpable and nonpumpable) hazardous waste feedrate for each location where hazardous waste is fed. [Reg.19.304 and 40 C.F.R. § 63.1209(j)(3)(i)]

- www. The permittee must specify operating parameters and limits to ensure that good operation of each hazardous waste firing system is maintained. [Reg.19.304 and 40 C.F.R. § 63.1209(j)(4)]
- xxx. The permittee must comply with the dioxin and furans emission standard by establishing and complying with the following operating parameter limits. You must base the limits on operations during the comprehensive performance test, unless the limits are based on manufacturer specifications. [Reg.19.304 and 40 C.F.R. § 63.1209(k)]
- yyy. The permittee must measure the temperature of each combustion chamber at a location that best represents, as practicable, the bulk gas temperature in the combustion zone. The permittee must document the temperature measurement location in the test plan and establish a minimum hourly rolling average limit as the average of the test runs. [Reg.19.304 and 40 C.F.R. § 63.1209(k)(2)(i) and (ii)]
- zzz. As an indicator of gas residence time in the control device, the permittee must establish and comply with a limit on the maximum flue gas flowrate, the maximum production rate, or another parameter which is an appropriate surrogate for residence time, as the hourly rolling averages for each run. Compliance with this limit is on an hourly rolling average basis. [Reg.19.304 and 40 C.F.R. § 63.1209(k)(3)(i) and (ii),]
- aaaa. The permittee must establish limits on the maximum pumpable and total (pumpable and nonpumpable) waste feedrate for each location where waste is fed and establish limits as the average of the maximum hourly rolling averages for each run. Compliance shall be based on an hourly rolling average basis. [Reg.19.304 and 40 C.F.R. § 63.1209(k)(4)(i B iii)]
- bbbb. The permittee shall ensure compliance with the mercury emission standard by establishing minimum mercury feed rate limit. The limit is established as a 12-hour rolling average limit for the total feedrate of mercury in all feedstreams as the average of the test run values, unless mercury feedrate limits are extrapolated from performance test feedrate levels, and maintaining the scrubber operating parameters described under §63.1209(1). [Reg.19.304 and 40 C.F.R. § 63.1209(1)]The permittee must comply with the particulate matter emission standard by establishing and complying with the operating parameter limits found in §63.1209(m) of this subpart. [Reg.19.304 and 40 C.F.R. § 63.1209(m)]
- cccc. The permittee must comply with the particulate matter emission standard by establishing and complying with the operating parameter limits found in §63.1209(m) of this subpart. [Reg.19.304 and 40 C.F.R. § 63.1209(m)]
- dddd. The permittee must establish a maximum ash feedrate limit as the average of the test run averages. [Reg.19.304 and 40 C.F.R. § 63.1209(m)(3)]
- eeee. The permittee must comply with the semivolatile metal (cadmium and lead) and low volatile metal (arsenic, beryllium, and chromium) emission standards by establishing and complying with the following operating parameter limits: [Reg.19.304 and 40 C.F.R. § 63.1209(n)]

- i. The permittee must establish feed rate limits for semivolatile metals and low volatile metals, with compliance based on 12-hour rolling average limits as the average of the test run averages. [Reg.19.304 and 40 C.F.R. § 63.1209(n)(2)(i)(A) and (B)]
- ii. The permittee must establish operating parameter limits on the particulate matter control device as specified by paragraph 63.1209(m)(1). [Reg.19.304 and 40 C.F.R. § 63.1209(n)(3)]
- iii. The permittee must establish a 12-hour rolling average limit for the feedrate of total chlorine and chloride in all feedstreams as the average of the test run averages. [Reg.19.304 and 40 C.F.R. § 63.1209(n)(4)]
- ffff. The permittee must establish a 12-hour rolling average limit for the total feedrate of chlorine in all feedstreams as the average of the test run averages. [Reg.19.304 and 40 C.F.R. § 63.1209(o)(1)]
- gggg. As an indicator of gas residence time in the control device, the permittee must establish a limit on the maximum flue gas flowrate, the maximum production rate, or another parameter documented in the site-specific test plan as an appropriate surrogate for gas residence time, as the average of the maximum hourly rolling averages for each run. This limit must be maintained on an hourly rolling average basis. [Reg.19.304 and 40 C.F.R. § 63.1209(o)(2)(i)]
- hhhh. The permittee must establish the following parameter limits for the wet scrubber: [Reg.19.304 and 40 C.F.R. § 63.1209(o)(3)]
 - i. Minimum pressure drop. The permittee must establish a limit on minimum pressure drop on an hourly rolling average as the average of the test run averages.
 - ii. Minimum pH. The permittee must establish a limit on minimum pH on an hourly rolling average as the average of the test run averages.
 - iii. Minimum scrubber liquid flow rate. The permittee must establish a minimum scrubber liquid flow rate on an hourly rolling average as the average of the test run averages.

Notification Requirements

- iiii. The permittee shall submit all of the applicable notifications prior to the deadlines established in this subpart. [Reg.19.304 and 40 C.F.R. § 63.1210(a)(1)]
- jjjj. The permittee must submit the required notifications outlined in this section to the Administrator in order to request or elect to comply with the alternative requirements contained in this subpart. [Reg.19.304 and 40 C.F.R. § 63.1210(a)(2)]
- kkkk. Upon postmark of the Notification of Compliance, the operating parameter limits identified in the Notification of Compliance, as applicable, shall be complied with, the limits identified in the Document of Compliance or a previous Notification of Compliance are no longer applicable. [Reg.19.304 and 40 C.F.R. § 63.1210(b)(2)]

Recordkeeping and Reporting Requirements

11. The permittee shall submit the reports required by this subpart to the Administrator prior to the deadlines set forth in this subpart. [Reg.19.304 and 40 C.F.R. § 63.1211]

Procedure for Extending the Compliance Date

mmmm. The permittee may request an extension of the compliance date to install pollution prevention or waste minimization controls provided that the conditions outlined in this section are met. [Reg.19.304 and 40 C.F.R. § 63.1213]

Solvent Recovery 4PSR-00 and SR-FUG

Source Description

FutureFuel Chemical Company operates dedicated Solvent Recovery equipment to recover solvents that become contaminated during the manufacturing processes. Individual streams from the chemical manufacturing processes are transferred to storage tanks in the solvent recovery area. These streams are pumped to a pH adjustment system and then to a series of distillation columns. After distillation, the solvents are reused in the manufacturing processes or are sold for other uses. Process emissions from the Solvent Recovery Facility are controlled with Regenerative Thermal Oxidizers (RTOs), 5N09-01.

Specific Conditions

SR 1. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by SR 3 and SR 5. [Reg.19.501 *et seq.* and 40 C.F.R. § 52 Subpart E]

SN	Description	Pollutant	lb/hr	tpy
4PSR-00	Solvent Recovery Facility	VOC	4.0	17.0
SR-FUG	Solvent Recovery Fugitive Emissions	VOC	5.0	21.9

SR 2. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by SR 3 and SR 5. The facility shall show compliance with the facility total ton/yr limits using the procedures outlined in Plantwide Conditions 12 through 16. Hourly Plantwide Hazardous Air Pollutant emissions are limited by Plantwide Conditions 16. Additional HAP limitations are included in Plantwide Condition 13. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

SN	Description	Pollutant	lb/hr	tpy
4PSR-00	Solvent Recovery Facility	Organic Pollutants**	4.0	17.0
SR-FUG	Solvent Recovery Fugitive Emissions	Organic Pollutants**	5.0	21.9

**Organic Pollutants are considered to be VOC, all organic HAPs, and organic Air Contaminates.

SR 3. The permittee shall not process more than 40 million pounds of VOC solvents at the solvent recovery facility in any consecutive 12-month period. [Reg.19.705, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 70.6]

- SR 4. The permittee shall keep monthly records of the amount of solvent throughput at 4PSR-00. These records shall be kept on site and made available upon request. [Reg.19.705 and 40 C.F.R. § 52 Subpart E]
- SR 5. The permittee is limited to 250 million gallons of biodiesel refining in the Solvent Recovery area in any consecutive 12-month period. [Reg.19.705, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 70.6]
- SR 6. Emissions from biodiesel production in the Solvent Recovery area shall be recalculated monthly and shall be based upon a 12-month rolling total. The records shall be updated by the last day of the month following the recorded 12-month period and shall be kept on site and made available for inspection upon request. [Reg.19.705 and 40 C.F.R. § 52 Subpart E and Reg.18.1004 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]

Wastewater Treatment Facility 7K01-01 and 7M01-02

Source Description

The Wastewater Treatment Plant (WWT) at FutureFuel Chemical Company treats process wastewater from various areas of the plant, sanitary sewage, and some storm water. The wastewater treatment system is an extended aeration activated sludge design consisting of equalization and neutralization, aeration, and clarification. The excess biomass is aerobically digested and either land applied on-site via a spray irrigation system or dewatered and burned in the coal-fired boilers. A diversion tank is used to reduce organic or hydraulic peaks in the untreated wastewater.

Specific Conditions

WWT 1. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with WWT 3. [Reg.19.501 *et seq.* and 40 C.F.R. § 52 Subpart E]

PES #	Description	Pollutant	lb/hr	tpy
7K01-01	WWT Facilities	VOC	28.6	125.3
7M01-02	EQ-C-03 Decant Tank	VOC	0.8	3.5
7M01-04	EQ-C-05 WWT Container	VOC	0.1	0.4

WWT 2. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall not exceed the emission rates set forth in the following table. Emission rates are based upon the maximum capacity of equipment. The facility shall show compliance with the facility total ton/yr limits using the procedures outlined in Plantwide Conditions 12 through 16. Hourly Plantwide Hazardous Air Pollutant emissions are limited by Plantwide Conditions 16. Additional HAP limitations are included in Plantwide Condition 13. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

PES #	Description	Pollutant	lb/hr	tpy
7K01-01	WWT Facilities	Organic Pollutants**	28.6	125.3
7M01-02	EQ-C-03 Decant Tank	Organic Pollutants**	0.8	3.5
7M01-03	EQ-C-04 pH Control Tank	Inorganics*	0.1	0.4
7М01-03-В	EQ-C-04-2 pH Control Tank	Inorganics*	0.1	0.1

PES #	Description	Pollutant	lb/hr	tpy
7M01-04	EQ-C-05 WWT Container	Organic Pollutants**	0.1	0.4

**Organic Pollutants are considered to be VOC, all organic HAPs, and organic Air Contaminates.

WWT 3. The permittee shall calculate the emissions of VOC from the wastewater basins (7K01-01) using a Department or EPA approved model once per year. Annual emissions shall be based on the most recent twelve consecutive months of operation. [Reg.19.703, 40 C.F.R. § 52 Subpart E, and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

Isopropyl Benzene Production (DIPB) 5NDIPB-TNK, 5N03-54, 5N03-48, 5N03-55, and DIPB-FUG

Source Description

The isopropyl benzene process consists of alkylation of benzene with propylene. A catalyst is used to promote the reaction. The intermediate, cumene, reacts with propylene to produce three isopropyl benzene variations. Subsequent to the reaction, the catalyst is removed by washing and decanting. Any benzene or intermediate generated that is not fully converted to product is recycled back into the process. 5N03-48 and 5N03-55 are scrubbers associated with the DIPB process.

NESHAP Subpart V (National Emission Standard for Equipment Leaks (Fugitive Emission Sources) applies to certain equipment installed after 1/5/81. Therefore, this regulation is applicable.

NESHAP Subpart J (Equipment Leaks of Benzene) applies to certain equipment in benzene service. Affected equipment does exist at the DIPB plant. Therefore, this regulation is applicable. This regulation requires affected facilities to comply with the requirements contained in NESHAP Subpart V (Equipment Leaks of VHAP).

NESHAP Subpart Y (Benzene Storage Vessels) applies to storage tank #T-210. A flare (5N03-54) controls emissions from this tank.

NESHAP Subpart FF (Benzene Waste Operations) applies to benzene waste streams at certain facilities, including chemical manufacturing plants. It is applicable to the DIPB plant. A flare (5N03-54) controls benzene emissions generated by the wastewater collection tank (T-9) and the wastewater steam stripper (D-9).

NESHAP Subpart FFFF applies to this part of the facility. The requirements are incorporated in a separate MON section.

Specific Conditions

IB 1. The permittee shall not exceed the emission rates set forth in the following table. These rates are based on maximum physical capacity of the equipment. [Reg.19.501 *et seq.* and 40 C.F.R. § 52 Subpart E]

PES #	Description	Pollutant	lb/hr	tpy
5NDIPB-TNK	Tank Bubble (10 tanks)	VOC	0.2	0.6
		PM_{10}	0.1	0.4
5N03-54	Flare	SO_2	0.5	1.9
		VOC	0.9	3.9
		CO	2.4	9.8
		NO _x	0.5	1.9
DIPB-FUG	Fugitive Emissions from Isopropyl Benzene Process	VOC	0.3	1.3

IB 2. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall not exceed the emission rates set forth in the following table. Emission rates are based upon the maximum capacity of equipment. The facility shall show compliance with the facility total ton/yr limits using the procedures outlined in Plantwide Conditions 12 through 16. Hourly Plantwide Hazardous Air Pollutant emissions are limited by Plantwide Conditions 16. Additional HAP limitations are included in Plantwide Condition 13. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

PES #	Description	Pollutant	lb/hr	tpy
5NDIPB-TNK	Tank Bubble (10 tanks)	Organic Pollutants** Inorganic Pollutants*	0.2 0.1	0.6 0.1
5N03-48	Scrubber	Inorganics*	0.1	0.4
5N03-54	Flare	PM Organic Pollutants**	0.1 0.9	0.4 3.9
5N03-55	Scrubber	Inorganics*	0.1	0.4
DIPB-FUG	Fugitive Emissions from Isopropyl Benzene Process	Organic Pollutants**	0.3	1.3

*Inorganics are considered to be non-organic Hazardous Air Pollutants

**Organic Pollutants are considered to be VOC, all organic HAPs, and organic Air Contaminates.

- IB 3. The permittee shall operate and maintain a control system on scrubbers 5N03-55 and 5N03-48 in accordance with most current version of the Facility Operating Plan. [Reg.19.303 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]
- IB 4. The permittee shall operate a control system which detects the presence of a flame on the flare (5N03-54) and gives an alarm if flame is not detected. The reactor process shall be shut down if the cause of the alarm is not corrected within 30 minutes. [Reg.19.303 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]
- IB 5. The permittee shall operate and monitor the DIPB off-gas flare (5N03-54) according to the requirements of §60.18(d), (e), and (f). Records shall be kept of all periods of operation during which the flare pilot flame is absent. [Reg.19.304 and 40 C.F.R. § 60.18]
- IB 6. The permittee shall monitor SN-5N03-54 as outlined in IB 4 and IB 5. The permittee shall submit reports as required by Plantwide Condition 33. [Rule 19.304 and 40 C.F.R. § 64]

40 CFR Part 61 Subpart FF - NESHAP for Benzene Waste Operations

IB 7. The permittee shall comply with all applicable benzene waste stream reporting requirements at the flare (5N03-54) (which controls benzene emissions generated by the wastewater steam stripper) of all applicable waste stream records as outlined by §61.356(b), and as outlined by §61.357(c). [Reg.19.304 and 40 C.F.R. §§ 61.356(b) and 61.357(c)]

- IB 8. Provisions of the Subpart FF NESHAP for Benzene Waste Operations shall apply to chemical manufacturing plants. [Reg.19.304 and 40 C.F.R. § 61.340(a)]
- IB 9. Subpart FF NESHAP for Benzene Waste Operations, §61.340(c) identifies wastes exempt from the regulatory requirements. The permittee may claim exemptions under §61.342(c)(2) and §61.342(c)(3) providing documentation is kept to support the exemptions identified. [Reg.19.304 and 40 C.F.R. §§ 61.340(c), 61.342(a), 61.342(c)(2), 61.342(c)(3)]
- IB 10. The permittee may claim exemptions as allowed in §61.342(a)(1) through (4), providing documentation of the benzene waste quantity is calculated as specified for the exemption. [Reg.19.304 and 40 C.F.R. § 61.342(a)(1) through (4)]
- IB 11. The permittee has elected to remove or destroy benzene in the waste using a treatment process or wastewater treatment system which complies with §61.348 (Treatment Processes) [Reg.19.304 and 40 C.F.R. § 61.342(c)(1)(i)]
- IB 12. The permittee shall comply with the standards specified in §61.343 through §61.347, as applicable, for each waste management unit. [Reg.19.304 and 40 C.F.R. § 61.342(c)(1)(ii)]
- IB 13. The permittee may elect to meet one of these additional compliance options identified in the citations noted. Subpart FF does not require prior approval for changing between options. The permittee may choose between compliance options as long as documentation is readily available for inspection to provide evidence of compliance with the applicable treatment standard. [Reg.19.304 and 40 C.F.R. § 61.342(d), (e), and (f)]
- IB 14. Compliance with this subpart will be determined by review of facility records and results from tests and inspections using methods and procedures specified in §61.355. [Reg.19.304 and 40 C.F.R. § 61.342(g)]

40 CFR Part 61 Subpart V - National Emission Standard for Equipment Leaks (Fugitive Emission Sources)

Applicability and designation of sources

- IB 15. The provisions of this subpart apply to each of the following sources that are intended to operate in volatile hazardous air pollutant (VHAP) service: pumps, compressors, pressure relief devices, sampling connection systems, open-ended valves or lines, valves, connectors, surge control vessels, bottoms receivers, and control devices or systems required by this subpart. [Reg.19.304 and 40 C.F.R. § 61.240(a)]
 - a. The provisions of this subpart apply to the sources listed in paragraph (a) after the date of promulgation of a specific subpart in part 61. [Reg.19.304 and 40 C.F.R. § 61.240(b)]
 - b. While the provisions of this subpart are effective, a source to which this subpart applies that is also subject to the provisions of 40 CFR part 60 only will be required to comply with the provisions of this subpart. [Reg.19.304 and 40 C.F.R. § 61.240(c)]
 - c. Alternative means of compliance —

- i. Option to comply with part 65. Owners or operators may choose to comply with 40 CFR part 65 to satisfy the requirements of §§ 61.242-1 through 61.247 for equipment that is subject to this subpart and that is part of the same process unit. When choosing to comply with 40 CFR part 65, the requirements of §§ 61.245(d) and 61.246(i) and (j) still apply. Other provisions applying to owners or operators who choose to comply with 40 CFR part 65 are provided in 40 CFR 65.1. [Reg.19.304 and 40 C.F.R. § 61.240(d)(1)]
- ii. *Part 65, subpart C or F.* For owners or operators choosing to comply with 40 CFR part 65, each surge control vessel and bottoms receiver subject to this subpart that meets the conditions specified in table 1 or table 2 of this subpart shall meet the requirements for storage vessels in 40 CFR part 65, subpart C; all other equipment subject to this subpart shall meet the requirements in 40 CFR part 65, subpart F. [Reg.19.304 and 40 C.F.R. § 61.240(d)(2)]
- iii. Part 61, subpart A. Owners or operators who choose to comply with 40 CFR part 65, subpart C or F, must also comply with §§ 61.01, 61.02, 61.05 through 61.08, 61.10(b) through (d), 61.11, and 61.15 for that equipment. All sections and paragraphs of subpart A of this part that are not mentioned in this paragraph (d)(3) do not apply to owners or operators of equipment subject to this subpart complying with 40 CFR part 65, subpart C or F, except that provisions required to be met prior to implementing 40 CFR part 65 still apply. Owners and operators who choose to comply with 40 CFR part 65, subpart C or F, must comply with 40 CFR part 65, subpart A. [Reg.19.304 and 40 C.F.R. § 61.240(d)(3)]
- *Rules referencing this subpart*. Owners or operators referenced to this subpart from subpart F or J of this part may choose to comply with 40 CFR part 65 for all equipment listed in paragraph (a) of this section. [Reg.19.304 and 40 C.F.R. § 61.240(d)(4)]

General Requirements

- IB 16. For equipment subject to these standards, the following General Requirements apply:
 - a. Each owner or operator subject to the provisions of this subpart shall demonstrate compliance with the requirements of §§ 61.242-1 to 61.242-11 for each new and existing source as required in 40 CFR 61.05, except as provided in §§ 61.243 and 61.244. [Reg.19.304 and 40 C.F.R. § 61.242-1(a)]
 - b. Compliance with this subpart will be determined by review of records, review of performance test results, and inspection using the methods and procedures specified in § 61.245. [Reg.19.304 and 40 C.F.R. § 61.242-1(b)]
 - c. An owner or operator may request a determination of alternative means of emission limitation to the requirements of §§ 61.242-2, 61.242-3, 61.242-5, 61.242-6, 61.242-7, 61.242-8, 61.242-9 and 61.242-11 as provided in § 61.244. [Reg.19.304 and 40 C.F.R. § 61.242-1(c)(1)]

- d. If the Administrator makes a determination that a means of emission limitation is at least a permissible alternative to the requirements of § 61.242-2, 61.242-3, 61.242-5, 61.242-6, 61.242-7, 61.242-8, 61.242-9 or 61.242-11, an owner or operator shall comply with the requirements of that determination. [Reg.19.304 and 40 C.F.R. § 61.242-1(c)(2)]
- e. Each piece of equipment to which this subpart applies shall be marked in such a manner that it can be distinguished readily from other pieces of equipment. [Reg.19.304 and 40 C.F.R. § 61.242-1(d)]
- f. Equipment that is in vacuum service is excluded from the requirements of § 61.242-2, to § 61.242-11 if it is identified as required in § 61.246(e)(5). [Reg.19.304 and 40 C.F.R. § 61.242-1(e)]

Standards: Pumps

- IB 17. For pumps, the permittee is subject to the following requirements:
 - a. Each pump shall be monitored monthly to detect leaks by the methods specified in § 61.245(b), except as provided in § 61.242-1(c) and paragraphs (d), (e), (f) and (g) of this section. [Reg.19.304 and 40 C.F.R. § 61.242-2(a)(1)]
 - Each pump shall be checked by visual inspection each calendar week for indications of liquids dripping from the pump seal. [Reg.19.304 and 40 C.F.R. § 61.242-2(a)(2)]
 - c. If an instrument reading of 10,000 ppm or greater is measured, a leak is detected. [Reg.19.304 and 40 C.F.R. § 61.242-2(b)(1)]
 - d. If there are indications of liquids dripping from the pump seal, a leak is detected. [Reg.19.304 and 40 C.F.R. § 61.242-2(b)(2)]
 - e. When a leak is detected, it shall be repaired as soon as practicable, but not later than 15 calendar days after it is detected, except as provided in § 61.242-10.
 [Reg.19.304 and 40 C.F.R. § 61.242-2(c)(1)]
 - f. A first attempt at repair shall be made no later than 5 calendar days after each leak is detected. [Reg.19.304 and 40 C.F.R. § 61.242-2(c)(2)]
 - g. Each pump equipped with a dual mechanical seal system that includes a barrier fluid system is exempt from the requirements of paragraphs (a) and (b) of this section, provided the following requirements are met:
 - i. Each dual mechanical seal system is:
 - 1. Operated with the barrier fluid at a pressure that is at all times greater than the pump stuffing box pressure; or
 - 2. Equipped with a barrier fluid degassing reservoir that is routed to a process or fuel gas system or connected by a closed-vent system to a control device that complies with the requirements of § 61.242-11; or

- 3. Equipped with a system that purges the barrier fluid into a process stream with zero VHAP emissions to atmosphere.
- ii. The barrier fluid is not in VHAP service and, if the pump is covered by standards under 40 CFR part 60, is not in VOC service.
- iii. Each barrier fluid system is equipped with a sensor that will detect failure of the seal system, the barrier fluid system, or both.
- iv. Each pump is checked by visual inspection each calendar week for indications of liquids dripping from the pump seal.
 - 1. If there are indications of liquid dripping from the pump seal at the time of the weekly inspection, the pump shall be monitored as specified in § 61.245 to determine the presence of VOC and VHAP in the barrier fluid.
 - 2. If the monitor reading (taking into account any background readings) indicates the presence of VHAP, a leak is detected. For the purpose of this paragraph, the monitor may be calibrated with VHAP, or may employ a gas chromatography column to limit the response of the monitor to VHAP, at the option of the owner or operator.
 - 3. If an instrument reading of 10,000 ppm or greater (total VOC) is measured, a leak is detected.
- v. Each sensor as described in paragraph (d)(3) of this section is checked daily or is equipped with an audible alarm.
- vi. The owner or operator determines, based on design considerations and operating experience, criteria applicable to the presence and frequency of drips and to the sensor that indicates failure of the seal system, the barrier fluid system, or both.
- vii. If indications of liquids dripping from the pump seal exceed the criteria established in paragraph (d)(6)(i) of this section, or if, based on the criteria established in paragraph (d)(6)(i) of this section, the sensor indicates failure of the seal system, the barrier fluid system, or both, a leak is detected.
- viii. When a leak is detected, it shall be repaired as soon as practicable, but no later than 15 calendar days after it is detected, except as provided in § 61.242-10.
 - ix. A first attempt at repair shall be made no later than five calendar days after each leak is detected.

[Reg.19.304 and 40 C.F.R. § 61.242-2(d)(1)-(d)(6)]

- h. Any pump that is designated, as described in § 61.246(e)(2), for no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, is exempt from the requirements of paragraphs (a), (c), and (d) if the pump:
 - i. Has no externally actuated shaft penetrating the pump housing,
 - ii. Is demonstrated to be operating with no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, as measured by the method specified in § 61.245(c), and
 - iii. Is tested for compliance with paragraph (e)(2) initially upon designation, annually, and at other times requested by the Administrator.

[Reg.19.304 and 40 C.F.R. § 61.242-2(e)(1-3)]

- If any pump is equipped with a closed-vent system capable of capturing and transporting any leakage from the seal or seals to a process or fuel gas system or to a control device that complies with the requirements of § 61.242-11, it is exempt from the requirements of paragraphs (a) through (e) of this section. [Reg.19.304 and 40 C.F.R. § 61.242-2(f)]
- j. Any pump that is designated, as described in § 61.246(f)(1), as an unsafe-tomonitor pump is exempt from the monitoring and inspection requirements of paragraphs (a) and (d)(4) through (6) of this section if:
 - i. The owner or operator of the pump demonstrates that the pump is unsafeto-monitor because monitoring personnel would be exposed to an immediate danger as a consequence of complying with paragraph (a) of this section; and
 - ii. The owner or operator of the pump has a written plan that requires monitoring of the pump as frequently as practicable during safe-tomonitor times but not more frequently than the periodic monitoring schedule otherwise applicable, and repair of the equipment according to the procedures in paragraph (c) of this section if a leak is detected.

[Reg.19.304 and 40 C.F.R. § 61.242-2(g)(1&2)]

k. Any pump that is located within the boundary of an unmanned plant site is exempt from the weekly visual inspection requirement of paragraphs (a)(2) and (d)(4) of this section, and the daily requirements of paragraph (d)(5) of this section, provided that each pump is visually inspected as often as practicable and at least monthly. [Reg.19.304 and 40 C.F.R. § 61.242-2(h)]

Standards: Compressors

IB 18. For compressors, the following requirements apply:

- a. Each compressor shall be equipped with a seal system that includes a barrier fluid system and that prevents leakage of process fluid to atmosphere, except as provided in § 61.242-1(c) and paragraphs (h) and (i) of this section. [Reg.19.304 and 40 C.F.R. § 61.242-3(a)]
- b. Each compressor seal system as required in paragraph (a) shall be:
 - i. Operated with the barrier fluid at a pressure that is greater than the compressor stuffing box pressure; or
 - ii. Equipped with a barrier fluid system degassing reservoir that is routed to a process or fuel gas system or connected by a closed-vent system to a control device that complies with the requirements of § 61.242-11; or
 - iii. Equipped with a system that purges the barrier fluid into a process stream with zero VHAP emissions to atmosphere.

[Reg.19.304 and 40 C.F.R. § 61.242-3(b)(1-3)]

- c. The barrier fluid shall not be in VHAP service and, if the compressor is covered by standards under 40 CFR part 60, shall not be in VOC service. [Reg.19.304 and 40 C.F.R. § 61.242-3(c)]
- d. Each barrier fluid system as described in paragraphs (a)-(c) of this section shall be equipped with a sensor that will detect failure of the seal system, barrier fluid system, or both. [Reg.19.304 and 40 C.F.R. § 61.242-3(d)]
- e. Each sensor as required in paragraph (d) of this section shall be checked daily or shall be equipped with an audible alarm unless the compressor is located within the boundary of an unmanned plant site. [Reg.19.304 and 40 C.F.R. § 61.242-3(e)(1)]
- f. The owner or operator shall determine, based on design considerations and operating experience, a criterion that indicates failure of the seal system, the barrier fluid system, or both. [Reg.19.304 and 40 C.F.R. § 61.242-3(e)(2)]
- g. If the sensor indicates failure of the seal system, the barrier fluid system, or both based on the criterion determined under paragraph (e)(2) of this section, a leak is detected. [Reg.19.304 and 40 C.F.R. § 61.242-3(f)]
- h. When a leak is detected, it shall be repaired as soon as practicable, but not later than 15 calendar days after it is detected, except as provided in § 61.242-10.
 [Reg.19.304 and 40 C.F.R. § 61.242-3(g)(1)]
- i. A first attempt at repair shall be made no later than 5 calendar days after each leak is detected. [Reg.19.304 and 40 C.F.R. § 61.242-3(g)(2)]

- j. A compressor is exempt from the requirements of paragraphs (a) and (b) of this section if it is equipped with a closed-vent system to capture and transport leakage from the compressor drive shaft back to a process or fuel gas system or to a control device that complies with the requirements of § 61.242-11, except as provided in paragraph (i) of this section. [Reg.19.304 and 40 C.F.R. § 61.242-3(h)]
- k. Any Compressor that is designated, as described in § 61.246(e)(2), for no detectable emission as indicated by an instrument reading of less than 500 ppm above background is exempt from the requirements of paragraphs (a)-(h) if the compressor:
 - i. Is demonstrated to be operating with no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, as measured by the method specified in § 61.245(c); and
 - ii. Is tested for compliance with paragraph (i)(1) initially upon designation, annually, and at other times requested by the Administrator.

[Reg.19.304 and 40 C.F.R. § 61.242-3(i)(1&2)]

Standards: Pressure relief devices in gas/vapor service

- IB 19. The following requirements apply to pressure relief devices in gas/vapor service:
 - a. Except during pressure releases, each pressure relief device in gas/vapor service shall be operated with no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, as measured by the method specified in § 61.245(c). [Reg.19.304 and 40 C.F.R. § 61.242-4(a)]
 - b. After each pressure release, the pressure relief device shall be returned to a condition of no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, as soon as practicable, but no later than 5 calendar days after each pressure release, except as provided in § 61.242-10. [Reg.19.304 and 40 C.F.R. § 61.242-4(b)(1)]
 - c. No later than 5 calendar days after the pressure release, the pressure relief device shall be monitored to confirm the condition of no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, as measured by the method specified in § 61.245(c). [Reg.19.304 and 40 C.F.R. § 61.242-4(b)(2)]
 - d. Any pressure relief device that is routed to a process or fuel gas system or equipped with a closed-vent system capable of capturing and transporting leakage from the pressure relief device to a control device as described in § 61.242-11 is exempt from the requirements of paragraphs (a) and (b) of this section. [Reg.19.304 and 40 C.F.R. § 61.242-4(c)]

- e. Any pressure relief device that is equipped with a rupture disk upstream of the pressure relief device is exempt from the requirements of paragraphs (a) and (b) of this section, provided the owner or operator complies with the requirements in paragraph (d)(2) of this section. [Reg.19.304 and 40 C.F.R. § 61.242-4(d)(1)]
- f. After each pressure release, a new rupture disk shall be installed upstream of the pressure relief device as soon as practicable, but no later than 5 calendar days after each pressure release, except as provided in § 61.242-10. [Reg.19.304 and 40 C.F.R. § 61.242-4(d)(2)]

Standards: Sampling connecting systems

- IB 20. For Sampling connection systems, the following requirements apply:
 - a. Each sampling connection system shall be equipped with a closed-purge, closedloop, or closed vent system, except as provided in § 61.242-1(c). Gases displaced during filling of the sample container are not required to be collected or captured. [Reg.19.304 and 40 C.F.R. § 61.242-5(a)]
 - b. Each closed-purge, closed-loop, or closed-vent system as required in paragraph
 (a) of this section shall comply with the requirements specified in paragraphs
 (b)(1) through (4) of this section.
 - i. Return the purged process fluid directly to the process line.
 - ii. Collect and recycle the purged process fluid to a process.
 - iii. Be designed and operated to capture and transport all the purged process fluid to a control device that complies with the requirements of § 61.242-11; or
 - iv. Collect, store, and transport the purged process fluid to any of the following systems or facilities:
 - 1. A waste management unit as defined in § 63.111, if the waste management unit is subject to and operated in compliance with the provisions of 40 CFR part 63, subpart G, applicable to Group 1 wastewater streams;
 - 2. A treatment, storage, or disposal facility subject to regulation under 40 CFR part 262, 264, 265, or 266;
 - 3. A facility permitted, licensed, or registered by a state to manage municipal or industrial solid waste, if the process fluids are not hazardous waste as defined in 40 CFR part 261;

[Reg.19.304 and 40 C.F.R. § 61.242-5(b)(1-4)]

c. In situ sampling systems and sampling systems without purges are exempt from the requirements of paragraphs (a) and (b) of this section. [Reg.19.304 and 40 C.F.R. § 61.242-5(c)]

Standards: Open-ended valves or lines

- IB 21. For open-ended valves or lines, the following requirements apply:
 - a. Each open-ended valve or line shall be equipped with a cap, blind flange, plug, or a second valve, except as provided in § 61.242-1(c). [Reg.19.304 and 40 C.F.R. § 61.242-6(a)(1)]
 - b. The cap, blind flange, plug, or second valve shall seal the open end at all times except during operations requiring process fluid flow through the open-ended valve or line. [Reg.19.304 and 40 C.F.R. § 61.242-6(a)(2)]
 - c. Each open-ended valve or line equipped with a second valve shall be operated in a manner such that the valve on the process fluid end is closed before the second valve is closed. [Reg.19.304 and 40 C.F.R. § 61.242-6(b)]
 - d. When a double block and bleed system is being used, the bleed valve or line may remain open during operations that require venting the line between the block valves but shall comply with paragraph (a) at all other times. [Reg.19.304 and 40 C.F.R. § 61.242-6(c)]
 - e. Open-ended valves or lines in an emergency shutdown system which are designed to open automatically in the event of a process upset are exempt from the requirements of paragraphs (a), (b) and (c) of this section. [Reg.19.304 and 40 C.F.R. § 61.242-6(d)]
 - f. Open-ended valves or lines containing materials which would auto- catalytically polymerize or would present an explosion, serious overpressure, or other safety hazard if capped or equipped with a double block and bleed system as specified in paragraphs (a) through (c) of this section are exempt from the requirements of paragraphs (a) through (c) of this section. [Reg.19.304 and 40 C.F.R. § 61.242-6(e)]

Standards: Valves

IB 22. For Valves, the following apply:

- a. Each valve shall be monitored monthly to detect leaks by the method specified in § 61.245(b) and shall comply with paragraphs (b)-(e), except as provided in paragraphs (f), (g), and (h) of this section, § 61.243-1 or § 61.243-2, and § 61.242-1(c). [Reg.19.304 and 40 C.F.R. § 61.242-7(a)]
- b. If an instrument reading of 10,000 ppm or greater is measured, a leak is detected. [Reg.19.304 and 40 C.F.R. § 61.242-7 (b)]
- c. Any valve for which a leak is not detected for 2 successive months may be monitored the first month of every quarter, beginning with the next quarter, until a leak is detected. [Reg.19.304 and 40 C.F.R. § 61.242-7 (c)(1)]
- d. If a leak is detected, the valve shall be monitored monthly until a leak is not detected for 2 successive months. [Reg.19.304 and 40 C.F.R. § 61.242-7 (c)(2)]

- e. When a leak is detected, it shall be repaired as soon as practicable, but no later than 15 calendar days after the leak is detected, except as provided in § 61.242-10. [Reg.19.304 and 40 C.F.R. § 61.242-7 (d)(1)]
- f. A first attempt at repair shall be made no later than 5 calendar days after each leak is detected. [Reg.19.304 and 40 C.F.R. § 61.242-7 (d)(2)]
- g. First attempts at repair include, but are not limited to, the following best practices where practicable:
 - i. Tightening of bonnet bolts;
 - ii. Replacement of bonnet bolts;
 - iii. Tightening of packing gland nuts;
 - iv. Injection of lubricant into lubricated packing.

[Reg.19.304 and 40 C.F.R. § 61.242-7 (e)(1-4)]

- h. Any valve that is designated, as described in § 61.246(e)(2), for no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, is exempt from the requirements of paragraph (a) if the valve:
 - i. Has no external actuating mechanism in contact with the process fluid,
 - ii. Is operated with emissions less than 500 ppm above background, as measured by the method specified in § 61.245(c), and
 - iii. Is tested for compliance with paragraph (f)(2) of this section initially upon designation, annually, and at other times requested by the Administrator.

[Reg.19.304 and 40 C.F.R. § 61.242-7 (f)(1-3)]

- i. Any valve that is designated, as described in \S 61.246(f)(1), as an unsafe-tomonitor valve is exempt from the requirements of paragraph (a) if:
 - i. The owner or operator of the valve demonstrates that the valve is unsafe to monitor because monitoring personnel would be exposed to an immediate danger as a consequence of complying with paragraph (a), and
 - ii. The owner or operator of the valve adheres to a written plan that requires monitoring of the valve as frequently as practicable during safe-to-monitor times.

[Reg.19.304 and 40 C.F.R. § 61.242-7 (g)(1&2)]

- j. Any valve that is designated, as described in 61.246(f)(2), as a difficult-tomonitor valve is exempt from the requirements of paragraph (a) if::
 - i. The owner or operator of the valve demonstrates that the valve cannot be monitored without elevating the monitoring personnel more than 2 meters above a support surface.

- ii. The process unit within which the valve is located either becomes an affected facility through § 60.14 or § 60.15 or the owner or operator designates less than 3.0 percent of the total number of valves as difficult-to-monitor, and
- iii. The owner or operator of the valve follows a written plan that requires monitoring of the valve at least once per calendar year.

[Reg.19.304 and 40 C.F.R. § 61.242-7 (g)(1-3)]

Standards: Pressure relief services in liquid service and connectors

- IB 23. For Pressure relief services in liquid service and connectors the following apply:
 - a. If evidence of a potential leak is found by visual, audible, olfactory, or any other detection method at pressure relief devices in liquid service and connectors, the owner or operator shall follow either one of the following procedures, except as provided in § 61.242-1(c):
 - i. The owner or operator shall monitor the equipment within 5 days by the method specified in § 61.245(b) and shall comply with the requirements of paragraphs (b) through (d) of this section.
 - ii. The owner or operator shall eliminate the visual, audible, olfactory, or other indication of a potential leak within 5 calendar days of detection.

[Reg.19.304 and 40 C.F.R. § 61.242-8 (a)(1&2)]

- b. If an instrument reading of 10,000 ppm or greater is measured, a leak is detected. [Reg.19.304 and 40 C.F.R. § 61.242-8 (b)]
- c. When a leak is detected, it shall be repaired as soon as practicable, but not later than 15 calendar days after it is detected, except as provided in § 61.242-10 [Reg.19.304 and 40 C.F.R. § 61.242-8 (c)(1)]
- d. The first attempt at repair shall be made no later than 5 calendar days after each leak is detected. [Reg.19.304 and 40 C.F.R. § 61.242-8 (c)(2)]
- e. First attempts at repair include, but are not limited to, the best practices described under § 61.242-7(e). [40 CFR§61.242-8 (d)]

Standards: Surge control vessels and bottoms receivers

IB 24. Each surge control vessel or bottoms receiver that is not routed back to the process and that meets the conditions specified in table 1 or table 2 of this subpart shall be equipped with a closed-vent system capable of capturing and transporting any leakage from the vessel back to the process or to a control device as described in § 61.242-11, except as provided in § 61.242-1(c); or comply with the requirements of 40 CFR 63.119(b) or (c). [40 CFR§61.242-9]

Standards: Delay of repair

IB 25. For leaks detected, the following are requirements for delay of repair:

- a. Delay of repair of equipment for which leaks have been detected will be allowed if repair within 15 days is technically infeasible without a process unit shutdown. Repair of this equipment shall occur before the end of the next process unit shutdown. [40 CFR§61.242-10 (a)]
- b. Delay of repair of equipment will be allowed for equipment which is isolated from the process and which does not remain in VOC service. [40 CFR§61.242-10 (b)]
- c. Delay of repair for valves will be allowed if:
 - i. The owner or operator demonstrates that emissions of purged material resulting from immediate repair are greater than the fugitive emissions likely to result from delay of repair, and
 - ii. When repair procedures are effected, the purged material is collected and destroyed or recovered in a control device complying with § 61.242-11.

[Reg.19.304 and 40 C.F.R. § 61.242-10 (c)(1 and 2)]

- d. Delay of repair for pumps will be allowed if:
 - i. Repair requires the use of a dual mechanical seal system that includes a barrier fluid system, and
 - ii. Repair is completed as soon as practicable, but not later than 6 months after the leak was detected.

[Reg.19.304 and 40 C.F.R. § 61.242-10 (d)(1 and 2)]

e. Delay of repair beyond a process unit shutdown will be allowed for a valve if valve assembly replacement is necessary during the process unit shutdown, valve assembly supplies have been depleted, and valve assembly supplies had been sufficiently stocked before the supplies were depleted. Delay of repair beyond the next process unit shutdown will not be allowed unless the next process unit shutdown occurs sooner than 6 months after the first process unit shutdown. [Reg.19.304 and 40 C.F.R. § 61.242-10 (e)]

Standards: Closed vent systems and control devices

- IB 26. The permittee shall comply with the following standards for closed vent systems and control devices subject to this subpart:
 - a. Owners or operators of closed-vent systems and control devices used to comply with provisions of this subpart shall comply with the provisions of this section, except as provided in §61.242-1(c). [Reg.19.304 and 40 C.F.R. § 61.242-11 (a)]
 - b. Vapor recovery systems (for example, condensers and absorbers) shall be designed and operated to recover the VOC emissions vented to them with an efficiency of 95 percent or greater, or to an exit concentration of 20 parts per million by volume, whichever is less stringent. [Reg.19.304 and 40 C.F.R. § 61.242-11 (b)]

- c. Enclosed combustion devices shall be designed and operated to reduce the VHAP emissions vented to them with an efficiency of 95 percent or greater, or to an exit concentration of 20 parts per million by volume, on a dry basis, corrected to 3 percent oxygen, whichever is less stringent, or to provide a minimum residence time of 0.50 seconds at a minimum temperature of 760 °C [Reg.19.304 and 40 C.F.R. § 61.242-11 (c)]
- d. Flares used to comply with this subpart shall comply with the requirements of § 60.18. [Reg.19.304 and 40 C.F.R. § 61.242-11(d)]
- e. Owners or operators of control devices used to comply with the provisions of this subpart shall monitor these control devices to ensure that they are operated and maintained in conformance with their designs. [Reg.19.304 and 40 C.F.R. § 61.242-11 (e)]
- f. Except as provided in paragraphs (i) through (k) of this section, each closed vent system shall be inspected according to the procedures and schedule specified in paragraph (f)(1) or (2) of this section, as applicable.
 - i. If the vapor collection system or closed vent system is constructed of hardpiping, the owner or operator shall comply with the following requirements:
 - Conduct an initial inspection according to the procedures in § 61.245(b); and
 - 2. Conduct annual visual inspections for visible, audible, or olfactory indications of leaks.
 - ii. If the vapor collection system or closed vent system is constructed of ductwork, the owner or operator shall:
 - Conduct an initial inspection according to the procedures in § 61.245(b); and
 - 2. Conduct annual inspections according to the procedures in § 61.245(b).

[Reg.19.304 and 40 C.F.R. § 61.242-11(f)(1&2)]

- g. Leaks, as indicated by an instrument reading greater than 500 parts per million by volume above background or by visual inspections, shall be repaired as soon as practicable except as provided in paragraph (h) of this section.
 - i. A first attempt at repair shall be made no later than 5 calendar days after the leak is detected.
 - ii. Repair shall be completed no later than 15 calendar days after the leak is detected.

[Reg.19.304 and 40 C.F.R. § 61.242-11(g)(1&2)]
- h. Delay of repair of a closed vent system for which leaks have been detected is allowed if the repair is technically infeasible without a process unit shutdown or if the owner or operator determines that emissions resulting from immediate repair would be greater than the fugitive emissions likely to result from delay of repair. Repair of such equipment shall be complete by the end of the next process unit shutdown. [Reg.19.304 and 40 C.F.R. § 61.242-11(h)]
- i. If a vapor collection system or closed vent system is operated under a vacuum, it is exempt from the inspection requirements of paragraphs (f)(1)(i) and (f)(2) of this section. [Reg.19.304 and 40 C.F.R. § 61.242-11(i)]
- j. Any parts of the closed vent system that are designated, as described in paragraph (l)(1) of this section, as unsafe-to-inspect are exempt from the inspection requirements of paragraphs (f)(1)(i) and (2) of this section if they comply with the following requirements:
 - i. The owner or operator determines that the equipment is unsafe to inspect because inspecting personnel would be exposed to an imminent or potential danger as a consequence of complying with paragraphs (f)(1)(i)or (f)(2) of this section; and
 - ii. The owner or operator has a written plan that requires inspection of the equipment as frequently as practicable during safe-to-inspect times.

[Reg.19.304 and 40 C.F.R. § 61.242-11(j)(1&2)]

- k. Any parts of the closed vent system that are designated, as described in paragraph (l)(2) of this section, as difficult-to-inspect are exempt from the inspection requirements of paragraphs (f)(1)(i) and (2) of this section if they comply with the following requirements:
 - i. The owner or operator determines that the equipment cannot be inspected without elevating the inspecting personnel more than 2 meters above a support surface; and
 - ii. The owner or operator has a written plan that requires inspection of the equipment at least once every 5 years. A closed vent system is exempt from inspection if it is operated under a vacuum.

[Reg.19.304 and 40 C.F.R. § 61.242-11(k)(1-3)]

- 1. The owner or operator shall record the following information:
 - i. Identification of all parts of the closed vent system that are designated as unsafe to inspect, an explanation of why the equipment is unsafe to inspect, and the plan for inspecting the equipment.
 - ii. Identification of all parts of the closed vent system that are designated as difficult to inspect, an explanation of why the equipment is difficult to inspect, and the plan for inspecting the equipment.

- iii. For each inspection during which a leak is detected, a record of the information specified in § 61.246(c).
- iv. For each inspection conducted in accordance with § 61.245(b) during which no leaks are detected, a record that the inspection was performed, the date of the inspection, and a statement that no leaks were detected.
- v. For each visual inspection conducted in accordance with paragraph (f)(1)(ii) of this section during which no leaks are detected, a record that the inspection was performed, the date of the inspection, and a statement that no leaks were detected.

[Reg.19.304 and 40 C.F.R. § 61.242-11(1)(1-5)]

m. Closed vent systems and control devices used to comply with provisions of this subpart shall be operated at all times when emissions may be vented to them.
 [Reg.19.304 and 40 C.F.R. § 61.242-11(m)]

Alternative standards for valves—allowable percentage of valves leaking

- IB 27. The permittee may comply with the following alternative standards for leaking valves:
 - a. An owner or operator may elect to have all valves within a process unit to comply with an allowable percentage of valves leaking of equal to or less than 2.0 percent. [Reg.19.304 and 40 C.F.R. § 61.243-1 (a)]
 - b. The following requirements shall be met if an owner or operator decides to comply with an allowable percentage of valves leaking:
 - i. An owner or operator must notify the Administrator that the owner or operator has elected to have all valves within a process unit to comply with the allowable percentage of valves leaking before implementing this alternative standard, as specified in § 61.247(d).
 - ii. A performance test as specified in paragraph (c) of this section shall be conducted initially upon designation, annually, and at other times requested by the Administrator.
 - iii. If a valve leak is detected, it shall be repaired in accordance with § 61.242-7(d) and (e).

[Reg.19.304 and 40 C.F.R. § 61.243-1(b)(1-3)]

- c. Performance tests shall be conducted in the following manner:
 - i. All valves in VHAP service within the process unit shall be monitored within 1 week by the methods specified in § 61.245(b).
 - ii. If an instrument reading of 10,000 ppm or greater is measured, a leak is detected.
 - iii. The leak percentage shall be determined by dividing the number of valves in VHAP service for which leaks are detected by the number of valves in VHAP service within the process unit.

[Reg.19.304 and 40 C.F.R. § 61.243-1 (c)(1-3)]

- d. Owner or operators who elect to have all valves comply with this alternative standard shall not have a process unit with a leak percentage greater than 2.0 percent. [Reg.19.304 and 40 C.F.R. § 61.243-1 (d)]
- e. If an owner or operator decides no longer to comply with § 61.243-1, the owner or operator must notify the Administrator in writing that the work practice standard described in § 61.242-7(a)-(e) will be followed. [Reg.19.304 and 40 C.F.R. § 61.243-1 (e)]

Alternative standards for valves—skip period leak detection and repair

- IB 28. The following requirements apply as alternative standards for valves—skip period leak detection and repair:
 - a. An owner or operator may elect to comply with one of the alternative work practices specified in paragraphs (b)(2) and (3) of this section. [Reg.19.304 and 40 C.F.R. § 61.243-2 (a)(1)]
 - An owner or operator must notify the Administrator before implementing one of the alternative work practices, as specified in § 61.247(d). [Reg.19.304 and 40 C.F.R. § 61.243-2 (a)(2)]
 - c. An owner or operator shall:
 - i. An owner or operator shall comply initially with the requirements for valves, as described in § 61.242-7.
 - ii. After 2 consecutive quarterly leak detection periods with the percentage of valves leaking equal to or less than 2.0, an owner or operator may begin to skip one of the quarterly leak detection periods for the valves in VHAP service.
 - iii. After five consecutive quarterly leak detection periods with the percentage of valves leaking equal to or less than 2.0, an owner or operator may begin to skip three of the quarterly leak detection periods for the valves in VHAP service.
 - iv. If the percentage of valves leaking is greater than 2.0, the owner or operator shall comply with the requirements as described in § 61.242-7 but may again elect to use this section.

[Reg.19.304 and 40 C.F.R. § 61.243-2 (b)(1-4)]

Alternative means of emission limitation

- IB 29. The permittee shall comply with the following alternative means of emission limitation:
 - a. Permission to use an alternative means of emission limitation under section 112(e)(3) of the Clean Air Act shall be governed by the following procedures (b) through (f) of this section. [Reg.19.304 and 40 C.F.R. § 61.244(a)]
 - b. Where the standard is an equipment, design, or operational requirement:

- i. Each owner or operator applying for permission shall be responsible for collecting and verifying test data for an alternative means of emission limitation to test data for the equipment, design, and operational requirements.
- ii. The Administrator may condition the permission on requirements that may be necessary to assure operation and maintenance to achieve the same emission reduction as the equipment, design, and operational requirements.

[Reg.19.304 and 40 C.F.R. § 61.244(b)(1&2)]

- c. Where the standard is a work practice:
 - i. Each owner or operator applying for permission shall be responsible for collecting and verifying test data for an alternative means of emission limitation.
 - ii. For each source for which permission is requested, the emission reduction achieved by the required work practices shall be demonstrated for a minimum period of 12 months.
 - iii. For each source for which permission is requested, the emission reduction achieved by the alternative means of emission limitation shall be demonstrated.
 - iv. Each owner or operator applying for permission shall commit in writing each source to work practices that provide for emission reductions equal to or greater than the emission reductions achieved by the required work practices.
 - v. The Administrator will compare the demonstrated emission reduction for the alternative means of emission limitation to the demonstrated emission reduction for the required work practices and will consider the commitment in paragraph (c)(4).
 - vi. The Administrator may condition the permission on requirements that may be necessary to assure operation and maintenance to achieve the same emission reduction as the required work practices of this subpart.

[Reg.19.304 and 40 C.F.R. § 61.244(c)(1-6)]

- d. An owner or operator may offer a unique approach to demonstrate the alternative means of emission limitation. [Reg.19.304 and 40 C.F.R. § 61.244(d)]
- e. Manufacturers of equipment used to control equipment leaks of a VHAP may apply to the Administrator for permission for an alternative means of emission limitation that achieves a reduction in emissions of the VHAP achieved by the equipment, design, and operational requirements of this subpart. [Reg.19.304 and 40 C.F.R. § 61.244(e)(1)]

f. The Administrator will grant permission according to the provisions of paragraphs (b), (c), and (d). [Reg.19.304 and 40 C.F.R. § 61.244(e)(2)]

Test methods and procedures

- IB 30. The permittee shall conduct performance testing in accordance with the provisions of this section.
 - a. Monitoring, as required in §§ 61.242, 61.243, 61.244, and 61.135, shall comply with the following requirements.
 - i. Monitoring shall comply with Method 21 of appendix A of 40 CFR part 60.
 - ii. The detection instrument shall meet the performance criteria of Method 21.
 - iii. The instrument shall be calibrated before use on each day of its use by the procedures specified in Method 21.
 - iv. Calibration gases shall be:
 - 1. Zero air (less than 10 ppm of hydrocarbon in air); and
 - 2. A mixture of methane or n-hexane and air at a concentration of about, but less than, 10,000 ppm methane or n-hexane.
 - v. The instrument probe shall be traversed around all potential leak interfaces as close to the interface as possible as described in Method 21.

[Reg.19.304 and 40 C.F.R. § 61.245(b)(1-5)]

- b. When equipment is tested for compliance with or monitored for no detectable emissions, the owner or operator shall comply with the following requirements:
 - i. The requirements of paragraphs (b) (1) through (4) shall apply.
 - ii. The background level shall be determined, as set forth in Method 21.
 - iii. The instrument probe shall be traversed around all potential leak interfaces as close to the interface as possible as described in Method 21.
 - iv. The arithmetic difference between the maximum concentration indicated by the instrument and the background level is compared with 500 ppm for determining compliance.

[Reg.19.304 and 40 C.F.R. § 61.245 (c)(1-4)]

- c. Each piece of equipment within a process unit that can conceivably contain equipment in VHAP service is presumed to be in VHAP service unless an owner or operator demonstrates that the piece of equipment is not in VHAP service. For a piece of equipment to be considered not in VHAP service, it must be determined that the percent VHAP content can be reasonably expected never to exceed 10 percent by weight. For purposes of determining the percent VHAP content of the process fluid that is contained in or contacts equipment, procedures that conform to the methods described in ASTM Method D-2267 (incorporated by the reference as specified in § 61.18) shall be used.
 - i. An owner or operator may use engineering judgment rather than the procedures in paragraph (d)(1) of this section to demonstrate that the percent VHAP content does not exceed 10 percent by weight, provided that the engineering judgment demonstrates that the VHAP content clearly does not exceed 10 percent by weight. When an owner or operator and the Administrator do not agree on whether a piece of equipment is not in VHAP service, however, the procedures in paragraph (d)(1) of this section shall be used to resolve the disagreement.
 - ii. If an owner or operator determines that a piece of equipment is in VHAP service, the determination can be revised only after following the procedures in paragraph (d)(1) of this section.
 - iii. Samples used in determining the percent VHAP content shall be representative of the process fluid that is contained in or contacts the equipment or the gas being combusted in the flare.

[Reg.19.304 and 40 C.F.R. § 61.245 (d)(1-3)]

- d. The owner or operator shall determine compliance with the standards of flares as follows:
 - i. Method 22 shall be used to determine visible emissions.
 - ii. A thermocouple or any other equivalent device shall be used to monitor the presence of a pilot flame in the flare.
 - iii. The net heating value (H_{τ}) of the gas being combusted in a flare shall be computed using the following equation:

$$H_{I} = K \sum_{i=1}^{n} C_{i} H_{i}$$

Where:

K = conversion constant, 1.740×10^{7} (g-mole) (MJ)/(ppm-scm-kcal) (metric units); or 4.674×10^{8} ((g-mole) (Btu)/(ppm-scf-kcal)) (English units)

> C_i = Concentration of sample component "i" in ppm, as measured by Method 18 of appendix A to 40 CFR part 60 and ASTM D2504-67, 77, or 88 (Reapproved 1993) (incorporated by reference as specified in § 61.18).H_i = Net heating value of the sample, MJ/scm (BTU/scf); where the net enthalpy per mole of offgas is based on combustion at 25 °C and 760 mm Hg (77 °F and 14.7 psi), but the standard temperature for determining the volume corresponding to one mole is 20 °C (68 °F).

- iv. The actual exit velocity of a flare shall be determined by dividing the volumetric flowrate (in units of standard temperature and pressure), as determined by Method 2, 2A, 2C, or 2D, as appropriate, by the unobstructed (free) cross section area of the flare tip.
- v. The maximum permitted velocity for air assisted flares shall be computed using the following equation:

$$V_{\max} = K_1 + K_2 H_T$$

Where:

 V_{max} = Maximum permitted velocity, m/sec (ft/sec)

 H_{T} = Net heating value of the gas being combusted, MJ/scm (Btu/scf).

 $K_1 = 8.706 \text{ m/sec} \text{ (metric units)}$

= 28.56 ft/sec (English units)

 $K_2 = 0.7084 \text{ m}_4 / (\text{MJ-sec}) \text{ (metric units)}$

= 0.087 ft⁴ /(Btu-sec) (English units)

[Reg.19.304 and 40 C.F.R. § 61.245 (e)(1-5)

Recordkeeping requirements

- IB 31. The permittee shall show compliance with the recordkeeping requirements of this subpart. [Reg.19.304 and 40 C.F.R. § 61.246(a)(1)]
 - a. An owner or operator of more than one process unit subject to the provisions of this subpart may comply with the recordkeeping requirements for these process units in one recordkeeping system if the system identifies each record by each process unit. [Reg.19.304 and 40 C.F.R. § 61.246 (a)(2)]
 - b. When each leak is detected as specified in §§ 61.242-2, 61.242-3, 61.242-7, 61.242-8, and 61.135, the following requirements apply:
 - i. A weatherproof and readily visible identification, marked with the equipment identification number, shall be attached to the leaking equipment.
 - ii. The identification on a valve may be removed after it has been monitored for 2 successive months as specified in § 61.242-7(c) and no leak has been detected during those 2 months

iii. The identification on equipment except on a valve, may be removed after it has been repaired.

[Reg.19.304 and 40 C.F.R. § 61.246 (b)(1-3)]

- c. When each leak is detected as specified in §§ 61.242-2, 61.242-3. 61.242-7, 61.242-8, and 61.135, the following information shall be recorded in a log and shall be kept for 2 years in a readily accessible location:
 - i. The instrument and operator identification numbers and the equipment identification number.
 - ii. The date the leak was detected and the dates of each attempt to repair the leak.
 - iii. Repair methods applied in each attempt to repair the leak.
 - iv. "Above 10,000" if the maximum instrument reading measured by the methods specified in § 61.245(a) after each repair attempt is equal to or greater than 10,000 ppm.
 - v. "Repair delayed" and the reason for the delay if a leak is not repaired within 15 calendar days after discovery of the leak.
 - vi. The signature of the owner or operator (or designate) whose decision it was that repair could not be effected without a process shutdown.
 - vii. The expected date of successful repair of the leak if a leak is not repaired within 15 days.
 - viii. Dates of process unit shutdowns that occur while the equipment is unrepaired.
 - ix. The date of successful repair of the leak.

[Reg.19.304 and 40 C.F.R. § 61.246(c)(1-9)]

- d. The following information pertaining to the design requirements for closed-vent systems and control devices described in § 61.242-11 shall be recorded and kept in a readily accessible location:
 - i. Detailed schematics, design specifications, and piping and instrumentation diagrams.
 - ii. The dates and descriptions of any changes in the design specifications.
 - iii. A description of the parameter or parameters monitored, as required in § 61.242-11(e), to ensure that control devices are operated and maintained in conformance with their design and an explanation of why that parameter (or parameters) was selected for the monitoring.
 - iv. Periods when the closed-vent systems and control devices required in §§ 61.242-2, 61.242-3, 61.242-4, 61.242-5 and 61.242-9 are not operated as designed, including periods when a flare pilot light does not have a flame.

v. Dates of startups and shutdowns of the closed-vent systems and control devices required in §§ 61.242-2, 61.242-3, 61.242-4, 61.242-5 and 61.242-9.

[Reg.19.304 and 40 C.F.R. § 61.246(d)(1-5)]

- e. The following information pertaining to all equipment to which a standard applies shall be recorded in a log that is kept in a readily accessible location:
 - i. A list of identification numbers for equipment subject to the requirements of this subpart. 2)(i) A list of identification numbers for equipment that the owner or operator elects to designate for no detectable emissions as indicated by an instrument reading of less than 500 ppm above background.
 - ii. The designation of this equipment for no detectable emissions shall be signed by the owner or operator.
 - iii. A list of equipment identification numbers for pressure relief devices required to comply with § 61.242-4(a).
 - iv. The dates of each compliance test required in §§ 61.242-2(e), 61.242-3(i), 61.242-4, 61.242-7(f), and 61.135(g).
 - v. The background level measured during each compliance test.
 - vi. The maximum instrument reading measured at the equipment during each compliance test.
 - vii. A list of identification numbers for equipment in vacuum service.

[Reg.19.304 and 40 C.F.R. § 61.246 (e)(1-5)]

- f. The following information pertaining to all valves subject to the requirements of § 61.242-7(g) and (h) and to all pumps subject to the requirements of § 61.242-2(g) shall be recorded in a log that is kept in a readily accessible location:
 - i. A list of identification numbers for valves and pumps that are designated as unsafe-to-monitor, an explanation for each valve or pump stating why the valve or pump is unsafe-to-monitor, and the plan for monitoring each valve or pump.
 - ii. A list of identification numbers for valves that are designated as difficultto-monitor, an explanation for each valve stating why the valve is difficult-to-monitor, and the schedule for monitoring each valve.

[Reg.19.304 and 40 C.F.R. § 61.246 (f)(1&2)]

- g. The following information shall be recorded for valves complying with § 61.243-2:
 - i. A schedule of monitoring.
 - ii. The percent of valves found leaking during each monitoring period.

[Reg.19.304 and 40 C.F.R. § 61.246 (g)(1&2)]

- h. The following information shall be recorded in a log that is kept in a readily accessible location:
 - i. Design criterion required in §§ 61.242-2(d)(5), 61.242-3(e)(2), and 61.135(e)(4) and an explanation of the design criterion; and
 - ii. Any changes to this criterion and the reasons for the changes.

[Reg.19.304 and 40 C.F.R. § 61.246 (h)(1&2)]

- i. The following information shall be recorded in a log that is kept in a readily accessible location for use in determining exemptions as provided in the applicability section of this subpart and other specific subparts:
 - i. An analysis demonstrating the design capacity of the affected facility,
 - ii. A statement listing the feed or raw materials and products from the affected facilities and an analysis demonstrating whether these chemicals are heavy liquids or beverage alcohol, and
 - iii. An analysis demonstrating that equipment is not in VHAP service.

[Reg.19.304 and 40 C.F.R. § 61.246 (i)(1-3)]

j. Information and data used to demonstrate that a piece of equipment is not in VHAP service shall be recorded in a log that is kept in a readily accessible location [40 CFR §§61.246 (j)]

Reporting requirements

- IB 32. The permittee shall comply with the reporting requirements of this section. [Reg.19.304 and 40 C.F.R. § 61.247]
 - a. An owner or operator of any piece of equipment to which this subpart applies shall
 - i. Submit a statement in writing notifying the Administrator that the requirements of §§ 61.242, 61.245, 61.246, and 61.247 are being implemented.
 - ii. In the case of an existing source or a new source which has an initial startup date preceding the effective date, the statement is to be submitted within 90 days of the effective date, unless a waiver of compliance is granted under § 61.11, along with the information required under § 61.10. If a waiver of compliance is granted, the statement is to be submitted on a date scheduled by the Administrator.
 - iii. In the case of new sources which did not have an initial startup date preceding December 14, 2000, the statement required under paragraph (a)(1) of this section shall be submitted with the application for approval of construction, as described in § 61.07.

- iv. For owners and operators complying with 40 CFR part 65, subpart C or F, the statement required under paragraph (a)(1) of this section shall notify the Administrator that the requirements of 40 CFR part 65, subpart C or F, are being implemented.
- v. The statement is to contain the following information for each source:
 - 1. Equipment identification number and process unit identification.
 - 2. Type of equipment (for example, a pump or pipeline valve).
 - 3. Percent by weight VHAP in the fluid at the equipment.
 - 4. Process fluid state at the equipment (gas/vapor or liquid).
 - 5. Method of compliance with the standard (for example, "monthly leak detection and repair" or "equipped with dual mechanical seals").

[Reg.19.304 and 40 C.F.R. § 61.247 (a)(1-5)]

- b. A report shall be submitted to the Administrator semiannually starting 6 months after the initial report required in paragraph (a) of this section, that includes the following information:
 - i. Process unit identification.
 - ii. For each month during the semiannual reporting period:
 - 1. Number of valves for which leaks were detected as described in § 61.242-7(b) of § 61.243-2.
 - 2. Number of valves for which leaks were not repaired as required in § 61.242-7(d).
 - 3. Number of pumps for which leaks were detected as described in § 61.242-2 (b) and (d)(6).
 - 4. Number of pumps for which leaks were not repaired as required in § 61.242-2 (c) and (d)(6).
 - 5. Number of compressors for which leaks were detected as described in § 61.242-3(f).
 - 6. Number of compressors for which leaks were not repaired as required in § 61.242-3(g).
 - 7. The facts that explain any delay of repairs and, where appropriate, why a process unit shutdown was technically infeasible.
 - iii. Dates of process unit shutdowns which occurred within the semiannual reporting period.

- iv. Revisions to items reported according to paragraph (a) if changes have occurred since the initial report or subsequent revisions to the initial report. NOTE: Compliance with the requirements of § 61.10(c) is not required for revisions documented under this paragraph.
- v. The results of all performance tests and monitoring to determine compliance with no detectable emissions and with §§ 61.243-1 and 61.243-2 conducted within the semiannual reporting period.

[Reg.19.304 and 40 C.F.R. § 61.247 (b)(1-5)]

- c. In the first report submitted as required in paragraph (a) of this section, the report shall include a reporting schedule stating the months that semiannual reports shall be submitted. Subsequent reports shall be submitted according to that schedule, unless a revised schedule has been submitted in a previous semiannual report. [Reg.19.304 and 40 C.F.R. § 61.247 (c)]
- d. An owner or operator electing to comply with the provisions of §§ 61.243-1 and 61.243-2 shall notify the Administrator of the alternative standard selected 90 days before implementing either of the provisions. [Reg.19.304 and 40 C.F.R. § 61.247 (d)]
- e. An application for approval of construction or modification, §§ 61.05(a) and 61.07, will not be required if
 - i. (1) The new source complies with the standard, § 61.242;
 - ii. (2) The new source is not part of the construction of a process unit; and
 - iii. (3) In the next semiannual report required by paragraph (b) of this section, the information in paragraph (a)(5) of this section is reported.

[Reg.19.304 and 40 C.F.R. § 61.247 (e)]

- f. For owners or operators choosing to comply with 40 CFR part 65, subpart C or F, an application for approval of construction or modification, as required under §§ 61.05 and 61.07 will not be required if:
 - i. The new source complies with 40 CFR 65.106 through 65.115 and with 40 CFR part 65, subpart C, for surge control vessels and bottoms receivers;
 - ii. The new source is not part of the construction of a process unit; and
 - iii. In the next semiannual report required by 40 CFR 65.120(b) and 65.48(b), the information in paragraph (a)(5) of this section is reported.

[40 CFR§61.247 (f)(1-3)]

40 CFR Part 61 Subpart J – National Emission Standard for Equipment Leaks (Fugitive Emission Sources of Benzene

- IB 33. The permittee shall comply with all applicable requirements in 40 CFR Part 61, Subpart J and Subpart V at all applicable sources in the DIPB process. The provisions of this subpart apply to each of the following sources that are intended to operate in benzene service: pumps, compressors, pressure relief devices, sampling connection systems, open-ended lines, valves, flanges, and other connectors, product accumulator vessels, and control devices or systems required by these subparts. [Reg.19.304 and 40 C.F.R. § 61, Subpart J and Subpart V]
- IB 34. NESHAP J National Emission Standard for Equipment Leaks (Fugitive Emission Sources of Benzene), applies to equipment in benzene service including: pumps, valves, flanges, compressors, pressure relief devices, sampling connections, open-ended valves or lines, other connectors, product accumulation vessels, and control devices or systems required by the subpart. [Reg.19.304 and 40 C.F.R. § 61.110(a)]
- IB 35. Each owner or operator subject to the provisions of this subpart shall comply with the provisions of NESHAP, Subpart V. [Reg.19.304 and 40 C.F.R. § 61.112(a)]
- IB 36. The owner/operator may elect to comply with the provisions of 61.243-1 and 61.243-2. [Reg.19.304 and 40 C.F.R. § 61.112(b)]
- IB 37. The permittee shall comply with all applicable parts of sections §61.240 through §61.247. [Reg.19.304 and 40 C.F.R. § 61.240 through §61.247]

40 CFR Part 61 Subpart Y - National Emission Standard for Benzene Storage Vessels

- IB 38. NESHAP Y National Emission Standard for Benzene Storage Vessels, defines applicability and designation of sources and defines exemptions. The condition applies to Tank T-210 which is vent to the DIPB flare (5N03-54). [Reg.19.304 and 40 C.F.R. § 61.270]
- IB 39. The storage vessel shall be equipped with a closed vent system and flare control device meeting the specifications of §61.271(d). [Reg.19.304 and 40 C.F.R. § 61.271(c)]
- IB 40. The closed vent system and flare shall meet the requirements as specified for general control devices in 40 CFR §60.18(e) and (f). [Reg.19.304 and 40 C.F.R. § 61.271(d)]
- IB 41. The specifications and requirements of §61.271(c)(1) and (2) do not apply during a control system malfunction. [Reg.19.304 and 40 C.F.R. § 61.271(c)(4)]
- IB 42. Excess emissions shall be reported as specified in §61.275(e). [40 CFR Part §61.275(e)]
- IB 43. The owner/operator shall keep copies of all reports and records required by §61.276(a). [Reg.19.304 and 40 C.F.R. § 61.276(a)]
- IB 44. The permittee shall keep readily assessable records showing the dimensions of the storage vessel and an analysis of the capacity. Each storage vessel with a design capacity of less than 10,000 gallons is subject to no provisions of this subpart other than this requirement. [Reg.19.304 and 40 C.F.R. § 61.276(b)]

Storage Tanks and Miscellaneous Sources 5N03TK-01, 6N01-02, and 6N01-03

Source Description

FutureFuel Chemical Company is a manufacturer of organic chemical intermediates. The primary business opportunities for this facility are producing chemicals that are put into the marketplace quickly. Many different chemicals can be manufactured; therefore many tanks are needed for raw material, intermediate, and product storage. Emissions from storage tanks that vent to the atmosphere and other miscellaneous sources are identified in the STMS section of the permit. With the exception of a fume scrubber TF-5FS on TF-5, there are no specific controls on these tanks other than conservation vents.

Source Number	TANK ID	AREA	Applicable Federal Regulation
5N01-41	TF-7	BULK	
5N03-39	TF-10	BULK	
5N03-45	TF-12	BULK	
5N01-27	TFV-4	BULK	
5N01-23	TFV-5	BULK	
5N01-26	TFV-6	BULK	
5N01-34	TFS-5	BULK	
5N01-37	TFS-10	BULK	
5N03-64	TFS-79	BULK	
5N01-48	WG-1	UTILITIES	
5N01-49	CG-1	UTILITIES	
4P94-12	PR-56A	4P-99	
4P94-13	PR-56B	4P-99	
4Q01-10	T-242	4Q	NSPS Kb
4Q01-11	T-243	4Q	NSPS Kb
4Q01-01	TFB-1	4Q	
4Q01-02	TFB-22	4Q	
4Q01-03	TFB-10	4Q	
4Q01-04	TFB-11	4Q	
4Q01-05	TFB-12	4Q	
4Q01-06	TFB-20	4Q	
4Q01-07	TFB-21	4Q	
4Q01-08	TFB-30	4Q	
4Q01-09	TFB-31	4Q	
5N01-36	TFS-7	BULK	
5N01-38	TFS-9	BULK	
5N03-18	PBV-50	DIPB	
5N01-42	TF-6	BULK	
5N03-50	PA-50	5P-99	

Source Number	TANK ID	AREA	Applicable Federal Regulation
5N03-40	TF-11	BULK	
5N01-32	TFS-1	BULK	
5N01-31	TFS-2	BULK	
5Q99-01	T-273	5Q	
5N01-33	TFS-6	BULK	
5N03-38	TF-8	BULK	
5P94-01	TF-5FS	BULK	NESHAP FFFF (MON MACT)

Specific Conditions

STMS 1. The permittee shall not exceed the emission rates set forth in the following table. These rates are based on maximum physical capacity of the equipment and Specific Condition STMS 3. [Reg.19.501 *et seq.* and 40 C.F.R. § 52 Subpart E]

SN	Description	Pollutant	lb/hr	tpy
6N01-02	Diesel Tank	VOC	0.1	0.4
6N01-03	Gas Tank	VOC	200.0	0.8
5N03TK-01	Process Tanks (36 Tanks)	VOC	4.0	17.5

STMS 2. The permittee shall not exceed the emission rates set forth in the following table. The facility shall show compliance with the facility total ton/yr limits using the procedures outlined in Plantwide Conditions 12 through 16. Hourly Plantwide Hazardous Air Pollutant emissions are limited by Plantwide Conditions 16. Additional HAP limitations are included in Plantwide Condition 13. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

SN	Description	Pollutant	lb/hr	tpy
6N01-02	Diesel Tank	Organic Pollutants**	***	0.4
6N01-03	Gas Tank	Organic Pollutants**	***	0.8
5N03TK-01	Process Tanks (36 Tanks)	Organic Pollutants** Inorganics *	*** 0.3	17.5 1.2

*Inorganics are considered to be non-organic Hazardous Air Pollutants

**Organic Pollutants are considered to be VOC, all organic HAPs, and organic Air Contaminates.

STMS 3. Emissions from Storage Tanks and Misc. Sources areas shall be recalculated monthly, and shall be based upon a 12-month rolling total. The records shall be updated by the last day of the month following the recorded 12-month period, and shall be kept on site and made available for inspection upon request. [Reg.19.705 and 40 C.F.R. § 52 Subpart E, and Reg.18.1004 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]

5N07 Production Facility

Source Description

The 5N07 production facility contains multi-purpose production equipment which may produce a variety of chemicals including biofuel. Biodiesel is the primary product from this facility. The contained or captured vapors from the equipment in this facility are vented through a collection system to the RTO (SN 5N09-01) units via a common duct. VOCs are destroyed by combustion.

Fugitive emissions are estimated based on components, monitoring data, and published emission leak factors. Fugitive emissions are designated as source OCI-FUG.

Biodiesel Washing

The reaction mixture from the Biodiesel production facilities is fed continuously to equipment where methanol, the catalyst and glycerin are removed by washing with water. After most of the water is removed, the washed Biodiesel flows to the Biodiesel Drying equipment. The used water streams flow to Methanol Recovery and Glycerin Recovery equipment.

Biodiesel Drying

Residual water and methanol are removed from the Biodiesel using heat. Vacuum is applied if needed. The dry Biodiesel is cooled and sent to storage.

Biodiesel Storage

Product from the Biodiesel Drying equipment flows to accumulation tanks. When an accumulation tank fills the contents are analyzed prior to transferring to sales tanks. Product from the sales tanks is loaded into tank trucks or railcars for sale.

Methanol Recovery and Storage

Emissions from methanol tank T-242 are accounted for in tank bubble 5N03TK-01. Methanolcontaining streams from the Biodiesel reactors and the aqueous stream from the Biodiesel Washing equipment are fed to the Methanol Recovery and Storage equipment. Methanol is separated from the other components (primarily water and glycerin) by distillation. The methanol is stored for recycle to the biodiesel reactors.

Glycerin Recovery and Storage

Glycerin is recovered and refined in the 4P (Solvent Recovery) production area. Glycerin streams from the Biodiesel reactors and the aqueous stream from the Biodiesel Washing equipment are fed to the Glycerin Recovery and Storage equipment. Methanol is separated from the glycerin by distillation and the glycerin is further refined to improve purity.

Specific Conditions

BD 1. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by BD 3. [Reg.19.501 *et seq.* and 40 C.F.R. § 52 Subpart E]

PES #	Description	Pollutant	lb/hr	tpy
5N07	Biodiesel Production	VOC	2.9	12.4

BD 2. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by BD 3. The facility shall show compliance with the facility total ton/yr limits using the procedures outlined in Plantwide Conditions 12 through 16. Hourly Plantwide Hazardous Air Pollutant emissions are limited by Plantwide Conditions 16. Additional HAP limitations are included in Plantwide Condition 13. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

PES #	Description	Pollutant	lb/hr	tpy
5N07	Biodiesel Production	Organic Pollutants**	2.9	12.4

**Organic Pollutants are considered to be VOC, all organic HAPs, and organic Air Contaminates.

- BD 3. The permittee is limited to 250 million gallons of biodiesel production. [Reg.18.1004 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]
- BD 4. The permittee shall maintain records to demonstrate compliance with the limits in BD 3. Emissions from biodiesel production shall be recalculated monthly and shall be based upon a 12-month rolling total. The records shall be updated by the last day of the month following the recorded 12-month period and shall be kept on site and made available for inspection upon request. [Reg.19.705 and 40 C.F.R. § 52 Subpart E and Reg.18.1004 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

Aldehyde Processing Section 4P05-01, 4P05-02, 4P05-03, 4PSR-FUG

Source Description

Raw materials are unloaded into storage tanks T-271, T-272, and TFS-75. TFS-75 vents to the RTO. Raw materials are transferred to the process as needed. Both columns and tanks vent to the RTOs (5N09-01).

The aldehyde process reactors are periodically cleaned and the vent gas is routed to a water scrubber, SV-03. The vent stream from this source discharges to the RTO. The water stream is routed to the wastewater treatment facilities. The reactors are heated with two 5 MMBTU/hr hot oil system, SN 4P05-01 and SN 4P05-03. The hot oil system SN 4P05-01 is designed to burn natural gas, fuel oil, biodiesel, and process vent streams. Hot oil system SN 4P05-03 is designed to burn natural gas and process vent streams.

Products are refined in distillation column SB-02 and extraction column SX-03. The distillation column and the extraction column both vent back to the hot oil system or the RTOs. The refined product is transferred to three lot tanks, VC-PT-01, VC-PT-02, and VC-PT-03. These tanks are equipped with vapor balancing and do not vent to the atmosphere. VC-ST-01 shutdown tank vents to the hot oil system for thermal recovery and is not an emission source itself. Tank WDT-03, a waste tank, is located in the Utilities section and is connected to the coal-fired boiler closed vent system and control device. VC-PT-03, VC-PT-01, and VC-PT-02 are all subject to NSPS Kb.

The emission points for the Aldehyde Processing Section are the hot oil system (4P05-01), hot oil system (4P05-03), equipment fugitives (4PSR-FUG), and the RTO (5N09-01).

Specific Conditions

AP 1. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by AP 3. [Reg.19.501 *et seq.* and 40 C.F.R. § 52 Subpart E]

SN #	Source Description	Polluant	lb/hr	tpy
		PM ₁₀	0.2	0.9
		SO_2	1.1	4.9
4P05-01	Hot Oil System	VOC	4.0	6.8
		CO	1.5	6.4
	NO _x	3.2	13.8	
		PM ₁₀	0.2	0.6
		SO_2	0.1	0.1
4P05-03	Hot Oil System #2	VOC	0.8	3.4
		CO	1.5	6.4
		NO _x	0.9	3.8

SN	#	Source Description	Polluant	lb/hr	tpy
4PSR-	FUG	Aldehyde Processing Fugitives	VOC	0.6	2.5

AP 2. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by AP 3. The facility shall show compliance with the facility total ton/yr limits using the procedures outlined in Plantwide Conditions 12 through 16. Hourly Plantwide Hazardous Air Pollutant emissions are limited by Plantwide Conditions 16. Additional HAP limitations are included in Plantwide Condition 13. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

SN #	Source Description	Polluant	lb/hr	tpy
4005-01		PM	0.2	0.9
4P05-01	Hot Oil System	Organic Pollutants**	4.0	6.8
4005.02		РМ	0.2	0.6
4P05-03	Hot Oil System #2	Organic Pollutants**	0.8	3.4
4PSR-FUG	Aldehyde Processing Fugitives	Organic Pollutants**	0.6	2.5

**Organic Pollutants are considered to be VOC, all organic HAPs, and organic Air Contaminates.

- AP 3. The permittee is limited to no more than 75.0 million pounds of vinyl compound products from the Aldehyde Processing Section per 12-month rolling total. [Reg.19.705, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 70.6]
- AP 4. Visible emissions from the Hot Oil System shall not exceed 5% opacity except during periods of fuel oil usage for SN 4P05-01, which the permittee is allowed opacity of 20% opacity. [Reg.18.501 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]
- AP 5. The permittee shall conduct weekly visual inspections for 4P05-01 and 4P05-03, for all fuels except fuel oil, for possible emissions using EPA Method 22 and monthly observations using EPA Method 9. In the event that fuel oil usage exceeds one week at 4P05-01, the permittee shall perform weekly observations of the Hot Oil system using EPA Method 9. The permittee shall record the presence of any excessive visible emissions and the subsequent actions taken to correct the exceedance. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]
- AP 6. The permittee shall keep monthly records of the amount of vinyl compounds produced via Aldehyde Processing equipment to demonstrate compliance with the limits in Specific Condition AP 3. These records shall be kept onsite and made available upon request. [Reg.19.705, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 70.6]

- AP 7. The permittee shall maintain a scrubber liquor flow rate in scrubber SV-03 as outlined in most current version of the Facility Operating Plan. [Reg.19.303 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]
- AP 8. The permittee shall keep daily records of the liquor flow rate at scrubber SV-03. These records shall be kept on site and made available upon request. [Reg.19.705 and 40 C.F.R. § 52 Subpart E]
- AP 9. The permittee shall equip and maintain the following tanks with the control equipment specified. [Reg.19.303 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]

Tank ID	Tank Size (gallons)	Control Device
TFS-51	42,302	RTO (5N09-01)
TFS-75	42,302	RTO (5N09-01)
TF-14	13,250	Vapor Balancing
TF-8	11,845	Vapor Balancing
VC-PT-01	29,660	Hot Oil System (4P05-01)
VC-PT-02	29,660	Hot Oil System (4P05-01)
VC-PT-03	45,000	Hot Oil System (4P05-01)
T-271	24,000	Vapor Balancing
T-272	24,000	Vapor Balancing

40 CFR Part 63 Subpart DDDDD—National Emission Standards for Hazardous Air Pollutants for Major Sources: Industrial, Commercial, and Institutional Boilers and Process Heaters

- AP 10. Source SN-4P05-01 and 4P05-03 are subject to the provisions of 40 CFR Part 63, Subpart DDDDD – National Emission Standards for Hazardous Air Pollutants for Major Sources: Industrial, Commercial, and Institutional Boilers and Process Heaters. The applicable provisions of this subpart include, but are not limited to, the items found in Specific Conditions AP-11 through AP-37. [Reg.19.304 and 40 C.F.R. §§ 63.7480, 63.7485, and 63.7490(a)]
- AP 11. The permittee must meet the requirements in paragraphs (a)(1) through (3) of §63.7500, except as provided in paragraphs (d) and (e) of §63.7500. The permittee must meet these requirements at all times the affected unit is operating, except as provided in paragraph (f) of §63.7500. [Reg.19.304 and 40 C.F.R. § 63.7500(a)]

- AP 12. The permittee must meet each emission limit and work practice standard in Table 3, and 11 through 13 to 40 CFR Part 63, Subpart DDDDD that applies to the permittee's boiler or process heater, for each boiler or process heater at the permittee's source, except as provided under §63.7522. [Reg.19.304 and 40 C.F.R. § 63.7500(a)(1)]
- AP 13. At all times, the permittee must operate and maintain any affected source (as defined in §63.7490), including associated air pollution control equipment and monitoring equipment, in a manner consistent with safety and good air pollution control practices for minimizing emissions. Determination of whether such operation and maintenance procedures are being used will be based on information available to the Administrator that may include, but is not limited to, monitoring results, review of operation and maintenance procedures, review of operation and maintenance records, and inspection of the source. [Reg.19.304 and 40 C.F.R. § 63.7500(a)(3)]
- AP 14. Boilers and process heaters with a heat input capacity of less than or equal to 5 million Btu per hour in the units designed to burn gas 2 (other) fuels subcategory or units designed to burn light liquid fuels subcategory must complete a tune-up every 5 years as specified in §63.7540. [Reg.19.304 and 40 C.F.R. § 63.7500(d)]
- AP 15. Boilers and process heaters in the units designed to burn gas 1 fuels subcategory with a heat input capacity of less than or equal to 5 million Btu per hour must complete a tuneup every 5 years as specified in §63.7540. Boilers and process heaters in the units designed to burn gas 1 fuels subcategory are not subject to the emission limits in Tables 1 and 2 or 11 through 13 to Subpart DDDDD, or the operating limits in Table 4 to Subpart DDDDD. [Reg.19.304 and 40 C.F.R. § 63.7500(e)]
- AP 16. These standards apply at all times the affected unit is operating, except during periods of startup and shutdown during which time the permittee must comply only with Table 3 to Subpart DDDDD. [Reg.19.304 and 40 C.F.R. § 63.7500(f)]
- AP 17. In response to an action to enforce the standards set forth in §63.7500 the permittee may assert an affirmative defense to a claim for civil penalties for violations of such standards that are caused by malfunction, as defined at §63.2. Appropriate penalties may be assessed if the permittee fail to meet the permittee's burden of proving all of the requirements in the affirmative defense. The affirmative defense shall not be available for claims for injunctive relief. [Reg.19.304 and 40 C.F.R. § 63.7501]
- AP 18. The permittee must be in compliance with the emission limits, work practice standards, and operating limits in this subpart. These limits apply to the permittee at all times the affected unit is operating except for the periods noted in §63.7500(f). [Reg.19.304 and 40 C.F.R. § 63.7505(a)]
- AP 19. For existing affected sources (as defined in §63.7490), the permittee must complete the initial compliance demonstration, as specified in paragraphs (a) through (d) of §63.7510, no later than 180 days after the compliance date that is specified for the permittee's

source in §63.7495 and according to the applicable provisions in §63.7(a)(2) as cited in Table 10 to Subpart DDDDD, except as specified in paragraph (j) of §63.7510. The permittee must complete an initial tune-up by following the procedures described in §63.7540(a)(10)(i) through (vi) no later than the compliance date specified in §63.7495, except as specified in paragraph (j) of §63.7510. The permittee must complete the one-time energy assessment specified in Table 3 to Subpart DDDDD no later than the compliance date specified in §63.7510. [Reg.19.304 and 40 C.F.R. § 63.7510(e)]

- AP 20. If the permittee is required to meet an applicable tune-up work practice standard, the permittee must conduct an annual, biennial, or 5-year performance tune-up according to §63.7540(a)(10), (11), or (12), respectively. Each annual tune-up specified in §63.7540(a)(10) must be no more than 13 months after the previous tune-up. Each biennial tune-up specified in §63.7540(a)(11) must be conducted no more than 25 months after the previous tune-up. Each 5-year tune-up specified in §63.7540(a)(12) must be conducted no more than 61 months after the previous tune-up. For a new or reconstructed affected source (as defined in §63.7490), the first annual, biennial, or 5-year tune-up must be no later than 13 months, 25 months, or 61 months, respectively, after the initial startup of the new or reconstructed affected source. [Reg.19.304 and 40 C.F.R. § 63.7515(d)]
- AP 21. For affected sources (as defined in §63.7490) that have not operated since the previous compliance demonstration and more than one year has passed since the previous compliance demonstration, the permittee must complete the subsequent compliance demonstration, if subject to the emission limits in Tables 1, 2, or 11 through 13 to Subpart DDDDD, no later than 180 days after the re-start of the affected source and according to the applicable provisions in §63.7(a)(2) as cited in Table 10 to Subpart DDDDD. The permittee must complete a subsequent tune-up by following the procedures described in §63.7540(a)(10)(i) through (vi) and the schedule described in §63.7540(a)(13) for units that are not operating at the time of their scheduled tune-up. [Reg.19.304 and 40 C.F.R. § 63.7515(g)]
- AP 22. If the permittee owns or operates an existing unit with a heat input capacity of less than 10 million Btu per hour or a unit in the unit designed to burn gas 1 subcategory, the permittee must submit a signed statement in the Notification of Compliance Status report that indicates that the permittee conducted a tune-up of the unit. [Reg.19.304 and 40 C.F.R. § 63.7530(d)]
- AP 23. The permittee must include with the Notification of Compliance Status a signed certification that the energy assessment was completed according to Table 3 to Subpart DDDDD and is an accurate depiction of the permittee's facility at the time of the assessment. [Reg.19.304 and 40 C.F.R. § 63.7530(e)]
- AP 24. The permittee must submit the Notification of Compliance Status containing the results of the initial compliance demonstration according to the requirements in §63.7545(e). [Reg.19.304 and 40 C.F.R. § 63.7530(f)]

- AP 25. The permittee must demonstrate continuous compliance with each emission limit in Tables 1 and 2 or 11 through 13 to Subpart DDDDD, the work practice standards in Table 3 to Subpart DDDDD, and the operating limits in Table 4 to Subpart DDDDD that applies to the permittee according to the methods specified in Table 8 to Subpart DDDDD and paragraphs (a)(1) through (19) of §63.7540. [Reg.19.304 and 40 C.F.R. § 63.7540(a)]
- AP 26. If the permittee's boiler or process heater has a continuous oxygen trim system that maintains an optimum air to fuel ratio, or a heat input capacity of less than or equal to 5 million Btu per hour and the unit is in the units designed to burn gas 1; units designed to burn gas 2 (other); or units designed to burn light liquid subcategories, or meets the definition of limited-use boiler or process heater in §63.7575, the permittee must conduct a tune-up of the boiler or process heater every 5 years as specified in paragraphs (a)(10)(i) through (vi) of §63.7540 to demonstrate continuous compliance. The permittee may delay the burner inspection specified in paragraph (a)(10)(i) of §63.7540 until the next scheduled or unscheduled unit shutdown, but the permittee must inspect each burner at least once every 72 months. [Reg.19.304 and 40 C.F.R. § 63.7540(a)(12)]
- AP 27. If the unit is not operating on the required date for a tune-up, the tune-up must be conducted within 30 calendar days of startup. [Reg.19.304 and 40 C.F.R. § 63.7540(a)(13)]
- AP 28. The permittee must submit to the Administrator all of the notifications in §§63.7(b) and (c), 63.8(e), (f)(4) and (6), and 63.9(b) through (h) that apply to the permittee by the dates specified. [Reg.19.304 and 40 C.F.R. § 63.7545(a)]
- AP 29. If the permittee is required to conduct an initial compliance demonstration as specified in §63.7530, the permittee must submit a Notification of Compliance Status according to §63.9(h)(2)(ii). For the initial compliance demonstration for each boiler or process heater, the permittee must submit the Notification of Compliance Status, including all performance test results and fuel analyses, before the close of business on the 60th day following the completion of all performance test and/or other initial compliance demonstrations for all boiler or process heaters at the facility according to §63.10(d)(2). The Notification of Compliance Status report must contain all the information specified in paragraphs (e)(1) through (8), as applicable. If the permittee is not required to conduct an initial compliance demonstration as specified in §63.7530(a), the Notification of Compliance Status must only contain the information specified in paragraphs (e)(1) and (8).
 - a. A signed certification that the permittee have met all applicable emission limits and work practice standards.
 - b. If the permittee had a deviation from any emission limit, work practice standard, or operating limit, the permittee must also submit a description of the deviation, the duration of the deviation, and the corrective action taken in the Notification of Compliance Status report.

- c. In addition to the information required in §63.9(h)(2), the permittee's notification of compliance status must include the following certification(s) of compliance, as applicable, and signed by a responsible official:
 - i. "This facility complies with the required initial tune-up according to the procedures in §63.7540(a)(10)(i) through (vi)."
 - ii. "This facility has had an energy assessment performed according to §63.7530(e)."

[Reg.19.304 and 40 C.F.R. § 63.7545(e)(6-8)]

- AP 30. The permittee must submit each report in Table 9 to Subpart DDDDD that applies to the permittee. [Reg.19.304 and 40 C.F.R. § 63.7550(a)]
- AP 31. Unless the EPA Administrator has approved a different schedule for submission of reports under §63.10(a), the permittee must submit each report, according to paragraph (h) of §63.7550, by the date in Table 9 to Subpart DDDDD and according to the requirements in paragraphs (b)(1) through (4) of §63.7550. For units that are subject only to a requirement to conduct an annual, biennial, or 5-year tune-up according to §63.7540(a)(10), (11), or (12), respectively, and not subject to emission limits or operating limits, the permittee may submit only an annual, biennial, or 5-year compliance report, as applicable, as specified in paragraphs (b)(1) through (4) of §63.7550, instead of a semi-annual compliance report.
 - a. The first compliance report must cover the period beginning on the compliance date that is specified for each boiler or process heater in §63.7495 and ending on July 31 or January 31, whichever date is the first date that occurs at least 180 days (or 1, 2, or 5 years, as applicable, if submitting an annual, biennial, or 5-year compliance report) after the compliance date that is specified for the permittee's source in §63.7495.
 - b. The first compliance report must be postmarked or submitted no later than July 31 or January 31, whichever date is the first date following the end of the first calendar half after the compliance date that is specified for each boiler or process heater in §63.7495. The first annual, biennial, or 5-year compliance report must be postmarked or submitted no later than January 31.
 - c. Each subsequent compliance report must cover the semiannual reporting period from January 1 through June 30 or the semiannual reporting period from July 1 through December 31. Annual, biennial, and 5-year compliance reports must cover the applicable 1-, 2-, or 5-year periods from January 1 to December 31.
 - d. Each subsequent compliance report must be postmarked or submitted no later than July 31 or January 31, whichever date is the first date following the end of the semiannual reporting period. Annual, biennial, and 5-year compliance reports must be postmarked or submitted no later than January 31.

[Reg.19.304 and 40 C.F.R. § 63.7550(b)(1-4)]

AP 32. A compliance report must contain the following information depending on how the facility chooses to comply with the limits set in this rule.

- a. If the facility is subject to a the requirements of a tune up they must submit a compliance report with the information in paragraphs (c)(5)(i) through (iv) and (xiv) of §63.7550.
 - [Reg.19.304 and 40 C.F.R. § 63.7550(c)(1)]
- AP 33. Company and Facility name and address.
 - a. Process unit information, emissions limitations, and operating parameter limitations.
 - b. Date of report and beginning and ending dates of the reporting period.
 - c. The total operating time during the reporting period.
 - d. Include the date of the most recent tune-up for each unit subject to only the requirement to conduct an annual, biennial, or 5-year tune-up according to §63.7540(a)(10), (11), or (12) respectively. Include the date of the most recent burner inspection if it was not done annually, biennially, or on a 5-year period and was delayed until the next scheduled or unscheduled unit shutdown. [Reg.19.304 and 40 C.F.R. § 63.7550(c)(5)(i-xiv)]
- AP 34. The permittee must keep records according to paragraphs (a)(1) and (2) of §63.7555.
 - a. A copy of each notification and report that the permittee submitted to comply with this subpart, including all documentation supporting any Initial Notification or Notification of Compliance Status or semiannual compliance report that the permittee submitted, according to the requirements in §63.10(b)(2)(xiv).
 - B. Records of performance tests, fuel analyses, or other compliance demonstrations and performance evaluations as required in §63.10(b)(2)(viii).
 [Reg.19.304 and 40 C.F.R. § 63.7555(a)(1-2)]
- AP 35. The permittee's records must be in a form suitable and readily available for expeditious review, according to §63.10(b)(1). [Reg.19.304 and 40 C.F.R. § 63.7560(a)]
- AP 36. As specified in §63.10(b)(1), the permittee must keep each record for 5 years following the date of each occurrence, measurement, maintenance, corrective action, report, or record. [Reg.19.304 and 40 C.F.R. § 63.7560(b)]
- AP 37. The permittee must keep each record on site, or they must be accessible from onsite (for example, through a computer network), for at least 2 years after the date of each occurrence, measurement, maintenance, corrective action, report, or record, according to §63.10(b)(1). The permittee can keep the records off site for the remaining 3 years. [Reg.19.304 and 40 C.F.R. § 63.7560(c)]

Anode Material Process

5M11-08, 5M11-09, and CP2-FUG

Source Description

The Anode Material Process (CP2), permitted as CP-2, consists of solids handling equipment, continuous stirred tank reactors, and dryers. Pelletized and granular material will be fed through a metering system into vessels where the material is heated and mixed. The formulated material is then dried and repackaged. All of this equipment is located inside the 5M11 building. Dust from all of the solids handling equipment is vented to a continuous dust control system (CDCS), 5M11-08. The CDCS consists of a baghouse for solid particle separation, a collection hopper, and an induced draft fan. A central vacuum cleaning system (CVC), 5M11-09 will be used to clean spills. All VOC vents will be routed to the RTOs, 5N09-01. Emissions from tanks and equipment containing VOC vent to the thermal oxidizer SN-5N09-01.

Neither CP2-T-003 or CP2-T-004 tanks are regulated by 40 CFR Subpart Kb (Standards of Performance for Volatile Organic Liquid Storage Vessels) per 40 CFR 60.11b(b). Both tanks contain liquids with vapor pressures less than 15.0 kPa. Both tanks are vented to the RTOs, 5N09-01. As a miscellaneous organic chemical manufacturing process unit (MCPU), this section is subject to 40 CFR Part 63 Subpart FFFF.

Specific Conditions

CP2 1. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by CP2 3. [Reg.19.501 *et seq.* and 40 C.F.R. § 52 Subpart E]

SN #	Description	Pollutant	lb/hr	tpy
5M11-08	Continuous Dust Collection System Baghouse	PM10	1.1	4.7
5M11-09	Central Vacuum System Baghouse	PM10	1.1	0.9
CP2-FUG	Anode Material Process VOC/HAP Fugitives	VOC	0.4	1.4

CP2 2. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by CP2 3. The facility shall show compliance with the facility total ton/yr limits using the procedures outlined in Plantwide Conditions 12 through 16. Hourly Plantwide Hazardous Air Pollutant emissions are limited by Plantwide Conditions 16. Additional HAP limitations are included in Plantwide Condition 13. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

SN #	Description	Pollutant	lb/hr	tpy
5M11-08	Continuous Dust Collection System Baghouse	PM	1.1	4.7
5M11-09	Central Vacuum System Baghouse	PM	1.1	0.9

CP2-FUG	Anode Material Process VOC/HAP Fugitives	Organic Pollutants**	0.4	1.4
	**Organic Pollutants are considered to be VOC, all	organic HAPs, and organic A	ir Contamina	ates.
CP2 3.	The permittee shall not produce more than 10.8 million pounds per consecutive 12- month of product (anode material) from the CP2 facility. [Reg.19.705, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 70.6 and Reg.18.1004 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]			2- rk.
CP2 4.	The permittee shall keep monthly records of the amount of anode material produced via CP2 equipment to demonstrate compliance with the limits in Specific Conditions CP2 3. The records shall be updated by the last day of the month and shall be kept on site and made available for inspection upon request. [Reg.19.705 and 40 C.F.R. § 52 Subpart E, and Reg.18.1004 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]			
CP2 5.	Visible emissions from the baghouses at 5M11-08 and 5M11-09 shall not exceed 5% opacity. [Reg.18.501 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]			
CP2 6.	The permittee shall conduct weekly visual inspections at 5M11-08 and 5M11-09 for possible emissions using EPA Method 22 and monthly observations of the system using EPA Method 9. The permittee shall record the presence of any excessive visible emissions and the subsequent actions taken to correct the exceedance. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]			
CP2 7.	The permittee shall continuously maintain a minimum pressure drop across the fabric filter at 5M11-08 and 5M11-09 as outlined in the most current version of the Facility Operating Plan. [Reg.19.705, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 70.6]			oric lity ode
CP2 8.	The permittee shall keep records on site of the pressure drop across 5M11-08 and 5M11-09. [Reg.19.705 and 40 C.F.R. § 52 Subpart E]			

Emergency Generators

Source Description

The facility has six emergency use generators onsite, two greater than 500 HP and four less than 500 HP. All of these generators except 4P-EG-01 are compression ignition generators and subject to applicable 40 CFR Part 60 Subpart IIII and/or 40 CFR Part 63 Subpart ZZZZ requirements. 4P-EG-01 is a spark ignition generator and is therefore subject to applicable 40 CFR Part 60 Subpart JJJJ and/or 40 CFR Part 63 Subpart ZZZZ requirements.

Engine	Size, BHP	Date of Manufacture	Stroke/Burn
5N01-WA-DIESEL	110	2002	4S/LB
7M04-HT-G01 DIESEL WASTE	234	1995	4S/LB
DISPOSAL PUMP			
7M04-HT-G04			
DIESEL WASTE	234	1995	4S/LB
DISPOSAL PUMP			
8M01 DIESEL FIRE	635	1076	28/PB
WATER PUMP	033	1970	25/ KD
6N02-EG Emergency	1400	2010	45
Generator	1490	2010	43
4P-EG-01	94	2022	4S/RB

Specific Conditions

EG 1. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by maintaining emission calculations for each emergency generator. The facility shall show compliance with the facility total ton/yr limits using the procedures outlined in Plantwide Condition 12. [Reg.19.501 *et seq.* and 40 C.F.R. § 52 Subpart E]

SN #	Description	Pollutant	lb/hr	tpy
		PM10	0.3	0.1
5N01-WA		SO_2	0.2	0.1
	Diesel Glycol Pump	VOC	0.3	0.1
		CO	0.8	0.1
		NO _x	3.8	0.2
7M04-HT-G01		PM10	0.6	0.1
		SO_2	0.5	0.1
	Diesel Waste Disposal Pump	VOC	0.6	0.1
		CO	1.7	$\begin{array}{c} 1 1 1 1 1 1 1 1$
		NO _x	8.0	0.4

SN #	Description	Pollutant	lb/hr	tpy	
		PM10	0.6	0.1	
7M04-HT-G04			SO_2	0.5	0.1
	Diesel Waste Disposal PumpVOC0.6CO1.7	VOC	0.6	0.1	
		0.1			
		NO _x	8.0	0.4	
		PM10	0.5	0.1	
		SO_2	5.7	0.3	
8M01	Diesel Fire Water Pump	VOC	0.5	0.1	
		CO 3.8 0.2			
		NO _x	16.8	tpy 0.1 0.1 0.1 0.1 0.1 0.4 0.1 0.2 0.8 0.1 0.4 0.1 0.2 0.8 0.1 0.4 0.1 0.4 0.1 0.4 0.1 0.4 0.1 0.4 0.1 0.1 0.1 0.1 0.1 0.1 0.1	
		PM ₁₀ 0.7 0.		0.1	
		SO_2	12.1	0.6	
6N02-EG	Emergency Diesel Generator	VOC	1.1	0.1 0.3 0.1 0.2 0.8 0.1 0.6 0.1 0.4 0.8 0.1 0.4 0.1 0.4 0.1 0.4 0.1 0.4 0.1 0.4 0.1 0.4 0.1 0.4 0.1 0.4 0.1 0.4 0.4 0.4 0.4 0.1 0.1 0.1 0.1 0.1 0.1 0.1 0.1	
		CO	8.5		
		NO _x	15.5	0.8	
		PM10	0.1	0.1	
	Emanganay Natural Gag	SO_2	0.1	0.1	
4P-EG-01	Emergency Natural Gas	VOC	0.1	$\begin{array}{c} 0.1 \\ 0.1 \\ 0.1 \\ 0.4 \\ 0.1 \\ 0.3 \\ 0.1 \\ 0.2 \\ 0.8 \\ 0.1 \\ 0.6 \\ 0.1 \\ 0.4 \\ 0.8 \\ 0.1 \\ 0.1 \\ 0.1 \\ 0.4 \\ 0.1 \\ 0.4 \\ 0.1 \\ \end{array}$	
	Generator	CO	3.2		
		NO _X	0.8	0.1	

EG 2. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall not exceed the emission rates set forth in the following table. Emission rates are based upon the maximum capacity of equipment. The facility shall show compliance with the facility total ton/yr limits using the procedures outlined in Plantwide Conditions 12 through 16. Hourly Plantwide Hazardous Air Pollutant emissions are limited by Plantwide Conditions 16. Additional HAP limitations are included in Plantwide Condition 13. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

SN #	Description	Pollutant	lb/hr	tpy
5N01-WA	Diagol Clysol Dump	PM	0.3	0.1
	Diesel Olycol Fullip	Organic Pollutants**	0.3	0.1
7M04-HT-G01	Diagal Wasta Dignagal Rump	PM	0.6	0.1
	Dieser waste Disposar Pump	Organic Pollutants**	0.6	0.1
7M04-HT-G04	Diagol Weste Dignogol Burn	PM	0.6	0.1 0.1 0.1 0.1 0.1
	Dieser waste Disposar Pump	Organic Pollutants**	0.6	0.1
8M01	Diagol Fire Water Dump	PM	0.5	0.1
	Diesei File water Fulip	Organic Pollutants**	0.5	0.1
6N02-EG	Emangeney Diagol Comentan	PM	0.7	0.1
	Emergency Dieser Generator	Organic Pollutants**	1.1	0.1
4P-EG-01	Emergency Natural Gas	PM	0.1	0.1
	Generator	Organic Pollutants**	0.1	0.1

**Organic Pollutants are considered to be VOC, all organic HAPs, and organic Air Contaminates.

- EG 3. Visible emissions from 4P-EG-01 shall not exceed 5% opacity, and all other generators shall not exceed 20% opacity. [Reg.19.503 and 40 C.F.R. § 52 Subpart E]
- EG 4. The permittee shall conduct daily visual inspections using EPA Method 9 once operation of any generator exceeds 24-hours of continuous use. The permittee shall record the presence of any excessive visible emissions and the subsequent actions taken to correct the exceedance. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]
- EG 5. The permittee shall not operate the emergency generator SN 6N02-EG in excess of 100 total hours (emergency and non-emergency) or Emergency Generator SN 4P-EG-01 in excess of 250 total hours (emergency and non-emergency) per calendar year in order to demonstrate compliance with the annual emission rate limits. Emergency operation in excess of these hours may be allowable but shall be reported and will be evaluated in accordance with Reg.19.602 and other applicable regulations. [Reg.19.705, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 70.6]
- EG 6. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #EG 6. The permittee shall update these records by the fifteenth day of the month following the month to which the records pertain. The calendar year totals and each individual month's data shall be maintained on-site, made available to Division of Environmental Quality personnel upon request, and submitted in accordance with General Provision #7. [Reg.19.705 and 40 C.F.R. § 52 Subpart E]

SN-8M01, Part 63 Subpart ZZZZ Conditions

- EG 7. As existing emergency stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, SN-8M01 RICE does not have to meet the requirements of this subpart and of subpart A of this part, including initial notification requirements. SN-8M01 does not operate and is not contractually obligated to be available for more than 15 hours per calendar year for the purposes specified in §63.6640(f)(2)(ii) and (iii). [Reg.19.304 and 40 C.F.R. § 63.6590(b)(3)(iii)]
- EG 8. As emergency stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions that was installed prior to June 12, 2006, the permittee must operate SN-8M01 according to the conditions described in paragraphs (f)(2)(i) through (iii) of this section. If you do not operate the engine according to the requirements in paragraphs (f)(2)(i) through (iii) of this section, the engine will not be considered an emergency engine under this subpart and will need to meet all requirements for non-emergency engines.
 - a. There is no time limit on the use of emergency stationary RICE in emergency situations.

- b. You may operate your emergency stationary RICE for the purpose of maintenance checks and readiness testing, provided that the tests are recommended by the manufacturer, the vendor, or the insurance company associated with the engine. Required testing of such units should be minimized, but there is no time limit on the use of emergency stationary RICE in emergency situations and for routine testing and maintenance.
- c. You may operate your emergency stationary RICE for an additional 50 hours per year in non-emergency situations. The 50 hours per year for non-emergency situations cannot be used for peak shaving or to generate income for a facility to supply power to an electric grid or otherwise supply power as part of a financial arrangement with another entity.

[Reg.19.304 and 40 C.F.R. § 63.6640(f)(2)(i-iii)]

SN-5N01-WA, SN-7M04-HT-G01, SN-7M04-HT-G04, Part 63 Subpart ZZZZ Conditions

- EG 9. 5N01-WA, 7M04-HT-G01, 7M04-HT-G04 are RICE engines with a site rating of less than 500 brake horsepower (HP) located at a major source of HAP emissions.
 [Reg.19.304 and 40 C.F.R. § 63.6590(a)]
- EG 10. 5N01-WA, 7M04-HT-G01, 7M04-HT-G04 are existing stationary RICE because they were constructed before June 12, 2006. [Reg.19.304 and 40 C.F.R. § 63.6590(a)(1)(i)(ii)]
- EG 11. The permittee must comply with the applicable emission limitations and operating limitations for 5N01-WA, 7M04-HT-G01, and 7M04-HT-G04 no later than May 3, 2013. [Reg.19.304 and 40 C.F.R. § 63.6595(a)]
- EG 12. The permittee must comply with the emission limitations and other requirements in Table 2c to this subpart which apply to 5N01-WA, 7M04-HT-G01, and 7M04-HT-G04. [Reg.19.304 and 40 C.F.R. § 63.6602 and Table 2(c)(1)]
 - a. Change oil and filter every 500 hours of operation or annually, whichever comes first;
 - b. Inspect air cleaner every 1,000 hours of operation or annually, whichever comes first;
 - c. Inspect all hoses and belts every 500 hours of operation or annually, whichever comes first, and replace as necessary.
 - d. During periods of startup, you must minimize the engine's time spent at idle and minimize the engine's startup time at startup to a period needed for appropriate and safe loading of the engine, not to exceed 30 minutes, after which time the non-startup emission limitations apply.
- EG 13. The permittee must be in compliance with the emission limitations, operating limitations, and other requirements for this subpart at all times.

> At all times the permittee must operate and maintain any affected source, including associated air pollution control equipment and monitoring equipment, in a manner consistent with safety and good air pollution control practices for minimizing emissions. The general duty to minimize emissions does not require any further efforts to reduce emissions if levels required by this standard have been achieved. Determination of whether such operation and maintenance procedures are being used will be based on information available to the Administrator which may include, but is not limited to, monitoring results, review of operation and maintenance procedures, review of operation and maintenance records, and inspection of the source.

[Reg.19.304 and 40 C.F.R. § 63.6605(a)&(b)]

- EG 14. The permittee must operate and maintain the stationary RICE 5N01-WA, 7M04-HT-G01, and 7M04-HT-G04 and their respective after-treatment control device (if any) according to the manufacturer's emission-related written instructions or develop a maintenance plan which must provide to the extent practicable for the maintenance and operation of the engine in a manner consistent with good air pollution control practice for minimizing emissions. [Reg.19.304 and 40 C.F.R. § 63.6625(e)(2)]
- EG 15. The permittee must install a non-resettable hour meter if one is not already installed at 5N01-WA, 7M04-HT-G01, and 7M04-HT-G04. [Reg.19.304 and 40 C.F.R. § 63.6625(f)]
- EG 16. The permittee must minimize the engine's, 5N01-WA, 7M04-HT-G01, and 7M04-HT-G04, time spent at idle during startup and minimize the engine's startup time to a period needed for appropriate and safe loading of the engine, not to exceed 30 minutes, after which time the emission standards applicable to all times other than startup in Table 2c. [Reg.19.304 and 40 C.F.R. § 63.6625(h)]
- EG 17. The permittee must operate 5N01-WA, 7M04-HT-G01, and 7M04-HT-G04 according to the requirements in paragraphs (f)(1)(i) through (iii) of this section. Any operation other than emergency operation, maintenance and testing, and operation in non-emergency situations for 50 hours per year, as described in paragraphs (f)(1)(i) through (iii) of this section, is prohibited. If you do not operate the engine according to the requirements in paragraphs (f)(1)(i) through (iii) of this section, the engine will not be considered an emergency engine under this subpart and will need to meet all requirements for non-emergency engines.
 - a. There is no time limit on the use of emergency stationary RICE in emergency situations.

- b. You may operate your emergency stationary RICE for the purpose of maintenance checks and readiness testing, provided that the tests are recommended by Federal, State or local government, the manufacturer, the vendor, or the insurance company associated with the engine. Maintenance checks and readiness testing of such units is limited to 100 hours per year. The owner or operator may petition the Administrator for approval of additional hours to be used for maintenance checks and readiness testing, but a petition is not required if the owner or operator maintains records indicating that Federal, State, or local standards require maintenance and testing of emergency RICE beyond 100 hours per year.
- c. You may operate your emergency stationary RICE up to 50 hours per year in nonemergency situations, but those 50 hours are counted towards the 100 hours per year provided for maintenance and testing. The 50 hours per year for nonemergency situations cannot be used for peak shaving or to generate income for a facility to supply power to an electric grid or otherwise supply power as part of a financial arrangement with another entity; except that owners and operators may operate the emergency engine for a maximum of 15 hours per year as part of a demand response program if the regional transmission organization or equivalent balancing authority and transmission operator has determined there are emergency conditions that could lead to a potential electrical blackout, such as unusually low frequency, equipment overload, capacity or energy deficiency, or unacceptable voltage level. The engine may not be operated for more than 30 minutes prior to the time when the emergency condition is expected to occur, and the engine operation must be terminated immediately after the facility is notified that the emergency condition is no longer imminent. The 15 hours per year of demand response operation are counted as part of the 50 hours of operation per year provided for non-emergency situations. The supply of emergency power to another entity or entities pursuant to financial arrangement is not limited by this paragraph (f)(1)(iii), as long as the power provided by the financial arrangement is limited to emergency power.

[Reg.19.304 and 40 C.F.R. § 63.6640(f)(1)(i-iii)]

- EG 18. The permittee must keep records of the maintenance conducted on 5N01-WA, 7M04-HT-G01, and 7M04-HT-G04 in order to demonstrate that you operated and maintained the stationary RICE and after-treatment control device (if any) according to your own maintenance plan. [Reg.19.304 and 40 C.F.R. § 63.6655(e)(2)]
- EG 19. The permittee must keep records of the hours of operation of operate 5N01-WA, 7M04-HT-G01, and 7M04-HT-G04 that is recorded through the non-resettable hour meter. The owner or operator must document how many hours are spent for emergency operation, including what classified the operation as emergency and how many hours are spent for non-emergency operation. If the engine is used for the purposes specified in § 63.6640(f)(1)(ii) or (iii), the owner or operator must keep records of the notification of the emergency situation, and the date, start time, and end time of engine operation for these purposes. [Reg.19.304 and 40 C.F.R. § 63.6655(f)(1)]

- EG 20. The permittee shall comply with the following for records required by this subpart for 5N01-WA, 7M04-HT-G01, and 7M04-HT-G04:
 - a. Records must be in a form suitable and readily available for expeditious review according to § 63.10(b)(1).
 - b. Keep each record for 5 years following the date of each occurrence, measurement, maintenance, corrective action, report, or record.
 - c. Keep each record readily accessible in hard copy or electronic form for at least 5 years after the date of each occurrence, measurement, maintenance, corrective action, report, or record, according to § 63.10(b)(1). [Reg.19.304 and 40 C.F.R. § 63.6660(a,b,&c)]

<u>SN-6N02 EG, Part 63 Subpart ZZZZ Conditions</u>

EG 21. SN-6N02 EG does not have to meet the requirements of this subpart and of Subpart A (except for initial notification requirements of §63.6645(f)) due to it being a new emergency stationary RICE with a site rating of more than 500 brake hp located at a major source of HAP emissions and does not operate or is not contractually obligated to be available for more than 15 hours per calendar year for the purposes specified in §63.6640(f)(2)(ii) and (iii). [Reg.19.304 and 40 C.F.R. § 63.6590(b)(1)(i)]

SN-6N02 EG, Part 60 Subpart IIII Conditions

EG 22. The permittee shall not exceed the limits in the following table for SN-6N02 EG:

6N02-EG	NMHC+NOx	СО	РМ
	6.4 (4.8)	3.5 (2.6)	0.2 (0.15)

Emission Standards in g/KW-hr (g/HP-hr)

[Reg.19.304 and 40 C.F.R. § 60.4205(b)]

- EG 23. The installed engine's (SN-6N02-EG) emission standards were certified by the manufacturer on 06/08/2010. EPA Certificate Number: CEX-STATCI-11-05. [40 CFR § 60.4202(a)(2), 60.4208(a)(b)(h), 60.4211(c)]
- EG 24. The permittee must use diesel fuel that meets the requirements of 40 CFR 80.510(b). [Reg 19.304 and 40 CFR § 60.4207(b)]

- EG 25. The permittee must operate and maintain SN-6N02 EG to achieve the emission standards as required in Condition EG 24 over the entire life of the engine. [Reg. 19.304 and 40 CFR § 60.4206]
- EG 26. SN 6N02-EG must comply with the emission standards specified in this subpart. The permittee must do all of the following, except as permitted under paragraph (g) of this section:
 - a. Operate and maintain SN 6N02-EG internal combustion engine and control device according to the manufacturer's emission-related written instructions;
 - b. Change only those emission-related settings that are permitted by the manufacturer; and
 - c. Meet the emission standards as specified in Condition EG 22.

[Reg 19.304 and 40 CFR § 60.4211(a)]

- EG 27. If the permittee does not install, configure, operate, and maintain SN 6N02-EG according to the manufacturer's emission-related written instructions, or you change emission-related settings in a way that is not permitted by the manufacturer the permittee must keep a maintenance plan and records of conducted maintenance and must, to the extent practicable, maintain and operate the engine in a manner consistent with good air pollution control practice for minimizing emissions. In addition, you must conduct an initial performance test to demonstrate compliance with the applicable emission standards within 1 year of start-up, or within 1 year after an engine and control device is no longer installed, configured, operated and maintained in accordance with the manufacturer's emission-related written instructions, or within 1 year after you change emission-related settings in a way that is not permitted by the manufacturer. You must conduct subsequent performance testing every 8,760 hours of engine operation or 3 years, whichever comes first, thereafter to demonstrate compliance with the applicable emission standards. [Reg 19.304 and 40 CFR § 60.4211(g)]
- EG 28. If the permittee does not install, configure, operate, and maintain the engine and control device according to the manufacturer's emission-related instructions, or change the emission-related settings in a way not permitted by the manufacturer, the permittee must conduct performance test according to 40 CFR §60.4212. [Reg 19.304 and 40 CFR § 60.4212]
- EG 29. The permittee must operate the emergency stationary ICE according to the requirements in paragraphs (a) through (c) of this section. In order for the engine to be considered an emergency stationary ICE under this subpart, any operation other than emergency operation, maintenance and testing, emergency demand response, and operation in nonemergency situations for 50 hours per year, as described in paragraphs (a) through (c) of this section, is prohibited. If you do not operate the engine according to the requirements in paragraphs (a) through (c) of this section, the engine will not be considered an emergency engine under this subpart and must meet all requirements for non-emergency engines.
- a. There is no limit on the use of emergency stationary ICE in emergency situations.
- b. The permittee may operate your emergency stationary ICE for any combination of the purposes specified in paragraphs (i) of this section for a maximum of 100 hours per calendar year. Any operation for non-emergency situations as allowed by paragraph (c) of this section counts as part of the 100 hours per calendar year allowed by this paragraph (b).
 - i. Emergency stationary ICE may be operated for maintenance checks and readiness testing, provided that the tests are recommended by federal, state or local government, the manufacturer, the vendor, the regional transmission organization or equivalent balancing authority and transmission operator , or the insurance company associated with the engine. The owner or operator may petition the Administrator for approval of additional hours to be used for maintenance checks and readiness testing, but a petition is not required if the owner or operator maintains records indicating that federal, state, or local standards require maintenance and testing of emergency ICE beyond 100 hours per calendar year.
- c. Emergency stationary ICE may be operated for up to 50 hours per calendar year in non-emergency situations. The 50 hours of operation in non-emergency situations are counted as part of the 100 hours per calendar year for maintenance and testing and emergency demand response provided in paragraph (b) of this section. The 50 hours per calendar year for non-emergency situations cannot be used for peak shaving or non-emergency demand response, or to generate income for a facility to an electric grid or otherwise supply power as part of a financial agreement with another entity.

[Reg 19.304 and 40 CFR § 60.4211(f)]

- EG 30. If SN 6N02-EG does not meet the standards applicable to non-emergency engines, the permittee must install a non-resettable hour meter prior to startup of SN 6N02-EG. [Reg 19.304 and 40 CFR § 60.4209(a)]
- EG 31. If SN 6N02-EG is an emergency stationary internal combustion engine, the permittee is not required to submit an initial notification. Starting with model year 2011, if the emergency engine does not meet the standards applicable to non-emergency engines in the applicable model year, the permittee must keep records of the operation of the engine in emergency and non-emergency service that are recorded through a non-resettable hour meter. The permittee must record the time of operation of the engine and the reason the engine was in operation during that time. [Reg 19.304 and 40 CFR § 60.4214(b)]

SN-4P-EG-01, Part 63 Subpart ZZZZ Conditions

EG 32. SN:4P-EG-01 meets the requirements of Part 63 subpart ZZZZ by complying with the applicable requirements of Part 60 subpart JJJJ. [Reg 19.304 and 40 CFR §

63.6590(c)(4)&(6)]

SN-4P-EG-01, Part 60 Subpart JJJJ Conditions

EG 33. The permittee shall not exceed the limits in the following table for SN: 4P-EG-01:

Emission Standards in g/KW-hr (g/HP-hr)		
4P-EG-01	NOx+HC	СО
	13.4 (10)	519 (387)

[Reg. 19.304 and 40 C.F.R. Part 60 Subpart JJJJ Table 1]

EG 34. The engine's (SN: 4P-EG-01) emission standards were certified by the manufacturer and shall comply with emission standards in Table 1. EPA Certificate Number:

NI<HXB06.2NNL-001. [40 CFR § 60.4233(d)]

- EG 35. The permittee must operate and maintain SN: 4P-EG-01 to achieve the emission standards as required in Condition EG 35 over the entire life of the engine. [Reg. 19.304 and 40 CFR § 60.4234]
- EG 36. The permittee must comply with the emission standards specified in EG 35 and must demonstrate compliance by purchasing a certified engine. [Reg. 19.304 and 40 CFR § 60.4243(b)(1)]
- EG 37. The permittee must operate the emergency stationary ICE according to the requirements below. In order for the engine to be considered an emergency stationary ICE under this subpart, any operation other than emergency operation, maintenance and testing, emergency demand response, and operation in non-emergency situations for 50 hours per year, as described in paragraphs (a) through (c) of this section, is prohibited. If you do not operate the engine according to the requirements in paragraphs (a) through (c) of this section, the engine will not be considered an emergency engine under this subpart and must meet all requirements for non-emergency engines.
 - a. There is no limit on the use of emergency stationary ICE in emergency situations.
 - b. The permittee may operate your emergency stationary ICE for any combination of the purposes specified in paragraphs (i) of this section for a

maximum of 100 hours per calendar year. Any operation for non-emergency situations as allowed by paragraph (c) of this section counts as part of the 100 hours per calendar year allowed by this paragraph (b).

- i. Emergency stationary ICE may be operated for maintenance checks and readiness testing, provided that the tests are recommended by federal, state or local government, the manufacturer, the vendor, the regional transmission organization or equivalent balancing authority and transmission operator, or the insurance company associated with the engine. The owner or operator may petition the Administrator for approval of additional hours to be used for maintenance checks and readiness testing, but a petition is not required if the owner or operator maintains records indicating that federal, state, or local standards require maintenance and testing of emergency ICE beyond 100 hours per calendar year.
- c. Emergency stationary ICE may be operated for up to 50 hours per calendar year in non-emergency situations. The 50 hours of operation in nonemergency situations are counted as part of the 100 hours per calendar year for maintenance and testing and emergency demand response provided in paragraph (b) of this section. The 50 hours per calendar year for nonemergency situations cannot be used for peak shaving or non-emergency demand response, or to generate income for a facility to an electric grid or otherwise supply power as part of a financial agreement with another entity.

[Reg 19.304 and 40 CFR § 60.4243(d)]

- EG 38. The permittee may operate their natural gas engine using propane for a maximum of 100 hours per year as an alternative fuel solely during emergency operations, but must keep records of such use. If propane is used for more than 100 hours per year in an engine that is not certified to the emission standards when using propane, the owners and operators are required to conduct a performance test to demonstrate compliance with the emission standards of § 60.4233. [Reg 19.304 and 40 CFR § 60.4243(e)]
- EG 39. The permittee of the stationary SI ICE must keep the following records:

(l) All notifications submitted to comply with this subpart and all documentation supporting any notification.

- (2) Maintenance conducted on the engine.
- (3) Documentation from the manufacturer that the engine is certified to meet the emission standards and information as required in 40 CFR parts 1048 1054 and 1060, as applicable. [Reg 19.304 and 40 CFR § 60.4245(a)]

EG 40. If SN 4P-EG-01 does not meet the standards applicable to non-emergency engines, the permittee must install a non-resettable hour meter prior to startup of SN 4P-EG-01. [Reg 19.304 and 40 CFR § 60.4245(b)]

Plantwide Sources Subject to MON

Source Description

Miscellaneous organic chemical manufacturing process units (MCPU) that are located at, or are part of, a major source of hazardous air pollutants (HAP) emissions as defined in section 112(a) of the Clean Air Act (CAA). [§63.2435(a)]

An MCPU includes equipment necessary to operate a miscellaneous organic chemical manufacturing process, as defined in §63.2550, that satisfies all of the conditions specified in paragraphs (b)(1) through (3) of this section. An MCPU also includes any assigned storage tanks and transfer racks; equipment in open systems that is used to convey or store water having the same concentration and flow characteristics as wastewater; and components such as pumps, compressors, agitators, pressure relief devices, sampling connection systems, open-ended valves or lines, valves, connectors, and instrumentation systems that are used to manufacture any material or family of materials described in paragraphs (b)(1)(i) through (v) of this section. [§63.2435(b)]

All or part of the following facilities are subject to the MON standards:

- Organic Chemical Intermediates
- Organic Sulfonation Process
- Solvent Recovery
- Isopropyl Benzene Production
- 5N07 Production Facility
- Aldehyde Processing Facility
- Storage Tanks and Miscellaneous Sources
- Anode Material Process

Affected Source Applicability

- MON 1. The permittee is an existing affected source subject to 40 CFR 63, Subpart FFFF as defined in §63.2435(a) through (e), and §63.2440. [Reg.19.304 and 40 C.F.R. §§ 63.2345 and 63.2440]
- MON 2. The CP-2 section contains equipment that are new affected sources subject to 40 CFR 63, Subpart FFFF as defined in §63.2435(a) through (e), and §63.2440(c)(1). [Reg.19.304 and 40 C.F.R. §§ 63.2335 and 63.2440]

Compliance Date

- MON 3. The permittee must comply with the requirements for existing sources in this subpart no later than May 10, 2008. [Reg.19.304 and 40 C.F.R. §§ 63.2440 and 63.2445(b), (g), and (h)]
- MON 4. The permittee must comply with the requirements for new affected sources in this subpart no later than upon startup of the affected new source. [Reg.19.304 and 40

C.F.R. §§ 63.2440 and 63.2445(a)(2)]

Notification Requirements

- MON 5. The permittee must meet the notification requirements in §63.2515 according to the dates specified in that section and in subpart A of part 63. [Reg.19.304 and 40 C.F.R. § 63.2445(c)]
- MON 6. If the permittee has a Group 2 emission point that becomes a Group 1 emission point after the compliance date for the affected source, the permittee must comply with the Group 1 requirements beginning on the date the switch occurs. An initial compliance demonstration as specified in this subpart must be conducted within 150 days after the switch occurs. [Reg.19.304 and 40 C.F.R. § 63.2445(d)]
- MON 7. If, after the compliance date for an affected source, hydrogen halide and halogen HAP emissions from process vents in a process increase to more than 1,000 lb/yr, or HAP metals emissions from a process at a new affected source increase to more than 150 lb/yr, the permittee must comply with the applicable emission limits specified in Table 3 to Subpart FFFF and the associated compliance requirements beginning on the date the emissions exceed the applicable threshold. An initial compliance demonstration as specified in this subpart must be conducted within 150 days after the switch occurs. [Reg.19.304 and 40 C.F.R. § 63.2445(e)]
- MON 8. If the permittee operates a small control device for process vents or transfer rack emissions that become a large control device, as defined in 63.2550(i), the permittee must comply with monitoring and associated recordkeeping and reporting requirements for large control devices beginning on the date the switch occurs. An initial compliance demonstration as specified in this subpart must be conducted within 150 days after the switch occurs. [Reg.19.304 and 40 C.F.R. § 63.2445(f)]
- MON 9. If after the compliance date of an affected source the permittee starts production of a Subpart FFFF covered process in existing equipment not previously subject to Subpart FFFF, the permittee must meet all Subpart FFFF applicable conditions on the process startup date and must conduct an initial compliance demonstration where applicable within 150 days after startup. [Reg.19.304 and 40 C.F.R. § 63.2445]

General Requirements

MON 10. The permittee must be in compliance with the emission limits and work practice standards in Tables 1 through 7 to Subpart FFFF at all times, and must meet the requirements specified in §§63.2455 through 63.2490 §63.2492 and §63.2493 (or the alternative means of compliance in §63.2495, §63.2500, or §63.2505), except as specified in paragraphs (b) through (v) excluding (p) of this section. The permittee must meet the notification, reporting, and recordkeeping requirements specified in §§63.2515, 63.2520, and 63.2525. [Reg.19.304 and 40 C.F.R. § 63.2450(a) and (u)]

Requirements for Continuous Process Vents

MON 11. The permittee must meet each emission limit in Table 1 to Subpart FFFF that applies to continuous process vents, and must meet each applicable requirement specified in paragraphs (b) through (c) of §63.2455. [Reg.19.304 and 40 C.F.R. § 63.2455(a)]

For each	For which	Then you must
1. Group 1 continuous process vent	a. Not applicable	 i. Reduce emissions of total organic HAP by ≥98 percent by weight or to an outlet process concentration ≤20 ppmv as organic HAP or TOC by venting emissions through a closed-vent system to any combination of control devices (except a flare); or ii. Reduce emissions of total organic HAP by venting emissions through a closed vent system to a flare; or iii. Use a recovery device to maintain the TRE above 1.9 for an existing source or above 5.0 for a new source.
2. Halogenated Group 1 continuous process vent stream	a. You use a combustion control device to control organic HAP emissions	i. Use a halogen reduction device after the combustion device to reduce emissions of hydrogen halide and halogen HAP by \geq 99 percent by weight, or to \leq 0.45 kg/hr, or to \leq 20 ppmv; or ii. Use a halogen reduction device before the combustion device to reduce the halogen atom mass emission rate to \leq 0.45 kg/hr or to a concentration \leq 20 ppmv.
3. Group 2 continuous process vent at an existing source	You use a recovery device to maintain the TRE level >1.9 but ≤5.0	Comply with the requirements in §63.2450(e)(4)and the requirements in §63.993 and the requirements referenced therein.
4. Group 2 continuous process vent at a new source	You use a recovery device to maintain the TRE level >5.0 but ≤8.0	Comply with the requirements in §63.2450(e)(4)and the requirements in §63.993 and the requirements referenced therein.
5. Continuous process vent	Beginning no later than the compliance dates specified in §63.2445(i), the continuous process vent contains ethylene oxide such that it is considered to be in ethylene oxide service as defined in §63.2550	Comply with the applicable emission limits specified in items 1 through 4 of this Table, and also: i. Reduce emissions of ethylene oxide by venting emissions through a closed-vent system to a flare; or Reduce emissions of ethylene oxide by venting emissions through a closed-vent system to a control device that reduces ethylene oxide by ≥99.9 percent by weight, or to a concentration <1 ppmv for each process vent or to <5 pounds per year for all combined process

For each	For which	Then you must
		vents.

Requirements for Batch Process Vents

MON 12. The permittee must meet each emission limit in Table 2 to this subpart that applies, and must meet each applicable requirement specified in paragraphs (b) and (c) of 63.2460. [Reg.19.304 and 40 C.F.R. § 63.2460(a)]

For each	Then you must	And you must
1. Process with Group 1 batch process vents	a. Reduce collective uncontrolled organic HAP emissions from the sum of all batch process vents within the process by ≥98 percent by weight by venting emissions from a sufficient number of the vents through one or more closed-vent systems to any combination of control devices (except a flare); or	Not applicable.
	b. Reduce collective uncontrolled organic HAP emissions from the sum of all batch process vents within the process by \geq 95 percent by weight by venting emissions from a sufficient number of the vents through one or more closed-vent systems to any combination of recovery devices or a biofilter, except you may elect to comply with the requirements of subpart WW of this part for any process tank; or	Not applicable.
	c. Reduce uncontrolled organic HAP emissions from one or more batch process vents within the process by venting through a closed-vent system to a flare or by venting through one or more closed-vent systems to any combination of control devices (excluding a flare) that reduce organic HAP to an outlet concentration ≤20 ppmv as TOC or total organic HAP.	For all other batch process vents within the process, reduce collective organic HAP emissions as specified in item 1.a and/or item 1.b of this table.
2. Halogenated Group 1 batch process vent for which you use a combustion device to control organic HAP emissions	a. Use a halogen reduction device after the combustion control device; or	i. Reduce overall emissions of hydrogen halide and halogen HAP by \geq 99 percent; or ii. Reduce overall emissions of hydrogen halide and halogen HAP to \leq 0.45 kg/hr; or iii. Reduce overall emissions of hydrogen halide and halogen HAP to a concentration \leq 20 ppmv.

For each	Then you must	And you must
	b. Use a halogen reduction device before the combustion control device	Reduce the halogen atom mass emission rate to ≤ 0.45 kg/hr or to a concentration ≤ 20 ppmv.
3. Batch process vent that contains ethylene oxide such that it is considered to be in ethylene oxide service as defined in §63.2550	 Beginning no later than the compliance dates specified in §63.2445(i), comply with the applicable emission limits specified in items 1 and 2 of this Table, and also: Reduce emissions of ethylene oxide by venting emissions through a closed-vent system to a flare; or Reduce emissions of ethylene oxide by venting emissions through a closed-vent system to a control device that reduces ethylene oxide by ≥99.9 percent by weight, or to a concentration <1 ppmv for each process vent or to a <5 pounds per year for all combined process vents. 	Not applicable.

Requirements for Process Vents that Emit Hydrogen Halide and Halogen HAP or HAP Metals

MON 13. The permittee must meet each emission limit in Table 3 to this subpart that applies and must meet each applicable requirement in paragraphs (b) through (d) of this section. [Reg.19.304 and 40 C.F.R. § 63.2465(a)]

For each	You must
1. Process with uncontrolled hydrogen halide and halogen HAP emissions from process vents \geq 1,000 lb/yr.	a. Reduce collective hydrogen halide and halogen HAP emissions by \geq 99 percent by weight or to an outlet concentration \leq 20 by venting through one or more closed-vent systems to any combination of control devices, or
	b. Reduce the halogen atom mass emission rate from the sum of all batch process vents and each individual continuous process vent to ≤ 0.45 kb/hr by venting through one or more closed-vent systems to a halogen reduction device.
2. Process at a new source with uncontrolled emissions from process vents ≥150 lb/yr of HAP metals	Reduce overall emission of HAP metals by \geq 97 percent by weight.

Requirements for Storage Tanks

MON 14. The permittee must meet each emission limit in Table 4 to Subpart FFFF that applies to its storage tanks, and must meet each applicable requirement specified in paragraphs (b) through (e) of §63.2470. [Reg.19.304 and 40 C.F.R. § 63.2470(a)]

For each	For which	Then you must
1. Group 1 storage a. tank to ≥′	a. The maximum true vapor pressure of total HAP at the storage temperature is ≥76.6 kilopascals	i. Reduce total HAP emissions by \geq 95 percent by weight or to \leq 20 ppmv of TOC or organic HAP and \leq 20 ppmv of hydrogen halide and halogen HAP by venting emissions through a closed vent system to any combination of control devices (excluding a flare); or
		ii. Reduce total organic HAP emissions by venting emissions through a closed vent system to a flare; or
		iii. Comply with the requirements in §63.2450(e)(4), as applicable; and reduce educe total HAP emissions by venting emissions to a fuel gas system or process in accordance with §63.982(d) and the requirements referenced therein.
	b. The maximum true vapor pressure of total HAP at the storage temperature is	i. Comply with the requirements of subpart WW of this part, except as specified in §63.2470; or
	6.6 kilopascals</td <td>ii. Reduce total HAP emissions by \geq95 percent by weight or to \leq20 ppmv of TOC or organic HAP and \leq20 ppmv of hydrogen halide and halogen HAP by venting emissions through a closed vent system to any combination of control devices (excluding a flare); or</td>	ii. Reduce total HAP emissions by \geq 95 percent by weight or to \leq 20 ppmv of TOC or organic HAP and \leq 20 ppmv of hydrogen halide and halogen HAP by venting emissions through a closed vent system to any combination of control devices (excluding a flare); or
		iii. Reduce total organic HAP emissions by venting emissions through a closed vent system to a flare; or
		iv. Comply with the requirements in §63.2450(e)(4), as applicable; and reduce total HAP emissions by venting emissions to a fuel gas system or process in accordance with §63.982(d) and the requirements referenced therein.
2. Halogenated vent stream from a Group 1 storage tank	You use a combustion control device to control organic HAP emissions	Meet one of the emission limit options specified in Item 2.a.i or ii. in Table 1 to this subpart.
3. Storage tank of any capacity and vapor pressure	Beginning no later than the compliance dates specified in §63.2445(i), the stored liquid contains ethylene oxide such that the storage tank is considered to be in ethylene oxide service as defined in §63.2550	 Comply with the applicable emission limits specified in items 1 and 2 of this Table, and also: Reduce emissions of ethylene oxide by venting emissions through a closed-vent system to a flare; or Reduce emissions of ethylene oxide by venting emissions through a closed-vent system to a control device that reduces ethylene oxide by ≥99.9 percent by weight, or to a concentration <1 ppm for each storage tank vent.

Requirements for Transfer Racks

MON 15. The permittee must comply with each emission limit and work practice standard in Table 5 to Subpart FFFF that applies to transfer racks, and must meet each applicable requirement in paragraphs (b) and (c) of §63.2475. [Reg.19.304 and 40 C.F.R. § 63.2475(a)]

For each	You must
1. Group 1 transfer rack	a. Reduce emissions of total organic HAP by \geq 98 percent by weight or to an outlet concentration \leq 20 ppmv as organic HAP or TOC by venting emissions through a closed-vent system to any combination of control devices (except a flare); or
	b. Reduce emissions of total organic HAP by venting emissions through a closed-vent system to a flare; or
	c. Comply with the requirements in §63.2450(e)(4), as applicable; and reduce emissions of total organic HAP by venting emissions to a fuel gas system or process in accordance with §63.982(d) and the requirements referenced therein; or
	d. Use a vapor balancing system designed and operated to collect organic HAP vapors displaced from tank trucks and railcars during loading and route the collected HAP vapors to the storage tank from which the liquid being loaded originated or to another storage tank connected by a common header.
2. Halogenated Group 1 transfer rack vent stream for which you use a combustion device to control organic HAP emissions	a. Use a halogen reduction device after the combustion device to reduce emissions of hydrogen halide and halogen HAP by \geq 99 percent by weight, to \leq 0.45 kg/hr, or to \leq 20 ppmv; or b. Use a halogen reduction device before the combustion device to reduce the halogen atom mass emission rate to \leq 0.45 kg/hr or to a concentration \leq 20 ppmv.

Requirements for Equipment Leaks

MON 16. The permittee must meet each requirement in Table 6 to Subpart FFFF that applies to equipment leaks, except as specified in paragraphs (b) through (f) of §63.2480. [Reg.19.304 and 40 C.F.R. § 63.2480(a)

For all	And that is part of	You must
1. Equipment that is in organic HAP service	a. Comply with the requirements of subpart UU of this part 63 and the requirements referenced therein, except as specified in §63.2480(b) and (d) through (f); or	
	b. Comply with the requirements of subpart H of this part 63 and the requirements referenced therein, except as specified in §63.2480(b) and (d) through (f); or	
	c. Comply with the requirements of 40 CFR part 65, subpart F and the requirements referenced therein, except as specified in §63.2480(c) and (d) through (f).	
2. Equipment that is in organic HAP service at a new	a. Any MCPU	i. Comply with the requirements of subpart UU of this part 63 and the requirements referenced therein; or
source		ii. Comply with the requirements of 40 CFR part 65, subpart F.
3. Equipment that is in ethylene oxide service as defined in §63.2550	a. Any MCPU	i. Beginning no later than the compliance dates specified in §63.2445(i), comply with the requirements of subpart UU of this part and the requirements referenced therein, except as specified in §63.2493(d) and (e); or
		ii. Beginning no later than the compliance dates specified in §63.2445(i), comply with the requirements of subpart H of this part and the requirements referenced therein, except as specified in §63.2493(d) and (e);
		iii. Beginning no later than the compliance dates specified in §63.2445(i), comply with the requirements of 40 CFR part 65, subpart F, and the requirements referenced therein, except as specified in §63.2493(d) and (e).

Requirements for Wastewater and Liquid Streams in Open Systems within an MCPU

MON 17. The permittee must meet each requirement in Table 7 to Subpart FFFF that applies to wastewater streams and liquid streams in open systems within an MCPU, except as specified in paragraphs (b) through (o) of §63.2485. [Reg.19.304 and 40 C.F.R. § 63.2485(a)]

For each	You must
1. Process wastewater stream	Comply with the requirements in §§63.132 through 63.148 and the requirements referenced therein, except as specified in §63.2485.
2. Maintenance wastewater stream	Comply with the requirements in §63.105 and the requirements referenced therein, except as specified in §63.2485.
3. Liquid streams in an open system within an MCPU	Comply with the requirements in §63.149 and the requirements referenced therein, except as specified in §63.2485.

Requirements for Heat Exchange Systems

MON 18. The permittee must comply with each requirement in Table 10 to Subpart FFFF that applies to heat exchange systems, except as specified in paragraphs (b) and (c) of §63.2490. [Reg.19.304 and 40 C.F.R. § 63.2490(a)]

For each	You must
Heat exchange system, as defined in §63.101	 a. Comply with the requirements of §63.104 and the requirements referenced therein, except as specified in §63.2490. b. Comply with the requirements in §63.2490(d).

Compliance with Pollution Prevention Standards

MON 19. The permittee may elect to comply with the pollution prevention alternative requirements specified in paragraphs (a)(1) and (2) of this section in lieu of the emission limitations and work practice standards contained in Tables 1 through 7 to this subpart for any MCPU for which initial startup occurred before April 4, 2002. [Reg.19.304 and 40 C.F.R. § 63.2595(a)]

Emissions Averaging

MON 20. For an existing source, the permittee may elect to comply with the percent reduction emission limitations in Tables 1, 2, 4, 5, and 7 to this subpart by complying with the emissions averaging provisions specified in 63.150, except as specified in paragraphs (b) through (f) of 63.2500. [Reg.19.304 and 40 C.F.R. § 63.2500(a)]

Alternative Standard

MON 21. As an alternative to complying with the emission limits and work practice standards for process vents and storage tanks in Tables 1 through 4 to this subpart and the requirements in §§63.2455 through 63.2470, the permittee may comply with the emission limits in paragraph (a) of this section and demonstrate compliance in accordance with the requirements in paragraph (b) of this section. [Reg.19.304 and 40

C.F.R. § 63.2505]

Notification, Reports, and Records

MON 22. The permittee must submit all of the notifications in §§63.6(h)(4) and (5), 63.7(b) and (c), 63.8(e), (f)(4) and (6), and 63.9(b) through (h) that apply by the dates specified. [Reg.19.304 and 40 C.F.R. § 63.2515(a) and (d)]

Reporting Requirements

MON 23. The permittee must submit each report in Table 11 to Subpart FFFF that applies. [Reg.19.304 and 40 C.F.R. § 63.2520(a) and (d)]

You must submit a(n)	The report must contain	You must submit the report
1. Precompliance report	The information specified in §63.2520(c)	At least 6 months prior to the compliance date; or for new sources, with the application for approval of construction or reconstruction.
2. Notification of compliance status report	The information specified in §63.2520(d)	No later than 150 days after the compliance date specified in §63.2445.
3. Compliance report	The information specified in §63.2520(e)	Semiannually according to the requirements in §63.2520(b).

- MON 24. Unless the Administrator has approved a different schedule for submission of reports under §63.10(a), the permittee must submit each report by the date in Table 11 to Subpart FFFF and according to paragraphs (b)(1) through (5) of §63.2520. [Reg.19.304 and 40 C.F.R. § 63.2520(b)]
- MON 25. The permittee shall follow the reporting requirements of §63.2520(c) through (e). [Reg.19.304 and 40 C.F.R. § 63.2520(c) through (e)]

Recordkeeping Requirements

- MON 26. The permittee shall keep the records specified in applicable paragraphs (a) through (u) of section §63.2525. [Reg.19.304 and 40 C.F.R. § 63.2525(a) through (u)]
- Compliance Options for Applicability to 40 CFR 63, Subpart FFFF and another Subpart
 - MON 27. For any equipment, emission stream, or wastewater stream subject to the provisions of both Subpart FFFF and another rule, the permittee may elect to comply

only with the provisions as specified in paragraphs (a) through (l) of this section. The permittee also must identify the subject equipment, emission stream, or wastewater stream, and the provisions with which the permittee will comply, in the notification of compliance status report required by §63.2520(d). [Reg.19.304 and 40 C.F.R. § 63.2535]

General Applicability

MON 28. The permittee shall comply with parts of the General Provisions in §§63.1 through 63.15 as referenced in §63.2540. [Reg.19.304 and 40 C.F.R. § 63.2540]

SECTION V: COMPLIANCE PLAN AND SCHEDULE

FutureFuel Chemical Company will continue to operate in compliance with those identified regulatory provisions. The facility will examine and analyze future rules and regulations that may apply and determine their applicability with any necessary action taken on a timely basis.

SECTION VI: PLANTWIDE CONDITIONS

- The permittee shall notify the Director in writing within thirty (30) days after commencing construction, completing construction, first placing the equipment and/or facility in operation, and reaching the equipment and/or facility target production rate. [Rule 19.704, 40 C.F.R. § 52 Subpart E, and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]
- 2. If the permittee fails to start construction within eighteen months or suspends construction for eighteen months or more, the Director may cancel all or part of this permit. [Rule 19.410(B) and 40 C.F.R. § 52 Subpart E]
- 3. The permittee must test any equipment scheduled for testing, unless otherwise stated in the Specific Conditions of this permit or by any federally regulated requirements, within the following time frames: (1) new equipment or newly modified equipment within sixty (60) days of achieving the maximum production rate, but no later than 180 days after initial start up of the permitted source or (2) operating equipment according to the time frames set forth by the Division of Environmental Quality or within 180 days of permit issuance if no date is specified. The permittee must notify the Division of Environmental Quality of the scheduled date of compliance testing at least fifteen (15) business days in advance of such test. The permittee shall submit the compliance test results to the Division of Environmental Quality within sixty (60) calendar days after completing the testing. [Rule 19.702 and/or Rule 18.1002 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]
- 4. The permittee must provide:
 - a. Sampling ports adequate for applicable test methods;
 - b. Safe sampling platforms;
 - c. Safe access to sampling platforms; and
 - d. Utilities for sampling and testing equipment.

[Rule 19.702 and/or Rule 18.1002 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

- 5. The permittee must operate the equipment, control apparatus and emission monitoring equipment within the design limitations. The permittee shall maintain the equipment in good condition at all times. [Rule 19.303 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]
- 6. This permit subsumes and incorporates all previously issued air permits for this facility. [Rule 26 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

- 7. Unless otherwise specified in the permit, approval to construct any new major stationary source or a major modification subject to 40 C.F.R. § 52.21 shall become invalid if construction is not commenced within 18 months after receipt of such approval, if construction is discontinued for a period of 18 months or more, or if construction is not completed within a reasonable time. The Division of Environmental Quality may extend the 18-month period upon a satisfactory showing that an extension is justified. [Rule 19.901 *et seq.* and 40 C.F.R. § 52 Subpart E]
- 8. In the absence of an averaging period stated within the permit or any enforceable federal requirement (NSPS, NESHAP, etc.), the lb/hr limits stated herein are considered to be based upon a 3-hour averaging period. [Reg.19.705 and 40 C.F.R. § 52 Subpart E and Reg.18.1004 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]
- 9. The permittee shall submit a compliance report with state-only enforceable terms and conditions contained in the permit, including emission limitations, standards, or work practices. This compliance report shall be submitted annually to the Department. All compliance reports required by this permit shall include the following [Reg.18.1004 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]:
 - a. The identification of each term or condition of the permit that is the basis of the certification;
 - b. The compliance status;
 - c. Whether compliance was continuous or intermittent;
 - d. The method(s) used for determining the compliance status of the source, currently and over the reporting period established by the monitoring requirements of this permit; and
 - e. Such other facts as the Department may require elsewhere in this permit.

This compliance report may be in the same format as, and may be included with, the annual compliance certification required by Section VI General Condition 21.

- For purposes of Section VI General Condition 8 of this permit and §§26.701(C)(3)(b) of Regulation #26, "prompt" or "prompt reporting" shall be construed to mean:
 - a. by the next business day, if deviations result in exceedances of applicable emission limitations lasting 30 or more minutes, in the aggregate during a 24-hour period, unless otherwise specified in an applicable permit or regulation (including, but not limited to, NSPS regulations); and
 - b. in the next semi-annual report for all other deviations.
 - [40 C.F.R. § 70.6(a)(3)(iii)(B), Reg.26.701(C)(3)(b), and Reg.19.601 and Reg.19.602]
- 11. The permittee may modify point source operating parameters (Facility Operating Plans) if the parameter modifications do not exceed emissions allowable under the

permit. Facility Operating Plan modifications must be provided to the Agency seven days in advance of the proposed changes. A shorter time frame may be allowed by the Agency in emergency cases. [Reg.26.802 and § 502(b)(10) of the Clean Air Act]

Plantwide Limits

12. The permittee shall not exceed the following emission rates at the facility during any consecutive 12-month period. [Reg.19.501 *et seq.* and 40 C.F.R. § 52 Subpart E]

Plantwide Limits		
Pollutant	ton/yr	
PM ₁₀	178.7	
SO ₂	6144.4	
VOC	491.0	
СО	1224.6	
NO _x	875.4	
Pb	3.5	

13. The permittee shall not exceed the following emission rates at the facility during any consecutive 12-month period. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

Pollutant	ton/yr
Inorganics*	1092.9
Organic Pollutants**	491.0

*Inorganics are considered to be non-organic Hazardous Air Pollutants

**Organic Pollutants are considered to be VOC, all organic HAPs, and organic Air Contaminates.

- 14. The permittee shall maintain records to demonstrate compliance with the criteria emission limits in 12. The emission records shall be recalculated monthly, and shall be based upon a 12-month rolling total. The records shall be updated by the last day of the month following the recorded 12-month period, and shall be kept on site and made available for inspection upon request. [Reg.19.705 and 40 C.F.R. § 52 Subpart E and Reg.18.1004 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]
- 15. The permittee shall determine the monthly emissions of each non-criteria air pollutant by material balance. This determination shall include each inorganic contaminant and each Hazardous Air Pollutant (HAP), as designated by Section 112 of the Clean Air Act. The material balance shall be recalculated monthly, and shall be based upon a 12-month rolling total. The records shall be updated by the last day of the month following the recorded 12-month period, and shall be kept on site and made available for inspection upon request. [Reg.18.801 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

16. The permittee shall use the emissions determined from 15 to show acceptable impacts in accordance with the Department's Non-Criteria Air Pollutant Control Strategy. Except for inorganic HCl, the permittee shall calculate the site-specific 30-day Presumptively Acceptable Emission Rate (PAER) for each non-criteria pollutant emitted at the facility using the equation presented below. [Reg.18.1004 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. § 8-4-304 and 8-4-311]

Except for inorganic HCl, this determination shall include each inorganic contaminant and each Hazardous Air Pollutant (HAP), as designated by Section 112 of the Clean Air Act. The permittee shall not emit more than the calculated 30-day PAER during any consecutive 30-day period. The permittee shall maintain on-site records of the emissions rates and the calculated 30-day site-specific PAER (lb/month) for each non-criteria pollutant emitted. These records shall be made available for inspection upon request.

Allowable site-specific PAER (lb/month) = 2.14 x (TLV in mg/m³ from ACGIH) x 720

Any exceedance of the site-specific PAER shall be reported to the Department within 24 hours of such discovery. A full report of the exceedance and subsequent corrective action shall be submitted to the Department within 5 business days.

For inorganic HCl, the permittee shall maintain documentation to confirm the monthly plantwide inorganic HCl emission rates are less than the allowable inorganic limit. These records shall be made available for inspection upon request.

The permittee shall review and update the TLV values used for each compound at least once annually, according to the most recent edition of the ACGIH Threshold Limit Values for Chemical Substances and Physical Agents.

17. This facility is a major stationary source as defined by 40 CFR §52.21. Any physical change or change in the method of operation which results in a significant emission increase, as defined by 40 CFR 52.21, shall require prior approval of a PSD netting exercise or a PSD permit before the event taking place, regardless of the plantwide emission rate. [Reg.19.901 *et seq.* and 40 C.F.R. § 52 Subpart E]

40 CFR Part 60, Subpart Kb Standards of Performance for Volatile Organic Liquid Storage Vessels

18. The permittee shall maintain documentation necessary to determine compliance with this subpart for all storage vessels having a capacity of greater than or equal to 75 cubic meters (19,813 gallons). Affected tanks include the following: [Reg.19.304 and 40 C.F.R. § 60.110b]

TF-13 (SN-5N03-43)	PM-50B	FAA-TF-01	VC-PT-03
WB-06 (SN-6M-03-08)	TBA-100	FAA-TF-02	VC-PT-01
WB-07 (SN-6M-03-09)	T-280 (SN-5N03-51)	FAA-TF-101	VC-PT-02
WB-08 (SN-6M-03-10)	T-265 (SN-5N03-53)	FAA-TF-102	T-243
WB-09 (SN-6M-03-11)	T-251	PROD-TF-02	T-271
TFS-60	T-220	PROD-TF-15	T-272
PT-60	T-211A	PROD-TF-302	T-273
PT-68	T-211B	RA-TF-01	

PT-69A	T-241	RA-TF-02
PT-69B	PA-50 T-270	SPS-TF-04
PB-51	RA-TF-01	SPS-TF-204
PB-52	AA-100	T-242
PM-50A	TBA-75	

- 19. The permittee shall maintain documentation identifying storage vessels complying with the requirements of 40 CFR §60.112b, including emission controls used, and all documentation to support compliance with the emission control used. [Reg.19.304 and 40 C.F.R. § 60.112b]
- 20. The permittee shall meet the specifications of this citation for closed vent systems and control devices used for tank emission abatement. [Reg.19.304 and 40 C.F.R. § 60.112b(a)(3)]
- 21. The permittee shall comply with all applicable testing and procedures as identified in §60.113b. The applicable requirement for a particular storage vessel depends on the control equipment installed to meet the requirements of §60.112b. [Reg.19.304 and 40 C.F.R. § 60.113b]
- 22. Each closed vent system and control device (other than a flare) is exempt from §60.8 of the General Provisions and shall comply with the requirements specified in this citation. [Reg.19.304 and 40 C.F.R. § 60.113b(c)]
- 23. Closed vent systems with flares shall comply with the requirements as specified in §60.18(e) and (f). Records shall be kept of all periods of operation during which the flare pilot flame is absent and shall be reported semiannually. [Reg.19.304 and 40 C.F.R. § 60.113b(d)(2) and (3)]
- 24. The permittee shall keep records and furnish reports as required, depending upon the control equipment installed, to meet the requirements of §60.112b. Copies of operating plans shall be kept for the life of the control equipment. [Reg.19.304 and 40 C.F.R. § 60.115b]
- 25. The permittee shall keep copies of all records required by Subpart Kb. [Reg.19.304 and 40 C.F.R. § 60.116b]
- 26. Each storage vessel equipped with a closed vent system and control device meeting the specifications of §60.112b is exempt from the requirements of paragraphs (c) and (d) of §60.116b. [Reg.19.304 and 40 C.F.R. § 60.116b(g)]

Acid Rain (Title IV)

27. The Director prohibits the permittee to cause any emissions exceeding any allowances the source lawfully holds under Title IV of the Act or the regulations promulgated under the Act. No permit revision is required for increases in emissions allowed by allowances acquired pursuant to the acid rain program, if such increases do not require a permit revision under any other applicable requirement. This permit establishes no limit on the number of allowances held by the permittee. However, the source may not use allowances as a defense for noncompliance with any other applicable requirement of this permit or the Act. The permittee will account for any

such allowance according to the procedures established in regulations promulgated under Title IV of the Act. A copy of the facility's Acid Rain Permit is attached in an appendix to this Title V permit. [Reg.26.701 and 40 C.F.R. § 70.6(a)(4)]

Title VI Provisions

- 28. The permittee must comply with the standards for labeling of products using ozonedepleting substances. [40 C.F.R. § 82 Subpart E]
 - a. All containers containing a class I or class II substance stored or transported, all products containing a class I substance, and all products directly manufactured with a class I substance must bear the required warning statement if it is being introduced to interstate commerce pursuant to § 82.106.
 - b. The placement of the required warning statement must comply with the requirements pursuant to § 82.108.
 - c. The form of the label bearing the required warning must comply with the requirements pursuant to § 82.110.
 - d. No person may modify, remove, or interfere with the required warning statement except as described in § 82.112.
- 29. The permittee must comply with the standards for recycling and emissions reduction, except as provided for MVACs in Subpart B. [40 C.F.R. § 82 Subpart F]
 - a. Persons opening appliances for maintenance, service, repair, or disposal must comply with the required practices pursuant to § 82.156.
 - b. Equipment used during the maintenance, service, repair, or disposal of appliances must comply with the standards for recycling and recovery equipment pursuant to § 82.158.
 - c. Persons performing maintenance, service repair, or disposal of appliances must be certified by an approved technician certification program pursuant to § 82.161.
 - d. Persons disposing of small appliances, MVACs, and MVAC like appliances must comply with record keeping requirements pursuant to § 82.166. ("MVAC like appliance" as defined at § 82.152)
 - e. Persons owning commercial or industrial process refrigeration equipment must comply with leak repair requirements pursuant to § 82.156.
 - f. Owners/operators of appliances normally containing 50 or more pounds of refrigerant must keep records of refrigerant purchased and added to such appliances pursuant to § 82.166.
- If the permittee manufactures, transforms, destroys, imports, or exports a class I or class II substance, the permittee is subject to all requirements as specified in 40 C.F.R. § 82 Subpart A, Production and Consumption Controls.
- 31. If the permittee performs a service on motor (fleet) vehicles when this service involves ozone depleting substance refrigerant (or regulated substitute substance) in

the motor vehicle air conditioner (MVAC), the permittee is subject to all the applicable requirements as specified in 40 C.F.R. § 82 Subpart B, Servicing of Motor Vehicle Air Conditioners.

The term "motor vehicle" as used in Subpart B does not include a vehicle in which final assembly of the vehicle has not been completed. The term "MVAC" as used in Subpart B does not include the air tight sealed refrigeration system used as refrigerated cargo, or the system used on passenger buses using HCFC 22 refrigerant.

- 32. The permittee can switch from any ozone depleting substance to any alternative listed in the Significant New Alternatives Program (SNAP) promulgated pursuant to 40 C.F.R. § 82 Subpart G.
- 33. The permittee shall submit reports as required by § 64.9(a). These records shall include: Summary information on the number, duration and cause (including unknown cause, if applicable) of excursions or exceedances, as applicable, and the corrective actions taken; summary information on the number, duration and cause (including unknown cause, if applicable) for monitor downtime incidents (other than downtime associated with zero and span or other daily calibration checks, if applicable); and a description of the actions taken to implement a QIP during the reporting period as specified in §64.8. Upon completion of a QIP, the owner or operator shall include in the next summary report documentation that the implementation of the plan has been completed and reduced the likelihood of similar levels of excursions or exceedances occurring. [Rule 19.304 and 40 C.F.R. § 64]

SECTION VII: INSIGNIFICANT ACTIVITIES

The Division of Environmental Quality deems the following types of activities or emissions as insignificant on the basis of size, emission rate, production rate, or activity in accordance with Group A of the Insignificant Activities list found in Rule 18 and Rule 19 Appendix A. Group B insignificant activities may be listed but are not required to be listed in permits. Insignificant activity emission determinations rely upon the information submitted by the permittee in an application dated May 17, 2023. [Rule 26.304 and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]

Source #	Description	Category
3N01-01	Storage Tank (Storage Tank Process), BD-01	A-13
4P02-02	Quenching (Solvent Recovery Process)	A-13
4P94-03	Storage Tank (Solvent Recovery Process)	A-3
4P94-04	Storage Tank (Solvent Recovery Process)	A-13
5M01-03	Vacuum System (Organic Sulfonation Process)	A-13
5M03-06	Vacuum System (Organic Sulfonation Process)	A-13
5M04-03	Storage Tank (Organic Sulfonation Process)	A-13
5M04-04	Storage Tank (Organic Sulfonation Process)	A-4
5M04-07	Storage Tank (Organic Sulfonation Process)	A-4
5M04-09	Storage Tank (Organic Sulfonation Process)	A-13
5M11-03	Vacuum System (Organic Sulfonation Process)	A-13
5M11-08	Vents (Organic Sulfonation Process)	A-13
5M11-09	Vents (Organic Sulfonation Process)	A-13
5N01-63	Storage Tank (Organic Chemical Intermediate Process)	A-3
5N01-64	Storage Tank (Organic Chemical Intermediate Process)	A-3
5N02-01	Storage Tank (Storage Tank Process)	A-13

Source #	Description	Category
5N02-02	Storage Tank (Storage Tank Process)	A-13
5N03-46	Unloading Station (Isopropyl Benzene Process)	A-13
5N03-47	Unloading Station (Isopropyl Benzene Process)	A-13
5N03-63	Storage Tank (Organic Chemical Intermediate Process)	A-3
5P99-01	US-2-VS Odor Absorber	A-13
6M03-15	Storage Tank (Chemical Destruction Process)	A-4
6N01-01	Storage Tank (Storage Tank Process)	A-3
	Caustic Tank (CL-01R)	A-4
	Railcar Loading and Unloading Racks	A-13
5P01-01	Storage Tank (Glycerin)	A-13
5P01-02	Storage Tank (Glycerin)	A-13
4Q01-12	Storage Tank (Glycerin)	A-13
4Q01-13	Storage Tank (Glycerin)	A-13

SECTION VIII: GENERAL PROVISIONS

- Any terms or conditions included in this permit which specify and reference Arkansas Pollution Control & Ecology Commission Rule 18 or the Arkansas Water and Air Pollution Control Act (Ark. Code Ann. § 8-4-101 *et seq.*) as the sole origin of and authority for the terms or conditions are not required under the Clean Air Act or any of its applicable requirements, and are not federally enforceable under the Clean Air Act. Arkansas Pollution Control & Ecology Commission Rule 18 was adopted pursuant to the Arkansas Water and Air Pollution Control Act (Ark. Code Ann. § 8-4-101 *et seq.*). Any terms or conditions included in this permit which specify and reference Arkansas Pollution Control & Ecology Commission Rule 18 or the Arkansas Water and Air Pollution Control & Ecology Commission Rule 18 or the Arkansas Water and Air Pollution Control & Ecology Commission Rule 18 or the Arkansas Water and Air Pollution Control Act (Ark. Code Ann. § 8-4-101 *et seq.*) as the origin of and authority for the terms or conditions are enforceable under this Arkansas statute. [40 C.F.R. § 70.6(b)(2)]
- 2. This permit shall be valid for a period of five (5) years beginning on the date this permit becomes effective and ending five (5) years later. [40 C.F.R. § 70.6(a)(2) and Rule 26.701(B)]
- 3. The permittee must submit a complete application for permit renewal at least six (6) months before permit expiration. Permit expiration terminates the permittee's right to operate unless the permittee submitted a complete renewal application at least six (6) months before permit expiration. If the permittee submits a complete application, the existing permit will remain in effect until the Division of Environmental Quality takes final action on the renewal application. The Division of Environmental Quality will not necessarily notify the permittee when the permit renewal application is due. [Rule 26.406]
- 4. Where an applicable requirement of the Clean Air Act, as amended, 42 U.S.C. 7401, *et seq.* (Act) is more stringent than an applicable requirement of regulations promulgated under Title IV of the Act, the permit incorporates both provisions into the permit, and the Director or the Administrator can enforce both provisions. [40 C.F.R. § 70.6(a)(1)(ii) and Rule 26.701(A)(2)]
- 5. The permittee must maintain the following records of monitoring information as required by this permit.
 - a. The date, place as defined in this permit, and time of sampling or measurements;
 - b. The date(s) analyses performed;
 - c. The company or entity performing the analyses;
 - d. The analytical techniques or methods used;
 - e. The results of such analyses; and
 - f. The operating conditions existing at the time of sampling or measurement.

[40 C.F.R. § 70.6(a)(3)(ii)(A) and Rule 26.701(C)(2)]

- 6. The permittee must retain the records of all required monitoring data and support information for at least five (5) years from the date of the monitoring sample, measurement, report, or application. Support information includes all calibration and maintenance records and all original strip chart recordings for continuous monitoring instrumentation, and copies of all reports required by this permit. [40 C.F.R. § 70.6(a)(3)(ii)(B) and Rule 26.701(C)(2)(b)]
- 7. The permittee must submit reports of all required monitoring every six (6) months. If the permit establishes no other reporting period, the reporting period shall end on the last day of the month six months after the issuance of the initial Title V permit and every six months thereafter. The report is due on the first day of the second month after the end of the reporting period. The first report due after issuance of the initial Title V permit shall contain six months of data and each report thereafter shall contain 12 months of data. The report shall contain data for all monitoring requirements in effect during the reporting period. If a monitoring requirement is not in effect for the entire reporting period, only those months of data in which the monitoring requirement was in effect are required to be reported. The report must clearly identify all instances of deviations from permit requirements. A responsible official as defined in Rule 26.2 must certify all required reports. The permittee will send the reports electronically using https://eportal.adeq.state.ar.us or mail them to the address below:

Division of Environmental Quality Office of Air Quality ATTN: Compliance Inspector Supervisor 5301 Northshore Drive North Little Rock, AR 72118-5317

[40 C.F.R. § 70.6(a)(3)(iii)(A) and Rule 26.701(C)(3)(a)]

- 8. The permittee shall report to the Division of Environmental Quality all deviations from permit requirements, including those attributable to upset conditions as defined in the permit.
 - a. For all upset conditions (as defined in Rule 19.601), the permittee will make an initial report to the Division of Environmental Quality by the next business day after the discovery of the occurrence. The initial report may be made by telephone and shall include:
 - i. The facility name and location;
 - ii. The process unit or emission source deviating from the permit limit;
 - iii. The permit limit, including the identification of pollutants, from which deviation occurs;
 - iv. The date and time the deviation started;

- v. The duration of the deviation;
- vi. The emissions during the deviation;
- vii. The probable cause of such deviations;
- viii. Any corrective actions or preventive measures taken or being taken to prevent such deviations in the future; and
 - ix. The name of the person submitting the report.

The permittee shall make a full report in writing to the Division of Environmental Quality within five (5) business days of discovery of the occurrence. The report must include, in addition to the information required by the initial report, a schedule of actions taken or planned to eliminate future occurrences and/or to minimize the amount the permit's limits were exceeded and to reduce the length of time the limits were exceeded. The permittee may submit a full report in writing (by facsimile, overnight courier, or other means) by the next business day after discovery of the occurrence, and the report will serve as both the initial report and full report.

b. For all deviations, the permittee shall report such events in semi-annual reporting and annual certifications required in this permit. This includes all upset conditions reported in 8a above. The semi-annual report must include all the information as required by the initial and full reports required in 8a.

[Rule 19.601, Rule 19.602, Rule 26.701(C)(3)(b), and 40 C.F.R. § 70.6(a)(3)(iii)(B)]

- 9. If any provision of the permit or the application thereof to any person or circumstance is held invalid, such invalidity will not affect other provisions or applications hereof which can be given effect without the invalid provision or application, and to this end, provisions of this Rule are declared to be separable and severable. [40 C.F.R. § 70.6(a)(5), Rule 26.701(E), and Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]
- 10. The permittee must comply with all conditions of this Part 70 permit. Any permit noncompliance with applicable requirements as defined in Rule 26 constitutes a violation of the Clean Air Act, as amended, 42 U.S.C. § 7401, *et seq.* and is grounds for enforcement action; for permit termination, revocation and reissuance, for permit modification; or for denial of a permit renewal application. [40 C.F.R. § 70.6(a)(6)(i) and Rule 26.701(F)(1)]
- 11. It shall not be a defense for a permittee in an enforcement action that it would have been necessary to halt or reduce the permitted activity to maintain compliance with the conditions of this permit. [40 C.F.R. § 70.6(a)(6)(ii) and Rule 26.701(F)(2)]
- 12. The Division of Environmental Quality may modify, revoke, reopen and reissue the permit or terminate the permit for cause. The filing of a request by the permittee for a permit modification, revocation and reissuance, termination, or of a notification of

planned changes or anticipated noncompliance does not stay any permit condition. [40 C.F.R. § 70.6(a)(6)(iii) and Rule 26.701(F)(3)]

- 13. This permit does not convey any property rights of any sort, or any exclusive privilege. [40 C.F.R. § 70.6(a)(6)(iv) and Rule 26.701(F)(4)]
- 14. The permittee must furnish to the Director, within the time specified by the Director, any information that the Director may request in writing to determine whether cause exists for modifying, revoking and reissuing, or terminating the permit or to determine compliance with the permit. Upon request, the permittee must also furnish to the Director copies of records required by the permit. For information the permittee claims confidentiality, the Division of Environmental Quality may require the permittee to furnish such records directly to the Director along with a claim of confidentiality. [40 C.F.R. § 70.6(a)(6)(v) and Rule 26.701(F)(5)]
- 15. The permittee must pay all permit fees in accordance with the procedures established in Rule 9. [40 C.F.R. § 70.6(a)(7) and Rule 26.701(G)]
- 16. No permit revision shall be required, under any approved economic incentives, marketable permits, emissions trading and other similar programs or processes for changes provided for elsewhere in this permit. [40 C.F.R. § 70.6(a)(8) and Rule 26.701(H)]
- 17. If the permit allows different operating scenarios, the permittee shall, contemporaneously with making a change from one operating scenario to another, record in a log at the permitted facility a record of the operational scenario. [40 C.F.R. § 70.6(a)(9)(i) and Rule 26.701(I)(1)]
- 18. The Administrator and citizens may enforce under the Act all terms and conditions in this permit, including any provisions designed to limit a source's potential to emit, unless the Division of Environmental Quality specifically designates terms and conditions of the permit as being federally unenforceable under the Act or under any of its applicable requirements. [40 C.F.R. § 70.6(b) and Rule 26.702(A) and (B)]
- Any document (including reports) required by this permit pursuant to 40 C.F.R. § 70 must contain a certification by a responsible official as defined in Rule 26.2. [40 C.F.R. § 70.6(c)(1) and Rule 26.703(A)]
- The permittee must allow an authorized representative of the Division of Environmental Quality, upon presentation of credentials, to perform the following: [40 C.F.R. § 70.6(c)(2) and Rule 26.703(B)]
 - a. Enter upon the permittee's premises where the permitted source is located or emissions related activity is conducted, or where records must be kept under the conditions of this permit;

- b. Have access to and copy, at reasonable times, any records required under the conditions of this permit;
- c. Inspect at reasonable times any facilities, equipment (including monitoring and air pollution control equipment), practices, or operations regulated or required under this permit; and
- d. As authorized by the Act, sample or monitor at reasonable times substances or parameters for assuring compliance with this permit or applicable requirements.
- 21. The permittee shall submit a compliance certification with the terms and conditions contained in the permit, including emission limitations, standards, or work practices. The permittee must submit the compliance certification annually. If the permit establishes no other reporting period, the reporting period shall end on the last day of the anniversary month of the initial Title V permit. The report is due on the first day of the second month after the end of the reporting period. The permittee must also submit the compliance certification to the Administrator as well as to the Division of Environmental Quality. All compliance certifications required by this permit must include the following: [40 C.F.R. § 70.6(c)(5) and Rule 26.703(E)(3)]
 - a. The identification of each term or condition of the permit that is the basis of the certification;
 - b. The compliance status;
 - c. Whether compliance was continuous or intermittent;
 - d. The method(s) used for determining the compliance status of the source, currently and over the reporting period established by the monitoring requirements of this permit; and
 - e. Such other facts as the Division of Environmental Quality may require elsewhere in this permit or by § 114(a)(3) and § 504(b) of the Act.
- 22. Nothing in this permit will alter or affect the following: [Rule 26.704(C)]
 - a. The provisions of Section 303 of the Act (emergency orders), including the authority of the Administrator under that section;
 - b. The liability of the permittee for any violation of applicable requirements prior to or at the time of permit issuance;
 - c. The applicable requirements of the acid rain program, consistent with § 408(a) of the Act; or
 - d. The ability of EPA to obtain information from a source pursuant to § 114 of the Act.
- 23. This permit authorizes only those pollutant emitting activities addressed in this permit. [Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311]
- 24. The permittee may request in writing and at least 15 days in advance of the deadline, an extension to any testing, compliance or other dates in this permit. No such extensions are authorized until the permittee receives written Division of Environmental Quality

approval. The Division of Environmental Quality may grant such a request, at its discretion in the following circumstances:

- a. Such an extension does not violate a federal requirement;
- b. The permittee demonstrates the need for the extension; and
- c. The permittee documents that all reasonable measures have been taken to meet the current deadline and documents reasons it cannot be met.

[Rule 18.314(A), Rule 19.416(A), Rule 26.1013(A), Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 52 Subpart E]

- 25. The permittee may request in writing and at least 30 days in advance, temporary emissions and/or testing that would otherwise exceed an emission rate, throughput requirement, or other limit in this permit. No such activities are authorized until the permittee receives written Division of Environmental Quality approval. Any such emissions shall be included in the facility's total emissions and reported as such. The Division of Environmental Quality may grant such a request, at its discretion under the following conditions:
 - a. Such a request does not violate a federal requirement;
 - b. Such a request is temporary in nature;
 - c. Such a request will not result in a condition of air pollution;
 - d. The request contains such information necessary for the Division of Environmental Quality to evaluate the request, including but not limited to, quantification of such emissions and the date/time such emission will occur;
 - e. Such a request will result in increased emissions less than five tons of any individual criteria pollutant, one ton of any single HAP and 2.5 tons of total HAPs; and
 - f. The permittee maintains records of the dates and results of such temporary emissions/testing.

[Rule 18.314(B), Rule 19.416(B), Rule 26.1013(B), Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 52 Subpart E]

- 26. The permittee may request in writing and at least 30 days in advance, an alternative to the specified monitoring in this permit. No such alternatives are authorized until the permittee receives written Division of Environmental Quality approval. The Division of Environmental Quality may grant such a request, at its discretion under the following conditions:
 - a. The request does not violate a federal requirement;
 - b. The request provides an equivalent or greater degree of actual monitoring to the current requirements; and
 - c. Any such request, if approved, is incorporated in the next permit modification application by the permittee.

[Rule 18.314(C), Rule 19.416(C), Rule 26.1013(C), Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 52 Subpart E]

27. Any credible evidence based on sampling, monitoring, and reporting may be used to determine violations of applicable emission limitations. [Rule 18.1001, Rule 19.701, Ark. Code Ann. § 8-4-203 as referenced by Ark. Code Ann. §§ 8-4-304 and 8-4-311, and 40 C.F.R. § 52 Subpart E]

Appendix A



CONTINUOUS EMISSION MONITORING SYSTEMS CONDITIONS

Division of Environmental Quality

Office of Air Quality

12/3/2020

PREAMBLE

These conditions are intended to outline the requirements for facilities required to operate Continuous Emission Monitoring Systems/Continuous Opacity Monitoring Systems (CEMS/COMS). Generally, there are three types of sources required to operate CEMS/COMS:

- 1. CEMS/COMS required by 40 C.F.R. § 60 or 63.
- 2. CEMS required by 40 C.F.R § 75.
- 3. CEMS/COMS required by permit for reasons other than § 60, 63 or 75.

These CEMS/COMS conditions are not intended to supersede 40 C.F.R. § 60, 63 or 75 requirements.

- Only CEMS/COMS in the third category (those required by the Arkansas Department of Energy and Environment's (Department) Division of Environmental Quality (DEQ) permit for reasons other than 40 C.F.R. § 60, 63 or 75) shall comply with SECTION II, <u>MONITORING</u> <u>REQUIREMENTS</u> and SECTION IV, <u>QUALITY ASSURANCE/QUALITY CONTROL</u>.
- All CEMS/COMS shall comply with Section III, <u>NOTIFICATION AND RECORDKEEPING.</u>

SECTION I

DEFINITIONS

Continuous Emission Monitoring System (CEMS) – The total equipment required for the determination of a gas concentration and/or emission rate so as to include sampling, analysis and recording of emission data.

Continuous Opacity Monitoring System (COMS) – The total equipment required for the determination of opacity as to include sampling, analysis and recording of emission data.

Calibration Drift (CD) – The difference in the CEMS output reading from the established reference value after a stated period of operation during which no uns cheduled maintenance, repair, or adjustments took place.

Back-up CEMS (Secondary CEMS) – A CEMS with the ability to sample, analyze and record stack pollutant to determine gas concentration and/or emission rate. This CEMS is to serve as a back-up to the primary CEMS to minimize monitor downtime.

Excess Emissions – Any period in which the emissions exceed the permit limits.

Monitor Downtime – Any period during which the CEMS/COMS is unable to sample, analyze and record a minimum of four evenly spaced data points over an hour, except during one daily zero-span check during which two data points per hour are sufficient.

Out-of-Control Period – Begins with the time corresponding to the completion of the fifth, consecutive, daily CD check with a CD in excess of two times the allowable limit, or the time corresponding to the completion of the daily CD check preceding the daily CD check that results in a CD in excess of four times the allowable limit and the time corresponding to the completion of the sampling for the Relative Accuracy Test Audit (RATA), Relative Accuracy Audit (RAA), or Cylinder Gas Audit (CGA) which exceeds the limits outlined in Section IV. Out-of-Control Period ends with the time corresponding to the completion of the CD check following corrective action with the results being within the allowable CD limit or the completion of the sampling of the subsequent successful RATA, RAA, or CGA.

Primary CEMS – The main reporting CEMS with the ability to sample, analyze, and record stack pollutant to determine gas concentration and/or emission rate.

Relative Accuracy (RA) – The absolute mean difference between the gas concentration or emission rate determined by the CEMS and the value determined by the reference method plus the 2.5 percent error confidence coefficient of a series of tests divided by the mean of the reference method tests of the applicable emission limit.

Span Value – The upper limit of a gas concentration measurement range.

DEQ CEM CONDITIONS
SECTION II

MONITORING REQUIREMENTS

** Only CEMS/COMS required by DEQ permit for reasons other than 40 C.F.R. § 60, 63 or 75 shall comply with this section.

- A. For new sources, the installation date for the CEMS/COMS shall be no later than thirty (30) days from the date of start-up of the source.
- B. For existing sources, the installation date for the CEMS/COMS shall be no later than sixty (60) days from the issuance of the permit unless the permit requires a specific date.
- C. Within sixty (60) days of installation of a CEMS/COMS, a performance specification test (PST) must be completed. PST's are defined in 40 C.F.R. § 60, Appendix B, PS 1-9. DEQ may accept alternate PST's for pollutants not covered by Appendix B on a case-by-case basis. Alternate PST's shall be approved, in writing, by the DEQ CEM Coordinator prior to testing.
- D. Each CEMS/COMS shall have, as a minimum, a daily zero-span check. The zero-span shall be adjusted whenever the 24-hour zero or 24-hour span drift exceeds two times the limits in the applicable performance specification in 40 C.F.R, § 60, Appendix B. Before any adjustments are made to either the zero or span drifts measured at the 24-hour interval, the excess zero and span drifts measured must be quantified and recorded.
- E. All CEMS/COMS shall be in continuous operation and shall meet minimum frequency of operation requirements of 95% up-time for each quarter for each pollutant measured. Percent of monitor down-time is calculated by dividing the total minutes the monitor is not in operation by the total time in the calendar quarter and multiplying by one hundred. Failure to maintain operation time shall constitute a violation of the CEMS conditions.
- F. Percent of excess emissions are calculated by dividing the total minutes of excess emissions by the total time the source operated and multiplying by one hundred. Failure to maintain compliance may constitute a violation of the CEMS conditions.
- G. All CEMS measuring emissions shall complete a minimum of one cycle of operation (sampling, analyzing, and data recording) for each successive fifteen-minute period unless more cycles are required by the permit. For each CEMS, one-hour averages shall be computed from four or more data points equally spaced over each one-hour period unless more data points are required by the permit.
- H. All COMS shall complete a minimum of one cycle of sampling and analyzing for each successive 10-second period and one cycle of data recording for each successive 6-minute period.

I. When the pollutant from a single affected facility is released through more than one point, a CEMS/COMS shall be installed on each point unless installation of fewer systems is approved, in writing, by the DEQ CEM Coordinator. When more than one CEM/COM is used to monitor emissions from one affected facility the owner or operator shall report the results as required from each CEMS/COMS.

SECTION III

NOTIFICATION AND RECORD KEEPING

** All CEMS/COMS shall comply with this section.

- A. When requested to do so by an owner or operator, the DEQ CEM Coordinator will review plans for installation or modification for the purpose of providing technical advice to the owner or operator.
- B. Each facility which operates a CEMS/COMS shall notify the DEQ CEM Coordinator of the date for which the demonstration of the CEMS/COMS performance will commence (i.e. PST, RATA, RAA, CGA). Notification shall be received in writing no less than 15 business days prior to testing. Performance test results shall be submitted to DEQ within thirty days after completion of testing.
- C. Each facility which operates a CEMS/COMS shall maintain records of the occurrence and duration of start up/shut down, cleaning/soot blowing, process problems, fuel problems, or other malfunction in the operation of the affected facility which causes excess emissions. This includes any malfunction of the air pollution control equipment or any period during which a continuous monitoring device/system is inoperative.
- D. Each facility required to install a CEMS/COMS shall submit an excess emission and monitoring system performance report to DEQ (Attention: DEQ, Office of Air Quality, CEM Coordinator) at least quarterly, unless more frequent submittals are warranted to assess the compliance status of the facility. Quarterly reports shall be postmarked no later than the 30th day of the month following the end of each calendar quarter.
- E. All excess emissions shall be reported in terms of the applicable standard. Each report shall be submitted on DEQ Quarterly Excess Emission Report Forms. Alternate forms may be used with prior written approval from DEQ.
- F. Each facility which operates a CEMS/COMS must maintain on site a file of CEMS/COMS data including all raw data, corrected and adjusted, repair logs, calibration checks, adjustments, and test audits. This file must be retained for a period of at least five years and is required to be maintained in such a condition that it can easily be audited by an inspector.
- G. Quarterly reports shall be used by DEQ to determine compliance with the permit.

SECTION IV

QUALITY ASSURANCE/QUALITY CONTROL

** Only CEMS/COMS required by DEQ permit for reasons other than 40 C.F.R. § 60, 63 or 75 shall comply with this section.

- A. For each CEMS/COMS a Quality Assurance/Quality Control (QA/QC) plan shall be submitted to DEQ (Attn.: DEQ, Office of Air Quality, CEM Coordinator). CEMS quality assurance procedures are defined in 40 C.F.R. § 60, Appendix F. This plan shall be submitted within 180 days of the CEMS/COMS installation. A QA/QC plan shall consist of procedure and practices which assures acceptable level of monitor data accuracy, precision, representativeness, and availability.
- B. The submitted QA/QC plan for each CEMS/COMS shall not be considered as accepted until the facility receives a written notification of acceptance from DEQ.
- C. Facilities responsible for one or more CEMS/COMS used for compliance monitoring shall meet these minimum requirements and are encouraged to develop and implement a more extensive QA/QC program, or to continue such programs where they already exist. Each QA/QC program must include written procedures which should describe in detail, complete, step-by-step procedures and operations for each of the following activities:
 - 1. Calibration of CEMS/COMS
 - a. Daily calibrations (including the approximate time(s) that the daily zero and span drifts will be checked, and the time required to perform these checks and return to stable operation)
 - 2. Calibration drift determination and adjustment of CEMS/COMS
 - a. Out-of-control period determination
 - b. Steps of corrective action
 - 3. Preventive maintenance of CEMS/COMS
 - a. CEMS/COMS information
 - 1) Manufacture
 - 2) Model number
 - 3) Serial number
 - b. Scheduled activities (check list)
 - c. Spare part inventory
 - 4. Data recording, calculations, and reporting
 - 5. Accuracy audit procedures including sampling and analysis methods
 - 6. Program of corrective action for malfunctioning CEMS/COMS

D. A Relative Accuracy Test Audit (RATA) shall be conducted at least once every four calendar quarters. A Relative Accuracy Audit (RAA), or a Cylinder Gas Audit (CGA), may be conducted in the other three quarters but in no more than three quarters in succession. The RATA should be conducted in accordance with the applicable test procedure in 40 C.F.R. § 60 Appendix A and calculated in accordance with the applicable performance specification in 40 C.F.R. § 60 Appendix B. CGA's and RAA's should be conducted and the data calculated in accordance with the procedure of 40 C.F.R. § 60 Appendix B. CGA's and RAA's should be conducted and the data calculated in accordance with the procedures outlined on 40 C.F.R. § 60 Appendix F.

If alternative testing procedures or methods of calculation are to be used in the RATA, RAA or CGA audits prior authorization must be obtained from the DEQ CEM Coordinator.

E. Criteria for excessive audit inaccuracy.

All Pollutants except Carbon Monoxide	> 20% Relative Accuracy		
Carbon Monoxide	> 10% Relative Accuracy		
All Pollutants except Carbon Monoxide	> 10% of the Applicable Standard		
Carbon Monoxide	> 5% of the Applicable Standard		
Diluent ($O_2 \& CO_2$)	> 1.0 % O ₂ or CO ₂		
Flow	> 20% Relative Accuracy		

RATA

CGA

Pollutant	> 15% of average audit value or 5 ppm difference
Diluent ($O_2 \& CO_2$)	> 15% of average audit value or 5 ppm difference

RAA

Pollutant	> 15% of the three-run average or $>$ 7.5 % of the applicable standard
Diluent ($O_2 \& CO_2$)	> 15% of the three-run average or > 7.5 % of the applicable standard

- F. If either the zero or span drift results exceed two times the applicable drift specification in 40 C.F.R. § 60, Appendix B for five consecutive, daily periods, the CEMS is out-of-control. If either the zero or span drift results exceed four times the applicable drift specification in Appendix B during a calibration drift check, the CEMS is out-of-control. If the CEMS exceeds the audit inaccuracies listed above, the CEMS is out-of-control. If a CEMS is out-of-control, the data from that out-of-control period is not counted towards meeting the minimum data availability as required and described in the applicable subpart. The end of the out-of-control period is the time corresponding to the completion of the successful daily zero or span drift or completion of the successful CGA, RAA or RATA.
- G. A back-up monitor may be placed on an emission source to minimize monitor downtime. This back-up CEMS is subject to the same QA/QC procedure and practices as the primary CEMS. The back-up CEMS shall be certified by a PST. Daily zero-span checks must be performed and recorded in accordance with standard practices. When the primary CEMS goes down, the back-up CEMS may then be engaged to sample, analyze, and record the emission source pollutant until repairs are made and the primary unit is placed back in service. Records must be maintained on site when the back-up CEMS is out of service, these records shall include at a minimum the reason the primary CEMS is out of service, the date and time the primary CEMS was placed back in service.

Appendix B

Dioxins/Furans	TEFs 1989	
1,2,3,4,7,8-HxCDF	0.1	
1,2,3,6,7,8-HxCDF	0.1	
1,2,3,7,8,9-HxCDF	0.1	
2.3.4.6.7.8-HxCDF	0.1	

§63.1360

83	\mathbf{FR}	35136,	July	25,	2018]	

AUTHENTICATED U.S. GOVERNMENT INFORMATION

Subpart MMM—National Emission Standards for Hazardous Air Pollutants for Pesticide Active Ingredient Production

SOURCE: 64 FR 33589, June 23, 1999, unless otherwise noted.

§63.1360 Applicability.

(a) Definition of affected source. The affected source subject to this subpart is the facility-wide collection of pesticide active ingredient manufacturing process units (PAI process units) that process, use, or produce HAP, and are located at a plant site that is a major source, as defined in section 112(a) of the CAA. An affected source also includes waste management units, heat exchange systems, and cooling towers that are associated with the PAI process units. Exemptions from an affected source are specified in paragraph (d) of this section.

(b) New source applicability. A new affected source subject to this subpart and to which the requirements for new sources apply is defined according to the criteria in paragraph (b)(1) or (2) of this section.

(1) An affected source for which construction or reconstruction commenced after November 10, 1997.

(2) Any dedicated PAI process unit that meets the criteria specified in paragraphs (b)(2)(i) and (ii) of this section.

(i) For which construction, as defined in §63.1361, commenced after November 10, 1997, or reconstruction commenced after September 20, 2002.

(ii) That has the potential to emit 10 tons/yr of any one HAP or 25 tons/yr of combined HAP.

(c) *General provisions*. Table 1 of this subpart specifies the provisions of subpart A of this part that apply to an owner or operator of an affected source subject to this subpart, and clarifies
 Dioxins/Furans
 TEFs 1989

 1,2,3,4,6,7,8-HpCDF
 0.01

 1,2,3,4,7,8,9-HpCDF
 0.01

 OCDF
 0.001

specific provisions in subpart A of this part as necessary for this subpart.

(d) Exemptions from the requirements of this subpart. The provisions of this subpart do not apply to:

(1) Research and development facilities;

(2) PAI process units that are subject to subpart F of this part;

(3) Production of ethylene;

(4) Coal tar distillation; and

(5) The following emission points listed:

(i) Storm water from segregated sewers;

(ii) Water from fire-fighting and deluge systems, including testing of such systems;

(iii) Spills;

(iv) Water from safety showers;(v) Noncontact steam boiler blow-

down and condensate;

(vi) Laundry water;

(vii) Vessels storing material that contains no organic HAP or contains organic HAP as impurities only; and

(viii) Equipment, as defined in §63.1363, that is intended to operate in organic HAP service for less than 300 hours during the calendar year.

(e) Applicability of this subpart. (1) Each provision set forth in this subpart shall apply at all times except during periods of non-operation of the affected source (or specific portion thereof) resulting in cessation of the emissions to which this subpart applies.

(i) The startup, shutdown, or malfunction precludes the ability of the owner or operator of an affected source to comply with one or more specific emission limitations to which a particular emission point is subject; and

(ii) The owner or operator follows the provisions for periods of startup, shutdown, and malfunction, as specified in $\S 63.1367(a)(3)$ and 63.1368(i).

(2) The provisions set forth in §63.1363 shall apply at all times except during periods of nonoperation of the PAI process unit (or specific portion thereof) in which the lines are drained and depressurized resulting in the cessation of the emissions to which §63.1363 applies.

(3) The owner or operator shall not shut down items of equipment that are required or utilized for compliance with the emissions limitations of this subpart during times when emissions (or, where applicable, wastewater streams or residuals) are being routed to such items of equipment, if the shutdown would contravene emissions limitations of this subpart applicable to such items of equipment.

(4) General duty. At all times, the owner or operator must operate and maintain any affected source, including associated air pollution control equipment and monitoring equipment, in a manner consistent with safety and good air pollution control practices for minimizing emissions. The general duty to minimize emissions does not require the owner or operator to make any further efforts to reduce emissions if levels required by the applicable standard have been achieved. Determination of whether a source is operating in compliance with operation and maintenance requirements will be based on information available to the Administrator, which may include, but is not limited to, monitoring results, review of operation and maintenance procedures, review of operation and maintenance records, and inspection of the source.

(f) Storage vessel applicability determination. An owner or operator shall follow the procedures specified in paragraphs (f)(1) through (5) of this section to determine whether a storage vessel is part of the affected source to which this subpart applies.

(1) If a storage vessel is already subject to another subpart of 40 CFR part 63 on June 23, 1999, the storage vessel shall belong to the process unit subject to the other subpart.

(2) Unless otherwise excluded under paragraph (f)(1) of this section, the storage vessel is part of a PAI process unit if either the input to the vessel from the PAI process unit is greater than or equal to the input from any other PAI or non-PAI process unit, or the output from the vessel to the PAI 40 CFR Ch. I (7–1–23 Edition)

process unit is greater than or equal to the output to any other PAI or non-PAI process unit. If the greatest input to and/or output from a shared storage vessel is the same for two or more process units, including one or more PAI process units, the owner or operator must assign the storage vessel to any one of the PAI process units that meet this condition.

(3) Unless otherwise excluded under paragraph (f)(1) of this section, where a storage vessel is located in a tank farm (including a marine tank farm), the applicability of this subpart shall be determined according to the provisions in paragraphs (f)(3)(i) through (iii) of this section.

(i) The storage vessel in the tank farm is not subject to the provisions of this subpart if the greatest input to or output from the storage vessel is for a non-PAI process unit. The input and output shall be determined among only those process units that share the storage vessel and that do not have an intervening storage vessel for that product (or raw material, as appropriate).

(ii) Except for storage vessels in a tank farm excluded in accordance with paragraph (f)(3)(i) of this section, applicability of this subpart shall be determined according to the provisions in paragraphs (f)(3)(i)(A) through (C) of this section.

(A) Except as specified in paragraph (f)(3)(ii)(C) of this section, this subpart does not apply to the storage vessel in a tank farm if each PAI process unit that receives material from or sends material to the storage vessel has an intervening storage vessel for that material.

(B) Except as specified in paragraph (f)(3)(ii)(C) of this section, a storage vessel in a tank farm shall be assigned to the PAI process unit that receives the greatest amount of material from or sends the greatest amount of material to the storage vessel and does not have an intervening storage vessel. If two or more PAI process units have the same input to or output from the storage vessel in the tank farm, then the storage vessel in the tank farm may be assigned to any one of the PAI process units that meet this condition.

(C) As an alternative to the requirements specified in paragraphs

(f)(3)(ii)(A) and (B) of this section, even if an intervening storage vessel is present, an owner or operator may elect to assign a storage vessel in a tank farm to the PAI process unit that sends the most material to or receives the most material from the storage vessel. If two or more PAI process units have the same input to or output from the storage vessel in the tank farm, then the storage vessel in the tank farm may be assigned to any one of the PAI process units that meet this condition.

(iii) With respect to a process unit, an intervening storage vessel means a storage vessel connected by hard-piping to the process unit and to the storage vessel in the tank farm so that the product or raw material entering or leaving the process flows into (or from) the intervening storage vessel and does not flow directly into (or from) the storage vessel in the tank farm.

(4) If use varies from year to year, then use for the purposes of this subpart for existing sources shall be based on the utilization that occurred during the year preceding June 23, 1999, or if the storage vessel was not in operation during that year, the use shall be based on the expected use in the 5 years after startup. This determination shall be reported as part of an operating permit application or as otherwise specified by the permitting authority.

(5) If the storage vessel begins receiving material from (or sending material to) another process unit, or ceasing to receive material from (or send material to) a PAI process unit, or if there is a significant change in the use of the storage vessel, the owner or operator shall reevaluate the ownership determination for the storage vessel.

(g) Designating production of an intermediate as a PAI process unit. Except as specified in paragraph (d) of this section, an owner or operator may elect to designate production of any intermediate that does not meet the definition of integral intermediate as a PAI process unit subject to this subpart. Any storage vessel containing the intermediate is assigned to a PAI process unit according to the procedures in paragraph (f) of this section. Any process tank containing the intermediate is part of the process unit used to produce the intermediate.

(h) Applicability of process units included in a process unit group. An owner or operator may elect to develop process unit groups in accordance with paragraph (h)(1) of this section. For the PAI process units in these process unit groups, the owner or operator may comply with the provisions in overlapping MACT standards, as specified in paragraphs (h)(2) through (4) of this section, as an alternative means of demonstrating compliance with the provisions of this subpart.

(1) Develop, revise, and document changes in a process unit group in accordance with the procedures specified in paragraphs (h)(1)(i) through (vi) of this section.

(i) Initially identify a non-dedicated PAI process unit that is operating on December 23, 2003 or a date after December 23, 2003, and identify all processing equipment that is part of this PAI process unit, based on descriptions in operating scenarios.

(ii) Add to the group any other nondedicated PAI and non-dedicated non-PAI process units expected to be operated in the 5 years after the date specified in paragraph (h)(1)(i) of this section, provided they satisfy the criteria specified in paragraphs (h)(1)(ii)(A)through (C) of this section. Also identify all of the processing equipment used for each process unit based on information from operating scenarios and other applicable documentation.

(A) Each PAI process unit that is added to a group must have some processing equipment that is part of one or more PAI process units that are already in the process unit group.

(B) Each non-PAI process unit that is added to a group must have some processing equipment that is also part of one or more of the PAI process units in the group.

(C) No process unit may be part of more than one process unit group.

(iii) The initial process unit group consists of all of the processing equipment for the process units identified in paragraphs (h)(1)(i) and (ii) of this section.

(iv) If compliance is to be demonstrated in accordance with paragraph (h)(3) of this section, determine the primary product of the process unit group according to the procedures specified in paragraphs (h)(1)(iv)(A) through (C) of this section.

(A) The primary product is the type of product (e.g., PAI, pharmaceutical product, thermoplastic resin, etc.) that is expected to be produced for the greatest operating time in the 5-year period specified in paragraph (h)(1)(i) of this section.

(B) If the process unit group produces multiple products equally based on operating time, then the primary product is the product with the greatest production on a mass basis over the 5-year period specified in paragraph (h)(1)(i) of this section.

(C) The primary product of the group must be redetermined if the owner or operator does not intend to make that product in the future or if it has not been made for 5 years. The results of the redetermination must be recorded as specified in §63.1367(b) and reported in a Periodic report no later than the report covering the period for the end of the 5th year as specified in §63.1368(g)(2). If the primary product changes, the owner or operator must either demonstrate compliance with the applicable subpart as specified in paragraph (h)(3) of this section or demonstrate compliance with the provisions of this subpart MMM.

(v) Add process units developed in the future in accordance with the conditions specified in paragraphs (h)(1)(ii)(A) through (C) of this section.

(vi) Maintain records of changes in the process units in each process unit group as specified in 63.1367(b)(9), and maintain reports as specified in 63.1368(f)(9) and (g)(2)(ix).

(2) If any of the products produced in the process unit group are subject to 40 CFR part 63, subpart GGG (Pharmaceuticals MACT), the owner or operator may elect to comply with the requirements of subpart GGG for the PAI process unit(s) within the process unit group, except for the following:

(i) The emission limit standard for process vents in 63.1362(b)(2)(i) shall apply in place of 63.1254(a)(2);

(ii) When the dates of April 2, 1997 and April 2, 2007 are provided in §63.1254(a)(3)(ii), the dates of November 10, 1997 and November 10, 2007, respec40 CFR Ch. I (7–1–23 Edition)

tively, shall apply for purposes of this subpart MMM; and

(iii) Requirements in $\S63.1367(a)(5)$ regarding application for approval of construction or reconstruction shall apply in place of the provisions in $\S63.1259(a)(5)$.

(3) If the primary product of a process unit group is determined to be a type of material that is subject to another subpart of 40 CFR part 63 on June 23, 1999 or startup of the first process unit after formation of the process unit group, whichever is later, the owner or operator may elect to comply with the other subpart for any PAI process unit within the process unit group, subject to the requirement in this paragraph (h)(3). Emissions from PAI Group 1 process vents, as defined in §63.1361, must be reduced in accordance with the control requirements for Group 1 vents as specified in the alternative subpart. The criteria in the alternative subpart for determining which process vents must be controlled do not apply for the purposes of this paragraph (h)(3).

(4) The requirements for new and reconstructed sources in the alternative subpart apply to all PAI process units in the process unit group if, and only if, the affected source under the alternative subpart meets the requirements for construction or reconstruction.

(i) Overlap with other regulations—(1) Compliance with other MACT standards.
(i) After the compliance dates specified in §63.1364, an affected source subject to the provisions of this subpart that is also subject to the provisions of any other subpart of 40 CFR part 63 may elect, to the extent the subparts are consistent, under which subpart to maintain records and report to EPA. The affected source shall identify in the Notification of Compliance Status report required by §63.1368(f) under which authority such records will be maintained.

(ii) After the compliance dates specified in §63.1364, at an offsite reloading or cleaning facility subject to §63.1362(b)(6), compliance with the emission standards and associated initial compliance monitoring, recordkeeping, and reporting provisions of any other subpart of 40 CFR part 63 constitutes compliance with the provisions of §63.1362(b)(6)(vii)(B) or (C). The

owner or operator of the affected storage vessel shall identify in the Notification of Compliance Status report required by §63.1368(f) the subpart of 40 CFR part 63 with which the owner or operator of the offsite reloading or cleaning facility complies.

(2) Overlap with RCRA subparts AA, BB, and/or CC. After the compliance dates specified in §63.1364, if any affected source subject to this subpart is also subject to monitoring, recordkeeping, and reporting requirements in 40 CFR part 264, subpart AA, BB, or CC, or is subject to monitoring and recordkeeping requirements in 40 CFR part 265, subpart AA, BB, or CC, and the owner or operator complies with the periodic reporting requirements under 40 CFR part 264, subpart AA, BB, or CC that would apply to the device if the facility had final-permitted status, the owner or operator may elect to comply either with the monitoring, recordkeeping, and reporting requirements of this subpart, or with the monitoring, recordkeeping, and reporting requirements in 40 CFR parts 264 and/or 265, as described in this paragraph, which shall constitute compliance with the monitoring, recordkeeping, and reporting requirements of this subpart. If the owner or operator elects to comply with the monitoring, recordkeeping, and reporting requirements in 40 CFR parts 264 and/or 265, the owner or operator shall report all excursions as required by §63.1368(g). The owner or operator shall identify in the Notification of Compliance Status report required by §63.1368(f) the monitoring, recordkeeping, and reporting authority under which the owner or operator will comply.

(3) Overlap with NSPS subpart Kb. After the compliance dates specified in §63.1364, a Group 1 or Group 2 storage vessel that is also subject to the provisions of 40 CFR part 60, subpart Kb, is required to comply only with the provisions of this subpart MMM.

(4) Overlap with subpart I. After the compliance dates specified in §63.1364, for all equipment within a process unit that contains equipment subject to subpart I of this part, an owner or operator may elect to comply with either the provisions of this subpart MMM or the provisions of subpart H of this part.

The owner or operator shall identify in the Notification of Compliance Status report required by §63.1368(f) the provisions with which the owner or operator elects to comply.

(5) Overlap with RCRA regulations for wastewater. After the compliance dates specified in §63.1364, the owner or operator of an affected wastewater stream that is also subject to provisions in 40 CFR parts 260 through 272 shall comply with the more stringent control requirements (e.g., waste management units, numerical treatment standards, etc.) and the more stringent testing, monitoring, recordkeeping, and reporting requirements that overlap between the provisions of this subpart and the provisions of 40 CFR parts 260 through 272. The owner or operator shall keep a record of the information used to determine which requirements were the most stringent and shall submit this information if requested by the Administrator.

(6) Overlap with NSPS subparts III, NNN, and RRR. After the compliance dates specified in §63.1364, if an owner or operator of a process vent subject to this subpart MMM that is also subject to the provisions of 40 CFR part 60, subpart III, or subpart NNN, or subpart RRR, elects to reduce organic HAP emissions from the process vent by 98 percent as specified in §63.1362(b)(2)(iii)(A), then the owner or operator is required to comply only with the provisions of this subpart MMM. Otherwise, the owner or operator shall comply with the provisions in both this subpart MMM and the provisions in 40 CFR part 60, subparts III, NNN, and RRR, as applicable.

(j) Meaning of periods of time. All terms in this subpart MMM that define a period of time for completion of required tasks (e.g., weekly, monthly, quarterly, annual), unless specified otherwise in the section or subsection that imposes the requirement, refer to the standard calendar periods.

(1) Notwithstanding time periods specified in the subpart MMM for completion of required tasks, such time periods may be changed by mutual agreement between the owner and operator and the Administrator, as specified in subpart A of this part (e.g., a period could begin on the compliance date or

§63.1360

another date, rather than on the first day of the standard period). For each time period that is changed by agreement, the revised period shall remain in effect until it is changed. A new request is not necessary for each recurring period.

(2) Where the period specified for compliance is a standard calendar period, if the initial compliance date occurs after the beginning of the period, compliance shall be required according to the schedule specified in paragraph (j)(2)(i) or (ii) of this section, as appropriate.

(i) Compliance shall be required before the end of the standard calendar period within which the compliance deadline occurs, if there remain at least 3 days for tasks that must be performed weekly, at least 2 weeks for tasks that must be performed monthly, at least 1 month for tasks that must be performed each quarter, or at least 3 months for tasks that must be performed annually; or

(ii) In all other cases, compliance shall be required before the end of the first full standard calendar period within which the initial compliance deadline occurs.

(3) In all instances where a provision of this subpart MMM requires completion of a task during each of multiple successive periods, an owner or operator may perform the required task at any time during the specified period, provided the task is conducted at a reasonable interval after completion of the task in the previous period.

(k) Affirmative defense for violation of emission standards during malfunction. In response to an action to enforce the standards set forth in this subpart, the owner or operator may assert an affirmative defense to a claim for civil penalties for violations of such standards that are caused by malfunction, as defined at §63.2. Appropriate penalties may be assessed if the owner or operator fails to meet their burden of proving all of the requirements in the affirmative defense. The affirmative defense shall not be available for claims for injunctive relief.

(1) Assertion of affirmative defense. To establish the affirmative defense in any action to enforce such a standard, the owner or operator must timely 40 CFR Ch. I (7–1–23 Edition)

meet the reporting requirements in paragraph (k)(2) of this section, and must prove by a preponderance of evidence that:

(i) The violation:

(A) Was caused by a sudden, infrequent, and unavoidable failure of air pollution control equipment, process equipment, or a process to operate in a normal or usual manner; and

(B) Could not have been prevented through careful planning, proper design or better operation and maintenance practices; and

(C) Did not stem from any activity or event that could have been foreseen and avoided, or planned for; and

(D) Was not part of a recurring pattern indicative of inadequate design, operation, or maintenance; and

(ii) Repairs were made as expeditiously as possible when a violation occurred; and

(iii) The frequency, amount, and duration of the violation (including any bypass) were minimized to the maximum extent practicable; and

(iv) If the violation resulted from a bypass of control equipment or a process, then the bypass was unavoidable to prevent loss of life, personal injury, or severe property damage; and

(v) All possible steps were taken to minimize the impact of the violation on ambient air quality, the environment, and human health; and

(vi) All emissions monitoring and control systems were kept in operation if at all possible, consistent with safety and good air pollution control practices; and

(vii) All of the actions in response to the violation were documented by properly signed, contemporaneous operating logs; and

(viii) At all times, the affected source was operated in a manner consistent with good practices for minimizing emissions; and

(ix) A written root cause analysis has been prepared, the purpose of which is to determine, correct, and eliminate the primary causes of the malfunction and the violation resulting from the malfunction event at issue. The analysis shall also specify, using best monitoring methods and engineering judgment, the amount of any emissions

that were the result of the malfunction.

(2) Report. The owner or operator seeking to assert an affirmative defense shall submit a written report to the Administrator, with all necessary supporting documentation, that explains how it has met the requirements set forth in paragraph (k)(1) of this section. This affirmative defense report shall be included in the first periodic compliance report, deviation report, or excess emission report otherwise required after the initial occurrence of the violation of the relevant standard (which may be the end of any applicable averaging period). If such compliance, deviation report or excess emission report is due less than 45 days after the initial occurrence of the violation, the affirmative defense report may be included in the second compliance, deviation report or excess emission report due after the initial occurrence of the violation of the relevant standard.

[64 FR 33589, June 23, 1999, as amended at 67 FR 59340, Sept. 20, 2002; 79 FR 17371, Mar. 27, 2014]

§63.1361 Definitions.

Terms used in this subpart are defined in the CAA, in subpart A of this part, or in this section. If the same term is defined in subpart A of this part and in this section, it shall have the meaning given in this section for the purposes of this subpart MMM.

Affirmative defense means, in the context of an enforcement proceeding, a response or defense put forward by a defendant, regarding which the defendant has the burden of proof, and the merits of which are independently and objectively evaluated in a judicial or administrative proceeding.

Air pollution control device or control device means equipment installed on a process vent, storage vessel, wastewater treatment exhaust stack, or combination thereof that reduces the mass of HAP emitted to the air. The equipment may consist of an individual device or a series of devices. Examples include incinerators, carbon adsorption units, condensers, flares, boilers, process heaters, and gas absorbers. Process condensers are not considered air pollution control devices or control devices.

Bag dump means equipment into which bags or other containers containing a powdered, granular, or other solid feedstock material are emptied. A bag dump is part of the process.

Batch emission episode means a discrete venting episode that is associated with a single unit operation. A unit operation may have more than one batch emission episode. For example, a batch distillation unit operation may consist of batch emission episodes associated with charging and heating. Charging the vessel with HAP will result in one discrete batch emission episode that will last through the duration of the charge and will have an average flowrate equal to the rate of the charge. Another discrete batch emission episode will result from the expulsion of expanded vapor as the contents of the vessel are heated.

Batch operation means a noncontinuous operation involving intermittent or discontinuous feed into PAI or integral intermediate manufacturing equipment, and, in general, involves the emptying of the equipment after the batch operation ceases and prior to beginning a new operation. Addition of raw material and withdrawal of product do not occur simultaneously in a batch operation. A batch process consists of a series of batch operations.

Bench-scale batch process means a batch process (other than a research and development facility) that is capable of being located on a laboratory bench top. This bench-scale equipment will typically include reagent feed vessels, a small reactor and associated product separator, recovery and holding equipment. These processes are only capable of producing small quantities of product.

Block means a time period equal to, at a maximum, the duration of a single batch.

Car seal means a seal that is placed on a device that is used to change the position of a valve (e.g., from opened to closed) in such a way that the position of the valve cannot be changed without breaking the seal.

§63.1361

Cleaning operation means routine rinsing, washing, or boil-off of equipment in batch operations between batches.

Closed-loop system means an enclosed system that returns process fluid to the process and is not vented to the atmosphere except through a closed-vent system.

Closed-purge system means a system or combination of system and portable containers, to capture purged liquids. Containers must be covered or closed when not being filled or emptied.

Closed-vent system means a system that is not open to the atmosphere and is composed of piping, ductwork, connections, and, if necessary, flow inducing devices that transport gas or vapor from an emission point to a control device.

Combustion device means an individual unit of equipment, such as a flare, incinerator, process heater, or boiler, used for the combustion of organic HAP vapors.

Connector means flanged, screwed, or other joined fittings used to connect two pipe lines or a pipe line and a piece of equipment. A common connector is a flange. Joined fittings welded completely around the circumference of the interface are not considered connectors for the purpose of this regulation. For the purpose of reporting and record keeping, connector means joined fittings that are not inaccessible, ceramic, or ceramic-lined as de-§63.1255(b)(1)(vii) scribed in and 63.1255(f)(3).

Construction means the onsite fabrication, erection, or installation of an affected source or dedicated PAI process unit. Addition of new equipment to an affected source does not constitute construction, provided the new equipment is not a dedicated PAI process unit with the potential to emit 10 tons/ yr of any one HAP or 25 tons/yr of combined HAP, but it may constitute reconstruction of the affected source or PAI process unit if it satisfies the definition of reconstruction in this section. At an affected source, changing raw materials processed and reconfiguring non-dedicated equipment to create a non-dedicated PAI process unit do not constitute construction.

40 CFR Ch. I (7–1–23 Edition)

Consumption means the quantity of all HAP raw materials entering a process in excess of the theoretical amount used as reactant, assuming 100 percent stoichiometric conversion. The raw materials include reactants, solvents, and any other additives. If HAP are generated in the process as well as added as raw material, consumption includes the quantity generated in the process.

Container, as used in the wastewater provisions, means any portable waste management unit that has a capacity greater than or equal to 0.1 m³ in which a material is stored, transported, treated, or otherwise handled. Examples of containers are drums, barrels, tank trucks, barges, dumpsters, tank cars, dump trucks, and ships.

Continuous process means a process where the inputs and outputs flow continuously throughout the duration of the process. Continuous processes typically approach steady state.

Continuous seal means a seal that forms a continuous closure that completely covers the space between the wall of the storage vessel and the edge of the floating roof. A continuous seal may be a vapor-mounted, liquidmounted, or metallic shoe seal.

Controlled HAP emissions means the quantity of HAP components discharged to the atmosphere from an air pollution control device.

Cover, as used in the wastewater provisions, means a device or system which is placed on or over a waste management unit containing wastewater or residuals so that the entire surface area is enclosed to minimize air emissions. A cover may have openings necessary for operation, inspection, and maintenance of the waste management unit such as access hatches, sampling ports, and gauge wells provided that each opening is closed when not in use. Examples of covers include a fixed roof installed on a wastewater tank, a lid installed on a container, and an air-supported enclosure installed over a waste management unit

Dedicated PAI process unit means a PAI process unit constructed from equipment that is fixed in place and designed and operated to produce only a single product or co-products. The

equipment is not designed to be reconfigured to create different process units, and it is not operated with different raw materials so as to produce different products.

Double block and bleed system means two block valves connected in series with a bleed valve or line that can vent the line between the two block valves.

Duct work means a conveyance system such as those commonly used for heating and ventilation systems. It is often made of sheet metal and often has sections connected by screws or crimping. Hard-piping is not ductwork.

Equipment, for purposes of §63.1363, means each pump, compressor, agitator, pressure relief device, sampling connection system, open-ended valve or line, valve, connector, and instrumentation system in organic hazardous air pollutant service.

External floating roof means a pontoon-type or double-deck type cover that rests on the liquid surface in a storage tank or waste management unit with no fixed roof.

FIFRA means the Federal Insecticide, Fungicide, and Rodenticide Act.

Fill or filling means the introduction of organic HAP into a storage tank or the introduction of a wastewater stream or residual into a waste management unit, but not necessarily to complete capacity.

First attempt at repair means to take action for the purpose of stopping or reducing leakage of organic material to the atmosphere.

Fixed roof means a cover that is mounted on a waste management unit or storage tank in a stationary manner and that does not move with fluctuations in liquid level.

Flame ionization detector (FID) means a device in which the measured change in conductivity of a standard flame (usually hydrogen) due to the insertion of another gas or vapor is used to detect the gas or vapor.

Floating roof means a cover consisting of a double deck, pontoon single deck, internal floating cover or covered floating roof, which rests upon and is supported by the liquid being contained, and is equipped with a continuous seal or seals to close the space between the roof edge and waste management unit or storage vessel wall. *Flow indicator* means a device that indicates whether gas flow is, or whether the valve position would allow gas flow to be, present in a line.

Formulation of pesticide products means the mixing, blending, or diluting of a PAI with one or more other PAI or inert ingredients.

Group 1 process vent means any process vent from a process at an existing or new affected source for which the uncontrolled organic HAP emissions from the sum of all process vents are greater than or equal to 0.15 Mg/yr and/ or the uncontrolled hydrogen chloride (HCl) and chlorine emissions from the sum of all process vents are greater than or equal to 6.8 Mg/yr.

Group 1 storage vessel means a storage vessel at an existing affected source with a capacity equal to or greater than 75 m^3 and storing material with a maximum true vapor pressure greater than or equal to 3.45 kPa, a storage vessel at a new affected source with a capacity equal to or greater than 40 m³ and storing material with a maximum true vapor pressure greater than or equal to 16.5 kPa, or a storage vessel at a new affected source with a capacity greater than or equal to 75 m³ and storing material with a maximum true vapor pressure greater than or equal to 3.45 kPa.

Group 1 wastewater stream means process wastewater at an existing or new source that meets the criteria for Group 1 status in §63.132(c) for compounds in Table 9 of subpart G of this part or a maintenance wastewater stream that contains 5.3 Mg of compounds in Table 9 of subpart G of this part per discharge event.

Group 2 process vent means any process vent that does not meet the definition of a Group 1 process vent.

Group 2 storage vessel means a storage vessel that does not meet the definition of a Group 1 storage vessel.

Group 2 wastewater stream means any wastewater stream that does not meet the definition of a Group 1 wastewater stream.

Group of processes means all of the equipment associated with processes in a building, processing area, or facility-wide. A group of processes may consist of a single process.

Halogenated compounds means organic compounds that contain chlorine atoms.

Halogenated vent stream means a process, storage vessel, or waste management unit vent stream determined to have a concentration of halogenated compounds of greater than 20 ppmv, as determined through process knowledge, test results using Method 18 of 40 CFR part 60, appendix A, or test results using any other test method that has been validated according to the procedures in Method 301 of appendix A of this part.

Hard-piping means piping or tubing that is manufactured and properly installed using good engineering judgment and standards, such as ANSI B31– 3.

Impurity means a substance that is produced coincidentally with the product(s), or is present in a raw material. An impurity does not serve a useful purpose in the production or use of the product(s) and is not isolated.

In gas/vapor service means that a piece of equipment in organic HAP service contains a gas or vapor at operating conditions.

In heavy liquid service means that a piece of equipment in organic HAP service is not in gas/vapor service or in light liquid service.

In light liquid service means that a piece of equipment in organic HAP service contains a liquid that meets the following conditions:

(1) The vapor pressure of one or more of the organic compounds is greater than 0.3 kPa at 20 °C;

(2) The total concentration of the pure organic compounds constituents having a vapor pressure greater than 0.3 kPa at 20 °C is equal to or greater than 20 percent by weight of the total process stream; and

(3) The fluid is a liquid at operating conditions.

NOTE: To definition of "In light liquid service: Vapor pressures may be determined by the methods described in 40 CFR 60.485(e)(1).

In liquid service means that a piece of equipment in organic HAP service is not in gas/vapor service.

In organic hazardous air pollutant or in organic HAP service means that a piece of equipment either contains or contacts a fluid (liquid or gas) that is at 40 CFR Ch. I (7–1–23 Edition)

least 5 percent by weight of total organic HAP as determined according to the provisions of §63.180(d) of subpart H of this part. The provisions of §63.180(d) of subpart H of this part also specify how to determine that a piece of equipment is not in organic HAP service.

In vacuum service means that equipment is operating at an internal pressure which is at least 5 kPa below ambient pressure.

In-situ sampling systems means nonextractive samplers or in-line samplers.

Individual drain system means the stationary system used to convey wastewater streams or residuals to a waste management unit or to discharge or disposal. The term includes: hard piping; all process drains and junction boxes; and associated sewer lines, other junction boxes, manholes, sumps, and lift stations conveying wastewater streams or residuals. A segregated stormwater sewer system, which is a drain and collection system designed and operated for the sole purpose of collecting rainfall-runoff at a facility, and which is segregated from all other individual drain systems, is excluded from this definition.

Instrumentation system means a group of equipment components used to condition and convey a sample of the process fluid to analyzers and instruments for the purpose of determining process operating conditions (e.g., composition, pressure, flow, etc.). Valves and connectors are the predominant type of equipment used in instrumentation systems; however, other types of equipment may also be included in these systems. Only valves nominally 0.5 inches and smaller and connectors nominally 0.75 inches and smaller in diameter are considered instrumentation systems for the purposes of this subpart. Valves greater than nominally 0.5 inches and connectors greater than nominally 0.75 inches associated with instrumentation systems are not considered part of instrumentation systems and must be monitored individnallv

Integral intermediate means an intermediate for which 50 percent or more of the annual production is used in onsite production of any PAI(s) and that is not stored before being used in the

production of another integral intermediate or the PAI(s). For the purposes of this definition, an intermediate is stored if it is discharged to a storage vessel and at least one of the following conditions is met: the processing equipment that discharges to the storage vessel is shutdown before the processing equipment that withdraws from the storage vessel is started up; during an annual period, the material must be stored in the vessel for at least 30 days before being used to make a PAI; or the processing equipment that discharges to the storage vessel is located in a separate building (or processing area) of the plant than the processing equipment that uses material from the storage vessel as a feedstock, and control equipment is not shared by the two processing areas. Any process unit that produces an intermediate and is subject to subpart F of this part is not an integral intermediate.

Intermediate means an organic compound that is manufactured in a process and that is further processed or modified in one or more additional steps to ultimately produce a PAI.

Internal floating roof means a cover that rests or floats on the liquid surface (but not necessarily in complete contact with it) inside a storage tank or waste management unit that has a permanently affixed roof.

Junction box means a manhole or access point to a wastewater sewer system line or a lift station.

Large control device means a control device that controls process vents, and the total HAP emissions into the control device from all sources are greater than or equal to 10 tons/yr.

Liquid-mounted seal means a foam- or liquid-filled seal mounted in contact with the liquid between the wall of the storage vessel or waste management unit and the floating roof. The seal is mounted continuously around the tank or unit.

Liquids dripping means any visible leakage from the seal including dripping, spraying, misting, clouding, and ice formation. Indications of liquid dripping include puddling or new stains that are indicative of an existing evaporated drip.

Maintenance wastewater means wastewater generated by the draining of process fluid from components in the PAI process unit into an individual drain system prior to or during maintenance activities. Maintenance wastewater can be generated through planned or unplanned shutdowns and during periods not associated with a shutdown. Examples of activities that can generate maintenance wastewaters include descaling of heat exchanger tubing bundles, cleaning of distillation column traps, draining of low legs and high point bleeds, draining of pumps into an individual drain system, and draining of portions of the PAI process unit for repair.

Malfunction means any sudden, infrequent, and not reasonably preventable failure of air pollution control equipment, emissions monitoring equipment, process equipment, or a process to operate in a normal or usual manner which causes, or has the potential to cause, the emission limitations in an applicable standard to be exceeded. Failures that are caused all or in part by poor maintenance or careless operation are not malfunctions.

Maximum true vapor pressure means the equilibrium partial pressure exerted by the total organic HAP in the stored or transferred liquid at the temperature equal to the highest calendarmonth average of the liquid storage or transferred temperature for liquids stored or transferred above or below the ambient temperature or at the local maximum monthly average temperature as reported by the National Weather Service for liquids stored or transferred at the ambient temperature, as determined:

(1) In accordance with methods described in Chapter 19.2 of the American Petroleum Institute's Manual of Petroleum Measurement Standards, Evaporative Loss From Floating-Roof Tanks (incorporated by reference as specified in §63.14 in subpart A of this part); or

(2) As obtained from standard reference texts; or

(3) As determined by the American Society for Testing and Materials Method D2879-97, Test Method for Vapor Pressure-Temperature Relationship and Initial Decomposition Temperature of Liquids by Isoteniscope (incorporated by reference as specified in §63.14 of subpart A of this part); or

§63.1361

(4) Any other method approved by the Administrator.

Metallic shoe seal or mechanical shoe seal means metal sheets that are held vertically against the wall of the storage tank by springs, weighted levers, or other mechanisms and connected to the floating roof by braces or other means. A flexible coated fabric (envelope) spans the annular space between the metal sheet and the floating roof.

Non-dedicated PAI process unit means a process unit that is not a dedicated PAI process unit.

Nonrepairable means that it is technically infeasible to repair a piece of equipment from which a leak has been detected without a process shutdown.

Open-ended valve or line means any valve, except pressure relief valves, having one side of the valve seat in contact with process fluid and one side open to atmosphere, either directly or through open piping.

Operating scenario, for the purposes of reporting and recordkeeping, means a description of a PAI process unit, including: identification of each wastewater point of determination (POD) and process vent, their associated emissions episodes and durations, and their associated level of control and control devices, as applicable; calculations and engineering analyses required to demonstrate compliance; and a description of operating and/or testing conditions for any associated control device.

Organic compound, as used in the definitions of intermediate and PAI, means any compound that contains both carbon and hydrogen with or without other elements.

Organic HAP means those HAP listed in section 112(b) of the CAA that are measured according to the procedures of Method 18 or Method 25A, 40 CFR part 60, appendix A.

Pesticide active ingredient or PAI means any material that is an active ingredient within the meaning of FIFRA section 2(a); that is used to produce an insecticide, herbicide, or fungicide end use pesticide product; that consists of one or more organic compounds; and that must be labeled in accordance with 40 CFR part 156 for transfer, sale, or distribution. These materials are typically described by

40 CFR Ch. I (7–1–23 Edition)

North American Industrial Classification System (NAICS) Codes 325199 and 32532 (i.e., previously known as Standard Industrial Classification System Codes 2869 and 2879). These materials are identified by product classification codes 01, 21, 02, 04, 44, 07, 08, and 16 in block 19 on the 1999 version of EPA form 3540-16, the Pesticides Report for Pesticide-Producing Establishments. The materials represented by these codes are: insecticides; insecticide-fungicides; fungicides; herbicides; herbicide-fungicides; plant regulators; defoliants. desiccants: or multi-use active ingredients.

Pesticide active ingredient manufacturing process unit (PAI process unit) means a process unit that is used to produce a material that is primarily used as a PAI or integral intermediate. A PAI process unit consists of: the process, as defined in this subpart: associated storage vessels, as determined by the procedures in §63.1360(f); equipment identified in §63.1362(k); connected piping and ducts; and components such as pumps, compressors, agitators, pressure relief devices, sampling connection systems, open-ended valves or lines, valves, connectors, and instrumentation systems. A material is primarily used as a PAI or integral intermediate if more than 50 percent of the projected annual production from a process unit in the 3 years after June 23, 1999 or startup, whichever is later, is used as a PAI or integral intermediate; recordkeeping is required if the material is used as a PAI or integral intermediate, but not as the primary use. If the primary use changes to a PAI or integral intermediate, the process unit becomes a PAI process unit unless it is already subject to the HON. If the primary use changes from a PAI or integral intermediate to another use, the process unit remains a PAI process unit. Any process tank containing an integral intermediate is part of the PAI process unit used to produce the integral intermediate. A process unit that produces an intermediate that is not an integral intermediate may be designated as a PAI process unit according to the procedures of §63.1360(g). Formulation of pesticide products is not considered

part of a PAI process unit. Quality assurance and quality control laboratories are not considered part of a PAI process unit.

Plant site means all contiguous or adjoining property that is under common control, including properties that are separated only by a road or other public right-of-way. Common control includes properties that are owned, leased, or operated by the same entity, parent entity, subsidiary, or any combination thereof.

Point of determination (POD) means each point where a wastewater stream exits the PAI process unit.

NOTE TO DEFINITION OF "POINT OF DETER-MINATION": The regulation allows determination of the characteristics of a wastewater stream: at the point of determination: or downstream of the point of determination if corrections are made for changes in flow rate and annual average concentration of Table 9 compounds as determined in §63.144 of subpart G of this part. Such changes include: losses by air emissions, reduction of annual average concentration or changes in flow rate by mixing with other water or wastewater streams, and reduction in flow rate or annual average concentration by treating or otherwise handling the wastewater stream to remove or destroy HAP.

Pressure release means the emission of materials resulting from the system pressure being greater than the set pressure of the pressure relief device. This release can be one release or a series of releases over a short time period due to a malfunction in the process.

Pressure relief device or valve means a safety device used to prevent operating pressures from exceeding the maximum allowable working pressure of the process equipment. A common pressure relief device is a spring-loaded pressure relief valve. Devices that are actuated either by a pressure of less than or equal to 2.5 pounds per square inch gauge or by a vacuum are not pressure relief devices.

Process means a logical grouping of processing equipment which collectively function to produce a product. For the purpose of this subpart, a PAI process includes all, or a combination of, reaction, recovery, separation, purification, treatment, cleaning, and other activities or unit operations which are used to produce a PAI or integral intermediate. Ancillary activi-

ties are not considered a PAI process or any part of a PAI process. Ancillary activities include boilers and incinerators (not used to comply with the provisions of §63.1362), chillers or refrigeration systems, and other equipment and activities that are not directly involved (i.e., they operate within a closed system and materials are not combined with process fluids) in the processing of raw materials or the manufacturing of a PAI. A PAI process and all integral intermediate processes for which 100 percent of the annual production is used in the production of the PAI may be linked together and defined as a single PAI process unit.

Process condenser means a condenser whose primary purpose is to recover material as an integral part of a unit operation. The condenser must cause a vapor-to-liquid phase change for periods during which the temperature of liquid in the process equipment is at or above its boiling or bubble point. Examples of process condensers include distillation condensers, reflux condensers, and condensers used in stripping or flashing operation. In a series of condensers, all condensers up to and including the first condenser with an exit gas temperature below the boiling or bubble point of the liquid in the process equipment are considered to be process condensers. All condensers in line prior to the vacuum source are included in this definition.

Process shutdown means a work practice or operational procedure that stops production from a process or part of a process during which it is technically feasible to clear process material from a process or part of a process consistent with safety constraints and during which repairs can be effected. An unscheduled work practice or operational procedure that stops production from a process or part of a process for less than 24 hours is not a process shutdown. An unscheduled work practice or operational procedure that would stop production from a process or part of a process for a shorter period of time than would be required to clear the process or part of the process of materials and start up the process, and would result in greater emissions than delay of repair of leaking components until the next scheduled process shutdown, is not a process shutdown. The use of spare equipment and technically feasible bypassing of equipment without stopping production are not process shutdowns.

Process tank means a tank that is used within a process to collect material discharged from a feedstock storage vessel or equipment within the process before the material is transferred to other equipment within the process or a product storage vessel. In many process tanks, unit operations such as reactions and blending are conducted. Other process tanks, such as surge control vessels and bottom receivers, however, may not involve unit operations.

Process unit means the equipment assembled and connected by pipes or ducts to process raw materials and to manufacture an intended product.

Process unit group means a group of process units that manufacture PAI and products other than PAI by alternating raw materials or operating conditions, or by reconfiguring process equipment. A process unit group is determined according to the procedures specified in §63.1360(g).

Process vent means a point of emission from processing equipment to the atmosphere or a control device. The vent may be the release point for an emission stream associated with an individual unit operation, or it may be the release point for emission streams from multiple unit operations that have been manifolded together into a common header. Examples of process vents include, but are not limited to, vents on condensers used for product recovery, bottom receivers, surge control vessels, reactors, filters, centrifuges, process tanks, and product dryers. A vent is not considered to be a process vent for a given emission episode if the undiluted and uncontrolled emission stream that is released through the vent contains less than 50 ppmv HAP, as determined through process knowledge that no HAP are present in the emission stream; using an engineering assessment as discussed in §63.1365(c)(2)(ii); from test data collected using Method 18 of 40 CFR part 60, appendix A; or from test data collected using any other test method

40 CFR Ch. I (7–1–23 Edition)

that has been validated according to the procedures in Method 301 of appendix A of this part. Process vents do not include vents on storage vessels regulated under §63.1362(c), vents on wastewater emission sources regulated under §63.1362(d), or pieces of equipment regulated under §63.1363.

Process wastewater means wastewater which, during manufacturing or processing, comes into direct contact with. or results from, the production or use of any raw material, intermediate product, finished product, by-product, or waste product. Examples include: product tank drawdown or feed tank drawdown; water formed during a chemical reaction or used as a reactant; water used to wash impurities from organic products \mathbf{or} reactants; water used to clean process equipment: water used to cool or quench organic vapor streams through direct contact; and condensed steam from jet ejector systems pulling vacuum on vessels containing organics.

Product means the compound(s) or chemical(s) that are produced or manufactured as the intended output of a process unit. Impurities and wastes are not considered products.

Product dryer means equipment that is used to remove moisture or other liquid from granular, powdered, or other solid PAI or integral intermediate products prior to storage, formulation, shipment, or other uses. The product dryer is part of the process.

Product dryer vent means a process vent from a product dryer through which a gas stream containing gaseous pollutants (i.e., organic HAP, HCl, or chlorine), particulate matter, or both are released to the atmosphere or are routed to a control device.

Production-indexed HAP consumption factor (HAP factor) is the result of dividing the annual consumption of total HAP by the annual production rate, per process.

Production-indexed VOC consumption factor (VOC factor) is the result of dividing the annual consumption of total VOC by the annual production rate, per process.

Publicly owned treatment works (POTW) is defined at 40 CFR part 403.3(0).

Reactor means a device or vessel in which one or more chemicals or reactants, other than air, are combined or decomposed in such a way that their molecular structures are altered and one or more new organic compounds are formed.

Reconfiguration means disassembly of processing equipment for a particular non-dedicated process unit and reassembly of that processing equipment in a different sequence, or in combination with other equipment, to create a different non-dedicated process unit.

Reconstruction, as used in §63.1360(b), shall have the meaning given in §63.2, except that "affected or previously unaffected stationary source" shall mean either "affected facility" or "PAI process unit."

Recovery device, as used in the wastewater provisions, means an individual unit of equipment capable of, and normally used for the purpose of, recovering chemicals for fuel value (i.e., net positive heating value), use, reuse, or for sale for fuel value, use, or reuse. Examples of equipment that may be recovery devices include organic removal devices such as decanters, strippers, or thin-film evaporation units. To be a recovery device, a decanter and any other equipment based on the operating principle of gravity separation must receive only multi-phase liquid streams.

Repaired means that equipment is adjusted, or otherwise altered, to eliminate a leak as defined in the applicable paragraphs of § 63.1363.

Research and development facility means any stationary source whose primary purpose is to conduct research and development, where the operations are under the close supervision of technically trained personnel, and is not engaged in the manufacture of products for commercial sale, except in a de minimis manner.

Residual means any liquid or solid material containing Table 9 compounds (as defined in §63.111 of subpart G of this part) that is removed from a wastewater stream by a waste management unit or treatment process that does not destroy organics (nondestructive unit). Examples of residuals from nondestructive wastewater management units include the organic layer and bottom residue removed by a decanter or organic-water separator and the overheads from a steam stripper or air stripper. Examples of materials which are not residuals include: silt; mud; leaves; bottoms from a steam stripper or air stripper; and sludges, ash, or other materials removed from wastewater being treated by destructive devices such as biological treatment units and incinerators.

Safety device means a closure device such as a pressure relief valve, frangible disc, fusible plug, or any other type of device which functions exclusively to prevent physical damage or permanent deformation to a unit or its air emission control equipment by venting gases or vapors directly to the atmosphere during unsafe conditions resulting from an unplanned, accidental, or emergency event. For the purposes of this subpart, a safety device is not used for routine venting of gases or vapors from the vapor headspace underneath a cover such as during filling of the unit or to adjust the pressure in this vapor headspace in response to normal daily diurnal ambient temperature fluctuations. A safety device is designed to remain in a closed position during normal operations and open only when the internal pressure, or another relevant parameter, exceeds the device threshold setting applicable to the air emission control equipment as determined by the owner or operator based on manufacturer recommendations, applicable regulations, fire protection and prevention codes, standard engineering codes and practices, or other requirements for the safe handling of flammable, combustible, explosive, reactive, or hazardous materials.

Sampling connection system means an assembly of equipment within a process unit used during periods of representative operation to take samples of the process fluid. Equipment used to take nonroutine grab samples is not considered a sampling connection system.

Sensor means a device that measures a physical quantity or the change in a physical quantity, such as temperature, pressure, flow rate, pH, or liquid level.

40 CFR Ch. I (7-1-23 Edition)

Set pressure means the pressure at which a properly operating pressure relief device begins to open to relieve atypical process system operating pressure.

Sever line means a lateral, trunk line, branch line, or other conduit including, but not limited to, grates, trenches, etc., used to convey wastewater streams or residuals to a downstream waste management unit.

Shutdown means the cessation of operation of a continuous PAI process unit for any purpose. Shutdown also means the cessation of a batch PAI process unit or any related individual piece of equipment required or used to comply with this part or for emptying and degassing storage vessels for periodic maintenance, replacement of equipment, repair, or any other purpose not excluded from this definition. Shutdown does not apply to cessation of a batch PAI process unit at the end of a campaign or between batches (e.g., for rinsing or washing equipment), for routine maintenance, or for other routine operations.

Small control device means a control device that controls process vents, and the total HAP emissions into the control device from all sources are less than 10 tons of HAP per year.

Startup means the setting in operation of a continuous PAI process unit for any purpose, the first time a new or reconstructed batch PAI process unit begins production, or, for new equipment added, including equipment used to comply with this subpart, the first time the equipment is put into operation. For batch process units, startup does not apply to the first time the equipment is put into operation at the start of a campaign to produce a product that has been produced in the past, after a shutdown for maintenance, or when the equipment is put into operation as part of a batch within a campaign. As used in §63.1363, startup means the setting in operation of a piece of equipment or a control device that is subject to this subpart.

Storage vessel means a tank or other vessel that is used to store organic liquids that contain one or more HAP and that has been assigned, according to the procedures in §63.1360(f) or (g), to a PAI process unit that is subject to this subpart MMM. The following are not considered storage vessels for the purposes of this subpart:

(1) Vessels permanently attached to motor vehicles such as trucks, railcars, barges, or ships;

(2) Pressure vessels designed to operate in excess of 204.9 kilopascals and without emissions to the atmosphere;

(3) Vessels storing material that contains no organic HAP or contains organic HAP only as impurities;

(4) Wastewater storage tanks;

(5) Process tanks; and

(6) Nonwastewater waste tanks.

Supplemental gases means any nonaffected gaseous streams (streams that are not from process vents, storage vessels, equipment or waste management units) that contain less than 50 ppmv TOC and less than 50 ppmv total HCl and chlorine, as determined through process knowledge, and are combined with an affected vent stream. Supplemental gases are often used to maintain pressures in manifolds or for fire and explosion protection and prevention. Air required to operate combustion device burner(s) is not considered a supplemental gas.

Surface impoundment means a waste management unit which is a natural topographic depression, manmade excavation, or diked area formed primarily of earthen materials (although it may be lined with manmade materials), which is designed to hold an accumulation of liquid wastes or waste containing free liquids. A surface impoundment is used for the purpose of treating, storing, or disposing of wastewater or residuals, and is not an injection well. Examples of surface impoundments are equalization, settling, and aeration pits, ponds, and lagoons.

Total organic compounds (TOC) means those compounds measured according to the procedures of Method 18 or Method 25A, 40 CFR part 60, appendix A.

Treatment process means a specific technique that removes or destroys the organics in a wastewater or residual stream such as a steam stripping unit, thin-film evaporation unit, waste incinerator, biological treatment unit, or any other process applied to wastewater streams or residuals to comply with §63.138 of subpart G of this part.

Most treatment processes are conducted in tanks. Treatment processes are a subset of waste management units.

Uncontrolled HAP emissions means a gas stream containing HAP which has exited the process (or process condenser, if any), but which has not yet been introduced into an air pollution control device to reduce the mass of HAP in the stream. If the process vent is not routed to an air pollution control device, uncontrolled emissions are those HAP emissions released to the atmosphere.

Unit operation means those processing steps that occur within distinct equipment that are used, among other things, to prepare reactants, facilitate reactions, separate and purify products, and recycle materials. Equipment used for these purposes includes, but is not limited to, reactors, distillation units, extraction columns, absorbers, decanters, dryers, condensers, and filtration equipment.

Vapor-mounted seal means a continuous seal that completely covers the annular space between the wall of the storage tank or waste management unit and the edge of the floating roof, and is mounted such that there is a vapor space between the stored liquid and the bottom of the seal.

Volatile organic compounds are defined in 40 CFR 51.100.

Waste management unit means the equipment, structure(s), and/or device(s) used to convey, store, treat, or dispose of wastewater streams or residuals. Examples of waste management units include wastewater tanks, surface impoundments, individual drain systems, and biological wastewater treatment units. Examples of equipment that may be waste management units include containers, air flotation units, oil-water separators or organic-water separators, or organic removal devices such as decanters, strippers, or thin-film evaporation units. If such equipment is a recovery device, then it is part of a PAI process unit and is not a waste management unit.

Wastewater means water that meets either of the conditions described in paragraph (1) or (2) of this definition and is discarded from a PAI process unit that is at an affected source: (1) Is generated from a PAI process or a scrubber used to control emissions from a PAI process and contains either:

(i) An annual average concentration of compounds in Table 9 of subpart G of this part of at least 5 ppmw and has an average flow rate of 0.02 L/min or greater; or

(ii) An annual average concentration of compounds in Table 9 of subpart G of this part of at least 10,000 ppmw at any flow rate;

(2) Is generated from a PAI process unit as a result of maintenance activities and contains at least 5.3 Mg of compounds listed in Table 9 of subpart G of this part per individual discharge event.

Wastewater tank means a stationary waste management unit that is designed to contain an accumulation of wastewater or residuals and is constructed primarily of nonearthen materials (e.g., wood, concrete, steel, plastic) which provide structural support. Wastewater tanks used for flow equalization are included in this definition.

Water seal controls means a seal pot, p-leg trap, or other type of trap filled with water (e.g., flooded sewers that maintain water levels adequate to prevent air flow through the system) that creates a water barrier between the sewer line and the atmosphere. The water level of the seal must be maintained in the vertical leg of a drain in order to be considered a water seal.

[64 FR 33589, June 23, 1999, as amended at Nov. 21, 2001; 67 FR 59343, Sept. 20, 2002; 71 FR 20460, Apr. 20, 2006; 79 FR 17372, Mar. 27, 2014]

§63.1362 Standards.

(a) On and after the compliance dates specified in §63.1364, each owner or operator of an affected source subject to the provisions of this subpart shall control HAP emissions to the levels specified in this section and in §63.1363, as summarized in Table 2 of this subpart.

(b) Process vents. (1) The owner or operator of an existing source shall comply with the requirements of paragraphs (b)(2) and (3) of this section. The owner or operator of a new source shall comply with the requirements of paragraphs (b)(4) and (5) of this section. Compliance with paragraphs (b)(2) through (b)(5) of this section shall be demonstrated through the applicable test methods and initial compliance procedures in $\S63.1365$ and the monitoring requirements in $\S63.1366$.

(2) Organic HAP emissions from existing sources. The owner or operator of an existing affected source must comply with the requirements in either paragraph (b)(2)(i) of this section or with the requirements in paragraphs (b)(2)(i) through (iv) of this section.

(i) The uncontrolled organic HAP emission rate shall not exceed 0.15 Mg/ yr from the sum of all process vents within a process.

(ii) (A) Except as provided in paragraph (b)(2)(ii)(B) of this section, uncontrolled organic HAP emissions from a process vent shall be reduced by 98 percent by weight or greater if the flow-weighted average flowrate for the vent as calculated using Equation 1 of this subpart is less than or equal to the flowrate calculated using Equation 2 of this subpart.

$$FR_{a} = \frac{\sum_{i=1}^{n} (D_{i})(FR_{i})}{\sum_{i=1}^{n} D_{i}}$$
(Eq. 1)

FR = 0.02 * (HL) - 1,000 (Eq. 2) Where:

 ${\rm FR}_{\rm a}$ = flow-weighted average flow rate for the vent, scfm

 D_i = duration of each emission event, min

 FR_i = flowrate of each emission event, scfm

n = number of emission events

FR = flowrate, scfm

HL = annual uncontrolled organic HAP emissions, lb/yr, as defined in §63.1361

(B) If the owner or operator can demonstrate that a control device, installed on or before November 10, 1997 on a process vent otherwise subject to the requirements of paragraph (b)(2)(ii)(A) of this section, reduces inlet emissions of total organic HAP by greater than or equal to 90 percent by weight but less than 98 percent by weight, then the control device must be operated to reduce inlet emissions of total organic HAP by 90 percent by weight or greater.

(iii) Excluding process vents that are subject to the requirements in paragraph (b)(2)(ii) of this section, uncon40 CFR Ch. I (7–1–23 Edition)

trolled organic HAP emissions from the sum of all process vents within a process shall be reduced by 90 percent or greater by weight.

(iv) As an alternative to the requirements in paragraphs (b)(2)(ii) and (iii) of this section, uncontrolled organic HAP emissions from any process vent may be reduced in accordance with any of the provisions in paragraphs (b)(2)(iv)(A) through (D) of this section. All remaining process vents within a process must be controlled in accordance with paragraphs (b)(2)(ii) and (iii) of this section.

(A) To outlet concentrations less than or equal to 20 ppmv; or

(B) By a flare that meets the requirements of §63.11(b); or

(C) By a control device specified in §63.1365(a)(4); or

(D) In accordance with the alternative standard specified in paragraph (b)(6) of this section.

(3) HCl and Cl_2 emissions from existing sources. For each process, the owner or operator of an existing source shall comply with the requirements of either paragraph (b)(3)(i) or (ii) of this section.

(i) The uncontrolled HCl and Cl_2 emissions, including HCl generated from the combustion of halogenated process vent emissions, from the sum of all process vents within a process shall not exceed 6.8 Mg/yr.

(ii) HCl and Cl_2 emissions, including HCl generated from combustion of halogenated process vent emissions, from the sum of all process vents within a process shall be reduced by 94 percent or greater or to outlet concentrations less than or equal to 20 ppmv.

(4) Organic HAP emissions from new sources. For each process, the owner or operator of a new source shall comply with the requirements of either paragraph (b)(4)(i) or (ii) of this section.

(i) The uncontrolled organic HAP emissions shall not exceed 0.15 Mg/yr from the sum of all process vents within a process.

(ii) The uncontrolled organic HAP emissions from the sum of all process vents within a process at a new affected source that are not controlled according to any of the requirements of paragraphs (b)(4)(ii)(A) through (C) or

(b)(6) of this section shall be reduced by 98 weight percent or greater.

(A) To outlet concentrations less than or equal to 20 ppmv; or

(B) By a flare that meets the requirements of §63.11(b); or

(C) By a control device specified in §63.1365(a)(4).

(5) HCl and Cl_2 emissions from new sources. For each process, the owner or operator of a new source shall comply with the requirements of either paragraph (b)(5)(i), (ii), or (iii) of this section.

(i) The uncontrolled HCl and Cl_2 emissions, including HCl generated from combustion of halogenated process vent emissions, from the sum of all process vents within a process shall not exceed 6.8 Mg/yr.

(ii) If HCl and Cl_2 emissions, including HCl generated from combustion of halogenated process vent emissions, from the sum of all process vents within a process are greater than 6.8 Mg/yr and less than or equal to 191 Mg/yr, these HCl and Cl_2 emissions shall be reduced by 94 percent or to an outlet concentration less than or equal to 20 ppmv.

(iii) If HCl and Cl_2 emissions, including HCl generated from combustion of halogenated process vent emissions, from the sum of all process vents within a process are greater than 191 Mg/yr, these HCl and Cl_2 emissions shall be reduced by 99 percent or greater or to an outlet concentration less than or equal to 20 ppmv.

(6) Alternative standard. As an alternative to the provisions in paragraphs (b)(2) through (5) of this section, the owner or operator may route emissions from a process vent to a combustion control device achieving an outlet TOC concentration, as calibrated on methane or the predominant HAP, of 20 ppmv or less, and an outlet concentration of HCl and Cl_2 of 20 ppmv or less. If the owner or operator is routing emissions to a non-combustion control device or series of control devices, the control device(s) must achieve an outlet TOC concentration, as calibrated on methane or the predominant HAP, of 50 ppmy or less, and an outlet concentration of HCl and Cl_2 of 50 ppmv or less. Any process vents within a process that are not routed to such a control

device or series of control devices must be controlled in accordance with the provisions of paragraph (b)(2)(ii), (iii), (iv), (b)(3)(ii), (b)(4)(ii), (b)(5)(ii) or (iii)of this section, as applicable.

(c) Storage vessels. (1) The owner or operator shall either determine the group status of a storage vessel or designate it as a Group 1 storage vessel. If the owner or operator elects to designate the storage vessel as a Group 1 storage vessel, the owner or operator is not required to determine the maximum true vapor pressure of the material stored in the storage vessel.

(2) Standard for existing sources. Except as specified in paragraphs (c)(4), (5), and (6) of this section, the owner or operator of a Group 1 storage vessel at an existing affected source, as defined in §63.1361, shall equip the affected storage vessel with one of the following:

(i) A fixed roof and internal floating roof, or

(ii) An external floating roof, or

(iii) An external floating roof converted to an internal floating roof, or

(iv) A closed vent system meeting the conditions of paragraph (j) of this section and a control device that meets any of the following conditions:

(A) Reduces organic HAP emissions by 95 percent by weight or greater; or

(B) Reduces organic HAP emissions to outlet concentrations of 20 ppmv or less; or

(C) Is a flare that meets the requirements of §63.11(b); or

(D) Is a control device specified in (63.1365(a))(4).

(3) Standard for new sources. Except as specified in paragraphs (c)(4), (5), and (6) of this section, the owner or operator of a Group 1 storage vessel at a new source, as defined in § 63.1361, shall equip the affected storage vessel in accordance with any one of paragraphs (c)(2)(i) through (iv) of this section.

(4) Alternative standard. As an alternative to the provisions in paragraphs (c)(2) and (3) of this section, the owner or operator of an existing or new affected source may route emissions from storage vessels to a combustion control device achieving an outlet TOC concentration, as calibrated on methane or the predominant HAP, of 20

ppmv or less, and an outlet concentration of hydrogen chloride and chlorine of 20 ppmv or less. If the owner or operator is routing emissions to a non-combustion control device or series of control devices, the control device(s) must achieve an outlet TOC concentration, as calibrated on methane or the predominant HAP, of 50 ppmv or less, and an outlet concentration of HCl and Cl_2 of 50 ppmv or less.

(5) Planned routine maintenance. The owner or operator is exempt from the specifications in paragraphs (c)(2) through (4) of this section during periods of planned routine maintenance of the control device that do not exceed 240 hr/yr. The owner or operator may submit an application to the Administrator requesting an extension of this time limit to a total of 360 hr/yr. The application must explain why the extension is needed, it must indicate that no material will be added to the storage vessel between the time the 240-hr limit is exceeded and the control device is again operational, and it must be submitted at least 60 days before the 240-hr limit will be exceeded.

(6) Vapor balancing alternative. As an alternative to the requirements in paragraphs (c)(2) and (3) of this section, the owner or operator of an existing or new affected source may implement vapor balancing in accordance with paragraphs (c)(6)(i) through (vii) of this section.

(i) The vapor balancing system must be designed and operated to route organic HAP vapors displaced from loading of the storage tank to the railcar or tank truck from which the storage tank is filled.

(ii) Tank trucks and railcars must have a current certification in accordance with the U.S. Department of Transportation pressure test requirements of 49 CFR part 180 for tank trucks and 49 CFR 173.31 for railcars.

(iii) Hazardous air pollutants must only be unloaded from tank trucks or railcars when vapor collection systems are connected to the storage tank's vapor collection system.

(iv) No pressure relief device on the storage tank or on the railcar or tank truck shall open during loading or as a result of diurnal temperature changes (breathing losses). 40 CFR Ch. I (7–1–23 Edition)

(v) Pressure relief devices on affected storage tanks must be set to no less than 2.5 psig at all times to prevent breathing losses. The owner or operator shall record the setting as specified in §63.1367(b)(8) and comply with the following requirements for each pressure relief valve:

(A) The pressure relief valve shall be monitored quarterly using the method described in §63.180(b).

(B) An instrument reading of 500 ppmv or greater defines a leak.

(vi) Railcars or tank trucks that deliver HAP to an affected storage tank must be reloaded or cleaned at a facility that utilizes one of the following control techniques:

(A) The railcar or tank truck must be connected to a closed vent system with a control device that reduces inlet emissions of HAP by 90 percent by weight or greater; or

(B) A vapor balancing system designed and operated to collect organic HAP vapor displaced from the tank truck or railcar during reloading must be used to route the collected HAP vapor to the storage tank from which the liquid being transferred originated.

(vii) The owner or operator of the facility where the railcar or tank truck is reloaded or cleaned must comply with the following requirements:

(A) Submit to the owner or operator of the affected storage tank and to the Administrator a written certification that the reloading or cleaning facility will meet the requirements of this section. The certifying entity may revoke the written certification by sending a written statement to the owner or operator of the affected storage tank giving at least 90 days notice that the certifying entity is rescinding acceptance of responsibility for compliance with the requirements of this paragraph (c)(6)(vii)(A).

(B) If complying with paragraph (c)(6)(vi)(A) of this section, demonstrate initial compliance in accordance with §63.1365(d), demonstrate continuous compliance in accordance with §63.1366, keep records as specified in §63.1367, and prepare reports as specified in §63.1368.

(C) If complying with paragraph (c)(6)(vi)(B) of this section, keep records of:

(1) The equipment to be used and the procedures to be followed when reloading the railcar or tank truck and displacing vapors to the storage tank from which the liquid originates, and

(2) Each time the vapor balancing system is used to comply with paragraph (c)(6)(vi)(B) of this section.

(7) Compliance with the provisions of paragraphs (c)(2) and (3) of this section is demonstrated using the initial compliance procedures in §63.1365(d) and the monitoring requirements in §63.1366. Compliance with the outlet concentrations in paragraph (c)(4) of this section shall be determined by the compliance provisions initial in §63.1365(a)(5) and the continuous emisof sion monitoring requirements §63.1366(b)(5).

(d) Wastewater. The owner or operator of each affected source shall comply with the requirements of \$ 63.132 through 63.147, with the differences noted in paragraphs (d)(1) through (16) of this section for the purposes of this subpart.

(1) When the determination of equivalence criteria in 63.102(b) is referred to in 863.132, 63.133, and 63.137 of subpart G of this part, the provisions in 63.6(g) of subpart A of this part shall apply.

(2) When the storage tank requirements contained in §§ 63.119 through 63.123 are referred to in §§ 63.132 through 63.147, §§ 63.119 through 63.123 are applicable, with the exception of the differences noted in paragraphs (d)(2)(i) through (iv) of this section.

(i) When the term "storage vessel" is used in §§63.119 through 63.123 of subpart G of this part, the definition of the term "storage vessel" in §63.1361 shall apply for the purposes of this subpart.

(ii) When December 31, 1992, is referred to in §63.119 of subpart G of this part, November 10, 1997 shall apply for the purposes of this subpart.

(iii) When April 22, 1994 is referred to in §63.119 of subpart G of this part, June 23, 1999 shall apply for the purposes of this subpart.

(iv) When the phrase "the compliance date specified in §63.100 of subpart F of this part" is referred to in §63.120 of subpart G of this part, the phrase "the compliance date specified in §63.1364" shall apply for the purposes of this subpart.

(3) To request approval to monitor alternative parameters, as referred to in §63.146(a) of subpart G of this part, the owner or operator shall comply with the procedures in §63.8(f) of subpart A of this part, as referred to in §63.1366(b)(4), instead of the procedures in §63.151(f) or (g) of subpart G of this part.

(4) When the Notification of Compliance Status report requirements contained in §63.152(b) of subpart G of this part are referred to in §63.146 of subpart G of this part, the Notification of Compliance Status report requirements in §63.1368(f) shall apply for the purposes of this subpart.

(5) When the recordkeeping requirements contained in $\S63.152(f)$ of subpart G of this part are referred to in $\S63.147(d)$ of subpart G of this part, the recordkeeping requirements in $\S63.1367$ shall apply for the purposes of this subpart.

(6) When the Periodic report requirements contained in 63.152(c) of subpart G of this part are referred to in 863.146 and 63.147 of subpart G of this part, the Periodic report requirements contained in 63.1368(g) shall apply for the purposes of this subpart.

(7) When the term "process wastewater" is referred to in §§ 63.132 through 63.147 of subpart G of this part, the term "wastewater" as defined in § 63.1361 shall apply for the purposes of this subpart.

(8) When the term "Group 1 wastewater stream" is used in §§ 63.132 through 63.147 of subpart G of this part, the definition of the term "Group 1 wastewater stream" in § 63.1361 shall apply for both new sources and existing sources for the purposes of this subpart.

40 CFR Ch. I (7–1–23 Edition)

(9) The requirements in \$ 63.132 through 63.147 for compounds listed on Table 8 of subpart G of this part shall not apply for the purposes of this subpart.

(10) When the total load of Table 9 compounds in the sum of all process wastewater from PAI process units at a new affected source is 2,100 Mg/yr (2,300 tons/yr) or more, the owner or operator shall reduce, by removal or destruction, the mass flow rate of all compounds in Table 9 of subpart G of this part in all wastewater (process and maintenance wastewater) by 99 percent or more. Alternatively, the owner or operator may treat the wastewater in a unit identified in and complying with §63.138(h) of subpart G of this part. The removal/destruction efficiency shall be determined by the procedures specified in §63.145(c) of subpart G of this part, noncombustion processes, for or §63.145(d) of subpart G of this part, for combustion processes.

(11) The compliance date for the affected source subject to the provisions of this section is specified in §63.1364.

(12) As an alternative to using Method 18 of 40 CFR part 60, as specified in §§ 63.139(c)(1)(ii) and 63.145(i)(2), the owner or operator may elect to use Method 25 or Method 25A of 40 CFR part 60, as specified in §63.1365(b).

(13) The requirement to correct outlet concentrations from combustion devices to 3 percent oxygen in $\S63.139(c)(1)(ii)$ shall apply only if supplemental gases are combined with affected vent streams, and the procedures in $\S63.1365(a)(7)(i)$ apply instead of the procedures in $\S63.145(i)(6)$ to determine the percent oxygen correction. If emissions are controlled with a vapor recovery system as specified in $\S63.139(c)(2)$, the owner or operator must correct for supplemental gases as specified in $\S63.1365(a)(7)(i)$.

(14) As an alternative to the management and treatment options specified in $\S63.132(g)(2)$, any Group 1 wastewater stream (or residual removed from a Group 1 wastewater stream) that contains less than 50 ppmw of HAP listed in Table 2 to subpart GGG of this part may be transferred offsite or to an onsite treatment operation not owned or operated by the owner or operator of the source generating the wastewater

(or residual) if the transferee manages and treats the wastewater stream or residual in accordance with paragraphs (d)(14)(i) through (iv) of this section.

(i) Treat the wastewater stream or residual in a biological treatment unit in accordance with §§ 63.138 and 63.145.

(ii) Cover the waste management units up to the activated sludge unit. Alternatively, covers are not required if the owner or operator demonstrates that less than 5 percent of the total HAP listed in Table 3 to subpart GGG of this part is emitted.

(iii) Inspect covers as specified in §63.1366(h).

(iv) The reference in (3.132(g))(2) to (3.102(b)) of subpart F'' does not apply for the purposes of this subpart.

(15) When §63.133 refers to Table 10 to subpart G of this part, the maximum true vapor pressures in the table shall be limited to the HAP listed in Table 9 to subpart G of this part.

(16) When the inspection, recordkeeping, and reporting requirements contained in $\S63.148$ are referred to in \$\$63.132 through 63.147, the inspection requirements in \$63.1366(h), the recordkeeping requirements in \$63.1367(f), and the reporting requirements in \$63.1368(g)(2)(iii) and (xi) shall apply for the purposes of this subpart.

(e) Bag dumps and product dryers. (1) The owner or operator shall reduce particulate matter emissions to a concentration not to exceed 0.01 gr/dscf from product dryers that dry a PAI or integral intermediate that is a HAP.

(2) The owner or operator shall reduce particulate matter emissions to a concentration not to exceed 0.01 gr/dscf from bag dumps that introduce to a PAI process unit a feedstock that is a solid material and a HAP, excluding bag dumps where the feedstock contains HAP only as impurities.

(3) Gaseous HAP emissions from product dryers and bag dumps shall be controlled in accordance with the provisions for process vent emissions in paragraph (b) of this section.

(f) *Heat exchange systems*. Unless one or more of the conditions specified in §63.104(a)(1) through (6) of subpart F of this part are met, an owner or operator shall monitor each heat exchange system that is used to cool process equipment in PAI process units that are part

of an affected source as defined in §63.1360(a) according to the provisions in either §63.104(b) or (c) of subpart F of this part. When the term "chemical manufacturing process unit" is used in §63.104(c) of subpart F of this part, the term "PAI process unit" shall apply for the purposes of this subpart. Whenever a leak is detected, the owner or operator shall comply with the requirements in §63.104(d) of subpart F of this part. Delay of repair of heat exchange systems for which leaks have been detected is allowed in accordance with the provisions of §63.104(e) of subpart F of this part.

(g) Pollution prevention alternative. Except as provided in paragraph (g)(1)of this section, for a process that has an initial startup before November 10, 1997, an owner or operator may choose to meet the pollution prevention alternative requirement specified in either paragraph (g)(2) or (3) of this section for any PAI process unit, in lieu of the requirements specified in paragraphs (b), (c), (d), and (e) of this section and in §63.1363. Compliance with the requirements of paragraphs (g)(2) and (3)of this section shall be demonstrated through the procedures in §§63.1365(g) and 63.1366(f).

(1) A HAP must be controlled according to the requirements of paragraphs (b), (c), (d), and (e) of this section and §63.1363 if it is generated in the PAI process unit or an associated control device and it is not part of the production-indexed HAP consumption factor (HAP factor).

(2) The HAP factor shall be reduced by at least 85 percent from a 3-year average baseline beginning no earlier than the 1987 through 1989 calendar years. Alternatively, for a process that has been operating for less than 3 years but more than 1 year, the baseline factor may be calculated for the time period from startup of the process until the present. For any reduction in the HAP factor achieved by reducing a HAP that is also a VOC, an equivalent reduction in the production-indexed VOC consumption factor (VOC factor) is also required (the equivalence is determined on a mass basis, not a percentage basis). For any reduction in the HAP factor that is achieved by reducing a HAP that is not a VOC, the VOC factor may not be increased.

(3) As an alternative to the provisions in paragraph (g)(2) of this section, the owner or operator may combine pollution prevention with emissions control as specified in paragraphs (g)(3)(i) and (ii) of this section.

(i) The HAP factor shall be reduced as specified in paragraph (g)(2) of this section except that a reduction of at least 50 percent shall apply for the purposes of this paragraph.

(ii) The total annual HAP emissions from the PAI process unit shall be reduced by an amount that, when divided by the annual production rate and added to the reduction of the HAP factor yields a value of at least 85 percent of the baseline HAP factor. The total annual VOC emissions from the process unit must be reduced by an amount equivalent to the reduction in HAP emissions for each HAP that is a VOC (the equivalence is determined on a mass basis). For HAP emissions reductions that are achieved by reducing a HAP that is not a VOC, the total annual VOC emissions may not be increased. The reduction in HAP air emissions must be achieved using one of the following control devices:

(A) Combustion control devices such as incinerators, flares, or process heaters.

(B) Control devices such as condensers and carbon adsorbers whose recovered product is destroyed or shipped offsite for destruction.

(C) Any control device that does not ultimately allow for recycling of material back to the PAI process unit.

(D) Any control device for which the owner or operator can demonstrate that the use of the device in controlling HAP emissions will have no effect on the HAP factor for the PAI process unit.

(h) Emissions averaging provisions. Except as provided in paragraphs (h)(1) through (7) of this section, the owner or operator of an existing affected facility may choose to comply with the emission standards in paragraphs (b), (c), and (d) of this section by using emissions averaging procedures specified in §63.1365(h) for organic HAP emissions from any storage vessel, process, or waste management unit

that is part of an affected source subject to this subpart.

(1) A State may restrict the owner or operator of an existing source to use only the procedures in paragraphs (b), (c), and (d) of this section to comply with the emission standards where State authorities prohibit averaging of HAP emissions.

(2) Group 1 emission points that are controlled as specified in paragraphs (h)(2)(i) through (iii) of this section may not be used to calculate emissions averaging credits, unless the equipment is approved for use in a different manner from that specified in paragraphs (b) through (d) of this section. and a nominal efficiency has been assigned according to the procedures in §63.150(i). The nominal efficiency must exceed the percent reduction required by paragraphs (b) and (c) of this section for process vents and storage vessels, respectively, exceed the percent reduction required in §63.139(c) for control devices used to control emissions vented from waste management units, and exceed the percent reduction required in §63.138(e) or (f) for wastewater treatment processes.

(i) Storage vessels controlled with an internal floating roof meeting the specifications of §63.119(b), an external floating roof meeting the specifications of §63.119(c), or an external floating roof converted to an internal floating meeting the specifications of §63.119(d). (ii) Emission points controlled with a

flare.

(iii) Wastewater streams that are managed in waste management units that are controlled as specified in §§ 63.133 through 63.137, treated using a steam stripper meeting the specifications of § 63.138(d), and emissions from the steam stripper are controlled in a control device that meets the percent reduction requirements specified in § 63.139(c).

(3) Process vents and storage vessels controlled with a control device to an outlet concentration of 20 ppmv or 50 ppmv, as specified in paragraph (b)(2)(iv)(A), (b)(3)(ii), (b)(6), (c)(2)(iv)(B), or (c)(4) of this section, and wastewater streams controlled in a treatment unit to an outlet concentration of 50 ppmw, may not be used in any averaging group.

40 CFR Ch. I (7–1–23 Edition)

(4) Maintenance wastewater streams, wastewater streams treated in biological treatment units, and Group 2 wastewater streams that are not managed as specified in §§ 63.133 through 63.137 may not be included in any averaging group.

(5) Processes which have been permanently shut down and storage vessels permanently taken out of HAP service may not be included in any averaging group.

(6) Emission points already controlled on or before November 15, 1990 may not be used to generate emissions averaging credits, unless the level of control has been increased after November 15, 1990. In these cases, credit will be allowed only for the increase in control after November 15, 1990.

(7) Emission points controlled to comply with a State or Federal rule other than this subpart may not be included in an emissions averaging group, unless the level of control has been increased after November 15, 1990, above what is required by the other State or Federal rule. Only the control above what is required by the other State or Federal rule will be credited. However, if an emission point has been used to generate emissions averaging credit in an approved emissions average, and the point is subsequently made subject to a State or Federal rule other than this subpart, the point can continue to generate emissions averaging credit for the purpose of complying with the previously approved average.

(i) Opening of a safety device. The owner or operator that opens a safety device, as defined in §63.1361, is not exempt from applicable standards in order to avoid unsafe conditions. If opening a safety device results in the failure to meet any applicable standard, the owner or operator must still comply with the general duty to minimize emissions. If opening a safety device results in a deviation or excess emissions, such events must be reported as specified in §63.1368(i). If the owner or operator attributes the event to a malfunction and intends to assert an affirmative defense, the owner or operator is subject to §63.1360(k).

(j) *Closed-vent systems*. The owner or operator of a closed-vent system that

contains bypass lines that could divert a vent stream away from a control device used to comply with the requirements in paragraphs (b) through (d) of this section shall comply with the requirements of Table 3 of this subpart and paragraph (j)(1) or (2) of this section. Equipment such as low leg drains, high point bleeds, analyzer vents, openended valves or lines, rupture disks and pressure relief valves needed for safety purposes are not subject to this paragraph.

(1) Install, calibrate, maintain, and operate a flow indicator that is capable of determining whether vent stream flow is present and taking frequent, periodic readings. Records shall be maintained as specified in §63.1367(f)(1). The flow indicator shall be installed at the entrance to any bypass line that could divert the vent stream away from the control device to the atmosphere; or

(2) Secure the bypass line valve in the closed position with a car-seal or lock-and-key type configuration. Records shall be maintained as specified in $\S63.1367(f)(2)$.

(k) Control requirements for certain liquid streams in open systems within a PAI process unit. (1) The owner or operator shall comply with the provisions of Table 4 of this subpart, for each item of equipment meeting all the criteria specified in paragraphs (k)(2) through (4) of this section and either paragraph (k)(5)(i) or (ii) of this section.

(2) The item of equipment is of a type identified in Table 4 of this subpart;

(3) The item of equipment is part of a PAI process unit as defined in §63.1361;

(4) The item of equipment is controlled less stringently than in Table 4 of this subpart, and the item of equipment is not otherwise exempt from controls by the provisions of this subpart or subpart A of this part;

(5) The item of equipment:

(i) Is a drain, drain hub, manhole, lift station, trench, pipe, or oil/water separator that conveys water with a total annual average concentration greater than or equal to 10,000 ppm by weight of compounds in Table 9 of subpart G of this part at any flowrate; or a total annual average concentration greater than or equal to 1,000 ppm by weight of compounds in Table 9 of subpart G of this part at an annual average flow rate greater than or equal to 10 liters per minute; or

(ii) Is a tank that receives one or more streams that contain water with a total annual average concentration greater than or equal to 1,000 ppm by weight of compounds in Table 9 of subpart G of this part at an annual average flowrate greater than or equal to 10 liters per minute. The owner or operator of the source shall determine the characteristics of the stream as specified in paragraphs (k)(5)(ii)(A) and (B) of this section.

(A) The characteristics of the stream being received shall be determined at the inlet to the tank.

(B) The characteristics shall be determined according to the procedures in §63.144(b) and (c) of subpart G of this part.

(1) Exemption for RCRA treatment units. An owner or operator shall be exempt from the initial compliance demonstrations and monitoring provisions in §§ 63.1365 and 63.1366 and the associated recordkeeping and reporting requirements in §§ 63.1367 and 63.1368 for emissions from process vents, storage vessels, and waste management units that are discharged to the following devices:

(1) A boiler or process heater burning hazardous waste for which the owner or operator:

(i) Has been issued a final permit under 40 CFR part 270 and complies with the requirements of 40 CFR part 266, subpart H; or

(ii) Has certified compliance with the interim status requirements of 40 CFR part 266, subpart H.

(2) A hazardous waste incinerator for which the owner or operator has been issued a final permit under 40 CFR part 270 and complies with the requirements of 40 CFR part 264, subpart O, or has certified compliance with the interim status requirements of 40 CFR part 265, subpart O.

[64 FR 33589, June 23, 1999, as amended at 67 FR 59343, Sept. 20, 2002; 68 FR 37358, June 23, 2003; 79 FR 17372, Mar. 27, 2014]

§63.1363 Standards for equipment leaks.

(a) General equipment leak requirements. (1) The provisions of this section apply to "equipment" as defined in §63.1361. The provisions of this section also apply to any closed-vent systems and control devices required by this section.

(2) Consistency with other regulations. After the compliance date for a process, equipment subject to both this section and either of the following will be required to comply only with the provisions of this subpart:

(i) 40 CFR part 60.

(ii) 40 CFR part 61.

(3) [Reserved]

(4) The provisions in $\S63.1(a)(3)$ of subpart A of this part do not alter the provisions in paragraph (a)(2) of this section.

(5) Lines and equipment not containing process fluids are not subject to the provisions of this section. Utilities, and other nonprocess lines, such as heating and cooling systems which do not combine their materials with those in the processes they serve, are not considered to be part of a process.

(6) The provisions of this section do not apply to bench-scale processes, regardless of whether the processes are located at the same plant site as a process subject to the provisions of this subpart MMM.

(7) Each piece of equipment to which this section applies shall be identified such that it can be distinguished readily from equipment that is not subject to this section. Identification of the equipment does not require physical tagging of the equipment. For example, the equipment may be identified on a plant site plan, in log entries, or by designation of process boundaries by some form of weatherproof identification. If changes are made to the affected source subject to the leak detection requirements, equipment identification for each type of component shall be updated, if needed, within 15 calendar days of the end of each monitoring period for that component.

(8) Equipment that is in vacuum service is excluded from the requirements of this section.

(9) Equipment that is in organic HAP service, but is in such service less than 300 hours per calendar year, is excluded from the requirements of this section if it is identified as required in paragraph (g)(9) of this section. 40 CFR Ch. I (7–1–23 Edition)

(10) When each leak is detected by visual, audible, or olfactory means, or by monitoring as described in §63.180(b) or (c) of subpart H of this part, the following requirements apply:

(i) A weatherproof and readily visible identification, marked with the equipment identification number, shall be attached to the leaking equipment.

(ii) The identification on a valve in light liquid or gas/vapor service may be removed after it has been monitored as specified in paragraph (e)(7)(iii) of this section, and no leak has been detected during the follow-up monitoring. If an owner or operator elects to comply with $\S63.174(c)(1)(i)$, the identification on a connector may be removed after it has been monitored as specified in $\S63.174(c)(1)(i)$ and no leak is detected during that monitoring.

(iii) The identification on equipment, except as specified in paragraph (a)(10)(ii) of this section, may be removed after it has been repaired.

(b) *References.* The owner or operator shall comply with the provisions of subpart H of this part as specified in paragraphs (b)(1) through (3) of this section and with paragraph (b)(4) of this section for pressure relief devices. When the term "process unit" is used in subpart H of this part, it shall mean any group of processes for the purposes of this subpart. Groups of processes as used in this subpart may be any individual process or combination of processes.

(1) Sections 63.160, 63.161, 63.162, 63.163, 63.167, 63.168, 63.170, 63.173, 63.175, 63.176, 63.181, and 63.182 of subpart H of this part shall not apply for the purposes of this subpart MMM. The owner or operator shall comply with the provisions specified in paragraphs (b)(1)(i) through (viii) of this section.

(i) Sections 63.160 and 63.162 of subpart H of this part shall not apply, instead the owner or operator shall comply with paragraph (a) of this section;

(ii) Section 63.161 of subpart H of this part shall not apply, instead the owner or operator shall comply with §63.1361;

(iii) Sections 63.163 and 63.173 of subpart H of this part shall not apply, instead the owner or operator shall comply with paragraph (c) of this section;

(iv) Section 63.167 of subpart H of this part shall not apply, instead the owner

or operator shall comply with paragraph (d) of this section;

(v) Section 63.168 of subpart H of this part shall not apply, instead the owner or operator shall comply with paragraph (e) of this section;

(vi) Section 63.170 of subpart H of this part shall not apply, instead the owner or operator shall comply with §63.1362(b);

(vii) Section 63.181 of subpart H of this part shall not apply, instead the owner or operator shall comply with paragraph (g) of this section; and

(viii) Section 63.182 of subpart H of this part shall not apply, instead the owner or operator shall comply with paragraph (h) of this section.

(2) The owner or operator shall comply with §§63.164, 63.166, 63.169, 63.177, and 63.179 of subpart H of this part in their entirety, except that when these sections reference other sections of subpart H of this part, the owner or operator shall comply with the revised sections as specified in paragraphs (b)(1), (3), and (4) of this section. Section 63.164 of subpart H of this part applies to compressors. Section 63.166 of subpart H of this part applies to sampling connection systems. Section 63.169 of subpart H of this part applies to: pumps, valves, connectors, and agitators in heavy liquid service; instrumentation systems; and pressure relief devices in liquid service. Section 63.177 of subpart H of this subpart applies to general alternative means of emission limitation. Section 63.179 of subpart H of this part applies to alternative means of emission limitation for enclosed-vented process units.

(3) The owner or operator shall comply with \S 63.171, 63.172, 63.174, 63.178, and 63.180 of subpart H of this part with the differences specified in paragraphs (b)(3)(i) through (v) of this section.

(i) Section 63.171, Delay of repair, shall apply except §63.171(a) shall not apply. Delay of repair of equipment for which leaks have been detected is allowed if one of the following conditions exist:

(A) The repair is technically infeasible without a process shutdown. Repair of this equipment shall occur by the end of the next scheduled process shutdown. (B) The owner or operator determines that repair personnel would be exposed to an immediate danger if attempting to repair without a process shutdown. Repair of this equipment shall occur by the end of the next scheduled process shutdown.

(ii) Section 63.172, Closed-vent systems and control devices, shall apply for closed-vent systems used to comply with this section, and for control devices used to comply with this section only, except:

(A) Section 63.172(k) and (l) shall not apply. The owner or operator shall instead comply with paragraph (f) of this section.

(B) Owners or operators may, instead of complying with the provisions of §63.172(f), design a closed-vent system to operate at a pressure below atmospheric pressure. The system shall be equipped with at least one pressure gauge or other pressure measurement device that can be read from a readily accessible location to verify that negative pressure is being maintained in the closed-vent system when the associated control device is operating.

(iii) Section 63.174, Connectors, shall apply except:

(A) Section 63.174(b), (f), (g), and (h) shall not apply. In place of §63.174(b), the owner or operator shall comply with paragraphs (b)(3)(ii)(C) through (G) of this section. In place of §63.174(f), (g), and (h), the owner or operator shall comply with paragraph (f) of this section.

(B) Days that the connectors are not in organic HAP service shall not be considered part of the 3-month period in §63.174(c).

(C) If the percent leaking connectors in a group of processes was greater than or equal to 0.5 percent during the initial monitoring period, monitoring shall be performed once per year until the percent leaking connectors is less than 0.5 percent.

(D) If the percent leaking connectors in the group of processes was less than 0.5 percent, but equal to or greater than 0.25 percent, during the last required monitoring period, monitoring shall be performed once every 4 years. An owner or operator may comply with the requirements of this paragraph by monitoring at least 40 percent of the connectors in the first 2 years and the remainder of the connectors within the next 2 years. The percent leaking connectors will be calculated for the total of all monitoring performed during the 4-year period.

(E) The owner or operator shall increase the monitoring frequency to once every 2 years for the next monitoring period if leaking connectors comprise at least 0.5 percent but less than 1.0 percent of the connectors monitored within either the 4 years specified in paragraph (b)(3)(iii)(D) of this section, the first 4 years specified in paragraph (b)(3)(iii)(G) of this section, or the entire 8 years specified in paragraph (b)(3)(iii)(G) of this section. At the end of that 2-year monitoring period, the owner or operator shall monitor once per year while the percent leaking connectors is greater than or equal to 0.5 percent: if the percent leaking connectors is less than 0.5 percent, the owner or operator may again elect to monitor in accordance with paragraph (b)(3)(iii)(D) or (G) of this section, as applicable.

(F) If an owner or operator complying with the requirements of paragraph (b)(3)(iii)(D) or (G) of this section for a group of processes determines that 1 percent or greater of the connectors are leaking, the owner or operator shall increase the monitoring frequency to one time per year. The owner or operator may again elect to use the provisions of paragraph (b)(3)(iii)(D) or (G) of this section after a monitoring period in which less than 0.5 percent of the connectors are determined to be leaking.

(G) Monitoring shall be required once every 8 years, if the percent leaking connectors in the group of process units was less than 0.25 percent during the last required monitoring period. An owner or operator shall monitor at least 50 percent of the connectors in the first 4 years and the remainder of the connectors within the next 4 years. If the percent leaking connectors in the first 4 years is equal to or greater than 0.35 percent, the monitoring program shall revert at that time to the appropriate monitoring frequency specified in paragraph (b)(3)(iii)(D), (E), or (F) of this section.

40 CFR Ch. I (7–1–23 Edition)

(iv) Section 63.178, shall apply, except as specified in paragraphs (b)(3)(iv)(A) and (B) of this section.

(A) Section 63.178(b), requirements for pressure testing, shall apply to all processes, not just batch processes.

(B) For pumps, the phrase "at the frequencies specified in Table 1 of this subpart" in §63.178(c)(3)(iii) shall mean "quarterly" for the purposes of this subpart.

(v) Section 63.180 of subpart H of this part, Test methods and procedures, shall apply except §63.180(b)(4)(ii)(A) through (C) of subpart H of this part shall not apply. Calibration gases shall be a mixture of methane and air at a concentration of approximately, but less than, 10,000 parts per million methane for agitators, 2,000 parts per million for pumps, and 500 parts per million for all other equipment, except as provided in §63.180(b)(4)(iii) of subpart H of this part.

(4) Requirements for pressure relief devices. Except as specified in paragraph (b)(4)(iv) of this section, the owner or operator must comply with the operating and pressure release requirements specified in paragraphs (b)(4)(i) and (ii) of this section for pressure relief devices in organic HAP gas or vapor service. Except as specified in paragraph (b)(4)(iv) of this section, the owner or operator must also comply with the pressure release management requirements specified in paragraph (b)(4)(iii) of this section for all pressure relief devices in organic HAP service.

(i) Operating requirements. Except during a pressure release event, operate each pressure relief device in organic HAP gas or vapor service with an instrument reading of less than 500 ppm above background as detected by Method 21 of 40 CFR part 60, appendix A.

(ii) *Pressure release requirements.* For pressure relief devices in organic HAP gas or vapor service, comply with paragraphs (b)(4)(ii)(A) or (B) of this section, as applicable.

(A) If the pressure relief device does not consist of or include a rupture disk, conduct instrument monitoring, as detected by Method 21 of 40 CFR part 60, appendix A, no later than 5 calendar days after the pressure release to verify that the pressure relief device is operating with an instrument reading

§63.1363

of less than 500 ppm above background, except as provided in §63.171.

(B) If the pressure relief device consists of or includes a rupture disk, install a replacement disk as soon as practicable after a pressure release, but no later than 5 calendar days after the pressure release, except as provided in §63.171.

(iii) Pressure release management. Except as specified in paragraph (b)(4)(iv) of this section, pressure releases to the atmosphere from pressure relief devices in organic HAP service are prohibited, and the owner or operator must comply with the requirements specified in paragraphs (b)(4)(ii)(A) and (B) of this section for all pressure relief devices in organic HAP service.

(A) For each pressure relief device in organic HAP service, the owner or operator must equip each pressure relief device with a device(s) or use a monitoring system that is capable of:

(1) Identifying the pressure release;

(2) Recording the time and duration of each pressure release; and

(3) Notifying operators immediately that a pressure release is occurring. The device or monitoring system may be either specific to the pressure relief device itself or may be associated with the process system or piping, sufficient to indicate a pressure release to the atmosphere. Examples of these types of devices and systems include, but are not limited to, a rupture disk indicator, magnetic sensor, motion detector on the pressure relief valve stem, flow monitor, or pressure monitor.

(B) If any pressure relief device in organic HAP service releases to atmosphere as a result of a pressure release event, the owner or operator must calculate the quantity of organic HAP released during each pressure release event and report this quantity as required in paragraph (h)(3)(v) of this section. Calculations may be based on data from the pressure relief device monitoring alone or in combination with process parameter monitoring data and process knowledge.

(iv) Pressure relief devices routed to a control device, process, or drain system. If a pressure relief device in organic HAP service is designed and operated to route all pressure releases through a closed vent system to a control device, process, or drain system, the owner or operator is not required to comply with paragraphs (b)(4)(i), (ii), or (iii) (if applicable) of this section. Both the closed vent system and control device (if applicable) must meet the requirements of 63.172. The drain system (if applicable) must meet the requirements of 63.136.

(c) Standards for pumps in light liquid service and agitators in gas/vapor service and in light liquid service. (1) The provisions of this section apply to each pump that is in light liquid service, and to each agitator in gas/vapor service or in light liquid service.

(2)(i) *Monitoring*. Each pump and agitator subject to this section shall be monitored quarterly to detect leaks by the method specified in $\S63.180(b)$, except as provided in $\S63.177$, 63.178, paragraph (f) of this section, and paragraphs (c)(5) through (9) of this section.

(ii) *Leak definition*. The instrument reading, as determined by the method as specified in §63.180(b) of subpart H of this part, that defines a leak is:

(A) For agitators, an instrument reading of 10,000 parts per million or greater.

(B) For pumps, an instrument reading of 2,000 parts per million or greater.

(iii) Visual inspections. Each pump and agitator shall be checked by visual inspection each calendar week for indications of liquids dripping from the pump or agitator seal. If there are indications of liquids dripping from the seal at the time of the weekly inspection, the owner or operator shall follow the procedure specified in either paragraph (c)(2)(iii)(A) or (B) of this section prior to the next weekly inspection.

(A) The owner or operator shall monitor the pump or agitator by the method specified in §63.180(b). If the instrument reading indicates a leak as specified in paragraph (c)(2)(ii) of this section, a leak is detected.

(B) The owner or operator shall eliminate the visual indications of liquids dripping.

(3) Repair provisions. (i) When a leak is detected pursuant to paragraph (c)(2)(i), (c)(2)(iii)(A), (c)(5)(iv)(A), or (c)(5)(vi)(B) of this section, it shall be repaired as soon as practicable, but not later than 15 calendar days after it is
detected, except as provided in paragraph (b)(3)(i) of this section.

(ii) A first attempt at repair shall be made no later than 5 calendar days after the leak is detected. First attempts at repair include, but are not limited to, the following practices where practicable:

(A) Tightening of packing gland nuts. (B) Ensuring that the seal flush is operating at design pressure and temperature.

(4) Calculation of percent leakers. (i) The owner or operator shall decide no later than the end of the first monitoring period what groups of processes will be developed. Once the owner or operator has decided, all subsequent percent calculations shall be made on the same basis.

40 CFR Ch. I (7–1–23 Edition)

(ii) If, calculated on a 1-year rolling average, 10 percent or more of the pumps in a group of processes (or 3 pumps in a group of processes with fewer than 30 pumps) leak, the owner or operator shall monitor each pump once per month, until the calculated 1year rolling average value drops below 10 percent (or three pumps in a group of processes with fewer than 30 pumps).

(iii) The number of pumps in a group of processes shall be the sum of all the pumps in organic HAP service, except that pumps found leaking in a continuous process within 1 quarter after startup of the pump shall not count in the percent leaking pumps calculation for that one monitoring period only.

(iv) Percent leaking pumps shall be determined using Equation 3 of this subpart:

$$P_{\rm L} = \left[\left(P_{\rm L} - P_{\rm S} \right) / \left(P_{\rm T} - P_{\rm S} \right) \right] \times 100$$
 (Eq. 3)

Where:

- $%P_{L}$ = percent leaking pumps
- P_L = number of pumps found leaking as determined through quarterly monitoring as required in paragraphs (c)(2)(i) and (ii) of this section.
- P_T = total pumps in organic HAP service, including those meeting the criteria in paragraphs (c)(5) and (6) of this section
- $P_{\rm S} = \text{number of pumps in a continuous proc-} \\ \text{ess leaking within 1 quarter of startup} \\ \text{during the current monitoring period}$

(5) *Exemptions*. Each pump or agitator equipped with a dual mechanical seal system that includes a barrier fluid system and meets the requirements specified in paragraphs (c)(5)(i) through (vii) is exempt from the requirements of paragraphs (c)(1) through (c)(4)(ii)of this section, except as specified in paragraphs (c)(5)(iv)(A) and (vii) of this section.

(i) Each dual mechanical seal system is:

(A) Operated with the barrier fluid at a pressure that is at all times greater than the pump/agitator stuffing box pressure; or

(B) Equipped with a barrier fluid degassing reservoir that is connected by a closed-vent system to a control device that complies with the requirements of paragraph (b)(3)(ii) of this section; or

(C) Equipped with a closed-loop system that purges the barrier fluid into a process stream.

(ii) The barrier fluid is not in light liquid service.

(iii) Each barrier fluid system is equipped with a sensor that will detect failure of the seal system, the barrier fluid system, or both.

(iv) Each pump/agitator is checked by visual inspection each calendar week for indications of liquids dripping from the pump/agitator seal. If there are indications of liquids dripping from the pump or agitator seal at the time of the weekly inspection, the owner or operator shall follow the procedures specified in either paragraph (c)(5)(iv)(A) or (B) of this section prior to the next required inspection.

(A) The owner or operator shall monitor the pump or agitator using the method specified in $\S63.180(b)$ to determine if there is a leak of organic HAP in the barrier fluid. If the instrument reading indicates a leak, as specified in paragraph (c)(2)(ii) of this section, a leak is detected.

(B) The owner or operator shall eliminate the visual indications of liquids dripping.

(v) Each sensor as described in paragraph (c)(5)(iii) of this section is observed daily or is equipped with an alarm unless the pump is located within the boundary of an unmanned plant site.

(vi)(A) The owner or operator determines, based on design considerations and operating experience, criteria applicable to the presence and frequency of drips and to the sensor that indicate failure of the seal system, the barrier fluid system, or both.

(B) If indications of liquids dripping from the pump/agitator seal exceed the criteria established in paragraph (c)(5)(vi)(A) of this section, or if, based on the criteria established in paragraph (c)(5)(vi)(A) of this section, the sensor indicates failure of the seal system, the barrier fluid system, or both, a leak is detected.

(vii) When a leak is detected pursuant to paragraph (c)(5)(iv)(A) or (vi)(B) of this section, the leak must be repaired as specified in paragraph (c)(3) of this section.

(6) Any pump/agitator that is designed with no externally actuated shaft penetrating the pump/agitator housing is exempt from the requirements of paragraphs (c)(1) through (3) of this section.

(7) Any pump/agitator equipped with a closed-vent system capable of capturing and transporting any leakage from the seal or seals back to the process or to a control device that complies with the requirements of paragraph (b)(3)(ii) of this section is exempt from the requirements of paragraphs (c)(2) through (5) of this section.

(8) Any pump/agitator that is located within the boundary of an unmanned plant site is exempt from the weekly visual inspection requirement of paragraphs (c)(2)(ii) and (c)(5)(iv) of this section, and the daily requirements of paragraph (c)(5)(v) of this section, provided that each pump/agitator is visually inspected as often as practicable and at least monthly.

(9) If more than 90 percent of the pumps in a group of processes meet the criteria in either paragraph (c)(5) or (6) of this section, the group of processes

is exempt from the requirements of paragraph (c)(4) of this section.

(d) Standards: open-ended values or lines. (1)(i) Each open-ended value or line shall be equipped with a cap, blind flange, plug, or a second value, except as provided in $\S63.177$ of subpart H of this part and paragraphs (d)(4) through (6) of this section.

(ii) The cap, blind flange, plug, or second valve shall seal the open end at all times except during operations requiring process fluid flow through the open-ended valve or line, or during maintenance or repair. The cap, blind flange, plug, or second valve shall be in place within 1 hour of cessation of operations requiring process fluid flow through the open-ended valve or line, or within 1 hour of cessation of maintenance or repair.

(2) Each open-ended valve or line equipped with a second valve shall be operated in a manner such that the valve on the process fluid end is closed before the second valve is closed.

(3) When a double block and bleed system is being used, the bleed valve or line may remain open during operations that require venting the line between the block valves but shall comply with paragraph (d)(1) of this section at all other times.

(4) Open-ended valves or lines in an emergency shutdown system which are designed to open automatically in the event of a process upset are exempt from the requirements of paragraphs (d)(1) through (3) of this section.

(5) Open-ended valves or lines containing materials which would autocatalytically polymerize are exempt from the requirements of paragraphs (d)(1) through (3) of this section.

(6) Open-ended valves or lines containing materials which could cause an explosion, serious overpressure, or other safety hazard if capped or equipped with a double block and bleed system as specified in paragraphs (d)(1) through (3) of this section are exempt from the requirements of paragraphs (d)(1) through (3) of this section.

(e) Standards: valves in gas/vapor service and in light liquid service. (1) The provisions of this section apply to valves that are either in gas/vapor service or in light liquid service. (2) For existing and new affected sources, all valves subject to this section shall be monitored, except as provided in paragraph (f) of this section and in §63.177 of subpart H of this part, by no later than 1 year after the compliance date.

(3) Monitoring. The owner or operator of a source subject to this section shall monitor all valves, except as provided in paragraph (f) of this section and in §63.177 of subpart H of this part, at the intervals specified in paragraph (e)(4) of this section and shall comply with all other provisions of this section, except as provided in paragraph (b)(3)(i) of this section and §§ 63.178 and 63.179 of subpart H of this part.

(i) The valves shall be monitored to detect leaks by the method specified in §63.180(b) of subpart H of this part.

(ii) An instrument reading of 500 parts per million or greater defines a leak.

(4) Subsequent monitoring frequencies. After conducting the initial survey required in paragraph (e)(2) of this section, the owner or operator shall monitor valves for leaks at the intervals specified below:

(i) For a group of processes with 2 percent or greater leaking valves, calculated according to paragraph (e)(6) of this section, the owner or operator shall monitor each valve once per month, except as specified in paragraph (e)(9) of this section.

(ii) For a group of processes with less than 2 percent leaking valves, the owner or operator shall monitor each valve once each quarter, except as provided in paragraphs (e)(4)(iii) through (v) of this section.

(iii) For a group of processes with less than 1 percent leaking valves, the owner or operator may elect to monitor each valve once every 2 quarters.

(iv) For a group of processes with less than 0.5 percent leaking valves, the owner or operator may elect to monitor each valve once every 4 quarters.

(v) For a group of processes with less than 0.25 percent leaking valves, the owner or operator may elect to monitor each valve once every 2 years.

(5) Calculation of percent leakers. For a group of processes to which this subpart applies, the owner or operator may choose to subdivide the valves in 40 CFR Ch. I (7–1–23 Edition)

the applicable group of processes and apply the provisions of paragraph (e)(4) of this section to each subgroup. If the owner or operator elects to subdivide the valves in the applicable group of processes, then the provisions of paragraphs (e)(5)(i) through (viii) of this section apply.

(i) The overall performance of total valves in the applicable group of processes must be less than 2 percent leaking valves, as detected according to paragraphs (e)(3)(i) and (ii) of this section and as calculated according to paragraphs (e)(6)(ii) and (iii) of this section.

(ii) The initial assignment or subsequent reassignment of valves to subgroups shall be governed by the provisions of paragraphs (e)(5)(ii) (A) through (C) of this section.

(A) The owner or operator shall determine which valves are assigned to each subgroup. Valves with less than 1 year of monitoring data or valves not monitored within the last 12 months must be placed initially into the most frequently monitored subgroup until at least 1 year of monitoring data have been obtained.

(B) Any valve or group of valves can be reassigned from a less frequently monitored subgroup to a more frequently monitored subgroup provided that the valves to be reassigned were monitored during the most recent monitoring period for the less frequently monitored subgroup. The monitoring results must be included with the less frequently monitored subgroup's monitoring event and associated next percent leaking valves calculation for that group.

(C) Any valve or group of valves can be reassigned from a more frequently monitored subgroup to a less frequently monitored subgroup provided that the valves to be reassigned have not leaked for the period of the less frequently monitored subgroup (e.g., for the last 12 months, if the valve or group of valves is to be reassigned to a subgroup being monitored annually). Nonrepairable valves may not be reassigned to a less frequently monitored subgroup.

(iii) The owner or operator shall determine every 6 months if the overall

performance of total valves in the applicable group of processes is less than 2 percent leaking valves and so indicate the performance in the next Periodic report. If the overall performance of total valves in the applicable group of processes is 2 percent leaking valves or greater, the owner or operator shall revert to the program required in paragraphs (e)(2) through (4) of this section. The overall performance of total valves in the applicable group of processes shall be calculated as a weighted average of the percent leaking valves of each subgroup according to Equation 4 of this subpart:

$$\%V_{LO} = \frac{\sum_{i=1}^{n} (\%V_{Li} \times V_{i})}{\sum_{i=1}^{n} V_{i}}$$
 (Eq. 4)

Where:

 $%V_{LO}$ = overall performance of total valves in the applicable group of processes

 $\% V_{Li}$ = percent leaking values in subgroup i, most recent value calculated according to the procedures in paragraphs (e)(6)(ii) and (iii) of this section

 V_i = number of valves in subgroup i

n = number of subgroups

(iv) *Records*. In addition to records required by paragraph (g) of this section, the owner or operator shall maintain records specified in paragraphs (e)(5)(iv)(A) through (D) of this section. (A) Which valves are assigned to each

subgroup,(B) Monitoring results and calculations made for each subgroup for each

(C) Which valves are reassigned and

when they were reassigned, and

(D) The results of the semiannual overall performance calculation required in paragraph (e)(5)(iii) of this section.

(v) The owner or operator shall notify the Administrator no later than 30 days prior to the beginning of the next monitoring period of the decision to subgroup valves. The notification shall identify the participating processes and the valves assigned to each subgroup.

(vi) Semiannual reports. In addition to the information required by paragraph (h)(3) of this section, the owner or oper-

ator shall submit in the Periodic reports the information specified in paragraphs (e)(5)(vi)(A) and (B) of this section.

(A) Valve reassignments occurring during the reporting period, and

(B) Results of the semiannual overall performance calculation required by paragraph (e)(5)(iii) of this section.

(vii) To determine the monitoring frequency for each subgroup, the calculation procedures of paragraph (e)(6)(iii) of this section shall be used.

(viii) Except for the overall performance calculations required by paragraphs (e)(5)(i) and (iii) of this section, each subgroup shall be treated as if it were a process for the purposes of applying the provisions of this section.

(6)(i) The owner or operator shall decide no later than the implementation date of this subpart or upon revision of an operating permit how to group the processes. Once the owner or operator has decided, all subsequent percentage calculations shall be made on the same basis.

(ii) Percent leaking values for each group of processes or subgroup shall be determined using Equation 5 of this subpart:

$$%V_{L} = [V_{L}/V_{T}] \times 100$$
 (Eq. 5)

Where:

 $%V_L$ = percent leaking values

- V_L = number of valves found leaking excluding nonrepairables as provided in paragraph (e)(6)(iv)(A) of this section
- V_T = total valves monitored, in a monitoring period excluding valves monitored as required by paragraph (e)(7)(iii) of this section

(iii) When determining monitoring frequency for each group of processes or subgroup subject to monthly, quarterly, or semiannual monitoring frequencies, the percent leaking valves shall be the arithmetic average of the percent leaking valves from the last two monitoring periods. When determining monitoring frequency for each group of processes or subgroup subject to annual or biennial (once every 2 years) monitoring frequencies, the percent leaking valves shall be the arithmetic average of the percent leaking valves from the last three monitoring periods.

(iv)(A) Nonrepairable valves shall be included in the calculation of percent leaking valves the first time the valve is identified as leaking and nonrepairable and as required to comply with paragraph (e)(6)(iv)(B) of this section. Otherwise, a number of nonrepairable valves (identified and included in the percent leaking calculation in a previous period) up to a maximum of 1 percent of the total number of valves in organic HAP service at a process may be excluded from calculation of percent leaking valves for subsequent monitoring periods.

(B) If the number of nonrepairable valves exceeds 1 percent of the total number of valves in organic HAP service at a process, the number of nonrepairable valves exceeding 1 percent of the total number of valves in organic HAP service shall be included in the calculation of percent leaking valves.

(7) *Repair provisions*. (i) When a leak is detected, it shall be repaired as soon as practicable, but no later than 15 calendar days after the leak is detected, except as provided in paragraph (b)(3)(i) of this section.

(ii) A first attempt at repair shall be made no later than 5 calendar days after each leak is detected.

(iii) When a leak is repaired, the valve shall be monitored at least once within the first 3 months after its repair. Days that the valve is not in organic HAP service shall not be considered part of this 3-month period. The monitoring required by this paragraph is in addition to the monitoring required to satisfy the definitions of "repaired" and "first attempt at repair."

(A) The monitoring shall be conducted as specified in §63.180(b) and (c) as appropriate, to determine whether the valve has resumed leaking.

(B) Periodic monitoring required by paragraphs (e)(2) through (4) of this section may be used to satisfy the requirements of paragraph (e)(7)(iii) of this section, if the timing of the monitoring period coincides with the time specified in paragraph (e)(7)(iii) of this section. Alternatively, other monitoring may be performed to satisfy the requirements of paragraph (e)(7)(ii) of this section, regardless of whether the timing of the monitoring period for periodic monitoring coincides with the 40 CFR Ch. I (7–1–23 Edition)

time specified in paragraph (e)(7)(iii) of this section.

(C) If a leak is detected by monitoring that is conducted pursuant to paragraph (e)(7)(iii) of this section, the owner or operator shall follow the provisions of paragraphs (e)(7)(iii)(C)(I) and (2) of this section to determine whether that valve must be counted as a leaking valve for purposes of paragraph (e)(6) of this section.

(1) If the owner or operator elects to use periodic monitoring required by paragraphs (e)(2) through (4) of this section to satisfy the requirements of paragraph (e)(7)(iii) of this section, then the valve shall be counted as a leaking valve.

(2) If the owner or operator elects to use other monitoring prior to the periodic monitoring required by paragraphs (e)(2) through (4) of this section to satisfy the requirements of paragraph (e)(7)(iii) of this section, then the valve shall be counted as a leaking valve unless it is repaired and shown by periodic monitoring not to be leaking.

(8) First attempts at repair include, but are not limited to, the following practices where practicable:

(i) Tightening of bonnet bolts,

(ii) Replacement of bonnet bolts,

(iii) Tightening of packing gland nuts, and

(iv) Injection of lubricant into lubricated packing.

(9) Any equipment located at a plant site with fewer than 250 valves in organic HAP service in the affected source is exempt from the requirements for monthly monitoring specified in paragraph (e)(4)(i) of this section. Instead, the owner or operator shall monitor each valve in organic HAP service for leaks once each quarter, or comply with paragraph (e)(4)(ii), (iv), or (v) of this section, except as provided in paragraph (f) of this section.

(f) Unsafe to monitor, difficult-to-monitor, and inaccessible equipment. (1) Equipment that is designated as unsafe-to-monitor, difficult-to-monitor, or inaccessible is exempt from the requirements as specified in paragraphs (f)(1)(i) through (iv) of this section provided the owner or operator meets the requirements specified in paragraph

(f)(2), (3), or (4) of this section, as applicable. All equipment, except connectors that meet the requirements in paragraph (f)(4) of this section, must be assigned to a group of processes. Ceramic or ceramic-lined connectors are subject to the same requirements as inaccessible connectors.

(i) For pumps and agitators, paragraphs (c)(2), (3), and (4) of this section do not apply.

(ii) For values, paragraphs (e)(2) through (7) of this section do not apply.

(iii) For connectors, §63.174(b) through (e) and paragraphs (b)(3)(iii)(C) through (G) of this section do not apply.

(iv) For closed-vent systems, (3.172(f)(1), (f)(2), and (g) do not apply.

(2) Equipment that is unsafe-to-monitor. (i) Valves, connectors, agitators, and any part of closed-vent systems may be designated as unsafe-to-monitor if the owner or operator determines that monitoring personnel would be exposed to an immediate danger as a consequence of complying with the monitoring requirements identified in paragraphs (f)(1)(i) through (iii) of this section, or the inspection requirements identified in paragraph (f)(1)(iv) of this section.

(ii) The owner or operator of equipment that is designated as unsafe-tomonitor must have a written plan that requires monitoring of the equipment as frequently as practicable during safe-to-monitor times. For valves, connectors, and agitators, monitoring shall not be more frequent than the periodic monitoring schedule otherwise applicable to the group of processes in which the equipment is located. For closed-vent systems, inspections shall not be more frequent than annually.

(3) Equipment that is difficult-to-monitor. (i) A valve, agitator, pump, or any part of a closed-vent system may be designated as difficult-to-monitor if the owner or operator determines that the equipment cannot be monitored or inspected without elevating the monitoring personnel more than 2 meters above a support surface or the equipment is not accessible in a safe manner when it is in organic HAP service;

(ii) At a new affected source, an owner or operator may designate no more than 3 percent of valves as difficult-to-monitor.

(iii) The owner or operator of valves, agitators, or pumps designated as difficult-to-monitor must have a written plan that requires monitoring of the equipment at least once per calendar year or on the periodic monitoring schedule otherwise applicable to the group of processes in which the equipment is located, whichever is less frequent. For any part of a closed-vent system designated as difficult-to-monitor, the owner or operator must have a written plan that requires inspection of the closed-vent system at least once every 5 years.

(4) Inaccessible, ceramic, or ceramiclined connectors. (i) A connector may be designated as inaccessible if it is:

(A) Buried;

(B) Insulated in a manner that prevents access to the equipment by a monitor probe;

(C) Obstructed by equipment or piping that prevents access to the equipment by a monitor probe;

(D) Unable to be reached from a wheeled scissor-lift or hydraulic-type scaffold which would allow access to equipment up to 7.6 meters above the ground; or

(E) Not able to be accessed at any time in a safe manner to perform monitoring. Unsafe access includes, but is not limited to, the use of a wheeled scissor-lift on unstable or uneven terrain, the use of a motorized man-lift basket in areas where an ignition potential exists, or access would require near proximity to hazards such as electrical lines, or would risk damage to equipment.

(F) Would require elevating the monitoring personnel more than 2 meters above a permanent support surface or would require the erection of scaffold.

(ii) At a new affected source, an owner or operator may designate no more than 3 percent of connectors as inaccessible.

(iii) If any inaccessible, ceramic, or ceramic-lined connector is observed by visual, audible, olfactory, or other means to be leaking, the leak shall be repaired as soon as practicable, but no later than 15 calendar days after the leak is detected, except as provided in paragraph (b)(3)(i) of this section.

40 CFR Ch. I (7–1–23 Edition)

(iv) Any connector that is inaccessible or that is ceramic or ceramiclined is exempt from the recordkeeping and reporting requirements of paragraphs (g) and (h) of this section.

(g) Recordkeeping requirements. (1) An owner or operator of more than one group of processes subject to the provisions of this section may comply with the record keeping requirements for the groups of processes in one recordkeeping system if the system identifies with each record the program being implemented (e.g., quarterly monitoring) for each type of equipment. All records and information required by this section shall be maintained in a manner that can be readily accessed at the plant site. This could include physically locating the records at the plant site or accessing the records from a central location by computer at the plant site.

(2) General record keeping. Except as provided in paragraph (g)(5) of this section, the following information pertaining to all equipment subject to the requirements in this section shall be recorded:

(i)(A) A list of identification numbers for equipment (except instrumentation systems) subject to the requirements of this section. Connectors, except those subject to paragraph (f) of this section, need not be individually identified if all connectors in a designated area or length of pipe subject to the provisions of this section are identified as a group, and the number of subject connectors is indicated. The list for each type of equipment shall be completed no later than the completion of the initial survey required for that component. The list of identification numbers shall be updated, if needed, to incorporate equipment changes within 15 calendar days of the completion of each monitoring survey for the type of equipment component monitored.

(B) A schedule for monitoring connectors subject to the provisions of $\S63.174(a)$ of subpart H of this part and valves subject to the provisions of paragraph (e)(4) of this section.

(C) Physical tagging of the equipment is not required to indicate that it is in organic HAP service. Equipment subject to the provisions of this section may be identified on a plant site plan, in log entries, or by other appropriate methods.

(ii)(A) A list of identification numbers for equipment that the owner or operator elects to equip with a closedvent system and control device, subject to the provisions of paragraphs (b)(4)(iv) or (c)(7) of this section or §63.164(h).

(B) A list of identification numbers for compressors that the owner or operator elects to designate as operating with an instrument reading of less than 500 parts per million above background, under the provisions of §63.164(i) of subpart H of this part.

(iii)(A) A list of identification numbers for pressure relief devices subject to the provisions in paragraph (b)(4)(i) of this section.

(B) A list of identification numbers for pressure relief devices equipped with rupture disks, subject to the provisions of paragraph (b)(4)(ii)(B) of this section.

(iv) Identification of instrumentation systems subject to the provisions of this section. Individual components in an instrumentation system need not be identified.

(v) The following information shall be recorded for each dual mechanical seal system:

(A) Design criteria required by paragraph (c)(5)(vi)(A) of this section and §63.164(e)(2) of subpart H of this part, and an explanation of the design criteria; and

(B) Any changes to these criteria and the reasons for the changes.

(vi) A list of equipment designated as unsafe-to-monitor or difficult-to-monitor under paragraph (f) of this section and a copy of the plan for monitoring this equipment.

(vii) A list of connectors removed from and added to the process, as described in $\S63.174(i)(1)$ of subpart H of this part, and documentation of the integrity of the weld for any removed connectors, as required in $\S63.174(j)$ of subpart H of this part. This is not required unless the net credits for removed connectors is expected to be used.

(viii) For batch processes that the owner or operator elects to monitor as provided under §63.178(c) of subpart H of this part, a list of equipment added

to batch product processes since the last monitoring period required in §63.178(c)(3)(ii) and (iii) of subpart H of this part. This list must be completed for each type of equipment within 15 calendar days of the completion of the each monitoring survey for the type of equipment monitored.

(3) Records of visual inspections. For visual inspections of equipment subject to the provisions of paragraphs (c)(2)(iii) and (c)(5)(iv) of this section, the owner or operator shall document that the inspection was conducted and the date of the inspection. The owner or operator shall maintain records as specified in paragraph (g)(4) of this section for leaking equipment identified in this inspection, except as provided in paragraph (g)(5) of this section. These records shall be retained for 5 years.

(4) Monitoring records. When each leak is detected as specified in paragraphs (c) and (e) of this section and §§ 63.164, 63.169, 63.172, and 63.174 of subpart H of this part, the owner or operator shall record the information specified in paragraphs (g)(4)(i) through (ix) of this section. All records shall be retained for 5 years, in accordance with the requirements of § 63.10(b)(1) of subpart A of this part.

(i) The instrument and the equipment identification number and the operator name, initials, or identification number.

(ii) The date the leak was detected and the date of first attempt to repair the leak.

(iii) The date of successful repair of the leak.

(iv) If postrepair monitoring is required, maximum instrument reading measured by Method 21 of 40 CFR part 60, appendix A, after it is successfully repaired or determined to be nonrepairable.

(v) "Repair delayed" and the reason for the delay if a leak is not repaired within 15 calendar days after discovery of the leak.

(A) The owner or operator may develop a written procedure that identifies the conditions that justify a delay of repair. The written procedures must be maintained at the plant site. Reasons for delay of repair may be documented by citing the relevant sections of the written procedure.

(B) If delay of repair was caused by depletion of stocked parts, there must be documentation that the spare parts were sufficiently stocked onsite before depletion and the reason for depletion.

(vi) If repairs were delayed, dates of process shutdowns that occur while the equipment is unrepaired.

(vii)(A) If the alternative in §63.174(c)(1)(ii) of subpart H of this part is not in use for the monitoring period, identification, either by list, location (area or grouping), or tagging of connectors disturbed since the last monitoring period required in §63.174(b) of subpart H of this part, as described in §63.174(c)(1) of subpart H of this part.

(B) The date and results of follow-up monitoring as required in §63.174(c) of subpart H of this part. If identification of disturbed connectors is made by location, then all connectors within the designated location shall be monitored.

(viii) The date and results of the monitoring required in $\S63.178(c)(3)(i)$ of subpart H of this part for equipment added to a batch process since the last monitoring period required in $\S63.178(c)(3)(i)$ and (iii) of subpart H of this part. If no leaking equipment is found in this monitoring, the owner or operator shall record that the inspection was performed. Records of the actual monitoring results are not required.

(ix) Copies of the periodic reports as specified in paragraph (h)(3) of this section, if records are not maintained on a computerized data base capable of generating summary reports from the records.

(5) Records of pressure tests. The owner or operator who elects to pressure test a process equipment train and supply lines between storage and processing areas to demonstrate compliance with this section is exempt from the requirements of paragraphs (g)(2), (3), (4), and (6) of this section. Instead, the owner or operator shall maintain records of the following information:

(i) The identification of each product, or product code, produced during the calendar year. It is not necessary to identify individual items of equipment in the process equipment train.

§63.1363

(ii) Records demonstrating the proportion of the time during the calendar year the equipment is in use in the process that is subject to the provisions of this subpart. Examples of suitable documentation are records of time in use for individual pieces of equipment or average time in use for the process unit. These records are not required if the owner or operator does not adjust monitoring frequency by the time in use, as provided in $\S63.178(c)(3)(iii)$ of subpart H of this part.

(iii) Physical tagging of the equipment to identify that it is in organic HAP service and subject to the provisions of this section is not required. Equipment in a process subject to the provisions of this section may be identified on a plant site plan, in log entries, or by other appropriate methods.

(iv) The dates of each pressure test required in §63.178(b) of subpart H of this part, the test pressure, and the pressure drop observed during the test.

(v) Records of any visible, audible, or olfactory evidence of fluid loss.

(vi) When a process equipment train does not pass two consecutive pressure tests, the following information shall be recorded in a log and kept for 2 years:

(A) The date of each pressure test and the date of each leak repair attempt.

(B) Repair methods applied in each attempt to repair the leak.

(C) The reason for the delay of repair.

(D) The expected date for delivery of the replacement equipment and the actual date of delivery of the replacement equipment.

(E) The date of successful repair.

(6) Records of compressor and pressure relief device compliance tests. The dates and results of each compliance test required for compressors subject to the provisions in §63.164(i) and the dates and results of the Method 21 of 40 CFR part 60, appendix A, monitoring following a pressure release for each pressure relief device subject to the provisions in paragraphs (b)(4)(i) and (ii) of this section. The results shall include:

(i) The background level measured during each compliance test.

40 CFR Ch. I (7–1–23 Edition)

(ii) The maximum instrument reading measured at each piece of equipment during each compliance test.

(7) Records for closed-vent systems. The owner or operator shall maintain records of the information specified in paragraphs (g)(7)(i) through (iii) of this section for closed-vent systems and control devices subject to the provisions of paragraph (b)(3)(ii) of this section. The records specified in paragraph (g)(7)(i) of this section shall be retained for the life of the equipment. The records specified in paragraphs (g)(7)(ii) and (iii) of this section shall be retained for 5 years.

(i) The design specifications and performance demonstrations specified in paragraphs (g)(7)(i)(A) through (D) of this section.

(A) Detailed schematics, design specifications of the control device, and piping and instrumentation diagrams.

(B) The dates and descriptions of any changes in the design specifications.

(C) The flare design (i.e., steam assisted, air assisted, or nonassisted) and the results of the compliance demonstration required by §63.11(b) of subpart A of this part.

(D) A description of the parameter or parameters monitored, as required in paragraph (b)(3)(ii) of this section, to ensure that control devices are operated and maintained in conformance with their design and an explanation of why that parameter (or parameters) was selected for the monitoring.

(ii) Records of operation of closedvent systems and control devices.

(A) Dates and durations when the closed-vent systems and control devices required in paragraph (c) of this section and §§63.164 through 63.166 of subpart H of this part are not operated as designed as indicated by the monitored parameters, including periods when a flare pilot light system does not have a flame.

(B) Dates and durations during which the monitoring system or monitoring device is inoperative.

(C) Dates and durations of startups and shutdowns of control devices required in paragraph (c) of this section and \S 63.164 through 63.166 of subpart H of this part.

(iii) Records of inspections of closedvent systems subject to the provisions of §63.172 of subpart H of this part.

(A) For each inspection conducted in accordance with the provisions of $\S63.172(f)(1)$ or (2) of subpart H of this part during which no leaks were detected, a record that the inspection was performed, the date of the inspection, and a statement that no leaks were detected.

(B) For each inspection conducted in accordance with the provisions of §63.172(f)(1) or (f)(2) of subpart H of this part during which leaks were detected, the information specified in paragraph (g)(4) of this section shall be recorded.

(8) Records for components in heavy liquid service. Information, data, and analysis used to determine that a piece of equipment or process is in heavy liquid service shall be recorded. Such a determination shall include an analysis or demonstration that the process fluids do not meet the criteria of "in light liquid or gas/vapor service." Examples of information that could document this include, but are not limited to, records of chemicals purchased for the process, analyses of process stream composition, engineering calculations, or process knowledge.

(9) *Records of exempt components*. Identification, either by list, location (area or group), or other method of equipment in organic HAP service less than 300 hr/yr subject to the provisions of this section.

(10) Records of alternative means of compliance determination. Owners and operators choosing to comply with the requirements of §63.179 of subpart H of this part shall maintain the following records:

(i) Identification of the process(es) and the organic HAP they handle.

(ii) A schematic of the process, enclosure, and closed-vent system.

(iii) A description of the system used to create a negative pressure in the enclosure to ensure that all emissions are routed to the control device.

(11) Records of pressure releases to the atmosphere from pressure relief devices. For pressure relief devices in organic HAP service subject to paragraph (b)(4)(iii) of this section, keep records of each pressure release to the atmos-

phere, including the following information:

(i) The source, nature, and cause of the pressure release.

(ii) The date, time, and duration of the pressure release.

(iii) The quantity of total HAP emitted during the pressure release and the calculations used for determining this quantity.

(iv) The actions taken to prevent this pressure release.

(v) The measures adopted to prevent future such pressure releases.

(h) Reporting Requirements. (1) Each owner or operator of a source subject to this section shall submit the reports listed in paragraphs (h)(1)(i) and (ii) of this section.

(i) A Notification of Compliance Status report described in paragraph (h)(2) of this section, and

(ii) Periodic reports described in paragraph (h)(3) of this section.

(2) Notification of compliance status report. Each owner or operator of a source subject to this section shall submit the information specified in paragraphs (h)(2)(i) through (iii) of this section in the Notification of Compliance Status report described in §63.1368(f). For pressure relief devices subject to requirements the of paragraph (b)(4)(iii) of this section, the owner or operator shall submit the information listed in paragraph (h)(2)(iv) of this section in the Notification of Compliance Status within 150 days after the first applicable compliance date for pressure relief device monitoring. Section 63.9(j) of subpart A of this part shall not apply to the Notification of Compliance Status report.

(i) The notification shall provide the information listed in paragraphs (h)(2)(i)(A) through (C) of this section for each group of processes subject to the requirements of paragraphs (b) through (g) of this section.

(A) Identification of the group of processes.

(B) Approximate number of each equipment type (e.g., valves, pumps) in organic HAP service, excluding equipment in vacuum service.

(C) Method of compliance with the standard (for example, "monthly leak detection and repair" or "equipped with dual mechanical seals").

(ii) The notification shall provide the information listed in paragraphs (h)(2)(ii)(A) and (B) of this section for each process subject to the requirements of paragraph (b)(3)(iv) of this section and §63.178(b) of subpart H of this part.

(A) Products or product codes subject to the provisions of this section, and

(B) Planned schedule for pressure testing when equipment is configured for production of products subject to the provisions of this section.

(iii) The notification shall provide the information listed in paragraphs (h)(2)(iii)(A) and (B) of this section for each process subject to the requirements in 63.179 of subpart H of this part.

(A) Process identification.

(B) A description of the system used to create a negative pressure in the enclosure and the control device used to comply with the requirements of paragraph (b)(3)(ii) of this section.

(iv) For pressure relief devices in organic HAP service, a description of the device or monitoring system to be implemented, including the pressure relief devices and process parameters to be monitored (if applicable), a description of the alarms or other methods by which operators will be notified of a pressure release, and a description of how the owner or operator will determine the information to be recorded under paragraphs (g)(11)(ii) and (iii) of this section (i.e., the duration of the pressure release and the methodology and calculations for determining of the quantity of total HAP emitted during the pressure release).

(3) *Periodic reports.* The owner or operator of a source subject to this section shall submit Periodic reports.

(i) A report containing the information in paragraphs (h)(3)(ii) through (v) of this section shall be submitted semiannually. The first Periodic report shall be submitted no later than 240 days after the date the Notification of Compliance Status report is due and shall cover the 6-month period beginning on the date the Notification of Compliance Status report is due. Each subsequent Periodic report shall cover the 6-month period following the preceding period. 40 CFR Ch. I (7–1–23 Edition)

(ii) For equipment complying with the provisions of paragraphs (b) through (g) of this section, the Periodic report shall contain the summary information listed in paragraphs (h)(3)(ii)(A) through (L) of this section for each monitoring period during the 6-month period.

(A) The number of valves for which leaks were detected as described in paragraph (e)(2) of this section, the percent leakers, and the total number of valves monitored;

(B) The number of valves for which leaks were not repaired as required in paragraph (e)(7) of this section, identifying the number of those that are determined nonrepairable;

(C) The number of pumps and agitators for which leaks were detected as described in paragraph (c)(2) of this section, the percent leakers, and the total number of pumps and agitators monitored;

(D) The number of pumps and agitators for which leaks were not repaired as required in paragraph (c)(3) of this section;

(E) The number of compressors for which leaks were detected as described in §63.164(f) of subpart H of this part;

(F) The number of compressors for which leaks were not repaired as required in §63.164(g) of subpart H of this part;

(G) The number of connectors for which leaks were detected as described in §63.174(a) of subpart H of this part, the percent of connectors leaking, and the total number of connectors monitored;

(H) The number of connectors for which leaks were not repaired as required in §63.174(d) of subpart H of this part, identifying the number of those that are determined nonrepairable;

(I) The facts that explain any delay of repairs and, where appropriate, why a process shutdown was technically infeasible.

(J) The results of all monitoring to show compliance with \S 63.164(i) and 63.172(f) conducted within the semiannual reporting period.

(K) If applicable, the initiation of a monthly monitoring program under either paragraph (c)(4)(i) or paragraph (e)(4)(i)(A) of this section.

(L) If applicable, notification of a change in connector monitoring alternatives as described in 63.174(c)(1) of subpart H of this part.

(iii) For owners or operators electing to meet the requirements of $\S63.178(b)$ of subpart H of this part, the Periodic report shall include the information listed in paragraphs (h)(3)(iii) (A) through (E) of this section for each process.

(A) Product process equipment train identification;

(B) The number of pressure tests conducted;

(C) The number of pressure tests where the equipment train failed either the retest or two consecutive pressure tests;

(D) The facts that explain any delay of repairs; and

(E) The results of all monitoring to determine compliance with §63.172(f) of subpart H of this part.

(iv) Any change in the information submitted under paragraph (h)(2) of this section shall be provided in the next Periodic report.

(v) For pressure relief devices in organic HAP service, Periodic Reports must include the information specified in paragraphs (h)(3)(v)(A) through (C) of this section.

(A) For pressure relief devices in organic HAP service subject to paragraph (b)(4) of this section, report confirmation that all monitoring to show compliance was conducted within the reporting period.

(B) For pressure relief devices in organic HAP gas or vapor service subject to paragraph (b)(4)(ii) of this section, report any instrument reading of 500 ppm above background or greater, more than 5 calendar days after the pressure release.

(C) For pressure relief devices in organic HAP service subject to paragraph (b)(4)(iii) of this section, report each pressure release to the atmosphere, including the following information:

(1) The source, nature, and cause of the pressure release.

(2) The date, time, and duration of the pressure release.

(3) The quantity of total HAP emitted during the pressure release and the method used for determining this quantity. $\left(4\right)$ The actions taken to prevent this pressure release.

(5) The measures adopted to prevent future such pressure releases.

[64 FR 33589, June 23, 1999, as amended at 67 FR 59345, Sept. 20, 2002; 79 FR 17372, Mar. 27, 2014]

§63.1364 Compliance dates.

(a) Compliance dates for existing sources. (1) An owner or operator of an existing affected source must comply with the provisions in this subpart (except §63.1363(b)(4)(iii)) by December 23, 2003. Compliance with the pressure relief device monitoring provisions of §63.1363(b)(4)(iii) shall occur no later than March 27, 2017.

(2) Pursuant to section 112(i)(3)(B) of the CAA, an owner or operator of an existing source may request an extension of up to 1 additional year to comply with the provisions of this subpart if the additional time is needed for the installation of controls.

(i) For purposes of this subpart, a request for an extension shall be submitted no later than 120 days prior to the compliance date specified in paragraph (a)(1) of this section, except as provided in paragraph (a)(2)(ii) of this section. The dates specified in \S 63.6(i) of subpart A of this part for submittal of requests for extensions shall not apply to sources subject to this subpart.

(ii) An owner or operator may submit a compliance extension request after the date specified in paragraph (a)(1)(i)of this section provided the need for the compliance extension arose after that date and before the otherwise applicable compliance date, and the need arose due to circumstances beyond reasonable control of the owner or operator. This request shall include the data described in §63.6(i)(8)(A), (B), and (D) of subpart A of this part.

(b) Compliance dates for new and reconstructed sources. An owner or operator of a new or reconstructed affected source must comply with the provisions of this subpart (except §63.1363(b)(4)(ii)) on June 23, 1999 or upon startup, whichever is later. New or reconstructed affected sources that commenced construction after November 10, 1997, but on or before January 9, 2012, must be in compliance with the pressure relief device monitoring provisions of §63.1363(b)(4)(iii) no later than March 27, 2017. New or reconstructed sources that commenced construction after January 9, 2012, must be in compliance with the pressure relief device monitoring provisions of §63.1363(b)(4)(iii) upon initial startup or by March 27, 2014, whichever is later.

[64 FR 33589, June 23, 1999, as amended at 67 FR 13511, Mar. 22, 2002; 67 FR 38203, June 3, 2002; 79 FR 17374, Mar. 27, 2014]

§63.1365 Test methods and initial compliance procedures.

(a) General. Except as specified in paragraph (a)(4) of this section, the procedures specified in paragraphs (c), (d), (e), (f), and (g) of this section are required to demonstrate initial compliance with §63.1362(b), (c), (d), (f), and (g), respectively. The provisions in paragraph (a)(1) of this section apply to design evaluations that are used to demonstrate compliance with the standards for process vents and storage vessels. The provisions in paragraph (a)(2) of this section apply to performance tests that are specified in paragraphs (c), (d), and (e) of this section. The provisions in paragraph (a)(3) of this section describe initial compliance procedures for flares. The provisions in paragraph (a)(5) of this section are used to demonstrate initial compliance with the alternative standards specified in (c)(4). The provisions in paragraph (a)(6) of this section are used to comply with the outlet concentration requirements specified in §63.1362(b)(2)(iv)(A), (b)(3)(ii). (b)(4)(ii)(A), (b)(5)(ii), and (b)(5)(iii).

(1) Design evaluation. To demonstrate that a control device meets the required control efficiency, a design evaluation must address the composition and HAP concentration of the vent stream entering the control device. A design evaluation also must address other vent stream characteristics and control device operating parameters as specified in any one of paragraphs (a)(1)(i) through (vii) of this section, depending on the type of control device that is used. If the vent stream is not the only inlet to the control device, the efficiency demonstration also must consider all other vapors, gases, and

40 CFR Ch. I (7–1–23 Edition)

liquids, other than fuels, received by the control device.

(i) For an enclosed combustion device used to comply with the provisions of (3.1362(b)(2)(iv), (b)(4)(ii), (c)(2)(iv)(B),or (c)(3) with a minimum residence time of 0.5 seconds and a minimum temperature of 760 °C, the design evaluation must document that these conditions exist.

(ii) For a combustion control device that does not satisfy the criteria in paragraph (a)(1)(i) of this section, the design evaluation must document control efficiency and address the following characteristics, depending on the type of control device:

(A) For a thermal vapor incinerator, the design evaluation must consider the autoignition temperature of the organic HAP, must consider the vent stream flow rate, and must establish the design minimum and average temperature in the combustion zone and the combustion zone residence time.

(B) For a catalytic vapor incinerator, the design evaluation must consider the vent stream flow rate and must establish the design minimum and average temperatures across the catalyst bed inlet and outlet.

(C) For a boiler or process heater, the design evaluation must consider the vent stream flow rate, must establish the design minimum and average flame zone temperatures and combustion zone residence time, and must describe the method and location where the vent stream is introduced into the flame zone.

(iii) For a condenser, the design evaluation must consider the vent stream flow rate, relative humidity, and temperature, and must establish the maximum temperature of the condenser exhaust vent stream and the corresponding outlet organic HAP compound concentration level or emission rate for which the required reduction is achieved.

(iv) For a carbon adsorption system that regenerates the carbon bed directly onsite in the control device such as a fixed-bed adsorber, the design evaluation must consider the vent stream flow rate, relative humidity, and temperature, and must establish the design exhaust vent stream organic

compound concentration level, adsorption cycle time, number of carbon beds and their capacities, type and working capacity of activated carbon used for the carbon beds, design total regeneration stream mass or volumetric flow over the period of each complete carbon bed regeneration cycle, design carbon bed temperature after regeneration, design carbon bed regeneration time, and design service life of carbon. For vacuum desorption, the pressure drop must be included.

(v) For a carbon adsorption system that does not regenerate the carbon bed directly onsite in the control device such as a carbon canister, the design evaluation must consider the vent stream mass or volumetric flow rate, relative humidity, and temperature, and must establish the design exhaust vent stream organic compound concentration level, capacity of the carbon bed, type and working capacity of activated carbon used for the carbon bed. and design carbon replacement interval based on the total carbon working capacity of the control device and source operating schedule.

(vi) For a scrubber, the design evaluation must consider the vent stream composition, constituent concentrations, liquid-to-vapor ratio, scrubbing liquid flow rate and concentration, temperature, and the reaction kinetics of the constituents with the scrubbing liquid. The design evaluation must establish the design evaluation must establish the design exhaust vent stream organic compound concentration level and must include the additional information in paragraphs (a)(1)(vi)(A) and (B) of this section for trays and a packed column scrubber.

(A) Type and total number of theoretical and actual trays;

(B) Type and total surface area of packing for entire column, and for individual packed sections if column contains more than one packed section.

(vii) For fabric filters, the design evaluation must include the pressure drop through the device and the net gas-to-cloth ratio (i.e., cubic feet of gas per square feet of cloth).

(2) Calculation of TOC or total organic HAP concentration. The TOC concentration or total organic HAP concentration is the sum of the concentrations of the individual components. If compliance is being determined based on TOC, the owner or operator shall compute TOC for each run using Equation 6 of this subpart. If compliance is being determined based on total organic HAP, the owner or operator shall compute total organic HAP using Equation 6 of this subpart, except that only organic HAP compounds shall be summed; when determining compliance with the wastewater provisions of §63.1362(d), the organic HAP compounds shall compounds sist of the organic HAP compounds in Table 9 of subpart G of this part.

$$CG_{T} = \frac{1}{m} \sum_{j=1}^{m} \left(\sum_{i=1}^{n} CGS_{i,j} \right) \qquad (Eq. 6)$$

Where:

- CG_T = total concentration of TOC or organic HAP in vented gas stream, average of samples, dry basis, ppmv
- CGS_{i, j} = concentration of sample components in vented gas stream for sample j, dry basis, ppmv

n = number of compounds in the sample

m = number of samples in the sample run.

(3) Initial compliance using flares. When a flare is used to comply with the standards, the owner or operator shall comply with the provisions in §63.11(b) of subpart A of this part.

(i) The initial compliance determination shall consist of a visible emissions determination using Method 22 of 40 CFR part 60, appendix A, as described in §63.11(b)(4) of subpart A of this part, and a determination of net heating value of gas being combusted and exit velocity to comply with the requirements of §63.11(b)(6) through (8) of subpart A of this part. The net heating value and exit velocity shall be based on the results of performance testing under the conditions described in paragraphs (b)(10) and (11) of this section.

(ii) An owner or operator is not required to conduct a performance test to determine percent emission reduction or outlet organic HAP or TOC concentration when a flare is used.

(4) Exemptions from compliance demonstrations. An owner or operator using any control device specified in paragraphs (a)(4)(i) through (ii) of this section is exempt from the initial compliance provisions in paragraphs (c), (d), and (e) of this section.

§63.1365

(i) A boiler or process heater with a design heat input capacity of 44 megawatts or greater.

(ii) A boiler or process heater into which the emission stream is introduced with the primary fuel.

(5) Initial compliance with alternative standard. Initial compliance with the alternative standards in §63.1362(b)(6) and (c)(4) for combustion devices is demonstrated when the outlet TOC concentration is 20 ppmy or less, and the outlet HCl and chlorine concentration is 20 ppmv or less. Initial compliance with the alternative standards in §63.1362(b)(6) and (c)(4) for noncombustion devices is demonstrated when the outlet TOC concentration is 50 ppmv or less, and the outlet HCl and chlorine concentration is 50 ppmv or less. To demonstrate initial compliance, the owner or operator shall be in compliance with the monitoring provisions in §63.1366(b)(5) on the initial compliance date. The owner or operator shall use Method 18 to determine the predominant organic HAP in the emission stream if the TOC monitor is calibrated on the predominant HAP.

(6) Initial compliance with the 20 ppmv outlet limit. Initial compliance with the 20 ppmv TOC or total organic HAP concentration is demonstrated when the outlet TOC or total organic HAP concentration is 20 ppmv or less. Initial compliance with the 20 ppmv HCl and chlorine concentration is demonstrated when the outlet HCl and chlorine concentration is 20 ppmv or less. To demonstrate initial compliance, the operator shall use applicable test methods described in paragraphs (b)(1) through (9) of this section, and test under conditions described in paragraph (b)(10) or (11) of this section, as applicable. The owner or operator shall comply with provisions the monitoring in §63.1366(b)(1) through (5) on the initial compliance date.

(7) Outlet concentration correction for supplemental gases. If supplemental gases are added to a vent stream for which compliance with an outlet concentration standard in 63.1362 or 63.1363 will be demonstrated, the owner or operator must correct the outlet concentration as specified in paragraphs (a)(7)(i) and (ii) of this section.

40 CFR Ch. I (7–1–23 Edition)

(i) Combustion device. Except as specified in $\S63.1366(b)(5)(ii)(A)$, if the vent stream is controlled with a combustion device, the owner or operator must comply with the provisions in paragraphs (a)(7)(i)(A) through (C) of this section.

(A) To comply with a TOC or total organic HAP outlet concentration standard in $\S63.1362(b)(2)(iv)(A)$, (b)(4)(ii)(A), (b)(6), (c)(2)(iv)(B), (c)(4), (d)(13), or $\S63.172$, the actual TOC outlet concentration must be corrected to 3 percent oxygen.

(B) If the inlet stream to the combustion device contains any HCl, chlorine, or halogenated compounds, and the owner or operator elects to comply with a total HCl and chlorine outlet concentration standard in $\S63.1362(b)(3)(ii)$, (b)(5)(ii), (b)(5)(ii), (b)(6), or (c)(4), the actual total HCl and chlorine outlet concentration must be corrected to 3 percent oxygen.

(C) The integrated sampling and analysis procedures of Method 3B of 40 CFR part 60, appendix A, shall be used to determine the actual oxygen concentration ((O_{2d})). The samples shall be taken during the same time that the TOC, total organic HAP, and total HCl and chlorine samples are taken. The concentration corrected to 3 percent oxygen (C_d) shall be computed using Equation 7 of this subpart:

$$C_{c} = C_{m} \left(\frac{17.9}{20.9 - \%O_{2d}} \right)$$
 (Eq. 7)

Where:

- C_c = concentration of TOC, total organic HAP, or total HCl and chlorine corrected to 3 percent oxygen, dry basis, ppmv
- C_m = total concentration of TOC, total organic HAP, or total HCl and chlorine in the vented gas stream, average of samples, dry basis, ppmv
- O_{2d} = concentration of oxygen measured in vented gas stream, dry basis, percent by volume.

(ii) Noncombustion devices. If a control device other than a combustion device, and not in series with a combustion device, is used to comply with a TOC, total organic HAP, or total HCl and chlorine outlet concentration standard, the owner or operator must correct the actual concentration for supplemental gases using Equation 8 of this subpart.

$$C_a = C_m \left(\frac{V_s + V_a}{V_a} \right) \quad (Eq. 8)$$

Where:

- $\begin{array}{l} C_m = actual \mbox{ TOC}, \mbox{ total organic HAP}, \mbox{ or total} \\ \mbox{ HCl and chlorine concentration measured} \\ at \mbox{ control device outlet, dry basis, ppmv} \end{array}$
- V_a = total volumetric flow rate of affected streams vented to the control device
- $V_{\rm s}$ = total volumetric flow rate of supplemental gases.

(b) Test methods and conditions. When testing is conducted to measure emissions from an affected source, the test methods specified in paragraphs (b)(1) through (9) of this section shall be used. Compliance and performance tests shall be performed under such conditions as the Administrator specifies to the owner or operator based on representative performance of the affected source for the period being tested and as specified in paragraphs (b)(10)and (11) of this section. Representative conditions exclude periods of startup and shutdown unless specified by the Administrator or an applicable subpart. The owner or operator may not conduct performance tests during periods of malfunction. The owner or operator must record the process information that is necessary to document operating conditions during the test and include in such record an explanation to support that such conditions represent normal operation. Upon request, the owner or operator shall make available to the Administrator such records as may be necessary to determine the conditions of performance tests.

(1) Method 1 or 1A of appendix A of 40 CFR part 60 shall be used for sample and velocity traverses.

(2) Method 2, 2A, 2C, or 2D of appendix A of 40 CFR part 60 shall be used for velocity and volumetric flow rates.

(3) Method 3 of appendix A of 40 CFR part 60 shall be used for gas analysis.

(4) Method 4 of appendix A of 40 CFR part 60 shall be used for stack gas moisture.

(5) Concentration measurements shall be adjusted to negate the dilution effects of introducing nonaffected gaseous streams into the vent streams prior to control or measurement. The following methods are specified for concentration measurements of organic compounds:

(i) Method 18 of appendix A of 40 CFR part 60 may be used to determine HAP concentration in any control device efficiency determination.

(ii) Method 25 of appendix A of 40 CFR part 60 may be used to determine total gaseous nonmethane organic concentration for control efficiency determinations in combustion devices.

(iii) Method 25A of appendix A of 40 CFR part 60 may be used to determine the HAP or TOC concentration for control device efficiency determinations under the conditions specified in Method 25 of appendix A of 40 CFR part 60 for direct measurement of an effluent with a flame ionization detector, or in demonstrating compliance with the 20 ppmv TOC outlet standard. If Method 25A of appendix A of 40 CFR part 60 is used to determine the concentration of TOC for the 20 ppmv standard, the instrument shall be calibrated on methane or the predominant HAP. If calibrating on the predominant HAP, the use of Method 25A of appendix A of 40 CFR part 60 shall comply with paragraphs (b)(5)(i)(A) through (C) of this section.

(A) The organic HAP used as the calibration gas for Method 25A, 40 CFR part 60, appendix A, shall be the single organic HAP representing the largest percent by volume.

(B) The use of Method 25A, 40 CFR part 60, appendix A, is acceptable if the response from the high level calibration gas is at least 20 times the standard deviation of the response from the zero calibration gas when the instrument is zeroed on the most sensitive scale.

(C) The span value of the analyzer must be less than 100 ppmv.

(6) The methods in either paragraph (b)(6)(i) or (ii) of this section shall be used to determine the concentration, in mg/dscm, of total HCl and chlorine. Concentration measurements shall be adjusted to negate the dilution effects of introducing nonaffected gaseous streams into the vent streams prior to control or measurement.

(i) Method 26 or 26A of 40 CFR part 60, appendix A.

(ii) Any other method if the method or data have been validated according to the applicable procedures of Method 301 of appendix A of this part.

(7) Method 5 of appendix A of 40 CFR part 60 shall be used to determine the concentration of particulate matter in exhaust gas streams from bag dumps and product dryers.

(8) Wastewater analysis shall be conducted in accordance with §63.144(b)(5)(i) through (iii) or as specified in paragraph (b)(8)(i) or (ii) of this section.

(i) As an alternative to the methods specified in §63.144(b)(5)(i), an owner or operator may conduct wastewater analyses using Method 1666 or 1671 of 40 CFR part 136, appendix A, and comply with the sampling protocol requirements specified in §63.144(b)(5)(ii). The validation requirements specified in §63.144(b)(5)(iii) do not apply if an owner or operator uses Method 1666 or 1671 of 40 CFR part 136, appendix A.

(ii) As an alternative to the methods specified in §63.144(b)(5)(i), an owner or operator may use procedures specified in Method 8260 or 8270 in "Test Methods for Evaluating Solid Waste, Physical/ Chemical Methods," EPA Publication No. SW-846, Third Edition, September 1986, as amended by Update I, November 15, 1992. An owner or operator also may use any more recent, updated version of Method 8260 or 8270 approved by EPA. For the purpose of using Method 8260 or 8270 to comply with this subpart, the owner or operator must maintain a formal quality assurance program consistent with either Section 8 of Method 8260 or Method 8270. This program must include the elements related to measuring the concentrations of volatile compounds that are specified in paragraphs (b)(8)(ii)(A) through (C) of this section.

(A) Documentation of site-specific procedures to minimize the loss of compounds due to volatilization, biodegradation, reaction, or sorption during the sample collection, storage, and preparation steps.

(B) Documentation of specific quality assurance procedures followed during sampling, sample preparation, sample introduction, and analysis.

(C) Measurement of the average accuracy and precision of the specific pro-

40 CFR Ch. I (7–1–23 Edition)

cedures, including field duplicates and field spiking of the material source before or during sampling with compounds having similar chemical characteristics to the target analytes.

(9) Method 22 of appendix A of 40 CFR part 60 shall be used to determine visible emissions from flares.

(10) Testing conditions for continuous processes. Testing of process vents on equipment operating as part of a continuous process shall consist of three one-hour runs. Gas stream volumetric flow rates shall be measured every 15 minutes during each 1-hour run. Organic HAP concentration shall be determined from samples collected in an integrated sample over the duration of each one-hour test run, or from grab samples collected simultaneously with the flow rate measurements (every 15 minutes). If an integrated sample is collected for laboratory analysis, the sampling rate shall be adjusted proportionally to reflect variations in flow rate. For continuous gas streams, the emission rate used to determine compliance shall be the average emission rate of the three test runs.

(11) Testing conditions for batch processes. Testing of emissions on equipment where the flow of gaseous emissions is intermittent (batch operations) shall be conducted at absolute peakcase conditions or hypothetical peakcase conditions, as specified in paragraphs (b)(11)(i) and (ii) of this section, respectively. Gas stream volumetric flow rates shall be measured at 15minute intervals. Organic HAP, TOC, or HCl and chlorine concentration shall be determined from samples collected in an integrated sample over the duration of the test, or from grab samples collected simultaneously with the flow rate measurements (every 15 minutes). If an integrated sample is collected for laboratory analysis, the sampling rate shall be adjusted proportionally to reflect variations in flow rate. In all cases, a site-specific test plan shall be submitted to the Administrator for approval prior to testing in accordance with §63.7(c). The test plan shall include the emissions profile described in paragraph (b)(11)(iii) of this section. The term "HAP mass loading" as used in paragraphs (b)(11)(i) through (iii) of this section refers to the class of HAP.

either organic or HCl and chlorine, that the control device is intended to control.

(i) Absolute peak-case. If the most challenging conditions for the control device occur under maximum HAP load, the absolute peak-case conditions shall be characterized by the criteria presented in paragraph (b)(11)(i)(A) or (B) of this section. Otherwise, absolute peak-case conditions are defined by the conditions in paragraph (b)(11)(i)(C) of this section.

(A) The period in which the inlet to the control device will contain at least 50 percent of the maximum HAP mass load that may be vented to the control device over any 8-hour period. An emission profile as described in paragraph (b)(11)(iii)(A) of this section shall be used to identify the 8-hour period that includes the maximum projected HAP load.

(B) A 1-hour period of time in which the inlet to the control device will contain the highest hourly HAP mass loading rate that may be vented to the control device. An emission profile as described in paragraph (b)(11)(iii)(A) of this section shall be used to identify the 1-hour period of maximum HAP loading.

(C) The period of time when a condition other than the maximum HAP load is most challenging for the control device. These conditions include, but are not limited to the following:

(1) Periods when the streams contain the highest combined VOC and HAP hourly load, as described by the emission profiles in paragraph (b)(11)(iii) of this section; or

(2) Periods when the streams contain HAP constituents that approach the limits of solubility for scrubbing media; or

(3) Periods when the streams contain HAP constituents that approach the limits of adsorptivity for carbon adsorption systems.

(ii) Hypothetical peak-case. Hypothetical peak-case conditions are simulated test conditions that, at a minimum, contain the highest total average hourly HAP load of emissions that would be predicted to be vented to the control device from the emissions profile described in either paragraph (b)(11)(iii)(B) or (C) of this section.

(iii) Emissions profile. The owner or operator may choose to perform tests only during those periods of the peakcase episode(s) that the owner or operator selects to control as part of achieving the required emission reduction. Except as specified in paragraph (b)(11)(iii)(D) of this section, the owner or operator shall develop an emission profile for the vent to the control device that describes the characteristics of the vent stream at the inlet to the control device under either absolute or hypothetical peak-case conditions. The emissions profile shall be developed based on the applicable procedures described in paragraphs (b)(11)(iii)(A) through (C) of this section, as required by paragraphs (b)(11)(i) and (ii) of this section.

(A) Emissions profile by process. The emissions profile must consider all emission episodes that could contribute to the vent stack for a period of time that is sufficient to include all processes venting to the stack and shall consider production scheduling. The profile shall describe the HAP load to the device that equals the highest sum of emissions from the episodes that can vent to the control device during the period of absolute peak-case conditions specified in paragraph (b)(11)(i)(A), (B), or (C) as appropriate. Emissions per episode shall be calculated using the procedures specified in paragraph (c)(2) of this section. complying with paragraph When (b)(11)(i)(B) of this section, emissions per episode shall be divided by the duration of the episode if the duration of the episode is longer than 1 hour.

(B) Emission profile by equipment. The emission profile must consist of emissions that meet or exceed the highest hourly HAP load that would be expected under actual processing conditions. The profile shall describe equipment configurations used to generate the emission events, volatility of materials processed in the equipment, and the rationale used to identify and characterize the emission events. The emissions may be based on using a compound more volatile than compounds actually used in the process(es), and the emissions may be generated from all equipment in the process(es) or only selected equipment.

40 CFR Ch. I (7–1–23 Edition)

(C) Emission profile by capture and control device limitation. The emission profile shall consider the capture and control system limitations and the highest hourly emissions that can be routed to the control device, based on maximum flow rate and concentrations possible because of limitations on conveyance and control equipment (e.g., fans, LEL alarms and safety bypasses).

(D) *Exemptions.* The owner or operator is not required to develop an emission profile under the circumstances described in paragraph (b)(11)(iii)(D)(1)or (2) of this section.

(1) If all process vents for a process are controlled using a control device or series of control devices that reduce HAP emissions by 98 percent or more, no other emission streams are vented to the control device when it is used to control emissions from the subject process, and the performance test is conducted over the entire batch cycle.

(2) If a control device is used to comply with the outlet concentration limit for process vent emission streams from a single process (but not necessarily all of the process vents from that process), no other emission streams are vented to the control device while it is used to control emissions from the subject process, and the performance test is conducted over the entire batch cycle.

(iv) Test duration. Three runs, at a minimum of 1 hour each, are required for performance testing. When complving with a percent reduction standard, each test run may be a maximum of either 24 hours or the duration of the longest batch controlled by the control device, whichever is shorter, and each run must include the same absolute or hypothetical peak-case conditions, as defined in paragraph (b)(11)(i) or (ii) of this section. When complying with an outlet concentration limit, each run must include the same absolute or hypothetical peak-case conditions, as defined in paragraph (b)(11)(i) or (ii) of this section, and the duration of each run may not exceed the duration of the applicable peak-case condition.

(c) Initial compliance with process vent provisions. The owner or operator of an affected source shall demonstrate compliance with the process vent standards in §63.1362(b) using the procedures described in paragraphs (c)(1) through (3) of this section.

(1) Compliance with the process vent standards in $\S63.1362(b)$ shall be demonstrated in accordance with the provisions specified in paragraphs (c)(1)(i) through (viii) of this section.

(i) Initial compliance with the emission limit cutoffs in $\S63.1362(b)(2)(i)$ and (b)(4)(i) is demonstrated when the uncontrolled organic HAP emissions from the sum of all process vents within a process are less than or equal to 0.15 Mg/yr. Uncontrolled HAP emissions shall be determined using the procedures described in paragraph (c)(2) of this section.

(ii) Initial compliance with the emission limit cutoffs in §63.1362(b)(3)(i) and (b)(5)(i) is demonstrated when the uncontrolled HCl and Cl₂ emissions from the sum of all process vents within a process are less than or equal to 6.8 Mg/ yr. Initial compliance with the emission limit cutoffs in §63.1362(b)(5)(ii) and (iii) is demonstrated when the uncontrolled HCl and Cl₂ emissions are greater than or equal to 6.8 Mg/yr or greater than or equal to 191 Mg/yr, respectively. Uncontrolled emissions shall be determined using the procedures described in paragraph (c)(2) of this section.

(iii) Initial compliance with the organic HAP percent reduction requirements specified in §63.1362(b)(2)(ii), (iii), and (b)(4)(ii) is demonstrated by determining controlled HAP emissions using the procedures described in paragraph (c)(3) of this section, determining uncontrolled HAP emissions using the procedures described in paragraph (c)(2) of this section, and calculating the applicable percent reduction. As an alternative, if the conditions specified in paragraph (b)(11)(iii)(D)(1) of this section are met, initial compliance may be demonstrated by showing the control device reduces emissions by 98 percent by weight or greater using the procedures specified in paragraph (c)(3) of this section.

(iv) Initial compliance with the HCl and Cl_2 percent reduction requirements specified in §63.1362(b)(3)(ii), (b)(5)(ii), and (b)(5)(iii) is demonstrated by determining controlled emissions of HCl and Cl_2 using the procedures described in

paragraph (c)(3) of this section, determining uncontrolled emissions of HCl and Cl_2 using the procedures described in paragraph (c)(2) of this section, and calculating the applicable percent reduction.

 $\begin{array}{c} (v) \mbox{ Initial compliance with the outlet} \\ \mbox{concentration} & \mbox{limits} & \mbox{in} \end{array}$

§63.1362(b)(2)(iv)(A), (b)(3)(ii), (b)(4)(ii)(A), (b)(5)(ii) and (iii) is demonstrated when the outlet TOC or total organic HAP concentration is 20 ppmv or less and the outlet HCl and chlorine concentration is 20 ppmv or less. The owner or operator shall demonstrate compliance by fulfilling the requirements in paragraph (a)(6) of this section. If an owner or operator elects to develop an emissions profile by process as described in paragraph (b)(11)(iii)(A) of this section, uncontrolled emissions shall be determined using the procedures in paragraph (c)(2) of this section.

(vi) Initial compliance with the alternative standard in 63.1362(b)(6) is demonstrated by fulfilling the requirements in paragraph (a)(5) of this section.

(vii) Initial compliance when using a flare is demonstrated by fulfilling the requirements in paragraph (a)(3) of this section.

(viii) No initial compliance demonstration is required for control devices specified in §63.1362(1).

(2) Uncontrolled emissions. The owner or operator referred to from paragraphs (c)(1)(i) through (v) of this section shall calculate uncontrolled emissions according to the procedures described in paragraph (c)(2)(i) or (ii) of this section, as appropriate.

(i) Emission estimation procedures. The owner or operator shall determine uncontrolled HAP emissions using emission measurements and/or calculations for each batch emission episode according to the engineering evaluation methodology in paragraphs (c)(2)(i)(A) through (H) of this section.

(A) Individual HAP partial pressures in multicomponent systems shall be determined in accordance with the methods specified in paragraphs (c)(2)(i)(A)(I) through (3) of this section. Chemical property data may be obtained from standard references.

(1) If the components are miscible in one another, use Raoult's law to calculate the partial pressures;

(2) If the solution is a dilute aqueous mixture, use Henry's law constants to calculate partial pressures;

(3) If Raoult's law or Henry's law are not appropriate or available, use any of the methods specified in paragraphs (c)(2)(i)(A) (3)(*i*) through (*iii*) of this section.

(*i*) Use experimentally obtained activity coefficients;

(*ii*) Use models such as the group-contribution models to predict activity coefficients;

(*iii*) Assume the components of the system behave independently and use the summation of all vapor pressures from the HAP as the total HAP partial pressure;

(B) *Charging or filling*. Emissions from vapor displacement due to transfer of material to a vessel shall be calculated using Equation 9 of this subpart:

$$E = \frac{(V)}{(R)(T)} \times \sum_{i=1}^{n} (P_i) (MW_i) \qquad (Eq. 9)$$

Where:

E = mass of HAP emitted

 P_i = partial pressure of the individual HAP

V = volume of gas displaced from the vessel R = ideal gas law constant

T = temperature of the vessel vapor space; absolute

 MW_{i} = molecular weight of the individual ${\rm HAP}$

(C) *Purging*. Emissions from purging shall be calculated using Equation 10 of this subpart, except that for purge flow rates greater than 100 scfm, the mole fraction of HAP will be assumed to be 25 percent of the saturated value.

$$E = \sum_{i=1}^{n} P_i M W_i \times \left(\frac{(V)(t)}{(R)(T)}\right) \times \frac{P_T}{P_T - \sum_{i=1}^{m} (P_j)}$$
(Eq. 10)

§63.1365

Where:

- E = mass of HAP emitted
- V = purge flow rate at the temperature and pressure of the vessel vapor space
- R = ideal gas law constant
- T = temperature of the vessel vapor space; absolute
- P_i = partial pressure of the individual HAP
- P_i = partial pressure of individual condensable compounds (including HAP)
- P_{T} = pressure of the vessel vapor space MW_i = molecular weight of the individual HAP
- t = time of purge
- n = number of HAP compounds in the emission stream
- m = number of condensable compounds (including HAP) in the emission stream.

(D) Heating. Emissions caused by heating the contents of a vessel to a

40 CFR Ch. I (7-1-23 Edition)

temperature less than the boiling point shall be calculated using the procedures in either paragraph (c)(2)(i)(D)(1), (2), or (4) of this section, as appropriate. If the contents of a vessel are heated to the boiling point, emissions while boiling are assumed to be zero if the owner or operator is complying with the provisions in paragraph (d)(2)(i)(C)(3) of this section.

(1) If the final temperature to which the vessel contents are heated is lower than 50 K below the boiling point of the HAP in the vessel, then emissions shall be calculated using Equations 11 through 14 of this subpart.

(i) The mass of HAP emitted per episode shall be calculated using Equation 11 of this subpart:

$$E = \frac{\frac{\sum_{i=1}^{n} (P_i)_{T1}}{Pa_1} + \frac{\sum_{i=1}^{n} (P_i)_{T2}}{Pa_2}}{2} \times \Delta \eta \times MW_{HAP}$$
(Eq. 11)

Where:

- E = mass of HAP vapor displaced from the vessel being heated
- $(P_i)_{Tn}$ = partial pressure of each HAP in the vessel headspace at initial (n = 1) and final (n = 2) temperatures
- Pa_1 = initial noncondensable gas pressure in the vessel, as calculated using Equation 13 of this subpart
- Pa_2 = final noncondensable gas pressure in the vessel, as calculated using Equation 13 of this subpart
- ΔH = number of moles of noncondensable gas displaced, as calculated using Equation 12 of this subpart
- MW_{HAP} = The average molecular weight of HAP present in the vessel, as calculated using Equation 14 of this subpart:
- n = number of HAP compounds in the displaced vapor

(ii) The moles of noncondensable gas displaced shall be calculated using Equation 12 of this subpart:

$$\Delta \eta = \frac{V}{R} \left[\left(\frac{Pa_1}{T_1} \right) - \left(\frac{Pa_2}{T_2} \right) \right] \qquad (Eq. 12)$$

Where:

 ΔH = number of moles of noncondensable gas displaced

V = volume of free space in the vessel

- R = ideal gas law constant
- $Pa_1 = initial noncondensable gas pressure in$ the vessel, as calculated using Equation 13 of this subpart
- Pa_2 = final noncondensable gas pressure in the vessel, as calculated using Equation 13 of this subpart
- = initial temperature of vessel contents, T_1 absolute
- T_2 = final temperature of vessel contents, absolute

(iii) The initial and final pressure of the noncondensable gas in the vessel shall be calculated according to Equation 13 of this subpart:

$$Pa_n = Pa_{atm} - \sum_{j=1}^{m} \left(P_j \right)_{Tn} \qquad (Eq. 13)$$

Where:

 $Pa_n = partial pressure of noncondensable gas$ in the vessel headspace at initial (n = 1)and final (n = 2) temperatures

 $P_{atm} = atmospheric pressure$

 $(P_i)_{Tn}$ = partial pressure of each condensable volatile organic compound (including HAP) in the vessel headspace at the initial temperature (n = 1) and final (n = 2)temperature

§63.1365

(*iv*) The average molecular weight of HAP in the displaced gas shall be cal-

culated using Equation 14 of this subpart:

$$MW_{HAP} = \sum_{i=1}^{n} \frac{\left((P_i)_{T_1} + (P_i)_{T_2} \right) MW_i}{\sum_{i=1}^{n} \left((P_i)_{T_1} + (P_i)_{T_2} \right)}$$
(Eq. 14)

Where:

- MW_{HAP} = average molecular weight of HAP in the displaced gas
- $(P_i)_{Tn}$ = partial pressure of each HAP in the vessel headspace at the initial (T_1) and final (T_2) temperatures

 MW_i = molecular weight of each HAP

n = number of HAP compounds in the emission stream

(2) If the vessel contents are heated to a temperature greater than 50 K below the boiling point, then emissions from the heating of a vessel shall be calculated as the sum of the emissions calculated in accordance with paragraphs (c)(2)(i)(D)(2)(i) and (ii) of this section.

(i) For the interval from the initial temperature to the temperature 50 K below the boiling point, emissions shall be calculated using Equation 11 of this subpart, where T_2 is the temperature 50 K below the boiling point.

(*ii*) For the interval from the temperature 50 K below the boiling point to the final temperature, emissions shall be calculated as the summation of emissions for each 5 K increment, where the emission for each increment shall be calculated using Equation 11 of this subpart. If the final temperature of the heatup is lower than 5 K below the boiling point, the final temperature for the last increment shall be the final temperature of the heatup, even if the last increment is less than 5 K. If the

final temperature of the heatup is higher than 5 K below the boiling point, the final temperature for the last increment shall be the temperature 5 K below the boiling point, even if the last increment is less than 5 K.

(3) While boiling, the vessel must be operated with a properly operated process condenser. An initial demonstration that a process condenser is properly operated is required for vessels that operate process condensers without secondary condensers that are air pollution control devices. The owner or operator must either measure the condenser exhaust gas temperature and show it is less than the boiling point of the substance(s) in the vessel, or perform a material balance around the vessel and condenser to show that at least 99 percent of the material vaporized while boiling is condensed. Uncontrolled emissions are assumed to be zero under these conditions. The initial demonstration shall be conducted for all appropriate operating scenarios and documented in the Notification of Compliance Status report as specified in §63.1368(f).

(4)(i) As an alternative to the procedures described in paragraphs (C)(2)(i)(D)(1) and (2) of this section, emissions caused by heating a vessel to any temperature less than the boiling point may be calculated using Equation 15 of this subpart.

$$E = MW_{HAP} \times \left(N_{avg} \times 1n \left(\frac{P_{T} - \sum_{i=1}^{m} (P_{i,1})_{\div}}{P_{T} - \sum_{i=1}^{m} (P_{i,2})_{\div}^{\div}} - (n_{HAP,2} - n_{HAP,1})_{\div}^{\div} \right) \right)$$
(Eq. 15)

§63.1365

Where:

E = mass of HAP vapor displaced from the vessel being heated

- $$\begin{split} N_{avg} &= average \; \text{gas space molar volume dur-} \\ & \text{ing the heating process, as calculated} \\ & \text{using Equation 16 of this subpart} \end{split}$$
- P_T = total pressure in the vessel
- $P_{i, 1}$ = partial pressure of the individual HAP compounds at T_1
- $P_{i,\ 2}$ = partial pressure of the individual HAP compounds at T_2
- MW_{HAP} = average molecular weight of the HAP compounds, as calculated using Equation 14 of this subpart
- $n_{HAP, 1}$ = number of moles of total HAP in the vessel headspace at T_1
- $n_{HAP, 2}$ = number of moles of total HAP in the vessel headspace at T_2
- m = number of HAP compounds in the emission stream.

(*ii*) The average gas space molar volume during the heating process is cal-

40 CFR Ch. I (7–1–23 Edition)

culated using Equation 16 of this subpart.

$$N_{avg} = \frac{VP_T}{2R} \left(\frac{1}{T_1} + \frac{1}{T_2} \right)$$
 (Eq. 16)

Where:

- N_{avg} = average gas space molar volume during the heating process
- V = volume of free space in vessel
- \mathbf{P}_{T} = total pressure in the vessel
- R = ideal gas law constant
- T_1 = initial temperature of the vessel contents, absolute
- T_2 = final temperature of the vessel contents, absolute

(*iii*) The difference in the number of moles of total HAP in the vessel headspace between the initial and final temperatures is calculated using Equation 17 of this subpart.

$$(n_{\text{HAP},2} - n_{\text{HAP},1}) = \frac{V}{(R)(T_2)} \sum_{i=1}^{n} P_{i,2} - \frac{V}{(R)(T_1)} \sum_{i=1}^{n} P_{i,1}$$
 (Eq. 17)

Where:

- $n_{HAP,\ 2}$ = number of moles of total HAP in the vessel headspace at T_2
- $n_{HAP, 1}$ = number of moles of total HAP in the vessel headspace at T_1
- V = volume of free space in vessel
- R = ideal gas law constant
- ${\rm T}_1$ = initial temperature of the vessel contents, absolute
- T_2 = final temperature of the vessel contents, absolute
- $P_{i, 1}$ = partial pressure of the individual HAP compounds at T_1
- $P_{i,\ 2}$ = partial pressure of the individual HAP compounds at T_2
- n = number of HAP compounds in the emission stream.

(E) Depressurization. Emissions from depressurization shall be calculated using the procedures in paragraphs (c)(2)(i)(E)(1) through (5) of this section. Alternatively, the owner or operator may elect to calculate emissions from depressurization using the procedures in paragraph (c)(2)(i)(E)(6) of this section.

(1) The moles of HAP vapor initially in the vessel are calculated using Equation 18 of this subpart:

$$n_{\text{HAP}} = \frac{V}{R T} \times \sum_{i=1}^{n} (P_i)$$
 (Eq. 18)

Where:

 n_{HAP} = moles of HAP vapor in the vessel

- P_i = partial pressure of each HAP in the vessel vapor space
- V = free volume in the vessel being depressurized
- R = ideal gas law constant
- T = absolute temperature in vessel
- n = number of HAP compounds in the emission stream

(2) The initial and final moles of noncondensable gas present in the vessel are calculated using Equations 19 and 20 of this subpart:

$$n_1 = \frac{VP_{nc_1}}{RT} \qquad (Eq. 19)$$
$$n_2 = \frac{VP_{nc_2}}{RT} \qquad (Eq. 20)$$

Where:

- n_1 = initial number of moles of noncondensable gas in the vessel
- n_2 = final number of moles of noncondensable gas in the vessel

- V = free volume in the vessel being depressurized
- P_{nc1} = initial partial pressure of the noncondensable gas, as calculated using Equation 21 of this subpart
- P_{nc2} = final partial pressure of the noncondensable gas, as calculated using Equation 22 of this subpart
- R = ideal gas law constant

T = temperature, absolute

(3) The initial and final partial pressures of the noncondensable gas in the vessel are determined using Equations 21 and 22 of this subpart.

$$P_{ncl} = P_1 - \sum_{j=1}^{m} (P_j *)(x_j)$$
 (Eq. 21)

$$P_{nc2} = P_2 - \sum_{j=1}^{m} (P_j *)(x_j)$$
 (Eq. 22)

Where:

- P_{nc1} = initial partial pressure of the noncondensable gas
- P_{nc2} = final partial pressure of the noncondensable gas
- P_1 = initial vessel pressure
- P_2 = final vessel pressure
- P_j^* = vapor pressure of each condensable compound (including HAP) in the emission stream
- = mole fraction of each condensable com-Xi pound (including HAP) in the liquid phase
- m = number of condensable compounds (including HAP) in the emission stream.

(4) The moles of HAP emitted during the depressurization are calculated by taking an approximation of the average ratio of moles of HAP to moles of noncondensable and multiplying by the total moles of noncondensables released during the depressurization, using Equation 23 of this subpart:

$$n_{\text{HAP,e}} = \frac{\left(\frac{n_{\text{HAP,1}}}{n_1} + \frac{n_{\text{HAP,2}}}{n_2}\right)}{2} [n_1 - n_2] \qquad (\text{Eq. 23})$$

Where:

- $n_{HAP, e}$ = moles of HAP emitted
- $n_{HAP, 1}$ = moles of HAP vapor in vessel at the initial pressure, as calculated using Equation 18 of this subpart
- $n_{HAP, 2}$ = moles of HAP vapor in vessel at the final pressure, as calculated using Equation 18 of this subpart
- n_1 = initial number of moles of noncondensable gas in the vessel, as calculated using Equation 19 of this subpart
- n_2 = final number of moles of noncondensable gas in the vessel, as calculated using Equation 19 of this subpart.

(5) Use Equation 24 of this subpart to calculate the mass of HAP emitted:

$$E = n_{HAP.e} * MW_{HAP} \qquad (Eq. 24)$$

Where: E = mass of HAP emitted

- $n_{\mathrm{HAP,\ e}}$ = moles of HAP emitted, as calculated
- using Equation 23 of this subpart MW_{HAP} = average molecular weight of the HAP as calculated using Equation 14 of this subpart

(6) As an alternative to the procedures in paragraphs (c)(2)(i)(E)(1)through (5) of this section, emissions from depressurization may be calculated using Equation 25 of this subpart:

$$E = \frac{V}{(R)(T)} \times \ln \left(\frac{P_1 - \sum_{j=1}^{m} (P_j)_{\div}^{+}}{P_2 - \sum_{j=1}^{m} (P_j)_{\div}^{+}} \times \sum_{i=1}^{n} (P_i) (MW_i) \right)$$
(Eq. 25)

§63.1365

Where:

- V = free volume in vessel being depressurized R = ideal gas law constant
- T = temperature of the vessel, absolute
- P_1 = initial pressure in the vessel
- P_2 = final pressure in the vessel
- $P_i = partial pressure of the individual HAP$
- compounds
- P_j = partial pressure of individual condensable VOC compounds (including HAP)

40 CFR Ch. I (7–1–23 Edition)

- MW_i = molecular weight of the individual HAP compounds
- n = number of HAP compounds in the emission stream
- m = number of condensable VOC compounds (including HAP) in the emission stream

(F) Vacuum systems. Calculate emissions from vacuum systems using Equation 26 of this subpart:

$$E = \frac{(MW_{HAP})(La)(t)}{MW_{nc}} \left(\frac{\sum_{i=1}^{n} P_{i}}{P_{T} - \sum_{j=1}^{m} P_{j} \frac{\dot{+}}{\dot{-}}} \right)$$
(Eq. 26)

Where:

- E = mass of HAP emitted
- $P_{\rm T}$ = absolute pressure of receiving vessel or ejector outlet conditions, if there is no receiver
- P_i = partial pressure of individual HAP at the receiver temperature or the ejector outlet conditions
- P_j = partial pressure of individual condensable compounds (including HAP) at the receiver temperature or the ejector outlet conditions
- La = total air leak rate in the system, mass/ time
- MW_{nc} = molecular weight of noncondensable gas

t = time of vacuum operation

- MW_{HAP} = average molecular weight of HAP in the emission stream, as calculated using Equation 14 of this subpart, with HAP partial pressures calculated at the temperature of the receiver or ejector outlet, as appropriate
- n = number of HAP components in the emission stream

 $\label{eq:main} \begin{array}{l} m \mbox{ = number of condensable compounds (including HAP) in the emission stream.} \end{array}$

(G) *Gas evolution*. Emissions from gas evolution shall be calculated using Equation 10 of this subpart with V calculated using Equation 27 of this subpart:

$$V = \frac{(W_g)(R)(T)}{(P_T)(MW_g)}$$
 (Eq. 27)

Where:

V = volumetric flow rate of gas evolution

 $W_{\rm g}$ = mass flow rate of gas evolution

R = ideal gas law constant

T = temperature at the exit, absolute

 P_T = vessel pressure

 MW_g = molecular weight of the evolved gas

(H) *Air drying*. Use Equation 28 of this subpart to calculate emissions from air drying:

$$E = B \times \left(\frac{PS_1}{100 - PS_1} - \frac{PS_2}{100 - PS_2}\right)$$
 (Eq. 28)

Where:

E = mass of HAP emitted

B = mass of dry solids

- PS_1 = HAP in material entering dryer, weight percent
- $PS_2 = HAP$ in material exiting dryer, weight percent.

(ii) Engineering assessments. The owner or operator shall conduct an engineering assessment to determine uncontrolled HAP emissions for each emission episode that is not due to vapor displacement, purging, heating, depressurization, vacuum systems, gas evolution, or air drying. For a given

emission episode caused by any of these seven types of activities, the owner or operator also may request approval to determine uncontrolled HAP emissions based on an engineering assessment. Except as specified in paragraph (c)(2)(ii)(A) of this section, all data, assumptions, and procedures used in the engineering assessment shall be documented in the Precompliance plan in accordance with §63.1367(b). An engineering assessment includes, but is not limited to, the information and procedures described in paragraphs (c)(2)(ii)(A) through (D) of this section.

(A) Test results, provided the tests are representative of current operating practices at the process unit. For process vents without variable emission stream characteristics, an engineering assessment based on the results of a previous test may be submitted in the Notification of Compliance Status report instead of the Precompliance plan. Results from a previous test of process vents with variable emission stream characteristics will be acceptable in place of values estimated using the procedures specified in paragraph (c)(2)(i) of this section if the test data show a greater than 20 percent discrepancy between the test value and the estimated value, and the results of the engineering assessment shall be included in the Notification of Compliance Status report. For other process vents with variable emission stream characteristics, engineering assessments based on the results of a previous test must be submitted in the Precompliance plan. For engineering assessments based on new tests, the owner or operator must comply with the test notification requirements in §63.1368(m), and the results of the engineering assessment may be submitted in the Notification of Compliance Status report rather than the Precompliance plan.

(B) Bench-scale or pilot-scale test data representative of the process under representative operating conditions.

(C) Maximum flow rate, HAP emission rate, concentration, or other relevant parameter specified or implied within a permit limit applicable to the process vent.

(D) Design analysis based on accepted chemical engineering principles, measurable process parameters, or physical or chemical laws or properties. Examples of analytical methods include, but are not limited to:

(1) Use of material balances based on process stoichiometry to estimate maximum organic HAP concentrations;

(2) Estimation of maximum flow rate based on physical equipment design such as pump or blower capacities; and

(3) Estimation of HAP concentrations based on saturation conditions.

(3) Controlled emissions. Except for condensers, the owner or operator shall determine controlled emissions using the procedures in either paragraph (c)(3)(i) or (ii) of this section, as applicable. For condensers, controlled emissions shall be calculated using the emission estimation equations described in paragraph (c)(3)(iii) of this section. The owner or operator is not required to calculate controlled emissions from devices described in paragraph (a)(4) of this section or from flares for which compliance is demonstrated in accordance with paragraph (a)(3) of this section. If the owner or operator is complying with an outlet concentration standard and the control device uses supplemental gases, the outlet concentrations shall be corrected in accordance with the procedures described in paragraph (a)(7) of this section.

(i) Small control devices, except condensers. Controlled emissions for each process vent that is controlled using a small control device, except for a condenser, shall be determined by using the design evaluation described in paragraph (c)(3)(i)(A) of this section, or by conducting a performance test in accordance with paragraph (c)(3)(i) of this section.

(A) Design evaluation. The design evaluation shall include documentation demonstrating that the control device being used achieves the required control efficiency under absolute or hypothetical peak-case conditions, as determined from the emission profile described in paragraph (b)(11)(iii) of this section. The control efficiency determined from this design evaluation shall be applied to uncontrolled emissions to estimate controlled emissions. The documentation must be conducted in accordance with the provisions in paragraph (a)(1) of this section. The design evaluation shall also include the value(s) and basis for the parameter(s) monitored under \S 63.1366.

(B) Whenever a small control device becomes a large control device, the owner or operator must comply with the provisions in paragraph (c)(3)(i) of this section and submit the test report in the next Periodic report.

(ii) Large control devices, except condensers. Controlled emissions for each process vent that is controlled using a large control device, except for a condenser, shall be determined by applying the control efficiency of the large control device to the estimated uncontrolled emissions. The control efficiency shall be determined by conducting a performance test on the control device as described in paragraphs (c)(3)(ii)(A) through (C) of this section, or by using the results of a previous performance test as described in paragraph (c)(3)(ii)(D) of this section. If the control device is intended to control only HCl and chlorine, the owner or operator may assume the control efficiency of organic HAP is 0 percent. If the control device is intended to control only organic HAP, the owner or operator may assume the control efficiency for HCl and chlorine is 0 percent.

(A) Performance test measurements shall be conducted at both the inlet and outlet of the control device for TOC, total organic HAP, and total HCl and chlorine, as applicable, using the test methods and procedures described in paragraph (b) of this section. Concentrations shall be calculated from the data obtained through emission testing according to the procedures in paragraph (a)(2) of this section.

(B) Performance testing shall be conducted under absolute or hypothetical peak-case conditions, as defined in paragraphs (b)(11)(i) and (ii) of this section.

(C) The owner or operator may elect to conduct more than one performance test on the control device for the purpose of establishing more than one operating condition at which the control 40 CFR Ch. I (7–1–23 Edition)

device achieves the required control efficiency.

(D) The owner or operator is not required to conduct a performance test for any control device for which a previous performance test was conducted, provided the test was conducted using the same procedures specified in paragraphs (b)(1) through (11) of this section over conditions typical of the absolute or hypothetical peak-case, as defined in paragraphs (b)(11)(i) and (ii) of this section. The results of the previous performance test shall be used to demonstrate compliance.

(iii) Condensers. The owner or operator using a condenser as a control device shall determine controlled emissions for each batch emission episode according to the engineering methodin paragraphs ology (c)(3)(iii)(A)through (G) of this section. The owner or operator must establish the maximum outlet gas temperature and calculate the controlled emissions using this temperature in the applicable equation. Individual HAP partial pressures shall be calculated as specified in paragraph (c)(2)(i) of this section.

(A) Emissions from vapor displacement due to transfer of material to a vessel shall be calculated using Equation 9 of this subpart with T set equal to the temperature of the receiver and the HAP partial pressures determined at the temperature of the receiver.

(B) Emissions from purging shall be calculated using Equation 10 of this subpart with T set equal to the temperature of the receiver and the HAP partial pressures determined at the temperature of the receiver.

(C) Emissions from heating shall be calculated using Equation 29 of this subpart. In Equation 29 of this subpart, $\Delta\eta$ is equal to the number of moles of noncondensable displaced from the vessel, as calculated using Equation 12 of this subpart. In Equation 29 of this subpart, the HAP average molecular weight shall be calculated using Equation 14 with the HAP partial pressures determined at the temperature of the receiver.

§63.1365

$$E = \Delta \eta \times \frac{\sum_{i=1}^{m} P_i}{P_T - \sum_{j=1}^{m} P_j} \times MW_{HAP} \qquad (Eq. 29)$$

n

Where:

- E = mass of HAP emitted
- $\Delta \eta$ = moles of noncondensable gas displaced
- P_{T} = pressure in the receiver P_{i} = partial pressure of the individual HAP
- at the receiver temperature P_j = partial pressure of the individual condensable VOC (including HAP) at the receiver temperature

n = number of HAP compounds in the emission stream

- $MW_{\rm HAP}$ = the average molecular weight of HAP in vapor exiting the receiver, as cal-
- culated using Equation 14 of this subpart m = number of condensable VOC (including HAP) in the emission stream
- (D)(1) Emissions from depressurization shall be calculated using Equation 30 of this subpart.

$$E = (V_{nc1} - V_{nc2}) \times \frac{\sum_{i=1}^{n} (P_i)}{P_T - \sum_{i=1}^{m} (P_j)} \times \frac{P_T}{RT} \times MW_{HAP}$$
(Eq. 30)

n

Where:

- E = mass of HAP vapor emitted
- V_{nc1} = initial volume of noncondensable in the vessel, corrected to the final pressure, as calculated using Equation 31 of this subpart
- $V_{\rm nc2}$ = final volume of noncondensable in the vessel, as calculated using Equation 32 of this subpart
- \mathbf{P}_i = partial pressure of each individual HAP at the receiver temperature
- P_j = partial pressure of each condensable VOC (including HAP) at the receiver temperature
- P_T = receiver pressure
- T = temperature of the receiver, absolute
- R = ideal gas law constant
- $\label{eq:n} \begin{array}{l} n \ = \ number \ of \ HAP \ compounds \ in \ the \ emission \ stream \end{array}$
- m = number of condensable VOC (including HAP) in the emission stream

(2) The initial and final volumes of noncondensable gas present in the vessel, adjusted to the pressure of the receiver, are calculated using Equations 31 and 32 of this subpart.

$$V_{ncl} = \frac{VP_{nc_1}}{P_T} \qquad (Eq. 31)$$
$$V_{nc2} = \frac{VP_{nc_2}}{P_T} \qquad (Eq. 32)$$

Where:

- V_{nc1} = initial volume of noncondensable gas in the vessel
- $V_{\rm nc2}$ = final volume of noncondensable gas in the vessel
- V = free volume in the vessel being depressurized
- P_{nc1} = initial partial pressure of the noncondensable gas, as calculated using Equation 33 of this subpart
- $P_{nc2} \mbox{ = final partial pressure of the non-condensable gas, as calculated using Equation 34 of this subpart }$
- P_T = pressure of the receiver

(3) Initial and final partial pressures of the noncondensable gas in the vessel are determined using Equations 33 and 34 of this subpart. §63.1365

$$P_{nc1} = P_1 - \sum_{j=1}^{m} P_j$$
 (Eq. 33)

$$P_{nc2} = P_2 - \sum_{j=1}^{m} P_j$$
 (Eq. 34)

Where:

 P_{nc1} = initial partial pressure of the non- condensable gas in the vessel

 P_{nc2} = final partial pressure of the noncondensable gas in the vessel

 $P_1 = initial vessel pressure$

 P_2 = final vessel pressure

 $\begin{array}{l} P_{j} \ = \ partial \ pressure \ of \ each \ condensable \\ VOC \ (including \ HAP) \ in \ the \ vessel \end{array}$

m = number of condensable VOC (including HAP) in the emission stream $% \left({{{\rm{AP}}} \right)_{\rm{AP}}} \right)$

(E) Emissions from vacuum systems shall be calculated using Equation 26 of this subpart.

(F) Emissions from gas evolution shall be calculated using Equation 8 with V calculated using Equation 27 of this subpart, T set equal to the receiver temperature, and the HAP partial pressures determined at the receiver temperature. The term for time, t, in Equation 10 of this subpart is not needed for the purposes of this calculation.

(G) Emissions from air drying shall be calculated using Equation 9 of this subpart with V equal to the air flow rate and P_i determined at the receiver temperature.

(d) Initial compliance with storage vessel provisions. The owner or operator of an existing or new affected source shall demonstrate initial compliance with storage vessel standards the in §63.1362(c)(2) through (4) by fulfilling the requirements in either paragraph (d)(1), (2), (3), (4), (5), or (6) of this section, as applicable. The owner or operator shall demonstrate initial compliance with the planned routine maintenance provision in §63.1362(c)(5) by fulfilling the requirements in paragraph (d)(7) of this section.

(1) Percent reduction requirement for control devices. If the owner or operator equips a Group 1 storage vessel with a closed vent system and control device, the owner or operator shall demonstrate initial compliance with the percent reduction requirement of $\S63.1362(c)(2)(iv)(A)$ or (c)(3) either by

40 CFR Ch. I (7–1–23 Edition)

calculating the efficiency of the control device using performance test data as specified in paragraph (d)(1)(i) of this section, or by preparing a design evaluation as specified in paragraph (d)(1)(i) of this section.

(i) Performance test option. If the owner or operator elects to demonstrate initial compliance based on performance test data, the efficiency of the control device shall be calculated as specified in paragraphs (d)(1)(i)(A) through (D) of this section.

(A) At the reasonably expected maximum filling rate, Equations 35 and 36 of this subpart shall be used to calculate the mass rate of total organic HAP or TOC at the inlet and outlet of the control device.

$$E_{i} = K_{2} \left(\sum_{j=1}^{n} C_{ij} M_{ij} \stackrel{}{\xrightarrow{}} Q_{i} \quad (Eq. 35) \right)$$
$$E_{o} = K_{2} \left(\sum_{j=1}^{n} C_{oj} M_{oj} \stackrel{}{\xrightarrow{}} Q_{o} \quad (Eq. 36) \right)$$

Where:

- $\begin{array}{l} C_{ij},\ C_{oj} = \text{concentration of sample component} \\ \text{j of the gas stream at the inlet and outlet} \\ \text{of the control device, respectively, dry} \\ \text{basis, ppmv} \end{array}$
- E_i, E_o = mass rate of total organic HAP or TOC at the inlet and outlet of the control device, respectively, dry basis, kg/hr
- M_{ij} , M_{oj} = molecular weight of sample component j of the gas stream at the inlet and outlet of the control device, respectively, g/gmole
- $Q_i, \; Q_o$ = flow rate of gas stream at the inlet and outlet of the control device, respectively, dscmm
- $\begin{array}{ll} K_2 \ = \ constant, \ 2.494 \ \times \ 10^{-6} \ (parts \ per \ million)^{-1} \ (gram-mole \ per \ standard \ cubic \ meter) \ (kilogram/gram) \ (minute/hour), \\ where \ standard \ temperature \ is \ 20 \ ^{\circ}C. \end{array}$

(B) The percent reduction in total organic HAP or TOC shall be calculated using Equation 37 of this subpart:

$$R = \frac{E_i - E_o}{E_i} (100)$$
 (Eq. 37)

Where:

- R = control efficiency of control device, percent
- $$\begin{split} E_i = mass \mbox{ rate of total organic HAP or TOC} \\ \mbox{ at the inlet to the control device as calculated under paragraph } (d)(l)(i)(A) \mbox{ of } \end{split}$$

this section, kilograms organic HAP per hour

 E_o = mass rate of total organic HAP or TOC at the outlet of the control device, as calculated under paragraph (d)(1)(i)(A) of this section, kilograms organic HAP per hour.

(C) A performance test is not required to be conducted if the control device used to comply with 63.1362(c)(storage tank provisions) is also used to comply with 63.1362(b) (process vent provisions), provided compliance with 63.1362(b) is demonstrated in accordance with paragraph (c) of this section and the demonstrated percent reduction is equal to or greater than 95 percent.

(D) A performance test is not required for any control device for which a previous test was conducted, provided the test was conducted using the same procedures specified in paragraph (b) of this section.

(ii) Design evaluation option. If the owner or operator elects to demonstrate initial compliance by conducting a design evaluation, the owner or operator shall prepare documentation in accordance with the design evaluation provisions in paragraph (a)(1) of this section, as applicable. The design evaluation shall demonstrate that the control device being used achieves the required control efficiency when the storage vessel is filled at the reasonably expected maximum filling rate.

(2) Outlet concentration requirement for control devices. If the owner or operator equips a Group 1 storage vessel with a closed vent system and control device, the owner or operator shall demonstrate initial compliance with the outlet concentration requirements of $\S 63.1362(c)(2)(iv)(B)$ or (c)(3) by fulfilling the requirements of paragraph (a)(6) of this section.

(3) Floating roof. If the owner or operator equips a Group 1 storage vessel with a floating roof to comply with the provisions in $\S63.1362(c)(2)$ or (c)(3), the owner or operator shall demonstrate initial compliance by complying with the procedures described in paragraphs (d)(3)(i) and (ii) of this section.

(i) Comply with 63.119(b), (c), or (d) of subpart G of this part, as applicable, with the differences specified in 63.1362(d)(2)(i) through (iii).

(ii) Comply with the procedures described in $\S63.120(a)$, (b), or (c), as applicable, with the differences specified in paragraphs (d)(3)(ii)(A) through (C) of this section.

(A) When the term "storage vessel" is used in §63.120, the definition of the term "storage vessel" in §63.1361 shall apply for the purposes of this subpart.

(B) When the phrase "the compliance date specified in §63.100 of subpart F of this part" is referred to in §63.120, the phrase "the compliance date specified in §63.1364" shall apply for the purposes of this subpart.

(C) When the phrase "the maximum true vapor pressure of the total organic HAP in the stored liquid falls below the values defining Group 1 storage vessels specified in Table 5 or Table 6 of this subpart" referred is to in §63.120(b)(1)(iv), the phrase "the maximum true vapor pressure of the total organic HAP in the stored liquid falls below the values defining Group 1 storage vessels specified in §63.1361" shall apply for the purposes of this subpart.

(4) Flares. If the owner or operator controls the emissions from a Group 1 storage vessel with a flare, initial compliance is demonstrated by fulfilling the requirements in paragraph (a)(3) of this section.

(5) Exemptions from initial compliance. No initial compliance demonstration is required for control devices specified in paragraph (a)(4) of this section.

(6) Initial compliance with alternative standard. If the owner or operator equips a Group 1 storage vessel with a closed-vent system and control device, the owner or operator shall demonstrate initial compliance with the alternative standard in \S 63.1362(c)(4) by fulfilling the requirements of paragraph (a)(5) of this section.

(7) Planned routine maintenance. The owner or operator shall demonstrate initial compliance with the planned routine maintenance provisions of $\S63.1362(c)(5)$ by including the anticipated periods of planned routine maintenance for the first reporting period in the Notification of Compliance Status report as specified in $\S63.1368(f)$.

(e) Initial compliance with wastewater provisions. The owner or operator shall demonstrate initial compliance with the wastewater requirements by complying with the applicable provisions in §63.145, except that the owner or operator need not comply with the requirement to determine visible emissions that is specified in §63.145(j)(1), and references to compounds in Table 8 of subpart G of this part are not applicable for the purposes of this subpart. When §63.145(i) refers to Method 18 of 40 CFR part 60, appendix A-6, the owner or operator may use any method specified in §63.1362(d)(12) to demonstrate initial compliance with this subpart.

(f) Initial compliance with the bag dump and product dryer provisions. Compliance with the particulate matter concentration limits specified in §63.1362(e) is demonstrated when the concentration of particulate matter is less than 0.01 gr/dscf, as measured using the method described in paragraph (b)(7) of this section.

(g) Initial compliance with the pollution prevention alternative standard. The owner or operator shall demonstrate initial compliance with §63.1362(g)(2) and (3) for a PAI process unit by preparing the demonstration summary in accordance with paragraph (g)(1) of this section and by calculating baseline and target annual HAP and VOC factors in accordance with paragraphs (g)(2) and (3) of this section. To demonstrate initial compliance with §63.1362(g)(3), the owner or operator must also comply with the procedures for add-on control devices that are specified in paragraph (g)(4) of this section.

(1) Demonstration summary. The owner or operator shall prepare a pollution prevention demonstration summary that shall contain, at a minimum, the information in paragraphs (g)(1)(i)through (iii) of this section. The demonstration summary shall be included in the Precompliance report as specified in §63.1368(e)(4).

(i) Descriptions of the methodologies and forms used to measure and record consumption of HAP and VOC compounds.

(ii) Descriptions of the methodologies and forms used to measure and record production of the product(s).

(iii) Supporting documentation for the descriptions provided in accordance with paragraphs (g)(1)(i) and (ii) of this 40 CFR Ch. I (7–1–23 Edition)

section including, but not limited to, operator log sheets and copies of daily, monthly, and annual inventories of materials and products. The owner or operator must show how this documentation will be used to calculate the annual factors required in $\S63.1366(f)(1)$.

(2) Baseline factors. The baseline HAP and VOC factors shall be calculated by dividing the consumption of total HAP and total VOC by the production rate, per process, for the first 3-year period in which the process was operational, beginning no earlier than the period consisting of the 1987 through 1989 calendar years. Alternatively, for a process that has been operational for less than 3 years, but more than 1 year, the baseline factors shall be established for the time period from startup of the process until the present.

(3) Target annual factors. The owner or operator must calculate target annual factors in accordance with either paragraph (g)(3)(i) or (ii) of this section.

(i) To demonstrate initial compliance with §63.1362(g)(2), the target annual HAP factor must be equal to or less than 15 percent of the baseline HAP factor. For each reduction in a HAP that is also a VOC, the target annual VOC factor must be lower than the baseline VOC factor by an equivalent amount on a mass basis. For each reduction in a HAP that is not a VOC, the target annual factor must be equal to or less than the baseline VOC factor.

(ii) To demonstrate initial compliance with $\S63.1362(g)(3)(i)$, the target annual HAP and VOC factors must be calculated as specified in paragraph (g)(3)(i) of this section, except that when "15 percent" is referred to in paragraph (g)(3)(i) of this section, "50 percent" shall apply for the purposes of this paragraph.

(4) Requirements for add-on control devices. Initial compliance with the requirements for add-on control devices in $\S63.1362(g)(3)(ii)$ is demonstrated when the requirements in paragraphs (g)(4)(i) through (iii) of this section are met.

(i) The yearly reductions associated with add-on controls that meet the criteria of $\S63.1362(g)(3)(ii)(A)$ through (D), must be equal to or greater than the

§63.1365

amounts calculated using Equations 38 and 39 of this subpart:

$$HAP_{reduced} = (HF_{base})(0.85 - R_{P2})(M_{prod})$$
(Eq. 38)
$$VOC_{reduced} = (VF_{base} - VF_{P2} - VF_{annual}) \times M_{prod}$$
(Eq. 39)

Where:

- $\label{eq:happendix} \begin{array}{l} {\rm HAP}_{\rm reduced} = {\rm the \ annual \ HAP \ emissions \ reduction \ required \ by \ add-on \ controls, \ kg/yr \end{array}$
- HF_{base} = the baseline HAP factor, kg HAP consumed/kg product
- R_{P2} = the fractional reduction in the annual HAP factor achieved using pollution prevention where R_{P2} is ≥ 0.5
- VOC_{reduced} = required VOC emission reduction from add-on controls, kg/yr
- VF_{base} = baseline VOC factor, kg VOC emitted/kg production
- VF_{P2} = reduction in VOC factor achieved by pollution prevention, kg VOC emitted/kg production

 $VF_{annual} = target annual VOC factor, kg VOC emitted/kg production$

 M_{prod} = production rate, kg/yr

(ii) Demonstration that the criteria in $\S63.1362(g)(3)(ii)(A)$ through (D) are met shall be accomplished through a description of the control device and of the material streams entering and exiting the control device.

(iii) The annual reduction achieved by the add-on control shall be quantified using the methods described in paragraph (c) of this section.

(h) Compliance with emissions averaging provisions. An owner or operator shall demonstrate compliance with the emissions averaging provisions of $\S63.1362(h)$ by fulfilling the requirements of paragraphs (h)(1) through (6) of this section.

(1) The owner or operator shall develop and submit for approval an Emissions Averaging Plan containing all the information required in §63.1367(d). The Emissions Averaging Plan shall be submitted no later than 18 months prior to the compliance date of the standard. The Administrator shall determine within 120 calendar days whether the Emissions Averaging Plan submitted by sources using emissions averaging presents sufficient information. The Administrator shall either approve the Emissions Averaging Plan, request changes, or request that the

owner or operator submit additional information. Once the Administrator receives sufficient information, the Administrator shall approve, disapprove, or request changes to the plan within 120 days. If the Emissions Averaging Plan is disapproved, the owner or operator must still be in compliance with the standard by the compliance date.

(2) For all points included in an emissions average, the owner or operator shall comply with the procedures that are specified in paragraphs (h)(2)(i)through (v) of this section.

(i) Calculate and record monthly debits for all Group 1 emission points that are controlled to a level less stringent than the standard for those emission points. Equations in paragraph (h)(5) of this section shall be used to calculate debits.

(ii) Calculate and record monthly credits for all Group 1 and Group 2 emission points that are overcontrolled to compensate for the debits. Equations in paragraph (h)(6) of this section shall be used to calculate credits. All process vent, storage vessel, and wastewater emission points except those specified in §63.1362(h)(1) through (6) may be included in the credit calculation.

(iii) Demonstrate that annual credits calculated according to paragraph (h)(6) of this section are greater than or equal to debits calculated according to paragraph (h)(5) of this section for the same annual compliance period. The initial demonstration in the Emissions Averaging Plan or operating permit application that credit-generating emission points will be capable of generating sufficient credits to offset the debit-generating emission points shall be made under representative operating conditions. After the compliance date, actual operating data shall be

§63.1365

used for all debit and credit calculations.

(iv) Demonstrate that debits calculated for a quarterly (3-month) period according to paragraph (h)(5) of this section are not more than 1.30 times the credits for the same period calculated according to paragraph (h)(6) of this section. Compliance for the quarter shall be determined based on the ratio of credits and debits from that quarter, with 30 percent more debits than credits allowed on a quarterly basis.

(v) Record and report quarterly and annual credits and debits as required in §§ 63.1367(d) and 63.1368(d).

(3) [Reserved]

(4) During periods of monitoring excursions, credits and debits shall be adjusted as specified in paragraphs (h)(4)(i) through (iii) of this section.

(i) No credits shall be assigned to the credit-generating emission point.

40 CFR Ch. I (7–1–23 Edition)

(ii) Maximum debits shall be assigned to the debit-generating emission point.

(iii) The owner or operator may demonstrate to the Administrator that full or partial credits or debits should be assigned using the procedures in §63.150(1) of subpart G of this part.

(5) Debits are generated by the difference between the actual emissions from a Group 1 emission point that is uncontrolled or controlled to a level less stringent than the applicable standard and the emissions allowed for the Group 1 emission point. Debits shall be calculated in accordance with the procedures specified in paragraphs (h)(5)(i) through (iv) of this section.

(i) Source-wide debits shall be calculated using Equation 40 of this subpart.

Debits and all terms of Equation 40 of this subpart are in units of Mg/month

$$Debits = \sum_{i=1}^{n} \left[EPV_{iA} - (0.10) (EPV_{iU}) \right] + \sum_{i=1}^{n} \left[ES_{iA} - (0.05) (ES_{iU}) \right] + \sum_{i=1}^{n} \left[EWW_{iA} - (EWW_{iC}) \right]$$
(Eq. 40)

Where:

- EPV_{iU} = uncontrolled emissions from process i calculated according to the procedures specified in paragraph (h)(5)(ii) of this section
- EPV_{iA} = actual emissions from each Group 1 process i that is uncontrolled or is controlled to a level less stringent than the applicable standard. EPV_{iA} is calculated using the procedures in paragraph (h)(5)(ii) of this section
- $$\begin{split} & \mathrm{ES}_{\mathrm{iU}} = \mathrm{uncontrolled\ emissions\ from\ storage} \\ & \mathrm{vessel\ i\ calculated\ according\ to\ the\ procedures\ specified\ in\ paragraph\ (h)(5)(iii) \\ & \mathrm{of\ this\ section\ } \end{split}$$
- $$\begin{split} \mathrm{ES}_{\mathrm{iA}} &= \mathrm{actual} \ \mathrm{emissions} \ \mathrm{from} \ \mathrm{each} \ \mathrm{Group} \ 1 \\ & \mathrm{storage} \ \mathrm{vessel} \ \mathrm{i} \ \mathrm{that} \ \mathrm{is} \ \mathrm{uncontrolled} \ \mathrm{or} \ \mathrm{is} \\ & \mathrm{controlled} \ \mathrm{to} \ \mathrm{a} \ \mathrm{level} \ \mathrm{less} \ \mathrm{stringent} \ \mathrm{than} \\ & \mathrm{the} \ \mathrm{applicable} \ \mathrm{standard}. \ \mathrm{ES}_{\mathrm{iA}} \ \mathrm{is} \ \mathrm{calculated} \ \mathrm{using} \ \mathrm{the} \ \mathrm{procedures} \ \mathrm{in} \ \mathrm{paragraph} \\ & \mathrm{culated} \ \mathrm{using} \ \mathrm{the} \ \mathrm{procedures} \ \mathrm{in} \ \mathrm{paragraph} \ \mathrm{standard}. \end{split}$$
- $$\begin{split} & EWW_{iC} = emissions from each Group 1 waste-\\ & water stream i if the standard had been applied to the uncontrolled emissions.\\ & EWW_{iC} is calculated using the procedures in paragraph (h)(5)(iv) of this section \end{split}$$
- $\begin{array}{l} {\rm EWW}_{iA} \mbox{ = actual emissions from each Group 1} \\ {\rm wastewater stream \ i \ that \ is \ uncontrolled} \\ {\rm or \ is \ controlled \ to \ a \ level \ less \ stringent} \\ {\rm than \ the \ applicable \ standard. \ EWW}_{iA} \ is \end{array}$

calculated using the procedures in paragraph (h)(5)(iv) of this section

n = the number of emission points being included in the emissions average; the value of n is not necessarily the same for process vents, storage tanks, and wastewater

(ii) Emissions from process vents shall be calculated in accordance with the procedures specified in paragraphs (h)(5)(ii)(A) through (C) of this section.

(A) Except as provided in paragraph (h)(5)(ii)(C) of this section, uncontrolled emissions for process vents shall be calculated using the procedures that are specified in paragraph (c)(2) of this section.

(B) Except as provided in paragraph (h)(5)(ii)(C) of this section, actual emissions for process vents shall be calculated using the procedures specified in paragraphs (c)(2) and (c)(3) of this section, as applicable.

(C) As an alternative to the procedures described in paragraphs (h)(5)(ii)(A) and (B) of this section, for continuous processes, uncontrolled and actual emissions may be calculated by

the procedures described in $\S63.150(g)(2)$ of subpart G of this part. For purposes of complying with this paragraph, a 90 percent reduction shall apply instead of the 98 percent reduction in $\S63.150(g)(2)(ii)$ of subpart G of this part, and the term "process condenser" shall apply instead of the term "recovery device" in $\S63.150(g)(2)$ for the purposes of this subpart.

(iii) Uncontrolled emissions from storage vessels shall be calculated in accordance with the procedures described in paragraph (d)(1) of this section. Actual emissions from storage vessels shall be calculated using the procedures specified in 63.150(g)(3)(i), (iii), or (iv) of subpart G of this subpart, as appropriate, except that when 63.150(g)(3)(i)(B) refers to the procedures in 63.120(d) for determining percent reduction for a control device, 63.1365(d)(2) or (3) shall apply for the purposes of this subpart.

(iv) Emissions from wastewater shall be calculated using the procedures specified in 63.150(g)(5) of subpart G of this part.

(6) Credits are generated by the difference between emissions that are allowed for each Group 1 and Group 2 emission point and the actual emissions from that Group 1 or Group 2 emission point that have been controlled after November 15, 1990 to a level more stringent than what is required in this subpart or any other State or Federal rule or statute. Credits shall be calculated in accordance with the procedures specified in paragraphs (h)(6)(i) through (v) of this section.

(i) Source-wide credits shall be calculated using Equation 41 of this subpart. Credits and all terms in Equation 41 of this subpart are in units of Mg/ month, the baseline date is November 15, 1990, the terms consisting of a constant multiplied by the uncontrolled emissions are the emissions from each emission point subject to the standards in §63.1362(b) and (c) that is controlled to a level more stringent than the standard.

$$Credits = D\sum_{i=1}^{n} [(0.10)(EPV1_{iU}) - EPV1_{iA}] + D\sum_{i=1}^{m} (EPV2_{iB} - EPV2_{iA}) + D\sum_{i=1}^{n} [(0.05)(ES1_{iU}) - ES1_{iA}] + D\sum_{i=1}^{m} (ES2_{iB} - ES2_{iA}) + D\sum_{i=1}^{n} (EWW1_{iC} - EWW1_{iA}) + D\sum_{i=1}^{m} (EWW2_{iB} - EWW2_{iA})$$
(Eq. 41)

Where:

- $$\begin{split} EPV1_{iU} &= uncontrolled \ emissions \ from \ each \\ Group 1 \ process \ i \ calculated \ according \ to \\ the \ procedures \ in \ paragraph \ (h)(6)(iii)(A) \\ of \ this \ section \end{split}$$
- $$\begin{split} EPV1_{iA} &= actual \ emissions \ from \ each \ Group \ 1 \\ process \ i \ that \ is \ controlled \ to \ a \ level \\ more \ stringent \ than \ the \ applicable \\ standard. \ EPV1_{iA} \ is \ calculated \ according \\ to \ the \ procedures \ in \ paragraph \\ (h)(6)(iii)(B) \ of \ this \ section \end{split}$$
- $$\begin{split} EPV2_{iB} &= emissions \mbox{ from each Group 2 proc-}\\ ess \mbox{ i at the baseline date. } EPV2_{iB} \mbox{ is cal-}\\ culated according to the procedures in \\ paragraph (h)(6)(iii)(C) \mbox{ of this section} \end{split}$$
- $$\begin{split} EPV2_{iA} &= actual \ emissions \ from \ each \ Group \ 2 \\ process \ i \ that \ is \ controlled. \ EPV2_{iA} \ is \\ calculated \ according \ to \ the \ procedures \ in \\ paragraph \ (h)(6)(iii)(C) \ of \ this \ section \end{split}$$
- $$\begin{split} \mathrm{ES1}_{iU} &= \text{ uncontrolled emissions from each} \\ \mathrm{Group 1 storage vessel i calculated according to the procedures in paragraph} \\ \mathrm{(h)(6)(iv) of this section} \end{split}$$

- $$\begin{split} \mathrm{ES1}_{iA} &= \mathrm{actual} \ \mathrm{emissions} \ \mathrm{from} \ \mathrm{each} \ \mathrm{Group} \ 1 \\ \mathrm{storage} \ \mathrm{vessel} \ i \ \mathrm{that} \ \mathrm{is} \ \mathrm{controlled} \ \mathrm{to} \ \mathrm{a} \\ \mathrm{level} \ \mathrm{more} \ \mathrm{stringent} \ \mathrm{that} \ \mathrm{the} \ \mathrm{applicable} \\ \mathrm{standard.} \ \mathrm{ES1}_{iA} \ \mathrm{is} \ \mathrm{calculated} \ \mathrm{according} \\ \mathrm{to} \ \mathrm{the} \ \mathrm{procedures} \ \mathrm{in} \ \mathrm{paragraph} \ (h)(6)(\mathrm{iv}) \\ \mathrm{of} \ \mathrm{this} \ \mathrm{section} \end{split}$$
- $$\begin{split} \mathrm{ES2}_{\mathrm{iB}} &= \mathrm{emissions} \ \mathrm{from} \ \mathrm{each} \ \mathrm{Group} \ 2 \ \mathrm{storage} \\ \mathrm{vessel} \ \mathrm{i} \ \mathrm{at} \ \mathrm{th} \ \mathrm{baseline} \ \mathrm{date} \ \mathrm{ES2}_{\mathrm{iB}} \ \mathrm{is} \ \mathrm{calculated} \\ \mathrm{culated} \ \mathrm{according} \ \mathrm{to} \ \mathrm{th} \ \mathrm{porcedures} \ \mathrm{in} \\ \mathrm{paragraph} \ (\mathrm{h})(6)(\mathrm{iv}) \ \mathrm{of} \ \mathrm{this} \ \mathrm{section} \end{split}$$
- $\mathrm{ES2}_{iA}$ = actual emissions from each Group 2 storage vessel i that is controlled. $\mathrm{ES2}_{iA}$ is calculated according to the procedures in paragraph (h)(6)(iv) of this section
- $\mathrm{EWW1_{iA}}$ = emissions from each Group 1 wastewater stream i that is controlled to a level more stringent that the applicable standard. $\mathrm{EWW1_{iA}}$ is calculated according

to the procedures in paragraph (h)(6)(v) of this section

- $\rm EWW2_{iB}$ = emissions from each Group 2 wastewater stream i at the baseline date. $\rm EWW2_{iB}$ is calculated according to the procedures in paragraph (h)(6)(v) of this section
- n = number of Group 1 emission points that are included in the emissions average. The value of n is not necessarily the same for process vents, storage tanks, and wastewater
- m = number of Group 2 emission points included in the emissions average. The value of m is not necessarily the same for process vents, storage tanks, and wastewater
- D = discount factor equal to 0.9 for all creditgenerating emission points except those controlled by a pollution prevention measure, which will not be discounted

40 CFR Ch. I (7–1–23 Edition)

(ii) For an emission point controlled using a pollution prevention measure, the nominal efficiency for calculating credits shall be as determined as described in §63.150(j) of subpart G of this part.

(iii) Emissions from process vents shall be calculated in accordance with the procedures specified in paragraphs (h)(6)(iii)(A) through (C) of this section.

(A) Uncontrolled emissions from Group 1 process vents shall be calculated according to the procedures in paragraph (h)(5)(ii)(A) or (C) of this section.

(B) Actual emissions from Group 1 process vents with a nominal efficiency greater than the applicable standard or a pollution prevention measure that achieves reductions greater than the applicable standard shall be calculated using Equation 42 of this subpart:

$$EPV1_{iA} = EPV1_{iU} \times [1 - N_{eff} / 100]$$
 (Eq. 42)

Where:

- $\mathrm{EPV1}_{iA}$ = actual emissions from each Group 1 process i that is controlled to a level more stringent than the applicable standard
- $EPV1_{iU}$ = uncontrolled emissions from each Group 1 process i
- $N_{\rm eff}$ = nominal efficiency of control device or pollution prevention measure, percent

(C) Baseline and actual emissions from Group 2 process vents shall be calculated according to the procedures in §63.150(h)(2)(iii) and (iv) with the following modifications:

(1) The term "90 percent reduction" shall apply instead of the term "98 percent reduction"; and

(2) When the phrase "paragraph (g)(2)" is referred to in §63.150(h)(2)(iii) and (iv), the provisions in paragraph (h)(5)(ii) of this section shall apply for the purposes of this subpart.

(iv) Uncontrolled emissions from storage vessels shall be calculated according to the procedures described in paragraph (d)(1) of this section. Actual and baseline emissions from storage tanks shall be calculated according to the procedures specified in \S 63.150(h)(3) of subpart G of this part, except when (3.150(h)(3)) refers to (3.150(g)(3)(i)), paragraph (d)(1) of this section shall apply for the purposes of this subpart.

(v) Emissions from wastewater shall be calculated using the procedures in §63.150(h)(5) of subpart G of this part.

[64 FR 33589, June 23, 1999, as amended at 67 FR 59347, Sept. 20, 2002; 79 FR 17374, Mar. 27, 2014]

§63.1366 Monitoring and inspection requirements.

(a) To provide evidence of continued compliance with the standard, the owner or operator of any existing or new affected source shall install, operate, and maintain monitoring devices as specified in this section. During the initial compliance demonstration, maximum or minimum operating parameter levels, or other design and operating characteristics, as appropriate, shall be established for emission sources that will indicate the source is in compliance. Test data, calculations, or information from the evaluation of

the control device design, as applicable, shall be used to establish the operating parameter level or characteristic.

(b) Monitoring for control devices—(1) Parameters to monitor. Except as specified in paragraph (b)(1)(i) of this section, for each control device, the owner or operator shall install and operate monitoring devices and operate within the established parameter levels to ensure continued compliance with the standard. Monitoring parameters are specified for control scenarios in paragraphs (b)(1)(ii) through (xii) of this section, and are summarized in Table 3 of this subpart.

(i) Periodic verification. For control devices that control vent streams containing total HAP emissions less than 0.91 Mg/yr, before control, monitoring shall consist of a periodic verification that the device is operating properly. This verification shall include, but not be limited to, a daily or more frequent demonstration that the unit is working as designed and may include the daily measurements of the parameters described in paragraphs (b)(1)(ii) through (xii) of this section. This demonstration shall be included in the Precompliance plan, to be submitted 6 months prior to the compliance date of the standard.

(ii) Scrubbers. For affected sources using liquid scrubbers, the owner or operator shall establish a minimum scrubber liquid flow rate or pressure drop as a site-specific operating parameter which must be measured and recorded at least once every 15 minutes during the period in which the scrubber is controlling HAP from an emission stream as required by the standards in §63.1362. If the scrubber uses a caustic solution to remove acid emissions, the pH of the effluent scrubber liquid shall also be monitored once a day. The minimum scrubber liquid flow rate or pressure drop shall be based on the conditions under which the initial compliance demonstration was conducted. Alternatively, for halogen scrubbers, the owner or operator may comply with the requirements specified in §63.994(c).

(A) The monitoring device used to determine the pressure drop shall be certified by the manufacturer to be accurate to within a gage pressure of ± 10 percent of the maximum pressure drop measured.

(B) The monitoring device used for measurement of scrubber liquid flowrate shall be certified by the manufacturer to be accurate to within ± 10 percent of the design scrubber liquid flowrate.

(C) The monitoring device shall be calibrated annually.

(iii) Condensers. For each condenser, the owner or operator shall establish the maximum condenser outlet gas temperature as a site-specific operating parameter which must be measured and recorded at least once every 15 minutes during the period in which the condenser is controlling HAP from an emission stream as required by the standards in §63.1362.

(A) The temperature monitoring device must be accurate to within ± 2 percent of the temperature measured in degrees Celsius or ± 2.5 °C, whichever is greater.

(B) The temperature monitoring device must be calibrated annually.

(iv) Regenerative carbon adsorbers. For each regenerative carbon adsorber, the owner or operator shall comply with the provisions in paragraphs (b)(1)(iv)(A) through (F) of this section.

(A) Establish the regeneration cycle characteristics specified in paragraphs (b)(1)(iv)(A) (1) through (4) of this section under absolute or hypothetical peak-case conditions, as defined in $\S63.1365(b)(11)(i)$ or (ii).

(1) Minimum regeneration frequency (i.e., operating time since last regeneration);

(2) Minimum temperature to which the bed is heated during regeneration;

(3) Maximum temperature to which the bed is cooled, measured within 15 minutes of completing the cooling phase; and

(4) Minimum regeneration stream flow.

(B) Monitor and record the regeneration cycle characteristics specified in paragraphs (b)(1)(iv)(B) (1) through (4) of this section for each regeneration cycle.

(1) Regeneration frequency (i.e., operating time since end of last regeneration);

(2) Temperature to which the bed is heated during regeneration;
(3) Temperature to which the bed is cooled, measured within 15 minutes of the completion of the cooling phase; and

(4) Regeneration stream flow.

(C) Use a temperature monitoring device that is accurate to within ± 2 percent of the temperature measured in degrees Celsius or ± 2.5 °C, whichever is greater.

(D) Use a regeneration stream flow monitoring device capable of recording the total regeneration stream flow to within ± 10 percent of the established value (i.e., accurate to within ± 10 percent of the reading).

(E) Calibrate the temperature and flow monitoring devices annually.

(F) Conduct an annual check for bed poisoning in accordance with manufacturer's specifications.

(v) Nonregenerative carbon adsorbers. For each nonregenerative carbon adsorption system such as a carbon canister that does not regenerate the carbon bed directly onsite in the control device, the owner or operator shall replace the existing carbon bed in the control device with fresh carbon on a regular schedule based on one of the following procedures:

(A) Monitor the TOC concentration level in the exhaust vent stream from the carbon adsorption system on a regular schedule, and replace the existing carbon with fresh carbon immediately when carbon breakthrough is indicated. The monitoring frequency shall be daily or at an interval no greater than 20 percent of the time required to consume the total carbon working capacity under absolute or hypothetical peak-case conditions as defined in §63.1365(b)(11)(i) or (ii), whichever is longer.

(B) Establish the maximum time interval between replacement, and replace the existing carbon before this time interval elapses. The time interval shall be established based on the conditions anticipated under absolute or hypothetical peak-case, as defined in §63.1365(b)(11)(i) or (ii).

(vi) *Flares.* For each flare, the presence of the pilot flame shall be monitored at least once every 15 minutes during the period in which the flare is controlling HAP from an emission stream subject to the standards in 40 CFR Ch. I (7–1–23 Edition)

§63.1362. The monitoring device shall be calibrated annually.

(vii) Thermal incinerators. For each thermal incinerator, the owner or operator shall monitor the temperature of the gases exiting the combustion chamber as the site-specific operating parameter which must be measured and recorded at least once every 15 minutes during the period in which the combustion device is controlling HAP from an emission stream subject to the standards in §63.1362.

(A) The temperature monitoring device must be accurate to within ± 0.75 percent of the temperature measured in degrees Celsius or ± 2.5 °C, whichever is greater.

(B) The monitoring device must be calibrated annually.

(viii) Catalytic incinerators. For each catalytic incinerator, the parameter levels that the owner or operator shall establish are the minimum temperature of the gas stream immediately before the catalyst bed and the minimum temperature difference across the catalyst bed. The owner or operator shall monitor the temperature of the gas stream immediately before and after the catalyst bed, and calculate the temperature difference across the catalyst bed, at least once every 15 minutes during the period in which the catalytic incinerator is controlling HAP from an emission stream subject to the standards in §63.1362.

(A) The temperature monitoring devices must be accurate to within ± 0.75 percent of the temperature measured in degrees Celsius or ± 2.5 °C, whichever is greater.

(B) The temperature monitoring devices must be calibrated annually.

(ix) Process heaters and boilers. (A) Except as specified in paragraph (b)(1)(ix)(B) of this section, for each boiler or process heater, the owner or operator shall monitor the temperature of the gases exiting the combustion chamber as the site-specific operating parameter which must be monitored and recorded at least every 15 minutes during the period in which the boiler or process heater is controlling HAP from an emission stream subject to the standards in §63.1362.

(1) The temperature monitoring device must be accurate to within ± 0.75

percent of the temperature measured in degrees Celsius or ± 2.5 °C, whichever is greater.

(2) The temperature monitoring device must be calibrated annually.

(B) The owner or operator is exempt from the monitoring requirements specified in paragraph (b)(1)(ix)(A) of this section if either:

(1) All vent streams are introduced with primary fuel; or

(2) The design heat input capacity of the boiler or process heater is 44 megawatts or greater.

(x) Continuous emission monitor. As an alternative to the parameters specified in paragraphs (b)(1)(ii) through (ix) of this section, an owner or operator may monitor and record the outlet HAP concentration or both the outlet TOC concentration and outlet total HCl and chlorine concentration at least every 15 minutes during the period in which the control device is controlling HAP from an emission stream subject to the standards in §63.1362. The owner or operator need not monitor the total HCl and chlorine concentration if the owner or operator determines that the emission stream does not contain HCl or chlorine. The owner or operator need not monitor the TOC concentration if the owner or operator determines the emission stream does not contain organic compounds. The HAP or TOC monitor must meet the requirements of Performance Specification 8 or 9 of appendix B of part 60 and must be installed, calibrated, and maintained, according to §63.8 of subpart A of this part. As part of the QA/QC Plan, calibration of the device must include, at a minimum, quarterly cylinder gas audits. If supplemental gases are introduced before the control device, the monitored concentration shall be corrected as specified in $\S63.1365(a)(7)$.

(xi) Fabric filters. For each fabric filter used to control particulate matter emissions from bag dumps and product dryers subject to $\S63.1362(e)$, the owner or operator shall install, calibrate, maintain, and continuously operate a bag leak detection system that meets the requirements in paragraphs (b)(1)(xi)(A) through (G) of this section.

(A) The bag leak detection system sensor must provide output of relative particulate matter emissions.

(B) The bag leak detection system must be equipped with an alarm system that will sound when an increase in particulate matter emissions over a preset level is detected.

(C) For positive pressure fabric filters, a bag leak detector must be installed in each fabric filter compartment or cell. If a negative pressure or induced air filter is used, the bag leak detector must be installed downstream of the fabric filter. Where multiple bag leak detectors are required (for either type of fabric filter), the system instrumentation and alarm may be shared among detectors.

(D) The bag leak detection system shall be installed, operated, calibrated and maintained in a manner consistent with available guidance from the U.S. Environmental Protection Agency or, in the absence of such guidance, the manufacturer's written specifications and instructions.

(E) Calibration of the system shall, at a minimum, consist of establishing the relative baseline output level by adjusting the range and the averaging period of the device and establishing the alarm set points and the alarm delay time.

(F) Following initial adjustment, the owner or operator shall not adjust the sensitivity or range, averaging period, alarm set points, or alarm delay time, except as established in an operation and maintenance plan that is to be submitted with the Precompliance plan. In no event shall the sensitivity be increased more than 100 percent or decreased by more than 50 percent over a 365-day period unless such adjustment follows a complete baghouse inspection which demonstrates the baghouse is in good operating condition.

(G) If the alarm on a bag leak detection system is triggered, the owner or operator shall, within 1 hour of an alarm, initiate the procedures to identify the cause of the alarm and take corrective action as specified in the corrective action plan.

(xii) For each waste management unit, treatment process, or control device used to comply with §63.1362(d), the owner or operator shall comply with the procedures specified in §63.143 of subpart G of this part, except that when the procedures to request approval to monitor alternative parameters according to the procedures in $\S63.151(f)$ are referred to in $\S63.143(d)(3)$, the procedures in paragraph (b)(4) of this section shall apply for the purposes of this subpart.

(xiii) Closed-vent system visual inspections. The owner or operator shall comply with the requirements in either paragraph (b)(1)(xiii)(A) or (B) of this section:

(A) Set the flow indicator at the entrance to any bypass line that could divert the stream away from the control device to the atmosphere to take a reading at least once every 15 minutes; or

(B) If the bypass device valve installed at the inlet to the bypass device is secured in the closed position with a car-seal or lock-and-key type configuration, visually inspect the seal or closure mechanism at least once every month to verify that the valve is maintained in the closed position and the vent stream is not diverted through the bypass line.

(2) Averaging periods. Averaging periods for parametric monitoring levels shall be established according to paragraphs (b)(2)(i) through (iii) of this section.

(i) Except as provided in paragraph (b)(2)(iii) of this section, a daily (24-hour) or block average shall be calculated as the average of all values for a monitored parameter level set according to the procedures in (b)(3)(ii) of this section recorded during the operating day or block.

(ii) The operating day or block shall be defined in the Notification of Compliance Status report. The operating day may be from midnight to midnight or another continuous 24-hour period. The operating block may be used as an averaging period only for vents from batch operations, and is limited to a period of time that is, at a maximum, equal to the time from the beginning to end of a series of consecutive batch operations.

(iii) Monitoring values taken during periods in which the control devices are not controlling HAP from an emission stream subject to the standards in §63.1362, as indicated by periods of no flow or periods when only streams that 40 CFR Ch. I (7–1–23 Edition)

are not subject to the standards in §63.1362 are controlled, shall not be considered in the averages. Where flow to the device could be intermittent, the owner or operator shall install, calibrate and operate a flow indicator at the inlet or outlet of the control device to identify periods of no flow.

(3) Procedures for setting parameter levels for control devices used to control emissions from process vents. (i) Small control devices. Except as provided in paragraph (b)(1)(i) of this section, for devices controlling less than 10 tons/yr of HAP for which a performance test is not required, the parameteric levels shall be set based on the design evaluation required in \S 63.1365(c)(3)(i)(A). If a performance test is conducted, the monitoring parameter level shall be established according to the procedures in paragraph (b)(3)(ii) of this section.

(ii) *Large control devices*. For devices controlling greater than or equal to 10 tons/yr of HAP for which a performance test is required, the parameter level must be established as follows:

(A) If the operating parameter level to be established is a maximum or minimum, it must be based on the average of the average values from each of the three test runs.

(B) The owner or operator may establish the parametric monitoring level(s) based on the performance test supplemented by engineering assessments and/or manufacturer's recommendations. Performance testing is not required to be conducted over the entire range of expected parameter values. The rationale for the specific level for each parameter, including any data and calculations used to develop the level(s) and a description of why the level indicates proper operation of the control device shall be provided in the Precompliance plan. Determination of the parametric monitoring level using these procedures is subject to review and approval by the Administrator.

(iii) Parameter levels for control devices controlling batch process vents. For devices controlling batch process vents alone or in combination with other streams, the level(s) shall be established in accordance with paragraph (b)(3)(iii)(A) or (B) of this section.

(A) A single level for the batch process(es) shall be calculated from the initial compliance demonstration.

(B) The owner or operator may establish separate levels for each batch emission episode or combination of emission episodes selected to be controlled. If separate monitoring levels are established, the owner or operator must provide a record indicating at what point in the daily schedule or log of processes required to be recorded per the requirements of §63.1367(b)(7), the parameter being monitored changes levels and must record at least one reading of the new parameter level, even if the duration of monitoring for the new parameter level is less than 15 minutes.

(4) Requesting approval to monitor alternative parameters. The owner or operator may request approval to monitor parameters other than those required by paragraphs (b)(1)(ii) through (xiii) of this section. The request shall be submitted according to the procedures specified in §63.8(f) of subpart A of this part or in the Precompliance report (as specified in §63.1368(e)).

(5) Monitoring for the alternative standards. (i) For control devices that are used to comply with the provisions of §63.1362(b)(6) and (c)(4), the owner or operator shall monitor and record the outlet TOC concentration and the outlet total HCl and chlorine concentration at least once every 15 minutes during the period in which the device is HAP controlling from emission streams subject to the standards in §63.1362. A TOC monitor meeting the requirements of Performance Specification 8 or 9 of appendix B of 40 CFR part 60 shall be installed, calibrated, and maintained, according to §63.8. The owner or operator need not monitor the total HCl and chlorine concentration if the owner or operator determines that the emission stream does not contain HCl or chlorine. The owner or operator need not monitor for TOC concentration if the owner or operator determines that the emission stream does not contain organic compounds.

(ii) If supplemental gases are introduced before the control device, the owner or operator must either correct for supplemental gases as specified in $\S63.1365(a)(7)$ or, if using a combustion control device, comply with the requirements of paragraph (b)(5)(ii)(A) of this section. If the owner or operator corrects for supplemental gases as specified in (3.1365(a))(7)(ii) for noncombustion control devices, the flow rates must be evaluated as specified in paragraph (b)(5)(ii)(B) of this section.

(A) Provisions for combustion devices. As an alternative to correcting for supplemental gases as specified in $\S63.1365(a)(7)$, the owner or operator may monitor residence time and firebox temperature according to the requirements of paragraphs (b)(5)(ii)(A)(1) and (2) of this section. Monitoring of residence time may be accomplished by monitoring flow rate into the combustion chamber.

(1) If complying with the alternative standard instead of achieving a control efficiency of 95 percent or less, the owner or operator must maintain a minimum residence time of 0.5 seconds and a minimum combustion chamber temperature of 760 °C.

(2) If complying with the alternative standard instead of achieving a control efficiency of 98 percent, the owner or operator must maintain a minimum residence time of 0.75 seconds and a minimum combustion chamber temperature of 816 °C.

(B) Flow rate evaluation for non-combustion devices. To demonstrate continuous compliance with the requirement to correct for supplemental gases as specified in §63.1365(a)(7)(ii) for noncombustion devices, the owner or operator must evaluate the volumetric flow rate of supplemental gases, V_s , and the volumetric flow rate of all gases, V_a, each time a new operating scenario is implemented based on process knowledge and representative operating data. The procedures used to evaluate the flow rates, and the resulting correction factor used in Equation 8 of this subpart, must be included in the Notification of Compliance Status report and in the next Periodic report submitted after an operating scenario change.

(6) Exceedances of operating parameters. An exceedance of an operating parameter is defined as one of the following:

(i) If the parameter level, averaged over the operating day or block, is below a minimum value established during the initial compliance demonstration.

(ii) If the parameter level, averaged over the operating day or block, is above the maximum value established during the initial compliance demonstration.

(iii) A loss of all pilot flames for a flare during an operating day or block. Multiple losses of all pilot flames during an operating day constitutes one exceedance.

(iv) Each operating day or block for which the time interval between replacement of a nonregenerative carbon adsorber exceeds the interval established in paragraph (b)(1)(v) of this section.

(v) Each instance in which procedures to initiate the response to a bag leak detector alarm within 1 hour of the alarm as specified in the corrective action plan.

(7) Excursions. Excursions are defined by either of the two cases listed in paragraph (b)(7)(i) or (ii) of this section. An excursion also occurs if the periodic verification for a small control device is not conducted as specified in paragraph (b)(1)(i) of this section.

(i) When the period of control device operation is 4 hours or greater in an operating day or block and monitoring data are insufficient to constitute a valid hour of data, as defined in paragraph (b)(7)(iii) of this section, for at least 75 percent of the operating hours.

(ii) When the period of control device operation is less than 4 hours in an operating day or block and more than 1 of the hours during the period of operation does not constitute a valid hour of data due to insufficient monitoring data.

(iii) Monitoring data are insufficient to constitute a valid hour of data, as used in paragraphs (b)(7)(i) and (ii) of this section, if measured values are unavailable for any of the required 15minute periods within the hour.

(8) Violations. Exceedances of parameters monitored according to the provisions of paragraphs (b)(1)(ii), (b)(1)(iv) through (ix), and (b)(5) of this section, or excursions as defined by paragraphs (b)(7)(i) and (ii) of this section, constitute violations of the operating limit according to paragraphs (b)(8)(i) 40 CFR Ch. I (7–1–23 Edition)

and (ii) of this section. Exceedances of the temperature limit monitored according to the provisions of paragraph (b)(1)(iii) of this section or exceedances of the outlet concentrations monitored according to the provisions of paragraph (b)(1)(x) of this section constitute violations of the emission limit according to paragraphs (b)(8)(i) and (ii) of this section. Exceedances of the outlet concentrations monitored according to the provisions of paragraph (b)(5) of this section constitute violations of the emission limit according to the provisions of paragraph (b)(8)(iii) of this section.

(i) For episodes occurring more than once per day, exceedances of established parameter limits or excursions will result in no more than one violation per operating day for each monitored item of equipment utilized in the process.

(ii) For control devices used for more than one process in the course of an operating day, exceedances or excursions will result in no more than one violation per operating day, per control device, for each process for which the control device is in service.

(iii) Exceedances of the 20 or 50 ppmv TOC outlet emission limit, averaged over the operating day, will result in no more than one violation per day per control device. Exceedances of the 20 or 50 ppmv HCl and chlorine outlet emission limit, averaged over the operating day, will result in no more than one violation per day per control device.

(c) Monitoring for uncontrolled emission rates. The owner or operator shall demonstrate continuous compliance with the emission limit in $\S63.1362$ (b)(2)(i) or (b)(4)(i) by calculating daily a 365day rolling summation of uncontrolled emissions based on the uncontrolled emissions per emission episode, as calculated using the procedures in $\S63.1365(c)(2)$, and records of the number of batches produced. Each day that the summation for a process exceeds 0.15 Mg/yr is considered a violation of the emission limit.

(d) Monitoring for equipment leaks. The standard for equipment leaks is based on monitoring. All monitoring requirements for equipment leaks are specified in §63.1363.

(e) Monitoring for heat exchanger systems. The standard for heat exchanger systems is based on monitoring. All monitoring requirements for heat exchanger systems are specified in §63.1362(f).

(f) Monitoring for the pollution prevention alternative standard. The owner or operator of an affected source that chooses to comply with the requirements of §63.1362(g) (2) or (3) shall calculate annual rolling average values of the HAP and VOC factors in accordance with the procedures specified in paragraph (f)(1) of this section. If complying with §63.1362(g)(3), the owner or operator shall also comply with the monitoring requirements specified in paragraph (b) of this section for the applicable add-on air pollution control device.

(1) Annual factors. The annual HAP and VOC factors shall be calculated in accordance with the procedures specified in paragraphs (f)(1) (i) through (iii) of this section.

(i) The consumption of both total HAP and total VOC shall be divided by the production rate, per process, for 12-month periods at the frequency specified in either paragraph (f)(1) (ii) or (iii) of this section, as applicable.

(ii) For continuous processes, the annual factors shall be calculated every 30 days for the 12-month period preceding the 30th day (annual rolling average calculated every 30 days). A process with both batch and continuous operations is considered a continuous process for the purposes of this section.

(iii) For batch processes, the annual factors shall be calculated every 10 batches for the 12-month period preceding the 10th batch (annual rolling average calculated every 10 batches). Additional annual factors shall be calculated every 12 months during the period before the 10th batch if more than 12 months elapse before the 10th batch is produced.

(2) Violations. Each rolling average that exceeds the target value established in (3.1365(g))(3) is considered a violation of the emission limit.

(g) Monitoring for emissions averaging. The owner or operator of an affected source that chooses to comply with the requirements of §63.1362(h) shall meet all monitoring requirements specified in paragraph (b) of this section, as applicable, for all processes, storage tanks, and waste management units included in the emissions average.

(h) Leak inspection provisions for vapor suppression equipment. (1) Except as provided in paragraphs (h)(9) and (10) of this section, for each vapor collection system, closed-vent system, fixed roof, cover, or enclosure required to comply with this section, the owner or operator shall comply with the requirements of paragraphs (h)(2) through (8) of this section.

(2) Except as provided in paragraphs (h)(6) and (7) of this section, each vapor collection system and closed-vent system shall be inspected according to the procedures and schedule specified in paragraphs (h)(2)(i) and (ii) of this section and each fixed roof, cover, and enclosure shall be inspected according to the procedures and schedule specified in paragraph (h)(2)(ii) of this section.

(i) If the vapor collection system or closed-vent system is constructed of hard-piping, the owner or operator shall:

(A) Conduct an initial inspection according to the procedures in paragraph (h)(3) of this section, and

(B) Conduct annual visual inspections for visible, audible, or olfactory indications of leaks.

(ii) If the vapor collection system or closed-vent system is constructed of ductwork, the owner or operator shall:

(A) Conduct an initial inspection according to the procedures in paragraph (h)(3) of this section,

(B) Conduct annual inspections according to the procedures in paragraph (h)(3) of this section, and

(C) Conduct annual visual inspections for visible, audible, or olfactory indications of leaks.

(iii) For each fixed roof, cover, and enclosure, the owner or operator shall:

(A) Conduct an initial inspection according to the procedures in paragraph (h)(3) of this section, and

(B) Conduct semiannual visual inspections for visible, audible, or olfactory indications of leaks.

(3) Each vapor collection system, closed-vent system, fixed roof, cover, and enclosure shall be inspected according to the procedures specified in paragraphs (h)(3)(i) through (vi) of this section.

(i) Inspections shall be conducted in accordance with Method 21 of 40 CFR part 60, appendix A.

(ii) Detection instrument performance criteria. (A) Except as provided in paragraph (h)(3)(ii)(B) of this section, the detection instrument shall meet the performance criteria of Method 21 of 40 CFR part 60, appendix A, except the instrument response factor criteria in section 3.1.2(a) of Method 21 shall be for the average composition of the process fluid not each individual VOC in the stream. For process streams that contain nitrogen, air, or other inerts which are not organic HAP or VOC, the average stream response factor shall be calculated on an inert-free basis.

(B) If no instrument is available at the plant site that will meet the performance criteria specified in paragraph (h)(3)(ii)(A) of this section, the instrument readings may be adjusted by multiplying by the average response factor of the process fluid, calculated on an inert-free basis as described in paragraph (h)(3)(ii)(A) of this section.

(iii) The detection instrument shall be calibrated before use on each day of its use by the procedures specified in Method 21 of 40 CFR part 60, appendix A.

(iv) Calibration gases shall be as follows:

(A) Zero air (less than 10 parts per million hydrocarbon in air); and

(B) Mixtures of methane in air at a concentration less than 10,000 parts per million. A calibration gas other than methane in air may be used if the instrument does not respond to methane or if the instrument does not meet the performance criteria specified in paragraph (h)(2)(ii)(A) of this section. In such cases, the calibration gas may be a mixture of one or more of the compounds to be measured in air.

(v) An owner or operator may elect to adjust or not adjust instrument readings for background. If an owner or operator elects to not adjust readings for background, all such instrument readings shall be compared directly to the applicable leak definition to determine whether there is a leak. If an owner or operator elects to adjust instrument readings for background, the

40 CFR Ch. I (7–1–23 Edition)

owner or operator shall measure background concentration using the procedures in §63.180(b) and (c). The owner or operator shall subtract background reading from the maximum concentration indicated by the instrument.

(vi) The arithmetic difference between the maximum concentration indicated by the instrument and the background level shall be compared with 500 parts per million for determining compliance.

(4) Leaks, as indicated by an instrument reading greater than 500 parts per million above background or by visual inspections, shall be repaired as soon as practicable, except as provided in paragraph (h)(5) of this section.

(i) A first attempt at repair shall be made no later than 5 calendar days after the leak is detected.

(ii) Repair shall be completed no later than 15 calendar days after the leak is detected.

(5) Delay of repair of a vapor collection system, closed-vent system, fixed roof, cover, or enclosure for which leaks have been detected is allowed if the repair is technically infeasible without a shutdown, as defined in §63.1361, or if the owner or operator determines that emissions resulting from immediate repair would be greater than the fugitive emissions likely to result from delay of repair. Repair of such equipment shall be complete by the end of the next shutdown.

(6) Any parts of the vapor collection system, closed-vent system, fixed roof, cover, or enclosure that are designated, as described in $\S63.1367(f)(1)$, as unsafeto-inspect are exempt from the inspection requirements of paragraphs (h)(2)(i), (ii), and (iii) of this section if:

(i) The owner or operator determines that the equipment is unsafe-to-inspect because inspecting personnel would be exposed to an imminent or potential danger as a consequence of complying with paragraph (h)(2)(i), (ii), or (iii) of this section; and

(ii) The owner or operator has a written plan that requires inspection of the equipment as frequently as practicable during safe-to-inspect times. Inspection is not required more than once annually.

(7) Any parts of the vapor collection system, closed-vent system, fixed roof,

cover, or enclosure that are designated, as described in $\S63.1367(f)(2)$, as difficult-to-inspect are exempt from the inspection requirements of paragraphs (h)(2)(i), (ii), and (iii)(A) of this section if:

(i) The owner or operator determines that the equipment cannot be inspected without elevating the inspecting personnel more than 2 meters above a support surface; and

(ii) The owner or operator has a written plan that requires inspection of the equipment at least once every 5 years.

(8) Records shall be maintained as specified in §63.1367(f).

(9) If a closed-vent system subject to this section is also subject to the equipment leak provisions of 63.1363, the owner or operator shall comply with the provisions of 63.1363 and is exempt from the requirements of this section.

(10) For any closed-vent system that is operated and maintained under negative pressure, the owner or operator is not required to comply with the requirements specified in paragraphs (h)(2) through (8) of this section.

[64 FR 33589, June 23, 1999, as amended at 67
FR 59352, Sept. 20, 2002; 68 FR 37358, June 23, 2003; 71 FR 20460, Apr. 20, 2006; 79 FR 17374, Mar. 27, 2014]

§63.1367 Recordkeeping requirements.

(a) Requirements of subpart A of this part. The owner or operator of an affected source shall comply with the recordkeeping requirements in subpart A of this part as specified in Table 1 of this subpart and in paragraphs (a)(1) through (5) of this section.

(1) Data retention. Each owner or operator of an affected source shall keep copies of all records and reports required by this subpart for at least 5 years, as specified in §63.10(b)(1) of subpart A of this part.

(2) Records of applicability determinations. The owner or operator of a stationary source that is not subject to this subpart shall keep a record of the applicability determination, as specified in 63.10(b)(3) of subpart A of this part.

(3) *Records of malfunctions*. (i) In the event that an affected unit fails to meet an applicable standard, record the number of failures. For each failure

record the date, time, and duration of each failure.

(ii) For each failure to meet an applicable standard, record and retain a list of the affected sources or equipment, an estimate of the quantity of each regulated pollutant emitted over any emission limit, and a description of the method used to estimate the emissions.

(iii) Record actions taken to minimize emissions in accordance with $\S63.1360(e)(4)$, and any corrective actions taken to return the affected unit to its normal or usual manner of operation.

(4) Recordkeeping requirements for sources with continuous monitoring systems. The owner or operator of an affected source who installs a continuous monitoring system to comply with the alternative standards in (3.1362(b)(6))or (c)(4) shall maintain records specified in (3.130(c))(1) through (14) of subpart A of this part.

(5) Application for approval of construction or reconstruction. For new affected sources, each owner or operator shall comply with the provisions regarding construction and reconstruction in §63.5 of subpart A of this part.

(b) Records of equipment operation. The owner or operator must keep the records specified in paragraphs (b)(1) through (11) of this section up-to-date and readily accessible.

(1) Each measurement of a control device operating parameter monitored in accordance with §63.1366 and each measurement of a treatment process parameter monitored in accordance with the provisions of §63.1362(d).

(2) For processes subject to §63.1362(g), records of consumption, production, and the rolling average values of the HAP and VOC factors.

(3) For each continuous monitoring system used to comply with the alternative standards in (3,1362(b))(6) and (c)(4), records documenting the completion of calibration checks and maintenance of the continuous monitoring systems.

(4) For processes in compliance with the 0.15 Mg/yr emission limit of $\S63.1362(b)(2)(i)$ or (b)(4)(i), daily records of the rolling annual calculations of uncontrolled emissions.

(5) For each bag leak detector used to monitor particulate HAP emissions

from a fabric filter, the owner or operator shall maintain records of any bag leak detection alarm, including the date and time, with a brief explanation of the cause of the alarm and the corrective action taken.

(6) The owner or operator of an affected source that complies with the standards for process vents, storage tanks, and wastewater systems shall maintain up-to-date, readily accessible records of the information specified in paragraphs (b)(6)(i) through (vii) of this section to document that HAP emissions or HAP loadings (for wastewater) are below the limits specified in $\S63.1362$:

(i) Except as specified in paragraph (b)(6)(ix) of this section, the initial calculations of uncontrolled and controlled emissions of gaseous organic HAP and HCl per batch for each process.

(ii) The wastewater concentrations and flow rates per POD and process.

(iii) The number of batches per year for each batch process.

(iv) The operating hours per year for continuous processes.

(v) The number of batches and the number of operating hours for processes that contain both batch and continuous operations.

(vi) The number of tank turnovers per year, if used in an emissions average or for determining applicability of a new PAI process unit.

(vii) A description of absolute or hypothetical peak-case operating conditions as determined using the procedures in §63.1365(b)(11).

(viii) Periods of planned routine maintenance as described in §63.1362(c)(5).

(ix) As an alternative to the records in paragraph (b)(6)(i) of this section, a record of the determination that the conditions in 63.1365(b)(11)(iii)(D)(1) or (2) are met.

(7) Daily schedule or log of each operating scenario updated daily or, at a minimum, each time a different operating scenario is put into operation.

(8) If the owner or operator elects to comply with the vapor balancing alternative in 63.1362(c)(6), the owner or operator must keep records of the DOT certification required by 63.1362(c)(6)(ii) and the pressure relief 40 CFR Ch. I (7–1–23 Edition)

vent setting and leak detection records specified in 63.1362(c)(6)(v).

(9) If the owner or operator elects to develop process unit groups, the owner or operator must keep records of the PAI and non-PAI process units in the process unit group, including records of the operating time for process units used to establish the process unit group. The owner or operator must also keep records of any redetermination of the primary product for the process unit group.

(10) All maintenance performed on the air pollution control equipment.

(11) If the owner or operator elects to comply with §63.1362(c) by installing a floating roof, the owner or operator must keep records of each inspection and seal gap measurement in accordance with §63.123(c) through (e) as applicable.

(c) Records of equipment leak detection and repair. The owner or operator of an affected source subject to the equipment leak standards in §63.1363 shall implement the recordkeeping requirements specified in §63.1363(g). All records shall be retained for a period of 5 years, in accordance with the requirements of §63.10(b)(1) of subpart A of this part.

(d) *Records of emissions averaging.* The owner or operator of an affected source that chooses to comply with the requirements of §63.1362(h) shall maintain up-to-date records of the following information:

(1) An Emissions Averaging Plan which shall include in the plan, for all emission points included in each of the emissions averages, the information listed in paragraphs (d)(1)(i) through (v) of this section.

(i) The identification of all emission points in each emissions average.

(ii) The values of all parameters needed for input to the emission debits and credits equations in §63.1365(h).

(iii) The calculations used to obtain the debits and credits.

(iv) The estimated values for all parameters required to be monitored under $\S63.1366(g)$ for each emission point included in an average. These parameter values, or as appropriate, limited ranges for parameter values, shall be specified as enforceable operating conditions for the operation of the

process, storage vessel, or waste management unit, as appropriate. Changes to the parameters must be reported as required by §63.1368(k).

(v) A statement that the compliance demonstration, monitoring, inspection, recordkeeping and reporting provisions in $\S63.1365(h)$, $\S63.1366(g)$, and $\S63.1368(k)$ that are applicable to each emission point in the emissions average will be implemented beginning on the date of compliance.

(2) The Emissions Averaging Plan shall demonstrate that the emissions from the emission points proposed to be included in the average will not result in greater hazard or, at the option of the operating permit authority, greater risk to human health or the environment than if the emission points were controlled according to the provisions in §63.1362(b) through (d).

(i) This demonstration of hazard or risk equivalency shall be made to the satisfaction of the operating permit authority.

(A) The Administrator may require an owner or operator to use specific methodologies and procedures for making a hazard or risk determination.

(B) The demonstration and approval of hazard or risk equivalency shall be made according to any guidance that the Administrator makes available for use or any other technically sound information or methods.

(ii) An Emissions Averaging Plan that does not demonstrate hazard or risk equivalency to the satisfaction of the Administrator shall not be approved. The Administrator may require such adjustments to the Emissions Averaging Plan as are necessary in order to ensure that the average will not result in greater hazard or risk to human health or the environment than would result if the emission points were controlled according to §63.1362(b) through (d).

(iii) A hazard or risk equivalency demonstration must satisfy the requirements specified in paragraphs (d)(2)(iii) (A) through (C) of this section.

(A) Be a quantitative, comparative chemical hazard or risk assessment;

(B) Account for differences between averaging and nonaveraging options in

chemical hazard or risk to human health or the environment; and

(C) Meet any requirements set by the Administrator for such demonstrations.

(3) Records as specified in paragraphs (a) and (b) of this section.

(4) A calculation of the debits and credits as specified in §63.1365(h) for the last quarter and the prior four quarters.

(e) The owner or operator of an affected source subject to the requirements for heat exchanger systems in $\S63.1362(g)$ shall retain the records as specified in $\S63.104(f)(1)(i)$ through (iv).

(f) Records of inspections. The owner or operator shall keep records specified in paragraphs (f)(1) through (6) of this section.

(1) Records identifying all parts of the vapor collection system, closedvent system, fixed roof, cover, or enclosure that are designated as unsafe to inspect in accordance with $\S63.1366(h)(6)$, an explanation of why the equipment is unsafe-to-inspect, and the plan for inspecting the equipment.

(2) Records identifying all parts of the vapor collection system, closedvent system, fixed roof, cover, or enclosure that are designated as difficult-toinspect in accordance with $\S63.1366(h)(7)$, an explanation of why the equipment is difficult-to-inspect, and the plan for inspecting the equipment.

(3) For each vapor collection system or closed-vent system that contains bypass lines that could divert a vent stream away from the control device and to the atmosphere, the owner or operator shall keep a record of the information specified in either paragraph (f)(3)(i) or (ii) of this section.

(i) Hourly records of whether the flow indicator specified under §63.1362(j)(1) was operating and whether a diversion was detected at any time during the hour, as well as records of the times and durations of all periods when the vent stream is diverted from the control device or the flow indicator is not operating.

(ii) Where a seal mechanism is used to comply with $\S63.1362(j)(2)$, hourly records of flow are not required. In such cases, the owner or operator shall record that the monthly visual inspection of the seals or closure mechanisms has been done and shall record the occurrence of all periods when the seal mechanism is broken, the bypass line valve position has changed, or the key for a lock-and-key type lock has been checked out, and records of any carseal that has broken.

(4) For each inspection conducted in accordance with 63.1366(h)(2) and (3) during which a leak is detected, a record of the information specified in paragraphs (f)(4)(i) through (ix) of this section.

(i) Identification of the leaking equipment.

(ii) The instrument identification numbers and operator name or initials, if the leak was detected using the procedures described in §63.1366(h)(3); or a record of that the leak was detected by sensory observations.

(iii) The date the leak was detected and the date of the first attempt to repair the leak.

(iv) Maximum instrument reading measured by the method specified in §63.1366(h)(4) after the leak is successfully repaired or determined to be nonrepairable.

(v) "Repair delayed" and the reason for the delay if a leak is not repaired within 15 calendar days after discovery of the leak.

(vi) The name, initials, or other form of identification of the owner or operator (or designee) whose decision it was that repair could not be effected without a shutdown.

(vii) The expected date of successful repair of the leak if a leak is not repaired within 15 calendar days.

(viii) Dates of shutdowns that occur while the equipment is unrepaired.

(ix) The date of successful repair of the leak.

(5) For each inspection conducted in accordance with \S 63.1366(h)(3) during which no leaks are detected, a record that the inspection was performed, the date of the inspection, and a statement that no leaks were detected.

(6) For each visual inspection conducted in accordance with §63.1366(h)(2)(i)(B) or (iii)(B) of this section during which no leaks are detected, a record that the inspection was performed, the date of the inspec40 CFR Ch. I (7–1–23 Edition)

tion, and a statement that no leaks were detected.

(g) Records of primary use. For a PAI process unit that is used to produce a given material for use as a PAI as well as for other purposes, the owner or operator shall keep records of the total production and the production for use as a PAI on a semiannual or more frequent basis if the use as a PAI is not the primary use.

[64 FR 33589, June 23, 1999, as amended at 67 FR 59353, Sept. 20, 2002; 71 FR 20460, Apr. 20, 2006; 79 FR 17374, Mar. 27, 2014]

§63.1368 Reporting requirements.

(a) The owner or operator of an affected source shall comply with the reporting requirements of paragraphs (b) through (l) of this section. The owner or operator shall also comply with applicable paragraphs of §§ 63.9 and 63.10 of subpart A of this part, as specified in Table 1 of this subpart.

(b) Initial notification. The owner or operator shall submit the applicable initial notification in accordance with §63.9(b) or (d) of subpart A of this part.

(c) Application for approval of construction or reconstruction. The owner or operator who is subject to (63.5(b)(3)) of subpart A of this part shall submit to the Administrator an application for approval of the construction of a new major source, the reconstruction of a major affected source, or the reconstruction of a major affected source subject to the standards. The application shall be prepared in accordance with (63.5(d)) of subpart A of this part.

(d) Notification of continuous monitoring system performance evaluation. An owner or operator who is required by the Administrator to conduct a performance evaluation for a continuous monitoring system that is used to comply with the alternative standard in (63.1362(b)(6) or (c)(4) shall notify the Administrator of the date of the performance evaluation as specified in (63.8(e)(2)) of subpart A of this part.

(e) Precompliance plan. The Precompliance plan shall be submitted at least 3 months prior to the compliance date of the standard. For new sources, the Precompliance plan shall be submitted to the Administrator with the application for approval of construction or reconstruction. The

Administrator shall have 90 days to approve or disapprove the Precompliance plan. The Precompliance plan shall be considered approved if the Administrator either approves it in writing, or fails to disapprove it in writing within the 90-day time period. The 90-day period shall begin when the Administrator receives the Precompliance plan. If the Precompliance plan is disapproved, the owner or operator must still be in compliance with the standard by the compliance date. To change any of the information submitted in the Precompliance plan or to submit a Precompliance plan for the first time after the compliance date, the owner or operator shall notify the Administrator at least 90 days before the planned change is to be implemented; the change shall be considered approved if the Administrator either approves the change in writing, or fails to disapprove the change in writing within 90 days of receipt of the change. The Precompliance plan shall include the information specified in paragraphs (e)(1) through (5) of this section.

(1) Requests for approval to use alternative monitoring parameters or requests to set monitoring parameters according to $\S63.1366(b)(4)$.

(2) Descriptions of the daily or per batch demonstrations to verify that control devices subject to §63.1366(b)(1)(i) are operating as designed.

(3) Data and rationale used to support the parametric monitoring level(s) that are set according to §63.1366(b)(3)(ii)(B).

(4) For owners and operators complying with the requirements of (63.1362(g)), the pollution prevention demonstration summary required in (63.1365(g)).

(5) Data and rationale used to support an engineering assessment to calculate uncontrolled emissions from process vents as required in $\S63.1365(c)(2)(ii)$.

(6) For fabric filters that are monitored with bag leak detectors, an operation and maintenance plan that describes proper operation and maintenance procedures, and a corrective action plan that describes corrective actions to be taken, and the timing of those actions, when the particulate matter concentration exceeds the setpoint and activates the alarm.

(f) Notification of compliance status report. The Notification of Compliance Status report required under (63.9(h))shall be submitted no later than 150 calendar days after the compliance date and shall include the information specified in paragraphs (f)(1) through (7) of this section.

(1) The results of any applicability determinations, emission calculations, or analyses used to identify and quantify HAP emissions from the affected source.

(2) The results of emissions profiles, performance tests, engineering analyses, design evaluations, or calculations used to demonstrate compliance. For performance tests, results should include descriptions of sampling and analysis procedures and quality assurance procedures.

(3) Descriptions of monitoring devices, monitoring frequencies, and the values of monitored parameters established during the initial compliance determinations, including data and calculations to support the levels established.

(4) Operating scenarios.

(5) Descriptions of absolute or hypothetical peak-case operating and/or testing conditions for control devices.

(6) Identification of emission points subject to overlapping requirements described in §63.1360(i) and the authority under which the owner or operator will comply, and identification of emission sources discharging to devices described by §63.1362(1).

(7) Anticipated periods of planned routine maintenance during which the owner or operator would not be in compliance with the provisions in $\S63.1362(c)(1)$ through (4).

(8) Percentage of total production from a PAI process unit that is anticipated to be produced for use as a PAI in the 3 years after either June 23, 1999 or startup, whichever is later.

(9) Records of the initial process units used to create each process unit group, if applicable.

(g) *Periodic reports.* The owner or operator shall prepare Periodic reports in accordance with paragraphs (g)(1) and (2) of this section and submit them to the Administrator.

§63.1368

(1) Submittal schedule. Except as provided in paragraphs (g)(1)(i) and (ii) of this section, the owner or operator shall submit Periodic reports semiannually. The first report shall be submitted no later than 240 days after the date the Notification of Compliance Status report is due and shall cover the 6-month period beginning on the date the Notification of Compliance Status report is due. Each subsequent Periodic report shall cover the 6-month period following the preceding period and shall be submitted no later than 60 days after the end of the applicable period.

(i) The Administrator may determine on a case-by-case basis that more frequent reporting is necessary to accurately assess the compliance status of the affected source.

(ii) Quarterly reports shall be submitted when the monitoring data are used to comply with the alternative standards in §63.1362(b)(6) or (c)(4) and the source experiences excess emissions. Once an affected source reports excess emissions, the affected source shall follow a quarterly reporting format until a request to reduce reporting frequency is approved. If an owner or operator submits a request to reduce the frequency of reporting, the provisions in §63.10(e)(3) (ii) and (iii) of subpart A of this part shall apply, except that the term "excess emissions and continuous monitoring system performance report and/or summary report" shall mean "Periodic report" for the purposes of this section.

(2) Content of periodic report. The owner or operator shall include the information in paragraphs (g)(2)(i) through (xii) of this section, as applicable.

(i) Each Periodic report must include the information in §63.10(e)(3)(vi)(A) through (M) of subpart A of this part, as applicable.

(ii) If the total duration of excess emissions, parameter exceedances, or excursions for the reporting period is 1 percent or greater of the total operating time for the reporting period, or the total continuous monitoring system downtime for the reporting period is 5 percent or greater of the total operating time for the reporting period, the Periodic report must include the

40 CFR Ch. I (7–1–23 Edition)

information in paragraphs (g)(2)(ii)(A) through (D) of this section.

(A) Monitoring data, including 15minute monitoring values as well as daily average values of monitored parameters, for all operating days when the average values were outside the ranges established in the Notification of Compliance Status report or operating permit.

(B) Duration of excursions, as defined in 63.1366(b)(7).

(C) Operating logs and operating scenarios for all operating days when the values are outside the levels established in the Notification of Compliance Status report or operating permit.

(D) When a continuous monitoring system is used, the information required in $\S63.10(c)(5)$ through (13) of subpart A of this part.

(iii) For each vapor collection system or closed vent system with a bypass line subject to $\S63.1362(j)(1)$, records required under $\S63.1366(f)$ of all periods when the vent stream is diverted from the control device through a bypass line. For each vapor collection system or closed vent system with a bypass line subject to $\S63.1362(j)(2)$, records required under $\S63.1362(j)(2)$, records required und

(iv) The information in paragraphs (g)(2)(iv)(A) through (D) of this section shall be stated in the Periodic report, when applicable.

(A) No excess emissions.

(B) No exceedances of a parameter.

(C) No excursions.

(D) No continuous monitoring system has been inoperative, out of control, repaired, or adjusted.

(v) For each storage vessel subject to control requirements:

(A) Actual periods of planned routine maintenance during the reporting period in which the control device does not meet the specifications of $\S63.1362(c)(5)$; and

(B) Anticipated periods of planned routine maintenance for the next reporting period.

(vi) For each PAI process unit that does not meet the definition of primary use, the percentage of the production

in the reporting period produced for use as a PAI.

(viii) Updates to the corrective action plan.

(ix) Records of process units added to each process unit group, if applicable.

(x) Records of redetermination of the primary product for a process unit group.

(xi) For each inspection conducted in accordance with 63.1366(h)(2) or (3) during which a leak is detected, the records specify in 63.1367(h)(4) must be included in the next Periodic report.

(xii) If the owner or operator elects to comply with the provisions of §63.1362(c) by installing a floating roof, the owner or operator shall submit the information specified in §63.122(d) through (f) as applicable. References to §63.152 in §63.122 shall not apply for the purposes of this subpart.

(h) Notification of process change. (1) Except as specified in paragraph (h)(2) of this section, whenever a process change is made, or any of the information submitted in the Notification of Compliance Status report changes, the owner or operator shall submit the information specified in paragraphs (h)(1)(i) through (iv) of this section with the next Periodic report required under paragraph (g) of this section. For the purposes of this section, a process change means the startup of a new process, as defined in §63.1361.

(i) A brief description of the process change;

(ii) A description of any modifications to standard procedures or quality assurance procedures;

(iii) Revisions to any of the information reported in the original Notification of Compliance Status report under paragraph (f) of this section; and

(iv) Information required by the Notification of Compliance Status report under paragraph (f) of this section for changes involving the addition of processes or equipment.

(2) The owner or operator must submit a report 60 days before the scheduled implementation date of either of the following:

(i) Any change in the activity covered by the Precompliance report.

(ii) A change in the status of a control device from small to large. (i) *Reports of malfunctions*. If a source fails to meet an applicable standard, report such events in the Periodic Report. Report the number of failures to meet an applicable standard. For each instance, report the date, time, and duration of each failure. For each failure the report must include a list of the affected sources or equipment, an estimate of the quantity of each regulated pollutant emitted over any emission limit, and a description of the method used to estimate the emissions.

(j) Reports of equipment leaks. The owner or operator of an affected source subject to the standards in §63.1363, shall implement the reporting requirements specified in §63.1363(h). Copies of all reports shall be retained as records for a period of 5 years, in accordance with the requirements of §63.10(b)(1) of subpart A of this part.

(k) Reports of emissions averaging. The owner or operator of an affected source that chooses to comply with the requirements of §63.1362(h) shall submit all information as specified in §63.1367(d) for all emission points included in the emissions average. The owner or operator shall also submit to the Administrator all information specified in paragraph (g) of this section for each emission point included in the emissions average.

(1) The reports shall also include the information listed in paragraphs (k)(1)(i) through (iv) of this section:

(i) Any changes to the processes, storage tanks, or waste management unit included in the average.

(ii) The calculation of the debits and credits for the reporting period.

(iii) Changes to the Emissions Averaging Plan which affect the calculation methodology of uncontrolled or controlled emissions or the hazard or risk equivalency determination.

(iv) Any changes to the parameters monitored according to §63.1366(g).

(2) Every second semiannual or fourth quarterly report, as appropriate, shall include the results according to $\S63.1367(d)(4)$ to demonstrate the emissions averaging provisions of $\S\$63.1362(h)$, 63.1365(h), 63.1366(g), and 63.1367(d) are satisfied.

(1) Reports of heat exchange systems. The owner or operator of an affected source subject to the requirements for heat exchange systems in $\S63.1362(f)$ shall submit information about any delay of repairs as specified in $\S63.104(f)(2)$ of subpart F of this part, except that when the phrase "periodic reports required by $\S63.152(c)$ of subpart G of this part" is referred to in $\S63.104(f)(2)$ of subpart F of this part, the periodic reports required in paragraph (g) of this section shall apply for the purposes of this subpart.

(m) Notification of performance test and test Plan. The owner or operator of an affected source shall notify the Administrator of the planned date of a performance test at least 60 days before the test in accordance with §63.7(b) of subpart A of this part. The owner or operator also must submit the test Plan required by §63.7(c) of subpart A of this part and the emission profile required by §63.1365(b)(11)(iii) with the notification of the performance test.

(n) Request for extension of compliance. The owner or operator may submit to the Administrator a request for an extension of compliance in accordance with $\S63.1364(a)(2)$.

(o) The owner or operator who submits an operating permit application before the date the Emissions Averaging Plan is due shall submit the information specified in paragraphs (o)(1) through (3) of this section with the operating permit application instead of the Emissions Averaging Plan.

(1) The information specified in §63.1367(d) for emission points included in the emissions average;

(2) The information specified in §63.9(h) of subpart A of this part, as applicable; and

(3) The information specified in paragraph (e) of this section, as applicable.

(p) Electronic reporting. Within 60 days after the date of completing each performance test (as defined in \S 63.2), the owner or operator must submit the results of the performance tests, including any associated fuel analyses, required by this subpart according to the methods specified in paragraphs (p)(1) or (2) of this section.

(1) For data collected using test methods supported by the EPA-provided software, the owner or operator shall submit the results of the performance test to the EPA by direct computer-to-computer electronic transfer 40 CFR Ch. I (7–1–23 Edition)

via EPA-provided software, unless otherwise approved by the Administrator. Owners or operators, who claim that some of the information being submitted for performance tests is confidential business information (CBI), must submit a complete file using EPA-provided software that includes information claimed to be CBI on a compact disk, flash drive, or other commonly used electronic storage media to the EPA. The electronic media must be clearly marked as CBI and mailed to U.S. EPA/OAPQS/CORE CBI Office, Attention: WebFIRE Administrator, MD C404-02, 4930 Old Page Rd., Durham, NC 27703. The same file with the CBI omitted must be submitted to the EPA by direct computerto-computer electronic transfer via EPA-provided software.

(2) For any performance test conducted using test methods that are not compatible with the EPA-provided software, the owner or operator shall submit the results of the performance test to the Administrator at the appropriate address listed in §60.4.

[64 FR 33589, June 23, 1999, as amended at 66 FR 58396, Nov. 21, 2001; 67 FR 59354, Sept. 20, 2002; 79 FR 17375, Mar. 27, 2014]

§63.1369 Implementation and enforcement.

(a) This subpart can be implemented and enforced by the U.S. EPA, or a delegated authority such as the applicable State, local, or Tribal agency. If the U.S. EPA Administrator has delegated authority to a State, local, or Tribal agency, then that agency, in addition to the U.S. EPA, has the authority to implement and enforce this subpart. Contact the applicable U.S. EPA Regional Office to find out if implementation and enforcement of this subpart is delegated to a State, local, or Tribal agency.

(b) In delegating implementation and enforcement authority of this subpart to a State, local, or Tribal agency under subpart E of this part, the authorities contained in paragraph (c) of this section are retained by the Administrator of U.S. EPA and cannot be transferred to the State, local, or Tribal agency.

(c) The authorities that cannot be delegated to State, local, or Tribal

Pt. 63, Subpt. MMM, Table 1

agencies are as specified in paragraphs (c)(1) through (4) of this section.

(1) Approval of alternatives to the requirements in §§ 63.1360 and 63.1362 through 63.1364. Where these standards reference another subpart, the cited provisions will be delegated according to the delegation provisions of the referenced subpart. Where these standards reference another subpart and modify the requirements, the requirements shall be modified as described in this subpart. Delegation of the modified requirements will also occur according to the delegation provisions of the referenced subpart.

(2) Approval of major alternatives to test methods for under 63.7(e)(2)(ii) and (f), as defined in 63.90, and as required in this subpart.

(3) Approval of major alternatives to monitoring under §63.8(f), as defined in §63.90, and as required in this subpart.

(4) Approval of major alternatives to recordkeeping and reporting under §63.10(f), as defined in §63.90, and as required in this subpart.

[68 FR 37358, June 23, 2003]

TABLE 1 TO	SUBPART	MMM 0	F PART 63-	-General	PROVISIONS	APPLICABILITY T	0
			SUBPAI	RT MMM			

Reference to subpart A	Applies to subpart MMM	Explanation
§63.1(a)(1)	Yes	Additional terms are defined in §63.1361.
§63.1(a)(2)–(3) §63.1(a)(4)	Yes	Subpart MMM (this table) specifies applicability of each paragraph in subpart A to subpart MMM
§63.1(a)(5) §63.1(a)(6)–(7)	N/A Yes	Reserved.
§63.1(a)(8) §63.1(a)(9)	No N/A	Discusses State programs. Reserved.
§63.1(a)(10)–(14) §63.1(b)(1)	Yes No	§63.1360 specifies applicability.
§63.1(c)(1)	Yes	Subpart MMM (this table) specifies the applicability of each paragraph in sub- part A to sources subject to subpart MMM
§63.1(c)(2) §63.1(c)(3)	No N/A	Area sources are not subject to subpart MMM. Reserved.
§63.1(c)(4)–(5) §63.1(c)(6)	Yes Yes.	
§63.1(d) §63.1(e)	N/A Yes	Reserved.
§63.2	Yes	Additional terms are defined in §63.1361; when overlap between subparts A and MMM occurs, subpart MMM takes precedence.
§63.3 §63.4(a)(1)–(3)	Yes Yes	Other units used in subpart MMM are defined in that subpart.
§63.4(a)(4) §63.4(a)(5)–(c) §63.5(a)	Yes Yes	reserved.
§ 63.5(b)(1)	Yes	"stationary source" in §63.5(a)(1) of subpart A.
§63.5(b)(2) §63.5(b)(3)–(5)	N/A Yes	Reserved.
§63.5(b)(6)	No	§63.1360(g) specifies requirements for determining applicability of added PAI equipment.
§63.5(c) §63.5(d)–(e)	N/A Yes	Reserved.
§63.5(f)(1)	Yes	Except "affected source" shall apply instead of "source" in §63.5(f)(1) of sub- part A.
§63.5(1)(2) §63.6(a)	Yes	8.62 1264 aposition compliance dates
§63.6(b)(1)–(2) §63.6(b)(3)–(4)	Yes Ves	§ 03. 1304 specifies compliance dates.
§63.6(b)(6) §63.6(b)(6)	N/A Yes	Reserved.
§63.6(c)(1)–(2)	Yes	Except "affected source" shall apply instead of "source" in §63.6(c)(1)-(2) of subpart A.
§63.6(c)(3)–(4) §63.6(c)(5)	N/A Yes	Reserved.
§63.6(d) §63.6(e)(1)(i)	N/A No	Reserved. See § 63.1360(e)(4) for general duty requirement.

Pt. 63, Subpt. MMM, Table 1

40 CFR Ch. I (7-1-23 Edition)

Reference to subpart A	Applies to subpart MMM	Explanation
§ 63.6(e)(1)(ii) § 63.6(e)(1)(iii) § 63.6(e)(3)	No Yes No	
§63.6(f)(1)	No	
§ 63.6(f)(2)–(3)	Yes	An alternative standard has been prepared, between effected services will be a
§63.6(g)	Yes	An alternative standard has been proposed; nowever, affected sources will have
8636(b)	No	Subpart MMM does not contain any onacity or visible emissions standards
863 6(i)(1)	Ves	Cubpart willing does not contain any opacity of visible christions standards.
§63.6(i)(2)	Yes	Except "affected source" shall apply instead of "source" in § 63.6(i)(2)(i) and (ii)
30010(1)(2)		of subpart A
§63.6(i)(3)–(14)	Yes	
§ 63.6(i)(15)	N/A	Reserved.
§63.6(i)(16)	Yes	
§63.6(j)	Yes	
§63.7(a)(1)	Yes	
§63.7(a)(2)(I)–(VI)	Yes	§ 63.1368 specifies that test results must be submitted in the Notification of
\$ 62 7(c)(2)(vii) (viii)	NI/A	Lompliance Status due 150 days after the compliance date.
$863.7(a)(2)(vii) - (viii) \dots $	N/A	Reserveu.
863 7(d)	Yes	Except "affected source" shall apply instead of "source" in §63.7(d) of subpart
3 (u)		A.
§63.7(e)(1)	No	See § 63.1365(b).
§63.7(e)(2)	Yes	
§63.7(e)(3)	Yes	Except § 63.1365 specifies less than 3 runs for certain tests.
§63.7(e)(4)	Yes.	
§63.7(f)	Yes	
§63.7(g)(1)	Yes	Except § 63.1368(a) specifies that the results of the performance test be sub- mitted with the Notification of □ompliance Status report
§63.7(g)(2)	N/A	Reserved.
§63.7(g)(3)	Yes	
§63.7(h)	Yes	
§63.8(a)(1)–(2)	Yes	Deserved
§ 63.8(a)(3)	N/A	Reserved.
863.8(b)(1)	Yes	
§ 63 8(b)(2)	No	8.63 1366 specifies □MS requirements
§ 63.8(b)(3)	Yes	2.00.1000 obcause Ture redarionario.
§63.8(c)(1)(i)	No	
§63.8(c)(1)(ii)	Yes	
§63.8(c)(1)(iii)	No	
§63.8(c)(2)–(3)	Yes	
§ 63.8(c)(4)	No	§ 63.1366 specifies monitoring frequencies.
§ 63.8(C)(5)–(8)	NO	Execut the last contained of $\S \in \mathbb{C}(d)(2)$, which shall be replaced with "The pro-
903.0(d)–(1)(3)	res	gram of corrective action should be included in the plan required under
§63.8(f)(4)	Yes	§63.8(d)(2)." for the purposes of this subpart. Except §63.1368(b) specifies that requests may also be included in the
• • • • • • • • • • • • • • • • • • • •		Precompliance report.
§63.8(f)(5)	Yes	
§63.8(f)(6)	No	Subpart MMM does not require □EM's.
§63.8(g)	No	§ 63.1366 specifies data reduction procedures.
8 63 9(a)-(a)	No	
\$ 63 0(f)	No	Subpart MMM does not contain onacity and visible emission standards
\$63.9(a)	No	Cubpart willing does not contain opacity and visible christion standards.
§63.9(h)(1)	Yes	
§63.9(h)(2)(i)	Yes	Except §63.1368(a)(1) specifies additional information to include in the Notifica-
0 ()()()		tion of □ompliance Status report.
§63.9(h)(2)(ii)	No	§63.1368 specifies the Notification of □ompliance Status report is to be sub- mitted within 150 days after the compliance date.
§63.9(h)(3)	Yes	
§63.9(h)(4)	N/A	Reserved.
§63.9(h)(5)–(6)	Yes	
63.9(i)	Yes.	
§63.9(j)	Yes	or change in major source status only, §63.1368(h) specifies procedures for other notification of changes.
§63.9(k)	Yes	Only as specified in §63.9(j).
§63.10(a)–(b)(1)	Yes	
§ 63.10(b)(2)	No	§ 63.1367 specifies recordkeeping requirements.
§ 63.10(b)(3)	Yes	
8 63 10(c)(1)-(14)	res	
803.10(0)(15)		1

Pt. 63, Subpt. MMM, Table 2

Reference to subpart A	Applies to subpart MMM	Explanation
§63.10(d)(1)	Yes	
§63.10(d)(2)	Yes	
§63.10(d)(3)	No	Subpart MMM does not include opacity and visible emission standards.
§63.10(d)(4)	Yes	
§63.10(d)(5)	No	See §63.1368(i) for malfunction reporting requirements.
§63.10(e)(1)-(2)(i)	Yes	
§63.10(e)(2)(ii)	No	Subpart MMM does not include opacity monitoring requirements.
§63.10(e)(3)	Yes	
§63.10(e)(4)	No	Subpart MMM does not include opacity monitoring requirements.
§63.10(f)	Yes	
§63.11-§63.15	Yes.	

[64 FR 33589, June 23, 1999, as amended at 67 FR 59355, Sept. 20, 2002; 79 FR 17375, Mar. 27, 2014; 85 FR 73899, Nov. 19, 2020]

TABLE 2 TO SUBPART	MMM 0	f Part 63—	-Standards	FOR NE	W AND	EXISTING	PAI
		Sou	RCES				

Emission source	Applicability	Requirement
Process vents	Existing Processes having uncontrolled organic □AP emissions ≥0.15 Mg/yr. Processes having uncontrolled □I and chlo- rine emissions ≥6.8 Mg/yr.	 90 for organic □AP per process or to outlet concentration of ≤20 ppmv □O 94 for □l and chlorine per process or to outlet □l and chlorine concentration of ≤20 ppmv.
	Individual process vents meeting flow and mass emissions criteria that have gaseous organic □AP emissions controlled to less than 90□ on or after November 10, 1997. New□	98□ gaseous organic □AP control per vent or ≤20 ppmv □O□ outlet limit.
	Processes having uncontrolled organic □AP emissions ≥0.15 Mg/yr.	98 for organic □AP per process or ≤20 ppmv □O□.
	Processes having uncontrolled □□I and chlo- rine emissions ≥6.8 Mg/yr and □191 Mg/yr.	94□ for □□ and chlorine per process or to out- let concentration of ≤20 ppmv □□I and chlo- rine.
	Processes having uncontrolled □□I and chlo- rine emissions ≥191 Mg/yr.	99□ for □□ and chlorine per process or to out- let concentration of ≤20 ppmv □□ and chlo- rine.
Storage vessels	Existing⊡≥75 m³ capacity and vapor pressure ≥3.45 kPa.	Install a floating roof, reduce □AP by 95□ per vessel, or to outlet concentration of ≤20 ppmv □O□.
	New $\supseteq \ge 38 \text{ m}^3$ capacity and vapor pressure $\ge 16.5 \text{ kPa}$.	Same as for existing sources.
□ astewater ^a	≥75 m ³ capacity and vapor pressure ≥3.45 kPa Existing Process wastewater with ≥10,000 ppmw □able 9 compounds at all of min, and maintenance wastewater with □AP load ≥5.3 Mg per discharge event.	Same as for existing sources. Reduce concentration of total ⊡able 9 com- pounds to ⊡50 ppmw (or other options).
	Same criteria as for existing sources	Reduce concentration of total able 9 compounds to 50 ppmw (or other options).
	□otal □AP load in wastewater POD streams ≥2,100 Mg/yr	99□ reduction of □able 9 compounds from all streams.
Equipment leaks	Subpart	Subpart with minor changes, including moni- toring frequencies consistent with the pro- posed AR.
Product dryers and bag dumps.	Dryers used to dry PAI that is also a a AP, and bag dumps used to introduce feedstock that is a solid and a AP.	Particulate matter concentration not to exceed 0.01 gr/dscf.
□eat exchange systems	Each heat exchange system used to cool proc- ess equipment in PAI manufacturing oper- ations.	Monitoring and leak repair program as in \Box ON.

^a \square able 9 is listed in the appendix to subpart \square of 40 $\square\square$ R part 63.

Pt. 63, Subpt. MMM, Table 3

40 CFR Ch. I (7-1-23 Edition)

TABLE 3 TO SUBPART MMM OF PART 63—MONITORING REQUIREMENTS FOR CONTROL	
DEVICES A	

□ontrol device	Monitoring equipment re- quired	Parameters to be monitored	□requency
All control devices	 □low indicator installed at all bypass lines to the at- mosphere and equipped with continuous recorder or. 	1. Presence of flow diverted from the control device to the atmosphere or.	Ourly records of whether the flow indicator was operating and whether a diversion was detected at any time
	2. □alves sealed closed with car-seal or lock-and-key configuration	2. Monthly inspections of sealed valves.	during each hour. Monthly.
Scrubber	□quid flow rate or pressure drop mounting device. Also a p □ monitor if the scrub- ber is used to control acid emissions	 □iquid flow rate into or out of the scrubber or the pres- sure drop across the scrub- ber 	1. Every 15 minutes.
		2. p□ of effluent scrubber liq- uid.	2. Once a day.
⊡hermal incinerator	□emperature monitoring de- vice installed in firebox or in ductwork immediately downstream of firebox ^b	□irebox temperature	Every 15 minutes.
□atalytic incinerator	emperature monitoring de- vice installed in gas stream immediately before and after catalyst bed	□emperature difference across catalyst bed.	Every 15 minutes.
□are	□eat sensing device installed	Presence of a flame at the pilot light	Every 15 minutes.
□oiler or process heater □44 megawatts and vent stream is not mixed with the primary fuel.	 Demperature monitoring de- vice installed in firebox^b. 	□ombustion temperature	Every 15 minutes.
□ondenser	□emperature monitoring de- vice installed at condenser exit	□ondenser exit (product side) temperature.	Every 15 minutes.
□arbon adsorber (nonregen-	None	Operating time since last re-	N/A.
arbon adsorber (regenera- tive).	Stream flow monitoring de- vice, and.	 Total regeneration stream mass or volumetric flow during carbon bed regen- eration cycle(s). 	 □or each regeneration cycle, record the total re- generation stream mass or volumetric flow.
	□arbon bed temperature monitoring device.	2. □emperature of carbon bed after regeneration.	 □or each regeneration cycle, record the maximum
		3. □emperature of carbon bed within 15 minutes of com- pleting any cooling cycle(s).	 3. □ ithin 15 minutes of completing any cooling cycle, record the carbon bed temperature.
		 Operating time since end of last regeneration. □heck for bed poisoning 	 Operating time to be based on worst-case conditions. Yearly.

^a As an alternative to the monitoring requirements specified in this table, the owner or operator may use a ⊡EM meeting the requirements of Performance Specifications 8 or 9 of appendix □ of part 60 to monitor □O□ every 15 minutes. ^b Monitor may be installed in the firebox or in the ductwork immediately downstream of the firebox before any substantial heat exchange is encountered.

TABLE 4 TO S	Subpart M	MM OF	PART 6	63—Con	vtrol R	REQUIRE	EMENTS	FOR	ITEMS	OF
	EQUIPMEN	T THAT	Meet	THE CI	RITERIA	OF §63	.1362(k)			

Item of equipment	□ontrol requirement ^a
1. Drain or drain hub	 (a) □ightly fitting solid cover (□□S□); or (b) □□S□ with a vent to either a process, or to a control device meeting the requirements of § 63.139(c); or (b) □ □ ot a control device meeting the process of borries to protect displaying from wind
2. Manhole ^b	 (c) a let sea with subneged discharge of barrier to protect discharge norm wind. (a) □SC; or (b) □SC with a vent to either a process or to a control device meeting the requirements of §63.139(c); or (c) If the item is vented to the atmosphere, use a □SC with a properly operating water seal at the entrance or exit to the item to restrict ventilation in the collection system. □he vent pipe shall be at least 90 cm in length and not exceeding 10.2 cm in nominal inside diameter.

§63.1381

Item of equipment	□ontrol requirement ^a
3. 🗆 ft station	 (a) □□S□; or (b) □□S□ with a vent to either a process, or to a control device meeting the requirements of §63.139(c); or (c) If the lift station is vented to the atmosphere, use a □□S□ with a properly operating water seal at the entrance or exit to the item to restrict ventilation in the collection system. □he vent pipe shall be at least 90 cm in length and not exceeding 10.2 cm in nominal inside diameter. □he lift station shall be level controlled to minimi⊡e changes in the liquid level.
4. □rench	 (a) □□S□; or (b) □□S□ with a vent to either a process, or to a control device meeting the requirements of §63.139(c); or (c) If the item is vented to the atmosphere, use a □□S□ with a properly operating water seal at the entrance or exit to the item to restrict ventilation in the collection system. □he vent pipe shall be at least 90 cm in length and not exceeding 10.2 cm in nominal inside diameter
5. Pipe 6. Oil/water separator	 Each pipe shall have outsided. (a) Equip with a fixed roof and route vapors to a process, or equip with a closed-vent system that routes vapors to a control device meeting the requirements of §63.139(c); or (b) Equip with a floating roof that meets the equipment specifications of §60.693
7. □ank	(a)(1)(i), (a)(1)(ii), (a)(2), (a)(3), and (a)(4). Maintain a fixed roof and consider vents as process vents. ^c

a □ here a tightly fitting solid cover is required, it shall be maintained with no visible gaps or openings, except during periods of sampling, inspection, or maintenance.
 b Manhole includes sumps and other points of access to a conveyance system.
 c A fixed roof may have openings necessary for proper venting of the tank, such as pressure/vacuum vent, j-pipe vent.

[67 FR 59355, Sept. 20, 2002]

Subpart NNN—National Emission Standards for Hazardous Air Pollutants for Wool Fiberglass Manufacturing

SOURCE: 64 FR 31709, June 14, 1999, unless otherwise noted.

§63.1380 Applicability.

(a) Except as provided in paragraphs (b) and (c) of this section, the requirements of this subpart apply to the owner or operator of each wool fiberglass manufacturing facility that is a major source or is located at a facility that is a major source.

(b) The requirements of this subpart apply to emissions of hazardous air pollutants (HAPs), as measured according to the methods and procedures in this subpart, emitted from the following new and existing sources at a wool fiberglass manufacturing facility subject to this subpart:

(1) Each new and existing glass-melting furnace located at a wool fiberglass manufacturing facility;

(2) Each new and existing rotary spin wool fiberglass manufacturing line producing a bonded wool fiberglass building insulation product; and

(3) Each new and existing flame attenuation wool fiberglass manufacturing line producing a bonded product.

(c) The requirements of this subpart do not apply to a wool fiberglass manufacturing facility that the owner or operator demonstrates to the Administrator is not a major source as defined in §63.2.

(d) The provisions of this part 63, subpart A that apply and those that do not apply to this subpart are specified in Table 1 of this subpart.

[64 FR 31709, June 14, 1999, as amended at 80 FR 45334, July 29, 2015]

§63.1381 Definitions.

Terms used in this subpart are defined in the Clean Air Act, in §63.2, or in this section as follows:

Aerospace and air filtration products means bonded wool fiberglass insulation manufactured for the thermal and acoustical insulation of aircraft and/or the air filtration markets. For the purposes of this subpart, a production line that manufactures these types of products for 75 percent or more of the line's annual operating hours is considered to be an aerospace and air filtration products line.

Bag leak detection system means systems that include, but are not limited Appendix C

AUTHENTICATED U.S. GOVERNMENT INFORMATION

§60.114a Alternative means of emission limitation.

(a) If, in the Administrator's judgment, an alternative means of emission limitation will achieve a reduction in emissions at least equivalent to the reduction in emissions achieved by any requirement in §60.112a, the Administrator will publish in the FEDERAL REGISTER a notice permitting the use of the alternative means for purposes of compliance with that requirement.

(b) Any notice under paragraph (a) of this section will be published only after notice and an opportunity for a hearing.

(c) Any person seeking permission under this section shall submit to the Administrator a written application including:

(1) An actual emissions test that uses a full-sized or scale-model storage vessel that accurately collects and measures all VOC emissions from a given control device and that accurately simulates wind and accounts for other emission variables such as temperature and barometric pressure.

(2) An engineering evaluation that the Administrator determines is an accurate method of determining equivalence.

(d) The Administrator may condition the permission on requirements that may be necessary to ensure operation and maintenance to achieve the same emissions reduction as specified in §60.112a.

(e) The primary vapor-mounted seal in the "Volume-Maximizing Seal" manufactured by R.F.I. Services Corporation is approved as equivalent to the vapor-mounted seal required by \S 60.112a(a)(1)(i) and must meet the gap criteria specified in \S 60.112a(a)(1)(i)(B). There shall be no gaps between the tank wall and any secondary seal used in conjunction with the primary seal in the "Volume-Maximizing Seal".

[52 FR 11429, Apr. 8, 1987]

§60.115a Monitoring of operations.

(a) Except as provided in paragraph (d) of this section, the owner or operator subject to this subpart shall maintain a record of the petroleum liquid stored, the period of storage, and the maximum true vapor pressure of that liquid during the respective storage period.

(b) Available data on the typical Reid vapor pressure and the maximum expected storage temperature of the stored product may be used to determine the maximum true vapor pressure from nomographs contained in API Bulletin 2517, unless the Administrator specifically requests that the liquid be sampled, the actual storage temperature determined, and the Reid vapor pressure determined from the sample(s).

(c) The true vapor pressure of each type of crude oil with a Reid vapor pressure less than 13.8 kPa (2.0 psia) or whose physical properties preclude determination by the recommended method is to be determined from available data and recorded if the estimated true vapor pressure is greater than 6.9 kPa (1.0 psia).

(d) The following are exempt from the requirements of this section:

(1) Each owner or operator of each storage vessel storing a petroleum liquid with a Reid vapor pressure of less than 6.9 kPa (1.0 psia) provided the maximum true vapor pressure does not exceed 6.9 kPa (1.0 psia).

(2) The owner or operator of each storage vessel equipped with a vapor recovery and return or disposal system in accordance with the requirements of (0,112a(a)(3)) and (b), or a closed vent system and control device meeting the specifications of 40 CFR (65.42(b)(4), (b)(5), or (c)).

[45 FR 23379, Apr. 4, 1980, as amended at 65 FR 78275, Dec. 14, 2000]

Subpart Kb—Standards of Performance for Volatile Organic Liquid Storage Vessels (Including Petroleum Liquid Storage Vessels) for Which Construction, Reconstruction, or Modification Commenced After July 23, 1984

SOURCE: 52 FR 11429, Apr. 8, 1987, unless otherwise noted.

§60.110b Applicability and designation of affected facility.

(a) Except as provided in paragraph (b) of this section, the affected facility

to which this subpart applies is each storage vessel with a capacity greater than or equal to 75 cubic meters (m^3) that is used to store volatile organic liquids (VOL) for which construction, reconstruction, or modification is commenced after July 23, 1984.

(b) This subpart does not apply to storage vessels with a capacity greater than or equal to 151 m^3 storing a liquid with a maximum true vapor pressure less than 3.5 kilopascals (kPa) or with a capacity greater than or equal to 75 m³ but less than 151 m³ storing a liquid with a maximum true vapor pressure less than 15.0 kPa.

(c) [Reserved]

(d) This subpart does not apply to the following:

(1) Vessels at coke oven by-product plants.

(2) Pressure vessels designed to operate in excess of 204.9 kPa and without emissions to the atmosphere.

(3) Vessels permanently attached to mobile vehicles such as trucks, railcars, barges, or ships.

(4) Vessels with a design capacity less than or equal to 1,589.874 m³ used for petroleum or condensate stored, processed, or treated prior to custody transfer.

(5) Vessels located at bulk gasoline plants.

(6) Storage vessels located at gasoline service stations.

 $\left(7\right)$ Vessels used to store be verage alcohol.

(8) Vessels subject to subpart GGGG of 40 CFR part 63.

(e) Alternative means of compliance—(1) Option to comply with part 65. Owners or operators may choose to comply with 40 CFR part 65, subpart C, to satisfy the requirements of \S 60.112b through 60.117b for storage vessels that are subject to this subpart that meet the specifications in paragraphs (e)(1)(i) and (ii) of this section. When choosing to comply with 40 CFR part 65, subpart C, the monitoring requirements of \S 60.116b(c), (e), (f)(1), and (g) still apply. Other provisions applying to owners or operators who choose to comply with 40 CFR part 65 are provided in 40 CFR 65.1.

(i) A storage vessel with a design capacity greater than or equal to 151 m^3 containing a VOL that, as stored, has a

40 CFR Ch. I (7–1–23 Edition)

maximum true vapor pressure equal to or greater than 5.2 kPa; or

(ii) A storage vessel with a design capacity greater than 75 m^3 but less than 151 m^3 containing a VOL that, as stored, has a maximum true vapor pressure equal to or greater than 27.6 kPa.

(2) Part 60, subpart A. Owners or operators who choose to comply with 40 CFR part 65, subpart C, must also comply with §§ 60.1, 60.2, 60.5, 60.6, 60.7(a)(1) and (4), 60.14, 60.15, and 60.16 for those storage vessels. All sections and paragraphs of subpart A of this part that are not mentioned in this paragraph (e)(2) do not apply to owners or operators of storage vessels complying with 40 CFR part 65, subpart C, except that provisions required to be met prior to implementing 40 CFR part 65 still apply. Owners and operators who choose to comply with 40 CFR part 65, subpart C, must comply with 40 CFR part 65, subpart A.

(3) Internal floating roof report. If an owner or operator installs an internal floating roof and, at initial startup, chooses to comply with 40 CFR part 65, subpart C, a report shall be furnished to the Administrator stating that the control equipment meets the specifications of 40 CFR 65.43. This report shall be an attachment to the notification required by 40 CFR 65.5(b).

(4) External floating roof report. If an owner or operator installs an external floating roof and, at initial startup, chooses to comply with 40 CFR part 65, subpart C, a report shall be furnished to the Administrator stating that the control equipment meets the specifications of 40 CFR 65.44. This report shall be an attachment to the notification required by 40 CFR 65.5(b).

(5) Option to comply with part 63, subpart WW, of this chapter. Except as specified in paragraphs (e)(5)(i) through (iv) of this section, owners or operators may choose to comply with 40 CFR part 63, subpart WW, to satisfy the requirements of §§ 60.112b through 60.117b for storage vessels either with a design capacity greater than or equal to 151 m³ containing a VOL that, as stored, has a maximum true vapor pressure equal to or greater than 5.2 kPa but less than 76.6 kPa, or with a design capacity greater than or equal to 75 m³

but less than 151 m^3 containing a VOL that, as stored, has a maximum true vapor pressure equal to or greater than 27.6 kPa but less than 76.6 kPa.

(i) The general provisions in subpart A of this part apply instead of the general provisions in subpart A of part 63 of this chapter.

(ii) Where terms are defined in both this subpart and 40 CFR part 63, subpart WW, the definitions in this subpart apply.

(iii) Owners or operators who choose to comply with 40 CFR part 63, subpart WW, also must comply with the monitoring requirements of 60.116b(a), (c), (e), and (f)(1), except as specified in paragraphs (e)(5)(iii)(A) through (C) of this section.

(A) The reference to all records applies only to the records required by §60.116b(c);

(B) The reference to §60.116b(b) does not apply; and

(C) The reference to §60.116b(g) does not apply.

(iv) Owners or operators who choose to comply with 40 CFR part 63, subpart WW, must also keep records and furnish reports as specified in paragraphs (e)(5)(iv)(A) through (F) of this section.

(A) For each affected facility, the owner or operator must notify the Administrator at least 30 days before the first inspection is conducted under 40 CFR part 63, subpart WW. After this notification is submitted to the Administrator, the owner or operator must continue to comply with the alternative standard described in this paragraph (e)(5) until the owner or operator submits another notification to the Administrator indicating the affected facility is using the requirements of §§60.112b through 60.117b instead of the alternative standard described in this paragraph (e)(5). The compliance schedule for events does not reset upon switching between compliance with this subpart and 40 CFR part 63, subpart WW.

(B) Keep a record of each affected facility using the alternative standard described in this paragraph (e)(5) when conducting an inspection required by §63.1063(c)(1) of this chapter.

(C) Keep a record of each affected facility using the alternative standard described in this paragraph (e)(5) when conducting an inspection required by §63.1063(c)(2) of this chapter.

(D) Copies of all records and reports kept pursuant to $\S60.115b(a)$ and (b) that have not met the 2-year record retention required by the introductory text of $\S60.115b$ must be kept for an additional 2 years after the date of submittal of the inspection notification specified in paragraph (e)(5)(iv)(A) of this section, indicating the affected facility is using the requirements of 40 CFR part 63, subpart WW.

(E) Copies of all records and reports kept pursuant to $\S63.1065$ of this chapter that have not met the 5-year record retention required by the introductory text of $\S63.1065$ must be kept for an additional 5 years after the date of submittal of the notification specified in paragraph (e)(5)(iv)(A) of this section, indicating the affected facility is using the requirements of \S 60.112b through 60.117b.

(F) The following exceptions to the reporting requirements of §63.1066 of this chapter apply:

(1) The notification of initial startup required under $\S63.1066(a)(1)$ and (2) of this chapter must be submitted as an attachment to the notification required by $\S960.7(a)(3)$ and 60.115b(a)(1);

(2) The reference in $\S63.1066(b)(2)$ of this chapter to periodic reports "when inspection failures occur" means to submit inspections results within 60 days of the initial gap measurements required by $\S63.1063(c)(2)(i)$ of this chapter and within 30 days of all other inspections required by $\S63.1063(c)(1)$ and (2) of this chapter.

[52 FR 11429, Apr. 8, 1987, as amended at 54
FR 32973, Aug. 11, 1989; 65 FR 78275, Dec. 14, 2000; 68 FR 59332, Oct. 15, 2003; 86 FR 5019, Jan. 19, 2021]

§60.111b Definitions.

Terms used in this subpart are defined in the Act, in subpart A of this part, or in this subpart as follows:

Bulk gasoline plant means any gasoline distribution facility that has a gasoline throughput less than or equal to 75,700 liters per day. Gasoline throughput shall be the maximum calculated design throughput as may be limited by compliance with an enforceable condition under Federal requirement or Federal, State or local law, and discoverable by the Administrator and any other person.

Condensate means hydrocarbon liquid separated from natural gas that condenses due to changes in the temperature or pressure, or both, and remains liquid at standard conditions.

Custody transfer means the transfer of produced petroleum and/or condensate, after processing and/or treatment in the producing operations, from storage vessels or automatic transfer facilities to pipelines or any other forms of transportation.

Fill means the introduction of VOL into a storage vessel but not necessarily to complete capacity.

Gasoline service station means any site where gasoline is dispensed to motor vehicle fuel tanks from stationary storage tanks.

Maximum true vapor pressure means the equilibrium partial pressure exerted by the volatile organic compounds (as defined in 40 CFR 51.100) in the stored VOL at the temperature equal to the highest calendar-month average of the VOL storage temperature for VOL's stored above or below the ambient temperature or at the local maximum monthly average temperature as reported by the National Weather Service for VOL's stored at the ambient temperature, as determined:

(1) In accordance with methods described in American Petroleum institute Bulletin 2517, Evaporation Loss From External Floating Roof Tanks, (incorporated by reference—see §60.17); or

(2) As obtained from standard reference texts; or

(3) As determined by ASTM D2879-83,96, or 97 (incorporated by reference—see §60.17);

(4) Any other method approved by the Administrator.

Petroleum means the crude oil removed from the earth and the oils derived from tar sands, shale, and coal.

Petroleum liquids means petroleum, condensate, and any finished or intermediate products manufactured in a petroleum refinery.

Process tank means a tank that is used within a process (including a solvent or raw material recovery process) to collect material discharged from a 40 CFR Ch. I (7-1-23 Edition)

feedstock storage vessel or equipment within the process before the material is transferred to other equipment within the process, to a product or by-product storage vessel, or to a vessel used to store recovered solvent or raw material. In many process tanks, unit operations such as reactions and blending are conducted. Other process tanks, such as surge control vessels and bottoms receivers, however, may not involve unit operations.

Reid vapor pressure means the absolute vapor pressure of volatile crude oil and volatile nonviscous petroleum liquids except liquified petroleum gases, as determined by ASTM D323-82 or 94 (incorporated by reference—see §60.17).

Storage vessel means each tank, reservoir, or container used for the storage of volatile organic liquids but does not include:

(1) Frames, housing, auxiliary supports, or other components that are not directly involved in the containment of liquids or vapors;

(2) Subsurface caverns or porous rock reservoirs; or

(3) Process tanks.

Volatile organic liquid (VOL) means any organic liquid which can emit volatile organic compounds (as defined in 40 CFR 51.100) into the atmosphere.

Waste means any liquid resulting from industrial, commercial, mining or agricultural operations, or from community activities that is discarded or is being accumulated, stored, or physically, chemically, or biologically treated prior to being discarded or recycled.

[52 FR 11429, Apr. 8, 1987, as amended at 54
FR 32973, Aug. 11, 1989; 65 FR 61756, Oct. 17, 2000; 68 FR 59333, Oct. 15, 2003]

§60.112b Standard for volatile organic compounds (VOC).

(a) The owner or operator of each storage vessel either with a design capacity greater than or equal to 151 m^3 containing a VOL that, as stored, has a maximum true vapor pressure equal to or greater than 5.2 kPa but less than 76.6 kPa or with a design capacity greater than or equal to 75 m³ but less than 151 m³ containing a VOL that, as stored, has a maximum true vapor pressure equal to or greater than 27.6 kPa but less than 27.6 kPa but less

each storage vessel with one of the following:

(1) A fixed roof in combination with an internal floating roof meeting the following specifications:

(i) The internal floating roof shall rest or float on the liquid surface (but not necessarily in complete contact with it) inside a storage vessel that has a fixed roof. The internal floating roof shall be floating on the liquid surface at all times, except during initial fill and during those intervals when the storage vessel is completely emptied or subsequently emptied and refilled. When the roof is resting on the leg supports, the process of filling, emptying, or refilling shall be continuous and shall be accomplished as rapidly as possible.

(ii) Each internal floating roof shall be equipped with one of the following closure devices between the wall of the storage vessel and the edge of the internal floating roof:

(A) A foam- or liquid-filled seal mounted in contact with the liquid (liquid-mounted seal). A liquid-mounted seal means a foam- or liquid-filled seal mounted in contact with the liquid between the wall of the storage vessel and the floating roof continuously around the circumference of the tank.

(B) Two seals mounted one above the other so that each forms a continuous closure that completely covers the space between the wall of the storage vessel and the edge of the internal floating roof. The lower seal may be vapor-mounted, but both must be continuous.

(C) A mechanical shoe seal. A mechanical shoe seal is a metal sheet held vertically against the wall of the storage vessel by springs or weighted levers and is connected by braces to the floating roof. A flexible coated fabric (envelope) spans the annular space between the metal sheet and the floating roof.

(iii) Each opening in a noncontact internal floating roof except for automatic bleeder vents (vacuum breaker vents) and the rim space vents is to provide a projection below the liquid surface.

(iv) Each opening in the internal floating roof except for leg sleeves, automatic bleeder vents, rim space vents, column wells, ladder wells, sample wells, and stub drains is to be equipped with a cover or lid which is to be maintained in a closed position at all times (i.e., no visible gap) except when the device is in actual use. The cover or lid shall be equipped with a gasket. Covers on each access hatch and automatic gauge float well shall be bolted except when they are in use.

(v) Automatic bleeder vents shall be equipped with a gasket and are to be closed at all times when the roof is floating except when the roof is being floated off or is being landed on the roof leg supports.

(vi) Rim space vents shall be equipped with a gasket and are to be set to open only when the internal floating roof is not floating or at the manufacturer's recommended setting.

(vii) Each penetration of the internal floating roof for the purpose of sampling shall be a sample well. The sample well shall have a slit fabric cover that covers at least 90 percent of the opening.

(viii) Each penetration of the internal floating roof that allows for passage of a column supporting the fixed roof shall have a flexible fabric sleeve seal or a gasketed sliding cover.

(ix) Each penetration of the internal floating roof that allows for passage of a ladder shall have a gasketed sliding cover.

(2) An external floating roof. An external floating roof means a pontoontype or double-deck type cover that rests on the liquid surface in a vessel with no fixed roof. Each external floating roof must meet the following specifications:

(i) Each external floating roof shall be equipped with a closure device between the wall of the storage vessel and the roof edge. The closure device is to consist of two seals, one above the other. The lower seal is referred to as the primary seal, and the upper seal is referred to as the secondary seal.

(A) The primary seal shall be either a mechanical shoe seal or a liquidmounted seal. Except as provided in §60.113b(b)(4), the seal shall completely cover the annular space between the edge of the floating roof and tank wall.

(B) The secondary seal shall completely cover the annular space between the external floating roof and the wall of the storage vessel in a continuous fashion except as allowed in (0,1)(4).

(ii) Except for automatic bleeder vents and rim space vents, each opening in a noncontact external floating roof shall provide a projection below the liquid surface. Except for automatic bleeder vents, rim space vents, roof drains, and leg sleeves, each opening in the roof is to be equipped with a gasketed cover, seal, or lid that is to be maintained in a closed position at all times (i.e., no visible gap) except when the device is in actual use. Automatic bleeder vents are to be closed at all times when the roof is floating except when the roof is being floated off or is being landed on the roof leg supports. Rim vents are to be set to open when the roof is being floated off the roof legs supports or at the manufacturer's recommended setting. Automatic bleeder vents and rim space vents are to be gasketed. Each emergency roof drain is to be provided with a slotted membrane fabric cover that covers at least 90 percent of the area of the opening.

(iii) The roof shall be floating on the liquid at all times (i.e., off the roof leg supports) except during initial fill until the roof is lifted off leg supports and when the tank is completely emptied and subsequently refilled. The process of filling, emptying, or refilling when the roof is resting on the leg supports shall be continuous and shall be accomplished as rapidly as possible.

(3) A closed vent system and control device meeting the following specifications:

(i) The closed vent system shall be designed to collect all VOC vapors and gases discharged from the storage vessel and operated with no detectable emissions as indicated by an instrument reading of less than 500 ppm above background and visual inspections, as determined in part 60, subpart VV, §60.485(b).

(ii) The control device shall be designed and operated to reduce inlet VOC emissions by 95 percent or greater. If a flare is used as the control device, it shall meet the specifications described in the general control device requirements (§60.18) of the General Provisions. 40 CFR Ch. I (7–1–23 Edition)

(4) A system equivalent to those described in paragraphs (a)(1), (a)(2), or (a)(3) of this section as provided in §60.114b of this subpart.

(b) The owner or operator of each storage vessel with a design capacity greater than or equal to 75 m^3 which contains a VOL that, as stored, has a maximum true vapor pressure greater than or equal to 76.6 kPa shall equip each storage vessel with one of the following:

(1) A closed vent system and control device as specified in (60.112b(a))(3).

(2) A system equivalent to that described in paragraph (b)(1) as provided in §60.114b of this subpart.

(c) Site-specific standard for Merck & Co., Inc.'s Stonewall Plant in Elkton, Virginia. This paragraph applies only to the pharmaceutical manufacturing facility, commonly referred to as the Stonewall Plant, located at Route 340 South, in Elkton, Virginia ("site").

(1) For any storage vessel that otherwise would be subject to the control technology requirements of paragraphs (a) or (b) of this section, the site shall have the option of either complying directly with the requirements of this subpart, or reducing the site-wide total criteria pollutant emissions cap (total emissions cap) in accordance with the procedures set forth in a permit issued pursuant to 40 CFR 52.2454. If the site chooses the option of reducing the total emissions cap in accordance with the procedures set forth in such permit, the requirements of such permit shall apply in lieu of the otherwise applicable requirements of this subpart for such storage vessel.

(2) For any storage vessel at the site not subject to the requirements of 40 CFR 60.112b (a) or (b), the requirements of 40 CFR 60.116b (b) and (c) and the General Provisions (subpart A of this part) shall not apply.

[52 FR 11429, Apr. 8, 1987, as amended at 62 FR 52641, Oct. 8, 1997]

§60.113b Testing and procedures.

The owner or operator of each storage vessel as specified in 60.112b(a)shall meet the requirements of paragraph (a), (b), or (c) of this section. The applicable paragraph for a particular storage vessel depends on the control

equipment installed to meet the requirements of §60.112b.

(a) After installing the control equipment required to meet §60.112b(a)(1) (permanently affixed roof and internal floating roof), each owner or operator shall:

(1) Visually inspect the internal floating roof, the primary seal, and the secondary seal (if one is in service), prior to filling the storage vessel with VOL. If there are holes, tears, or other openings in the primary seal, the secondary seal, or the seal fabric or defects in the internal floating roof, or both, the owner or operator shall repair the items before filling the storage vessel.

(2) For Vessels equipped with a liquid-mounted or mechanical shoe primary seal, visually inspect the internal floating roof and the primary seal or the secondary seal (if one is in service) through manholes and roof hatches on the fixed roof at least once every 12 months after initial fill. If the internal floating roof is not resting on the surface of the VOL inside the storage vessel, or there is liquid accumulated on the roof, or the seal is detached, or there are holes or tears in the seal fabric, the owner or operator shall repair the items or empty and remove the storage vessel from service within 45 days. If a failure that is detected during inspections required in this paragraph cannot be repaired within 45 days and if the vessel cannot be emptied within 45 days, a 30-day extension may be requested from the Administrator in the inspection report required in (60.115b(a)(3)). Such a request for an extension must document that alternate storage capacity is unavailable and specify a schedule of actions the company will take that will assure that the control equipment will be repaired or the vessel will be emptied as soon as possible.

(3) For vessels equipped with a double-seal system as specified in §60.112b(a)(1)(ii)(B):

(i) Visually inspect the vessel as specified in paragraph (a)(4) of this section at least every 5 years; or

(ii) Visually inspect the vessel as specified in paragraph (a)(2) of this section.

(4) Visually inspect the internal floating roof, the primary seal, the secondary seal (if one is in service), gaskets, slotted membranes and sleeve seals (if any) each time the storage vessel is emptied and degassed. If the internal floating roof has defects, the primary seal has holes, tears, or other openings in the seal or the seal fabric, or the secondary seal has holes, tears, or other openings in the seal or the seal fabric, or the gaskets no longer close off the liquid surfaces from the atmosphere, or the slotted membrane has more than 10 percent open area, the owner or operator shall repair the items as necessary so that none of the conditions specified in this paragraph exist before refilling the storage vessel with VOL. In no event shall inspections conducted in accordance with this provision occur at intervals greater than 10 years in the case of vessels conducting the annual visual inspection as specified in paragraphs (a)(2) and (a)(3)(ii) of this section and at intervals no greater than 5 years in the case of vessels specified in paragraph (a)(3)(i) of this section.

(5) Notify the Administrator in writing at least 30 days prior to the filling or refilling of each storage vessel for which an inspection is required by paragraphs (a)(1) and (a)(4) of this section to afford the Administrator the opportunity to have an observer present. If the inspection required by paragraph (a)(4) of this section is not planned and the owner or operator could not have known about the inspection 30 days in advance or refilling the tank, the owner or operator shall notify the Administrator at least 7 days prior to the refilling of the storage vessel. Notification shall be made by telephone immediately followed by written documentation demonstrating why the inspection was unplanned. Alternatively, this notification including the written documentation may be made in writing and sent by express mail so that it is received by the Administrator at least 7 days prior to the refilling.

(b) After installing the control equipment required to meet $\S60.112b(a)(2)$ (external floating roof), the owner or operator shall: (1) Determine the gap areas and maximum gap widths, between the primary seal and the wall of the storage vessel and between the secondary seal and the wall of the storage vessel according to the following frequency.

(i) Measurements of gaps between the tank wall and the primary seal (seal gaps) shall be performed during the hydrostatic testing of the vessel or within 60 days of the initial fill with VOL and at least once every 5 years thereafter.

(ii) Measurements of gaps between the tank wall and the secondary seal shall be performed within 60 days of the initial fill with VOL and at least once per year thereafter.

(iii) If any source ceases to store VOL for a period of 1 year or more, subsequent introduction of VOL into the vessel shall be considered an initial fill for the purposes of paragraphs (b)(1)(i) and (b)(1)(i) of this section.

(2) Determine gap widths and areas in the primary and secondary seals individually by the following procedures:

(i) Measure seal gaps, if any, at one or more floating roof levels when the roof is floating off the roof leg supports.

(ii) Measure seal gaps around the entire circumference of the tank in each place where a 0.32-cm diameter uniform probe passes freely (without forcing or binding against seal) between the seal and the wall of the storage vessel and measure the circumferential distance of each such location.

(iii) The total surface area of each gap described in paragraph (b)(2)(ii) of this section shall be determined by using probes of various widths to measure accurately the actual distance from the tank wall to the seal and multiplying each such width by its respective circumferential distance.

(3) Add the gap surface area of each gap location for the primary seal and the secondary seal individually and divide the sum for each seal by the nominal diameter of the tank and compare each ratio to the respective standards in paragraph (b)(4) of this section.

(4) Make necessary repairs or empty the storage vessel within 45 days of identification in any inspection for seals not meeting the requirements listed in (b)(4) (i) and (ii) of this section: 40 CFR Ch. I (7-1-23 Edition)

(i) The accumulated area of gaps between the tank wall and the mechanical shoe or liquid-mounted primary seal shall not exceed 212 cm² per meter of tank diameter, and the width of any portion of any gap shall not exceed 3.81 cm.

(A) One end of the mechanical shoe is to extend into the stored liquid, and the other end is to extend a minimum vertical distance of 61 cm above the stored liquid surface.

(B) There are to be no holes, tears, or other openings in the shoe, seal fabric, or seal envelope.

(ii) The secondary seal is to meet the following requirements:

(A) The secondary seal is to be installed above the primary seal so that it completely covers the space between the roof edge and the tank wall except as provided in paragraph (b)(2)(iii) of this section.

(B) The accumulated area of gaps between the tank wall and the secondary seal shall not exceed 21.2 cm^2 per meter of tank diameter, and the width of any portion of any gap shall not exceed 1.27 cm.

(C) There are to be no holes, tears, or other openings in the seal or seal fabric.

(iii) If a failure that is detected during inspections required in paragraph (b)(1) of $\S60.113b(b)$ cannot be repaired within 45 days and if the vessel cannot be emptied within 45 days, a 30-day extension may be requested from the Administrator in the inspection report required in $\S60.115b(b)(4)$. Such extension request must include a demonstration of unavailability of alternate storage capacity and a specification of a schedule that will assure that the control equipment will be repaired or the vessel will be emptied as soon as possible.

(5) Notify the Administrator 30 days in advance of any gap measurements required by paragraph (b)(1) of this section to afford the Administrator the opportunity to have an observer present.

(6) Visually inspect the external floating roof, the primary seal, secondary seal, and fittings each time the vessel is emptied and degassed.

(i) If the external floating roof has defects, the primary seal has holes, tears, or other openings in the seal or

the seal fabric, or the secondary seal has holes, tears, or other openings in the seal or the seal fabric, the owner or operator shall repair the items as necessary so that none of the conditions specified in this paragraph exist before filling or refilling the storage vessel with VOL.

(ii) For all the inspections required by paragraph (b)(6) of this section, the owner or operator shall notify the Administrator in writing at least 30 days prior to the filling or refilling of each storage vessel to afford the Administrator the opportunity to inspect the storage vessel prior to refilling. If the inspection required by paragraph (b)(6) of this section is not planned and the owner or operator could not have known about the inspection 30 days in advance of refilling the tank, the owner or operator shall notify the Administrator at least 7 days prior to the refilling of the storage vessel. Notification shall be made by telephone immediately followed by written documentation demonstrating why the inspection was unplanned. Alternatively, this notification including the written documentation may be made in writing and sent by express mail so that it is received by the Administrator at least 7 days prior to the refilling.

(c) The owner or operator of each source that is equipped with a closed vent system and control device as required in 60.112b (a)(3) or (b)(2) (other than a flare) is exempt from 60.8 of the General Provisions and shall meet the following requirements.

(1) Submit for approval by the Administrator as an attachment to the notification required by 60.7(a)(1) or, if the facility is exempt from 60.7(a)(1), as an attachment to the notification required by 60.7(a)(2), an operating plan containing the information listed below.

(i) Documentation demonstrating that the control device will achieve the required control efficiency during maximum loading conditions. This documentation is to include a description of the gas stream which enters the control device, including flow and VOC content under varying liquid level conditions (dynamic and static) and manufacturer's design specifications for the control device. If the control device or

the closed vent capture system receives vapors, gases, or liquids other than fuels from sources that are not designated sources under this subpart, the efficiency demonstration is to include consideration of all vapors, gases, and liquids received by the closed vent capture system and control device. If an enclosed combustion device with a minimum residence time of 0.75 seconds and a minimum temperature of 816 °C is used to meet the 95 percent requirement, documentation that those conditions will exist is sufficient to meet the requirements of this paragraph.

(ii) A description of the parameter or parameters to be monitored to ensure that the control device will be operated in conformance with its design and an explanation of the criteria used for selection of that parameter (or parameters).

(2) Operate the closed vent system and control device and monitor the parameters of the closed vent system and control device in accordance with the operating plan submitted to the Administrator in accordance with paragraph (c)(1) of this section, unless the plan was modified by the Administrator during the review process. In this case, the modified plan applies.

(d) The owner or operator of each source that is equipped with a closed vent system and a flare to meet the requirements in (0,12) (a)(3) or (b)(2) shall meet the requirements as specified in the general control device requirements, (0,12) (e) and (f).

[52 FR 11429, Apr. 8, 1987, as amended at 54 FR 32973, Aug. 11, 1989]

§60.114b Alternative means of emission limitation.

(a) If, in the Administrator's judgment, an alternative means of emission limitation will achieve a reduction in emissions at least equivalent to the reduction in emissions achieved by any requirement in §60.112b, the Administrator will publish in the FEDERAL REGISTER a notice permitting the use of the alternative means for purposes of compliance with that requirement.

(b) Any notice under paragraph (a) of this section will be published only after notice and an opportunity for a hearing. (c) Any person seeking permission under this section shall submit to the Administrator a written application including:

(1) An actual emissions test that uses a full-sized or scale-model storage vessel that accurately collects and measures all VOC emissions from a given control device and that accurately simulates wind and accounts for other emission variables such as temperature and barometric pressure.

(2) An engineering evaluation that the Administrator determines is an accurate method of determining equivalence.

(d) The Administrator may condition the permission on requirements that may be necessary to ensure operation and maintenance to achieve the same emissions reduction as specified in §60.112b.

§60.115b Reporting and recordkeeping requirements.

The owner or operator of each storage vessel as specified in §60.112b(a) shall keep records and furnish reports as required by paragraphs (a), (b), or (c) of this section depending upon the control equipment installed to meet the requirements of §60.112b. The owner or operator shall keep copies of all reports and records required by this section, except for the record required by (c)(1), for at least 2 years. The record required by (c)(1) will be kept for the life of the control equipment.

(a) After installing control equipment in accordance with §60.112b(a)(1) (fixed roof and internal floating roof), the owner or operator shall meet the following requirements.

(1) Furnish the Administrator with a report that describes the control equipment and certifies that the control equipment meets the specifications of (0.112b(a)(1)) and (0.113b(a)(1)). This report shall be an attachment to the notification required by (0.7(a)(3)).

(2) Keep a record of each inspection performed as required by 60.113b(a)(1), (a)(2), (a)(3), and (a)(4). Each record shall identify the storage vessel on which the inspection was performed and shall contain the date the vessel was inspected and the observed condition of each component of the control

40 CFR Ch. I (7–1–23 Edition)

equipment (seals, internal floating roof, and fittings).

(3) If any of the conditions described in (3, 113b(a))(2) are detected during the annual visual inspection required by (3, 0, 113b(a))(2), a report shall be furnished to the Administrator within 30 days of the inspection. Each report shall identify the storage vessel, the nature of the defects, and the date the storage vessel was emptied or the nature of and date the repair was made.

(4) After each inspection required by (4) After each inspection required by (4) and (3) that finds holes or tears in the seal or seal fabric, or defects in the internal floating roof, or other control equipment defects listed in (6) and (3) and (3) are port shall be furnished to the Administrator within 30 days of the inspection. The report shall identify the storage vessel and the reason it did not meet the specifications of (6) and list each repair made.

(b) After installing control equipment in accordance with §60.112b(a)(2) (external floating roof), the owner or operator shall meet the following requirements.

(1) Furnish the Administrator with a report that describes the control equipment and certifies that the control equipment meets the specifications of (0.112b(a)(2)) and (0.113b(b)(2)), (0.112b(a)(2)), and (0.113b(b)(2)), (0.112b(a)(2)), and (0.113b(b)(2)), (0.112b(a)(2)), and (0.113b(b)(2)), (0.112b(a)(2)), and (0.112b(a)(2)).

(2) Within 60 days of performing the seal gap measurements required by §60.113b(b)(1), furnish the Administrator with a report that contains:

(i) The date of measurement.

(ii) The raw data obtained in the measurement.

(iii) The calculations described in §60.113b (b)(2) and (b)(3).

(3) Keep a record of each gap measurement performed as required by §60.113b(b). Each record shall identify the storage vessel in which the measurement was performed and shall contain:

(i) The date of measurement.

(ii) The raw data obtained in the measurement.

(iii) The calculations described in §60.113b (b)(2) and (b)(3).

(4) After each seal gap measurement that detects gaps exceeding the limitations specified by 60.113b(b)(4), submit a report to the Administrator within 30 days of the inspection. The report will identify the vessel and contain the information specified in paragraph (b)(2) of this section and the date the vessel was emptied or the repairs made and date of repair.

(c) After installing control equipment in accordance with $\S60.112b$ (a)(3) or (b)(1) (closed vent system and control device other than a flare), the owner or operator shall keep the following records.

(1) A copy of the operating plan.

(2) A record of the measured values of the parameters monitored in accordance with 60.113b(c)(2).

(d) After installing a closed vent system and flare to comply with §60.112b, the owner or operator shall meet the following requirements.

A report containing the measurements required by §60.18(f) (1), (2), (3),
 (4), (5), and (6) shall be furnished to the Administrator as required by §60.8 of the General Provisions. This report shall be submitted within 6 months of the initial start-up date.

(2) Records shall be kept of all periods of operation during which the flare pilot flame is absent.

(3) Semiannual reports of all periods recorded under 60.115b(d)(2) in which the pilot flame was absent shall be furnished to the Administrator.

[52 FR 11429, Apr. 8, 1987, as amended at 86 FR 5019, Jan. 19, 2021]

§60.116b Monitoring of operations.

(a) The owner or operator shall keep copies of all records required by this section, except for the record required by paragraph (b) of this section, for at least 2 years. The record required by paragraph (b) of this section will be kept for the life of the source.

(b) The owner or operator of each storage vessel as specified in §60.110b(a) shall keep readily accessible records showing the dimension of the storage vessel and an analysis showing the capacity of the storage vessel.

(c) Except as provided in paragraphs (f) and (g) of this section, the owner or operator of each storage vessel either with a design capacity greater than or equal to 151 m³ storing a liquid with a maximum true vapor pressure greater than or equal to 3.5 kPa or with a design capacity greater than or equal to 75 m^3 but less than 151 m³ storing a liquid with a maximum true vapor pressure greater than or equal to 15.0 kPa shall maintain a record of the VOL stored, the period of storage, and the maximum true vapor pressure of that VOL during the respective storage period.

(d) Except as provided in paragraph (g) of this section, the owner or operator of each storage vessel either with a design capacity greater than or equal to 151 m³ storing a liquid with a maximum true vapor pressure that is normally less than 5.2 kPa or with a design capacity greater than or equal to 75 m³ but less than 151 m³ storing a liquid with a maximum true vapor pressure that is normally less than 27.6 kPa shall notify the Administrator within 30 days when the maximum true vapor pressure of the liquid exceeds the respective maximum true vapor vapor pressure values for each volume range.

(e) Available data on the storage temperature may be used to determine the maximum true vapor pressure as determined below.

(1) For vessels operated above or below ambient temperatures, the maximum true vapor pressure is calculated based upon the highest expected calendar-month average of the storage temperature. For vessels operated at ambient temperatures, the maximum true vapor pressure is calculated based upon the maximum local monthly average ambient temperature as reported by the National Weather Service.

(2) For crude oil or refined petroleum products the vapor pressure may be obtained by the following:

(i) Available data on the Reid vapor pressure and the maximum expected storage temperature based on the highest expected calendar-month average temperature of the stored product may be used to determine the maximum true vapor pressure from nomographs contained in API Bulletin 2517 (incorporated by reference—see §60.17), unless the Administrator specifically requests that the liquid be sampled, the actual storage temperature determined, and the Reid vapor pressure determined from the sample(s).

(ii) The true vapor pressure of each type of crude oil with a Reid vapor pressure less than 13.8 kPa or with physical properties that preclude determination by the recommended method is to be determined from available data and recorded if the estimated maximum true vapor pressure is greater than 3.5 kPa.

(3) For other liquids, the vapor pressure:

(i) May be obtained from standard reference texts, or

(ii) Determined by ASTM D2879-83,96, or 97 (incorporated by reference—see §60.17); or

(iii) Measured by an appropriate method approved by the Administrator; or

(iv) Calculated by an appropriate method approved by the Administrator.

(f) The owner or operator of each vessel storing a waste mixture of indeterminate or variable composition shall be subject to the following requirements.

(1) Prior to the initial filling of the vessel, the highest maximum true vapor pressure for the range of anticipated liquid compositions to be stored will be determined using the methods described in paragraph (e) of this section.

(2) For vessels in which the vapor pressure of the anticipated liquid composition is above the cutoff for monitoring but below the cutoff for controls as defined in (0.112b(a)), an initial physical test of the vapor pressure is required; and a physical test at least once every 6 months thereafter is required as determined by the following methods:

(i) ASTM D2879-83, 96, or 97 (incorporated by reference—see §60.17); or

(ii) ASTM D323-82 or 94 (incorporated by reference—see §60.17); or

(iii) As measured by an appropriate method as approved by the Administrator.

(g) The owner or operator of each vessel equipped with a closed vent system and control device meeting the specification of §60.112b or with emissions reductions equipment as specified

40 CFR Ch. I (7–1–23 Edition)

in 40 CFR 65.42(b)(4), (b)(5), (b)(6), or (c) is exempt from the requirements of paragraphs (c) and (d) of this section.

[52 FR 11429, Apr. 8, 1987, as amended at 65
FR 61756, Oct. 17, 2000; 65 FR 78276, Dec. 14, 2000; 68 FR 59333, Oct. 15, 2003]

§60.117b Delegation of authority.

(a) In delegating implementation and enforcement authority to a State under section 111(c) of the Act, the authorities contained in paragraph (b) of this section shall be retained by the Administrator and not transferred to a State.

(b) Authorities which will not be delegated to States: \$60.111b(f)(4), 60.114b, 60.116b(e)(3)(iii), 60.116b(e)(3)(iv), and 60.116b(f)(2)(iii).

 $[52\ {\rm FR}$ 11429, Apr. 8, 1987, as amended at 52 ${\rm FR}$ 22780, June 16, 1987]

Subpart L—Standards of Performance for Secondary Lead Smelters

§60.120 Applicability and designation of affected facility.

(a) The provisions of this subpart are applicable to the following affected facilities in secondary lead smelters: Pot furnaces of more than 250 kg (550 lb) charging capacity, blast (cupola) furnaces, and reverberatory furnaces.

(b) Any facility under paragraph (a) of this section that commences construction or modification after June 11, 1973, is subject to the requirements of this subpart.

[42 FR 37937, July 25, 1977]

§60.121 Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Act and in subpart A of this part.

(a) *Reverberatory furnace* includes the following types of reverberatory furnaces: stationary, rotating, rocking, and tilting.

(b) Secondary lead smelter means any facility producing lead from a leadbearing scrap material by smelting to the metallic form.

(c) *Lead* means elemental lead or alloys in which the predominant component is lead.

Appendix D

§60.245 Recordkeeping.

AUTHENTICATED U.S. GOVERNMENT INFORMATION

> Any facility under 60.240(a) that commences construction, modification, or reconstruction after November 7, 2014 is subject to the requirements of this section. You must maintain the records identified as specified in 60.7(f) and in paragraphs (a) and (b) of this section. All records required by this subpart must be maintained onsite for at least 5 years.

(a) Records of the daily average pressure drop through the absorber.

(b) Records of deviations. A deviation is determined to have occurred when the monitoring data or lack of monitoring data result in any one of the criteria specified in paragraphs (b)(1) and (2) of this section being met.

(1) A deviation occurs when the daily average value of a monitored operating parameter is less than the minimum pressure drop, or greater than the maximum pressure drop established in $\S60.243(e)(3)$.

(2) A deviation occurs when the monitoring data are not available for at least 75 percent of the operating hours in a day.

[80 FR 50436, Aug. 19, 2015]

Subpart Y—Standards of Performance for Coal Preparation and Processing Plants

SOURCE: 74 FR 51977, Oct. 8, 2009, unless otherwise noted.

§60.250 Applicability and designation of affected facility.

(a) The provisions of this subpart apply to affected facilities in coal preparation and processing plants that process more than 181 megagrams (Mg) (200 tons) of coal per day.

(b) The provisions in §§ 60.251, 60.252(a), 60.253(a), 60.254(a), 60.255(a), and 60.256(a) of this subpart are applicable to any of the following affected facilities that commenced construction, reconstruction or modification after October 27, 1974, and on or before April 28, 2008: Thermal dryers, pneumatic coal-cleaning equipment (air tables), coal processing and conveying equipment (including breakers and crushers), and coal storage systems, transfer and loading systems.

The provisions in §§ 60.251, (c) 60.252(b)(1) and (c), 60.253(b), 60.254(b), 60.255(b) through (h), 60.256(b) and (c), 60.257, and 60.258 of this subpart are applicable to any of the following affected facilities that commenced construction, reconstruction or modification after April 28, 2008, and on or before May 27, 2009: Thermal dryers, pneumatic coal-cleaning equipment (air tables), coal processing and conveying equipment (including breakers and crushers), and coal storage systems, transfer and loading systems.

The provisions in §§60.251, (d) 60.252(b)(1) through (3), and (c), 60.253(b), 60.254(b) and (c), 60.255(b) through (h), 60.256(b) and (c), 60.257, and 60.258 of this subpart are applicable to any of the following affected facilities that commenced construction, reconstruction or modification after May 27, 2009: Thermal dryers, pneumatic coalcleaning equipment (air tables), coal processing and conveying equipment (including breakers and crushers), coal storage systems, transfer and loading systems, and open storage piles.

§60.251 Definitions.

As used in this subpart, all terms not defined herein have the meaning given them in the Clean Air Act (Act) and in subpart A of this part.

(a) Anthracite means coal that is classified as anthracite according to the American Society of Testing and Materials in ASTM D388 (incorporated by reference, see §60.17).

(b) Bag leak detection system means a system that is capable of continuously monitoring relative particulate matter (dust loadings) in the exhaust of a fabric filter to detect bag leaks and other upset conditions. A bag leak detection system includes, but is not limited to, an instrument that operates on triboelectric, light scattering, light transmittance, or other effect to continuously monitor relative particulate matter loadings.

(c) *Bituminous coal* means solid fossil fuel classified as bituminous coal by ASTM D388 (incorporated by reference—see §60.17).

(d) Coal means:

(1) For units constructed, reconstructed, or modified on or before May 27, 2009, all solid fossil fuels classified

40 CFR Ch. I (7-1-23 Edition)

as anthracite, bituminous, subbituminous, or lignite by ASTM D388 (incorporated by reference—see §60.17).

§ 60.251

(2) For units constructed, reconstructed, or modified after May 27, 2009, all solid fossil fuels classified as anthracite, bituminous, subbituminous, or lignite by ASTM D388 (incorporated by reference—see §60.17), and coal refuse.

(e) Coal preparation and processing plant means any facility (excluding underground mining operations) which prepares coal by one or more of the following processes: breaking, crushing, screening, wet or dry cleaning, and thermal drying.

(f) Coal processing and conveying equipment means any machinery used to reduce the size of coal or to separate coal from refuse, and the equipment used to convey coal to or remove coal and refuse from the machinery. This includes, but is not limited to, breakers, crushers, screens, and conveyor belts. Equipment located at the mine face is not considered to be part of the coal preparation and processing plant.

(g) *Coal refuse* means waste products of coal mining, physical coal cleaning, and coal preparation operations (e.g., culm, gob, *etc.*) containing coal, matrix material, clay, and other organic and inorganic material.

(h) *Coal storage system* means any facility used to store coal except for open storage piles.

(i) Design controlled potential PM emissions rate means the theoretical particulate matter (PM) emissions (Mg) that would result from the operation of a control device at its design emissions rate (grams per dry standard cubic meter (g/dscm)), multiplied by the maximum design flow rate (dry standard cubic meter per minute (dscm/ min)), multiplied by 60 (minutes per hour (min/hr)), multiplied by 8,760 (hours per year (hr/yr)), divided by 1,000,000 (megagrams per gram (Mg/g)).

(j) Indirect thermal dryer means a thermal dryer that reduces the moisture content of coal through indirect heating of the coal through contact with a heat transfer medium. If the source of heat (the source of combustion or furnace) is subject to another subpart of this part, then the furnace and the associated emissions are not part of the affected facility. However, if the source of heat is not subject to another subpart of this part, then the furnace and the associated emissions are part of the affected facility.

(k) *Lignite* means coal that is classified as lignite A or B according to the American Society of Testing and Materials in ASTM D388 (incorporated by reference, *see* §60.17).

(1) *Mechanical vent* means any vent that uses a powered mechanical drive (machine) to induce air flow.

(m) Open storage pile means any facility, including storage area, that is not enclosed that is used to store coal, including the equipment used in the loading, unloading, and conveying operations of the facility.

(n) Operating day means a 24-hour period between 12 midnight and the following midnight during which coal is prepared or processed at any time by the affected facility. It is not necessary that coal be prepared or processed the entire 24-hour period.

(o) *Pneumatic coal-cleaning equipment* means:

(1) For units constructed, reconstructed, or modified on or before May 27, 2009, any facility which classifies bituminous coal by size or separates bituminous coal from refuse by application of air stream(s).

(2) For units constructed, reconstructed, or modified after May 27, 2009, any facility which classifies coal by size or separates coal from refuse by application of air stream(s).

(p) Potential combustion concentration means the theoretical emissions (nanograms per joule (ng/J) or pounds per million British thermal units (lb/ MMBtu) heat input) that would result from combustion of a fuel in an uncleaned state without emission control systems, as determined using Method 19 of appendix A-7 of this part.

(q) Subbituminous coal means coal that is classified as subbituminous A, B, or C according to the American Society of Testing and Materials in ASTM D388 (incorporated by reference, see §60.17).

(r) *Thermal dryer* means:

(1) For units constructed, reconstructed, or modified on or before May
27, 2009, any facility in which the moisture content of bituminous coal is reduced by contact with a heated gas stream which is exhausted to the atmosphere.

(2) For units constructed, reconstructed, or modified after May 27, 2009, any facility in which the moisture content of coal is reduced by either contact with a heated gas stream which is exhausted to the atmosphere or through indirect heating of the coal through contact with a heated heat transfer medium.

(s) *Transfer and loading system* means any facility used to transfer and load coal for shipment.

§60.252 Standards for thermal dryers.

(a) On and after the date on which the performance test is conducted or required to be completed under §60.8, whichever date comes first, an owner or operator of a thermal dryer constructed, reconstructed, or modified on or before April 28, 2008, subject to the provisions of this subpart must meet the requirements in paragraphs (a)(1) and (a)(2) of this section.

(1) The owner or operator shall not cause to be discharged into the atmosphere from the thermal dryer any gases which contain PM in excess of 0.070 g/ dscm (0.031 grains per dry standard cubic feet (gr/dscf)); and

(2) The owner or operator shall not cause to be discharged into the atmosphere from the thermal dryer any gases which exhibit 20 percent opacity or greater.

(b) Except as provided in paragraph (c) of this section, on and after the date on which the performance test is conducted or required to be completed under §60.8, whichever date comes first, an owner or operator of a thermal dryer constructed, reconstructed, or modified after April 28, 2008, subject to the provisions of this subpart must meet the applicable standards for PM and opacity, as specified in paragraph (b)(1) of this section. In addition, and except as provided in paragraph (c) of this section, on and after the date on which the performance test is conducted or required to be completed under §60.8, whichever date comes first, an owner or operator of a thermal dryer constructed, reconstructed, or

modified after May 29, 2009, subject to the provisions of this subpart must also meet the applicable standards for sulfur dioxide (SO₂), and combined nitrogen oxides (NO_X) and carbon monoxide (CO) as specified in paragraphs (b)(2) and (b)(3) of this section.

(1) The owner or operator must meet the requirements for PM emissions in paragraphs (b)(1)(i) through (iii) of this section, as applicable to the affected facility.

(i) For each thermal dryer constructed or reconstructed after April 28, 2008, the owner or operator must meet the requirements of (b)(1)(i)(A)and (b)(1)(i)(B).

(A) The owner or operator must not cause to be discharged into the atmosphere from the thermal dryer any gases that contain PM in excess of 0.023 g/ dscm (0.010 grains per dry standard cubic feet (gr/dscf)); and

(B) The owner or operator must not cause to be discharged into the atmosphere from the thermal dryer any gases that exhibit 10 percent opacity or greater.

(ii) For each thermal dryer modified after April 28, 2008, the owner or operator must meet the requirements of paragraphs (b)(1)(ii)(A) and (b)(1)(ii)(B) of this section.

(A) The owner or operator must not cause to be discharged to the atmosphere from the affected facility any gases which contain PM in excess of 0.070 g/dscm (0.031 gr/dscf); and

(B) The owner or operator must not cause to be discharged into the atmosphere from the affected facility any gases which exhibit 20 percent opacity or greater.

(2) Except as provided in paragraph (b)(2)(iii) of this section, for each thermal dryer constructed, reconstructed, or modified after May 27, 2009, the owner or operator must meet the requirements for SO_2 emissions in either paragraph (b)(2)(i) or (b)(2)(ii) of this section.

(i) The owner or operator must not cause to be discharged into the atmosphere from the affected facility any gases that contain SO_2 in excess of 85 ng/J (0.20 lb/MMBtu) heat input; or

(ii) The owner or operator must not cause to be discharged into the atmosphere from the affected facility any gases that either contain SO_2 in excess of 520 ng/J (1.20 lb/MMBtu) heat input or contain SO_2 in excess of 10 percent of the potential combustion concentration (*i.e.*, the facility must achieve at least a 90 percent reduction of the potential combustion concentration and may not exceed a maximum emissions rate of 1.2 lb/MMBtu (520 ng/J)).

(iii) Thermal dryers that receive all of their thermal input from a source other than coal or residual oil, that receive all of their thermal input from a source subject to an SO_2 limit under another subpart of this part, or that use waste heat or residual from the combustion of coal or residual oil as their only thermal input are not subject to the SO_2 limits of this section.

(3) Except as provided in paragraph (b)(3)(iii) of this section, the owner or operator must meet the requirements for combined NO_X and CO emissions in paragraph (b)(3)(i) or (b)(3)(ii) of this section, as applicable to the affected facility.

(i) For each thermal dryer constructed after May 27, 2009, the owner or operator must not cause to be discharged into the atmosphere from the affected facility any gases which contain a combined concentration of NO_X and CO in excess of 280 ng/J (0.65 lb/ MMBtu) heat input.

(ii) For each thermal dryer reconstructed or modified after May 27, 2009, the owner or operator must not cause to be discharged into the atmosphere from the affected facility any gases which contain combined concentration of NO_x and CO in excess of 430 ng/J (1.0 lb/MMBtu) heat input.

(iii) Thermal dryers that receive all of their thermal input from a source other than coal or residual oil, that receive all of their thermal input from a source subject to a NO_X limit and/or CO limit under another subpart of this part, or that use waste heat or residual from the combustion of coal or residual oil as their only thermal input, are not subject to the combined NO_X and CO limits of this section.

(c) Thermal dryers receiving all of their thermal input from an affected facility covered under another 40 CFR Part 60 subpart must meet the applicable requirements in that subpart but 40 CFR Ch. I (7–1–23 Edition)

are not subject to the requirements in this subpart.

§60.253 Standards for pneumatic coalcleaning equipment.

(a) On and after the date on which the performance test is conducted or required to be completed under §60.8, whichever date comes first, an owner or operator of pneumatic coal-cleaning equipment constructed, reconstructed, or modified on or before April 28, 2008, must meet the requirements of paragraphs (a)(1) and (a)(2) of this section.

(1) The owner or operator must not cause to be discharged into the atmosphere from the pneumatic coal-cleaning equipment any gases that contain PM in excess of 0.040 g/dscm (0.017 gr/ dscf); and

(2) The owner or operator must not cause to be discharged into the atmosphere from the pneumatic coal-cleaning equipment any gases that exhibit 10 percent opacity or greater.

(b) On and after the date on which the performance test is conducted or required to be completed under 60.8, whichever date comes first, an owner or operator of pneumatic coal-cleaning equipment constructed, reconstructed, or modified after April 28, 2008, must meet the requirements in paragraphs (b)(1) and (b)(2) of this section.

(1) The owner of operator must not cause to be discharged into the atmosphere from the pneumatic coal-cleaning equipment any gases that contain PM in excess or 0.023 g/dscm (0.010 gr/dscf); and

(2) The owner or operator must not cause to be discharged into the atmosphere from the pneumatic coal-cleaning equipment any gases that exhibit greater than 5 percent opacity.

§ 60.254 Standards for coal processing and conveying equipment, coal storage systems, transfer and loading systems, and open storage piles.

(a) On and after the date on which the performance test is conducted or required to be completed under §60.8, whichever date comes first, an owner or operator shall not cause to be discharged into the atmosphere from any coal processing and conveying equipment, coal storage system, or coal transfer and loading system processing

coal constructed, reconstructed, or modified on or before April 28, 2008, gases which exhibit 20 percent opacity or greater.

(b) On and after the date on which the performance test is conducted or required to be completed under §60.8, whichever date comes first, an owner or operator of any coal processing and conveying equipment, coal storage system, or coal transfer and loading system processing coal constructed, reconstructed, or modified after April 28, 2008, must meet the requirements in paragraphs (b)(1) through (3) of this section, as applicable to the affected facility.

(1) Except as provided in paragraph (b)(3) of this section, the owner or operator must not cause to be discharged into the atmosphere from the affected facility any gases which exhibit 10 percent opacity or greater.

(2) The owner or operator must not cause to be discharged into the atmosphere from any mechanical vent on an affected facility gases which contain particulate matter in excess of 0.023 g/dscm (0.010 gr/dscf).

(3) Equipment used in the loading, unloading, and conveying operations of open storage piles are not subject to the opacity limitations of paragraph (b)(1) of this section.

(c) The owner or operator of an open storage pile, which includes the equipment used in the loading, unloading, and conveying operations of the affected facility, constructed, reconstructed, or modified after May 27, 2009, must prepare and operate in accordance with a submitted fugitive coal dust emissions control plan that is appropriate for the site conditions as specified in paragraphs (c)(1) through (6) of this section.

(1) The fugitive coal dust emissions control plan must identify and describe the control measures the owner or operator will use to minimize fugitive coal dust emissions from each open storage pile.

(2) For open coal storage piles, the fugitive coal dust emissions control plan must require that one or more of the following control measures be used to minimize to the greatest extent practicable fugitive coal dust: Locating the source inside a partial enclosure, installing and operating a water spray or fogging system, applying appropriate chemical dust suppression agents on the source (when the provisions of paragraph (c)(6) of this section are met), use of a wind barrier, compaction, or use of a vegetative cover. The owner or operator must select, for inclusion in the fugitive coal dust emissions control plan, the control measure or measures listed in this paragraph that are most appropriate for site conditions. The plan must also explain how the measure or measures selected are applicable and appropriate for site conditions. In addition, the plan must be revised as needed to reflect any changing conditions at the source.

(3) Any owner or operator of an affected facility that is required to have a fugitive coal dust emissions control plan may petition the Administrator to approve, for inclusion in the plan for the affected facility, alternative control measures other than those specified in paragraph (c)(2) of this section as specified in paragraphs (c)(3)(i) through (iv) of this section.

(i) The petition must include a description of the alternative control measures, a copy of the fugitive coal dust emissions control plan for the affected facility that includes the alternative control measures, and information sufficient for EPA to evaluate the demonstrations required by paragraph (c)(3)(ii) of this section.

(ii) The owner or operator must either demonstrate that the fugitive coal dust emissions control plan that includes the alternate control measures will provide equivalent overall environmental protection or demonstrate that it is either economically or technically infeasible for the affected facility to use the control measures specifically identified in paragraph (c)(2).

(iii) While the petition is pending, the owner or operator must comply with the fugitive coal dust emissions control plan including the alternative control measures submitted with the petition. Operation in accordance with the plan submitted with the petition shall be deemed to constitute compliance with the requirement to operate in accordance with a fugitive coal dust emissions control plan that contains one of the control measures specifically identified in paragraph (c)(2) of this section while the petition is pending.

(iv) If the petition is approved by the Administrator, the alternative control measures will be approved for inclusion in the fugitive coal dust emissions control plan for the affected facility. In lieu of amending this subpart, a letter will be sent to the facility describing the specific control measures approved. The facility shall make any such letters and the applicable fugitive coal dust emissions control plan available to the public. If the Administrator determines it is appropriate, the conditions and requirements of the letter can be reviewed and changed at any point.

(4) The owner or operator must submit the fugitive coal dust emissions control plan to the Administrator or delegated authority as specified in paragraphs (c)(4)(i) and (c)(4)(ii) of this section.

(i) The plan must be submitted to the Administrator or delegated authority prior to startup of the new, reconstructed, or modified affected facility, or 30 days after the effective date of this rule, whichever is later.

(ii) The plan must be revised as needed to reflect any changing conditions at the source. Such revisions must be dated and submitted to the Administrator or delegated authority before a source can operate pursuant to these revisions. The Administrator or delegated authority may also object to such revisions as specified in paragraph (c)(5) of this section.

(5) The Administrator or delegated authority may object to the fugitive coal dust emissions control plan as specified in paragraphs (c)(5)(i) and (c)(5)(ii) of this section.

(i) The Administrator or delegated authority may object to any fugitive coal dust emissions control plan that it has determined does not meet the requirements of paragraphs (c)(1) and (c)(2) of this section.

(ii) If an objection is raised, the owner or operator, within 30 days from receipt of the objection, must submit a revised fugitive coal dust emissions control plan to the Administrator or delegated authority. The owner or op40 CFR Ch. I (7–1–23 Edition)

erator must operate in accordance with the revised fugitive coal dust emissions control plan. The Administrator or delegated authority retain the right, under paragraph (c)(5) of this section, to object to the revised control plan if it determines the plan does not meet the requirements of paragraphs (c)(1) and (c)(2) of this section.

(6) Where appropriate chemical dust suppression agents are selected by the owner or operator as a control measure to minimize fugitive coal dust emissions, (1) only chemical dust suppressants with Occupational Safety and Health Administration (OSHA)-compliant material safety data sheets (MSDS) are to be allowed; (2) the MSDS must be included in the fugitive coal dust emissions control plan; and (3) the owner or operator must consider and document in the fugitive coal dust emissions control plan the site-specific impacts associated with the use of such chemical dust suppressants.

§60.255 Performance tests and other compliance requirements.

(a) An owner or operator of each affected facility that commenced construction, reconstruction, or modification on or before April 28, 2008, must conduct all performance tests required by $\S60.8$ to demonstrate compliance with the applicable emission standards using the methods identified in $\S60.257$.

(b) An owner or operator of each affected facility that commenced construction, reconstruction, or modification after April 28, 2008, must conduct performance tests according to the requirements of 60.8 and the methods identified in 60.257 to demonstrate compliance with the applicable emissions standards in this subpart as specified in paragraphs (b)(1) and (2) of this section.

(1) For each affected facility subject to a PM, SO₂, or combined NO_X and CO emissions standard, an initial performance test must be performed. Thereafter, a new performance test must be conducted according the requirements in paragraphs (b)(1)(i) through (iii) of this section, as applicable.

(i) If the results of the most recent performance test demonstrate that emissions from the affected facility are

greater than 50 percent of the applicable emissions standard, a new performance test must be conducted within 12 calendar months of the date that the previous performance test was required to be completed.

(ii) If the results of the most recent performance test demonstrate that emissions from the affected facility are 50 percent or less of the applicable emissions standard, a new performance test must be conducted within 24 calendar months of the date that the previous performance test was required to be completed.

(iii) An owner or operator of an affected facility that has not operated for the 60 calendar days prior to the due date of a performance test is not required to perform the subsequent performance test until 30 calendar days after the next operating day.

(2) For each affected facility subject to an opacity standard, an initial performance test must be performed. Thereafter, a new performance test must be conducted according to the requirements in paragraphs (b)(2)(i) through (iii) of this section, as applicable, except as provided for in paragraphs (e) and (f) of this section. Performance test and other compliance requirements for coal truck dump operations are specified in paragraph (h) of this section.

(i) If any 6-minute average opacity reading in the most recent performance test exceeds half the applicable opacity limit, a new performance test must be conducted within 90 operating days of the date that the previous performance test was required to be completed.

(ii) If all 6-minute average opacity readings in the most recent performance test are equal to or less than half the applicable opacity limit, a new performance test must be conducted within 12 calendar months of the date that the previous performance test was required to be completed.

(iii) An owner or operator of an affected facility continuously monitoring scrubber parameters as specified in 60.256(b)(2) is exempt from the requirements in paragraphs (b)(2)(i) and (ii) if opacity performance tests are conducted concurrently with (or within a 60-minute period of) PM performance tests. (c) If any affected coal processing and conveying equipment (*e.g.*, breakers, crushers, screens, conveying systems), coal storage systems, or coal transfer and loading systems that commenced construction, reconstruction, or modification after April 28, 2008, are enclosed in a building, and emissions from the building do not exceed any of the standards in §60.254 that apply to the affected facility, then the facility shall be deemed to be in compliance with such standards.

(d) An owner or operator of an affected facility (other than a thermal dryer) that commenced construction, reconstruction, or modification after April 28, 2008, is subject to a PM emission standard and uses a control device with a design controlled potential PM emissions rate of 1.0 Mg (1.1 tons) per year or less is exempted from the requirements of paragraphs (b)(1)(i) and (ii) of this section provided that the owner or operator meets all of the conditions specified in paragraphs (d)(1) through (3) of this section. This exemption does not apply to thermal dryers.

(1) PM emissions, as determined by the most recent performance test, are less than or equal to the applicable limit,

(2) The control device manufacturer's recommended maintenance procedures are followed, and

(3) All 6-minute average opacity readings from the most recent performance test are equal to or less than half the applicable opacity limit or the monitoring requirements in paragraphs (e) or (f) of this section are followed.

(e) For an owner or operator of a group of up to five of the same type of affected facilities that commenced construction, reconstruction, or modification after April 28, 2008, that are subject to PM emissions standards and use identical control devices, the Administrator or delegated authority may allow the owner or operator to use a single PM performance test for one of the affected control devices to demonstrate that the group of affected facilities is in compliance with the applicable emissions standards provided that the owner or operator meets all of the conditions specified in paragraphs (e)(1) through (3) of this section.

(1) PM emissions from the most recent performance test for each individual affected facility are 90 percent or less of the applicable PM standard;

(2) The manufacturer's recommended maintenance procedures are followed for each control device; and

(3) A performance test is conducted on each affected facility at least once every 5 calendar years.

(f) As an alternative to meeting the requirements in paragraph (b)(2) of this section, an owner or operator of an affected facility that commenced construction, reconstruction, or modification after April 28, 2008, may elect to comply with the requirements in paragraph (f)(1) or (f)(2) of this section.

(1) Monitor visible emissions from each affected facility according to the requirements in paragraphs (f)(1)(i)through (iii) of this section.

(i) Conduct one daily 15-second observation each operating day for each affected facility (during normal operation) when the coal preparation and processing plant is in operation. Each observation must be recorded as either visible emissions observed or no visible emissions observed. Each observer determining the presence of visible emissions must meet the training requirements specified in §2.3 of Method 22 of appendix A-7 of this part. If visible emissions are observed during any 15second observation, the owner or operator must adjust the operation of the affected facility and demonstrate within 24 hours that no visible emissions are observed from the affected facility. If visible emissions are observed, a Method 9, of appendix A-4 of this part, performance test must be conducted within 45 operating days.

(ii) Conduct monthly visual observations of all process and control equipment. If any deficiencies are observed, the necessary maintenance must be performed as expeditiously as possible.

(iii) Conduct a performance test using Method 9 of appendix A-4 of this part at least once every 5 calendar years for each affected facility.

(2) Prepare a written site-specific monitoring plan for a digital opacity compliance system for approval by the Administrator or delegated authority. The plan shall require observations of at least one digital image every 15 sec40 CFR Ch. I (7–1–23 Edition)

onds for 10-minute periods (during normal operation) every operating day. An approvable monitoring plan must include a demonstration that the occurrences of visible emissions are not in excess of 5 percent of the observation period. For reference purposes in preparing the monitoring plan, see OAQPS 'Determination of Visible Emission Opacity from Stationary Sources Using Computer-Based Photographic Analysis Systems." This document is available from the U.S. Environmental Protection Agency (U.S. EPA); Office of Air Quality and Planning Standards; Sector Policies and Programs Division; Measurement Group (D243-02), Research Triangle Park, NC 27711. This document is also available on the Technology Transfer Network (TTN) under Emission Measurement Center Preliminary Methods. The monitoring plan approved by the Administrator or delegated authority shall be implemented by the owner or operator.

(g) As an alternative to meeting the requirements in paragraph (b)(2) of this section, an owner or operator of an affected facility that commenced construction, reconstruction, or modification after April 28, 2008, subject to a visible emissions standard under this subpart may install, operate, and maintain a continuous opacity monitoring system (COMS). Each COMS used to comply with provisions of this subpart must be installed, calibrated, maintained, and continuously operated according to the requirements in paragraphs (g)(1) and (2) of this section.

(1) The COMS must meet Performance Specification 1 in 40 CFR part 60, appendix B.

(2) The COMS must comply with the quality assurance requirements in paragraphs (g)(2)(i) through (v) of this section.

(i) The owner or operator must automatically (intrinsic to the opacity monitor) check the zero and upscale (span) calibration drifts at least once daily. For particular COMS, the acceptable range of zero and upscale calibration materials is as defined in the applicable version of Performance Specification 1 in 40 CFR part 60, appendix B.

(ii) The owner or operator must adjust the zero and span whenever the 24-

hour zero drift or 24-hour span drift exceeds 4 percent opacity. The COMS must allow for the amount of excess zero and span drift measured at the 24hour interval checks to be recorded and quantified. The optical surfaces exposed to the effluent gases must be cleaned prior to performing the zero and span drift adjustments, except for systems using automatic zero adjustments. For systems using automatic zero adjustments, the optical surfaces must be cleaned when the cumulative automatic zero compensation exceeds 4 percent opacity.

(iii) The owner or operator must apply a method for producing a simulated zero opacity condition and an upscale (span) opacity condition using a certified neutral density filter or other related technique to produce a known obscuration of the light beam. All procedures applied must provide a system check of the analyzer internal optical surfaces and all electronic circuitry including the lamp and photodetector assembly.

(iv) Except during periods of system breakdowns, repairs, calibration checks, and zero and span adjustments, the COMS must be in continuous operation and must complete a minimum of one cycle of sampling and analyzing for each successive 10-second period and one cycle of data recording for each successive 6-minute period.

(v) The owner or operator must reduce all data from the COMS to 6minute averages. Six-minute opacity averages must be calculated from 36 or more data points equally spaced over each 6-minute period. Data recorded during periods of system breakdowns, repairs, calibration checks, and zero and span adjustments must not be included in the data averages. An arithmetic or integrated average of all data may be used.

(h) The owner or operator of each affected coal truck dump operation that commenced construction, reconstruction, or modification after April 28, 2008, must meet the requirements specified in paragraphs (h)(1) through (3) of this section.

(1) Conduct an initial performance test using Method 9 of appendix A-4 of this part according to the requirements in paragraphs (h)(1)(i) and (ii).

(i) Opacity readings shall be taken during the duration of three separate truck dump events. Each truck dump event commences when the truck bed begins to elevate and concludes when the truck bed returns to a horizontal position.

(ii) Compliance with the applicable opacity limit is determined by averaging all 15-second opacity readings made during the duration of three separate truck dump events.

(2) Conduct monthly visual observations of all process and control equipment. If any deficiencies are observed, the necessary maintenance must be performed as expeditiously as possible.

(3) Conduct a performance test using Method 9 of appendix A-4 of this part at least once every 5 calendar years for each affected facility.

§60.256 Continuous monitoring requirements.

(a) The owner or operator of each affected facility constructed, reconstructed, or modified on or before April 28, 2008, must meet the monitoring requirements specified in paragraphs (a)(1) and (2) of this section, as applicable to the affected facility.

(1) The owner or operator of any thermal dryer shall install, calibrate, maintain, and continuously operate monitoring devices as follows:

(i) A monitoring device for the measurement of the temperature of the gas stream at the exit of the thermal dryer on a continuous basis. The monitoring device is to be certified by the manufacturer to be accurate within ± 1.7 °C (± 3 °F).

(ii) For affected facilities that use wet scrubber emission control equipment:

(A) A monitoring device for the continuous measurement of the pressure loss through the venturi constriction of the control equipment. The monitoring device is to be certified by the manufacturer to be accurate within ± 1 inch water gauge.

(B) A monitoring device for the continuous measurement of the water supply pressure to the control equipment. The monitoring device is to be certified by the manufacturer to be accurate within ± 5 percent of design water supply pressure. The pressure sensor or tap must be located close to the water discharge point. The Administrator shall have discretion to grant requests for approval of alternative monitoring locations.

(2) All monitoring devices under paragraph (a) of this section are to be recalibrated annually in accordance with procedures under §60.13(b).

(b) The owner or operator of each affected facility constructed, reconstructed, or modified after April 28, 2008, that has one or more mechanical vents must install, calibrate, maintain, and continuously operate the monitoring devices specified in paragraphs (b)(1) through (3) of this section, as applicable to the mechanical vent and any control device installed on the vent.

(1) For mechanical vents with fabric filters (baghouses) with design controlled potential PM emissions rates of 25 Mg (28 tons) per year or more, a bag leak detection system according to the requirements in paragraph (c) of this section.

(2) For mechanical vents with wet scrubbers, monitoring devices according to the requirements in paragraphs (b)(2)(i) through (iv) of this section.

(i) A monitoring device for the continuous measurement of the pressure loss through the venturi constriction of the control equipment. The monitoring device is to be certified by the manufacturer to be accurate within ± 1 inch water gauge.

(ii) A monitoring device for the continuous measurement of the water supply flow rate to the control equipment. The monitoring device is to be certified by the manufacturer to be accurate within ± 5 percent of design water supply flow rate.

(iii) A monitoring device for the continuous measurement of the pH of the wet scrubber liquid. The monitoring device is to be certified by the manufacturer to be accurate within ±5 percent of design pH.

(iv) An average value for each monitoring parameter must be determined during each performance test. Each monitoring parameter must then be maintained within 10 percent of the value established during the most recent performance test on an operating day average basis. 40 CFR Ch. I (7–1–23 Edition)

(3) For mechanical vents with control equipment other than wet scrubbers, a monitoring device for the continuous measurement of the reagent injection flow rate to the control equipment, as applicable. The monitoring device is to be certified by the manufacturer to be accurate within ±5 percent of design injection flow rate. An average reagent injection flow rate value must be determined during each performance test. The reagent injection flow rate must then be maintained within 10 percent of the value established during the most recent performance test on an operating day average basis.

(c) Each bag leak detection system used to comply with provisions of this subpart must be installed, calibrated, maintained, and continuously operated according to the requirements in paragraphs (c)(1) through (3) of this section.

(1) The bag leak detection system must meet the specifications and requirements in paragraphs (c)(1)(i)through (viii) of this section.

(i) The bag leak detection system must be certified by the manufacturer to be capable of detecting PM emissions at concentrations of 1 milligram per dry standard cubic meter (mg/ dscm) (0.00044 grains per actual cubic foot (gr/acf)) or less.

(ii) The bag leak detection system sensor must provide output of relative PM loadings. The owner or operator shall continuously record the output from the bag leak detection system using electronic or other means (e.g., using a strip chart recorder or a data logger).

(iii) The bag leak detection system must be equipped with an alarm system that will sound when the system detects an increase in relative particulate loading over the alarm set point established according to paragraph (c)(1)(iv) of this section, and the alarm must be located such that it can be heard by the appropriate plant personnel.

(iv) In the initial adjustment of the bag leak detection system, the owner or operator must establish, at a minimum, the baseline output by adjusting the sensitivity (range) and the averaging period of the device, the alarm set points, and the alarm delay time.

(v) Following initial adjustment, the owner or operator must not adjust the averaging period, alarm set point, or alarm delay time without approval from the Administrator or delegated authority except as provided in paragraph (c)(2)(vi) of this section.

(vi) Once per quarter, the owner or operator may adjust the sensitivity of the bag leak detection system to account for seasonal effects, including temperature and humidity, according to the procedures identified in the sitespecific monitoring plan required by paragraph (c)(2) of this section.

(vii) The owner or operator must install the bag leak detection sensor downstream of the fabric filter.

(viii) Where multiple detectors are required, the system's instrumentation and alarm may be shared among detectors.

(2) The owner or operator must develop and submit to the Administrator or delegated authority for approval a site-specific monitoring plan for each bag leak detection system. This plan must be submitted to the Administrator or delegated authority 30 days prior to startup of the affected facility. The owner or operator must operate and maintain the bag leak detection system according to the site-specific monitoring plan at all times. Each monitoring plan must describe the items in paragraphs (c)(2)(i) through (vi) of this section.

(i) Installation of the bag leak detection system;

(ii) Initial and periodic adjustment of the bag leak detection system, including how the alarm set-point will be established;

(iii) Operation of the bag leak detection system, including quality assurance procedures;

(iv) How the bag leak detection system will be maintained, including a routine maintenance schedule and spare parts inventory list;

(v) How the bag leak detection system output will be recorded and stored; and

(vi) Corrective action procedures as specified in paragraph (c)(3) of this section. In approving the site-specific monitoring plan, the Administrator or delegated authority may allow the owner and operator more than 3 hours to alleviate a specific condition that causes an alarm if the owner or operator identifies in the monitoring plan this specific condition as one that could lead to an alarm, adequately explains why it is not feasible to alleviate this condition within 3 hours of the time the alarm occurs, and demonstrates that the requested time will ensure alleviation of this condition as expeditiously as practicable.

(3) For each bag leak detection system, the owner or operator must initiate procedures to determine the cause of every alarm within 1 hour of the alarm. Except as provided in paragraph (C)(2)(vi) of this section, the owner or operator must alleviate the cause of the alarm within 3 hours of the alarm by taking whatever corrective action(s) are necessary. Corrective actions may include, but are not limited to the following:

(i) Inspecting the fabric filter for air leaks, torn or broken bags or filter media, or any other condition that may cause an increase in PM emissions;

(ii) Sealing off defective bags or filter media;

(iii) Replacing defective bags or filter media or otherwise repairing the control device;

(iv) Sealing off a defective fabric filter compartment;

(v) Cleaning the bag leak detection system probe or otherwise repairing the bag leak detection system; or

(vi) Shutting down the process producing the PM emissions.

§60.257 Test methods and procedures.

(a) The owner or operator must determine compliance with the applicable opacity standards as specified in paragraphs (a)(1) through (3) of this section.

(1) Method 9 of appendix A-4 of this part and the procedures in 60.11 must be used to determine opacity, with the exceptions specified in paragraphs (a)(1)(i) and (ii).

(i) The duration of the Method 9 of appendix A-4 of this part performance test shall be 1 hour (ten 6-minute averages).

(ii) If, during the initial 30 minutes of the observation of a Method 9 of appendix A-4 of this part performance test, all of the 6-minute average opacity readings are less than or equal to half the applicable opacity limit, then the observation period may be reduced from 1 hour to 30 minutes.

(2) To determine opacity for fugitive coal dust emissions sources, the additional requirements specified in paragraphs (a)(2)(i) through (iii) must be used.

(i) The minimum distance between the observer and the emission source shall be 5.0 meters (16 feet), and the sun shall be oriented in the 140-degree sector of the back.

(ii) The observer shall select a position that minimizes interference from other fugitive coal dust emissions sources and make observations such that the line of vision is approximately perpendicular to the plume and wind direction.

(iii) The observer shall make opacity observations at the point of greatest opacity in that portion of the plume where condensed water vapor is not present. Water vapor is not considered a visible emission.

(3) A visible emissions observer may conduct visible emission observations for up to three fugitive, stack, or vent emission points within a 15-second interval if the following conditions specified in paragraphs (a)(3)(i) through (iii) of this section are met.

(i) No more than three emissions points may be read concurrently.

(ii) All three emissions points must be within a 70 degree viewing sector or angle in front of the observer such that the proper sun position can be maintained for all three points.

(iii) If an opacity reading for any one of the three emissions points is within 5 percent opacity from the applicable standard (excluding readings of zero opacity), then the observer must stop taking readings for the other two points and continue reading just that single point.

(b) The owner or operator must conduct all performance tests required by 60.8 to demonstrate compliance with the applicable emissions standards specified in 60.252 according to the requirements in 60.8 using the applicable test methods and procedures in paragraphs (b)(1) through (8) of this section. 40 CFR Ch. I (7–1–23 Edition)

(1) Method 1 or 1A of appendix A-4 of this part shall be used to select sampling port locations and the number of traverse points in each stack or duct. Sampling sites must be located at the outlet of the control device (or at the outlet of the emissions source if no control device is present) prior to any releases to the atmosphere.

(2) Method 2, 2A, 2C, 2D, 2F, or 2G of appendix A-4 of this part shall be used to determine the volumetric flow rate of the stack gas.

(3) Method 3, 3A, or 3B of appendix A-4 of this part shall be used to determine the dry molecular weight of the stack gas. The owner or operator may use ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses (incorporated by reference—see §60.17) as an alternative to Method 3B of appendix A-2 of this part.

(4) Method 4 of appendix A-4 of this part shall be used to determine the moisture content of the stack gas.

(5) Method 5, 5B or 5D of appendix A-4 of this part or Method 17 of appendix A-7 of this part shall be used to determine the PM concentration as follows:

(i) The sampling time and sample volume for each run shall be at least 60 minutes and 0.85 dscm (30 dscf). Sampling shall begin no less than 30 minutes after startup and shall terminate before shutdown procedures begin. A minimum of three valid test runs are needed to comprise a PM performance test.

(ii) Method 5 of appendix A of this part shall be used only to test emissions from affected facilities without wet flue gas desulfurization (FGD) systems.

(iii) Method 5B of appendix A of this part is to be used only after wet FGD systems.

(iv) Method 5D of appendix A-4 of this part shall be used for positive pressure fabric filters and other similar applications (*e.g.*, stub stacks and roof vents).

(v) Method 17 of appendix A-6 of this part may be used at facilities with or without wet scrubber systems provided the stack gas temperature does not exceed a temperature of 160 °C (320 °F). The procedures of sections 8.1 and 11.1 of Method 5B of appendix A-3 of this

part may be used in Method 17 of appendix A-6 of this part only if it is used after a wet FGD system. Do not use Method 17 of appendix A-6 of this part after wet FGD systems if the effluent is saturated or laden with water droplets.

(6) Method 6, 6A, or 6C of appendix A-4 of this part shall be used to determine the SO_2 concentration. A minimum of three valid test runs are needed to comprise an SO_2 performance test.

(7) Method 7 or 7E of appendix A–4 of this part shall be used to determine the NO_X concentration. A minimum of three valid test runs are needed to comprise an NO_X performance test.

(8) Method 10 of appendix A-4 of this part shall be used to determine the CO concentration. A minimum of three valid test runs are needed to comprise a CO performance test. CO performance tests are conducted concurrently (or within a 60-minute period) with NO_X performance tests.

§60.258 Reporting and recordkeeping.

(a) The owner or operator of a coal preparation and processing plant that commenced construction, reconstruction, or modification after April 28, 2008, shall maintain in a logbook (written or electronic) on-site and make it available upon request. The logbook shall record the following:

(1) The manufacturer's recommended maintenance procedures and the date and time of any maintenance and inspection activities and the results of those activities. Any variance from manufacturer recommendation, if any, shall be noted.

(2) The date and time of periodic coal preparation and processing plant visual observations, noting those sources with visible emissions along with corrective actions taken to reduce visible emissions. Results from the actions shall be noted.

(3) The amount and type of coal processed each calendar month.

(4) The amount of chemical stabilizer or water purchased for use in the coal preparation and processing plant.

(5) Monthly certification that the dust suppressant systems were operational when any coal was processed and that manufacturer's recommendations were followed for all control systems. Any variance from the manufacturer's recommendations, if any, shall be noted.

(6) Monthly certification that the fugitive coal dust emissions control plan was implemented as described. Any variance from the plan, if any, shall be noted. A copy of the applicable fugitive coal dust emissions control plan and any letters from the Administrator providing approval of any alternative control measures shall be maintained with the logbook. Any actions, e.g., objections, to the plan and any actions relative to the alternative control measures, e.g., approvals, shall be noted in the logbook as well.

(7) For each bag leak detection system, the owner or operator must keep the records specified in paragraphs (a)(7)(i) through (iii) of this section.

(i) Records of the bag leak detection system output;

(ii) Records of bag leak detection system adjustments, including the date and time of the adjustment, the initial bag leak detection system settings, and the final bag leak detection settings; and

(iii) The date and time of all bag leak detection system alarms, the time that procedures to determine the cause of the alarm were initiated, the cause of the alarm, an explanation of the actions taken, the date and time the cause of the alarm was alleviated, and whether the cause of the alarm was alleviated within 3 hours of the alarm.

(8) A copy of any applicable monitoring plan for a digital opacity compliance system and monthly certification that the plan was implemented as described. Any variance from plan, if any, shall be noted.

(9) During a performance test of a wet scrubber, and each operating day thereafter, the owner or operator shall record the measurements of the scrubber pressure loss, water supply flow rate, and pH of the wet scrubber liquid.

(10) During a performance test of control equipment other than a wet scrubber, and each operating day thereafter, the owner or operator shall record the measurements of the reagent injection flow rate, as applicable.

(b) For the purpose of reports required under section 60.7(c), any owner operator subject to the provisions of this subpart also shall report semiannually periods of excess emissions as follow:

(1) The owner or operator of an affected facility with a wet scrubber shall submit semiannual reports to the Administrator or delegated authority of occurrences when the measurements of the scrubber pressure loss, water supply flow rate, or pH of the wet scrubber liquid vary by more than 10 percent from the average determined during the most recent performance test.

(2) The owner or operator of an affected facility with control equipment other than a wet scrubber shall submit semiannual reports to the Administrator or delegated authority of occurrences when the measurements of the reagent injection flow rate, as applicable, vary by more than 10 percent from the average determined during the most recent performance test.

(3) All 6-minute average opacities that exceed the applicable standard.

(c) The owner or operator of an affected facility shall submit the results of initial performance tests to the Administrator or delegated authority, consistent with the provisions of section 60.8. The owner or operator who elects to comply with the reduced performance testing provisions of sections 60.255(c) or (d) shall include in the performance test report identification of each affected facility that will be subject to the reduced testing. The owner or operator electing to comply with section 60.255(d) shall also include information which demonstrates that the control devices are identical.

(d) After July 1, 2011, within 60 days after the date of completing each performance evaluation conducted to demonstrate compliance with this subpart, the owner or operator of the affected facility must submit the test data to EPA by successfully entering the data electronically into EPA's WebFIRE data base available at http:// cfpub.epa.gov/oarweb/index.cfm?action = fire.main. For performance tests that cannot be entered into WebFIRE (i.e., Method 9 of appendix A-4 of this part opacity performance tests) the owner or operator of the affected facility must mail a summary copy to United

40 CFR Ch. I (7–1–23 Edition)

States Environmental Protection Agency; Energy Strategies Group; 109 TW Alexander DR; mail code: D243-01; RTP, NC 27711.

Subpart Z—Standards of Performance for Ferroalloy Production Facilities

SOURCE: 41 FR 18501, May 4, 1976, unless otherwise noted.

§60.260 Applicability and designation of affected facility.

(a) The provisions of this subpart are applicable to the following affected facilities: Electric submerged arc furnaces which produce silicon metal, ferrosilicon, calcium silicon. silicomanganese zirconium. ferrochrome silicon, silvery iron, high-carbon ferrochrome, charge chrome, standard ferromanganese, silicomanganese, ferromanganese silicon, or calcium carbide; and dust-handling equipment.

(b) Any facility under paragraph (a) of this section that commences construction or modification after October 21, 1974, is subject to the requirements of this subpart.

[42 FR 37938, July 25, 1977]

§60.261 Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Act and in subpart A of this part.

(a) *Electric submerged arc furnace* means any furnace wherein electrical energy is converted to heat energy by transmission of current between electrodes partially submerged in the furnace charge.

(b) *Furnace charge* means any material introduced into the electric submerged arc furnace, and may consist of, but is not limited to, ores, slag, carbonaceous material, and limestone.

(c) *Product change* means any change in the composition of the furnace charge that would cause the electric submerged arc furnace to become subject to a different mass standard applicable under this subpart. Appendix E

§60.480

AUTHENTICATED U.S. GOVERNMENT INFORMATION

> standard, the Administrator will establish and promulgate in the FEDERAL REGISTER an opacity standard for the blowing still that will be the opacity standard when fuel oil is used to fire the afterburner. When the afterburner is fired with natural gas, the zero percent opacity remains the applicable opacity standard.

> [54 FR 6677, Feb. 14, 1989, as amended 54 FR 27016, June 27, 1989; 65 FR 61762, Oct. 17, 2000]

Subpart VV—Standards of Performance for Equipment Leaks of VOC in the Synthetic Organic Chemicals Manufacturing Industry for which Construction, Reconstruction, or Modification Commenced After January 5, 1981, and on or Before November 7, 2006

SOURCE: 48 FR 48335, Oct. 18, 1983, unless otherwise noted.

§60.480 Applicability and designation of affected facility.

(a)(1) The provisions of this subpart apply to affected facilities in the synthetic organic chemicals manufacturing industry.

(2) The group of all equipment (defined in §60.481) within a process unit is an affected facility.

(b) Any affected facility under paragraph (a) of this section that commences construction, reconstruction, or modification after January 5, 1981, and on or before November 7, 2006, shall be subject to the requirements of this subpart.

(c) Addition or replacement of equipment for the purpose of process improvement which is accomplished without a capital expenditure shall not by itself be considered a modification under this subpart.

(d)(1) If an owner or operator applies for one or more of the exemptions in this paragraph, then the owner or operator shall maintain records as required in 60.486(i).

(2) Any affected facility that has the design capacity to produce less than 1,000 Mg/yr (1,102 ton/yr) of a chemical listed in §60.489 is exempt from §§60.482-1 through 60.482-10.

40 CFR Ch. I (7–1–23 Edition)

(3) If an affected facility produces heavy liquid chemicals only from heavy liquid feed or raw materials, then it is exempt from §§60.482–1 through 60.482–10.

(4) Any affected facility that produces beverage alcohol is exempt from §§ 60.482-1 through 60.482-10.

(5) Any affected facility that has no equipment in volatile organic compounds (VOC) service is exempt from §§ 60.482-1 through 60.482-10.

(e) Alternative means of compliance—(1) Option to comply with part 65. (i) Owners or operators may choose to comply with the provisions of 40 CFR part 65, subpart F, to satisfy the requirements of \S 60.482 through 60.487 for an affected facility. When choosing to comply with 40 CFR part 65, subpart F, the requirements of \S 60.485(d), (e), and (f) and 60.486(i) and (j) still apply. Other provisions applying to an owner or operator who chooses to comply with 40 CFR part 65 are provided in 40 CFR 65.1.

(ii) Part 60, subpart A. Owners or operators who choose to comply with 40 CFR part 65, subpart F must also comply with §§ 60.1, 60.2, 60.5, 60.6, 60.7(a)(1) and (4), 60.14, 60.15, and 60.16 for that equipment. All sections and paragraphs of subpart A of this part that are not mentioned in this paragraph (e)(1)(ii)do not apply to owners and operators of equipment subject to this subpart complying with 40 CFR part 65, subpart F, except that provisions required to be met prior to implementing 40 CFR part 65 still apply. Owners and operators who choose to comply with 40 CFR part 65, subpart F, must comply with 40 CFR part 65, subpart A.

(2) Subpart VVa. Owners or operators may choose to comply with the provisions of subpart VVa of this part 60 to satisfy the requirements of this subpart VV for an affected facility.

(f) Stay of standards. Owners or operators are not required to comply with the definition of "process unit" in §60.481 and the requirements in §60.482– 1(g) of this subpart until the EPA takes final action to require compliance and publishes a document in the FEDERAL REGISTER. While the definition of "process unit" is stayed, owners or operators should use the following definition:

Process unit means components assembled to produce, as intermediate or final products, one or more of the chemicals listed in §60.489 of this part. A process unit can operate independently if supplied with sufficient feed or raw materials and sufficient storage facilities for the product.

[48 FR 48335, Oct. 18, 1983, as amended at 49 FR 22607, May 30, 1984; 65 FR 61762, Oct. 17, 2000; 65 FR 78276, Dec. 14, 2000; 72 FR 64879, Nov. 16, 2007, 73 FR 31379, June 2, 2008; 73 FR 31375, June 2, 2008]

§60.481 Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Act or in subpart A of part 60, and the following terms shall have the specific meanings given them.

Capital expenditure means, in addition to the definition in 40 CFR 60.2, an expenditure for a physical or operational change to an existing facility that:

(a) Exceeds P, the product of the facility's replacement cost, R, and an adjusted annual asset guideline repair allowance, A, as reflected by the following equation: $P = R \times A$, where

(1) The adjusted annual asset guideline repair allowance, A, is the product of the percent of the replacement cost, Y, and the applicable basic annual asset guideline repair allowance, B, divided by 100 as reflected by the following equation:

$\mathbf{A} = \mathbf{Y} \times (\mathbf{B} \div 100);$

(2) The percent Y is determined from the following equation: Y = 1.0 - 0.575 log X, where X is 1982 minus the year of construction: and

(3) The applicable basic annual asset guideline repair allowance, B, is selected from the following table consistent with the applicable subpart:

TABLE FOR DETER	VALDE
FOR B	

Su⊡part appli⊡a⊡e to fa⊡lity	Value of B to ⊑e u⊑e⊒ in e⊡uation
VV	12.5 12.5
	4.5

Closed-loop system means an enclosed system that returns process fluid to the process.

Closed-purge system means a system or combination of systems and portable containers to capture purged liquids. Containers for purged liquids must be covered or closed when not being filled or emptied.

Closed vent system means a system that is not open to the atmosphere and that is composed of hard-piping, ductwork, connections, and, if necessary, flow-inducing devices that transport gas or vapor from a piece or pieces of equipment to a control device or back to a process.

Connector means flanged, screwed, or other joined fittings used to connect two pipe lines or a pipe line and a piece of process equipment or that close an opening in a pipe that could be connected to another pipe. Joined fittings welded completely around the circumference of the interface are not considered connectors for the purpose of this subpart.

Control device means an enclosed combustion device, vapor recovery system, or flare.

Distance piece means an open or enclosed casing through which the piston rod travels, separating the compressor cylinder from the crankcase.

Double block and bleed system means two block valves connected in series with a bleed valve or line that can vent the line between the two block valves.

Duct work means a conveyance system such as those commonly used for heating and ventilation systems. It is often made of sheet metal and often has sections connected by screws or crimping. Hard-piping is not ductwork.

Equipment means each pump, compressor, pressure relief device, sampling connection system, open-ended valve or line, valve, and flange or other connector in VOC service and any devices or systems required by this subpart.

First attempt at repair means to take action for the purpose of stopping or reducing leakage of organic material to the atmosphere using best practices.

Fuel gas means gases that are combusted to derive useful work or heat.

Fuel gas system means the offsite and onsite piping and flow and pressure

control system that gathers gaseous stream(s) generated by onsite operations, may blend them with other sources of gas, and transports the gaseous stream for use as fuel gas in combustion devices or in-process combustion equipment, such as furnaces and gas turbines, either singly or in combination.

Hard-piping means pipe or tubing that is manufactured and properly installed using good engineering judgment and standards such as ASME B31.3, Process Piping (available from the American Society of Mechanical Engineers, PO Box 2300, Fairfield, NJ 07007-2300).

In gas/vapor service means that the piece of equipment contains process fluid that is in the gaseous state at operating conditions.

In heavy liquid service means that the piece of equipment is not in gas/vapor service or in light liquid service.

In light liquid service means that the piece of equipment contains a liquid that meets the conditions specified in $\S60.485(e)$.

In-situ sampling systems means nonextractive samplers or in-line samplers.

In vacuum service means that equipment is operating at an internal pressure which is at least 5 kilopascals (kPa)(0.7 psia) below ambient pressure.

In VOC service means that the piece of equipment contains or contacts a process fluid that is at least 10 percent VOC by weight. (The provisions of $\S60.485(d)$ specify how to determine that a piece of equipment is not in VOC service.)

Liquids dripping means any visible leakage from the seal including spraying, misting, clouding, and ice formation.

Open-ended valve or line means any valve, except safety relief valves, having one side of the valve seat in contact with process fluid and one side open to the atmosphere, either directly or through open piping.

Pressure release means the emission of materials resulting from system pressure being greater than set pressure of the pressure relief device.

Process improvement means routine changes made for safety and occupational health requirements, for energy 40 CFR Ch. I (7-1-23 Edition)

savings, for better utility, for ease of maintenance and operation, for correction of design deficiencies, for bottleneck removal, for changing product requirements, or for environmental control.

Process unit means the components assembled and connected by pipes or ducts to process raw materials and to produce, as intermediate or final products, one or more of the chemicals listed in §60.489. A process unit can operate independently if supplied with sufficient feed or raw materials and sufficient storage facilities for the product. For the purpose of this subpart, process unit includes any feed, intermediate and final product storage vessels (except as specified in §60.482-1(g)), product transfer racks, and connected ducts and piping. A process unit includes all equipment as defined in this subpart.

Process unit shutdown means a work practice or operational procedure that stops production from a process unit or part of a process unit during which it is technically feasible to clear process material from a process unit or part of a process unit consistent with safety constraints and during which repairs can be accomplished. The following are not considered process unit shutdowns:

(1) An unscheduled work practice or operational procedure that stops production from a process unit or part of a process unit for less than 24 hours.

(2) An unscheduled work practice or operational procedure that would stop production from a process unit or part of a process unit for a shorter period of time than would be required to clear the process unit or part of the process unit of materials and start up the unit, and would result in greater emissions than delay of repair of leaking components until the next scheduled process unit shutdown.

(3) The use of spare equipment and technically feasible bypassing of equipment without stopping production.

Quarter means a 3-month period; the first quarter concludes on the last day of the last full month during the 180 days following initial startup.

Repaired means that equipment is adjusted, or otherwise altered, in order to eliminate a leak as defined in the applicable sections of this subpart and,

except for leaks identified in accordance with \$ 60.482-2(b)(2)(ii) and (d)(6)(ii) and (iii), 60.482-3(f), and 60.482-10(f)(1)(ii), is re-monitored as specified in \$ 60.485(b) to verify that emissions from the equipment are below the applicable leak definition.

Replacement cost means the capital needed to purchase all the depreciable components in a facility.

Sampling connection system means an assembly of equipment within a process unit used during periods of representative operation to take samples of the process fluid. Equipment used to take nonroutine grab samples is not considered a sampling connection system.

Sensor means a device that measures a physical quantity or the change in a physical quantity such as temperature, pressure, flow rate, pH, or liquid level.

Storage vessel means a tank or other vessel that is used to store organic liquids that are used in the process as raw material feedstocks, produced as intermediates or final products, or generated as wastes. Storage vessel does not include vessels permanently attached to motor vehicles, such as trucks, railcars, barges, or ships.

Synthetic organic chemicals manufacturing industry means the industry that produces, as intermediates or final products, one or more of the chemicals listed in §60.489.

Transfer rack means the collection of loading arms and loading hoses, at a single loading rack, that are used to fill tank trucks and/or railcars with organic liquids.

Volatile organic compounds or VOC means, for the purposes of this subpart, any reactive organic compounds as defined in §60.2 Definitions.

[48 FR 48335, Oct. 18, 1983, as amended at 49
FR 22607, May 30, 1984; 49 FR 26738, June 29, 1984; 60 FR 43258, Aug. 18, 1995; 65 FR 61762, Oct. 17, 2000; 65 FR 78276, Dec. 14, 2000; 72 FR 64879, Nov. 16, 2007]

EFFECTIVE DATE NOTE: At 73 FR 31375, June 2, 2008, in §60.481, the definition of "process unit" was stayed until further notice.

§60.482-1 Standards: General.

(a) Each owner or operator subject to the provisions of this subpart shall demonstrate compliance with the requirements of §§ 60.482–1 through 60.482– 10 or §60.480(e) for all equipment within 180 days of initial startup.

(b) Compliance with §§ 60.482-1 to 60.482-10 will be determined by review of records and reports, review of performance test results, and inspection using the methods and procedures specified in § 60.485.

(c)(1) An owner or operator may request a determination of equivalence of a means of emission limitation to the requirements of \$ 60.482–2, 60.482–3, 60.482–5, 60.482–6, 60.482–7, 60.482–8, and 60.482–10 as provided in \$ 60.484.

(2) If the Administrator makes a determination that a means of emission limitation is at least equivalent to the requirements of 60.482-2, 60.482-3, 60.482-5, 60.482-6, 60.482-7, 60.482-8, or 60.482-10, an owner or operator shall comply with the requirements of that determination.

(d) Equipment that is in vacuum service is excluded from the requirements of \$ 60.482-2 to 60.482-10 if it is identified as required in \$ 60.486(e)(5).

(e) Equipment that an owner or operator designates as being in VOC service less than 300 hours (hr)/yr is excluded from the requirements of \$60.482-2through 60.482-10 if it is identified as required in \$60.486(e)(6) and it meets any of the conditions specified in paragraphs (e)(1) through (3) of this section.

(1) The equipment is in VOC service only during startup and shutdown, excluding startup and shutdown between batches of the same campaign for a batch process.

(2) The equipment is in VOC service only during process malfunctions or other emergencies.

(3) The equipment is backup equipment that is in VOC service only when the primary equipment is out of service.

(f)(1) If a dedicated batch process unit operates less than 365 days during a year, an owner or operator may monitor to detect leaks from pumps and valves at the frequency specified in the following table instead of monitoring as specified in §§ 60.482-2, 60.482-7, and 60.483-2:

Operating	E_ui_alent monitoring fre_uen_y time in u_e		
of hour⊡ ⊡ur⊡ ing year⊡	□ onthly	Quarterly	Semiannually
0 to <25	Quarterly	Annually	Annually.

§60.482-2

Operating	E lui alent monitoring fre luen y time in u e		y time in u⊡e
of hour⊡ ⊡ur⊡ ing year⊡	□ onthly	Quarterly	Semiannually
25 to <50	Quarterly	Semiannually	Annually.
50 to <75	Bimonthly	Three ⊡uar⊡ ter⊡	Semiannu⊡ ally.
75 to 100	□ onthly	Quarterly	Semiannu⊡ ally.

(2) Pumps and valves that are shared among two or more batch process units that are subject to this subpart may be monitored at the frequencies specified in paragraph (f)(1) of this section, provided the operating time of all such process units is considered.

(3) The monitoring frequencies specified in paragraph (f)(1) of this section are not requirements for monitoring at specific intervals and can be adjusted to accommodate process operations. An owner or operator may monitor at any time during the specified monitoring period (e.g., month, quarter, year), provided the monitoring is conducted at a reasonable interval after completion of the last monitoring campaign. Reasonable intervals are defined in paragraphs (f)(3)(i) through (iv) of this section.

(i) When monitoring is conducted quarterly, monitoring events must be separated by at least 30 calendar days.

(ii) When monitoring is conducted semiannually (*i.e.*, once every 2 quarters), monitoring events must be separated by at least 60 calendar days.

(iii) When monitoring is conducted in 3 quarters per year, monitoring events must be separated by at least 90 calendar days.

(iv) When monitoring is conducted annually, monitoring events must be separated by at least 120 calendar days.

(g) If the storage vessel is shared with multiple process units, the process unit with the greatest annual amount of stored materials (predominant use) is the process unit the storage vessel is assigned to. If the storage vessel is shared equally among process units, and one of the process units has equipment subject to subpart VVa of this part, the storage vessel is assigned to that process unit. If the storage vessel is shared equally among process units, none of which have equipment subject to subpart VVa of this part, the storage vessel is assigned to any process unit subject to this subpart. If the predominant use of the storage vessel

40 CFR Ch. I (7–1–23 Edition)

varies from year to year, then the owner or operator must estimate the predominant use initially and reassess every 3 years. The owner or operator must keep records of the information and supporting calculations that show how predominant use is determined. All equipment on the storage vessel must be monitored when in VOC service.

[48 FR 48335, Oct. 18, 1983, as amended at 49 FR 22608, May 30, 1984; 65 FR 78276, Dec. 14, 2000; 72 FR 64880, Nov. 16, 2007]

EFFECTIVE DATE NOTE: At 73 FR 31375, June 2, 2008, in 60.482-1, paragraph (g) was stayed until further notice.

§60.482–2 Standards: Pumps in light liquid service.

(a)(1) Each pump in light liquid service shall be monitored monthly to detect leaks by the methods specified in $\S60.485(b)$, except as provided in $\S60.482-$ 1(c) and (f) and paragraphs (d), (e), and (f) of this section. A pump that begins operation in light liquid service after the initial startup date for the process unit must be monitored for the first time within 30 days after the end of its startup period, except for a pump that replaces a leaking pump and except as provided in $\S60.482-1(c)$ and (f) and paragraphs (d), (e), and (f) of this section.

(2) Each pump in light liquid service shall be checked by visual inspection each calendar week for indications of liquids dripping from the pump seal, except as provided in §60.482–1(f).

(b)(1) If an instrument reading of 10,000 ppm or greater is measured, a leak is detected.

(2) If there are indications of liquids dripping from the pump seal, the owner or operator shall follow the procedure specified in either paragraph (b)(2)(i) or (ii) of this section. This requirement does not apply to a pump that was monitored after a previous weekly inspection if the instrument reading for that monitoring event was less than 10,000 ppm and the pump was not repaired since that monitoring event.

(i) Monitor the pump within 5 days as specified in §60.485(b). If an instrument reading of 10,000 ppm or greater is measured, a leak is detected. The leak shall be repaired using the procedures in paragraph (c) of this section.

(ii) Designate the visual indications of liquids dripping as a leak, and repair the leak within 15 days of detection by eliminating the visual indications of liquids dripping.

(c)(1) When a leak is detected, it shall be repaired as soon as practicable, but not later than 15 calendar days after it is detected, except as provided in §60.482-9.

(2) A first attempt at repair shall be made no later than 5 calendar days after each leak is detected. First attempts at repair include, but are not limited to, the practices described in paragraphs (c)(2)(i) and (ii) of this section, where practicable.

(i) Tightening the packing gland nuts;

(ii) Ensuring that the seal flush is operating at design pressure and temperature.

(d) Each pump equipped with a dual mechanical seal system that includes a barrier fluid system is exempt from the requirements of paragraph (a) of this section, provided the requirements specified in paragraphs (d)(1) through (6) of this section are met.

(1) Each dual mechanical seal system is—

(i) Operated with the barrier fluid at a pressure that is at all times greater than the pump stuffing box pressure; or

(ii) Equipped with a barrier fluid degassing reservoir that is routed to a process or fuel gas system or connected by a closed vent system to a control device that complies with the requirements of §60.482–10; or

(iii) Equipped with a system that purges the barrier fluid into a process stream with zero VOC emissions to the atmosphere.

(2) The barrier fluid system is in heavy liquid service or is not in VOC service.

(3) Each barrier fluid system is equipped with a sensor that will detect failure of the seal system, the barrier fluid system, or both.

(4)(i) Each pump is checked by visual inspection, each calendar week, for indications of liquids dripping from the pump seals.

(ii) If there are indications of liquids dripping from the pump seal at the time of the weekly inspection, the owner or operator shall follow the procedure specified in either paragraph (d)(4)(ii)(A) or (B) of this section.

(A) Monitor the pump within 5 days as specified in §60.485(b) to determine if there is a leak of VOC in the barrier fluid. If an instrument reading of 10,000 ppm or greater is measured, a leak is detected.

(B) Designate the visual indications of liquids dripping as a leak.

(5)(i) Each sensor as described in paragraph (d)(3) of this section is checked daily or is equipped with an audible alarm.

(ii) The owner or operator determines, based on design considerations and operating experience, a criterion that indicates failure of the seal system, the barrier fluid system, or both.

(iii) If the sensor indicates failure of the seal system, the barrier fluid system, or both, based on the criterion established in paragraph (d)(5)(ii) of this section, a leak is detected.

(6)(i) When a leak is detected pursuant to paragraph (d)(4)(ii)(A) of this section, it shall be repaired as specified in paragraph (c) of this section.

(ii) A leak detected pursuant to paragraph (d)(5)(iii) of this section shall be repaired within 15 days of detection by eliminating the conditions that activated the sensor.

(iii) A designated leak pursuant to paragraph (d)(4)(ii)(B) of this section shall be repaired within 15 days of detection by eliminating visual indications of liquids dripping.

(e) Any pump that is designated, as described in 60.486(e)(1) and (2), for no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, is exempt from the requirements of paragraphs (a), (c), and (d) of this section if the pump:

(1) Has no externally actuated shaft penetrating the pump housing,

(2) Is demonstrated to be operating with no detectable emissions as indicated by an instrument reading of less than 500 ppm above background as measured by the methods specified in $\S60.485(c)$, and

(3) Is tested for compliance with paragraph (e)(2) of this section initially upon designation, annually, and at other times requested by the Administrator.

(f) If any pump is equipped with a closed vent system capable of capturing and transporting any leakage from the seal or seals to a process or to a fuel gas system or to a control device that complies with the requirements of §60.482–10, it is exempt from paragraphs (a) through (e) of this section.

(g) Any pump that is designated, as described in 60.486(f)(1), as an unsafeto-monitor pump is exempt from the monitoring and inspection requirements of paragraphs (a) and (d)(4) through (6) of this section if:

(1) The owner or operator of the pump demonstrates that the pump is unsafe-to-monitor because monitoring personnel would be exposed to an immediate danger as a consequence of complying with paragraph (a) of this section; and

(2) The owner or operator of the pump has a written plan that requires monitoring of the pump as frequently as practicable during safe-to-monitor times but not more frequently than the periodic monitoring schedule otherwise applicable, and repair of the equipment according to the procedures in paragraph (c) of this section if a leak is detected.

(h) Any pump that is located within the boundary of an unmanned plant site is exempt from the weekly visual inspection requirement of paragraphs (a)(2) and (d)(4) of this section, and the daily requirements of paragraph (d)(5)of this section, provided that each pump is visually inspected as often as practicable and at least monthly.

[48 FR 48335, Oct. 18, 1983, as amended at 65 FR 61762, Oct. 17, 2000; 65 FR 78276, Dec. 14, 2000; 72 FR 64880, Nov. 16, 2007]

§60.482-3 Standards: Compressors.

(a) Each compressor shall be equipped with a seal system that includes a barrier fluid system and that prevents leakage of VOC to the atmosphere, except as provided in 60.482-1(c)and paragraphs (h), (i), and (j) of this section.

(b) Each compressor seal system as required in paragraph (a) shall be:

(1) Operated with the barrier fluid at a pressure that is greater than the compressor stuffing box pressure; or

(2) Equipped with a barrier fluid system degassing reservoir that is routed 40 CFR Ch. I (7–1–23 Edition)

to a process or fuel gas system or connected by a closed vent system to a control device that complies with the requirements of §60.482–10; or

(3) Equipped with a system that purges the barrier fluid into a process stream with zero VOC emissions to the atmosphere.

(c) The barrier fluid system shall be in heavy liquid service or shall not be in VOC service.

(d) Each barrier fluid system as described in paragraph (a) shall be equipped with a sensor that will detect failure of the seal system, barrier fluid system, or both.

(e)(1) Each sensor as required in paragraph (d) shall be checked daily or shall be equipped with an audible alarm.

(2) The owner or operator shall determine, based on design considerations and operating experience, a criterion that indicates failure of the seal system, the barrier fluid system, or both.

(f) If the sensor indicates failure of the seal system, the barrier system, or both based on the criterion determined under paragraph (e)(2), a leak is detected.

(g)(1) When a leak is detected, it shall be repaired as soon as practicable, but not later than 15 calendar days after it is detected, except as provided in 60.482-9.

(2) A first attempt at repair shall be made no later than 5 calendar days after each leak is detected.

(h) A compressor is exempt from the requirements of paragraphs (a) and (b) of this section, if it is equipped with a closed vent system to capture and transport leakage from the compressor drive shaft back to a process or fuel gas system or to a control device that complies with the requirements of §60.482–10, except as provided in paragraph (i) of this section.

(i) Any compressor that is designated, as described in 60.486(e) (1) and (2), for no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, is exempt from the requirements of paragraphs (a)-(h) if the compressor:

(1) Is demonstrated to be operating with no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, as

measured by the methods specified in (0.485(c)); and

(2) Is tested for compliance with paragraph (i)(1) of this section initially upon designation, annually, and at other times requested by the Administrator.

(j) Any existing reciprocating compressor in a process unit which becomes an affected facility under provisions of §60.14 or §60.15 is exempt from paragraphs (a) through (e) and (h) of this section, provided the owner or operator demonstrates that recasting the distance piece or replacing the compressor are the only options available to bring the compressor into compliance with the provisions of paragraphs (a) through (e) and (h) of this section.

[48 FR 48335, Oct. 18, 1983, as amended at 65 FR 61762, Oct. 17, 2000; 65 FR 78277, Dec. 14, 2000; 72 FR 64881, Nov. 16, 2007]

§60.482-4 Standards: Pressure relief devices in gas/vapor service.

(a) Except during pressure releases, each pressure relief device in gas/vapor service shall be operated with no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, as determined by the methods specified in §60.485(c).

(b)(1) After each pressure release, the pressure relief device shall be returned to a condition of no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, as soon as practicable, but no later than 5 calendar days after the pressure release, except as provided in §60.482–9.

(2) No later than 5 calendar days after the pressure release, the pressure relief device shall be monitored to confirm the conditions of no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, by the methods specified in 60.485(c).

(c) Any pressure relief device that is routed to a process or fuel gas system or equipped with a closed vent system capable of capturing and transporting leakage through the pressure relief device to a control device as described in $\S60.482-10$ is exempted from the requirements of paragraphs (a) and (b) of this section. (d)(1) Any pressure relief device that is equipped with a rupture disk upstream of the pressure relief device is exempt from the requirements of paragraphs (a) and (b) of this section, provided the owner or operator complies with the requirements in paragraph (d)(2) of this section.

(2) After each pressure release, a new rupture disk shall be installed upstream of the pressure relief device as soon as practicable, but no later than 5 calendar days after each pressure release, except as provided in §60.482–9.

[48 FR 48335, Oct. 18, 1983, as amended at 65 FR 61762, Oct. 17, 2000; 65 FR 78277, Dec. 14, 2000]

§60.482–5 Standards: Sampling connection systems.

(a) Each sampling connection system shall be equipped with a closed-purge, closed-loop, or closed-vent system, except as provided in 60.482-1(c) and paragraph (c) of this section.

(b) Each closed-purge, closed-loop, or closed-vent system as required in paragraph (a) of this section shall comply with the requirements specified in paragraphs (b)(1) through (4) of this section.

(1) Gases displaced during filling of the sample container are not required to be collected or captured.

(2) Containers that are part of a closed-purge system must be covered or closed when not being filled or emptied.

(3) Gases remaining in the tubing or piping between the closed-purge system valve(s) and sample container valve(s) after the valves are closed and the sample container is disconnected are not required to be collected or captured.

(4) Each closed-purge, closed-loop, or closed-vent system shall be designed and operated to meet requirements in either paragraph (b)(4)(i), (ii), (ii), or (iv) of this section.

(i) Return the purged process fluid directly to the process line.

(ii) Collect and recycle the purged process fluid to a process.

(iii) Capture and transport all the purged process fluid to a control device that complies with the requirements of 60.482-10.

§60.482-6

(iv) Collect, store, and transport the purged process fluid to any of the following systems or facilities:

(A) A waste management unit as defined in §63.111, if the waste management unit is subject to and operated in compliance with the provisions of 40 CFR part 63, subpart G, applicable to Group 1 wastewater streams;

(B) A treatment, storage, or disposal facility subject to regulation under 40 CFR part 262, 264, 265, or 266;

(C) A facility permitted, licensed, or registered by a state to manage municipal or industrial solid waste, if the process fluids are not hazardous waste as defined in 40 CFR part 261;

(D) A waste management unit subject to and operated in compliance with the treatment requirements of §61.348(a), provided all waste management units that collect, store, or transport the purged process fluid to the treatment unit are subject to and operated in compliance with the management requirements of §§61.343 through 61.347; or

(E) A device used to burn off-specification used oil for energy recovery in accordance with 40 CFR part 279, subpart G, provided the purged process fluid is not hazardous waste as defined in 40 CFR part 261.

(c) In situ sampling systems and sampling systems without purges are exempt from the requirements of paragraphs (a) and (b) of this section.

[60 FR 43258, Aug. 18, 1995, as amended at 65 FR 61762, Oct. 17, 2000; 65 FR 78277, Dec. 14, 2000; 72 FR 64881, Nov. 16, 2007]

\$60.482-6 Standards: Open-ended valves or lines.

(a)(1) Each open-ended valve or line shall be equipped with a cap, blind flange, plug, or a second valve, except as provided in 60.482-1(c) and paragraphs (d) and (e) of this section.

(2) The cap, blind flange, plug, or second valve shall seal the open end at all times except during operations requiring process fluid flow through the open-ended valve or line.

(b) Each open-ended valve or line equipped with a second valve shall be operated in a manner such that the valve on the process fluid end is closed before the second valve is closed.

40 CFR Ch. I (7–1–23 Edition)

(c) When a double block-and-bleed system is being used, the bleed valve or line may remain open during operations that require venting the line between the block valves but shall comply with paragraph (a) at all other times.

(d) Open-ended valves or lines in an emergency shutdown system which are designed to open automatically in the event of a process upset are exempt from the requirements of paragraphs (a), (b) and (c) of this section.

(e) Open-ended valves or lines containing materials which would autocatalytically polymerize or would present an explosion, serious overpressure, or other safety hazard if capped or equipped with a double block and bleed system as specified in paragraphs (a) through (c) of this section are exempt from the requirements of paragraphs (a) through (c) of this section.

[48 FR 48335, Oct. 18, 1983, as amended at 49 FR 22607, May 30, 1984; 65 FR 78277, Dec. 14, 2000; 72 FR 64881, Nov. 16, 2007]

§60.482–7 Standards: Valves in gas/ vapor service and in light liquid service.

(a)(1) Each valve shall be monitored monthly to detect leaks by the methods specified in 60.485(b) and shall comply with paragraphs (b) through (e) of this section, except as provided in paragraphs (f), (g), and (h) of this section, 60.482-1(c) and (f), and 80.483-1 and 60.483-2.

(2) A valve that begins operation in gas/vapor service or light liquid service after the initial startup date for the process unit must be monitored according to paragraphs (a)(2)(i) or (ii), except for a valve that replaces a leaking valve and except as provided in paragraphs (f), (g), and (h) of this section, §60.482–1(c), and §§60.483–1 and 60.483–2.

(i) Monitor the valve as in paragraph (a)(1) of this section. The valve must be monitored for the first time within 30 days after the end of its startup period to ensure proper installation.

(ii) If the valves on the process unit are monitored in accordance with §60.483-1 or §60.483-2, count the new valve as leaking when calculating the percentage of valves leaking as described in §60.483-2(b)(5). If less than 2.0

percent of the valves are leaking for that process unit, the valve must be monitored for the first time during the next scheduled monitoring event for existing valves in the process unit or within 90 days, whichever comes first.

(b) If an instrument reading of 10,000 ppm or greater is measured, a leak is detected.

(c)(1)(i) Any valve for which a leak is not detected for 2 successive months may be monitored the first month of every quarter, beginning with the next quarter, until a leak is detected.

(ii) As an alternative to monitoring all of the valves in the first month of a quarter, an owner or operator may elect to subdivide the process unit into 2 or 3 subgroups of valves and monitor each subgroup in a different month during the quarter, provided each subgroup is monitored every 3 months. The owner or operator must keep records of the valves assigned to each subgroup.

(2) If a leak is detected, the valve shall be monitored monthly until a leak is not detected for 2 successive months.

(d)(1) When a leak is detected, it shall be repaired as soon as practicable, but no later than 15 calendar days after the leak is detected, except as provided in 60.482-9.

(2) A first attempt at repair shall be made no later than 5 calendar days after each leak is detected.

(e) First attempts at repair include, but are not limited to, the following best practices where practicable:

(1) Tightening of bonnet bolts;

(2) Replacement of bonnet bolts;

(3) Tightening of packing gland nuts;

(4) Injection of lubricant into lubricated packing.

(f) Any valve that is designated, as described in 60.486(e)(2), for no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, is exempt from the requirements of paragraph (a) if the valve:

(1) Has no external actuating mechanism in contact with the process fluid,

(2) Is operated with emissions less than 500 ppm above background as determined by the method specified in §60.485(c), and (3) Is tested for compliance with paragraph (f)(2) of this section initially upon designation, annually, and at other times requested by the Administrator.

(g) Any value that is designated, as described in 60.486(f)(1), as an unsafeto-monitor value is exempt from the requirements of paragraph (a) if:

(1) The owner or operator of the valve demonstrates that the valve is unsafe to monitor because monitoring personnel would be exposed to an immediate danger as a consequence of complying with paragraph (a), and

(2) The owner or operator of the valve adheres to a written plan that requires monitoring of the valve as frequently as practicable during safe-to-monitor times.

(h) Any valve that is designated, as described in 60.486(f)(2), as a difficult-to-monitor valve is exempt from the requirements of paragraph (a) if:

(1) The owner or operator of the valve demonstrates that the valve cannot be monitored without elevating the monitoring personnel more than 2 meters above a support surface.

(2) The process unit within which the valve is located either becomes an affected facility through 60.14 or 60.15 or the owner or operator designates less than 3.0 percent of the total number of valves as difficult-to-monitor, and

(3) The owner or operator of the valve follows a written plan that requires monitoring of the valve at least once per calendar year.

[48 FR 48335, Oct. 18, 1983, as amended at 49 FR 22608, May 30, 1984; 65 FR 61762, Oct. 17, 2000; 72 FR 64881, Nov. 16, 2007]

§60.482–8 Standards: Pumps and valves in heavy liquid service, pressure relief devices in light liquid or heavy liquid service, and connectors.

(a) If evidence of a potential leak is found by visual, audible, olfactory, or any other detection method at pumps and valves in heavy liquid service, pressure relief devices in light liquid or heavy liquid service, and connectors, the owner or operator shall follow either one of the following procedures:

(1) The owner or operator shall monitor the equipment within 5 days by the method specified in 60.485(b) and shall comply with the requirements of paragraphs (b) through (d) of this section.

(2) The owner or operator shall eliminate the visual, audible, olfactory, or other indication of a potential leak within 5 calendar days of detection.

(b) If an instrument reading of 10,000 ppm or greater is measured, a leak is detected.

(c)(1) When a leak is detected, it shall be repaired as soon as practicable, but not later than 15 calendar days after it is detected, except as provided in §60.482-9.

(2) The first attempt at repair shall be made no later than 5 calendar days after each leak is detected.

(d) First attempts at repair include, but are not limited to, the best practices described under \$60.482-2(c)(2) and 60.482-7(e).

[48 CFR 48335, Oct. 18, 1983, as amended at 65 FR 78277, Dec. 14, 2000; 72 FR 64882, Nov. 16, 2007]

§60.482-9 Standards: Delay of repair.

(a) Delay of repair of equipment for which leaks have been detected will be allowed if repair within 15 days is technically infeasible without a process unit shutdown. Repair of this equipment shall occur before the end of the next process unit shutdown. Monitoring to verify repair must occur within 15 days after startup of the process unit.

(b) Delay of repair of equipment will be allowed for equipment which is isolated from the process and which does not remain in VOC service.

(c) Delay of repair for valves will be allowed if:

(1) The owner or operator demonstrates that emissions of purged material resulting from immediate repair are greater than the fugitive emissions likely to result from delay of repair, and

(2) When repair procedures are effected, the purged material is collected and destroyed or recovered in a control device complying with §60.482–10.

(d) Delay of repair for pumps will be allowed if:

(1) Repair requires the use of a dual mechanical seal system that includes a barrier fluid system, and

40 CFR Ch. I (7–1–23 Edition)

(2) Repair is completed as soon as practicable, but not later than 6 months after the leak was detected.

(e) Delay of repair beyond a process unit shutdown will be allowed for a valve, if valve assembly replacement is necessary during the process unit shutdown, valve assembly supplies have been depleted, and valve assembly supplies had been sufficiently stocked before the supplies were depleted. Delay of repair beyond the next process unit shutdown will not be allowed unless the next process unit shutdown occurs sooner than 6 months after the first process unit shutdown.

(f) When delay of repair is allowed for a leaking pump or valve that remains in service, the pump or valve may be considered to be repaired and no longer subject to delay of repair requirements if two consecutive monthly monitoring instrument readings are below the leak definition.

[48 FR 48335, Oct. 18, 1983, as amended at 65 FR 78277, Dec. 14, 2000; 72 FR 64882, Nov. 16, 2007]

§60.482-10 Standards: Closed vent systems and control devices.

(a) Owners or operators of closed vent systems and control devices used to comply with provisions of this subpart shall comply with the provisions of this section.

(b) Vapor recovery systems (for example, condensers and absorbers) shall be designed and operated to recover the VOC emissions vented to them with an efficiency of 95 percent or greater, or to an exit concentration of 20 parts per million by volume, whichever is less stringent.

(c) Enclosed combustion devices shall be designed and operated to reduce the VOC emissions vented to them with an efficiency of 95 percent or greater, or to an exit concentration of 20 parts per million by volume, on a dry basis, corrected to 3 percent oxygen, whichever is less stringent or to provide a minimum residence time of 0.75 seconds at a minimum temperature of 816 °C.

(d) Flares used to comply with this subpart shall comply with the requirements of §60.18.

(e) Owners or operators of control devices used to comply with the provisions of this subpart shall monitor

these control devices to ensure that they are operated and maintained in conformance with their designs.

(f) Except as provided in paragraphs (i) through (k) of this section, each closed vent system shall be inspected according to the procedures and schedule specified in paragraphs (f)(1) and (f)(2) of this section.

(1) If the vapor collection system or closed vent system is constructed of hard-piping, the owner or operator shall comply with the requirements specified in paragraphs (f)(1)(i) and (f)(1)(i) of this section:

(i) Conduct an initial inspection according to the procedures in 60.485(b); and

(ii) Conduct annual visual inspections for visible, audible, or olfactory indications of leaks.

(2) If the vapor collection system or closed vent system is constructed of ductwork, the owner or operator shall:

(i) Conduct an initial inspection according to the procedures in 60.485(b); and

(ii) Conduct annual inspections according to the procedures in 60.485(b).

(g) Leaks, as indicated by an instrument reading greater than 500 parts per million by volume above background or by visual inspections, shall be repaired as soon as practicable except as provided in paragraph (h) of this section.

(1) A first attempt at repair shall be made no later than 5 calendar days after the leak is detected.

(2) Repair shall be completed no later than 15 calendar days after the leak is detected.

(h) Delay of repair of a closed vent system for which leaks have been detected is allowed if the repair is technically infeasible without a process unit shutdown or if the owner or operator determines that emissions resulting from immediate repair would be greater than the fugitive emissions likely to result from delay of repair. Repair of such equipment shall be complete by the end of the next process unit shutdown.

(i) If a vapor collection system or closed vent system is operated under a vacuum, it is exempt from the inspection requirements of paragraphs (f)(1)(i) and (f)(2) of this section.

(j) Any parts of the closed vent system that are designated, as described in paragraph (1)(1) of this section, as unsafe to inspect are exempt from the inspection requirements of paragraphs (f)(1)(i) and (f)(2) of this section if they comply with the requirements specified in paragraphs (j)(1) and (j)(2) of this section:

(1) The owner or operator determines that the equipment is unsafe to inspect because inspecting personnel would be exposed to an imminent or potential danger as a consequence of complying with paragraphs (f)(1)(i) or (f)(2) of this section; and

(2) The owner or operator has a written plan that requires inspection of the equipment as frequently as practicable during safe-to-inspect times.

(k) Any parts of the closed vent system that are designated, as described in paragraph (1)(2) of this section, as difficult to inspect are exempt from the inspection requirements of paragraphs (f)(1)(i) and (f)(2) of this section if they comply with the requirements specified in paragraphs (k)(1) through (k)(3) of this section:

(1) The owner or operator determines that the equipment cannot be inspected without elevating the inspecting personnel more than 2 meters above a support surface; and

(2) The process unit within which the closed vent system is located becomes an affected facility through §§ 60.14 or 60.15, or the owner or operator designates less than 3.0 percent of the total number of closed vent system equipment as difficult to inspect; and

(3) The owner or operator has a written plan that requires inspection of the equipment at least once every 5 years. A closed vent system is exempt from inspection if it is operated under a vacuum.

(1) The owner or operator shall record the information specified in paragraphs (1)(1) through (1)(5) of this section.

(1) Identification of all parts of the closed vent system that are designated as unsafe to inspect, an explanation of why the equipment is unsafe to inspect, and the plan for inspecting the equipment.

(2) Identification of all parts of the closed vent system that are designated as difficult to inspect, an explanation

of why the equipment is difficult to inspect, and the plan for inspecting the equipment.

(3) For each inspection during which a leak is detected, a record of the information specified in 60.486(c).

(4) For each inspection conducted in accordance with §60.485(b) during which no leaks are detected, a record that the inspection was performed, the date of the inspection, and a statement that no leaks were detected.

(5) For each visual inspection conducted in accordance with paragraph (f)(1)(ii) of this section during which no leaks are detected, a record that the inspection was performed, the date of the inspection, and a statement that no leaks were detected.

(m) Closed vent systems and control devices used to comply with provisions of this subpart shall be operated at all times when emissions may be vented to them.

[48 FR 48335, Oct. 18, 1983, as amended at 51 FR 2702, Jan. 21, 1986; 60 FR 43258, Aug. 18, 1995; 61 FR 29878, June 12, 1996; 65 FR 78277, Dec. 14, 2000]

§60.483–1 Alternative standards for valves—allowable percentage of valves leaking.

(a) An owner or operator may elect to comply with an allowable percentage of valves leaking of equal to or less than 2.0 percent.

(b) The following requirements shall be met if an owner or operator wishes to comply with an allowable percentage of valves leaking:

(1) An owner or operator must notify the Administrator that the owner or operator has elected to comply with the allowable percentage of valves leaking before implementing this alternative standard, as specified in §60.487(d).

(2) A performance test as specified in paragraph (c) of this section shall be conducted initially upon designation, annually, and at other times requested by the Administrator.

(3) If a valve leak is detected, it shall be repaired in accordance with 60.482-7(d) and (e).

(c) Performance tests shall be conducted in the following manner:

(1) All valves in gas/vapor and light liquid service within the affected facil-

40 CFR Ch. I (7–1–23 Edition)

ity shall be monitored within 1 week by the methods specified in §60.485(b).

(2) If an instrument reading of 10,000 ppm or greater is measured, a leak is detected.

(3) The leak percentage shall be determined by dividing the number of valves for which leaks are detected by the number of valves in gas/vapor and light liquid service within the affected facility.

(d) Owners and operators who elect to comply with this alternative standard shall not have an affected facility with a leak percentage greater than 2.0 percent, determined as described in §60.485(h).

[48 FR 48335, Oct. 18, 1983, as amended at 65 FR 61762, Oct. 17, 2000; 65 FR 78278, Dec. 14, 2000; 72 FR 64882, Nov. 16, 2007]

§60.483-2 Alternative standards for valves—skip period leak detection and repair.

(a)(1) An owner or operator may elect to comply with one of the alternative work practices specified in paragraphs (b)(2) and (3) of this section.

(2) An owner or operator must notify the Administrator before implementing one of the alternative work practices, as specified in §60.487(d).

(b)(1) An owner or operator shall comply initially with the requirements for valves in gas/vapor service and valves in light liquid service, as described in §60.482-7.

(2) After 2 consecutive quarterly leak detection periods with the percent of valves leaking equal to or less than 2.0, an owner or operator may begin to skip 1 of the quarterly leak detection periods for the valves in gas/vapor and light liquid service.

(3) After 5 consecutive quarterly leak detection periods with the percent of valves leaking equal to or less than 2.0, an owner or operator may begin to skip 3 of the quarterly leak detection periods for the valves in gas/vapor and light liquid service.

(4) If the percent of valves leaking is greater than 2.0, the owner or operator shall comply with the requirements as described in §60.482–7 but can again elect to use this section.

(5) The percent of valves leaking shall be determined as described in §60.485(h).

(6) An owner or operator must keep a record of the percent of valves found leaking during each leak detection period.

(7) A valve that begins operation in gas/vapor service or light liquid service after the initial startup date for a process unit following one of the alternative standards in this section must be monitored in accordance with $\S60.482-7(a)(2)(i)$ or (ii) before the provisions of this section can be applied to that valve.

[48 FR 48335, Oct. 18, 1983, as amended at 65 FR 61762, Oct. 17, 2000; 65 FR 78278, Dec. 14, 2000; 72 FR 64882, Nov. 16, 2007]

§60.484 Equivalence of means of emission limitation.

(a) Each owner or operator subject to the provisions of this subpart may apply to the Administrator for determination of equivalence for any means of emission limitation that achieves a reduction in emissions of VOC at least equivalent to the reduction in emissions of VOC achieved by the controls required in this subpart.

(b) Determination of equivalence to the equipment, design, and operational requirements of this subpart will be evaluated by the following guidelines:

(1) Each owner or operator applying for an equivalence determination shall be responsible for collecting and verifying test data to demonstrate equivalence of means of emission limitation.

(2) The Administrator will compare test data for demonstrating equivalence of the means of emission limitation to test data for the equipment, design, and operational requirements.

(3) The Administrator may condition the approval of equivalence on requirements that may be necessary to assure operation and maintenance to achieve the same emission reduction as the equipment, design, and operational requirements.

(c) Determination of equivalence to the required work practices in this subpart will be evaluated by the following guidelines:

(1) Each owner or operator applying for a determination of equivalence shall be responsible for collecting and verifying test data to demonstrate equivalence of an equivalent means of emission limitation.

(2) For each affected facility for which a determination of equivalence is requested, the emission reduction achieved by the required work practice shall be demonstrated.

(3) For each affected facility, for which a determination of equivalence is requested, the emission reduction achieved by the equivalent means of emission limitation shall be demonstrated.

(4) Each owner or operator applying for a determination of equivalence shall commit in writing to work practice(s) that provide for emission reductions equal to or greater than the emission reductions achieved by the required work practice.

(5) The Administrator will compare the demonstrated emission reduction for the equivalent means of emission limitation to the demonstrated emission reduction for the required work practices and will consider the commitment in paragraph (c)(4).

(6) The Administrator may condition the approval of equivalence on requirements that may be necessary to assure operation and maintenance to achieve the same emission reduction as the required work practice.

(d) An owner or operator may offer a unique approach to demonstrate the equivalence of any equivalent means of emission limitation.

(e)(1) After a request for determination of equivalence is received, the Administrator will publish a notice in the FEDERAL REGISTER and provide the opportunity for public hearing if the Administrator judges that the request may be approved.

(2) After notice and opportunity for public hearing, the Administrator will determine the equivalence of a means of emission limitation and will publish the determination in the FEDERAL REG-ISTER.

(3) Any equivalent means of emission limitations approved under this section shall constitute a required work practice, equipment, design, or operational standard within the meaning of section 111(h)(1) of the Clean Air Act.

(f)(1) Manufacturers of equipment used to control equipment leaks of VOC may apply to the Administrator for determination of equivalence for any equivalent means of emission limitation that achieves a reduction in emissions of VOC achieved by the equipment, design, and operational requirements of this subpart.

(2) The Administrator will make an equivalence determination according to the provisions of paragraphs (b), (c), (d), and (e) of this section.

[48 FR 48335, Oct. 18, 1983, as amended at 65 FR 61762, Oct. 17, 2000; 72 FR 64882, Nov. 16, 2007]

§60.485 Test methods and procedures.

(a) In conducting the performance tests required in 60.8, the owner or operator shall use as reference methods and procedures the test methods in appendix A of this part or other methods and procedures as specified in this section, except as provided in 60.8(b).

(b) The owner or operator shall determine compliance with the standards in §§ 60.482-1 through 60.482-10, 60.483, and 60.484 as follows:

(1) Method 21 shall be used to determine the presence of leaking sources. The instrument shall be calibrated before use each day of its use by the procedures specified in Method 21. The following calibration gases shall be used:

(i) Zero air (less than 10 ppm of hydrocarbon in air); and

(ii) A mixture of methane or nhexane and air at a concentration of about, but less than, 10,000 ppm methane or n-hexane.

(c) The owner or operator shall determine compliance with the no detectable emission standards in \S 60.482–2(e), 60.482–3(i), 60.482–4, 60.482–7(f), and 60.482–10(e) as follows:

(1) The requirements of paragraph (b) shall apply.

(2) Method 21 shall be used to determine the background level. All potential leak interfaces shall be traversed as close to the interface as possible. The arithmetic difference between the maximum concentration indicated by the instrument and the background level is compared with 500 ppm for determining compliance.

(d) The owner or operator shall test each piece of equipment unless he demonstrates that a process unit is not in VOC service, i.e., that the VOC content would never be reasonably expected to 40 CFR Ch. I (7–1–23 Edition)

exceed 10 percent by weight. For purposes of this demonstration, the following methods and procedures shall be used:

(1) Procedures that conform to the general methods in ASTM E260-73, 91, or 96, E168-67, 77, or 92, E169-63, 77, or 93 (incorporated by reference—see \S 60.17) shall be used to determine the percent VOC content in the process fluid that is contained in or contacts a piece of equipment.

(2) Organic compounds that are considered by the Administrator to have negligible photochemical reactivity may be excluded from the total quantity of organic compounds in determining the VOC content of the process fluid.

(3) Engineering judgment may be used to estimate the VOC content, if a piece of equipment had not been shown previously to be in service. If the Administrator disagrees with the judgment, paragraphs (d) (1) and (2) of this section shall be used to resolve the disagreement.

(e) The owner or operator shall demonstrate that a piece of equipment is in light liquid service by showing that all the following conditions apply:

(1) The vapor pressure of one or more of the organic components is greater than 0.3 kPa at 20 °C (1.2 in. H₂O at 68 °F). Standard reference texts or ASTM D2879-83, 96, or 97 (incorporated by reference—see §60.17) shall be used to determine the vapor pressures.

(2) The total concentration of the pure organic components having a vapor pressure greater than 0.3 kPa at 20 °C (1.2 in. H₂O at 68 °F) is equal to or greater than 20 percent by weight.

(3) The fluid is a liquid at operating conditions.

(f) Samples used in conjunction with paragraphs (d), (e), and (g) of this section shall be representative of the process fluid that is contained in or contacts the equipment or the gas being combusted in the flare.

(g) The owner or operator shall determine compliance with the standards of flares as follows:

(1) Method 22 shall be used to determine visible emissions.

(2) A thermocouple or any other equivalent device shall be used to monitor the presence of a pilot flame in the flare.

(3) The maximum permitted velocity for air assisted flares shall be computed using the following equation:

$$V_{max} = K_1 + K_2 H_T$$

Where:

V_{max} = Maximum permitted velocity, m/sec (ft/sec)

 H_T = Net heating value of the gas being combusted, MJ/scm (Btu/scf).

 $K_1 = 8.706 \text{ m/sec}$ (metric units)

= 28.56 ft/sec (English units)

 $K_2 = 0.7084 \text{ m}^4/(\text{MJ-sec}) \text{ (metric units)}$

= 0.087 ft⁴/(Btu-sec) (English units)

(4) The net heating value (H_T) of the gas being combusted in a flare shall be computed using the following equation:

$$\mathbf{H}_{\mathrm{T}} = \mathbf{K} \sum_{i=1}^{n} \mathbf{C}_{i} \mathbf{H}_{i}$$

Where:

- K = Conversion constant, 1.740×10^{-7} (gmole)(MJ)/(ppm-scm-kcal) (metric units) = 4.674×10^{-6} [(g-mole)(Btu)/(ppm-scfkcal)] (English units)
- C_i = Concentration of sample component "i," ppm
- H_i = Net heat of combustion of sample component "i" at 25 °C and 760 mm Hg (77 °F and 14.7 psi), kcal/g-mole

(5) Method 18 or ASTM D6420-99 (2004) (where the target compound(s) are those listed in Section 1.1 of ASTM D6420-99, and the target concentration is between 150 parts per billion by volume and 100 parts per million by volume) and ASTM D2504-67, 77 or 88 (Reapproved 1993) (incorporated by reference—see §60.17) shall be used to determine the concentration of sample component "i."

(6) ASTM D2382-76 or 88 or D4809-95 (incorporated by reference—see §60.17) shall be used to determine the net heat of combustion of component "i" if published values are not available or cannot be calculated.

(7) Method 2, 2A, 2C, or 2D, as appropriate, shall be used to determine the actual exit velocity of a flare. If needed, the unobstructed (free) cross-sectional area of the flare tip shall be used.

(h) The owner or operator shall determine compliance with §60.483-1 or §60.483–2 as follows:

(1) The percent of valves leaking shall be determined using the following equation:

 $V_{\rm L} = (V_{\rm L}/V_{\rm T}) * 100$

Where:

 $%V_{L}$ = Percent leaking values

V_L = Number of valves found leaking

 V_T = The sum of the total number of valves monitored

(2) The total number of valves monitored shall include difficult-to-monitor and unsafe-to-monitor valves only during the monitoring period in which those valves are monitored.

(3) The number of valves leaking shall include valves for which repair has been delayed.

(4) Any new valve that is not monitored within 30 days of being placed in service shall be included in the number of valves leaking and the total number of valves monitored for the monitoring period in which the valve is placed in service.

(5) If the process unit has been subdivided in accordance with §60.482-7(c)(1)(ii), the sum of valves found leaking during a monitoring period includes all subgroups.

(6) The total number of valves monitored does not include a valve monitored to verify repair.

[54 FR 6678, Feb. 14, 1989, as amended at 54 FR 27016, June 27, 1989; 65 FR 61763, Oct. 17, 2000; 72 FR 64882, Nov. 16, 2007]

§60.486 Recordkeeping requirements.

(a)(1) Each owner or operator subject to the provisions of this subpart shall comply with the recordkeeping requirements of this section.

(2) An owner or operator of more than one affected facility subject to the provisions of this subpart may comply with the recordkeeping requirements for these facilities in one recordkeeping system if the system identifies each record by each facility.

(b) When each leak is detected as specified in §§60.482-2, 60.482-3, 60.482-7, 60.482-8, and 60.483-2, the following requirements apply:

LAPBH6H6L3 with DISTILLER

§60.486

(1) A weatherproof and readily visible identification, marked with the equipment identification number, shall be attached to the leaking equipment.

(2) The identification on a valve may be removed after it has been monitored for 2 successive months as specified in §60.482–7(c) and no leak has been detected during those 2 months.

(3) The identification on equipment except on a valve, may be removed after it has been repaired.

(c) When each leak is detected as specified in §§ 60.482-2, 60.482-3, 60.482-7, 60.482-8, and 60.483-2, the following information shall be recorded in a log and shall be kept for 2 years in a readily accessible location:

(1) The instrument and operator identification numbers and the equipment identification number.

(2) The date the leak was detected and the dates of each attempt to repair the leak.

(3) Repair methods applied in each attempt to repair the leak.

(4) "Above 10,000" if the maximum instrument reading measured by the methods specified in §60.485(a) after each repair attempt is equal to or greater than 10,000 ppm.

(5) "Repair delayed" and the reason for the delay if a leak is not repaired within 15 calendar days after discovery of the leak.

(6) The signature of the owner or operator (or designate) whose decision it was that repair could not be effected without a process shutdown.

(7) The expected date of successful repair of the leak if a leak is not repaired within 15 days.

(8) Dates of process unit shutdowns that occur while the equipment is unrepaired.

(9) The date of successful repair of the leak.

(d) The following information pertaining to the design requirements for closed vent systems and control devices described in 60.482-10 shall be recorded and kept in a readily accessible location:

(1) Detailed schematics, design specifications, and piping and instrumentation diagrams.

(2) The dates and descriptions of any changes in the design specifications.

40 CFR Ch. I (7–1–23 Edition)

(3) A description of the parameter or parameters monitored, as required in $\S60.482-10(e)$, to ensure that control devices are operated and maintained in conformance with their design and an explanation of why that parameter (or parameters) was selected for the monitoring.

(4) Periods when the closed vent systems and control devices required in §§ 60.482-2, 60.482-3, 60.482-4, and 60.482-5 are not operated as designed, including periods when a flare pilot light does not have a flame.

(5) Dates of startups and shutdowns of the closed vent systems and control devices required in \$ 60.482–2, 60.482–3, 60.482–4, and 60.482–5.

(e) The following information pertaining to all equipment subject to the requirements in \$60.482-1 to 60.482-10shall be recorded in a log that is kept in a readily accessible location:

(1) A list of identification numbers for equipment subject to the requirements of this subpart.

(2)(i) A list of identification numbers for equipment that are designated for no detectable emissions under the provisions of §§ 60.482-2(e), 60.482-3(i) and 60.482-7(f).

(ii) The designation of equipment as subject to the requirements of 60.482-2(e), 60.482-3(i), or 60.482-7(f) shall be signed by the owner or operator. Alternatively, the owner or operator may establish a mechanism with their permitting authority that satisfies this requirement.

(3) A list of equipment identification numbers for pressure relief devices required to comply with §60.482-4.

(4)(i) The dates of each compliance test as required in \$ 60.482–2(e), 60.482–3(i), 60.482–4, and 60.482–7(f).

(ii) The background level measured during each compliance test.

(iii) The maximum instrument reading measured at the equipment during each compliance test.

(5) A list of identification numbers for equipment in vacuum service.

(6) A list of identification numbers for equipment that the owner or operator designates as operating in VOC service less than 300 hr/yr in accordance with $\S 60.482-1(e)$, a description of the conditions under which the equipment is in VOC service, and rationale

supporting the designation that it is in VOC service less than 300 hr/yr.

(f) The following information pertaining to all valves subject to the requirements of 60.482-7(g) and (h) and to all pumps subject to the requirements of 60.482-2(g) shall be recorded in a log that is kept in a readily accessible location:

(1) A list of identification numbers for valves and pumps that are designated as unsafe-to-monitor, an explanation for each valve or pump stating why the valve or pump is unsafe-tomonitor, and the plan for monitoring each valve or pump.

(2) A list of identification numbers for valves that are designated as difficult-to-monitor, an explanation for each valve stating why the valve is difficult-to-monitor, and the schedule for monitoring each valve.

(g) The following information shall be recorded for valves complying with §60.483-2:

(1) A schedule of monitoring.

(2) The percent of valves found leaking during each monitoring period.

(h) The following information shall be recorded in a log that is kept in a readily accessible location:

(1) Design criterion required in §§ 60.482-2(d)(5) and 60.482-3(e)(2) and explanation of the design criterion; and

(2) Any changes to this criterion and the reasons for the changes.

(i) The following information shall be recorded in a log that is kept in a readily accessible location for use in determining exemptions as provided in §60.480(d):

(1) An analysis demonstrating the design capacity of the affected facility,

(2) A statement listing the feed or raw materials and products from the affected facilities and an analysis demonstrating whether these chemicals are heavy liquids or beverage alcohol, and

(3) An analysis demonstrating that equipment is not in VOC service.

(j) Information and data used to demonstrate that a piece of equipment is not in VOC service shall be recorded in a log that is kept in a readily accessible location. (k) The provisions of §60.7 (b) and (d) do not apply to affected facilities subject to this subpart.

[48 FR 48335, Oct. 18, 1983, as amended at 65
 FR 61763, Oct. 17, 2000; 65 FR 78278, Dec. 14, 2000; 72 FR 64883, Nov. 16, 2007]

§60.487 Reporting requirements.

(a) Each owner or operator subject to the provisions of this subpart shall submit semiannual reports to the Administrator beginning six months after the initial startup date.

(b) The initial semiannual report to the Administrator shall include the following information:

(1) Process unit identification.

(2) Number of valves subject to the requirements of §60.482–7, excluding those valves designated for no detectable emissions under the provisions of §60.482–7(f).

(3) Number of pumps subject to the requirements of §60.482-2, excluding those pumps designated for no detectable emissions under the provisions of §60.482-2(e) and those pumps complying with §60.482-2(f).

(4) Number of compressors subject to the requirements of 60.482-3, excluding those compressors designated for no detectable emissions under the provisions of 60.482-3(i) and those compressors complying with 60.482-3(h).

(c) All semiannual reports to the Administrator shall include the following information, summarized from the information in §60.486:

(1) Process unit identification.

(2) For each month during the semiannual reporting period,

(i) Number of valves for which leaks were detected as described in §60.482– 7(b) or §60.483–2,

(ii) Number of valves for which leaks were not repaired as required in §60.482-7(d)(1),

(iii) Number of pumps for which leaks were detected as described in §60.482– 2(b), (d)(4)(ii)(A) or (B), or (d)(5)(iii),

(iv) Number of pumps for which leaks were not repaired as required in §60.482-2(c)(1) and (d)(6),

(v) Number of compressors for which leaks were detected as described in §60.482-3(f),

(vi) Number of compressors for which leaks were not repaired as required in 60.482-3(g)(1), and

LAPBH6H6L3 with DISTILLER

§60.488

(vii) The facts that explain each delay of repair and, where appropriate, why a process unit shutdown was technically infeasible.

(3) Dates of process unit shutdowns which occurred within the semiannual reporting period.

(4) Revisions to items reported according to paragraph (b) if changes have occurred since the initial report or subsequent revisions to the initial report.

(d) An owner or operator electing to comply with the provisions of §60.483-1 or §60.483-2 shall notify the Administrator of the alternative standard selected 90 days before implementing either of the provisions.

(e) An owner or operator shall report the results of all performance tests in accordance with §60.8 of the General Provisions. The provisions of §60.8(d) do not apply to affected facilities subject to the provisions of this subpart except that an owner or operator must notify the Administrator of the schedule for the initial performance tests at least 30 days before the initial performance tests.

(f) The requirements of paragraphs (a) through (c) of this section remain in force until and unless EPA, in delegating enforcement authority to a State under section 111(c) of the Act, approves reporting requirements or an alternative means of compliance surveillance adopted by such State. In that event, affected sources within the State will be relieved of the obligation to comply with the requirements of paragraphs (a) through (c) of this section, provided that they comply with the requirements established by the State.

[48 FR 48335, Oct. 18, 1983, as amended at 49
 FR 22608, May 30, 1984; 65 FR 61763, Oct. 17, 2000; 72 FR 64883, Nov. 16, 2007]

§60.488 Reconstruction.

For the purposes of this subpart:

(a) The cost of the following frequently replaced components of the facility shall not be considered in calculating either the "fixed capital cost of the new components" or the "fixed capital costs that would be required to construct a comparable new facility" under §60.15: pump seals, nuts and bolts, rupture disks, and packings.

40 CFR Ch. I (7–1–23 Edition)

(b) Under §60.15, the "fixed capital cost of new components" includes the fixed capital cost of all depreciable components (except components specified in §60.488 (a)) which are or will be replaced pursuant to all continuous programs of component replacement which are commenced within any 2year period following the applicability date for the appropriate subpart. (See the "Applicability and designation of affected facility" section of the appropriate subpart.) For purposes of this paragraph, "commenced" means that an owner or operator has undertaken a continuous program of component replacement or that an owner or operator has entered into a contractual obligation to undertake and complete, within a reasonable time, a continuous program of component replacement.

[49 FR 22608, May 30, 1984]

§60.489 List of chemicals produced by affected facilities.

The following chemicals are produced, as intermediates or final products, by process units covered under this subpart. The applicability date for process units producing one or more of these chemicals is January 5, 1981.

CAS □o. ª	Chemi⊡al
105=57=7	A⊑etal.
75_07_0	A⊑etal⊒ehy⊒e.
107 89 1	A⊑etal ⊡ol.
60 35 5	A⊡etami⊡e.
103 84 4	A⊡etanili⊡e.
64 19 7	A⊑eti⊟a⊡
108 24 7	A⊑eti⊟ anhy⊡ri⊑e.
67 64 1	A⊡etone.
75 86 5	A⊑etone ⊒yanohy⊒rin.
75 05 8	A etonitrile.
98 86 2	A etophenone.
75_36_5	A⊑etyl ⊡hlori⊑e.
74 86 2	A etylene.
107 02 8	A⊡rolein.
79 06 1	A⊡rylami⊒e.
79 10 7	Alryli all.
107 13 1	A rylonitrile.
124 04 9	A⊡pi□a⊡⊡
111_69_3	A⊑iponitrile.
£0	Alkyl naphthalene□
107 18 6	Allyl al ohol.
107 05 1	Allyl ⊑hlori⊑e.
1321 11 5	Amino en oi a i .
11104101	Aminoethylethanolamine.
123 30 8	p Aminophenol.
628 63 7, 123	Amyl a etate □
92 2.	
71□41□0 □	Amyl al⊡ohol⊡
110 58 7	Amyl amine.
543 59 9	Amyl ⊡hlori⊒e.
110_66_7	Amyl mer⊑aptan □
1322 06 1	Amyl phenol.
62 53 3	Aniline.

666

§60.489

CAS □o. ª	Chemi⊡al	CAS □o. ª	Chemi⊡al
142-04-1	Aniline hy ro blori e	108-39-4	m Tre Tol
29191 52 4		95-48-7	
100 66 3	Ani⊑ole.	106 44 5	
118 92 3	Anthranili a 🖬 🗆	1319 77 3	
84_65_1	Anthra uinone.	1319 77 3	Cre⊡yli⊡a⊡⊡
100 52 7	Ben⊡al⊡ehy⊡e.	4170 30 0	Crotonal ⊒ehy⊒e.
55 21 0	Ben ami e.	3724_65_0	Crotonia
/1_43_2		98_82_8	
98 40 0		372 09 8	Cvanoa eti a a i a
134 81 6	Ben II.	506 77 4	Cyanogen ⊡hlori⊡e.
76_93_7	Ben ili a i □.	108 80 5	Cyanuri⊡a⊡⊡
65 85 0	Ben⊡oi□a⊡⊡	108 77 0	Cyanuri□ ⊡hlori⊡e.
119 53 9	Ben Doin.	110 82 7	Cy⊡ohe⊡ane.
100_47_0	Ben onitrile.	108_93_0	Cyllohe anol.
08-07-7		110 83 8	
98 88 4	Ben ovl Thlori e.	108 91 8	Cyllohe Vlamine.
100 51 6	Ben⊡yl al⊡ohol.	111 78 4	Cy⊡oo⊡ta⊡ene.
100□46□9	Ben⊡ylamine.	112 30 1	De⊡anol.
120 51 4	Ben⊡yl ⊑en⊡oate.	123 42 2	Dia etone al ohol.
100 44 7		27576 04 1	
98_87_3		95_/6_1, 95_62_	Di∟nioroaniiine.
92_52_4 80_05_7	Binhenol A	9, 554±00±7, 608±27±5	
10_86_1	Bromo en ene.	608 31 1.	
27497 51 4	Bromonaphthalene.	626 43 7,	
106 99 0	Buta⊡ene.	27134 27 6,	
106 98 9	1 ⊡utene.	57311 92 9 .	
123 86 4	n⊡utyl a⊡etate.	541 73 1	m i hloro en ene.
141_32_2	nillutyi alliylate.	95_50_1	
78-92-2		75 71 8	Dichloro ifluoromethane
75 65 0	tTutyl allohol.	111 44 4	Di hloroethyl ether.
109 73 9	n⊡utylamine.	107 06 2	1,2 III hloroethane EDC
13952 84 6	□IIutylamine.	96 23 1	Di⊡hlorohy⊡rin.
75_64_9	t⊡utylamine.	26952 23 8	Di⊡hloropropene.
98 73 7	pîtertîlutyl Den Coil a îl l	101 83 7	Di⊡y⊡ohe⊡ylamine.
107_88_0	1,3 Lutylene gly_ol.		Diethylamine.
107-92-6		112 36 7	Diethylene gly⊡ol. Diethylene gly⊡ol ⊑iethyl ether
106-31-0	Butyri anhy ri le	111_96_6	Diethylene gly ol fimethyl ether
109 74 0	Butyronitrile.	112 34 5	Diethylene gly ol mono utyl ether.
105 60 2	Caprola⊡tam.	124 17 4	Diethylene gly ol mono utyl ether a e
75 1 50	Car⊡on ⊡⊔lfi⊡e.		tate.
558 13 4	Car⊡on tetra ⊡romi ⊡e.	111 90 0	Diethylene gly⊡ol monoethyl ether.
56_23_5	CarLon tetraLhloriLe.	112_15_2	biethylene gly∟ol monoethyl ether a⊥e⊔
79_11_8		111 77 3	late. Diethylene alv⊡ol monomethyl ether
108 42 9	mThloroaniline.	64_67_5	Diethyl Fulfate.
95□51□2	o⊡hloroaniline.	75_37_6	Difluoroethane.
106□47□8	p⊞hloroaniline.	25167 70 8	Dii⊡o⊡utylene.
35913 09 8	Chloro en al ehy e.	26761 40 0	Dii⊡o ⊒e ⊒yl phthalate.
108 90 7	Chloro en ene.	27554 26 3	Dii⊡oo⊡tyl phthalate.
118_91_2, 535_ 90_9_74_11		124 40 2	Diketene.
3		124_40_3	
2136 81 4.	Chloro ⊡en ⊡otri ⊡hlori ⊡e.	115 10 6	□.□ III imethyl ether.
2136 89 2,		68 12 2	□, □ IIImethylformami □e.
5216 25 1 .		57 14 7	Dimethylhy a line.
1321 03 5	Chloro en oyl hlori e.	77□78□1	Dimethyl Iulfate.
25497 29 4	Chloro Itiuoromethane.	75 18 3	Dimethyl Culfice.
/ 0_40_0	Chloroform	120-61-6	Dimentyl LuliolLe.
25586-43-0	Chloronaphthalene	99-34-3	
88_73_3	o Thloronitro en ene.	51 28 5	Dinitrophenol.
100_00_5	pIhloronitro en ene.	25321 14 6	Dinitrotoluene.
25167 80 0	Chlorophenol 🗆	123_91_1	Dio⊡ane.
126 99 8	Chloroprene.	646_06_0	Dio⊡lane.
7790 94 5		122 39 4	Diphenylamine.
108_41_8	millinorotoluene.	101_84_8	Diphenyl oll_e.
55_49_6 106_43_4	p Thlorotoluene	25265 71 8	Dipropylene gly ol
75 72 9	Chlorotrifluoromethane	25378 22 7	Dorerene.

667

§60.489

40 CFR Ch. I (7-1-23 Edition)

CAS □o. ª	Chemi⊡al	CAS □o. ª	Chemi⊡al
28675 17 4	Do⊡e⊒ylaniline.	463□51□4	□etene.
27193 86 8	Do⊡e⊡ylphenol.	Œ0	Linear alkyl 💷lfonate.
106 89 8	Epi⊡hlorohy⊡rin.	123□01□3	Linear alkyl en ene dinear
64L17L5	Ethanol.	110-16-7	
141 78 6	Ethyl a etate.	108 31 6	alei anhy ri e.
141_97_9	Ethyl a etoa etate.	6915 15 7	
140 88 5	Ethyl a⊡rylate.	141_79_7	□e⊑ityl o⊑i⊒e.
75047	Ethylamine.	121 47 1	
100_41_4 74_96_4	Ethyl_en_ene.	79L41L4	
9004 57 3	Ethyl ellulo e.	67 56 1	ethanol.
75=00=3	Ethyl ⊡hlori⊡e.	79_20_9	□ ethyl a ⊡etate.
105 39 5	Ethyl hloroa etate.	105 45 3	□ethyl a⊡etoa⊡etate.
105_56_6 74_85_1	Ethyl_yanoa_etate.	74_89_5	□ etnylamine.
96-49-1	Ethylene FarFonate.	74 83 9	ethyl romire.
107 07 3	Ethylene ⊡hlorohy⊡rin.	37365 71 2	□ ethyl □utynol.
107 15 3	Ethylene⊡amine.	74 87 3	□ ethyl □hlori □e.
106 93 4	Ethylene 🗇 romi 🕒.	108 87 2	ethyl_y_ohe_ane.
111_55_7	Ethylene glyLol.	75-09-2	□ethylene □hlori□e
110 71 4	Ethylene gly ol imethyl ether.	101 77 9	□ethylene ⊑ianiline.
111 76 2	Ethylene gly ol mono utyl ether.	101 68 8	□ethylene ⊡phenyl ⊡i⊡o⊡yanate.
112 07 2	Ethylene gly ol mono utyl ether a etate.	78_93_3	ethyl ethyl ketone.
111_15_9	Ethylene gly_of monoethyl ether.	107_31_3	ethyl iconutyl carcinol
109 86 4	Ethylene gly ol monomethyl ether.	108 10 1	ethyl iooutyl ketone.
110 49 6	Ethylene gly ol monomethyl ether a e	80 62 6	□ ethyl metha⊡rylate.
400-00-0	tate.	77 75 8	ethylpentynol.
122 99 6	Ethylene gly ol monophenyl ether.	98_83_9	almethylltyrene.
2607 <u>□</u> 30 <u>□</u> 9 75 21 8	Ethylene orice	85-47-2	amaphthalene ⊓ulfoni⊓ari⊓
60 29 7	Ethyl ether	120 18 3	□naphthalene □ulfoni□a□.
104 76 7	2 ethylhe anol.	90□15□3	a haphthol.
122 51 0	Ethyl orthoformate.	135 19 3	Inaphthol.
95_92_1 /1802_71_1	Etnyl ollalate.	75_98_9 88_74_4	
50-00-0	Formal ehve.	100_01_6	p Thitroaniline.
75 12 7	Formami⊒e.	91 23 6	o'nitroani⊡ole.
64 18 6	Formi□a□	100 17 4	p⊡itroani⊡ole.
110 17 8	Fumariaa	98_95_3	
96∟01⊔1 56⊏81⊏5	Futural.	27170L03L2 79-24-3	Litroethane
26545 73 7	□ly⊡rol ⊡_hlorohy⊡rin.	75 52 5	□itromethane.
25791 96 2	□ly erol triether.	8807505	2 itrophenol.
56 40 6	□ly⊑ne.	25322 01 4	□itropropane.
107_22_2		1321L12L6 27215-05-8	
67 72 1	He a hloroethane.	25154 52 3	onviphenol.
36653 82 4	He a e yl al ohol.	27193 28 8	O⊡tylphenol.
124 09 4	He amethylene iamine.	123 63 7	Paral ⊡ehy ⊡e.
629 11 8	He amethylene gly ol.		Pentaerythritol.
74 90 8	Hy rogen vani e.	109_67_1	1 pentene
123 31 9	Hy Tro Luinone.	127 18 4	Per⊡hloroethylene.
99_96_7	p[hy]ro]y]en_oi] a[i].	594 42 3	Per⊡hloromethyl mer⊡aptan.
26760 64 5	l oamylene.	94 70 2	opheneti⊡ne.
/o_oo⊔⊺ 110⊡19⊡0	Lo utanoi.	100_43_4	Phenol
115□11□7	Io_utylene.	98_67_9, 585	Phenol_ulfoni a i =
78 84 2	I o utyral ehy e.	38 6, 609 46	
79 31 2	I⊡o_utyri⊡ a ⊡.	1, 1333 39 7 .	
∠5339⊔17□7 26052□21□6	Lo∟e∟anol.	91_40_7	
78 78 4	Topentane.	75-44-5	Phologene.
78_59_1	I ophorone.	85 44 9	Phthali□ anhy⊡ri⊡e.
121_91_5	I⊡ophthali□ a ⊡.	85_41_6	Phthalimi⊡e.
78 79 5	loprene.	108 99 6	
07∟03∟0 108⊏21⊏4	i⊡opropanoi. I⊺opropvl a⊺etate	9003-29-6	Polv⊡itene⊡
75_31_0	I opropylamine.	25036 29 7	
75 29 6	I⊡opropyl ⊡hlori⊡e.	25322 68 3	Polyethylene gly ol.
25168 06 3	I opropylphenol.	25322 69 4	Polypropylene gly⊡ol.

668

CAS 🗆 o. ª	Chemi⊡al
123□38□6	Propional ehv e.
79_09_4	Propioni □ a □ □.
71 23 8	n propyl al ohol.
107 10 8	Propylamine.
540 54 5	Propyl ⊑hlori⊒e.
115_07_1	Propylene. Bronylong Thloroby Trin
78-87-5	
57 55 6	Propylene gly_ol.
75 56 9	Propylene o i ⊡e.
110 86 1	Pyri ine.
106□51□4	Quinone.
108 46 3	
69-72-7	
127 09 3	So ium a etate.
532 32 1	So ium en oate.
9004 32 4	So ium ar o ymethyl ellulo e.
3926 62 3	So ium Inloroa etate.
141 53 7	So ium formate.
139_02_6	Solum phenate.
100-44-1	Styrene
110-15-6	
110_61_2	Su⊡inonitrile.
121 57 3	Sulfanili□a⊡
126 33 0	Sulfolane.
1401 55 4	Tanni a a a
100_21_0	
117 08 8	
78-00-2	Tetraethyl lea
119_64_2	Tetrahy ronaphthalene.
85 43 8	Tetrahy ⊡rophthali □ anhy ⊡ri □e.
7507401	Tetramethyl lea□
110 60 1	Tetramethylene ⊡amine.
110_18_9	Tetramethylethylene⊔amine.
95-80-7	Toluene 2 4 Tiamine
584 84 9	Toluene 2.4 iii o vanate.
26471 62 5	Toluene iioyanate miture
1333 07 9	Toluene ⊒ulfonami ⊒e.
104 🗆 15 🗆 4 🗠	Toluene ulfoni a i
98 59 9	I oluene ultonyl hlori e.
20915L12L8	
1 .	
71□55□6	1,1,1 tri⊡hloroethane.
79=00=5	1,1,2 Itri Inloroethane.
79□01□6	Tri⊡hloroethylene.
75 69 4	Trichlorofluoromethane.
96_18_4	1,2,3 III Inforopropane.
121 44 8	Triethylamine
112 27 6	Triethylene gly ol.
112 49 2	Triethylene gly ol imethyl ether.
7756 94 7	Trii 🗅 Lutylene.
75□50□3	Trimethylamine.
57 13 6	
100_05_4	Viriyi aLetate.
75_35_4	Vinyi ⊑iion∟e. Vinyli⊂ene ⊑hlori⊏e
25013 15 4	Vinvl toluene.
1330 20 7	Xylene mi e
95 47 6	o ylene.
106 42 3	p⊡ylene.
1300 71 6	Xylenol.
13001/318	∧yıı⊔ne.

a CAS num⊡er efer to the Chemical A□tratta Registry num⊡er a⊡gne to petfit themical iomer or miture of themical Some iomer or miture that are to bere by the tiantarta cont hate CAS num⊡er aligne to them. The tiantarta apply to all of the themical little, whether CAS num⊡er hate been a⊡gne or not. §60.480a

□o CAS num⊡er⊡hale ⊡een a⊡gnel to thil themilal, it⊡icomeril or militurel ontaining thele themilala □CAS num⊡eril for iome of the icomeril are litte⊡ the Itaniarti apply to all of the icomeril and militurel, ecen if CAS num⊡eril hale not icen a⊡ignel

 $[48\ {\rm FR}\ 48335,\ {\rm Oct.}\ 18,\ 1983,\ {\rm as}\ {\rm amended}\ {\rm at}\ 65\ {\rm FR}\ 61763,\ {\rm Oct.}\ 17,\ 2000]$

Subpart VVa—Standards of Performance for Equipment Leaks of VOC in the Synthetic Organic Chemicals Manufacturing Industry for Which Construction, Reconstruction, or Modification Commenced After November 7, 2006

SOURCE: $72\,$ FR 64883, Nov. 16, 2007, unless otherwise noted.

§60.480a Applicability and designation of affected facility.

(a)(1) The provisions of this subpart apply to affected facilities in the synthetic organic chemicals manufacturing industry.

(2) The group of all equipment (defined in 60.481a) within a process unit is an affected facility.

(b) Any affected facility under paragraph (a) of this section that commences construction, reconstruction, or modification after November 7, 2006, shall be subject to the requirements of this subpart.

(c) Addition or replacement of equipment for the purpose of process improvement which is accomplished without a capital expenditure shall not by itself be considered a modification under this subpart.

(d)(1) If an owner or operator applies for one or more of the exemptions in this paragraph, then the owner or operator shall maintain records as required in 60.486a(i).

(2) Any affected facility that has the design capacity to produce less than 1,000 Mg/yr (1,102 ton/yr) of a chemical listed in $\S 60.489$ is exempt from \$\$ 60.482-1a through 60.482-11a.

(3) If an affected facility produces heavy liquid chemicals only from heavy liquid feed or raw materials, then it is exempt from §§60.482-1a through 60.482-11a.

(4) Any affected facility that produces beverage alcohol is exempt from \$\$ 60.482-1a through 60.482-11a.

Appendix F
AUTHENTICATED U.S. GOVERNMENT INFORMATION

> (f) Greater sensitivity can be attained if a 500 ml capacity Tutwiler burette is used with a more dilute (0.001N)iodine solution. Concentrations less than 1.0 grains per 100 cubic foot can be determined in this way. Usually, the starch-iodine end point is much less distinct, and a blank determination of end point, with H₂S-free gas or air, is required.



FIGURE 1. TUTWILER BURETTE (LETTERED ITEMS MENTIONED IN TEXT)

Subpart MMM [Reserved]

Subpart NNN—Standards of Performance for Volatile Organic Compound (VOC) Emissions From Synthetic Organic Chemical Manufacturing Industry (SOCMI) Distillation Operations

SOURCE: 55 FR 26942, June 29, 1990, unless otherwise noted.

§60.660 Applicability and designation of affected facility.

(a) The provisions of this subpart apply to each affected facility designated in paragraph (b) of this section that is part of a process unit that produces any of the chemicals listed in $\S60.667$ as a product, co-product, byproduct, or intermediate, except as provided in paragraph (c).

(b) The affected facility is any of the following for which construction, modification, or reconstruction commenced after December 30, 1983:

(1) Each distillation unit not discharging its vent stream into a recovery system.

(2) Each combination of a distillation unit and the recovery system into which its vent stream is discharged.

(3) Each combination of two or more distillation units and the common recovery system into which their vent streams are discharged.

(c) Exemptions from the provisions of paragraph (a) of this section are as follows:

(1) Any distillation unit operating as part of a process unit which produces coal tar or beverage alcohols, or which uses, contains, and produces no VOC is not an affected facility.

(2) Any distillation unit that is subject to the provisions of subpart DDD is not an affected facility.

(3) Any distillation unit that is designed and operated as a batch operation is not an affected facility.

(4) Each affected facility that has a total resource effectiveness (TRE) index value greater than 8.0 is exempt from all provisions of this subpart except for $\S 60.662$; 60.664 (e), (f), and (g); and 60.665 (h) and (l).

(5) Each affected facility in a process unit with a total design capacity for all chemicals produced within that unit of less than one gigagram per year is exempt from all provisions of this subpart except for the recordkeeping and reporting requirements in paragraphs (j), (l)(6), and (n) of § 60.665.

(6) Each affected facility operated with a vent stream flow rate less than 0.008 scm/min is exempt from all provisions of this subpart except for the test method and procedure and the recordkeeping and reporting requirements in 60.664(g) and paragraphs (i), (1)(5), and (o) of 60.665.

(d) Alternative means of compliance— (1) Option to comply with part 65. Owners or operators of process vents that are subject to this subpart may choose to comply with the provisions of 40 CFR part 65, subpart D, to satisfy the requirements of §§60.662 through 60.665 and 60.668. The provisions of 40 CFR part 65 also satisfy the criteria of paragraphs (c)(4) and (6) of this section. Other provisions applying to an owner or operator who chooses to comply with 40 CFR part 65 are provided in 40 CFR 65.1.

(2) Part 60, subpart A. Owners or operators who choose to comply with 40 CFR part 65, subpart D, must also comply with §§ 60.1, 60.2, 60.5, 60.6, 60.7(a)(1) and (4), 60.14, 60.15, and 60.16 for those process vents. All sections and paragraphs of subpart A of this part that are not mentioned in this paragraph (d)(2) do not apply to owners or operators of process vents complying with 40 CFR part 65, subpart D, except that provisions required to be met prior to implementing 40 CFR part 65 still apply. Owners and operators who choose to comply with 40 CFR part 65, subpart D, must comply with 40 CFR part 65, subpart A.

(3) Compliance date. Owners or operators who choose to comply with 40 CFR part 65, subpart D, at initial startup shall comply with paragraphs (d)(1) and (2) of this section for each vent stream on and after the date on which the initial performance test is completed, but not later than 60 days after achieving the maximum production rate at which the affected facility will be operated, or 180 days after the initial startup, whichever date comes first.

(4) Initial startup notification. Each owner or operator subject to the provisions of this subpart that chooses to comply with 40 CFR part 65, subpart D, at initial startup shall notify the Administrator of the specific provisions of 40 CFR 65.63(a)(1), (2), or (3), with which the owner or operator has elected to comply. Notification shall be submitted with the notifications of initial startup required by 40 CFR 65.5(b).

NOTE: The intent of these standards is to minimize the emissions of VOC through the application of best demonstrated technology 40 CFR Ch. I (7–1–23 Edition)

(BDT). The numerical emission limits in these standards are expressed in terms of total organic compounds (TOC), measured as TOC less methane and ethane. This emission limit reflects the performance of BDT.

[55 FR 26942, June 29, 2000, as amended at 65 FR 78279, Dec. 14, 2000; 79 FR 11251, Feb. 27, 2014]

§60.661 Definitions.

As used in this subpart, all terms not defined here shall have the meaning given them in the Act and in subpart A of part 60, and the following terms shall have the specific meanings given them.

Batch distillation operation means a noncontinuous distillation operation in which a discrete quantity or batch of liquid feed is charged into a distillation unit and distilled at one time. After the initial charging of the liquid feed, no additional liquid is added during the distillation operation.

Boiler means any enclosed combustion device that extracts useful energy in the form of steam.

By compound means by individual stream components, not carbon equivalents.

Continuous recorder means a data recording device recording an instantaneous data value at least once every 15 minutes.

Distillation operation means an operation separating one or more feed stream(s) into two or more exit stream(s), each exit stream having component concentrations different from those in the feed stream(s). The separation is achieved by the redistribution of the components between the liquid and vapor-phase as they approach equilibrium within the distillation unit.

Distillation unit means a device or vessel in which distillation operations occur, including all associated internals (such as trays or packing) and accessories (such as reboiler, condenser, vacuum pump, steam jet, etc.), plus any associated recovery system.

Flame zone means the portion of the combustion chamber in a boiler occupied by the flame envelope.

Flow indicator means a device which indicates whether gas flow is present in a vent stream.

Halogenated vent stream means any vent stream determined to have a total

concentration (by volume) of compounds containing halogens of 20 ppmv (by compound) or greater.

Incinerator means any enclosed combustion device that is used for destroying organic compounds and does not extract energy in the form of steam or process heat.

Process heater means a device that transfers heat liberated by burning fuel to fluids contained in tubes, including all fluids except water that is heated to produce steam.

Process unit means equipment assembled and connected by pipes or ducts to produce, as intermediates or final products, one or more of the chemicals in $\S60.667$. A process unit can operate independently if supplied with sufficient fuel or raw materials and sufficient product storage facilities.

Product means any compound or chemical listed in §60.667 that is produced for sale as a final product as that chemical, or for use in the production of other chemicals or compounds. Byproducts, co-products, and intermediates are considered to be products.

Recovery device means an individual unit of equipment, such as an absorber, carbon adsorber, or condenser, capable of and used for the purpose of recovering chemicals for use, reuse, or sale.

Recovery system means an individual recovery device or series of such devices applied to the same vent stream.

Total organic compounds (TOC) means those compounds measured according to the procedures in §60.664(b)(4). For the purposes of measuring molar composition as required in (60.664(d)(2)(i)); hourly emissions rate as required in §60.664(d)(5) and §60.664(e); and TOC concentration asrequired in 60.665(b)(4) and 60.665(g)(4), those compounds which the Administrator has determined do not contribute appreciably to the formation of ozone are to be excluded. The compounds to be excluded are identified in Environmental Protection Agency's statements on ozone abatement policy for State Implementation Plans (SIP) revisions (42 FR 35314; 44 FR 32042; 45 FR 32424; 45 FR 48942).

TRE index value means a measure of the supplemental total resource requirement per unit reduction of TOC associated with an individual distillation vent stream, based on vent stream flow rate, emission rate of TOC net heating value, and corrosion properties (whether or not the vent stream is halogenated), as quantified by the equation given under §60.664(e).

Vent stream means any gas stream discharged directly from a distillation facility to the atmosphere or indirectly to the atmosphere after diversion through other process equipment. The vent stream excludes relief valve discharges and equipment leaks including, but not limited to, pumps, compressors, and valves.

§60.662 Standards.

Each owner or operator of any affected facility shall comply with paragraph (a), (b), or (c) of this section for each vent stream on and after the date on which the initial performance test required by §§60.8 and 60.664 is completed, but not later than 60 days after achieving the maximum production rate at which the affected facility will be operated, or 180 days after the initial start-up, whichever date comes first. Each owner or operator shall either:

(a) Reduce emissions of TOC (less methane and ethane) by 98 weight-percent, or to a TOC (less methane and ethane) concentration of 20 ppmv, on a dry basis corrected to 3 percent oxygen, whichever is less stringent. If a boiler or process heater is used to comply with this paragraph, then the vent stream shall be introduced into the flame zone of the boiler or process heater; or

(b) Combust the emissions in a flare that meets the requirements of 60.18; or

(c) Maintain a TRE index value greater than 1.0 without use of VOC emission control devices.

§60.663 Monitoring of emissions and operations.

(a) The owner or operator of an affected facility that uses an incinerator to seek to comply with the TOC emission limit specified under §60.662(a) shall install, calibrate, maintain, and operate according to manufacturer's specifications the following equipment:

(1) A temperature monitoring device equipped with a continuous recorder and having an accuracy of ± 1 percent of the temperature being monitored expressed in degrees Celsius or ± 0.5 °C, whichever is greater.

(i) Where an incinerator other than a catalytic incinerator is used, a temperature monitoring device shall be installed in the firebox.

(ii) Where a catalytic incinerator is used, temperature monitoring devices shall be installed in the gas stream immediately before and after the catalyst bed.

(2) A flow indicator that provides a record of vent stream flow to the incinerator at least once every hour for each affected facility. The flow indicator shall be installed in the vent stream from each affected facility at a point closest to the inlet of each incinerator and before being joined with any other vent stream.

(b) The owner or operator of an affected facility that uses a flare to seek to comply with §60.662(b) shall install, calibrate, maintain and operate according to manufacturer's specifications the following equipment:

(1) A heat sensing device, such as an ultra-violet beam sensor or thermocouple, at the pilot light to indicate the continuous presence of a flame.

(2) A flow indicator that provides a record of vent stream flow to the flare at least once every hour for each affected facility. The flow indicator shall be installed in the vent stream from each affected facility at a point closest to the flare and before being joined with any other vent stream.

(c) The owner or operator of an affected facility that uses a boiler or process heater to seek to comply with §60.662(a) shall install, calibrate, maintain and operate according to the manufacturer's specifications the following equipment:

(1) A flow indicator that provides a record of vent stream flow to the boiler or process heater at least once every hour for each affected facility. The flow indicator shall be installed in the vent stream from each distillation unit within an affected facility at a point closest to the inlet of each boiler or process heater and before being joined with any other vent stream.

(2) A temperature monitoring device in the firebox equipped with a contin40 CFR Ch. I (7–1–23 Edition)

uous recorder and having an accuracy of ± 1 percent of the temperature being measured expressed in degrees Celsius or ± 0.5 °C, whichever is greater, for boilers or process heaters of less than 44 MW (150 million Btu/hr) heat input design capacity.

(d) Monitor and record the periods of operation of the boiler or process heater if the design heat input capacity of the boiler or process heater is 44 MW (150 million Btu/hr) or greater. The records must be readily available for inspection.

(e) The owner or operator of an affected facility that seeks to comply with the TRE index value limit specified under §60.662(c) shall install, calibrate, maintain, and operate according to manufacturer's specifications the following equipment, unless alternative monitoring procedures or requirements are approved for that facility by the Administrator:

(1) Where an absorber is the final recovery device in the recovery system:

(i) A scrubbing liquid temperature monitoring device having an accuracy of ± 1 percent of the temperature being monitored expressed in degrees Celsius or ± 0.5 °C, whichever is greater, and a specific gravity monitoring device having an accuracy of ± 0.02 specific gravity units, each equipped with a continuous recorder, or

(ii) An organic monitoring device used to indicate the concentration level of organic compounds exiting the recovery device based on a detection principle such as infrared, photoionization, or thermal conductivity, each equipped with a continuous recorder.

(2) Where a condenser is the final recovery device in the recovery system:

(i) A condenser exit (product side) temperature monitoring device equipped with a continuous recorder and having an accuracy of ± 1 percent of the temperature being monitored expressed in degrees Celsius or ± 0.5 °C, whichever is greater, or

(ii) An organic monitoring device used to monitor organic compounds exiting the recovery device based on a detection principle such as infra-red, photoionization, or thermal conductivity, each equipped with a continuous recorder.

(3) Where a carbon adsorber is the final recovery device unit in the recovery system:

(i) An integrating steam flow monitoring device having an accuracy of ± 10 percent, and a carbon bed temperature monitoring device having an accuracy of ± 1 percent of the temperature being monitored expressed in degrees Celsius or ± 0.5 °C, whichever is greater, both equipped with a continuous recorder, or

(ii) An organic monitoring device used to indicate the concentration level of organic compounds exiting the recovery device based on a detection principle such as infra-red, photoionization, or thermal conductivity, each equipped with a continuous recorder.

(f) An owner or operator of an affected facility seeking to demonstrate compliance with the standards specified under §60.662 with control devices other than incinerator, boiler, process heater, or flare; or recovery device other than an absorber, condenser, or carbon adsorber shall provide to the Administrator information describing the operation of the control device or recovery device and the process parameter(s) which would indicate proper operation and maintenance of the device. The Administrator may request further information and will specify appropriate monitoring procedures or requirements.

 $[55\ {\rm FR}\ 26942,\ {\rm June}\ 29,\ 1990,\ {\rm as}\ {\rm amended}\ {\rm at}\ 65\ {\rm FR}\ 61774,\ {\rm Oct.}\ 17,\ 2000]$

§60.664 Test methods and procedures.

(a) For the purpose of demonstrating compliance with §60.662, all affected facilities shall be run at full operating conditions and flow rates during any performance test.

(b) The following methods in appendix A to this part, except as provided under §60.8(b), shall be used as reference methods to determine compliance with the emission limit or percent reduction efficiency specified under §60.662(a).

(1) Method 1 or 1A, as appropriate, for selection of the sampling sites. The control device inlet sampling site for determination of vent stream molar composition or TOC (less methane and ethane) reduction efficiency shall be

LAPBH6H6L3 with DISTILLER

prior to the inlet of the control device and after the recovery system.

§60.664

(2) Method 2, 2A, 2C, or 2D, as appropriate, for determination of the gas volumetric flow rates.

(3) The emission rate correction factor, integrated sampling and analysis procedure of Method 3 shall be used to determine the oxygen concentration $(\%O_{2d})$ for the purposes of determining compliance with the 20 ppmv limit. The sampling site shall be the same as that of the TOC samples, and the samples shall be taken during the same time that the TOC samples are taken.

The TOC concentration corrected to 3 percent 0_2 (C_c) shall be computed using the following equation:

$$C_c = C_{TOC} \frac{17.9}{20.9 - \%O_{2d}}$$

where:

 C_c = Concentration of TOC corrected to 3 percent O₂, dry basis, ppm by volume.

C_{TOC} = Concentration of TOC (minus methane and ethane), dry basis, ppm by volume.

 O_{2d} = Concentration of O_2 , dry basis, percent by volume.

(4) Method 18 to determine the concentration of TOC in the control device outlet and the concentration of TOC in the inlet when the reduction efficiency of the control device is to be determined.

(i) The sampling time for each run shall be 1 hour in which either an integrated sample or four grab samples shall be taken. If grab sampling is used then the samples shall be taken at 15minute intervals.

(ii) The emission reduction (R) of TOC (minus methane and ethane) shall be determined using the following equation:

$$R = \frac{E_i - E_o}{E_i} \times 100$$

where:

- $$\label{eq:rescaled} \begin{split} R &= Emission \mbox{ reduction, percent by weight.} \\ E_i &= Mass \mbox{ rate of TOC entering the control} \end{split}$$
- device, kg/hr (lb/hr). E_o = Mass rate of TOC discharged to the atmosphere, kg/hr (lb/hr).
- (iii) The mass rates of TOC (E_i, E_o) shall be computed using the following equations:

EC16 091.097 < TH:

EC16001

§60.664

$$E_{i} = K_{2} \left(\sum_{j=1}^{n} C_{ij} M_{ij} \right) Q_{i}$$
$$E_{O} = K_{2} \left(\sum_{j=1}^{n} C_{oj} M_{oj} \right) Q_{O}$$

where:

- $$\begin{split} M_{ij},\,M_{oj} &= Molecular \mbox{ weight of sample compo-}\\ &nent ``j'' \mbox{ of the gas stream at the inlet}\\ &and \mbox{ outlet of the control device, respectively, g/g-mole} \ (lb/lb-mole). \end{split}$$
- Q_i, Q_o = Flow rate of gas stream at the inlet and outlet of the control device, respectively, dscm/min (dscf/min).
- $\begin{array}{l} {\rm K_2~=~2.494~\times~10^{-6}~(1/ppm)(g-mole/scm)~(kg/g)}\\ {\rm (min/hr)~(metric~units),~where~standard}\\ {\rm temperature~for~(g-mole/scm)~is~20~^{\circ}C}. \end{array}$
- = 1.557×10^{-7} (l/ppm) (lb-mole/scf) (min/hr) (English units), where standard temperature for (lb-mole/scf) is 68 °F.

(iv) The TOC concentration (C_{TOC}) is the sum of the individual components and shall be computed for each run using the following equation:

$$C_{TOC} = \sum_{i=1}^{n} C_{i}$$

where:

- C_{TOC} = Concentration of TOC (minus methane and ethane), dry basis, ppm by volume
- C_j = Concentration of sample components "j", dry basis, ppm by volume.

n = Number of components in the sample.

(c) When a boiler or process heater with a design heat input capacity of 44 MW (150 million Btu/hour) or greater is used to seek to comply with 60.662(a), the requirement for an initial performance test is waived, in accordance with 60.8(b). However, the Administrator reserves the option to require testing at such other times as may be required, as provided for in section 114 of the Act.

(d) When a flare is used to seek to comply with 60.662(b), the flare shall comply with the requirements of 60.18.

(e) The following test methods in appendix A to this part, except as provided under 60.8(b), shall be used for determining the net heating value of

40 CFR Ch. I (7–1–23 Edition)

the gas combusted to determine compliance under §60.662(b) and for determining the process vent stream TRE index value to determine compliance under §60.662(c).

(1)(i) Method 1 or 1A, as appropriate, for selection of the sampling site. The sampling site for the vent stream flow rate and molar composition determination prescribed in $\S60.664(e)(2)$ and (3) shall be, except for the situations outlined in paragraph (e)(1)(ii) of this section, prior to the inlet of any control device, prior to any post-distillation dilution of the stream with air, and prior to any post-distillation introduction of halogenated compounds into the process vent stream. No transverse site selection method is needed for vents smaller than 10 centimeters (4 inches) in diameter.

(ii) If any gas stream other than the distillation vent stream from the affected facility is normally conducted through the final recovery device.

(A) The sampling site for vent stream flow rate and molar composition shall be prior to the final recovery device and prior to the point at which the nondistillation stream is introduced.

(B) The efficiency of the final recovery device is determined by measuring the TOC concentration using Method 18 at the inlet to the final recovery device after the introduction of any nondistillation vent stream and at the outlet of the final recovery device.

(C) This efficiency is applied to the TOC concentration measured prior to the final recovery device and prior to the introduction of the nondistillation stream to determine the concentration of TOC in the distillation vent stream from the final recovery device. This concentration of TOC is then used to perform the calculations outlined in §60.664(e)(4) and (5).

(2) The molar composition of the process vent stream shall be determined as follows:

(i) Method 18 to measure the concentration of TOC including those containing halogens.

(ii) ASTM D1946-77 or 90 (Reapproved 1994) (incorporation by reference as specified in §60.17 of this part) to measure the concentration of carbon monoxide and hydrogen.

(iii) Method 4 to measure the content of water vapor.

(3) The volumetric flow rate shall be determined using Method 2, 2A, 2C, or 2D, as appropriate.

(4) The net heating value of the vent stream shall be calculated using the following equation:

$$H_{T} = K_{1} \left(\sum_{j=1}^{n} C_{j} H_{j} \right)$$

where:

- $H_{\rm T}$ = Net heating value of the sample, MJ/ scm (Btu/scf), where the net enthalpy per mole of vent stream is based on combustion at 25 °C and 760 mm Hg (77 °F and 30 in. Hg), but the standard temperature for determining the volume corresponding to one mole is 20 °C (68 °F).
- $\begin{array}{ll} {\rm K_1} &= 1.74 \, \times \, 10^{-7} \, \left({\rm 1/ppm} \right) \, \left({\rm g-mole/scm} \right) \, \left({\rm MJ} \right) \\ & {\rm kcal} \left({\rm metric \ units} \right) , \, {\rm where \ standard \ temperature \ for \ (g-mole/scm) \ is \ 20 \ ^{\circ}{\rm C} . \\ &= 1.03 \, \times \, 10^{-11} \, \left({\rm 1/ppm} \right) \, \left({\rm lb-mole/scf} \right) \, \left({\rm Btu} \right) \end{array}$
- = 1.03×10^{-11} (1/ppm) (lb-mole/scf) (Btu/ kcal) (English units) where standard temperature for (lb/mole/scf) is 68 °F.
- C_j = Concentration on a wet basis of compound j in ppm, as measured for organics by Method 18 and measured for hydrogen and carbon monoxide by ASTM D1946-77 or 90 (Reapproved 1994) (incorporation by reference as specified in §60.17 of this part) as indicated in §60.664(e)(2).
- $\rm H_{j}$ = Net heat of combustion of compound j, kcal/(g-mole) [kcal/(lb-mole)], based on combustion at 25 °C and 760 mm Hg (77 °F and 30 in. Hg).

The heats of combustion of vent stream components would be required to be determined using ASTM D2382-76 (incorporation by reference as specified in §60.17 of this part) if published values are not available or cannot be calculated.

(5) The emission rate of TOC in the vent stream shall be calculated using the following equation:

$$E_{TOC} = K_2 \left[\sum_{j=1}^{n} C_j M_j \right] Q_s$$

§ 60.664

where:

- $\mathrm{E}_{\mathrm{TOC}}$ = Measured emission rate of TOC, kg/hr (lb/hr).
- $\begin{array}{l} K_2 = 2.494 \times 10^{-6} \ (1/ppm) \ (g-mole/scm) \ (kg/g) \\ (min/hr) \ (metric \ units), \ where \ standard \\ temperature \ for \ (g-mole/scm) \ is \ 20 \ ^{\circ}C. \end{array}$
- = 1.557×10^{-7} (1/ppm) (lb-mole/scf) (min/hr) (English units), where standard temperature for (lb-mole/scf) is 68 °F.
- $\begin{array}{l} C_{j} = Concentration \mbox{ on a wet basis of compound } j \mbox{ in ppm, as measured by Method} \\ 18 \mbox{ as indicated in } \$60.664(e)(2). \end{array}$
- $$\begin{split} M_{j} &= Molecular \text{ weight of sample j, g/g-mole} \\ & (lb/lb-mole). \end{split}$$
- Q_s = Vent stream flow rate, scm/min (scf/ min), at a temperature of 20 $^\circ C$ (68 $^\circ F).$

(6) The total process vent stream concentration (by volume) of compounds containing halogens (ppmv, by compound) shall be summed from the individual concentrations of compounds containing halogens which were measured by Method 18.

(f) For purposes of complying with $\S60.662(c)$ the owner or operator of a facility affected by this subpart shall calculate the TRE index value of the vent stream using the equation for incineration in paragraph (e)(1) of this section for halogenated vent streams. The owner or operator of an affected facility with a nonhalogenated vent stream shall determine the TRE index value by calculating values using both the incinerator equation in (e)(1) and the flare equation in (e)(2) of the two values.

(1) The equation for calculating the TRE index value of a vent stream controlled by an incinerator is as follows:

$$TRE = \frac{1}{E_{TOC}} \left[a + b(Q_s)^{0.88} + c(Q_s) + d(Q_s)(H_T) + e(Q_s)^{0.88} (H_T)^{0.88} + f(Y_s)^{0.5} \right]$$

(i) Where for a vent stream flow rate that is greater than or equal to 14.2 scm/min (501 scf/min) at a standard temperature of 20 °C (68 °F):

 Q_s = Vent stream flow rate, scm/min (scf/ min), at a temperature of 20 $^\circ C$ (68 $^\circ F).$

 $\rm H_T$ = Vent stream net heating value, MJ/scm (Btu/scf), where the net enthalpy per mole of vent stream is based on combustion at 25 °C and 760 mm Hg (68 °F and 30

TRE = TRE index value.

LAPBH6H6L3 with DISTILLER

§60.664

40 CFR Ch. I (7-1-23 Edition)

in. Hg), but the standard temperature for determining the volume corresponding to one mole is 20 °C (68 °F) as in the definition of $Q_{\rm s.}$

 $E_{\rm TOC}$ = Hourly emissions of TOC, kg/hr (lb/ hr).

a, b, c, d, e, and f are coefficients.

The set of coefficients that apply to a vent stream can be obtained from table 1.

TABLE 1. DISTILLATION NSPS TRE COEFFICIENTS FOR VENT STREAMS

CONTROLLED BY AN INCINERATOR

DESIGN CATEGORY A1. FOR HALOGENATED PROCESS VENT STREAMS, IF 0 < NET HEATING VALUE (MJ/scm) < 3.5 OR IF 0 < NET HEATING VALUE (Btu/scf) < 94:

Q _s = Vent Stream Flow rate scm/min (scf/min)	а	ь	с	d	e	f
14.2 ≤ Q _s ≤ 18.8	18.84466	0.26742	-0.20044	0	0	0.01025
(501 < Q _s ≤ 664)	(41.54494)	(0.016696)	(-0.019194)	(0)	(0)	(0.003803)
18.8 < Q _s ≤ 699	19.66658	0.26742	-0.25332	0	0	0.01025
(664 < Q _s ≤ 24,700)	(43.35694)	(0.016696)	(-0.024258)	(0)	(0)	(0.003803)
699 < Q _s ≤ 1400	39.19213	0.29062	-0.25332	0	0	0.01449
(24,700 < Q _s ≤ 49,000)	(86.40297)	(0.018145)	(-0.024258)	(0)	(0)	(0.005376)
1400 < Q _s ≲ 2100	58.71768	0.30511	-0.25332	0	0	0.01775
(49,000 < Q _s ≤ 74,000)	(129.4490)	(0.019050)	(-0.024258)	(0)	(0)	(0.006585)
2100 < Q _s ≤ 2800	78.24323	0.31582	-0.25332	0	0	0.02049
(74,000 < Q _S < 99,000)	(172.4950)	(0.019718)	(-0.024258)	(0)	(0)	(0.007602)
2800 < Q _s < 3500	97.76879	0.32439	-0.25332	0	0	0.02291
(99,000 < Q _S = 120,000)	(215.5411)	(0.020253)	(-0.024258)	(0)	(0)	(0.008500)

§60.664

DESIGN CATEGORY A2.

FOR HALOGENATED PROCESS VENT STREAMS, IF NET HEATING VALUE < 3.5 (MJ/scm) OR IF NET HEATING VALUE < 94 (Btu/scf):

Q ₅ = Vent Stream Flow rate scm/min(scf/min)	а	b	с	đ	e	f
14.2 ≤ Q _s ≤ 18.8	18.84466	0.26742	-0.20044	0	0	0.01025
(501 ≤ Q _s ≤ 664)	(41.54494)	(0.016696)	(-0.019194)	(0)	(0)	(0.003803)
18.8 < Q _s < 699	19.66658	0.26742	-0.25332	0	0	0.01025
(664 < Q _s ≤ 24,700)	(43.35694)	(0.016696)	(-0.024258)	(0)	(0)	(0.003803)
699 < Q _s < 1400	39.19213	0.29062	-0.25332	0	0	0.01449
(24,700 < Q ₅ ≤ 49,000)	(86.40297)	(0.018145)	(-0.024258)	(0)	(0)	(0.005376)
1400 < Q _s ≤ 2100	58.71768	0.30511	-0.25332	0	0	0.01775
(49,000 < Q _s ≤ 74,000)	(129.4490)	(0.019050)	(-0.024258)	(0)	(0)	(0.006585)
2100 < Q _s < 2800	78.24323	0.31582	-0.25332	0	0	0.02049
(74,000 < Q _s ≤ 99,000)	(172.4950)	(0.019718)	(-0.024258)	(0)	(0)	(0.007602)
2800 < Q _s ≤ 3500	97.76879	0.32439	-0.25332	0	0	0.02291
(99,000 < Q _s < 120,000)	(215.5411)	(0.020253)	(-0.024258)	(0)	(0)	(0.008500)

DESIGN CATEGORY B. FOR NONHALOGENATED PROCESS VENT STREAMS, IF 0 ≤ NET HEATING VALUE (MJ/scm) ≤ 0.48 OR IF 0 ≤ NET HEATING VALUE (Btu/scf) ≤ 13:

Q _s = Vent Stream Flow rate scm/min(scf/min)	а	b	с	đ	e	f
$\begin{array}{l} 14.2 \leq Q_{\rm S} \leq 1340 \\ (501 \leq Q_{\rm S} \leq 47,300) \\ 1340 < Q_{\rm S} \leq 2690 \\ (47,300 < Q_{\rm S} \leq 95,000) \\ 2690 < Q_{\rm S} \leq 4040 \\ (95,000 < Q_{\rm S} \leq 143,000) \end{array}$	8.54245 (18.83268) 16.94386 (37.35443) 25.34528 (55.87620)	0.10555 (0.0065901) 0.11470 (0.0071614) 0.12042 (0.0075185)	0.09030 (0.008647) 0.09030 (0.008647) 0.09030 (0.008647)	-0.17109 (-0.00039762) -0.17109 (-0.00039762) -0.17109 (-0.00039762)	0 (0) (0) 0 (0)	0.01025 (0.003803) 0.01449 (0.005376) 0.01775 (0.006585)

§60.664

40 CFR Ch. I (7-1-23 Edition)

DESIGN CATEGORY C. FOR NONHALOGENTED PROCESS VENT STREAMS, IF 0.48 < NET HEATING VALUE (MJ/scm) ≤ 1.9 OR IF 13 < NET HEATING VALUE (Btu/scf) < 51:

Q _s = Vent Stream Flow rate scm/min(scf/min)	а	b	с	đ	е	f
14.2 ≤ Q _s ≤ 1340	9.25233	0.06105	0.31937	-0.16181	0	0.01025
(501 ≤ Q _s ≤ 47,300)	(20.39769)	(0.003812)	(0.030582)	(-0.00037605)	(0)	(0.003803)
1340 < Q _s < 2690	18.36363	0.06635	0.31937	-0.16181	0	0.01449
(47,300 < Q _s < 95,000)	(40.48446)	(0.004143)	(0.030582)	(-0.00037605)	(0)	(0.005376)
2690 < Q _S ≤ 4040	27.47492	0.06965	0.31937	-0.16181	0	0.01775
(95,000 < Q ₅ ≤ 143,000)	(60.57121)	(0.004349)	(0.030582)	(-0.00037605)	(0)	(0.006585)

DESIGN CATEGORY D. FOR NONHALOGENATED PROCESS VENT STREAMS, IF 1.9 < NET HEATING VALUE (MJ/scm) ≤ 3.6 OR IF 51 < NET HEATING VALUE (Btu/scf) is 97:

Q _s = Vent Stream Flow rate scm/min(scf/min)	а	b	с	d	e	f
$\begin{array}{c} \text{schrink}(\text{schrink})\\ 14.2 \leq Q_{\text{S}} \leq 1180\\ (501 \pm Q_{\text{S}} \leq 41,700)\\ 1180 < Q_{\text{S}} \leq 2370\\ (41,700 < Q_{\text{S}} \leq 83,700)\\ 2370 < Q_{\text{S}} \leq 3550 \end{array}$	6.67868 (14.72382) 13.21633 (29.13672) 19.75398	0.06943 (0.004335) 0.07546 (0.004711) 0.07922	0.02582 (0.002472) 0.02582 (0.002472) 0.02582	0 (0) 0 (0) 0	0 (0) 0 (0) 0	0.01025 (0.003803) 0.01449 (0.005376) 0.01775
(83,700 < Q _S ≤ 125,000)	(43.54962)	(0.004946)	(0.002472)	(0)	(0)	(0.006585)

DESIGN CATEGORY E. FOR NONHALOGENATED PROCESS VENT STREAMS, IF NET HEATING VALUE > 3.6 MJ/scm OR IF NET HEATING VALUE > 97 (Btu/scf):

Q _s = Vent Stream Flow rate scm/min(scf/min)	а	b	С	d	е	f
14.2 ≤ Y _S ≤ 1180	6.67868	0	0	-0.00707	0.02220	0.01025
(501 ≤ Y _s ≤ 41,700)	(14.72382)	(0)	(0)	(-0.0000164)	(0.0001174)	(0.003803)
1180 < Y _s ≲ 2370	13.21633	0	0	-0.00707	0.02412	0.01449
(41,700 < Y _s ≤ 83,700)	(29.13672)	(0)	(0)	(-0.0000164)	(0.0001276)	(0.005376)
2370 < Y _s < 3550	19.75398	0	0	-0.00707	0.02533	0.01775
(83,700 < Y _s ≤ 125,000)	(43.54962)	(0)	(0)	(-0.0000164)	(0.0001340)	(0.006585)
-						

(ii) Where for a vent stream flow rate that is less than 14.2 scm/min (501 scf/ min) at a standard temperature of 20 °C (68 °F):

TRE = TRE index value.

on LAPBH6H6L3 with DISTILLER

 $Q_{\rm s}$ = 14.2 scm/min (501 scf/min). $H_T = (FLOW) (HVAL)/Q_s.$

Where the following inputs are used:

FLOW = Vent stream flow rate, scm/min (scf/ min), at a temperature of 20 °C (68 °F).

 $\rm HVAL$ = Vent stream net heating value, $\rm MJ/$ scm (Btu/scf), where the net enthalpy per mole of vent stream is based on combustion at 25 $^{\circ}\mathrm{C}$ and 760 mm Hg (68 $^{\circ}\mathrm{F}$ and 30

in. Hg), but the standard temperature for determining the volume corresponding to one mole is 20 $^{\circ}\mathrm{C}$ (68 $^{\circ}\mathrm{F})$ as in the definition of $Q_{\rm s}.$

- $Y_s = Q_s$ for all vent stream categories listed in table 1 except for Category E vent streams where $Y_s = Q_s H_T/3.6$.
- $E_{\rm TOC}$ = Hourly emissions of TOC, kg/hr (lb/ hr).

a, b, c, d, e, and f are coefficients

The set of coefficients that apply to a vent stream can be obtained from table 1.

§60.665

(2) The equation for calculating the TRE index value of a vent stream controlled by a flare is as follows:

$$TRE = \frac{1}{E_{TOC}} \Big[a(Q_s) + b(Q_s)^{0.8} + c(Q_s)(H_T) + d(E_{TOC}) + e \Big]$$

where:

TRE = TRE index value.

- $E_{\rm TOC}$ = Hourly emissions of TOC, kg/hr (lb/ hr).
- Q_s = Vent stream flow rate, scm/min (scf/min), at a standard temperature of 20 °C (68 °F).
- H_T = Vent stream net heating value, MJ/scm $({\rm Btu/scf}),$ where the net enthalpy per mole of vent stream is based on combus-

tion at 25 °C and 760 mm Hg (68 °F and 30 in. Hg), but the standard temperature for determining the volume corresponding to one mole is 20 °C (68 °F) as in the definition of Q_s .

a, b, c, d, and e are coefficients.

The set of coefficients that apply to a vent stream shall be obtained from table 2.

TABLE 2 DISTILLATIO DSPS TRE COEFFICIEDTS FOR VEDT STREADS CODTROLLED BD A FLARE

	а				е
$\begin{array}{l} H_T < 11.2 \ \ \Box \ \ J \blacksquare \square m \\ H_T < 301 \ Btu \blacksquare \square f \\ H_T \ge 11.2 \ \ \Box \ \ J \blacksquare \square m \\ H_T \ge 301 \ Btu \blacksquare \square f \\ \end{array}$	2.25	0.288	-0.193	-0.0051	2.08
	0.140	10.0367	-0.000448	-0.0051	[4.59]
	0.309	0.0619	-0.0043	-0.0034	2.08
	0.0193	10.00788	-0.000010	-0.0034	[4.59]

149

(g) Each owner or operator of an affected facility seeking to comply with §60.660(c)(4) or §60.662(c) shall recalculate the TRE index value for that affected facility whenever process changes are made. Examples of process changes include changes in production capacity, feedstock type, or catalyst type, or whenever there is replacement, removal, or addition of recovery equipment. The TRE index value shall be recalculated based on test data, or on best engineering estimates of the effects of the change to the recovery system.

(1) Where the recalculated TRE index value is less than or equal to 1.0, the owner or operator shall notify the Administrator within 1 week of the recalculation and shall conduct a performance test according to the methods and procedures required by §60.664 in order to determine compliance with §60.662(a). Performance tests must be conducted as soon as possible after the process change but no later than 180 days from the time of the process change.

(2) Where the initial TRE index value is greater than 8.0 and the recalculated

TRE index value is less than or equal to 8.0 but greater than 1.0, the owner or operator shall conduct a performance test in accordance with §§ 60.8 and 60.664 and shall comply with §§ 60.663, 60.664 and 60.665. Performance tests must be conducted as soon as possible after the process change but no later than 180 days from the time of the process change.

(h) Any owner or operator subject to the provisions of this subpart seeking to demonstrate compliance with §60.660(c)(6) shall use Method 2, 2A, 2C, or 2D as appropriate, for determination of volumetric flow rate.

 $[55\ {\rm FR}$ 26942, June 29, 1990, as amended at 65 ${\rm FR}$ 61774, Oct. 17, 2000]

§60.665 Reporting and recordkeeping requirements.

(a) Each owner or operator subject to $\S60.662$ shall notify the Administrator of the specific provisions of $\S60.662$ ($\S60.662$ (a), (b), or (c)) with which the owner or operator has elected to comply. Notification shall be submitted with the notification of initial start-up required by $\S60.7(a)(3)$. If an owner or operator elects at a later date to use an

alternative provision of §60.662 with which he or she will comply, then the Administrator shall be notified by the owner or operator 90 days before implementing a change and, upon implementing the change, a performance test shall be performed as specified by §60.664 within 180 days.

(b) Each owner or operator subject to the provisions of this subpart shall keep an up-to-date, readily accessible record of the following data measured during each performance test, and also include the following data in the report of the initial performance test required under §60.8. Where a boiler or process heater with a design heat input capacity of 44 MW (150 million Btu/hour) or greater is used to comply with §60.662(a), a report containing performance test data need not be submitted, but a report containing the information in §60.665(b)(2)(i) is required. The same data specified in this section shall be submitted in the reports of all subsequently required performance tests where either the emission control efficiency of a control device, outlet concentration of TOC, or the TRE index value of a vent stream from a recovery system is determined.

(1) Where an owner or operator subject to the provisions of this subpart seeks to demonstrate compliance with §60.662(a) through use of either a thermal or catalytic incinerator:

(i) The average firebox temperature of the incinerator (or the average temperature upstream and downstream of the catalyst bed for a catalytic incinerator), measured at least every 15 minutes and averaged over the same time period of the performance testing, and

(ii) The percent reduction of TOC determined as specified in §60.664(b) achieved by the incinerator, or the concentration of TOC (ppmv, by compound) determined as specified in §60.664(b) at the outlet of the control device on a dry basis corrected to 3 percent oxygen.

(2) Where an owner or operator subject to the provisions of this subpart seeks to demonstrate compliance with §60.662(a) through use of a boiler or process heater:

(i) A description of the location at which the vent stream is introduced into the boiler or process heater, and 40 CFR Ch. I (7–1–23 Edition)

(ii) The average combustion temperature of the boiler or process heater with a design heat input capacity of less than 44 MW (150 million Btu/hr) measured at least every 15 minutes and averaged over the same time period of the performance testing.

(3) Where an owner or operator subject to the provisions of this subpart seeks to demonstrate compliance with §60.662(b) through use of a smokeless flare, flare design (i.e., steam-assisted, air-assisted or nonassisted), all visible emission readings, heat content determinations, flow rate measurements, and exit velocity determinations made during the performance test, continuous records of the flare pilot flame monitoring, and records of all periods of operations during which the pilot flame is absent.

(4) Where an owner or operator subject to the provisions of this subpart seeks to demonstrate compliance with §60.662(c):

(i) Where an absorber is the final recovery device in the recovery system, the exit specific gravity (or alternative parameter which is a measure of the degree of absorbing liquid saturation, if approved by the Administrator), and average exit temperature, of the absorbing liquid measured at least every 15 minutes and averaged over the same time period of the performance testing (both measured while the vent stream is normally routed and constituted), or

(ii) Where a condenser is the final recovery device in the recovery system, the average exit (product side) temperature measured at least every 15 minutes and averaged over the same time period of the performance testing while the vent stream is routed and constituted normally, or

(iii) Where a carbon adsorber is the final recovery device in the recovery system, the total steam mass flow measured at least every 15 minutes and averaged over the same time period of the performance test (full carbon bed cycle), temperature of the carbon bed after regeneration (and within 15 minutes of completion of any cooling cycle(s)), and duration of the carbon bed steaming cycle (all measured while the vent stream is routed and constituted normally), or

(iv) As an alternative to §60.665(b)(4) ((i), (ii) or (iii), the concentration level or reading indicated by the organics monitoring device at the outlet of the absorber, condenser, or carbon adsorber, measured at least every 15 minutes and averaged over the same time period of the performance testing while the vent stream is normally routed and constituted.

(v) All measurements and calculations performed to determine the TRE index value of the vent stream.

(c) Each owner or operator subject to the provisions of this subpart shall keep up-to-date, readily accessible continuous records of the equipment operating parameters specified to be monitored under §60.663 (a) and (c) as well as up-to-date, readily accessible records of periods of operation during which the parameter boundaries established during the most recent performance test are exceeded. The Administrator may at any time require a report of these data. Where a combustion device is used to comply with §60.662(a), periods of operation during which the parameter boundaries established during the most recent performance tests are exceeded are defined as follows:

(1) For thermal incinerators, all 3hour periods of operation during which the average combustion temperature was more than 28 °C (50 °F) below the average combustion temperature during the most recent performance test at which compliance with 60.662(a)was determined.

(2) For catalytic incinerators, all 3hour periods of operation during which the average temperature of the vent stream immediately before the catalyst bed is more than 28 °C (50 °F) below the average temperature of the vent stream during the most recent performance test at which compliance with §60.662(a) was determined. The owner or operator also shall record all 3-hour periods of operation during which the average temperature difference across the catalyst bed is less than 80 percent of the average temperature difference of the device during the most recent performance test at which compliance with §60.662(a) was determined.

(3) All 3-hour periods of operation during which the average combustion temperature was more than 28 °C (50 °F) below the average combustion temperature during the most recent performance test at which compliance with $\S60.662(a)$ was determined for boilers or process heaters with a design heat input capacity of less than 44 MW (150 million Btu/hr).

(4) For boilers or process heaters, whenever there is a change in the location at which the vent stream is introduced into the flame zone as required under 60.662(a).

(d) Each owner or operator subject to the provisions of this subpart shall keep up to date, readily accessible continuous records of the flow indication specified under $\{60.663(a)(2),$ $\{60.663(b)(2)$ and $\{60.663(c)(1), as well as$ up-to-date, readily accessible records of all periods when the vent stream is diverted from the control device or has no flow rate.

(e) Each owner or operator subject to the provisions of this subpart who uses a boiler or process heater with a design heat input capacity of 44 MW (150 million Btu/hour) or greater to comply with §60.662(a) shall keep an up-todate, readily accessible record of all periods of operation of the boiler or process heater. (Examples of such records could include records of steam use, fuel use, or monitoring data collected pursuant to other State or Federal regulatory requirements.)

(f) Each owner or operator subject to the provisions of this subpart shall keep up-to-date, readily accessible continuous records of the flare pilot flame monitoring specified under 60.663(b), as well as up-to-date, readily accessible records of all periods of operations in which the pilot flame is absent.

(g) Each owner or operator subject to the provisions of this subpart shall keep up-to-date, readily accessible continuous records of the equipment operating parameters specified to be monitored under § 60.663(e), as well as up-todate, readily accessible records of periods of operation during which the parameter boundaries established during the most recent performance test are exceeded. The Administrator may at any time require a report of these data. Where an owner or operator seeks to comply with §60.662(c), periods of operation during which the parameter boundaries established during the most recent performance tests are exceeded are defined as follows:

(1) Where an absorber is the final recovery device in a recovery system, and where an organic compound monitoring device is not used:

(i) All 3-hour periods of operation during which the average absorbing liquid temperature was more than 11 °C (20 °F) above the average absorbing liquid temperature during the most recent performance test, or

(ii) All 3-hour periods of operation during which the average absorbing liquid specific gravity was more than 0.1 unit above, or more than 0.1 unit below, the average absorbing liquid specific gravity during the most recent performance test (unless monitoring of an alternative parameter, which is a measure of the degree of absorbing liquid saturation, is approved by the Administrator, in which case he will define appropriate parameter boundaries and periods of operation during which they are exceeded).

(2) Where a condenser is the final recovery device in a system, and where an organic compound monitoring device is not used, all 3-hour periods of operation during which the average exit (product side) condenser operating temperature was more than 6 °C (11 °F) above the average exit (product side) operating temperature during the most recent performance test.

(3) Where a carbon adsorber is the final recovery device in a system, and where an organic compound monitoring device is not used:

(i) All carbon bed regeneration cycles during which the total mass steam flow was more than 10 percent below the total mass steam flow during the most recent performance test, or

(ii) All carbon bed regeneration cycles during which the temperature of the carbon bed after regeneration (and after completion of any cooling cycle(s)) was more than 10 percent greater than the carbon bed temperature (in degrees Celsius) during the most recent performance test.

(4) Where an absorber, condenser, or carbon adsorber is the final recovery device in the recovery system and 40 CFR Ch. I (7-1-23 Edition)

where an organic compound monitoring device is used, all 3-hour periods of operation during which the average organic compound concentration level or reading of organic compounds in the exhaust gases is more than 20 percent greater than the exhaust gas organic compound concentration level or reading measured by the monitoring device during the most recent performance test.

(h) Each owner or operator of an affected facility subject to the provisions of this subpart and seeking to demonstrate compliance with §60.662(c) shall keep up-to-date, readily accessible records of:

(1) Any changes in production capacity, feedstock type, or catalyst type, or of any replacement, removal or addition of recovery equipment or a distillation unit;

(2) Any recalculation of the TRE index value performed pursuant to §60.664(g); and

(3) The results of any performance test performed pursuant to the methods and procedures required by §60.664(e).

(i) Each owner or operator of an affected facility that seeks to comply with the requirements of this subpart by complying with the flow rate cutoff in §60.660(c)(6) shall keep up-to-date, readily accessible records to indicate that the vent stream flow rate is less than 0.008 scm/min (0.3 scf/min) and of any change in equipment or process operation that increases the operating vent stream flow rate, including a measurement of the new vent stream flow rate.

(j) Each owner or operator of an affected facility that seeks to comply with the requirements of this subpart by complying with the design production capacity provision in §60.660(c)(5) shall keep up-to-date, readily accessible records of any change in equipment or process operation that increases the design production capacity of the process unit in which the affected facility is located.

(k) Each owner and operator subject to the provisions of this subpart is exempt from the quarterly reporting requirements contained in §60.7(c) of the General Provisions.

(1) Each owner or operator that seeks to comply with the requirements of this subpart by complying with the requirements of 60.660 (c)(4), (c)(5), or (c)(6) or 60.662 shall submit to the Administrator semiannual reports of the following recorded information. The initial report shall be submitted within 6 months after the initial start-up date.

(1) Exceedances of monitored parameters recorded under §60.665 (c) and (g).

(2) All periods recorded under §60.665(d) when the vent stream is diverted from the control device or has no flow rate.

(3) All periods recorded under §60.665(e) when the boiler or process heater was not operating.

(4) All periods recorded under §60.665(f) in which the pilot flame of the flare was absent.

(5) Any change in equipment or process operation that increases the operating vent stream flow rate above the low flow exemption level in §60.660(c)(6), including a measurement of the new vent stream flow rate, as recorded under §60.665(i). These must be reported as soon as possible after the change and no later than 180 days after the change. These reports may be submitted either in conjunction with semiannual reports or as a single separate report. A performance test must be completed with the same time period to verify the recalculated flow value and to obtain the vent stream characteristics of heating value and E_{TOC} . The performance test is subject to the requirements of §60.8 of the General Provisions. Unless the facility qualifies for an exemption under the low capacity exemption status in §60.660(c)(5), the facility must begin compliance with the requirements set forth in §60.662.

(6) Any change in equipment or process operation, as recorded under paragraph (j) of this section, that increases the design production capacity above the low capacity exemption level in §60.660(c)(5) and the new capacity resulting from the change for the distillation process unit containing the affected facility. These must be reported as soon as possible after the change and no later than 180 days after the change. These reports may be submitted either in conjunction with semiannual reports or as a single separate report. A performance test must be completed within the same time period to obtain the vent stream flow rate, heating value, and E_{TOC} . The performance test is subject to the requirements of §60.8. The facility must begin compliance with the requirements set forth in §60.660(d) or §60.662. If the facility chooses to comply with §60.662, the facility may qualify for an exemption in §60.660(c)(4) or (6).

(7) Any recalculation of the TRE index value, as recorded under §60.665(h).

(m) The requirements of §60.665(1) remain in force until and unless EPA, in delegating enforcement authority to a State under section 111(c) of the Act, approves reporting requirements or an alternative means of compliance surveillance adopted by such State. In that event, affected sources within the State will be relieved of the obligation to comply with §60.665(1), provided that they comply with the requirements established by the State.

(n) Each owner or operator that seeks to demonstrate compliance with §60.660(c)(5) must submit to the Administrator an initial report detailing the design production capacity of the process unit.

(o) Each owner or operator that seeks to demonstrate compliance with §60.660(c)(6) must submit to the Administrator an initial report including a flow rate measurement using the test methods specified in §60.664.

(p) The Administrator will specify appropriate reporting and recordkeeping requirements where the owner or operator of an affected facility complies with the standards specified under §60.662 other than as provided under §60.663(a), (b), (c) and (d).

[55 FR 26922, June 29, 1990; 55 FR 36932, Sept.
7, 1990, as amended at 60 FR 58237, Nov. 27, 1995; 65 FR 61778, Oct. 17, 2000; 65 FR 78279, Dec. 14, 2000; 79 FR 11251, Feb. 27, 2014]

§60.666 Reconstruction.

For purposes of this subpart "fixed capital cost of the new components," as used in §60.15, includes the fixed capital cost of all depreciable components which are or will be replaced pursuant

§60.666

§60.667

to all continuous programs of component replacement which are commenced within any 2-year period following December 30, 1983. For purposes of this paragraph, "commenced" means that an owner or operator has undertaken a continuous program of component replacement or that an owner or operator has entered into a contractual obligation to undertake and complete, within a reasonable time, a continuous program of component replacement.

§60.667 Chemicals affected by subpart NNN.

Chemi⊡al name	CAS □o.□
A⊡etal⊡ehy⊒e	75=07=0
A⊡etal⊡ol	107 89 1
A⊡eti⊡a⊡⊡	64 19 7
A⊡eti⊡anhv⊡ri⊡e	108 24 7
A done	67 64 1
Andone vanohvrin	75 86 5
Andrylene	74 86 2
Anryinarin	79 10 7
Anrybnitrile	107 13 1
Aripinarin	124 04 9
Aliponitrile	111_69_3
Alcohol C11 or lower micture	
Alcohol C12 or higher micture	
	107-05-1
	513 35 0
	010_00_0
	62-53-3
Pon Dong	71 71 72 72
	00-11-2
	9011113
	60001 -01 -0
	00001_01_2
	00_00_0
	100_44_7
Bipnenyi	92_52_4
Bicphenol A	80_05_7
Brometone	76_08_4
1,3 Buta lene	106_99_0
Butallene and Lutene traction	
n Butane	106_97_8
1,4 Butane iol	110=63=4
Butane , mile	
1 Butene	106_98_9
2 Butene	25167 67 3
Butene , mi e .	
n Butyl a etate	123 86 4
Butyl a rylate	141 32 2
n Butyl al ohol	71_36_3
⊡e⊡Butyl al⊡ohol	78 92 2
tert.Butyl al⊡ohol	75_65_0
Butyl en lyl phthalate	85 68 7
Butylene gly ol	107 88 0
tert Butyl hy ropero i e	75 91 2
2 Butyne 1,4 iol	110_65_6
Butyral ⊇ehy ⊇e	123 72 8
Butvri⊟anhv⊡ri⊡e	106_31_0
Caprola tam	105 60 2
CarTon TiTulfiTe	75 15 0
Car on tetra romi e	558-13-4
Car on tetra hlori e	56 23 5
Chloro en ene	108 90 7
2 Chloro 4 Tethylamino T6 Ti opropylamino	100_00_/
tria line	1012-24-0
Chloroform	67 66 3
n Chloronitro En Ene	
	126-00-0
	120_99_8

40 CFR Ch. I (7-1-23 Edition)

	77 92 9
Craterial Tabu Ta	
Grotonai∟eny∟e	4170 30 0
Crotoni 🗆 a 🖬 🗆	3724 65 0
Cumene	98 82 8
	80_15_6
	68512 15 7
	108 93 0
Cy⊡ohe⊑anone	108 94 1
Cy⊡ohe⊡anone o⊡me	100_64_1
Cylohelene	110 83 8
1,3 ℃y □openta ⊡ene	542 92 7
Cy⊡opropane	75 19 4
	123_42_2
	110-57-6
3 4 Di hloro 1 Tutene	64037 54 6
Di⊡hloro ifluoromethane	7507108
Di⊡hloro⊡methyl⊡lane	75 78 5
Dichlorofluoromethane	75□43□4
Di⊡hlorohy⊡rin	96 23 1
Diethylen en en	25340_17_4
Dia hentyl in honyl un re vl obthalate	85_68_7
	26761 40 0
Dii ononyl phthalate	28553 12 0
Dimethylamine	124 40 3
Dimethyl terephthalate	120_61_6
2,4 Dinitrotoluene	
2,4 an 2,6 Initrotoiuene	
Dio Ityl phthalate	
Do e ene	25378 22 7
Do⊡e⊡yl⊡en⊡ene, non linear.	
	27176 87 0
Dole ylen ene ulfoni al, olum alt	25155 30 0
Epilnioronyllin	
Ethanolamine	141-43-5
Ethyl a⊡etate	141 78 6
Ethyl a⊡ylate	140 88 5
Ethyl⊡en⊡ene	100 41 4
Ethyl Chlorice	75_00_3
Ethyl Lyani Le	
	106-03-0
	107 06 2
Ethylene gly⊡ol	107 21 1
Ethylene gly ol mono utyl	111_76_2
Ethylene gly ol monoethyl ether	110 80 5
Ethylene gly ol monoethyl ether a etate	111_15_9
Ethylene gly_ol monomethyl ether	109_86_4
	26266_68_6
2 Ethylhe VI al ohol	104 76 7
2 Ethylhe JI amine	104 75 6
Ethylmethyl en ene	25550 14 5
6 Ethyl 1,2,3,4 tetrahy ro 9,10	
anthra ene ione	15547 17 8
Formai∟eny∟e	50_00_0
n Heptane	142 82 5
Heptene mile	
He⊡a⊡e⊒yl ⊡hlori⊡e.	
He⊡amethylene ⊡amine	124_09_4
He amethylene i amine a i pate	3323 53 3
He_amethylenetetramine	
ne∟ane 21He⊺ene⊺initrile	130420020
3 He ene initrile	1119 85
Hy⊡rogen ⊒yani⊒e	74_90_8
I⊡o⊡utane	75 28 5

154

Chemi⊡al name	CAS □o.□
Doutanol Doutyral chyle Doutyral chyle Doelyl allohol Dopentane Dophthalil all Dopponol Deprene Dopponol Detene Linear allohol Dispertene Dispertene	78 83 1 115 11 7 78 84 2 25339 7 7 26952 21 6 78 78 4 121 91 5 78 79 5 67 63 0 463 51 4
Jum [alt, mi]e] inear alcohol _ ulfate _ colum [alt, mi]e] inear alcohol _ ulfate _ agne Jum a letate alci anhy [r]e elamine ethyloitrile ethyloitrile ethylonic ethyloni	123:01:3 142:72:3 188:31:6 108:78:1 141:79:7 767:56:1 74:89:5 25376:45:8 74:87:3 75:09:2 78:93:3 74:88:4 108:10:1 80:62:6 107:83:55 872:50:4
aphthalene	91 20 3 98 95 3 27215 95 8 143 08 8 25154 52 3 9016 45 9 25377 83 7
Pentaerythritol	115 77 5 109 66 0 4635 87 4 109 67 1 127 18 4 108 95 2 3071 32 7 103 65 1
PhoTgene	75_44_5 85_44_9 74_98_6 123_38_6 79_09_4 71_23_8 115_07_1 78_89_7 57_55_6
Propylene o le Solum lyanile SorItol Styrene Terephthalila 1,1,2,2 Tetra Inloroethane Tetraethyl lea Tetrahylrofuran Tetra Interhyliethyllea	75_56_9 143_33_9 50_70_4 100_42_5 100_21_0 79_34_5 78_00_2 109_99_9
Tetramethyl lea⊡ Toluene Toluene !2,4⊞amine Toluene !2,4⊞am⊡, 2,6⊞ii⊡o_yanate 180/20 mi⊡ture	7507401 10808803 9508007 2647106205 7502502
1 1 1 Tri bloroethane	71-55-6

§60.670

Chemi⊑al name	CAS □o.□
1,1,2⊡ri⊡hloroethane	79_00_5
Tri hloroethylene	79⊡01⊡6
Trichlorofluoromethane	75_69_4
1,1,2 Tri hloro 1,2,2 Irifluoroethane	76 13 1
Triethanolamine	102 71 6
Triethylene gly ol	112 27 6
Vinyl a etate	108_05_4
Vinyl ⊡hlori⊡e	75_01_4
Vinyli⊡ene ⊡hlori⊡e	75_35_4
m Xyene	108_38_3
o Xylene	95 47 6
p Xylene	106 42 3
Xylene Imi e	1330 20 7
m⊡Xyehol	576 26 1

CAS num er refer to the Chemial A tratt Registry num er a tigne to petfic themial, itomer, or miture of themial. Some itomer or miture that are correctly the tantar on the CAS num er alignet to them. The tantar apply to all of the themial little, whether CAS num er has been adigned or not.

[55 FR 26942, June 29, 1990, as amended at 60 FR 58237, 58238, Nov. 27, 1995]

§60.668 Delegation of authority.

(a) In delegating implementation and enforcement authority to a State under §111(c) of the Act, the authorities contained in paragraph (b) of this section shall be retained by the Administrator and not transferred to a State.

(b) Authorities which will not be delegated to States: §60.663(e).

Subpart OOO—Standards of Performance for Nonmetallic Mineral Processing Plants

SOURCE: 74 FR 19309, Apr. 28, 2009, unless otherwise noted.

§60.670 Applicability and designation of affected facility.

(a)(1) Except as provided in paragraphs (a)(2), (b), (c), and (d) of this section, the provisions of this subpart are applicable to the following affected facilities in fixed or portable nonmetallic mineral processing plants: each crusher, grinding mill, screening operation, bucket elevator, belt conveyor, bagging operation, storage bin, enclosed truck or railcar loading station. Also, crushers and grinding mills at hot mix asphalt facilities that reduce the size of nonmetallic minerals embedded in recycled asphalt pavement and subsequent affected facilities up to, but not including, the first storage silo or bin are subject to the provisions of this subpart.

Appendix G

AUTHENTICATED U.S. GOVERNMENT INFORMATION

Subpart J—National Emission Standard for Equipment Leaks (Fugitive Emission Sources) of Benzene

SOURCE: 49 FR 23513, June 6, 1984, unless otherwise noted.

§61.110 Applicability and designation of sources.

(a) The provisions of this subpart apply to each of the following sources that are intended to operate in benzene service: pumps, compressors, pressure relief devices, sampling connection systems, open-ended valves or lines, valves, connectors, surge control vessels, bottoms receivers, and control devices or systems required by this subpart.

(b) The provisions of this subpart do not apply to sources located in coke by-product plants.

(c)(1) If an owner or operator applies for one of the exemptions in this paragraph, then the owner or operator shall maintain records as required in $\S61.246(i)$.

(2) Any equipment in benzene service that is located at a plant site designed to produce or use less than 1,000 megagrams (1,102 tons) of benzene per year is exempt from the requirements of §61.112.

(3) Any process unit (defined in §61.241) that has no equipment in benzene service is exempt from the requirements of §61.112.

(d) While the provisions of this subpart are effective, a source to which this subpart applies that is also subject to the provisions of 40 CFR part 60 only will be required to comply with the provisions of this subpart.

[49 FR 23513, June 6, 1984, as amended at 65 FR 62156, Oct. 17, 2000; 65 FR 78280, Dec. 14, 2000]

§61.111 Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Act, in subpart A of part 61, or in subpart V of part 61, and the following terms shall have the specific meanings given them:

In benzene service means that a piece of equipment either contains or contacts a fluid (Liquid or gas) that is at least 10 percent benzene by weight as determined according to the provisions of §61.245(d). The provisions of §61.245(d) also specify how to determine that a piece of equipment is not in benzene service.

Semiannual means a 6-month period; the first semiannual period concludes on the last day of the last month during the 180 days following initial startup for new sources; and the first semiannual period concludes on the last day of the last full month during the 180 days after June 6, 1984 for existing sources.

§61.112 Standards.

(a) Each owner or operator subject to the provisions of this subpart shall comply with the requirements of subpart V of this part.

(b) An owner or operator may elect to comply with the requirements of $\S 61.243-1$ and 61.243-2.

(c) An owner or operator may apply to the Administrator for a determination of an alternative means of emission limitation that achieves a reduction in emissions of benzene at least equivalent to the reduction in emissions of benzene achieved by the controls required in this subpart. In doing so, the owner or operator shall comply with requirements of §61.244.

Subpart K—National Emission Standards for Radionuclide Emissions From Elemental Phosphorus Plants

SOURCE: 54 FR 51699, Dec. 15, 1989, unless otherwise noted.

§61.120 Applicability.

The provisions of this subpart are applicable to owners or operators of calciners and nodulizing kilns at elemental phosphorus plants.

§61.121 Definitions.

(a) Elemental phosphorus plant or plant means any facility that processes phosphate rock to produce elemental phosphorus. A plant includes all buildings, structures, operations, calciners and nodulizing kilns on one contiguous site. Appendix H

AUTHENTICATED U.S. GOVERNMENT INFORMATION

Subpart U [Reserved]

Subpart V—National Emission Standard for Equipment Leaks (Fugitive Emission Sources)

SOURCE: 49 FR 23513, June 6, 1984, unless otherwise noted.

§61.240 Applicability and designation of sources.

(a) The provisions of this subpart apply to each of the following sources that are intended to operate in volatile hazardous air pollutant (VHAP) service: pumps, compressors, pressure relief devices, sampling connection systems, open-ended valves or lines, valves, connectors, surge control vessels, bottoms receivers, and control devices or systems required by this subpart.

(b) The provisions of this subpart apply to the sources listed in paragraph (a) after the date of promulgation of a specific subpart in part 61.

(c) While the provisions of this subpart are effective, a source to which this subpart applies that is also subject to the provisions of 40 CFR part 60 only will be required to comply with the provisions of this subpart.

(d) Alternative means of compliance— (1) Option to comply with part 65. Owners or operators may choose to comply with 40 CFR part 65 to satisfy the requirements of §§ 61.242–1 through 61.247 for equipment that is subject to this subpart and that is part of the same process unit. When choosing to comply with 40 CFR part 65, the requirements of §§ 61.245(d) and 61.246(i) and (j) still apply. Other provisions applying to owners or operators who choose to comply with 40 CFR part 65 are provided in 40 CFR 65.1.

(2) Part 65, subpart C or F. For owners or operators choosing to comply with 40 CFR part 65, each surge control vessel and bottoms receiver subject to this subpart that meets the conditions specified in table 1 or table 2 of this subpart shall meet the requirements for storage vessels in 40 CFR part 65, subpart C; all other equipment subject to this subpart shall meet the requirements in 40 CFR part 65, subpart F.

(3) Part 61, subpart A. Owners or operators who choose to comply with 40

CFR part 65, subpart C or F, must also comply with §§ 61.01, 61.02, 61.05 through 61.08, 61.10(b) through (d), 61.11, and 61.15 for that equipment. All sections and paragraphs of subpart A of this part that are not mentioned in this paragraph (d)(3) do not apply to owners or operators of equipment subject to this subpart complying with 40 CFR part 65, subpart C or F, except that provisions required to be met prior to implementing 40 CFR part 65 still apply. Owners and operators who choose to comply with 40 CFR part 65, subpart C or F, must comply with 40 CFR part 65, subpart A.

(4) Rules referencing this subpart. Owners or operators referenced to this subpart from subpart F or J of this part may choose to comply with 40 CFR part 65 for all equipment listed in paragraph (a) of this section.

[49 FR 23513, June 6, 1984, as amended at 65 FR 78280, Dec. 14, 2000]

§61.241 Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Act, in subpart A of part 61, or in specific subparts of part 61; and the following terms shall have specific meaning given them:

Bottoms receiver means a tank that collects distillation bottoms before the stream is sent for storage or for further downstream processing.

Closed-vent system means a system that is not open to atmosphere and that is composed of hard-piping, ductwork, connections, and, if necessary, flow-inducing devices that transport gas or vapor from a piece or pieces of equipment to a control device or back to a process.

Connector means flanged, screwed, welded, or other joined fittings used to connect two pipe lines or a pipe line and a piece of equipment. For the purpose of reporting and recordkeeping, connector means flanged fittings that are not covered by insulation or other materials that prevent location of the fittings.

Control device means an enclosed combustion device, vapor recovery system, or flare.

Double block and bleed system means two block valves connected in series with a bleed valve or line that can vent the line between the two block valves.

Duct work means a conveyance system such as those commonly used for heating and ventilation systems. It is often made of sheet metal and often has sections connected by screws or crimping. Hard-piping is not ductwork.

Equipment means each pump, compressor, pressure relief device, sampling connection system, open-ended valve or line, valve, connector, surge control vessel, bottoms receiver in VHAP service, and any control devices or systems required by this subpart.

First attempt at repair means to take rapid action for the purpose of stopping or reducing leakage of organic material to atmosphere using best practices.

In gas/vapor service means that a piece of equipment contains process fluid that is in the gaseous state at operating conditions.

Fuel gas means gases that are combusted to derive useful work or heat.

Fuel gas system means the offsite and onsite piping and flow and pressure control system that gathers gaseous stream(s) generated by onsite operations, may blend them with other sources of gas, and transports the gaseous stream for use as fuel gas in combustion devices or in-process combustion equipment, such as furnaces and gas turbines, either singly or in combination.

Hard-piping means pipe or tubing that is manufactured and properly installed using good engineering judgement and standards such as ASME B31.3, Process Piping (available from the American Society of Mechanical Engineers, PO Box 2900, Fairfield, NJ 07007-2900).

In liquid service means that a piece of equipment is not in gas/vapor service.

In-situ sampling systems means nonextractive samplers or in-line samplers.

In vacuum service means that equipment is operating at an internal pressure which is at least 5 kilopascals (kPa) (0.7 psia) below ambient pressure.

In VHAP service means that a piece of equipment either contains or contacts a fluid (liquid or gas) that is at least 10 percent by weight a volatile hazardous air pollutant (VHAP) as determined according to the provisions of §61.245(d). The provisions of §61.245(d) also specify how to determine that a piece of equipment is not in VHAP service.

In VOC service means, for the purposes of this subpart, that (a) the piece of equipment contains or contacts a process fluid that is at least 10 percent VOC by weight (see 40 CFR 60.2 for the definition of volatile organic compound or VOC and 40 CFR 60.485(d) to determine whether a piece of equipment is not in VOC service) and (b) the piece of equipment is not in heavy liquid service as defined in 40 CFR 60.481.

Maximum true vapor pressure means the equilibrium partial pressure exerted by the total VHAP in the stored or transferred liquid at the temperature equal to the highest calendarmonth average of the liquid storage or transfer temperature for liquids stored or transferred above or below the ambient temperature or at the local maximum monthly average temperature as reported by the National Weather Service for liquids stored or transferred at the ambient temperature, as determined:

(1) In accordance with methods described in American Petroleum Institute Publication 2517, Evaporative Loss From External Floating-Roof Tanks (incorporated by reference as specified in §61.18); or

(2) As obtained from standard reference texts; or

(3) As determined by the American Society for Testing and Materials Method D2879-83, Standard Test Method for Vapor Pressure-Temperature Relationship and Initial Decomposition Temperature of Liquids by Isoteniscope (incorporated by reference as specified in §61.18); or

(4) Any other method approved by the Administrator.

Open-ended valve or line means any valve, except pressure relief valves, having one side of the valve seat in contact with process fluid and one side open to atmosphere, either directly or through open piping.

Pressure release means the emission of materials resulting from the system pressure being greater than the set pressure of the pressure relief device.

Process unit means equipment assembled to produce a VHAP or its derivatives as intermediates or final products, or equipment assembled to use a VHAP in the production of a product. A process unit can operate independently if supplied with sufficient feed or raw materials and sufficient product storage facilities.

Process unit shutdown means a work practice or operational procedure that stops production from a process unit or part of a process unit. An unscheduled work practice or operational procedure that stops production from a process unit or part of a process unit for less than 24 hours is not a process unit shutdown. The use of spare equipment and technically feasible bypassing of equipment without stopping production are not process unit shutdowns.

Repaired means that equipment is adjusted, or otherwise altered, to eliminate a leak.

Sampling connection system means an assembly of equipment within a process unit used during periods of representative operation to take samples of the process fluid. Equipment used to take non-routine grab samples is not considered a sampling connection system.

Semiannual means a 6-month period; the first semiannual period concludes on the last day of the last month during the 180 days following initial startup for new sources; and the first semiannual period concludes on the last day of the last full month during the 180 days after the effective date of a specific subpart that references this subpart for existing sources.

Sensor means a device that measures a physical quantity or the change in a physical quantity, such as temperature, pressure, flow rate, pH, or liquid level.

Stuffing box pressure means the fluid (liquid or gas) pressure inside the casing or housing of a piece of equipment, on the process side of the inboard seal.

Surge control vessel means feed drums, recycle drums, and intermediate vessels. Surge control vessels are used within a process unit when in-process storage, mixing, or management of flow rates of volumes is needed on a recurring or ongoing basis to assist in production of a product. Volatile hazardous air pollutant or VHAP means a substance regulated under this part for which a standard for equipment leaks of the substance has been proposed and promulgated. Benzene is a VHAP. Vinyl chloride is a VHAP.

[49 FR 23513, June 6, 1984; 49 FR 38946, Oct. 2, 1984, as amended at 51 FR 34915, Sept. 30, 1986; 54 FR 38076, Sept. 14, 1989; 65 FR 62158, Oct. 17, 2000; 65 FR 78280, Dec. 14, 2000]

§61.242-1 Standards: General.

(a) Each owner or operator subject to the provisions of this subpart shall demonstrate compliance with the requirements of §§61.242–11 to 61.242–11 for each new and existing source as required in 40 CFR 61.05, except as provided in §§61.243 and 61.244.

(b) Compliance with this subpart will be determined by review of records, review of performance test results, and inspection using the methods and procedures specified in §61.245.

(c)(1) An owner or operator may request a determination of alternative means of emission limitation to the requirements of \$ 61.242–2, 61.242–3, 61.242–5, 61.242–6, 61.242–7, 61.242–8, 61.242–9 and 61.242–11 as provided in \$ 61.244.

(2) If the Administrator makes a determination that a means of emission limitation is at least a permissible alternative to the requirements of §61.242-2, 61.242-3, 61.242-5, 61.242-6, 61.242-7, 61.242-8, 61.242-9 or 61.242-11, an owner or operator shall comply with the requirements of that determination.

(d) Each piece of equipment to which this subpart applies shall be marked in such a manner that it can be distinguished readily from other pieces of equipment.

(e) Equipment that is in vacuum service is excluded from the requirements of 61.242-2, to 61.242-11 if it is identified as required in 61.246(e)(5).

[49 FR 23513, June 6, 1984; 49 FR 38946, Oct. 2, 1984]

§61.242–2 Standards: Pumps.

(a)(1) Each pump shall be monitored monthly to detect leaks by the methods specified in 61.245(b), except as provided in 61.242-1(c) and paragraphs (d), (e), (f) and (g) of this section.

§61.242-2

40 CFR Ch. I (7–1–23 Edition)

(2) Each pump shall be checked by visual inspection each calendar week for indications of liquids dripping from the pump seal.

(b)(1) If an instrument reading of 10,000 ppm or greater is measured, a leak is detected.

(2) If there are indications of liquids dripping from the pump seal, a leak is detected.

(c)(1) When a leak is detected, it shall be repaired as soon as practicable, but not later than 15 calendar days after it is detected, except as provided in §61.242-10.

(2) A first attempt at repair shall be made no later than 5 calendar days after each leak is detected.

(d) Each pump equipped with a dual mechanical seal system that includes a barrier fluid system is exempt from the requirements of paragraphs (a) and (b) of this section, provided the following requirements are met:

(1) Each dual mechanical seal system is:

(i) Operated with the barrier fluid at a pressure that is at all times greater than the pump stuffing box pressure; or

(ii) Equipped with a barrier fluid degassing reservoir that is routed to a process or fuel gas system or connected by a closed-vent system to a control device that complies with the requirements of §61.242-11; or

(iii) Equipped with a system that purges the barrier fluid into a process stream with zero VHAP emissions to atmosphere.

(2) The barrier fluid is not in VHAP service and, if the pump is covered by standards under 40 CFR part 60, is not in VOC service.

(3) Each barrier fluid system is equipped with a sensor that will detect failure of the seal system, the barrier fluid system, or both.

(4) Each pump is checked by visual inspection each calendar week for indications of liquids dripping from the pump seal.

(i) If there are indications of liquid dripping from the pump seal at the time of the weekly inspection, the pump shall be monitored as specified in §61.245 to determine the presence of VOC and VHAP in the barrier fluid.

(ii) If the monitor reading (taking into account any background readings)

indicates the presence of VHAP, a leak is detected. For the purpose of this paragraph, the monitor may be calibrated with VHAP, or may employ a gas chromatography column to limit the response of the monitor to VHAP, at the option of the owner or operator.

(iii) If an instrument reading of 10,000 ppm or greater (total VOC) is measured, a leak is detected.

(5) Each sensor as described in paragraph (d)(3) of this section is checked daily or is equipped with an audible alarm.

(6)(i) The owner or operator determines, based on design considerations and operating experience, criteria applicable to the presence and frequency of drips and to the sensor that indicates failure of the seal system, the barrier fluid system, or both.

(ii) If indications of liquids dripping from the pump seal exceed the criteria established in paragraph (d)(6)(i) of this section, or if, based on the criteria established in paragraph (d)(6)(i) of this section, the sensor indicates failure of the seal system, the barrier fluid system, or both, a leak is detected.

(iii) When a leak is detected, it shall be repaired as soon as practicable, but no later than 15 calendar days after it is detected, except as provided in §61.242-10.

(iv) A first attempt at repair shall be made no later than five calendar days after each leak is detected.

(e) Any pump that is designated, as described in $\S61.246(e)(2)$, for no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, is exempt from the requirements of paragraphs (a), (c), and (d) if the pump:

(1) Has no externally actuated shaft penetrating the pump housing,

(2) Is demonstrated to be operating with no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, as measured by the method specified in §61.245(c), and

(3) Is tested for compliance with paragraph (e)(2) initially upon designation, annually, and at other times requested by the Administrator.

(f) If any pump is equipped with a closed-vent system capable of capturing and transporting any leakage

from the seal or seals to a process or fuel gas system or to a control device that complies with the requirements of §61.242-11, it is exempt from the requirements of paragraphs (a) through (e) of this section.

(g) Any pump that is designated, as described in 61.246(f)(1), as an unsafeto-monitor pump is exempt from the monitoring and inspection requirements of paragraphs (a) and (d)(4) through (6) of this section if:

(1) The owner or operator of the pump demonstrates that the pump is unsafe-to-monitor because monitoring personnel would be exposed to an immediate danger as a consequence of complying with paragraph (a) of this section; and

(2) The owner or operator of the pump has a written plan that requires monitoring of the pump as frequently as practicable during safe-to-monitor times but not more frequently than the periodic monitoring schedule otherwise applicable, and repair of the equipment according to the procedures in paragraph (c) of this section if a leak is detected.

(h) Any pump that is located within the boundary of an unmanned plant site is exempt from the weekly visual inspection requirement of paragraphs (a)(2) and (d)(4) of this section, and the daily requirements of paragraph (d)(5)of this section, provided that each pump is visually inspected as often as practicable and at least monthly.

[49 FR 23513, June 6, 1984, as amended at 49 FR 38946, Oct. 2, 1984; 55 FR 28349, July 10, 1990; 65 FR 78281, Dec. 14, 2000]

§61.242-3 Standards: Compressors.

(a) Each compressor shall be equipped with a seal system that includes a barrier fluid system and that prevents leakage of process fluid to atmosphere, except as provided in $\S61.242-1(c)$ and paragraphs (h) and (i) of this section.

(b) Each compressor seal system as required in paragraph (a) shall be:

(1) Operated with the barrier fluid at a pressure that is greater than the compressor stuffing box pressure; or

(2) Equipped with a barrier fluid system degassing reservoir that is routed to a process or fuel gas system or connected by a closed-vent system to a control device that complies with the requirements of §61.242–11; or

(3) Equipped with a system that purges the barrier fluid into a process stream with zero VHAP emissions to atmosphere.

(c) The barrier fluid shall not be in VHAP service and, if the compressor is covered by standards under 40 CFR part 60, shall not be in VOC service.

(d) Each barrier fluid system as described in paragraphs (a)–(c) of this section shall be equipped with a sensor that will detect failure of the seal system, barrier fluid system, or both.

(e)(1) Each sensor as required in paragraph (d) of this section shall be checked daily or shall be equipped with an audible alarm unless the compressor is located within the boundary of an unmanned plant site.

(2) The owner or operator shall determine, based on design considerations and operating experience, a criterion that indicates failure of the seal system, the barrier fluid system, or both.

(f) If the sensor indicates failure of the seal system, the barrier fluid system, or both based on the criterion determined under paragraph (e)(2) of this section, a leak is detected.

(g)(1) When a leak is detected, it shall be repaired as soon as practicable, but not later than 15 calendar days after it is detected, except as provided in §61.242-10.

(2) A first attempt at repair shall be made no later than 5 calendar days after eack leak is detected.

(h) A compressor is exempt from the requirements of paragraphs (a) and (b) of this section if it is equipped with a closed-vent system to capture and transport leakage from the compressor drive shaft back to a process or fuel gas system or to a control device that complies with the requirements of §61.242–11, except as provided in paragraph (i) of this section.

(i) Any Compressor that is designated, as described in 61.246(e)(2), for no detectable emission as indicated by an instrument reading of less than 500 ppm above background is exempt from the requirements of paragraphs (a)-(h) if the compressor:

(1) Is demonstrated to be operating with no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, as measured by the method specified in §61.245(c); and

(2) Is tested for compliance with paragraph (i)(1) initially upon designation, annually, and at other times requested by the Administrator.

[49 FR 23513, June 6, 1984; 49 FR 38946, Oct. 2, 1984, as amended at 65 FR 78281, Dec. 14, 2000]

§61.242–4 Standards: Pressure relief devices in gas/vapor service.

(a) Except during pressure releases, each pressure relief device in gas/vapor service shall be operated with no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, as measured by the method specified in §61.245(c).

(b)(1) After each pressure release, the pressure relief device shall be returned to a condition of no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, as soon as practicable, but no later than 5 calendar days after each pressure release, except as provided in §61.242–10.

(2) No later than 5 calendar days after the pressure release, the pressure relief device shall be monitored to confirm the condition of no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, as measured by the method specified in §61.245(c).

(c) Any pressure relief device that is routed to a process or fuel gas system or equipped with a closed-vent system capable of capturing and transporting leakage from the pressure relief device to a control device as described in $\S61.242-11$ is exempt from the requirements of paragraphs (a) and (b) of this section.

(d)(1) Any pressure relief device that is equipped with a rupture disk upstream of the pressure relief device is exempt from the requirements of paragraphs (a) and (b) of this section, provided the owner or operator complies with the requirements in paragraph (d)(2) of this section.

(2) After each pressure release, a new rupture disk shall be installed upstream of the pressure relief device as soon as practicable, but no later than 5 40 CFR Ch. I (7–1–23 Edition)

calendar days after each pressure release, except as provided in §61.242–10.

[49 FR 23513, June 6, 1984; 49 FR 38946, Oct. 2, 1984, as amended at 65 FR 78281, Dec. 14, 2000]

§61.242-5 Standards: Sampling connecting systems.

(a) Each sampling connection system shall be equipped with a closed-purge, closed-loop, or closed vent system, except as provided in §61.242–1(c). Gases displaced during filling of the sample container are not required to be collected or captured.

(b) Each closed-purge, closed-loop, or closed vent system as required in paragraph (a) of this section shall comply with the requirements specified in paragraphs (b)(1) through (4) of this section:

(1) Return the purged process fluid directly to the process line; or

(2) Collect and recycle the purged process fluid; or

(3) Be designed and operated to capture and transport all the purged process fluid to a control device that complies with the requirements of §61.242– 11; or

(4) Collect, store, and transport the purged process fluid to any of the following systems or facilities:

(i) A waste management unit as defined in 40 CFR 63.111 if the waste management unit is subject to and operated in compliance with the provisions of 40 CFR part 63, subpart G, applicable to Group 1 wastewater streams; or

(ii) A treatment, storage, or disposal facility subject to regulation under 40 CFR part 262, 264, 265, or 266; or

(iii) A facility permitted, licensed, or registered by a State to manage municipal or industrial solid waste, if the process fluids are not hazardous waste as defined in 40 CFR part 261.

(c) In-situ sampling systems and sampling systems without purges are exempt from the requirements of paragraphs (a) and (b) of this section.

[65 FR 78281, Dec. 14, 2000]

§61.242–6 Standards: Open-ended valves or lines.

(a)(1) Each open-ended value or line shall be equipped with a cap, blind flange, plug, or a second value, except as provided in 61.242-1(c).

(2) The cap, blind flange, plug, or second valve shall seal the open end at all times except during operations requiring process fluid flow through the open-ended valve or line.

(b) Each open-ended valve or line equipped with a second valve shall be operated in a manner such that the valve on the process fluid end is closed before the second valve is closed.

(c) When a double block and bleed system is being used, the bleed valve or line may remain open during operations that require venting the line between the block valves but shall comply with paragraph (a) at all other times.

(d) Open-ended valves or lines in an emergency shutdown system which are designed to open automatically in the event of a process upset are exempt from the requirements of paragraphs (a), (b) and (c) of this section.

(e) Open-ended valves or lines containing materials which would autocatalytically polymerize or would present an explosion, serious overpressure, or other safety hazard if capped or equipped with a double block and bleed system as specified in paragraphs (a) through (c) of this section are exempt from the requirements of paragraphs (a) through (c) of this section.

[49 FR 23513, June 6, 1984, as amended at 65 FR 78282, Dec. 14, 2000]

§61.242–7 Standards: Valves.

(a) Each valve shall be monitored monthly to detect leaks by the method specified in 61.245(b) and shall comply with paragraphs (b)–(e), except as provided in paragraphs (f), (g), and (h) of this section, 61.243-1 or 61.243-2, and 61.242-1(c).

(b) If an instrument reading of 10,000 ppm or greater is measured, a leak is detected.

(c)(1) Any valve for which a leak is not detected for 2 successive months may be monitored the first month of every quarter, beginning with the next quarter, until a leak is detected.

(2) If a leak is detected, the valve shall be monitored monthly until a leak is not detected for 2 successive months.

(d)(1) When a leak is detected, it shall be repaired as soon as practicable,

but no later than 15 calendar days after the leak is detected, except as provided in §61.242–10.

(2) A first attempt at repair shall be made no later than 5 calendar days after each leak is detected.

(e) First attempts at repair include, but are not limited to, the following best practices where practicable:

(1) Tightening of bonnet bolts;

(2) Replacement of bonnet bolts;

(3) Tightening of packing gland nuts; and

(4) Injection of lubricant into lubricated packing.

(f) Any valve that is designated, as described in $\S61.246(e)(2)$, for no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, is exempt from the requirements of paragraph (a) if the valve:

(1) Has no external actuating mechanism in contact with the process fluid;

(2) Is operated with emissions less than 500 ppm above background, as measured by the method specified in §61.245(c); and

(3) Is tested for compliance with paragraph (f)(2) initially upon designation, annually, and at other times requested by the Administrator.

(g) Any valve that is designated, as described in 61.246(f)(1), as an unsafeto-monitor valve is exempt from the requirements of paragraph (a) if:

(1) The owner or operator of the valve demonstrates that the valve is unsafe to monitor because monitoring personnel would be exposed to an immediate danger as a consequence of complying with paragraph (a); and

(2) The owner or operator of the valve has a written plan that requires monitoring of the valve as frequent as practicable during safe-to-monitor times.

(h) Any valve that is designated, as described in 61.246(f)(2), as a difficult-to-monitor valve is exempt from the requirements of paragraph (a) if:

(1) The owner or operator of the valve demonstrates that the valve cannot be monitored without elevating the monitoring personnel more than 2 meters above a support surface;

(2) The process unit within which the valve is located is an existing process unit; and

§61.242-8

(3) The owner or operator of the valve follows a written plan that requires monitoring of the valve at least once per calendar year.

§61.242-8 Standards: Pressure relief services in liquid service and connectors.

(a) If evidence of a potential leak is found by visual, audible, olfactory, or any other detection method at pressure relief devices in liquid service and connectors, the owner or operator shall follow either one of the following procedures, except as provided in 61.242-1(c):

(1) The owner or operator shall monitor the equipment within 5 days by the method specified in §61.245(b) and shall comply with the requirements of paragraphs (b) through (d) of this section.

(2) The owner or operator shall eliminate the visual, audible, olfactory, or other indication of a potential leak.

(b) If an instrument reading of 10,000 ppm or greater is measured, a leak is detected.

(c)(1) When a leak is detected, it shall be repaired as soon as practicable, but not later than 15 calendar days after it is detected, except as provided in §61.242-10.

(2) The first attempt at repair shall be made no later than 5 calendar days after each leak is detected.

(d) First attempts at repair include, but are not limited to, the best practices described under §61.242-7(e).

[49 FR 23513, June 6, 1984; 49 FR 38946, Oct. 2, 1984, as amended at 65 FR 78282, Dec. 14, 2000]

§61.242-9 Standards: Surge control vessels and bottoms receivers.

Each surge control vessel or bottoms receiver that is not routed back to the process and that meets the conditions specified in table 1 or table 2 of this subpart shall be equipped with a closed-vent system capable of capturing and transporting any leakage from the vessel back to the process or to a control device as described in \S 61.242–11, except as provided in \S 61.242– 1(c); or comply with the requirements of 40 CFR 63.119(b) or (c).

[65 FR 78282, Dec. 14, 2000]

40 CFR Ch. I (7–1–23 Edition)

§61.242-10 Standards: Delay of repair.

(a) Delay of repair of equipment for which leaks have been detected will be allowed if repair within 15 days is technically infeasible without a process unit shutdown. Repair of this equipment shall occur before the end of the next process unit shutdown.

(b) Delay of repair of equipment for which leaks have been detected will be allowed for equipment that is isolated from the process and that does not remain in VHAP service.

(c) Delay of repair for valves will be allowed if:

(1) The owner or operator demonstrates that emissions of purged material resulting from immediate repair are greater than the fugitive emissions likely to result from delay of repair, and

(2) When repair procedures are effected, the purged material is collected and destroyed or recovered in a control device complying with §61.242–11.

(d) Delay of repair for pumps will be allowed if:

(1) Repair requires the use of a dual mechanical seal system that includes a barrier fluid system, and

(2) Repair is completed as soon as practicable, but not later than 6 months after the leak was detected.

(e) Delay of repair beyond a process unit shutdown will be allowed for a valve if valve assembly replacement is necessary during the process unit shutdown, valve assembly supplies have been depleted, and valve assembly supplies had been sufficiently stocked before the supplies were depleted. Delay of repair beyond the next process unit shutdown will not be allowed unless the next process unit shutdown occurs sooner than 6 months after the first process unit shutdown.

[49 FR 23513, June 6, 1984, as amended at 65 FR 78282, Dec. 14, 2000]

§61.242-11 Standards: Closed-vent systems and control devices.

(a) Owners or operators of closedvent systems and control devices used to comply with provisions of this subpart shall comply with the provisions of this section, except as provided in §61.242-1(c).

(b) Vapor recovery systems (for example, condensers and absorbers) shall

be designed and operated to recover the organic vapors vented to them with an efficiency of 95 percent or greater, or to an exit concentration of 20 parts per million by volume, whichever is less stringent.

(c) Enclosed combustion devices shall be designed and operated to reduce the VHAP emissions vented to them with an efficiency of 95 percent or greater, or to an exit concentration of 20 parts per million by volume, on a dry basis, corrected to 3 percent oxygen, whichever is less stringent, or to provide a minimum residence time of 0.50 seconds at a minimum temperature of 760 °C.

(d) Flares shall used to comply with this subpart shall comply with the requirements of 60.18.

(e) Owners or operators of control devices that are used to comply with the provisions of this subpart shall monitor these control devices to ensure that they are operated and maintained in conformance with their design.

(f) Except as provided in paragraphs (i) through (k) of this section, each closed vent system shall be inspected according to the procedures and schedule specified in paragraph (f)(1) or (2) of this section, as applicable.

(1) If the vapor collection system or closed vent system is constructed of hard-piping, the owner or operator shall comply with the following requirements:

(i) Conduct an initial inspection according to the procedures in 61.245(b); and

(ii) Conduct annual visual inspections for visible, audible, or olfactory indications of leaks.

(2) If the vapor collection system or closed vent system is constructed of ductwork, the owner or operator shall:

(i) Conduct an initial inspection according to the procedures in 61.245(b); and

(ii) Conduct annual inspections according to the procedures in §61.245(b).

(g) Leaks, as indicated by an instrument reading greater than 500 parts per million by volume above background or by visual inspections, shall be repaired as soon as practicable except as provided in paragraph (h) of this section. (1) A first attempt at repair shall be made no later than 5 calendar days after the leak is detected.

(2) Repair shall be completed no later than 15 calendar days after the leak is detected.

(h) Delay of repair of a closed vent system for which leaks have been detected is allowed if the repair is technically infeasible without a process unit shutdown, or if the owner or operator determines that emissions resulting from immediate repair would be greater than the fugitive emissions likely to result from delay of repair. Repair of such equipment shall be complete by the end of the next process unit shutdown.

(i) If a vapor collection system or closed vent system is operated under a vacuum, it is exempt from the inspection requirements of paragraphs (f)(1)(i) and (2) of this section.

(j) Any parts of the closed vent system that are designated, as described in paragraph (1)(1) of this section, as unsafe-to-inspect are exempt from the inspection requirements of paragraphs (f)(1)(i) and (2) of this section if they comply with the following requirements:

(1) The owner or operator determines that the equipment is unsafe-to-inspect because inspecting personnel would be exposed to an imminent or potential danger as a consequence of complying with paragraph (f)(1)(i) or (2) of this section; and

(2) The owner or operator has a written plan that requires inspection of the equipment as frequently as practicable during safe-to-inspect times.

(k) Any parts of the closed vent system that are designated, as described in paragraph (1)(2) of this section, as difficult-to-inspect are exempt from the inspection requirements of paragraphs (f)(1)(i) and (2) of this section if they comply with the following requirements:

(1) The owner or operator determines that the equipment cannot be inspected without elevating the inspecting personnel more than 2 meters above a support surface; and

(2) The owner or operator has a written plan that requires inspection of the equipment at least once every 5 years. A closed vent system is exempt from

40 CFR Ch. I (7–1–23 Edition)

inspection if it is operated under a vacuum.

(1) The owner or operator shall record the following information:

(1) Identification of all parts of the closed vent system that are designated as unsafe-to-inspect, an explanation of why the equipment is unsafe-to-inspect, and the plan for inspecting the equipment.

(2) Identification of all parts of the closed vent system that are designated as difficult-to-inspect, an explanation of why the equipment is difficult-to-inspect, and the plan for inspecting the equipment.

(3) For each inspection during which a leak is detected, a record of the information specified in §61.246(c).

(4) For each inspection conducted in accordance with §61.245(b) during which no leaks are detected, a record that the inspection was performed, the date of the inspection, and a statement that no leaks were detected.

(5) For each visual inspection conducted in accordance with paragraph (f)(1)(ii) of this section during which no leaks are detected, a record that the inspection was performed, the date of the inspection, and a statement that no leaks were detected.

(m) Closed vent systems and control devices used to comply with provisions of this subpart shall be operated at all times when emissions may be vented to them.

[49 FR 23513, June 6, 1984; 49 FR 38946, Oct. 2, 1984, as amended at 51 FR 2702, Jan. 21, 1986;
65 FR 62158, Oct. 17, 2000; 65 FR 78282, Dec. 14, 2000]

§61.243–1 Alternative standards for valves in VHAP service—allowable percentage of valves leaking.

(a) An owner or operator may elect to have all valves within a process unit to comply with an allowable percentage of valves leaking of equal to or less than 2.0 percent.

(b) The following requirements shall be met if an owner or operator decides to comply with an allowable percentage of valves leaking:

(1) An owner or operator must notify the Administrator that the owner or operator has elected to have all valves within a process unit to comply with the allowable percentage of valves leaking before implementing this alternative standard, as specified in §61.247(d).

(2) A performance test as specified in paragraph (c) of this section shall be conducted initially upon designation, annually, and at other times requested by the Administrator.

(3) If a valve leak is detected, it shall be repaired in accordance with §61.242–7(d) and (e).

(c) Performance tests shall be conducted in the following manner:

(1) All valves in VHAP service within the process unit shall be monitored within 1 week by the methods specified in §61.245(b).

(2) If an instrument reading of 10,000 ppm or greater is measured, a leak is detected.

(3) The leak percentage shall be determined by dividing the number of valves in VHAP service for which leaks are detected by the number of valves in VHAP service within the process unit.

(d) Owner or operators who elect to have all valves comply with this alternative standard shall not have a process unit with a leak percentage greater than 2.0 percent.

(e) If an owner or operator decides no longer to comply with §61.243–1, the owner or operator must notify the Administrator in writing that the work practice standard described in §61.242– 7(a)-(e) will be followed.

§61.243-2 Alternative standards for valves in VHAP service—skip period leak detection and repair.

(a)(1) An owner or operator may elect for all valves within a process unit to comply with one of the alternative work practices specified in paragraphs (b)(2) and (3) of this section.

(2) An owner or operator must notify the Administrator before implementing one of the alternative work practices, as specified in §61.247(d).

(b)(1) An owner or operator shall comply initially with the requirements for valves, as described in §61.242-7.

(2) After 2 consecutive quarterly leak detection periods with the percentage of valves leaking equal to or less than 2.0, an owner or operator may begin to skip one of the quarterly leak detection periods for the valves in VHAP service.

(3) After five consecutive quarterly leak detection periods with the percentage of valves leaking equal to or less than 2.0, an owner or operator may begin to skip three of the quarterly leak detection periods for the valves in VHAP service.

(4) If the percentage of valves leaking is greater than 2.0, the owner or operator shall comply with the requirements as described in §61.242–7 but may again elect to use this section.

[49 FR 23513, June 6, 1984, as amended at 65 FR 62158, Oct. 17, 2000]

§61.244 Alternative means of emission limitation.

(a) Permission to use an alternative means of emission limitation under section 112(e)(3) of the Clean Air Act shall be governed by the following procedures:

(b) Where the standard is an equipment, design, or operational requirement:

(1) Each owner or operator applying for permission shall be responsible for collecting and verifying test data for an alternative means of emission limitation to test data for the equipment, design, and operational requirements.

(2) The Administrator may condition the permission on requirements that may be necessary to assure operation and maintenance to achieve the same emission reduction as the equipment, design, and operational requirements.

(c) Where the standard is a work practice:

(1) Each owner or operator applying for permission shall be responsible for collecting and verifying test data for an alternative means of emission limitation.

(2) For each source for which permission is requested, the emission reduction achieved by the required work practices shall be demonstrated for a minimum period of 12 months.

(3) For each source for which permission is requested, the emission reduction achieved by the alternative means of emission limitation shall be demonstrated.

(4) Each owner or operator applying for permission shall commit in writing each source to work practices that provide for emission reductions equal to or greater than the emission reductions achieved by the required work practices.

(5) The Administrator will compare the demonstrated emission reduction for the alternative means of emission limitation to the demonstrated emission reduction for the required work practices and will consider the commitment in paragraph (c)(4).

(6) The Administrator may condition the permission on requirements that may be necessary to assure operation and maintenance to achieve the same emission reduction as the required work practices of this subpart.

(d) An owner or operator may offer a unique approach to demonstrate the alternative means of emission limitation.

(e)(1) Manufacturers of equipment used to control equipment leaks of a VHAP may apply to the Administrator for permission for an alternative means of emission limitation that achieves a reduction in emissions of the VHAP achieved by the equipment, design, and operational requirements of this subpart.

(2) The Administrator will grant permission according to the provisions of paragraphs (b), (c), and (d).

[49 FR 23513, June 6, 1984, as amended at 65 FR 62158, Oct. 17, 2000]

§ 61.245 Test methods and procedures.

(a) Each owner or operator subject to the provisions of this subpart shall comply with the test methods and procedures requirements provided in this section.

(b) Monitoring, as required in §§ 61.242, 61.243, 61.244, and 61.135, shall comply with the following requirements:

(1) Monitoring shall comply with Method 21 of appendix A of 40 CFR part 60.

(2) The detection instrument shall meet the performance criteria of Method 21.

(3) The instrument shall be calibrated before use on each day of its use by the procedures specified in Method 21.

(4) Calibration gases shall be:

(i) Zero air (less than 10 ppm of hydrocarbon in air); and

(ii) A mixture of methane or nhexane and air at a concentration of approximately, but less than, 10,000 ppm methane or n-hexane.

(5) The instrument probe shall be traversed around all potential leak interfaces as close to the interface as possible as described in Method 21.

(c) When equipment is tested for compliance with or monitored for no detectable emissions, the owner or operator shall comply with the following requirements:

(1) The requirements of paragraphs(b) (1) through (4) shall apply.

(2) The background level shall be determined, as set forth in Method 21.

(3) The instrument probe shall be traversed around all potential leak interfaces as close to the interface as possible as described in Method 21.

(4) The arithmetic difference between the maximum concentration indicated by the instrument and the background level is compared with 500 ppm for determining compliance.

(d)(1) Each piece of equipment within a process unit that can conceivably contain equipment in VHAP service is presumed to be in VHAP service unless an owner or operator demonstrates that the piece of equipment is not in VHAP service. For a piece of equipment to be considered not in VHAP service, it must be determined that the percent VHAP content can be reasonably expected never to exceed 10 percent by weight. For purposes of determining the percent VHAP content of the process fluid that is contained in or contacts equipment, procedures that conform to the methods described in ASTM Method D-2267 (incorporated by the reference as specified in §61.18) shall be used.

(2)(i) An owner or operator may use engineering judgment rather than the procedures in paragraph (d)(1) of this section to demonstrate that the percent VHAP content does not exceed 10 percent by weight, provided that the engineering judgment demonstrates that the VHAP content clearly does not exceed 10 percent by weight. When an owner or operator and the Administrator do not agree on whether a piece of equipment is not in VHAP service, however, the procedures in paragraph (d)(1) of this section shall be used to resolve the disagreement.

40 CFR Ch. I (7–1–23 Edition)

(ii) If an owner or operator determines that a piece of equipment is in VHAP service, the determination can be revised only after following the procedures in paragraph (d)(1) of this section.

(3) Samples used in determining the percent VHAP content shall be representative of the process fluid that is contained in or contacts the equipment or the gas being combusted in the flare.

(e)(1) Method 22 of appendix A of 40 CFR part 60 shall be used to determine compliance of flares with the visible emission provisions of this subpart.

(2) The presence of a flare pilot flame shall be monitored using a thermocouple or any other equivalent device to detect the presence of a flame.

(3) The net heating value of the gas being combusted in a flare shall be calculated using the following equation:

$$\mathbf{H}_{\mathrm{T}} = \mathbf{K} \left(\sum_{i=1}^{n} \mathbf{C}_{i} \mathbf{H}_{i} \right)$$

Where:

- H_T = Net heating value of the sample, MJ/ scm (BTU/scf); where the net enthalpy per mole of offgas is based on combustion at 25 °C and 760 mm Hg (77 °F and 14.7 psi), but the standard temperature for determining the volume corresponding to one mole is 20 °C (68 °F).
- $\begin{array}{l} \mathrm{K} = \mathrm{conversion\ constant,\ 1.740 \times 10^{\,7}\ (g-mole)} \\ \mathrm{(MJ)/(ppm-scm-kcal)\ (metric\ units);\ or} \\ \mathrm{4.674 \times 10^{8}\ ((g-mole)\ (Btu)/(ppm-scf-kcal))} \\ \mathrm{(English\ units)} \end{array}$
- Ci = Concentration of sample component "i" in ppm, as measured by Method 18 of appendix A to 40 CFR part 60 and ASTM D2504-67, 77, or 88 (Reapproved 1993) (incorporated by reference as specified in §61.18).

(4) The actual exit velocity of a flare shall be determined by dividing the volumetric flowrate (in units of standard temperature and pressure), as determined by Method 2, 2A, 2C, or 2D, as appropriate, by the unobstructed (free) cross section area of the flare tip.

(5) The maximum permitted velocity, V_{max} , for air-assisted flares shall be determined by the following equation:

$$V_{max} = K_1 + K_2 H_T$$

Where:

 V_{max} = Maximum permitted velocity, m/sec (ft/sec).

 H_T = Net heating value of the gas being combusted, as determined in paragraph (e)(3) of this section, MJ/scm (Btu/scf).

 $K_1 = 8.706 \text{ m/sec} (\text{metric units})$

= 28.56 ft/sec (English units)

 $K_2 = 0.7084 \text{ m}^4/(\text{MJ-sec}) \text{ (metric units)}$ = 0.087 ft⁴/(Btu-sec) (English units)

[49 FR 23513, June 6, 1984, as amended at 49 FR 38946, Oct. 2, 1984; 49 FR 43647, Oct. 31, 1984; 53 FR 36972, Sept. 23, 1988; 54 FR 38077, Sept. 14, 1989; 65 FR 62158, Oct. 17, 2000]

§61.246 Recordkeeping requirements.

(a)(1) Each owner or operator subject to the provisions of this subpart shall comply with the recordkeeping requirements of this section.

(2) An owner or operator of more than one process unit subject to the provisions of this subpart may comply with the recordkeeping requirements for these process units in one recordkeeping system if the system identifies each record by each process unit.

(b) When each leak is detected as specified in §§61.242-2, 61.242-3, 61.242-7, 61.242-8, and 61.135, the following requirements apply:

(1) A weatherproof and readily visible identification, marked with the equipment identification number, shall be attached to the leaking equipment.

(2) The identification on a valve may be removed after it has been monitored for 2 successive months as specified in §61.242–7(c) and no leak has been detected during those 2 months.

(3) The identification on equipment, except on a valve, may be removed after it has been repaired.

(c) When each leak is detected as specified in §§ 61.242–2, 61.242–3, 61.242–7, 61.242–8, and 61.135, the following information shall be recorded in a log and shall be kept for 2 years in a readily accessible location:

(1) The instrument and operator identification numbers and the equipment identification number.

(2) The date the leak was detected and the dates of each attempt to repair the leak. (3) Repair methods applied in each attempt to repair the leak.

(4) "Above 10,000" if the maximum instrument reading measured by the methods specified in §61.245(a) after each repair attempt is equal to or greater than 10,000 ppm.

(5) "Repair delayed" and the reason for the delay if a leak is not repaired within 15 calendar days after discovery of the leak.

(6) The signature of the owner or operator (or designate) whose decision it was that repair could not be effected without a process shutdown.

(7) The expected date of successful repair of the leak if a leak is not repaired within 15 calendar days.

(8) Dates of process unit shutdowns that occur while the equipment is unrepaired.

(9) The date of successful repair of the leak.

(d) The following information pertaining to the design requirements for closed-vent systems and control devices described in §61.242–11 shall be recorded and kept in a readily accessible location:

(1) Detailed schematics, design specifications, and piping and instrumentation diagrams.

(2) The dates and descriptions of any changes in the design specifications.

(3) A description of the parameter or parameters monitored, as required in $\S61.242-11(e)$, to ensure that control devices are operated and maintained in conformance with their design and an explanation of why that parameter (or parameters) was selected for the monitoring.

(4) Periods when the closed-vent systems and control devices required in §§ 61.242-2, 61.242-3, 61.242-4, 61.242-5 and 61.242-9 are not operated as designed, including periods when a flare pilot light does not have a flame.

(5) Dates of startups and shutdowns of the closed-vent systems and control devices required in §§61.242-2, 61.242-3, 61.242-4, 61.242-5 and 61.242-9.

(e) The following information pertaining to all equipment to which a standard applies shall be recorded in a log that is kept in a readily accessible location:

(1) A list of identification numbers for equipment (except welded fittings) subject to the requirements of this subpart.

(2)(i) A list of identification numbers for equipment that the owner or operator elects to designate for no detectable emissions as indicated by an instrument reading of less than 500 ppm above background.

(ii) The designation of this equipment for no detectable emissions shall be signed by the owner or operator.

(3) A list of equipment identification numbers for pressure relief devices required to comply with 61.242-4(a).

(4)(i) The dates of each compliance test required in §§ 61.242–2(e), 61.242–3(i), 61.242–4, 61.242–7(f), and 61.135(g).

(ii) The background level measured during each compliance test.

(iii) The maximum instrument reading measured at the equipment during each compliance test.

(5) A list of identification numbers for equipment in vacuum service.

(f) The following information pertaining to all valves subject to the requirements of 61.242-7(g) and (h) and to all pumps subject to the requirements of 61.242-2(g) shall be recorded in a log that is kept in a readily accessible location:

(1) A list of identification numbers for valves and pumps that are designated as unsafe to monitor, an explanation for each valve or pump stating why the valve or pump is unsafe to monitor, and the plan for monitoring each valve or pump.

(2) A list of identification numbers for valves that are designated as difficult to monitor, an explanation for each valve stating why the valve is difficult to monitor, and the planned schedule for monitoring each valve.

(g) The following information shall be recorded for valves complying with §61.243-2:

(1) A schedule of monitoring.

(2) The percent of valves found leaking during each monitoring period.

(h) The following information shall be recorded in a log that is kept in a readily accessible location:

(1) Design criterion required in \$ 61.242-2(d)(5), 61.242-3(e)(2), and 61.135(e)(4) and an explanation of the design criterion; and

(2) Any changes to this criterion and the reasons for the changes.

40 CFR Ch. I (7–1–23 Edition)

(i) The following information shall be recorded in a log that is kept in a readily accessible location for use in determining exemptions as provided in the applicability section of this subpart and other specific subparts:

(1) An analysis demonstrating the design capacity of the process unit, and

(2) An analysis demonstrating that equipment is not in VHAP service.

(j) Information and data used to demonstrate that a piece of equipment is not in VHAP service shall be recorded in a log that is kept in a readily accessible location.

[49 FR 23513, June 6, 1984, as amended at 49 FR 38946, Oct. 2, 1984; 54 FR 38077, Sept. 14, 1989; 65 FR 78283, Dec. 14, 2000]

§61.247 Reporting requirements.

(a)(1) An owner or operator of any piece of equipment to which this subpart applies shall submit a statement in writing notifying the Administrator that the requirements of \S 61.242, 61.245, 61.246, and 61.247 are being implemented.

(2) In the case of an existing source or a new source which has an initial startup date preceding the effective date, the statement is to be submitted within 90 days of the effective date, unless a waiver of compliance is granted under $\S61.11$, along with the information required under $\S61.10$. If a waiver of compliance is granted, the statement is to be submitted on a date scheduled by the Administrator.

(3) In the case of new sources which did not have an initial startup date preceding December 14, 2000, the statement required under paragraph (a)(1) of this section shall be submitted with the application for approval of construction, as described in §61.07.

(4) For owners and operators complying with 40 CFR part 65, subpart C or F, the statement required under paragraph (a)(1) of this section shall notify the Administrator that the requirements of 40 CFR part 65, subpart C or F, are being implemented.

(5) The statement is to contain the following information for each source:

(i) Equipment identification number and process unit identification.

(ii) Type of equipment (for example, a pump or pipeline valve).

(iii) Percent by weight VHAP in the fluid at the equipment.

(iv) Process fluid state at the equipment (gas/vapor or liquid).

(v) Method of compliance with the standard (for example, "monthly leak detection and repair" or "equipped with dual mechanical seals").

(b) A report shall be submitted to the Administrator semiannually starting 6 months after the initial report required in paragraph (a) of this section, that includes the following information:

(1) Process unit identification.

(2) For each month during the semiannual reporting period,

(i) Number of valves for which leaks were detected as described in §61.242–7(b) of §61.243–2.

(ii) Number of valves for which leaks were not repaired as required in §61.242-7(d).

(iii) Number of pumps for which leaks were detected as described in §61.242–2(b) and (d)(6).

(iv) Number of pumps for which leaks were not repaired as required in §61.242-2 (c) and (d)(6).

(v) Number of compressors for which leaks were detected as described in §61.242-3(f).

(vi) Number of compressors for which leaks were not repaired as required in §61.242-3(g).

(vii) The facts that explain any delay of repairs and, where appropriate, why a process unit shutdown was technically infeasible.

(3) Dates of process unit shutdowns which occurred within the semiannual reporting period.

(4) Revisions to items reported according to paragraph (a) if changes have occurred since the initial report or subsequent revisions to the initial report.

NOTE: Compliance with the requirements of §61.10(c) is not required for revisions documented under this paragraph.

(5) The results of all performance tests and monitoring to determine compliance with no detectable emissions and with §§ 61.243–1 and 61.243–2 conducted within the semiannual reporting period.

(c) In the first report submitted as required in paragraph (a) of this section, the report shall include a reporting schedule stating the months that semiannual reports shall be submitted. Subsequent reports shall be submitted according to that schedule, unless a revised schedule has been submitted in a previous semiannual report.

(d) An owner or operator electing to comply with the provisions of §§ 61.243– 1 and 61.243–2 shall notify the Administrator of the alternative standard selected 90 days before implementing either of the provisions.

(e) An application for approval of construction or modification, §§ 61.05(a) and 61.07, will not be required if—

(1) The new source complies with the standard, §61.242;

(2) The new source is not part of the construction of a process unit; and

(3) In the next semiannual report required by paragraph (b) of this section, the information in paragraph (a)(5) of this section is reported.

(f) For owners or operators choosing to comply with 40 CFR part 65, subpart C or F, an application for approval of construction or modification, as required under §§ 61.05 and 61.07 will not be required if:

(1) The new source complies with 40 CFR 65.106 through 65.115 and with 40 CFR part 65, subpart C, for surge control vessels and bottoms receivers;

(2) The new source is not part of the construction of a process unit; and

(3) In the next semiannual report required by 40 CFR 65.120(b) and 65.48(b), the information in paragraph (a)(5) of this section is reported.

[49 FR 23513, June 6, 1984, as amended at 49
FR 38947, Oct. 2, 1984; 54 FR 38077, Sept. 14, 1989; 65 FR 78283, Dec. 14, 2000]

TABLE 1 TO SUBPART V OF PART 61— SURGE CONTROL VESSELS AND BOT-TOMS RECEIVERS AT EXISTING SOURCES

Vesse⊡capacity □c⊡ic meters□	Vapor press⊡re ¹ □k̄bpascaīs□
⊑5 ≤capacity <151	≥13.1
151 ≤capacity	≥5.2

¹Ma⊡m⊡m tr⊡e ⊡apor press⊡re as ⊡efine⊟ in ⊡61.241.

[65 FR 78283, Dec. 14, 2000]

Pt. 61, Subpt. V, Table 1

Pt. 61, Subpt. V, Table 2

TABLE 2 TO SUBPART V OF PART 61— SURGE CONTROL VESSELS AND BOT-TOMS RECEIVERS AT NEW SOURCES

Vesse⊑capacity □c⊡ic meters□	Vapor press⊡re ¹ □kõpascaເs⊡
38 ≤capacity <151	≥13.1
151 ≤capacity	≥0.□

¹Ma im m tre apor press re as efine in 61.241.

[65 FR 78283, Dec. 14, 2000]

Subpart W—National Emission Standards for Radon Emissions From Operating Mill Tailings

SOURCE: 54 FR 51703, Dec. 15, 1989, unless otherwise noted.

§61.250 Designation of facilities.

The provisions of this subpart apply to owners or operators of facilities licensed to manage uranium byproduct materials during and following the processing of uranium ores, commonly referred to as uranium mills and their associated tailings. This subpart does not apply to the disposal of tailings.

§61.251 Definitions.

As used in this subpart, all terms not defined here have the meaning given them in the Clean Air Act or 40 CFR part 61, subpart A. The following terms shall have the following specific meanings:

(a) Area means the vertical projection of the pile upon the earth's surface.

(b) Continuous disposal means a method of uranium byproduct material or tailings management and disposal in which uranium byproduct material or tailings are dewatered by mechanical methods immediately after generation. The dried uranium byproduct material or tailings are then placed in trenches or other disposal areas and immediately covered to limit emissions consistent with applicable Federal standards.

(c) *Dewatered* means to remove the water from recently produced uranium byproduct material or tailings by mechanical or evaporative methods such that the water content of the uranium byproduct material or tailings does not exceed 30 percent by weight.

40 CFR Ch. I (7-1-23 Edition)

(d) Existing conventional impoundment means any conventional uranium byproduct material or tailings impoundment which is licensed to accept additional uranium byproduct material or tailings and is in existence on December 15, 1989.

(e) Operation. Operation means that an impoundment is being used for the continued placement of uranium byproduct material or tailings or is in standby status for such placement. An impoundment is in operation from the day that uranium byproduct material or tailings are first placed in the impoundment until the day that final closure begins.

(f) *Phased disposal* means a method of uranium byproduct material or tailings management and disposal which uses lined impoundments which are filled and then immediately dried and covered to meet all applicable Federal standards.

(g) Uranium byproduct material or tailings means the waste produced by the extraction or concentration of uranium from any ore processed primarily for its source material content. Ore bodies depleted by uranium solution extraction and which remain underground do not constitute byproduct material for the purposes of this subpart.

(h) Conventional impoundment. A conventional impoundment is a permanent structure located at any uranium recovery facility which contains mostly solid uranium byproduct material or tailings from the extraction of uranium from uranium ore. These impoundments are left in place at facility closure.

(i) Non-conventional impoundment. A non-conventional impoundment is used for managing liquids from uranium recovery operations and contains uranium byproduct material or tailings suspended in and/or covered by liquids. These structures are commonly known as holding ponds or evaporation ponds and can be located at any uranium recovery facility. They are typically not permanent structures unless they transition to become used as conventional impoundments. Impoundments constructed for the purpose of managing
Appendix I

AUTHENTICATED U.S. GOVERNMENT INFORMATION

§61.255 Recordkeeping requirements.

(a) The owner or operator of any uranium recovery facility must maintain records that confirm that the conventional impoundment(s), non-conventional impoundment(s) and heap leach pile(s) subject to this subpart at the facility meet the requirements in 40 CFR 192.32(a)(1). These records shall include, but not be limited to, the results of liner compatibility tests.

(b) The owner or operator of any uranium recovery facility with non-conventional impoundments must maintain written records from daily inspections and other records confirming that any sediments have remained saturated in the non-conventional impoundments at the facility. Periodic digital photographic evidence, with embedded date stamp and other identifying metadata, shall be collected no less frequently than weekly to demonstrate compliance with the requirements of §61.252(b). Should inspection reveal that a non-conventional impoundment is not in compliance with the requirements of §61.252(b), the owner or operator shall collect photographic evidence before and after the non-compliance is corrected.

(c) The records required in paragraphs (a) and (b) in this section must be kept at the uranium recovery facility for the operational life of the facility and must be made available for inspection by the Administrator, or his authorized representative.

(1) Digital photographs taken to demonstrate compliance with the requirements of §61.252(c) shall be submitted electronically using the Subpart W Impoundment Photographic Reporting (SWIPR) system that is accessed through EPA's Central Data Exchange (CDX) (cdx.epa.gov) at least monthly.

(i) Owners and operators must also submit information identifying the facility and facility location, the name or other designation of each impoundment, and the date and time of each photograph.

(ii) If the reporting form specific to this subpart is not available in SWIPR, the owner or operator must retain the digital photographs at the facility and provide them to the EPA or authorized State upon request, with the supporting information required in paragraph (c)(1)(i) of this section. (2) [Reserved]

[82 FR 5179, Jan. 17, 2017]

§61.256 Exemption from the reporting and testing requirements of 40 CFR 61.10.

All facilities designated under this subpart are exempt from the reporting requirements of 40 CFR 61.10.

Subpart X [Reserved]

Subpart Y—National Emission Standard for Benzene Emissions From Benzene Storage Vessels

SOURCE: 54 FR 38077, Sept. 14, 1989, unless otherwise noted.

§61.270 Applicability and designation of sources.

(a) The source to which this subpart applies is each storage vessel that is storing benzene having a specific gravity within the range of specific gravities specified in ASTM D836-84 for Industrial Grade Benzene, ASTM D835-85 for Refined Benzene-485, ASTM D2359-85a or 93 for Refined Benzene-535, and ASTM D4734-87 or 96 for Refined Benzene-545. These specifications are incorporated by reference as specified in §61.18. See §61.18 for acceptable versions of these methods.

(b) Except for paragraph (b) in §61.276, storage vessels with a design storage capacity less than 38 cubic meters (10,000 gallons) are exempt from the provisions of this subpart.

(c) This subpart does not apply to storage vessels used for storing benzene at coke by-product facilities.

(d) This subpart does not apply to vessels permanently attached to motor vehicles such as trucks, rail cars, barges, or ships.

(e) This subpart does not apply to pressure vessels designed to operate in excess of 204.9 kPa (29.72 psia) and without emissions to the atmosphere.

(f) A designated source subject to the provisions of this subpart that is also subject to applicable provisions of 40 CFR part 60 subparts K, Ka, and Kb shall be required to comply only with the subpart that contains the most stringent requirements for that source.

(g) Alternative means of compliance— (1) Option to comply with part 65. Owners or operators may choose to comply with 40 CFR part 65, subpart C, to satisfy the requirements of $\S 61.271$ through 61.277, except for $\S 61.271(d)(2)$ and 61.274(a) for storage vessels that are subject to this subpart. Other provisions applying to owners or operators who choose to comply with 40 CFR part 65 are provided in 40 CFR 65.1.

(2) Part 61, subpart A. Owners or operators who choose to comply with 40 CFR part 65, subpart C, must also comply with §§61.01, 61.02, 61.05 through 61.08, 61.10(b) through (d), 61.11, and 61.15 for those storage vessels. All sections and paragraphs of subpart A of this part that are not mentioned in this paragraph (g)(2) do not apply for storage vessels complying with 40 CFR part 65, subpart C, except that provisions required to be met prior to implementing 40 CFR part 65 still apply. Owners and operators who choose to comply with 40 CFR part 65, subpart C, must comply with 40 CFR part 65, subpart A.

[54 FR 38077, Sept. 14, 1989, as amended at 65 FR 62159, Oct. 17, 2000; 65 FR 78283, Dec. 14, 2000]

§61.271 Emission standard.

The owner or operator of each storage vessel with a design storage capacity greater than or equal to 38 cubic meters (10,000 gallons) to which this subpart applies shall comply with the requirements in paragraph (d) of this section and with the requirements either in paragraph (a), (b), or (c) of this section, or equivalent as provided in §61.273.

(a) The storage vessel shall be equipped with a fixed roof and an internal floating roof.

(1) An internal floating roof means a cover that rests on the liquid surface (but not necessarily in complete contact with it) inside a storage vessel that has a permanently affixed roof. The internal floating roof shall be floating on the liquid surface at all times, except during initial fill and during those intervals when the storage vessel is completely emptied or subsequently emptied and refilled. 40 CFR Ch. I (7–1–23 Edition)

When the roof is resting on the leg supports, the process of filling, emptying, or refilling shall be continuous and shall be accomplished as rapidly as possible.

(2) Each internal floating roof shall be equipped with one of the closure devices listed in paragraphs (a)(2) (i), (ii), or (iii) of this section between the wall of the storage vessel and the edge of the internal floating roof. This requirement does not apply to each existing storage vessel for which construction of an internal floating roof equipped with a continuous seal commenced on or before July 28, 1988. A continuous seal means a seal that forms a continuous closure that completely covers the space between the wall of the storage vessel and the edge of the internal floating roof.

(i) A foam- or liquid-filled seal mounted in contact with the liquid (liquid-mounted seal). A liquid-mounted seal means a foam- or liquid-filled seal mounted in contact with the liquid between the wall of the storage vessel and the floating roof continuously around the circumference of the vessel.

(ii) Two seals mounted one above the other so that each forms a continuous closure that completely covers the space between the wall of the storage vessel and the edge of the internal floating roof. The lower seal may be vapor-mounted, but both must be continuous.

(iii) A metallic shoe seal. A metallic shoe seal (also referred to as a mechanical shoe seal) is, but is not limited to, a metal sheet held vertically against the wall of the storage vessel by springs or weighted levers and is connected by braces to the floating roof. A flexible coated fabric (envelope) spans the annular space between the metal sheet and the floating roof.

(3) Automatic bleeder vents are to be closed at all times when the roof is floating, except when the roof is being floated off or is being landed on the roof leg supports.

(4) Each opening in a noncontact internal floating roof except for automatic bleeder vents (vacuum breaker vents) and the rim space vents is to provide a projection below the liquid surface.

(5) Each internal floating roof shall meet the specifications listed below. If an existing storage vessel had an internal floating roof with a continuous seal as of July 28, 1988, the requirements listed below do not have to be met until the first time after September 14, 1989, the vessel is emptied and degassed or September 14, 1999, whichever occurs first,

(i) Each opening in the internal floating roof except for leg sleeves, automatic bleeder vents, rim space vents, column wells, ladder wells, sample wells, and stub drains is to be equipped with a cover or lid. The cover or lid shall be equipped with a gasket. Covers on each access hatch and automatic gauge float well shall be bolted.

(ii) Each penetration of the internal floating roof for the purposes of sampling shall be a sample well. Each sample well shall have a slit fabric cover that covers at least 90 percent of the opening.

(iii) Each automatic bleeder vent shall be gasketed.

(iv) Rim space vents shall be equipped with a gasket.

(v) Each penetration of the internal floating roof that allows for passage of a ladder shall have a gasketed sliding cover.

(vi) Each penetration of the internal floating roof that allows for passage of a column supporting the fixed roof shall have a flexible fabric sleeve seal or a gasketed sliding cover.

(6) Each cover or lid on any opening in the internal floating roof shall be closed (i.e., no visible gaps), except when a device is in actual use Covers on each access hatch and each automatic gauge float well which are equipped with bolts shall be bolted when they are not in use. Rim space vents are to be set to open only when the internal floating roof is not floating or at the manufacturer's recommended setting.

(b) The storage vessel shall have an external floating roof.

(1) An external floating roof means a pontoon-type or double-deck-type cover that rests on the liquid surface in a vessel with no fixed roof.

(2) Each external floating roof shall be equipped with a closure device between the wall of the storage vessel and the roof edge. Except as provided in paragraph (b)(5) of this section, the closure device is to consist of two seals, one above the other. The lower seal is referred to as the primary seal and the upper seal is referred to as the secondary seal.

(i) The primary seal shall be either a metallic shoe seal or a liquid-mounted seal. A liquid-mounted seal means a foam- or liquid-filled seal mounted in contact with the liquid between the wall of the storage vessel and the floating roof continuously around the circumference of the vessel. A metallic shoe seal (which can also be referred to as a mechanical shoe seal) is, but is not limited to, a metal sheet held vertically against the wall of the storage vessel by springs or weighted levers and is connected by braces to the floating roof. A flexible coated fabric (envelope) spans the annular space between the metal sheet and the floating roof. Except as provided in §61.272(b)(4), the primary seal shall completely cover the annular space between the edge of the floating roof and the vessel wall.

(ii) The secondary seal shall completely cover the annular space between the external floating roof and the wall of the storage vessel in a continuous fashion except as allowed in §61.272(b)(4).

(3) Except for automatic bleeder vents and rim space vents, each opening in the noncontact external floating roof shall provide a projection below the liquid surface. Except for automatic bleeder vents, rim space vents, roof drains, and leg sleeves, each opening in the roof is to be equipped with a gasketed cover, seal or lid which is to be maintained in a closed position at all times (i.e., no visible gap) except when the device is in actual use. Automatic bleeder vents are to be closed at all times when the roof is floating, except when the roof is being floated off or is being landed on the roof leg supports. Rim vents are to be set to open when the roof is being floated off the roof leg supports or at the manufacturer's recommended setting. Automatic bleeder vents and rim space vents are to be gasketed. Each emergency roof drain is to be provided with a slotted membrane fabric cover that covers at

least 90 percent of the area of the opening.

(4) The roof shall be floating on the liquid at all times (i.e., off the roof leg supports) except during initial fill until the roof is lifted off leg supports and when the vessel is completely emptied and subsequently refilled. The process of emptying and refilling when the roof is resting on the leg supports shall be continuous and shall be accomplished as rapidly as possible.

(5) The requirement for a secondary seal does not apply to each existing storage vessel that was equipped with a liquid-mounted primary seal as of July 28, 1988, until after the first time after September 14, 1989, when the vessel is emptied and degassed or 10 years from September 14, 1989, whichever occurs first.

(c) The storage vessel shall be equipped with a closed vent system and a control device.

(1) The closed vent system shall be designed to collect all benzene vapors and gases discharged from the storage vessel and operated with no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background and visual inspections, as determined in §61.242-11 (subpart V).

(2) The control device shall be designed and operated to reduce inlet benzene emissions by 95 percent or greater. If a flare is used as the control device, it shall meet the specifications described in the general control device requirements of 40 CFR 60.18.

(3) The specifications and requirements listed in paragraphs (c)(1) and (c)(2) of this section for closed vent systems and control devices do not apply during periods of routine maintenance. During periods of routine maintenance, the benzene level in the storage vessel(s) serviced by the control device subject to the provisions of §61.271(c) may be lowered but not raised. Periods of routine maintenance shall not exceed 72 hours as outlined in the maintenance plan required bv §61.272(c)(1)(iii).

(4) The specifications and requirements listed in paragraphs (c)(1) and (c)(2) of this section for closed vents and control devices do not apply during a control system malfunction. A con-

40 CFR Ch. I (7–1–23 Edition)

trol system malfunction means any sudden and unavoidable failure of air pollution control equipment. A failure caused entirely or in part by design deficiencies, poor maintenance, careless operation, or other preventable upset condition or equipment breakdown is not considered a malfunction.

(d) The owner or operator of each affected storage vessel shall meet the requirements of paragraph (a), (b), or (c) of this section or §61.270(g) as follows:

(1) The owner or operator of each existing benzene storage vessel shall meet the requirements of paragraph (a), (b), or (c) of this section no later than 90 days after September 14, 1989, with the exceptions noted in paragraphs (a)(5) and (b)(5), unless a waiver of compliance has been approved by the Administrator in accordance with §61.11.

(2) The owner or operator of each benzene storage vessel upon which construction commenced after September 14, 1989 shall meet the requirements of paragraph (a), (b), or (c) of this section or 61.270(g) prior to filling (*i.e.*, roof is lifted off leg supports) the storage vessel with benzene.

(3) The owner or operator of each benzene storage vessel upon which construction commenced on or after July 28, 1988, and before September 14, 1989, shall meet the requirements of paragraph (a), (b), or (c) of this section on September 14, 1989.

[54 FR 38077, Sept. 14, 1989; 54 FR 50887, Dec. 11, 1989, as amended at 65 FR 78284, Dec. 14, 2000]

§61.272 Compliance provisions.

(a) For each vessel complying with §61.271(a) (fixed roof and internal floating roof) each owner or operator shall:

(1) After installing the control equipment required to comply with §61.271(a), visually inspect the internal floating roof, the primary seal, and the secondary seal (if one is in service), prior to filling the storage vessel with benzene. If there are holes, tears or other openings in the primary seal, the secondary seal, or the seal fabric, or defects in the internal floating roof, the owner or operator shall repair the items before filling the storage vessel.

(2) Visually inspect the internal floating roof and the primary seal or

the secondary seal (if one is in service) through manholes and roof hatches on the fixed roof at least once every 12 months after initial fill, or at least once every 12 months after September 14, 1989, except as provided in paragraph (a)(4)(i) of this section. If the internal floating roof is not resting on the surface of the benzene liquid inside the storage vessel, or there is liquid on the roof, or the seal is detached, or there are holes or tears in the seal fabric, the owner or operator shall repair the items or empty and remove the storage vessel from service within 45 days. If a failure that is detected during inspections required in this paragraph cannot be repaired within 45 days and if the vessel cannot be emptied within 45 days, an extension of up to 30 additional days may be requested from the Administrator in the inspection report required in §61.275(a). Such a request for an extension must document that alternate storage capacity is unavailable and specify a schedule of actions the company will take that will ensure that the control equipment will be repaired or the vessel will be emptied as soon as possible.

(3) Visually inspect the internal floating roof, the primary seal, the secondary seal (if one is in service), gaskets, slotted membranes and sleeve seals (if any) each time the storage vessel is emptied and degassed. In no event shall inspections conducted in accordance with this provision occur at intervals greater than 10 years in the case of vessels conducting the annual visual inspections as specified in paragraph (a)(2) of this section and at intervals greater than 5 years in the case of vessels specified in paragraph (a)(4)(i) of this section.

(i) For all the inspections required by paragraphs (a)(1) and (a)(3) of this section, the owner or operator shall notify the Administrator in writing at least 30 days prior to the refilling of each storage vessel to afford the Administrator the opportunity to have an observer present. If the inspection required by paragraph (a)(3) of this section is not planned and the owner or operator could not have known about the inspection 30 days in advance of refilling the vessel, the owner or operator shall notify the Administrator at least 7 days prior to the refilling of the storage vessel. Notification shall be made by telephone immediately followed by written documentation demonstrating why the inspection was unplanned. Alternatively, the notification including the written documentation may be made in writing and sent by express mail so that it is received by the Administrator at least 7 days prior to refilling.

(ii) If the internal floating roof has defects, the primary seal has holes, tears, or other openings in the seal or the seal fabric, or the secondary seal has holes, tears, or other openings in the seal or the seal fabric, or the gaskets no longer close off the liquid surfaces from the atmosphere, or the slotted membrane has more than 10 percent open area, the owner or operator shall repair the items as necessary so that none of the conditions specified in this paragraph exist before refilling the storage vessel with benzene.

(4) For vessels equipped with a double-seal system as specified in §61.271(a)(2)(ii):

(i) Visually inspect the vessel as specified in paragraph (a)(3) of this section at least every 5 years; or

(ii) Visually inspect the vessel annually as specified in paragraph (a)(2) of this section, and at least every 10 years as specified in paragraph (a)(3) of this section.

(b) For each vessel complying with §61.271(b) (external floating roof) the owner or operator shall:

(1) Determine the gap areas and maximum gap widths between the primary seal and the wall of the storage vessel, and the secondary seal and the wall of the storage vessel according to the following frequency.

(i) For an external floating roof vessel equipped with primary and secondary seals, measurements of gaps between the vessel wall and the primary seal (seal gaps) shall be performed during the hydrostatic testing of the vessel or within 90 days of the initial fill with benzene or within 90 days of September 14, 1989, whichever occurs last, and at least once every 5 years thereafter, except as provided in paragraph (b)(1)(ii) of this section. (ii) For an external floating roof vessel equipped with a liquid-mounted primary seal and without a secondary seal as provided for in §61.271(b)(5), measurement of gaps between the vessel wall and the primary seal (seal gaps) shall be performed within 90 days of September 14, 1989, and at least once per year thereafter. When a secondary seal is installed over the primary seal, measurement of primary seal gaps shall be performed within 90 days of installation and at least once every 5 years thereafter.

(iii) For an external floating roof vessel equipped with primary and secondary seals, measurements of gaps between the vessel wall and the secondary seal shall be performed within 90 days of the initial fill with benzene, within 90 days of installation of the secondary seal, or within 90 days after September 14, 1989, whichever occurs last, and at least once per year thereafter.

(iv) If any source ceases to store benzene for a period of 1 year or more, subsequent introduction of benzene into the vessel shall be considered an initial fill for the purposes of paragraphs (b)(1)(i), (b)(1)(ii), and (b)(1)(iii) of this section.

(2) Determine gap widths and areas in the primary and secondary seals individually by the following procedures:

(i) Measure seal gaps, if any, at one or more floating roof levels when the roof is floating off the roof leg supports.

(ii) Measure seal gaps around the entire circumference of the vessel in each place where a 0.32 centimeter (cm) (1/8 in) diameter uniform probe passes freely (without forcing or binding against the seal) between the seal and the wall of the storage vessel and measure the circumferential distance of each such location.

(iii) The total surface area of each gap described in paragraph (b)(2)(ii) of this section shall be determined by using probes of various widths to measure accurately the actual distance from the vessel wall to the seal and multiplying each such width by its respective circumferential distance.

(3) Add the gap surface area of each gap location for the primary seal and the secondary seal individually. Divide

40 CFR Ch. I (7–1–23 Edition)

the sum for each seal by the nominal diameter of the vessel and compare each ratio to the respective standards in 61.272(b)(4) and 61.272(b)(5).

(4) Repair conditions that do not meet requirements listed in paragraph (b)(4) (i) and (ii) within 45 days of identification in any inspection or empty and remove the storage vessel from service within 45 days.

(i) The accumulated area of gaps between the vessel wall and the metallic shoe seal or the liquid-mounted primary seal shall not exceed 212 cm² per meter of vessel diameter (10.0 in² per foot of vessel diameter) and the width of any portion of any gap shall not exceed 3.81 cm (1½ in).

(A) One end of the metallic shoe is to extend into the stored liquid and the other end is to extend a minimum vertical distance of 61 cm (24 in) above the stored liquid surface.

(B) There are to be no holes, tears, or other openings in the shoe, seal fabric, or seal envelope.

(ii) The secondary seal is to meet the following requirements:

(A) The secondary seal is to be installed above the primary seal so that it completely covers the space between the roof edge and the vessel wall except as provided in paragraph (b)(4)(ii)(B) of this section.

(B) The accumulated area of gaps between the vessel wall and the secondary seal shall not exceed 21.2 cm² per meter of vessel diameter (1.0 in² per foot of vessel diameter) or the width of any portion of any gap shall not exceed 1.27 cm ($\frac{1}{2}$ in). These seal gap requirements may be exceeded during the measurement of primary seal gaps as required by paragraph (b)(1)(i) or (b)(1)(ii) of this section.

(C) There are to be no holes, tears, or other openings in the seal or seal fabric.

(iii) If a failure that is detected during inspections required in this paragraph cannot be repaired within 45 days and if the vessel cannot be emptied within 45 days, an extension of up to 30 additional days may be requested from the Administrator in the inspection report required in §61.275(d). Such extension request must include a

demonstration of unavailability of alternate storage capacity and a specification of a schedule that will assure that the control equipment will be repaired or the vessel will be emptied as soon as possible.

(5) The owner or operator shall notify the Administrator 30 days in advance of any gap measurements required by paragraph (b)(1) of this section to afford the Administrator the opportunity to have an observer present.

(6) Visually inspect the external floating roof, the primary seal, secondary seal, and fittings each time the vessel is emptied and degassed.

(i) If the external floating roof has defects, the primary seal has holes, tears, or other openings in the seal or the seal fabric, or the secondary seal has holes, tears, or other openings in the seal or the seal fabric, the owner or operator shall repair the items as necessary so that none of the conditions specified in this paragraph exist before filling or refilling the storage vessel with benzene.

(ii) For all the inspections required by paragraph (b)(6) of this section, the owner or operator shall notify the Administrator in writing at least 30 days prior to filling or refilling of each storage vessel to afford the Administrator the opportunity to inspect the storage vessel prior to refilling. If the inspection required by paragraph (b)(6) of this section is not planned and the owner or operator could not have known about the inspection 30 days in advance of refilling the vessel, the owner or operator shall notify the Administrator at least 7 days prior to refilling of the storage vessel. Notification shall be made by telephone immediately followed by written documentation demonstrating why the inspection was unplanned. Alternatively, this notification including the written documentation may be made in writing and sent by express mail so that it is received by the Administrator at least 7 days prior to the refilling.

(c) The owner or operator of each source that is equipped with a closed vent system and control device as required in 60.271(c), other than a flare, shall meet the following requirements.

(1) Within 90 days after initial fill or after September 14, 1989, whichever

comes last, submit for approval by the Administrator, an operating plan containing the information listed below.

(i) Documentation demonstrating that the control device being used achieves the required control efficiency during reasonably expected maximum loading conditions. This documentation is to include a description of the gas stream which enters the control device, including flow and benzene content under varying liquid level conditions (dynamic and static) and manufacturer's design specifications for the control device. If the control device or the closed vent capture system receives vapors, gases or liquids, other than fuels, from sources that are not designated sources under this subpart, the efficiency demonstration is to include consideration of all vapors, gases and liquids received by the closed vent capture system and control device. If an enclosed combustion device with a minimum residence time of 0.75 seconds and a minimum temperature of 816 °C (1,500 °F) is used to meet the 95 percent requirement, documentation that those conditions exist is sufficient to meet the requirements of this paragraph.

(ii) A description of the parameter or parameters to be monitored to ensure that the control device is operated and maintained in conformance with its design and an explanation of the criteria used for selection of that parameter (or parameters).

(iii) A maintenance plan for the system including the type of maintenance necessary, planned frequency of maintenance, and lengths of maintenance periods for those operations that would require the closed vent system or the control device to be out of compliance with §61.271(c). The maintenance plan shall require that the system be out of compliance with §61.271(c) for no more than 72 hours per year.

(2) Operate, monitor the parameters, and maintain the closed vent system and control device in accordance with the operating plan submitted to the Administrator in accordance with paragraph (c)(1) of this section, unless the plan was modified by the Administrator during the approval process. In this case, the modified plan applies.

§61.273

(d) The owner or operator of each source that is equipped with a closed vent system and a flare to meet the requirements in 61.271(c) shall meet the requirements as specified in the general control device requirements in 40 CFR 60.18 (e) and (f).

 $[54\ {\rm FR}\ 38077,\ {\rm Sept.}\ 14,\ 1989,\ as\ amended\ at\ 65\ {\rm FR}\ 62159,\ {\rm Oct.}\ 17,\ 2000]$

§61.273 Alternative means of emission limitation.

(a) Upon written application from any person, the Administrator may approve the use of alternative means of emission limitation which have been demonstrated to his satisfaction to achieve a reduction in benzene emissions at least equivalent to the reduction in emissions achieved by any requirement in §61.271 (a), (b), or (c) of this subpart.

(b) Determination of equivalence to the reduction in emissions achieved by the requirements of §61.271 (a), (b), or (c) will be evaluated using the following information to be included in the written application to the Administrator:

(1) Actual emissions tests that use full-size or scale-model storage vessels that accurately collect and measure all benzene emissions from a given control device, and that accurately simulate wind and account for other emission variables such as temperature and barometric pressure.

(2) An engineering evaluation that the Administrator determines is an accurate method of determining equivalence.

(c) The Administrator may condition approval of equivalency on requirements that may be necessary to ensure operation and maintenance to achieve the same emission reduction as the requirements of 61.271 (a), (b), or (c).

(d) If, in the Administrator's judgment, an application for equivalence may be approvable, the Administrator will publish a notice of preliminary determination in the FEDERAL REGISTER and provide the opportunity for public hearing. After notice and opportunity for public hearing, the Administrator will determine the equivalence of the alternative means of emission limitation and will publish the final determination in the FEDERAL REGISTER.

40 CFR Ch. I (7–1–23 Edition)

§61.274 Initial report.

(a) The owner or operator of each storage vessel to which this subpart applies and which has a design capacity greater than or equal to 38 cubic meters (10.000 gallons) shall submit an initial report describing the controls which will be applied to meet the equipment requirements of §61.271 or §61.270(g). For an existing storage vessel or a new storage vessel for which construction and operation commenced prior to September 14, 1989, this report shall be submitted within 90 days of September 14, 1989 and can be combined with the report required by §61.10. For a new storage vessel for which construction or operation commenced on or after September 14, 1989, the report shall be combined with the report required by §61.07 or 40 CFR 65.5(b). In the case where the owner or operator seeks to comply with 61.271(c), with a control device other than a flare, this information may consist of the information required by 61.272(c)(1).

(b) The owner or operator of each storage vessel seeking to comply with §61.271(c) with a flare, shall submit a report containing the measurements required by 40 CFR 60.18(f) (1), (2), (3), (4), (5), and (6). For the owner or operator of an existing storage vessel not seeking to obtain a waiver or a new storage vessel for which construction and operation commenced prior to September 14, 1989, this report shall be combined with the report required by paragraph (a) of this section. For the owner or operator of an existing storage vessel seeking to obtain a waiver, the reporting date will be established in the response to the waiver request. For the owner or operator of a new storage vessel for which construction or operation commenced after September 14, 1989, the report shall be submitted within 90 days of the date the vessel is initially filled (or partially filled) with benzene.

[54 FR 38077, Sept. 14, 1989, as amended at 65 FR 78284, Dec. 14, 2000]

§61.275 Periodic report.

(a) The owner or operator of each storage vessel to which this subpart applies after installing control equipment in accordance with §61.271(a) (fixed roof

and internal floating roof) shall submit a report describing the results of each inspection conducted in accordance with 61.272(a). For vessels for which annual inspections are required under 61.272(a)(2), the first report is to be submitted no more than 12 months after the initial report submitted in accordance with 61.274, and each report is to be submitted within 60 days of each annual inspection.

(1) Each report shall include the date of the inspection of each storage vessel and identify each storage vessel in which:

(i) The internal floating roof is not resting on the surface of the benzene liquid inside the storage vessel, or there is liquid on the roof, or the seal is detached from the internal floating roof, or there are holes, tears or other openings in the seal or seal fabric; or

(ii) There are visible gaps between the seal and the wall of the storage vessel.

(2) Where an annual report identifies any condition in paragraph (a)(1) of this section the annual report shall describe the nature of the defect, the date the storage vessel was emptied, and the nature of and date the repair was made, except as provided in paragraph (a)(3) of this section.

(3) If an extension is requested in an annual periodic report in accordance with §61.272(a)(2), a supplemental periodic report shall be submitted within 15 days of repair. The supplemental periodic report shall identify the vessel and describe the date the storage vessel was emptied and the nature of and date the repair was made.

(b) The owner or operator of each storage vessel to which this subpart applies after installing control equipment in accordance with §61.271(a) (fixed roof and internal floating roof) shall submit a report describing the results of each inspection conducted in accordance with §61.272(a) (3) or (4).

(1) The report is to be submitted within 60 days of conducting each inspection required by 61.272(a) (3) or (4).

(2) Each report shall identify each storage vessel in which the owner or operator finds that the internal floating roof has defects, the primary seal has holes, tears, or other openings in the seal or the seal fabric, or the secondary seal (if one has been installed) has holes, tears, or other openings in the seal or the seal fabric, or the gaskets no longer close off the liquid surfaces from the atmosphere, or the slotted membrane has more than 10 percent open area. The report shall also describe the nature of the defect, the date the storage vessel was emptied, and the nature of and date the repair was made.

(c) Any owner or operator of an existing storage vessel which had an internal floating roof with a continuous seal as of July 28, 1988, and which seeks to comply with the requirements of (61.271(a)(5)) during the first time after September 14, 1989, when the vessel is emptied and degassed but no later than 10 years from September 14, 1989, shall notify the Administrator 30 days prior to the completion of the installation of such controls and the date of refilling of the vessel so the Administrator has an opportunity to have an observer present to inspect the storage vessel before it is refilled. This report can be combined with the one required by §61.275(b).

(d) The owner or operator of each storage vessel to which this subpart applies after installing control equipment in accordance with §61.271(b) (external floating roof) shall submit a report describing the results of each seal gap measurement made in accordance with §61.272(b). The first report is to be submitted no more than 12 months after the initial report submitted in accordance with §61.274(a), and each annual periodic report is to be submitted within 60 days of each annual inspection.

(1) Each report shall include the date of the measurement, the raw data obtained in the measurement, and the calculations described in §61.272(b) (2) and (3), and shall identify each storage vessel which does not meet the gap specifications of §61.272(b). Where an annual report identifies any vessel not meeting the seal gap specifications of §61.272(b) the report shall describe the date the storage vessel was emptied, the measures used to correct the condition and the date the storage vessel was brought into compliance.

(2) If an extension is requested in an annual periodic report in accordance with 61.272(b)(4)(iii), a supplemental

periodic report shall be submitted within 15 days of repair. The supplemental periodic report shall identify the vessel and describe the date the vessel was emptied and the nature of and date the repair was made.

(e) Excess emission report.

(1) The owner or operator of each source seeking to comply with §61.271(c) (vessels equipped with closed vent systems with control devices) shall submit a quarterly report informing the Administrator of each occurrence that results in excess emissions. Excess emissions are emissions that occur at any time when compliance with the specifications and requirements of §61.271(c) are not achieved, as evidenced by the parameters being in accordance measured with §61.272(c)(1)(ii) if a control device other than a flare is used, or by the measurements required in §61.272(d) and the general control device requirements in 40 CFR 60.18(f) (1) and (2) if a flare is used.

(2) The owner or operator shall submit the following information as a minimum in the report required by (e)(1) of this section:

(i) Identify the stack and other emission points where the excess emissions occurred;

(ii) A statement of whether or not the owner or operator believes a control system malfunction has occurred.

(3) If the owner or operator states that a control system malfunction has occurred, the following information as a minimum is also to be included in the report required under paragraph (e)(1) of this section:

(i) Time and duration of the control system malfunction as determined by continuous monitoring data (if any), or the inspections or monitoring done in accordance with the operating plan required by 61.272(c).

(ii) Cause of excess emissions.

§61.276 Recordkeeping.

(a) Each owner or operator with a storage vessel subject to this subpart shall keep copies of all the reports and records required by this subpart for at least 2 years, except as specified in paragraphs (b) and (c)(1) of this section.

(b) Each owner or operator with a storage vessel, including any vessel

40 CFR Ch. I (7–1–23 Edition)

which has a design storage capacity less than 38 cubic meters (10,000 gallons), shall keep readily accessible records showing the dimensions of the storage vessel and an analysis showing the capacity of the storage vessel. This record shall be kept as long as the storage vessel is in operation. Each storage vessel with a design capacity of less than 38 cubic meters (10,000 gallons) is subject to no provisions of this subpart other than those required by this paragraph.

(c) The following information pertaining to closed vent system and control devices shall be kept in a readily accessible location.

(1) A copy of the operating plan. This record shall be kept as long as the closed vent system and control device is in use.

(2) A record of the measured values of the parameters monitored in accordance with $\S61.272(c)(1)(ii)$ and $\S61.272(c)(2)$.

(3) A record of the maintenance performed in accordance with §61.272(c)(1)(iii) of the operating plan, including the following:

(i) The duration of each time the closed vent system and control device does not meet the specifications of §61.271(c) due to maintenance, including the following:

(A) The first time of day and date the requirements of 61.271(c) were not met at the beginning of maintenance.

(B) The first time of day and date the requirements of §61.271(c) were met at the conclusion of maintenance.

(C) A continuous record of the liquid level in each storage vessel that the closed vent system and control device receive vapors from during the interval between the times specified by (c)(3)(i)(A) and (c)(3)(i)(B). Pumping records (simultaneous input and output) may be substituted for records of the liquid level.

§61.277 Delegation of authority.

(a) In delegating implementation and enforcement authority to a State under section 112(d) of the Act, the authorities contained in paragraph (b) of this section shall be retained by the Administrator and not transferred to a State.

(b) Authorities which will not be delegated to States: §61.273.

Subparts Z-AA [Reserved]

Subpart BB—National Emission Standard for Benzene Emissions From Benzene Transfer Operations

SOURCE: 55 FR 8341, Mar. 7, 1990, unless otherwise noted.

§61.300 Applicability.

(a) The affected facility to which this subpart applies is the total of all loading racks at which benzene is loaded into tank trucks, railcars, or marine vessels at each benzene production facility and each bulk terminal. However, specifically exempted from this regulation are loading racks at which only the following are loaded: Benzeneladen waste (covered under subpart FF of this part), gasoline, crude oil, natural gas liquids, petroleum distillates (e.g., fuel oil, diesel, or kerosene), or benzene-laden liquid from coke byproduct recovery plants.

(b) Any affected facility under paragraph (a) of this section which loads only liquid containing less than 70 weight-percent benzene is exempt from the requirements of this subpart, except for the recordkeeping and reporting requirements in §61.305(i).

(c) Comply with standards at each loading rack. Any affected facility under paragraph (a) of this section shall comply with the standards in §61.302 or as specified in paragraph (f) of this section, if applicable, at each loading rack that is handling a liquid containing 70 weight-percent or more benzene.

(d) Any affected facility under paragraph (a) of this section whose annual benzene loading is less than 1.3 million liters of 70 weight-percent or more benzene is exempt from the requirements of this subpart, except for the recordkeeping and reporting requirements in §61.305(i).

(e) The owner or operator of an affected facility, as defined in §61.300(a) that loads a marine vessel shall be in compliance with the provisions of this subpart on and after July 23, 1991. If an affected facility that loads a marine vessel also loads a tank truck or railcar, the marine vessel loading racks shall be in compliance with the provisions of this subpart on and after July 23, 1991, while the tank truck loading racks and the railcar loading racks shall be in compliance as required by §61.12.

(f) Alternative means of compliance—(1) Option to comply with part 65. Owners or operators may choose to comply with 40 CFR part 65, subpart E, to satisfy the requirements of §§ 61.302 through 61.306 for all tank truck or railcar loading racks that are subject to this subpart. Loading racks are referred to as transfer racks in 40 CFR part 65, subpart E. Other provisions applying to owners or operators who choose to comply with 40 CFR part 65 are provided in 40 CFR 65.1. All marine vessel loading racks shall comply with the provisions in §§ 61.302 through 61.306.

(2) Part 61, subpart A. Owners or operators who choose to comply with 40 CFR part 65, subpart E, must also comply with §§61.01, 61.02, 61.05 through 61.08, 61.10(b) through (d), 61.11, and 61.15 for those loading racks. All sections and paragraphs of subpart A of this part that are not mentioned in this paragraph (f)(2) do not apply to owners or operators of loading racks complying with 40 CFR part 65, subpart E, except that provisions required to be met prior to implementing 40 CFR part 65 still apply. Owners and operators who choose to comply with 40 CFR part 65, subpart E, must comply with 40 CFR part 65, subpart A.

[55 FR 8341, Mar. 7, 1990, as amended at 55 FR 45804, Oct. 31, 1990; 65 FR 78284, Dec. 14, 2000]

§61.301 Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Act, or in subpart A or subpart V of part 61.

Bulk terminal means any facility which receives liquid product containing benzene by pipelines, marine vessels, tank trucks, or railcars, and loads the product for further distribution into tank trucks, railcars, or marine vessels.

Car-sealed means having a seal that is placed on the device used to change the position of a valve (e.g., from open to closed) such that the position of the Appendix J

§61.306

AUTHENTICATED U.S. GOVERNMENT INFORMATION

> (d) and (e) for each tank truck, railcar, or marine vessel at least once per year to reflect current test results as determined by the appropriate method. The owner or operator shall include, as a minimum, the following information in this documentation:

(1) Test title;

(2) Tank truck, railcar, or marine vessel owner and address;

(3) Tank truck, railcar, or marine vessel identification number;

(4) Testing location;

(5) Date of test;

(6) Tester name and signature;

(7) Witnessing inspector: name, signature, and affiliation; and

(8) Test results, including, for railcars and tank trucks, the initial pressure up to which the tank was pressured at the start of the test.

(i) Each owner or operator of an affected facility complying with §61.300(b) or §61.300(d) shall record the following information. The first year after promulgation the owner or operator shall submit a report containing the requested information to the Director of the Emission Standards Division, (MD-13), U.S. Environmental Protection Agency, Research Triangle Park, North Carolina 27711. After the first year, the owner or operator shall continue to record: however, no reporting is required. The information shall be made available if requested. The information shall include, as a minimum:

(1) The affected facility's name and address;

(2) The weight percent of the benzene loaded;

(3) The type of vessel loaded (i.e., tank truck, railcar, or marine vessel); and

(4) The annual amount of benzene loaded into each type of vessel.

 $[55\ {\rm FR}\ 8341,\ {\rm Mar.}\ 7,\ 1990,\ {\rm as}\ {\rm amended}\ {\rm at}\ 65\ {\rm FR}\ 62159,\ {\rm Oct.}\ 17,\ 2000]$

§61.306 Delegation of authority.

(a) In delegating implementation and enforcement authority to a State under section 112(d) of the Act, the authorities contained in paragraph (b) of this section shall be retained by the Administrator and not transferred to a State.

(b) Authorities which will not be delegated to States: No restrictions.

40 CFR Ch. I (7–1–23 Edition)

Subparts CC-EE [Reserved]

Subpart FF—National Emission Standard for Benzene Waste Operations

SOURCE: 55 FR 8346, Mar. 7, 1990, unless otherwise noted.

§61.340 Applicability.

(a) The provisions of this subpart apply to owners and operators of chemical manufacturing plants, coke byproduct recovery plants, and petroleum refineries.

(b) The provisions of this subpart apply to owners and operators of hazardous waste treatment, storage, and disposal facilities that treat, store, or dispose of hazardous waste generated by any facility listed in paragraph (a) of this section. The waste streams at hazardous waste treatment, storage. and disposal facilities subject to the provisions of this subpart are the benzene-containing hazardous waste from any facility listed in paragraph (a) of this section. A hazardous waste treatment, storage, and disposal facility is a facility that must obtain a hazardous waste management permit under subtitle C of the Solid Waste Disposal Act.

(c) At each facility identified in paragraph (a) or (b) of this section, the following waste is exempt from the requirements of this subpart:

(1) Waste in the form of gases or vapors that is emitted from process fluids:

(2) Waste that is contained in a segregated stormwater sewer system.

(d) At each facility identified in paragraph (a) or (b) of this section, any gaseous stream from a waste management unit, treatment process, or wastewater treatment system routed to a fuel gas system, as defined in §61.341, is exempt from this subpart. No testing, monitoring, recordkeeping, or reporting is required under this subpart for any gaseous stream from a waste management unit, treatment process, or wastewater treatment unit routed to a fuel gas system.

[55 FR 8346, Mar. 7, 1990, as amended at 55 FR 37231, Sept. 10, 1990; 58 FR 3095, Jan. 7, 1993;
67 FR 68531, Nov. 12, 2002]

§61.341

§61.341 Definitions.

Benzene concentration means the fraction by weight of benzene in a waste as determined in accordance with the procedures specified in §61.355 of this subpart.

Car-seal means a seal that is placed on a device that is used to change the position of a valve (e.g., from opened to closed) in such a way that the position of the valve cannot be changed without breaking the seal.

Chemical manufacturing plant means any facility engaged in the production of chemicals by chemical, thermal, physical, or biological processes for use as a product, co-product, by-product, or intermediate including but not limited to industrial organic chemicals, organic pesticide products, pharmaceutical preparations, paint and allied products, fertilizers, and agricultural chemicals. Examples of chemical manufacturing plants include facilities at which process units are operated to produce one or more of the following chemicals: benzenesulfonic acid, benzene. chlorobenzene, cumene. cyclohexane, ethylene, ethylbenzene, hydroquinone, linear alklylbenzene, nitrobenzene, resorcinol, sulfolane, or styrene.

Closed-vent system means a system that is not open to the atmosphere and is composed of piping, ductwork, connections, and, if necessary, flow inducing devices that transport gas or vapor from an emission source to a control device.

Coke by-product recovery plant means any facility designed and operated for the separation and recovery of coal tar derivatives (by-products) evolved from coal during the coking process of a coke oven battery.

Container means any portable waste management unit in which a material is stored, transported, treated, or otherwise handled. Examples of containers are drums, barrels, tank trucks, barges, dumpsters, tank cars, dump trucks, and ships.

Control device means an enclosed combustion device, vapor recovery system, or flare.

Cover means a device or system which is placed on or over a waste placed in a waste management unit so that the entire waste surface area is enclosed and sealed to minimize air emissions. A cover may have openings necessary for operation, inspection, and maintenance of the waste management unit such as access hatches, sampling ports, and gauge wells provided that each opening is closed and sealed when not in use. Example of covers include a fixed roof installed on a tank, a lid installed on a container, and an airsupported enclosure installed over a waste management unit.

External floating roof means a pontoon-type or double-deck type cover with certain rim sealing mechanisms that rests on the liquid surface in a waste management unit with no fixed roof.

Facility means all process units and product tanks that generate waste within a stationary source, and all waste management units that are used for waste treatment, storage, or disposal within a stationary source.

Fixed roof means a cover that is mounted on a waste management unit in a stationary manner and that does not move with fluctuations in liquid level.

Floating roof means a cover with certain rim sealing mechanisms consisting of a double deck, pontoon single deck, internal floating cover or covered floating roof, which rests upon and is supported by the liquid being contained, and is equipped with a closure seal or seals to close the space between the roof edge and unit wall.

Flow indicator means a device which indicates whether gas flow is present in a line or vent system.

Fuel gas system means the offsite and onsite piping and control system that gathers gaseous streams generated by facility operations, may blend them with sources of gas, if available, and transports the blended gaseous fuel at suitable pressures for use as fuel in heaters, furnaces, boilers, incinerators, gas turbines, and other combustion devices located within or outside the facility. The fuel is piped directly to each individual combustion device, and the system typically operates at pressures over atmospheric.

Individual drain system means the system used to convey waste from a process unit, product storage tank, or waste management unit to a waste management unit. The term includes all process drains and common junction boxes, together with their associated sewer lines and other junction boxes, down to the receiving waste management unit.

Internal floating roof means a cover that rests or floats on the liquid surface inside a waste management unit that has a fixed roof.

Liquid-mounted seal means a foam or liquid-filled primary seal mounted in contact with the liquid between the waste management unit wall and the floating roof continuously around the circumference.

Loading means the introduction of waste into a waste management unit but not necessarily to complete capacity (also referred to as filling).

Maximum organic vapor pressure means the equilibrium partial pressure exerted by the waste at the temperature equal to the highest calendarmonth average of the waste storage temperature for waste stored above or below the ambient temperature or at the local maximum monthly average temperature as reported by the National Weather Service for waste stored at the ambient temperature, as determined:

(1) In accordance with §60.17(c); or

(2) As obtained from standard reference texts; or

(3) In accordance with §60.17(a)(37); or(4) Any other method approved by the Administrator.

No detectable emissions means less than 500 parts per million by volume (ppmv) above background levels, as measured by a detection instrument reading in accordance with the procedures specified in §61.355(h) of this subpart.

Oil-water separator means a waste management unit, generally a tank or surface impoundment, used to separate oil from water. An oil-water separator consists of not only the separation unit but also the forebay and other separator basins, skimmers, weirs, grit chambers, sludge hoppers, and bar screens that are located directly after the individual drain system and prior to additional treatment units such as an air flotation unit, clarifier, or biological treatment unit. Examples of an oil-water separator incude an API sepa40 CFR Ch. I (7–1–23 Edition)

rator, parallel-plate interceptor, and corrugated-plate interceptor with the associated ancillary equipment.

Petroleum refinery means any facility engaged in producing gasoline, kerosene, distillate fuel oils, residual fuel oils, lubricants, or other products through the distillation of petroleum, or through the redistillation, cracking, or reforming of unfinished petroleum derivatives.

Petroleum means the crude oil removed from the earth and the oils derived from tar sands, shale, and coal.

Point of waste generation means the location where the waste stream exits the process unit component or storage tank prior to handling or treatment in an operation that is not an integral part of the production process, or in the case of waste management units that generate new wastes after treatment, the location where the waste stream exits the waste management unit component.

Process unit means equipment assembled and connected by pipes or ducts to produce intermediate or final products. A process unit can be operated independently if supplied with sufficient fuel or raw materials and sufficient product storage facilities.

Process unit turnaround means the shutting down of the operations of a process unit, the purging of the contents of the process unit, the maintenance or repair work, followed by restarting of the process.

Process unit turnaround waste means a waste that is generated as a result of a process unit turnaround.

Process wastewater means water which comes in contact with benzene during manufacturing or processing operations conducted within a process unit. Process wastewater is not organic wastes, process fluids, product tank drawdown, cooling tower blowdown, steam trap condensate, or landfill leachate.

Process wastewater stream means a waste stream that contains only process wastewater.

Product tank means a stationary unit that is designed to contain an accumulation of materials that are fed to or

produced by a process unit, and is constructed primarily of non-earthen materials (e.g., wood, concrete, steel, plastic) which provide structural support.

Product tank drawdown means any material or mixture of materials discharged from a product tank for the purpose of removing water or other contaminants from the product tank.

Safety device means a closure device such as a pressure relief valve, frangible disc, fusible plug, or any other type of device which functions exclusively to prevent physical damage or permanent deformation to a unit or its air emission control equipment by venting gases or vapors directly to the atmosphere during unsafe conditions resulting from an unplanned, accidental, or emergency event. For the purpose of this subpart, a safety device is not used for routine venting of gases or vapors from the vapor headspace underneath a cover such as during filling of the unit or to adjust the pressure in this vapor headspace in response to normal daily diurnal ambient temperature fluctuations. A safety device is designed to remain in a closed position during normal operations and open only when the internal pressure, or another relevant parameter, exceeds the device threshold setting applicable to the air emission control equipment as determined by the owner or operator based on manufacturer recommendations, applicable regulations, fire protection and prevention codes, standard engineering codes and practices, or other requirements for the safe handling of flammable, ignitable, explosive, reactive, or hazardous materials.

Segregated stormwater sewer system means a drain and collection system designed and operated for the sole purpose of collecting rainfall runoff at a facility, and which is segregated from all other individual drain systems.

Sewer line means a lateral, trunk line, branch line, or other enclosed conduit used to convey waste to a downstream waste management unit.

Slop oil means the floating oil and solids that accumulate on the surface of an oil-water separator.

Sour water stream means a stream that:

(1) Contains ammonia or sulfur compounds (usually hydrogen sulfide) at concentrations of 10 ppm by weight or more;

(2) Is generated from separation of water from a feed stock, intermediate, or product that contained ammonia or sulfur compounds; and

(3) Requires treatment to remove the ammonia or sulfur compounds.

Sour water stripper means a unit that: (1) Is designed and operated to remove ammonia or sulfur compounds

(usually hydrogen sulfide) from sour water streams;

(2) Has the sour water streams transferred to the stripper through hard piping or other enclosed system; and

(3) Is operated in such a manner that the offgases are sent to a sulfur recovery unit, processing unit, incinerator, flare, or other combustion device.

Surface impoundment means a waste management unit which is a natural topographic depression, man-made excavation, or diked area formed primarily of earthen materials (although it may be lined with man-made materials), which is designed to hold an accumulation of liquid wastes or waste containing free liquids, and which is not an injection well. Examples of surface impoundments are holding, storage, settling, and aeration pits, ponds, and lagoons.

Tank means a stationary waste management unit that is designed to contain an accumulation of waste and is constructed primarily of nonearthen materials (e.g., wood, concrete, steel, plastic) which provide structural support.

Treatment process means a stream stripping unit, thin-film evaporation unit, waste incinerator, or any other process used to comply with §61.348 of this subpart.

Vapor-mounted seal means a foamfilled primary seal mounted continuously around the perimeter of a waste management unit so there is an annular vapor space underneath the seal. The annular vapor space is bounded by the bottom of the primary seal, the unit wall, the liquid surface, and the floating roof.

Waste means any material resulting from industrial, commercial, mining or agricultural operations, or from community activities that is discarded or is being accumulated, stored, or physically, chemically, thermally, or biologically treated prior to being discarded, recycled, or discharged.

Waste management unit means a piece of equipment, structure, or transport mechanism used in handling, storage, treatment, or disposal of waste. Examples of a waste management unit include a tank, surface impoundment, container, oil-water separator, individual drain system, steam stripping unit, thin-film evaporation unit, waste incinerator, and landfill.

Waste stream means the waste generated by a particular process unit, product tank, or waste management unit. The characteristics of the waste stream (e.g., flow rate, benzene concentration, water content) are determined at the point of waste generation. Examples of a waste stream include process wastewater, product tank drawdown, sludge and slop oil removed from waste management units, and landfill leachate.

Wastewater treatment system means any component, piece of equipment, or installation that receives, manages, or treats process wastewater, product tank drawdown, or landfill leachate prior to direct or indirect discharge in accordance with the National Pollutant Discharge Elimination System permit regulations under 40 CFR part 122. These systems typically include individual drain systems, oil-water separators, air flotation units, equalization tanks, and biological treatment units.

Water seal controls means a seal pot, p-leg trap, or other type of trap filled with water (e.g., flooded sewers that maintain water levels adequate to prevent air flow through the system) that creates a water barrier between the sewer line and the atmosphere. The water level of the seal must be maintained in the vertical leg of a drain in order to be considered a water seal.

[55 FR 8346, Mar. 7, 1990; 55 FR 12444, Apr. 3, 1990, as amended at 58 FR 3095, Jan. 7, 1993; 67 FR 68531, Nov. 12, 2002]

§61.342 Standards: General.

(a) An owner or operator of a facility at which the total annual benzene quantity from facility waste is less than 10 megagrams per year (Mg/yr) (11 ton/yr) shall be exempt from the re-

40 CFR Ch. I (7–1–23 Edition)

quirements of paragraphs (b) and (c) of this section. The total annual benzene quantity from facility waste is the sum of the annual benzene quantity for each waste stream at the facility that has a flow-weighted annual average water content greater than 10 percent or that is mixed with water, or other wastes, at any time and the mixture has an annual average water content greater than 10 percent. The benzene quantity in a waste stream is to be counted only once without multiple counting if other waste streams are mixed with or generated from the original waste stream. Other specific requirements for calculating the total annual benzene waste quantity are as follows:

(1) Wastes that are exempted from control under \$ 61.342(c)(2) and 61.342(c)(3) are included in the calculation of the total annual benzene quantity if they have an annual average water content greater than 10 percent, or if they are mixed with water or other wastes at any time and the mixture has an annual average water content greater than 10 percent.

(2) The benzene in a material subject to this subpart that is sold is included in the calculation of the total annual benzene quantity if the material has an annual average water content greater than 10 percent.

(3) Benzene in wastes generated by remediation activities conducted at the facility, such as the excavation of contaminated soil, pumping and treatment of groundwater, and the recovery of product from soil or groundwater, are not included in the calculation of total annual benzene quantity for that facility. If the facility's total annual benzene quantity is 10 Mg/yr (11 ton/yr) or more, wastes generated by remediation activities are subject to the requirements of paragraphs (c) through (h) of this section. If the facility is managing remediation waste generated offsite, the benzene in this waste shall be included in the calculation of total annual benzene quantity in facility waste, if the waste streams have an annual average water content greater than 10 percent, or if they are mixed with water or other wastes at any time and the mixture has an annual average water content greater than 10 percent.

(4) The total annual benzene quantity is determined based upon the quantity of benzene in the waste before any waste treatment occurs to remove the benzene except as specified in $\S61.355(c)(1)(i)$ (A) through (C).

(b) Each owner or operator of a facility at which the total annual benzene quantity from facility waste is equal to or greater than 10 Mg/yr (11 ton/yr) as determined in paragraph (a) of this section shall be in compliance with the requirements of paragraphs (c) through (h) of this section no later than 90 days following the effective date, unless a waiver of compliance has been obtained under fdl.11, or by the initial startup for a new source with an initial startup after the effective date.

(1) The owner or operator of an existing source unable to comply with the rule within the required time may request a waiver of compliance under §61.10.

(2) As part of the waiver application, the owner or operator shall submit to the Administrator a plan under (10,0) that is an enforceable commitment to obtain environmental benefits to mitigate the benzene emissions that result from extending the compliance date. The plan shall include the following information:

(i) A description of the method of compliance, including the control approach, schedule for installing controls, and quantity of the benzene emissions that result from extending the compliance date;

(ii) If the control approach involves a compliance strategy designed to obtain integrated compliance with multiple regulatory requirements, a description of the other regulations involved and their effective dates; and

(iii) A description of the actions to be taken at the facility to obtain mitigating environmental benefits, including how the benefits will be obtained, the schedule for these actions, and an estimate of the quantifiable benefits that directly result from these actions.

(c) Each owner or operator of a facility at which the total annual benzene quantity from facility waste is equal to or greater than 10 Mg/yr (11 ton/yr) as determined in paragraph (a) of this section shall manage and treat the facility waste as follows: (1) For each waste stream that contains benzene, including (but not limited to) organic waste streams that contain less than 10 percent water and aqueous waste streams, even if the wastes are not discharged to an individual drain system, the owner or operator shall:

(i) Remove or destroy the benzene contained in the waste using a treatment process or wastewater treatment system that complies with the standards specified in §61.348 of this subpart.

(ii) Comply with the standards specified in \$ 61.343 through 61.347 of this subpart for each waste management unit that receives or manages the waste stream prior to and during treatment of the waste stream in accordance with paragraph (c)(1)(i) of this section.

(iii) Each waste management unit used to manage or treat waste streams that will be recycled to a process shall comply with the standards specified in §§ 61.343 through 61.347. Once the waste stream is recycled to a process, including to a tank used for the storage of production process feed, product, or product intermediates, unless this tank is used primarily for the storage of wastes, the material is no longer subject to paragraph (c) of this section.

(2) A waste stream is exempt from paragraph (c)(1) of this section provided that the owner or operator demonstrates initially and, thereafter, at least once per year that the flowweighted annual average benzene concentration for the waste stream is less than 10 ppmw as determined by the procedures specified in 61.355(c)(2) or 61.355(c)(3).

(3) A waste stream is exempt from paragraph (c)(1) of this section provided that the owner or operator demonstrates initially and, thereafter, at least once per year that the conditions specified in either paragraph (c)(3)(i) or (c)(3)(ii) of this section are met.

(i) The waste stream is process wastewater that has a flow rate less than 0.02 liters per minute (0.005 gallons per minute) or an annual wastewater quantity of less than 10 Mg/yr (11 ton/yr); or

(ii) All of the following conditions are met:

§61.342

(A) The owner or operator does not choose to exempt process wastewater under paragraph (c)(3)(i) of this section,

(B) The total annual benzene quantity in all waste streams chosen for exemption in paragraph (c)(3)(ii) of this section does not exceed 2.0 Mg/yr (2.2 ton/yr) as determined in the procedures in 61.355(j), and

(C) The total annual benzene quantity in a waste stream chosen for exemption, including process unit turnaround waste, is determined for the year in which the waste is generated.

(d) As an alternative to the requirements specified in paragraphs (c) and (e) of this section, an owner or operator of a facility at which the total annual benzene quantity from facility waste is equal to or greater than 10 Mg/yr (11 ton/yr) as determined in paragraph (a) of this section may elect to manage and treat the facility waste as follows:

(1) The owner or operator shall manage and treat facility waste other than process wastewater in accordance with the requirements of paragraph (c)(1) of this section.

(2) The owner or operator shall manage and treat process wastewater in accordance with the following requirements:

(i) Process wastewater shall be treated to achieve a total annual benzene quantity from facility process wastewater less than 1 Mg/yr (1.1 ton/yr). Total annual benzene from facility process wastewater shall be determined by adding together the annual benzene quantity at the point of waste generation for each untreated process wastewater stream plus the annual benzene quantity exiting the treatment process for each process wastewater stream treated in accordance with the requirements of paragraph (c)(1)(i) of this section.

(ii) Each treated process wastewater stream identified in paragraph (d)(2)(i)of this section shall be managed and treated in accordance with paragraph (c)(1) of this section.

(iii) Each untreated process wastewater stream identified in paragraph (d)(2)(i) of this section is exempt from the requirements of paragraph (c)(1) of this section.

40 CFR Ch. I (7–1–23 Edition)

(e) As an alternative to the requirements specified in paragraphs (c) and (d) of this section, an owner or operator of a facility at which the total annual benzene quantity from facility waste is equal to or greater than 10 Mg/ yr (11 ton/yr) as determined in paragraph (a) of this section may elect to manage and treat the facility waste as follows:

(1) The owner or operator shall manage and treat facility waste with a flow-weighted annual average water content of less than 10 percent in accordance with the requirements of paragraph (c)(1) of this section; and

(2) The owner or operator shall manage and treat facility waste (including remediation and process unit turnaround waste) with a flow-weighted annual average water content of 10 percent or greater, on a volume basis as total water, and each waste stream that is mixed with water or wastes at any time such that the resulting mixture has an annual water content greater than 10 percent, in accordance with the following:

(i) The benzene quantity for the wastes described in paragraph (e)(2) of this section must be equal to or less than 6.0 Mg/yr (6.6 ton/yr), as determined in $\S61.355(k)$. Wastes as described in paragraph (e)(2) of this section that are transferred offsite shall be included in the determination of benzene quantity as provided in $\S61.355(k)$. The provisions of paragraph (f) of this section shall not apply to any owner or operator who elects to comply with the provisions of paragraph (e) of this section.

(ii) The determination of benzene quantity for each waste stream defined in paragraph (e)(2) of this section shall be made in accordance with 61.355(k).

(f) Rather than treating the waste onsite, an owner or operator may elect to comply with paragraph (c)(1)(i) of this section by transferring the waste offsite to another facility where the waste is treated in accordance with the requirements of paragraph (c)(1)(i) of this section. The owner or operator transferring the waste shall:

(1) Comply with the standards specified in \$61.343 through 61.347 of this subpart for each waste management unit that receives or manages the

waste prior to shipment of the waste offsite.

(2) Include with each offsite waste shipment a notice stating that the waste contains benzene which is required to be managed and treated in accordance with the provisions of this subpart.

(g) Compliance with this subpart will be determined by review of facility records and results from tests and inspections using methods and procedures specified in §61.355 of this subpart.

(h) Permission to use an alternative means of compliance to meet the requirements of §§ 61.342 through 61.352 of this subpart may be granted by the Administrator as provided in §61.353 of this subpart.

[55 FR 8346, Mar. 7, 1990, as amended at 58 FR 3095, Jan. 7, 1993; 65 FR 62159, 62160, Oct. 17, 2000]

§61.343 Standards: Tanks.

(a) Except as provided in paragraph (b) of this section and in §61.351, the owner or operator must meet the standards in paragraph (a)(1) or (2) of this section for each tank in which the waste stream is placed in accordance with §61.342 (c)(1)(ii). The standards in this section apply to the treatment and storage of the waste stream in a tank, including dewatering.

(1) The owner or operator shall install, operate, and maintain a fixedroof and closed-vent system that routes all organic vapors vented from the tank to a control device.

(i) The fixed-roof shall meet the following requirements:

(A) The cover and all openings (e.g., access hatches, sampling ports, and gauge wells) shall be designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods specified in §61.355(h) of this subpart.

(B) Each opening shall be maintained in a closed, sealed position (e.g., covered by a lid that is gasketed and latched) at all times that waste is in the tank except when it is necessary to use the opening for waste sampling or removal, or for equipment inspection, maintenance, or repair. (C) If the cover and closed-vent system operate such that the tank is maintained at a pressure less than atmospheric pressure, then paragraph (a)(1)(i)(B) of this section does not apply to any opening that meets all of thefollowing conditions:

(1) The purpose of the opening is to provide dilution air to reduce the explosion hazard;

(2) The opening is designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods specified in §61.355(h); and

(3) The pressure is monitored continuously to ensure that the pressure in the tank remains below atmospheric pressure.

(ii) The closed-vent system and control device shall be designed and operated in accordance with the requirements of §61.349 of this subpart.

(2) The owner or operator must install, operate, and maintain an enclosure and closed-vent system that routes all organic vapors vented from the tank, located inside the enclosure, to a control device in accordance with the requirements specified in paragraph (e) of this section.

(b) For a tank that meets all the conditions specified in paragraph (b)(1) of this section, the owner or operator may elect to comply with paragraph (b)(2) of this section as an alternative to the requirements specified in paragraph (a)(1) of this section.

(1) The waste managed in the tank complying with paragraph (b)(2) of this section shall meet all of the following conditions:

(i) Each waste stream managed in the tank must have a flow-weighted annual average water content less than or equal to 10 percent water, on a volume basis as total water.

(ii) The waste managed in the tank either:

(A) Has a maximum organic vapor pressure less than 5.2 kilopascals (kPa) (0.75 pounds per square inch (psi));

(B) Has a maximum organic vapor pressure less than 27.6 kPa (4.0 psi) and is managed in a tank having design capacity less than 151 m^3 (40,000 gal); or

(C) Has a maximum organic vapor pressure less than 76.6 kPa (11.1 psi) and is managed in a tank having a design capacity less than 75 m³ (20,000 gal).

(2) The owner or operator shall install, operate, and maintain a fixed roof as specified in paragraph (a)(1)(i).

(3) For each tank complying with paragraph (b) of this section, one or more devices which vent directly to the atmosphere may be used on the tank provided each device remains in a closed, sealed position during normal operations except when the device needs to open to prevent physical damage or permanent deformation of the tank or cover resulting from filling or emptying the tank, diurnal temperature changes, atmospheric pressure changes or malfunction of the unit in accordance with good engineering and safety practices for handling flammable, explosive, or other hazardous materials.

(c) Each fixed-roof, seal, access door, and all other openings shall be checked by visual inspection initially and quarterly thereafter to ensure that no cracks or gaps occur and that access doors and other openings are closed and gasketed properly.

(d) Except as provided in §61.350 of this subpart, when a broken seal or gasket or other problem is identified, or when detectable emissions are measured, first efforts at repair shall be made as soon as practicable, but not later than 45 calendar days after identification.

(e) Each owner or operator who controls air pollutant emissions by using an enclosure vented through a closedvent system to a control device must meet the requirements specified in paragraphs (e)(1) through (4) of this section.

(1) The tank must be located inside a total enclosure. The enclosure must be designed and operated in accordance with the criteria for a permanent total enclosure as specified in "Procedure T—Criteria for and Verification of a Permanent or Temporary Total Enclosure" in 40 CFR 52.741, appendix B. The enclosure may have permanent or temporary openings to allow worker access; passage of material into or out of the enclosure by conveyor, vehicles, or

40 CFR Ch. I (7–1–23 Edition)

other mechanical means; entry of permanent mechanical or electrical equipment; or direct airflow into the enclosure. The owner or operator must perform the verification procedure for the enclosure as specified in section 5.0 of Procedure T initially when the enclosure is first installed and, thereafter, annually. A facility that has conducted an initial compliance demonstration and that performs annual compliance demonstrations in accordance with the requirements for Tank Level 2 control requirements 40 CFR 264.1084(i) or 40 CFR 265(i) is not required to make repeat demonstrations of initial and continuous compliance for the purposes of this subpart.

(2) The enclosure must be vented through a closed-vent system to a control device that is designed and operated in accordance with the standards for control devices specified in §61.349.

(3) Safety devices, as defined in this subpart, may be installed and operated as necessary on any enclosure, closed-vent system, or control device used to comply with the requirements of paragraphs (e)(1) and (2) of this section.

(4) The closed-vent system must be designed and operated in accordance with the requirements of §61.349.

[55 FR 8346, Mar. 7, 1990, as amended at 55 FR
18331, May 2, 1990; 58 FR 3096, Jan. 7, 1993; 67
FR 68532, Nov. 12, 2002; 68 FR 6082, Feb. 6, 2003; 68 FR 67935, Dec. 4, 2003]

§61.344 Standards: Surface impoundments.

(a) The owner or operator shall meet the following standards for each surface impoundment in which waste is placed in accordance with §61.342(c)(1)(ii) of this subpart:

(1) The owner or operator shall install, operate, and maintain on each surface impoundment a cover (e.g., airsupported structure or rigid cover) and closed-vent system that routes all organic vapors vented from the surface impoundment to a control device.

(i) The cover shall meet the following requirements:

(A) The cover and all openings (e.g., access hatches, sampling ports, and gauge wells) shall be designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background,

initially and thereafter at least once per year by the methods specified in §61.355(h) of this subpart.

(B) Each opening shall be maintained in a closed, sealed position (e.g., covered by a lid that is gasketed and latched) at all times that waste is in the surface impoundment except when it is necessary to use the opening for waste sampling or removal, or for equipment inspection, maintenance, or repair.

(C) If the cover and closed-vent system operate such that the enclosure of the surface impoundment is maintained at a pressure less than atmospheric pressure, then paragraph (a)(1)(i)(B) of this section does not apply to any opening that meets all of the following conditions:

(1) The purpose of the opening is to provide dilution air to reduce the explosion hazard;

(2) The opening is designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods specified in §61.355(h) of this subpart; and

(3) The pressure is monitored continuously to ensure that the pressure in the enclosure of the surface impoundment remains below atmospheric pressure.

(D) The cover shall be used at all times that waste is placed in the surface impoundment except during removal of treatment residuals in accordance with 40 CFR 268.4 or closure of the surface impoundment in accordance with 40 CFR 264.228. (Note: the treatment residuals generated by these activities may be subject to the requirements of this part.)

(ii) The closed-vent system and control device shall be designed and operated in accordance with §61.349 of this subpart.

(b) Each cover seal, access hatch, and all other openings shall be checked by visual inspection initially and quarterly thereafter to ensure that no cracks or gaps occur and that access hatches and other openings are closed and gasketed properly.

(c) Except as provided in §61.350 of this subpart, when a broken seal or

gasket or other problem is identified, or when detectable emissions are measured, first efforts at repair shall be made as soon as practicable, but not later than 15 calendar days after identification.

[55 FR 8346, Mar. 7, 1990, as amended at 58 FR 3097, Jan. 7, 1993]

§61.345 Standards: Containers.

(a) The owner or operator shall meet the following standards for each container in which waste is placed in accordance with 61.342(c)(1)(ii) of this subpart:

(1) The owner or operator shall install, operate, and maintain a cover on each container used to handle, transfer, or store waste in accordance with the following requirements:

(i) The cover and all openings (e.g., bungs, hatches, and sampling ports) shall be designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, initially and thereafter at least once per year by the methods specified in §61.355(h) of this subpart.

(ii) Except as provided in paragraph (a)(4) of this section, each opening shall be maintained in a closed, sealed position (e.g., covered by a lid that is gasketed and latched) at all times that waste is in the container except when it is necessary to use the opening for waste loading, removal, inspection, or sampling.

(2) When a waste is transferred into a container by pumping, the owner or operator shall perform the transfer using a submerged fill pipe. The submerged fill pipe outlet shall extend to within two fill pipe diameters of the bottom of the container while the container is being loaded. During loading of the waste, the cover shall remain in place and all openings shall be maintained in a closed, sealed position except for those openings required for the submerged fill pipe, those openings required for venting of the container to prevent physical damage or permanent deformation of the container or cover, and any openings complying with paragraph (a)(4) of this section.

(3) Treatment of a waste in a container, including aeration, thermal or other treatment, must be performed by the owner or operator in a manner such that while the waste is being treated the container meets the standards specified in paragraphs (a)(3)(i) through (iii) of this section, except for covers and closed-vent systems that meet the requirements in paragraph (a)(4) of this section.

(i) The owner or operator must either:

(A) Vent the container inside a total enclosure which is exhausted through a closed-vent system to a control device in accordance with the requirements of paragraphs (a)(3)(ii)(A) and (B) of this section; or

(B) Vent the covered or closed container directly through a closed-vent system to a control device in accordance with the requirements of paragraphs (a)(3)(ii)(B) and (C) of this section.

(ii) The owner or operator must meet the following requirements, as applicable to the type of air emission control equipment selected by the owner or operator:

(A) The total enclosure must be designed and operated in accordance with the criteria for a permanent total enclosure as specified in section 5 of the "Procedure T-Criteria for and Verification of a Permanent or Temporary Total Enclosure" in 40 CFR 52.741, appendix B. The enclosure may have permanent or temporary openings to allow worker access; passage of containers through the enclosure by conveyor or other mechanical means; entry of permanent mechanical or electrical equipment; or direct airflow into the enclosure. The owner or operator must perform the verification procedure for the enclosure as specified in section 5.0 of "Procedure T-Criteria for and Verification of a Permanent or Temporary Total Enclosure" initially when the enclosure is first installed and, thereafter, annually. A facility that has conducted an initial compliance demonstration and that performs annual compliance demonstrations in accordance with the Container Level 3 control requirements in 40 CFR. 264.1086(e)(2)(i) 40 CFR or 265.1086(e)(2)(i) is not required to make repeat demonstrations of initial and continuous compliance for the purposes of this subpart.

40 CFR Ch. I (7–1–23 Edition)

(B) The closed-vent system and control device must be designed and operated in accordance with the requirements of §61.349.

(C) For a container cover, the cover and all openings (e.g., doors, hatches)must be designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, initially and thereafter at least once per year by the methods specified in §61.355(h).

(iii) Safety devices, as defined in this subpart, may be installed and operated as necessary on any container, enclosure, closed-vent system, or control device used to comply with the requirements of paragraph (a)(3)(i) of this section.

(4) If the cover and closed-vent system operate such that the container is maintained at a pressure less than atmospheric pressure, the owner or operator may operate the system with an opening that is not sealed and kept closed at all times if the following conditions are met:

(i) The purpose of the opening is to provide dilution air to reduce the explosion hazard;

(ii) The opening is designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by methods specified in §61.355(h); and

(iii) The pressure is monitored continuously to ensure that the pressure in the container remains below atmospheric pressure.

(b) Each cover and all openings shall be visually inspected initially and quarterly thereafter to ensure that they are closed and gasketed properly.

(c) Except as provided in §61.350 of this subpart, when a broken seal or gasket or other problem is identified, first efforts at repair shall be made as soon as practicable, but not later than 15 calendar days after identification.

[55 FR 8346, Mar. 7, 1990, as amended at 58 FR 3097, Jan. 7, 1993; 67 FR 68532, Nov. 12, 2002; 68 FR 67936, Dec. 4, 2003]

§61.346 Standards: Individual drain systems.

(a) Except as provided in paragraph (b) of this section, the owner or operator shall meet the following standards for each individual drain system in which waste is placed in accordance with 61.342(c)(1)(ii) of this subpart:

(1) The owner or operator shall install, operate, and maintain on each drain system opening a cover and closed-vent system that routes all organic vapors vented from the drain system to a control device.

(i) The cover shall meet the following requirements:

(A) The cover and all openings (e.g., access hatches, sampling ports) shall be designed to operate with no detactable emissions as indicated by an instrument reading of less than 500 ppmv above background, initially and thereafter at least once per year by the methods specified in §61.355(h) of this subpart.

(B) Each opening shall be maintained in a closed, sealed position (e.g., covered by a lid that is gasketed and latched) at all times that waste is in the drain system except when it is necessary to use the opening for waste sampling or removal, or for equipment inspection, maintenance, or repair.

(C) If the cover and closed-vent system operate such that the individual drain system is maintained at a pressure less than atmospheric pressure, then paragraph (a)(1)(i)(B) of this section does not apply to any opening that meets all of the following conditions:

(1) The purpose of the opening is to provide dilution air to reduce the explosion hazard;

(2) The opening is designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods specified in §61.355(h); and

(3) The pressure is monitored continuously to ensure that the pressure in the individual drain system remains below atmospheric pressure.

(ii) The closed-vent system and control device shall be designed and operated in accordance with §61.349 of this subpart. (2) Each cover seal, access hatch, and all other openings shall be checked by visual inspection initially and quarterly thereafter to ensure that no cracks or gaps occur and that access hatches and other openings are closed and gasketed properly.

(3) Except as provided in §61.350 of this subpart, when a broken seal or gasket or other problem is identified, or when detectable emissions are measured, first efforts at repair shall be made as soon as practicable, but not later than 15 calendar days after identification.

(b) As an alternative to complying with paragraph (a) of this section, an owner or operator may elect to comply with the following requirements:

(1) Each drain shall be equipped with water seal controls or a tightly sealed cap or plug.

(2) Each junction box shall be equipped with a cover and may have a vent pipe. The vent pipe shall be at least 90 cm (3 ft) in length and shall not exceed 10.2 cm (4 in) in diameter.

(i) Junction box covers shall have a tight seal around the edge and shall be kept in place at all times, except during inspection and maintenance.

(ii) One of the following methods shall be used to control emissions from the junction box vent pipe to the atmosphere:

(A) Equip the junction box with a system to prevent the flow of organic vapors from the junction box vent pipe to the atmosphere during normal operation. An example of such a system includes use of water seal controls on the junction box. A flow indicator shall be installed, operated, and maintained on each junction box vent pipe to ensure that organic vapors are not vented from the junction box to the atmosphere during normal operation.

(B) Connect the junction box vent pipe to a closed-vent system and control device in accordance with §61.349 of this subpart.

(3) Each sewer line shall not be open to the atmosphere and shall be covered or enclosed in a manner so as to have no visual gaps or cracks in joints, seals, or other emission interfaces.

(4) Equipment installed in accordance with paragraphs (b)(1), (b)(2), or

(b)(3) of this section shall be inspected as follows:

(i) Each drain using water seal controls shall be checked by visual or physical inspection initially and thereafter quarterly for indications of low water levels or other conditions that would reduce the effectiveness of water seal controls.

(ii) Each drain using a tightly sealed cap or plug shall be visually inspected initially and thereafter quarterly to ensure caps or plugs are in place and properly installed.

(iii) Each junction box shall be visually inspected initially and thereafter quarterly to ensure that the cover is in place and to ensure that the cover has a tight seal around the edge.

(iv) The unburied portion of each sewer line shall be visually inspected initially and thereafter quarterly for indication of cracks, gaps, or other problems that could result in benzene emissions.

(5) Except as provided in §61.350 of this subpart, when a broken seal, gap, crack or other problem is identified, first efforts at repair shall be made as soon as practicable, but not later than 15 calendar days after identification.

[55 FR 8346, Mar. 7, 1990, as amended at 55 FR 37231, Sept. 10, 1990; 58 FR 3097, Jan. 7, 1993]

§61.347 Standards: Oil-water separators.

(a) Except as provided in §61.352 of this subpart, the owner or operator shall meet the following standards for each oil-water separator in which waste is placed in accordance with §61.342(c)(1)(ii) of this subpart:

(1) The owner or operator shall install, operate, and maintain a fixedroof and closed-vent system that routes all organic vapors vented from the oil-water separator to a control device.

(i) The fixed-roof shall meet the following requirements:

(A) The cover and all openings (e.g., access hatches, sampling ports, and gauge wells) shall be designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods specified in §61.355(h) of this subpart. 40 CFR Ch. I (7–1–23 Edition)

(B) Each opening shall be maintained in a closed, sealed position (e.g., covered by a lid that is gasketed and latched) at all times that waste is in the oil-water separator except when it is necessary to use the opening for waste sampling or removal, or for equipment inspection, maintenance, or repair.

(C) If the cover and closed-vent system operate such that the oil-water separator is maintained at a pressure less than atmospheric pressure, then paragraph (a)(1)(i)(B) of this section does not apply to any opening that meets all of the following conditions:

(1) The purpose of the opening is to provide dilution air to reduce the explosion hazard;

(2) The opening is designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods specified in §61.355(h); and

(3) The pressure is monitored continuously to ensure that the pressure in the oil-water separator remains below atmospheric pressure.

(ii) The closed-vent system and control device shall be designed and operated in accordance with the requirements of §61.349 of this subpart.

(b) Each cover seal, access hatch, and all other openings shall be checked by visual inspection initially and quarterly thereafter to ensure that no cracks or gaps occur between the cover and oil-water separator wall and that access hatches and other openings are closed and gasketed properly.

(c) Except as provided in §61.350 of this subpart, when a broken seal or gasket or other problem is identified, or when detectable emissions are measured, first efforts at repair shall be made as soon as practicable, but not later than 15 calendar days after identification.

[55 FR 8346, Mar. 7, 1990, as amended at 58 FR 3098, Jan. 7, 1993]

§61.348 Standards: Treatment processes.

(a) Except as provided in paragraph (a)(5) of this section, the owner or operator shall treat the waste stream in accordance with the following requirements:

(1) The owner or operator shall design, install, operate, and maintain a treatment process that either:

(i) Removes benzene from the waste stream to a level less than 10 parts per million by weight (ppmw) on a flowweighted annual average basis,

(ii) Removes benzene from the waste stream by 99 percent or more on a mass basis, or

(iii) Destroys benzene in the waste stream by incinerating the waste in a combustion unit that achieves a destruction efficiency of 99 percent or greater for benzene.

(2) Each treatment process complying with paragraphs (a)(1)(i) or (a)(1)(ii) of this section shall be designed and operated in accordance with the appropriate waste management unit standards specified in §§ 61.343 through 61.347 of this subpart. For example, if a treatment process is a tank, then the owner or operator shall comply with §61.343 of this subpart.

(3) For the purpose of complying with the requirements specified in paragraph (a)(1)(i) of this section, the intentional or unintentional reduction in the benzene concentration of a waste stream by dilution of the waste stream with other wastes or materials is not allowed.

(4) An owner or operator may aggregate or mix together individual waste streams to create a combined waste stream for the purpose of facilitating treatment of waste to comply with the requirements of paragraph (a)(1) of this section except as provided in paragraph (a)(5) of this section.

(5) If an owner or operator aggregates or mixes any combination of process wastewater, product tank drawdown, or landfill leachate subject to §61.342(c)(1) of this subpart together with other waste streams to create a combined waste stream for the purpose of facilitating management or treatment of waste in a wastewater treatment system, then the wastewater treatment system shall be operated in accordance with paragraph (b) of this section. These provisions apply to above-ground wastewater treatment systems as well as those that are at or below ground level.

(b) Except for facilities complying with §61.342(e), the owner or operator that aggregates or mixes individual waste streams as defined in paragraph (a)(5) of this section for management and treatment in a wastewater treatment system shall comply with the following requirements:

(1) The owner or operator shall design and operate each waste management unit that comprises the wastewater treatment system in accordance with the appropriate standards specified in \S 61.343 through 61.347 of this subpart.

(2) The provisions of paragraph (b)(1) of this section do not apply to any waste management unit that the owner or operator demonstrates to meet the following conditions initially and, thereafter, at least once per year:

(i) The benzene content of each waste stream entering the waste management unit is less than 10 ppmw on a flow-weighted annual average basis as determined by the procedures specified in §61.355(c) of this subpart; and

(ii) The total annual benzene quantity contained in all waste streams managed or treated in exempt waste management units comprising the facility wastewater treatment systems is less than 1 Mg/yr (1.1 ton/yr). For this determination, total annual benzene quantity shall be calculated as follows:

(A) The total annual benzene quantity shall be calculated as the sum of the individual benzene quantities determined at each location where a waste stream first enters an exempt waste management unit. The benzene quantity discharged from an exempt waste management unit shall not be included in this calculation.

(B) The annual benzene quantity in a waste stream managed or treated in an enhanced biodegradation unit shall not be included in the calculation of the total annual benzene quantity, if the enhanced biodegradation unit is the first exempt unit in which the waste is managed or treated. A unit shall be considered enhanced biodegradation if it is a suspended-growth process that generates biomass, uses recycled biomass, and periodically removes biomass from the process. An enhanced biodegradation unit typically operates at a food-to-microorganism ratio in the range of 0.05 to 1.0 kg of biological oxygen demand per kg of biomass per day, a mixed liquor suspended solids ratio in the range of 1 to 8 grams per liter (0.008 to 0.7 pounds per liter), and a residence time in the range of 3 to 36 hours.

(c) The owner and operator shall demonstrate that each treatment process or wastewater treatment system unit, except as provided in paragraph (d) of this section, achieves the appropriate conditions specified in paragraphs (a) or (b) of this section in accordance with the following requirements:

(1) Engineering calculations in accordance with requirements specified in §61.356(e) of this subpart; or

(2) Performance tests conducted using the test methods and procedures that meet the requirements specified in §61.355 of this subpart.

(d) A treatment process or waste stream is in compliance with the requirements of this subpart and exempt from the requirements of paragraph (c) of this section provided that the owner or operator documents that the treatment process or waste stream is in compliance with other regulatory requirements as follows:

(1) The treatment process is a hazardous waste incinerator for which the owner or operator has been issued a final permit under 40 CFR part 270 and complies with the requirements of 40 CFR part 264, subpart O;

(2) The treatment process is an industrial furnace or boiler burning hazardous waste for energy recovery for which the owner or operator has been issued a final permit under 40 CFR part 270 and complies with the requirements of 40 CFR part 266, subpart D;

(3) The waste stream is treated by a means or to a level that meets benzenespecific treatment standards in accordance with the Land Disposal Restrictions under 40 CFR part 268, and the treatment process is designed and operated with a closed-vent system and control device meeting the requirements of §61.349 of this subpart; 40 CFR Ch. I (7–1–23 Edition)

(4) The waste stream is treated by a means or to a level that meets benzenespecific effluent limitations or performance standards in accordance with the Effluent Guidelines and Standards under 40 CFR parts 401-464, and the treatment process is designed and operated with a closed-vent system and control device meeting the requirements of §61.349 of this subpart; or

(5) The waste stream is discharged to an underground injection well for which the owner or operator has been issued a final permit under 40 CFR part 270 and complies with the requirements of 40 CFR part 122.

(e) Except as specified in paragraph (e)(3) of this section, if the treatment process or wastewater treatment system unit has any openings (e.g., access doors, hatches, etc.), all such openings shall be sealed (e.g., gasketed, latched, etc.) and kept closed at all times when waste is being treated, except during inspection and maintenance.

(1) Each seal, access door, and all other openings shall be checked by visual inspections initially and quarterly thereafter to ensure that no cracks or gaps occur and that openings are closed and gasketed properly.

(2) Except as provided in §61.350 of this subpart, when a broken seal or gasket or other problem is identified, first efforts at repair shall be made as soon as practicable, but not later than 15 calendar days after identification.

(3) If the cover and closed-vent system operate such that the treatment process and wastewater treatment system unit are maintained at a pressure less than atmospheric pressure, the owner or operator may operate the system with an opening that is not sealed and kept closed at all times if the following conditions are met:

(i) The purpose of the opening is to provide dilution air to reduce the explosion hazard;

(ii) The opening is designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods specified in §61.355(h); and

(iii) The pressure is monitored continuously to ensure that the pressure

in the treatment process and wastewater treatment system unit remain below atmospheric pressure.

(f) Except for treatment processes complying with paragraph (d) of this section, the Administrator may request at any time an owner or operator demonstrate that a treatment process or wastewater treatment system unit meets the applicable requirements specified in paragraphs (a) or (b) of this section by conducting a performance test using the test methods and procedures as required in §61.355 of this subpart.

(g) The owner or operator of a treatment process or wastewater treatment system unit that is used to comply with the provisions of this section shall monitor the unit in accordance with the applicable requirements in §61.354 of this subpart.

[55 FR 8346, Mar. 7, 1990, as amended at 55 FR 37231, Sept. 10, 1990; 58 FR 3098, Jan. 7, 1993;
65 FR 62160, Oct. 17, 2000]

§61.349 Standards: Closed-vent systems and control devices.

(a) For each closed-vent system and control device used to comply with standards in accordance with §§61.343 through 61.348 of this subpart, the owner or operator shall properly design, install, operate, and maintain the closed-vent system and control device in accordance with the following requirements:

(1) The closed-vent system shall:

(i) Be designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods specified in §61.355(h) of this subpart.

(ii) Vent systems that contain any bypass line that could divert the vent stream away from a control device used to comply with the provisions of this subpart shall install, maintain, and operate according to the manufacturer's specifications a flow indicator that provides a record of vent stream flow away from the control device at least once every 15 minutes, except as provided in paragraph (a)(1)(ii)(B) of this section.

(A) The flow indicator shall be installed at the entrance to any bypass line that could divert the vent stream away from the control device to the atmosphere.

(B) Where the bypass line valve is secured in the closed position with a carseal or a lock-and-key type configuration, a flow indicator is not required.

(iii) All gauging and sampling devices shall be gas-tight except when gauging or sampling is taking place.

(iv) For each closed-vent system complying with paragraph (a) of this section, one or more devices which vent directly to the atmosphere may be used on the closed-vent system provided each device remains in a closed, sealed position during normal operations except when the device needs to open to prevent physical damage or permanent deformation of the closedvent system resulting from malfunction of the unit in accordance with good engineering and safety practices for handling flammable, explosive, or other hazardous materials.

(2) The control device shall be designed and operated in accordance with the following conditions:

(i) An enclosed combustion device (e.g., a vapor incinerator, boiler, or process heater) shall meet one of the following conditions:

(A) Reduce the organic emissions vented to it by 95 weight percent or greater;

(B) Achieve a total organic compound concentration of 20 ppmv (as the sum of the concentrations for individual compounds using Method 18) on a dry basis corrected to 3 percent oxygen; or

(C) Provide a minimum residence time of 0.5 seconds at a minimum temperature of 760 °C (1,400 °F). If a boiler or process heater issued as the control device, then the vent stream shall be introduced into the flame zone of the boiler or process heater.

(ii) A vapor recovery system (e.g., a carbon adsorption system or a condenser) shall recover or control the organic emissions vented to it with an efficiency of 95 weight percent or greater, or shall recover or control the benzene emissions vented to it with an efficiency of 98 weight percent or greater.

(iii) A flare shall comply with the requirements of 40 CFR 60.18.

§61.349

(iv) A control device other than those described in paragraphs (a)(2) (i) through (iii) of this section may be used provided that the following conditions are met:

(A) The device shall recover or control the organic emissions vented to it with an efficiency of 95 weight percent or greater, or shall recover or control the benzene emissions vented to it with an efficiency of 98 weight percent or greater.

(B) The owner or operator shall develop test data and design information that documents the control device will achieve an emission control efficiency of either 95 percent or greater for organic compounds or 98 percent or greater for benzene.

(C) The owner or operator shall identify:

(1) The critical operating parameters that affect the emission control performance of the device;

(2) The range of values of these operating parameters that ensure the emission control efficiency specified in paragraph (a)(2)(iv)(A) of this section is maintained during operation of the device; and

(3) How these operating parameters will be monitored to ensure the proper operation and maintenance of the device.

(D) The owner or operator shall submit the information and data specified in paragraphs (a)(2)(iv) (B) and (C) of this section to the Administrator prior to operation of the alternative control device.

(E) The Administrator will determine, based on the information submitted under paragraph (a)(2)(iv)(D) of this section, if the control device subject to paragraph (a)(2)(iv) of this section meets the requirements of §61.349. The control device subject to paragraph (a)(2)(iv) of this section may be operated prior to receiving approval from the Administrator. However, if the Administrator determines that the control device does not meet the requirements of §61.349, the facility may be subject to enforcement action beginning from the time the control device began operation.

(b) Each closed-vent system and control device used to comply with this subpart shall be operated at all times 40 CFR Ch. I (7–1–23 Edition)

when waste is placed in the waste management unit vented to the control device except when maintenance or repair of the waste management unit cannot be completed without a shutdown of the control device.

(c) An owner and operator shall demonstrate that each control device, except for a flare, achieves the appropriate conditions specified in paragraph (a)(2) of this section by using one of the following methods:

(1) Engineering calculations in accordance with requirements specified in §61.356(f) of this subpart; or

(2) Performance tests conducted using the test methods and procedures that meet the requirements specified in §61.355 of this subpart.

(d) An owner or operator shall demonstrate compliance of each flare in accordance with paragraph (a)(2)(iii) of this section.

(e) The Administrator may request at any time an owner or operator demonstrate that a control device meets the applicable conditions specified in paragraph (a)(2) of this section by conducting a performance test using the test methods and procedures as required in §61.355, and for control devices subject to paragraph (a)(2)(iv) of this section, the Administrator may specify alternative test methods and procedures, as appropriate.

(f) Each closed-vent system and control device shall be visually inspected initially and quarterly thereafter. The visual inspection shall include inspection of ductwork and piping and connections to covers and control devices for evidence of visable defects such as holes in ductwork or piping and loose connections.

(g) Except as provided in §61.350 of this subpart, if visible defects are observed during an inspection, or if other problems are identified, or if detectable emissions are measured, a first effort to repair the closed-vent system and control device shall be made as soon as practicable but no later than 5 calendar days after detection. Repair shall be completed no later than 15 calendar days after the emissions are detected or the visible defect is observed.

(h) The owner or operator of a control device that is used to comply with the provisions of this section shall

monitor the control device in accordance with §61.354(c) of this subpart.

[55 FR 8346, Mar. 7, 1990; 55 FR 12444, Apr. 3, 1990, as amended at 55 FR 37231, Sept. 10, 1990; 58 FR 3098, Jan. 7, 1993; 65 FR 62160, Oct. 17, 2000]

§61.350 Standards: Delay of repair.

(a) Delay of repair of facilities or units that are subject to the provisions of this subpart will be allowed if the repair is technically impossible without a complete or partial facility or unit shutdown.

(b) Repair of such equipment shall occur before the end of the next facility or unit shutdown.

§61.351 Alternative standards for tanks.

(a) As an alternative to the standards for tanks specified in §61.343 of this subpart, an owner or operator may elect to comply with one of the following:

(1) A fixed roof and internal floating roof meeting the requirements in 40 CFR 60.112b(a)(1);

(2) An external floating roof meeting the requirements of 40 CFR 60.112b (a)(2); or

(3) An alternative means of emission limitation as described in 40 CFR 60.114b.

(b) If an owner or operator elects to comply with the provisions of this section, then the owner or operator is exempt from the provisions of §61.343 of this subpart applicable to the same facilities.

[55 FR 8346, Mar. 7, 1990, as amended at 55 FR 37231, Sept. 10, 1990]

§61.352 Alternative standards for oilwater separators.

(a) As an alternative to the standards for oil-water separators specified in §61.347 of this subpart, an owner or operator may elect to comply with one of the following:

(1) A floating roof meeting the requirements in 40 CFR 60.693-2(a); or

(2) An alternative means of emission limitation as described in 40 CFR 60.694.

(b) For portions of the oil-water separator where it is infeasible to construct and operate a floating roof, such as over the weir mechanism, a fixed roof vented to a vapor control device that meets the requirements in §§61.347 and 61.349 of this subpart shall be installed and operated.

(c) Except as provided in paragraph (b) of this section, if an owner or operator elects to comply with the provisions of this section, then the owner or operator is exempt from the provisions in §61.347 of this subpart applicable to the same facilities.

§61.353 Alternative means of emission limitation.

(a) If, in the Administrator's judgment, an alternative means of emission limitation will achieve a reduction in benzene emissions at least equivalent to the reduction in benzene emissions from the source achieved by the applicable design, equipment, work practice, or operational requirements in §§61.342 through 61.349, the Administrator will publish in the FEDERAL REGISTER a notice permitting the use of the alternative means for purposes of compliance with that requirement. The notice may condition the permission on requirements related to the operation and maintenance of the alternative means.

(b) Any notice under paragraph (a) of this section shall be published only after public notice and an opportunity for a hearing.

(c) Any person seeking permission under this section shall collect, verify, and submit to the Administrator information showing that the alternative means achieves equivalent emission reductions.

 $[55\ {\rm FR}\ 8346,\ {\rm Mar.}\ 7,\ 1990,\ {\rm as}\ {\rm amended}\ {\rm at}\ 58\ {\rm FR}\ 3099,\ {\rm Jan.}\ 7,\ 1993]$

§61.354 Monitoring of operations.

(a) Except for a treatment process or waste stream complying with §61.348(d), the owner or operator shall monitor each treatment process or wastewater treatment system unit to ensure the unit is properly operated and maintained by one of the following monitoring procedures:

(1) Measure the benzene concentration of the waste stream exiting the treatment process complying with $\S61.348(a)(1)(i)$ at least once per month by collecting and analyzing one or more samples using the procedures specified in §61.355(c)(3).

(2) Install, calibrate, operate, and maintain according to manufacturer's specifications equipment to continuously monitor and record a process parameter (or parameters) for the treatment process or wastewater treatment system unit that indicates proper system operation. The owner or operator shall inspect at least once each operating day the data recorded by the monitoring equipment (e.g., temperature monitor or flow indicator) to ensure that the unit is operating properly.

(b) If an owner or operator complies with the requirements of §61.348(b), then the owner or operator shall monitor each wastewater treatment system to ensure the unit is properly operated and maintained by the appropriate monitoring procedure as follows:

(1) For the first exempt waste management unit in each waste treatment train, other than an enhanced biodegradation unit, measure the flow rate, using the procedures of (61.355(b)), and the benzene concentration of each waste stream entering the unit at least once per month by collecting and analyzing one or more samples using the procedures specified in (61.355(c))

(2) For each enhanced biodegradation unit that is the first exempt waste management unit in a treatment train, measure the benzene concentration of each waste stream entering the unit at least once per month by collecting and analyzing one or more samples using the procedures specified in §61.355(c)(3).

(c) An owner or operator subject to the requirements in §61.349 of this subpart shall install, calibrate, maintain, and operate according to the manufacturer's specifications a device to continuously monitor the control device operation as specified in the following paragraphs, unless alternative monitoring procedures or requirements are approved for that facility by the Ad-ministrator. The owner or operator shall inspect at least once each operating day the data recorded by the monitoring equipment (e.g., temperature monitor or flow indicator) to ensure that the control device is operating properly.

40 CFR Ch. I (7–1–23 Edition)

(1) For a thermal vapor incinerator, a temperature monitoring device equipped with a continuous recorder. The device shall have an accuracy of ± 1 percent of the temperature being monitored in °C or ± 0.5 °C, whichever is greater. The temperature sensor shall be installed at a representative location in the combustion chamber.

(2) For a catalytic vapor incinerator. ิล temperature monitoring device equipped with a continuous recorder. The device shall be capable of monitoring temperature at two locations, and have an accuracy of ±1 percent of the temperature being monitored in °C or ±0.5 °C, whichever is greater. One temperature sensor shall be installed in the vent stream at the nearest feasible point to the catalyst bed inlet and a second temperature sensor shall be installed in the vent stream at the nearest feasible point to the catalyst bed outlet.

(3) For a flare, a monitoring device in accordance with 40 CFR 60.18(f)(2) equipped with a continuous recorder.

(4) For a boiler or process heater having a design heat input capacity less than 44 MW (150×10^6 BTU/hr), a temperature monitoring device equipped with a continuous recorder. The device shall have an accuracy of ±1 percent of the temperature being monitored in °C or ±0.5 °C, whichever is greater. The temperature sensor shall be installed at a representative location in the combustion chamber.

(5) For a boiler or process heater having a design heat input capacity greater than or equal to 44 MW (150×10^6 BTU/hr), a monitoring device equipped with a continuous recorder to measure a parameter(s) that indicates good combustion operating practices are being used.

(6) For a condenser, either:

(i) A monitoring device equipped with a continuous recorder to measure either the concentration level of the organic compounds or the concentration level of benzene in the exhaust vent stream from the condenser; or

(ii) A temperature monitoring device equipped with a continuous recorder. The device shall be capable of monitoring temperature at two locations, and have an accuracy of ± 1 percent of the temperature being monitored in °C

or ± 0.5 °C, whichever is greater. One temperature sensor shall be installed at a location in the exhaust stream from the condenser, and a second temperature sensor shall be installed at a location in the coolant fluid exiting the condenser.

(7) For a carbon adsorption system that regenerates the carbon bed directly in the control device such as a fixed-bed carbon adsorber, either:

(i) A monitoring device equipped with a continuous recorder to measure either the concentration level of the organic compounds or the benzene concentration level in the exhaust vent stream from the carbon bed; or

(ii) A monitoring device equipped with a continuous recorder to measure a parameter that indicates the carbon bed is regenerated on a regular, predetermined time cycle.

(8) For a vapor recovery system other than a condenser or carbon adsorption system, a monitoring device equipped with a continuous recorder to measure either the concentration level of the organic compounds or the benzene concentration level in the exhaust vent stream from the control device.

(9) For a control device subject to the requirements of (1.349(a)(2))(iv), devices to monitor the parameters as specified in (1.349(a)(2))(iv)(C).

(d) For a carbon adsorption system that does not regenerate the carbon bed directly on site in the control device (e.g., a carbon canister), either the concentration level of the organic compounds or the concentration level of benzene in the exhaust vent stream from the carbon adsorption system shall be monitored on a regular schedule, and the existing carbon shall be replaced with fresh carbon immediately when carbon breakthrough is indicated. The device shall be monitored on a daily basis or at intervals no greater than 20 percent of the design carbon replacement interval, whichever is greater. As an alternative to conducting this monitoring, an owner or operator may replace the carbon in the carbon adsorption system with fresh carbon at a regular predetermined time interval that is less than the carbon replacement interval that is determined by the maximum design flow rate and either the organic concentration or the benzene concentration in the gas stream vented to the carbon adsorption system.

(e) An alternative operation or process parameter may be monitored if it can be demonstrated that another parameter will ensure that the control device is operated in conformance with these standards and the control device's design specifications.

(f) Owners or operators using a closed-vent system that contains any bypass line that could divert a vent stream from a control device used to comply with the provisions of this sub-part shall do the following:

(1) Visually inspect the bypass line valve at least once every month, checking the position of the valve and the condition of the car-seal or closure mechanism required under $\S 61.349(a)(1)(ii)$ to ensure that the valve is maintained in the closed position and the vent stream is not diverted through the bypass line.

(2) Visually inspect the readings from each flow monitoring device required by §61.349(a)(1)(ii) at least once each operating day to check that vapors are being routed to the control device as required.

(g) Each owner or operator who uses a system for emission control that is maintained at a pressure less than atmospheric pressure with openings to provide dilution air shall install, calibrate, maintain, and operate according to the manufacturer's specifications a device equipped with a continuous recorder to monitor the pressure in the unit to ensure that it is less than atmospheric pressure.

[55 FR 8346, Mar. 7, 1990, as amended at 58 FR 3099, Jan. 7, 1993; 65 FR 62160, Oct. 17, 2000]

§61.355 Test methods, procedures, and compliance provisions.

(a) An owner or operator shall determine the total annual benzene quantity from facility waste by the following procedure:

(1) For each waste stream subject to this subpart having a flow-weighted annual average water content greater than 10 percent water, on a volume basis as total water, or is mixed with water or other wastes at any time and the resulting mixture has an annual average water content greater than 10

40 CFR Ch. I (7–1–23 Edition)

percent as specified in §61.342(a), the owner or operator shall:

(i) Determine the annual waste quantity for each waste stream using the procedures specified in paragraph (b) of this section.

(ii) Determine the flow-weighted annual average benzene concentration for each waste stream using the procedures specified in paragraph (c) of this section.

(iii) Calculate the annual benzene quantity for each waste stream by multiplying the annual waste quantity of the waste stream times the flowweighted annual average benzene concentration.

(2) Total annual benzene quantity from facility waste is calculated by adding together the annual benzene quantity for each waste stream generated during the year and the annual benzene quantity for each process unit turnaround waste annualized according to paragraph (b)(4) of this section.

(3) If the total annual benzene quantity from facility waste is equal to or greater than 10 Mg/yr (11 ton/yr), then the owner or operator shall comply with the requirements of §61.342 (c), (d), or (e).

(4) If the total annual benzene quantity from facility waste is less than 10 Mg/yr (11 ton/yr) but is equal to or greater than 1 Mg/yr (1.1 ton/yr), then the owner or operator shall:

(i) Comply with the record keeping requirements of 61.356 and reporting requirements of 61.357 of this subpart; and

(ii) Repeat the determination of total annual benzene quantity from facility waste at least once per year and whenever there is a change in the process generating the waste that could cause the total annual benzene quantity from facility waste to increase to 10 Mg/yr (11 ton/yr) or more.

(5) If the total annual benzene quantity from facility waste is less than 1 Mg/yr (1.1 ton/yr), then the owner or operator shall:

(i) Comply with the recordkeeping requirements of §61.356 and reporting requirements of §61.357 of this subpart; and

(ii) Repeat the determination of total annual benzene quantity from facility waste whenever there is a change in the process generating the waste that could cause the total annual benzene quantity from facility waste to increase to 1 Mg/yr (1.1 ton/yr) or more.

(6) The benzene quantity in a waste stream that is generated less than one time per year, except as provided for process unit turnaround waste in paragraph (b)(4) of this section, shall be included in the determination of total annual benzene quantity from facility waste for the year in which the waste is generated unless the waste stream is otherwise excluded from the determination of total annual benzene quantity from facility waste in accordance with paragraphs (a) through (c) of this section. The benzene quantity in this waste stream shall not be annualized or averaged over the time interval between the activities that resulted in generation of the waste, for purposes of determining the total annual benzene quantity from facility waste.

(b) For purposes of the calculation required by paragraph (a) of this section, an owner or operator shall determine the annual waste quantity at the point of waste generation, unless otherwise provided in paragraphs (b) (1), (2), (3), and (4) of this section, by one of the methods given in paragraphs (b) (5) through (7) of this section.

(1) The determination of annual waste quantity for sour water streams that are processed in sour water strippers shall be made at the point that the water exits the sour water stripper.

(2) The determination of annual waste quantity for wastes at coke byproduct plants subject to and complying with the control requirements of §61.132, 61.133, 61.134, or 61.139 of subpart L of this part shall be made at the location that the waste stream exits the process unit component or waste management unit controlled by that subpart or at the exit of the ammonia still, provided that the following conditions are met:

(i) The transfer of wastes between units complying with the control requirements of subpart L of this part, process units, and the ammonia still is made through hard piping or other enclosed system.

(ii) The ammonia still meets the definition of a sour water stripper in §61.341.

§61.355

(3) The determination of annual waste quantity for wastes that are received at hazardous waste treatment, storage, or disposal facilities from offsite shall be made at the point where the waste enters the hazardous waste treatment, storage, or disposal facility.

(4) The determination of annual waste quantity for each process unit turnaround waste generated only at 2 year or greater intervals, may be made by dividing the total quantity of waste generated during the most recent process unit turnaround by the time period (in the nearest tenth of a year) between the turnaround resulting in generation of the waste and the most recent preceding process turnaround for the unit. The resulting annual waste quantity shall be included in the calculation of the annual benzene quantity as provided in paragraph (a)(1)(iii) of this section for the year in which the turnaround occurs and for each subsequent year until the unit undergoes the next process turnaround. For estimates of total annual benzene quantity as specified in the 90-day report, required under §61.357(a)(1), the owner or operator shall estimate the waste quantity generated during the most recent turnaround, and the time period between turnarounds in accordance with good engineering practices. If the owner or operator chooses not to annualize process unit turnaround waste, as specified in this paragraph, then the process unit turnaround waste quantity shall be included in the calculation of the annual benzene quantity for the year in which the turnaround occurs.

(5) Select the highest annual quantity of waste managed from historical records representing the most recent 5 years of operation or, if the facility has been in service for less than 5 years but at least 1 year, from historical records representing the total operating life of the facility;

(6) Use the maximum design capacity of the waste management unit; or

(7) Use measurements that are representative of maximum waste generation rates.

(c) For the purposes of the calculation required by §§61.355(a) of this subpart, an owner or operator shall determine the flow-weighted annual average ben-zene concentration in a manner that meets the requirements given in paragraph (c)(1) of this section using either of the methods given in paragraphs (c)(2) and (c)(3) of this section.

(1) The determination of flow-weighted annual average benzene concentration shall meet all of the following criteria:

(i) The determination shall be made at the point of waste generation except for the specific cases given in paragraphs (c)(1)(i)(A) through (D) of this section.

(A) The determination for sour water streams that are processed in sour water strippers shall be made at the point that the water exits the sour water stripper.

(B) The determination for wastes at coke by-product plants subject to and complying with the control requirements of §61.132, 61.133, 61.134, or 61.139 of subpart L of this part shall be made at the location that the waste stream exits the process unit component or waste management unit controlled by that subpart or at the exit of the ammonia still, provided that the following conditions are met:

(1) The transfer of wastes between units complying with the control requirements of subpart L of this part, process units, and the ammonia still is made through hard piping or other enclosed system.

(2) The ammonia still meets the definition of a sour water stripper in §61.341.

(C) The determination for wastes that are received from offsite shall be made at the point where the waste enters the hazardous waste treatment, storage, or disposal facility.

(D) The determination of flowweighted annual average benzene concentration for process unit turnaround waste shall be made using either of the methods given in paragraph (c)(2) or (c)(3) of this section. The resulting flow-weighted annual average benzene concentration shall be included in the calculation of annual benzene quantity as provided in paragraph (a)(1)(iii) of this section for the year in which the turnaround occurs and for each subsequent year until the unit undergoes the next process unit turnaround.

(ii) Volatilization of the benzene by exposure to air shall not be used in the

40 CFR Ch. I (7–1–23 Edition)

determination to reduce the benzene concentration.

(iii) Mixing or diluting the waste stream with other wastes or other materials shall not be used in the determination—to reduce the benzene concentration.

(iv) The determination shall be made prior to any treatment of the waste that removes benzene, except as specified in paragraphs (c)(1)(i)(A) through (D) of this section.

(v) For wastes with multiple phases, the determination shall provide the weighted-average benzene concentration based on the benzene concentration in each phase of the waste and the relative proportion of the phases.

(2) Knowledge of the waste. The owner or operator shall provide sufficient information to document the flowweighted annual average benzene concentration of each waste stream. Examples of information that could constitute knowledge include material balances, records of chemicals purchases, or previous test results provided the results are still relevant to the current waste stream conditions. If test data are used, then the owner or operator shall provide documentation describing the testing protocol and the means by which sampling variability and analytical variability were accounted for in the determination of the flow-weighted annual average benzene concentration for the waste stream. When an owner or operator and the Administrator do not agree on determinations of the flow-weighted annual average benzene concentration based on knowledge of the waste, the procedures under paragraph (c)(3) of this section shall be used to resolve the disagreement.

(3) Measurements of the benzene concentration in the waste stream in accordance with the following procedures:

(i) Collect a minimum of three representative samples from each waste stream. Where feasible, samples shall be taken from an enclosed pipe prior to the waste being exposed to the atmosphere.

(ii) For waste in enclosed pipes, the following procedures shall be used:

(A) Samples shall be collected prior to the waste being exposed to the atmosphere in order to minimize the loss of benzene prior to sampling.

(B) A static mixer shall be installed in the process line or in a by-pass line unless the owner or operator demonstrates that installation of a static mixer in the line is not necessary to accurately determine the benzene concentration of the waste stream.

(C) The sampling tap shall be located within two pipe diameters of the static mixer outlet.

(D) Prior to the initiation of sampling, sample lines and cooling coil shall be purged with at least four volumes of waste.

(E) After purging, the sample flow shall be directed to a sample container and the tip of the sampling tube shall be kept below the surface of the waste during sampling to minimize contact with the atmosphere.

(F) Samples shall be collected at a flow rate such that the cooling coil is able to maintain a waste temperature less than 10 $^{\circ}$ C (50 $^{\circ}$ F).

(G) After filling, the sample container shall be capped immediately (within 5 seconds) to leave a minimum headspace in the container.

(H) The sample containers shall immediately be cooled and maintained at a temperature below 10 $^{\circ}$ C (50 $^{\circ}$ F) for transfer to the laboratory.

(iii) When sampling from an enclosed pipe is not feasible, a minimum of three representative samples shall be collected in a manner to minimize exposure of the sample to the atmosphere and loss of benzene prior to sampling.

(iv) Each waste sample shall be analyzed using one of the following test methods for determining the benzene concentration in a waste stream:

(A) Method 8020, Aromatic Volatile Organics, in "Test Methods for Evaluating Solid Waste, Physical/Chemical Methods," EPA Publication No. SW-846 (incorporation by reference as specified in §61.18 of this part);

(B) Method 8021, Volatile Organic Compounds in Water by Purge and Trap Capillary Column Gas Chromatography with Photoionization and Electrolytic Conductivity Detectors in Series in "Test Methods for Evaluating
Solid Waste, Physical/Chemical Methods," EPA Publication No. SW-846 (incorporation by reference as specified in §61.18 of this part);

(C) Method 8240, Gas Chromatography/Mass Spectrometry for Volatile Organics in "Test Methods for Evaluating Solid Waste, Physical/ Chemical Methods," EPA Publication No. SW-846 (incorporation by reference as specified in §61.18 of this part);

(D) Method 8260, Gas Chromatography/Mass Spectrometry for Volatile Organics: Capillary Column Technique in "Test Methods for Evaluating Solid Waste, Physical/Chemical Methods," EPA Publication No. SW-846 (incorporation by reference as specified in §61.18 of this part);

(E) Method 602, Purgeable Aromatics, as described in 40 CFR part 136, appendix A, Test Procedures for Analysis of Organic Pollutants, for wastewaters for which this is an approved EPA methods; or

(F) Method 624, Purgeables, as described in 40 CFR part 136, appendix A, Test Procedures for Analysis of Organic Pollutants, for wastewaters for which this is an approved EPA method.

(v) The flow-weighted annual average benzene concentration shall be calculated by averaging the results of the sample analyses as follows:

$$\overline{\mathbf{C}} = \frac{1}{\mathbf{Q}_{t}} \times \sum_{i=1}^{n} (\mathbf{Q}_{i}) (\mathbf{C}_{i})$$

Where:

 $\bar{\mathbf{C}}$ = Flow-weighted annual average benzene concentration for waste stream, ppmw.

 Q_t = Total annual waste quantity for waste stream, kg/yr (lb/yr).

n = Number of waste samples (at least 3).

 Q_i = Annual waste quantity for waste stream represented by C_i , kg/yr (lb/yr).

 C_i = Measured concentration of benzene in waste sample i, ppmw.

(d) An owner or operator using performance tests to demonstrate compliance of a treatment process with §61.348 (a)(1)(i) shall measure the flowweighted annual average benzene concentration of the waste stream exiting the treatment process by collecting and analyzing a minimum of three representative samples of the waste stream using the procedures in paragraph (c)(3) of this section. The test shall be conducted under conditions that exist when the treatment process is operating at the highest inlet waste stream flow rate and benzene content expected to occur. Operations during periods of startup, shutdown, and malfunction shall not constitute representative conditions for the purpose of a test. The owner or operator shall record all process information as is necessary to document the operating conditions during the test.

(e) An owner or operator using performance tests to demonstrate compliance of a treatment process with $\S61.348(a)(1)(ii)$ of this subpart shall determine the percent reduction of benzene in the waste stream on a mass basis by the following procedure:

(1) The test shall be conducted under conditions that exist when the treatment process is operating at the highest inlet waste stream flow rate and benzene content expected to occur. Operations during periods of startup, shutdown, and malfunction shall not constitute representative conditions for the purpose of a test. The owner or operator shall record all process information as is necessary to document the operating conditions during the test.

(2) All testing equipment shall be prepared and installed as specified in the appropriate test methods.

(3) The mass flow rate of benzene entering the treatment process (E_b) shall be determined by computing the product of the flow rate of the waste stream entering the treatment process, as determined by the inlet flow meter, and the benzene concentration of the waste stream, as determined using the sampling and analytical procedures specified in paragraph (c)(2) or (c)(3) of this section. Three grab samples of the waste shall be taken at equally spaced time intervals over a 1-hour period. Each 1-hour period constitutes a run, and the performance test shall consist of a minimum of 3 runs conducted over a 3-hour period. The mass flow rate of benzene entering the treatment process is calculated as follows:

$$E_b = \frac{K}{n \times 10^6} \left[\sum_{i=1}^n V_i C_i \right]$$

Where:

§61.355

- $E_{\rm b}$ = Mass flow rate of benzene entering the treatment process, kg/hr (lb/hr).
- K = Density of the waste stream, kg/m^3 (lb/ ft^3).
- V_i = Average volume flow rate of waste entering the treatment process during each run i, m³/hr (ft³/hr).
- C_i = Average concentration of benzene in the waste stream entering the treatment process during each run i, ppmw.
- n = Number of runs.

 10^6 = Conversion factor for ppmw.

(4) The mass flow rate of benzene exiting the treatment process (E_a) shall be determined by computing the product of the flow rate of the waste stream exiting the treatment process, as determined by the outlet flow meter or the inlet flow meter, and the benzene concentration of the waste stream, as determined using the sampling and analytical procedures specified in paragraph (c)(2) or (c)(3) of this section. Three grab samples of the waste shall be taken at equally spaced time intervals over a 1-hour period. Each 1-hour period constitutes a run, and the performance test shall consist of a minimum of 3 runs conducted over the same 3-hour period at which the mass flow rate of benzene entering the treatment process is determined. The mass flow rate of benzene exiting the treatment process is calculated as follows:

$$=\frac{1}{n\times 10^6}\left[\sum_{i=1}^n i i\right]$$

Where:

- E_a = Mass flow rate of benzene exiting the treatment process, kg/hr (lb/hr).
- K = Density of the waste stream, kg/m³ (lb/ ft^3).
- V_i = Average volume flow rate of waste exiting the treatment process during each run i, m³/hr (ft³/hr).
- C_i = Average concentration of benzene in the waste stream exiting the treatment process during each run i, ppmw.
- n = Number of runs.
- 10^6 = Conversion factor for ppmw.

(f) An owner or operator using performance tests to demonstrate compliance of a treatment process with $\S61.348(a)(1)(iii)$ of this subpart shall determine the benzene destruction efficiency for the combustion unit by the following procedure:

(1) The test shall be conducted under conditions that exist when the combus-

40 CFR Ch. I (7–1–23 Edition)

tion unit is operating at the highest inlet waste stream flow rate and benzene content expected to occur. Operations during periods of startup, shutdown, and malfunction shall not constitute representative conditions for the purpose of a test. The owner or operator shall record all process information necessary to document the operating conditions during the test.

(2) All testing equipment shall be prepared and installed as specified in the appropriate test methods.

(3) The mass flow rate of benzene entering the combustion unit shall be determined by computing the product of the flow rate of the waste stream entering the combustion unit, as determined by the inlet flow meter, and the benzene concentration of the waste stream, as determined using the sampling procedures in paragraph (c)(2) or (c)(3) of this section. Three grab samples of the waste shall be taken at equally spaced time intervals over a 1hour period. Each 1-hour period constitutes a run, and the performance test shall consist of a minimum of 3 runs conducted over a 3-hour period. The mass flow rate of benzene into the combustion unit is calculated as follows:

$$_{b} = \frac{1}{n \times 10^{6}} \left[\sum_{i=1}^{n} i i \right]$$

Where:

- E_b = Mass flow rate of benzene entering the combustion unit, kg/hr (lb/hr).
- K = Density of the waste stream, kg/m³ (lb/ ft^3).
- V_i = Average volume flow rate of waste entering the combustion unit during each run i, m³/hr (ft³/hr).
- C_i = Average concentration of benzene in the waste stream entering the combustion unit during each run i, ppmw.

n = Number of runs.

 $10^6 =$ Conversion factor for ppmw.

(4) The mass flow rate of benzene exiting the combustion unit exhaust stack shall be determined as follows:

(i) The time period for the test shall not be less than 3 hours during which at least 3 stack gas samples are collected and be the same time period at which the mass flow rate of benzene entering the treatment process is determined. Each sample shall be collected

over a 1-hour period (e.g., in a tedlar bag) to represent a time-integrated composite sample and each 1-hour period shall correspond to the periods when the waste feed is sampled.

(ii) A run shall consist of a 1-hour period during the test. For each run:

(A) The reading from each measurement shall be recorded;

(B) The volume exhausted shall be determined using Method 2, 2A, 2C, or 2D from appendix A of 40 CFR part 60, as appropriate.

(C) The average benzene concentration in the exhaust downstream of the combustion unit shall be determined using Method 18 from appendix A of 40 CFR part 60.

(iii) The mass of benzene emitted during each run shall be calculated as follows:

$$M_i = D_b VC(10^{-6})$$

Where:

 M_{i} = Mass of benzene emitted during run i, kg (lb).

V = Volume of air-vapor mixture exhausted at standard conditions, m³ (ft³).

C = Concentration of benzene measured in the exhaust, ppmv.

 $D_b = Density of benzene, 3.24 \text{ kg/m}^3 (0.202 \text{ lb/} ft^3).$

 10^6 = Conversion factor for ppmv.

(iv) The benzene mass emission rate in the exhaust shall be calculated as follows:

$$\mathbf{E}_{a} = \left(\sum_{i=1}^{n} \mathbf{M}_{i}\right) / \mathbf{T}$$

Where:

 E_a = Mass flow rate of benzene emitted from the combustion unit, kg/hr (lb/hr).

 M_i = Mass of benzene emitted from the combustion unit during run i, kg (lb).

T = Total time of all runs, hr.

n = Number of runs.

(5) The benzene destruction efficiency for the combustion unit shall be calculated as follows:

$$R = \frac{E_b - E_a}{E_b} \times 100$$

Where:

R = Benzene destruction efficiency for the combustion unit, percent.

 E_b = Mass flow rate of benzene entering the combustion unit, kg/hr (lb/hr).

 E_a = Mass flow rate of benzene emitted from the combustion unit, kg/hr (lb/hr).

(g) An owner or operator using performance tests to demonstrate compliance of a wastewater treatment system unit with §61.348(b) shall measure the flow-weighted annual average benzene concentration of the wastewater stream where the waste stream enters an exempt waste management unit by collecting and analyzing a minimum of three representative samples of the waste stream using the procedures in paragraph (c)(3) of this section. The test shall be conducted under conditions that exist when the wastewater treatment system is operating at the highest inlet wastewater stream flow rate and benzene content expected to occur. Operations during periods of startup, shutdown, and malfunction shall not constitute representative conditions for the purpose of a test. The owner or operator shall record all process information as is necessary to document the operating conditions during the test.

(h) An owner or operator shall test equipment for compliance with no detectable emissions as required in §§ 61.343 through 61.347, and §61.349 of this subpart in accordance with the following requirements:

(1) Monitoring shall comply with Method 21 from appendix A of 40 CFR part 60.

(2) The detection instrument shall meet the performance criteria of Method 21.

(3) The instrument shall be calibrated before use on each day of its use by the procedures specified in Method 21.

(4) Calibration gases shall be:

(i) Zero air (less than 10 ppm of hydrocarbon in air); and

(ii) A mixture of methane or nhexane and air at a concentration of approximately, but less than, 10,000 ppm methane or n-hexane.

(5) The background level shall be determined as set forth in Method 21.

(6) The instrument probe shall be traversed around all potential leak interfaces as close as possible to the interface as described in Method 21.

§61.355

(7) The arithmetic difference between the maximum concentration indicated by the instrument and the background level is compared to 500 ppm for determining compliance.

(i) An owner or operator using a performance test to demonstrate compliance of a control device with either the organic reduction efficiency requirement or the benzene reduction efficiency requirement specified under $\S 61.349(a)(2)$ shall use the following procedures:

(1) The test shall be conducted under conditions that exist when the waste management unit vented to the control device is operating at the highest load or capacity level expected to occur. Operations during periods of startup, shutdown, and malfunction shall not constitute representative conditions for the purpose of a test. The owner or operator shall record all process information necessary to document the operating conditions during the test.

(2) Sampling sites shall be selected using Method 1 or 1A from appendix A of 40 CFR part 60, as appropriate.

(3) The mass flow rate of either the organics or benzene entering and exiting the control device shall be determined as follows:

(i) The time period for the test shall not be less than 3 hours during which at least 3 stack gas samples are collected. Samples of the vent stream entering and exiting the control device shall be collected during the same time period. Each sample shall be collected over a 1-hour period (e.g., in a tedlar bag) to represent a time-integrated composite sample.

(ii) A run shall consist of a 1-hour period during the test. For each run:

(A) The reading from each measurement shall be recorded;

(B) The volume exhausted shall be determined using Method 2, 2A, 2C, or 2D from appendix A of 40 CFR part 60, as appropriate;

(C) The organic concentration or the benzene concentration, as appropriate, in the vent stream entering and exiting the control shall be determined using Method 18 from appendix A of 40 CFR part 60.

(iii) The mass of organics or benzene entering and exiting the control device

40 CFR Ch. I (7–1–23 Edition)

during each run shall be calculated as follows:

$$\begin{split} \mathbf{M}_{aj} &= \frac{\mathbf{K}_{1} \mathbf{V}_{aj}}{10^{6}} \Bigg(\sum_{i=1}^{n} \mathbf{C}_{ai} \mathbf{M} \mathbf{W}_{i} \Bigg) \\ \mathbf{M}_{bj} &= \frac{\mathbf{K}_{1} \mathbf{V}_{bj}}{10^{6}} \Bigg(\sum_{i=1}^{n} \mathbf{C}_{bi} \mathbf{M} \mathbf{W}_{i} \Bigg) \end{split}$$

- M_{aj} = Mass of organics or benzene in the vent stream entering the control device during run j, kg (lb).
- M_{bj} = Mass of organics or benzene in the vent stream exiting the control device during run j, kg (lb).
- V_{aj} = Volume of vent stream entering the control device during run j, at standard conditions, m³ (ft³).
- V_{bj} = Volume of vent stream exiting the control device during run j, at standard conditions, m³ (ft³).
- C_{ai} = Organic concentration of compound i or the benzene concentration measured in the vent stream entering the control device as determined by Method 18, ppm by volume on a dry basis.
- C_{bi} = Organic concentration of compound i or the benzene concentration measured in the vent stream exiting the control device as determined by Method 18, ppm by volume on a dry basis.
- MW_i = Molecular weight of organic compound i in the vent stream, or the molecular weight of benzene, kg/kg-mol (lb/lb-mole).
- n = Number of organic compounds in the vent stream; if benzene reduction efficiency is being demonstrated, then n = 1.
- $K_1 = \mbox{Conversion factor for molar volume at standard conditions (293 K and 760 mm Hg (527 R and 14.7 psia)) }$
- = $0.0416 \text{ kg-mol/m}^3 (0.00118 \text{ lb-mol/ft}^3)$
- 10^{-6} = Conversion factor for ppmv.

(iv) The mass flow rate of organics or benzene entering and exiting the control device shall be calculated as follows:

$$-\left(\sum_{i=1}^{n} \mathbf{M}_{i}\right) / b = \left(\sum_{i=1}^{n} b_{i}\right) /$$

Where:

- $\begin{array}{l} E_a = Mass \mbox{ flow rate of organics or benzene} \\ \mbox{ entering the control device, kg/hr (lb/hr).} \\ E_b = Mass \mbox{ flow rate of organics or benzene} \end{array}$
- exiting the control device, kg/hr (lb/hr). M_{aj} = Mass of organics or benzene in the vent
- stream entering the control device during run j, kg (lb).
- M $_{bj}$ = Mass of organics or benzene in the vent stream exiting the control device during run j, kg (lb).
- T = Total time of all runs, hr.

n = Number of runs.

(4) The organic reduction efficiency or the benzene reduction efficiency for the control device shall be calculated as follows:

$$R = \frac{E_a - E_b}{E_a} \times 100$$

Where:

- R = Total organic reduction of efficiency or benzene reduction efficiency for the control device, percent.
- $E_b = Mass \ flow \ rate \ of \ organics \ or \ benzene \\ entering \ the \ control \ device, \ kg/hr \ (lb/hr).$
- E_a = Mass flow rate of organic or benzene emitted from the control device, kg/hr (lb/hr).

(j) An owner or operator shall determine the benzene quantity for the purposes of the calculation required by 61.342 (c)(3)(ii)(B) according to the provisions of paragraph (a) of this section, except that the procedures in paragraph (a) of this section shall also apply to wastes with a water content of 10 percent or less.

(k) An owner or operator shall determine the benzene quantity for the purposes of the calculation required by §61.342(e)(2) by the following procedure:

(1) For each waste stream that is not controlled for air emissions in accordance with §61.343. 61.344, 61.345, 61.346, 61.347, or 61.348(a), as applicable to the waste management unit that manages the waste, the benzene quantity shall be determined as specified in paragraph (a) of this section, except that paragraph (b)(4) of this section shall not apply, i.e., the waste quantity for process unit turnaround waste is not annualized but shall be included in the determination of benzene quantity for the year in which the waste is generated for the purposes of the calculation required by \$61.342(e)(2).

(2) For each waste stream that is controlled for air emissions in accordance with §61.343. 61.344, 61.345, 61.346,

61.347, or 61.348(a), as applicable to the waste management unit that manages the waste, the determination of annual waste quantity and flow-weighted annual average benzene concentration shall be made at the first applicable location as described in paragraphs (k)(2)(i), (k)(2)(i), and (k)(2)(ii) of this section and prior to any reduction of benzene concentration through volatilization of the benzene, using the methods given in (k)(2)(iv) and (k)(2)(v)

(i) Where the waste stream enters the first waste management unit not complying with \$ 61.343, 61.344, 61.345, 61.346, 61.347, and 61.348(a) that are applicable to the waste management unit.

(ii) For each waste stream that is managed or treated only in compliance with §§ 61.343 through 61.348(a) up to the point of final direct discharge from the facility, the determination of benzene quantity shall be prior to any reduction of benzene concentration through volatilization of the benzene, or

(iii) For wastes managed in units controlled for air emissions in accordance with §§61.343, 61.344, 61.345, 61.346, 61.347, and 61.348(a), and then transferred offsite, facilities shall use the first applicable offsite location as described in paragraphs (k)(2)(i) and (k)(2)(i) of this section if they have documentation from the offsite facility of the benzene quantity at this location. Facilities without this documentation for offsite wastes shall use the benzene quantity determined at the point where the transferred waste leaves the facility.

(iv) Annual waste quantity shall be determined using the procedures in paragraphs (b)(5), (6), or (7) of this section, and

(v) The flow-weighted annual average benzene concentration shall be determined using the procedures in paragraphs (c)(2) or (3) of this section.

(3) The benzene quantity in a waste stream that is generated less than one time per year, including process unit turnaround waste, shall be included in the determination of benzene quantity as determined in paragraph (k)(6) of this section for the year in which the waste is generated. The benzene quantity in this waste stream shall not be annualized or averaged over the time interval between the activities that resulted in generation of the waste for purposes of determining benzene quantity as determined in paragraph (k)(6)of this section.

(4) The benzene in waste entering an enhanced biodegradation unit, as defined in 61.348(b)(2)(ii)(B), shall not be included in the determination of benzene quantity, determined in paragraph (k)(6) of this section, if the following conditions are met:

(i) The benzene concentration for each waste stream entering the enhanced biodegradation unit is less than 10 ppmw on a flow-weighted annual average basis, and

(ii) All prior waste management units managing the waste comply with \$ 61.343, 61.344, 61.345, 61.346, 61.347 and 61.348(a).

(5) The benzene quantity for each waste stream in paragraph (k)(2) of this section shall be determined by multiplying the annual waste quantity of each waste stream times its flow-weighted annual average benzene concentration.

(6) The total benzene quantity for the purposes of the calculation required by $\S61.342(e)(2)$ shall be determined by adding together the benzene quantities determined in paragraphs (k)(1) and (k)(5) of this section for each applicable waste stream.

(7) If the benzene quantity determined in paragraph (6) of this section exceeds 6.0 Mg/yr (6.6 ton/yr) only because of multiple counting of the benzene quantity for a waste stream, the owner or operator may use the following procedures for the purposes of the calculation required by §61.342(e)(2):

(i) Determine which waste management units are involved in the multiple counting of benzene;

(ii) Determine the quantity of benzene that is emitted, recovered, or removed from the affected units identified in paragraph (k)(7)(i) of this section, or destroyed in the units if applicable, using either direct measurements or the best available estimation techniques developed or approved by the Administrator.

(iii) Adjust the benzene quantity to eliminate the multiple counting of ben-

40 CFR Ch. I (7–1–23 Edition)

zene based on the results from paragraph (k)(7)(ii) of this section and determine the total benzene quantity for the purposes of the calculation required by §61.342(e)(2).

(iv) Submit in the annual report required under 61.357(a) a description of the methods used and the resulting calculations for the alternative procedure under paragraph (k)(7) of this section, the benzene quantity determination from paragraph (k)(6) of this section, and the adjusted benzene quantity determination from paragraph (k)(7)(iii) of this section.

[55 FR 8346, Mar. 7, 1990; 55 FR 12444, Apr. 3, 1990, as amended at 55 FR 37231, Sept. 10, 1990; 58 FR 3099, Jan. 7, 1993; 65 FR 62160, Oct. 17, 2000]

§61.356 Recordkeeping requirements.

(a) Each owner or operator of a facility subject to the provisions of this subpart shall comply with the recordkeeping requirements of this section. Each record shall be maintained in a readily accessible location at the facility site for a period not less than two years from the date the information is recorded unless otherwise specified.

(b) Each owner or operator shall maintain records that identify each waste stream at the facility subject to this subpart, and indicate whether or not the waste stream is controlled for benzene emissions in accordance with this subpart. In addition the owner or operator shall maintain the following records:

(1) For each waste stream not controlled for benzene emissions in accordance with this subpart, the records shall include all test results, measurements, calculations, and other documentation used to determine the following information for the waste stream: waste stream identification, water content, whether or not the waste stream is a process wastewater stream, annual waste quantity, range of benzene concentrations, annual average flow-weighted benzene quantity.

(2) For each waste stream exempt from 61.342(c)(1) in accordance with 61.342(c)(3), the records shall include:

(i) All measurements, calculations, and other documentation used to determine that the continuous flow of process wastewater is less than 0.02 liters (0.005 gallons) per minute or the annual waste quantity of process wastewater is less than 10 Mg/yr (11 ton/yr) in accordance with 61.342(c)(3)(i), or

(ii) All measurements, calculations, and other documentation used to determine that the sum of the total annual benzene quantity in all exempt waste streams does not exceed 2.0 Mg/yr (2.2 ton/yr) in accordance with $\S61.342(c)(3)(ii)$.

(3) For each facility where process wastewater streams are controlled for benzene emissions in accordance with §61.342(d) of this subpart, the records shall include for each treated process wastewater stream all measurements, calculations, and other documentation used to determine the annual benzene quantity in the process wastewater stream exiting the treatment process.

(4) For each facility where waste streams are controlled for benzene emissions in accordance with $\S61.342(e)$, the records shall include for each waste stream all measurements, including the locations of the measurements, calculations, and other documentation used to determine that the total benzene quantity does not exceed 6.0 Mg/yr (6.6 ton/yr).

(5) For each facility where the annual waste quantity for process unit turnaround waste is determined in accordance with §61.355(b)(5), the records shall include all test results, measurements, calculations, and other documentation used to determine the following information: identification of each process unit at the facility that undergoes turnarounds, the date of the most recent turnaround for each process unit, identification of each process unit turnaround waste, the water content of each process unit turnaround waste, the annual waste quantity determined in accordance with §61.355(b)(5), the range of benzene concentrations in the waste, the annual average flow-weighted benzene concentration of the waste, and the annual benzene quantity calculated in accordance with §61.355(a)(1)(iii) of this section.

(6) For each facility where wastewater streams are controlled for benzene emissions in accordance with §61.348(b)(2), the records shall include all measurements, calculations, and other documentation used to determine the annual benzene content of the waste streams and the total annual benzene quantity contained in all waste streams managed or treated in exempt waste management units.

(c) An owner or operator transferring waste off-site to another facility for treatment in accordance with §61.342(f) shall maintain documentation for each offsite waste shipment that includes the following information: Date waste is shipped offsite, quantity of waste shipped offsite, name and address of the facility receiving the waste, and a copy of the notice sent with the waste shipment.

(d) An owner or operator using control equipment in accordance with §§ 61.343 through 61.347 shall maintain engineering design documentation for all control equipment that is installed on the waste management unit. The documentation shall be retained for the life of the control equipment. If a control device is used, then the owner or operator shall maintain the control device records required by paragraph (f) of this section.

(e) An owner or operator using a treatment process or wastewater treatment system unit in accordance with §61.348 of this subpart shall maintain the following records. The documentation shall be retained for the life of the unit.

(1) A statement signed and dated by the owner or operator certifying that the unit is designed to operate at the documented performance level when the waste stream entering the unit is at the highest waste stream flow rate and benzene content expected to occur.

(2) If engineering calculations are used to determine treatment process or wastewater treatment system unit performance, then the owner or operator shall maintain the complete design analysis for the unit. The design analysis shall include for example the following information: Design specifications, drawings, schematics, piping and instrumentation diagrams, and other documentation necessary to demonstrate the unit performance.

(3) If performance tests are used to determine treatment process or wastewater treatment system unit performance, then the owner or operator shall maintain all test information necessary to demonstrate the unit performance.

(i) A description of the unit including the following information: type of treatment process; manufacturer name and model number; and for each waste stream entering and exiting the unit, the waste stream type (e.g., process wastewater, sludge, slurry, etc.), and the design flow rate and benzene content.

(ii) Documentation describing the test protocol and the means by which sampling variability and analytical variability were accounted for in the determination of the unit performance. The description of the test protocol shall include the following information: sampling locations, sampling method, sampling frequency, and analytical procedures used for sample analysis.

(iii) Records of unit operating conditions during each test run including all key process parameters.

(iv) All test results.

(4) If a control device is used, then the owner or operator shall maintain the control device records required by paragraph (f) of this section.

(f) An owner or operator using a closed-vent system and control device in accordance with §61.349 of this subpart shall maintain the following records. The documentation shall be retained for the life of the control device.

(1) A statement signed and dated by the owner or operator certifying that the closed-vent system and control device is designed to operate at the documented performance level when the waste management unit vented to the control device is or would be operating at the highest load or capacity expected to occur.

(2) If engineering calculations are used to determine control device performance in accordance with §61.349(c), then a design analysis for the control device that includes for example:

40 CFR Ch. I (7–1–23 Edition)

(i) Specifications, drawings, schematics, and piping and instrumentation diagrams prepared by the owner or operator, or the control device manufacturer or vendor that describe the control device design based on acceptable engineering texts. The design analysis shall address the following vent stream characteristics and control device operating parameters:

(A) For a thermal vapor incinerator, the design analysis shall consider the vent stream composition, constituent concentrations, and flow rate. The design analysis shall also establish the design minimum and average temperature in the combustion zone and the combustion zone residence time.

(B) For a catalytic vapor incinerator, the design analysis shall consider the vent stream composition, constituent concentrations, and flow rate. The design analysis shall also establish the design minimum and average temperatures across the catalyst bed inlet and outlet.

(C) For a boiler or process heater, the design analysis shall consider the vent stream composition, constituent concentrations, and flow rate. The design analysis shall also establish the design minimum and average flame zone temperatures, combustion zone residence time, and description of method and location where the vent stream is introduced into the flame zone.

(D) For a flare, the design analysis shall consider the vent stream composition, constituent concentrations, and flow rate. The design analysis shall also consider the requirements specified in 40 CFR 60.18.

(E) For a condenser, the design analysis shall consider the vent stream composition, constituent concentration, flow rate, relative humidity, and temperature. The design analysis shall also establish the design outlet organic compound concentration level or the design outlet benzene concentration level, design average temperature of the condenser exhaust vent stream, and the design average temperatures of the coolant fluid at the condenser inlet and outlet.

(F) For a carbon adsorption system that regenerates the carbon bed directly on-site in the control device such as a fixed-bed adsorber, the design

analysis shall consider the vent stream composition, constituent concentration, flow rate, relative humidity, and temperature. The design analysis shall also establish the design exhaust vent stream organic compound concentration level or the design exhaust vent stream benzene concentration level, number and capacity of carbon beds, type and working capacity of activated carbon used for carbon beds, design total steam flow over the period of each complete carbon bed regeneration cycle, duration of the carbon bed steaming and cooling/drying cycles, design carbon bed temperature after regeneration, design carbon bed regeneration time, and design service life of carbon.

(G) For a carbon adsorption system that does not regenerate the carbon bed directly on-site in the control device, such as a carbon canister, the design analysis shall consider the vent stream composition, constituent concentration, flow rate, relative humid-ity, and temperature. The design analysis shall also establish the design exhaust vent stream organic compound concentration level or the design exhaust vent stream benzene concentration level, capacity of carbon bed, type and working capacity of activated carbon used for carbon bed, and design carbon replacement interval based on the total carbon working capacity of the control device and source operating schedule.

(H) For a control device subject to the requirements of $\S61.349(a)(2)(iv)$, the design analysis shall consider the vent stream composition, constituent concentration, and flow rate. The design analysis shall also include all of the information submitted under \$61.349(a)(2)(iv).

(ii) [Reserved]

(3) If performance tests are used to determine control device performance in accordance with §61.349(c) of this subpart:

(i) A description of how it is determined that the test is conducted when the waste management unit or treatment process is operating at the highest load or capacity level. This description shall include the estimated or design flow rate and organic content of each vent stream and definition of the acceptable operating ranges of key process and control parameters during the test program.

(ii) A description of the control device including the type of control device, control device manufacturer's name and model number, control device dimensions, capacity, and construction materials.

(iii) A detailed description of sampling and monitoring procedures, including sampling and monitoring locations in the system, the equipment to be used, sampling and monitoring frequency, and planned analytical procedures for sample analysis.

(iv) All test results.

(g) An owner or operator shall maintain a record for each visual inspection required by §§61.343 through 61.347 of this subpart that identifies a problem (such as a broken seal, gap or other problem) which could result in benzene emissions. The record shall include the date of the inspection, waste management unit and control equipment location where the problem is identified, a description of the problem, a description of the corrective action taken, and the date the corrective action was completed.

(h) An owner or operator shall maintain a record for each test of no detectable emissions required by §§61.343 through 61.347 and §61.349 of this subpart. The record shall include the following information: date the test is performed, background level measured during test, and maximum concentration indicated by the instrument reading measured for each potential leak interface. If detectable emissions are measured at a leak interface, then the record shall also include the waste management unit, control equipment, and leak interface location where detectable emissions were measured, a description of the problem, a description of the corrective action taken, and the date the corrective action was completed.

(i) For each treatment process and wastewater treatment system unit operated to comply with §61.348, the owner or operator shall maintain documentation that includes the following information regarding the unit operation: (1) Dates of startup and shutdown of the unit.

(2) If measurements of waste stream benzene concentration are performed in accordance with 61.354(a)(1) of this subpart, the owner or operator shall maintain records that include date each test is performed and all test results.

(3) If a process parameter is continuously monitored in accordance with $\S61.354(a)(2)$ of this subpart, the owner or operator shall maintain records that include a description of the operating parameter (or parameters) to be monitored to ensure that the unit will be operated in conformance with these standards and the unit's design specifications, and an explanation of the criteria used for selection of that parameter (or parameters). This documentation shall be kept for the life of the unit.

(4) If measurements of waste stream benzene concentration are performed in accordance with §61.354(b), the owner or operator shall maintain records that include the date each test is performed and all test results.

(5) Periods when the unit is not operated as designed.

(j) For each control device, the owner or operator shall maintain documentation that includes the following information regarding the control device operation:

(1) Dates of startup and shutdown of the closed-vent system and control device.

(2) A description of the operating parameter (or parameters) to be monitored to ensure that the control device will be operated in conformance with these standards and the control device's design specifications and an explanation of the criteria used for selection of that parameter (or parameters). This documentation shall be kept for the life of the control device.

(3) Periods when the closed-vent system and control device are not operated as designed including all periods and the duration when:

(i) Any valve car-seal or closure mechanism required under §61.349(a)(1)(ii) is broken or the by-pass line valve position has changed.

(ii) The flow monitoring devices required under §61.349(a)(1)(ii) indicate 40 CFR Ch. I (7–1–23 Edition)

that vapors are not routed to the control device as required.

(4) If a thermal vapor incinerator is used, then the owner or operator shall maintain continuous records of the temperature of the gas stream in the combustion zone of the incinerator and records of all 3-hour periods of operation during which the average temperature of the gas stream in the combustion zone is more than 28 °C (50 °F) below the design combustion zone temperature.

(5) If a catalytic vapor incinerator is used, then the owner or operator shall maintain continuous records of the temperature of the gas stream both upstream and downstream of the catalyst bed of the incinerator, records of all 3hour periods of operation during which the average temperature measured before the catalyst bed is more than 28 °C (50 °F) below the design gas stream temperature, and records of all 3-hour periods of operation during which the average temperature difference across the catalyst bed is less than 80 percent of the design temperature difference.

(6) If a boiler or process heater is used, then the owner or operator shall maintain records of each occurrence when there is a change in the location at which the vent stream is introduced into the flame zone as required by §61.349(a)(2)(i)(C). For a boiler or process heater having a design heat input capacity less than 44 MW (150 \times 106 BTU/hr), the owner or operator shall maintain continuous records of the temperature of the gas stream in the combustion zone of the boiler or process heater and records of all 3-hour periods of operation during which the average temperature of the gas stream in the combustion zone is more than 28 °C (50 °F) below the design combustion zone temperature. For a boiler or process heater having a design heat input capacity greater than or equal to 44 MW (150 \times 106 BTU/hr), the owner or operator shall maintain continuous records of the parameter(s) monitored in accordance with the requirements of §61.354(c)(5).

(7) If a flare is used, then the owner or operator shall maintain continuous records of the flare pilot flame monitoring and records of all periods during which the pilot flame is absent.

(8) If a condenser is used, then the owner or operator shall maintain records from the monitoring device of the parameters selected to be monitored in accordance with 61.354(c)(6). If concentration of organics or concentration of benzene in the control device outlet gas stream is monitored, then the owner or operator shall record all 3-hour periods of operation during which the concentration of organics or the concentration of benzene in the exhaust stream is more than 20 percent greater than the design value. If the temperature of the condenser exhaust stream and coolant fluid is monitored, then the owner or operator shall record all 3-hour periods of operation during which the temperature of the condenser exhaust vent stream is more than 6 °C (11 °F) above the design average exhaust vent stream temperature, or the temperature of the coolant fluid exiting the condenser is more than 6 °C (11 °F) above the design average coolant fluid temperature at the condenser outlet.

(9) If a carbon adsorber is used, then the owner or operator shall maintain records from the monitoring device of the concentration of organics or the concentration of benzene in the control device outlet gas stream. If the concentration of organics or the concentration of benzene in the control device outlet gas stream is monitored, then the owner or operator shall record all 3-hour periods of operation during which the concentration of organics or the concentration of benzene in the exhaust stream is more than 20 percent greater than the design value. If the carbon bed regeneration interval is monitored, then the owner or operator shall record each occurrence when the vent stream continues to flow through the control device beyond the predetermined carbon bed regeneration time.

(10) If a carbon adsorber that is not regenerated directly on site in the control device is used, then the owner or operator shall maintain records of dates and times when the control device is monitored, when breakthrough is measured, and shall record the date and time then the existing carbon in the control device is replaced with fresh carbon. (11) If an alternative operational or process parameter is monitored for a control device, as allowed in 61.354(e) of this subpart, then the owner or operator shall maintain records of the continuously monitored parameter, including periods when the device is not operated as designed.

(12) If a control device subject to the requirements of $\S61.349(a)(2)(iv)$ is used, then the owner or operator shall maintain records of the parameters that are monitored and each occurrence when the parameters monitored are outside the range of values specified in \$61.349(a)(2)(iv)(C), or other records as specified by the Administrator.

(k) An owner or operator who elects to install and operate the control equipment in §61.351 of this subpart shall comply with the recordkeeping requirements in 40 CFR 60.115b.

(1) An owner or operator who elects to install and operate the control equipment in $\S61.352$ of this subpart shall maintain records of the following:

(1) The date, location, and corrective action for each visual inspection required by 40 CFR 60.693-2(a)(5), during which a broken seal, gap, or other problem is identified that could result in benzene emissions.

(2) Results of the seal gap measurements required by 40 CFR 60.693-2(a).

(m) If a system is used for emission control that is maintained at a pressure less than atmospheric pressure with openings to provide dilution air, then the owner or operator shall maintain records of the monitoring device and records of all periods during which the pressure in the unit is operated at a pressure that is equal to or greater than atmospheric pressure.

(n) Each owner or operator using a total enclosure to comply with control requirements for tanks in §61.343 or the control requirements for containers in §61.345 must keep the records required in paragraphs (n)(1) and (2) of this section. Owners or operators may use records as required in 40 CFR. 264.1089(b)(2)(iv) 40 CFR. or 265.1090(b)(2)(iv) for a tank or as required in 40 CFR 264.1089(d)(1) or 40 CFR 265.1090(d)(1) for a container to meet the recordkeeping requirement in paragraph (n)(1) of this section. The owner or operator must make the

records of each verification of a total enclosure available for inspection upon request.

(1) Records of the most recent set of calculations and measurements performed to verify that the enclosure meets the criteria of a permanent total enclosure as specified in "Procedure T—Criteria for and Verification of a Permanent or Temporary Total Enclosure" in 40 CFR 52.741, appendix B;

(2) Records required for a closed-vent system and control device according to the requirements in paragraphs (d) (f), and (j) of this section.

[55 FR 8346, Mar. 7, 1990; 55 FR 12444, Apr. 3, 1990; 55 FR 18331, May 2, 1990, as amended at 58 FR 3103, Jan. 7, 1993; 65 FR 62161, Oct. 17, 2000; 67 FR 68533, Nov. 12, 2002]

§61.357 Reporting requirements.

(a) Each owner or operator of a chemical plant, petroleum refinery, coke byproduct recovery plant, and any facility managing wastes from these industries shall submit to the Administrator within 90 days after January 7, 1993, or by the initial startup for a new source with an initial startup after the effective date, a report that summarizes the regulatory status of each waste stream subject to §61.342 and is determined by the procedures specified in §61.355(c) to contain benzene. Each owner or operator subject to this subpart who has no benzene onsite in wastes, products, byproducts, or intermediates shall submit an initial report that is a statement to this effect. For all other owners or operators subject to this subpart, the report shall include the following information:

(1) Total annual benzene quantity from facility waste determined in accordance with 61.355(a) of this subpart.

(2) A table identifying each waste stream and whether or not the waste stream will be controlled for benzene emissions in accordance with the requirements of this subpart.

(3) For each waste stream identified as not being controlled for benzene emissions in accordance with the requirements of this subpart the following information shall be added to the table: 40 CFR Ch. I (7–1–23 Edition)

(i) Whether or not the water content of the waste stream is greater than 10 percent;

(ii) Whether or not the waste stream is a process wastewater stream, product tank drawdown, or landfill leachate;

(iii) Annual waste quantity for the waste stream;

(iv) Range of benzene concentrations for the waste stream;

(v) Annual average flow-weighted benzene concentration for the waste stream; and

(vi) Annual benzene quantity for the waste stream.

(4) The information required in paragraphs (a) (1), (2), and (3) of this section should represent the waste stream characteristics based on current configuration and operating conditions. An owner or operator only needs to list in the report those waste streams that contact materials containing benzene. The report does not need to include a description of the controls to be installed to comply with the standard or other information required in §61.10(a).

(b) If the total annual benzene quantity from facility waste is less than 1 Mg/yr (1.1 ton/yr), then the owner or operator shall submit to the Administrator a report that updates the information listed in paragraphs (a)(1) through (a)(3) of this section whenever there is a change in the process generating the waste stream that could cause the total annual benzene quantity from facility waste to increase to 1 Mg/yr (1.1 ton/yr) or more.

(c) If the total annual benzene quantity from facility waste is less than 10 Mg/yr (11 ton/yr) but is equal to or greater than 1 Mg/yr (1.1 ton/yr), then the owner or operator shall submit to the Administrator a report that updates the information listed in paragraphs (a)(1) through (a)(3) of this section. The report shall be submitted annually and whenever there is a change in the process generating the waste stream that could cause the total annual benzene quantity from facility waste to increase to 10 Mg/yr (11 ton/yr) or more. If the information in the annual report required by paragraphs (a)(1) through (a)(3) of this section is not changed in the following year, the

owner or operator may submit a statement to that effect.

(d) If the total annual benzene quantity from facility waste is equal to or greater than 10 Mg/yr (11 ton/yr), then the owner or operator shall submit to the Administrator the following reports:

(1) Within 90 days after January 7, 1993, unless a waiver of compliance under §61.11 of this part is granted, or by the date of initial startup for a new source with an initial startup after the effective date, a certification that the equipment necessary to comply with these standards has been installed and that the required initial inspections or tests have been carried out in accordance with this subpart. If a waiver of compliance is granted under §61.11, the certification of equipment necessary to comply with these standards shall be submitted by the date the waiver of compliance expires.

(2) Beginning on the date that the equipment necessary to comply with these standards has been certified in accordance with paragraph (d)(1) of this section, the owner or operator shall submit annually to the Administrator a report that updates the information listed in paragraphs (a)(1) through (a)(3) of this section. If the information in the annual report required by paragraphs (a)(1) through (a)(3) of this section is not changed in the following year, the owner or operator may submit a statement to that effect.

(3) If an owner or operator elects to comply with the requirements of (1,342(c)) (3)(ii), then the report required by paragraph (d)(2) of this section shall include a table identifying each waste stream chosen for exemption and the total annual benzene quantity in these exempted streams.

(4) If an owner or operator elects to comply with the alternative requirements of §61.342(d) of this subpart, then he shall include in the report required by paragraph (d)(2) of this section a table presenting the following information for each process wastewater stream:

(i) Whether or not the process wastewater stream is being controlled for benzene emissions in accordance with the requirements of this subpart; (ii) For each process wastewater stream identified as not being controlled for benzene emissions in accordance with the requirements of this subpart, the table shall report the following information for the process wastewater stream as determined at the point of waste generation: annual waste quantity, range of benzene concentrations, annual average flowweighted benzene concentration, and annual benzene quantity;

(iii) For each process wastewater stream identified as being controlled for benzene emissions in accordance with the requirements of this subpart, the table shall report the following information for the process wastewater stream as determined at the exit to the treatment process: Annual waste quantity, range of benzene concentrations, annual average flow-weighted benzene concentration, and annual benzene quantity.

(5) If an owner or operator elects to comply with the alternative requirements of 61.342(e), then the report required by paragraph (d)(2) of this section shall include a table presenting the following information for each waste stream:

(i) For each waste stream identified as not being controlled for benzene emissions in accordance with the requirements of this subpart; the table shall report the following information for the waste stream as determined at the point of waste generation: annual waste quantity, range of benzene concentrations, annual average flowweighted benzene concentration, and annual benzene quantity;

(ii) For each waste stream identified as being controlled for benzene emissions in accordance with the requirements of this subpart; the table shall report the following information for the waste stream as determined at the applicable location described in §61.355(k)(2): Annual waste quantity, range of benzene concentrations, annual average flow-weighted benzene concentration, and annual benzene quantity.

(6) Beginning 3 months after the date that the equipment necessary to comply with these standards has been certified in accordance with paragraph (d)(1) of this section, the owner or operator shall submit quarterly to the Administrator a certification that all of the required inspections have been carried out in accordance with the requirements of this subpart.

(7) Beginning 3 months after the date that the equipment necessary to comply with these standards has been certified in accordance with paragraph (d)(1) of this section, the owner or operator shall submit a report quarterly to the Administrator that includes:

(i) If a treatment process or wastewater treatment system unit is monitored in accordance with $\S61.354(a)(1)$ of this subpart, then each period of operation during which the concentration of benzene in the monitored waste stream exiting the unit is equal to or greater than 10 ppmw.

(ii) If a treatment process or wastewater treatment system unit is monitored in accordance with $\S61.354(a)(2)$ of this subpart, then each 3-hour period of operation during which the average value of the monitored parameter is outside the range of acceptable values or during which the unit is not operating as designed.

(iii) If a treatment process or wastewater treatment system unit is monitored in accordance with §61.354(b), then each period of operation during which the flow-weighted annual average concentration of benzene in the monitored waste stream entering the unit is equal to or greater than 10 ppmw and/or the total annual benzene quantity is equal to or greater than 1.0 mg/yr.

(iv) For a control device monitored in accordance with §61.354(c) of this subpart, each period of operation monitored during which any of the following conditions occur, as applicable to the control device:

(A) Each 3-hour period of operation during which the average temperature of the gas stream in the combustion zone of a thermal vapor incinerator, as measured by the temperature monitoring device, is more than $28 \,^{\circ}\text{C}$ ($50 \,^{\circ}\text{F}$) below the design combustion zone temperature.

(B) Each 3-hour period of operation during which the average temperature of the gas stream immediately before the catalyst bed of a catalytic vapor 40 CFR Ch. I (7–1–23 Edition)

incinerator, as measured by the temperature monitoring device, is more than 28 °C (50 °F) below the design gas stream temperature, and any 3-hour period during which the average temperature difference across the catalyst bed (i.e., the difference between the temperatures of the gas stream immediately before and after the catalyst bed), as measured by the temperature monitoring device, is less than 80 percent of the design temperature difference.

(C) Each 3-hour period of operation during which the average temperature of the gas stream in the combustion zone of a boiler or process heater having a design heat input capacity less than 44 MW (150×106 BTU/hr), as mesured by the temperature monitoring device, is more than 28 °C (50 °F) below the design combustion zone temperature.

(D) Each 3-hour period of operation during which the average concentration of organics or the average concentration of benzene in the exhaust gases from a carbon adsorber, condenser, or other vapor recovery system is more than 20 percent greater than the design concentration level of organics or benzene in the exhaust gas.

(E) Each 3-hour period of operation during which the temperature of the condenser exhaust vent stream is more than 6 °C (11 °F) above the design average exhaust vent stream temperature, or the temperature of the coolant fluid exiting the condenser is more than 6 °C (11 °F) above the design average coolant fluid temperature at the condenser outlet.

(F) Each period in which the pilot flame of a flare is absent.

(G) Each occurrence when there is a change in the location at which the vent stream is introduced into the flame zone of a boiler or process heater as required by 61.349(a)(2)(i)(C) of this subpart.

(H) Each occurrence when the carbon in a carbon adsorber system that is regenerated directly on site in the control device is not regenerated at the predetermined carbon bed regeneration time.

(I) Each occurrence when the carbon in a carbon adsorber system that is not

regenerated directly on site in the control device is not replaced at the predetermined interval specified in §61.354(c) of this subpart.

(J) Each 3-hour period of operation during which the parameters monitored are outside the range of values specified in 61.349(a)(2)(iv)(C), or any other periods specified by the Administrator for a control device subject to the requirements of 61.349(a)(2)(iv).

(v) For a cover and closed-vent system monitored in accordance with §61.354(g), the owner or operator shall submit a report quarterly to the Administrator that identifies any period in which the pressure in the waste management unit is equal to or greater than atmospheric pressure.

(8) Beginning one year after the date that the equipment necessary to comply with these standards has been certified in accordance with paragraph (d)(1) of this section, the owner or operator shall submit annually to the Administrator a report that summarizes all inspections required by \$61.342through 61.354 during which detectable emissions are measured or a problem (such as a broken seal, gap or other problem) that could result in benzone emissions is identified, including information about the repairs or corrective action taken.

(e) An owner or operator electing to comply with the provisions of \$ 61.351

or 61.352 of this subpart shall notify the Administrator of the alternative standard selected in the report required under §61.07 or §61.10 of this part.

(f) An owner or operator who elects to install and operate the control equipment in §61.351 of this subpart shall comply with the reporting requirements in 40 CFR 60.115b.

(g) An owner or operator who elects to install and operate the control equipment in §61.352 of this subpart shall submit initial and quarterly reports that identify all seal gap measurements, as required in 40 CFR 60.693-2(a), that are outside the prescribed limits.

[55 FR 8346, Mar. 7, 1990; 55 FR 12444, Apr. 3, 1990, as amended at 55 FR 37231, Sept. 10, 1990; 58 FR 3105, Jan. 7, 1993; 65 FR 62161, Oct. 17, 2000]

§61.358 Delegation of authority.

(a) In delegating implementation and enforcement authority to a State under section 112(d) of the Clean Air Act, the authorities contained in paragraph (b) of this section shall be retained by the Administrator and not transferred to a State.

(b) Alternative means of emission limitation under §61.353 of this subpart will not be delegated to States.

§61.359 [Reserved]

§61.359

Appendix K

§63.680

AUTHENTICATED U.S. GOVERNMENT INFORMATION

40 CFR Ch. I (7-1-23 Edition)

TABLE 13 CALIBRATION AND CALITY CONTROL RECEIREMENTS FOR CPMS Continie

Parameter	Minim m acc racy re ire ments	Caî⊡ration re⊡rements
Net Heatin⊟ Va⊞e _y Gas Chromato⊡raph.	As specifie⊟ in Performance Specification 9 of 40 CFR part 60, appen⊡⊡ B	For the procetter in Performance Specification 9 of 40 CFR part 60, appentil B, etcept that a sinter taily mitt etceloattration check can be the tailer than tripicate analysis; the mittipoint cattration can be conducted thartery tather than monthy, and the sampind ine temp perattre mist be maintained at a minimum temperattre of 60 °C rather than 20 °C
Hy⊡o⊡en anaÿ⊡er	±2 percent o er the concentra tion meas⊡re or 0.1 to⊡me percent, whiche er is ereat er.	Specify call ration re irements in yoir site specific CPMS monitorin plan. Call ration re irements sho in follow man factorers recommendations at a minim m. Where feasile, select the samplin location at least two eliaent int diameters from the nearest controlleice, point of pointant eneration, air in leakales, or other point at which a chanie in the point at concentration occirs.

[60 FR 43260, Aug. 18, 1995, as amended at 61 FR 29881, 29882, June 12, 1996; 63 FR 44142, 44143, Aug. 18, 1998; 74 FR 55688, Oct. 28, 2009; 75 FR 37731, June 30, 2010; 80 FR 75269, Dec. 1, 2015; 81 FR 45241, July 13, 2016; 83 FR 60722, Nov. 26, 2018; 85 FR 73893, Nov. 19, 2020]

Subpart DD—National Emission Standards for Hazardous Air Pollutants from Off-Site Waste and Recovery Operations

SOURCE: 61 FR 34158, July 1, 1996, unless otherwise noted.

§63.680 Applicability and designation of affected sources.

(a) The provisions of this subpart apply to the owner and operator of a plant site for which both of the conditions specified in paragraphs (a)(1) and (a)(2) of this section are applicable. If either one of these conditions does not apply to the plant site, then the owner and operator of the plant site are not subject to the provisions of this subpart.

(1) The plant site is a major source of hazardous air pollutant (HAP) emissions as defined in 40 CFR 63.2.

(2) At the plant site is located one or more of operations that receives offsite materials as specified in paragraph (b) of this section and the operations is one of the following waste management operations or recovery operations as specified in paragraphs (a)(2)(i) through (a)(2)(vi) of this section.

(i) A waste management operation that receives off-site material and the operation is regulated as a hazardous waste treatment, storage, and disposal facility (TSDF) under either 40 CFR part 264 or part 265. (ii) A waste management operation that treats wastewater which is an offsite material and the operation is exempted from regulation as a hazardous waste treatment, storage, and disposal facility under 40 CFR 264.1(g)(6) or 40 CFR 265.1(c)(10).

(iii) A waste management operation that treats wastewater which is an offsite material and the operation meets both of the following conditions:

(A) The operation is subject to regulation under either section 402 or 307(b) of the Clean Water Act but is not owned by a "state" or "municipality" as defined by section 502(3) and 502(4), respectively, of the Clean Water Act; and

(B) The treatment of wastewater received from off-site is the predominant activity performed at the plant site.

(iv) A recovery operation that recycles or reprocesses hazardous waste which is an off-site material and the operation is exempted from regulation as a hazardous waste treatment, disposal, and storage facility under 40 CFR 264.1(g)(2) or 40 CFR 265.1(c)(6).

(v) A recovery operation that recycles or reprocesses used solvent which is an off-site material and the operation is not part of a chemical, petroleum, or other manufacturing process that is required to use air emission controls by another subpart of 40 CFR part 63 or 40 CFR part 61.

(vi) A recovery operation that re-refines or reprocesses used oil which is an off-site material and the operation is regulated under 40 CFR 279 subpart F— Standards for Used Oil Processors and Refiners.

(b) For the purpose of implementing this subpart, an off-site material is a material that meets all of the criteria specified in paragraph (b)(1) of this section but is not one of the materials specified in paragraph (b)(2) of this section.

(1) An off-site material is a material that meets all of the criteria specified in paragraphs (b)(1)(i) through (b)(1)(ii) of this section. If any one of these criteria do not apply to the material, then the material is not an off-site material subject to this subpart.

(i) The material is a waste, used oil, or used solvent as defined in §63.681 of this subpart;

(ii) The waste, used oil, or used solvent is not produced or generated within the plant site, but the material is delivered, transferred, or otherwise moved to the plant site from a location outside the boundaries of the plant site; and

(iii) The waste, used oil, or used solvent contains one or more of the hazardous air pollutants (HAP) listed in Table 1 of this subpart based on the composition of the material at the point-of-delivery, as defined in §63.681 of this subpart.

(2) For the purpose of implementing this subpart, the following materials are not off-site materials:

(i) Household waste as defined in 40 CFR 258.2.

(ii) Radioactive mixed waste managed in accordance with all applicable regulations under Atomic Energy Act and Nuclear Waste Policy Act authorities.

(iii) Waste that is generated as a result of implementing remedial activities required under the Resource Conservation and Recovery Act (RCRA) corrective action authorities (RCRA sections 3004(u), 3004(v), or 3008(h)), Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA) authorities, or similar Federal or State authorities.

(iv) Waste containing HAP that is generated by residential households

(e.g., old paint, home garden pesticides) and subsequently is collected as a community service by government agencies, businesses, or other organizations for the purpose of promoting the proper disposal of this waste.

(v) Waste that is transferred from a chemical manufacturing plant or other facility for which the owner or operator of the facility from which the waste is transferred has complied with the provisions of the air emission control standards for process wastewater specified by another subpart of this part. This exemption does not apply to a source which complies with another subpart of this part by transferring its wastewater off-site for control.

(vi) Waste that is transferred from a chemical manufacturing plant, petroleum refinery, or coke by-product recovery plant which is subject to 40 CFR part 61, subpart FF—National Emission Standards for Benzene Waste Operations, and for which both of the following conditions apply to the waste:

(A) The waste is generated at a facility that is not exempted under the provisions of 40 CFR 61.342(a) from meeting the air emission control standards of 40 CFR part 61, subpart FF; and

(B) The owner or operator of the facility from which the waste is transferred has complied with the provisions of 40 CFR 61.342(f)(2).

(vii) Ship ballast water pumped from a ship to an onshore wastewater treatment facility.

(viii) Hazardous waste that is stored for 10 days or less at a transfer facility in compliance with the provisions of 40 CFR 263.12.

(c) Affected sources-(1) Off-site material management units. For each operation specified in paragraphs (a)(2)(i) through (a)(2)(vi) of this section that is located at the plant site, the affected source is the entire group of off-site material management units associated with the operation. An off-site material management unit is a tank, container, surface impoundment, oil-water separator, organic-water separator, or transfer system used to manage off-site material. For the purpose of implementing the standards under this subpart, a unit that meets the definition of a tank or container but also is equipped with a vent that serves as a

process vent for any of the processes listed in paragraphs (c)(2)(i) through (c)(2)(vi) of this section is not an offsite material management unit but instead is a process vent and is to be included in the appropriate affected source group under paragraph (c)(2) of this section. Examples of such a unit may include, but are not limited to, a distillate receiver vessel, a primary condenser, a bottoms receiver vessel, a surge control tank, a separator tank, and a hot well.

(2) Process vents. For each operation specified in paragraphs (a)(2)(i) through (a)(2)(vi) of this section that is located at the plant site, the affected source is the entire group of process equipment associated with the process vents for the processes listed in paragraphs (c)(2)(i) through (c)(2)(vi) of this section.

(i) Distillation process used for the treatment, recycling, or recovery of off-site material. Distillation means a process, either batch or continuous, separating one or more off-site material feed streams into two or more exit streams having different component concentrations from those in the feed stream or streams. The separation is achieved by the redistribution of the components between the liquid and vapor phases as they approach equilibrium within the distillation unit.

(ii) Fractionation process used for the treatment, recycling, or recovery of off-site material. Fractionation means a liquid mixture separation process or method used to separate a mixture of several volatile components of different boiling points in successive stages, each stage removing from the mixture some proportion of one of the components.

(iii) Thin-film evaporation process used for the treatment, recycling, or recovery of off-site material. Thin-film evaporation means a liquid mixture separation process or method that uses a heating surface consisting of a large diameter tube that may be either straight or tapered, horizontal or vertical. Liquid is spread on the tube wall by a rotating assembly of blades that maintain a close clearance from the wall or actually ride on the film of liquid on the wall.

40 CFR Ch. I (7–1–23 Edition)

(iv) Solvent extraction process used for the treatment, recycling, or recovery of off-site material. Solvent extraction means a separation process or method in which a solid or a solution is contacted with a liquid solvent (the material and the solvent being relatively insoluble in each other) to preferentially dissolve and transfer one or more components into the solvent.

(v) Steam stripping process used for the treatment, recycling, or recovery of off-site material. Steam stripping means a liquid mixture separation process or method in which vaporization of the volatile components of a liquid mixture occurs by the introduction of steam directly into the process.

(vi) Gas stripping process used for the treatment, recycling, or recovery of off-site material. Gas stripping means a desorption process or method used to transfer one or more volatile components from a liquid mixture into a gas stream either with or without the application of heat to the liquid. Packed towers, spray towers, and bubble-cap, sieve, or valve-type plate towers are examples of the process configurations used for contacting the gas and a liquid.

(3) Equipment leaks. For each operation specified in paragraphs (a)(2)(i)through (a)(2)(vi) of this section that is located at the plant site, the affected source is the entire group of equipment components for which each component meets all of the conditions specified in paragraphs (c)(3)(i) through (c)(3)(ii) of this section. If any one of these conditions do not apply to an equipment component, then that component is not part of the affected source for equipment leaks.

(i) The equipment component is a pump, compressor, agitator, pressure relief device, sampling connection system, open-ended valve or line, valve, connector, or instrumentation system:

(ii) The equipment component contains or contacts off-site material having a total HAP concentration equal to or greater than 10 percent by weight; and

(iii) The equipment component is intended to operate for 300 hours or more during a calendar year in off-site material service, as defined in §63.681 of this subpart.

(d) Facility-wide exemption. The owner or operator of affected sources subject to this subpart is exempted from the requirements of §§ 63.682 through 63.699 of this subpart in situations when the total annual quantity of the HAP that is contained in the off-site material received at the plant site is less than 1 megagram per year. For a plant site to be exempted under the provisions of this paragraph (d), the owner or operator must meet the requirements in paragraphs (d)(1) through (d)(3) of this section.

(1) The owner or operator must prepare an initial determination of the total annual HAP quantity in the offsite material received at the plant site. This determination is based on the total quantity of the HAP listed in Table 1 of this subpart as determined at the point-of-delivery for each offsite material stream.

(2) The owner or operator must prepare a new determination whenever the extent of changes to the quantity or composition of the off-site material received at the plant site could cause the total annual HAP quantity in the offsite material received at the plant site to exceed the limit of 1 megagram per year.

(3) The owner or operator must maintain documentation to support the owner's or operator's determination of the total annual HAP quantity in the off-site material received at the plant site. This documentation must include the basis and data used for determining the HAP content of the off-site material.

(e) Compliance dates—(1) Existing sources. The owner or operator of an affected source that commenced construction or reconstruction before October 13, 1994, must achieve compliance with the provisions of this subpart on or before the date specified in paragraphs (e)(1)(i), (ii), or (iii) of this section as applicable to the affected source.

(i) For an affected source that commenced construction or reconstruction before October 13, 1994 and receives offsite material for the first time before February 1, 2000, the owner or operator of this affected source must achieve compliance with the provisions of the subpart (except §§63.685(b)(1)(ii), 63.691(b)(2), and 63.691(c)(3)(i) and (ii)) on or before February 1, 2000 unless an extension has been granted by the Administrator as provided in §63.6(i). These existing affected sources shall be in compliance with the tank requirements of §63.685(b)(1)(ii) 2 years after the publication date of the final amendments on March 18, 2015, the equipment leak requirements of §63.691(b)(2) 1 year after the publication date of the final amendments on March 18, 2015, and the pressure relief device monitoring requirements of 63.691(c)(3)(i) and (ii) 3 years after the publication date of the final amendments on March 18, 2015.

(ii) For an affected source that commenced construction or reconstruction before October 13, 1994, but receives offsite material for the first time on or after February 1, 2000, but before March 18, 2015, the owner or operator of the affected source must achieve compliance with the provisions of this subpart (except §§63.685(b)(1)(ii), 63.691(b)(2), and 63.691(c)(3)(i) and (ii)) upon the first date that the affected source begins to manage off-site material. These existing affected sources shall be in compliance with the tank requirements of §63.685(b)(1)(ii) 2 years after the publication date of the final amendments on March 18, 2015, the equipment leak requirements of §63.691(b)(2) 1 year after the publication date of the final amendments on March 18, 2015, and the pressure relief device monitoring requirements of 63.691(c)(3)(i) and (ii) 3 years after the publication date of the final amendments on March 18, 2015.

(iii) For an affected source that commenced construction or reconstruction before October 13, 1994, but receives offsite material for the first time on or after March 18, 2015, the owner or operator of the affected source must achieve compliance with the provisions of $_{\mathrm{this}}$ subpart (except §§ 63.685 (b)(1)(ii), 63.691(b)(2), and 63.691(c)(3)(i) and (ii)) upon the first date that the affected source begins to manage off-site material. These existing affected sources shall be in compliance with the tank requirements of §63.685(b)(1)(ii) 2 years after the publication date of the final amendments on March 18, 2015, the equipment leak requirements of §63.691(b)(2) 1 year after the publication date of the final amendments on March 18, 2015, and the pressure relief device monitoring requirements of §63.691(c)(3)(i) and (ii) 3 years after the publication date of the final amendments on March 18, 2015.

(2) New sources. The owner or operator of an affected source for which construction or reconstruction commences on or after October 13, 1994, must achieve compliance with the provisions of this subpart (except §§63.685(b)(2), 63.691(b)(2), and 63.691(c)(3)(i) and (ii)) on or before July 1, 1996, or upon initial startup of operations, whichever date is later as provided in 40 CFR 63.6(b). New affected sources that commenced construction or reconstruction after October 13, 1994, but on or before July 2, 2014, shall be in compliance with the tank requirements of §63.685(b)(2) 2 years after the publication date of the final amendments, the equipment leak requirements of §63.691(b)(2) 1 year after the publication date of the final amendments, and the pressure relief device monitoring requirements of §63.691(c)(3)(i) and (ii) 3 years after the effective date of the final amendments. New affected sources that commence construction or reconstruction after July 2, 2014, shall be in compliance with the tank requirements of §63.685(b)(2), the equipment leak requirements of §63.691(b)(2), and the pressure relief device monitoring requirements of §63.691(c)(3)(i) and (ii) upon initial startup or by the effective date of the final amendments, whichever is later.

(f) The provisions of 40 CFR part 63, subpart A—General Provisions that apply and those that do not apply to this subpart are specified in Table 2 of this subpart.

(g) Applicability of this subpart. (1) The emission limitations set forth in this subpart and the emission limitations referred to in this subpart shall apply at all times except during periods of non-operation of the affected source (or specific portion thereof) resulting in cessation of the emissions to which this subpart applies.

(2) The owner or operator shall not shut down items of equipment that are required or utilized for compliance with this subpart during times when 40 CFR Ch. I (7–1–23 Edition)

emissions are being routed to such items of equipment, if the shutdown would contravene requirements of this subpart applicable to such items of equipment.

[61 FR 34158, July 1, 1996, as amended at 65 FR 38963, July 20, 1999; 80 FR 14271, Mar. 18, 2015]

§63.681 Definitions.

All terms used in this subpart shall have the meaning given to them in this section, 40 CFR 63.2 of this part, and the Act.

Boiler means an enclosed combustion device that extracts useful energy in the form of steam and is not an incinerator or a process heater.

Bypass means diverting a process vent or closed vent system stream to the atmosphere such that it does not first pass through an emission control device.

Closed-vent system means a system that is not open to the atmosphere and is composed of hard-piping, ductwork, connections, and, if necessary, fans, blowers, or other flow-inducing devices that conveys gas or vapor from an emission point to a control device.

Closure device means a cap, hatch, lid, plug, seal, valve, or other type of fitting that prevents or reduces air pollutant emissions to the atmosphere by blocking an opening in a cover when the device is secured in the closed position. Closure devices include devices that are detachable from the cover (e.g., a sampling port cap), manually operated (e.g., a hinged access lid or hatch), or automatically operated (e.g., a spring-loaded pressure relief valve).

Container means a portable unit used to hold material. Examples of containers include but are not limited to drums, dumpsters, roll-off boxes, bulk cargo containers commonly known as "portable tanks" or "totes", cargo tank trucks, and tank rail cars.

Continuous record means documentation of data values measured at least once every 15 minutes and recorded at the frequency specified in this subpart.

Continuous recorder means a data recording device that either records an instantaneous data value at least once every 15 minutes or records 15-minutes or more frequent block averages.

Continuous seal means a seal that forms a continuous closure that completely covers the space between the edge of the floating roof and the wall of a tank. A continuous seal may be a vapor-mounted seal, liquid-mounted seal, or metallic shoe seal. A continuous seal may be constructed of fastened segments so as to form a continuous seal.

Control device means equipment used for recovering, removing, oxidizing, or destroying organic vapors. Examples of such equipment include but are not limited to carbon adsorbers, condensers, vapor incinerators, flares, boilers, and process heaters.

Cover means a device or system that provides a continuous barrier over the material managed in an off-site material management unit to prevent or reduce air pollutant emissions to the atmosphere. A cover may have openings needed for operation, inspection, sampling, maintenance, and repair of the unit provided that each opening is closed when not in use (e.g., access hatches, sampling ports). A cover may be a separate piece of equipment which can be detached and removed from the unit or a *cover* may be formed by structural features permanently integrated into the design of the unit.

Emission point means an individual tank, surface impoundment, container, oil-water or organic-water separator, transfer system, process vent, or enclosure.

Enclosure means a structure that surrounds a tank or container, captures organic vapors emitted from the tank or container, and vents the captured vapor through a closed vent system to a control device.

External floating roof means a pontoon-type or double-deck type cover that rests on the liquid surface in a tank with no fixed roof.

Fixed roof means a cover that is mounted on a unit in a stationary position and does not move with fluctuations in the level of the liquid managed in the unit.

Flame zone means the portion of the combustion chamber in a boiler or process heater occupied by the flame envelope.

Floating roof means a cover consisting of a double deck, pontoon single

deck, or internal floating cover which rests upon and is supported by the liquid being contained, and is equipped with a continuous seal.

Flow indicator means a device that indicates whether gas is flowing, or whether the valve position would allow gas to flow in a bypass line.

Hard-piping means pipe or tubing that is manufactured and properly installed in accordance with relevant standards and good engineering practices.

Hazardous air pollutants or HAP means the specific organic chemical compounds, isomers, and mixtures listed in Table 1 of this subpart.

Hazardous waste means a waste that is determined to be hazardous under the Resource Conservation and Recovery Act (PL 94-580) (RCRA), as implemented by 40 CFR parts 260 and 261.

In gas/vapor service means that a piece of equipment in off-site material service contains or contacts a gas or vapor at operating conditions.

In heavy liquid service means that a piece of equipment in off-site material service is not in gas/vapor service or in light liquid service.

In light liquid service means that a piece of equipment in off-site material service contains or contacts a liquid that meets the following conditions:

(1) The vapor pressure of one or more of the organic compounds is greater than 0.3 kilopascals at 20 °C;

(2) The total concentration of the pure organic compounds constituents having a vapor pressure greater than 0.3 kilopascals at 20 °C is equal to or greater than 20 percent by weight of the total process stream; and

(3) The fluid is a liquid at operating conditions. Note to *In light liquid service:* Vapor pressures may be determined by the methods described in 40 CFR 60.485(e)(1).

In liquid service means that a piece of equipment in off-site material service is not in gas/vapor service.

Individual drain system means a stationary system used to convey wastewater streams or residuals to a waste management unit or to discharge or disposal. The term includes hard-piping, all drains and junction boxes, together with their associated sewer lines and other junction boxes (e.g.,

40 CFR Ch. I (7-1-23 Edition)

manholes, sumps, and lift stations) conveying wastewater streams or residuals. For the purpose of this subpart, an individual drain system is not a drain and collection system that is designed and operated for the sole purpose of collecting rainfall runoff (e.g., stormwater sewer system) and is segregated from all other individual drain systems.

§63.681

Internal floating roof means a cover that rests or floats on the liquid surface (but not necessarily in complete contact with it inside a tank that has a fixed roof).

Light-material service means the container is used to manage an off-site material for which both of the following conditions apply: the vapor pressure of one or more of the organic constituents in the off-site material is greater than 0.3 kilopascals (kPa) at 20 °C; and the total concentration of the pure organic constituents having a vapor pressure greater than 0.3 kPa at 20 °C is equal to or greater than 20 percent by weight.

Liquid-mounted seal means a foam- or liquid-filled continuous seal mounted in contact with the liquid in a unit.

Maximum HAP vapor pressure means the sum of the individual HAP equilibrium partial pressure exerted by an off-site material at the temperature equal to either: the local maximum monthly average temperature as reported by the National Weather Service when the off-site material is stored or treated at ambient temperature; or the highest calendar-month average temperature of the off-site material when the off-site material is stored at temperatures above the ambient temperature or when the off-site material is stored or treated at temperatures below the ambient temperature. For the purpose of this subpart, maximum HAP vapor pressure is determined using the procedures specified in §63.694(j) of this subpart.

Metallic shoe seal means a continuous seal that is constructed of metal sheets which are held vertically against the wall of the tank by springs, weighted levers, or other mechanisms and is connected to the floating roof by braces or other means. A flexible coated fabric (envelope) spans the annular space between the metal sheet and the floating roof.

No detectable organic emissions means no escape of organics to the atmosphere as determined using the procedure specified in §63.694(k) of this subpart.

Off-site material means a material that meets all of the criteria specified in paragraph §63.680(b)(1) of this subpart but is not one of the materials specified in §63.680(b)(2) of this subpart.

Off-site material management unit means a tank, container, surface impoundment, oil-water separator, organic-water separator, or transfer system used to manage off-site material.

Off-site material service means any time when a pump, compressor, agitator, pressure relief device, sampling connection system, open-ended valve or line, valve, connector, or instrumentation system contains or contacts offsite material.

Off-site material stream means an offsite material produced or generated by a particular process or source such that the composition and form of the material comprising the stream remain consistent. An off-site material stream may be delivered, transferred, or otherwise moved to the plant site in a continuous flow of material (e.g., wastewater flowing through a pipeline) or in a series of discrete batches of material (e.g., a truckload of drums all containing the same off-site material or multiple bulk truck loads of an off-site material produced by the same process).

Oil-water separator means a separator as defined for this subpart that is used to separate oil from water.

Operating parameter value means a minimum or maximum value established for a control device or treatment process parameter which, if achieved by itself or in combination with one or more other operating parameter values, determines that an owner or operator has complied with an applicable emission limitation or standard.

Organic-water separator means a separator as defined for this subpart that is used to separate organics from water.

Plant site means all contiguous or adjoining property that is under common control including properties that are

separated only by a road or other public right-of-way. Common control includes properties that are owned, leased, or operated by the same entity, parent entity, subsidiary, or any combination thereof. A unit or group of units within a contiguous property that are not under common control (e.g., a wastewater treatment unit or solvent recovery unit located at the site but is sold to a different company) is a different plant site.

Point-of-delivery means the point at the boundary or within the plant site where the owner or operator first accepts custody, takes possession, or assumes responsibility for the management of an off-site material stream managed in a waste management operation or recovery operation specified in $\S63.680$ (a)(2)(i) through (a)(2)(vi) of this subpart. The characteristics of an offsite material stream are determined prior to combining the off-site material stream with other off-site material streams or with any other materials.

Point-of-treatment means a point after the treated material exits the treatment process but before the first point downstream of the treatment process exit where the organic constituents in the treated material have the potential to volatilize and be released to the atmosphere. For the purpose of applying this definition to this subpart, the first point downstream of the treatment process exit is not a fugitive emission point due to an equipment leak from any of the following equipment components: Pumps, compressors, valves, connectors, instrumentation systems, or pressure relief devices.

Pressure release means the emission of materials resulting from the system pressure being greater than the set pressure of the pressure relief device. This release can be one release or a series of releases over a short time period.

Pressure relief device or valve means a safety device used to prevent operating pressures from exceeding the maximum allowable working pressure of the process equipment. A common pressure relief device is a spring-loaded pressure relief valve. Devices that are actuated either by a pressure of less than or equal to 2.5 pounds per square inch gauge or by a vacuum are not pressure relief devices.

Process heater means an enclosed combustion device that transfers heat released by burning fuel directly to process streams or to heat transfer liquids other than water.

Process vent means an open-ended pipe, stack, or duct through which a gas stream containing HAP is continuously or intermittently discharged to the atmosphere from any of the processes listed in §63.680(c)(2)(i) through (vi). For the purpose of this subpart, a process vent is none of the following: a pressure relief device; an open-ended line or other vent that is subject to the equipment leak control requirements under §63.691; or a stack or other vent that is used to exhaust combustion products from a boiler, furnace, process heater, incinerator, or other combustion device.

Recovery operation means the collection of off-site material management units, process vents, and equipment components used at a plant site to manage an off-site material stream from the point-of-delivery through the point where the material has been recycled, reprocessed, or re-refined to obtain the intended product or to remove the physical and chemical impurities of concern.

Separator means a waste management unit, generally a tank, used to separate oil or organics from water. A separator consists of not only the separation unit but also the forebay and other separator basins, skimmers, weirs, grit chambers, sludge hoppers, and bar screens that are located directly after the individual drain system and prior to any additional treatment units such as an air flotation unit clarifier or biological treatment unit. Examples of a separator include, but are not limited to, an API separator, parallel-plate interceptor, and corrugated-plate interceptor with the associated ancillary equipment.

Single-seal system means a floating roof having one continuous seal. This seal may be vapor-mounted, liquidmounted, or a metallic shoe seal.

Surface impoundment means a unit that is a natural topographical depression, man-made excavation, or diked

area formed primarily of earthen materials (although it may be lined with man-made materials), which is designed to hold an accumulation of liquids. Examples of surface impoundments include holding, storage, settling, and aeration pits, ponds, and lagoons.

Tank means a stationary unit that is constructed primarily of nonearthen materials (such as wood, concrete, steel, fiberglass, or plastic) which provide structural support and is designed to hold an accumulation of liquids or other materials.

Transfer system means a stationary system for which the predominant function is to convey liquids or solid materials from one point to another point within a waste management operation or recovery operation. For the purpose of this subpart, the conveyance of material using a container (as defined for this subpart) or a self-propelled vehicle (e.g., a front-end loader) is not a transfer system. Examples of a transfer system include but are not limited to a pipeline, an individual drain system, a gravity-operated conveyor (such as a chute), and a mechanically-powered conveyor (such as a belt or screw conveyor).

Temperature monitoring device means a piece of equipment used to monitor temperature and having an accuracy of ± 1 percent of the temperature being monitored expressed in degrees Celsius (°C) or ± 1.2 degrees °C, whichever value is greater.

Treatment process means a process in which an off-site material stream is physically, chemically, thermally, or biologically treated to destroy, degrade, or remove hazardous air pollutants contained in the off-site material. A treatment process can be composed of a single unit (e.g., a steam stripper) or a series of units (e.g., a wastewater treatment system). A treatment process can be used to treat one or more off-site material streams at the same time.

Used oil means any oil refined from crude oil or any synthetic oil that has been used and as a result of such use is contaminated by physical or chemical impurities. This definition is the same definition of "used oil" in 40 CFR 279.1. 40 CFR Ch. I (7-1-23 Edition)

Used solvent means a mixture of aliphatic hydrocarbons or a mixture of one and two ring aromatic hydrocarbons that has been used as a solvent and as a result of such use is contaminated by physical or chemical impurities.

Vapor-mounted seal means a continuous seal that is mounted such that there is a vapor space between the liquid in the unit and the bottom of the seal.

Volatile organic hazardous air pollutant concentration or VOHAP concentration means the fraction by weight of those compounds listed in Table 1 of this subpart that are in an off-site material as measured using Method 305 in appendix A of this part and expressed in terms of parts per million (ppm). As an alternative to using Method 305, an owner or operator may determine the HAP concentration of an off-site material using any one of the other test methods specified in §63.694(b)(2)(ii) of this subpart. When a test method specified in §63.694(b)(2)(ii) of this subpart other than Method 305 is used to determine the speciated HAP concentration of an off-site material, the individual compound concentration may be adjusted by the corresponding f_{m305} value listed in Table 1 of this subpart to determine a VOHAP concentration.

Waste means a material generated from industrial, commercial, mining, or agricultural operations or from community activities that is discarded, discharged, or is being accumulated, stored, or physically, chemically, thermally, or biologically treated prior to being discarded or discharged.

Waste management operation means the collection of off-site material management units, process vents, and equipment components used at a plant site to manage an off-site material stream from the point-of-delivery to the point where the waste exits or is discharged from the plant site or the waste is placed for on-site disposal in a unit not subject to this subpart (e.g., a waste incinerator, a land disposal unit).

Waste stabilization process means any physical or chemical process used to either reduce the mobility of hazardous constituents in a waste or eliminate free liquids as determined by Test

Method 9095-Paint Filter Liquids Test in "Test Methods for Evaluating Solid Waste, Physical/Chemical Methods, EPA Publication No. SW-846, Third Edition, September 1986, as amended by Update I, November 15, 1992. (As an alternative, an owner or operator may use any more recent, updated version of Method 9095 approved by the EPA.) A waste stabilization process includes mixing the waste with binders or other materials and curing the resulting waste and binder mixture. Other synonymous terms used to refer to this process are "waste fixation" or "waste solidification." A waste stabilization process does not include the adding of absorbent materials to the surface of a waste, without mixing, agitation, or subsequent curing, to absorb free liquid.

[61 FR 34158, July 1, 1996, as amended at 64 FR 38964, July 20, 1999; 80 FR 14272, Mar. 18, 2015]

§63.682 [Reserved]

§63.683 Standards: General.

(a) The general standards under this section apply to owners and operators of affected sources as designated in §63.680(c) of this subpart.

(b) Off-site material management units. (1) For each off-site material management unit that is part of an affected source, the owner or operator must meet the requirements in either paragraph (b)(1)(i), (b)(1)(i), or (b)(1)(ii) of this section except for those off-site material management units exempted under paragraph (b)(2) of this section.

(i) The owner or operator controls air emissions from the off-site material management unit in accordance with the applicable standards specified in §§ 63.685 through 63.689 of this subpart.

(ii) The owner or operator removes or destroys HAP in the off-site material before placing the material in the offsite material management unit by treating the material in accordance with the standards specified in §63.684 of this subpart.

(iii) The owner or operator determines before placing off-site material in the off-site material management unit that the average VOHAP concentration of the off-site material is less than 500 parts per million by

weight (ppmw) at the point-of-delivery. The owner or operator must perform an initial determination of the average VOHAP concentration of the off-site material using the procedures specified in §63.694(b) of this subpart. This initial determination must be performed either before the first time any portion of the off-site material stream is placed in the unit or by the compliance date, whichever date is later. Thereafter, the owner or operator must review and update, as necessary, this determination at least once every calendar year following the date of the initial determination for the off-site material stream.

(2) An off-site material management unit is exempted from the requirements in paragraph (b)(1) of this section when the owner or operator meets one of the exemptions provided in paragraphs (b)(2)(i) through (b)(2)(iv) of this section as applicable to the unit.

(i) An off-site material management unit is exempted from the requirements in paragraph (b)(1) of this section if the off-site material management unit is also subject to another subpart under 40 CFR part 63 or 40 CFR part 61, and the owner or operator is controlling the HAP listed in Table 1 of this subpart that are emitted from the unit in compliance with the provisions specified in the other applicable subpart of1.

(ii) At the discretion of the owner or operator, one or a combination of offsite material management units may be exempted from the requirements in paragraph (b)(1) of this section when these units meet the condition that the total annual quantity of HAP contained in the off-site material placed in the units exempted under this paragraph (b)(2)(ii) is less than 1 megagram per year. For the off-site material management units selected by the owner or operator to be exempted from the requirements in paragraph (b)(1) of this section, the owner or operator must meet the requirements in paragraphs (b)(2)(ii)(A) and (b)(2)(ii)(B) of this section. An owner or operator may change the off-site material management units selected to be exempted under this paragraph (b)(2)(ii) by preparing a new designation for the exempt-units as required by paragraph (b)(2)(ii)(A) of this section and performing a new determination as required by paragraph (b)(2)(ii)(B) of this section.

(A) The owner or operator must designate each of the off-site material management units selected by the owner or operator to be exempt under paragraph (b)(2)(ii) of this section by either submitting to the Administrator a written notification identifying the exempt-units or permanently marking the exempt-units at the plant site. If an owner or operator chooses to prepare and submit a written notification, this notification must include a site plan, process diagram, or other appropriate documentation identifying each of the exempt-units. If an owner or operator chooses to permanently mark the exempt-units, each exempt-unit must be marked in such a manner that it can be readily identified as an exempt-unit from the other off-site material management units located at the plant site.

(B) The owner or operator must prepare an initial determination of the total annual HAP quantity in the offsite material placed in the units exempted under this paragraph (b)(2)(ii). This determination is based on the total quantity of the HAP listed in Table 1 of this subpart as determined at the point where the off-site material is placed in each exempted unit. The owner or operator must perform a new determination whenever the extent of changes to the quantity or composition of the off-site material placed in the exempted units could cause the total annual HAP content in the off-site material to exceed 1 megagram per year. The owner or operator must maintain documentation to support the most recent determination of the total annual HAP quantity. This documentation must include the basis and data used for determining the HAP content of the off-site material.

(iii) A tank or surface impoundment is exempted from the requirements in paragraph (b)(1) of this section if the unit is used for a biological treatment process that meets the requirements in either paragraph (b)(2)(iii)(A) or (b)(2)(iii)(B) of this section and the owner or operator complies with the 40 CFR Ch. I (7–1–23 Edition)

 $\begin{array}{ll} \mbox{monitoring} & \mbox{requirements} & \mbox{in} \\ \S 63.684(e)(4) \mbox{ of this subpart.} \end{array}$

(A) The HAP biodegradation efficiency (R_{bio}) for the biological treatment process is equal to or greater than 95 percent. The HAP biodegradation efficiency (R_{bio}) shall be determined in accordance with the requirements of §63.694(h) of this subpart.

(B) The total actual HAP mass removal rate (MR_{bio}) for the off-site material treated by the biological treatment process is equal to or greater than the required HAP mass removal rate (RMR) for the off-site material. The total actual HAP mass removal rate (MR_{bio}) must be determined in accordance with the requirements of $\S63.694(i)$ of this subpart. The required HAP mass removal rate (RMR) must be determined in accordance with the requirements of $\S63.694(e)$ of this subpart.

(iv) An off-site material management unit is exempted from the requirements in paragraph (b)(1) of this section if the off-site material placed in the unit is a hazardous waste that meets the conditions specified in either paragraph (b)(2)(iv)(A) or (b)(2)(iv)(B) of this section.

(A) The hazardous waste meets the numerical organic concentration limits, applicable to the hazardous waste, as specified in 40 CFR part 268—Land Disposal Restrictions, listed in the table, "Treatment Standards for Hazardous Waste" in 40 CFR 268.40.

(B) The organic hazardous constituents in the hazardous waste have been treated by the treatment technology established by the EPA for the hazardous waste in 40 CFR 268.42(a), or have been removed or destroyed by an equivalent method of treatment approved by the EPA under 40 CFR 268.42(b).

(v) A tank used for bulk feed of offsite material to a waste incinerator is exempted from the requirements specified in paragraph (b)(1) of this section if the tank meets all of the conditions specified in paragraphs (b)(2)(v)(A) through (b)(2)(v)(C) of this section.

(A) The tank is located inside an enclosure vented to a control device that is designed and operated in accordance with all applicable requirements specified under 40 CFR part 61, subpart FF—

National Emission Standards for Benzene Waste Operations for a facility at which the total annual benzene quantity from the facility waste is equal to or greater than 10 megagrams per year;

(B) The enclosure and control device serving the tank were installed and began operation prior to July 1, 1996; and

(C) The enclosure is designed and operated in accordance with the criteria for a permanent total enclosure as specified in "Procedure T-Criteria for and Verification of a Permanent or Temporary Total Enclosure" under 40 CFR 52.741, appendix B. The enclosure may have permanent or temporary openings to allow worker access; passage of material into or out of the enclosure by conveyor, vehicles, or other mechanical or electrical equipment; or to direct air flow into the enclosure. The owner or operator must annually perform the verification procedure for the enclosure as specified in Section 5.0 to "Procedure T-Criteria for and Verification of a Permanent or Temporary Total Enclosure."

(c) *Process vents.* (1) For each process vent that is part of an affected source, the owner or operator must meet the requirements in either paragraph (c)(1)(i) or (c)(1)(i) of this section except for those process vents exempted under paragraph (c)(2) of this section.

(i) The owner or operator controls air emissions from the process vent in accordance with the standards specified in §63.690 of this subpart.

(ii) The owner or operator determines before placing off-site material in the process equipment associated with the process vent that the average VOHAP concentration of the off-site material is less than 500 ppmw at the point-ofdelivery. The owner or operator must perform an initial determination of the average VOHAP concentration of the off-site material using the procedures specified in §63.694(b) of this subpart before any portion of the off-site material stream is placed in the unit. Thereafter, the owner or operator must review and update, as necessary, this determination at least once every calendar year following the date of the initial determination for the off-site material stream.

(2) A process vent is exempted from the requirements of paragraph (c)(1) of this section when the owner or operator meets one of the exemptions provided in paragraphs (c)(2)(i) through (c)(2)(iii) of this section.

(i) A process vent is exempted from the requirements in paragraph (c)(1) of this section if the process vent is also subject to another subpart under part 63 or 40 CFR part 61, and the owner or operator is controlling the HAP listed in Table 1 of this subpart that are emitted from the process vent in compliance with the provisions specified in the other applicable subpart under part 61 or part 63.

(ii) A process vent is exempted from the requirements specified in paragraph (c)(1) of this section if the owner or operator determines that the process vent stream flow rate is less than 0.005 cubic meters per minute (m³/min) at standard conditions (as defined in 40 CFR 63.2). The process vent stream flow rate shall be determined in accordance with the procedures specified in §63.694(m) of this subpart. Documentation must be prepared by the owner or operator and maintained at the plant site to support the determination of the process vent stream flow rate. This documentation must include identification of each process vent exempted under this paragraph and the test results used to determine the process vent stream flow rate.

(iii) A process vent is exempted from the requirements specified in paragraph (c)(1) of this section if the owner or operator determines that the process vent stream flow rate is less than 6.0 m³/min at standard conditions (as defined in 40 CFR 63.2) and the total HAP concentration is less than 20 ppmv. The process vent stream flow rate and total HAP concentration shall be determined in accordance with the procedures specified in §63.694(m) of this subpart. Documentation must be prepared by the owner or operator and maintained at the plant site to support the determination of the process vent stream flow rate and total HAP concentration. This documentation must include identification of each process vent exempted under this paragraph (c)(2)(iii) and the test results used to determine the process vent stream flow

rate and total HAP concentration. The owner or operator must perform a new determination of the process vent stream flow rate and total HAP concentration when the extent of changes to operation of the unit on which the process vent is used could cause either the process vent stream flow rate to exceed the limit of 6.0 m³/min or the total HAP concentration to exceed the limit of 20 ppmv.

(d) Equipment leaks. The owner or operator must control equipment leaks from each equipment component that is part of the affected source specified in 63.680(c)(3) of this subpart by implementing leak detection and control measures in accordance with the standards specified in 63.691 of this subpart.

(e) General duty. At all times, the owner or operator must operate and maintain any affected source, including associated air pollution control equipment and monitoring equipment, in a manner consistent with safety and good air pollution control practices for minimizing emissions. The general duty to minimize emissions does not require the owner operator to make any further efforts to reduce emissions if levels required by the applicable standard have been achieved. Determination of whether a source is operating in compliance with operation and maintenance requirements will be based on information available to the Administrator, which may include, but is not limited to, monitoring results, review of operation and maintenance procedures, review of operation and maintenance records, and inspection of the source.

(f) In addition to the cases listed in (63.695(e))(4), deviation means any of the cases listed in paragraphs (f)(1) through (6) of this section.

(1) Any instance in which an affected source subject to this subpart, or an owner or operator of such a source, fails to meet any requirement or obligation established by this subpart, including, but not limited to, any emission limit, operating limit or work practice standard.

(2) When a performance test indicates that emissions of a pollutant in Table 1 to this subpart are exceeding the emission standard for the pollutant specified in Table 1 to this subpart. 40 CFR Ch. I (7–1–23 Edition)

(3) When the average value of a monitored operating parameter, based on the data averaging period for compliance specified in §63.695, does not meet the operating limit specified in §63.693.

(4) When an affected source discharges directly into the atmosphere from any of the sources specified in paragraphs (f)(4)(i) and (ii) of this section.

(i) A pressure relief device, as defined in §63.681.

(ii) A bypass, as defined in §63.681.

(5) Any instance in which the affected source subject to this subpart, or an owner or operator of such a source, fails to meet any term or condition specified in paragraph (f)(5)(i) or (ii) of this section.

(i) Any term or condition that is adopted to implement an applicable requirement in this subpart.

(ii) Any term or condition relating to compliance with this subpart that is included in the operating permit for an affected source to obtain such a permit.

(6) Any failure to collect required data, except for periods of monitoring system malfunctions, repairs associated with monitoring system malfunctions, and required monitoring system quality assurance or quality control activities (including, as applicable, calibration checks and required zero and span adjustments).

[64 FR 38965, July 20, 1999, as amended at 80 FR 14272, Mar. 18, 2015]

§63.684 Standards: Off-site material treatment.

(a) The provisions of this section apply to the treatment of off-site material to remove or destroy HAP for which §63.683(b)(1)(ii) of this subpart references the requirements of this section for such treatment.

(b) The owner or operator shall remove or destroy the HAP contained in off-site material streams to be managed in the off-site material management unit in accordance with §63.683(b)(1)(ii) of this subpart using a treatment process that continuously achieves, under normal operations, one or more of the performance levels specified in paragraphs (b)(1) through (b)(5) of this section (as applicable to the type of treatment process) for the

range of off-site material stream compositions and quantities expected to be treated.

(1) VOHAP concentration. The treatment process shall reduce the VOHAP concentration of the off-site material using a means, other than by dilution, to achieve one of the following performance levels, as applicable:

(i) In the case when every off-site material stream entering the treatment process has an average VOHAP concentration equal to or greater than 500 ppmw at the point-of-delivery, then the VOHAP concentration of the off-site material shall be reduced to a level that is less than 500 ppmw at the pointof-treatment.

(ii) In the case when off-site material streams entering the treatment process are a mixture of off-site material streams having an average VOHAP concentration equal to or greater than 500 ppmw at the point-of-delivery with off-site material streams having average VOHAP concentrations less than 500 ppmw at the point-of-delivery, then the VOHAP concentration of the offsite material must be reduced to a level at the point-of-treatment that meets the performance level specified in either paragraph (b)(1)(ii)(A) or (B) of this section.

(A) Less than the VOHAP concentration limit (C_R) established for the treatment process using the procedure specified in §63.694(d); or

(B) Less than the lowest VOHAP concentration determined for each of the off-site material streams entering the treatment process as determined by the VOHAP concentration of the offsite material at the point-of-delivery.

(2) HAP mass removal. The treatment process shall achieve a performance level such that the total quantity of HAP actually removed from the offsite material stream (MR) is equal to or greater than the required mass removal (RMR) established for the offsite material stream using the procedure specified in $\S63.694(e)$ of this subpart. The MR for the off-site material streams shall be determined using the procedures specified in $\S63.694(f)$ of this subpart.

(3) HAP reduction efficiency. For any treatment process except a treatment process that uses biological degrada-

tion and is performed in an open tank or surface impoundment, the treatment process must achieve the applicable performance level specified in either paragraph (b)(3)(i) or (b)(3)(ii) of this section.

(i) In the case when the owner or operator determines that off-site material stream entering the treatment process has an average VOHAP concentration less than 10,000 ppmw at the point-of-delivery, then the treatment process shall achieve a performance level such that the total quantity of HAP in the off-site material stream is reduced by 95 percent or more. The HAP reduction efficiency (R) for the treatment process shall be determined using the procedure specified in §63.694(g) of this subpart. The average VOHAP concentration of the off-site material stream at the point-of-delivery shall be determined using the procedure specified in §63.694(b) of this subpart.

(ii) In the case when the off-site material stream entering the treatment process has an average VOHAP concentration equal to or greater than 10,000 ppmw at the point-of-delivery, then the treatment process shall achieve a performance level such that the total quantity of HAP in the offsite material stream is reduced by 95 percent or more, and the average VOHAP concentration of the off-site material at the point-of-treatment is less than 100 parts per million by weight (ppmw). The HAP reduction efficiency (R) for the treatment process shall be determined using the procedure specified in §63.694(g) of this subpart. The average VOHAP concentration of the off-site material stream at the point-of-treatment shall be determined using the procedure specified in §63.694(c) of this subpart.

(4) Biological degradation performed in an open tank or surface impoundment. A treatment process using biological degradation and performed in an open tank or surface impoundment must achieve the performance level specified in either paragraph (b)(4)(i) or (b)(4)(ii) of this section.

(i) The HAP reduction efficiency (R) for the treatment process is equal to or greater than 95 percent, and the HAP biodegradation efficiency (R_{bio}) for the

treatment process is equal to or greater than 95 percent. The HAP reduction efficiency (R) shall be determined using the procedure specified in $\S 63.694(g)$ of this subpart. The HAP biodegradation efficiency (R_{bio}) shall be determined in accordance with the requirements of $\S 63.694(h)$ of this subpart.

(ii) The total quantity of HAP actually removed from the off-site material stream by biological degradation (MR_{bio}) shall be equal to or greater than the required mass removal (RMR) established for the off-site material stream using the procedure specified in $\S63.694(e)$ of this subpart. The MR_{bio} of the off-site material stream shall be determined using the procedures specified in $\S63.694(i)$ of this subpart.

(5) Incineration. The treatment process must destroy the HAP contained in the off-site material stream using one of the combustion devices specified in paragraphs (b)(5)(i) through (v) of this section.

(i) An incinerator for which the owner or operator has been issued a final permit under 40 CFR part 270, and the incinerator is designed and operated in accordance with the requirements of 40 CFR part 264, subpart O— Incinerators, or

(ii) An incinerator for which the owner or operator has certified compliance with the interim status requirements of 40 CFR part 265, subpart O— Incinerators.

(iii) A boiler or industrial furnace for which the owner or operator has been issued a final permit under 40 CFR part 270, and the combustion unit is designed and operated in accordance with the requirements of 40 CFR part 266, subpart H—Hazardous Waste Burned in Boilers and Industrial Furnaces.

(iv) A boiler or industrial furnace for which the owner or operator has certified compliance with the interim status requirements of 40 CFR part 266, subpart H Hazardous Waste Burned in Boilers and Industrial Furnaces.

(v) An incinerator, boiler, or industrial furnace for which the owner or operator has submitted a Notification of Compliance under \$63.1207(j) and 63.1210(d) and complies with the requirements of subpart EEE of this part at all times (including times when nonhazardous waste is being burned). 40 CFR Ch. I (7-1-23 Edition)

(c) For a treatment process that removes the HAP from the off-site material by a means other than thermal destruction or biological degradation to achieve one of the performances levels specified in paragraph (b)(1), (b)(2), or (b)(3) of this section, the owner or operator shall manage the HAP removed from the off-site material in such a manner to minimize release of these HAP to the atmosphere, to the extent practical. Examples of HAP emission control measures that meet the requirements of this paragraph include managing the HAP removed from the off-site material in units that use air emission controls in accordance with the standards specified in §§63.685 through 63.689 of this subpart, as applicable to the unit.

(d) When the owner or operator treats the off-site material to meet one of the performance levels specified in paragraphs (b)(1) through (b)(4) of this section, the owner or operator shall demonstrate that the treatment process achieves the selected performance level for the range of expected off-site material stream compositions expected to be treated. An initial demonstration shall be performed as soon as possible but no later than 30 days after first time an owner or operator begins using the treatment process to manage offsite material streams in accordance with the requirements of either §63.683(b)(1)(ii) or §63.683(b)(2)(ii) of this subpart as applicable to the affected off-site material management unit or process equipment. Thereafter, the owner or operator shall review and update, as necessary, this demonstration at least once every calendar year following the date of the initial demonstration.

(e) When the owner or operator treats the off-site material to meet one of the performance levels specified in paragraphs (b)(1) through (b)(4) of this section, the owner or operator shall ensure that the treatment process is achieving the applicable performance requirements by continuously monitoring the operation of the process when it is used to treat off-site material by complying with paragraphs (e)(1) through (e)(3) or, for biological treatment units, paragraph (e)(4) of this section:

(1) A continuous monitoring system shall be installed and operated for each treatment that measures operating parameters appropriate for the treatment process technology. This system shall include a continuous recorder that records the measured values of the selected operating parameters. The monitoring equipment shall be installed, calibrated, and maintained in accordance with the equipment manufacturer's specifications. The continuous recorder shall be a data recording device that is capable of recording either an instantaneous data value at least once every 15 minutes or an average value for intervals of 15 minutes or less.

(2) For each monitored operating parameter, the owner or operator shall establish a minimum operating parameter value or a maximum operating parameter value, as appropriate, to define the range of conditions at which the treatment process must be operated to continuously achieve the applicable performance requirements of this section.

(3) When the treatment process is operating to treat off-site material, the owner or operator shall inspect the data recorded by the continuous monitoring system on a routine basis and operate the treatment process such that the actual value of each monitored operating parameter is greater than the minimum operating parameter value or less than the maximum operating parameter value, as appropriate, established for the treatment process.

(4) When the treatment process is a biological treatment process that is complying with paragraph (b)(4) of this section, the owner or operator must establish and implement a written procedure to monitor the appropriate parameters that demonstrate proper operation of the biological treatment unit in accordance with the evaluation required in §63.694(h) of this subpart. The written procedure must list the operating parameters that will be monitored and state the frequency of monitoring to ensure that the biological treatment unit is operating between the minimum operating parameter values and maximum operating parameter values to establish that the biological

treatment unit is continuously achieving the performance requirement.

(f) The owner or operator must maintain records for each treatment process in accordance with the requirements of §63.696(a) of this subpart.

(g) The owner or operator must prepare and submit reports for each treatment process in accordance with the requirements of §63.697(a) of this subpart.

(h) The Administrator may at any time conduct or require that the owner or operator conduct testing necessary to demonstrate that a treatment process is achieving the applicable performance requirements of this section. The testing shall be conducted in accordance with the applicable requirements of this section. The Administrator may elect to have an authorized representative observe testing conducted by the owner or operator.

[61 FR 34158, July 1, 1996, as amended at 64 FR 38967, July 20, 1999; 66 FR 1266, Jan. 8, 2001; 68 FR 37351, June 23, 2003; 80 FR 14273, Mar. 18, 2015]

§63.685 Standards: Tanks.

(a) The provisions of this section apply to the control of air emissions from tanks for which 63.683(b)(1)(i) of this subpart references the use of this section for such air emission control.

(b) According to the date an affected source commenced construction or reconstruction and the date an affected source receives off-site material for the first time as established in \S 63.680(e)(i) through (iii), the owner or operator shall control air emissions from each tank subject to this section in accordance with either paragraph (b)(1)(i) or (ii) of this section.

(1)(i) For a tank that is part of an existing affected source but the tank is not used for a waste stabilization process as defined in §63.681, the owner or operator shall determine whether the tank is required to use either Tank Level 1 controls or Tank Level 2 controls as specified for the tank by Table 3 of this subpart based on the off-site material maximum HAP vapor pressure and the tank's design capacity. The owner or operator shall control air emissions from a tank required by Table 3 to use Tank Level 1 controls in accordance with the requirements of paragraph (c) of this section. The owner or operator shall control air emissions from a tank required by Table 3 to use Tank Level 2 controls in accordance with the requirements of paragraph (d) of this section.

(ii) For a tank that is part of an existing affected source but the tank is not used for a waste stabilization process as defined in §63.681, the owner or operator shall determine whether the tank is required to use either Tank Level 1 controls or Tank Level 2 controls as specified for the tank by Table 4 of this subpart based on the off-site material maximum HAP vapor pressure and the tank's design capacity. The owner or operator shall control air emissions from a tank required by Table 4 to use Tank Level 1 controls in accordance with the requirements of paragraph (c) of this section. The owner or operator shall control air emissions from a tank required by Table 4 to use Tank Level 2 controls in accordance with the requirements of paragraph (d) of this section.

(2) For a tank that is part of a new affected source but the tank is not used for a waste stabilization process as defined in §63.681, the owner or operator shall determine whether the tank is required to use either Tank Level 1 controls or Tank Level 2 controls as specified for the tank by Table 5 of this subpart based on the off-site material maximum HAP vapor pressure and the tank's design capacity. The owner or operator shall control air emissions from a tank required by Table 5 to use Tank Level 1 controls in accordance with the requirements of paragraph (c) of this section. The owner or operator shall control air emissions from a tank required by Table 5 to use Tank Level 2 controls in accordance with the requirements of paragraph (d) of this section.

(3) For a tank that is used for a waste stabilization process, the owner or operator shall control air emissions from the tank by using Tank Level 2 controls in accordance with the requirements of paragraph (d) of this section.

(c) Owners and operators controlling air emissions from a tank using Tank Level 1 controls shall meet the following requirements: 40 CFR Ch. I (7–1–23 Edition)

(1) The owner or operator shall determine the maximum HAP vapor pressure for an off-site material to be managed in the tank using Tank Level 1 controls before the first time the offsite material is placed in the tank. The maximum HAP vapor pressure shall be determined using the procedures specified in §63.694(j). Thereafter, the owner or operator shall perform a new determination whenever changes to the offsite material managed in the tank could potentially cause the maximum HAP vapor pressure to increase to a level that is equal to or greater than the maximum HAP vapor pressure limit for the tank design capacity category specified in Table 3, Table 4, or Table 5 of this subpart, as applicable to the tank.

(2) The owner or operator must control air emissions from the tank in accordance with the requirements in either paragraph (c)(2)(i), (c)(2)(ii), or (c)(2)(iii) of this section, as applicable to the tank.

(i) The owner or operator controls air emissions from the tank in accordance with the provisions specified in subpart OO of this part—National Emission Standards for Tanks—Level 1, except that $\S63.902(c)(2)$ and (3) shall not apply for the purposes of this subpart.

(ii) As an alternative to meeting the requirements in paragraph (c)(2)(i) of this section, an owner or operator may control air emissions from the tank in accordance with the provisions for Tank Level 2 controls as specified in paragraph (d) of this section.

(iii) As an alternative to meeting the requirements in paragraph (c)(2)(i) of this section when a tank is used as an interim transfer point to transfer offsite material from containers to another off-site material management unit, an owner or operator may control air emissions from the tank in accordance with the requirements in paragraphs (c)(2)(iii)(A) and (c)(2)(iii)(B) of this section. An example of such a tank is an in-ground tank into which organic-contaminated debris is dumped from roll-off boxes or dump trucks, and then this debris is promptly transferred from the tank to a macroencapsulation unit by a backhoe. (A) During those periods of time

(A) During those periods of time when the material transfer activity is

occurring, the tank may be operated without a cover.

(B) At all other times, air emissions from the tank must be controlled in accordance with the provisions specified in subpart OO of this part—National Emission Standards for Tanks—Level 1, with the exceptions specified in paragraphs (c)(2)(iii)(B)(1) and (2) of this section.

(1) Where 63.902(c)(2) provides an exception for a spring-loaded pressurevacuum relief valve, conservation vent, or similar type of pressure relief device which vents to the atmosphere, only a conservation vent shall be eligible for the exception for the purposes of this subpart.

(2) Section 63.902(c)(3) shall not apply for the purposes of this subpart.

(d) Owners and operators controlling air emissions from a tank using Tank Level 2 controls shall use one of the following tanks:

(1) A fixed-roof tank equipped with an internal floating roof in accordance with the requirements specified in paragraph (e) of this section;

(2) A tank equipped with an external floating roof in accordance with the requirements specified in paragraph (f) of this section;

(3) A tank vented through a closedvent system to a control device in accordance with the requirements specified in paragraph (g) of this section;

(4) A pressure tank designed and operated in accordance with the requirements specified in paragraph (h) of this section; or

(5) A tank located inside an enclosure that is vented through a closed-vent system to an enclosed combustion control device in accordance with the requirements specified in paragraph (i) of this section.

(e) The owner or operator who elects to control air emissions from a tank using a fixed-roof with an internal floating roof shall meet the requirements specified in paragraphs (e)(1) through (e)(3) of this section.

(1) The tank shall be equipped with a fixed roof and an internal floating roof in accordance with the following requirements:

(i) The internal floating roof shall be designed to float on the liquid surface

except when the floating roof must be supported by the leg supports.

(ii) The internal floating roof shall be equipped with a continuous seal between the wall of the tank and the floating roof edge that meets either of the following requirements:

(A) A single continuous seal that is either a liquid-mounted seal or a metallic shoe seal, as defined in §63.681 of this subpart; or

(B) Two continuous seals mounted one above the other. The lower seal may be a vapor-mounted seal.

(iii) The internal floating roof shall meet the following specifications:

(A) Each opening in a noncontact internal floating roof except for automatic bleeder vents (vacuum breaker vents) and the rim space vents is to provide a projection below the liquid surface.

(B) Each opening in the internal floating roof shall be equipped with a gasketed cover or a gasketed lid except for leg sleeves, automatic bleeder vents, rim space vents, column wells, ladder wells, sample wells, and stub drains.

(C) Each penetration of the internal floating roof for the purpose of sampling shall have a slit fabric cover that covers at least 90 percent of the opening.

(D) Each automatic bleeder vent and rim space vent shall be gasketed.

(E) Each penetration of the internal floating roof that allows for passage of a ladder shall have a gasketed sliding cover.

(F) Each penetration of the internal floating roof that allows for passage of a column supporting the fixed roof shall have a flexible fabric sleeve seal or a gasketed sliding cover.

(2) The owner or operator shall operate the tank in accordance with the following requirements:

(i) When the floating roof is resting on the leg supports, the process of filling, emptying, or refilling shall be continuous and shall be accomplished as soon as practical.

(ii) Automatic bleeder vents are to be set closed at all times when the roof is floating, except when the roof is being floated off or is being landed on the leg supports. (iii) Prior to filling the tank, each cover, access hatch, gauge float well or lid on any opening in the internal floating roof shall be bolted or fastened closed (i.e., no visible gaps). Rim spaces vents are to be set to open only when the internal floating roof is not floating or when the pressure beneath the rim exceeds the manufacturer's recommended setting.

(3) The owner or operator shall inspect the internal floating roof in accordance with the procedures specified in §63.695(b) of this subpart.

(f) The owner or operator who elects to control tank emissions by using an external floating roof shall meet the requirements specified in paragraphs (f)(1) through (f)(3) of this section.

(1) The owner or operator shall design the external floating roof in accordance with the following requirements:

(i) The external floating roof shall be designed to float on the liquid surface except when the floating roof must be supported by the leg supports.

(ii) The floating roof shall be equipped with two continuous seals, one above the other, between the wall of the tank and the roof edge. The lower seal is referred to as the primary seal, and the upper seal is referred to as the secondary seal.

(A) The primary seal shall be a liquid-mounted seal or a metallic shoe seal, as defined in §63.681 of this subpart. The total area of the gaps between the tank wall and the primary seal shall not exceed 212 square centimeters (cm2) per meter of tank diameter, and the width of any portion of these gaps shall not exceed 3.8 centimeters (cm). If a metallic shoe seal is used for the primary seal, the metallic shoe seal shall be designed so that one end extends into the liquid in the tank and the other end extends a vertical distance of at least 61 centimeters (24 inches) above the liquid surface.

(B) The secondary seal shall be mounted above the primary seal and cover the annular space between the floating roof and the wall of the tank. The total area of the gaps between the tank wall and the secondary seal shall not exceed 21.2 square centimeters (cm^2) per meter of tank diameter, and 40 CFR Ch. I (7–1–23 Edition)

the width of any portion of these gaps shall not exceed 1.3 centimeters (cm).

(iii) The external floating roof shall be meet the following specifications:

(A) Except for automatic bleeder vents (vacuum breaker vents) and rim space vents, each opening in a noncontact external floating roof shall provide a projection below the liquid surface.

(B) Except for automatic bleeder vents, rim space vents, roof drains, and leg sleeves, each opening in the roof shall be equipped with a gasketed cover, seal, or lid.

(C) Each access hatch and each gauge float wells shall be equipped with covers designed to be bolted or fastened when the cover is secured in the closed position.

(D) Each automatic bleeder vent and each rim space vents shall be equipped with a gasket.

(E) Each roof drain that empties into the liquid managed in the tank shall be equipped with a slotted membrane fabric cover that covers at least 90 percent of the area of the opening.

(F) Each unslotted and slotted guide pole well shall be equipped with a gasketed sliding cover or a flexible fabric sleeve seal.

(G) Each unslotted guide pole shall be equipped with a gasketed cap on the end of the pole.

(H) Each slotted guide pole shall be equipped with a gasketed float or other device which closes off the surface from the atmosphere.

(I) Each gauge hatch and each sample well shall be equipped with a gasketed cover.

(2) The owner or operator shall operate the tank in accordance with the following requirements:

(i) When the floating roof is resting on the leg supports, the process of filling, emptying, or refilling shall be continuous and shall be accomplished as soon as practical.

(ii) Except for automatic bleeder vents, rim space vents, roof drains, and leg sleeves, each opening in the roof shall be secured and maintained in a closed position at all times except when the closure device must be open for access.

(iii) Covers on each access hatch and each gauge float well shall be bolted or

fastened when secured in the closed position.

(iv) Automatic bleeder vents shall be set closed at all times when the roof is floating, except when the roof is being floated off or is being landed on the leg supports.

(v) Rim space vents shall be set to open only at those times that the roof is being floated off the roof leg supports or when the pressure beneath the rim seal exceeds the manufacturer's recommended setting.

(vi) The cap on the end of each unslotted guide pole shall be secured in the closed position at all times except when measuring the level or collecting samples of the liquid in the tank.

(vii) The cover on each gauge hatch or sample well shall be secured in the closed position at all times except when the hatch or well must be opened for access.

(viii) Both the primary seal and the secondary seal shall completely cover the annular space between the external floating roof and the wall of the tank in a continuous fashion except during inspections.

(3) The owner or operator shall inspect the external floating roof in accordance with the procedures specified in §63.695(b) of this subpart.

(g) The owner or operator who controls tank air emissions by venting to a control device shall meet the requirements specified in paragraphs (g)(1)through (g)(3) of this section.

(1) The tank shall be covered by a fixed roof and vented directly through a closed-vent system to a control device in accordance with the following requirements:

(i) The fixed roof and its closure devices shall be designed to form a continuous barrier over the entire surface area of the liquid in the tank.

(ii) Each opening in the fixed roof not vented to the control device shall be equipped with a closure device. If the pressure in the vapor headspace underneath the fixed roof is less than atmospheric pressure when the control device is operating, the closure devices shall be designed to operate such that when the closure device is secured in the closed position there are no visible cracks, holes, gaps, or other open spaces in the closure device or between the perimeter of the cover opening and the closure device. If the pressure in the vapor headspace underneath the fixed roof is equal to or greater than atmospheric pressure when the control device is operating, the closure device shall be designed to operate with no detectable organic emissions.

(iii) The fixed roof and its closure devices shall be made of suitable materials that will minimize exposure of the off-site material to the atmosphere, to the extent practical, and will maintain the integrity of the equipment throughout its intended service life. Factors to be considered when selecting the materials for and designing the fixed roof and closure devices shall include: organic vapor permeability, the effects of any contact with the liquid and its vapor managed in the tank; the effects of outdoor exposure to wind, moisture, and sunlight; and the operating practices used for the tank on which the fixed roof is installed.

(iv) The closed-vent system and control device shall be designed and operated in accordance with the requirements of §63.693 of this subpart.

(2) Whenever an off-site material is in the tank, the fixed roof shall be installed with each closure device secured in the closed position and the vapor headspace underneath the fixed roof vented to the control device except that venting to the control device is not required, and opening of closure devices or removal of the fixed roof is allowed at the following times:

(i) To provide access to the tank for performing routine inspection, maintenance, or other activities needed for normal operations. Examples of such activities include those times when a worker needs to open a port to sample liquid in the tank, or when a worker needs to open a hatch to maintain or repair equipment. Following completion of the activity, the owner or operator shall promptly secure the closure device in the closed position or reinstall the cover, as applicable, to the tank.

(ii) To remove accumulated sludge or other residues from the bottom of the tank.

(3) The owner or operator shall inspect and monitor the air emission control equipment in accordance with
the procedures specified in §63.695 of this subpart.

(h) The owner or operator who elects to control tank air emissions by using a pressure tank shall meet the following requirements.

(1) The tank shall be designed not to vent to the atmosphere as a result of compression of the vapor headspace in the tank during filling of the tank to its design capacity.

(2) All tank openings shall be equipped with closure devices designed to operate with no detectable organic emissions as determined using the procedure specified in §63.694(k) of this subpart.

(3) Whenever an off-site material is in the tank, the tank shall be operated as a closed system that does not vent to the atmosphere except at those times when purging of inerts from the tank is required and the purge stream is routed to a closed-vent system and control device designed and operated in accordance with the requirements of §63.693.

(i) The owner or operator who elects to control air emissions by using an enclosure vented through a closed-vent system to an enclosed combustion control device shall meet the requirements specified in paragraphs (i)(1) through (3) of this section.

(1) The tank shall be located inside an enclosure. The enclosure shall be designed and operated in accordance with the criteria for a permanent total enclosure as specified in "Procedure T-Criteria for and Verification of a Permanent or Temporary Total Enclosure" under 40 CFR 52.741, appendix B. The enclosure may have permanent or temporary openings to allow worker access; passage of material into or out of the enclosure by conveyor, vehicles, or other mechanical means; entry of permanent mechanical or electrical equipment; or to direct airflow into the enclosure. The owner or operator shall perform the verification procedure for the enclosure as specified in Section 5.0 "Procedure T-Criteria for and to Verification of a Permanent or Temporary Total Enclosure" initially when the enclosure is first installed and, thereafter, annually.

(2) The enclosure shall be vented through a closed-vent system to an en-

40 CFR Ch. I (7–1–23 Edition)

closed combustion control device that is designed and operated in accordance with the standards for either a vapor incinerator, boiler, or process heater specified in §63.693 of this subpart.

(3) The owner or operator shall inspect and monitor the closed-vent system and control device as specified in §63.693.

[61 FR 34158, July 1, 1996, as amended at 64 FR 38968, July 20, 1999; 66 FR 1266, Jan. 8, 2001; 80 FR 14273, Mar. 18, 2015]

§63.686 Standards: Oil-water and organic-water separators.

(a) The provisions of this section apply to the control of air emissions from oil-water separators and organicwater separators for which $\S63.683(b)(1)(i)$ of this subpart references the use of this section for such air emission control.

(b) The owner or operator shall control air emissions from each separator subject to this section by using one of the following:

(1) A floating roof in accordance with all applicable provisions specified in subpart VV of this part—National Emission Standards for Oil-Water Separators and Organic-Water Separators, except that $\S 63.1043(c)(2)$, 63.1044(c)(2), and 63.1045(b)(3)(i) shall not apply for the purposes of this subpart. For portions of the separator where it is infeasible to install and operate a floating roof, such as over a weir mechanism, the owner or operator shall comply with the requirements specified in paragraph (b)(2) of this section.

(2) A fixed-roof that is vented through a closed-vent system to a control device in accordance with all applicable provisions specified in subpart VV of this part—National Emission Standards for Oil-Water Separators and Organic-Water Separators, except that §§ 63.1043(c)(2), 63.1044(c)(2), and 63.1045(b)(3)(i) shall not apply for the purposes of this subpart.

(3) A pressurized separator that operates as a closed system in accordance with all applicable provisions specified in subpart VV of this part—National Emission Standards for Oil-Water Separators and Organic-Water Separators, except that §§ 63.1043(c)(2), 63.1044(c)(2),

and 63.1045(b)(3)(i) shall not apply for the purposes of this subpart.

[61 FR 34158, July 1, 1996, as amended at 64 FR 38969, July 20, 1999; 80 FR 14274, Mar. 18, 2015]

§63.687 Standards: Surface impoundments.

(a) The provisions of this section apply to the control of air emissions from surface impoundments for which $\S63.683(b)(1)(i)$ of this subpart references the use of this section for such air emission control.

(b) The owner or operator shall control air emissions from each surface impoundment subject to this section by using one of the following:

(1) A floating membrane cover in accordance with the applicable provisions specified in subpart QQ of this part— National Emission Standards for Surface Impoundments, except that $\S 63.942(c)(2)$ and (3) and 63.943(c)(2)shall not apply for the purposes of this subpart; or

(2) A cover that is vented through a closed-vent system to a control device in accordance with all applicable provisions specified in subpart QQ of this part—National Emission Standards for Surface Impoundments, except that \S 63.942(c)(2) and (3) and 63.943(c)(2) shall not apply for the purposes of this subpart.

[61 FR 34158, July 1, 1996, as amended at 64 FR 38969, July 20, 1999; 80 FR 14274, Mar. 18, 2015]

§63.688 Standards: Containers.

(a) The provisions of this section apply to the control of air emissions from containers for which §63.683(b)(1)(i) of this subpart references the use of this section for such air emission control.

(b) The owner or operator shall control air emissions from each container subject to this section in accordance with the following requirements, as applicable to the container, except when the special provisions for waste stabilization processes specified in paragraph (c) of this section apply to the container.

(1) For a container having a design capacity greater than 0.1 m^3 and less than or equal to 0.46 m^3 , the owner or operator must control air emissions

from the container in accordance with the requirements in either paragraph (b)(1)(i) or (b)(1)(i) of this section.

(i) The owner or operator controls air emissions from the container in accordance with the standards for Container Level 1 controls as specified in subpart PP of this part—National Emission Standards for Containers, except that §§63.922(d)(4) and (5) and 63.923(d)(4) and (5) shall not apply for the purposes of this subpart.

(ii) As an alternative to meeting the requirements in paragraph (b)(1)(i) of this section, an owner or operator may choose to control air emissions from the container in accordance with the standards for either Container Level 2 controls or Container Level 3 controls as specified in subpart PP of this part— National Emission Standards for Containers, except that §§63.922(d)(4) and (5) and 63.923(d)(4) and (5) shall not apply for the purposes of this subpart.

(2) For a container having a design capacity greater than 0.46 m^3 and the container is not in light-material service as defined in §63.681 of this subpart, the owner or operator must control air emissions from the container in accordance with the requirements in either paragraph (b)(1)(i) or (b)(1)(ii) of this section.

(3) For a container having a design capacity greater than 0.46 m^3 and the container is in light-material service as defined in §63.681 of this subpart, the owner or operator must control air emissions from the container in accordance with the requirements in either paragraph (b)(3)(i) or (b)(3)(ii) of this section.

(i) The owner or operator controls air emissions from the container in accordance with the standards for Container Level 2 controls as specified in subpart PP of this part—National Emission Standards for Containers, except that $\S 63.922(d)(4)$ and (5) and 63.923(d)(4) and (5) shall not apply for the purposes of this subpart.

(ii) As an alternative to meeting the requirements in paragraph (b)(3)(i) of this section, an owner or operator may choose to control air emissions from the container in accordance with the standards for Container Level 3 controls as specified in 40 CFR part 63,

subpart PP—National Emission Standards for Containers.

(c) When a container subject to this subpart and having a design capacity greater than 0.1 m^3 is used for treatment of an off-site material by a waste stabilization process as defined in § 63.681 of this subpart, the owner or operator shall control air emissions from the container at those times during the process when the off-site material in the container is exposed to the atmosphere in accordance with the standards for Container Level 3 controls as specified in 40 CFR part 63, subpart PP—National Emission Standards for Containers.

[61 FR 34158, July 1, 1996, as amended at 64 FR 38969, July 20, 1999; 80 FR 14274, Mar. 18, 2015]

§63.689 Standards: Transfer systems.

(a) The provisions of this section apply to the control of air emissions from transfer systems for which 63.683(b)(1)(i) of this subpart references the use of this section for such air emission control.

(b) For each transfer system that is subject to this section and is an individual drain system, the owner or operator shall control air emissions in accordance with the standards specified in 40 CFR part 63, subpart RR—National Emission Standards for Individual Drain Systems.

(c) For each transfer system that is subject to this section but is not an individual drain system, the owner or operator shall control air emissions by using one of the transfer systems specified in paragraphs (c)(1) through (c)(3) of this section.

(1) A transfer system that uses covers in accordance with the requirements specified in paragraph (d) of this section.

(2) A transfer system that consists of continuous hard-piping. All joints or seams between the pipe sections shall be permanently or semi-permanently sealed (e.g., a welded joint between two sections of metal pipe or a bolted and gasketed flange).

(3) A transfer system that is enclosed and vented through a closed-vent system to a control device in accordance with the requirements specified in 40 CFR Ch. I (7–1–23 Edition)

paragraphs (c)(3)(i) and (c)(3)(ii) of this section.

(i) The transfer system is designed and operated such that an internal pressure in the vapor headspace in the enclosure is maintained at a level less than atmospheric pressure when the control device is operating, and

(ii) The closed-vent system and control device are designed and operated in accordance with the requirements of §63.693 of this subpart.

(d) Owners and operators controlling air emissions from a transfer system using covers in accordance with the provisions of paragraph (c)(1) of this section shall meet the requirements specified in paragraphs (d)(1) through (d)(6) of this section.

(1) The cover and its closure devices shall be designed to form a continuous barrier over the entire surface area of the off-site material as it is conveyed by the transfer system except for the openings at the inlet and outlet to the transfer system through which the offsite material passes. The inlet and outlet openings used for passage of the offsite material through the transfer system shall be the minimum size required for practical operation of the transfer system.

(2) The cover shall be installed in a manner such that there are no visible cracks, holes, gaps, or other open spaces between cover section joints or between the interface of the cover edge and its mounting.

(3) Except for the inlet and outlet openings to the transfer system through which the off-site material passes, each opening in the cover shall be equipped with a closure device designed to operate such that when the closure device is secured in the closed position there are no visible cracks, holes, gaps, or other open spaces in the closure device or between the perimeter of the opening and the closure device.

(4) The cover and its closure devices shall be made of suitable materials that will minimize exposure of the offsite material to the atmosphere, to the extent practical, and will maintain the integrity of the equipment throughout its intended service life. Factors to be considered when selecting the materials for and designing the cover and

closure devices shall include: organic vapor permeability; the effects of any contact with the material or its vapors conveyed in the transfer system; the effects of outdoor exposure to wind, moisture, and sunlight; and the operating practices used for the transfer system on which the cover is installed.

(5) Whenever an off-site material is in the transfer system, the cover shall be installed with each closure device secured in the closed position, except the opening of closure devices or removal of the cover is allowed to provide access to the transfer system for performing routine inspection, maintenance, repair, or other activities needed for normal operations. Examples of such activities include those times when a worker needs to open a hatch or remove the cover to repair conveyance equipment mounted under the cover or to clear a blockage of material inside the system. Following completion of the activity, the owner or operator shall promptly secure the closure device in the closed position or reinstall the cover, as applicable.

(6) The owner or operator shall inspect the air emission control equipment in accordance with the requirements specified in §63.695 of this subpart.

[61 FR 34158, July 1, 1996, as amended at 64 FR 38970, July 20, 1999; 80 FR 14275, Mar. 18, 2015]

§63.690 Standards: Process vents.

(a) The provisions of this section apply to the control of air emissions from process vents for which §63.683(c)(1)(i) of this subpart references the use of this section for such air emission control.

(b) The owner or operator must route the vent stream from each affected process vent through a closed-vent system to a control device that meets the standards specified in §63.693 of this subpart. For the purpose of complying with this paragraph (b), a primary condenser is not a control device; however, a second condenser or other organic recovery device that is operated downstream of the primary condenser is considered a control device.

[64 FR 38970, July 20, 1999]

§63.691 Standards: Equipment leaks.

(a) The provisions of this section apply to the control of air emissions from equipment leaks for which §63.683(d) references the use of this section for such air emissions control.

(b) According to the date an affected source commenced construction or reconstruction and the date an affected source receives off-site material for the first time, as established in \S 63.680(e)(i) through (iii), the owner or operator shall control the HAP emitted from equipment leaks in accordance with the applicable provisions specified in either paragraph (b)(1) or (2) of this section.

(1)(i) The owner or operator controls the HAP emitted from equipment leaks in accordance with §§61.241 through 61.247 in 40 CFR part 61, subpart V—National Emission Standards for Equipment Leaks, with the difference noted in paragraphs (b)(1)(ii) and (iv) of this section for the purposes of this subpart; or

(ii) The owner or operator controls the HAP emitted from equipment leaks in accordance with §§63.161 through 63.182 in subpart H of this part—National Emission Standards for Organic Hazardous Air Pollutants from Equipment Leaks, with the differences noted in paragraphs (b)(2)(i) through (iv) of this section for the purposes of this subpart.

(iii) On or after March 18, 2015, for the purpose of complying with the requirements of 40 CFR 61.242-6(a)(2) or the requirements of 63.167(a)(2), the open end is sealed when instrument monitoring of the open-ended valve or line conducted according to Method 21 of 40 CFR part 60, appendix A indicates no readings of 500 ppm or greater.

(iv) On or after March 18, 2015, for the purpose of complying with the requirements of 40 CFR 61.242–6(d) or the requirements of §63.167(d), open-ended valves or lines in an emergency shutdown system which are designed to open automatically in the event of a process upset and that are exempt from the requirements in 40 CFR 61.242–6(a), (b), and (c) or §63.167(a), (b), and (c) must comply with the requirements in §63.693(c)(2).

(2) The owner or operator controls the HAP emitted from equipment leaks

§63.691

40 CFR Ch. I (7-1-23 Edition)

in accordance with §§ 63.161 through 63.183 in subpart H of this part—National Emission Standards for Organic Hazardous Air Pollutants for Equipment Leaks, with the differences noted in paragraphs (b)(2)(i) through (v) of this section for the purposes of this subpart.

§63.691

(i) For each valve in gas/vapor or in light liquid service, as defined in §63.681, that is part of an affected source under this subpart, an instrument reading that defines a leak is 500 ppm or greater as detected by Method 21 of 40 CFR part 60, appendix A.

(ii) For each pump in light liquid service, as defined in §63.681, that is part of an affected source under this subpart, an instrument reading that defines a leak is 1,000 ppm or greater as detected by Method 21 of 40 CFR part 60, appendix A. Repair is not required unless an instrument reading of 2,000 ppm or greater is detected.

(iii) On or after March 18, 2015, for the purpose of complying with the requirements of $\S63.167(a)(2)$, the open end is sealed when instrument monitoring of the open-ended valve or line conducted according to Method 21 of 40 CFR part 60, appendix A indicates no readings of 500 ppm or greater.

(iv) On or after March 18, 2015, for the purpose of complying with the requirements of $\S63.167(d)$, open-ended valves or lines in an emergency shutdown system which are designed to open automatically in the event of a process upset and that are exempt from the requirements in $\S63.167(a)$, (b), and (c) must comply with the requirements in $\S63.693(c)(2)$.

(v) For the purposes of this subpart, the pressure relief device requirements of 63.691(c) of this subpart rather than those of 63.165 or of 40 CFR 61.242-4, as applicable, shall apply. The pressure relief device requirements of 63.691(c)(3) and (4) apply in addition to the requirements of 63.169 or of 40 CFR 61.242-8, as applicable, for pressure relief devices in liquid service.

(c) Requirements for pressure relief devices. Except as provided in paragraph (c)(4) of this section, the owner or operator must comply with the requirements specified in paragraphs (c)(1) through (3) of this section for pressure

relief devices in off-site material service.

(1) Operating requirements. Except during a pressure release event, operate each pressure relief device in gas/ vapor service with an instrument reading of less than 500 ppm above background as detected by Method 21 of 40 CFR part 60, appendix A.

(2) Pressure release requirements. For pressure relief devices in gas/vapor service, the owner or operator must comply with either paragraph (c)(2)(i)or (ii) of this section following a pressure release, as applicable.

(i) If the pressure relief device does not consist of or include a rupture disk, the pressure relief device shall be returned to a condition indicated by an instrument reading of less than 500 ppm above background, as detected by Method 21 of 40 CFR part 60, appendix A, no later than 5 calendar days after the pressure release device returns to off-site material service following a pressure release, except as provided in §63.171.

(ii) If the pressure relief device consists of or includes a rupture disk, except as provided in §63.171, install a replacement disk as soon as practicable but no later than 5 calendar days after the pressure release.

(3) Pressure release management. Except as provided in paragraph (c)(4) of this section, emissions of HAP listed in Table 1 of this subpart may not be discharged directly to the atmosphere from pressure relief devices in off-site material service, and according to the date an affected source commenced construction or reconstruction and the date an affected source receives off-site material for the first time, as established in $\S63.680(e)(1)(i)$ through (iii), the owner or operator must comply with the requirements specified in paragraphs (c)(3)(i) and (ii) of this section for all pressure relief devices in off-site material service, except that containers are not subject to the obligations in paragraph (c)(3)(i) of this section.

(i) The owner or operator must equip each pressure relief device in off-site material service with a device(s) or use a monitoring system. The device or monitoring system may be either specific to the pressure release device

itself or may be associated with the process system or piping, sufficient to indicate a pressure release to the atmosphere. Examples of these types of devices or monitoring systems include, but are not limited to, a rupture disk indicator, magnetic sensor, motion detector on the pressure relief valve stem, flow monitor, pressure monitor, or parametric monitoring system. The devices or monitoring systems must be capable of meeting the requirements specified in paragraphs (c)(3)(i)(A) through (C) of this section.

(A) Identifying the pressure release;

(B) Recording the time and duration of each pressure release; and

(C) Notifying operators immediately that a pressure release is occurring.

(ii) If any pressure relief device in off-site material service releases directly to the atmosphere as a result of a pressure release event, the owner or operator must calculate the quantity of HAP listed in Table 1 of this subpart released during each pressure release event and report this quantity as required in §63.697(b)(5). Calculations may be based on data from the pressure relief device monitoring alone or in combination with process parameter monitoring data and process knowledge. For containers, the calculations may be based on process knowledge and information alone.

(4) Pressure relief devices routed to a drain system, fuel gas system, process or control device. If a pressure relief device in off-site material service is designed and operated to route all pressure releases through a closed vent system to a drain system, fuel gas system, process or control device, paragraphs (c)(1), (2), and (3) of this section do not apply. The fuel gas system or closed vent system and the process or control device (if applicable) must meet the requirements of §63.693. The drain system (if applicable) must meet the requirements of §63.689.

[64 FR 38970, July 20, 1999, as amended at 66 FR 1266, Jan. 8, 2001; 80 FR 14275, Mar. 18, 2015; 83 FR 3992, Jan. 29, 2018]

§63.692 [Reserved]

§63.693 Standards: Closed-vent systems and control devices.

(a) The provisions of this section apply to closed-vent systems and control devices used to control air emissions for which another standard references the use of this section for such air emission control.

(b) For each closed-vent system and control device used to comply with this section, the owner or operator shall meet the following requirements:

(1) The owner or operator must use a closed-vent system that meets the requirements specified in paragraph (c) of this section.

(2) The owner or operator must use a control device that meets the requirements specified in paragraphs (d) through (h) of this section as applicable to the type and design of the control device selected by the owner or operator to comply with the provisions of this section.

(3) Whenever gases or vapors containing HAP are routed from a tank through a closed-vent system connected to a control device used to comply with the requirements of $\S63.685(b)(1)$, (2), or (3), the control device must be operating except as provided for in paragraphs (b)(3)(i) and (ii) of this section.

(i) The control device may only be bypassed for the purpose of performing planned routine maintenance of the closed-vent system or control device in situations when the routine maintenance cannot be performed during periods that tank emissions are vented to the control device.

(ii) On an annual basis, the total time that the closed-vent system or control device is bypassed to perform routine maintenance shall not exceed 240 hours per each calendar year.

(4) The owner or operator must inspect and monitor each closed-vent system in accordance with the requirements specified in either paragraph (b)(4)(i) or (b)(4)(ii) of this section.

(i) The owner or operator inspects and monitors the closed-vent system in accordance with the requirements specified in §63.695(c) of this subpart, and complies with the applicable recordkeeping requirements in §63.696 of this

§63.693

subpart and the applicable reporting requirements in §63.697 of this subpart.

(ii) As an alternative to meeting the requirements specified in paragraph (b)(4)(i) of this section, the owner or operator may choose to inspect and monitor the closed-vent system in accordance with the requirements under 40 CFR part 63, subpart H—National Emission Standards for Organic Hazardous Air Pollutants for Equipment Leaks as specified in 40 CFR 63.172(f) through (h), and complies with the applicable record keeping requirements in 40 CFR 63.181 and the applicable reporting requirements in 40 CFR 63.182.

(5) The owner or operator must monitor the operation of each control device in accordance with the requirements specified in paragraphs (d) through (h) of this section as applicable to the type and design of the control device selected by the owner or operator to comply with the provisions of this section.

(6) The owner or operator shall maintain records for each control device in accordance with the requirements of §63.696 of this subpart.

(7) The owner or operator shall prepare and submit reports for each control device in accordance with the requirements of 63.697 of this subpart.

(8) In the case when an owner or operator chooses to use a design analysis to demonstrate compliance of a control device with the applicable performance requirements specified in this section as provided for in paragraphs (d) through (g) of this section, the Administrator may require that the design analysis be revised or amended by the owner or operator to correct any deficiencies identified by the Administrator. If the owner or operator and the Administrator do not agree on the acceptability of using the design analysis (including any changes required by the Administrator) to demonstrate that the control device achieves the applicable performance requirements, then the disagreement must be resolved using the results of a performance test conducted by the owner or operator in accordance with the requirements of §63.694(1). The Administrator may choose to have an authorized representative observe the performance test conducted by the owner or oper40 CFR Ch. I (7–1–23 Edition)

ator. Should the results of this performance test not agree with the determination of control device performance based on the design analysis, then the results of the performance test will be used to establish compliance with this subpart.

(c) Closed-vent system requirements.

(1) The vent stream required to be controlled shall be conveyed to the control device by either of the following closed-vent systems:

(i) A closed-vent system that is designed to operate with no detectable organic emissions using the procedure specified in §63.694(k) of this subpart; or

(ii) A closed-vent system that is designed to operate at a pressure below atmospheric pressure. The system shall be equipped with at least one pressure gauge or other pressure measurement device that can be read from a readily accessible location to verify that negative pressure is being maintained in the closed-vent system when the control device is operating.

(2) In situations when the closed-vent system includes bypass devices that could be used to divert a vent stream from the closed-vent system to the atmosphere at a point upstream of the control device inlet, each bypass device must be equipped with either a flow indicator as specified in paragraph (c)(2)(i) of this section or a seal or locking device as specified in paragraph (c)(2)(i) of this section, except as provided for in paragraph (c)(2)(i) of this section.

(i) If a flow indicator is used, the indicator must be installed at the entrance to the bypass line used to divert the vent stream from the closed-vent system to the atmosphere. The flow indicator must indicate a reading at least once every 15 minutes. The owner or operator must maintain records of the following information: hourly records of whether the flow indicator was operating and whether flow was detected at any time during the hour; and records of all periods when flow is detected or the flow indicator is not operating.

(ii) If a seal or locking device is used to comply with paragraph (c)(2) of this section, the device shall be placed on

the mechanism by which the bypass device position is controlled (*e.g.*, valve handle, damper lever) when the bypass device is in the closed position such that the bypass device cannot be opened without breaking the seal or removing the lock. Examples of such devices include, but are not limited to, a car-seal or a lock-and-key configuration valve.

(iii) Equipment needed for safety reasons, including low leg drains, openended valves and lines not in emergency shutdown systems, and pressure relief devices subject to the requirements of $\S63.691(c)$ are not subject to the requirements of paragraphs (c)(2)(i) and (ii) of this section.

(d) Carbon adsorption control device requirements.

(1) The carbon adsorption system must achieve the performance specifications in either paragraph (d)(1)(i) or (d)(1)(i) of this section.

(i) Recover 95 percent or more, on a weight-basis, of the total organic compounds (TOC), less methane and ethane, contained in the vent stream entering the carbon adsorption system; or

(ii) Recover 95 percent or more, on a weight-basis, of the total HAP listed in Table 1 of this subpart contained in the vent stream entering the carbon adsorption system.

(2) The owner or operator must demonstrate that the carbon adsorption system achieves the performance requirements in paragraph (d)(1) of this section by either performing a performance test as specified in paragraph (d)(2)(i) of this section or a design analysis as specified in paragraph (d)(2)(ii) of this section.

(i) An owner or operator choosing to use a performance test to demonstrate compliance must conduct the test in accordance with the requirements of §63.694(1) of this subpart.

(ii) An owner or operator choosing to use a design analysis to demonstrate compliance must include as part of this design analysis the information specified in either paragraph (d)(2)(ii)(A) or (d)(2)(ii)(B) of this section as applicable to the carbon adsorption system design.

(A) For a regenerable carbon adsorption system, the design analysis shall

address the vent stream composition, constituent concentrations, flow rate, relative humidity, and temperature and shall establish the design exhaust vent stream organic compound concentration, adsorption cycle time, number and capacity of carbon beds, type and working capacity of activated carbon used for carbon beds, design total regeneration steam flow over the period of each complete carbon bed regeneration cycle, design carbon bed temperature after regeneration, design carbon bed regeneration time, and design service life of the carbon.

(B) For a nonregenerable carbon adsorption system (e.g., a carbon canister), the design analysis shall address the vent stream composition, constituent concentrations, flow rate, relative humidity, and temperature and shall establish the design exhaust vent stream organic compound concentration, carbon bed capacity, activated carbon type and working capacity, and design carbon replacement interval based on the total carbon working capacity of the control device and emission point operating schedule.

(3) The owner or operator must monitor the operation of the carbon adsorption system in accordance with the requirements of §63.695(e) using one of the continuous monitoring systems specified in paragraphs (d)(3)(i) through (iii) of this section. Monitoring the operation of a nonregenerable carbon adsorption system (e.g., a carbon canister) using a continuous monitoring system is not required when the carbon canister or the carbon in the control device is replaced on a regular basis according to the requirements in paragraph (d)(4)(iii) of this section.

(i) For a regenerative-type carbon adsorption system:

(A) A continuous parameter monitoring system to measure and record the average total regeneration stream mass flow or volumetric flow during each carbon bed regeneration cycle. The integrating regenerating stream flow monitoring device must have an accuracy of ±10 percent; and

(B) A continuous parameter monitoring system to measure and record the average carbon bed temperature for the duration of the carbon bed steaming cycle and to measure the actual carbon bed temperature after regeneration and within 15 minutes of completing the cooling cycle. The accuracy of the temperature monitoring device must be ± 1 percent of the temperature being measured, expressed in degrees Celsius or ± 5 °C, whichever is greater.

(ii) A continuous monitoring system to measure and record the daily average concentration level of organic compounds in the exhaust gas stream from the control device. The organic monitoring system must comply either with Performance Specification 8 or 9 in 40 CFR part 60, appendix B. The relative accuracy provision of Performance Specification 8, Sections 2.4 and 3 need not be conducted.

(iii) A continuous monitoring system that measures other alternative operating parameters upon approval of the Administrator as specified in 40 CFR 63.8(f)(1) through (f)(5) of this part.

(4) The owner or operator shall manage the carbon used for the carbon adsorption system, as follows:

(i) Following the initial startup of the control device, all carbon in the control device shall be replaced with fresh carbon on a regular, predetermined time interval that is no longer than the carbon service life established for the carbon adsorption system. The provisions of this paragraph (d)(4)(i) do not apply to a nonregenerable carbon adsorption system (e.g., a carbon canister) for which the carbon canister or the carbon in the control device is replaced on a regular basis according to requirements in the paragraph (d)(4)(iii) of this section.

(ii) The spent carbon removed from the carbon adsorption system must be either regenerated, reactivated, or burned in one of the units specified in paragraphs (d)(4)(ii)(A) through (d)(4)(ii)(G) of this section.

(A) Regenerated or reactivated in a thermal treatment unit for which the owner or operator has been issued a final permit under 40 CFR part 270 that implements the requirements of 40 CFR part 264, subpart X.

(B) Regenerated or reactivated in a thermal treatment unit equipped with and operating air emission controls in accordance with this section.

(C) Regenerated or reactivated in a thermal treatment unit equipped with

40 CFR Ch. I (7-1-23 Edition)

and operating organic air emission controls in accordance with a national emission standard for hazardous air pollutants under another subpart in 40 CFR part 63 or 40 CFR part 61.

(D) Burned in a hazardous waste incinerator for which the owner or operator has been issued a final permit under 40 CFR part 270 that implements the requirements of 40 CFR part 264, subpart O.

(E) Burned in a hazardous waste incinerator for which the owner or operator has designed and operates the incinerator in accordance with the interim status requirements of 40 CFR part 265, subpart O.

(F) Burned in a boiler or industrial furnace for which the owner or operator has been issued a final permit under 40 CFR part 270 that implements the requirements of 40 CFR part 266, subpart H.

 (\hat{G}) Burned in a boiler or industrial furnace for which the owner or operator has designed and operates the unit in accordance with the interim status requirements of 40 CFR part 266, subpart H.

(iii) As an alternative to meeting the requirements in paragraphs (d)(3) and (d)(4)(i) of this section, an owner or operator of a nonregenerable carbon adsorption system may choose to replace on a regular basis the carbon canister or the carbon in the control device using the procedures in either paragraph (d)(4)(iii)(A) or (d)(4)(iii)(B) of this section. For the purpose of complying with this paragraph (d)(4)(iii), a nonregenerable carbon adsorption system means a carbon adsorption system that does not regenerate the carbon bed directly onsite in the control device, such as a carbon canister. The spent carbon removed from the nonregenerable carbon adsorption system must be managed according to the requirements in paragraph (d)(4)(ii) of this section.

(A) Monitor the concentration level of the organic compounds in the exhaust vent from the carbon adsorption system on a regular schedule, and when carbon breakthrough is indicated, immediately replace either the existing carbon canister with a new carbon canister or replace the existing carbon in the control device with fresh carbon.

Measurement of the concentration level of the organic compounds in the exhaust vent stream must be made with a detection instrument that is appropriate for the composition of organic constituents in the vent stream and is routinely calibrated to measure the organic concentration level expected to occur at breakthrough. The monitoring frequency must be daily or at an interval no greater than 20 percent of the time required to consume the total carbon working capacity established as a requirement of paragraph (d)(2)(ii)(B) of this section, whichever is longer.

(B) Replace either the existing carbon canister with a new carbon canister or replace the existing carbon in the control device with fresh carbon at a regular, predetermined time interval that is less than the design carbon replacement interval established as a requirement of paragraph (d)(2)(ii)(B) of this section.

(e) Condenser control device requirements.

(1) The condenser must achieve the performance specifications in either paragraph (e)(1)(i) or (e)(1)(i) of this section.

(i) Recover 95 percent or more, on a weight-basis, of the total organic compounds (TOC), less methane and ethane, contained in the vent stream entering the condenser; or

(ii) Recover 95 percent or more, on a weight-basis, of the total HAP, listed in Table 1 of this subpart, contained in the vent stream entering the condenser.

(2) The owner or operator must demonstrate that the condenser achieves the performance requirements in paragraph (e)(1) of this section by either performing a performance test as specified in paragraph (e)(2)(i) of this section or a design analysis as specified in paragraph (e)(2)(ii) of this section.

(i) An owner or operator choosing to use a performance tests to demonstrate compliance must conduct the test in accordance with the requirements of §63.694(1) of this subpart.

(ii) An owner or operator choosing to use a design analysis to demonstrate compliance must include as part of this design analysis the following information: description of the vent stream composition, constituent concentrations, flow rate, relative humidity, and temperature; and specification of the design outlet organic compound concentration level, design average temperature of the condenser exhaust vent stream, and the design average temperatures of the coolant fluid at the condenser inlet and outlet.

(3) The owner or operator must monitor the operation of the condenser in accordance with the requirements of 63.695(e) of this subpart using one of the continuous monitoring systems specified in paragraphs (e)(3)(i) through (e)(3)(iii) of this section.

(i) A continuous parameter monitoring system to measure and record the daily average temperature of the exhaust gases from the control device. The accuracy of the temperature monitoring device shall be ± 1 percent of the temperature being measured, expressed in degrees Celsius or ± 5 °C, whichever is greater.

(ii) A continuous monitoring system to measure and record the daily average concentration level of organic compounds in the exhaust gas stream from the control device. The organic monitoring system must comply either with Performance Specification 8 or 9 in 40 CFR part 60, appendix B. The relative accuracy provision of Performance Specification 8, Sections 2.4 and 3 need not be conducted.

(iii) A continuous monitoring system that measures other alternative operating parameters upon approval of the Administrator as specified in 40 CFR 63.8(f)(1) through (f)(5) of this part.

(f) Vapor incinerator control device requirements.

(1) The vapor incinerator must achieve the performance specifications in either paragraph (f)(1)(i), (f)(1)(ii), or (f)(1)(iii) of this section.

(i) Destroy the total organic compounds (TOC), less methane and ethane, contained in the vent stream entering the vapor incinerator either:

(A) By 95 percent or more, on a weight-basis, or

(B) To achieve a total incinerator outlet concentration for the TOC, less methane and ethane, of less than or equal to 20 ppmv on a dry basis corrected to 3 percent oxygen.

40 CFR Ch. I (7-1-23 Edition)

(ii) Destroy the HAP listed in Table 1 of this subpart contained in the vent stream entering the vapor incinerator either:

(A) By 95 percent or more, on a total HAP weight-basis, or

(B) To achieve a total incinerator outlet concentration for the HAP, listed in Table 1 of this subpart, of less than or equal to 20 ppmv on a dry basis corrected to 3 percent oxygen.

(iii) Maintain the conditions in the vapor incinerator combustion chamber at a residence time of 0.5 seconds or longer and at a temperature of 760° C or higher.

(2) The owner or operator must demonstrate that the vapor incinerator achieves the performance requirements in paragraph (f)(1) of this section by conducting either a performance test as specified in paragraph (f)(2)(i) of this section or a design analysis as specified in paragraph (f)(2)(ii) of this section, except as provided for in paragraph (f)(2)(iii) of this section.

(i) An owner or operator choosing to use a performance test to demonstrate compliance must conduct the test in accordance with the requirements of §63.694(1) of this subpart.

(ii) An owner or operator choosing to use a design analysis to demonstrate compliance must include as part of this design analysis the information specified in either paragraph (f)(2)(ii)(A) or (f)(2)(ii)(B) of this section as applicable to the vapor incinerator design.

(A) For a thermal vapor incinerator, the design analysis shall address the vent stream composition, constituent concentrations, and flow rate and shall establish the design minimum and average temperatures in the combustion chamber and the combustion chamber residence time.

(B) For a catalytic vapor incinerator, the design analysis shall address the vent stream composition, constituent concentrations, and flow rate and shall establish the design minimum and average temperatures across the catalyst bed inlet and outlet, and the design service life of the catalyst.

(iii) An owner or operator is not required to conduct a performance test or design analysis if the incinerator has been issued a final permit under 40 CFR part 270 and complies with the requirements of 40 CFR part 264, subpart O, or has certified compliance with the interim status requirements of 40 CFR part 265, subpart O.

(3) The owner or operator must monitor the operation of the vapor incinerator in accordance with the requirements of 63.695(e) of this subpart using one of the continuous monitoring systems specified in paragraphs (f)(3)(i) through (f)(3)(iv) of this section as applicable to the type of vapor incinerator used.

(i) For a thermal vapor incinerator, a continuous parameter monitoring system to measure and record the daily average temperature of the exhaust gases from the control device. The accuracy of the temperature monitoring device must be ± 1 percent of the temperature being measured, expressed in degrees Celsius of ± 0.5 °C, whichever is greater.

(ii) For a catalytic vapor incinerator, a temperature monitoring device capable of monitoring temperature at two locations equipped with a continuous recorder. One temperature sensor shall be installed in the vent stream at the nearest feasible point to the catalyst bed inlet and a second temperature sensor shall be installed in the vent stream at the nearest feasible point to the catalyst bed outlet.

(iii) For either type of vapor incinerator, a continuous monitoring system to measure and record the daily average concentration of organic compounds in the exhaust vent stream from the control device. The organic monitoring system must comply either with Performance Specification 8 or 9 in 40 CFR part 60, appendix B. The relative accuracy provision of Performance Specification 8, Sections 2.4 and 3 need not be conducted.

(iv) For either type of vapor incinerator, a continuous monitoring system that measures alternative operating parameters other than those specified in paragraph (f)(3)(i) or (f)(3)(ii) of this section upon approval of the Administrator as specified in 40 CFR 63.8(f)(1)through (f)(5) of this part.

(g) Boilers and process heaters control device requirements.

(1) The boiler or process heater must achieve the performance specifications in either paragraph (g)(1)(i), (g)(1)(i),

 $(g)(1)(iii),\ (g)(1)(iv),\ or\ (g)(1)(v)$ of this section.

(i) Destroy the total organic compounds (TOC), less methane and ethane, contained in the vent stream introduced into the flame zone of the boiler or process heater either:

(A) By 95 percent or more, on a weight-basis, or

(B) To achieve in the exhausted combustion gases a total concentration for the TOC, less methane and ethane, of less than or equal to 20 parts ppmv on a dry basis corrected to 3 percent oxygen.

(ii) Destroy the HAP listed in Table 1 of this subpart contained in the vent stream entering the vapor incinerator either:

(A) By 95 percent or more, on a total HAP weight-basis, or

(B) To achieve in the exhausted combustion gases a total concentration for the HAP, listed in Table 1 of the subpart, of less than or equal to 20 ppmv on a dry basis corrected to 3 percent oxygen.

(iii) Introduce the vent stream into the flame zone of the boiler or process heater and maintain the conditions in the combustion chamber at a residence time of 0.5 seconds or longer and at a temperature of 760°C or higher.

(iv) Introduce the vent stream with the fuel that provides the predominate heat input to the boiler or process heater (i.e., the primary fuel); or

(v) Introduce the vent stream to a boiler or process heater for which the owner or operator either has been issued a final permit under 40 CFR part 270 and complies with the requirements of 40 CFR part 266, subpart H; or has certified compliance with the interim status requirements of 40 CFR part 266, subpart H; or has submitted a Notification of Compliance under §§63.1207(j) and 63.1210(d) and complies with the requirements of subpart EEE of this part at all times (including times when nonhazardous waste is being burned).

(2) The owner or operator must demonstrate that the boiler or process heater achieves the performance specifications in paragraph (g)(1) of this section chosen by the owner or operator using the applicable method specified in paragraph (g)(2)(i) or (g)(2)(ii) of this section. (i) If an owner or operator chooses to comply with the performance specifications in either paragraph (g)(1)(i), (ii), or (iii) of this section, the owner or operator must demonstrate compliance with the applicable performance specifications by conducting either a performance test as specified in paragraph (g)(2)(i)(A) of this section or a design analysis as specified in paragraph (g)(2)(i)(B) of this section, except as provided for in paragraph (g)(2)(i)(C) of this section.

(A) An owner or operator choosing to use a performance test to demonstrate compliance must conduct the test in accordance with the requirements of §63.694(1) of this subpart.

(B) An owner or operator choosing to use a design analysis to demonstrate compliance must include as part of this design analysis the following information: description of the vent stream composition, constituent concentrations, and flow rate; specification of the design minimum and average flame zone temperatures and combustion zone residence time; and description of the method and location by which the vent stream is introduced into the flame zone.

(C) An owner or operator is not required to conduct a performance test or design analysis if the boiler or process heater has been issued a final permit under 40 CFR part 270 and complies with the requirements of 40 CFR part 266, subpart H; or has certified compliance with the interim status requirements of 40 CFR part 266, subpart H.

(ii) If an owner or operator chooses to comply with the performance specifications in either paragraph (g)(1)(iv) or (g)(1)(v) of this section, the owner or operator must demonstrate compliance by maintaining the records that document that the boiler or process heater is designed and operated in accordance with the applicable requirements of this section.

(3) For a boiler or process heater complying with the performance specifications in either paragraph (g)(1)(i), (g)(1)(i), or (g)(1)(ii) of this section, the owner or operator must monitor the operation of a boiler or process heater in accordance with the requirements of §63.695(e) of this subpart using one of the continuous monitoring systems specified in paragraphs (g)(3)(i) through (g)(3)(ii) of this section.

(i) A continuous parameter monitoring system to measure and record the daily average combustion zone temperature. The accuracy of the temperature sensor must be ± 1 percent of the temperature being measured, expressed in degrees Celsius or ± 0.5 °C, whichever is greater;

(ii) A continuous monitoring system to measure and record the daily average concentration of organic compounds in the exhaust vent stream from the control device. The organic monitoring system must comply either with Performance Specification 8 or 9 in 40 CFR part 60, appendix B. The relative accuracy provision of Performance Specification 8, Sections 2.4 and 3 need not be conducted.

(iii) A continuous monitoring system that measures alternative operating parameters other than those specified in paragraph (g)(3)(i) or (g)(3)(i) of this section upon approval of the Administrator as specified in 40 CFR 63.8(f)(1) through (f)(5) of this part.

(h) Flare control device requirements.

(1) The flare must be designed and operated in accordance with the requirements in 40 CFR 63.11(b).

(2) The owner or operator must demonstrate that the flare achieves the requirements in paragraph (h)(1) of this section by performing the procedures specified in paragraph (h)(2)(i) of this section. A previous compliance demonstration for the flare that meets all of the conditions specified in paragraph (h)(2)(ii) of this section may be used by an owner or operator to demonstrate compliance with this paragraph (h)(2).

(i) To demonstrate that a flare achieves the requirements in paragraph (h)(1) of this section, the owner or operator performs all of the procedures specified in paragraphs (h)(2)(i)(A) through (h)(2)(i)(C) of this section.

(A) The owner or operator conducts a visible emission test for the flare in accordance with the requirements specified in 40 CFR 63.11(b)(4).

(B) The owner or operator determines the net heating value of the gas being combusted in the flare in accordance 40 CFR Ch. I (7-1-23 Edition)

with the requirements specified in 40 CFR 63.11(b)(6); and

(C) The owner or operator determines the flare exit velocity in accordance with the requirements applicable to the flare design as specified in 40 CFR 63.11(b)(7) or 40 CFR 63.11(b)(8).

(ii) A previous compliance demonstration for the flare may be used by an owner or operator to demonstrate compliance with paragraph (h)(2) of this section provided that all conditions for the compliance determination and subsequent flare operation are met as specified in paragraphs (h)(2)(ii)(A)and (h)(2)(ii)(B) of this section.

(A) The owner or operator conducted the compliance determination using the procedures specified in paragraph (h)(2)(i) of this section.

(B) No flare operating parameter or process changes have occurred since completion of the compliance determination which could affect the compliance determination results.

(3) The owner or operator must monitor the operation of the flare using a heat sensing monitoring device (including but not limited to a thermocouple, ultraviolet beam sensor, or infrared sensor) that continuously detects the presence of a pilot flame. The owner or operator must record, for each 1-hour period, whether the monitor was continuously operating and whether a pilot flame was continuously present during each hour as required in §63.696(b)(3) of this subpart.

[64 FR 38970, July 20, 1999, as amended at 66 FR 1266, Jan. 8, 2001; 68 FR 37351, June 23, 2003; 80 FR 14276, Mar. 18, 2015]

§63.694 Testing methods and procedures.

(a) This section specifies the testing methods and procedures required for this subpart to perform the following:

(1) To determine the average VOHAP concentration for off-site material streams at the point-of-delivery for compliance with standards specified §63.683 of this subpart, the testing methods and procedures are specified in paragraph (b) of this section.

(2) To determine the average VOHAP concentration for treated off-site material streams at the point-of-treatment for compliance with standards specified §63.684 of this subpart, the testing

methods and procedures are specified in paragraph (c) of this section.

(3) To determine the treatment process VOHAP concentration limit (C_R) for compliance with standards specified §63.684 of this subpart, the testing methods and procedures are specified in paragraph (d) of this section.

(4) To determine treatment process required HAP removal rate (RMR) for compliance with standards specified §63.684 of this subpart, the testing methods and procedures are specified in paragraph (e) of this section.

(5) To determine treatment process actual HAP removal rate (MR) for compliance with standards specified §63.684 of this subpart, the testing methods and procedures are specified in paragraph (f) of this section.

(6) To determine treatment process required HAP reduction efficiency (R) for compliance with standards specified in §63.684 of this subpart, the testing methods and procedures are specified in paragraph (g) of this section.

(7) To determine treatment process required HAP biodegradation efficiency (R_{bio}) for compliance with standards specified in §63.684 of this subpart, the testing methods and procedures are specified in paragraph (h) of this section.

(8) To determine treatment process required actual HAP mass removal rate (MR_{bio}) for compliance with standards specified in§63.684 of this subpart, the testing methods and procedures are specified in paragraph (i) of this section.

(9) To determine maximum organic HAP vapor pressure of off-site materials in tanks for compliance with the standards specified in §63.685 of this subpart, the testing methods and procedures are specified in paragraph (j) of this section.

(10) To determine no detectable organic emissions, the testing methods and procedures are specified in paragraph (k) of this section.

(11) To determine closed-vent system and control device performance for compliance with the standards specified in §63.693 of this subpart, the testing methods and procedures are specified in paragraph (1) of this section.

(12) To determine process vent stream flow rate and total organic

HAP concentration for compliance with the standards specified in §63.693 of this subpart, the testing methods and procedures are specified in paragraph (m) of this section.

(b) Testing methods and procedures to determine average VOHAP concentration of an off-site material stream at the point-of-delivery.

(1) The average VOHAP concentration of an off-site material at the point-of-delivery shall be determined using either direct measurement as specified in paragraph (b)(2) of this section or by knowledge as specified in paragraph (b)(3) of this section.

(2) Direct measurement to determine VOHAP concentration—(i) Sampling. Samples of the off-site material stream shall be collected from the container, pipeline, or other device used to deliver the off-site material stream to the plant site in a manner such that volatilization of organics contained in the sample is minimized and an adequately representative sample is collected and maintained for analysis by the selected method.

(A) The averaging period to be used for determining the average VOHAP concentration for the off-site material stream on a mass-weighted average basis shall be designated and recorded. The averaging period can represent any time interval that the owner or operator determines is appropriate for the off-site material stream but shall not exceed 1 year.

(B) A sufficient number of samples, but no less than four samples, shall be collected to represent the complete range of HAP compositions and HAP quantities that occur in the off-site material stream during the entire averaging period due to normal variations in the operating conditions for the source or process generating the offsite material stream. Examples of such normal variations are seasonal variations in off-site material quantity or fluctuations in ambient temperature.

(C) All samples shall be collected and handled in accordance with written procedures prepared by the owner or operator and documented in a site sampling plan. This plan shall describe the procedure by which representative samples of the off-site material stream are collected such that a minimum loss of organics occurs throughout the sample collection and handling process and by which sample integrity is maintained. A copy of the written sampling plan shall be maintained on-site in the plant site operating records. An example of an acceptable sampling plan includes a plan incorporating sample collection and handling procedures in accordance with the requirements specified in "Test Methods for Evaluating Solid Waste, Physical/Chemical Methods," EPA Publication No. SW-846 or Method 25D in 40 CFR part 60, appendix A.

(ii) *Analysis.* Each collected sample must be prepared and analyzed in accordance with one of the following methods as applicable to the sampled off-site material for the purpose of measuring the HAP listed in Table 1 of this subpart:

(A) Method 305 in 40 CFR part 63, appendix A.

(B) Method 25D in 40 CFR part 60, appendix A.

(C) Method 624 in 40 CFR part 136, appendix A. If this method is used to analyze one or more compounds that are not on the method's published list of approved compounds, the Alternative Test Procedure specified in 40 CFR 136.4 and 40 CFR 136.5 must be followed.

(D) Method 625 in 40 CFR part 136, appendix A. For the purpose of using this method to comply with this subpart, the owner or operator must perform corrections to these compounds based on the "accuracy as recovery" using the factors in Table 7 of the method. If this method is used to analyze one or more compounds that are not on the method's published list of approved compounds, the Alternative Test Procedure specified in 40 CFR 136.4 and 40 CFR 136.5 must be followed.

(E) Method 1624 in 40 CFR part 136, appendix A.

(F) Method 1625 in 40 CFR part 136, appendix A.

(G) Method 8260 in "Test Methods for Evaluating Solid Waste, Physical/ Chemical Methods," EPA Publication No. SW-846, Third Edition, September 1986, as amended by Update I, November 15, 1992. As an alternative, an owner or operator may use any more recent, updated version of Method 8260 approved by the EPA. For the purpose of 40 CFR Ch. I (7-1-23 Edition)

using Method 8260 to comply with this subpart, the owner or operator must maintain a formal quality assurance program consistent with section 8 of Method 8260, and this program must include the following elements related to measuring the concentrations of volatile compounds:

(1) Documentation of site-specific procedures to minimize the loss of compounds due to volatilization, biodegradation, reaction, or sorption during the sample collection, storage, and preparation steps.

(2) Documentation of specific quality assurance procedures followed during sampling, sample preparation, sample introduction, and analysis.

(3) Measurement of the average accuracy and precision of the specific procedures, including field duplicates and field spiking of the off-site material source before or during sampling with compounds having similar chemical characteristics to the target analytes.

(H) Method 8270 in "Test Methods for Evaluating Solid Waste, Physical/ Chemical Methods," EPA Publication No. SW-846, Third Edition, September 1986, as amended by Update I, November 15, 1992. As an alternative, an owner or operator may use any more recent, updated version of Method 8270 approved by the EPA. For the purpose of using Method 8270 to comply with this subpart, the owner or operator must maintain a formal quality assurance program consistent with Method 8270, and this program must include the following elements related to measuring the concentrations of volatile compounds:

(1) Documentation of site-specific procedures to minimize the loss of compounds due to volatilization, bio-degradation, reaction, or sorption during the sample collection, storage, and preparation steps.

(2) Documentation of specific quality assurance procedures followed during sampling, sample preparation, sample introduction, and analysis.

(3) Measurement of the average accuracy and precision of the specific procedures, including field duplicates and field spiking of the off-site material source before or during sampling with compounds having similar chemical characteristics to the target analytes.

(I) Any other analysis method that has been validated in accordance with the procedures specified in section 5.1 and section 5.3 and the corresponding calculations in section 6.1 or section 6.3 of Method 301 in appendix A in 40 CFR part 63. The data are acceptable if they meet the criteria specified in section 6.1.5 or section 6.3.3 of Method 301. If correction is required under section 6.3.3 of Method 301, the data are acceptable if the correction factor is within the range of 0.7 to 1.30. Other sections of Method 301 are not required.

(iii) Calculations. The average VOHAP concentration (C) on a mass-weighted basis shall be calculated by using the results for all samples analyzed in accordance with paragraph (b)(2)(ii) of this section and the following equation. An owner or operator using a test method that provides species-specific chemical concentrations may adjust the measured concentrations to the corresponding concentration values which would be obtained had the offsite material samples been analyzed using Method 305. To adjust these data, the measured concentration for each individual HAP chemical species contained in the off-site material is multiplied by the appropriate species-specific adjustment factor (f_{m305}) listed in Table 1 of this subpart.

$$C = \frac{1}{Q_T} \times \sum_{i=1}^{n} (Q_i \times C_i)$$

Where:

- C = Average VOHAP concentration of the off-site material at the point-of-delivery on a mass-weighted basis, ppmw.
- i = Individual sample "i" of the off-site material.
- $\label{eq:n} \begin{array}{l} \mbox{= Total number of samples of the off-site} \\ \mbox{material collected (at least 4) for the} \\ \mbox{averaging period (not to exceed 1 year).} \end{array}$
- Q_i = Mass quantity of off-site material stream represented by C_i , kg/hr.
- $\label{eq:QT} \begin{array}{l} Q_T = {\rm Total\ mass\ quantity\ of\ off-site\ material} \\ {\rm during\ the\ averaging\ period,\ kg/hr.} \end{array}$
- $C_i = Measured \ VOHAP \ concentration \ of \ sample \ ``i'' \ as \ determined \ in \ accordance \ with \ the \ requirements \ of \ \$63.694(a), \ ppmw.$
- (3) Knowledge of the off-site material to determine VOHAP concentration.

(i) Documentation shall be prepared that presents the information used as the basis for the owner's or operator's knowledge of the off-site material stream's average VOHAP concentration. Examples of information that may be used as the basis for knowledge include: material balances for the source or process generating the offsite material stream; species-specific chemical test data for the off-site material stream from previous testing that are still applicable to the current off-site material stream; previous test data for other locations managing the same type of off-site material stream; or other knowledge based on information in documents such as manifests, shipping papers, or waste certification notices.

(ii) If test data are used as the basis for knowledge, then the owner or operator shall document the test method, sampling protocol, and the means by which sampling variability and analytical variability are accounted for in the determination of the average VOHAP concentration. For example, an owner or operator may use HAP concentration test data for the off-site material stream that are validated in accordance with Method 301 in 40 CFR part 63, appendix A of this part as the basis for knowledge of the off-site material.

(iii) An owner or operator using species-specific chemical concentration test data as the basis for knowledge of the off-site material may adjust the test data to the corresponding average VOHAP concentration value which would be obtained had the off-site material samples been analyzed using Method 305. To adjust these data, the measured concentration for each individual HAP chemical species contained in the off-site material is multiplied by the appropriate species-specific adjustment factor (f_{m305}) listed in Table 1 of this subpart.

(iv) In the event that the Administrator and the owner or operator disagree on a determination of the average VOHAP concentration for an offsite material stream using knowledge, then the results from a determination of VOHAP concentration using direct measurement as specified in paragraph (b)(2) of this section shall be used to establish compliance with the applicable requirements of this subpart. The Administrator may perform or require that the owner or operator perform this determination using direct measurement.

(c) Determination of average VOHAP concentration of an off-site material stream at the point-of-treatment.

(1) Sampling. Samples of the off-site material stream shall be collected at the point-of-treatment in a manner such that volatilization of organics contained in the sample is minimized and an adequately representative sample is collected and maintained for analysis by the selected method.

(i) The averaging period to be used for determining the average VOHAP concentration for the off-site material stream on a mass-weighted average basis shall be designated and recorded. The averaging period can represent any time interval that the owner or operator determines is appropriate for the off-site material stream but shall not exceed 1 year.

(ii) A sufficient number of samples, but no less than four samples, shall be collected to represent the complete range of HAP compositions and HAP quantities that occur in the off-site material stream during the entire averaging period due to normal variations in the operating conditions for the treatment process. Examples of such normal variations are seasonal variations in off-site material quantity or fluctuations in ambient temperature.

(iii) All samples shall be collected and handled in accordance with written procedures prepared by the owner or operator and documented in a site sampling plan. This plan shall describe the procedure by which representative samples of the off-site material stream are collected such that a minimum loss of organics occurs throughout the sample collection and handling process and by which sample integrity is maintained. A copy of the written sampling plan shall be maintained on-site in the plant site operating records. An example of an acceptable sampling plan includes a plan incorporating sample collection and handling procedures in accordance with the requirements specified in "Test Methods for Evaluating Solid Waste, Physical/Chemical Methods." EPA Publication No. SW-846 or Method 25D in 40 CFR part 60, appendix Α.

40 CFR Ch. I (7–1–23 Edition)

(2) Analysis. Each collected sample must be prepared and analyzed in accordance with one of the methods specified in paragraphs (b)(2)(ii)(A) through (b)(2)(ii)(I) of this section, as applicable to the sampled off-site material, for the purpose of measuring the HAP listed in Table 1 of this subpart.

(3) Calculations. The average VOHAP concentration (\overline{C}) a mass-weighted basis shall be calculated by using the results for all samples analyzed in accordance with paragraph (c)(2) of this section and the following equation. An owner or operator using a test method that provides species-specific chemical concentrations may adjust the measured concentrations to the corresponding concentration values which would be obtained had the off-site material samples been analyzed using Method 305. To adjust these data, the measured concentration for each individual HAP chemical species contained in the off-site material is multiplied by the appropriate species-specific adjustment factor $(f_{\rm m305})$ listed in Table 1 of this subpart.

$$\overline{\mathbf{C}} = \frac{1}{\mathbf{Q}_{\mathrm{T}}} \times \sum_{i=1}^{n} (\mathbf{Q}_{i} \times \mathbf{C}_{i})$$

Where:

- $\bar{\mathbf{C}}$ = Average VOHAP concentration of the off-site material on a mass-weighted basis, ppmw.
- i = Individual sample "i" of the off-site material.
- n = Total number of samples of the off-site material collected (at least 4) for the averaging period (not to exceed 1 year).
- $\begin{aligned} Q_i &= Mass \ quantity \ of \ off-site \ material \\ stream \ represented \ by \ C_i, \ kg/hr. \end{aligned}$
- $\label{eq:QT} \begin{aligned} Q_T &= \text{Total mass quantity of off-site material} \\ & \text{during the averaging period, kg/hr.} \end{aligned}$
- $\begin{array}{l} C_i = Measured \; VOHAP \; concentration \; of \; sample \; ``i'' as \; determined in \; accordance \; with \; the \; requirements \; of §63.694(a), \; ppmw. \end{array}$

(d) Determination of treatment process VOHAP concentration limit (C_R). (1) All of the off-site material streams entering the treatment process shall be identified.

(2) The average VOHAP concentration of each off-site material stream at the point-of-delivery shall be determined using the procedures specified in paragraph (b) of this section.

(3) The VOHAP concentration limit (C_R) shall be calculated by using the results determined for each individual

off-site material stream and the following equation:

$$C_{R} = \frac{\sum_{x=1}^{m} (Q_{x} \times \overline{C}_{x}) + \sum_{y=1}^{n} (Q_{y} \times 500 \text{ ppmw})}{\sum_{x=1}^{m} Q_{x} + \sum_{y=1}^{n} Q_{y}}$$

where:

- C_R = VOHAP concentration limit, ppmw.
- x = Individual off-site material stream "x" that has a VOHAP concentration less than 500 ppmw at the point-of-delivery.
- y = Individual off-site material stream "y" that has a VOHAP concentration equal to or greater than 500 ppmw at the pointof-delivery.
- m = Total number of "x" off-site material streams treated by process.
- = Total number of "y" off-site material streams treated by process.
- Q_x = Total mass quantity of off-site material stream "x", kg/yr.
- Q_v = Total mass quantity of off-site material stream "y", kg/yr.
- \bar{C}_x = VOHAP concentration of off-site material stream "x" at the point-of-delivery, ppmw.

(e) Determination of required HAP mass removal rate (RMR).

(1) Each individual stream containing HAP that enters the treatment process shall be identified.

(2) The average VOHAP concentration at the point-of-delivery for each stream identified in paragraph (e)(1) of this section shall be determined using the test methods and procedures specified in paragraph (b) of this section.

(3) For each stream identified in paragraph (e)(1) of this section that has an average VOHAP concentration equal to or greater than 500 ppmw at the point-of-delivery, the average volumetric flow rate and the density of the off-site material stream at the pointof-delivery shall be determined.

(4) The required HAP mass removal rate (RMR) shall be calculated by using the average VOHAP concentration, average volumetric flow rate, and density determined in paragraph (e)(3) of this section for each stream and the following equation:

$$RMR = \sum_{y=1}^{n} \left[V_y \times k_y \times \frac{\left(\overline{C_y} - 500 \text{ ppmw}\right)}{10^6} \right]$$

Where:

- RMR = Required HAP mass removal rate, kg/ hr.
- y = Individual stream "y" that has a VOHAP concentration equal to or greater than 500 ppmw at the point-of-delivery as determined in §63.694(b).
- n = Total number of "y" streams treated by process. $V_{\boldsymbol{y}}$ = Average volumetric flow rate of stream
- ''y'' at the point-of-delivery, $m^{3\!/}hr.$
- determined in §63.694(b)(2), ppmw.

(f) Determination of actual HAP mass removal rate (MR).

(1) The actual HAP mass removal rate (MR) shall be determined based on results for a minimum of three consecutive runs. The sampling time for each run shall be at least 1 hour.

(2) The HAP mass flow entering the process (E_b) and the HAP mass flow exiting the process (E_a) shall be determined using the test methods and procedures specified in paragraphs (g)(2)through (g)(4) of this section.

(3) The actual mass removal rate shall be calculated using the HAP mass flow rates determined in paragraph (f)(2) of this section and the following equation:

 $MR = E_b - E_a$

where:

MR = Actual HAP mass removal rate, kg/ hr.

 $E_{\rm b} = {\rm HAP}$ mass flow entering process as determined in paragraph (f)(2) of this section, kg/hr.

LAPBH6H6L3 with DISTILLER

§ 63.694

 $E_{\rm a}$ = HAP mass flow exiting process as determined in paragraph (f)(2) of this section, kg/hr.

(g) Determination of treatment process HAP reduction efficiency (R).

(1) The HAP reduction efficiency (R) for a treatment process shall be determined based on results for a minimum of three consecutive runs.

(2) Each individual stream containing HAP that enters the treatment process shall be identified. Each individual stream containing HAP that exits the treatment process shall be identified. The owner or operator shall prepare a sampling plan for measuring the identified streams that accurately reflects the retention time of the material in the process.

(3) For each run, information shall be determined for each stream identified in paragraph (g)(2) of this section as specified in paragraphs (g)(3)(i) through (g)(3)(iii) of this section.

(i) The mass quantity shall be determined for each stream identified in paragraph (g)(2) of this section as entering the process (Q_b). The mass quantity shall be determined for each stream identified in paragraph (g)(2) of this section as exiting the process (Q_a).

(ii) The average VOHAP concentration at the point-of-delivery shall be determined for each stream entering the process (C_b) (as identified in paragraph (g)(2) of this section) using the test methods and procedures specified in paragraph (b) of this section.

(iii) The average VOHAP concentration at the point-of-treatment shall be determined for each stream exiting the process (C_a) (as identified in paragraph (g)(2) of this section) using the test methods and procedures specified in paragraph (c) of this section.

(4) The HAP mass flow entering the process (E_b) and the HAP mass flow exiting the process (E_a) shall be calculated using the results determined in paragraph (g)(3) of this section and the following equations:

$$E_{a} = \frac{1}{10^{6}} \sum_{j=1}^{m} \left(Q_{aj} \times \overline{C_{aj}} \right)$$
$$E_{b} = \frac{1}{10^{6}} \sum_{j=1}^{m} \left(Q_{bj} \times \overline{C_{bj}} \right)$$

LAPBH6H6L3 with DISTILLER

40 CFR Ch. I (7-1-23 Edition)

Where:

 E_{b} = HAP mass flow entering process, kg/hr.

 $E_a = HAP$ mass flow exiting process, kg/hr. m = Total number of runs (at least 3)

- i = Individual run "i"
- Q_{bj} = Mass quantity of material entering
- process during run "j", kg/hr. Q_{aj} = Average mass quantity of material exiting process during run "j", kg/hr.
- C_{aj} = Average VOHAP concentration of material exiting process during run "j" as determined in §63.694(c), ppmw.
- C_{bj} = Average VOHAP concentration of material entering process during run "j" as determined in §63.694(b)(2), ppmw.

(5) The HAP reduction efficiency (R) shall be calculated using the HAP mass flow rates determined in paragraph (g)(4) of this section and the following equation:

$$R = \frac{E_b - E_a}{E_b} \times 100$$

Where:

- R = HAP reduction efficiency, percent.
 E_b = HAP mass flow entering process as determined in paragraph (g)(4) of this section. kg/hr.
- E_a = HAP mass flow exiting process as determined in accordance with the requirements of paragraph (g)(4) of this section, kg/hr.

(h) Determination of HAP biodegradation efficiency (R_{bio}) .

(1) The fraction of HAP biodegraded (F_{bio}) shall be determined using one of the procedures specified in appendix C of this part 63.

(2) The HAP biodegradation efficiency ($R_{\rm bio}$) shall be calculated by using the following equation:

 $R_{\rm bio} - F_{\rm bio} \times 100$

where:

 $R_{\rm bio}$ = HAP biodegradation efficiency, percent.

 $F_{\rm bio}$ = Fraction of HAP biodegraded as determined in paragraph (h)(1) of this section.

(i) Determination of actual HAP mass removal rate ($MR_{\rm bio}$). (1) The actual HAP mass removal rate ($MR_{\rm bio}$) shall be determined based on results for a minimum of three consecutive runs. The sampling time for each run shall be at least 1 hour.

(2) The HAP mass flow entering the process (E_b) shall be determined using

the test methods and procedures specified in paragraphs (g)(2) through (g)(4) of this section.

(3) The fraction of HAP biodegraded ($F_{\rm bio}$) shall be determined using the procedure specified in 40 CFR part 63, appendix C of this part.

(4) The actual mass removal rate shall be calculated by using the HAP mass flow rates and fraction of HAP biodegraded determined in paragraphs (i)(2) and (i)(3), respectively, of this section and the following equation:

 $MR_{\rm bio}$ = $E^{\rm b} \times F_{\rm bio}$

Where:

MR_{bio} = Actual HAP mass removal rate, kg/ hr.

 E_b = HAP mass flow entering process, kg/hr. F_{bio} = Fraction of HAP biodegraded.

(j) Determination of maximum HAP vapor pressure for off-site material in a tank. (1) The maximum HAP vapor pressure of the off-site material composition managed in a tank shall be determined using either direct measurement as specified in paragraph (j)(2) of this section or by knowledge of the offsite material as specified by paragraph (j)(3) of this section.

(2) Direct measurement to determine the maximum HAP vapor pressure of an off-site material.

(i) Sampling. A sufficient number of samples shall be collected to be representative of the off-site material contained in the tank. All samples shall be collected and handled in accordance with written procedures prepared by the owner or operator and documented in a site sampling plan. This plan shall describe the procedure by which representative samples of the off-site material is collected such that a minimum loss of organics occurs throughout the sample collection and handling process and by which sample integrity is maintained. A copy of the written sampling plan shall be maintained onsite in the plant site operating records. An example of an acceptable sampling plan includes a plan incorporating sample collection and handling procedures in accordance with the requirements specified in "Test Methods for Evaluating Solid Waste, Physical/Chemical Methods," EPA Publication No. SW-846 or Method 25D in 40 CFR part 60, appendix A.

(ii) Analysis. Any one of the following methods may be used to analyze the samples and compute the maximum HAP vapor pressure of the offsite material:

(A) Method 25E in 40 CFR part 60 appendix A;

(B) Methods described in American Petroleum Institute Bulletin 2517, "Evaporation Loss from External Floating Roof Tanks,";

(C) Methods obtained from standard reference texts;

(D) ASTM Method 2879-83; or

(E) Any other method approved by the Administrator.

(3) Use of knowledge to determine the maximum HAP vapor pressure of the offsite material. Documentation shall be prepared and recorded that presents the information used as the basis for the owner's or operator's knowledge that the maximum HAP vapor pressure of the off-site material is less than the maximum vapor pressure limit listed in Table 3, Table 4, or Table 5 of this subpart for the applicable tank design capacity category. Examples of information that may be used include: the off-site material is generated by a process for which at other locations it previously has been determined by direct measurement that the off-site material maximum HAP vapor pressure is less than the maximum vapor pressure limit for the appropriate tank design capacity category. In the event that the Administrator and the owner or operator disagree on a determination of the maximum HAP vapor pressure for an off-site material stream using knowledge, then the results from a determination of HAP vapor pressure using direct measurement as specified in paragraph (j)(2) of this section shall be used to establish compliance with the applicable requirements of this subpart. The Administrator may perform or require that the owner or operator perform this determination using direct measurement.

(k) Procedure for determining no detectable organic emissions for the purpose of complying with this subpart.

(1) The test shall be conducted in accordance with the procedures specified in Method 21 of 40 CFR part 60, appendix A. Each potential leak interface (i.e., a location where organic vapor leakage could occur) on the cover and associated closure devices shall be checked. Potential leak interfaces that are associated with covers and closure devices include, but are not limited to: the interface of the cover and its foundation mounting; the periphery of any opening on the cover and its associated closure device; and the sealing seat interface on a spring-loaded pressurerelief valve.

(2) The test shall be performed when the unit contains a material having a total organic concentration representative of the range of concentrations for the materials expected to be managed in the unit. During the test, the cover and closure devices shall be secured in the closed position.

(3) The detection instrument shall meet the performance criteria of Method 21 of 40 CFR part 60, appendix A, except the instrument response factor criteria in section 8.1.1 of Method 21 shall be for the weighted average composition of the organic constituents in the material placed in the unit at the time of monitoring, not for each individual organic constituent.

(4) The detection instrument shall be calibrated before use on each day of its use by the procedures specified in Method 21 of 40 CFR part 60, appendix A.

(5) Calibration gases shall be as follows:

(i) Zero air (less than 10 ppmv hydrocarbon in air); and

(ii) A mixture of methane or nhexane in air at a concentration of approximately, but less than, 10,000 ppmv.

(6) An owner or operator may choose to adjust or not adjust the detection instrument readings to account for the background organic concentration level. If an owner or operator chooses to adjust the instrument readings for the background level, the background level value must be determined according to the procedures in Method 21 of 40 CFR part 60, appendix A.

(7) Each potential leak interface shall be checked by traversing the instrument probe around the potential leak interface as close to the interface as possible, as described in Method 21. In the case when the configuration of the cover or closure device prevents a 40 CFR Ch. I (7–1–23 Edition)

complete traverse of the interface, all accessible portions of the interface shall be sampled. In the case when the configuration of the closure device prevents any sampling at the interface and the device is equipped with an enclosed extension or horn (e.g., some pressure relief devices), the instrument probe inlet shall be placed at approximately the center of the exhaust area to the atmosphere.

(8) An owner or operator must determine if a potential leak interface operates with no detectable emissions using the applicable procedure specified in paragraph (k)(8)(i) or (k)(8)(ii) of this section.

(i) If an owner or operator chooses not to adjust the detection instrument readings for the background organic concentration level, then the maximum organic concentration value measured by the detection instrument is compared directly to the applicable value for the potential leak interface as specified in paragraph (k)(9) of this section.

(ii) If an owner or operator chooses to adjust the detection instrument readings for the background organic concentration level, the value of the arithmetic difference between the maximum organic concentration value measured by the instrument and the background organic concentration value as determined in paragraph (k)(6) of this section is compared with the applicable value for the potential leak interface as specified in paragraph (k)(9) of this section.

(9) A potential leak interface is determined to operate with no detectable emissions using the applicable criteria specified in paragraphs (k)(9)(i) and (k)(9)(i) of this section.

(i) For a potential leak interface other than a seal around a shaft that passes through a cover opening, the potential leak interface is determined to operate with no detectable organic emissions if the organic concentration value determined in paragraph (k)(8) is less than 500 ppmv.

(ii) For a seal around a shaft that passes through a cover opening, the potential leak interface is determined to operate with no detectable organic emissions if the organic concentration

value determined in paragraph (k)(8) is less than 10,000 ppmv.

(1) Control device performance test procedures. Performance tests shall be based on representative performance (i.e., performance based on normal operating conditions) and shall exclude periods of startup and shutdown unless specified by the Administrator. The owner or operator may not conduct performance tests during periods of malfunction. The owner or operator must record the process information that is necessary to document operating conditions during the test and include in such record an explanation to support that such conditions represent normal operation. Upon request, the owner or operator shall make available to the Administrator such records as may be necessary to determine the conditions of performance tests.

(1) Method 1 or 1A of 40 CFR part 60, appendix A, as appropriate, shall be used for selection of the sampling sites at the inlet and outlet of the control device.

(i) To determine compliance with a control device percent reduction requirement, sampling sites shall be located at the inlet of the control device as specified in paragraphs (1)(1)(i)(A)and (1)(1)(i)(B) of this section, and at the outlet of the control device.

(A) The control device inlet sampling site shall be located after the final product recovery device.

(B) If a vent stream is introduced with the combustion air or as an auxiliary fuel into a boiler or process heater, the location of the inlet sampling sites shall be selected to ensure that the measurement of total HAP concentration or TOC concentration, as applicable, includes all vent streams and primary and secondary fuels introduced into the boiler or process heater.

(ii) To determine compliance with an enclosed combustion device concentration limit, the sampling site shall be located at the outlet of the device.

(2) The gas volumetric flow rate shall be determined using Method 2, 2A, 2C, or 2D, 2F, or 2G of 40 CFR part 60, appendix A, as appropriate.

(3) To determine compliance with the control device percent reduction requirement, the owner or operator shall use Method 18 of 40 CFR part 60, appendix A to measure the HAP in Table 1 of this subpart or Method 25A of 40 CFR part 60, appendix A to measure TOC. Method 18 may be used to measure methane and ethane, and the measured concentration may be subtracted from the Method 25A measurement. Alternatively, any other method or data that has been validated according to the applicable procedures in Method 301 in appendix A of this part may be used. The following procedures shall be used to calculate percent reduction efficiency:

(i) A minimum of three sample runs must be performed. The minimum sampling time for each run shall be 1 hour. For Method 18, either an integrated sample or a minimum of four grab samples shall be taken. If grab sampling is used, then the samples shall be taken at approximately equal intervals in time such as 15 minute intervals during the run.

(ii) The mass rate of either TOC (minus methane and ethane) or total HAP (E_i and E_o ,respectively) shall be computed.

(A) The following equations shall be used:

$$\begin{split} \mathbf{E}_{i} &= \mathbf{K}_{2} \times \mathbf{Q}_{i} \times \sum_{j=1}^{n} \left(\mathbf{C}_{ij} \times \mathbf{M}_{ij} \right) \\ \mathbf{E}_{o} &= \mathbf{K}_{2} \times \mathbf{Q}_{o} \times \sum_{i=1}^{n} \left(\mathbf{C}_{oj} \times \mathbf{M}_{oj} \right) \end{split}$$

Where:

- $\begin{array}{l} C_{ij}, \ C_{oj} = Concentration \ of \ sample \ component \\ j \ of \ the \ gas \ stream \ at \ the \ inlet \ and \ outlet \\ of \ the \ control \ device, \ respectively, \ dry \\ basis, \ parts \ per \ million \ by \ volume. \end{array}$
- E_i , E_o = Mass rate of TOC (minus methane and ethane) or total HAP at the inlet and outlet of the control device, respectively, dry basis, kilogram per hour.
- $$\begin{split} M_{ij},\,M_{oj} &= Molecular \mbox{ weight of sample component } j \mbox{ of the gas stream at the inlet and outlet of the control device, respectively, gram/gram-mole.} \end{split}$$
- Q_i, Q_o = Flow rate of gas stream at the inlet and outlet of the control device, respectively, dry standard cubic meter per minute.
- $\begin{array}{ll} K_2 = Constant, \ 2.494 \times 10^{-6} \ (parts \ per \ million)^{-1} \ (gram-mole \ per \ standard \ cubic \ meter) \ (kilogram/gram) \ (minute/hour), \\ where \ standard \ temperature \ (gram-mole \ per \ standard \ cubic \ meter) \ is \ 20 \ ^{\circ}C. \end{array}$

(B) When the TOC mass rate is calculated, the average concentration reading (minus methane and ethane) measured by Method 25A of 40 CFR part 60, appendix A shall be used in the equation in paragraph (1)(3)(ii)(A) of this section.

(C) When the total HAP mass rate is calculated, only the HAP constituents shall be summed using the equation in paragraph (1)(3)(ii)(A) of this section.

(iii) The percent reduction in TOC (minus methane and ethane) or total HAP shall be calculated as follows:

$$R_{cd} = \frac{E_i - E_o}{E_i} \times 100$$

where:

- R_{cd} = Control efficiency of control device, percent.
- E_i = Mass rate of TOC (minus methane and ethane) or total HAP at the inlet to the control device as calculated under paragraph (1)(3)(ii) of this section, kilograms TOC per hour or kilograms HAP per hour.
- $E_{\rm o}$ = Mass rate of TOC (minus methane and ethane) or total HAP at the outlet of the control device, as calculated under paragraph (1)(3)(ii) of this section, kilograms TOC per hour or kilograms HAP per hour.

(iv) If the vent stream entering a boiler or process heater is introduced with the combustion air or as a secondary fuel, the weight-percent reduction of total HAP or TOC (minus methane and ethane) across the device shall be determined by comparing the TOC (minus methane and ethane) or total HAP in all combusted vent streams and primary and secondary fuels with the TOC (minus methane and ethane) or 40 CFR Ch. I (7–1–23 Edition)

total HAP exiting the device, respectively.

(4) To determine compliance with the enclosed combustion device total HAP concentration limit of this subpart, the owner or operator shall use Method 18 of 40 CFR part 60, appendix A to measure the total HAP in Table 1 of this subpart or Method 25A of 40 CFR part 60, appendix A to measure TOC. Method 18 may be used to measure methane and ethane and the measured concentration may be subtracted from the Method 25A measurement. Alternatively, any other method or data that has been validated according to Method 301 in appendix A of this part, may be used. The following procedures shall be used to calculate parts per million by volume concentration, corrected to 3 percent oxygen:

(i) A minimum of three sample runs must be performed. The minimum sampling time for each run shall be 1 hour. For Method 18, either an integrated sample or a minimum of four grab samples shall be taken. If grab sampling is used, then the samples shall be taken at approximately equal intervals in time, such as 15 minute intervals during the run.

(ii) The TOC concentration or total HAP concentration shall be calculated according to paragraph (m)(4)(ii)(A) or (m)(4)(ii)(B) of this section.

(A) The TOC concentration (C_{TOC}) is the average concentration readings provided by Method 25 A of 40 CFR part 60, appendix A, minus the concentration of methane and ethane.

(B) The total HAP concentration (C_{HAP}) shall be computed according to the following equation:

$$C_{HAP} = \sum_{i=1}^{x} \frac{\sum_{j=1}^{n} C_{ji}}{x}$$

where:

LAPBH6H6L3 with DISTILLER

- C_{HAP} = Total concentration of HAP compounds listed in Table 1 of this subpart, dry basis, parts per million by volume.
- C_{ij} = Concentration of sample components j of sample i, dry basis, parts per million by volume.
- n = Number of components in the sample.

x = Number of samples in the sample run.

(iii) The measured TOC concentration or total HAP concentration shall be corrected to 3 percent oxygen as follows:

208

(A) The emission rate correction factor or excess air, integrated sampling and analysis procedures of Method 3B of 40 CFR part 60, appendix A shall be used to determine the oxygen concentration ($\%O_{2dry}$). Alternatively, the owner or operator may use Method 3A of 40 CFR part 60, appendix A to determine the oxygen concentration. The samples shall be collected during the same time that the samples are collected for determining TOC concentration or total HAP concentration.

(B) The concentration corrected to 3 percent oxygen (C_c) shall be computed using the following equation:

$$C_{c} = C_{m} \left(\frac{17.9}{20.9 - \%0_{2 \, dry}} \right)$$

where:

- $C_{\rm c}$ = TOC concentration or total HAP concentration corrected to 3 percent oxygen, dry basis, parts per million by volume.
- C_m = Measured TOC concentration or total HAP concentration, dry basis, parts per million by volume.
- O_{2dry} = Concentration of oxygen, dry basis, percent by volume.

(m) Determination of process vent stream flow rate and total HAP concentration.

(1) Method 1 or 1A of 40 CFR part 60, appendix A, as appropriate, must be used for selection of the sampling site.

(2) No traverse site selection method is needed for vents smaller than 0.10 meter in diameter. For vents smaller than 0.10 meter in diameter, sample at the center of the vent.

(3) Process vent stream gas volumetric flow rate must be determined using Method 2, 2A, 2C, 2D, 2F, or 2G of 40 CFR part 60, appendix A, as appropriate.

(4) Process vent stream total HAP concentration must be measured using the following procedures:

(i) Method 18 of 40 CFR part 60, appendix A, must be used to measure the total HAP concentration. Alternatively, any other method or data that has been validated according to the protocol in Method 301 of appendix A of this part may be used.

(ii) Where Method 18 of 40 CFR part 60, appendix A, is used, the following procedures must be used to calculate parts per million by volume concentration:

(A) The minimum sampling time for each run must be 1 hour in which either an integrated sample or four grab samples must be taken. If grab sampling is used, then the samples must be taken at approximately equal intervals in time, such as 15 minute intervals during the run.

(B) The total HAP concentration (C_{HAP}) must be computed according to the following equation:

$$C_{HAP} = \frac{\displaystyle{\sum_{i=l}^{x}} \left(\sum_{j=l}^{n} C_{ji} \right)}{X}$$

Where:

- C_{HAP} = Total concentration of HAP compounds listed in Table 1 of this subpart, dry basis, parts per million by volume.
- $C_{ji} = \mbox{Concentration of sample component } j \mbox{ of the sample } i, \mbox{ dry basis, parts per million } by \ volume.$
- n = Number of components in the sample.
- x = Number of samples in the sample run.

[61 FR 34158, July 1, 1996, as amended at 64 FR 38974, July 20, 1999; 66 FR 1267, Jan. 8, 2001; 80 FR 14277, Mar. 18, 2015]

§63.695 Inspection and monitoring requirements.

(a) The owner or operator must install, calibrate, maintain, and operate all monitoring system components according to §§ 63.8, 63.684(e), 63.693(d)(3), (e)(3), (f)(3), (g)(3), and (h)(3), and paragraph (a)(5) of this section and perform the inspection and monitoring procedures specified in paragraphs (a)(1) through (4) of this section.

(1) To inspect tank fixed roofs and floating roofs for compliance with the Tank Level 2 controls standards specified in §63.685 of this subpart, the inspection procedures are specified in paragraph (b) of this section.

(2) To inspect and monitor closedvent systems for compliance with the standards specified in §63.693 of this subpart, the inspection and monitoring procedures are specified in paragraph (c) of this section.

(3) To inspect and monitor transfer system covers for compliance with the standards specified in $\S63.689(c)(1)$ of

209

this subpart, the inspection and monitoring procedures are specified in paragraph (d) of this section.

(4) To monitor and record off-site material treatment processes for compliance with the standards specified in 63.684(e), the monitoring procedures are specified in paragraph (e) of this section.

(5)(i) Except for periods of monitoring system malfunctions, repairs associated with monitoring system malfunctions and required monitoring system quality assurance or quality control activities (including, as applicable, calibration checks and required zero and span adjustments), the owner or operator must operate the continuous monitoring system at all times the affected source is operating. A monitoring system malfunction is any sudden, infrequent, not reasonably preventable failure of the monitoring system to provide data. Monitoring system failures that are caused in part by poor maintenance or careless operation are not malfunctions. The owner or operator is required to complete monitoring system repairs in response to monitoring system malfunctions and to return the monitoring system to operation as expeditiously as practicable.

(ii) The owner or operator may not use data recorded during monitoring system malfunctions, repairs associated with monitoring system malfunctions, or required monitoring system quality assurance or control activities in calculations used to report emissions or operating levels. The owner or operator must use all the data collected during all other required data collection periods in assessing the operation of the control device and associated control system. The owner or operator must report any periods for which the monitoring system failed to collect required data.

(b) Tank Level 2 fixed roof and floating roof inspection requirements.

(1) Owners and operators that use a tank equipped with an internal floating roof in accordance with the provisions of §63.685(e) of this subpart shall meet the following inspection requirements:

(i) The floating roof and its closure devices shall be visually inspected by the owner or operator to check for defects that could result in air emissions. 40 CFR Ch. I (7–1–23 Edition)

Defects include, but are not limited to, the internal floating roof is not floating on the surface of the liquid inside the tank; liquid has accumulated on top of the internal floating roof; any portion of the roof seals have detached from the roof rim; holes, tears, or other openings are visible in the seal fabric; the gaskets no longer close off the waste surfaces from the atmosphere; or the slotted membrane has more than 10 percent open area.

(ii) The owner or operator shall inspect the internal floating roof components as follows except as provided for in paragraph (b)(1)(iii) of this section:

(A) Visually inspect the internal floating roof components through openings on the fixed-roof (e.g., manholes and roof hatches) at least once every calendar year after initial fill, and

(B) Visually inspect the internal floating roof, primary seal, secondary seal (if one is in service), gaskets, slotted membranes, and sleeve seals (if any) each time the tank is emptied and degassed and at least every 10 years. Prior to each inspection, the owner or operator shall notify the Administrator in accordance with the reporting requirements specified in §63.697 of this subpart.

(iii) As an alternative to performing the inspections specified in paragraph (b)(1)(ii) of this section for an internal floating roof equipped with two continuous seals mounted one above the other, the owner or operator may visually inspect the internal floating roof, primary and secondary seals, gaskets, slotted membranes, and sleeve seals (if any) each time the tank is emptied and degassed and at least every 5 years. Prior to each inspection, the owner or operator shall notify the Administrator in accordance with the reporting requirements specified in §63.697 of this subpart.

(iv) In the event that a defect is detected, the owner or operator shall repair the defect in accordance with the requirements of paragraph (b)(4) of this section.

(v) The owner or operator shall maintain a record of the inspection in accordance with the requirements specified in §63.696 of this subpart.

(2) Owners and operators that use a tank equipped with an external floating roof in accordance with the provisions of §63.685(f) of this subpart shall meet the following requirements:

(i) The owner or operator shall measure the external floating roof seal gaps in accordance with the following requirements:

(A) The owner or operator shall perform measurements of gaps between the tank wall and the primary seal within 60 days after initial operation of the tank following installation of the floating roof and, thereafter, at least once every 5 years. Prior to each inspection, the owner or operator shall notify the Administrator in accordance with the reporting requirements specified in §63.697 of this subpart.

(B) The owner or operator shall perform measurements of gaps between the tank wall and the secondary seal within 60 days after initial operation of the separator following installation of the floating roof and, thereafter, at least once every year. Prior to each inspection, the owner or operator shall notify the Administrator in accordance with the reporting requirements specified in §63.697 of this subpart.

(C) If a tank ceases to hold off-site material for a period of 1 year or more, subsequent introduction of off-site material into the tank shall be considered an initial operation for the purposes of paragraphs (b)(2)(i)(A) and (b)(2)(i)(B) of this section.

(D) The owner shall determine the total surface area of gaps in the primary seal and in the secondary seal individually using the following procedure.

(1) The seal gap measurements shall be performed at one or more floating roof levels when the roof is floating off the roof supports.

(2) Seal gaps, if any, shall be measured around the entire perimeter of the floating roof in each place where a 0.32centimeter (cm) ($\frac{1}{6}$ -inch) diameter uniform probe passes freely (without forcing or binding against the seal) between the seal and the wall of the tank and measure the circumferential distance of each such location.

(3) For a seal gap measured under paragraph (b)(2) of this section, the gap surface area shall be determined by

using probes of various widths to measure accurately the actual distance from the tank wall to the seal and multiplying each such width by its respective circumferential distance.

(4) The total gap area shall be calculated by adding the gap surface areas determined for each identified gap location for the primary seal and the secondary seal individually, and then dividing the sum for each seal type by the nominal diameter of the tank. These total gap areas for the primary seal and secondary seal are then compared to the respective standards for the seal type as specified in §63.685(f)(1) of this subpart.

(E) In the event that the seal gap measurements do not conform to the specifications in $\S63.685(f)(1)$ of this subpart, the owner or operator shall repair the defect in accordance with the requirements of paragraph (b)(4) of this section.

(F) The owner or operator shall maintain a record of the inspection in accordance with the requirements specified in §63.696 of this subpart.

(ii) The owner or operator shall visually inspect the external floating roof in accordance with the following requirements:

(A) The floating roof and its closure devices shall be visually inspected by the owner or operator to check for defects that could result in air emissions. Defects include, but are not limited to: holes, tears, or other openings in the rim seal or seal fabric of the floating roof; a rim seal detached from the floating roof; all or a portion of the floating roof deck being submerged below the surface of the liquid in the tank; broken, cracked, or otherwise damaged seals or gaskets on closure devices; and broken or missing hatches, access covers, caps, or other closure devices.

(B) The owner or operator shall perform the inspections following installation of the external floating roof and, thereafter, at least once every year.

(C) In the event that a defect is detected, the owner or operator shall repair the defect in accordance with the requirements of paragraph (b)(4) of this section.

(D) The owner or operator shall maintain a record of the inspection in

accordance with the requirements specified in §63.696(d) of this subpart.

(3) Owners and operators that use a tank equipped with a fixed roof in accordance with the provisions of §63.685(g) of this subpart shall meet the following requirements:

(i) The fixed roof and its closure devices shall be visually inspected by the owner or operator to check for defects that could result in air emissions. Defects include, but are not limited to. visible cracks, holes, or gaps in the roof sections or between the roof and the separator wall; broken, cracked, or otherwise damaged seals or gaskets on closure devices; and broken or missing hatches, access covers, caps, or other closure devices. In the case when a tank is buried partially or entirely underground, inspection is required only for those portions of the cover that extend to or above the ground surface, and those connections that are on such portions of the cover (e.g., fill ports, access hatches, gauge wells, etc.) and can be opened to the atmosphere.

(ii) The owner or operator must perform an initial inspection following installation of the fixed roof. Thereafter, the owner or operator must perform the inspections at least once every calendar year except as provided for in paragraph (f) of this section.

(iii) In the event that a defect is detected, the owner or operator shall repair the defect in accordance with the requirements of paragraph (b)(4) of this section.

(iv) The owner or operator shall maintain a record of the inspection in accordance with the requirements specified in §63.696(e) of this subpart.

(4) The owner or operator shall repair each defect detected during an inspection performed in accordance with the requirements of paragraph (b)(1), (b)(2), or (b)(3) of this section in the following manner:

(i) The owner or operator shall within 45 calendar days of detecting the defect either repair the defect or empty the tank and remove it from service. If within this 45-day period the defect cannot be repaired or the tank cannot be removed from service without disrupting operations at the plant site, the owner or operator is allowed two 30-day extensions. In cases when an 40 CFR Ch. I (7–1–23 Edition)

owner or operator elects to use a 30-day extension, the owner or operator shall prepare and maintain documentation describing the defect, explaining why alternative storage capacity is not available, and specify a schedule of actions that will ensure that the control equipment will be repaired or the tank emptied as soon as possible.

(ii) When a defect is detected during an inspection of a tank that has been emptied and degassed, the owner or operator shall repair the defect before refilling the tank.

(c) Owners and operators that use a closed-vent system in accordance with the provisions of §63.693 of this subpart shall meet the following inspection and monitoring requirements:

(1) Each closed-vent system that is used to comply with 63.693(c)(1)(i) of this subpart shall be inspected and monitored in accordance with the following requirements:

(i) At initial startup, the owner or operator shall monitor the closed-vent system components and connections using the procedures specified in $\S63.694(k)$ of this subpart to demonstrate that the closed-vent system operates with no detectable organic emissions.

(ii) After initial startup, the owner or operator shall inspect and monitor the closed-vent system as follows:

(A) Closed-vent system joints, seams, or other connections that are permanently or semi-permanently sealed (e.g., a welded joint between two sections of hard piping or a bolted and gasketed ducting flange) shall be visually inspected at least once per year to check for defects that could result in air emissions. The owner or operator shall monitor a component or connection using the procedures specified in §63.694(k) of this subpart to demonstrate that it operates with no detectable organic emissions following any time the component is repaired or replaced (e.g., a section of damaged hard piping is replaced with new hard piping) or the connection is unsealed (e.g., a flange is unbolted).

(B) Closed-vent system components or connections other than those specified in paragraph (c)(1)(ii)(A) of this section, shall be monitored at least once per year using the procedures

specified in §63.694(k) of this subpart to demonstrate that components or connections operate with no detectable organic emissions.

(C) The continuous monitoring system required by §63.693(b)(4)(i) shall monitor and record either an instantaneous data value at least once every 15 minutes or an average value for intervals of 15 minutes or less.

(D) The owner or operator shall visually inspect the seal or closure mechanism required by 63.693(c)(2)(ii) at least once every month to verify that the bypass mechanism is maintained in the closed position.

(iii) In the event that a defect or leak is detected, the owner or operator shall repair the defect or leak in accordance with the requirements of paragraph (c)(3) of this section.

(iv) The owner or operator shall maintain a record of the inspection and monitoring in accordance with the requirements specified in §63.696 of this subpart.

(2) Each closed-vent system that is used to comply with 63.693(c)(1)(ii) of this subpart shall be inspected and monitored in accordance with the following requirements:

(i) The closed-vent system shall be visually inspected by the owner or operator to check for defects that could result in air emissions. Defects include, but are not limited to, visible cracks, holes, or gaps in ductwork or piping; loose connections; or broken or missing caps or other closure devices.

(ii) The owner or operator must perform an initial inspection following installation of the closed-vent system. Thereafter, the owner or operator must perform the inspections at least once every calendar year except as provided for in paragraph (f) of this section.

(iii) In the event that a defect is detected, the owner or operator shall repair the defect in accordance with the requirements of paragraph (c)(3) of this section.

(iv) The owner or operator shall maintain a record of the inspection in accordance with the requirements specified in §63.696 of this subpart.

(3) The owner or operator shall repair all detected defects as follows:

(i) The owner or operator shall make first efforts at repair of the defect no

later than 5 calendar days after detection and repair shall be completed as soon as possible but no later than 45 calendar days after detection.

(ii) Repair of a defect may be delayed beyond 45 calendar days if either of the conditions specified in paragraph (c)(3)(ii)(A) or (c)(3)(ii)(B) occurs. In this case, the owner or operator must repair the defect the next time the process or unit that vents to the closed-vent system is shutdown. Repair of the defect must be completed before the process or unit resumes operation.

(A) Completion of the repair is technically infeasible without the shutdown of the process or unit that vents to the closed-vent system.

(B) The owner or operator determines that the air emissions resulting from the repair of the defect within the specified period would be greater than the fugitive emissions likely to result by delaying the repair until the next time the process or unit that vents to the closed-vent system is shutdown.

(iii) The owner or operator shall maintain a record of the defect repair in accordance with the requirements specified in §63.696 of this subpart.

(d) Owners and operators that use a transfer system equipped with a cover in accordance with the provisions of §63.689(c)(1) of this subpart shall meet the following inspection requirements:

(1) The cover and its closure devices shall be visually inspected by the owner or operator to check for defects that could result in air emissions. Defects include, but are not limited to, visible cracks, holes, or gaps in the cover sections or between the cover and its mounting; broken, cracked, or otherwise damaged seals or gaskets on closure devices; and broken or missing hatches, access covers, caps, or other closure devices. In the case when a transfer system is buried partially or entirely underground, inspection is required only for those portions of the cover that extend to or above the ground surface, and those connections that are on such portions of the cover (e.g., access hatches, etc.) and can be opened to the atmosphere.

(2) The owner or operator must perform an initial inspection following installation of the cover. Thereafter, the owner or operator must perform the inspections at least once every calendar year except as provided for in paragraph (f) of this section.

(3) In the event that a defect is detected, the owner or operator shall repair the defect in accordance with the requirements of paragraph (d)(5) of this section.

(4) The owner or operator shall maintain a record of the inspection in accordance with the requirements specified in §63.696 of this subpart.

(5) The owner or operator shall repair all detected defects as follows:

(i) The owner or operator shall make first efforts at repair of the defect no later than 5 calendar days after detection and repair shall be completed as soon as possible but no later than 45 calendar days after detection except as provided in paragraph (d)(5)(ii) of this section.

(ii) Repair of a defect may be delayed beyond 45 calendar days if the owner or operator determines that repair of the defect requires emptying or temporary removal from service of the transfer system and no alternative transfer system is available at the site to accept the material normally handled by the system. In this case, the owner or operator shall repair the defect the next time the process or unit that is generating the material handled by the transfer system stops operation. Repair of the defect must be completed before the process or unit resumes operation.

(iii) The owner or operator shall maintain a record of the defect repair in accordance with the requirements specified in §63.696 of this subpart.

(e) Control device monitoring requirements. For each control device required under §63.693 to be monitored in accordance with the provisions of this paragraph (e), the owner or operator must ensure that each control device operates properly by monitoring the control device in accordance with the requirements specified in paragraphs (e)(1) through (5) of this section.

(1) A continuous parameter monitoring system must be used to measure the operating parameter or parameters specified for the control device in §63.693(d) through §63.693(g) of this subpart as applicable to the type and design of the control device. The contin40 CFR Ch. I (7–1–23 Edition)

uous parameter monitoring system must meet the following specifications and requirements:

(i) The continuous parameter monitoring system must measure either an instantaneous value at least once every 15 minutes or an average value for intervals of 15 minutes or less and continuously record either:

(A) Each measured data value; or

(B) Each block average value for each 1-hour period or shorter periods calculated from all measured data values during each period. If values are measured more frequently than once per minute, a single value for each minute may be used to calculate the hourly (or shorter period) block average instead of all measured values.

(ii) The monitoring system must be installed, calibrated, operated, and maintained in accordance with the manufacturer's specifications or other written procedures that provide reasonable assurance that the monitoring equipment is operating properly.

(2) Using the data recorded by the monitoring system, the owner or operator must calculate the daily average value for each monitored operating parameter for each operating day. If operation of the control device is continuous, the operating day is a 24-hour period. If control device operation is not continuous, the operating day is the total number of hours of control device operation per 24-hour period. Valid data points must be available for 75 percent of the operating hours in an operating day to compute the daily average.

(3) For each monitored operating parameter, the owner or operator must establish a minimum operating parameter value or a maximum operating parameter value, as appropriate, to define the range of conditions at which the control device must be operated to continuously achieve the applicable performance requirements specified in $\S 63.693(b)(2)$ of this subpart. Each minimum or maximum operating parameter value must be established in accordance with the requirements in paragraphs (e)(3)(i) and (e)(3)(ii) of this section.

(i) If the owner or operator conducts a performance test to demonstrate control device performance, then the minimum or maximum operating parameter value must be established based on values measured during the performance test and supplemented, as necessary, by the control device design specifications, manufacturer recommendations, or other applicable information.

(ii) If the owner or operator uses a control device design analysis to demonstrate control device performance, then the minimum or maximum operating parameter value must be established based on the control device design analysis and supplemented, as necessary, by the control device manufacturer recommendations or other applicable information.

(4) A deviation for a given control device is determined to have occurred when the monitoring data or lack of monitoring data result in any one of the criteria specified in paragraphs (e)(4)(i) through (iii) of this section being met. When multiple operating parameters are monitored for the same control device and during the same operating day more than one of these operating parameters meets a deviation criterion specified in paragraphs (e)(4)(i) through (iii) of this section, then a single deviation is determined to have occurred for the control device for that operating day.

(i) A deviation occurs when the daily average value of a monitored operating parameter is less than the minimum operating parameter limit (or, if applicable, greater than the maximum operating parameter limit) established for the operating parameter in accordance with the requirements of paragraph (e)(3) of this section.

(ii) A deviation occurs when the period of control device operation is 4 hours or greater in an operating day and the monitoring data are insufficient to constitute a valid hour of data for at least 75 percent of the operating hours. Monitoring data are insufficient to constitute a valid hour of data if measured values are unavailable for any of the 15-minute periods within the hour.

(iii) A deviation occurs when the period of control device operation is less than 4 hours in an operating day and more than 1 of the hours during the period does not constitute a valid hour of data due to insufficient monitoring data. Monitoring data are insufficient to constitute a valid hour of data if measured values are unavailable for any of the 15-minute periods within the hour.

(5) For each deviation, except when the deviation occurs during periods of non-operation of the unit or the process that is vented to the control device (resulting in cessation of HAP emissions to which the monitoring applies), the owner or operator shall be deemed to have failed to have applied control in a manner that achieves the required operating parameter limits. Failure to achieve the required operating parameter limits is a violation of this standard.

(f) Alternative inspection and monitoring interval. Following the initial inspection and monitoring of a piece of air pollution control equipment in accordance with the applicable provisions of this section, subsequent inspection and monitoring of the equipment may be performed at intervals longer than 1 year when an owner or operator determines that performing the required inspection or monitoring procedures would expose a worker to dangerous, hazardous, or otherwise unsafe conditions and the owner or operator complies with the requirements specified in paragraphs (f)(1) and (f)(2) of this section.

(1) The owner or operator must prepare and maintain at the plant site written documentation identifying the specific air pollution control equipment designated as "unsafe to inspect and monitor." The documentation must include for each piece of air pollution control equipment designated as such a written explanation of the reasons why the equipment is unsafe to inspect or monitor using the applicable procedures under this section.

(2) The owner or operator must develop and implement a written plan and schedule to inspect and monitor the air pollution control equipment using the applicable procedures specified in this section during times when a worker can safely access the air pollution control equipment. The required inspections and monitoring must be performed as frequently as practicable but do not need to be performed more frequently than the periodic schedule that would be otherwise applicable to the air pollution control equipment under the provisions of this section. A copy of the written plan and schedule must be maintained at the plant site.

[64 FR 38977, July 20, 1999, as amended at 68 FR 37352, June 23, 2003; 71 FR 20457, Apr. 20, 2006; 80 FR 14278, Mar. 18, 2015]

§63.696 Recordkeeping requirements.

(a) The owner or operator subject to this subpart shall comply with the recordkeeping requirements in §63.10 under 40 CFR 63 subpart A—General Provisions that are applicable to this subpart as specified in Table 2 of this subpart.

(b) The owner or operator of a control device subject to this subpart shall maintain the records in accordance with the requirements of 40 CFR 63.10 of this part.

(c) [Reserved]

(d) Each owner or operator using an internal floating roof to comply with the tank control requirements specified in 63.685(e) of this subpart or using an external floating roof to comply with the tank control requirements specified in 63.685(f) of this subpart shall prepare and maintain the following records:

(1) Documentation describing the floating roof design and the dimensions of the tank.

(2) A record for each inspection required by §63.695(b) of this subpart, as applicable to the tank, that includes the following information: a tank identification number (or other unique identification description as selected by the owner or operator) and the date of inspection.

(3) The owner or operator shall record for each defect detected during inspections required by 63.695(b) of this subpart the following information: the location of the defect, a description of the defect, the date of detection, and corrective action taken to repair the defect. In the event that repair of the defect is delayed in accordance with the provisions of 63.695(b)(4) of this section, the owner or operator shall also record the reason for the delay and 40 CFR Ch. I (7–1–23 Edition)

the date that completion of repair of the defect is expected.

(4) Owners and operators that use a tank equipped with an external floating roof in accordance with the provisions of §63.685(f) of this subpart shall prepare and maintain records for each by seal gap inspection required §63.695(b) describing the results of the seal gap measurements. The records shall include the date of that the measurements are performed, the raw data obtained for the measurements, and the calculations of the total gap surface area. In the event that the seal gap measurements do not conform to the specifications in §63.695(b) of this subpart, the records shall include a description of the repairs that were made, the date the repairs were made, and the date the separator was emptied, if necessary.

(e) Each owner or operator using a fixed roof to comply with the tank control requirements specified in §63.685(g) of this subpart shall prepare and maintain the following records:

(1) A record for each inspection required by §63.695(b) of this subpart, as applicable to the tank, that includes the following information: a tank identification number (or other unique identification description as selected by the owner or operator) and the date of inspection.

(2) The owner or operator shall record for each defect detected during inspections required by §63.695(b) of this subpart the following information: the location of the defect, a description of the defect, the date of detection, and corrective action taken to repair the defect. In the event that repair of the defect is delayed in accordance with the provisions of §63.695(b)(4) of this section, the owner or operator shall also record the reason for the delay and the date that completion of repair of the defect is expected.

(f) Each owner or operator using an enclosure to comply with the tank control requirements specified in §63.685(i) of this subpart shall prepare and maintain records for the most recent set of calculations and measurements performed by the owner or operator to verify that the enclosure meets the criteria of a permanent total enclosure as specified in "Procedure T—Criteria for

and Verification of a Permanent or Temporary Total Enclosure'' under 40 CFR 52.741, appendix B.

(g) An owner or operator shall record, on a semiannual basis, the information specified in paragraphs (g)(1) and (g)(2) of this section for those planned routine maintenance operations that would require the control device not to meet the requirements of 63.693(d)through (h) of this subpart, as applicable.

(1) A description of the planned routine maintenance that is anticipated to be performed for the control device during the next 6 months. This description shall include the type of maintenance necessary, planned frequency of maintenance, and lengths of maintenance periods.

(2) A description of the planned routine maintenance that was performed for the control device during the previous 6 months. This description shall include the type of maintenance performed and the total number of hours during these 6 months that the control device did not meet the requirement of $\S63.693$ (d) through (h) of this subpart, as applicable, due to planned routine maintenance.

(h) An owner or operator shall record the malfunction information specified in paragraphs (h)(1) through (3) of this section.

(1) In the event that an affected unit fails to meet an applicable standard, record the number of failures. For each failure, record the date, time and duration of the failure.

(2) For each failure to meet an applicable standard, record and retain a list of the affected sources or equipment, an estimate of the volume of each regulated pollutant emitted over any emission limit and a description of the method used to estimate the emissions.

(3) Record actions taken to minimize emissions in accordance with §63.683(e) and any corrective actions taken to return the affected unit to its normal or usual manner of operation.

(i) For pressure relief devices in offsite material service, keep records of the information specified in paragraphs (i)(1) through (5) of this section, as applicable.

(1) A list of identification numbers for pressure relief devices that the

owner or operator elects to route emissions through a closed-vent system to a control device, process or drain system under the provisions in $\S63.691(c)(4)$.

(2) A list of identification numbers for pressure relief devices that do not consist of or include a rupture disk, subject to the provisions in $\S63.691(c)(2)(i)$.

(3) A list of identification numbers for pressure relief devices equipped with rupture disks, subject to the provisions in 63.691(c)(2)(i).

(4) The dates and results of the Method 21 of 40 CFR part 60, appendix A, monitoring following a pressure release for each pressure relief device subject to the provisions in $\S63.691(c)(2)(i)$. The results of each monitoring event shall include:

(i) The measured background level.

(ii) The maximum instrument reading measured at each pressure relief device.

(5) For pressure relief devices in offsite material service subject to §63.691(c)(3), keep records of each pressure release to the atmosphere, including the following information:

(i) The source, nature, and cause of the pressure release.

(ii) The date, time, and duration of the pressure release.

(iii) An estimate of the quantity of HAP listed in Table 1 of this subpart emitted during the pressure release and the calculations used for determining this quantity.

(iv) The actions taken to prevent this pressure release.

(v) The measures adopted to prevent future such pressure releases.

(j) (1) For pressure tank closure devices, as specified in 63.685(h)(2), keep records of each release to the atmosphere, including the information specified in paragraphs (j)(3) though (7) of this section.

(2) For each closed vent system that includes bypass devices that could divert a stream away from the control device and into the atmosphere, as specified in §63.693(c)(2), and each openended valve or line in an emergency shutdown system which is designed to open automatically in the event of a process upset, as specified in §63.167(d)or 40 CFR 61.242-6(d), keep records of each release to the atmosphere, including the information specified in paragraphs (j)(3) though (9) of this section.

(3) The source, nature, and cause of the release.

(4) The date, time, and duration of the release.

(5) An estimate of the quantity of HAP listed in Table 1 of this subpart emitted during the release and the calculations used for determining this quantity.

(6) The actions taken to prevent this release.

(7) The measures adopted to prevent future such release.

(8) Hourly records of whether the bypass flow indicator specified under $\S63.693(c)(2)$ was operating and whether a diversion was detected at any time during the hour, as well as records of the times of all periods when the vent stream is diverted from the control device or the flow indicator is not operating.

(9) Where a seal mechanism is used to comply with $\S63.693(c)(2)$, hourly records of flow are not required. In such cases, the owner or operator shall record that the monthly visual inspection of the seals or closure mechanism has been done, and shall record the duration of all periods when the seal mechanism is broken, the bypass line valve position has changed, or the key for a lock-and-key type lock has been checked out, and records of any carseal that has broken.

[61 FR 34158, July 1, 1996, as amended at 80 FR 14279, Mar. 18, 2015]

§63.697 Reporting requirements.

(a) Each owner or operator of an affected source subject to this subpart must comply with the notification requirements specified in paragraph (a)(1) of this section and the reporting requirements specified in paragraphs (a)(2) and (3) of this section.

(1) The owner or operator of an affected source must submit notices to the Administrator in accordance with the applicable notification requirements in 40 CFR 63.9 as specified in Table 2 of this subpart. For the purpose of this subpart, an owner or operator subject to the initial notification requirements under 40 CFR 63.9(b)(2) must submit the required notification 40 CFR Ch. I (7–1–23 Edition)

on or before October 19, 1999, or no later than 120 days after the source becomes subject to this subpart, whichever is later.

(i) For pressure relief devices in offsite material service subject to the requirements of $\S63.691(c)$, the owner or operator must submit the information listed in paragraph (a)(1)(ii) of this section in the notification of compliance status required under $\S63.9(h)$ within 150 days after the first applicable compliance date for pressure relief device monitoring.

(ii) For pressure relief devices in offsite material service, a description of the device or monitoring system to be implemented, including the pressure relief devices and process parameters to be monitored (if applicable), a description of the alarms or other methods by which operators will be notified of a pressure release, and a description of how the owner or operator will determine the information to be recorded under §63.696(i)(5)(ii) through (iii) (i.e., the duration of the pressure release and the methodology and calculations for determining the quantity of HAP listed in Table 1 of this subpart emitted during the pressure release).

(2) The owner or operator of an affected source must submit reports to the Administrator in accordance with the applicable reporting requirements in 40 CFR 63.10 as specified in Table 2 of this subpart.

(3) Electronic reporting. Within 60 days after the date of completing each performance test (as defined in §63.2) required by this subpart, the owner or operator must submit the results of the performance test according to the manner specified by either paragraph (a)(3)(i) or (ii) of this section.

(i) For data collected using test methods supported by the EPA's Electronic Reporting Tool (ERT) as listed on the EPA's ERT Web site (http:// www.epa.gov/ttn/chief/ert/index.html), the owner or operator must submit the results of the performance test to the EPA via the Compliance and Emissions Data Reporting Interface (CEDRI) accessed through the EPA's Central Data Exchange (CDX) (http:// cdx.epa.gov/epa_home.asp). Performance test data must be submitted in a file format generated through the use of

the EPA's ERT. Owners or operators who claim that some of the performance test information being submitted is confidential business information (CBI) must submit a complete file generated through the use of the EPA's ERT, including information claimed to be CBI, on a compact disc, flash drive, or other commonly used electronic storage media to the EPA. The electronic media must be clearly marked as CBI and mailed to U.S. EPA/OAPQS/ CORE CBI Office, Attention: Group Leader, Measurement Policy Group, MD C404-02, 4930 Old Page Road, Durham, NC 27703. The same ERT file with the CBI omitted must be submitted to the EPA via the EPA's CDX as described earlier in this paragraph (a)(3)(i)

(ii) For data collected using test methods that are not supported by the EPA's ERT as listed on the EPA's ERT Web site, the owner or operator must submit the results of the performance test to the Administrator at the appropriate address listed in 40 CFR 60.4.

(b) The owner or operator of a control device used to meet the requirements of §63.693 of this subpart shall submit the following notifications and reports to the Administrator:

(1) A Notification of Performance Tests specified in 63.7 and 63.9(g) of this part,

(2) Performance test reports specified in 63.10(d)(2) of this part, and

(3) Reports of malfunctions. If a source fails to meet an applicable standard, report such events in the Periodic Report. Report the number of failures to meet an applicable standard. For each instance, report the date, time and duration of each failure. For each failure the report must include a list of the affected sources or equipment, an estimate of the volume of each regulated pollutant emitted over any emission limit, and a description of the method used to estimate the emissions.

(4) A summary report specified in $\S63.10(e)(3)$ shall be submitted on a semiannual basis (*i.e.*, once every 6-month period). The summary report must include a description of all deviations as defined in $\S\S63.683(f)$ and 63.695(e) that have occurred during the 6-month reporting period. For each deviation caused when the daily average

value of a monitored operating parameter is less than the minimum operating parameter limit (or, if applicable, greater than the maximum operating parameter limit), the report must include the daily average values of the monitored parameter, the applicable operating parameter limit, and the date and duration of the period that the deviation occurred. For each deviation caused by lack of monitoring data, the report must include the date and duration of period when the monitoring data were not collected and the reason why the data were not collected.

(5) For pressure relief devices in offsite material service subject to 63.691(c), Periodic Reports must include the information specified in paragraphs (b)(5)(i) through (iii) of this section.

(i) For pressure relief devices in offsite material service subject to §63.691(c), report the results of all monitoring conducted within the reporting period.

(ii) For pressure relief devices in gas/ vapor service subject to §63.691(c)(2)(i), report any instrument reading of 500 ppm above background or greater, if detected more than 5 days after the pressure release.

(iii) For pressure relief devices in offsite material service subject to §63.691(c)(3), report each pressure release to the atmosphere, including the following information:

(A) The source, nature, and cause of the pressure release.

(B) The date, time, and duration of the pressure release.

(C) An estimate of the quantity of HAP listed in Table 1 of this subpart emitted during the pressure release and the method used for determining this quantity.

(D) The actions taken to prevent this pressure release.

(E) The measures adopted to prevent future such pressure releases.

(6) Pressure tank closure device or bypass deviation report. The owner or operator must submit to the Administrator the information specified in paragraph (b)(6)(iv) of this section when any of the conditions in paragraphs (b)(6)(i) through (iii) of this section are met. (i) Any pressure tank closure device, as specified in §63.685(h)(2), has released to the atmosphere.

(ii) Any closed vent system that includes bypass devices that could divert a vent a stream away from the control device and into the atmosphere, as specified in §63.693(c)(2), has released directly to the atmosphere.

(iii) Any open-ended valve or line in an emergency shutdown system which is designed to open automatically in the event of a process upset, as specified in §63.167(d) or 40 CFR 61.242-6(d), has released directly to the atmosphere.

(iv) The pressure tank closure device or bypass deviation report must include the information specified in paragraphs (b)(6)(iv)(A) through (E) of this section.

(A) The source, nature and cause of the release.

(B) The date, time and duration of the discharge.

(C) An estimate of the quantity of HAP listed in Table 1 of this subpart emitted during the release and the method used for determining this quantity.

(D) The actions taken to prevent this release.

(E) The measures adopted to prevent future such releases.

(c) Each owner or operator using an internal floating roof or external floating roof to comply with the Tank Level 2 control requirements specified in §63.685(d) of this subpart shall notify the Administrator in advance of each inspection required under §63.695(b) of this subpart to provide the Administrator with the opportunity to have an observer present during the inspection. The owner or operator shall notify the Administrator of the date and location of the inspection as follows:

(1) Prior to each inspection to measure external floating roof seal gaps as required under §63.695(b) of this subpart, written notification shall be prepared and sent by the owner or operator so that it is received by the Administrator at least 30 calendar days before the date the measurements are scheduled to be performed.

(2) Prior to each visual inspection of an internal floating roof or external floating roof in a tank that has been emptied and degassed, written notification shall be prepared and sent by the owner or operator so that it is received by the Administrator at least 30 calendar days before refilling the tank except when an inspection is not planned as provided for in paragraph (c)(3) of this section.

(3) When a visual inspection is not planned and the owner or operator could not have known about the inspection 30 calendar days before refilling the tank, the owner or operator shall notify the Administrator as soon as possible, but no later than 7 calendar days before refilling of the tank. This notification may be made by telephone and immediately followed by a written explanation for why the inspection is unplanned. Alternatively, written notification, including the explanation for the unplanned inspection, may be sent so that it is received by the Administrator at least 7 calendar days before refilling the tank.

[61 FR 34158, July 1, 1996, as amended at 64
FR 38981, July 20, 1999; 80 FR 14279, Mar. 18, 2015; 85 FR 73893, Nov. 19, 2020]

§63.698 Implementation and enforcement.

(a) This subpart can be implemented and enforced by the U.S. EPA, or a delegated authority such as the applicable State, local, or Tribal agency. If the U.S. EPA Administrator has delegated authority to a State, local, or Tribal agency, then that agency, in addition to the U.S. EPA, has the authority to implement and enforce this subpart. Contact the applicable U.S. EPA Regional Office to find out if this subpart is delegated to a State, local, or Tribal agency.

(b) In delegating implementation and enforcement authority of this subpart to a State, local, or Tribal agency under subpart E of this part, the authorities contained in paragraph (c) of this section are retained by the Administrator of U.S. EPA and cannot be transferred to the State, local, or Tribal agency.

(c) The authorities that cannot be delegated to State, local, or Tribal agencies are as specified in paragraphs (c)(1) through (5) of this section.

(1) Approval of alternatives to the requirements in \$ 63.680, 63.683 through

63.691, and 63.693. Where these standards reference another subpart, the cited provisions will be delegated according to the delegation provisions of the referenced subpart.

(2) Approval of major alternatives to test methods under §63.7(e)(2)(ii) and (f), as defined in §63.90, and as required in this subpart.

(3) Approval of major alternatives to monitoring under §63.8(f), as defined in §63.90, and as required in this subpart.

Pt. 63, Subpt. DD, Table 1

0.499

1.000

(4) Approval of major alternatives to recordkeeping and reporting under §63.10(f), as defined in §63.90, and as required in this subpart.

(5) Approval of alternatives to the electronic reporting requirements in §63.697(a)(3).

[68 FR 37352, June 23, 2003, as amended at 80 FR 14280, Mar. 18, 2015]

CAS No. a	Chemical name	$f_{\rm m}$ 305
75–07–0	Acetaldehyde	1.000
75–05–8	Acetonitrile	0.989
98–86–2	Acetophenone	0.314
107–02–8	Acrolein	1.000
107–13–1	Acrylonitrile	0.999
107–05–1	Allyl chloride	1.000
71–43–2	Benzene (includes benzene in gasoline)	1.000
98–07–7	Benzotrichloride (isomers and mixture)	0.958
100-44-7	Benzyl chloride	1.000
92–52–4	Biphenyl	0.864
542-88-1	Bis(chloromethyl)ether ^b	0.999
75–25–2	Bromoform	0.998
106–99–0	1.3-Butadiene	1.000
75–15–0	Carbon disulfide	1.000
56–23–5	Carbon tetrachloride	1.000
43-58-1	Carbonyl sulfide	1.000

TABLE 1 TO SUBPART DD OF PART 63—LIST OF HAZARDOUS AIR POLLUTANTS (HAP) FOR SUBPART DD

(5-05-8	Acetonitrie	0.989
98–86–2	Acetophenone	0.314
107–02–8	Acrolein	1.000
107–13–1	Acrylonitrile	0.999
107–05–1	Allyl chloride	1.000
71–43–2	Benzene (includes benzene in gasoline)	1.000
98–07–7	Benzotrichloride (isomers and mixture)	0.958
100-44-7	Benzyl chloride	1.000
92-52-4	Biphenyl	0.864
542-88-1	Bis(chloromethyl)ether ^b	0.999
75_25_2	Bromoform	0.000
106_00_0	1 3-Butadiene	1 000
75_15_0	Carbon disulfide	1.000
56 23 5	Carbon disalide	1.000
12 50 1		1.000
122 00 4		1.000
100 00 7		0.033
108–90–7	Chlorobenzene	1.000
07-00-3	Chiorotorm	1.000
107–30–2		1.000
126–99–8	Chloroprene	1.000
98–82–8	Cumene	1.000
94–75–7	2,4-D, salts and esters	0.167
334–88–3	Diazomethane c	0.999
132–64–9	Dibenzofurans	0.967
96–12–8	1,2-Dibromo-3-chloropropane	1.000
106–46–7	1,4-Dichlorobenzene(p)	1.000
107–06–2	Dichloroethane (Ethylene dichloride)	1.000
111–44–4	Dichloroethyl ether (Bis(2-chloroethyl ether)	0.757
542–75–6	1,3-Dichloropropene	1.000
79–44–7	Dimethyl carbamoyl chloride ^c	0.150
64–67–5	Diethyl sulfate	0.0025
77–78–1	Dimethyl sulfate	0.086
121-69-7	N.N-Dimethylaniline	0.0008
51-28-5	2 4-Dinitrophenol	0.0077
121_14_2	2.4-Dinitrotoluene	0.0848
123_91_1	1 4-Dioxane (1 4-Diethyleneoxide)	0.869
106_89_8	Enichlorobydrin (1-Chloro-2 3-enovypronane)	0.000
106 88 7	1.2 Epotybutane	1 000
140 99 5	The conducto	1.000
100 44 4	Lifyi aciyiate	1.000
75 00 0	Ethyl belavida (Ohlassathana)	1.000
(5-00-3	Ethyl chloride (Chloroethane)	1.000
106-93-4		0.999
107–06–2	Ethylene dichloride (1,2-Dichloroethane)	1.000
151–56–4	Ethylene imine (Aziridine)	0.867
(5-21-8	Ethylene oxide	1.000
75–34–3	Ethylidene dichloride (1,1-Dichloroethane)	1.000
	Glycol ethers ^d that have a Henry's Law constant value equal to or greater than 0.1 Y/X (1.8×10^{-6} atm/gm-mole/m ³) at 25°C.	(c)
118–74–1	Hexachlorobenzene	0.97
37–68–3	Hexachlorobutadiene	0.88

Hexachloroethane 110-54-3 ... Hexane ..

67-72-1
Pt. 63, Subpt. DD, Table 2

40 CFR Ch. I (7-1-23 Edition)

CAS No. ^a	Chemical name	f _m 305
78–59–1	Isophorone	0.506
58–89–9	Lindane (all isomers)	1.000
67–56–1	ethanol	0.855
74-83-9	□ethyl bromide (Bromomethane)	1.000
74–87–3	□ ethyl chloride (Choromethane)	1.000
71–55–6	ethyl chloroform (1,1,1-Trichloroethane)	1.000
78–93–3	□ethyl ethyl ketone (2-Butanone)	0.990
74–88–4	□ ethyl iodide (lodomethane)	1.0001
108–10–1	□ethyl isobutyl ketone (Hexone)	0.9796
624-83-9	□ ethyl isocvanate	1.000
80–62–6	□ ethyl methacrylate	0.916
1634-04-4	□ethyl tert butyl ether	1 000
75–09–2	□ ethylene chloride (Dichloromethane)	1.000
91–20–3	Naphthalene	0.994
98-95-3	Nitrobenzene	0.394
79–46–9	2-Nitropropane	0.989
82–68–8	Pentachloronitrobenzene (Quintobenzene)	0.839
87–86–5	Pentachlorophenol	0.0898
75–44–5	Phosene ^c	1 000
123_38_6	Pronionaldehyde	0.999
78–87–5	Propylene dichloride (1 2–Dichloropropane)	1 000
75–56–9	Pronylene oxide	1 000
75–55–8	1 2–Propylenimine (2–□ethyl aziridine)	0.945
100-42-5	Styrene	1 000
96-09-3	Styrene oxide	0.830
79–34–5	1 1 2 2-Tetrachloroethane	0.999
127_18_4	Tetrachloroethylene (Perchloroethylene)	1 000
108-88-3		1 000
95-53-4	o-Toluidine	0 152
120_82_1	1 2 4-Trichlorobenzene	1 000
71_55_6	1 1 1–Trichloroethane (□ ethyl chlorform)	1 000
79_00_5	1.1.2-Trichloroethane (Vinyl trichloride)	1 000
79_01_6	Trichloroethylene	1 000
95_95_4	2.4.5-Trichlorophenol	0 108
88–06–2	2.4.6-Trichlorophenol	0.132
121_44_8	Triethylamine	1 000
540-84-1	2.2.4-Trimethylpentane	1 000
108-05-4	Vinvl acetate	1 000
593_60_2	Vinyl bromide	1 000
75_01_4	Vinyl chloride	1.000
75–35–4	Vinylidene chloride (1 1–Dichloroethylene)	1.000
1330–20–7	Xylenes (isomers and mixture)	1.000
95-47-6	o_Xylenes	1.000
108_38_3	m-Yylenes	1.000
106_12_3	n-Xylanas	1.000
100 12-0	p Aylonoo	1.000

NoTES: m 305 □ ethod 305 fraction measure factor. a. CAS numbers refer to the Chemical Abstracts Services registry number assigned to specific compounds, isomers, or mix-tures of compounds. b. Denotes a HAP that hydrolyzes quickly in water, but the hydrolysis products are also HAP chemicals. c. Denotes a HAP that may react violently with water, exercise caustic is an expected analyte. d. Denotes a HAP that hydrolyzes slowly in water. e. The f_{m 305} factors for some of the more common glycol ethers can be obtained by contacting the □ aste and Chemical Proc-esses Group, Office of Air Quality Planning and Standards, Research Triangle Park, NC 27711.

[64 FR 38981, July 20, 1999]

TABLE 2 TO	SUBPART I	D of Pai	RT 63—АР	PLICABILITY	OF	PARAGRAPHS	$_{\rm IN}$	SUBPART
	A of This	Part 63—	-GENERAL	PROVISIONS	з тс	SUBPART DD		

Subpart A reference	Applies to Subpart DD	Explanation
· · ·		
63.1(a)(1)	Yes	
63.1(a)(2)	Yes	
63.1(a)(3)	Yes	
63.1(a)(4)	No	Subpart DD (this table) specifies applicability of each para- graph in subpart A to subpart DD.
63.1(a)(5)-63.1(a)(9)	No	
63.1(a)(10)	Yes	
63.1(a)(11)	Yes	
63.1(a)(12)	Yes	
63.1(b)(1)	No	Subpart DD specifies its own applicability.
63.1(b)(2)	No	Reserved.
63.1(b)(3)	No	

Pt. 63, Subpt. DD, Table 2

Subpart A reference	Applies to Subpart DD	Explanation
63.1(c)(1)	No	Subpart DD explicitly specifies requirements that apply.
63.1(c)(2)	No	Area sources are not subject to subpart DD.
63.1(c)(3)	No	Reserved.
63.1(c)(4)	No	Reserved.
63.1(c)(5)	Yes	Except that sources are not required to submit notifications
63 1(c)(6)	Yes	overhuden by this table.
63.1(d)	No	
63.1(e)	No	
63.2	Yes	□63.681 of subpart DD specifies that if the same term is de-
		fined in subparts A and DD, it shall have the meaning given in subpart DD.
63.3	Yes	g
63.4(a)(1)-63.4(a)(2)	Yes	
63.4(a)(3)	No	Reserved.
63.4(a)(4)	No	Reserved.
63.4(a)(5)	No	Reserved.
63.4(b)	Yes	
63.4(c)	Yes	
63.5(a)(1)	Yes	
63.5(a)(2)	Yes	
63.5(b)(1)	Yes	
63.5(b)(2)	No	Reserved.
63.5(D)(3)	Yes	
63.5(b)(4)	Yes	Except the cross-reference to □63.9(b) is changed to □63.9(b)(4) and (5). Subpart DD overrides □63.9(b)(2) and (b)(3).
63.5(b)(5)	No	Reserved.
63.5(b)(6)	Yes	
63.5(c)	No	Reserved.
63.5(d)(1)(i)	Yes	
63.5(d)(1)(ii)	Yes	
63.5(d)(1)(iii)	Yes	
63.5(d)(2)	No	
63.5(d)(3)	Yes	
63.5(d)(4)	Yes	
63.5(e)	Yes	
63.5(f)(1)	Yes	
63.5(T)(Z)	Yes	
63.6(b)(1)	No	Subpart DD specifies compliance dates for sources subject to
00.0(b)(1)		subpart DD.
03.0(D)(∠)	No	
63.6(b)(4)	No.	
63 6(b)(5)	No.	□63 697 of subpart DD includes notification requirements
63.6(b)(6)	No	
63.6(b)(7)	No	
63.6(c)(1)	No	□63.680 of subpart DD specifies the compliance date.
63.6(c)(2)-63.6(c)(4)	No	
63.6(c)(5)	Yes	
63.6(d)	No	
63.6(e)(1)(i)	No	See □63.683(e) for general duty requirement.
63.6(e)(1)(ii)	No	
63.6(e)(1)(III)	Yes	
63.6(e)(2)	NO	Reserved.
b3.b(e)(3)	NO No	
63.6(f)(1)	NO	
03.0(1)(2)(1)	Yes	Subpart DD aposition the use of manitaring data in datas
63.6(1)(2)(ii) (A) (B) and (C)	Yes	mining compliance with subpart DD.
63.6(f)(2)(iii) (D)	No	
63.6(f)(2)(iv)	Yes	
63.6(f)(2)(v)	Yes	
63.6(f)(3)	Yes	
63.6(g)	Yes	
63.6(h)	No	Subpart DD does not require opacity and visible emission
		standards.
63.6(i)	Yes	Except for
63.6()	Yes	
63.7(a)(1)	No	Subpart DD specifies required testing and compliance dem-
(2, 7(-))(2)	Vaa	onstration procedures.
os.≀(a)(∠)	res	

Pt. 63, Subpt. DD, Table 2

40 CFR Ch. I (7-1-23 Edition)

Subpart A reference	Applies to Subpart DD	Explanation
63.7(a)(3)	Yes	
63.7(a)(4)	Yes	
63.7(b)	Yes	
63.7(c)	Yes	
63.7(d)	Yes	
63.7(e)(1)	No	See □63.694(I).
63.7(e)(2)	Yes	
63.7(e)(3)	No	Subpart DD specifies test methods and procedures.
63.7(e)(4)	Yes	
63.7(a)	Yes	
63.7(h)(1)	Yes	
63.7(h)(2)	Yes	
63.7(h)(3)	Yes	
63.7(h)(4)	No	
63.7(h)(5)	Yes	
63.8(a)	No	
63.8(b)(2)	No	Subpart DD specifies locations to conduct monitoring
63.8(b)(2)	Yes	Subpart DD specifies locations to conduct monitoring.
63.8(c)(1)(i)	Yes	
63.8(c)(1)(ii)	Yes	
63.8(c)(1)(iii)	No	
63.8(c)(2)	Yes	
63.8(c)(3)	Yes	Submat DD enceifies menitoring for more set
0.0(C)(4)	No	Subbait OD specifies monitoring frequency
63.8(d)	No	
63.8(e)	No	
63.8(f)(1)	Yes	
63.8(f)(2)	Yes	
63.8(f)(3)	Yes	
63.8(f)(4)(i)	Yes	
63 8(f)(4)(iii)	No	
63.8(f)(5)(i)	Yes	
63.8(f)(5)(ii)	No	
63.8(f)(5)(iii)	Yes	
63.8(f)(6)	Yes	
63.8(g)	Yes	
63 9(b)(1)(i)	Yes	
63.9(b)(1)(ii)	No	
63.9(b)(2)	Yes	
63.9(b)(3)	No	
63.9(b)(4)	Yes	
63.9(D)(5)	Yes	
63.9(d)	Yes	
63.9(e)	Yes	
63.9(f)	No	
63.9(g)	Yes	
63.9(h)	Yes	
03.9(I)	Yes	For change in major source status only
63.9(k)	Yes	Only as specified in $\Box 63.9(1)$
63.10(a)	Yes	
63.10(b)(1)	Yes	
63.10(b)(2)(i)	No	
63.10(b)(2)(ii)	No	See 63.696(h) for recordkeeping of (1) date, time and dura-
		tion (2) listing of affected source or equipment, and an es- timate of the volume of each regulated pollutant emitted over the standard (3) actions to minimize emissions and correct the failure.
63.10(b)(2)(iii)	Yes	
63.10(b)(2)(iv)	No	
63.10(b)(2)(v)	No	
63.10(b)(2)(v)–(ix)	Yes	
(2)(2)(2)(3) - (3)(3)(2)(3)(2)(3)(3)(3)(3)(3)(3)(3)(3)(3)(3)(3)(3)(3)	No	
63.10(b)(3)	Yes	
63.10(c)(1)–(6)	No	
63.10(c)(7)–(8)	Yes	
63.10(c)(9)–(15)	No	

orley on LAPBH6H6L3 with DISTILLER

Pt. 63, Subpt. DD, Table 4

Subpart A reference	Applies to Subpart DD	Explanation
63.10(d)(1) 63.10(d)(2) 63.10(d)(3) 63.10(d)(4) 63.10(e)(1) 63.10(e)(1) 63.10(e)(2) 63.10(e)(4) 63.10(e)(4) 63.10(e)(4) 63.10(e)(5) 63.10(e)(1) 63.10(e)(3) 63.10(e)(4) 63.10(f) 63.11 63.15 63.16	No Yes No Yes No Yes No Yes No	See □63.697(b)(3) for reporting of malfunctions.

a □ herever subpart A specifies ⊡postmark" dates, submittals may be sent by methods other than the □.S. □ ail (e.g., by fax or courier). Submittals shall be sent by the specified dates, but a postmark is not required.

 $[64~{\rm FR}$ 38983, July 20, 1999, as amended at 66 FR 1267, Jan. 8, 2001; 80 FR 14280, Mar. 18, 2015; 85 FR 73893, Nov. 19, 2020]

TABLE 3 TO SUBPART DD OF PART 63—TANK CONTROL LEVELS FOR TANKS AT EXISTING AFFECTED SOURCES AS REQUIRED BY 40 CFR 63.685(b)(1)

Tank design capacity (cubic meters)	□ aximum HAP vapor pressure of off-site material managed in tank (kilopascals)	Tank control level
Design capacity less than 75 m ³	□ aximum HAP vapor pressure less than 76.6 kPa.	Level 1.
Design capacity less than 75 m ³	aximum HAP vapor pressure equal to or greater than 76.6 kPa.	Level 2, except that fixed roof tanks equipped with an internal floating roof and tanks equipped with an externa floating roof as provided for in □63.685(d)(1) and (2) shall not be used.
Design capacity equal to or greater than 75 m^3 and less than 151 m^3 .	aximum HAP vapor pressure less than 27.6 kPa.	Level 1.
	aximum HAP vapor pressure equal to or greater than 27.6 kPa.	Level 2.
Design capacity equal to or greater than 151 $\ensuremath{\text{m}}^3.$	aximum HAP vapor pressure less than 5.2 kPa.	Level 1.
	aximum HAP vapor pressure equal to or greater than 5.2 kPa.	Level 2.

[80 FR 14282, Mar. 18, 2015]

TABLE 4 TO SUBPART DD OF PART 63—TANK CONTROL LEVELS FOR TANKS AT EXISTING AFFECTED SOURCES AS REQUIRED BY 40 CFR 63.685(b)(1)(ii)

Tank design capacity (cubic meters)	□ aximum HAP vapor pressure of off-site material managed in tank (kilopascals)	Tank control level
Design capacity less than 75 m ³	□ aximum HAP vapor pressure less than 76.6 kPa.	Level 1.
Design capacity less than 75 m ³	□ aximum HAP vapor pressure equal to or greater than 76.6 kPa.	Level 2, except that fixed roof tanks equipped with an internal floating roof and tanks equipped with an external floating roof as provided for in \square 63.685(d)(1) and (2) shall not be used.
Design capacity equal to or greater than 75 m ³ and less than 151 m ³ .	aximum HAP vapor pressure less than 13.1 kPa.	Level 1.
	aximum HAP vapor pressure equal to or greater than 13.1 kPa.	Level 2.
Design capacity equal to or greater than 151 m ³ .	□ aximum HAP vapor pressure less than 5.2 kPa.	Level 1.
	□ aximum HAP vapor pressure equal to or greater than 5.2 kPa.	Level 2.

[80 FR 14283, Mar. 18, 2015]

Pt. 63, Subpt. DD, Table 5

40 CFR Ch. I (7-1-23 Edition)

 TABLE 5 TO SUBPART DD OF PART 63—TANK CONTROL LEVELS FOR TANKS AT NEW

 AFFECTED SOURCES AS REQUIRED BY 40 CFR 63.685(b)(2)

Tank design capacity (cubic meters)	aximum HAP vapor pressure of off-site material managed in tank (kilopascals)	Tank control level
Design capacity less than 38 m ³	□ aximum HAP vapor pressure less than 76.6 kPa.	Level 1.
Design capacity less than 38 m ³	aximum HAP vapor pressure equal to or greater than 76.6 kPa.	Level 2, except that fixed roof tanks equipped with an internal floating roof and tanks equipped with an external floating roof as provided for in $\Box 63.685(d)(1)$ and (2) shall not be used.
Design capacity equal to or greater than 38 m^3 and less than 151 m^3 .	aximum HAP vapor pressure less than 13.1 kPa.	Level 1.
	aximum HAP vapor pressure equal to or greater than 13.1 kPa.	Level 2.
Design capacity equal to or greater than 151 m^3 .	□ aximum HAP vapor pressure less than 0.7 kPa.	Level 1.
	□ aximum HAP vapor pressure equal to or greater than 0.7 kPa.	Level 2.

[80 FR 14283, Mar. 18, 2015]

Subpart EE—National Emission Standards for Magnetic Tape Manufacturing Operations

SOURCE: 59 FR 64596, Dec. 15, 1994, unless otherwise noted.

§63.701 Applicability.

(a) Except as specified in paragraph(b) of this section, the provisions of this subpart apply to:

(1) Each new and existing magnetic tape manufacturing operation located at a major source of hazardous air pollutant (HAP) emissions; and

(2) A magnetic tape manufacturing operation for which the owner or operator chooses to use the provisions of §63.703(b) and (h) to obtain a Federally enforceable limit on its potential to emit HAP.

EXPLANATORY NOTE: A reason the owner or operator would make the choice described in paragraph (a)(2) of this section is if the plant site, without this limit, would be a major source. The owner or operator could use this limit, which would establish the potential to emit from magnetic tape manufacturing operations, in conjunction with the potential to emit from the other HAP emission points at the stationary source, to be an area source. Note, however, that an owner or operator is not required to use the provisions in 63.703(b) and (h) to determine the potential to emit HAP from magnetic tape manufacturing operations. (b) This subpart does not apply to the following:

(1) Research or laboratory facilities; and

(2) Any coating operation that produces a quantity of magnetic tape that is 1 percent or less of total production (in terms of total square footage coated) from that coating operation in any 12-month period.

(c) The affected source subject to this standard is the magnetic tape manufacturing operation, as defined in §63.702.

(d) An owner or operator of an existing affected source subject to the provisions of this subpart shall comply according to the following schedule:

(1) Within 3 years after the effective date of the standard, if the owner or operator is required to install a new add-on air pollution control device to meet the requirements of §63.703(c) or (g); or

(2) Within 2 years after the effective date of the standard, if a new add-on air pollution control device is not needed to comply with §63.703(c) or (g) of these standards.

(e) The compliance date for an owner or operator of a new affected source subject to the provisions of this subpart is immediately upon startup of the affected source.

(f) The provisions of this subpart apply during periods of startup and shutdown, and whenever magnetic tape manufacturing operations are taking place. Appendix L

AUTHENTICATED U.S. GOVERNMENT INFORMATION

- TABLE 6 TO SUBPART PPP OF PART 63—PROC-ESS VENTS FROM CONTINUOUS UNIT OPER-ATIONS—MONITORING, RECORDKEEPING, AND REPORTING REQUIREMENTS
- TABLE 7 TO SUBPART PPP OF PART 63—OPER-ATING PARAMETERS FOR WHICH MONI-TORING LEVELS ARE REQUIRED TO BE ES-TABLISHED FOR PROCESS VENT STREAMS
- TABLE 8 TO SUBPART PPP of PART 63—Routine Reports Required by This Subpart

AUTHORITY: 42 U.S.C. 7401 et seq.

SOURCE: 57 FR 61992, Dec. 29, 1992, unless otherwise noted.

Subpart EEE—National Emission Standards for Hazardous Air Pollutants from Hazardous Waste Combustors

SOURCE: 64 FR 53038, Sept. 30, 1999, unless otherwise noted.

General

§63.1200 Who is subject to these regulations?

The provisions of this subpart apply to all hazardous waste combustors:

hazardous waste incinerators, hazardous waste cement kilns, hazardous waste lightweight aggregate kilns, hazardous waste solid fuel boilers, hazardous waste liquid fuel boilers, and hazardous waste hydrochloric acid production furnaces. Hazardous waste combustors are also subject to applicable requirements under parts 260 through 270 of this chapter.

(a) What if I am an area source? (1) Both area sources and major sources are subject to this subpart.

(2) Both area sources and major sources subject to this subpart, but not previously subject to title V, are immediately subject to the requirement to apply for and obtain a title V permit in all States, and in areas covered by part 71 of this chapter.

(b) These regulations in this subpart do not apply to sources that meet the criteria in Table 1 of this Section, as follows:

TABLE 1 TO 63.1200 HACARDOCS ASTE	COBSTORS EXESPT FROS SEPART EEE
-----------------------------------	---------------------------------

lf	And if	Then
(1) You are a previously af- fected source.	 (i) You ceased feeding hazardous waste for a period of time greater than the hazardous waste residence time (i.e., hazardous waste no longer resides in the combustion chamber)□ (ii) You have initiated the closure requirements of subpart G, parts 264 or 265 of this chapter□ (iii) You begin complying with the requirements of all other applicable standards of this part (Part 63)□and. (iv) You notify the Administrator in writing that you are no longer an affected source under this subpart (Subpart EEE). 	You are no longer sublect to this subpart (Subpart EEE).
(2) You are a research, devel- opment, and demonstration source.	You operate for no longer than one year after first burning hazardous waste (Note that the Administrator can extend this one-year restriction on a case-by-case basis upon your written request documenting when you first burned haz- ardous waste and the üstification for needing additional time to perform research, development, or demonstration operations)	You are not sub ect to this subpart (Subpart EEE). This exemption applies even if there is a hazardous waste combustor at the plant site that is regulated under this subpart. You still, however, remain sub ect to □270.65 of this chapter.
(3) The only hazardous wastes you burn are exempt from regulation under □266.100(c) of this chapter.		You are not subject to the re- quirements of this subpart (Subpart EEE).
 (4) You meet the definition of a small quantity burner under □266.108 of this chapter. 		You are not sublect to the re- quirements of this subpart (Subpart EEE).

§63.1200

§63.1201

(c) Table 1 of this section specifies the provisions of subpart A (General Provisions, §§ 63.1-63.15) that apply and those that do not apply to sources affected by this subpart.

[64 FR 53038, Sept. 30, 1999, as amended at 65 FR 42297, July 10, 2000; 67 FR 6986, Feb. 14, 2002; 70 FR 59540, Oct. 12, 2005]

§63.1201 Definitions and acronyms used in this subpart.

(a) The terms used in this subpart are defined in the Act, in subpart A of this part, or in this section as follows:

Air pollution control system means the equipment used to reduce the release of particulate matter and other pollutants to the atmosphere.

Automatic waste feed cutoff (AWFCO) system means a system comprised of cutoff valves, actuator, sensor, data manager, and other necessary components and electrical circuitry designed, operated and maintained to stop the flow of hazardous waste to the combustion unit automatically and immediately (except as provided by §63.1206(c)(3)(viii)) when any operating requirement is exceeded.

Btu means British Thermal Units.

By-pass duct means a device which diverts a minimum of 10 percent of a cement kiln's off gas, or a device which the Administrator determines on a case-by-case basis diverts a sample of kiln gas that contains levels of carbon monoxide or hydrocarbons representative of the levels in the kiln.

Combustion chamber means the area in which controlled flame combustion of hazardous waste occurs.

Continuous monitor means a device which continuously samples the regulated parameter specified in §63.1209 without interruption, evaluates the detector response at least once every 15 seconds, and computes and records the average value at least every 60 seconds, except during allowable periods of calibration and except as defined otherwise by the CEMS Performance Specifications in appendix B, part 60 of this chapter.

Dioxin/furan and dioxins and furans mean tetra-, penta-, hexa-, hepta-, and octa-chlorinated dibenzo dioxins and furans.

Existing source means any affected source that is not a new source.

40 CFR Ch. I (7–1–23 Edition)

Feedrate operating limits means limits on the feedrate of materials (e.g., metals, chlorine) to the combustor that are established based on comprehensive performance testing. The limits are established and monitored by knowing the concentration of the limited material (e.g., chlorine) in each feedstream and the flowrate of each feedstream.

Feedstream means any material fed into a hazardous waste combustor, including, but not limited to, any pumpable or nonpumpable solid, liquid, or gas.

Flowrate means the rate at which a feedstream is fed into a hazardous waste combustor.

Hazardous waste is defined in §261.3 of this chapter.

Hazardous waste burning cement kiln means a rotary kiln and any associated preheater or precalciner devices that produce clinker by heating limestone and other materials for subsequent production of cement for use in commerce, and that burns hazardous waste at any time.

Hazardous waste combustor means a hazardous waste incinerator, hazardous waste burning cement kiln, hazardous waste burning lightweight aggregate kiln, hazardous waste liquid fuel boiler, or hazardous waste solid fuel boiler, or hazardous waste hydrochloric acid production furnace.

Hazardous waste hydrochloric acid production furnace and Hazardous Waste HCl production furnace mean a halogen acid furnace defined under §260.10 of this chapter that produces aqueous hydrochloric acid (HCl) product and that burns hazardous waste at any time.

Hazardous waste incinerator means a device defined as an incinerator in §260.10 of this chapter and that burns hazardous waste at any time. For purposes of this subpart, the hazardous waste incinerator includes all associated firing systems and air pollution control devices, as well as the combustion chamber equipment.

Hazardous waste lightweight aggregate kiln means a rotary kiln that produces clinker by heating materials such as slate, shale and clay for subsequent production of lightweight aggregate used in commerce, and that burns hazardous waste at any time.

Hazardous waste liquid fuel boiler means a boiler defined under §260.10 of this chapter that does not burn solid fuels and that burns hazardous waste at any time. Liquid fuel boiler includes boilers that only burn gaseous fuel.

Hazardous waste residence time means the time elapsed from cutoff of the flow of hazardous waste into the combustor (including, for example, the time required for liquids to flow from the cutoff valve into the combustor) until solid, liquid, and gaseous materials from the hazardous waste (excluding residues that may adhere to combustion chamber surfaces and excluding waste-derived recycled materials such as cement kiln dust and internally recycled metals) exit the combustion chamber. For combustors with multiple firing systems whereby the residence time may vary for the firing systems, the hazardous waste residence time for purposes of complying with this subpart means the longest residence time for any firing system in use at the time of the waste cutoff.

Hazardous waste solid fuel boiler means a boiler defined under §260.10 of this chapter that burns a solid fuel and that burns hazardous waste at any time.

Initial comprehensive performance test means the comprehensive performance test that is used as the basis for initially demonstrating compliance with the standards.

In-line kiln raw mill means a hazardous waste burning cement kiln design whereby kiln gas is ducted through the raw material mill for portions of time to facilitate drying and heating of the raw material.

Instantaneous monitoring for combustion system leak control means detecting and recording pressure, without use of an averaging period, at a frequency adequate to detect combustion system leak events from hazardous waste combustion.

Monovent means an exhaust configuration of a building or emission control device (e.g. positive pressure fabric filter) that extends the length of the structure and has a width very small in relation to its length (i.e., length to width ratio is typically greater than 5:1). The exhaust may be an open vent with or without a roof, louvered vents, or a combination of such features.

MTEC means maximum theoretical emissions concentration of metals or HCl/Cl, expressed as $\mu g/dscm$, and is calculated by dividing the feedrate by the gas flowrate.

New source means any affected source the construction or reconstruction of which is commenced after the dates specified under \S 63.1206(a)(1)(i)(B), (a)(1)(ii)(B), and (a)(2)(ii).

One-minute average means the average of detector responses calculated at least every 60 seconds from responses obtained at least every 15 seconds.

Operating record means a documentation retained at the facility for ready inspection by authorized officials of all information required by the standards to document and maintain compliance with the applicable regulations, including data and information, reports, notifications, and communications with regulatory officials.

Operating requirements means operating terms or conditions, limits, or operating parameter limits developed under this subpart that ensure compliance with the emission standards.

Preheater tower combustion gas monitoring location means a location within the preheater tower of a dry process cement kiln downstream (in terms of gas flow) of all hazardous waste firing locations and where a representative sample of combustion gas to measure combustion efficiency can be monitored.

Raw material feed means the prepared and mixed materials, which include but are not limited to materials such as limestone, clay, shale, sand, iron ore, mill scale, cement kiln dust and flyash, that are fed to a cement or lightweight aggregate kiln. Raw material feed does not include the fuels used in the kiln to produce heat to form the clinker product.

Research, development, and demonstration source means a source engaged in laboratory, pilot plant, or prototype demonstration operations:

(1) Whose primary purpose is to conduct research, development, or shortterm demonstration of an innovative and experimental hazardous waste treatment technology or process; and

§63.1202

(2) Where the operations are under the close supervision of technicallytrained personnel.

Rolling average means the average of all one-minute averages over the averaging period.

Run means the net period of time during which an air emission sample is collected under a given set of operating conditions. Three or more runs constitutes a test. Unless otherwise specified, a run may be either intermittent or continuous.

Run average means the average of the one-minute average parameter values for a run.

System removal efficiency means [1 – Emission Rate (mass/time) / Feedrate (mass/time)] X 100.

TEQ means the international method of expressing toxicity equivalents for dioxins and furans as defined in U.S. EPA, Interim Procedures for Estimating Risks Associated with Exposures to Mixtures of Chlorinated Dibenzo-p-dioxins and -dibenzofurans (CDDs and CDFs) and 1989 Update, March 1989.

You means the owner or operator of a hazardous waste combustor.

(b) The acronyms used in this subpart refer to the following:

AWFCO means automatic waste feed cutoff.

CAS means chemical abstract services registry.

CEMS means continuous emissions monitoring system.

CMS means continuous monitoring system.

DRE means destruction and removal efficiency.

MACT means maximum achievable control technology.

MTEC means maximum theoretical emissions concentration.

NIC means notification of intent to comply.

[64 FR 53038, Sept. 30, 1999, as amended at 65 FR 42297, July 10, 2000; 65 FR 67271, Nov. 9, 2000; 66 FR 35103, July 3, 2001; 67 FR 6986, Feb. 14, 2002; 67 FR 77691, Dec. 19, 2002; 70 FR 59540, Oct. 12, 2005]

40 CFR Ch. I (7–1–23 Edition)

§63.1202 [Reserved]

INTERIM EMISSIONS STANDARDS AND OP-ERATING LIMITS FOR INCINERATORS, CEMENT KILNS, AND LIGHTWEIGHT AG-GREGATE KILNS

§63.1203 What are the standards for hazardous waste incinerators that are effective until compliance with the standards under §63.1219?

(a) *Emission limits for existing sources.* You must not discharge or cause combustion gases to be emitted into the atmosphere that contain:

(1) For dioxins and furans:

(i) Emissions in excess of 0.20 ng TEQ/ dscm corrected to 7 percent oxygen; or

(ii) Emissions in excess of 0.40 ng TEQ/dscm corrected to 7 percent oxygen provided that the combustion gas temperature at the inlet to the initial particulate matter control device is 400 °F or lower based on the average of the test run average temperatures. (For purposes of compliance, operation of a wet particulate control device is presumed to meet the 400 °F or lower requirement);

(2) Mercury in excess of 130 μ g/dscm corrected to 7 percent oxygen;

(3) Lead and cadmium in excess of 240 µg/dscm, combined emissions, corrected to 7 percent oxygen;

(4) Arsenic, beryllium, and chromium in excess of 97 μ g/dscm, combined emissions, corrected to 7 percent oxygen;

(5) For carbon monoxide and hydrocarbons, either:

(i) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (a)(5)(ii) of this section, you must also document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons do not exceed 10 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or

(ii) Hydrocarbons in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane;

(6) Hydrochloric acid and chlorine gas in excess of 77 parts per million by volume, combined emissions, expressed as hydrochloric acid equivalents, dry basis and corrected to 7 percent oxygen; and

(7) Particulate matter in excess of 34 mg/dscm corrected to 7 percent oxygen.

(b) *Emission limits for new sources.* You must not discharge or cause combustion gases to be emitted into the atmosphere that contain:

(1) Dioxins and furans in excess of 0.20 ng TEQ/dscm, corrected to 7 percent oxygen;

(2) Mercury in excess of 45 μ g/dscm corrected to 7 percent oxygen;

(3) Lead and cadmium in excess of 120 µg/dscm, combined emissions, corrected to 7 percent oxygen;

(4) Arsenic, beryllium, and chromium in excess of 97 μg/dscm, combined emissions, corrected to 7 percent oxygen;

(5) For carbon monoxide and hydrocarbons, either:

(i) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (b)(5)(ii) of this section, you must also document that, during the destruction and removal efficiency (DRE) test runs or equivalent as provided by their §63.1206(b)(7), hydrocarbons do not exceed 10 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane: or

(ii) Hydrocarbons in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane;

(6) Hydrochloric acid and chlorine gas in excess of 21 parts per million by volume, combined emissions, expressed as hydrochloric acid equivalents, dry basis and corrected to 7 percent oxygen; and

(7) Particulate matter in excess of 34 mg/dscm corrected to 7 percent oxygen.

(c) Destruction and removal efficiency (DRE) standard—(1) 99.99% DRE. Except as provided in paragraph (c)(2) of this section, you must achieve a destruction and removal efficiency (DRE) of 99.99% for each principle organic hazardous constituent (POHC) designated under paragraph (c)(3) of this section. You must calculate DRE for each POHC from the following equation:

DRE = $[1 - (W_{out} / W_{in})] \times 100\%$

Where:

- $W_{\rm in}$ = mass feedrate of one principal organic hazardous constituent (POHC) in a waste feedstream; and
- W_{out} = mass emission rate of the same POHC present in exhaust emissions prior to release to the atmosphere.

(2) 99.9999% DRE. If you burn the dioxin-listed hazardous wastes F020, F021, F022, F023, F026, or F027 (see §261.31 of this chapter), you must achieve a destruction and removal efficiency (DRE) of 99.9999% for each principle organic hazardous constituent (POHC) that you designate under paragraph (c)(3) of this section. You must demonstrate this DRE performance on POHCs that are more difficult to incinerate than tetra-, penta-, and hexachlorodibenzo-p-dioxins and dibenzofurans. You must use the equation in paragraph (c)(1) of this section to calculate DRE for each POHC. In addition, you must notify the Administrator of your intent to incinerate hazardous wastes F020, F021, F022, F023, F026. or F027.

(3) Principal organic hazardous constituents (POHCs). (i) You must treat the Principal Organic Hazardous Constituents (POHCs) in the waste feed that you specify under paragraph (c)(3)(ii) of this section to the extent required by paragraphs (c)(1) and (c)(2) of this section.

(ii) You must specify one or more POHCs that are representative of the

most difficult to destroy organic compounds in your hazardous waste feedstream. You must base this specification on the degree of difficulty of incineration of the organic constituents in the hazardous waste and on their concentration or mass in the hazardous waste feed, considering the results of hazardous waste analyses or other data and information.

(d) Significant figures. The emission limits provided by paragraphs (a) and (b) of this section are presented with two significant figures. Although you must perform intermediate calculations using at least three significant figures, you may round the resultant emission levels to two significant figures to document compliance.

(e) The provisions of this section no longer apply after any of the following dates, whichever occurs first:

(1) The date that your source begins to comply with §63.1219 by placing a Documentation of Compliance in the operating record pursuant to §63.1211(c):

(2) The date that your source begins to comply with §63.1219 by submitting a Notification of Compliance pursuant to §63.1210(b); or

(3) The date for your source to comply with §63.1219 pursuant to §63.1206 and any extensions granted there under.

[67 FR 6809, Feb. 13, 2002, as amended at 70 FR 59541, Oct. 12, 2005; 73 FR 18979, Apr. 8, 2008]

§63.1204 What are the standards for hazardous waste burning cement kilns that are effective until compliance with the standards under §63.1220?

(a) *Emission limits for existing sources*. You must not discharge or cause combustion gases to be emitted into the atmosphere that contain:

(1) For dioxins and furans:

(i) Emissions in excess of 0.20 ng TEQ/ dscm corrected to 7 percent oxygen; or

(ii) Emissions in excess of 0.40 ng TEQ/dscm corrected to 7 percent oxygen provided that the combustion gas temperature at the inlet to the initial dry particulate matter control device is 400 °F or lower based on the average of the test run average temperatures; 40 CFR Ch. I (7–1–23 Edition)

(2) Mercury in excess of 120 µg/dscm corrected to 7 percent oxygen;

(3) Lead and cadmium in excess of 330 µg/dscm, combined emissions, corrected to 7 percent oxygen;

(4) Arsenic, beryllium, and chromium in excess of 56 μ g/dscm, combined emissions, corrected to 7 percent oxygen;

(5) Carbon monoxide and hydrocarbons.(i) For kilns equipped with a by-pass duct or midkiln gas sampling system, either:

(A) Carbon monoxide in the by-pass duct or mid-kiln gas sampling system in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continyous emissions monitoring system). dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (a)(5)(i)(B) of this section, you must also document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons in the by-pass duct or mid-kiln gas sampling system do not exceed 10 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or

(B) Hydrocarbons in the by-pass duct or midkiln gas sampling system in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane;

(ii) For kilns not equipped with a bypass duct or midkiln gas sampling system, either:

(A) Hydrocarbons in the main stack in excess of 20 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or

(B) Carbon monoxide in the main stack in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a

continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (a)(5)(ii)(A) of this section, you also must document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons in the main stack do not exceed 20 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane.

(6) Hydrochloric acid and chlorine gas in excess of 130 parts per million by volume, combined emissions, expressed as hydrochloric acid equivalents, dry basis, corrected to 7 percent oxygen; and

(7) Particulate matter in excess of 0.15 kg/Mg dry feed and opacity greater than 20 percent.

(i) You must use suitable methods to determine the kiln raw material feedrate.

(ii) Except as provided in paragraph (a)(7)(iii) of this section, you must compute the particulate matter emission rate, E, from the following equation:

 $E = (C_s \times Q_{sd})/P$

Where:

E = emission rate of particulate matter, kg/ Mg of kiln raw material feed;

 $C_s = concentration of particulate matter, kg/dscm;$

 Q_{sd} = volumetric flowrate of effluent gas, dscm/hr: and

P = total kiln raw material feed (dry basis), Mg/hr.

(iii) If you operate a preheater or preheater/precalciner kiln with dual stacks, you must test simultaneously and compute the combined particulate matter emission rate, E_c , from the following equation:

$$E_{c} = (C_{sk} \times Q_{sdk} + C_{sb} \times Q_{sdb})/P$$

Where:

- $$\begin{split} E_{\rm c} = the \ combined \ emission \ rate \ of \ particulate \ matter \ from \ the \ kiln \ and \ bypass \ stack, \ kg/Mg \ of \ kiln \ raw \ material \ feed; \end{split}$$
- $$\label{eq:csk} \begin{split} C_{\rm sk} &= \mbox{concentration of particulate matter in} \\ & \mbox{the kiln effluent, kg/dscm;} \end{split}$$

- Q_{sdk} = volumetric flowrate of kiln effluent gas, dscm/hr;
- $C_{sb} = \text{concentration of particulate matter in} \\ \text{the bypass stack effluent, kg/dscm;}$
- Q_{sdb} = volumetric flowrate of bypass stack effluent gas, dscm/hr; and
- ${\rm P}$ = total kiln raw material feed (dry basis), Mg/hr.

(b) *Emission limits for new sources*. You must not discharge or cause combustion gases to be emitted into the atmosphere that contain:

(1) For dioxins and furans:

(i) Emissions in excess of $0.20~{\rm ng}$ TEQ/ dscm corrected to 7 percent oxygen; or

(ii) Emissions in excess of 0.40 ng TEQ/dscm corrected to 7 percent oxygen provided that the combustion gas temperature at the inlet to the initial dry particulate matter control device is 400 °F or lower based on the average of the test run average temperatures;

(2) Mercury in excess of 120 μ g/dscm corrected to 7 percent oxygen;

(3) Lead and cadmium in excess of 180 µg/dscm, combined emissions, corrected to 7 percent oxygen;

(4) Arsenic, beryllium, and chromium in excess of 54 μ g/dscm, combined emissions, corrected to 7 percent oxygen;

(5) Carbon monoxide and hydrocarbons.
(i) For kilns equipped with a by-pass duct or midkiln gas sampling system, carbon monoxide and hydrocarbons emissions are limited in both the by-pass duct or midkiln gas sampling system and the main stack as follows:

(A) Emissions in the by-pass or midkiln gas sampling system are limited to either:

(1) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (b)(5)(i)(A)(2)of this section, you also must document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons do not exceed 10 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to

§63.1204

40 CFR Ch. I (7–1–23 Edition)

7 percent oxygen, and reported as propane; or

(2) Hydrocarbons in the by-pass duct or midkiln gas sampling system in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; and

(B) Hydrocarbons in the main stack are limited, if construction of the kiln commenced after April 19, 1996 at a plant site where a cement kiln (whether burning hazardous waste or not) did not previously exist, to 50 parts per million by volume, over a 30-day block average (monitored continuously with a continuous monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane.

(ii) For kilns not equipped with a bypass duct or midkiln gas sampling system, hydrocarbons and carbon monoxide are limited in the main stack to either:

(A) Hydrocarbons not exceeding 20 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or

(B)(1) Carbon monoxide not exceeding 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen; and

(2) Hydrocarbons not exceeding 20 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane at any time during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7); and

(3) If construction of the kiln commenced after April 19, 1996 at a plant site where a cement kiln (whether burning hazardous waste or not) did not previously exist, hydrocarbons are limited to 50 parts per million by volume, over a 30-day block average (monitored continuously with a continuous monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane.

(6) Hydrochloric acid and chlorine gas in excess of 86 parts per million, combined emissions, expressed as hydrochloric acid equivalents, dry basis and corrected to 7 percent oxygen; and

(7) Particulate matter in excess of 0.15 kg/Mg dry feed and opacity greater than 20 percent.

(i) You must use suitable methods to determine the kiln raw material feedrate.

(ii) Except as provided in paragraph (a)(7)(iii) of this section, you must compute the particulate matter emission rate, E, from the equation specified in paragraph (a)(7)(ii) of this section.

(iii) If you operate a preheater or preheater/precalciner kiln with dual stacks, you must test simultaneously and compute the combined particulate matter emission rate, E_c , from the equation specified in paragraph (a)(7)(iii) of this section.

(c) Destruction and removal efficiency (DRE) standard—(1) 99.99% DRE. Except as provided in paragraph (c)(2) of this section, you must achieve a destruction and removal efficiency (DRE) of 99.99% for each principle organic hazardous constituent (POHC) designated under paragraph (c)(3) of this section. You must calculate DRE for each POHC from the following equation:

 $DRE=[1-(W_{out}/W_{in})]\times 100\%$

Where:

- W_{in} = mass feedrate of one principal organic hazardous constituent (POHC) in a waste feedstream; and
- W_{out} = mass emission rate of the same POHC present in exhaust emissions prior to release to the atmosphere.

(2) 99.9999% DRE. If you burn the dioxin-listed hazardous wastes F020, F021, F022, F023, F026, or F027 (see §261.31 of this chapter), you must achieve a destruction and removal efficiency (DRE) of 99.9999% for each principle organic hazardous constituent (POHC) that you designate under paragraph (c)(3) of this section. You must demonstrate this DRE performance on POHCs that are more difficult to incinerate than tetra-, penta-, and hexachlorodibenzo-p-dioxins and

dibenzofurans. You must use the equation in paragraph (c)(1) of this section to calculate DRE for each POHC. In addition, you must notify the Administrator of your intent to incinerate hazardous wastes F020, F021, F022, F023, F026, or F027.

(3) Principal organic hazardous constituents (POHCs). (i) You must treat the Principal Organic Hazardous Constituents (POHCs) in the waste feed that you specify under paragraph (c)(3)(ii) of this section to the extent required by paragraphs (c)(1) and (c)(2) of this section.

(ii) You must specify one or more POHCs that are representative of the most difficult to destroy organic compounds in your hazardous waste feedstream. You must base this specification on the degree of difficulty of incineration of the organic constituents in the hazardous waste and on their concentration or mass in the hazardous waste feed, considering the results of hazardous waste analyses or other data and information.

(d) Cement kilns with in-line kiln raw mills—(1) General. (i) You must conduct performance testing when the raw mill is on-line and when the mill is off-line to demonstrate compliance with the emission standards, and you must establish separate operating parameter limits under 63.1209 for each mode of operation, except as provided by paragraph (d)(1)(iv) of this section.

(ii) You must document in the operating record each time you change from one mode of operation to the alternate mode and begin complying with the operating parameter limits for that alternate mode of operation.

(iii) You must calculate rolling averages for operating parameter limits as provided by 63.1209(q)(2).

(iv) If your in-line kiln raw mill has dual stacks, you may assume that the dioxin/furan emission levels in the bypass stack and the operating parameter limits determined during performance testing of the by-pass stack when the raw mill is off-line are the same as when the mill is on-line.

(2) *Emissions averaging*. You may comply with the mercury, semivolatile metal, low volatile metal, and hydrochloric acid/chlorine gas emission standards on a time-weighted average basis under the following procedures:

(i) Averaging methodology. You must calculate the time-weighted average emission concentration with the following equation:

 $\begin{array}{l} C_{total}{=}\{C_{mill-off}\times(T_{mill-off}/(T_{mill-off}+T_{mill-on}))\}\\ + & \{C_{mill-on}\times(T_{mill-on}/(T_{mill-off}+T_{mill-on}))\}\end{array}$

Where:

- $\begin{array}{l} C_{total} = time-weighted \ average \ concentration \\ of \ a \ regulated \ constituent \ considering \\ both \ raw \ mill \ on \ time \ and \ off \ time; \end{array}$
- $C_{mill-off}$ = average performance test concentration of regulated constituent with the raw mill off-line;
- $C_{mill-on}$ = average performance test concentration of regulated constituent with the raw mill on-line;

T_{mill-off} = time when kiln gases are not routed through the raw mill; and

T_{mill-on} = time when kiln gases are routed through the raw mill.

(ii) Compliance. (A) If you use this emission averaging provision, you must document in the operating record compliance with the emission standards on an annual basis by using the equation provided by paragraph (d)(2) of this section.

(B) Compliance is based on one-year block averages beginning on the day you submit the initial notification of compliance.

(iii) Notification. (A) If you elect to document compliance with one or more emission standards using this emission averaging provision, you must notify the Administrator in the initial comprehensive performance test plan submitted under §63.1207(e).

(B) You must include historical raw mill operation data in the performance test plan to estimate future raw mill down-time and document in the performance test plan that estimated emissions and estimated raw mill down-time will not result in an exceedance of an emission standard on an annual basis.

(C) You must document in the notification of compliance submitted under §63.1207(j) that an emission standard will not be exceeded based on the documented emissions from the performance test and predicted raw mill downtime.

(e) Preheater or preheater/precalciner kilns with dual stacks—(1) General. You

must conduct performance testing on each stack to demonstrate compliance with the emission standards, and you must establish operating parameter limits under §63.1209 for each stack, except as provided by paragraph (d)(1)(iv) of this section for dioxin/furan emissions testing and operating parameter limits for the by-pass stack of in-line raw mills.

(2) *Emissions averaging*. You may comply with the mercury, semivolatile metal, low volatile metal, and hydrochloric acid/chlorine gas emission standards specified in this section on a gas flowrate-weighted average basis under the following procedures:

(i) Averaging methodology. You must calculate the gas flowrate-weighted average emission concentration using the following equation:

Where:

C_{tot} = gas flowrate-weighted average concentration of the regulated constituent;

 C_{main} = average performance test concentration demonstrated in the main stack;

 C_{bypass} = average performance test concentration demonstrated in the bypass stack;

 $Q_{\rm main}$ = volumetric flowrate of main stack effluent gas; and

Q_{bypass} = volumetric flowrate of bypass effluent gas.

(ii) Compliance. (A) You must demonstrate compliance with the emission standard(s) using the emission concentrations determined from the performance tests and the equation provided by paragraph (e)(1) of this section; and

(B) You must develop operating parameter limits for bypass stack and main stack flowrates that ensure the emission concentrations calculated with the equation in paragraph (e)(1) of this section do not exceed the emission standards on a 12-hour rolling average basis. You must include these flowrate limits in the Notification of Compliance.

(iii) *Notification*. If you elect to document compliance under this emissions averaging provision, you must:

(A) Notify the Administrator in the initial comprehensive performance test plan submitted under §63.1207(e). The performance test plan must include, at a minimum, information describing the

40 CFR Ch. I (7–1–23 Edition)

flowrate limits established under paragraph (e)(2)(ii)(B) of this section; and

(B) Document in the Notification of Compliance submitted under $\S63.1207(j)$ the demonstrated gas flowrate-weighted average emissions that you calculate with the equation provided by paragraph (e)(2) of this section.

(f) Significant figures. The emission limits provided by paragraphs (a) and (b) of this section are presented with two significant figures. Although you must perform intermediate calculations using at least three significant figures, you may round the resultant emission levels to two significant figures to document compliance.

(g) [Reserved]

(h) When you comply with the particulate matter requirements of paragraphs (a)(7) or (b)(7) of this section, you are exempt from the New Source Performance Standard for particulate matter and opacity under 60.60 of this chapter.

(i) The provisions of this section no longer apply after any of the following dates, whichever occurs first:

(1) The date that your source begins to comply with §63.1220 by placing a Documentation of Compliance in the operating record pursuant to §63.1211(c);

(2) The date that your source begins to comply with 63.1220 by submitting a Notification of Compliance pursuant to 63.1210(b); or

(3) The date for your source to comply with §63.1220 pursuant to §63.1206 and any extensions granted there under.

[67 FR 6809, Feb. 13, 2002, as amended at 67
 FR 6987, Feb. 14, 2002; 70 FR 59541, Oct. 12, 2005; 73 FR 18979, Apr. 8, 2008]

§63.1205 What are the standards for hazardous waste burning lightweight aggregate kilns that are effective until compliance with the standards under §63.1221?

(a) *Emission limits for existing sources*. You must not discharge or cause combustion gases to be emitted into the atmosphere that contain:

(1) For dioxins and furans:

(i) Emissions in excess of 0.20 ng TEQ/ dscm corrected to 7 percent oxygen; or

(ii) Rapid quench of the combustion gas temperature at the exit of the

(last) combustion chamber (or exit of any waste heat recovery system) to 400 °F or lower based on the average of the test run average temperatures. You must also notify in writing the RCRA authority that you are complying with this option;

(2) Mercury in excess of 120 µg/dscm corrected to 7 percent oxygen;

(3) Lead and cadmium in excess of 250 µg/dscm, combined emissions, corrected to 7 percent oxygen;

(4) Arsenic, beryllium, and chromium in excess of 110 μ g/dscm, combined emissions, corrected to 7 percent oxygen;

(5) Carbon monoxide and hydrocarbons. (i) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (a)(5)(ii) of this section, you also must document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons do not exceed 20 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane: or

(ii) Hydrocarbons in excess of 20 parts per million by volume, over an hourly rolling average, dry basis, corrected to 7 percent oxygen, and reported as propane;

(6) Hydrochloric acid and chlorine gas in excess of 600 parts per million by volume, combined emissions, expressed as hydrochloric acid equivalents, dry basis and corrected to 7 percent oxygen; and

(7) Particulate matter in excess of 57 mg/dscm corrected to 7 percent oxygen.

(b) *Emission limits for new sources*. You must not discharge or cause combustion gases to be emitted into the atmosphere that contain:

(1) For dioxins and furans:

(i) Emissions in excess of 0.20 ng TEQ/ dscm corrected to 7 percent oxygen; or (ii) Rapid quench of the combustion gas temperature at the exit of the (last) combustion chamber (or exit of any waste heat recovery system) to 400 °F or lower based on the average of the test run average temperatures. You must also notify in writing the RCRA authority that you are complying with this option;

(2) Mercury in excess of 120 μ g/dscm corrected to 7 percent oxygen;

(3) Lead and cadmium in excess of 43 µg/dscm, combined emissions, corrected to 7 percent oxygen;

(4) Arsenic, beryllium, and chromium in excess of 110 μ g/dscm, combined emissions, corrected to 7 percent oxygen;

(5) Carbon monoxide and hydrocarbons. (i) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (b)(5)(ii) of this section, you also must document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons do not exceed 20 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane: or

(ii) Hydrocarbons in excess of 20 parts per million by volume, over an hourly rolling average, dry basis, corrected to 7 percent oxygen, and reported as propane;

(6) Hydrochloric acid and chlorine gas in excess of 600 parts per million by volume, combined emissions, expressed as hydrochloric acid equivalents, dry basis and corrected to 7 percent oxygen; and

(7) Particulate matter in excess of 57 mg/dscm corrected to 7 percent oxygen.

(c) Destruction and removal efficiency (DRE) standard—(1) 99.99% DRE. Except as provided in paragraph (c)(2) of this section, you must achieve a destruction and removal efficiency (DRE) of

99.99% for each principal organic hazardous constituent (POHC) designated under paragraph (c)(3) of this section. You must calculate DRE for each POHC from the following equation:

 $\mathrm{DRE} = [1-(\mathrm{W}_{\mathrm{out}} \ / \ \mathrm{W}_{\mathrm{in}})] \times 100\%$

Where:

- W_{in} = mass feedrate of one principal organic hazardous constituent (POHC) in a waste feedstream; and
- W_{out} = mass emission rate of the same POHC present in exhaust emissions prior to release to the atmosphere.

(2) 99.9999% DRE. If you burn the dioxin-listed hazardous wastes F020, F021, F022, F023, F026, or F027 (see §261.31 of this chapter), you must achieve a destruction and removal efficiency (DRE) of 99.9999% for each principal organic hazardous constituent (POHC) that you designate under paragraph (c)(3) of this section. You must demonstrate this DRE performance on POHCs that are more difficult to incinthan tetra-, erate penta-. and hexachlorodibenzo-dioxins and dibenzofurans. You must use the equation in paragraph (c)(1) of this section to calculate DRE for each POHC. In addition, you must notify the Administrator of your intent to burn hazardous wastes F020, F021, F022, F023, F026, or F027.

(3) Principal organic hazardous constituents (POHCs). (i) You must treat the Principal Organic Hazardous Constituents (POHCs) in the waste feed that you specify under paragraph (c)(3)(ii) of this section to the extent required by paragraphs (c)(1) and (c)(2) of this section.

(ii) You must specify one or more POHCs that are representative of the most difficult to destroy organic compounds in your hazardous waste feedstream. You must base this specification on the degree of difficulty of incineration of the organic constituents in the hazardous waste and on their concentration or mass in the hazardous waste feed, considering the results of hazardous waste analyses or other data and information.

(d) *Significant figures*. The emission limits provided by paragraphs (a) and (b) of this section are presented with two significant figures. Although you must perform intermediate calcula-

40 CFR Ch. I (7–1–23 Edition)

tions using at least three significant figures, you may round the resultant emission levels to two significant figures to document compliance.

(e) The provisions of this section no longer apply after any of the following dates, whichever occurs first:

(1) The date that your source begins to comply with §63.1221 by placing a Documentation of Compliance in the operating record pursuant to §63.1211(c):

(2) The date that your source begins to comply with §63.1221 by submitting a Notification of Compliance pursuant to §63.1210(b); or

(3) The date for your source to comply with §63.1221 pursuant to §63.1206 and any extensions granted there under.

[67 FR 6812, Feb. 13, 2002, as amended at 67
 FR 77691, Dec. 19, 2002; 70 FR 59541, Oct. 12, 2005; 73 FR 18979, Apr. 8, 2008]

MONITORING AND COMPLIANCE PROVISIONS

§63.1206 When and how must you comply with the standards and operating requirements?

(a) Compliance dates—(1) Compliance dates for incinerators, cement kilns, and lightweight aggregate kilns that burn hazardous waste—(i) Compliance date for standards under §§ 63.1203, 63.1204, and 63.1205—(A) Compliance dates for existing sources. You must comply with the emission standards under §§ 63.1203, 63.1204, and 63.1205 and the other requirements of this subpart no later than the compliance date, September 30, 2003, unless the Administrator grants you an extension of time under § 63.6(i) or § 63.1213, except:

(1) Cement kilns are exempt from the bag leak detection system requirements under paragraph (c)(8) of this section:

(2) The bag leak detection system required under 63.1206(c)(8) must be capable of continuously detecting and recording particulate matter emissions at concentrations of 1.0 milligram per actual cubic meter unless you demonstrate under 63.1209(g)(1) that a higher detection limit would adequately detect bag leaks, in lieu of the requirement for the higher detection

limit under paragraph (c)(8)(ii)(A) of this section; and

(3) The excessive exceedances notification requirements for bag leak detection systems under paragraph (c)(8)(iv)of this section are waived.

(B) New or reconstructed sources. (1) If you commenced construction or reconstruction of your hazardous waste combustor after April 19, 1996, you must comply with the emission standards under §§ 63.1203, 63.1204, and 63.1205 and the other requirements of this subpart by the later of September 30, 1999 or the date the source starts operations, except as provided by paragraphs (a)(1)(i)(A)(1)through (3)and (a)(1)(i)(B)(2) of this section. The costs of retrofitting and replacement of equipment that is installed specifically to comply with this subpart, between April 19, 1996 and a source's compliance date, are not considered to be reconstruction costs.

(2) For a standard under \S 63.1203, 63.1204, and 63.1205 that is more stringent than the standard proposed on April 19, 1996, you may achieve compliance no later than September 30, 2003 if you comply with the standard proposed on April 19, 1996 after September 30, 1999. This exception does not apply, however, to new or reconstructed area source hazardous waste combustors that become major sources after September 30, 1999. As provided by \S 63.6(b)(7), such sources must comply with the standards under \S 63.1203, 63.1204, and 63.1205 at startup.

(ii) Compliance date for standards under §§ 63.1219, 63.1220, and 63.1221—(A) Compliance dates for existing sources. You must comply with the emission standards under §§ 63.1219, 63.1220, and 63.1221 and the other requirements of this subpart no later than the compliance date, October 14, 2008, unless the Administrator grants you an extension of time under § 63.6(i) or § 63.1213.

(B) New or reconstructed sources. (1) If you commenced construction or reconstruction of your hazardous waste combustor after April 20, 2004, you must comply with the new source emission standards under \S 63.1219, 63.1220, and 63.1221 and the other requirements of this subpart by the later of October 12, 2005 or the date the source starts operations, except as provided by paragraphs (a)(1)(ii)(B)(2) and (a)(1)(ii)(B)(3)of this section. The costs of retrofitting and replacement of equipment that is installed specifically to comply with this subpart, between April 20, 2004, and a source's compliance date, are not considered to be reconstruction costs.

(2) For a standard under §§ 63.1219, 63.1220, and 63.1221 that is more stringent than the standard proposed on April 20, 2004, you may achieve compliance no later than October 14, 2008, if you comply with the standard proposed on April 20, 2004, after October 12, 2005. This exception does not apply, however, to new or reconstructed area source hazardous waste combustors that become major sources after October 14, 2008. As provided by § 63.6(b)(7), such sources must comply with the standards under §§ 63.1219, 63.1220, and 63.1221 at startup.

(3) If you commenced construction or reconstruction of a cement kiln after April 20, 2004, you must comply with the new source emission standard for particulate matter under $\S 63.1220(b)(7)(i)$ by the later of October 28, 2008 or the date the source starts operations.

(2) Compliance date for solid fuel boilers, liquid fuel boilers, and hydrochloric acid production furnaces that burn hazardous waste for standards under \$ 63.1216, 63.1217, and 63.1218. (i) Compliance date for existing sources. You must comply with the standards of this subpart no later than the compliance date, October 14, 2008, unless the Administrator grants you an extension of time under \$ 63.6(i) or \$ 63.1213.

(ii) New or reconstructed sources. (A) If you commenced construction or reconstruction of your hazardous waste combustor after April 20, 2004, you must comply with the new source emission standards of this subpart by the later of October 12, 2005, or the date the source starts operations, except as provided by paragraph (a)(2)(i)(B) of this section. The costs of retrofitting and replacement of equipment that is installed specifically to comply with this subpart, between April 20, 2004, and a source's compliance date, are not considered to be reconstruction costs. (B) For a standard in the subpart that is more stringent than the standard proposed on April 20, 2004, you may achieve compliance no later than October 14, 2008, if you comply with the standard proposed on April 20, 2004, after October 12, 2005. This exception does not apply, however, to new or reconstructed area source hazardous waste combustors that become major sources after October 14, 2008. As provided by §63.6(b)(7), such sources must comply with this subpart at startup.

(3) Early compliance. If you choose to comply with the emission standards of this subpart prior to the dates specified in paragraphs (a)(1) and (a)(2) of this section, your compliance date is the earlier of the date you postmark the Notification of Compliance under $\S63.1207(j)(1)$ or the dates specified in paragraphs (a)(1) and (a)(2) of this section.

(b) Compliance with standards—(1) Applicability. The emission standards and operating requirements set forth in this subpart apply at all times except:

(i) During periods of startup, shutdown, and malfunction; and

(ii) When hazardous waste is not in the combustion chamber (i.e., the hazardous waste feed to the combustor has been cut off for a period of time not less than the hazardous waste residence time) and you have documented in the operating record that you are complying with all otherwise applicable requirements and standards promulgated under authority of sections 112 (e.g., 40 CFR part 63, subparts LLL, DDDDD, and NNNNN) or 129 of the Clean Air Act in lieu of the emission under §§63.1203, standards 63.1204. 63.1205, 63.1215, 63.1216, 63.1217, 63.1218. 63.1219, 63.1220, and 63.1221; the monitoring and compliance standards of this section and §§63.1207 through 63.1209, except the modes of operation requirements of 63.1209(q); and the notification, reporting, and recordkeeping requirements §§ 63.1210 of through 63.1212.

(2) Methods for determining compliance. The Administrator will determine compliance with the emission standards of this subpart as provided by \S 63.6(f)(2). Conducting performance testing under operating conditions representative of the extreme range of normal conditions

40 CFR Ch. I (7–1–23 Edition)

is consistent with the requirements of \$ 63.6(f)(2)(iii)(B) and 63.7(e)(1) to conduct performance testing under representative operating conditions.

(3) Finding of compliance. The Administrator will make a finding concerning compliance with the emission standards and other requirements of this subpart as provided by $\S63.6(f)(3)$.

(4) Extension of compliance with emission standards. The Administrator may grant an extension of compliance with the emission standards of this subpart as provided by \S 63.6(i) and 63.1213.

(5) Changes in design, operation, or maintenance—(i) Changes that may adversely affect compliance. If you plan to change (as defined in paragraph (b)(5)(iii) of this section) the design, operation, or maintenance practices of the source in a manner that may adversely affect compliance with any emission standard that is not monitored with a CEMS:

(A) Notification. You must notify the Administrator at least 60 days prior to the change, unless you document circumstances that dictate that such prior notice is not reasonably feasible. The notification must include:

(1) A description of the changes and which emission standards may be affected; and

(2) A comprehensive performance test schedule and test plan under the requirements of §63.1207(f) that will document compliance with the affected emission standard(s);

(B) *Performance test.* You must conduct a comprehensive performance test under the requirements of §§ 63.1207(f)(1) and (g)(1) to document compliance with the affected emission standard(s) and establish operating parameter limits as required under §63.1209, and submit to the Administrator a Notification of Compliance under §§ 63.1207(j) and 63.1210(d); and

(C) Restriction on waste burning. (I) Except as provided by paragraph (b)(5)(i)(C)(2) of this section, after the change and prior to submitting the notification of compliance, you must not burn hazardous waste for more than a total of 720 hours (renewable at the discretion of the Administrator) and only

for the purposes of pretesting or comprehensive performance testing. Pretesting is defined at $\S63.1207(h)(2)(i)$ and (ii).

(2) You may petition the Administrator to obtain written approval to burn hazardous waste in the interim prior to submitting a Notification of Compliance for purposes other than testing or pretesting. You must specify operating requirements, including limits on operating parameters, that you determine will ensure compliance with the emission standards of this subpart based on available information. The Administrator will review, modify as necessary, and approve if warranted the interim operating requirements.

(ii) Changes that will not affect compliance. If you determine that a change will not adversely affect compliance with the emission standards or operating requirements, you must document the change in the operating record upon making such change. You must revise as necessary the performance test plan, Documentation of Compliance, Notification of Compliance, and start-up, shutdown, and malfunction plan to reflect these changes.

(iii) Definition of "change." For purposes of paragraph (b)(5) of this section, "change" means any change in design, operation, or maintenance practices that were documented in the comprehensive performance test plan, Notification of Compliance, or startup, shutdown, and malfunction plan.

(6) Compliance with the carbon monoxide and hydrocarbon emission standards. This paragraph applies to sources that elect to comply with the carbon monoxide and hydrocarbon emissions standards of this subpart by documenting continuous compliance with the carbon monoxide standard using a continuous emissions monitoring system and documenting compliance with the hydrocarbon standard during the destruction and removal efficiency (DRE) performance test or its equivalent.

(i) If a DRE test performed pursuant to §63.1207(c)(2) is acceptable as documentation of compliance with the DRE standard, you may use the highest hourly rolling average hydrocarbon level achieved during the DRE test runs to document compliance with the hydrocarbon standard. An acceptable DRE test is any test for which the data and results are determined to meet quality assurance objectives (on a sitespecific basis) such that the results adequately demonstrate compliance with the DRE standard.

(ii) If during this acceptable DRE test you did not obtain hydrocarbon emissions data sufficient to document compliance with the hydrocarbon standard, you must either:

(A) Perform, as part of the performance test, an "equivalent DRE test" to document compliance with the hydrocarbon standard. An equivalent DRE test is comprised of a minimum of three runs each with a minimum duration of one hour during which you operate the combustor as close as reasonably possible to the operating parameter limits that you established based on the initial DRE test. You must use the highest hourly rolling average hydrocarbon emission level achieved during the equivalent DRE test to document compliance with the hydrocarbon standard; or

(B) Perform a DRE test as part of the performance test.

(7) Compliance with the DRE standard.(i) Except as provided in paragraphs(b)(7)(ii) and (b)(7)(iii) of this section:

(A) You must document compliance with the Destruction and Removal Efficiency (DRE) standard under this subpart only once provided that you do not modify the source after the DRE test in a manner that could affect the ability of the source to achieve the DRE standard.

(B) You may use any DRE test data that documents that your source achieves the required level of DRE provided:

(1) You have not modified the design or operation of your source in a manner that could effect the ability of your source to achieve the DRE standard since the DRE test was performed; and,

(2) The DRE test data meet quality assurance objectives determined on a site-specific basis.

(ii) Sources that feed hazardous waste at locations other than the normal flame zone. (A) Except as provided by paragraph (b)(7)(ii)(B) of this section, if you feed hazardous waste at a location in the combustion system other than the normal flame zone, then you must demonstrate compliance with the DRE standard during each comprehensive performance test;

(B)(1) A cement kiln that feeds hazardous waste at a location other than the normal flame zone need only demonstrate compliance with the DRE standard during three consecutive comprehensive performance tests provided that:

(*i*) All three tests achieve the DRE standard in this subpart; and

(*ii*) The design, operation, and maintenance features of each of the three tests are similar;

(*iii*) The data in lieu restriction of (3.1207(c)(2)(iv)) does not apply when complying with the provisions of paragraph (b)(7)(ii)(B) of this section;

(2) If at any time you change your design, operation, and maintenance features in a manner that could reasonably be expected to affect your ability to meet the DRE standard, then you must comply with the requirements of paragraph (b)(7)(ii)(A) of this section.

(iii) For sources that do not use DRE previous testing to document conformance with the DRE standard pursuant to \$63.1207(c)(2), you must perform DRE testing during the initial comprehensive performance test.

(8) Applicability of particulate matter and opacity standards during particulate matter CEMS correlation tests. (i) Any particulate matter and opacity standards of parts 60, 61, 63, 264, 265, and 266 of this chapter (i.e., any title 40 particulate or opacity standards) applicable to a hazardous waste combustor do not apply while you conduct particulate matter continuous emissions monitoring system (CEMS) correlation tests (i.e., correlation with manual stack methods) under the conditions of paragraphs (b)(8)(iii) through (vii) of this section.

(ii) Any permit or other emissions or operating parameter limits or conditions, including any limitation on workplace practices, that are applicable to hazardous waste combustors to ensure compliance with any particulate matter and opacity standards of parts 60, 61, 63, 264, 265, and 266 of this chapter (i.e., any title 40 particulate or opacity standards) do not apply while you conduct particulate matter CEMS 40 CFR Ch. I (7–1–23 Edition)

correlation tests under the conditions of paragraphs (b)(8)(iii) through (vii) of this section.

(iii) For the provisions of this section to apply, you must:

(A) Develop a particulate matter CEMS correlation test plan that includes the following information. This test plan may be included as part of the comprehensive performance test plan required under \$63.1207(e) and (f):

(1) Number of test conditions and number of runs for each test condition;

(2) Target particulate matter emission level for each test condition;

(3) How you plan to modify operations to attain the desired particulate matter emission levels; and

(4) Anticipated normal particulate matter emission levels; and

(B) Submit the test plan to the Administrator for approval at least 90 calendar days before the correlation test is scheduled to be conducted.

(iv) The Administrator will review and approve/disapprove the correlation test plan under the procedures for review and approval of the site-specific test plan provided by 63.7(c)(3)(i) and (iii). If the Administrator fails to approve or disapprove the correlation test plan within the time period specified by 63.7(c)(3)(i), the plan is considered approved, unless the Administrator has requested additional information.

(v) The particulate matter and opacity standards and associated operating limits and conditions will not be waived for more than 96 hours, in the aggregate, for a correlation test, including all runs of all test conditions, unless more time is approved by the Administrator.

(vi) The stack sampling team must be on-site and prepared to perform correlation testing no later than 24 hours after you modify operations to attain the desired particulate matter emissions concentrations, unless you document in the correlation test plan that a longer period of conditioning is appropriate.

(vii) You must return to operating conditions indicative of compliance with the applicable particulate matter and opacity standards as soon as possible after correlation testing is completed.

(9) Alternative standards for existing or new hazardous waste burning lightweight aggregate kilns using MACT. (i) You may petition the Administrator to request alternative standards to the mercury or hydrogen chloride/chlorine gas emission standards of this subpart, to the semivolatile metals emission standards under \$ 63.1205, 63.1221(a)(3)(ii), or 63.1221(b)(3)(ii), or to the low volatile metals emissions standards under §§ 63.1205, 63.1221(a)(4)(ii), or 63.1221(b)(4)(ii) if:

(A) You cannot achieve one or more of these standards while using maximum achievable control technology (MACT) because of raw material contributions to emissions of mercury, semivolatile metals, low volatile metals, or hydrogen chloride/chlorine gas; or

(B) You determine that mercury is not present at detectable levels in your raw material.

(ii) The alternative standard that you recommend under paragraph (b)(9)(i)(A) of this section may be an operating requirement, such as a hazardous waste feedrate limitation for metals and/or chlorine, and/or an emission limitation.

(iii) The alternative standard must include a requirement to use MACT, or better, applicable to the standard for which the source is seeking relief, as defined in paragraphs (b)(9)(viii) and (ix) of this section.

(iv) Documentation required. (A) The alternative standard petition you submit under paragraph (b)(9)(i)(A) of this section must include data or information documenting that raw material contributions to emissions prevent you from complying with the emission standard even though the source is using MACT, as defined under paragraphs (b)(9)(vii) and (ix) of this section, for the standard for which you are seeking relief.

(B) Alternative standard petitions that you submit under paragraph (b)(9)(i)(B) of this section must include data or information documenting that mercury is not present at detectable levels in raw materials.

(v) You must include data or information with semivolatile metal and low volatility metal alternative standard petitions that you submit under paragraph (b)(9)(i)(A) of this section documenting that increased chlorine feedrates associated with the burning of hazardous waste, when compared to non-hazardous waste operations, do not significantly increase metal emissions attributable to raw materials.

(vi) You must include data or information with semivolatile metals, low volatile metals, and hydrogen chloride/ chlorine gas alternative standard petitions that you submit under paragraph (b)(9)(i)(A) of this section documenting that semivolatile metals, low volatile metals, and hydrogen chloride/chlorine gas emissions attributable to the hazardous waste only will not exceed the emission standards of this subpart.

(vii) You must not operate pursuant to your recommended alternative standards in lieu of emission standards specified in this subpart:

(A) Unless the Administrator approves the provisions of the alternative standard petition request or establishes other alternative standards; and

(B) Until you submit a revised Notification of Compliance that incorporates the revised standards.

(viii) For purposes of this alternative standard provision, MACT for existing hazardous waste burning lightweight aggregate kilns is defined as:

(A) For mercury, a hazardous waste feedrate corresponding to an MTEC of $24 \mu g/dscm$ or less;

(B) For semivolatile metals, a hazardous waste feedrate corresponding to an MTEC of 280,000 μ g/dscm or less, and use of a particulate matter control device that achieves particulate matter emissions of 57 mg/dscm or less;

(C) For low volatile metals, a hazardous waste feedrate corresponding to an MTEC of 120,000 μ g/dscm or less, and use of a particulate matter control device that achieves particulate matter emissions of 57 mg/dscm or less; and

(D) For hydrogen chloride/chlorine gas, a hazardous waste chlorine feedrate corresponding to an MTEC of 2,000,000 μ gm/dscm or less, and use of an air pollution control device with a hydrogen chloride/chlorine gas removal efficiency of 85 percent or greater.

(ix) For purposes of this alternative standard provision, MACT for new hazardous waste burning lightweight aggregate kilns is defined as:

§63.1206

(A) For mercury, a hazardous waste feedrate corresponding to an MTEC of 4 μ g/dscm or less;

(B) For semivolatile metals, a hazardous waste feedrate corresponding to an MTEC of 280,000 μ g/dscm or less, and use of a particulate matter control device that achieves particulate matter emissions of 57 mg/dscm or less;

(C) For low volatile metals, a hazardous waste feedrate corresponding to an MTEC of 46,000 μ g/dscm or less, and use of a particulate matter control device that achieves particulate matter emissions of 57 mg/dscm or less;

(D) For hydrogen chloride/chlorine gas, a hazardous waste chlorine feedrate corresponding to an MTEC of 14,000,000 μ gm/dscm or less, and use of an air pollution control device with a hydrogen chloride/chlorine gas removal efficiency of 99.6 percent or greater.

(10) Alternative standards for existing or new hazardous waste burning cement kilns using MACT. (i) You may petition the Administrator to request alternative standards to the mercury or hydrogen chloride/chlorine gas emission standards of this subpart, to the semivolatile metals emission standards under §§63.1204, 63.1220(a)(3)(ii), or 63.1220(b)(3)(ii), or to the low volatile metals emissions standards under 63.1220(a)(4)(ii), §§ 63.1204. or 63.1220(b)(4)(ii) if:

(A) You cannot achieve one or more of these standards while using maximum achievable control technology (MACT) because of raw material contributions to emissions of mercury, semivolatile metals, low volatile metals, or hydrogen chloride/chlorine gas; or

(B) You determine that mercury is not present at detectable levels in your raw material.

(ii) The alternative standard that you recommend under paragraph (b)(10)(i)(A) of this section may be an operating requirement, such as a hazardous waste feedrate limitation for metals and/or chlorine, and/or an emission limitation.

(iii) The alternative standard must include a requirement to use MACT, or better, applicable to the standard for which the source is seeking relief, as defined in paragraphs (b)(10)(viii) and (ix) of this section.

40 CFR Ch. I (7–1–23 Edition)

(iv) Documentation required. (A) The alternative standard petition you submit under paragraph (b)(10)(i)(A) of this section must include data or information documenting that raw material contributions to emissions prevent you from complying with the emission standard even though the source is using MACT, as defined in paragraphs (b)(10)(viii) and (ix) of this section, for the standard for which you are seeking relief.

(B) Alternative standard petitions that you submit under paragraph (b)(10)(i)(B) of this section must include data or information documenting that mercury is not present at detectable levels in raw materials.

(v) You must include data or information with semivolatile metal and low volatile metal alternative standard petitions that you submit under paragraph (b)(10)(i)(A) of this section documenting that increased chlorine feedrates associated with the burning of hazardous waste, when compared to non-hazardous waste operations, do not significantly increase metal emissions attributable to raw materials.

(vi) You must include data or information with semivolatile metals, low volatile metals, and hydrogen chloride/ chlorine gas alternative standard petitions that you submit under paragraph (b)(10)(i)(A) of this section documenting that emissions of the regulated metals and hydrogen chloride/ chlorine gas attributable to the hazardous waste only will not exceed the emission standards in this subpart.

(vii) You must not operate pursuant to your recommended alternative standards in lieu of emission standards specified in this subpart:

(A) Unless the Administrator approves the provisions of the alternative standard petition request or establishes other alternative standards: and

(B) Until you submit a revised Notification of Compliance that incorporates the revised standards.

(viii) For purposes of this alternative standard provision, MACT for existing hazardous waste burning cement kilns is defined as:

(A) For mercury, a hazardous waste feedrate corresponding to an MTEC of $88 \mu g/dscm$ or less;

(B) For semivolatile metals, a hazardous waste feedrate corresponding to an MTEC of 31,000 μ g/dscm or less, and use of a particulate matter control device that achieves particulate matter emissions of 0.15 kg/Mg dry feed or less;

(C) For low volatile metals, a hazardous waste feedrate corresponding to an MTEC of 54,000 $\mu g/dscm$ or less, and use of a particulate matter control device that achieves particulate matter emissions of 0.15 kg/Mg dry feed or less; and

(D) For hydrogen chloride/chlorine gas, a hazardous waste chlorine feedrate corresponding to an MTEC of 720.000 μ gm/dscm or less.

(ix) For purposes of this alternative standard provision, MACT for new hazardous waste burning cement kilns is defined as:

(A) For mercury, a hazardous waste feedrate corresponding to an MTEC of 7 μ g/dscm or less;

(B) For semivolatile metals, a hazardous waste feedrate corresponding to an MTEC of 31,000 μ g/dscm or less, and use of a particulate matter control device that achieves particulate matter emissions of 0.15 kg/Mg dry feed or less;

(C) For low volatile metals, a hazardous waste feedrate corresponding to an MTEC of 15,000 μ g/dscm or less, and use of a particulate matter control device that achieves particulate matter emissions of 0.15 kg/Mg dry feed or less;

(D) For hydrogen chloride/chlorine gas, a hazardous waste chlorine feedrate corresponding to an MTEC of $420,000 \ \mu gm/dscm$ or less.

(11) Calculation of hazardous waste residence time. You must calculate the hazardous waste residence time and include the calculation in the performance test plan under §63.1207(f) and the operating record. You must also provide the hazardous waste residence time in the Documentation of Compliance under §63.1211(c) and the Notification of Compliance under §§63.1207(j) and 63.1210(d).

(12) Documenting compliance with the standards based on performance testing.
(i) You must conduct a minimum of three runs of a performance test required under §63.1207 to document compliance with the emission standards of this subpart.

(ii) You must document compliance with the emission standards based on the arithmetic average of the emission results of each run, except that you must document compliance with the destruction and removal efficiency standard for each run of the comprehensive performance test individually.

(13) Cement kilns and lightweight aggregate kilns that feed hazardous waste at a location other than the end where products are normally discharged and where fuels are normally fired. (i) Cement kilns that feed hazardous waste at a location other than the end where products are normally discharged and where fuels are normally fired must comply with the carbon monoxide and hydrocarbon standards of this subpart as follows:

(A) For existing sources, you must not discharge or cause combustion gases to be emitted into the atmosphere that contain either:

(1) Hydrocarbons in the main stack in excess of 20 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or

(2) Hydrocarbons both in the by-pass duct and at a preheater tower combustion gas monitoring location in excess of 10 parts per million by volume, at each location, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or

(3) If the only firing location of hazardous waste upstream (in terms of gas flow) of the point where combustion gases are diverted into the bypass duct is at the kiln end where products are normally discharged, then both hydrocarbons at the preheater tower combustion gas monitoring location in excess of 10 parts per million by volume. over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane, and either hydrocarbons in the by-pass duct in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry

basis, corrected to 7 percent oxygen, and reported as propane, or carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), drv basis, and corrected to 7 percent oxygen. If you comply with the carbon monoxide standard of 100 parts per million by volume in the by-pass duct, then you must also not discharge or cause combustion gases to be emitted into the atmosphere that contain hydrocarbons in the by-pass duct in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane, at any time during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7).

(B) For new sources, you must not discharge or cause combustion gases to be emitted into the atmosphere that contain either:

(1) Hydrocarbons in the main stack in excess of 20 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or

(2)(i) Hydrocarbons both in the bypass duct and at a preheater tower combustion gas monitoring location in excess of 10 parts per million by volume, at each location, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane, and

(*ii*) Hydrocarbons in the main stack, if construction of the kiln commenced after April 19, 1996 at a plant site where a cement kiln (whether burning hazardous waste or not) did not previously exist, to 50 parts per million by volume, over a 30-day block average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or

(3)(i) If the only firing location of hazardous waste upstream (in terms of gas flow) of the point where combustion gases are diverted into the bypass

40 CFR Ch. I (7–1–23 Edition)

duct is at the kiln end where products are normally discharged, then both hydrocarbons at the preheater tower combustion gas monitoring location in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane, and either hydrocarbons in the by-pass duct in excess of 10 parts per million by volume. over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane, or carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, and corrected to 7 percent oxygen. If you comply with the carbon monoxide standard of 100 parts per million by volume in the by-pass duct, then you must also not discharge or cause combustion gases to be emitted into the atmosphere that contain hydrocarbons in the by-pass duct in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane, at any time during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7).

(*ii*) If construction of the kiln commenced after April 19, 1996 at a plant site where a cement kiln (whether burning hazardous waste or not) did not previously exist, hydrocarbons are limited to 50 parts per million by volume, over a 30-day block average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane.

(ii) Lightweight aggregate kilns that feed hazardous waste at a location other than the end where products are normally discharged and where fuels are normally fired must comply with the hydrocarbon standards of this subpart as follows:

(A) Existing sources must comply with the 20 parts per million by volume hydrocarbon standard of this subpart;

(B) New sources must comply with the 20 parts per million by volume hydrocarbon standard of this subpart.

(14) Alternative to the particulate matter standard for incinerators—(i) General. In lieu of complying with the particulate matter standards under §63.1203, you may elect to comply with the following alternative metal emission control requirements:

(ii) Alternative metal emission control requirements for existing incinerators. (A) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain cadmium, lead, and selenium in excess of 240 μ gm/dscm, combined emissions, corrected to 7 percent oxygen; and,

(B) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain antimony, arsenic, beryllium, chromium, cobalt, manganese, and nickel in excess of 97 µgm/dscm, combined emissions, corrected to 7 percent oxygen.

(iii) Alternative metal emission control requirements for new incinerators. (A) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain cadmium, lead, and selenium in excess of 24 µgm/dscm, combined emissions, corrected to 7 percent oxygen; and,

(B) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain antimony, arsenic, beryllium, chromium, cobalt, manganese, and nickel in excess of 97 µgm/dscm, combined emissions, corrected to 7 percent oxygen.

(iv) Operating limits. Semivolatile and low volatile metal operating parameter limits must be established to ensure compliance with the alternative emission limitations described in paragraphs (b)(14)(ii) and (iii) of this section pursuant to §63.1209(n), except that semivolatile metal feedrate limits apply to lead, cadmium, and selenium, combined, and low volatile metal feedrate limits apply to arsenic, beryllium, chromium, antimony, cobalt, manganese, and nickel, combined.

(15) Alternative to the interim standards for mercury for cement and lightweight aggregate kilns—(i) General. In lieu of complying with the applicable mercury standards of \S 63.1204(a)(2) and (b)(2) for existing and new cement kilns and §§ 63.1205(a)(2) and (b)(2) for existing and new lightweight aggregate kilns, you may instead elect to comply with the alternative mercury standard described in paragraphs (b)(15)(ii) through (b)(15)(v) of this section.

(ii) Operating requirement. You must not exceed a hazardous waste feedrate corresponding to a maximum theoretical emission concentration (MTEC) of $120 \ \mu g/dscm$ on a twelve-hour rolling average.

(iii) To document compliance with the operating requirement of paragraph (b)(15)(ii) of this section, you must:

(A) Monitor and record the feedrate of mercury for each hazardous waste feedstream according to §63.1209(c);

(B) Monitor with a CMS and record in the operating record the gas flowrate (either directly or by monitoring a surrogate parameter that you have correlated to gas flowrate);

(C) Continuously calculate and record in the operating record a MTEC assuming mercury from all hazardous waste feedstreams is emitted;

(D) Interlock the MTEC calculated in paragraph (b)(15)(iii)(C) of this section to the AWFCO system to stop hazardous waste burning when the MTEC exceeds the operating requirement of paragraph (b)(15)(ii) of this section.

(iv) In lieu of the requirement in paragraph (b)(15)(iii) of this section, you may:

(A) Identify in the Notification of Compliance a minimum gas flowrate limit and a maximum feedrate limit of mercury from all hazardous waste feedstreams that ensures the MTEC calculated in paragraph (b)(15)(iii)(C) of this section is below the operating requirement of paragraph (b)(15)(ii) of this section; and

(B) Interlock the minimum gas flowrate limit and maximum feedrate limits in paragraph (b)(15)(iv)(A) of this section to the AWFCO system to stop hazardous waste burning when the gas flowrate or mercury feedrate exceeds the limits in paragraph (b)(15)(iv)(A) of this section.

(v) *Notification requirement*. You must notify in writing the RCRA authority that you intend to comply with the alternative standard.

(16) Compliance with subcategory standards for liquid fuel boilers. You must comply with the mercury, semivolatile metals, low volatile metals, and hydrogen chloride and chlorine standards for liquid fuel boilers under §63.1217 as follows:

(i) You must determine the as-fired heating value of each batch of hazardous waste fired by each firing system of the boiler so that you know the mass-weighted heating value of the hazardous waste fired at all times.

(ii) If the as-fired heating value of the hazardous waste is 10,000 Btu per pound or greater, you are subject to the thermal emission concentration standards (lb/million Btu) under §63.1217.

(iii) If the as-fired heating value of the hazardous waste is less than 10,000 Btu/lb, you are subject to the mass or volume emission concentration standards (μ gm/dscm or ppmv) under §63.1217.

(iv) If the as-fired heating value of hazardous wastes varies above and below 10,000 Btu/lb over time, you are subject to the thermal concentration standards when the heating value is 10,000 Btu/lb or greater and the mass concentration standards when the heating value is less than 10,000 Btu/lb. You may elect to comply at all times with the more stringent operating requirements that ensure compliance with both the thermal emission concentration standards and the mass or volume emission concentration standards.

(c) Operating requirements—(1) General. (i) You must operate only under the operating requirements specified in the Documentation of Compliance under §63.1211(c) or the Notification of Compliance under §§63.1207(j) and 63.1210(d), except:

(A) During performance tests under approved test plans according to §63.1207(e), (f), and (g), and

(B) Under the conditions of paragraph (b)(1)(i) or (ii) of this section;

(ii) The Documentation of Compliance and the Notification of Compliance must contain operating requirements including, but not limited to, the operating requirements in this section and §63.1209 40 CFR Ch. I (7–1–23 Edition)

(iii) Failure to comply with the operating requirements is failure to ensure compliance with the emission standards of this subpart;

(iv) Operating requirements in the Notification of Compliance are applicable requirements for purposes of parts 70 and 71 of this chapter;

(v) The operating requirements specified in the Notification of Compliance will be incorporated in the title V permit.

(2) Startup, shutdown, and malfunction plan. (i) You are subject to the startup, shutdown, and malfunction plan requirements of 63.6(e)(3).

(ii) If you elect to comply with §§ 270.235(a)(1)(iii), 270.235(a)(2)(iii), or 270.235(b)(1)(ii) of this chapter to address RCRA concerns that you minimize emissions of toxic compounds from startup, shutdown, and malfunction events (including releases from emergency safety vents):

(A) The startup, shutdown, and malfunction plan must include a description of potential causes of malfunctions, including releases from emergency safety vents, that may result in significant releases of hazardous air pollutants, and actions the source is taking to minimize the frequency and severity of those malfunctions.

(B) You must submit the startup, shutdown, and malfunction plan to the Administrator for review and approval.

(1) Approval procedure. The Administrator will notify you of approval or intention to deny approval of the startup, shutdown, and malfunction plan within 90 calendar days after receipt of the original request and within 60 calendar days after receipt of any supplemental information that you submit. Before disapproving the plan, the Administrator will notify you of the Administrator's intention to disapprove the plan together with:

(*i*) Notice of the information and findings on which intended disapproval is based; and

(*ii*) Notice of opportunity for you to present additional information to the Administrator before final action on disapproval of the plan. At the time the Administrator notifies you of intention to disapprove the plan, the Administrator will specify how much time you will have after being notified

on the intended disapproval to submit additional information.

(2) Responsibility of owners and operators. You are responsible for ensuring that you submit any supplementary and additional information supporting your plan in a timely manner to enable the Administrator to consider whether to approve the plan. Neither your submittal of the plan, nor the Administrator's failure to approve or disapprove the plan, relieves you of the responsibility to comply with the provisions of this subpart.

(C) Changes to the plan that may significantly increase emissions. (1) You must request approval in writing from the Administrator within 5 days after making a change to the startup, shutdown, and malfunction plan that may significantly increase emissions of hazardous air pollutants.

(2) To request approval of such changes to the startup, shutdown, and malfunction plan, you must follow the procedures provided by paragraph (c)(2)(ii)(B) of this section for initial approval of the plan.

(iii) You must identify in the plan a projected oxygen correction factor based on normal operations to use during periods of startup and shutdown.

(iv) You must record the plan in the operating record.

(v) Operating under the startup, shutdown, and malfunction plan-(A) Compliance with AWFCO requirements during malfunctions. (1) During malfunctions, the automatic waste feed cutoff requirements of §63.1206(c)(3) continue to apply, except for paragraphs (c)(3)(v)and (c)(3)(vi) of this section. If you exceed a part 63, subpart EEE, of this chapter emission standard monitored by a CEMS or COMs or operating limit specified under §63.1209, the automatic waste feed cutoff system must immediately and automatically cutoff the hazardous waste feed, except as provided by paragraph (c)(3)(viii) of this section. If the malfunction itself prevents immediate and automatic cutoff of the hazardous waste feed, however, you must cease feeding hazardous waste as quickly as possible.

(2) Although the automatic waste feed cutoff requirements continue to apply during a malfunction, an exceedance of an emission standard monitored by a CEMS or COMS or operating limit specified under §63.1209 is not a violation of this subpart if you take the corrective measures prescribed in the startup, shutdown, and malfunction plan.

(3) Excessive exceedances during malfunctions. For each set of 10 exceedances of an emission standard or operating requirement while hazardous waste remains in the combustion chamber (i.e., when the hazardous waste residence time has not transpired since the hazardous waste feed was cutoff) during a 60-day block period, you must:

(i) Within 45 days of the 10th exceedance, complete an investigation of the cause of each exceedance and evaluation of approaches to minimize the frequency, duration, and severity of each exceedance, and revise the startup, shutdown, and malfunction plan as warranted by the evaluation to minimize the frequency, duration, and severity of each exceedance; and

(*ii*) Record the results of the investigation and evaluation in the operating record, and include a summary of the investigation and evaluation, and any changes to the startup, shutdown, and malfunction plan, in the excess emissions report required under $\S63.10(e)(3)$.

(B) Compliance with AWFCO requirements when burning hazardous waste during startup and shutdown. (1) If you feed hazardous waste during startup or shutdown, you must include waste feed restrictions (e.g., type and quantity), and other appropriate operating conditions and limits in the startup, shutdown, and malfunction plan.

(2) You must interlock the operating limits you establish under paragraph (c)(2)(v)(B)(1) of this section with the automatic waste feed cutoff system required under §63.1206(c)(3), except for paragraphs (c)(3)(v) and (c)(3)(vi) of this section.

(3) When feeding hazardous waste during startup or shutdown, the automatic waste feed cutoff system must immediately and automatically cutoff the hazardous waste feed if you exceed the operating limits you establish under paragraph (c)(2)(v)(B)(I) of this section, except as provided by paragraph (c)(3)(viii) of this section. (4) Although the automatic waste feed cutoff requirements of this paragraph apply during startup and shutdown, an exceedance of an emission standard or operating limit is not a violation of this subpart if you comply with the operating procedures prescribed in the startup, shutdown, and malfunction plan.

(3) Automatic waste feed cutoff (AWFCO)—(i) General. Upon the compliance date, you must operate the hazardous waste combustor with a functioning system that immediately and automatically cuts off the hazardous waste feed, except as provided by paragraph (c)(3)(viii) of this section:

(A) When any of the following are exceeded: Operating parameter limits specified under §63.1209; an emission standard monitored by a CEMS; and the allowable combustion chamber pressure;

(B) When the span value of any CMS detector, except a CEMS, is met or exceeded;

(C) Upon malfunction of a CMS monitoring an operating parameter limit specified under §63.1209 or an emission level; or

(D) When any component of the automatic waste feed cutoff system fails.

(ii) Ducting of combustion gases. During an AWFCO, you must continue to duct combustion gasses to the air pollution control system while hazardous waste remains in the combustion chamber (i.e., if the hazardous waste residence time has not transpired since the hazardous waste feed cutoff system was activated).

(iii) Restarting waste feed. You must continue to monitor during the cutoff the operating parameters for which limits are established under §63.1209 and the emissions required under that section to be monitored by a CEMS, and you must not restart the hazardous waste feed until the operating parameters and emission levels are within the specified limits.

(iv) Failure of the AWFCO system. If the AWFCO system fails to automatically and immediately cutoff the flow of hazardous waste upon exceedance of a parameter required to be interlocked with the AWFCO system under paragraph (c)(3)(i) of this section, you have failed to comply with the AWFCO re40 CFR Ch. I (7–1–23 Edition)

quirements of paragraph (c)(3) of this section. If an equipment or other failure prevents immediate and automatic cutoff of the hazardous waste feed, however, you must cease feeding hazardous waste as quickly as possible.

(v) Corrective measures. If, after any AWFCO, there is an exceedance of an emission standard or operating requirement, irrespective of whether the exceedance occurred while hazardous waste remained in the combustion chamber (i.e., whether the hazardous waste residence time has transpired since the hazardous waste feed cutoff system was activated), you must investigate the cause of the AWFCO, take appropriate corrective measures to minimize future AWFCOs, and record the findings and corrective measures in the operating record.

(vi) Excessive exceedance reporting. (A) For each set of 10 exceedances of an emission standard or operating requirement while hazardous waste remains in the combustion chamber (i.e., when the hazardous waste residence time has not transpired since the hazardous waste feed was cutoff) during a 60-day block period, you must submit to the Administrator a written report within 5 calendar days of the 10th exceedance documenting the exceedances and results of the investigation and corrective measures taken.

(B) On a case-by-case basis, the Administrator may require excessive exceedance reporting when fewer than 10 exceedances occur during a 60-day block period.

(vii) *Testing*. The AWFCO system and associated alarms must be tested at least weekly to verify operability, unless you document in the operating record that weekly inspections will unduly restrict or upset operations and that less frequent inspection will be adequate. At a minimum, you must conduct operability testing at least monthly. You must document and record in the operating record AWFCO operability test procedures and results.

(viii) Ramping down waste feed. (A) You may ramp down the waste feedrate of pumpable hazardous waste over a period not to exceed one minute, except as provided by paragraph (c)(3)(viii)(B) of this section. If you elect to ramp

down the waste feed, you must document ramp down procedures in the operating and maintenance plan. The procedures must specify that the ramp down begins immediately upon initiation of automatic waste feed cutoff and the procedures must prescribe a bona fide ramping down. If an emission standard or operating limit is exceeded during the ramp down, you have failed to comply with the emission standards or operating requirements of this subpart.

(B) If the automatic waste feed cutoff is triggered by an exceedance of any of the following operating limits, you may not ramp down the waste feed cutoff: Minimum combustion chamber temperature, maximum hazardous waste feedrate, or any hazardous waste firing system operating limits that may be established for your combustor.

(4) ESV openings-(i) Failure to meet standards. If an emergency safety vent (ESV) opens when hazardous waste remains in the combustion chamber (i.e., when the hazardous waste residence time has not expired) during an event other than a malfunction as defined in the startup, shutdown, and malfunction plan such that combustion gases are not treated as during the most recent comprehensive performance test (e.g., if the combustion gas by-passes any emission control device that was operating during the performance test), you must document in the operating record whether you remain in compliance with the emission standards of this subpart considering emissions during the ESV opening event.

(ii) *ESV operating plan*. (A) You must develop an ESV operating plan, comply with the operating plan, and keep the plan in the operating record.

(B) The ESV operating plan must provide detailed procedures for rapidly stopping the waste feed, shutting down the combustor, and maintaining temperature and negative pressure in the combustion chamber during the hazardous waste residence time, if feasible. The plan must include calculations and information and data documenting the effectiveness of the plan's procedures for ensuring that combustion chamber temperature and negative pressure are maintained as is reasonably feasible. (iii) Corrective measures. After any ESV opening that results in a failure to meet the emission standards as defined in paragraph (c)(4)(i) of this section, you must investigate the cause of the ESV opening, take appropriate corrective measures to minimize such future ESV openings, and record the findings and corrective measures in the operating record.

(iv) Reporting requirements. You must submit to the Administrator a written report within 5 days of an ESV opening that results in failure to meet the emission standards of this subpart (as determined in paragraph (c)(4)(i) of this section) documenting the result of the investigation and corrective measures taken.

(5) Combustion system leaks. (i) Combustion system leaks of hazardous air pollutants must be controlled by:

(A) Keeping the combustion zone sealed to prevent combustion system leaks; or

(B) Maintaining the maximum combustion zone pressure lower than ambient pressure using an instantaneous monitor; or

(C) Upon prior written approval of the Administrator, an alternative means of control to provide control of combustion system leaks equivalent to maintenance of combustion zone pressure lower than ambient pressure; or

(D) Upon prior written approval of the Administrator, other technique(s) which can be demonstrated to prevent fugitive emissions without use of instantaneous pressure limits; and

(ii) You must specify in the performance test workplan and Notification of Compliance the method that will be used to control combustion system leaks. If you control combustion system leaks by maintaining the combustion zone pressure lower than ambient pressure using an instantaneous monitor, you must also specify in the performance test workplan and Notification of Compliance the monitoring and recording frequency of the pressure monitor, and specify how the monitoring approach will be integrated into the automatic waste feed cutoff system.

(6) Operator training and certification.(i) You must establish training programs for all categories of personnel

whose activities may reasonably be expected to directly affect emissions of hazardous air pollutants from the source. Such persons include, but are not limited to, chief facility operators, control room operators, continuous monitoring system operators, persons that sample and analyze feedstreams, persons that manage and charge feedstreams to the combustor, persons that operate emission control devices, and ash and waste handlers. Each training program shall be of a technical level commensurate with the person's job duties specified in the training manual. Each commensurate training program shall require an examination to be administered by the instructor at the end of the training course. Passing of this test shall be deemed the "certification" for personnel, except that, for control room operators, the training and certification program shall be as specified in paragraphs (c)(6)(iii) through (c)(6)(vi) of this section.

(ii) You must ensure that the source is operated and maintained at all times by persons who are trained and certified to perform these and any other duties that may affect emissions of hazardous air pollutants. A certified control room operator must be on duty at the site at all times the source is in operation.

(iii) Hazardous waste incinerator control room operators must:

(A) Be trained and certified under a site-specific, source-developed and implemented program that meets the requirements of paragraph (c)(6)(v) of this section; or

(B) Be trained under the requirements of, and certified under, one of the following American Society of Mechanical Engineers (ASME) standards: QHO-1-1994, QHO-1a-1996, or QHO-1-2004 (Standard for the Qualification and Certification of Hazardous Waste Incinerator Operators). If you elect to use the ASME program:

(1) Control room operators must, prior to the compliance date, achieve provisional certification, and must submit an application to ASME and be scheduled for the full certification exam. Within one year of the compliance date, control room operators must achieve full certification; 40 CFR Ch. I (7–1–23 Edition)

(2) New operators and operators of new sources must, before assuming their duties, achieve provisional certification, and must submit an application to ASME, and be scheduled for the full certification exam. Within one year of assuming their duties, these operators must achieve full certification; or

(C) Be trained and certified under a State program.

(iv) Control room operators of cement kilns, lightweight aggregate kilns, solid fuel boilers, liquid fuel boilers, and hydrochloric acid production furnaces must be trained and certified under:

(A) A site-specific, source-developed and implemented program that meets the requirements of paragraph (c)(6)(v)of this section; or

(B) A State program.

(v) Site-specific, source developed and implemented training programs for control room operators must include the following elements:

(A) Training on the following subjects:

(1) Environmental concerns, including types of emissions;

(2) Basic combustion principles, including products of combustion;

(3) Operation of the specific type of combustor used by the operator, including proper startup, waste firing, and shutdown procedures;

(4) Combustion controls and continuous monitoring systems;

(5) Operation of air pollution control equipment and factors affecting performance;

(6) Inspection and maintenance of the combustor, continuous monitoring systems, and air pollution control devices;

(7) Actions to correct malfunctions or conditions that may lead to malfunction;

 (δ) Residue characteristics and handling procedures; and

(9) Applicable Federal, state, and local regulations, including Occupational Safety and Health Administration workplace standards; and

(B) An examination designed and administered by the instructor; and

(C) Written material covering the training course topics that may serve as reference material following completion of the course.

(vi) To maintain control room operator qualification under a site-specific, source developed and implemented training program as provided by paragraph (c)(6)(v) of this section, control room operators must complete an annual review or refresher course covering, at a minimum, the following topics:

(A) Update of regulations;

(B) Combustor operation, including startup and shutdown procedures, waste firing, and residue handling;

(C) Inspection and maintenance;

(D) Responses to malfunctions or conditions that may lead to malfunction; and

(E) Operating problems encountered by the operator.

(vii) You must record the operator training and certification program in the operating record.

(7) Operation and maintenance plan— (i) You must prepare and at all times operate according to an operation and maintenance plan that describes in detail procedures for operation, inspection, maintenance, and corrective measures for all components of the combustor, including associated pollution control equipment, that could affect emissions of regulated hazardous air pollutants.

(ii) The plan must prescribe how you will operate and maintain the combustor in a manner consistent with good air pollution control practices for minimizing emissions at least to the levels achieved during the comprehensive performance test.

(iii) This plan ensures compliance with the operation and maintenance requirements of §63.6(e) and minimizes emissions of pollutants, automatic waste feed cutoffs, and malfunctions.

(iv) You must record the plan in the operating record.

(8) Bag leak detection system requirements. (i) If your combustor is equipped with a baghouse (fabric filter), you must continuously operate either:

(A) A bag leak detection system that meets the specifications and requirements of paragraph (c)(8)(ii) of this section and you must comply with the corrective measures and notification requirements of paragraphs (c)(8)(ii)and (iv) of this section; or (B) A particulate matter detection system under paragraph (c)(9) of this section.

(ii) Bag leak detection system specification and requirements. (A) The bag leak detection system must be certified by the manufacturer to be capable of continuously detecting and recording particulate matter emissions at concentrations of 1.0 milligrams per actual cubic meter unless you demonstrate, under 63.1209(g)(1), that a higher detection limit would routinely detect particulate matter loadings during normal operations;

(B) The bag leak detection system shall provide output of relative or absolute particulate matter loadings;

(C) The bag leak detection system shall be equipped with an alarm system that will sound an audible alarm when an increase in relative particulate loadings is detected over a preset level;

(D) The bag leak detection system shall be installed and operated in a manner consistent with available written guidance from the U.S. Environmental Protection Agency or, in the absence of such written guidance, the manufacturer's written specifications and recommendations for installation, operation, and adjustment of the system;

(E) The initial adjustment of the system shall, at a minimum, consist of establishing the baseline output by adjusting the sensitivity (range) and the averaging period of the device, and establishing the alarm set points and the alarm delay time;

(F) Following initial adjustment, you must not adjust the sensitivity or range, averaging period, alarm set points, or alarm delay time, except as detailed in the operation and maintenance plan required under paragraph (c)(7) of this section. You must not increase the sensitivity by more than 100 percent or decrease the sensitivity by more than 50 percent over a 365 day period unless such adjustment follows a complete baghouse inspection which demonstrates the baghouse is in good operating condition;

(G) For negative pressure or induced air baghouses, and positive pressure baghouses that are discharged to the atmosphere through a stack, the bag leak detector shall be installed downstream of the baghouse and upstream of any wet acid gas scrubber; and

(H) Where multiple detectors are required, the system's instrumentation and alarm system may be shared among the detectors.

(iii) Bag leak detection system corrective measures requirements. The operating and maintenance plan required by paragraph (c)(7) of this section must include a corrective measures plan that specifies the procedures you will follow in the case of a bag leak detection system alarm or malfunction. The corrective measures plan must include, at a minimum, the procedures used to determine and record the time and cause of the alarm or bag leak detection system malfunction in accordance with of the requirements paragraph (c)(8)(iii)(A) of this section as well as the corrective measures taken to correct the control device or bag leak detection system malfunction or to minimize emissions in accordance with the requirements of paragraph (c)(8)(iii)(B) of this section. Failure to initiate the corrective measures required by this paragraph is failure to ensure compliance with the emission standards in this subpart.

(A) You must initiate the procedures used to determine the cause of the alarm or bag leak detection system malfunction within 30 minutes of the time the alarm first sounds; and

(B) You must alleviate the cause of the alarm or bag leak detection system malfunction by taking the necessary corrective measure(s) which may include, but are not to be limited to, the following:

(1) Inspecting the baghouse for air leaks, torn or broken filter elements, or any other malfunction that may cause an increase in emissions;

(2) Sealing off defective bags or filter media;

(3) Replacing defective bags or filter media, or otherwise repairing the control device;

(4) Sealing off a defective baghouse compartment;

(5) Cleaning the bag leak detection system probe, or otherwise repairing the bag leak detection system; or

(6) Shutting down the combustor.

40 CFR Ch. I (7–1–23 Edition)

(iv) Excessive exceedances notification. If you operate the combustor when the detector response exceeds the alarm set-point or the bag leak detection system is malfunctioning more than 5 percent of the time during any 6-month block time period, you must submit a notification to the Administrator within 30 days of the end of the 6-month block time period that describes the causes of the exceedances and bag leak detection system malfunctions and the revisions to the design, operation, or maintenance of the combustor, baghouse, or bag leak detection system taking vou are to minimize exceedances and bag leak detection system malfunctions. To document compliance with this requirement:

(A) You must keep records of the date, time, and duration of each alarm and bag leak detection system malfunction, the time corrective action was initiated and completed, and a brief description of the cause of the alarm or bag leak detection system malfunction and the corrective action taken;

(B) You must record the percent of the operating time during each 6month period that the alarm sounds and the bag leak detection system malfunctions;

(C) If inspection of the fabric filter demonstrates that no corrective action is required, then no alarm time is counted; and

(D) If corrective action is required, each alarm shall be counted as a minimum of 1 hour. Each bag leak detection system malfunction shall also be counted as a minimum of 1 hour.

(9) Particulate matter detection system *requirements*. You must continuously operate a particulate matter detection system (PMDS) that meets the specifications and requirements of paragraphs (c)(9)(i) through (v) of this section and you must comply with the corrective measures and notification requirements of paragraphs (c)(9)(vii) and (viii) of this section if your combustor either: Is equipped with an electrostatic precipitator or ionizing wet scrubber and you do not establish sitespecific control device operating parameter limits under §63.1209(m)(1)(iv) that are linked to the automatic waste feed cutoff system under paragraph

(c)(3) of this section, or is equipped with a baghouse (fabric filter) and you do not operate a bag leak detection system as provided by paragraph (c)(8)(i)(B) of this section.

(i) *PMDS requirements*. (A) The PMDS must be certified by the manufacturer to be capable of continuously detecting and recording particulate matter emissions at concentrations of 1.0 milligrams per actual cubic meter unless you demonstrate, under §63.1209(g)(1), that a higher detection limit would routinely detect particulate matter loadings during normal operations;

(B) The particulate matter detector shall provide output of relative or absolute particulate matter loadings;

(C) The PMDS shall be equipped with an alarm system that will sound an audible alarm when an increase in relative or absolute particulate loadings is detected over the set-point;

(D) You must install, operate, and maintain the PMDS in a manner consistent with the provisions of paragraph (c)(9) of this section and available written guidance from the U.S. Environmental Protection Agency or, in the absence of such written guidance, the manufacturer's written specifications and recommendations for installation, operation, maintenance and quality assurance of the system.

(1) Set-points established without extrapolation. If you establish the alarm set-point without extrapolation under paragraph (c)(9)(iii)(A) of this section, you must request approval from the regulatory authority, in the continuous monitoring system test plan, of the quality assurance procedures that will reasonably ensure that PMDS response values below the alarm setpoint correspond to PM emission concentrations below those demonstrated during the comprehensive performance test. Your recommended quality assurance procedures may include periodic testing under as-found conditions (i.e., normal operations) to obtain additional PM concentration and PMDS response run pairs, as warranted.

(2) Set-points established with extrapolation. If you establish the alarm setpoint by extrapolation under paragraph (c)(9)(iii)(B) of this section, you must request approval from the regulatory authority, in the continuous monitoring system test plan, of the quality assurance procedures that will reasonably ensure that PMDS response values below the alarm set-point correspond to PM emission concentrations below the value that correlates to the alarm set-point.

(E) You must include procedures for installation, operation, maintenance, and quality assurance of the PMDS in the site-specific continuous monitoring system test plan required under §§ 63.1207(e) and 63.8(e)(3);

(F) Where multiple detectors are required to monitor multiple control devices, the system's instrumentation and alarm system may be shared among the detectors.

(G) You must establish the alarm setpoint as a 6-hour rolling average as provided by paragraphs (c)(9)(ii), (c)(9)(iii), and (c)(9)(iv) of this section;

(H) Your PMDS must complete a minimum of one cycle of operation (sampling, analyzing, and data recording) for each successive 15-minute period. You must update the 6-hour rolling average of the detector response each hour with a one-hour block average that is the average of the detector responses over each 15-minute block; and

(I) If you exceed the alarm set-point (or if your PMDS malfunctions), you must comply with the corrective measures under paragraph (c)(9)(vii) of this section.

(ii) Establishing the alarm set-point for operations under the Documentation of Compliance. You must establish the alarm set-point for operations under the Documentation of Compliance (i.e., after the compliance date but prior to submitting a Notification of Compliance subsequent to conducting the initial comprehensive performance test) of an existing source as follows:

(A) You must obtain a minimum of three pairs of Method 5 or 5I data, provided in appendix A-3 to part 60 of this chapter, and PMDS data to establish an approximate correlation curve. Data obtained up to 60 months prior to the compliance date may be used provided that the design and operation of the combustor or PMDS has not changed in a manner that may adversely affect the correlation of PM concentrations and PMDS response.

§63.1206

(B) You must request approval from the regulatory authority, in the continuous monitoring system test plan, of your determination whether multiple correlation curves are needed considering the design and operation of your combustor and PMDS.

(C) You must approximate the correlation of the reference method data to the PMDS data.

(1) You may assume a linear correlation of the PMDS response to particulate matter emission concentrations;

(2) You may include a zero point correlation value. To establish a zero point, you must follow one or more of the following steps:

(*i*) Zero point data for in-situ instruments should be obtained, to the extent possible, by removing the instrument from the stack and monitoring ambient air on a test bench;

(*ii*) Zero point data for extractive instruments should be obtained by removing the extractive probe from the stack and drawing in clean ambient air;

(*iii*) Zero point data also can be obtained by performing manual reference method measurements when the flue gas is free of PM emissions or contains very low PM concentrations (e.g., when your process is not operating, but the fans are operating or your source is combusting only natural gas); and

(iv) If none of the steps in paragraphs (c)(9)(ii)(B)(2)(i) through (iii) of this section are possible, you must estimate the monitor response when no PM is in the flue gas (e.g., 4 mA = 0 mg/acm).

(3) For reference method data that were obtained from runs during a test condition where controllable operating factors were held constant, you must average the test run averages of PM concentrations and PMDS responses to obtain a single pair of data for PM concentration and PMDS response. You may use this pair of data and the zero point to define a linear correlation model for the PMDS.

(D) You must establish the alarm setpoint as the PMDS response that corresponds to a PM concentration that is 50% of the PM emission standard or 125% of the highest PM concentration used to develop the correlation, whichever is greater. For reference method data that were obtained from runs dur40 CFR Ch. I (7–1–23 Edition)

ing a test condition where controllable operating factors were held constant, you must use the average of the test run averages of PM concentrations for extrapolating the alarm set-point. The PM emission concentration used to extrapolate the alarm set-point must not exceed the PM emission standard, however.

(iii) Establishing the initial alarm setpoint for operations under the Notification of Compliance. You must establish the initial alarm set-point for operations under the Notification of Compliance as provided by either paragraph (c)(9)(iii)(A) or paragraph (c)(9)(iii)(B)of this section. You must periodically revise the alarm set-point as provided by paragraph (c)(9)(iv) of this section.

(A) Establishing the initial set-point without extrapolation. (I) If you establish the initial alarm set-point without extrapolation, the alarm set-point is the average of the test run averages of the PMDS response during the runs of the comprehensive performance test that document compliance with the PM emission standard.

(2) During the comprehensive performance test, you may simulate PM emission concentrations at the upper end of the range of normal operations by means including feeding high levels of ash and detuning the emission control equipment.

(B) Establishing the initial set-point by extrapolation. You may extrapolate the particulate matter detector response to establish the alarm set-point under the following procedures:

(1) You must request approval from the regulatory authority, in the continuous monitoring system test plan, of the procedures you will use to establish an approximate correlation curve using the three pairs of Method 5 or 5I data (see methods in appendix A-3 of part 60 of this chapter) and PMDS data from the comprehensive performance test, the data pairs used to establish the correlation curve for the Documentation of Compliance under paragraph (c)(9)(ii) of this section, and additional data pairs, as warranted.

(2) You must request approval from the regulatory authority, in the continuous monitoring system test plan,
of your determination of whether multiple correlation curves are needed considering the design and operation of your combustor and PMDS. If so, you must recommend the number of data pairs needed to establish those correlation curves and how the data will be obtained.

(3) During the comprehensive performance test, you may simulate PM emission concentrations at the upper end of the range of normal operations by means including feeding high levels of ash and detuning the emission control equipment.

(4) Data obtained up to 60 months prior to the comprehensive performance test may be used provided that the design and operation of the combustor or PMDS has not changed in a manner that may adversely affect the correlation of PM concentrations and PMDS response.

(5) You may include a zero point correlation value. To establish a zero point, you must follow the procedures under paragraph (c)(9)(ii)(C)(2) of this section.

(6) You must use a least-squares regression model to correlate PM concentrations to PMDS responses for data pairs. You may assume a linear regression model approximates the relationship between PM concentrations and PMDS responses.

(7) You must establish the alarm setpoint as the PMDS response that corresponds to a PM concentration that is 50% of the PM emission standard or 125% of the highest PM concentration used to develop the correlation, whichever is greater. The emission concentration used to extrapolate the PMDS response must not exceed the PM emission standard.

(iv) Revising the Notification of Compliance alarm set-point—(A) Revising setpoints established without extrapolation. If you establish the alarm set-point without extrapolation under paragraph (c)(9)(iii)(A) of this section, you must establish a new alarm set-point in the Notification of Compliance following each comprehensive performance test as the average of the test run averages of the PMDS response during the runs of the comprehensive performance test that document compliance with the PM emission standard. (B) Revising set-points established with extrapolation. If you establish the alarm set-point by extrapolation under paragraph (c)(9)(iii)(B) of this section, you must request approval from the regulatory authority, in the continuous monitoring system test plan, of the procedures for periodically revising the alarm set-point, considering the additional data pairs obtained during periodic comprehensive performance tests and data pairs obtained from other tests, such as for quality assurance.

(v) Quality assurance—(A) Set-points established without extrapolation. If you establish the alarm set-point without extrapolation under paragraph (c)(9)(iii)(A) of this section, you must request approval from the regulatory authority, in the continuous monitoring system test plan, of the quality assurance procedures that reasonably ensure that PMDS response values below the alarm set-point correspond to PM emission concentrations below the average of the PM concentrations demonstrated during the comprehensive performance test. Your recommended quality assurance procedures may include periodic testing under as-found conditions (i.e., normal operations) to obtain additional PM concentration and PMDS response run pairs, as warranted.

(B) Set-points established with extrapolation. If you establish the alarm setpoint by extrapolation under paragraph (c)(9)(iii)(B) of this section, you must request approval from the regulatory authority, in the continuous monitoring system test plan, of the quality assurance procedures that reasonably ensure that PMDS response values below the alarm set-point correspond to PM emission concentrations below the value that correlated to the alarm set-point.

(vi) *PMDS* are used for compliance assurance only. For a PMDS for which the alarm set-point is established by extrapolation using a correlation curve under paragraphs (c)(9)(ii), (c)(9)(iii)(B), and (c)(9)(iv)(B) of this section, an exceedance of the PMDS response that appears to correlate with a PM concentration that exceeds the PM emission standard is not by itself evidence that the standard has been exceeded.

§63.1207

(vii) PMDS corrective measures require*ments*. The operating and maintenance plan required by paragraph (c)(7) of this section must include a corrective measures plan that specifies the procedures you will follow in the case of a PMDS alarm or malfunction. The corrective measures plan must include, at a minimum, the procedures used to determine and record the time and cause of the alarm or PMDS malfunction as well as the corrective measures taken to correct the control device or PMDS malfunction or minimize emissions as specified below. Failure to initiate the corrective measures required by this paragraph is failure to ensure compliance with the emission standards in this subpart.

(A) You must initiate the procedures used to determine the cause of the alarm or PMDS malfunction within 30 minutes of the time the alarm first sounds or the PMDS malfunctions; and

(B) You must alleviate the cause of the alarm or the PMDS malfunction by taking the necessary corrective measure(s) which may include shutting down the combustor.

(viii) Excessive exceedances notification. If you operate the combustor when the detector response exceeds the alarm set-point or when the PMDS is malfunctioning more than 5 percent of the time during any 6-month block time period, you must submit a notification to the Administrator within 30 days of the end of the 6-month block time period that describes the causes of the exceedances and the revisions to the design, operation, or maintenance of the combustor, emission control device, or PMDS you are taking to minimize exceedances. To document compliance with this requirement:

(A) You must keep records of the date, time, and duration of each alarm and PMDS malfunction, the time corrective action was initiated and completed, and a brief description of the cause of the alarm or PMDS malfunction and the corrective action taken:

(B) You must record the percent of the operating time during each 6month period that the alarm sounds and the PMDS malfunctions;

(C) If inspection of the emission control device demonstrates that no cor40 CFR Ch. I (7–1–23 Edition)

rective action is required, then no alarm time is counted; and

(D) If corrective action to the emission control device is required, each alarm shall be counted as a minimum of 1 hour. Each PMDS malfunction shall also be counted as a minimum of 1 hour.

[64 FR 53038, Sept. 30, 1999, as amended at 65
FR 42298, July 10, 2000; 65 FR 67271, Nov. 9,
2000; 66 FR 24272, May 14, 2001; 66 FR 35103,
July 3, 2001; 66 FR 63317, Dec. 7, 2001; 67 FR
6813, Feb. 13, 2002; 67 FR 6989, Feb. 14, 2002; 67
FR 77691, Dec. 19, 2002; 70 FR 59541, Oct. 12,
2005; 70 FR 75047, Dec. 19, 2005; 71 FR 20459,
Apr. 20, 2006; 71 FR 62393, Oct. 25, 2006; 73 FR
18979, Apr. 8, 2008; 73 FR 64094, Oct. 28, 2008]

§63.1207 What are the performance testing requirements?

(a) *General*. The provisions of §63.7 apply, except as noted below.

(b) Types of performance tests—(1) Comprehensive performance test. You must conduct comprehensive performance tests to demonstrate compliance with the emission standards provided by this subpart, establish limits for the operating parameters provided by $\S63.1209$, and demonstrate compliance with the performance specifications for continuous monitoring systems.

(2) Confirmatory performance test. You must conduct confirmatory performance tests to:

(i) Demonstrate compliance with the dioxin/furan emission standard when the source operates under normal operating conditions; and

(ii) Conduct a performance evaluation of continuous monitoring systems required for compliance assurance with the dioxin/furan emission standard under §63.1209(k).

(3) One-Time Dioxin/Furan Test for Sources Not Subject to a Numerical Dioxin/Furan Standard. For solid fuel boilers and hydrochloric acid production furnaces, for lightweight aggregate kilns that are not subject to a numerical dioxin/furan emission standard under §63.1221, and liquid fuel boilers that are not subject to a numerical dioxin/furan emission standard under §63.1217, you must conduct a one-time emission test for dioxin/furan under feed and operating conditions that are most likely to reflect daily maximum operating variability, similar to a

dioxin/furan comprehensive performance test.

(i) You must conduct the dioxin/furan emissions test no later than the deadline for conducting the initial comprehensive performance test.

(ii) You may use dioxin/furan emissions data from previous testing to meet this requirement, provided that:

(A) The testing was conducted under feed and operating conditions that are most likely to reflect daily maximum operating variability, similar to a dioxin/furan compliance test;

(B) You have not changed the design or operation of the source in a manner that could significantly affect stack gas dioxin/furan emission concentrations; and

(C) The data meet quality assurance objectives that may be determined on a site-specific basis.

(iii) You may use dioxin/furan emissions data from a source to represent emissions from another on-site source in lieu of testing (i.e., data in lieu of testing) if the design and operation, including hazardous waste feed and other feedstreams, of the sources are identical.

(iv) You must include the results of the one-time dioxin/furan emissions test with the results of the initial comprehensive performance test in the Notification of Compliance.

(v) You must repeat the dioxin/furan emissions test if you change the design or operation of the source in a manner that may increase dioxin/furan emissions.

(vi) Sources that are required to perform the one-time dioxin/furan test pursuant to paragraph (b)(3) of this section are not required to perform confirmatory performance tests.

(c) Initial comprehensive performance test—(1) Test date. Except as provided by paragraphs (c)(2) and (c)(3) of this section, you must commence the initial comprehensive performance test not later than six months after the compliance date.

(2) Data in lieu of the initial comprehensive performance test. (i) You may request that previous emissions test data serve as documentation of conformance with the emission standards of this subpart provided that the previous testing: (A) Was initiated after 54 months prior to the compliance date, except as provided by paragraphs (c)(2)(iii) or (c)(2)(iv) of this section;

(B) Results in data that meet quality assurance objectives (determined on a site-specific basis) such that the results demonstrate compliance with the applicable standards;

(C) Was in conformance with the requirements of paragraph (g)(1) of this section; and

(D) Was sufficient to establish the applicable operating parameter limits under §63.1209.

(ii) You must submit data in lieu of the initial comprehensive performance test in lieu of (i.e., if the data are in lieu of all performance testing) or with the notification of performance test required under paragraph (e) of this section.

(iii) The data in lieu test age restriction provided in paragraph (c)(2)(i)(A)of this section does not apply for the duration of the interim standards (i.e., the standards published in the FEDERAL REGISTER on February 13, 2002, 67 FR 6792). See 40 CFR parts 63, 264, 265, 266, 270, and 271 revised as of July 1, 2002. Paragraph (c)(2)(i)(A) of this section does not apply until EPA promulgates permanent replacement standards pursuant to the Settlement Agreement noticed in the FEDERAL REGISTER on November 16, 2001 (66 FR 57715).

(iv) The data in lieu test age restriction provided in paragraph (c)(2)(i)(A)of this section does not apply to DRE data provided you do not feed hazardous waste at a location in the combustion system other than the normal flame zone.

(3) For incinerators, cement kilns, and lightweight aggregate kilns, you must commence the initial comprehensive performance test to demonstrate compliance with the standards under §§ 63.1219, 63.1220, and 63.1221 not later than 12 months after the compliance date.

(d) Frequency of testing. Except as otherwise specified in paragraph (d)(4) of this section, you must conduct testing periodically as prescribed in paragraphs (d)(1) through (d)(3) of this section. The date of commencement of the initial comprehensive performance test

is the basis for establishing the deadline to commence the initial confirmatory performance test and the next comprehensive performance test. You may conduct performance testing at any time prior to the required date. The deadline for commencing subsequent confirmatory and comprehensive performance testing is based on the date of commencement of the previous comprehensive performance test. Unless the Administrator grants a time extension under paragraph (i) of this section, you must conduct testing as follows:

(1) Comprehensive performance testing. Except as otherwise specified in paragraph (d)(4) of this section, you must commence testing no later than 61 months after the date of commencing the previous comprehensive performance test used to show compliance with $\S63.1216$, $\S63.1217$, $\S63.1218$, $\S63.1219$, $\S63.1220$, or $\S63.1221$. If you submit data in lieu of the initial performance test, you must commence the subsequent comprehensive performance test within 61 months of commencing the test used to provide the data in lieu of the initial performance test.

(2) Confirmatory performance testing. Except as otherwise specified in paragraph (d)(4) of this section, you must commence confirmatory performance testing no later than 31 months after the date of commencing the previous comprehensive performance test used to show compliance with §63.1217, §63.1219, §63.1220, or §63.1221. If you submit data in lieu of the initial performance test, you must commence the initial confirmatory performance test within 31 months of the date six months after the compliance date. To ensure that the confirmatory test is conducted approximately midway between comprehensive performance tests, the Administrator will not approve a test plan that schedules testing within 18 months of commencing the previous comprehensive performance test.

(3) Duration of testing. You must complete performance testing within 60 days after the date of commencement, unless the Administrator determines that a time extension is warranted based on your documentation in writing of factors beyond your control that 40 CFR Ch. I (7–1–23 Edition)

prevent you from meeting the 60-day deadline.

(4) Applicable testing requirements under the interim standards—(i) Waiver of periodic comprehensive performance tests. Except as provided by paragraph (c)(2) of this section, you must conduct only an initial comprehensive performance test under the interim standards (§§63.1203 through 63.1205); all subsequent comprehensive performance testing requirements are waived under the interim standards. The provisions in the introductory text to paragraph (d) and in paragraph (d)(1) of this section apply only to tests used to demonstrate compliance with the standards under §§ 63.1219 through 63.1221.

(ii) Waiver of confirmatory performance tests. You are not required to conduct a confirmatory test under the interim standards (\S 63.1203 through 63.1205). The confirmatory testing requirements in the introductory text to paragraph (d) and in paragraph (d)(2) of this section apply only after you have demonstrated compliance with the standards under \S 63.1219 through 63.1221.

(e) Notification of performance test and CMS performance evaluation, and approval of test plan and CMS performance evaluation plan. (1) The provisions of §63.7(b) and (c) and §63.8(e) apply, except:

(i) Comprehensive performance test. You must submit to the Administrator a notification of your intention to conduct a comprehensive performance test and CMS performance evaluation and a site-specific test plan and CMS performance evaluation test plan at least one year before the performance test and performance evaluation are scheduled to begin.

(A) The Administrator will notify you of approval or intent to deny approval of the site-specific test plan and CMS performance evaluation test plan within 9 months after receipt of the original plan.

(B) You must submit to the Administrator a notification of your intention to conduct the comprehensive performance test at least 60 calendar days before the test is scheduled to begin.

(ii) Confirmatory performance test. You must submit to the Administrator a notification of your intention to conduct a confirmatory performance test

and CMS performance evaluation and a site-specific test plan and CMS performance evaluation test plan at least 60 calendar days before the performance test is scheduled to begin. The Administrator will notify you of approval or intent to deny approval of the sitespecific test plan and CMS performance evaluation test plan within 30 calendar days after receipt of the original test plans.

(2) You must make your site-specific test plan and CMS performance evaluation test plan available to the public for review no later than 60 calendar days before initiation of the test. You must issue a public notice to all persons on your facility/public mailing list (developed pursuant to 40 CFR 70.7(h), 71.11(d)(3)(i)(E) and 124.10(c)(1)(ix)) announcing the availability of the test plans and the location where the test plans are available for review. The test plans must be accessible to the public for 60 calendar days, beginning on the date that you issue your public notice. The location must be unrestricted and provide access to the public during reasonable hours and provide a means for the public to obtain copies. The notification must include the following information at a minimum:

(i) The name and telephone number of the source's contact person;

(ii) The name and telephone number of the regulatory agency's contact person;

(iii) The location where the test plans and any necessary supporting documentation can be reviewed and copied:

(iv) The time period for which the test plans will be available for public review; and

(v) An expected time period for commencement and completion of the performance test and CMS performance evaluation test.

(3) Petitions for time extension if Administrator fails to approve or deny test plans. You may petition the Administrator under §63.7(h) to obtain a "waiver" of any performance test—initial or periodic performance test; comprehensive or confirmatory test. The "waiver" would be implemented as an extension of time to conduct the performance test at a later date. (i) Qualifications for the waiver. (A) You may not petition the Administrator for a waiver under this section if the Administrator has issued a notification of intent to deny your test plan(s) under $\S63.7(c)(3)(i)(B)$;

(B) You must submit a site-specific emissions testing plan and a continuous monitoring system performance evaluation test plan at least one year before a comprehensive performance test is scheduled to begin as required by paragraph (c)(1) of this section, or at least 60 days before a confirmatory performance test is scheduled to begin as required by paragraph (d) of this section. The test plans must include all required documentation, including the substantive content requirements of paragraph (f) of this section and §63.8(e); and

(C) You must make a good faith effort to accommodate the Administrator's comments on the test plans.

(ii) Procedures for obtaining a waiver and duration of the waiver. (A) You must submit to the Administrator a waiver petition or request to renew the petition under 63.7(h) separately for each source at least 60 days prior to the scheduled date of the performance test;

(B) The Administrator will approve or deny the petition within 30 days of receipt and notify you promptly of the decision;

(C) The Administrator will not approve an individual waiver petition for a duration exceeding 6 months;

(D) The Administrator will include a sunset provision in the waiver ending the waiver within 6 months;

(E) You may submit a revised petition to renew the waiver under §63.7(h)(3)(iii) at least 60 days prior to the end date of the most recently approved waiver petition;

(F) The Administrator may approve a revised petition for a total waiver period up to 12 months.

(iii) Content of the waiver. (A) You must provide documentation to enable the Administrator to determine that the source is meeting the relevant standard(s) on a continuous basis as required by $\S63.7(h)(2)$. For extension requests for the initial comprehensive performance test, you must submit your Documentation of Compliance to

assist the Administrator in making this determination.

(B) You must include in the petition information justifying your request for a waiver, such as the technical or economic infeasibility, or the impracticality, of the affected source performing the required test, as required by §63.7(h)(3)(iii).

(iv) Public notice. At the same time that you submit your petition to the Administrator, you must notify the public (e.g., distribute a notice to the facility/public mailing list developed pursuant to 40 CFR 70.7(h), 71.11(d)(3)(i)(E) and 124.10(c)(1)(ix)) of your petition to waive a performance test. The notification must include all of the following information at a minimum:

(A) The name and telephone number of the source's contact person;

(B) The name and telephone number of the regulatory agency's contact person;

(C) The date the source submitted its site-specific performance test plan and CMS performance evaluation test plans; and

(D) The length of time requested for the waiver.

(f) Content of performance test plan. The provisions of \S 63.7(c)(2)(i)–(iii) and (v) regarding the content of the test plan apply. In addition, you must include the following information in the test plan:

(1) Content of comprehensive performance test plan. (i) An analysis of each feedstream, including hazardous waste, other fuels, and industrial furnace feedstocks, as fired, that includes:

(A) Heating value, levels of ash (for hazardous waste incinerators only), levels of semivolatile metals, low volatile metals, mercury, and total chlorine (organic and inorganic); and

(B) Viscosity or description of the physical form of the feedstream;

(ii) For organic hazardous air pollutants established by 42 U.S.C. 7412(b)(1), excluding caprolactam (CAS number 105602) as provided by §63.60:

(A) Except as provided by paragraph (f)(1)(ii)(D) of this section, an identification of such organic hazardous air pollutants that are present in each hazardous waste feedstream. You need not analyze for organic hazardous air pol-

40 CFR Ch. I (7–1–23 Edition)

lutants that would reasonably not be expected to be found in the feedstream. You must identify any constituents you exclude from analysis and explain the basis for excluding them. You must conduct the feedstream analysis according to §63.1208(b)(8);

(B) An approximate quantification of such identified organic hazardous air pollutants in the hazardous waste feedstreams, within the precision produced by analytical procedures of §63.1208(b)(8); and

(C) A description of blending procedures, if applicable, prior to firing the hazardous waste feedstream, including a detailed analysis of the materials prior to blending, and blending ratios.

(D) The Administrator may approve on a case-by-case basis a hazardous waste feedstream analysis for organic hazardous air pollutants in lieu of the analysis required under paragraph (f)(1)(ii)(A) of this section if the reduced analysis is sufficient to ensure that the POHCs used to demonstrate compliance with the applicable DRE standards of this subpart continue to be representative of the most difficult to destroy organic compounds in your hazardous waste feedstreams;

(iii) A detailed engineering description of the hazardous waste combustor, including:

(A) Manufacturer's name and model number of the hazardous waste combustor;

(B) Type of hazardous waste combustor;

(C) Maximum design capacity in appropriate units;

(D) Description of the feed system for each feedstream;

(E) Capacity of each feed system;

(F) Description of automatic hazardous waste feed cutoff system(s);

(G) Description of the design, operation, and maintenance practices for any air pollution control system; and

(H) Description of the design, operation, and maintenance practices of any stack gas monitoring and pollution control monitoring systems;

(iv) A detailed description of sampling and monitoring procedures including sampling and monitoring locations in the system, the equipment to

be used, sampling and monitoring frequency, and planned analytical procedures for sample analysis;

(v) A detailed test schedule for each hazardous waste for which the performance test is planned, including date(s), duration, quantity of hazardous waste to be burned, and other relevant factors;

(vi) A detailed test protocol, including, for each hazardous waste identified, the ranges of hazardous waste feedrate for each feed system, and, as appropriate, the feedrates of other fuels and feedstocks, and any other relevant parameters that may affect the ability of the hazardous waste combustor to meet the emission standards;

(vii) A description of, and planned operating conditions for, any emission control equipment that will be used;

(viii) Procedures for rapidly stopping the hazardous waste feed and controlling emissions in the event of an equipment malfunction;

(ix) A determination of the hazardous
waste residence time as required by
§63.1206(b)(11);

(x) If you are requesting to extrapolate metal feedrate limits from comprehensive performance test levels under $\S 63.1209(1)(1)(v)$ or 63.1209(n)(2)(vii):

(A) A description of the extrapolation methodology and rationale for how the approach ensures compliance with the emission standards;

(B) Documentation of the historical range of normal (i.e., other than during compliance testing) metals feedrates for each feedstream;

(C) Documentation that the level of spiking recommended during the performance test will mask sampling and analysis imprecision and inaccuracy to the extent that the extrapolated feedrate limits adequately assure compliance with the emission standards;

(xi) If you do not continuously monitor regulated constituents in natural gas, process air feedstreams, and feedstreams from vapor recovery systems under §63.1209(c)(5), you must include documentation of the expected levels of regulated constituents in those feedstreams;

(xii) Documentation justifying the duration of system conditioning required to ensure the combustor has achieved steady-state operations under performance test operating conditions, as provided by paragraph (g)(1)(iii) of this section;

(xiii) For cement kilns with in-line raw mills, if you elect to use the emissions averaging provision of this subpart, you must notify the Administrator of your intent in the initial (and subsequent) comprehensive performance test plan, and provide the information required by the emission averaging provision;

(xiv) For preheater or preheater/ precalciner cement kilns with dual stacks, if you elect to use the emissions averaging provision of this subpart, you must notify the Administrator of your intent in the initial (and subsequent) comprehensive performance test plan, and provide the information required by the emission averaging provision;

(xv) If you request to use Method 23 for dioxin/furan you must provide the information required under §63.1208(b)(1)(i)(B);

(xvi) If you are not required to conduct performance testing to document compliance with the mercury, semivolatile metals, low volatile metals, or hydrogen chloride/chlorine gas emission standards under paragraph (m) of this section, you must include with the comprehensive performance test plan documentation of compliance with the provisions of that section.

(xvii) If you propose to use a surrogate for measuring or monitoring gas flowrate, you must document in the comprehensive performance test plan that the surrogate adequately correlates with gas flowrate, as required by paragraph (m)(7) of this section, and $\S63.1209(j)(2)$, (k)(3), (m)(2)(i), (n)(5)(i), and (o)(2)(i).

(xviii) You must submit an application to request alternative monitoring under 63.1209(g)(1) not later than with the comprehensive performance test plan, as required by 63.1209(g)(1)(iii)(A).

(xix) You must document the temperature location measurement in the comprehensive performance test plan, as required by $\S 63.1209(j)(1)(i)$ and 63.1209(k)(2)(i).

(xx) If your source is equipped with activated carbon injection, you must

document in the comprehensive performance test plan:

(A) The manufacturer specifications for minimum carrier fluid flowrate or pressure drop, as required by §63.1209(k)(6)(ii); and

(B) Key parameters that affect carbon adsorption, and the operating limits you establish for those parameters based on the carbon used during the performance test, if you elect not to specify and use the brand and type of carbon used during the comprehensive performance test, as required by §63.1209(k)(6)(iii).

(xxi) If your source is equipped with a carbon bed system, and you elect not to specify and use the brand and type of carbon used during the comprehensive performance test, you must include in the comprehensive performance test plan key parameters that affect carbon adsorption, and the operating limits you establish for those parameters based on the carbon used during the performance test, as required by §63.1209(k)(7)(ii).

(xxii) If you feed a dioxin/furan inhibitor into the combustion system, you must document in the comprehensive performance test plan key parameters that affect the effectiveness of the inhibitor, and the operating limits you establish for those parameters based on the inhibitor fed during the performance test, if you elect not to specify and use the brand and type of inhibitor used during the comprehensive performance test. as required bv §63.1209(k)(9)(ii).

(xxiii) If your source is equipped with a wet scrubber and you elect to monitor solids content of the scrubber liquid manually but believe that hourly monitoring of solids content is not warranted, you must support an alternative monitoring frequency in the comprehensive performance test plan, as required by $\S63.1209(m)(1)(i)(B)(1)(i)$.

(xxiv) If your source is equipped with a particulate matter control device other than a wet scrubber, baghouse, or electrostatic precipitator, you must include in the comprehensive performance test plan:

(A) Documentation to support the operating parameter limits you establish for the control device, as required by $\S63.1209(m)(1)(iv)(A)(4)$; and 40 CFR Ch. I (7–1–23 Edition)

(B) Support for the use of manufacturer specifications if you recommend such specifications in lieu of basing operating limits on performance test operating levels, as required by $\S63.1209(m)(1)(iv)(D)$.

(xxv) If your source is equipped with a dry scrubber to control hydrogen chloride and chlorine gas, you must document in the comprehensive performance test plan key parameters that affect adsorption, and the limits you establish for those parameters based on the sorbent used during the performance test, if you elect not to specify and use the brand and type of sorbent used during the comprehensive performance test, as required by §63.1209(0)(4)(iii)(A); and

(xxvi) For purposes of calculating semivolatile metal, low volatile metal, mercury, and total chlorine (organic and inorganic), and ash feedrate limits, a description of how you will handle performance test feedstream analytical results that determines these constituents are not present at detectable levels.

(xxvii) Such other information as the Administrator reasonably finds necessary to determine whether to approve the performance test plan.

(2) Content of confirmatory test plan. (i) A description of your normal hydrocarbon or carbon monoxide operating levels, as specified in paragraph (g)(2)(i) of this section, and an explanation of how these normal levels were determined;

(ii) A description of your normal applicable operating parameter levels, as specified in paragraph (g)(2)(ii) of this section, and an explanation of how these normal levels were determined;

(iii) A description of your normal chlorine operating levels, as specified in paragraph (g)(2)(iii) of this section, and an explanation of how these normal levels were determined;

(iv) If you use carbon injection or a carbon bed, a description of your normal cleaning cycle of the particulate matter control device, as specified in paragraph (g)(2)(iv) of this section, and an explanation of how these normal levels were determined;

(v) A detailed description of sampling and monitoring procedures including sampling and monitoring locations in

§63.1207

the system, the equipment to be used, sampling and monitoring frequency, and planned analytical procedures for sample analysis;

(vi) A detailed test schedule for each hazardous waste for which the performance test is planned, including date(s), duration, quantity of hazardous waste to be burned, and other relevant factors;

(vii) A detailed test protocol, including, for each hazardous waste identified, the ranges of hazardous waste feedrate for each feed system, and, as appropriate, the feedrates of other fuels and feedstocks, and any other relevant parameters that may affect the ability of the hazardous waste combustor to meet the dioxin/furan emission standard;

(viii) A description of, and planned operating conditions for, any emission control equipment that will be used;

(ix) Procedures for rapidly stopping the hazardous waste feed and controlling emissions in the event of an equipment malfunction; and

(x) Such other information as the Administrator reasonably finds necessary to determine whether to approve the confirmatory test plan.

(g) Operating conditions during testing. You must comply with the provisions of 63.7(e). Conducting performance testing under operating conditions representative of the extreme range of normal conditions is consistent with the requirement of 63.7(e)(1) to conduct performance testing under representative operating conditions.

(1) Comprehensive performance testing—(i) Operations during testing. For the following parameters, you must operate the combustor during the performance test under normal conditions (or conditions that will result in higher than normal emissions):

(A) *Chlorine feedrate*. You must feed normal (or higher) levels of chlorine during the dioxin/furan performance test;

(B) Ash feedrate. For hazardous waste incinerators, you must conduct the following tests when feeding normal (or higher) levels of ash: The semivolatile metal and low volatile metal performance tests; and the dioxin/furan and mercury performance tests if activated carbon injection or a carbon bed is used; and

(C) Cleaning cycle of the particulate matter control device. You must conduct the following tests when the particulate matter control device undergoes its normal (or more frequent) cleaning cycle: The particulate matter, semivolatile metal, and low volatile metal performance tests; and the dioxin/furan and mercury performance tests if activated carbon injection or a carbon bed is used.

(ii) Modes of operation. Given that you must establish limits for the applicable operating parameters specified in § 63.1209 based on operations during the comprehensive performance test, you may conduct testing under two or more operating modes to provide operating flexibility.

(iii) Steady-state conditions. (A) Prior to obtaining performance test data, you must operate under performance test conditions until you reach steadystate operations with respect to emissions of pollutants you must measure during the performance test and operating parameters under §63.1209 for which you must establish limits. During system conditioning, you must ensure that each operating parameter for which you must establish a limit is held at the level planned for the performance test. You must include documentation in the performance test plan under paragraph (f) of this section justifying the duration of system conditioning.

(B) If you own or operate a hazardous waste cement kiln that recycles collected particulate matter (i.e., cement kiln dust) into the kiln, you must sample and analyze the recycled particulate matter prior to obtaining performance test data for levels of selected metals that must be measured during performance testing to document that the system has reached steady-state conditions (i.e., that metals levels have stabilized). You must document the rationale for selecting metals that are indicative of system equilibrium and include the information in the performance test plan under paragraph (f) of this section. To determine system equilibrium, you must sample and analyze the recycled particulate matter hourly

for each selected metal, unless you submit in the performance test plan a justification for reduced sampling and analysis and the Administrator approves in writing a reduced sampling and analysis frequency.

(2) Confirmatory performance testing. You must conduct confirmatory performance testing for dioxin/furan under normal operating conditions for the following parameters:

(i) Carbon monoxide (or hydrocarbon) CEMS emissions levels must be within the range of the average value to the maximum value allowed, except as provided by paragraph (g)(2)(v) of this section. The average value is defined as the sum of the hourly rolling average values recorded (each minute) over the previous 12 months, divided by the number of rolling averages recorded during that time. The average value must not include calibration data, startup data, shutdown data, malfunction data, and data obtained when not burning hazardous waste;

(ii) Each operating limit (specified in §63.1209) established to maintain compliance with the dioxin/furan emission standard must be held within the range of the average value over the previous 12 months and the maximum or minimum, as appropriate, that is allowed, except as provided by paragraph (g)(2)(v) of this section. The average value is defined as the sum of the rolling average values recorded over the previous 12 months, divided by the number of rolling averages recorded during that time. The average value must not include calibration data, startup data, shutdown data, malfunction data, and data obtained when not burning hazardous waste:

(iii) You must feed chlorine at normal feedrates or greater; and

(iv) If the combustor is equipped with carbon injection or carbon bed, normal cleaning cycle of the particulate matter control device.

(v) The Administrator may approve an alternative range to that required by paragraphs (g)(2)(i) and (ii) of this section if you document in the confirmatory performance test plan that it may be problematic to maintain the required range during the test. In addition, when making the finding of compliance, the Administrator may con40 CFR Ch. I (7–1–23 Edition)

sider test conditions outside of the range specified in the test plan based on a finding that you could not reasonably maintain the range specified in the test plan and considering factors including whether the time duration and level of the parameter when operations were out of the specified range were such that operations during the confirmatory test are determined to be reasonably representative of normal operations. In addition, the Administrator will consider the proximity of the emission test results to the standard.

(h) Operating conditions during subsequent testing. (1) Current operating parameter limits established under §63.1209 are waived during subsequent comprehensive performance testing.

(2) Current operating parameter limits are also waived during pretesting prior to comprehensive performance testing for an aggregate time not to exceed 720 hours of operation (renewable at the discretion of the Administrator) under an approved test plan or if the source records the results of the pretesting. Pretesting means:

(i) Operations when stack emissions testing for dioxin/furan, mercury, semivolatile metals, low volatile metals, particulate matter, or hydrogen chloride/chlorine gas is being performed; and

(ii) Operations to reach steady-state operating conditions prior to stack emissions testing under paragraph (g)(1)(iii) of this section.

(i) Time extension for subsequent performance tests. After the initial comprehensive performance test, you may request up to a one-year time extension for conducting a comprehensive or confirmatory performance test to consolidate performance testing with other state or federally required emission testing, or for other reasons deemed acceptable by the Administrator. If the Administrator grants a time extension for a comprehensive performance test, the deadlines for commencing the next comprehensive and confirmatory tests are based on the date that the subject comprehensive performance test commences.

(1) You must submit in writing to the Administrator any request under this

§63.1207

paragraph for a time extension for conducting a performance test.

(2) You must include in the request for an extension for conducting a performance test the following:

(i) A description of the reasons for requesting the time extension;

(ii) The date by which you will commence performance testing.

(3) The Administrator will notify you in writing of approval or intention to deny approval of your request for an extension for conducting a performance test within 30 calendar days after receipt of sufficient information to evaluate your request. The 30-day approval or denial period will begin after you have been notified in writing that your application is complete. The Administrator will notify you in writing whether the application contains sufficient information to make a determination within 30 calendar days after receipt of the original application and within 30 calendar days after receipt of any supplementary information that you submit.

(4) When notifying you that your application is not complete, the Administrator will specify the information needed to complete the application. The Administrator will also provide notice of opportunity for you to present, in writing, within 30 calendar days after notification of the incomplete application, additional information or arguments to the Administrator to enable further action on the application.

(5) Before denying any request for an extension for performance testing, the Administrator will notify you in writing of the Administrator's intention to issue the denial, together with:

(i) Notice of the information and findings on which the intended denial is based; and

(ii) Notice of opportunity for you to present in writing, within 15 calendar days after notification of the intended denial, additional information or arguments to the Administrator before further action on the request.

(6) The Administrator's final determination to deny any request for an extension will be in writing and will set forth specific grounds upon which the denial is based. The final determination will be made within 30 calendar days after the presentation of additional information or argument (if the application is complete), or within 30 calendar days after the final date specified for the presentation if no presentation is made.

(j) Notification of compliance—(1) Comprehensive performance test. (i) Except as provided by paragraphs (j)(4) and (j)(5) of this section, within 90 days of completion of a comprehensive performance test, you must postmark a Notification of Compliance documenting compliance with the emission standards and continuous monitoring system requirements, and identifying operating parameter limits under §63.1209.

(ii) Upon postmark of the Notification of Compliance, you must comply with all operating requirements specified in the Notification of Compliance in lieu of the limits specified in the Documentation of Compliance required under §63.1211(c).

(2) Confirmatory performance test. Except as provided by paragraph (j)(4) of this section, within 90 days of completion of a confirmatory performance test, you must postmark a Notification of Compliance documenting compliance or noncompliance with the applicable dioxin/furan emission standard.

(3) See §§ 63.7(g), 63.9(h), and 63.1210(d) for additional requirements pertaining to the Notification of Compliance (e.g., you must include results of performance tests in the Notification of Compliance).

(4) *Time extension*. You may submit a written request to the Administrator for a time extension documenting that, for reasons beyond your control, you may not be able to meet the 90-day deadline for submitting the Notification of Compliance after completion of testing. The Administrator will determine whether a time extension is warranted.

(5) Early compliance. If you conduct the initial comprehensive performance test prior to the compliance date, you must postmark the Notification of Compliance within 90 days of completion of the performance test or by the compliance date, whichever is later.

(k) Failure to submit a timely notification of compliance. (1) If you fail to postmark a Notification of Compliance by the specified date, you must cease hazardous waste burning immediately.

(2) Prior to submitting a revised Notification of Compliance as provided by paragraph (k)(3) of this section, you may burn hazardous waste only for the purpose of pretesting or comprehensive performance testing and only for a maximum of 720 hours (renewable at the discretion of the Administrator).

(3) You must submit to the Administrator a Notification of Compliance subsequent to a new comprehensive performance test before resuming hazardous waste burning.

(1) Failure of performance test—(1) Comprehensive performance test. The provisions of this paragraph do not apply to the initial comprehensive performance test if you conduct the test prior to your compliance date.

(i) If you determine (based on CEM recordings, results of analyses of stack samples, or results of CMS performance evaluations) that you have exceeded any emission standard during a comprehensive performance test for a mode of operation, you must cease hazardous waste burning immediately under that mode of operation. You must make this determination within 90 days following completion of the performance test.

(ii) If you have failed to demonstrate compliance with the emission standards for any mode of operation:

(A) Prior to submitting a revised Notification of Compliance as provided by paragraph (1)(1)(ii)(C) of this section, you may burn hazardous waste only for the purpose of pretesting or comprehensive performance testing under revised operating conditions, and only for a maximum of 720 hours (renewable at the discretion of the Administrator), except as provided by paragraph (1)(3)of this section;

(B) You must conduct a comprehensive performance test under revised operating conditions following the requirements for performance testing of this section; and

(C) You must submit to the Administrator a Notification of Compliance subsequent to the new comprehensive performance test.

(2) Confirmatory performance test. If you determine (based on CEM recordings, results of analyses of stack sam40 CFR Ch. I (7–1–23 Edition)

ples, or results of CMS performance evaluations) that you have failed the dioxin/furan emission standard during a confirmatory performance test, you must cease burning hazardous waste immediately. You must make this determination within 90 days following completion of the performance test. To burn hazardous waste in the future:

(i) You must submit to the Administrator for review and approval a test plan to conduct a comprehensive performance test to identify revised limits on the applicable dioxin/furan operating parameters specified in §63.1209(k):

(ii) You must submit to the Administrator a Notification of Compliance with the dioxin/furan emission standard under the provisions of paragraphs (j) and (k) of this section and this paragraph (l). You must include in the Notification of Compliance the revised limits on the applicable dioxin/furan operating parameters specified in §63.1209(k); and

(iii) Until the Notification of Compliance is submitted, you must not burn hazardous waste except for purposes of pretesting or confirmatory performance testing, and for a maximum of 720 hours (renewable at the discretion of the Administrator), except as provided by paragraph (1)(3) of this section.

(3) You may petition the Administrator to obtain written approval to burn hazardous waste in the interim prior to submitting a Notification of Compliance for purposes other than testing or pretesting. You must specify operating requirements, including limits on operating parameters, that you determine will ensure compliance with the emission standards of this subpart based on available information including data from the failed performance test. The Administrator will review, modify as necessary, and approve if warranted the interim operating requirements. An approval of interim operating requirements will include a schedule for submitting a Notification of Compliance.

(m) Waiver of performance test. You are not required to conduct performance tests to document compliance with the mercury, semivolatile metals, low volatile metals, or hydrogen chloride/chlorine gas emission standards

under the conditions specified in paragraphs (m)(1) or (m)(2) of this section. The waiver provisions of this paragraph apply in addition to the provisions of §63.7(h).

(1) Emission standards based on exhaust gas flow rate. (i) You are deemed to be in compliance with an emission standard based on the volumetric flow rate of exhaust gas (i.e., $\mu g/dscm$ or ppmv) if the maximum theoretical emission concentration (MTEC) does not exceed the emission standard over the relevant averaging period specified under §63.1209(1), (n), and (o) of this section for the standard:

(A) Determine the feedrate of mercury, semivolatile metals, low volatile metals, or total chlorine and chloride from all feedstreams;

(B) Determine the stack gas flowrate; and

(C) Calculate a MTEC for each standard assuming all mercury, semivolatile metals, low volatile metals, or total chlorine (organic and inorganic) from all feedstreams is emitted;

(ii) To document compliance with this provision, you must:

(A) Monitor and record the feedrate of mercury, semivolatile metals, low volatile metals, and total chlorine and chloride from all feedstreams according to §63.1209(c);

(B) Monitor with a CMS and record in the operating record the gas flowrate (either directly or by monitoring a surrogate parameter that you have correlated to gas flowrate);

(C) Continuously calculate and record in the operating record the MTEC under the procedures of paragraph (m)(1)(i) of this section; and

(D) Interlock the MTEC calculated in paragraph (m)(1)(i)(C) of this section to the AWFCO system to stop hazardous waste burning when the MTEC exceeds the emission standard.

(iii) In lieu of the requirement in paragraphs (m)(1)(ii)(C) and (D) of this section, you may:

(A) Identify in the Notification of Compliance a minimum gas flowrate limit and a maximum feedrate limit of mercury, semivolatile metals, low volatile metals, and/or total chlorine and chloride from all feedstreams that ensures the MTEC as calculated in paragraph (m)(1)(i)(C) of this section is below the applicable emission standard; and

(B) Interlock the minimum gas flowrate limit and maximum feedrate limit of paragraph (m)(1)(iii)(A) of this section to the AWFCO system to stop hazardous waste burning when the gas flowrate or mercury, semivolatile metals, low volatile metals, and/or total chlorine and chloride feedrate exceeds the limits of paragraph (m)(1)(iii)(A) of this section.

(2) Emission standards based on hazardous waste thermal concentration. (i) You are deemed to be in compliance with an emission standard specified on a hazardous waste thermal concentration basis (i.e., pounds emitted per million Btu of heat input) if the HAP thermal concentration in the waste feed does not exceed the allowable HAP thermal concentration emission rate.

(ii) To document compliance with this provision, you must:

(A) Monitor and record the feedrate of mercury, semivolatile metals, low volatile metals, and total chlorine and chloride from all hazardous waste feedstreams in accordance with §63.1209(c);

(B) Determine and record the higher heating value of each hazardous waste feed;

(C) Continuously calculate and record the thermal feed rate of all hazardous waste feedstreams by summing the products of each hazardous waste feed rate multiplied by the higher heating value of that hazardous waste;

(D) Continuously calculate and record the total HAP thermal feed concentration for each constituent by dividing the HAP feedrate determined in paragraph (m)(2)(ii)(A) of this section by the thermal feed rate determined in paragraph (m)(2)(ii)(C) of this section for all hazardous waste feedstreams;

(E) Interlock the HAP thermal feed concentration for each constituent with the AWFCO to stop hazardous waste feed when the thermal feed concentration exceeds the applicable thermal emission standard.

(3) When you determine the feedrate of mercury, semivolatile metals, low volatile metals, or total chlorine and chloride for purposes of this provision, except as provided by paragraph (m)(4) of this section, you must assume that

§63.1208

the analyte is present at the full detection limit when the feedstream analysis determines that the analyte in not detected in the feedstream.

(4) Owners and operators of hazardous waste burning cement kilns and lightweight aggregate kilns may assume that mercury is present in raw material at half the detection limit when the raw material feedstream analysis determines that mercury is not detected.

(5) You must state in the site-specific test plan that you submit for review and approval under paragraph (e) of this section that you intend to comply with the provisions of this paragraph. You must include in the test plan documentation that any surrogate that is proposed for gas flowrate adequately correlates with the gas flowrate.

[64 FR 53038, Sept. 30, 1999, as amended at 65 FR 42299, July 10, 2000; 65 FR 67271, Nov. 9, 2000; 66 FR 35106, July 3, 2001; 66 FR 63318, Dec. 6, 2001; 67 FR 6814, Feb. 13, 2002; 67 FR 6990, Feb. 14, 2002; 67 FR 77691, Dec. 19, 2002; 70 FR 59546, Oct. 12, 2005; 73 FR 18980, Apr. 8, 2008; 73 FR 64096, Oct. 28, 2008]

§63.1208 What are the test methods?

(a) [Reserved]

(b) *Test methods*. You must use the following test methods to determine compliance with the emissions standards of this subpart:

(1) *Dioxins and furans*. (i) To determine compliance with the emission standard for dioxins and furans, you must use:

(A) Method 0023A, Sampling Method for Polychlorinated Dibenzo-p-Dioxins and Polychlorinated Dibenzofurans emissions from Stationary Sources, EPA Publication SW-846 (incorporated by reference—see §63.14); or

(B) Method 23, provided in Appendix A, Part 60 of this chapter.

(ii) You must sample for a minimum of three hours, and you must collect a minimum sample volume of 2.5 dscm.

(iii) You may assume that nondetects are present at zero concentration.

40 CFR Ch. I (7–1–23 Edition)

(2) *Mercury*. You must use Method 29, provided in appendix A, part 60 of this chapter, to demonstrate compliance with emission standard for mercury.

(3) Cadmium and lead. You must use Method 29, provided in appendix A, part 60 of this chapter, to determine compliance with the emission standard for cadmium and lead (combined).

(4) Arsenic, beryllium, and chromium. You must use Method 29, provided in appendix A, part 60 of this chapter, to determine compliance with the emission standard for arsenic, beryllium, and chromium (combined).

(5) Hydrogen chloride and chlorine gas—(i) Compliance with MACT standards. To determine compliance with the emission standard for hydrogen chloride and chlorine gas (combined), you must use:

(A) Method 26/26A as provided in appendix A, part 60 of this chapter; or

(B) Methods 320 or 321 as provided in appendix A, part 63 of this chapter, or

(C) ASTM D 6735-01, Standard Test Method for Measurement of Gaseous Chlorides and Fluorides from Mineral Calcining Exhaust Sources-Impinger Method to measure emissions of hydrogen chloride, and Method 26/26A to measure emissions of chlorine gas, provided that you follow the provisions in paragraphs (b)(5)(C)(1) through (6) of this section. ASTM D 6735-01 is available for purchase from at least one of the following addresses: American Society for Testing and Materials (ASTM), 100 Barr Harbor Drive, Post Office Box C700, West Conshohocken, PA 19428-2959; or ProQuest, 300 North Zeeb Road, Ann Arbor, MI 48106.

(1) A test must include three or more runs in which a pair of samples is obtained simultaneously for each run according to section 11.2.6 of ASTM Method D6735-01.

(2) You must calculate the test run standard deviation of each set of paired samples to quantify data precision, according to Equation 1 of this section:

$$RSD_{a} = (100) AbsoluteValue\left[\frac{Cl_{a} - C2_{a}}{Cl_{a} + C2_{a}}\right] (Eq. 1)$$

Where:

 RSD_a = The test run relative standard deviation of sample pair a, percent.

 $C1_a$ and $C2_a$ = The HCl concentrations, milligram/dry standard cubic meter (mg/dscm), from the paired samples.

(3) You must calculate the test average relative standard deviation according to Equation 2 of this section:

$$RSD_{TA} = \frac{\sum_{a=1}^{p} RSD_{a}}{p} \qquad (Eq. 2)$$

Where:

 RSD_{TA} = The test average relative standard deviation, percent.

 RSD_a = The test run relative standard deviation for sample pair a.

p = The number of test runs, ≥ 3 .

(4) If RSDTA is greater than 20 percent, the data are invalid and the test must be repeated.

(5) The post-test analyte spike procedure of section 11.2.7 of ASTM Method D6735–01 is conducted, and the percent recovery is calculated according to section 12.6 of ASTM Method D6735–01.

(6) If the percent recovery is between 70 percent and 130 percent, inclusive, the test is valid. If the percent recovery is outside of this range, the data are considered invalid, and the test must be repeated.

(ii) Compliance with risk-based limits under §63.1215. To demonstrate compliance with emission limits established under §63.1215, you must use Method 26/ 26A as provided in appendix A, part 60 of this chapter, Method 320 as provided in appendix A, part 63 of this chapter, Method 321 as provided in appendix A, part 63 of this chapter, or ASTM D 6735-01, Standard Test Method for Measurement of Gaseous Chlorides and Fluorides from Mineral Calcining Exhaust Sources-Impinger Method (following the provisions of paragraphs (b)(5)(C)(1) through (6) of this section), except:

(A) For cement kilns and sources equipped with a dry acid gas scrubber, you must use Methods 320 or 321 as provided in appendix A, part 63 of this chapter, or ASTM D 6735-01 to measure hydrogen chloride, and the back-half, caustic impingers of Method 26/26A as provided in appendix A, part 60 of this chapter to measure chlorine gas; and

(B) For incinerators, boilers, and lightweight aggregate kilns, you must use Methods 320 or 321 as provided in appendix A, part 63 of this chapter, or ASTM D 6735-01 to measure hydrogen chloride, and Method 26/26A as provided in appendix A, part 60 of this chapter to measure total chlorine, and calculate chlorine gas by difference if:

(1) The bromine/chlorine ratio in feedstreams is greater than 5 percent; or

(2) The sulfur/chlorine ratio in feedstreams is greater than 50 percent.

(6) Particulate matter. You must use Methods 5 or 5I, provided in appendix A, part 60 of this chapter, to demonstrate compliance with the emission standard for particulate matter.

(7) Other test methods. You may use applicable test methods in EPA Publication SW-846, as incorporated by reference in paragraph (a) of this section, as necessary to demonstrate compliance with requirements of this subpart, except as otherwise specified in paragraphs (b)(2)-(b)(6) of this section.

(8) Feedstream analytical methods. You may use any reliable analytical method to determine feedstream concentrations of metals, chlorine, and other constituents. It is your responsibility to ensure that the sampling and analysis procedures are unbiased, precise, and that the results are representative of the feedstream.

(9) Opacity. If you determine compliance with the opacity standard under the monitoring requirements of \$ 63.1209(a)(1)(iv) and (a)(1)(v), you must use Method 9, provided in appendix A, part 60 of this chapter.

[64 FR 53038, Sept. 30, 1999, as amended at 69
FR 18803, Apr. 9, 2004; 70 FR 34555, June 14, 2005; 70 FR 59547, Oct. 12, 2005; 87 FR 16673, Mar. 20, 2023]

§63.1209 What are the monitoring requirements?

(a) Continuous emissions monitoring systems (CEMS) and continuous opacity monitoring systems (COMS). (1)(i) You must use either a carbon monoxide or hydrocarbon CEMS to demonstrate and monitor compliance with the carbon monoxide and hydrocarbon standard under this subpart. You must also use an oxygen CEMS to continuously correct the carbon monoxide or hydrocarbon level to 7 percent oxygen.

(ii) (A) Cement kilns under §63.1204. Except as provided by paragraphs (a)(1)(iv) and (a)(1)(v) of the section, you must use a COMS to demonstrate and monitor compliance with the opacity standard under §§63.1204(a)(7) and (b)(7) at each point where emissions are vented from these affected sources including the bypass stack of a preheater or preheater/precalciner kiln with dual stacks.

(B) Cement kilns under §63.1220. Except as provided by paragraphs (a)(1)(iv) and (a)(1)(v) of the section and unless your source is equipped with a bag leak detection system under §63.1206(c)(8) or a particulate matter detection system under §63.1206(c)(9), you must use a COMS to demonstrate and monitor compliance with the opacity standard under §§63.1220(a)(7) and (b)(7) at each point where emissions are vented from these affected sources including the bypass stack of a preheater or preheater/precalciner kiln with dual stacks.

(C) You must maintain and operate each COMS in accordance with the requirements of 63.8(c) except for the requirements under 63.8(c)(3). The requirements of 63.1211(c) shall be complied with instead of 63.8(c)(3); and

(D) Compliance is based on a sixminute block average.

(iii) You must install, calibrate, maintain, and operate a particulate matter CEMS to demonstrate and monitor compliance with the particulate matter standards under this subpart. However, compliance with the requirements in this section to install, calibrate, maintain and operate the PM CEMS is not required until such time that the Agency promulgates all performance specifications and operational requirements applicable to PM CEMS.

(iv) If you operate a cement kiln subject to the provisions of this subpart and use a fabric filter with multiple stacks or an electrostatic precipitator with multiple stacks, you may, in lieu of installing the COMS required by paragraph (a)(1)(ii) of this section, comply with the opacity standard in

40 CFR Ch. I (7–1–23 Edition)

accordance with the procedures of Method 9 to part 60 of this chapter:

(A) You must conduct the Method 9 test while the affected source is operating at the highest load or capacity level reasonably expected to occur within the day;

(B) The duration of the Method 9 test shall be at least 30 minutes each day;

(C) You must use the Method 9 procedures to monitor and record the average opacity for each six-minute block period during the test; and

(D) To remain in compliance, all sixminute block averages must not exceed the opacity standard.

(v) If you operate a cement kiln subject to the provisions of this subpart and use a particulate matter control device that exhausts through a monovent, or if the use of a COMS in accordance with the installation specification of Performance Specification 1 (PS-1) of appendix B to part 60 of this chapter is not feasible, you may, in lieu of installing the COMS required by paragraph (a)(1)(i) of this section, comply with the opacity standard in accordance with the procedures of Method 9 to part 60 of this chapter:

(A) You must conduct the Method 9 test while the affected source is operating at the highest load or capacity level reasonably expected to occur within the day;

(B) The duration of the Method 9 test shall be at least 30 minutes each day;

(C) You must use the Method 9 procedures to monitor and record the average opacity for each six-minute block period during the test; and

(D) To remain in compliance, all sixminute block averages must not exceed the opacity standard.

(2) Performance specifications. You must install, calibrate, maintain, and continuously operate the CEMS and COMS in compliance with the quality assurance procedures provided in the appendix to this subpart and Performance Specifications 1 (opacity), 4B (carbon monoxide and oxygen), and 8A (hydrocarbons) in appendix B, part 60 of this chapter.

(3) Carbon monoxide readings exceeding the span. (i) Except as provided by paragraph (a)(3)(ii) of this section, if a

carbon monoxide CEMS detects a response that results in a one-minute average at or above the 3,000 ppmv span level required by Performance Specification 4B in appendix B, part 60 of this chapter, the one-minute average must be recorded as 10,000 ppmv. The one-minute 10,000 ppmv value must be used for calculating the hourly rolling average carbon monoxide level.

(ii) Carbon monoxide CEMS that use a span value of 10,000 ppmv when oneminute carbon monoxide levels are equal to or exceed 3,000 ppmv are not subject to paragraph (a)(3)(i) of this section. Carbon monoxide CEMS that use a span value of 10,000 are subject to the same CEMS performance and equipment specifications when operating in the range of 3,000 ppmv to 10,000 ppmv that are provided by Performance Specification 4B for other carbon monoxide CEMS, except:

(A) Calibration drift must be less than 300 ppmv; and

(B) Calibration error must be less than 500 ppmv.

(4) Hydrocarbon readings exceeding the span. (i) Except as provided by paragraph (a)(4)(ii) of this section, if a hydrocarbon CEMS detects a response that results in a one-minute average at or above the 100 ppmv span level required by Performance Specification 8A in appendix B, part 60 of this chapter, the one-minute average must be recorded as 500 ppmv. The one-minute 500 ppmv value must be used for calculating the hourly rolling average HC level.

(ii) Hydrocarbon CEMS that use a span value of 500 ppmv when oneminute hydrocarbon levels are equal to or exceed 100 ppmv are not subject to paragraph (a)(4)(i) of this section. Hydrocarbon CEMS that use a span value of 500 ppmv are subject to the same CEMS performance and equipment specifications when operating in the range of 100 ppmv to 500 ppmv that are provided by Performance Specification 8A for other hydrocarbon CEMS, except:

(A) The zero and high-level calibration gas must have a hydrocarbon level of between 0 and 100 ppmv, and between 250 and 450 ppmv, respectively;

(B) The strip chart recorder, computer, or digital recorder must be capable of recording all readings within the CEM measurement range and must have a resolution of 2.5 ppmv;

(C) The CEMS calibration must not differ by more than ± 15 ppmv after each 24-hour period of the seven day test at both zero and high levels;

(D) The calibration error must be no greater than 25 ppmv; and

(E) The zero level, mid-level, and high level calibration gas used to determine calibration error must have a hydrocarbon level of 0-200 ppmv, 150-200 ppmv, and 350-400 ppmv, respectively.

(5) Petitions to use CEMS for other standards. You may petition the Administrator to use CEMS for compliance monitoring for particulate matter, mercury, semivolatile metals, low volatile metals, and hydrogen chloride and chlorine gas under §63.8(f) in lieu of compliance with the corresponding operating parameter limits under this section.

(6) Calculation of rolling averages—(i) Calculation of rolling averages initially. The carbon monoxide or hydrocarbon CEMS must begin recording oneminute average values by 12:01 a.m. and hourly rolling average values by 1:01 a.m., when 60 one-minute values will be available for calculating the initial hourly rolling average for those sources that come into compliance on the regulatory compliance date. Sources that elect to come into compliance before the regulatory compliance date must begin recording oneminute and hourly rolling average values within 60 seconds and 60 minutes (when 60 one-minute values will be available for calculating the initial hourly rolling average), respectively, from the time at which compliance begins.

(ii) Calculation of rolling averages upon intermittent operations. You must ignore periods of time when one-minute values are not available for calculating the hourly rolling average. When oneminute values become available again, the first one-minute value is added to the previous 59 values to calculate the hourly rolling average.

(iii) Calculation of rolling averages when the hazardous waste feed is cutoff.
(A) Except as provided by paragraph
(a)(6)(iii)(B) of this section, you must continue monitoring carbon monoxide and hydrocarbons when the hazardous waste feed is cutoff if the source is operating. You must not resume feeding hazardous waste if the emission levels exceed the standard.

(B) You are not subject to the CEMS requirements of this subpart during periods of time you meet the requirements of §63.1206(b)(1)(ii) (compliance with emissions standards for nonhazardous waste burning sources when you are not burning hazardous waste).

(7) Operating parameter limits for hy*drocarbons*. If you elect to comply with the carbon monoxide and hydrocarbon emission standard by continuously monitoring carbon monoxide with a CEMS, you must demonstrate that hydrocarbon emissions during the comprehensive performance test do not exceed the hydrocarbon emissions standard. In addition, the limits you establish on the destruction and removal efficiency (DRE) operating parameters required under paragraph (j) of this section also ensure that you maintain compliance with the hydrocarbon emission standard. If you do not conduct the hydrocarbon demonstration and DRE tests concurrently, you must establish separate operating parameter limits under paragraph (j) of this section based on each test and the more restrictive of the operating parameter limits applies.

(b) Other continuous monitoring systems (CMS). (1) You must use CMS (e.g., thermocouples, pressure transducers, flow meters) to document compliance with the applicable operating parameter limits under this section.

(2) Except as specified in paragraphs (b)(2)(i) and (ii) of this section, you must install and operate continuous monitoring systems other than CEMS in conformance with §63.8(c)(3) that requires you, at a minimum, to comply with the manufacturer's written specifications or recommendations for installation, operation, and calibration of the system:

(i) Calibration of thermocouples and pyrometers. The calibration of thermocouples must be verified at a frequency and in a manner consistent with manufacturer specifications, but no less frequent than once per year. You must operate and maintain optical

40 CFR Ch. I (7–1–23 Edition)

pyrometers in accordance with manufacturer specifications unless otherwise approved by the Administrator. You must calibrate optical pyrometers in accordance with the frequency and procedures recommended by the manufacturer, but no less frequent than once per year, unless otherwise approved by the Administrator. And,

(ii) Accuracy and calibration of weight measurement devices for activated carbon injection systems. If you operate a carbon injection system, the accuracy of the weight measurement device must be ± 1 percent of the weight being measured. The calibration of the device must be verified at least once each calendar quarter at a frequency of approximately 120 days.

(3) CMS must sample the regulated parameter without interruption, and evaluate the detector response at least once each 15 seconds, and compute and record the average values at least every 60 seconds.

(4) The span of the non-CEMS CMS detector must not be exceeded. You must interlock the span limits into the automatic waste feed cutoff system required by 63.1206(c)(3).

(5) Calculation of rolling averages—(i) Calculation of rolling averages initially. Continuous monitoring systems must begin recording one-minute average values by 12:01 a.m., hourly rolling average values by 1:01 a.m. (e.g., when 60 one-minute values will be available for calculating the initial hourly rolling average), and twelve-hour rolling averages by 12:01 p.m. (e.g., when 720 oneminute averages are available to calculate a 12-hour rolling average), for those sources that come into compliance on the regulatory compliance date. Sources that elect to come into compliance before the regulatory compliance date must begin recording oneminute, hourly rolling average, and 12hour rolling average values within 60 seconds, 60 minutes (when 60 oneminute values will be available for calculating the initial hourly rolling average), and 720 minutes (when 720 oneminute values will be available for calculating the initial 12-hour hourly rolling average) respectively, from the time at which compliance begins.

(ii) Calculation of rolling averages upon intermittent operations. You must ignore

periods of time when one-minute values are not available for calculating rolling averages. When one-minute values become available again, the first one-minute value is added to the previous one-minute values to calculate rolling averages.

(iii) Calculation of rolling averages when the hazardous waste feed is cutoff. (A) Except as provided by paragraph (b)(5)(iii)(B) of this section, you must continue monitoring operating parameter limits with a CMS when the hazardous waste feed is cutoff if the source is operating. You must not resume feeding hazardous waste if an operating parameter exceeds its limit.

(B) You are not subject to the CMS requirements of this subpart during periods of time you meet the requirements of §63.1206(b)(1)(ii) (compliance with emissions standards for nonhazardous waste burning sources when you are not burning hazardous waste).

(c) Analysis of feedstreams—(1) General. Prior to feeding the material, you must obtain an analysis of each feedstream that is sufficient to document compliance with the applicable feedrate limits provided by this section.

(2) *Feedstream analysis plan*. You must develop and implement a feedstream analysis plan and record it in the operating record. The plan must specify at a minimum:

(i) The parameters for which you will analyze each feedstream to ensure compliance with the operating parameter limits of this section;

(ii) Whether you will obtain the analysis by performing sampling and analysis or by other methods, such as using analytical information obtained from others or using other published or documented data or information;

(iii) How you will use the analysis to document compliance with applicable feedrate limits (e.g., if you blend hazardous wastes and obtain analyses of the wastes prior to blending but not of the blended, as-fired, waste, the plan must describe how you will determine the pertinent parameters of the blended waste);

(iv) The test methods which you will use to obtain the analyses;

(v) The sampling method which you will use to obtain a representative

sample of each feedstream to be analyzed using sampling methods described in appendix IX, part 266 of this chapter, or an equivalent method; and

(vi) The frequency with which you will review or repeat the initial analysis of the feedstream to ensure that the analysis is accurate and up to date.

(3) *Review and approval of analysis plan.* You must submit the feedstream analysis plan to the Administrator for review and approval, if requested.

(4) Compliance with feedrate limits. To comply with the applicable feedrate limits of this section, you must monitor and record feedrates as follows:

(i) Determine and record the value of the parameter for each feedstream by sampling and analysis or other method;

(ii) Determine and record the mass or volume flowrate of each feedstream by a CMS. If you determine flowrate of a feedstream by volume, you must determine and record the density of the feedstream by sampling and analysis (unless you report the constituent concentration in units of weight per unit volume (e.g., mg/l)); and

(iii) Calculate and record the mass feedrate of the parameter per unit time.

(5) Waiver of monitoring of constituents in certain feedstreams. You are not required to monitor levels of metals or chlorine in the following feedstreams to document compliance with the feedrate limits under this section provided that you document in the comprehensive performance test plan the expected levels of the constituent in the feedstream and account for those assumed feedrate levels in documenting compliance with feedrate limits: natural gas, process air, and feedstreams from vapor recovery systems.

(d) Performance evaluations. (1) The requirements of §§ 63.8(d) (Quality control program) and (e) (Performance evaluation of continuous monitoring systems) apply, except that you must conduct performance evaluations of components of the CMS under the frequency and procedures (for example, submittal of performance evaluation test plan for review and approval) applicable to performance tests as provided by §63.1207.

§63.1209

(2) You must comply with the quality assurance procedures for CEMS prescribed in the appendix to this subpart.

(e) *Conduct of monitoring*. The provisions of §63.8(b) apply.

(f) Operation and maintenance of continuous monitoring systems. The provisions of §63.8(c) apply except:

(1) Section 63.8(c)(3). The requirements of §63.1211(c), that requires CMSs to be installed, calibrated, and operational on the compliance date, shall be complied with instead of section 63.8(c)(3);

(2) Section 63.8(c)(4)(ii). The performance specifications for carbon monoxide, hydrocarbon, and oxygen CEMSs in subpart B, part 60 of this chapter that requires detectors to measure the sample concentration at least once every 15 seconds for calculating an average emission rate once every 60 seconds shall be complied with instead of section 63.8(c)(4)(ii); and

(3) Sections 63.8(c)(4)(i), (c)(5), and (c)(7)(i)(C) pertaining to COMS apply only to owners and operators of hazardous waste burning cement kilns.

(g) Alternative monitoring requirements other than continuous emissions monitoring systems (CEMS)-(1) Requests to use alternatives to operating parameter monitoring requirements. (i) You may submit an application to the Administrator under this paragraph for approval of alternative operating parameter monitoring requirements to document compliance with the emission standards of this subpart. For requests to use additional CEMS, however, you must use paragraph (a)(5) of this section and §63.8(f). Alternative requests to operating parameter monitoring requirements that include unproven monitoring methods may not be made under this paragraph and must be made under §63.8(f).

(ii) You may submit an application to waive an operating parameter limit specified in this section based on documentation that neither that operating parameter limit nor an alternative operating parameter limit is needed to ensure compliance with the emission standards of this subpart.

(iii) You must comply with the following procedures for applications submitted under paragraphs (g)(1)(i) and (ii) of this section:

40 CFR Ch. I (7–1–23 Edition)

(A) *Timing of the application*. You must submit the application to the Administrator not later than with the comprehensive performance test plan.

(B) *Content of the application*. You must include in the application:

(1) Data or information justifying your request for an alternative monitoring requirement (or for a waiver of an operating parameter limit), such as the technical or economic infeasibility or the impracticality of using the required approach;

(2) A description of the proposed alternative monitoring requirement, including the operating parameter to be monitored, the monitoring approach/ technique (e.g., type of detector, monitoring location), the averaging period for the limit, and how the limit is to be calculated; and

(3) Data or information documenting that the alternative monitoring requirement would provide equivalent or better assurance of compliance with the relevant emission standard, or that it is the monitoring requirement that best assures compliance with the standard and that is technically and economically practicable.

(C) Approval of request to use an alternative monitoring requirement or waive an operating parameter limit. The Administrator will notify you of approval or intention to deny approval of the request within 90 calendar days after receipt of the original request and within 60 calendar days after receipt of any supplementary information that you submit. The Administrator will not approve an alternative monitoring request unless the alternative monitoring requirement provides equivalent or better assurance of compliance with the relevant emission standard, or is the monitoring requirement that best assures compliance with the standard and that is technically and economically practicable. Before disapproving any request, the Administrator will notify you of the Administrator's intention to disapprove the request together with:

(1) Notice of the information and findings on which the intended disapproval is based; and

(2) Notice of opportunity for you to present additional information to the Administrator before final action on

the request. At the time the Administrator notifies you of intention to disapprove the request, the Administrator will specify how much time you will have after being notified of the intended disapproval to submit the additional information.

(D) Responsibility of owners and operators. You are responsible for ensuring that you submit any supplementary and additional information supporting your application in a timely manner to enable the Administrator to consider your application during review of the comprehensive performance test plan. Neither your submittal of an application, nor the Administrator's failure to approve or disapprove the application, relieves you of the responsibility to comply with the provisions of this subpart.

(iv) Dual standards that incorporate the interim standards for HAP metals— (A) Semivolatile and low volatile metals. You may petition the Administrator to waive a feedrate operating parameter limit under paragraph (n)(2) of this section for either the emission standards expressed in a thermal emissions format or the interim standards based on documentation that the feedrate operating parameter limit is not needed to ensure compliance with the relevant standard on a continuous basis.

(B) Mercury. You may petition the Administrator to waive a feedrate operating parameter limit under paragraph (1)(1) of this section for either the feed concentration standard under \$ (3)(3)(2)(1) and (b)(2)(1) or the interim standards based on documentation that the feedrate operating parameter limit is not needed to ensure compliance with the relevant standard on a continuous basis.

(2) Administrator's discretion to specify additional or alternative requirements. The Administrator may determine on a case-by-case basis at any time (e.g., during review of the comprehensive performance test plan, during compliance certification review) that you may need to limit additional or alternative operating parameters (e.g., opacity in addition to or in lieu of operating parameter limits on the particulate matter control device) or that alternative approaches to establish limits on operating parameters may be necessary to document compliance with the emission standards of this subpart.

(h) Reduction of monitoring data. The provisions of 63.8(g) apply.

(i) When an operating parameter is applicable to multiple standards. Paragraphs (j) through (p) of this section require you to establish limits on operating parameters based on comprehensive performance testing to ensure you maintain compliance with the emission standards of this subpart. For several parameters, you must establish a limit for the parameter to ensure compliance with more than one emission standard. An example is a limit on minimum combustion chamber temperature to ensure compliance with both the DRE standard of paragraph (j) of this section and the dioxin/furan standard of paragraph (k) of this section. If the performance tests for such standards are not performed simultaneously, the most stringent limit for a parameter derived from independent performance tests applies.

(j) *DRE*. To remain in compliance with the destruction and removal efficiency (DRE) standard, you must establish operating limits during the comprehensive performance test (or during a previous DRE test under provisions of §63.1206(b)(7)) for the following parameters, unless the limits are based on manufacturer specifications, and comply with those limits at all times that hazardous waste remains in the combustion chamber (i.e., the hazardous waste residence time has not transpired since the hazardous waste feed cutoff system was activated):

(1) Minimum combustion chamber temperature. (i) You must measure the temperature of each combustion chamber at a location that best represents, as practicable, the bulk gas temperature in the combustion zone. You must document the temperature measurement location in the test plan you submit under §63.1207(e);

(ii) You must establish a minimum hourly rolling average limit as the average of the test run averages;

(2) Maximum flue gas flowrate or production rate. (i) As an indicator of gas residence time in the control device, you must establish and comply with a limit on the maximum flue gas flowrate, the maximum production rate, or another parameter that you document in the site-specific test plan as an appropriate surrogate for gas residence time, as the average of the maximum hourly rolling averages for each run.

(ii) You must comply with this limit on a hourly rolling average basis;

(3) Maximum hazardous waste feedrate.(i) You must establish limits on the maximum pumpable and total (i.e., pumpable and nonpumpable) hazardous waste feedrate for each location where hazardous waste is fed.

(ii) You must establish the limits as the average of the maximum hourly rolling averages for each run.

(iii) You must comply with the feedrate limit(s) on a hourly rolling average basis;

(4) Operation of waste firing system. You must specify operating parameters and limits to ensure that good operation of each hazardous waste firing system is maintained.

(k) Dioxins and furans. You must comply with the dioxin and furans emission standard by establishing and complying with the following operating parameter limits. You must base the limits on operations during the comprehensive performance test, unless the limits are based on manufacturer specifications.

(1) Gas temperature at the inlet to a dry particulate matter control device. (i) For sources other than a lightweight aggregate kiln, if the combustor is equipped with an electrostatic precipitator, baghouse (fabric filter), or other dry emissions control device where particulate matter is suspended in contact with combustion gas, you must establish a limit on the maximum temperature of the gas at the inlet to the device on an hourly rolling average. You must establish the hourly rolling average limit as the average of the test run averages.

(ii) For hazardous waste burning lightweight aggregate kilns, you must establish a limit on the maximum temperature of the gas at the exit of the (last) combustion chamber (or exit of any waste heat recovery system) on an hourly rolling average. The limit must be established as the average of the test run averages:

40 CFR Ch. I (7–1–23 Edition)

(2) Minimum combustion chamber temperature. (i) For sources other than cement kilns, you must measure the temperature of each combustion chamber at a location that best represents, as practicable, the bulk gas temperature in the combustion zone. You must document the temperature measurement location in the test plan you submit under §§ 63.1207(e) and (f);

(ii) You must establish a minimum hourly rolling average limit as the average of the test run averages.

(3) Maximum flue gas flowrate or production rate. (i) As an indicator of gas residence time in the control device, you must establish and comply with a limit on the maximum flue gas flowrate, the maximum production rate, or another parameter that you document in the site-specific test plan as an appropriate surrogate for gas residence time, as the average of the maximum hourly rolling averages for each run.

(ii) You must comply with this limit on a hourly rolling average basis;

(4) Maximum hazardous waste feedrate. (i) You must establish limits on the maximum pumpable and total (pumpable and nonpumpable) hazardous waste feedrate for each location where waste is fed.

(ii) You must establish the limits as the average of the maximum hourly rolling averages for each run.

(iii) You must comply with the feedrate limit(s) on a hourly rolling average basis;

(5) Particulate matter operating limit. If your combustor is equipped with an activated carbon injection system, you must establish operating parameter limits on the particulate matter control device as specified by paragraph (m)(1) of this section;

(6) Activated carbon injection parameter limits. If your combustor is equipped with an activated carbon injection system:

(i) Carbon feedrate. You must establish a limit on minimum carbon injection rate on an hourly rolling average calculated as the average of the test run averages. If your carbon injection system injects carbon at more than one location, you must establish a carbon feedrate limit for each location.

(ii) Carrier fluid. You must establish a limit on minimum carrier fluid (gas or liquid) flowrate or pressure drop as an hourly rolling average based on the manufacturer's specifications. You must document the specifications in the test plan you submit under §§ 63.1207(e) and (f);

(iii) Carbon specification. (A) You must specify and use the brand (i.e., manufacturer) and type of carbon used during the comprehensive performance test until a subsequent comprehensive performance test is conducted, unless you document in the site-specific performance test plan required under §§ 63.1207(e) and (f) key parameters that affect adsorption and establish limits on those parameters based on the carbon used in the performance test.

(B) You may substitute at any time a different brand or type of carbon provided that the replacement has equivalent or improved properties compared to the carbon used in the performance test and conforms to the key sorbent parameters you identify under paragraph (k)(6)(ii)(A) of this section. You must include in the operating record documentation that the substitute carbon will provide the same level of control as the original carbon.

(7) Carbon bed parameter limits. If your combustor is equipped with a carbon bed system:

(i) Monitoring bed life. You must:

(A) Monitor performance of the carbon bed consistent with manufacturer's specifications and recommendations to ensure the carbon bed (or bed segment for sources with multiple segments) has not reached the end of its useful life to minimize dioxin/furan and mercury emissions at least to the levels required by the emission standards;

(B) Document the monitoring procedures in the operation and maintenance plan;

(C) Record results of the performance monitoring in the operating record; and

(D) Replace the bed or bed segment before it has reached the end of its useful life to minimize dioxin/furan and mercury emissions at least to the levels required by the emission standards.

(ii) Carbon specification. (A) You must specify and use the brand (i.e., manufacturer) and type of carbon used dur-

ing the comprehensive performance test until a subsequent comprehensive performance test is conducted, unless you document in the site-specific performance test plan required under §§ 63.1207(e) and (f) key parameters that affect adsorption and establish limits on those parameters based on the carbon used in the performance test.

(B) You may substitute at any time a different brand or type of carbon provided that the replacement has equivalent or improved properties compared to the carbon used in the performance test. You must include in the operating record documentation that the substitute carbon will provide an equivalent or improved level of control as the original carbon.

(iii) Maximum temperature. You must measure the temperature of the carbon bed at either the bed inlet or exit and you must establish a maximum temperature limit on an hourly rolling average as the average of the test run averages.

(8) Catalytic oxidizer parameter limits. If your combustor is equipped with a catalytic oxidizer, you must establish limits on the following parameters:

(i) Minimum flue gas temperature at the entrance of the catalyst. You must establish a limit on minimum flue gas temperature at the entrance of the catalyst on an hourly rolling average as the average of the test run averages.

(ii) *Maximum time in-use*. You must replace a catalytic oxidizer with a new catalytic oxidizer when it has reached the maximum service time specified by the manufacturer.

(iii) Catalyst replacement specifications. When you replace a catalyst with a new one, the new catalyst must be equivalent to or better than the one used during the previous comprehensive test, as measured by:

(A) Catalytic metal loading for each metal;

(B) Space time, expressed in the units s^{-1} , the maximum rated volumetric flow of combustion gas through the catalyst divided by the volume of the catalyst; and

(C) Substrate construction, including materials of construction, washcoat type, and pore density.

(iv) Maximum flue gas temperature. You must establish a maximum flue gas temperature limit at the entrance of the catalyst as an hourly rolling average, based on manufacturer's specifications.

(9) Inhibitor feedrate parameter limits. If you feed a dioxin/furan inhibitor into the combustion system, you must establish limits for the following parameters:

(i) *Minimum inhibitor feedrate*. You must establish a limit on minimum inhibitor feedrate on an hourly rolling average as the average of the test run averages.

(ii) Inhibitor specifications. (A) You must specify and use the brand (i.e., manufacturer) and type of inhibitor used during the comprehensive performance test until a subsequent comprehensive performance test is conducted, unless you document in the site-specific performance test plan required under §§ 63.1207(e) and (f) key parameters that affect the effectiveness of the inhibitor and establish limits on those parameters based on the inhibitor used in the performance test.

(B) You may substitute at any time a different brand or type of inhibitor provided that the replacement has equivalent or improved properties compared to the inhibitor used in the performance test and conforms to the key parameters you identify under paragraph (k)(9)(ii)(A) of this section. You must include in the operating record documentation that the substitute inhibitor will provide the same level of control as the original inhibitor.

(1) *Mercury*. You must comply with the mercury emission standard by establishing and complying with the following operating parameter limits. You must base the limits on operations during the comprehensive performance test, unless the limits are based on manufacturer specifications.

(1) Feedrate of mercury. (i) For incinerators and solid fuel boilers, when complying with the mercury emission standards under \S 63.1203, 63.1216 and 63.1219, you must establish a 12-hour rolling average limit for the total feedrate of mercury in all feedstreams as the average of the test run averages.

(ii) For liquid fuel boilers, when complying with the mercury emission standards of §63.1217, you must establish a rolling average limit for the mer40 CFR Ch. I (7–1–23 Edition)

cury feedrate as follows on an averaging period not to exceed an annual rolling average:

(A) You must calculate a mercury system removal efficiency for each test run and calculate the average system removal efficiency of the test run averages. If emissions exceed the mercury emission standard during the comprehensive performance test, it is not a violation because the averaging period for the mercury emission standard is (not-to-exceed) one year and compliance is based on compliance with the mercury feedrate limit with an averaging period not-to-exceed one year.

(B) If you burn hazardous waste with a heating value of 10,000 Btu/lb or greater, you must calculate the mercury feedrate limit as follows:

(1) The mercury feedrate limit is the emission standard divided by [1 - system removal efficiency].

(2) The mercury feedrate limit is a hazardous waste thermal concentration limit expressed as pounds of mercury in hazardous waste feedstreams per million Btu of hazardous waste fired.

(3) You must comply with the hazardous waste mercury thermal concentration limit by determining the feedrate of mercury in all hazardous waste feedstreams (lb/hr) at least once a minute and the hazardous waste thermal feedrate (MM Btu/hr) at least once a minute to calculate a 60-minute average thermal emission concentration as [hazardous waste mercury feedrate (lb/ hr) / hazardous waste thermal feedrate (MM Btu/hr)].

(4) You must calculate a rolling average hazardous waste mercury thermal concentration that is updated each hour.

(5) If you select an averaging period for the feedrate limit that is greater than a 12-hour rolling average, you must calculate the initial rolling average as though you had selected a 12hour rolling average, as provided by paragraph (b)(5)(i) of this section. Thereafter, you must calculate rolling averages using either one-minute or one-hour updates. Hourly updates shall be calculated using the average of the one-minute average data for the preceding hour. For the period beginning with initial operation under this standard until the source has operated for

§63.1209

the full averaging period that you select, the average feedrate shall be based only on actual operation under this standard.

(C) If you burn hazardous waste with a heating value of less than 10,000 Btu/ lb, you must calculate the mercury feedrate limit as follows:

(1) You must calculate the mercury feedrate limit as the mercury emission standard divided by [1 – System Removal Efficiency].

(2) The feedrate limit is expressed as a mass concentration per unit volume of stack gas (μ gm/dscm) and is converted to a mass feedrate (lb/hr) by multiplying it by the average stack gas flowrate of the test run averages.

(3) You must comply with the feedrate limit by determining the mercury feedrate (lb/hr) at least once a minute to calculate a 60-minute average feedrate.

(4) You must update the rolling average feedrate each hour with this 60minute feedrate measurement.

(5) If you select an averaging period for the feedrate limit that is greater than a 12-hour rolling average, you must calculate the initial rolling average as though you had selected a 12hour rolling average, as provided by paragraph (b)(5)(i) of this section. Thereafter, you must calculate rolling averages using either one-minute or one-hour updates. Hourly updates shall be calculated using the average of the one-minute average data for the preceding hour. For the period beginning with initial operation under this standard until the source has operated for the full averaging period that you select, the average feedrate shall be based only on actual operation under this standard.

(D) If your boiler is equipped with a wet scrubber, you must comply with the following unless you document in the performance test plan that you do not feed chlorine at rates that may substantially affect the system removal efficiency of mercury for purposes of establishing a mercury feedrate limit based on the system removal efficiency during the test:

(1) Scrubber blowdown must be minimized during a pretest conditioning period and during the performance test: (2) Scrubber water must be preconditioned so that mercury in the water is at equilibrium with stack gas at the mercury feedrate level of the performance test; and

(3) You must establish an operating limit on minimum pH of scrubber water as the average of the test run averages and comply with the limit on an hourly rolling average.

(iii) For cement kilns:

(A) When complying with the emission standards under §§63.1220(a)(2)(i) and (b)(2)(i), you must:

(1) Comply with the mercury hazardous waste feed concentration operating requirement on a twelve-hour rolling average;

(2) Monitor and record in the operating record the as-fired mercury concentration in the hazardous waste (or the weighted-average mercury concentration for multiple hazardous waste feedstreams);

(3) Initiate an automatic waste feed cutoff that immediately and automatically cuts off the hazardous waste feed when the as-fired mercury concentration operating requirement is exceeded;

(B) When complying with the emission standards under \S 63.1204 and 63.1220(a)(2)(ii)(A) and (b)(2)(ii)(A), you must establish a 12-hour rolling average limit for the feedrate of mercury in all feedstreams as the average of the test run averages;

(C) Except as provided by paragraph (1)(1)(iii)(D) of this section, when complying with the hazardous waste maximum theoretical emission concentration (MTEC) under §63.1220(a)(2)(ii)(B) and (b)(2)(ii)(B), you must:

(1) Comply with the MTEC operating requirement on a twelve-hour rolling average;

(2) Monitor and record the feedrate of mercury for each hazardous waste feedstream according to §63.1209(c);

(3) Monitor with a CMS and record in the operating record the gas flowrate (either directly or by monitoring a surrogate parameter that you have correlated to gas flowrate);

(4) Continuously calculate and record in the operating record a MTEC assuming mercury from all hazardous waste feedstreams is emitted;

§63.1209

40 CFR Ch. I (7–1–23 Edition)

(5) Initiate an automatic waste feed cutoff that immediately and automatically cuts off the hazardous waste feed when the MTEC operating requirement is exceeded:

(D) In lieu of complying with paragraph (1)(1)(iii)(C) of this section, you may:

(1) Identify in the Notification of Compliance a minimum gas flowrate limit and a maximum feedrate limit of mercury from all hazardous waste feedstreams that ensures the MTEC calculated in paragraph (1)(1)(iii)(C)(4)of this section is below the operating requirement under paragraphs §§ 63.1220(a)(2)(ii)(B) and (b)(2)(ii)(B); and

(2) Initiate an automatic waste feed cutoff that immediately and automatically cuts off the hazardous waste feed when either the gas flowrate or mercury feedrate exceeds the limits identified in paragraph (1)(1)(iii)(D)(I) of this section.

 $(\ensuremath{\text{iv}})$ For lightweight aggregate kilns:

(A) When complying with the emission standards under \S 63.1205, 63.1221(a)(2)(i) and (b)(2)(i), you must establish a 12-hour rolling average limit for the total feedrate of mercury in all feedstreams as the average of the test run averages;

(B) Except as provided by paragraph (1)(1)(iv)(C) of this section, when complying with the hazardous waste feedrate corresponding to a maximum theoretical emission concentration (MTEC) under §§ 63.1221(a)(2)(ii) and (b)(2)(ii), you must:

(1) Comply with the MTEC operating requirement on a twelve-hour rolling average;

(2) Monitor and record the feedrate of mercury for each hazardous waste feedstream according to §63.1209(c);

(3) Monitor with a CMS and record in the operating record the gas flowrate (either directly or by monitoring a surrogate parameter that you have correlated to gas flowrate);

(4) Continuously calculate and record in the operating record a MTEC assuming mercury from all hazardous waste feedstreams is emitted;

(5) Initiate an automatic waste feed cutoff that immediately and automatically cuts off the hazardous waste feed when the MTEC operating requirement is exceeded;

(C) In lieu of complying with paragraph (l)(l)(iv)(B) of this section, you may:

(1) Identify in the Notification of Compliance a minimum gas flowrate limit and a maximum feedrate limit of mercury from all hazardous waste feedstreams that ensures the MTEC calculated in paragraph (1)(1)(iv)(B)(4)of this section is below the operating requirement under paragraphs §§ 63.1221(a)(2)(ii) and (b)(2)(ii); and

(2) Initiate an automatic waste feed cutoff that immediately and automatically cuts off the hazardous waste feed when either the gas flowrate or mercury feedrate exceeds the limits identified in paragraph (1)(1)(iv)(C)(1) of this section.

(v) Extrapolation of feedrate levels. In lieu of establishing mercury feedrate limits as specified in paragraphs (l)(1)(i) through (iv) of this section, you may request as part of the performance test plan under §§63.7(b) and (c) and §§63.1207 (e) and (f) to use the mercury feedrates and associated emission rates during the comprehensive performance test to extrapolate to higher allowable feedrate limits and emission rates. The extrapolation methodology will be reviewed and approved, as warranted, by the Administrator. The review will consider in particular whether:

(A) Performance test metal feedrates are appropriate (i.e., whether feedrates are at least at normal levels; depending on the heterogeneity of the waste, whether some level of spiking would be appropriate; and whether the physical form and species of spiked material is appropriate); and

(B) Whether the extrapolated feedrates you request are warranted considering historical metal feedrate data.

(2) Wet scrubber. If your combustor is equipped with a wet scrubber, you must establish operating parameter limits prescribed by paragraph (o)(3) of this section, except for paragraph (o)(3)(iv).

(3) Activated carbon injection. If your combustor is equipped with an activated carbon injection system, you must establish operating parameter

§63.1209

limits prescribed by paragraphs (k)(5) and (k)(6) of this section.

(4) Activated carbon bed. If your combustor is equipped with an activated carbon bed system, you must comply with the requirements of (k)(7) of this section to assure compliance with the mercury emission standard.

(m) Particulate matter. You must comply with the particulate matter emission standard by establishing and complying with the following operating parameter limits. You must base the limits on operations during the comprehensive performance test, unless the limits are based on manufacturer specifications.

(1) Control device operating parameter limits (OPLs)—(i) Wet scrubbers. For sources equipped with wet scrubbers, including ionizing wet scrubbers, high energy wet scrubbers such as venturi, hydrosonic, collision, or free jet wet scrubbers, and low energy wet scrubbers such as spray towers, packed beds, or tray towers, you must establish limits on the following parameters:

(A) For high energy scrubbers only, minimum pressure drop across the wet scrubber on an hourly rolling average, established as the average of the test run averages;

(B) For all wet scrubbers:

(1) To ensure that the solids content of the scrubber liquid does not exceed levels during the performance test, you must either:

(i) Establish a limit on solids content of the scrubber liquid using a CMS or by manual sampling and analysis. If you elect to monitor solids content manually, you must sample and analyze the scrubber liquid hourly unless you support an alternative monitoring frequency in the performance test plan that you submit for review and approval; or

(*ii*) Establish a minimum blowdown rate using a CMS and either a minimum scrubber tank volume or liquid level using a CMS.

(2) For maximum solids content monitored with a CMS, you must establish a limit on a twelve-hour rolling average as the average of the test run averages.

(3) For maximum solids content measured manually, you must establish an hourly limit, as measured at least once per hour, unless you support an alternative monitoring frequency in the performance test plan that you submit for review and approval. You must establish the maximum hourly limit as the average of the manual measurement averages for each run.

(4) For minimum blowdown rate and either a minimum scrubber tank volume or liquid level using a CMS, you must establish a limit on an hourly rolling average as the average of the test run averages.

(C) For high energy wet scrubbers only, you must establish limits on either the minimum liquid to gas ratio or the minimum scrubber water flowrate and maximum flue gas flowrate on an hourly rolling average. If you establish limits on maximum flue gas flowrate under this paragraph, you need not establish a limit on maximum flue gas flowrate under paragraph (m)(2) of this section. You must establish these hourly rolling average limits as the average of the test run averages; and

(ii)-(iii) [Reserved]

(iv) Other particulate matter control devices. For each particulate matter control device that is not a fabric filter or high energy wet scrubber, or is not an electrostatic precipitator or ionizing wet scrubber for which you elect to monitor particulate matter loadings under \S 63.1206(c)(9) of this chapter for process control, you must ensure that the control device is properly operated and maintained as required by \S 63.1206(c)(7) and by monitoring the operation of the control device as follows:

(A) During each comprehensive performance test conducted to demonstrate compliance with the particulate matter emissions standard, you must establish a range of operating values for the control device that is a representative and reliable indicator that the control device is operating within the same range of conditions as during the performance test. You must establish this range of operating values as follows:

(1) You must select a set of operating parameters appropriate for the control device design that you determine to be a representative and reliable indicator of the control device performance.

§63.1209

(2) You must measure and record values for each of the selected operating parameters during each test run of the performance test. A value for each selected parameter must be recorded using a continuous monitor.

(3) For each selected operating parameter measured in accordance with the requirements of paragraph (m)(1)(iv)(A)(I) of this section, you must establish a minimum operating parameter limit or a maximum operating parameter limit, as appropriate for the parameter, to define the operating limits within which the control device can operate and still continuously achieve the same operating conditions as during the performance test.

(4) You must prepare written documentation to support the operating parameter limits established for the control device and you must include this documentation in the performance test plan that you submit for review and approval. This documentation must include a description for each selected parameter and the operating range and monitoring frequency required to ensure the control device is being properly operated and maintained.

(B) You must install, calibrate, operate, and maintain a monitoring device equipped with a recorder to measure the values for each operating parameter selected in accordance with the requirements paragraph of (m)(1)(iv)(A)(1) of this section. You must install, calibrate, and maintain the monitoring equipment in accordance with the equipment manufacturer's specifications. The recorder must record the detector responses at least every 60 seconds, as required in the definition of continuous monitor.

(C) You must regularly inspect the data recorded by the operating parameter monitoring system at a sufficient frequency to ensure the control device is operating properly. An excursion is determined to have occurred any time that the actual value of a selected operating parameter is less than the minimum operating limit (or, if applicable, greater than the maximum operating limit) established for the parameter in accordance with the requirements of paragraph (m)(1)(iv)(A)(3) of this section.

40 CFR Ch. I (7–1–23 Edition)

(D) Operating parameters selected in accordance with paragraph (m)(1)(iv) of this section may be based on manufacturer specifications provided you support the use of manufacturer specifications in the performance test plan that you submit for review and approval.

(2) Maximum flue gas flowrate or production rate. (i) As an indicator of gas residence time in the control device, you must establish a limit on the maximum flue gas flowrate, the maximum production rate, or another parameter that you document in the site-specific test plan as an appropriate surrogate for gas residence time, as the average of the maximum hourly rolling averages for each run.

(ii) You must comply with this limit on a hourly rolling average basis;

(3) Maximum ash feedrate. Owners and operators of hazardous waste incinerators, solid fuel boilers, and liquid fuel boilers must establish a maximum ash feedrate limit as a 12-hour rolling average based on the average of the test run averages. This requirement is waived, however, if you comply with the particulate matter detection system requirements under §63.1206(c)(9).

(n) Semivolatile metals and low volatility metals. You must comply with the semivolatile metal (cadmium and lead) and low volatile metal (arsenic, beryllium, and chromium) emission standards by establishing and complying with the following operating parameter limits. You must base the limits on operations during the comprehensive performance test, unless the limits are based on manufacturer specifications.

(1) Maximum inlet temperature to dry particulate matter air pollution control device. You must establish a limit on the maximum inlet temperature to the primary dry metals emissions control device (e.g., electrostatic precipitator, baghouse) on an hourly rolling average basis as the average of the test run averages.

(2) Maximum feedrate of semivolatile and low volatile metals—(i) General. You must establish feedrate limits for semivolatile metals (cadmium and lead) and low volatile metals (arsenic, beryllium, and chromium) as follows, except as provided by paragraph (n)(2)(vii) of this section.

(ii) For incinerators, cement kilns, and lightweight aggregate kilns, when complying with the emission standards under §§63.1203, 63.1204, 63.1205, and 63.1219, and for solid fuel boilers when complying with the emission standards under §63.1216, you must establish 12hour rolling average limits for the total feedrate of semivolatile and low volatile metals in all feedstreams as the average of the test run averages.

(iii) Cement kilns under §63.1220. (A) When complying with the emission §63.1220(a)(3)(i), standards under (a)(4)(i), (b)(3)(i), and (b)(4)(i), you must establish 12-hour rolling average feedrate limits for semivolatile and low volatile metals as the thermal concentration of semivolatile metals or low volatile metals in all hazardous waste feedstreams. You must calculate hazardous waste thermal concentrations for semivolatile metals and low volatile metals for each run as the total mass feedrate of semivolatile metals or low volatile metals for all hazardous waste feedstreams divided by the total heat input rate for all hazardous waste feedstreams. The 12-hour rolling average feedrate limits for semivolatile metals and low volatile metals are the average of the test run averages, calculated on a thermal concentration basis, for all hazardous waste feeds.

(B) When complying with the emission standards under \$ 63.1220(a)(3)(ii), (a)(4)(ii), (b)(3)(ii), and (b)(4)(ii), you must establish 12-hour rolling average limits for the total feedrate of semivolatile and low volatile metals in all feedstreams as the average of the test run averages.

(iv) Lightweight aggregate kilns under §63.1221. (A) When complying with the emission standards under §§63.1221(a)(3)(i), (a)(4)(i), (b)(3)(i), and (b)(4)(i), you must establish 12-hour rolling average feedrate limits for semivolatile and low volatile metals as thermal concentration the of semivolatile metals or low volatile metals in all hazardous waste feedstreams as specified in paragraphs (n)(2)(iii)(A) of this section.

(B) When complying with the emission standards under \S 63.1221(a)(3)(ii), (a)(4)(ii), (b)(3)(ii), and (b)(4)(ii), you must establish 12-hour rolling average

limits for the total feedrate of semivolatile and low volatile metals in all feedstreams as the average of the test run averages.

(v) Liquid fuel boilers under §63.1217—
(A) Semivolatile metals. You must establish a rolling average limit for the semivolatile metal feedrate as follows on an averaging period not to exceed an annual rolling average.

(1) System removal efficiency. You must calculate a semivolatile metal system removal efficiency for each test run and calculate the average system removal efficiency of the test run averages. If emissions exceed the semivolatile metal emission standard during the comprehensive performance test, it is not a violation because the averaging period for the semivolatile metal emission standard is one year and compliance is based on compliance with the semivolatile metal feedrate limit that has an averaging period not to exceed an annual rolling average.

(2) Boilers that feed hazardous waste with a heating value of 10,000 Btu/lb or greater. You must calculate the semivolatile metal feedrate limit as the semivolatile metal emission standard divided by [1 – System Removal Efficiency].

(i) The feedrate limit is a hazardous waste thermal concentration limit expressed as pounds of semivolatile metals in all hazardous waste feedstreams per million Btu of hazardous waste fed to the boiler.

(ii) You must comply with the hazardous waste semivolatile metal thermal concentration limit by determining the feedrate of semivolatile all hazardous metal in waste feedstreams (lb/hr) and the hazardous waste thermal feedrate (MM Btu/hr) at least once a minute to calculate a 60minute average thermal emission concentration as [hazardous waste semivolatile metal feedrate (lb/hr) / hazardous waste thermal feedrate (MM Btu/hr)].

(*iii*) You must calculate a rolling average hazardous waste semivolatile metal thermal concentration that is updated each hour.

(*iv*) If you select an averaging period for the feedrate limit that is greater than a 12-hour rolling average, you must calculate the initial rolling average as though you had selected a 12hour rolling average, as provided by paragraph (b)(5)(i) of this section. Thereafter, you must calculate rolling averages using either one-minute or one-hour updates. Hourly updates shall be calculated using the average of the one-minute average data for the preceding hour. For the period beginning with initial operation under this standard until the source has operated for the full averaging period that you select, the average feedrate shall be based only on actual operation under this standard.

(3) Boilers that feed hazardous waste with a heating value less than 10,000 Btu/ lb. (i) You must calculate the semivolatile metal feedrate limit as the semivolatile metal emission standard divided by [1 - System Removal Efficiency].

(*ii*) The feedrate limit is expressed as a mass concentration per unit volume of stack gas (μ gm/dscm) and is converted to a mass feedrate (lb/hr) by multiplying it by the average stack gas flowrate (dscm/hr) of the test run averages.

(*iii*) You must comply with the feedrate limit by determining the semivolatile metal feedrate (lb/hr) at least once a minute to calculate a 60-minute average feedrate.

(*iv*) You must update the rolling average feedrate each hour with this 60-minute feedrate measurement.

(v) If you select an averaging period for the feedrate limit that is greater than a 12-hour rolling average, you must calculate the initial rolling average as though you had selected a 12hour rolling average, as provided by paragraph (b)(5)(i) of this section. Thereafter, you must calculate rolling averages using either one-minute or one-hour updates. Hourly updates shall be calculated using the average of the one-minute average data for the preceding hour. For the period beginning with initial operation under this standard until the source has operated for the full averaging period that you select, the average feedrate shall be based only on actual operation under this standard.

(B) Chromium—(1) Boilers that feed hazardous waste with a heating value of

40 CFR Ch. I (7–1–23 Edition)

10,000 Btu/lb or greater. (i) The 12-hour rolling average feedrate limit is a hazardous waste thermal concentration limit expressed as pounds of chromium in all hazardous waste feedstreams per million Btu of hazardous waste fed to the boiler. You must establish the 12hour rolling average feedrate limit as the average of the test run averages.

(*ii*) You must comply with the hazardous waste chromium thermal concentration limit by determining the feedrate of chromium in all hazardous waste feedstreams (lb/hr) and the hazardous waste thermal feedrate (MMBtu/hr) at least once each minute as [hazardous waste chromium feedrate (lb/hr)/hazardous waste thermal feedrate (MMBtu/hr)].

(2) Boilers that feed hazardous waste with a heating value less than 10,000 Btu/ lb. You must establish a 12-hour rolling average limit for the total feedrate (lb/ hr) of chromium in all feedstreams as the average of the test run averages.

(vi) LVM limits for pumpable wastes. You must establish separate feedrate limits for low volatile metals in pumpable feedstreams using the procedures prescribed above for total low volatile metals. Dual feedrate limits for both pumpable and total feedstreams are not required, however, if you base the total feedrate limit solely on the feedrate of pumpable feedstreams.

(vii) Extrapolation of feedrate levels. In lieu of establishing feedrate limits as specified in paragraphs (n)(2)(ii)through (vi) of this section, you may request as part of the performance test plan under §§63.7(b) and (c) and §§63.1207(e) and (f) to use the semivolatile metal and low volatile metal feedrates and associated emission rates during the comprehensive performance test to extrapolate to higher allowable feedrate limits and emission rates. The extrapolation methodology will be reviewed and approved, as warranted, by the Administrator. The review will consider in particular whether:

(A) Performance test metal feedrates are appropriate (i.e., whether feedrates are at least at normal levels; depending on the heterogeneity of the waste, whether some level of spiking would be appropriate; and whether the physical

§63.1209

form and species of spiked material is appropriate); and

(B) Whether the extrapolated feedrates you request are warranted considering historical metal feedrate data.

(3) Control device operating parameter limits (OPLs). You must establish operating parameter limits on the particulate matter control device as specified by paragraph (m)(1) of this section;

(4) Maximum total chlorine and chloride feedrate. You must establish a 12-hour rolling average limit for the feedrate of total chlorine and chloride in all feedstreams as the average of the test run averages.

(5) Maximum flue gas flowrate or production rate. (i) As an indicator of gas residence time in the control device, you must establish a limit on the maximum flue gas flowrate, the maximum production rate, or another parameter that you document in the site-specific test plan as an appropriate surrogate for gas residence time, as the average of the maximum hourly rolling averages for each run.

(ii) You must comply with this limit on a hourly rolling average basis.

(0) Hydrogen chloride and chlorine gas. You must comply with the hydrogen chloride and chlorine gas emission standard by establishing and complying with the following operating parameter limits. You must base the limits on operations during the comprehensive performance test, unless the limits are based on manufacturer specifications.

(1) Feedrate of total chlorine and chloride—(i) Incinerators, cement kilns, lightweight aggregate kilns, solid fuel boilers, and hydrochloric acid production furnaces. You must establish a 12-hour rolling average limit for the total feedrate of chlorine (organic and inorganic) in all feedstreams as the average of the test run averages.

(ii) Liquid fuel boilers—(A) Boilers that feed hazardous waste with a heating value not less than 10,000 Btu/lb. (1) The feedrate limit is a hazardous waste thermal concentration limit expressed as pounds of chlorine (organic and inorganic) in all hazardous waste feedstreams per million Btu of hazardous waste fed to the boiler. (2) You must establish a 12-hour rolling average feedrate limit as the average of the test run averages.

(3) You must comply with the feedrate limit by determining the mass feedrate of hazardous waste feedstreams (lb/hr) at least once a minute and by knowing the chlorine content (organic and inorganic, lb of chlorine/lb of hazardous waste) and heating value (Btu/lb) of hazardous waste feedstreams at all times to calculate a 1-minute average feedrate measurement as [hazardous waste chlorine content (lb of chlorine/lb of hazardous waste feed)/hazardous waste heating value (Btu/lb of hazardous waste)]. You must update the rolling average feedrate each hour with this 60-minute average feedrate measurement.

(B) Boilers that feed hazardous waste with a heating value less than 10,000 Btu/ lb. You must establish a 12-hour rolling average limit for the total feedrate of chlorine (organic and inorganic) in all feedstreams as the average of the test run averages. You must update the rolling average feedrate each hour with a 60-minute average feedrate measurement.

(2) Maximum flue gas flowrate or production rate. (i) As an indicator of gas residence time in the control device, you must establish a limit on the maximum flue gas flowrate, the maximum production rate, or another parameter that you document in the site-specific test plan as an appropriate surrogate for gas residence time, as the average of the maximum hourly rolling averages for each run.

(ii) You must comply with this limit on a hourly rolling average basis;

(3) Wet scrubber. If your combustor is equipped with a wet scrubber:

(i) If your source is equipped with a high energy wet scrubber such as a venturi, hydrosonic, collision, or free jet wet scrubber, you must establish a limit on minimum pressure drop across the wet scrubber on an hourly rolling average as the average of the test run averages;

(ii) If your source is equipped with a low energy wet scrubber such as a spray tower, packed bed, or tray tower, you must establish a minimum pressure drop across the wet scrubber based on manufacturer's specifications. You must comply with the limit on an hourly rolling average;

(iii) If your source is equipped with a low energy wet scrubber, you must establish a limit on minimum liquid feed pressure to the wet scrubber based on manufacturer's specifications. You must comply with the limit on an hourly rolling average;

(iv) You must establish a limit on minimum pH on an hourly rolling average as the average of the test run averages;

(v) You must establish limits on either the minimum liquid to gas ratio or the minimum scrubber water flowrate and maximum flue gas flowrate on an hourly rolling average as the average of the test run averages. If you establish limits on maximum flue gas flowrate under this paragraph, you need not establish a limit on maximum flue gas flowrate under paragraph (o)(2) of this section; and

(4) *Dry scrubber*. If your combustor is equipped with a dry scrubber, you must establish the following operating parameter limits:

(i) *Minimum sorbent feedrate*. You must establish a limit on minimum sorbent feedrate on an hourly rolling average as the average of the test run averages.

(ii) Minimum carrier fluid flowrate or nozzle pressure drop. You must establish a limit on minimum carrier fluid (gas or liquid) flowrate or nozzle pressure drop based on manufacturer's specifications.

(iii) Sorbent specifications. (A) You must specify and use the brand (i.e., manufacturer) and type of sorbent used during the comprehensive performance test until a subsequent comprehensive performance test is conducted, unless you document in the site-specific performance test plan required under §§ 63.1207(e) and (f) key parameters that affect adsorption and establish limits on those parameters based on the sorbent used in the performance test.

(B) You may substitute at any time a different brand or type of sorbent provided that the replacement has equivalent or improved properties compared to the sorbent used in the performance test and conforms to the key sorbent parameters you identify under para-

40 CFR Ch. I (7–1–23 Edition)

graph (0)(4)(iii)(A) of this section. You must record in the operating record documentation that the substitute sorbent will provide the same level of control as the original sorbent.

(p) Maximum combustion chamber pressure. If you comply with the requirements for combustion system leaks under §63.1206(c)(5) by maintaining the maximum combustion chamber zone pressure lower than ambient pressure to prevent combustion systems leaks from hazardous waste combustion, you must perform instantaneous monitoring of pressure and the automatic waste feed cutoff system must be engaged when negative pressure is not adequately maintained.

(q) Operating under different modes of operation. If you operate under different modes of operation, you must establish operating parameter limits for each mode. You must document in the operating record when you change a mode of operation and begin complying with the operating limits for an alternative mode of operation.

(1) Operating under otherwise applicable standards after the hazardous waste residence time has transpired. As provided by §63.1206(b)(1)(ii), you may operate under otherwise applicable requirements promulgated under sections 112 and 129 of the Clean Air Act in lieu of the substantive requirements of this subpart.

(i) The otherwise applicable requirements promulgated under sections 112 and 129 of the Clean Air Act are applicable requirements under this subpart.

(ii) You must specify (e.g., by reference) the otherwise applicable requirements as a mode of operation in your Documentation of Compliance under §63.1211(c), your Notification of Compliance under §63.1207(j), and your title V permit application. These requirements include the otherwise applicable requirements governing emission standards, monitoring and compliance, and notification, reporting, and recordkeeping.

(2) Calculating rolling averages under different modes of operation. When you transition to a different mode of operation, you must calculate rolling averages as follows:

(i) *Retrieval approach*. Calculate rolling averages anew using the continuous monitoring system values previously recorded for that mode of operation (i.e., you ignore continuous monitoring system values subsequently recorded under other modes of operation when you transition back to a mode of operation); or

(ii) *Start anew*. Calculate rolling averages anew without considering previous recordings.

(A) Rolling averages must be calculated as the average of the available one-minute values for the parameter until enough one-minute values are available to calculate hourly or 12-hour rolling averages, whichever is applicable to the parameter.

(B) You may not transition to a new mode of operation using this approach if the most recent operation in that mode resulted in an exceedance of an applicable emission standard measured with a CEMS or operating parameter limit prior to the hazardous waste residence time expiring; or

(iii) Seamless transition. Continue calculating rolling averages using data from the previous operating mode provided that both the operating limit and the averaging period for the parameter are the same for both modes of operation.

(r) Averaging periods. The averaging periods specified in this section for operating parameters are not-to-exceed averaging periods. You may elect to use shorter averaging periods. For example, you may elect to use a 1-hour rolling average rather than the 12-hour rolling average specified in paragraph (l)(1)(i) of this section for mercury.

[64 FR 53038, Sept. 30, 1999, as amended at 65
FR 42300, July 10, 2000; 65 FR 67271, Nov. 9,
2000; 66 FR 24272, May 14, 2001; 66 FR 35106,
July 3, 2001; 67 FR 6815, Feb. 13, 2002; 67 FR
6991, Feb. 14, 2002; 67 FR 77691, Dec. 19, 2002;
70 FR 59548, Oct. 12, 2005; 73 FR 18981, Apr. 8,
2008]

NOTIFICATION, REPORTING AND RECORDKEEPING

§63.1210 What are the notification requirements?

(a) *Summary of requirements*. (1) You must submit the following notifications to the Administrator:

Reference	Notification
63.9(b) 63.9(d) 63.9(d) 63.1206(b)(5)(i) 63.1206(c)(8)(iv) 63.1206(c)(9)(v) 63.1206(c)(9)(v) 63.1207(e), 63.9(e) 63.9(g)(1) and (3). 63.1210(b) 63.1207(k), 63.1207(l), 63.1207(k), 63.1207(l), 63.1207(k), 63.1207(l), 63.120(c)(2).	Initial notifications that you are sub≊ct to Subpart EEE of this Part. Notification that you are subïect to special compliance requirements. Notification and documentation of any change in information already provided under □63.9. Notification of changes in design, operation, or maintenance. Notification of excessive bag leak detection system exceedances. Notification of excessive particulate matter detection system exceedances. Notification of performance test and continuous monitoring system evaluation, including the perform- ance test plan and C□S performance evaluation plan. ¹ Notification of compliance, including results of performance tests and continuous monitoring system performance evaluations.

¹You may also be required on a case-by-case basis to submit a feedstream analysis plan under □63.1209(c)(3).

(2) You mu	ist subi	mit the follow	ing	you request or elect to comply with al-
notifications	to the	Administrator	r if	ternative requirements:

Reference	Notification, request, petition, or application
63.9(i)	You may request an ad@stment to time periods or postmark deadlines for submittal and review of required information.
63.10(e)(3)(ii)	You may request to reduce the frequency of excess emissions and C□S performance reports.
63.10(f)	You may request to waive recordkeeping or reporting requirements.
63.1204(d)(2)(iii), 63.1220(d)(2)(iii).	Notification that you elect to comply with the emission averaging requirements for cement kilns with in-line raw mills.
63.1204(e)(2)(iii), 63.1220(e)(2)(iii).	Notification that you elect to comply with the emission averaging requirements for preheater or pre- heater/precalciner kilns with dual stacks.
63.1206(b)(4), 63.1213, 63.6(i), 63.9(c).	You may request an extension of the compliance date for up to one year.

§63.1210

40 CFR Ch. I (7-1-23 Edition)

Reference	Notification, request, petition, or application
63.1206(b)(5)(i)(C)	You may request to burn hazardous waste for more than 720 hours and for purposes other than testing or pretesting after making a change in the design or operation that could affect compliance with emission standards and prior to submitting a revised Notification of Compliance
63.1206(b)(8)(iii)(B)	If you elect to conduct particulate matter CE_S correlation testing and wish to have federal particu- late matter and opacity standards and associated operating limits waived during the testing, you must polify the Administrator by submittion the correlation test loan for review and approval
63.1206(b)(8)(v)	You may request approval to have the particulate matter and opacity standards and associated op- erating limits and conditions waived for more than 96 hours for a correlation test.
63.1206(b)(9)	Owners and operators of lightweight aggregate kilns may request approval of alternative emission standards for mercury, semivolatile metal, low volatile metal, and hydrogen chloride/chlorine gas under certain conditions.
63.1206(b)(10)	Owners and operators of cement kilns may request approval of alternative emission standards for mercury, semivolatile metal, low volatile metal, and hydrogen chloride/chlorine gas under certain conditions.
63.1206(b)(14)	Owners and operators of incinerators may elect to comply with an alternative to the particulate mat- ter standard.
63.1206(b)(15)	Owners and operators of cement and lightweight aggregate kilns may request to comply with the al- ternative to the interim standards for mercury.
63.1206(c)(2)(ii)(C)	You may request to make changes to the startup, shutdown, and malfunction plan.
63.1206(c)(5)(i)(C)	You may request an alternative means of control to provide control of combustion system leaks.
63.1206(c)(5)(I)(D)	You may request other techniques to prevent fugitive emissions without use or instantaneous pres- sure limits.
63.1207(c)(2) 63.1207(d)(3)	You may request to base initial compliance on data in lieu of a comprehensive performance test. You may request more than 60 days to complete a performance test if additional time is needed for reasons beyond your control.
63.1207(e)(3), 63.7(h) 63.1207(h)(2)	You may request a time extension if the Administrator fails to approve or deny your test plan. You may request to waive current operating parameter limits during pretesting for more than 720 hours.
63.1207(f)(1)(ii)(D)	You may request a reduced hazardous waste feedstream analysis for organic hazardous air pollut- ants if the reduced analysis continues to be representative of organic hazardous air pollutants in your hazardous waste feedstreams.
63.1207(g)(2)(v)	You may request to operate under a wider operating range for a parameter during confirmatory per- formance testing.
63.1207(i)	You may request up to a one-year time extension for conducting a performance test (other than the initial comprehensive performance test) to consolidate testing with other state or federally-required testing.
63.1207())(4)	You may request more than 90 days to submit a Notification of Compliance after completing a per- formance test if additional time is needed for reasons beyond your control.
63.1207(I)(3)	After failure of a performance test, you may request to burn hazardous waste for more than 720 hours and for purposes other than testing or pretesting.
63.1209(a)(5), 63.8(f)	You may request: (1) Approval of alternative monitoring methods for compliance with standards that are monitored with a CE□S□and (2) approval to use a CE□S in lieu of operating parameter lim- its.
63.1209(g)(1)	You may request approval of: (1) Alternatives to operating parameter monitoring requirements, ex- cept for standards that you must monitor with a continuous emission monitoring system (CE□S) and except for requests to use a CE□S in lieu of operating parameter limits⊡or (2) a waiver of an operating parameter limit.
63.1209(I)(1) 63.1209(n)(2) 63.1211(d)	You may request to extrapolate mercury feedrate limits. You may request to extrapolate semivolatile and low volatile metal feedrate limits. You may request to use data compression techniques to record data on a less frequent basis than
	required by □63.1209.

(b) Notification of intent to comply (NIC). These procedures apply to sources that have not previously complied with the requirements of paragraphs (b) and (c) of this section, and to sources that previously complied with the NIC requirements of \S 63.1210 and 63.1212(a), which were in effect prior to October 11, 2000, that must make a technology change requiring a Class 1 permit modification to meet the standards of \S 63.1219, 63.1220, and 63.1221.

(1) You must prepare a Notification of Intent to Comply that includes all of the following information:

(i) General information:

(A) The name and address of the owner/operator and the source;

(B) Whether the source is a major or an area source;

(C) Waste minimization and emission control technique(s) being considered;

(D) Emission monitoring technique(s) you are considering;

(E) Waste minimization and emission control technique(s) effectiveness;

(F) A description of the evaluation criteria used or to be used to select waste minimization and/or emission control technique(s); and

(G) A general description of how you intend to comply with the emission standards of this subpart.

(ii) As applicable to each source, information on key activities and estimated dates for these activities that will bring the source into compliance with emission control requirements of this subpart. You must include all of the following key activities and dates in your NIC:

(A) The dates by which you anticipate you will develop engineering designs for emission control systems or process changes for emissions;

(B) The date by which you anticipate you will commit internal or external resources for installing emission control systems or making process changes for emission control, or the date by which you will issue orders for the purchase of component parts to accomplish emission control or process changes.

(C) The date by which you anticipate you will submit construction applications;

(D) The date by which you anticipate you will initiate on-site construction, installation of emission control equipment, or process change;

(E) The date by which you anticipate you will complete on-site construction, installation of emission control equipment, or process change; and

(F) The date by which you anticipate you will achieve final compliance. The individual dates and milestones listed in paragraphs (b)(1)(ii)(A) through (F) of this section as part of the NIC are not requirements and therefore are not enforceable deadlines; the requirements of paragraphs (b)(1)(ii)(A) through (F) of this section must be included as part of the NIC only to inform the public of how you intend to comply with the emission standards of this subpart.

(iii) A summary of the public meeting required under paragraph (c) of this section;

(iv) If you intend to cease burning hazardous waste prior to or on the compliance date, the requirements of paragraphs (b)(1)(ii) and (b)(1)(iii) of this section do not apply. You must include in your NIC a schedule of key dates for the steps to be taken to stop hazardous waste activity at your combustion unit. Key dates include the date for submittal of RCRA closure documents required under subpart G, part 264 or subpart G, part 265 of this chapter.

(2) You must make a draft of the NIC available for public review no later than 30 days prior to the public meeting required under paragraph (c)(1) of this section or no later than 9 months after the effective date of the rule if you intend to cease burning hazardous waste prior to or on the compliance date.

(3) You must submit the final NIC to the Administrator:

(i) *Existing units*. No later than one year following the effective date of the emission standards of this subpart; or

(ii) *New units*. No later than 60 days following the informal public meeting.

(c) *NIC public meeting and notice*. (1) Prior to the submission of the NIC to the permitting agency and:

(i) Existing units. No later than 10 months after the effective date of the emission standards of this subpart, you must hold at least one informal meeting with the public to discuss the anticipated activities described in the draft NIC for achieving compliance with the emission standards of this subpart. You must post a sign-in sheet or otherwise provide a voluntary opportunity for attendees to provide their names and addresses.

(ii) New units. No earlier than thirty (30) days following notice of the informal public meeting, you must hold at least one informal meeting with the public to discuss the anticipated activities described in the draft NIC for achieving compliance with the emission standards of this subpart. You must post a sign-in sheet or otherwise provide a voluntary opportunity for attendees to provide their names and addresses.

(2) You must submit a summary of the meeting, along with the list of attendees and their addresses developed under paragraph (b)(1) of this section, and copies of any written comments or materials submitted at the meeting, to the Administrator as part

of the final NIC, in accordance with paragraph (b)(1)(iii) of this section;

(3) You must provide public notice of the NIC meeting at least 30 days prior to the meeting and you must maintain, and provide to the Administrator upon request, documentation of the notice. You must provide public notice in all of the following forms:

(i) Newspaper advertisement. You must publish a notice in a newspaper of general circulation in the county or equivalent jurisdiction of your facility. In addition, you must publish the notice in newspapers of general circulation in adjacent counties or equivalent jurisdiction where such publication would be necessary to inform the affected public. You must publish the notice as a display advertisement.

(ii) Visible and accessible sign. You must post a notice on a clearly marked sign at or near the source. If you place the sign on the site of the hazardous waste combustor, the sign must be large enough to be readable from the nearest spot where the public would pass by the site.

(iii) Broadcast media announcement. You must broadcast a notice at least once on at least one local radio station or television station.

(iv) Notice to the facility mailing list. You must provide a copy of the notice to the facility mailing list in accordance with 124.10(c)(1)(ix) of this chapter.

(4) You must include all of the following in the notices required under paragraph (c)(3) of this section:

(i) The date, time, and location of the meeting;

(ii) A brief description of the purpose of the meeting:

(iii) A brief description of the source and proposed operations, including the address or a map (e.g., a sketched or copied street map) of the source location;

(iv) A statement encouraging people to contact the source at least 72 hours 40 CFR Ch. I (7–1–23 Edition)

before the meeting if they need special access to participate in the meeting;

(v) A statement describing how the draft NIC (and final NIC, if requested) can be obtained; and

(vi) The name, address, and telephone number of a contact person for the NIC.

(5) The requirements of this paragraph do not apply to sources that intend to cease burning hazardous waste prior to or on the compliance date.

(d) Notification of compliance. (1) The Notification of Compliance status requirements of §63.9(h) apply, except that:

(i) The notification is a Notification of Compliance, rather than compliance status;

(ii) The notification is required for the initial comprehensive performance test and each subsequent comprehensive and confirmatory performance test; and

(iii) You must postmark the notification before the close of business on the 90th day following completion of relevant compliance demonstration activity specified in this subpart rather than the 60th day as required by $\S63.9(h)(2)(ii)$.

(2) Upon postmark of the Notification of Compliance, the operating parameter limits identified in the Notification of Compliance, as applicable, shall be complied with, the limits identified in the Documentation of Compliance or a previous Notification of Compliance are no longer applicable.

(3) The Notification of Compliance requirements of §63.1207(j) also apply.

[64 FR 53038, Sept. 30, 1999, as amended at 64
FR 63211, Nov. 19, 1999; 65 FR 42301, July 10, 2000; 66 FR 24272, May 14, 2001; 67 FR 6992, Feb. 14, 2002; 70 FR 59552, Oct. 12, 2005; 73 FR 18982, Apr. 8, 2008; 73 FR 64097, Oct. 28, 2008]

§ 63.1211 What are the recordkeeping and reporting requirements?

(a) *Summary of reporting requirements*. You must submit the following reports to the Administrator:

Reference	Report
63.10(d)(4)	Compliance progress reports, if required as a condition of an extension of the compliance date granted under □63.6(i)
63.10(d)(5)(i)	Periodic startup, shutdown, and malfunction reports.
§63.1211

Reference	Report
63.10(d)(5)(ii)	Immediate startup, shutdown, and malfunction reports.
63.10(e)(3)	Excessive emissions and continuous monitoring system performance re-
	port and summary report.
63.1206(c)(2)(ii)(B)	Startup, shutdown, and malfunction plan.
63.1206(c)(3)(vi)	Excessive exceedances reports.
63.1206(c)(4)(iv)	Emergency safety vent opening reports.

(b) *Summary of recordkeeping requirements*. You must retain the following in the operating record:

Reference	Document, Data, or Information
63.1200, 63.10(b) and (c)	General. Information required to document and maintain compliance with the regulations of Subpart EEE, including data recorded by continuous monitoring systems (C□S), and copies of all notifications, reports, plans, and other documents submitted to the Administrator.
63.1204(d)(1)(ii), 63.1220(d)(1)(ii).	Documentation of mode of operation changes for cement kilns with in-line raw mills.
63.1204(d)(2)(ii), 63.1220(d)(2)(ii).	Documentation of compliance with the emission averaging requirements for cement kilns with in-line raw mills.
63.1204(e)(2)(ii), 63.1220(e)(2)(ii).	Documentation of compliance with the emission averaging requirements for preheater or preheater/ precalciner kilns with dual stacks.
63.1206(b)(1)(ii)	If you elect to comply with all applicable requirements and standards promulgated under authority of the Clean Air Act, including Sections 112 and 129, in lieu of the requirements of Subpart EEE when not burning hazardous waste, you must document in the operating record that you are in compliance with those requirements.
63.1206(b)(5)(ii)	Documentation that a change will not adversely affect compliance with the emission standards or operating requirements.
63.1206(b)(11)	Calculation of hazardous waste residence time.
63.1206(c)(2)	Startup, shutdown, and malfunction plan.
63.1206(c)(2)(v)(A)	Documentation of your investigation and evaluation of excessive exceedances during malfunctions.
63.1206(c)(3)(v)	Corrective measures for any automatic waste feed cutoff that results in an exceedance of an emis- sion standard or operating parameter limit.
63.1206(c)(3)(vii)	Documentation and results of the automatic waste feed cutoff operability testing.
63.1206(c)(4)(ii)	Emergency safety vent operating plan.
63.1206(c)(4)(iii)	Corrective measures for any emergency safety vent opening.
63.1206(c)(5)(ii)	ethod used for control of combustion system leaks.
63.1206(c)(6)	Operator training and certification program.
63.1206(c)(7)(i)(D)	Operation and maintenance plan.
63.1209(c)(2)	Feedstream analysis plan.
63.1209(k)(6)(iii),	Documentation that a substitute activated carbon, dioxin/furan formation reaction inhibitor, or dry
63.1209(k)(7)(ii), 63.1209(k)(9)(ii).	scrubber sorbent will provide the same level of control as the original material.
$63 1209(\alpha)(4)(iii)$	
63.1209(k)(7)(i)(C)	Results of carbon bed performance monitoring.
63.1209(g)	Documentation of changes in modes of operation.
63.1211(c)	Documentation of compliance.

(c) Documentation of compliance. (1) By the compliance date, you must develop and include in the operating record a Documentation of Compliance. You are not subject to this requirement, however, if you submit a Notification of Compliance under §63.1207(j) prior to the compliance date. Upon inclusion of the Documentation of Compliance in the operating record, hazardous waste burning incinerators, cement kilns, and lightweight aggregate kilns regulated under the interim standards of §§63.1203, 63.1204, and 63.1205 are no longer subject to compliance with the previously applicable Notification of Compliance.

(2) The Documentation of Compliance must identify the applicable emission standards under this subpart and the limits on the operating parameters under $\S63.1209$ that will ensure compliance with those emission standards.

(3) You must include a signed and dated certification in the Documentation of Compliance that:

(i) Required CEMs and CMS are installed, calibrated, and continuously operating in compliance with the requirements of this subpart; and

(ii) Based on an engineering evaluation prepared under your direction or supervision in accordance with a system designed to ensure that qualified personnel properly gathered and evaluated the information and supporting documentation, and considering at a minimum the design, operation, and maintenance characteristics of the combustor and emissions control equipment, the types, quantities, and characteristics of feedstreams, and available emissions data:

(A) You are in compliance with the emission standards of this subpart; and

(B) The limits on the operating parameters under §63.1209 ensure compliance with the emission standards of this subpart.

(4) You must comply with the emission standards and operating parameter limits specified in the Documentation of Compliance.

(d) Data compression. You may submit a written request to the Administrator for approval to use data compression techniques to record data from CMS, including CEMS, on a frequency less than that required by §63.1209. You must submit the request for review and approval as part of the comprehensive performance test plan.

(1) You must record a data value at least once each ten minutes.

(2) For each CEMS or operating parameter for which you request to use data compression techniques, you must recommend:

(i) A fluctuation limit that defines the maximum permissible deviation of a new data value from a previously generated value without requiring you to revert to recording each one-minute value.

(A) If you exceed a fluctuation limit, you must record each one-minute value for a period of time not less than ten minutes.

(B) If neither the fluctuation limit nor the data compression limit are exceeded during that period of time, you may reinitiate recording data values on a frequency of at least once each ten minutes; and

(ii) A data compression limit defined as the closest level to an operating parameter limit or emission standard at which reduced data recording is allowed.

(A) Within this level and the operating parameter limit or emission standard, you must record each oneminute average. 40 CFR Ch. I (7–1–23 Edition)

(B) The data compression limit should reflect a level at which you are unlikely to exceed the specific operating parameter limit or emission standard, considering its averaging period, with the addition of a new oneminute average.

[64 FR 53038, Sept. 30, 1999, as amended at 64 FR 63212, Nov. 19, 1999; 65 FR 42301, July 10, 2000; 66 FR 24272, May 14, 2001; 66 FR 35106, July 3, 2001; 67 FR 6993, Feb. 14, 2002; 70 FR 59554, Oct. 12, 2005]

OTHER

§63.1212 What are the other requirements pertaining to the NIC?

(a) Certification of intent to complu. The Notice of Intent to Comply (NIC) must contain the following certification signed and dated by a responsible official as defined under §63.2 of this chapter: I certify under penalty of law that I have personally examined and am familiar with the information submitted in this document and all attachments and that, based on my inquiry of those individuals immediately responsible for obtaining the information, I believe that the information is true, accurate, and complete. I am aware that there are significant penalties for submitting false information, including the possibility of fine and imprisonment.

(b) *New units.* Any source that files a RCRA permit application or permit modification request for construction of a hazardous waste combustion unit after October 12, 2005 must:

(1) Prepare a draft NIC pursuant to §63.1210(b) and make it available to the public upon issuance of the notice of public meeting pursuant to §63.1210(c)(3);

(2) Prepare a draft comprehensive performance test plan pursuant to the requirements of §63.1207 and make it available for public review upon issuance of the notice of NIC public meeting;

(3) Provide notice to the public of a pre-application meeting pursuant to §124.31 of this chapter or notice to the public of a permit modification request pursuant to §270.42 of this chapter;

(4) Hold an informal public meeting [pursuant to 63.1210(c)(1) and (c)(2)] no earlier than 30 days following notice of

the NIC public meeting and notice of the pre-application meeting or notice of the permit modification request to discuss anticipated activities described in the draft NIC and pre-application or permit modification request for achieving compliance with the emission standards of this subpart; and

(5) Submit a final NIC pursuant to §63.1210(b)(3).

(c) Information Repository specific to new combustion units. (1) Any source that files a RCRA permit application or modification request for construction of a new hazardous waste combustion unit after October 12, 2005 may be required to establish an information repository if deemed appropriate.

(2) The Administrator may assess the need, on a case-by-case basis for an information repository. When assessing the need for a repository, the Administrator shall consider the level of public interest, the presence of an existing repository, and any information available via the New Source Review and Title V permit processes. If the Administrator determines a need for a repository, then the Administrator shall notify the facility that it must establish and maintain an information repository.

(3) The information repository shall contain all documents, reports, data, and information deemed necessary by the Administrator. The Administrator shall have the discretion to limit the contents of the repository.

(4) The information repository shall be located and maintained at a site chosen by the source. If the Administrator finds the site unsuitable for the purposes and persons for which it was established, due to problems with location, hours of availability, access, or other relevant considerations, then the Administrator shall specify a more appropriate site.

(5) The Administrator shall require the source to provide a written notice about the information repository to all individuals on the source mailing list.

(6) The source shall be responsible for maintaining and updating the repository with appropriate information throughout a period specified by the Administrator. The Administrator may close the repository at his or her discretion based on the considerations in paragraph (c)(2) of this section.

[70 FR 59555, Oct. 12, 2005, as amended at 73 FR 18982, Apr. 8, 2008]

§ 63.1213 How can the compliance date be extended to install pollution prevention or waste minimization controls?

(a) Applicability. You may request from the Administrator or State with an approved Title V program an extension of the compliance date of up to one year. An extension may be granted if you can reasonably document that the installation of pollution prevention or waste minimization measures will significantly reduce the amount and/or toxicity of hazardous wastes entering the feedstream(s) of the hazardous waste combustor(s), and that you could not install the necessary control measures and comply with the emission standards and operating requirements of this subpart by the compliance date.

(b) Requirements for requesting an extension. (1) You must make your requests for an (up to) one-year extension in writing in accordance with §63.6(i)(4)(B) and (C). The request must contain the following information:

(i) A description of pollution prevention or waste minimization controls that, when installed, will significantly reduce the amount and/or toxicity of hazardous wastes entering the feedstream(s) of the hazardous waste combustor(s). Pollution prevention or waste minimization measures may include: equipment or technology modifications, reformulation or redesign of products, substitution of raw materials, improvements in work practices, maintenance, training, inventory control, or recycling practices conducted as defined in §261.1(c) of this chapter;

(ii) A description of other pollution controls to be installed that are necessary to comply with the emission standards and operating requirements;

(iii) A reduction goal or estimate of the annual reductions in quantity and/ or toxicity of hazardous waste(s) entering combustion feedstream(s) that you will achieve by installing the proposed pollution prevention or waste minimization measures; (iv) A comparison of reductions in the amounts and/or toxicity of hazardous wastes combusted after installation of pollution prevention or waste minimization measures to the amounts and/or toxicity of hazardous wastes combusted prior to the installation of these measures. If the difference is less than a fifteen percent reduction, include a comparison to pollution prevention and waste minimization reductions recorded during the previous five years;

(v) Reasonable documentation that installation of the pollution prevention or waste minimization changes will not result in a net increase (except for documented increases in production) of hazardous constituents released to the environment through other emissions, wastes or effluents;

(vi) Reasonable documentation that the design and installation of waste minimization and other measures that are necessary for compliance with the emission standards and operating requirements of this subpart cannot otherwise be installed within the three year compliance period, and

(vii) The information required in §63.6(i)(6)(i)(B) through (D).

(2) You may enclose documentation prepared under an existing State-required pollution prevention program that contains the information prescribed in paragraph (b) of this section with a request for extension in lieu of complying with the time extension requirements of that paragraph.

(c) Approval of request for extension of compliance date. Based on the information provided in any request made under paragraph (a) of this section, the Administrator or State with an approved title V program may grant an extension of the compliance date of this subpart. The extension will be in writing in accordance with \S 63.6(i)(10)(i) through 63.6(i)(10)(v)(A).

[57 FR 61992, Dec. 29, 1992, as amended at 67 FR 6994, Feb. 14, 2002; 67 FR 77691, Dec. 19, 2002]

§63.1214 Implementation and enforcement.

(a) This subpart can be implemented and enforced by the U.S. EPA, or a delegated authority such as the applicable State, local, or Tribal agency. If the

40 CFR Ch. I (7–1–23 Edition)

U.S. EPA Administrator has delegated authority to a State, local, or Tribal agency, then that agency, in addition to the U.S. EPA, has the authority to implement and enforce this subpart. Contact the applicable U.S. EPA Regional Office to find out if this subpart is delegated to a State, local, or Tribal agency.

(b) In delegating implementation and enforcement authority of this subpart to a State, local, or Tribal agency under subpart E of this part, the authorities contained in paragraph (c) of this section are retained by the Administrator of U.S. EPA and cannot be transferred to the State, local, or Tribal agency.

(c) The authorities that cannot be delegated to State, local, or Tribal agencies are as specified in paragraphs (c)(1) through (4) of this section.

(1) Approval of alternatives to requirements in \S 63.1200, 63.1203, 63.1204, 63.1205, 63.1206(a), 63.1215, 63.1216, 63.1217, 63.1218, 63.1219, 63.1220, and 63.1221.

(2) Approval of major alternatives to test methods under \$ 63.7(e)(2)(ii) and (f), 63.1208(b), and 63.1209(a)(1), as defined under \$ 63.90, and as required in this subpart.

(3) Approval of major alternatives to monitoring under $\S 63.8(f)$ and 63.1209(a)(5), as defined under $\S 63.90$, and as required in this subpart.

(4) Approval of major alternatives to recordkeeping and reporting under §§ 63.10(f) and 63.1211(a) through (c), as defined under §63.90, and as required in this subpart.

[68 FR 37356, June 23, 2003, as amended at 70 FR 59555, Oct. 12, 2005]

§63.1215 What are the health-based compliance alternatives for total chlorine?

(a) General—(1) Overview. You may establish and comply with health-based compliance alternatives for total chlorine under the procedures prescribed in this section for your hazardous waste combustors other than hydrochloric acid production furnaces. You may comply with these health-based compliance alternatives in lieu of the emission standards for total chlorine provided under §§ 63.1216, 63.1217, 63.1219,

63.1220, and 63.1221. To identify and comply with the limits, you must:

(i) Identify a total chlorine emission concentration (ppmv) expressed as chloride (Cl(-)) equivalent for each on site hazardous waste combustor. You may select total chlorine emission concentrations as you choose to demonstrate eligibility for the risk-based limits under this section, except as provided by paragraph (b)(7) of this section;

(ii) Apportion the total chlorine emission concentration between HCl and Cl₂ according to paragraph (b)(6)(i) of this section, and calculate HCl and Cl₂ emission rates (lb/hr) using the gas flowrate and other parameters from the most recent regulatory compliance test.

(iii) Calculate the annual average HCl-equivalent emission rate as prescribed in paragraph (b)(2) of this section.

(iv) Perform an eligibility demonstration to determine if your HClequivalent emission rate meets the national exposure standard and thus is below the annual average HCl-equivalent emission rate limit, as prescribed by paragraph (c) of this section;

(v) Submit your eligibility demonstration for review and approval, as prescribed by paragraph (e) of this section, which must include information to ensure that the 1-hour average HClequivalent emission rate limit is not exceeded, as prescribed by paragraph (d) of this section;

(vi) Demonstrate compliance with the annual average HCl-equivalent emission rate limit during the comprehensive performance test, as prescribed by the testing and monitoring requirements under paragraph (e) of this section;

(vii) Comply with compliance monitoring requirements, including establishing feedrate limits on total chlorine and chloride, and operating parameter limits on emission control equipment, as prescribed by paragraph (f) of this section; and

(viii) Comply with the requirements for changes, as prescribed by paragraph (h) of this section.

(2) *Definitions*. In addition to the definitions under §63.1201, the following definitions apply to this section:

1-Hour Average HCl-Equivalent Emission Rate means the HCl-equivalent emission rate (lb/hr) determined by equating the toxicity of chlorine to HCl using aRELs as the health risk metric for acute exposure.

1-Hour Average HCl-Equivalent Emission Rate Limit means the HCl-equivalent emission rate (lb/hr) determined by equating the toxicity of chlorine to HCl using aRELs as the health risk metric for acute exposure and which ensures that maximum 1-hour average ambient concentrations of HCl-equivalents do not exceed a Hazard Index of 1.0, rounded to the nearest tenths decimal place (0.1), at an off-site receptor location.

Acute Reference Exposure Level (aREL) means health thresholds below which there would be no adverse health effects for greater than once in a lifetime exposures of one hour. ARELs are developed by the California Office of Health Hazard Assessment and are available at http://www.oehha.ca.gov/air/ acute_rels/acuterel.html.

Annual Average HCl-Equivalent Emission Rate means the HCl-equivalent emission rate (lb/hr) determined by equating the toxicity of chlorine to HCl using RfCs as the health risk metric for long-term exposure.

Annual Average HCl-Equivalent Emission Rate Limit means the HCl-equivalent emission rate (lb/hr) determined by equating the toxicity of chlorine to HCl using RfCs as the health risk metric for long-term exposure and which ensures that maximum annual average ambient concentrations of HCl equivalents do not exceed a Hazard Index of 1.0, rounded to the nearest tenths decimal place (0.1), at an off-site receptor location.

Hazard Index (HI) means the sum of more than one Hazard Quotient for multiple substances and/or multiple exposure pathways. In this section, the Hazard Index is the sum of the Hazard Quotients for HCl and chlorine.

Hazard Quotient (HQ) means the ratio of the predicted media concentration of a pollutant to the media concentration at which no adverse effects are expected. For chronic inhalation exposures, the HQ is calculated under this section as the air concentration divided by the RfC. For acute inhalation exposures, the HQ is calculated under this section as the air concentration divided by the aREL.

Look-up table analysis means a risk screening analysis based on comparing the HCl-equivalent emission rate from the affected source to the appropriate HCl-equivalent emission rate limit specified in Tables 1 through 4 of this section.

Reference Concentration (RfC) means an estimate (with uncertainty spanning perhaps an order of magnitude) of a continuous inhalation exposure to the human population (including sensitive subgroups) that is likely to be without an appreciable risk of deleterious effects during a lifetime. It can be derived from various types of human or animal data, with uncertainty factors generally applied to reflect limitations of the data used.

(b) *HCl-equivalent emission rates*. (1) You must express total chlorine emission rates for each hazardous waste combustor as HCl-equivalent emission rates.

(2) Annual average rates. You must calculate annual average toxicityweighted HCl-equivalent emission rates for each combustor as follows:

 $ER_{LTtw} = ER_{HC1} + ER_{Cl_2} \times (RfC_{HCl}/RfC_{Cl_2})$

Where:

ER_{LTtw} is the annual average HCl toxicityweighted emission rate (HCl-equivalent emission rate) considering long-term exposures, lb/hr

 $\mathrm{ER}_{\mathrm{HCl}}$ is the emission rate of HCl in lbs/hr

 $\mathrm{ER}_{\mathrm{Cl}_2}$ is the emission rate of chlorine in lbs/ hr

 ${\rm RfC}_{\rm HCI}$ is the reference concentration of HCl ${\rm RfC}_{\rm Cl_2}$ is the reference concentration of chlorine

(3) *1-hour average rates.* You must calculate 1-hour average toxicity-weighted HCl-equivalent emission rates for each combustor as follows:

 $\begin{array}{rcl} ER_{STtw} &= ER_{HCl} + ER_{Cl_2} \times (aREL_{HCl} / \\ aREL_{Cl_2}) \end{array}$

Where:

 $\mathrm{ER}_{\mathrm{STtw}}$ is the 1-hour average HCl-toxicityweighted emission rate (HCl-equivalent emission rate) considering 1-hour (shortterm) exposures, lb/hr

 $\mathrm{ER}_{\mathrm{HCl}}$ is the emission rate of HCl in lbs/hr

 $\mathrm{ER}_{\mathrm{Cl}_2}$ is the emission rate of chlorine in lbs/ hr

 $aREL_{HCl}$ is the aREL for HCl

 $a\mathrm{REL}_{Cl_2}$ is the aREL for chlorine

(4) You must use the RfC values for hydrogen chloride and chlorine found at *http://epa.gov/ttn/atw/toxsource/ summary.html*.

(5) You must use the aREL values for hydrogen chloride and chlorine found at *http://www.oehha.ca.gov/air/ acute_rels/acuterel.html.*

(6) Cl_2HCl ratios—(i) Ratio for calculating annual average HCl-equivalent emission rates. (A) To calculate the annual average HCl-equivalent emission rate (lb/hr) for each combustor, you must apportion the total chlorine emission concentration (ppmv chloride ($Cl^{(-)}$) equivalent) between HCl and chlorine according to the historical average Cl_2/HCl volumetric ratio for all regulatory compliance tests.

(B) You must calculate HCl and Cl_2 emission rates (lb/hr) using the apportioned emission concentrations and the gas flowrate and other parameters from the most recent regulatory compliance test.

(C) You must calculate the annual average HCl-equivalent emission rate using these HCl and Cl_2 emission rates and the equation in paragraph (b)(2) of this section.

(ii) Ratio for calculating 1-hour average HCl-equivalent emission rates. (A) To calculate the 1-hour average HCl-equivalent emission rate for each combustor as a criterion for you to determine under paragraph (d) of this section if an hourly rolling average feedrate limit on total chlorine and chloride may be waived, you must apportion the total chlorine emission concentration (ppmv chloride (Cl⁽⁻⁾) equivalent) between HCl and chlorine according to the historical highest Cl₂/HCl volumetric ratio for all regulatory compliance tests.

(B) You must calculate HCl and Cl_2 emission rates (lb/hr) using the apportioned emission concentrations and the gas flowrate and other parameters from the most recent regulatory compliance test.

(C) You must calculate the 1-hour average HCl-equivalent emission rate using these HCl and Cl_2 emission rates and the equation in paragraph (b)(3) of this section.

(iii) Ratios for new sources. (A) You must use engineering information to estimate the Cl_2/HCl volumetric ratio

for a new source for the initial eligibility demonstration.

(B) You must use the Cl₂/HCl volumetric ratio demonstrated during the initial comprehensive performance test to demonstrate in the Notification of Compliance that your HCl-equivalent emission rate does not exceed your HCl-equivalent emission rate limit.

(C) When approving the test plan for the initial comprehensive performance test, the permitting authority will establish a periodic testing requirement, such as every 3 months for 1 year, to establish a record of representative $Cl_2/$ HCl volumetric ratios.

(1) You must revise your HCl-equivalent emission rates and HCl-equivalent emission rate limits after each such test using the procedures prescribed in paragraphs (b)(6)(i) and (ii) of this section.

(2) If you no longer are eligible for the health-based compliance alternative, you must notify the permitting authority immediately and either:

(i) Submit a revised eligibility demonstration requesting lower HCl-equivalent emission rate limits, establishing lower HCl-equivalent emission rates, and establishing by downward extrapolation lower feedrate limits for total chlorine and chloride; or

(*ii*) Request a compliance schedule of up to three years to demonstrate compliance with the emission standards under \S 63.1216, 63.1217, 63.1219, 63.1220, and 63.1221.

(iv) Unrepresentative or inadequate historical Cl_2/HCl volumetric ratios. (A) If you believe that the Cl_2/HCl volumetric ratio for one or more historical regulatory compliance tests is not representative of the current ratio, you may request that the permitting authority allow you to screen those ratios from the analysis of historical ratios.

(B) If the permitting authority believes that too few historical ratios are available to calculate a representative average ratio or establish a maximum ratio, the permitting authority may require you to conduct periodic testing to establish representative ratios.

(v) Updating Cl_2/HCl ratios. You must include the Cl_2/HCl volumetric ratio demonstrated during each performance test in your data base of historical $Cl_2/$

HCl ratios to update the ratios you establish under paragraphs (b)(6)(i) and (ii) of this section for subsequent calculations of the annual average and 1hour average HCl-equivalent emission rates.

(7) Emission rates are capped. The hydrogen chloride and chlorine emission rates you use to calculate the HClequivalent emission rate limit for incinerators, cement kilns, and lightweight aggregate kilns must not result in total chlorine emission concentrations exceeding:

(i) For incinerators that were existing sources on April 19, 1996: 77 parts per million by volume, combined emissions, expressed as chloride $(Cl^{(-)})$ equivalent, dry basis and corrected to 7 percent oxygen;

(ii) For incinerators that are new or reconstructed sources after April 19, 1996: 21 parts per million by volume, combined emissions, expressed as chloride ($Cl^{(-)}$) equivalent, dry basis and corrected to 7 percent oxygen;

(iii) For cement kilns that were existing sources on April 19, 1996: 130 parts per million by volume, combined emissions, expressed as chloride $(Cl^{(-)})$ equivalent, dry basis and corrected to 7 percent oxygen;

(iv) For cement kilns that are new or reconstructed sources after April 19, 1996: 86 parts per million by volume, combined emissions, expressed as chloride ($Cl^{(-)}$) equivalent, dry basis and corrected to 7 percent oxygen;

(v) For lightweight aggregate kilns that were existing sources on April 19, 1996: 600 parts per million by volume, combined emissions, expressed as chloride ($Cl^{(-)}$) equivalent, dry basis and corrected to 7 percent oxygen;

(vi) For lightweight aggregate kilns that are new or reconstructed sources after April 19, 1996: 600 parts per million by volume, combined emissions, expressed as chloride $(Cl^{(-)})$ equivalent, dry basis and corrected to 7 percent oxygen.

(c) Eligibility demonstration—(1) General. (i) You must perform an eligibility demonstration to determine whether the total chlorine emission rates you select for each on-site hazardous waste combustor meet the national exposure standards using either a look-up table analysis prescribed by paragraph (c)(3) of this section, or a site-specific compliance demonstration prescribed by paragraph (c)(4) of this section.

(ii) You must also determine in your eligibility demonstration whether each combustor may exceed the 1-hour HCIequivalent emission rate limit absent an hourly rolling average limit on the feedrate of total chlorine and chloride, as provided by paragraph (d) of this section.

(2) Definition of eligibility. (i) Eligibility for the risk-based total chlorine standard is determined by comparing the annual average HCl-equivalent emission rate for the total chlorine emission rate you select for each combustor to the annual average HClequivalent emission rate limit.

(ii) The annual average HCl-equivalent emission rate limit ensures that the Hazard Index for chronic exposure from HCl and chlorine emissions from all on-site hazardous waste combustors is less than or equal to 1.0, rounded to the nearest tenths decimal place (0.1), for the actual individual most exposed to the facility's emissions, considering off-site locations where people reside and where people congregate for work, school, or recreation.

(iii) Your facility is eligible for the health-based compliance alternative for total chlorine if either:

(A) The annual average HCl-equivalent emission rate for each on-site hazardous waste combustor is below the appropriate value in the look-up table determined under paragraph (c)(3) of this section; or

(B) The annual average HCl-equivalent emission rate for each on-site hazardous waste combustor is below the annual average HCl-equivalent emission rate limit you calculate based on a site-specific compliance demonstra40 CFR Ch. I (7–1–23 Edition)

tion under paragraph (c)(4) of this section.

(3) *Look-up table analysis*. Look-up tables for the eligibility demonstration are provided as Tables 1 and 2 to this section.

(i) Table 1 presents annual average HCl-equivalent emission rate limits for sources located in flat terrain. For purposes of this analysis, flat terrain is terrain that rises to a level not exceeding one half the stack height within a distance of 50 stack heights.

(ii) Table 2 presents annual average HCl-equivalent emission rate limits for sources located in simple elevated terrain. For purposes of this analysis, simple elevated terrain is terrain that rises to a level exceeding one half the stack height, but that does not exceed the stack height, within a distance of 50 stack heights.

(iii) To determine the annual average HCl-equivalent emission rate limit for a source from the look-up table, you must use the stack height and stack diameter for your hazardous waste combustors and the distance between the stack and the property boundary.

(iv) If any of these values for stack height, stack diameter, and distance to nearest property boundary do not match the exact values in the look-up table, you must use the next lowest table value.

(v) Adjusted HCl-equivalent emission rate limit for multiple on-site combustors.
(A) If you have more than one hazardous waste combustor on site, the sum across all hazardous waste combustors of the ratio of the adjusted HCl-equivalent emission rate limit to the HCl-equivalent emission rate limit provided by Tables 1 or 2 cannot exceed 1.0, according to the following equation:

$\sum_{i=1}^{n} \frac{\text{HC1-Equivalent Emission Rate Limit Adjusted}_{i}}{\text{HC1-Equivalent Emission Rate Limit Table}_{i}} \leq 1.0$

Where:

- i = number of on-site hazardous waste combustors;
- HCl-Equivalent Emission Rate Limit Adjusted, means the apportioned, allowable

HCl-equivalent emission rate limit for combustor i, and

 $\begin{array}{l} HCl\text{-}Equivalent \ Emission \ Rate \ Limit \ Table_i \\ means \ the \ HCl\text{-}equivalent \ emission \ rate \end{array}$

limit from Table 1 or 2 to §63.1215 for combustor i.

(B) The adjusted HCl-equivalent emission rate limit becomes the HCl-equivalent emission rate limit.

(4) Site-specific compliance demonstration. (i) You may use any scientificallyaccepted peer-reviewed risk assessment methodology for your site-specific compliance demonstration to calculate an annual average HCl-equivalent emission rate limit for each on-site hazardous waste combustor. An example of one approach for performing the demonstration for air toxics can be found in the EPA's "Air Toxics Risk Assessment Reference Library, Volume 2. Site-Specific Risk Assessment Technical Resource Document," which may be obtained through the EPA's Air Toxics Web site at http://www.epa.gov/ ttn/fera/risk_atra_main.html.

(ii) The annual average HCl-equivalent emission rate limit is the HClequivalent emission rate that ensures that the Hazard Index associated with maximum annual average exposures is not greater than 1.0 rounded to the nearest tenths decimal place (0.1).

(iii) To determine the annual average HCl-equivalent emission rate limit, your site-specific compliance demonstration must, at a minimum:

(A) Estimate long-term inhalation exposures through the estimation of annual or multi-year average ambient concentrations;

(B) Estimate the inhalation exposure for the actual individual most exposed to the facility's emissions from hazardous waste combustors, considering off-site locations where people reside and where people congregate for work, school, or recreation;

(C) Use site-specific, quality-assured data wherever possible;

(D) Use health-protective default assumptions wherever site-specific data are not available, and:

(E) Contain adequate documentation of the data and methods used for the assessment so that it is transparent and can be reproduced by an experienced risk assessor and emissions measurement expert.

(iv) Your site-specific compliance demonstration need not:

(A) Assume any attenuation of exposure concentrations due to the penetration of outdoor pollutants into indoor exposure areas;

(B) Assume any reaction or deposition of the emitted pollutants during transport from the emission point to the point of exposure.

(d) Assurance that the 1-hour HClequivalent emission rate limit will not be exceeded. To ensure that the 1-hour HCl-equivalent emission rate limit will not be exceeded when complying with the annual average HCl-equivalent emission rate limit, you must establish a 1-hour average HCl-equivalent emission rate for each combustor, establish a 1-hour average HCl-equivalent emission rate limit for each combustor, and consider site-specific factors including prescribed criteria to determine if the 1-hour average HCl-equivalent emission rate limit may be exceeded absent an hourly rolling average limit on the feedrate of total chlorine and chloride. If the 1-hour average HCl-equivalent emission rate limit may be exceeded, you must establish an hourly rolling average feedrate limit on total chlorine as provided by paragraph (f)(3) of this section.

(1) 1-hour average HCl-equivalent emission rate. You must calculate the 1hour average HCl-equivalent emission rate from the total chlorine emission concentration you select for each source as prescribed in paragraph (b)(6)(ii)(C) of this section.

(2) 1-hour average HCl-equivalent emission rate limit. You must establish the 1-hour average HCl-equivalent emission rate limit for each affected source using either a look-up table analysis or site-specific analysis:

(i) Look-up table analysis. Look-up tables are provided for 1-hour average HCl-equivalent emission rate limits as Table 3 and Table 4 to this section. Table 3 provides limits for facilities located in flat terrain. Table 4 provides limits for facilities located in simple elevated terrain. You must use the Tables to establish 1-hour average HCl-equivalent emission rate limits as prescribed in paragraphs (c)(3)(ii) through (c)(3)(v) of this section for annual average HCl-equivalent emission rate limits.

(ii) *Site-specific analysis.* The 1-hour average HCl-equivalent emission rate limit is the HCl-equivalent emission

rate that ensures that the Hazard Index associated with maximum 1-hour average exposures is not greater than 1.0 rounded to the nearest tenths decimal place (0.1). You must follow the risk assessment procedures under paragraph (c)(4) of this section to estimate shortterm inhalation exposures through the estimation of maximum 1-hour average ambient concentrations.

(3) Criteria for determining whether the 1-hour HCl-equivalent emission rate may be exceeded absent an hourly rolling average limit on the feedrate of total chlorine and chloride. An hourly rolling average feedrate limit on total chlorine and chloride is waived if you determine considering the criteria listed below that the long-term feedrate limit (and averaging period) established under paragraph (c)(4)(i) of this section will also ensure that the 1-hour average HCl-equivalent emission rate will not exceed the 1-hour average HCl-equivalent emission rate limit you calculate for each combustor.

(i) The ratio of the 1-hour average HCl-equivalent emission rate based on the total chlorine emission rate you select for each hazardous waste combustor to the 1-hour average HCl-equivalent emission rate limit for the combustor; and

(ii) The potential for the source to vary total chlorine and chloride feedrates substantially over the averaging period for the feedrate limit established under paragraph (c)(4)(i) of this section.

(e) Review and approval of eligibility demonstrations—(1) Content of the eligibility demonstration—(i) General. The eligibility demonstration must include the following information, at a minimum:

(A) Identification of each hazardous waste combustor combustion gas emission point (e.g., generally, the flue gas stack);

(B) The maximum and average capacity at which each combustor will operate, and the maximum rated capacity for each combustor, using the metric of stack gas volume (under both actual and standard conditions) emitted per unit of time, as well as any other metric that is appropriate for the combustor (e.g., million Btu/hr heat input 40 CFR Ch. I (7–1–23 Edition)

for boilers; tons of dry raw material feed/hour for cement kilns);

(C) Stack parameters for each combustor, including, but not limited to stack height, stack diameter, stack gas temperature, and stack gas exit velocity;

(D) Plot plan showing all stack emission points, nearby residences and property boundary line;

(E) Identification of any stack gas control devices used to reduce emissions from each combustor;

(F) Identification of the RfC values used to calculate annual average HClequivalent emission rates and the aREL values used to calculate 1-hour average HCl-equivalent emission rates;

(G) Calculations used to determine the annual average and 1-hour average HCl-equivalent emission rates and rate limits, including calculation of the $Cl_2/$ HCl ratios as prescribed by paragraph (b)(6) of this section;

(ii) Additional content to implement the annual average HCl-equivalent emission rate limit. You must include the following in your eligibility demonstration to implement the annual average HCl-equivalent emission rate limit:

(A) For incinerators, cement kilns, and lightweight aggregate kilns, calculations to confirm that the annual average HCl-equivalent emission rate that you calculate from the total chlorine emission rate you select for each combustor does not exceed the limits provided by paragraph (b)(7) of this section;

(B) Comparison of the annual average HCl-equivalent emission rate limit for each combustor to the annual average HCl-equivalent emission rate for the total chlorine emission rate you select for each combustor;

(C) The annual average HCl-equivalent emission rate limit for each hazardous waste combustor, and the limits on operating parameters required under paragraph (g)(1) of this section;

(D) Determination of the long-term chlorine feedrate limit, including the total chlorine system removal efficiency for sources that establish an (up to) annual rolling average feedrate limit under paragraph (g)(2)(ii) of this section;

(iii) Additional content to implement the I-hour average HCl-equivalent emission rate limit. You must include the following in your eligibility demonstration to implement the 1-hour average HCl-equivalent emission rate limit:

(A) Determination of whether the combustor may exceed the 1-hour HClequivalent emission rate limit absent an hourly rolling average chlorine feedrate limit, including:

(1) Determination of the 1-hour average HCl-equivalent emission rate from the total chlorine emission rate you select for the combustor;

(2) Determination of the 1-hour average HCl-equivalent emission rate limit using either look-up Tables 3 and 4 to this section or site-specific risk analysis;

(3) Determination of the ratio of the 1-hour average HCl-equivalent emission rate to the 1-hour average HClequivalent emission rate limit for the combustor; and

(4) The potential for the source to vary total chlorine and chloride feedrates substantially over the averaging period for the long-term feedrate limit established under paragraphs (g)(2)(i) and (g)(2)(i) of this section; and

(B) Determination of the hourly rolling average chlorine feedrate limit, including the total chlorine system removal efficiency.

(iv) Additional content of a look-up table demonstration. If you use the lookup table analysis to establish HClequivalent emission rate limits, your eligibility demonstration must also contain, at a minimum, the following:

(A) Documentation that the facility is located in either flat or simple elevated terrain; and

(B) For facilities with more than one on-site hazardous waste combustor, documentation that the sum of the ratios for all such combustors of the HClequivalent emission rate to the HClequivalent emission rate limit does not exceed 1.0.

(v) Additional content of a site-specific compliance demonstration. If you use a site-specific compliance demonstration, your eligibility demonstration must also contain, at a minimum, the following information to support your determination of the annual average HCl-equivalent emission rate limit for each combustor:

(A) Identification of the risk assessment methodology used;

(B) Documentation of the fate and transport model used;

(C) Documentation of the fate and transport model inputs, including the stack parameters listed in paragraph (d)(1)(i)(C) of this section converted to the dimensions required for the model;

(D) As applicable:

(1) Meteorological data;

(2) Building, land use, and terrain data;

(3) Receptor locations and population data, including areas where people congregate for work, school, or recreation; and

(4) Other facility-specific parameters input into the model;

(E) Documentation of the fate and transport model outputs; and

(F) Documentation of any exposure assessment and risk characterization calculations.

(2) Review and approval—(i) Existing sources. (A) If you operate an existing source, you must submit the eligibility demonstration to your permitting authority for review and approval not later than 12 months prior to the compliance date. You must also submit a separate copy of the eligibility demonstration to: U.S. EPA, Risk and Exposure Assessment Group, Emission Standards Division (C404–01), Attn: Group Leader, Research Triangle Park, North Carolina 27711, electronic mail address REAG@epa.gov.

(B) Your permitting authority should notify you of approval or intent to disapprove your eligibility demonstration within 6 months after receipt of the original demonstration, and within 3 months after receipt of any supplemental information that you submit. A notice of intent to disapprove your eligibility demonstration, whether before or after the compliance date, will identify incomplete or inaccurate information or noncompliance with prescribed procedures and specify how much time vou will have to submit additional information or to achieve the MACT standards for total chlorine under §63.1216, 63.1217, 63.1219, 63.1220, and 63.1221. If your eligibility demonstration is disapproved, the permitting authority may extend the compliance date of the total chlorine standards up to one year to allow you to make changes to the design or operation of the combustor or related systems as quickly as practicable to enable you to achieve compliance with the MACT total chlorine standards.

(C) If your permitting authority has not approved your eligibility demonstration by the compliance date, and has not issued a notice of intent to disapprove your demonstration, you may begin complying, on the compliance date, with the HCl-equivalent emission rate limits you present in your eligibility demonstration provided that you have made a good faith effort to provide complete and accurate information and to respond to any requests for additional information in a timely manner. If the permitting authority believes that you have not made a good faith effort to provide complete and accurate information or to respond to any requests for additional information, however, the authority may notify you in writing by the compliance date that you have not met the conditions for complying with the healthbased compliance alternative without prior approval. Such notice will explain the basis for concluding that you have not made a good faith effort to comply with the health-based compliance alternative by the compliance date.

(D) If your permitting authority issues a notice of intent to disapprove your eligibility demonstration after the compliance date, the authority will identify the basis for that notice and specify how much time you will have to submit additional information or to comply with the MACT standards for total chlorine under §§63.1216, 63.1217, 63.1219, 63.1220, and 63.1221. The permitting authority may extend the compliance date of the total chlorine standards up to one-year to allow you to make changes to the design or operation of the combustor or related systems as quickly as practicable to enable you to achieve compliance with the MACT standards for total chlorine.

(ii) New or reconstructed sources—(A) General. The procedures for review and

40 CFR Ch. I (7–1–23 Edition)

approval of eligibility demonstrations applicable to existing sources under paragraph (e)(2)(i) of this section also apply to new or reconstructed sources, except that the date you must submit the eligibility demonstration is as prescribed in this paragraph (e)(2)(ii).

(B) If you operate a new or reconstructed source that starts up before April 12, 2007, or a solid fuel boiler or liquid fuel boiler that is an area source that increases its emissions or its potential to emit such that it becomes a major source of HAP before April 12, 2007, you must either:

(1) Comply with the final total chlorine emission standards under \S 63.1216, 63.1217, 63.1219, 63.1220, and 63.1221, by October 12, 2005, or upon startup, whichever is later, except for a standard that is more stringent than the standard proposed on April 20, 2004 for your source. If a final standard is more stringent than the proposed standard, you may comply with the proposed standard until October 14, 2008, after which you must comply with the final standard; or

(2) Submit an eligibility demonstration for review and approval under this section by April 12, 2006, and comply with the HCl-equivalent emission rate limits and operating requirements you establish in the eligibility demonstration.

(C) If you operate a new or reconstructed source that starts up on or after April 12, 2007, or a solid fuel boiler or liquid fuel boiler that is an area source that increases its emissions or its potential to emit such that it becomes a major source of HAP on or after April 12, 2007, you must either:

(1) Comply with the final total chlorine emission standards under §§ 63.1216, 63.1217, 63.1219, 63.1220, and 63.1221 upon startup. If the final standard is more stringent than the standard proposed for your source on April 20, 2004, however, and if you start operations before October 14, 2008, you may comply with the proposed standard until October 14, 2008, after which you must comply with the final standard; or

(2) Submit an eligibility demonstration for review and approval under this section 12 months prior to startup.

(3) The operating requirements in the eligibility demonstration are applicable requirements for purposes of parts 70 and 71 of this chapter and will be incorporated in the title V permit.

(f) Testing requirements—(1) General. You must comply with the requirements for comprehensive performance testing under §63.1207.

(2) System removal efficiency. (i) You must calculate the total chlorine removal efficiency of the combustor during each run of the comprehensive performance test.

(ii) You must calculate the average system removal efficiency as the average of the test run averages.

(iii) If your source does not control emissions of total chlorine, you must assume zero system removal efficiency.

(3) Annual average HCl-equivalent emission rate limit. If emissions during the comprehensive performance test exceed the annual average HCl-equivalent emission rate limit, eligibility for emission limits under this section is not affected. This emission rate limit is an annual average limit even though compliance is based on a 12-hour or (up to) an annual rolling average feedrate limit on total chlorine and chloride because the feedrate limit is also used for assurance compliance for the semivolatile metal emission standard

(4) 1-hour average HCl-equivalent emission rate limit. Total chlorine emissions during each run of the comprehensive performance test cannot exceed the 1hour average HCl-equivalent emission rate limit.

(5) Test methods. (i) If you operate a cement kiln or a combustor equipped with a dry acid gas scrubber, you must use EPA Method 320/321 or ASTM D 6735-01, or an equivalent method, to measure hydrogen chloride, and the back-half (caustic impingers) of Method 26/26A, or an equivalent method, to measure chlorine gas.

(ii) Bromine and sulfur considerations. If you operate an incinerator, boiler, or lightweight aggregate kiln and your feedstreams contain bromine or sulfur during the comprehensive performance test at levels specified under paragraph (e)(2)(ii)(B) of this section, you must use EPA Method 320/321 or ASTM D 6735-01, or an equivalent method, to measure hydrogen chloride, and Method 26/26A, or an equivalent method, to measure chlorine and hydrogen chloride, and determine your chlorine emissions as follows:

(A) You must determine your chlorine emissions to be the higher of the value measured by Method 26/26A as provided in appendix A-8, part 60 of this chapter, or an equivalent method, or the value calculated by the difference between the combined hydrogen chloride and chlorine levels measured by Method 26/26A as provided in appendix A-8, part 60 of this chapter, or an equivalent method, and the hydrogen chloride measurement from EPA Method 320/321 as provided in appendix A, part 63 of this chapter, or ASTM D 6735 - 01as described under §63.1208(b)(5)(i)(C), or an equivalent method.

(B) The procedures under paragraph (f)(2)(ii) of this section for determining hydrogen chloride and chlorine emissions apply if you feed bromine or sulfur during the performance test at the levels specified in this paragraph (f)(5)(ii)(B):

(1) If the bromine/chlorine ratio in feedstreams is greater than 5 percent by mass; or

(2) If the sulfur/chlorine ratio in feedstreams is greater than 50 percent by mass.

(g) Monitoring requirements—(1) General. You must establish and comply with limits on the same operating parameters that apply to sources complying with the MACT standard for total chlorine under $\S63.1209(0)$, except that feedrate limits on total chlorine and chloride must be established according to paragraphs (g)(2) and (g)(3) of this section:

(2) Feedrate limit to ensure compliance with the annual average HCl-equivalent emission rate limit. (i) For sources subject to the feedrate limit for total chlorine and chloride under $\S63.1209(n)(4)$ to ensure compliance with the semivolatile metals standard:

(A) The feedrate limit (and averaging period) for total chlorine and chloride to ensure compliance with the annual average HCl-equivalent emission rate limit is the same as required by $\S63.1209(n)(4)$, except as provided by paragraph (g)(2)(i)(B) of this section.

(B) The numerical value of the total chlorine and chloride feedrate limit (i.e., not considering the averaging period) you establish under 63.1209(n)(4) must not exceed the value you calculate as the annual average HCl-equivalent emission rate limit (lb/hr) divided by [1 - system removal efficiency], where the system removal efficiency is calculated as prescribed by paragraph (f)(2) of this section.

(ii) For sources exempt from the feedrate limit for total chlorine and chloride under 63.1209(n)(4) because they comply with 63.1207(m)(2), the feedrate limit for total chlorine and chloride to ensure compliance with the annual average HCl-equivalent emission rate must be established as follows:

(A) You must establish an average period for the feedrate limit that does not exceed an annual rolling average;

(B) The numerical value of the total chlorine and chloride feedrate limit (i.e., not considering the averaging period) must not exceed the value you calculate as the annual average HCl-equivalent emission rate limit (lb/hr) divided by [1 - system removal efficiency], where the system removal efficiency is calculated as prescribed by paragraph (f)(2) of this section.

(C) You must calculate the initial rolling average as though you had selected a 12-hour rolling average, as provided by paragraph (b)(5)(i) of this section. You must calculate rolling averages thereafter as the average of the available one-minute values until enough one-minute values are available to calculate the rolling average period you select. At that time and thereafter, you update the rolling average feedrate each hour with a 60-minute average feedrate.

(3) Feedrate limit to ensure compliance with the 1-hour average HCl-equivalent emission rate limit. (i) You must establish an hourly rolling average feedrate limit on total chlorine and chloride to ensure compliance with the 1-hour average HCl-equivalent emission rate limit unless you determine that the hourly rolling average feedrate limit is waived under paragraph (d) of this section.

(ii) You must calculate the hourly rolling average feedrate limit for total

40 CFR Ch. I (7–1–23 Edition)

chlorine and chloride as the 1-hour average HCl-equivalent emission rate limit (lb/hr) divided by [1 - system removal efficiency], where the system removal efficiency is calculated as prescribed by paragraph (f)(2)(ii) of this section.

(h) Changes—(1) Changes over which you have control—(i) Changes that would affect the HCl-equivalent emission rate limit. (A) If you plan to change the design, operation, or maintenance of the facility in a manner than would decrease the annual average or 1-hour average HCl-equivalent emission rate limit, you must submit to the permitting authority prior to the change a revised eligibility demonstration documenting the lower emission rate limits and calculations of reduced total chlorine and chloride feedrate limits.

(B) If you plan to change the design, operation, or maintenance of the facility in a manner than would increase the annual average or 1-hour average HCl-equivalent emission rate limit, and you elect to increase your total chlorine and chloride feedrate limits. You must also submit to the permitting authority prior to the change a revised eligibility demonstration documenting the increased emission rate limits and calculations of the increased feedrate limits prior to the change.

(ii) Changes that could affect system removal efficiency. (A) If you plan to change the design, operation, or maintenance of the combustor in a manner than could decrease the system removal efficiency, you are subject to the requirements of §63.1206(b)(5) for conducting a performance test to reestablish the combustor's system removal efficiency and you must submit a revised eligibility demonstration documenting the lower system removal efficiency and the reduced feedrate limits on total chlorine and chloride.

(B) If you plan to change the design, operation, or maintenance of the combustor in a manner than could increase the system removal efficiency, and you elect to document the increased system removal efficiency to establish higher feedrate limits on total chlorine and chloride, you are subject to the requirements of §63.1206(b)(5) for conducting a performance test to reestablish the combustor's system removal

efficiency. You must also submit to the permitting authority a revised eligibility demonstration documenting the higher system removal efficiency and the increased feedrate limits on total chlorine and chloride.

(2) Changes over which you do not have control that may decrease the HCl-equivalent emission rate limits. These requirements apply if you use a site-specific risk assessment under paragraph (c)(4) of this section to demonstrate eligibility for the health-based limits.

(i) *Proactive review*. You must submit for review and approval with each comprehensive performance test plan either a certification that the information used in your eligibility demonstration has not changed in a manner that would decrease the annual average or 1-hour average HCl-equivalent emission rate limit, or a revised eligibility demonstration. (ii) *Reactive review*. If in the interim between your comprehensive performance tests you have reason to know of changes that would decrease the annual average or 1-hour average HClequivalent emission rate limit, you must submit a revised eligibility demonstration as soon as practicable but not more frequently than annually.

(iii) Compliance schedule. If you determine that you cannot demonstrate compliance with a lower annual average HCl-equivalent emission rate limit during the comprehensive performance test because you need additional time to complete changes to the design or operation of the source, you may request that the permitting authority grant you additional time to make those changes as quickly as practicable.

§63.1215

§63.	1215
------	------

40 CFR Ch. I (7-1-23 Edition)

	able 1 of	563.1215	Annual	Average	HCI-Equ	e to prope	rty bound:	tate Lim	its (Ib/hr)	-Flat Ter	rain	
Stack Diameter =	= 0.3 m											
Stack Height (m)	30	50	70	100	200	300	500	700	1000	2000	3000	5000
9	3.7E-01	4,86-01	7.36-01	9.1E-01	1,6E+00	2.36+00	4.1E+00	5.7E+00	6.1E+00	1.06+01	1.6E+01	2.96+01
1	1.0€+00	1,0E+00	1.1E+00	1,56+00	Z.1E+00	2.7E+00	4.8E+00	5.7E+00	6.5E+00	1.1E+01	1.8E+01	3.2E+01
50	2.3E+00	2,3E+00	2.36+00	2.3E+00	2.7E+00	3.7E+00	5,65+00	7.4E+00	1,06+01	1.96+01	2,96+01	5,26+01
8	4.1E+00	4.1E+00	4.1E+00	4.2E+00	4.7E+00	6.0E+00	3.5E+00	1,36401	1.86+01	3.35+01	4,86+01	7,86+01
8	1.2E+01	1,2E+01	1.2E+01	1.2E+01	1.36+01	1.5E+01	2,0E+01	2.8E+01	3.8E+01	7.1E+01	1:0E+02	1,66+02
Stack Diameter	m 0.0 m										-	
Stack Height (m)	30	20	70	100	200	300	500	700	1000	2000	3000	2000
5	6.5E-01	9.36-01	1.4E400	1.BE+00	3.0E+00	4.4E+00	7,2E+00	9.2E+00	1,36+01	1.5E+01	2,0E+01	3.4E+01
10	1.4E+00	1.4E+00	1.6E+00	2.1E+00	3.9E+00	5.4E+00	8,3E+00	1.0E+01	1.36+01	1.7E+01	2,3E+01	3,85+01
20	3,7E+00	3,7E+00	3.7E+00	3.96+00	4.9E+00	6.5E+00	8.5E+00	1.05+01	1.3E+01	2.2E+01	3.2E+01	5.5E+01
80	5.5E+00	5.5E+00	5,5E+00	5,5E+00	5.6E+00	6,7E+00.	1.05+01	1.46+01	1.96+01	3.4E+01	4.9E+01	8.1E+01
Stark Diamotor	1.4E+01	1.4E+01	1.4E+01	1.4E+01	1.46+01	1,66+01	2.1E+01	2,86+01	3,96+01	7.2E+01	1,0E+02	1.6E+02
Stack Union tere	m 01 -			444							4048	
full wifield where		20	2	not	200	nor	noc	100	0001	0007	2000	0000
10	3.2E+00	3.65+00	4.0E+00	5.4E+00	9.6E+00	1,3E+01	1.8E+01	2.3E+01	2.8E+01	4.5E+01	5.3E+01	6.5E+01
02	5.9E+00	5,9E+00	5.0E+00	6.1E+00	8,65+00	1.3E+01	1.86+01	2.3E+01	2.8E+01	4.5E+01	5.3E+01	7,56+01
00	1.0E+01	1.06+01	1.05+01	1.0E+01	1,2E+01	1.3E+01	1.8E+01	2.3E+01	2.8E+01	4.5E+01	6.1E+01	9.3E+01
00	1.85+01	1.86+01	1,86+01	1.8E+01	1.85+01-	1.86+01	2.3E+01	3.1E+01	4.2E+01	7.7E+01	1.1E+02	1.7E+02
0	7.4E+01	7.46+01	7.45+01	7.4E+01	7.4E+01	7.4E+01	8.0E+01	1.0E+02	1.4E+02	2.1E+02	2.7E+02	4,0E+02
Stack Diamoter	= 1.0 m	~								-		-
In tufficut view	141100	00	o server	100 ×	200	000	200	00/	DODL .	2000	2000	2000
90	4.1EHUU	0,36100	0.45+00	00+38"/	1.35401	216401	2./E+01	3.6E+01	4.86+01	10+30"L	9.16+01	1.1E+02
20	100+00	1.04-30.7	1.66400	('BE+00	1.36+01	2.1E+01	2.7E+01	3.6E+01	4.85+01	7.65+01	9,1E+01	1.2E+02
8	1.35+01	1.35+01	1.36+01	1.35+01	1.6E+01	21E+01	2.7E+01	3.6E+01	4,85+01	7,6E+01	9,1E+01	1,2E+02
8	2.36+01	2.3E+01	2,3E+01	2.3E+01	2.3E+01	2.36+01	2.7E+01	3.6E+01	4.8E+01	8.65+01	1.2E+02	1.8E+02
DI DI	1.00=+02	1.05+02	1.0E+02	1.0E+02	1,0E+02	1.0E+02	1.1E+02	1.4E+02	1.8E+02	3.0E+02	4.0E+02	5,8E+02
Stack Ulameter	= 2.0 m											
STACK Height (m)	30	20	2	100	200	300	200	100	1000	2000	3000	2000
10	5.0E+00	6.3E+00	7.7E+00	9.8E+00	1.76+01	2.8E+01	3.3E+01	4.4E+01	5.9E+01	1.0€+02	1.4E+02	1.6E+02
87	9.3E+00	00+3E+00	9.4E+00	1.0E+01	1.7E+01	2.8E+01	3.3E+01	4,4E+01	5.9E+01	1.06+02	1,4E+02	1.8E+02
00	1.05+01	1.0E+01	1,6E+01	1.66+01	1.06+01	2.8E+01	3.3E+01	4.4E+01	5.9E+01	1.06+02	1.4E+02	1.8E+02
8	2.96+01	2.9E+01	2,9E+01	2,96+01	2.96+01	2.9E+01	3.3E+01	4.4E+01	5.9E+01	1.0E+02	1.4E+02	2.0E+02
0	1.45+02	1.45+02	1.4E+U2	1.4E+02	1,45+02	1.4E+02	1.4E+02	1,8E+02	2.3E+02	3.4E+02	4,3E+02	6,4E+02
101	3.05102	3.0E+02	3.0E+02	3.0E+02	3.0E+02	3.0E+02	3.0E+02	3.0E+02	3,56+02	5.2E+02	6,85+02	8,2E+02
Stack Ulameter	= 3.0 m											
Stack Height (m)	30	8	70	100	200	300	500	700	1000	2000	3000	2000
01	6.5E+00	6.9E+00	7.7E+00	00+38'6	2.2E+01	3,45+01	5,4E+01	7.4E+01	9.85+01	1,3E+02	1,6E+02	1.6€+02
02	1.62+01	1.66+01	1.76+01	2.0E+01	2.5E+01	3,76+01	5,65+01	7.4E+01	9.85+01	1,5E+02	2.1E+02	20430'8
or	2.0E+01	2,0E+01	2.0E+01	2.06+01	2.5E+01	3.7E+01	5.6E+01	7.4E+01	9.8E+01	1.7E+02	2.2E+02	3.06+02
05	4.2E+01	4.2E+01	4.25+01	4.2E+01	4.45+01	5.16+01	5.66+01	7.4E+01	9.8E+01	1.7E+02	2.2E+02	3.0E+02
02	2 3E+02	2.3E+02	2.3E+02	2.3E+02	2.3E+02	2.4E+02	2.4E+02	2.9E+02	3.66+02	4.1E+02	5.0E+02	7.0€+02
100	3.5E+02	3.5E+02	3.6E+02	3,5E+02	3,5E+02	3,5E+02	3.5E+02	3.5E+02	3.9E+02	6,3E+02	7,5E+02	8.7E+02
Stack Diameter	= 4.0 m											
Stack Height (m)	30	50	20	100	200	300	500	700	1000	2000	3000	5000
30	2,56+01	2.5E+01	2.5E+01	2.56+01	3.4E+01	5.6E+01	8.1E+01	1.1E+02	1.4E+02	2.2E+02	2.8E+02	4,3E+02
50	5.1E+01	5.1E+01	5.1E+01	5.1E+01	6.3E+01	6.2E+01	8,1E+01	1.1E+02	1.4E+02	2.4E+02	3.1E+02	4.4E+02
70	2.6E+02	2.6E+02	2.6E+02	2.6E+02	2.76+02	2.85+02	3.3E+02-	4.8E+02	4,8E+02	5.0E+02	5.7E+02	7.7E+02
100	5.7E+02	5.7E+02	5.7E+02	5.7E+02	5.7E+02	5.7E+02	5.7E+02	5.7E+02	5.8E+02	B.6E+02	9.3E+02	1.15+03

§63.1215

				0	Dist	ance to prope	rty boundary	(III)	Andreas (1 11 411	
Stack Diameter = 0.3 m												
Stack height (m)	30	50	, 70	100	200	300	500	100	1000	2000	3000	5000
8	1.3E-01	1.8E-01	2.5E-01	3.7E-01	6.4E-01	8.9E-01	1.4E+00	2.0E+00	3.1E+00	7.7E+00	1.3E+01	2.6E+01
10	3.8E-01	3.8E-01	4.4E-01	6.1E-01	6.4E-01	10-36.8	1.4E+00	2.0E+00	3.1E+00	7.7E+00	1.3E+01	2.6E+01
07	1,1E+00.	1.1E+00	1.1E+00	1.2E+00	1.2E+00	1.5E+00	2.3E+00	3.4E+00	5.2E+00	1.2E+01	2.0E+01	3.9E+01
50	7.78:400	7 7E+00	7.75+00	7.78400	7 78400	3.5E+00	4.2E+00	5.2E+00	7.0E+00	1.5E+01	2.6E+01	4.9E+01
Stack Diameter = 0.5 m				AA-911	M-71	0.0110.0	0.02100	001000	0.051100	2.05701	3.45701	10120.0
Stack height (m)	30	50	70	100	200	300	500	700	1000	2000	3000	5000
5	1.8E-01	2.6E-01	3.5E-01	5.6E-01	1.4E+00	1.6E+00	2.3E+00	3.4E+00	5.2E+00	9.6E+00	1.5E+01	2.8E+01
10	5.3E-01	5.3E-01	6.1E-01	8.5E-01	1.4E+00	1.6E+00	2.3E+00	3.4E+00	5.2E+00	9.6E+00	1.5E+01	2.8E+01
20	1.5E+00	1.5E+00	1.5E+00	1.5E+00	1.5E+00	1.6E+00	2.3E+00	3,4E+00	5.2E+00	1.2E+01	2.0E+01	3.9E+01
30	2.9E+00	2.9E+00	2.9E+00	2.9E+00	2.9E+00	3.5E+00	4.2E+00	5.5E+00	8.1E+00	1.7E+01	2.8E+01	5.2E+01
Stack Diameter = 1.0 m	0,012100	0.015100	0.00100	8.012+00	8.00+00	8.8E+00	1.2E+01	1.2E+01	1.2E+01	2.3E+01	3.7E+01	6.9E+01
Stack height (m)	30	50	70	100	200	300	500	700	1000	0006	1000	6000
10	9.7E-01	9.7E-01	1.1E+00	1.7E+00	3.7E+00	3.7E+00	4.2E+00	5 5E+00	7.5E+00	1 5P+01	10+46.6	4 1 F + 01
20	2.7E+00	2.7E+00	2.7E+00	3.0E+00	3.7E+00	3.7E+00	4.2E+00	5.5E+00	7.5E+00	1.5E+01	2.3E+01	4.3E+01
30	4.3E+00	4.3E+00	4.3E+00	4.3E+00	4.3E+00	4.3E+00	4.3E+00	5.5E+00	8.1E+00	1.7E+01	2.8E+01	5.2E+01
50	9.5E+00	9.5E+00	9.5E+00	9.5E+00	9.5E+00	9.5E+00	1.2E+01	1.4E+01	1.6E+01	3.1E+01	4.8E+01	8.3E+01
70	4.0E+01	4.0E+01	4.0E+01	4.0E+01	4.0E+01	4.0E+01	4.0E+01	4.1E+01	4.1E+01	4.1E+01	5.8E+01	· 9.8E+01
Stack Diameter = 1.5 m												
Stack height (m)	30	50	70	100	200	300	500	700	1000	2000	3000	5000
10	2.0E+00	2.0E+00	2.3E+00	3.4E+00	5.1E+00	6.0E+00	6.0E+00	6.6E+00	9.3E+00	1.9E+01	3.0E+01	5.4E+01
07	3.5E+00	3.5E+00	3.5E+00	3.9E+00	5.1E+00	6.0E+00	6.0E+00	6.6E+00	9.3E+00	1.9E+01	3.0E+01	5.4E+01
00	0.0E+00	0.0E+00	6.0E+00	6.0E+00	6.0E+00	6.0E+00	6.0E+00	6.6E+00	9.3E+00	1.9E+01	3.0E+01	5.5E+01
02	10+31.1	10+31-1	1.1E+01	1.1E+01	1.1E+01	1.1E+01	1.2E+01	1.4E+01	1.6E+01	3.1E+01	4.8E+01	8.3E+01
Stack Diameter = 2.0 m	INTEL:C	10-31°c	10431.0	10+31.c	5.1E+01	5.IE+01	5.1E+01	5.1E+01	5.1E+01	6.2E+01	7.8E+01	1.2E+02
Stack height (m)	30	50	70	100	100	000	004	000	~~~~			
10	C KETOO	NATES C	N AD AD	1 OD TOO	007	300	000	0.0/	1000	2000	3000	5000
20	4.2E+00	4.2E+00	4.2E+00	4.7E+00	6.3E+00	9.28+00	9.2E+00	1.0E+01	1.4E+01	2.5E+01	3.7E+01	6.3E+01
30	8.4E+00	8.4E+00	8.4E+00	8.4E+00	9.2E+00	9.2E+00	9.2E+00	1.0E+01	1.4E+01	2.58+01	3.7E+01	638401
50	1.4E+01	1.4E+01	1.4E+01	1.4E+01	1.4E+01	1.4E+01	1.4E+01	1.5E+01	1.6E+01	3.1E+01	4.8E+01	8.3E+01
70	5.9E+01	5.9E+01	5.9E+01	5.9E+01	5.9E+01	5.9E+01	5.9E+01	5.9E+01	5.9E+01	7.0E+01	1.0E+02	1.5E+02
100	8.2E+01	8.2E+01	8.2E+01	8.2E+01	8.2E+01	8.2E+01	8.2E+01	8.2E+01	8.2E+01	8.2E+01	1.1E+02	1.7E+02
Stack Diameter = 3.0 m												
Stack height (m)	30	50	70	100	200	300	500	700	1000	2000	3000	5000
10	3.3E+00	3.4E+00	3.9E+00	5.5E+00	1.IE+01	1.7E+01	1.7E+01	1.7E+01	1.7B+01	3.3E+01	5.0E+01	8.6E+01
07	112101	00+30.0	004959	7.6E+00	1.1E+01	1.7E+01	1.7E+01	1.7E+01	1.7E+01	3.3E+01	5.0E+01	8.6E+01
50	1.76401	101211	1 78401	1010101	1.75+01	1.75+01	1.78+01	1.7E+01	1.7E+01	3.3E+01	5.0E+01	8.6E+01
70	8.0E+01	8.0E+01	8.0F+01	8 0F401	8 0E+01	101211	010101	10/E+01	0.05101	3.35+01	0.0001	8.06+01
100	1.3E+02	1.3E+02	1.3E+02	· 1.3E+02	1.3E+02	1.3E+02	1 3F+02	1 38402	1 36+02	1 35407	01-07-1	2 48402
Stack Diameter = 4.0 m												
Stack height (m)	30	50	70	100	200	300	500	700	1000	2000	3000	5000
30	- 1.3E+01	1.3E+01	1.3E+01	1.3E+01	1.5E+01	2.1E+01	2.1E+01	2.1E+01	2.1E+01	4.0E+01	6.0E+01	9.8E+01
50	2.1E+01	2.1E+0]	2.1E+01	2.1E+01	2.1E+01	2.1E+01	2.1E+01	2.1E+01	2.1E+01	4.0E+01	6.0E+01	9.8E+01
100	1.15+02	1.1E+02	1.1E+02	1.1E+02	1.1E+02	1.1E+02	1.1E+02	1.1E+02	1.1E+02	1.1E+02	1.5E+02	2.3E+02
Abr	20126.1	2012C.1	1.35702	1.58+02	1.5E+02	1.5E+02	1.5E+02	1.5E+02	1.5E+02	1.5E+02	2.2E+02	3.4E+02

§63.1215

40 CFR Ch. I (7-1-23 Edition)

	Table 3	of §63.1.	215: 1-Hc	our Avera	Ige HCI-E	quivalen	t Emissic	on Rates	(lb/hr)-F	lat Terra	.=	
Ctack Dismeter					Distano	ce to prope	erty bound	ary (m)				
Stack Diameter	m 20.3 m											
Stack Height (m)	30	50	22.102	100	200	300	500	200	1000	2000	3000	5000
	0.454.0	0.1E+00	1.00+00	8.6E+00	1.66+01	2.4E+01	4.3E+01	5.3E+01	6.2E+01	1.1E+02	1.7E+02	3.1E+02
20	9.7E+00	9.8E+00	1.1E+01	1.4E+01	2.0E+01	2.6E+01	4.6E+01	5.3E+01	6.2E+01	1.1E+02	1.7E+02	3.1E+02
30	3.96+01	3.96+01	3.95+01	4 0F+01	4 4E+01	5.7E404	0.05101	1 26403	1.75100	1.85+02	2.85+02	4.9E+02
3	1 2E+02	1.2E+02	1 25+02	1 26402	1 26402	1 46405	1 06400	2 661.00	2 6E 100	3.7E+UZ	4.0E+U2	1.05+02
Stack Diameter =	- 0.5 m			20. 22.		1.441.44	201261	2.00102	3.DETUK	D.I ETUZ	A.I.CTUZ	1.05703
Stack Height (m)	30	50	02.	100	200	300	500	700	1000	2000	3000	5000
5	6.9E+00	9.8E+00	1.5E+01	1.8E+01	3.2E+01	4.6E+01	7.5E+01	9.7E+01	1.2E+02	1.6E+02	2.1E+02	3.6E+02
10	1.3E+01	1.4E+01	1.5E+01	2.0E+01	3.7E+01	5.1E+01	7.9E+01	9.7E+01	1.2E+02	1.6E+02	2.2E+02	3.6E+02
20	3.5E+01	3.5E+01	3.5E+01	3.6E+01	4.6E+01	6.2E+01	8.1E+01	9.7E+01	1.2E+02	2.1E+02	3.0E+02	5.2E+02
30	5.2E+01	5.2E+01	5.2E+01	5.2E+01	5.3E+01	6.4E+01	9.8E+01	1.3E+02	1.6E+02	3.2E+02	4.7E+02	7.7E+02
50 Stack Diameter =	1.3E+02	1.3E+02	1.3E+02	1.3E+02	1.3E+02	1.4E+02	2.0E+02	2.7E+02	3.7E+02	6.8E+02	9.7E+02	1.5E+03
Stack Helaht (m)	30	92	70	100	VUC	006	002	AAA		0000		
10	3.0E+01	3.46+01	3.8F+01	5 1F+01	9 DF+01	1 25400	1 76400	M/ DC C	MUL C	A SCLOD	2002	0000
20	5.5E+01	5.5E+01	5.5E+01	5.85+01	9 0F+04	1 25402	1 76402	0.00400	2.7E400	4 36-00	201210	211210
30	9.6E+01	9.6E+01	9.66+01	9.6E+01	1 1E+02	1 25402	1 7F402	2 2E400	27E100	A 36-00	COLETON	0.051/0
50	1.7E+02	1.7E+02	1.7E+02	1.7E+02	1.7E+02	1.7E+02	2 2F402	0 05400	4 0F402	7 36402	1 05103	1 65103
20	7.0E+02	7.0E+02	7.0E+02	7.0E+02	7.0E+02	7.0E+02	7.6E+02	0.95+02	135+03	2.0F-103	2 6F403	3.85403
Stack Diameter -	= 1.5 m								20. 10.			20.122
Stack Height (m)	30	50	02	100	200	300	500	700	1000	2000	3000	5000
10	3.9E+01	5.0E+01	6.1E+01	- 7.5E+01	1.2E+02	2.0E+02	2.5E+02	3.4E+02	4.6E+02	7.2E+02	8.6E+02	1.0E+03
20	7.1E+01	7.1E+01	7.2E+01	7.5E+01	1.2E+02	2.0E+02	2.5E+02	3.4E+02	4.6E+02	7.2E+02	8.6E+02	1.1E+03
30	1.2E+02	1.2E+02	1.2E+02	1.2E+02	1.5E+02	2.0E+02	2.5E+02	3.4E+02	4.6E+02	7.2E+02	8.6E+02	1.1E+03
50	2.2E+02	2.2E+02	2.2E+02	2.2E+02	2.2E+02	2.2E+02	2.5E+02	3.4E+02	4.6E+02	~8.1E+02	1.1E+03	1.7E+03
20.1.0	9.6E+02	9.6E+02	9.6E+02	9.6E+02	9.6E+02	9.6E+02	1.0E+03	1.3E+03	1.7E+03	2.9E+03	3.8E+03	5.5E+03
Stack Diameter	= 2.0 m											
Stack Height (m)	8	20	2	100	200	300	500	700	1000	2000	3000	5000
10	4.7E+01	6.0E+01	7.3E+01	9.2E+01	1.7E+02	2.6E+02	3.2E+02	4.2E+02	5.6E+02	9.7E+02	1.3E+03	1.5E+03
70	8,8E+01	8.85+01	8.8E+01	9.4E+01	1.7E+02	2.6E+02	3.2E+02	4.2E+02	5.6E+02	9.7E+02	1.3E+03	1.7E+03
05	1.0E+02	1.5E+02	1.5E+02	1.5E+02	1.8E+02	2.6E+02	3.2E+02	4.2E+02	5.6€+02	9.7E+02	1.3E+03	1.7E+03
70	1.3F+03	1 35+03	1 36403	1 36103	1 26403	1 95100	3.2E+UZ	4.2E+02	5.0E+02	9.7E+02	1.3E+03	1.9E+03
100	2.8E+03	2.8E+03	2.8E+03	2.8E+03	2.8E+03	2 8E+03	2.8E+03	2 85403	3 35403	5.0E403	R READS	7 76-00
Stack Diameter -	= 3.0 m								20.100	~~~~~	201-102	20111
Stack Height (m)	30	20	70	100	200	300	500	700	1000	2000	3000	5000
10	6.2E+01	6.5E+01	7.3E+01	9.2E+01	2.1E+02	3.3E+02	5.1E+02	7.0E+02	9.3E+02	1.2E+03	1.5E+03	1.5E+03
20	1.5E+02	1.5E+02	1.6E+02	1.9E+02	2.4E+02	3.5E+02	5.3E+02	7.0E+02	9.3E+02	1.4E+03	2.0E+03	2.8E+03
30	1.9E+02	1.9E+02	1.9E+02	1.9E+02	2.4E+02	3.5E+02	5.3E+02	7.0E+02	9.3E+02	1.6E+03	2.1E+03	2.8E+03
8	4.0E+02	4.0E+02	4.0E+02	4.0E+02	4.2E+02	4.8E+02	5.3E+02	7.0E+02	9.3E+02	1.6E+03	2.1E+03	2.8E+03
02	2.2E+03	2.2E+03	2.2E+03	2.2E+03	2.2E+03	2.3E+03	2.3E+03	2.8E+03	3.4E+03	3.9E+03	4.7E+03	6.6E+03
Chack Dismeter	3.3E+03	3.3E+03	3.3E+03	3.3E+03	3.3E+03	3.3E+03	3.3E+03	3.3E+03	3.7E+03	6.0E+03	7.1E+03	8.2E+03
Stack Diameter	= 4.0 m											
Stack Height (m)	8	8	20	10	200	300	500	200	1000	2000	3000	5000
8	2.3E+02	2.3E+02	2.3E+02	2.4E+02	3.2E+02	5.3E+02	7.7E+02	1.0E+03	1.3E+03	2.1E+03	2.6E+03,	4.1E+03
02	9.0E#04	4.8E+UZ	4.85402	4.8E+02	5.0E+02	5,8E+02	7.7E+02	1.0E+03	1.3E+03	2.3E+03	3.0E+03	4.2E+03
404	2.45400	2.45+03	Z 4E+03	2.4E+03	2.5E+03	2.6E+03	3.2E+03	4.3E+03	4.5E+03	4.7E+03	5.4E+03	7.2E+03
DUL	5.4E+03	5.4E+03	5.4E+03	5.4E+03	5.4E+03	5.4F+03	545+03	F.4F+03	5 KEANS	0 45103	0.95403	* DELVA

§63.1216

Table 4	of §63.12	215: 1-Ho	our Avera	ge HCI-E	quivalen	t Emissic	n Rate L	imits (Ib/	hr)Simp	le Elevat	ed Terrai	_
Stack Diameter =	0.3 m				Distanc	ce to prope	rty bounds	ary (m)				
Stack Height (m)	30	50	20	100	200	300	200	700	1000	2000	3000	5000
22	1.4E+00	1.9E+00 -	2.6E+00	3.8E+00	6.8E+00	9.4E+00	1.5E+01	2.1E+01	3.3E+01	8.1E+01	1.4E+02	2.7E+02
10	4.0E+00	4.0E+00	4.6E+00	6.4E+00	6.8E+00	9.4E+00	1.5E+01	2.1E+01	3.3E+01	8.1E+01	1.4E+02	2.7E+02
20	1.1E+01	1.1E+01	1.1E+01	1.1E+01	1.2E+01	1.5E+01	2.4E+01	3.5E+01	5.4E+01	1.3E+02	2.1E+02	4.0E+02
30	2.3E+01	2.3E+01	2.3E+01	2.3E+01	2.5E+01	3.3E+01	4.4E+01	5.5E+01	7.3E+01	1.6E+02	2.7E+02	5.2E+02
8	7.3E+01	7.3E+01	7.3E+01	7.3E+01	7.3E+01	8.3E+01	9.0E+01	9.0E+01	9.0E+01	2.1E+02	3.5E+02	6.8E+02
Stack Diameter =	: 0.5 m											
Stack Height (m)	30	50	70	100	200	300	500	700	1000	2000	3000	5000
5	1.9E+00	2.7E+00	3.7E+00	5.9E+00	1.4E+01	1.7E+01	2.4E+01	3.5E+01	5.4E+01	1.0E+02	1.6E+02	3.0E+02
10	5.6E+00	5.6E+00	6.4E+00	8.9E+00	1.4E+01	1.7E+01	2.4E+01	3.5E+01	5.4E+01	1.0E+02	1.6E+02	3.0E+02
20	1.6E+01	1.6E+01	1.6E+01	1.6E+01	1.6E+01	1.7E+01	2.4E+01	3.5E+01	5.4E+01	1.3E+02	2.1E+02	4.0E+02
30	2.7E+01	2.7E+01	2.7E+01	2.7E+01	2.7E+01	3.3E+01	4.4E+01	5.8E+01	8.5E+01	1.8E+02	2.0E+02	5.5E+02
50	7.6E+01	7.6E+01	7.6E+01	7.6E+01	7.6E+01	8.3E+01	1.1E+02	1.3E+02	1.3E+02	2.4E+02	3.9E+02	7.2E+02
Stack Diameter =	: 1.0 m											
Stack Height (m)	30	50	70	100	200	300	500	700	1000	2000	3000	5000
10	1.0E+01	1.0E+01	1.2E+01	1.7E+01	3.9E+01	3.9E+01	4.5E+01	5.8E+01	7.9E+01	1.6E+02	2.4E+02	4.4E+02
20	2.6E+01	2.6E+01	2.6E+01	2.8E+01	3.9E+01	3.9E+01	4.5E+01	5.8E+01	7.9E+01	1.6E+02	2.4E+02	4,5E+02
30	4.2E+01	4.2E+01	4.2E+01	4.2E+01	4.2E+01	4.2E+01	4.5E+01	5.8E+01	8.5E+01	1.8E+02	2.9E+02	5.5E+02
80	8.9E+01	8.9E+01	8.9E+01	8.9E+01	8.9E+01	8.9E+01	1.1E+02	1.4E+02	1.7E+02	3.3E+02	5.0E+02	8.7E+02
20	3.8E+02	3.8E+02	3.8E+02	3.8E+02	3.8E+02	3.8E+02	3.8E+02	4.0E+02	4.1E+02	4.3E+02	6.1E+02	1.0E+03-
Stack Diameter =	: 1.5 m									į.		
Stack Height (m)	30	50	70	100	200	300	500	700	1000	2000	3000	5000
10	2.1E+01	2.1E+01	2.5E+01	3.6E+01	5.4E+01	6.3E+01	6.3E+01	6.9E+01	9.8E+01	2.0E+02	3.2E+02	5.7E+02
20	3.3E+01	3.3E+01	3.3E+01	3.7E+01	5.4E+01	6.3E+01	6.3E+01	6.9E+01	9.8E+01	2.0E+02	3.2E+02	5.7E+02
30	6.3E+01	6.3E+01 ·	6.3E+01	6.3E+01	6.3E+01	6.3E+01	6.3E+01	6.9E+01	9.8E+01	2.0E+02	3.2E+02	5.8E+02
80	1.0E+02	1.0E+02	1.0E+02	1.0E+02	1.0E+02	1.0E+02	1.2E+02	1.4E+02	1.7E+02	3.3E+02	5.0E+02	8.7E+02
70	4.8E+02	4.8E+02	4.8E+02	4.8E+02	4.8E+02	4.8E+02	4.8E+02	4.8E+02	4.8E+02	6.5E+02	8.2E+02	1.3E+03
Stack Diameter =	2.0 m											
Stack Height (m)	30	50	70	100	200	300	500	200	1000	2000	3000	5000
10	2.7E+01	2.7E+01	3.2E+01	4.4E+01	6.6E+01	9.7E+01	9.7E+01	1.1E+02	1.5E+02	2.6E+02	3.9E+02	6.6E+02
20	4.0E+01	4.0E+01	4.0E+01	4.4E+01	6.6E+01	9.7E+01	9.7E+01	1.1E+02	1.5E+02	2.6E+02	3.9E+02	6.6E+02
30	7.9E+01	7.9E+01	7.9E+01	7.9E+01	9.1E+01	9.7E+01	9.7E+01	1.1E+02	1.5E+02	2.6E+02	3.9E+02	6.6E+02
50	1.3E+02	1.3E+02	1.3E+02	1.3E+02	1.3E+02	1.3E+02	1.3E+02	1.4E+02	1.7E+02	3.3E+02	5.0E+02	8.7E+02
70	5.6E+02	5.6E+02	5.6E+02	5.6E+02	5.6E+02	5.6E+02	5.6E+02	5.6E+02	5.6E+02	7.3E+02	1.1E+03	1.5E+03
100	8.6E+02	8.6E+02	8.6E+02	8.6E+02	8.6E+02	8.6E+02	8.6E+02	8.6E+02	8.6E+02	8.6E+02	1.2E+03	1.7E+03
Stack Diameter =	= 3.0 m											
Stack Height (m)	30	50	02	100	200	300	200	700	1000	2000	3000	5000
10	3.5E+01	3.5E+01	4.1E+01	5.8E+01	1.2E+02	1.6E+02	1.8E+02	1.8E+02	1.8E+02	3.5E+02	5.2E+02	9.0E+02
20	6.2E+01	6.2E+01	6.2E+01	7.2E+01	1.2E+02	1.6E+02	1.8E+02	1.8E+02	1.8E+02	3.5E+02	5.2E+02	9.0E+02
30	1.0E+02	1.0E+02	1.0E+02	1.0E+02	1.2E+02	1.6E+02	1.8E+02	1.8E+02	1.8E+02	3.5E+02	5.2E+02	9.0E+02
8	1.8E+02	1.8E+02	1.8E+02	1.8E+02	1.8E+02 -	1.8E+02	1.8E+02	1.8E+02	1.8E+02	3.5E+02	5.2E+02	9.0E+02
20	7.5E+02	7.5E+02	7.5E+02	7.5E+02	7.5E+02	7,5E+02	7.5E+02	7.5E+02	7.5E+02	8.9E+02	1.3E+03	2.0E+03
100	1.4E+03	1.4E+03	1.4E+03	1.4E+03	1.4E+03	1.4E+03	1.4E+03	1.4E+03	1.4E+03	1.4E+03	2.0E+03	2.6E+03
Stack Diameter a	= 4.0 m											
Stack Height (m)	30	50	02	100	200	300	500	700	1000	2000	3000	5000
8	1.2E+02	1.2E+02	1.2E+02	1.2E+02	1.4E+02	2.0E+02	2.2E+02	2.2E+02	2.2E+02	4.2E+02	6.3E+02	1.0E+03
20	2.2E+02	2.2E+02	2.2E+02	2.2E+02	2.2E+02	2.2E+02	2.2E+02	2.2E+02	2.2E+02	4.2E+02	6.3E+02	1.0E+03
P.	1.0E+03	1.0E+03	1.0E+03	1.0E+03	1.0E+03	1.0E+03	1.0E+03	1.0E+03	1.0E+03	1.1E+03	1.6E+03	2.4E+03
100	1.6E+03	1.6E+03	1.6E+03	1.6E+03	1.6F+03	1 6 4 03	1 8 4 4 13	1.65403	1.6E+03	1.85403	2 3E+03	3 READS

[70 FR 59565, Oct. 12, 2005, as amended at 73 FR 18982, Apr. 8, 2008; 73 FR 64097, Oct. 28, 2008]

EMISSIONS STANDARDS AND OPERATING LIMITS FOR SOLID FUEL BOILERS, LIQ-UID FUEL BOILERS, AND HYDROCHLORIC ACID PRODUCTION FURNACES

§63.1216 What are the standards for solid fuel boilers that burn hazardous waste?

(a) *Emission limits for existing sources*. You must not discharge or cause combustion gases to be emitted into the atmosphere that contain:

(1) For dioxins and furans, either carbon monoxide or hydrocarbon emissions in excess of the limits provided by paragraph (a)(5) of this section;

(2) Mercury in excess of 11 $\mu gm/dscm$ corrected to 7 percent oxygen;

(3) For cadmium and lead combined, except for an area source as defined under 63.2, emissions in excess of 180 µgm/dscm, corrected to 7 percent oxygen;

(4) For arsenic, beryllium, and chromium combined, except for an area source as defined under §63.2, emissions in excess of 380 μ gm/dscm, corrected to 7 percent oxygen;

(5) For carbon monoxide and hydrocarbons, either:

(i) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (a)(5)(ii) of this section, you must also document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons do not exceed 10 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane: or

(ii) Hydrocarbons in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane;

(6) For hydrogen chloride and chlorine combined, except for an area source as defined under §63.2, emissions in excess of 440 parts per million by volume, expressed as a chloride $(Cl(^{-}))$ equivalent, dry basis and corrected to 7 percent oxygen; and

(7) For particulate matter, except for an area source as defined under §63.2 or as provided by paragraph (e) of this section, emissions in excess of 68 mg/ dscm corrected to 7 percent oxygen.

(b) *Emission limits for new sources*. You must not discharge or cause combustion gases to be emitted into the atmosphere that contain:

(1) For dioxins and furans, either carbon monoxide or hydrocarbon emissions in excess of the limits provided by paragraph (b)(5) of this section;

(2) Mercury in excess of 11 $\mu gm/dscm$ corrected to 7 percent oxygen;

(3) For cadmium and lead combined, except for an area source as defined under §63.2, emissions in excess of 180 µgm/dscm, corrected to 7 percent oxygen;

(4) For arsenic, beryllium, and chromium combined, except for an area source as defined under §63.2, emissions 40 CFR Ch. I (7–1–23 Edition)

in excess of 190 µgm/dscm, corrected to 7 percent oxygen;

(5) For carbon monoxide and hydrocarbons, either:

(i) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (b)(5)(ii) of this section, you must also document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons do not exceed 10 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane: or

(ii) Hydrocarbons in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane;

(6) For hydrogen chloride and chlorine combined, except for an area source as defined under §63.2, emissions in excess of 73 parts per million by volume, expressed as a chloride $(Cl^{(-)})$ equivalent, dry basis and corrected to 7 percent oxygen; and

(7) For particulate matter, except for an area source as defined under §63.2 or as provided by paragraph (e) of this section, emissions in excess of 34 mg/ dscm corrected to 7 percent oxygen.

(c) Destruction and removal efficiency (DRE) standard—(1) 99.99% DRE. Except as provided in paragraph (c)(2) of this section, you must achieve a DRE of 99.99% for each principle organic hazardous constituent (POHC) designated under paragraph (c)(3) of this section. You must calculate DRE for each POHC from the following equation:

 $DRE = [1 - (W_{out} \div W_{in})] \times 100\%$

Where:

 W_{in} = mass feedrate of one POHC in a waste feedstream; and

W_{out} = mass emission rate of the same POHC present in exhaust emissions prior to release to the atmosphere.

(2) 99.9999% DRE. If you burn the dioxin-listed hazardous wastes F020, F021, F022, F023, F026, or F027 (see §261.31 of this chapter), you must achieve a DRE of 99.9999% for each POHC that you designate under paragraph (c)(3) of this section. You must demonstrate this DRE performance on POHCs that are more difficult to incinand erate than tetra-, penta-, hexachlorodibenzo-*p*-dioxins and dibenzofurans. You must use the equation in paragraph (c)(1) of this section to calculate DRE for each POHC. In addition, you must notify the Administrator of your intent to incinerate hazardous wastes F020, F021, F022, F023, F026, or F027.

(3) Principal organic hazardous constituents (POHCs). (i) You must treat the POHCs in the waste feed that you specify under paragraph (c)(3)(ii) of this section to the extent required by paragraphs (c)(1) and (c)(2) of this section.

(ii) You must specify one or more POHCs that are representative of the most difficult to destroy organic compounds in your hazardous waste feedstream. You must base this specification on the degree of difficulty of incineration of the organic constituents in the hazardous waste and on their concentration or mass in the hazardous waste feed, considering the results of hazardous waste analyses or other data and information.

(d) Significant figures. The emission limits provided by paragraphs (a) and (b) of this section are presented with two significant figures. Although you must perform intermediate calculations using at least three significant figures, you may round the resultant emission levels to two significant figures to document compliance.

(e) Alternative to the particulate matter standard—(1) General. In lieu of complying with the particulate matter standards of this section, you may elect to comply with the following alternative metal emission control requirement:

(2) Alternative metal emission control requirements for existing solid fuel boilers.

(i) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain cadmium, lead, and selenium in excess of 180 µgm/ dscm, combined emissions, corrected to 7 percent oxygen; and,

(ii) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain antimony, arsenic, beryllium, chromium, cobalt, manganese, and nickel in excess of 380 µgm/dscm, combined emissions, corrected to 7 percent oxygen.

(3) Alternative metal emission control requirements for new solid fuel boilers. (i) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain cadmium, lead, and selenium in excess of 180 μgm/dscm, combined emissions, corrected to 7 percent oxygen; and,

(ii) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain antimony, arsenic, beryllium, chromium, cobalt, manganese, and nickel in excess of 190 µgm/dscm, combined emissions, corrected to 7 percent oxygen.

(4) Operating limits. Semivolatile and low volatile metal operating parameter limits must be established to ensure compliance with the alternative emission limitations described in paragraphs (e)(2) and (e)(3) of this section pursuant to §63.1209(n), except that semivolatile metal feedrate limits apply to lead, cadmium, and selenium, combined, and low volatile metal feedrate limits apply to arsenic, beryllium, chromium, antimony, cobalt, manganese, and nickel, combined.

(f) Elective standards for area sources. Area sources as defined under §63.2 are subject to the standards for cadmium and lead, the standards for arsenic, beryllium, and chromium, the standards for hydrogen chloride and chlorine, and the standards for particulate matter under this section if they elect under §266.100(b)(3) of this chapter to comply with those standards in lieu of the standards under 40 CFR 266.105, 266.106, and 266.107 to control those pollutants.

[70 FR 59565, Oct. 12, 2005]

§63.1217

§63.1217 What are the standards for liquid fuel boilers that burn hazardous waste?

(a) *Emission limits for existing sources.* You must not discharge or cause combustion gases to be emitted into the atmosphere that contain:

(1)(i) Dioxins and furans in excess of 0.40 ng TEQ/dscm, corrected to 7 percent oxygen, for liquid fuel boilers equipped with a dry air pollution control system; or

(ii) Either carbon monoxide or hydrocarbon emissions in excess of the limits provided by paragraph (a)(5) of this section for sources not equipped with a dry air pollution control system;

(iii) A source equipped with a wet air pollution control system followed by a dry air pollution control system is not considered to be a dry air pollution control system, and a source equipped with a dry air pollution control system followed by a wet air pollution control system is considered to be a dry air pollution control system for purposes of this emission limit;

(2) For mercury, except as provided for in paragraph (a)(2)(iii) of this section:

(i) When you burn hazardous waste with an as-fired heating value less than 10,000 Btu/lb, emissions in excess of 19 μ gm/dscm, corrected to 7 percent oxygen, on an (not-to-exceed) annual averaging period;

(ii) When you burn hazardous waste with an as-fired heating value 10,000 Btu/lb or greater, emissions in excess of 4.2×10^{-5} lbs mercury attributable to the hazardous waste per million Btu heat input from the hazardous waste on an (not-to-exceed) annual averaging period;

(iii) The boiler operated by Diversified Scientific Services, Inc. with EPA identification number TND982109142, and which burns radioactive waste mixed with hazardous waste, must comply with the mercury emission standard under §63.1219(a)(2);

(3) For cadmium and lead combined, except for an area source as defined under §63.2,

(i) When you burn hazardous waste with an as-fired heating value less than 10,000 Btu/lb, emissions in excess of 150 μ gm/dscm, corrected to 7 percent oxy-

gen, on an (not-to-exceed) annual averaging period;

(ii) When you burn hazardous waste with an as-fired heating value of 10,000 Btu/lb or greater, emissions in excess of 8.2×10^{-5} lbs combined cadmium and lead emissions attributable to the hazardous waste per million Btu heat input from the hazardous waste on an (not-to-exceed) annual averaging period;

(4) For chromium, except for an area source as defined under §63.2:

(i) When you burn hazardous waste with an as-fired heating value less than 10,000 Btu/lb, emissions in excess of 370 μ gm/dscm, corrected to 7 percent oxygen;

(ii) When you burn hazardous waste with an as-fired heating value of 10,000 Btu/lb or greater, emissions in excess of 1.3×10^{-4} lbs chromium emissions attributable to the hazardous waste per million Btu heat input from the hazardous waste;

(5) For carbon monoxide and hydrocarbons, either:

(i) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (a)(5)(ii) of this section, you must also document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons do not exceed 10 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane: or

(ii) Hydrocarbons in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane;

(6) For hydrogen chloride and chlorine, except for an area source as defined under §63.2:

(i) When you burn hazardous waste with an as-fired heating value less than 10,000 Btu/lb, emissions in excess of 31 parts per million by volume, combined emissions, expressed as a chloride $(Cl^{(-)})$ equivalent, dry basis and corrected to 7 percent oxygen;

(ii) When you burn hazardous waste with an as-fired heating value of 10,000 Btu/lb or greater, emissions in excess of 5.1×10^{-2} lbs combined emissions of hydrogen chloride and chlorine gas attributable to the hazardous waste per million Btu heat input from the hazardous waste;

(7) For particulate matter, except for an area source as defined under $\S63.2$ or as provided by paragraph (e) of this section, emissions in excess of 80 mg/dscm corrected to 7 percent oxygen.

(b) *Emission limits for new sources*. You must not discharge or cause combustion gases to be emitted into the atmosphere that contain:

(1)(i) Dioxins and furans in excess of 0.40 ng TEQ/dscm, corrected to 7 percent oxygen, for liquid fuel boilers equipped with a dry air pollution control system; or

(ii) Either carbon monoxide or hydrocarbon emissions in excess of the limits provided by paragraph (b)(5) of this section for sources not equipped with a dry air pollution control system;

(iii) A source equipped with a wet air pollution control system followed by a dry air pollution control system is not considered to be a dry air pollution control system, and a source equipped with a dry air pollution control system followed by a wet air pollution control system is considered to be a dry air pollution control system for purposes of this emission limit;

(2) For mercury:

(i) When you burn hazardous waste with an as-fired heating value less than 10,000 Btu/lb, emissions in excess of 6.8 µgm/dscm, corrected to 7 percent oxygen, on an (not-to-exceed) annual averaging period;

(ii) When you burn hazardous waste with an as-fired heating value of 10,000 Btu/lb or greater, emissions in excess of 1.2×10^{-6} lbs mercury emissions attributable to the hazardous waste per million Btu heat input from the hazardous waste on an (not-to-exceed) annual averaging period;

(3) For cadmium and lead combined, except for an area source as defined under §63.2:

(i) When you burn hazardous waste with an as-fired heating value less than 10,000 Btu/lb, emissions in excess of 78 µgm/dscm, corrected to 7 percent oxygen, on an (not-to-exceed) annual averaging period;

(ii) When you burn hazardous waste with an as-fired heating value greater than or equal to 10,000 Btu/lb, emissions in excess of 6.2×10^{-6} lbs combined cadmium and lead emissions attributable to the hazardous waste per million Btu heat input from the hazardous waste on an (not-to-exceed) annual averaging period;

(4) For chromium, except for an area source as defined under §63.2:

(i) When you burn hazardous waste with an as-fired heating value less than 10,000 Btu/lb, emissions in excess of 12 μ gm/dscm, corrected to 7 percent oxygen:

(ii) When you burn hazardous waste with an as-fired heating value of 10,000 Btu/lb or greater, emissions in excess of 1.4×10^{-5} lbs chromium emissions attributable to the hazardous waste per million Btu heat input from the hazardous waste;

(5) For carbon monoxide and hydrocarbons, either:

(i) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (b)(5)(ii) of this section, you must also document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons do not exceed 10 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or

(ii) Hydrocarbons in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane;

(6) For hydrogen chloride and chlorine, except for an area source as defined under §63.2:

(i) When you burn hazardous waste with an as-fired heating value less than 10,000 Btu/lb, emissions in excess of 31 parts per million by volume, combined emissions, expressed as a chloride (Cl(-)) equivalent, dry basis and corrected to 7 percent oxygen;

(ii) When you burn hazardous waste with an as-fired heating value of 10,000 Btu/lb or greater, emissions in excess of $5.1 \times ^{-2}$ lbs combined emissions of hydrogen chloride and chlorine gas attributable to the hazardous waste per million Btu heat input from the hazardous waste;

(7) For particulate matter, except for an area source as defined under §63.2 or as provided by paragraph (e) of this section, emissions in excess of 20 mg/ dscm corrected to 7 percent oxygen.

(c) Destruction and removal efficiency (DRE) standard—(1) 99.99% DRE. Except as provided in paragraph (c)(2) of this section, you must achieve a DRE of 99.99% for each principle organic hazardous constituent (POHC) designated under paragraph (c)(3) of this section. You must calculate DRE for each POHC from the following equation:

 $\mathrm{DRE} = [1 - (\mathrm{W}_{\mathrm{out}} \div \mathrm{W}_{\mathrm{in}})] \times 100\%$

Where:

 W_{in} = mass feedrate of one POHC in a waste feedstream; and

W_{out} = mass emission rate of the same POHC present in exhaust emissions prior to release to the atmosphere.

(2) 99.9999% DRE. If you burn the dioxin-listed hazardous wastes F020, F021, F022, F023, F026, or F027 (see §261.31 of this chapter), you must achieve a DRE of 99.9999% for each POHC that you designate under paragraph (c)(3) of this section. You must demonstrate this DRE performance on POHCs that are more difficult to incinpenta-, erate than tetra-, and hexachlorodibenzo-p-dioxins and dibenzofurans. You must use the equation in paragraph (c)(1) of this section to calculate DRE for each POHC. In addition, you must notify the Administrator of your intent to incinerate haz40 CFR Ch. I (7–1–23 Edition)

ardous wastes F020, F021, F022, F023, F026, or F027.

(3) Principal organic hazardous constituents (POHCs). (i) You must treat the POHCs in the waste feed that you specify under paragraph (c)(3)(i) of this section to the extent required by paragraphs (c)(1) and (c)(2) of this section.

(ii) You must specify one or more POHCs that are representative of the most difficult to destroy organic compounds in your hazardous waste feedstream. You must base this specification on the degree of difficulty of incineration of the organic constituents in the hazardous waste and on their concentration or mass in the hazardous waste feed, considering the results of hazardous waste analyses or other data and information.

(d) Significant figures. The emission limits provided by paragraphs (a) and (b) of this section are presented with two significant figures. Although you must perform intermediate calculations using at least three significant figures, you may round the resultant emission levels to two significant figures to document compliance.

(e) Alternative to the particulate matter standard—(1) General. In lieu of complying with the particulate matter standards of this section, you may elect to comply with the following alternative metal emission control requirement:

(2) Alternative metal emission control requirements for existing liquid fuel boilers. (i) When you burn hazardous waste with a heating value less than 10,000 Btu/lb:

(A) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain cadmium, lead, and selenium, combined, in excess of 150 µgm/dscm, corrected to 7 percent oxygen; and

(B) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain antimony, arsenic, beryllium, chromium, cobalt, manganese, and nickel, combined, in excess of 370 µgm/dscm, corrected to 7 percent oxygen;

(ii) When you burn hazardous waste with a heating value of 10,000 Btu/lb or greater:

(A) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain in excess of 8.2×10^{-5} lbs combined emissions of cadmium, lead, and selenium attributable to the hazardous waste per million Btu heat input from the hazardous waste; and

(B) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain either in excess of 1.3×10^{-4} lbs combined emissions of antimony, arsenic, beryllium, chromium, cobalt, manganese, and nickel attributable to the hazardous waste per million Btu heat input from the hazardous waste;

(3) Alternative metal emission control requirements for new liquid fuel boilers.(i) When you burn hazardous waste with a heating value less than 10,000 Btu/lb:

(A) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain cadmium, lead, and selenium, combined, in excess of 78 μ gm/dscm, corrected to 7 percent oxygen; and

(B) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain antimony, arsenic, beryllium, chromium, cobalt, manganese, and nickel, combined, in excess of 12 µgm/dscm, corrected to 7 percent oxygen;

(ii) When you burn hazardous waste with a heating value greater than or equal to 10,000 Btu/lb:

(A) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain in excess of 6.2×10^{-6} lbs combined emissions of cadmium, lead, and selenium attributable to the hazardous waste per million Btu heat input from the hazardous waste; and

(B) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain either in excess of 1.4×10^{-5} lbs combined emissions of antimony, arsenic, beryllium, chromium, cobalt, manganese, and nickel attributable to the hazardous waste per million Btu heat input from the hazardous waste;

(4) Operating limits. Semivolatile and low volatile metal operating parameter limits must be established to ensure compliance with the alternative emission limitations described in paragraphs (e)(2) and (e)(3) of this section pursuant to §63.1209(n), except that semivolatile metal feedrate limits apply to lead, cadmium, and selenium, combined, and low volatile metal feedrate limits apply to arsenic, beryllium, chromium, antimony, cobalt, manganese, and nickel, combined.

(f) Elective standards for area sources. Area sources as defined under §63.2 are subject to the standards for cadmium and lead, the standards for chromium, the standards for hydrogen chloride and chlorine, and the standards for particulate matter under this section if they elect under §266.100(b)(3) of this chapter to comply with those standards in lieu of the standards under 40 CFR 266.105, 266.106, and 266.107 to control those pollutants.

[70 FR 59567, Oct. 12, 2005, as amended at 73 FR 18983, Apr. 8, 2008]

§63.1218 What are the standards for hydrochloric acid production furnaces that burn hazardous waste?

(a) *Emission limits for existing sources*. You must not discharge or cause combustion gases to be emitted into the atmosphere that contain:

(1) For dioxins and furans, either carbon monoxide or hydrocarbon emissions in excess of the limits provided by paragraph (a)(5) of this section;

(2) For mercury, hydrogen chloride and chlorine gas emissions in excess of the levels provided by paragraph (a)(6)of this section;

(3) For lead and cadmium, except for an area source as defined under §63.2, hydrogen chloride and chlorine gas emissions in excess of the levels provided by paragraph (a)(6) of this section;

(4) For arsenic, beryllium, and chromium, except for an area source as defined under $\S63.2$, hydrogen chloride and chlorine gas emissions in excess of the levels provided by paragraph (a)(6) of this section;

(5) For carbon monoxide and hydrocarbons, either:

(i) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (a)(5)(ii) of this section, you must also document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons do not exceed 10 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or

(ii) Hydrocarbons in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane;

(6) For hydrogen chloride and chlorine gas, either:

(i) Emission in excess of 150 parts per million by volume, combined emissions, expressed as a chloride ($Cl^{(-)}$ equivalent, dry basis and corrected to 7 percent oxygen; or

(ii) Emissions greater than the levels that would be emitted if the source is achieving a system removal efficiency (SRE) of less than 99.923 percent for total chlorine and chloride fed to the combustor. You must calculate SRE from the following equation:

 $SRE = [1 - (Cl_{out} / Cl_{in})] \times 100\%$

Where:

- Cl in = mass feedrate of total chlorine or chloride in all feedstreams, reported as chloride; and
- Cl out = mass emission rate of hydrogen chloride and chlorine gas, reported as chloride, in exhaust emissions prior to release to the atmosphere.

(7) For particulate matter, except for an area source as defined under §63.2, hydrogen chloride and chlorine gas emissions in excess of the levels provided by paragraph (a)(6) of this section.

(b) *Emission limits for new sources*. You must not discharge or cause combustion gases to be emitted into the atmosphere that contain:

(1) For dioxins and furans, either carbon monoxide or hydrocarbon emis40 CFR Ch. I (7–1–23 Edition)

sions in excess of the limits provided by paragraph (b)(5) of this section;

(2) For mercury, hydrogen chloride and chlorine gas emissions in excess of the levels provided by paragraph (b)(6) of this section;

(3) For lead and cadmium, except for an area source as defined under §63.2, hydrogen chloride and chlorine gas emissions in excess of the levels provided by paragraph (b)(6) of this section;

(4) For arsenic, beryllium, and chromium, except for an area source as defined under §63.2, hydrogen chloride and chlorine gas emissions in excess of the levels provided by paragraph (b)(6) of this section;

(5) For carbon monoxide and hydrocarbons, either:

(i) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (b)(5)(ii) of this section, you must also document that, during the destruction and removal efficiency (DRE) test runs or equivalent as provided by their §63.1206(b)(7), hydrocarbons do not exceed 10 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or

(ii) Hydrocarbons in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane;

(6) For hydrogen chloride and chlorine gas, either:

(i) Emission in excess of 25 parts per million by volume, combined emissions, expressed as a chloride ($Cl^{(-)}$ equivalent, dry basis and corrected to 7 percent oxygen; or

(ii) Emissions greater than the levels that would be emitted if the source is achieving a system removal efficiency (SRE) of less than 99.987 percent for

total chlorine and chloride fed to the combustor. You must calculate SRE from the following equation:

 $SRE = [1 - (Cl_{out} / Cl_{in})] \times 100\%$

Where:

- Cl in = mass feedrate of total chlorine or chloride in all feedstreams, reported as chloride; and
- Cl out = mass emission rate of hydrogen chloride and chlorine gas, reported as chloride, in exhaust emissions prior to release to the atmosphere.

(7) For particulate matter, except for an area source as defined under 63.2, hydrogen chloride and chlorine gas emissions in excess of the levels provided by paragraph (b)(6) of this section.

(c) Destruction and removal efficiency (DRE) standard—(1) 99.99% DRE. Except as provided in paragraph (c)(2) of this section, you must achieve a DRE of 99.99% for each principle organic hazardous constituent (POHC) designated under paragraph (c)(3) of this section. You must calculate DRE for each POHC from the following equation:

DRE = $[1 - (W_{out} / W_{in})] \times 100\%$

Where:

Win = mass feedrate of one POHC in a waste feedstream; and

Wout = mass emission rate of the same POHC present in exhaust emissions prior to release to the atmosphere.

(2) 99.9999% DRE. If you burn the dioxin-listed hazardous wastes F020, F021, F022, F023, F026, or F027 (see §261.31 of this chapter), you must achieve a DRE of 99.9999% for each POHC that you designate under paragraph (c)(3) of this section. You must demonstrate this DRE performance on POHCs that are more difficult to incinerate than tetra-, penta-, and hexachlorodibenzo-p-dioxins and dibenzofurans. You must use the equation in paragraph (c)(1) of this section to calculate DRE for each POHC. In addition, you must notify the Administrator of your intent to incinerate hazardous wastes F020, F021, F022, F023, F026. or F027.

(3) Principal organic hazardous constituents (POHCs). (i) You must treat the POHCs in the waste feed that you specify under paragraph (c)(3)(i) of this section to the extent required by paragraphs (c)(1) and (c)(2) of this section.

(ii) You must specify one or more POHCs that are representative of the most difficult to destroy organic compounds in your hazardous waste feedstream. You must base this specification on the degree of difficulty of incineration of the organic constituents in the hazardous waste and on their concentration or mass in the hazardous waste feed, considering the results of hazardous waste analyses or other data and information.

(d) Significant figures. The emission limits provided by paragraphs (a) and (b) of this section are presented with two significant figures. Although you must perform intermediate calculations using at least three significant figures, you may round the resultant emission levels to two significant figures to document compliance.

(e) Elective standards for area sources. Area sources as defined under §63.2 are subject to the standards for cadmium and lead, the standards for arsenic, beryllium, and chromium, the standards for hydrogen chloride and chlorine, and the standards for particulate matter under this section if they elect under \$266.100(b)(3) of this chapter to comply with those standards in lieu of the standards under 40 CFR 266.105, 266.106, and 266.107 to control those pollutants.

[70 FR 59569, Oct. 12, 2005]

REPLACEMENT EMISSIONS STANDARDS AND OPERATING LIMITS FOR INCINER-ATORS, CEMENT KILNS, AND LIGHT-WEIGHT AGGREGATE KILNS

§63.1219 What are the replacement standards for hazardous waste incinerators?

(a) *Emission limits for existing sources.* You must not discharge or cause combustion gases to be emitted into the atmosphere that contain:

(1) For dioxins and furans:

(i) For incinerators equipped with either a waste heat boiler or dry air pollution control system, either:

(A) Emissions in excess of 0.20 ng TEQ/dscm, corrected to 7 percent oxygen; or

(B) Emissions in excess of 0.40 ng TEQ/dscm, corrected to 7 percent oxygen, provided that the combustion gas temperature at the inlet to the initial particulate matter control device is 400 °F or lower based on the average of the test run average temperatures. (For purposes of compliance, operation of a wet particulate matter control device is presumed to meet the 400 °F or lower requirement);

(ii) Emissions in excess of 0.40 ng TEQ/dscm, corrected to 7 percent oxygen, for incinerators not equipped with either a waste heat boiler or dry air pollution control system;

(iii) A source equipped with a wet air pollution control system followed by a dry air pollution control system is not considered to be a dry air pollution control system, and a source equipped with a dry air pollution control system followed by a wet air pollution control system is considered to be a dry air pollution control system for purposes of this standard;

(2) Mercury in excess of 130 μ gm/dscm, corrected to 7 percent oxygen;

(3) Cadmium and lead in excess of 230 µgm/dscm, combined emissions, corrected to 7 percent oxygen;

(4) Arsenic, beryllium, and chromium in excess of 92 μ gm/dscm, combined emissions, corrected to 7 percent oxygen;

(5) For carbon monoxide and hydrocarbons, either:

(i) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (a)(5)(ii) of this section, you must also document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons do not exceed 10 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane: or

(ii) Hydrocarbons in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions 40 CFR Ch. I (7–1–23 Edition)

monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane;

(6) Hydrogen chloride and chlorine gas (total chlorine) in excess of 32 parts per million by volume, combined emissions, expressed as a chloride $(Cl^{(-)})$ equivalent, dry basis and corrected to 7 percent oxygen; and

(7) Except as provided by paragraph (e) of this section, particulate matter in excess of 0.013 gr/dscf corrected to 7 percent oxygen.

(b) *Emission limits for new sources.* You must not discharge or cause combustion gases to be emitted into the atmosphere that contain:

(1)(i) Dioxins and furans in excess of 0.11 ng TEQ/dscm corrected to 7 percent oxygen for incinerators equipped with either a waste heat boiler or dry air pollution control system; or

(ii) Dioxins and furans in excess of 0.20 ng TEQ/dscm corrected to 7 percent oxygen for sources not equipped with either a waste heat boiler or dry air pollution control system;

(iii) A source equipped with a wet air pollution control system followed by a dry air pollution control system is not considered to be a dry air pollution control system, and a source equipped with a dry air pollution control system followed by a wet air pollution control system is considered to be a dry air pollution control system for purposes of this standard;

(2) Mercury in excess of 8.1 µgm/dscm, corrected to 7 percent oxygen;

(3) Cadmium and lead in excess of 10 μgm/dscm, combined emissions, corrected to 7 percent oxygen;

(4) Arsenic, beryllium, and chromium in excess of 23 μgm/dscm, combined emissions, corrected to 7 percent oxygen;

(5) For carbon monoxide and hydrocarbons, either:

(i) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (b)(5)(ii) of this section, you must also document

that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons do not exceed 10 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or

(ii) Hydrocarbons in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane;

(6) Hydrogen chloride and chlorine gas in excess of 21 parts per million by volume, combined emissions, expressed as a chloride $(Cl^{(-)})$ equivalent, dry basis and corrected to 7 percent oxygen; and

(7) Except as provided by paragraph (e) of this section, particulate matter emissions in excess of 0.0016 gr/dscf corrected to 7 percent oxygen.

(c) Destruction and removal efficiency (DRE) standard—(1) 99.99% DRE. Except as provided in paragraph (c)(2) of this section, you must achieve a destruction and removal efficiency (DRE) of 99.99% for each principle organic hazardous constituent (POHC) designated under paragraph (c)(3) of this section. You must calculate DRE for each POHC from the following equation:

DRE = $[1 - (W_{out} / W_{in})] \times 100\%$

Where:

 W_{in} = mass feedrate of one POHC in a waste feedstream; and

 W_{out} = mass emission rate of the same POHC present in exhaust emissions prior to release to the atmosphere.

(2) 99.9999% DRE. If you burn the dioxin-listed hazardous wastes F020, F021, F022, F023, F026, or F027 (see §261.31 of this chapter), you must achieve a DRE of 99.9999% for each POHC that you designate under paragraph (c)(3) of this section. You must demonstrate this DRE performance on POHCs that are more difficult to incinthan tetra-, penta-, and erate hexachlorodibenzo-p-dioxins and dibenzofurans. You must use the equation in paragraph (c)(1) of this section to calculate DRE for each POHC. In addition, you must notify the Administrator of your intent to incinerate hazardous wastes F020, F021, F022, F023, F026, or F027.

(3) Principal organic hazardous constituent (POHC). (i) You must treat each POHC in the waste feed that you specify under paragraph (c)(3)(ii) of this section to the extent required by paragraphs (c)(1) and (c)(2) of this section.

(ii) You must specify one or more POHCs that are representative of the most difficult to destroy organic compounds in your hazardous waste feedstream. You must base this specification on the degree of difficulty of incineration of the organic constituents in the hazardous waste and on their concentration or mass in the hazardous waste feed, considering the results of hazardous waste analyses or other data and information.

(d) Significant figures. The emission limits provided by paragraphs (a) and (b) of this section are presented with two significant figures. Although you must perform intermediate calculations using at least three significant figures, you may round the resultant emission levels to two significant figures to document compliance.

(e) Alternative to the particulate matter standard—(1) General. In lieu of complying with the particulate matter standards of this section, you may elect to comply with the following alternative metal emission control requirement:

(2) Alternative metal emission control requirements for existing incinerators. (i) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain cadmium, lead, and selenium in excess of 230 μ gm/dscm, combined emissions, corrected to 7 percent oxygen; and,

(ii) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain antimony, arsenic, beryllium, chromium, cobalt, manganese, and nickel in excess of 92 µgm/dscm, combined emissions, corrected to 7 percent oxygen.

(3) Alternative metal emission control requirements for new incinerators. (i) You

must not discharge or cause combustion gases to be emitted into the atmosphere that contain cadmium, lead, and selenium in excess of 10 μ gm/dscm, combined emissions, corrected to 7 percent oxygen; and,

(ii) You must not discharge or cause combustion gases to be emitted into the atmosphere that contain antimony, arsenic, beryllium, chromium, cobalt, manganese, and nickel in excess of 23 µgm/dscm, combined emissions, corrected to 7 percent oxygen.

(4) Operating limits. Semivolatile and low volatile metal operating parameter limits must be established to ensure compliance with the alternative emission limitations described in paragraphs (e)(2) and (e)(3) of this section pursuant to $\S63.1209(n)$, except that semivolatile metal feedrate limits apply to lead, cadmium, and selenium, combined, and low volatile metal feedrate limits apply to arsenic, beryllium, chromium, antimony, cobalt, manganese, and nickel, combined.

[70 FR 59570, Oct. 12, 2005, as amended at 73 FR 64097, Oct. 28, 2008]

§63.1220 What are the replacement standards for hazardous waste burning cement kilns?

(a) Emission and hazardous waste feed limits for existing sources. You must not discharge or cause combustion gases to be emitted into the atmosphere or feed hazardous waste that contain:

(1) For dioxins and furans, either:

(i) Emissions in excess of 0.20 ng TEQ/ dscm corrected to 7 percent oxygen; or

(ii) Emissions in excess of 0.40 ng TEQ/dscm corrected to 7 percent oxygen provided that the combustion gas temperature at the inlet to the initial dry particulate matter control device is 400 °F or lower based on the average of the test run average temperatures;

(2) For mercury, both:

(i) An average as-fired concentration of mercury in all hazardous waste feedstreams in excess of 3.0 parts per million by weight; and

(ii) Either:

 (A) Emissions in excess of 120 μg/ dscm, corrected to 7 percent oxygen, or
 (B) A hazardous waste feed maximum

(B) A hazardous waste reed maximum theoretical emission concentration (MTEC) in excess of 120 μ g/dscm;

(3) For cadmium and lead, both:

(i) Emissions in excess of 7.6×10^{-4} lbs combined emissions of cadmium and lead attributable to the hazardous waste per million Btu heat input from the hazardous waste; and

(ii) Emissions in excess of 330 µgm/ dscm, combined emissions, corrected to 7 percent oxygen;

(4) For arsenic, beryllium, and chromium, both:

(i) Emissions in excess of 2.1×10^{-5} lbs combined emissions of arsenic, beryllium, and chromium attributable to the hazardous waste per million Btu heat input from the hazardous waste; and

(ii) Emissions in excess of 56 µgm/ dscm, combined emissions, corrected to 7 percent oxygen;

(5) Carbon monoxide and hydrocarbons.(i) For kilns equipped with a by-pass duct or midkiln gas sampling system, either:

(A) Carbon monoxide in the by-pass duct or mid-kiln gas sampling system in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (a)(5)(i)(B) of this section, you must also document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons in the by-pass duct or mid-kiln gas sampling system do not exceed 10 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or

(B) Hydrocarbons in the by-pass duct or midkiln gas sampling system in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane;

(ii) For kilns not equipped with a bypass duct or midkiln gas sampling system, either:

(A) Hydrocarbons in the main stack in excess of 20 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or

(B) Carbon monoxide in the main stack in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (a)(5)(ii)(A) of this section, you also must document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons in the main stack do not exceed 20 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane.

(6) Hydrogen chloride and chlorine gas in excess of 120 parts per million by volume, combined emissions, expressed as a chloride $(Cl^{(-)})$ equivalent, dry basis, corrected to 7 percent oxygen; and

(7) For particulate matter, both:

(i) Emissions in excess of 0.028 gr/dscf corrected to 7 percent oxygen; and

(ii) Opacity greater than 20 percent, unless your source is equipped with a bag leak detection system under §63.1206(c)(8) or a particulate matter detection system under §63.1206(c)(9).

(b) Emission and hazardous waste feed limits for new sources. You must not discharge or cause combustion gases to be emitted into the atmosphere or feed hazardous waste that contain:

(1) For dioxins and furans, either:

(i) Emissions in excess of 0.20 ng TEQ/ dscm corrected to 7 percent oxygen; or

(ii) Emissions in excess of 0.40 ng TEQ/dscm corrected to 7 percent oxygen provided that the combustion gas temperature at the inlet to the initial dry particulate matter control device is 400 °F or lower based on the average of the test run average temperatures;

(2) For mercury, both:

(i) An average as-fired concentration of mercury in all hazardous waste feedstreams in excess of 1.9 parts per million by weight; and

(ii) Either:

(A) Emissions in excess of 120 $\mu g/$ dscm, corrected to 7 percent oxygen, or

(B) A hazardous waste feed maximum theoretical emission concentration (MTEC) in excess of 120 µg/dscm;

(3) For cadmium and lead, both:

(i) Emissions in excess of 6.2×10^{-5} lbs combined emissions of cadmium and lead attributable to the hazardous waste per million Btu heat input from the hazardous waste; and

(ii) Emissions in excess of 180 µgm/ dscm, combined emissions, corrected to 7 percent oxygen;

(4) For arsenic, beryllium, and chromium, both:

(i) Emissions in excess of 1.5×10^{-5} lbs combined emissions of arsenic, beryllium, and chromium attributable to the hazardous waste per million Btu heat input from the hazardous waste; and

(ii) Emissions in excess of 54 µgm/ dscm, combined emissions, corrected to 7 percent oxygen;

(5) Carbon monoxide and hydrocarbons.
(i) For kilns equipped with a by-pass duct or midkiln gas sampling system, carbon monoxide and hydrocarbons emissions are limited in both the by-pass duct or midkiln gas sampling system and the main stack as follows:

(A) Emissions in the by-pass or midkiln gas sampling system are limited to either:

(1) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (b)(5)(i)(A)(2)of this section, you also must document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons do not exceed 10 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to

7 percent oxygen, and reported as propane; or

(2) Hydrocarbons in the by-pass duct or midkiln gas sampling system in excess of 10 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; and

(B) Hydrocarbons in the main stack are limited, if construction of the kiln commenced after April 19, 1996 at a plant site where a cement kiln (whether burning hazardous waste or not) did not previously exist, to 50 parts per million by volume, over a 30-day block average (monitored continuously with a continuous monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane.

(ii) For kilns not equipped with a bypass duct or midkiln gas sampling system, hydrocarbons and carbon monoxide are limited in the main stack to either:

(A) Hydrocarbons not exceeding 20 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane; or

(B)(1) Carbon monoxide not exceeding 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen; and

(2) Hydrocarbons not exceeding 20 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane at any time during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7); and

(3) If construction of the kiln commenced after April 19, 1996 at a plant site where a cement kiln (whether burning hazardous waste or not) did not previously exist, hydrocarbons are limited to 50 parts per million by volume, over a 30-day block average (monitored continuously with a continuous monitoring system), dry basis, cor40 CFR Ch. I (7–1–23 Edition)

rected to 7 percent oxygen, and reported as propane.

(6) Hydrogen chloride and chlorine gas in excess of 86 parts per million by volume, combined emissions, expressed as a chloride $(Cl^{(-)})$ equivalent, dry basis and corrected to 7 percent oxygen; and

(7) For particulate matter, both:

(i) Emissions in excess of 0.0069 gr/ dscf corrected to 7 percent oxygen; and

(ii) Opacity greater than 20 percent, unless your source is equipped with a bag leak detection system under §63.1206(c)(8) or a particulate matter detection system under §63.1206(c)(9).

(c) Destruction and removal efficiency (DRE) standard—(1) 99.99% DRE. Except as provided in paragraph (c)(2) of this section, you must achieve a destruction and removal efficiency (DRE) of 99.99% for each principle organic hazardous constituent (POHC) designated under paragraph (c)(3) of this section. You must calculate DRE for each POHC from the following equation:

DRE = $[1 - (W_{out} / W_{in})] \times 100\%$

Where:

- W_{in} = mass feedrate of one POHC in a waste feedstream; and
- W_{out} = mass emission rate of the same POHC present in exhaust emissions prior to release to the atmosphere.

(2) 99.9999%~DRE. If you burn the dioxin-listed hazardous wastes F020, F021, F022, F023, F026, or F027 (see §261.31 of this chapter), you must achieve a DRE of 99.9999% for each POHC that you designate under paragraph (c)(3) of this section. You must demonstrate this DRE performance on POHCs that are more difficult to incinerate than tetra-. penta-. and hexachlorodibenzo-p-dioxins and dibenzofurans. You must use the equation in paragraph (c)(1) of this section to calculate DRE for each POHC. In addition, you must notify the Administrator of your intent to incinerate hazardous wastes F020, F021, F022, F023, F026, or F027.

(3) Principal organic hazardous constituent (POHC). (i) You must treat each POHC in the waste feed that you specify under paragraph (c)(3)(ii) of this section to the extent required by paragraphs (c)(1) and (c)(2) of this section.

(ii) You must specify one or more POHCs that are representative of the most difficult to destroy organic compounds in your hazardous waste feedstream. You must base this specification on the degree of difficulty of incineration of the organic constituents in the hazardous waste and on their concentration or mass in the hazardous waste feed, considering the results of hazardous waste analyses or other data and information.

(d) Cement kilns with in-line kiln raw mills—(1) General. (i) You must conduct performance testing when the raw mill is on-line and when the mill is off-line to demonstrate compliance with the emission standards, and you must establish separate operating parameter limits under 63.1209 for each mode of operation, except as provided by paragraphs (d)(1)(iv) and (d)(1)(v) of this section.

(ii) You must document in the operating record each time you change from one mode of operation to the alternate mode and begin complying with the operating parameter limits for that alternate mode of operation.

(iii) You must calculate rolling averages for operating parameter limits as provided by 63.1209(q)(2).

(iv) If your in-line kiln raw mill has dual stacks, you may assume that the dioxin/furan emission levels in the bypass stack and the operating parameter limits determined during performance testing of the by-pass stack when the raw mill is off-line are the same as when the mill is on-line.

(v) In lieu of conducting a performance test to demonstrate compliance with the dioxin/furan emission standards for the mode of operation when the raw mill is on-line, you may specify in the performance test workplan and Notification of Compliance the same operating parameter limits required under §63.1209(k) for the mode of operation when the raw mill is on-line as you establish during performance testing for the mode of operation when the raw mill is off-line.

(2) *Emissions averaging*. You may comply with the mercury, semivolatile metal, low volatile metal, and hydrogen chloride/chlorine gas emission standards on a time-weighted average basis under the following procedures: (i) Averaging methodology. You must calculate the time-weighted average emission concentration with the following equation:

$$\begin{array}{l} C_{total} \ = \ \{C_{mill-off} \ \times \ (T_{mill-off} \ / \ (T_{mill-off} \ + \\ T_{mill-on}))\} \ + \ \{C_{mill-on} \ \times \ (T_{mill-on} \ / \ (T_{mill-off} \ + \\ T_{mill-on}))\} \end{array}$$

Where:

- C_{total} = time-weighted average concentration of a regulated constituent considering both raw mill on time and off time;
- $C_{mill-off}$ = average performance test concentration of regulated constituent with the raw mill off-line:
- $C_{mill-on} = average \ performance \ test \ concentration \ of \ regulated \ constituent \ with \ the \ raw \ mill \ on-line;$
- $T_{\rm mill-off}$ = time when kiln gases are not routed through the raw mill; and

 $T_{mill-on}$ = time when kiln gases are routed through the raw mill.

(ii) Compliance. (A) If you use this emission averaging provision, you must document in the operating record compliance with the emission standards on an annual basis by using the equation provided by paragraph (d)(2) of this section.

(B) Compliance is based on one-year block averages beginning on the day you submit the initial notification of compliance.

(iii) Notification. (A) If you elect to document compliance with one or more emission standards using this emission averaging provision, you must notify the Administrator in the initial comprehensive performance test plan submitted under §63.1207(e).

(B) You must include historical raw mill operation data in the performance test plan to estimate future raw mill down-time and document in the performance test plan that estimated emissions and estimated raw mill down-time will not result in an exceedance of an emission standard on an annual basis.

(C) You must document in the notification of compliance submitted under §63.1207(j) that an emission standard will not be exceeded based on the documented emissions from the performance test and predicted raw mill downtime.

(e) Preheater or preheater/precalciner kilns with dual stacks—(1) General. You must conduct performance testing on each stack to demonstrate compliance with the emission standards, and you must establish operating parameter limits under 63.1209 for each stack, except as provided by paragraph (d)(1)(iv) of this section for dioxin/furan emissions testing and operating parameter limits for the by-pass stack of in-line raw mills.

(2) Emissions averaging. You may comply with the mercury, semivolatile metal, low volatile metal, and hydrogen chloride/chlorine gas emission standards specified in this section on a gas flowrate-weighted average basis under the following procedures:

(i) Averaging methodology. You must calculate the gas flowrate-weighted average emission concentration using the following equation:

 $\begin{array}{l} C_{tot} \ = \ \{C_{main} \times (Q_{main} \ / \ (Q_{main} \ + \ Q_{bypass}))\} \ + \\ \{C_{bypass} \times (Q_{bypass} \ / \ (Q_{main} \ + \ Q_{bypass}))\} \end{array}$

Where:

C_{tot} = gas flowrate-weighted average concentration of the regulated constituent:

C_{main} = average performance test concentration demonstrated in the main stack:

 C_{bypass} = average performance test concentration demonstrated in the bypass stack;

 Q_{main} = volumetric flowrate of main stack effluent gas; and

 Q_{bypass} = volumetric flowrate of bypass effluent gas.

(ii) Compliance. (A) You must demonstrate compliance with the emission standard(s) using the emission concentrations determined from the performance tests and the equation provided by paragraph (e)(1) of this section; and

(B) You must develop operating parameter limits for bypass stack and main stack flowrates that ensure the emission concentrations calculated with the equation in paragraph (e)(1) of this section do not exceed the emission standards on a 12-hour rolling average basis. You must include these flowrate limits in the Notification of Compliance.

(iii) *Notification*. If you elect to document compliance under this emissions averaging provision, you must:

(A) Notify the Administrator in the initial comprehensive performance test plan submitted under §63.1207(e). The performance test plan must include, at a minimum, information describing the flowrate limits established under paragraph (e)(2)(ii)(B) of this section; and

40 CFR Ch. I (7–1–23 Edition)

(B) Document in the Notification of Compliance submitted under 63.1207(j)the demonstrated gas flowrate-weighted average emissions that you calculate with the equation provided by paragraph (e)(2) of this section.

(f) Significant figures. The emission limits provided by paragraphs (a) and (b) of this section are presented with two significant figures. Although you must perform intermediate calculations using at least three significant figures, you may round the resultant emission levels to two significant figures to document compliance.

(g) [Reserved]

(h) When you comply with the particulate matter requirements of paragraphs (a)(7) or (b)(7) of this section, you are exempt from the New Source Performance Standard for particulate matter and opacity under 60.60 of this chapter.

[70 FR 59571, Oct. 12, 2005, as amended at 71
 FR 62394, Oct. 25, 2006; 73 FR 18983, Apr. 8, 2008; 73 FR 64097, Oct. 28, 2008]

§63.1221 What are the replacement standards for hazardous waste burning lightweight aggregate kilns?

(a) Emission and hazardous waste feed limits for existing sources. You must not discharge or cause combustion gases to be emitted into the atmosphere or feed hazardous waste that contain:

(1) For dioxins and furans, either:

(i) Emissions in excess of $0.20~\mathrm{ng}$ TEQ/ dscm corrected to 7 percent oxygen; or

(ii) Rapid quench of the combustion gas temperature at the exit of the (last) combustion chamber (or exit of any waste heat recovery system that immediately follows the last combustion chamber) to 400 °F or lower based on the average of the test run average temperatures. You must also notify in writing the RCRA authority that you are complying with this option;

(2) For mercury, either:

(i) Emissions in excess of 120 $\mu gm/$ dscm, corrected to 7 percent oxygen; or

(ii) A hazardous waste feedrate corresponding to a maximum theoretical emission concentration (MTEC) in excess of 120 µgm/dscm;

(3) For cadmium and lead, both:

(i) Emissions in excess of 3.0×10^{-4} lbs combined emissions of cadmium

and lead attributable to the hazardous waste per million Btu heat input from the hazardous waste; and

(ii) Emissions in excess of 250 µgm/ dscm, combined emissions, corrected to 7 percent oxygen;

(4) For arsenic, beryllium, and chromium, both:

(i) In excess of 9.5×10^{-5} lbs combined emissions of arsenic, beryllium, and chromium attributable to the hazardous waste per million Btu heat input from the hazardous waste;

(ii) Emissions in excess of 110 µgm/ dscm, combined emissions, corrected to 7 percent oxygen;

(5) Carbon monoxide and hydrocarbons. (i) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (a)(5)(ii) of this section, you also must document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons do not exceed 20 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to 7 percent oxygen, and reported as propane: or

(ii) Hydrocarbons in excess of 20 parts per million by volume, over an hourly rolling average, dry basis, corrected to 7 percent oxygen, and reported as propane;

(6) Hydrogen chloride and chlorine gas in excess of 600 parts per million by volume, combined emissions, expressed as a chloride $(Cl^{(-)})$ equivalent, dry basis and corrected to 7 percent oxygen; and

(7) Particulate matter emissions in excess of 0.025 gr/dscf, corrected to 7 percent oxygen.

(b) Emission and hazardous waste feed limits for new sources. You must not discharge or cause combustion gases to be emitted into the atmosphere or feed hazardous waste that contain:

(1) For dioxins and furans, either:

(i) Emissions in excess of $0.20~\mathrm{ng}$ TEQ/ dscm corrected to 7 percent oxygen; or

(ii) Rapid quench of the combustion gas temperature at the exit of the (last) combustion chamber (or exit of any waste heat recovery system that immediately follows the last combustion chamber) to $400 \, ^\circ\mathrm{F}$ or lower based on the average of the test run average temperatures. You must also notify in writing the RCRA authority that you are complying with this option;

(2) For mercury, either:

(i) Emissions in excess of 120 $\mu gm/$ dscm, corrected to 7 percent oxygen; or

(ii) A hazardous waste feedrate corresponding to a maximum theoretical emission concentration (MTEC) in excess of 120 μgm/dscm;

(3) For cadmium and lead, both:

(i) Emissions in excess of 3.7×10^{-5} lbs combined emissions of cadmium and lead attributable to the hazardous waste per million Btu heat input from the hazardous waste; and

(ii) Emissions in excess of 43 μgm/ dscm, combined emissions, corrected to 7 percent oxygen;

(4) For arsenic, beryllium, and chromium, both:

(i) In excess of 3.3×10^{-5} lbs combined emissions of arsenic, beryllium, and chromium attributable to the hazardous waste per million Btu heat input from the hazardous waste;

(ii) Emissions in excess of 110 µgm/ dscm, combined emissions, corrected to 7 percent oxygen;

(5) Carbon monoxide and hydrocarbons. (i) Carbon monoxide in excess of 100 parts per million by volume, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis and corrected to 7 percent oxygen. If you elect to comply with this carbon monoxide standard rather than the hydrocarbon standard under paragraph (b)(5)(ii) of this section, you also must document that, during the destruction and removal efficiency (DRE) test runs or their equivalent as provided by §63.1206(b)(7), hydrocarbons do not exceed 20 parts per million by volume during those runs, over an hourly rolling average (monitored continuously with a continuous emissions monitoring system), dry basis, corrected to

Pt. 63, Subpt. EEE, Table 1

7 percent oxygen, and reported as propane; or

(ii) Hydrocarbons in excess of 20 parts per million by volume, over an hourly rolling average, dry basis, corrected to 7 percent oxygen, and reported as propane;

(6) Hydrogen chloride and chlorine gas in excess of 600 parts per million by volume, combined emissions, expressed as a chloride ($Cl^{(-)}$) equivalent, dry basis and corrected to 7 percent oxygen; and

(7) Particulate matter emissions in excess of 0.0098 gr/dscf corrected to 7 percent oxygen.

(c) Destruction and removal efficiency (DRE) standard—(1) 99.99% DRE. Except as provided in paragraph (c)(2) of this section, you must achieve a destruction and removal efficiency (DRE) of 99.99% for each principal organic hazardous constituent (POHC) designated under paragraph (c)(3) of this section. You must calculate DRE for each POHC from the following equation:

 $DRE = [1 - (W_{out} / Win)] \times 100\%$

Where:

- W_{in} = mass feedrate of one POHC in a waste feedstream; and
- W_{out} = mass emission rate of the same POHC present in exhaust emissions prior to release to the atmosphere.

(2) 99.9999% DRE. If you burn the dioxin-listed hazardous wastes F020, F021, F022, F023, F026, or F027 (see §261.31 of this chapter), you must achieve a destruction and removal efficiency (DRE) of 99.9999% for each POHC that you designate under paragraph (c)(3) of this section. You must

40 CFR Ch. I (7–1–23 Edition)

demonstrate this DRE performance on POHCs that are more difficult to incinerate than tetra-, penta-, and hexachlorodibenzo-dioxins and dibenzofurans. You must use the equation in paragraph (c)(1) of this section to calculate DRE for each POHC. In addition, you must notify the Administrator of your intent to burn hazardous wastes F020, F021, F022, F023, F026, or F027.

(3) Principal organic hazardous constituents (POHCs). (i) You must treat each POHC in the waste feed that you specify under paragraph (c)(3)(ii) of this section to the extent required by paragraphs (c)(1) and (c)(2) of this section.

(ii) You must specify one or more POHCs that are representative of the most difficult to destroy organic compounds in your hazardous waste feedstream. You must base this specification on the degree of difficulty of incineration of the organic constituents in the hazardous waste and on their concentration or mass in the hazardous waste feed, considering the results of hazardous waste analyses or other data and information.

(d) Significant figures. The emission limits provided by paragraphs (a) and (b) of this section are presented with two significant figures. Although you must perform intermediate calculations using at least three significant figures, you may round the resultant emission levels to two significant figures to document compliance.

[70 FR 59574, Oct. 12, 2005]

TABLE 1 TO SUBPART EEE OF PART 63—GENERAL PROVISIONS APPLICABLE TO SUBPART EEE

Reference	Applies to subpart EEE	Explanation
63.1	Yes.	
63.2	Yes.	
63.3	Yes.	
63.4	Yes	
63.5	Yes.	
63.6(a), (b), (c), (d), and (e)	Yes.	
63.6(f)	Yes	Except that the performance test requirements of Sec. 63.1207 apply instead of □63.6(f)(2)(iii)(B).
63.6(g) and (h)	Yes.	
63.6(i)	Yes	Section 63.1213 specifies that the compliance date may also be extended for inability to install necessary emis- sion control equipment by the compliance date because of implementation of pollution prevention or waste mini- mization controls.
Pt. 63, Subpt. EEE, Table 1

Reference	Applies to subpart EEE	Explanation	
63.6())	Yes.		
63.7(a)	Yes	Except _63.1207(e)(3) allows you to petition the Adminis- trator under _63.7(h) to provide an extension of time to conduct a performance test.	
63.7(b)	Yes	Except □63.1207(e) requires you to submit the site-spe- cific test plan for approval at least one year before the comprehensive performance test is scheduled to begin.	
63.7(c)	Yes	Except □63.1207(e) requires you to submit the site-spe- cific test plan (including the quality assurance provi- sions under □63.7(c)) for approval at least one year be- fore the comprehensive performance test is scheduled to begin.	
63.7(e)	Yes. Yes	Except _63.1207 prescribes operations during perform- ance testing and _63.1209 specifies operating limits that will be established during performance testing (such that testing is likely to be representative of the extreme range of normal performance).	
63.7(f)	Yes.		
63.7(g)	Yes	Except $\square 63.120/(1)$ requiring that you submit the results of the performance test (and the notification of compli- ance) within 90 days of completing the test, unless the Administrator grants a time extension, applies instead of $\square 63.7(g)(1)$.	
63.7(h)	Yes	Except □63.1207(c)(2) allows data in lieu of the initial comprehensive performance test, and □63.1207(m) provides a waiver of certain performance tests. You must submit requests for these waivers with the site-specific test plan.	
63.8(a) and (b)	Yes.		
63.8(c)	Yes	Except: (1) □63.1211(c) that requires you to install, cali- brate, and operate C□S by the compliance date ap- plies instead of □63.8(c)(3)□and (2) the performance specifications for CO, HC, and O2 CE□S in subpart B,	
		of this chapter requiring that the detectors measure the sample concentration at least once every 15 seconds for calculating an average emission level once every 60 seconds apply instead of □63.8(c)(4)(ii).	
63.8(d)	Yes.		
63.8(e)	Yes	Except □63.1207(e) requiring you to submit the site-spe- cific comprehensive performance test plan and the C□S performance evaluation test plan for approval at least one year prior to the planned test date applies in- stead of □063.8(e)(2) and (3)(iii).	
63.8(f) and (g)	Yes.		
63.9(b)	Yes	Note: Section 63.9(b)(1)(ii) pertains to notification require- ments for area sources that become a major source, and §63.9(b)(2)(v) requires a major source determina- tion. Although area sources are subject to all provisions of this subpart (Subpart EEE), these sections nonethe- less apply because the major source determination may affect the applicability or part 62 cherdardh er title	
		V permit requirements to other sources (i.e., other than a hazardous waste combustor) of hazardous air pollut- ants at the facility.	
63.9(c) and (d) 63.9(e)	Yes. Yes	Except §63.1207(e) which requires you to submit the comprehensive performance test plan for approval one year prior to the planned performance test date applies instead of 8.63.9(e).	
63.9(f)	Yes	Section 63.9(f) applies if you are allowed under §63.1209(a)(1)(v) to use visible determination of opac- ity for compliance in lieu of a COMS.	
63.9(g) 63.9(h)	Yes Yes	Except §63.9(g)(2) pertaining to COMS does not apply. Except §63.1207(j) requiring you to submit the notification of compliance within 90 days of completing a perform- ance test unless the Administrator grants a time exten- sion applies instead of §63.9(h)(2)(iii). Note: Even though area sources are subject to this subpart, the major source determination required by §63.9(h)(2)(i)(E) is applicable to hazardous waste com-	
63.9(i) and (j)	Yes.	pustors for the reasons discussed above.	

Pt. 63, Subpt. EEE, App.

40 CFR Ch. I (7-1-23 Edition)

□eference	Applies to subpart EEE	Explanation
63.9(□) 63.10	Yes Yes	Only as specified in § 63.9(j). Except reports of performance test results required under § 63.10(d)(2) may be submitted up to 90 days after completion of the test.
63.11 63.12⊑63.1□	No. Yes.	

[67 FR 6994, Feb. 14, 2002, as amended at 85 FR 73897, Nov. 19, 2020]

APPENDIX A TO SUBPART EEE OF PART 63—QUALITY ASSURANCE PROCE-DURES FOR CONTINUOUS EMISSIONS MONITORS USED FOR HAZARDOUS WASTE COMBUSTORS

1. Applicability and Principle

1.1 Applicability. These quality assurance requirements are used to evaluate the effectiveness of quality control (QC) and quality assurance (QA) procedures and the quality of data produced by continuous emission monitoring systems (CEMS) that are used for determining compliance with the emission standards on a continuous basis as specified in the applicable regulation. The QA procedures specified by these requirements represent the minimum requirements necessary for the control and assessment of the quality of CEMS data used to demonstrate compliance with the emission standards provided under this subpart EEE of part 63. Owners and operators must meet these minimum requirements and are encouraged to develop and implement a more extensive QA program. These requirements supersede those found in part 60, Appendix F, of this chapter. Appendix F does not apply to hazardous waste-burning devices.

1.2 Principle. The QA procedures consist of two distinct and equally important functions. One function is the assessment of the quality of the CEMS data by estimating accuracy. The other function is the control and improvement of the quality of the CEMS data by implementing QC policies and corrective actions. These two functions form a control loop. When the assessment function indicates that the data quality is inadequate, the source must immediately stop burning hazardous waste. The CEM data control effort must be increased until the data quality is acceptable before hazardous waste burning can resume.

a. In order to provide uniformity in the assessment and reporting of data quality, this procedure explicitly specifies the assessment methods for response drift and accuracy. The methods are based on procedures included in the applicable performance specifications provided in appendix B to part 60 of this chapter. These procedures also require the analysis of the EPA audit samples concurrent with certain reference method (RM) analyses as specified in the applicable RM's.

b. Because the control and corrective action function encompasses a variety of policies, specifications, standards, and corrective measures, this procedure treats QC requirements in general terms to allow each source owner or operator to develop a QC system that is most effective and efficient for the circumstances.

2. Definitions

2.1 Continuous Emission Monitoring System (CEMS). The total equipment required for the determination of a pollutant concentration. The system consists of the following major subsystems:

2.1.1 Sample Interface. That portion of the CEMS used for one or more of the following: sample acquisition, sample transport, and sample conditioning, or protection of the monitor from the effects of the stack effluent.

2.1.2 *Pollutant Analyzer*. That portion of the CEMS that senses the pollutant concentration and generates a proportional output.

2.1.3 *Diluent Analyzer*. That portion of the CEMS that senses the diluent gas (O2) and generates an output proportional to the gas concentration.

2.1.4 Data Recorder. That portion of the CEMS that provides a permanent record of the analyzer output. The data recorder may provide automatic data reduction and CEMS control capabilities.

2.2 Relative Accuracy (RA). The absolute mean difference between the pollutant concentration determined by the CEMS and the value determined by the reference method (RM) plus the 2.5 percent error confidence coefficient of a series of test divided by the mean of the RM tests or the applicable emission limit.

2.3 Calibration Drift (CD). The difference in the CEMS output readings from the established reference value after a stated period of operation during which no unscheduled maintenance, repair, or adjustment took place.

2.4 Zero Drift (ZD). The difference in CEMS output readings at the zero pollutant level after a stated period of operation during which no unscheduled maintenance, repair, or adjustment took place.

2.5 Calibration Standard. Calibration standards produce a known and unchanging response when presented to the pollutant analyzer portion of the CEMS, and are used to calibrate the drift or response of the analyzer.

2.6 Relative Accuracy Test Audit (RATA). Comparison of CEMS measurements to reference method measurements in order to evaluate relative accuracy following procedures and specification given in the appropriate performance specification.

2.7 Absolute Calibration Audit (ACA). Equivalent to calibration error (CE) test defined in the appropriate performance specification using NIST traceable calibration standards to challenge the CEMS and assess accuracy.

2.8 *Rolling Average.* The average emissions, based on some (specified) time period, calculated every minute from a one-minute average of four measurements taken at 15-second intervals.

3. QA/QC Requirements

3.1 QC Requirements. a. Each owner or operator must develop and implement a QC program. At a minimum, each QC program must include written procedures describing in detail complete, step-by-step procedures and operations for the following activities.

1. Checks for component failures, leaks, and other abnormal conditions.

2. Calibration of CEMS.

3. CD determination and adjustment of CEMS.

4. Integration of CEMS with the automatic waste feed cutoff (AWFCO) system.

5. Preventive Maintenance of CEMS (including spare parts inventory).

6. Data recording, calculations, and report-

7. Checks of record keeping.

8. Accuracy audit procedures, including sampling and analysis methods.

9. Program of corrective action for malfunctioning CEMS.

10. Operator training and certification.

11. Maintaining and ensuring current certification or naming of cylinder gasses, metal solutions, and particulate samples used for audit and accuracy tests, daily checks, and calibrations.

b. Whenever excessive inaccuracies occur for two consecutive quarters, the current written procedures must be revised or the CEMS modified or replaced to correct the deficiency causing the excessive inaccuracies. These written procedures must be kept on record and available for inspection by the enforcement agency.

3.2 QA Requirements. Each source owner or operator must develop and implement a QA plan that includes, at a minimum, the following.

Pt. 63, Subpt. EEE, App.

1. QA responsibilities (including maintaining records, preparing reports, reviewing reports).

2. Schedules for the daily checks, periodic audits, and preventive maintenance.

3. Check lists and data sheets.

4. Preventive maintenance procedures.

5. Description of the media, format, and location of all records and reports.

6. Provisions for a review of the CEMS data at least once a year. Based on the results of the review, the owner or operator must revise or update the QA plan, if necessary.

4. CD and ZD Assessment and Daily System Audit

4.1 CD and ZD Requirement. Owners and operators must check, record, and quantify the ZD and the CD at least once daily (approximately 24 hours) in accordance with the method prescribed by the manufacturer. The CEMS calibration must, at a minimum, be adjusted whenever the daily ZD or CD exceeds the limits in the Performance Specifications. If, on any given ZD and/or CD check the ZD and/or CD exceed(s) two times the limits in the Performance Specifications, or if the cumulative adjustment to the ZD and/or CD (see Section 4.2) exceed(s) three times the limits in the Performance Specifications, hazardous waste burning must immediately cease and the CEMS must be serviced and recalibrated. Hazardous waste burning cannot resume until the owner or operator documents that the CEMS is in compliance with the Performance Specifications by carrying out an ACA.

4.2 Recording Requirements for Automatic ZD and CD Adjusting Monitors. Monitors that automatically adjust the data to the corrected calibration values must record the unadjusted concentration measurement prior to resetting the calibration, if performed, or record the amount of the adjustment.

4.3 Daily System Audit. The audit must include a review of the calibration check data, an inspection of the recording system, an inspection of the control panel warning lights, and an inspection of the sample transport and interface system (e.g., flowmeters, filters, etc.) as appropriate.

4.4 Data Recording and Reporting. All measurements from the CEMS must be retained in the operating record for at least 5 years.

5. Performance Evaluation for CO, O₂, and HC CEMS

Carbon Monoxide (CO), Oxygen (O_2) , and Hydrocarbon (HC) CEMS. An Absolute Calibration Audit (ACA) must be conducted quarterly, and a Relative Accuracy Test Audit (RATA) (if applicable, see sections 5.1 and 5.2 of this method) must be conducted

Pt. 63, Subpt. EEE, App.

yearly. When a performance test is also required under §63.1207 to document compliance with emission standards, the RATA must coincide with the performance test. The audits must be conducted as follows.

5.1 Relative Accuracy Test Audit (RATA). This requirement applies to O_2 and CO CEMS. The RATA must be conducted at least yearly. Conduct the RATA as described in the RA test procedure (or alternate procedures section) described in the applicable performance specifications. In addition, analyze the appropriate performance audit samples received from the EPA as described in the applicable sampling methods.

5.2 Absolute Calibration Audit (ACA). The ACA must be conducted at least quarterly except in a quarter when a RATA (if applicable, see section 5.1 of this method) is conducted instead. Conduct an ACA as described in the calibration error (CE) test procedure described in the applicable performance specifications.

5.3 Excessive Audit Inaccuracy. If the RA from the RATA or the CE from the ACA exceeds the criteria in the applicable performance specifications, hazardous waste burning must cease immediately. Hazardous waste burning cannot resume until the owner or operator takes corrective measures and audit the CEMS with a RATA to document that the CEMS is operating within the specifications.

6. Other Requirements

6.1 *Performance Specifications*. CEMS used by owners and operators of HWCs must comply with the following performance specifications in appendix B to part 60 of this chapter:

CEMS	□er- form- ance speci- fication
Carbon monoxide Oxygen ⊡otal hydrocarbons	

6.2 Downtime due to Calibration. Facilities may continue to burn hazardous waste for a maximum of 20 minutes while calibrating the CEMS. If all CEMS are calibrated at once, the facility must have twenty minutes to calibrate all the CEMS. If CEMS are calibrated individually, the facility must have twenty minutes to calibrate each CEMS. If the CEMS are calibrated individually, other CEMS must be operational while the individual CEMS is being calibrated.

6.3 Span of the CEMS.

6.3.1 CO CEMS. The CO CEM must have two ranges, a low range with a span of 200 ppmv and a high range with a span of 3000

40 CFR Ch. I (7–1–23 Edition)

ppmv at an oxygen correction factor of 1. A one-range CEM may be used, but it must meet the performance specifications for the low range in the specified span of the low range.

6.3.2 $O_{2 \text{ CEMS}}$. The O_2 CEM must have a span of 25 percent. The span may be higher than 25 percent if the O_2 concentration at the sampling point is greater than 25 percent.

6.3.3 *HC CEMS*. The HC CEM must have a span of 100 ppmv, expressed as propane, at an oxygen correction factor of 1.

6.9.4 CEMS Span Values. When the Oxygen Correction Factor is Greater than 2. When an owner or operator installs a CEMS at a location of high ambient air dilution, i.e., where the maximum oxygen correction factor as determined by the permitting agency is greater than 2, the owner or operator must install a CEM with a lower span(s), proportionate to the larger oxygen correction factor, than those specified above.

6.3.5 Use of Alternative Spans. Owner or operators may request approval to use alternative spans and ranges to those specified. Alternate spans must be approved in writing in advance by the Administrator. In considering approval of alternative spans and ranges, the Administrator will consider that measurements beyond the span will be recorded as values at the maximum span for purposes of calculating rolling averages.

6.3.6 Documentation of Span Values. The span value must be documented by the CEMS manufacturer with laboratory data.

6.4.1 Moisture Correction. Method 4 of appendix A, part 60 of this chapter, must be used to determine moisture content of the stack gasses.

6.4.2 Oxygen Correction Factor. Measured pollutant levels must be corrected for the amount of oxygen in the stack according to the following formula:

$$P_c = P_m \times 14/(E - Y)$$

Where:

- P_c = concentration of the pollutant or standard corrected to 7 percent oxygen, dry basis;
- P_m = measured concentration of the pollutant, dry basis;
- E = volume fraction of oxygen in the combustion air fed into the device, on a dry basis (normally 21 percent or 0.21 if only air is fed);
- Y = measured fraction of oxygen on a dry basis at the sampling point.

The oxygen correction factor is:

$$OCF = \frac{14}{E - Y}$$

6.4.3 *Temperature Correction*. Correction values for temperature are obtainable from standard reference materials.

6.5 *Rolling Average*. A rolling average is the arithmetic average of all one-minute averages over the averaging period.

6.5.1 One-Minute Average for CO and HHC CEMS. One-minute averages are the arithmetic average of the four most recent 15-second observations and must be calculated using the following equation:

$$\bar{c} = \sum_{i=1}^4 \frac{c_i}{4}$$

Where:

 \bar{c} = the one minute average

 c_{i} = a fifteen-second observation from the $C\mathrm{EM}$

Fifteen second observations must not be rounded or smoothed. Fifteen-second observations may be disregarded only as a result of a failure in the CEMS and allowed in the source's quality assurance plan at the time of the CEMS failure. One-minute averages must not be rounded, smoothed, or disregarded.

6.5.2 Ten Minute Rolling Average Equation. The ten minute rolling average must be calculated using the following equation:

$$C_{RA} = \sum_{i=1}^{10} \frac{\overline{c}_i}{10}$$

Where:

 $C_{\rm RA}$ = The concentration of the standard, expressed as a rolling average

 $\bar{c}_i = a$ one minute average

6.5.3 Hourly Rolling Average Equation for CO and THC CEMS and Operating Parameter Limits. The rolling average, based on a specific number integer of hours, must be calculated using the following equation:

$$C_{RA} = \sum_{i=1}^{60} \frac{\overline{c}_i}{60}$$

Where:

 $c_{\rm RA}$ = The concentration of the standard, expressed as a rolling average

 \bar{c}_i = a one minute average

6.5.4 Averaging Periods for CEMS other than CO and THC. The averaging period for CEMS other than CO and THC CEMS must be calculated as a rolling average of all onehour values over the averaging period. An hourly average is comprised of 4 measurements taken at equally spaced time intervals, or at most every 15 minutes. Fewer than 4 measurements might be available within an hour for reasons such as facility downtime or CEMS calibration. If at least two measurements (30 minutes of data) are available, an hourly average must be calculated. The *n*-hour rolling average is calculated by averaging the n most recent hourly averages.

6.6 Units of the Standards for the Purposes of Recording and Reporting Emissions. Emissions must be recorded and reported expressed after correcting for oxygen, temperature, and moisture. Emissions must be reported in metric, but may also be reported in the English system of units, at 7 percent oxygen, 20 °C, and on a dry basis.

6.7 Rounding and Significant Figures. Emissions must be rounded to two significant figures using ASTM procedure E-29-90 or its successor. Rounding must be avoided prior to rounding for the reported value.

7. Bibliography

1. 40 CFR part 60, appendix F, "Quality Assurance Procedures: Procedure 1. Quality Assurance Requirements for Gas continuous Emission Monitoring Systems Used For Compliance Determination".

[64 FR 53038, Sept. 30, 1999, as amended at 65 FR 42301, July 10, 2000; 88 FR 18412, Mar. 29, 2023]

Subpart FFF [Reserved]

Subpart GGG—National Emission Standards for Pharmaceuticals Production

SOURCE: 63 FR 50326, Sept. 21, 1998, unless otherwise noted.

§63.1250 Applicability.

(a) Definition of affected source. (1) The affected source subject to this subpart consists of the pharmaceutical manufacturing operations as defined in \S 63.1251. Except as specified in paragraph (d) of this section, the provisions of this subpart apply to pharmaceutical manufacturing operations that meet the criteria specified in paragraphs (a)(1) (i) through (iii) of this section:

(i) Manufacture a pharmaceutical product as defined in §63.1251;

(ii) Are located at a plant site that is a major source as defined in section 112(a) of the Act; and

(iii) Process, use, or produce HAP.

(2) Determination of the applicability of this subpart shall be reported as part of an operating permit application or as otherwise specified by the permitting authority.

(b) *New source applicability*. A new affected source subject to this subpart and to which the requirements for new

Appendix M

AUTHENTICATED U.S. GOVERNMENT INFORMATION

Subpart FFF—Emission Guidelines and Compliance Times for Other Solid Waste Incineration Units That Commenced Construction On or Before December 9, 2004

SOURCE: 70 FR 74907, Dec. 16, 2005, unless otherwise noted.

INTRODUCTION

§60.2980 What is the purpose of this subpart?

This subpart establishes emission guidelines and compliance schedules for the control of emissions from other solid waste incineration (OSWI) units. The pollutants addressed by these emission guidelines are listed in table 2 of this subpart. These emission guidelines are developed in accordance with sections 111(d) and 129 of the Clean Air Act and subpart B of this part.

§60.2981 Am I affected by this subpart?

(a) If you are the Administrator of an air quality program in a State or United States protectorate with one or more existing OSWI units or air curtain incinerators subject to this subpart as described in §60.2994(b) that commenced construction on or before December 9, 2004, you must submit a State plan to the U.S. Environmental Protection Agency (EPA) that implements the emission guidelines contained in this subpart.

(b) You must submit the State plan to EPA by December 18, 2006.

§60.2982 Is a State plan required for all States?

No, you are not required to submit a State plan if there are no existing OSWI units or air curtain incinerators subject to this subpart as described in $\S60.2994(b)$ in your State and you submit a negative declaration letter in place of the State plan.

§60.2983 What must I include in my State plan?

(a) You must include the following nine items in your State plan:

(1) Inventory of affected incineration units, including those that have ceased operation but have not been dismantled.

(2) Inventory of emissions from affected incineration units in your State.(3) Compliance schedules for each af-

(a) compliance schedules for each af fected incineration unit.

(4) For each affected incineration unit, emission limitations, operator training and qualification requirements, a waste management plan, and operating parameter requirements that are at least as protective as the emission guidelines contained in this subpart.

(5) Stack testing, recordkeeping, and reporting requirements.

(6) Transcript of the public hearing on the State plan.

(7) Provision for State progress reports to EPA.

(8) Identification of enforceable State mechanisms that you selected for implementing the emission guidelines of this subpart.

(9) Demonstration of your State's legal authority to carry out the sections 111(d) and 129 in your State plan.

(b) Your State plan may deviate from the format and content of the emission guidelines contained in this subpart. However, if your State plan does deviate, you must demonstrate that your State plan is at least as protective as the emission guidelines contained in this subpart. Your State plan must address regulatory applicability, compliance schedule, operator training and qualification, a waste management plan, emission limitations, stack testing, operating parameter requirements, monitoring, recordkeeping and reporting, and air curtain incinerator requirements.

(c) You must follow the requirements of subpart B of this part (Adoption and Submittal of State Plans for Designated Facilities) in your State plan.

§60.2984 Is there an approval process for my State plan?

Yes, EPA will review your State plan according to §60.27.

§60.2985 What if my State plan is not approvable?

If you do not submit an approvable State plan (or a negative declaration letter) by December 17, 2007, EPA will develop a Federal plan according to §60.27 to implement the emission guidelines contained in this subpart. Owners and operators of incineration units not covered by an approved State plan must comply with the Federal plan. The Federal plan is an interim action and applies to units until a State plan covering those units is approved and becomes effective.

§60.2986 Is there an approval process for a negative declaration letter?

No, EPA has no formal review process for negative declaration letters. Once we receive your negative declaration letter, we will place a copy in the public docket and publish a notice in the FEDERAL REGISTER. If, at a later date, an existing incineration unit is found in your State, the Federal plan implementing the emission guidelines contained in this subpart would automatically apply to that unit until your State plan is approved.

§60.2987 What compliance schedule must I include in my State plan?

Your State plan must include compliance schedules that require OSWI units and air curtain incinerators subject to this subpart as described in §60.2994(b) to achieve final compliance as expeditiously as practicable after approval of the State plan but not later than the earlier of the following two dates:

(a) December 16, 2010.

(b) Three years after the effective date of State plan approval.

§60.2988 Are there any State plan requirements for this subpart that apply instead of the requirements specified in subpart B of this part?

Yes, subpart B of this part establishes general requirements for developing and processing section 111(d) plans. This subpart applies instead of the requirements in subpart B of this part for the following:

(a) State plans developed to implement this subpart must be as protective as the emission guidelines contained in this subpart. State plans must require all OSWI units and air curtain incinerators subject to this subpart as described in $\S60.2994$ (b) to comply by December 16, 2010 or 3 years after the effective date of State plan approval, whichever is sooner. This applies instead of the option for case-bycase less stringent emission standards and longer compliance schedules in $\S60.24(f)$.

(b) State plans developed to implement this subpart are required to include only one increment of progress for the affected incineration units. This increment is the final compliance date in (0.21(h))(5). This applies instead of the requirement of (0.24(e))(1).

§60.2989 Does this subpart directly affect incineration unit owners and operators in my State?

(a) No, this subpart does not directly affect incineration unit owners and operators in your State. However, unit owners and operators must comply with the State plan you develop to implement the emission guidelines contained in this subpart.

(b) If you do not submit an approvable plan to implement and enforce the guidelines contained in this subpart by December 17, 2007, EPA will implement and enforce a Federal plan, as provided in §60.2985, to ensure that each unit within your State reaches compliance with all the provisions of this subpart by December 16, 2010.

§ 60.2990 What Authorities are withheld by EPA?

The following authorities are withheld by EPA and not transferred to the State, local or tribal agency:

(1) Approval of alternatives to the emission limitations in table 2 of this subpart and operating limits established under §60.3023 and table 3 of this subpart.

(2) Approval of petitions for specific operating limits in §60.3024.

(3) Approval of major alternatives to test methods.

(4) Approval of major alternatives to monitoring.

(5) Approval of major alternatives to recordkeeping and reporting.

(6) The status report requirements in §60.3020(c)(2).

Applicability of State Plans

§60.2991 What incineration units must I address in my State plan?

Your State plan must address all incineration units in your State that meet all the requirements specified in

paragraphs (a) through (c) of this section.

(a) The incineration unit is an existing incineration unit as defined in §60.2992.

(b) The incineration unit is an OSWI unit as defined in 60.3078 or an air curtain incinerator subject to this subpart as described in 60.2994(b). OSWI units are very small municipal waste combustion units and institutional waste incineration units as defined in 60.3078.

(c) The incineration unit is not excluded under §60.2993.

§60.2992 What is an existing incineration unit?

An existing incineration unit is an OSWI unit or air curtain incinerator subject to this subpart that commenced construction on or before December 9, 2004, except as provided in paragraph (a) of this section.

(a) If the owner or operator of an incineration unit makes changes that meet the definition of modification or reconstruction on or after June 16, 2006, the unit becomes subject to subpart EEEE of this part (New Source Performance Standards for Other Solid Waste Incineration Units) and the State plan no longer applies to that unit.

(b) If the owner or operator of an existing incineration unit makes physical or operational changes to the unit primarily to comply with the State plan, then subpart EEEE of this part does not apply to that unit. Such changes do not qualify as modifications or reconstructions under subpart EEEE of this part.

§ 60.2993 Are any combustion units excluded from my State plan?

This subpart excludes the types of units described in paragraphs (a) through (q) of this section, as long as the owner/operator meets the requirements of this section.

(a) Cement kilns. The unit is excluded if it is regulated under subpart LLL of part 63 of this chapter (National Emission Standards for Hazardous Air Pollutants from the Portland Cement Manufacturing Industry).

(b) Co-fired combustors. The unit, that would otherwise be considered a very

small municipal waste combustion unit, is excluded if the owner/operator of the unit meets the five requirements specified in paragraphs (b)(1) through (5) of this section.

(1) Has a Federally enforceable permit limiting the combustion of municipal solid waste to 30 percent of the total fuel input by weight.

(2) Notifies the Administrator that the unit qualifies for the exclusion.

(3) Provides the Administrator with a copy of the Federally enforceable permit.

(4) Records the weights, each calendar quarter, of municipal solid waste and of all other fuels combusted.

(5) Keeps each report for 5 years. These records must be kept on site for at least 2 years, but may be kept off site for the remaining 3 years.

(c) Cogeneration facilities. The unit is excluded if it meets the three requirements specified in paragraphs (c)(1)through (3) of this section.

(1) The unit qualifies as a cogeneration facility under section 3(18)(B) of the Federal Power Act (16 U.S.C. 796(18)(B)).

(2) The unit burns homogeneous waste (not including refuse-derived fuel) to produce electricity and steam or other forms of energy used for industrial, commercial, heating, or cooling purposes.

(3) The owner/operator of the unit notifies the Administrator that the unit meets all of these criteria.

(d) Commercial and industrial solid waste incineration units. The unit is excluded if it is regulated under subparts CCCC or DDDD of this part or subpart III of part 62 and is required to meet the emission limitations established in those subparts.

(e) Hazardous waste combustion units. The unit is excluded if it meets either of the two criteria specified in paragraph (e)(1) or (2) of this section.

(1) The owner/operator of the unit is required to get a permit for the unit under section 3005 of the Solid Waste Disposal Act.

(2) The unit is regulated under 40 CFR part 63, subpart EEE (National Emission Standards for Hazardous Air Pollutants from Hazardous Waste Combustors). (f) Hospital/medical/infectious waste incinerators. The unit is excluded if it is regulated under subparts Ce or Ec of this part (New Source Performance Standards and Emission Guidelines for Hospital/Medical/Infectious Waste Incinerators) or subpart HHH of part 62 (Federal Plan for Hospital/Medical/Infectious Waste Incinerators constructed on or before June 20, 1996).

(g) Incinerators and air curtain incinerators in isolated areas of Alaska. The incineration unit is excluded if it is used at a solid waste disposal site in Alaska that is classified as a Class II or Class III municipal solid waste landfill, as defined in §60.3078.

(h) Rural institutional waste incinerators. The incineration unit is excluded if it is an institutional waste incinerator, as defined in 60.3078, and the application for exclusion described in paragraphs (h)(1) and (2) of this section has been approved by the Administrator.

(1) Prior to 1 year before the final compliance date, an application and supporting documentation demonstrating that the institutional waste incineration unit meets the two requirements specified in paragraphs (h)(1)(i) and (ii) of this section must be submitted to the Administrator for approval.

(i) The unit is located more than 50 miles from the boundary of the nearest Metropolitan Statistical Area,

(ii) Alternative disposal options are not available or are economically infeasible.

(2) The application described in paragraph (h)(1) of this section must be revised and resubmitted to the Administrator for approval every 5 years following the initial approval of the exclusion for your unit.

(3) If you re-applied for an exclusion pursuant to paragraph (h)(2) of this section and were denied exclusion by the Administrator, you have 3 years from the expiration date of the current exclusion to comply with the emission limits and all other applicable requirements of this subpart.

(i) Institutional boilers and process heaters. The unit is excluded if it is regulated under 40 CFR part 63, subpart DDDDD (National Emission Standards for Hazardous Air Pollutants for Indus40 CFR Ch. I (7–1–23 Edition)

trial, Commercial, and Institutional Boilers and Process Heaters).

(j) Laboratory Analysis Units. The unit is excluded if it burns samples of materials only for the purpose of chemical or physical analysis.

(k) Materials recovery units. The unit is excluded if it combusts waste for the primary purpose of recovering metals. Examples include primary and secondary smelters.

(1) Pathological waste incineration units. The institutional waste incineration unit or very small municipal waste combustion unit is excluded from this subpart if it burns 90 percent or more by weight (on a calendar quarter basis and excluding the weight of auxiliary fuel and combustion air) of pathological waste, low-level radioactive waste, and/or chemotherapeutic waste as defined in §60.3078 and the owner/operator of the unit notifies the Administrator that the unit meets these criteria.

(m) Small or large municipal waste combustion units. The unit is excluded if it is regulated under subparts AAAA, BBBB, Ea, Eb, or Cb, of this part or subparts FFF or JJJ of part 62 and is required to meet the emission limitations established in those subparts.

(n) Small power production facilities. The unit is excluded if it meets the three requirements specified in paragraphs (n)(1) through (3) of this section.

(1) The unit qualifies as a small power-production facility under section 3(17)(C) of the Federal Power Act (16 U.S.C. 796(17)(C)).

(2) The unit burns homogeneous waste (not including refuse-derived fuel) to produce electricity.

(3) The owner/operator of the unit notifies the Administrator that the unit meets all of these criteria.

(o) Temporary-use incinerators and air curtain incinerators used in disaster recovery. The incineration unit is excluded if it is used on a temporary basis to combust debris from a disaster or emergency such as a tornado, hurricane, flood, ice storm, high winds, or act of bioterrorism and you comply with the requirements in §60.3061.

(p) Units that combust contraband or prohibited goods. The incineration unit is excluded if the unit is owned or operated by a government agency such as

police, customs, agricultural inspection, or a similar agency to destroy only illegal or prohibited goods such as illegal drugs, or agricultural food products that can not be transported into the country or across state lines to prevent biocontamination. The exclusion does not apply to items either confiscated or incinerated by private, industrial, or commercial entities.

(q) Incinerators used for national security. Your incineration unit is excluded if it meets the requirements specified in either (q)(1) or (2) of this section.

(1) The incineration unit is used solely during military training field exercises to destroy national security materials integral to the field exercises.

(2) The incineration unit is used solely to incinerate national security materials, its use is necessary to safeguard national security, you follow the exclusion request requirements in paragraphs (q)(2)(i) and (ii) of this section, and the Administrator has approved your request for exclusion.

(i) The request for exclusion and supporting documentation must demonstrate both that the incineration unit is used solely to destroy national security materials and that a reliable alternative to incineration that ensures acceptable destruction of national security materials is unavailable, on either a permanent or temporary basis.

(ii) The request for exclusion must be submitted to the Administrator prior to 1 year before the final compliance date.

§60.2994 Are air curtain incinerators regulated under this subpart?

(a) Air curtain incinerators that burn less than 35 tons per day of municipal solid waste or air curtain incinerators located at institutional facilities burning any amount of institutional waste generated at that facility are subject to all requirements of this subpart, including the emission limitations specified in table 2 of this subpart.

(b) Air curtain incinerators that burn only less than 35 tons per day of the materials listed in paragraphs (b)(1) through (4) of this section collected from the general public and from residential, commercial, institutional, and industrial sources; or, air curtain incinerators located at institutional facilities that burn only the materials listed in paragraphs (b)(1) through (4) of this section generated at that facility, are required to meet only the requirements in §§ 60.3062 through 60.3069 and are exempt from all other requirements of this subpart.

(1) 100 percent wood waste.

- (2) 100 percent clean lumber.
- (3) 100 percent yard waste.

(4) 100 percent mixture of only wood waste, clean lumber, and/or yard waste.

MODEL RULE—USE OF MODEL RULE

§60.2996 What is the purpose of the "model rule" in this subpart?

(a) The model rule provides the emission guidelines requirements in a standard regulation format. You must develop a State plan that is at least as protective as the model rule. You may use the model rule language as part of your State plan. Alternative language may be used in your State plan if you demonstrate that the alternative language is at least as protective as the model rule contained in this subpart.

(b) In the "model rule" of \$ 60.3000 through 60.3078, "you" means the owner or operator of an OSWI unit or air curtain incinerator subject to this subpart.

§60.2997 How does the model rule relate to the required elements of my State plan?

Use the model rule to satisfy the State plan requirements specified in (60.2983(a)(4)) and (5).

§60.2998 What are the principal components of the model rule?

The model rule contains nine major components, as follows:

(a) Compliance schedule.

(b) Waste management plan.

(c) Operator training and qualification.

 $\left(d\right)$ Emission limitations and operating limits.

(e) Performance testing.

(f) Initial compliance requirements.

(g) Continuous compliance requirements.

(h) Monitoring.

(i) Record keeping and reporting.

§60.3000

MODEL RULE—COMPLIANCE SCHEDULE

§60.3000 When must I comply?

Table 1 of this subpart specifies the final compliance date. You must submit a notification to the Administrator stating whether final compliance has been achieved, postmarked within 10 business days after the final compliance date in table 1 of this subpart.

§60.3001 What must I do if I close my OSWI unit and then restart it?

(a) If you close your OSWI unit but will reopen it prior to the final compliance date in your State plan, you must meet the final compliance date specified in table 1 of this subpart.

(b) If you close your OSWI unit but will restart it after your final compliance date, you must complete emission control retrofit and meet the emission limitations on the date your OSWI unit restarts operation. You must conduct your initial performance test within 30 days of restarting your OSWI unit.

§60.3002 What must I do if I plan to permanently close my OSWI unit and not restart it?

You must close the unit before the final compliance date specified in table 1 of this subpart.

MODEL RULE—WASTE MANAGEMENT PLAN

§60.3010 What is a waste management plan?

A waste management plan is a written plan that identifies both the feasibility and the methods used to reduce or separate certain components of solid waste from the waste stream in order to reduce or eliminate toxic emissions from incinerated waste.

§60.3011 When must I submit my waste management plan?

You must submit a waste management plan no later than 60 days following the initial performance test as specified in table 5 of this subpart. Section 60.3031 specifies the date by which you are required to conduct your performance test.

40 CFR Ch. I (7-1-23 Edition)

§60.3012 What should I include in my waste management plan?

A waste management plan must include consideration of the reduction or separation of waste-stream elements such as paper, cardboard, plastics, glass, batteries, or metals; or the use of recyclable materials. The plan must identify any additional waste management measures and implement those measures the source considers practical and feasible, considering the effectiveness of waste management measures already in place, the costs of additional measures, the emissions reductions expected to be achieved, and any other environmental or energy impacts they might have.

MODEL RULE—OPERATOR TRAINING AND QUALIFICATION

§60.3014 What are the operator training and qualification requirements?

(a) No OSWI unit can be operated unless a fully trained and qualified OSWI unit operator is accessible, either at the facility or can be at the facility within 1 hour. The trained and qualified OSWI unit operator may operate the OSWI unit directly or be the direct supervisor of one or more other plant personnel who operate the unit. If all qualified OSWI unit operators are temporarily not accessible, you must follow the procedures in §60.3020.

(b) Operator training and qualification must be obtained through a Stateapproved program or by completing the requirements included in paragraph (c) of this section.

(c) Training must be obtained by completing an incinerator operator training course that includes, at a minimum, the three elements described in paragraphs (c)(1) through (3) of this section.

(1) Training on the 13 subjects listed in paragraphs (c)(1)(i) through (xiii) of this section.

(i) Environmental concerns, including types of emissions.

(ii) Basic combustion principles, including products of combustion.

(iii) Operation of the specific type of incinerator to be used by the operator, including proper startup, waste charging, and shutdown procedures.

(iv) Combustion controls and monitoring.

(v) Operation of air pollution control equipment and factors affecting performance (if applicable).

(vi) Inspection and maintenance of the incinerator and air pollution control devices.

(vii) Methods to monitor pollutants (including monitoring of incinerator and control device operating parameters) and monitoring equipment calibration procedures, where applicable.

(viii) Actions to correct malfunctions or conditions that may lead to malfunction.

(ix) Bottom and fly ash characteristics and handling procedures.

(x) Applicable Federal, State, and local regulations, including Occupational Safety and Health Administration workplace standards.

(xi) Pollution prevention.

(xii) Waste management practices.

(xiii) Recordkeeping requirements.

(2) An examination designed and administered by the instructor.

(3) Written material covering the training course topics that may serve as reference material following completion of the course.

§60.3015 When must the operator training course be completed?

The operator training course must be completed by the latest of the three dates specified in paragraphs (a) through (c) of this section.

(a) The final compliance date specified in table 1 of this subpart.

(b) Six months after your OSWI unit startup.

(c) Six months after an employee assumes responsibility for operating the OSWI unit or assumes responsibility for supervising the operation of the OSWI unit.

§60.3016 How do I obtain my operator qualification?

(a) You must obtain operator qualification by completing a training course that satisfies the criteria under 60.3014(c).

(b) Qualification is valid from the date on which the training course is completed and the operator successfully passes the examination required under 60.3014(c)(2).

§60.3017 How do I maintain my operator qualification?

To maintain qualification, you must complete an annual review or refresher course covering, at a minimum, the five topics described in paragraphs (a) through (e) of this section.

(a) Update of regulations.

(b) Incinerator operation, including startup and shutdown procedures, waste charging, and ash handling.

(c) Inspection and maintenance.

(d) Responses to malfunctions or conditions that may lead to malfunction.

(e) Discussion of operating problems encountered by attendees.

§60.3018 How do I renew my lapsed operator qualification?

You must renew a lapsed operator qualification by one of the two methods specified in paragraphs (a) and (b) of this section.

(a) For a lapse of less than 3 years, you must complete a standard annual refresher course described in §60.3017.

(b) For a lapse of 3 years or more, you must repeat the initial qualification requirements in §60.3016(a).

§60.3019 What site-specific documentation is required?

(a) Documentation must be available at the facility and readily accessible for all OSWI unit operators that addresses the nine topics described in paragraphs (a)(1) through (9) of this section. You must maintain this information and the training records required by paragraph (c) of this section in a manner that they can be readily accessed and are suitable for inspection upon request.

(1) Summary of the applicable standards under this subpart.

(2) Procedures for receiving, handling, and charging waste.

(3) Incinerator startup, shutdown, and malfunction procedures.

(4) Procedures for maintaining proper combustion air supply levels.

(5) Procedures for operating the incinerator and associated air pollution control systems within the standards established under this subpart.

(6) Monitoring procedures for demonstrating compliance with the operating limits established under this subpart. (7) Reporting and recordkeeping procedures.

(8) The waste management plan required under §§ 60.3010 through 60.3012.

(9) Procedures for handling ash.

(b) You must establish a program for reviewing the information listed in paragraph (a) of this section with each incinerator operator.

(1) The initial review of the information listed in paragraph (a) of this section must be conducted by the latest of three dates specified in paragraphs (b)(1)(i) through (iii) of this section.

(i) The final compliance date specified in table 1 of this subpart.

(ii) Six months after your OSWI unit startup.

(iii) Six months after an employee assumes responsibility for operating the OSWI unit or assumes responsibility for supervising the operation of the OSWI unit.

(2) Subsequent annual reviews of the information listed in paragraph (a) of this section must be conducted not later than 12 months following the previous review.

(c) You must also maintain the information specified in paragraphs (c)(1) through (3) of this section.

(1) Records showing the names of OSWI unit operators who have completed review of the information in paragraph (a) of this section as required by paragraph (b) of this section, including the date of the initial review and all subsequent annual reviews.

(2) Records showing the names of the OSWI unit operators who have completed the operator training requirements under 60.3014, met the criteria for qualification under 60.3016, and maintained or renewed their qualification under 60.3017 or 60.3018. Records must include documentation of training, the dates of the initial and refresher training, and the dates of their qualification and all subsequent renewals of such qualifications.

(3) For each qualified operator, the phone and/or pager number at which they can be reached during operating hours.

40 CFR Ch. I (7–1–23 Edition)

§60.3020 What if all the qualified operators are temporarily not accessible?

If all qualified operators are temporarily not accessible (*i.e.*, not at the facility and not able to be at the facility within 1 hour), you must meet one of the three criteria specified in paragraphs (a) through (c) of this section, depending on the length of time that a qualified operator is not accessible.

(a) When all qualified operators are not accessible for 12 hours or less, the OSWI unit may be operated by other plant personnel familiar with the operation of the OSWI unit who have completed review of the information specified in §60.3019(a) within the past 12 months. You do not need to notify the Administrator or include this as a deviation in your annual report.

(b) When all qualified operators are not accessible for more than 12 hours, but less than 2 weeks, the OSWI unit may be operated by other plant personnel familiar with the operation of the OSWI unit who have completed a review of the information specified in $\S60.3019(a)$ within the past 12 months. However, you must record the period when all qualified operators were not accessible and include this deviation in the annual report as specified under \$60.3051.

(c) When all qualified operators are not accessible for 2 weeks or more, you must take the two actions that are described in paragraphs (c)(1) and (2) of this section.

(1) Notify the Administrator of this deviation in writing within 10 days. In the notice, state what caused this deviation, what you are doing to ensure that a qualified operator is accessible, and when you anticipate that a qualified operator will be accessible.

(2) Submit a status report to EPA every 4 weeks outlining what you are doing to ensure that a qualified operator is accessible, stating when you anticipate that a qualified operator will be accessible and requesting approval from EPA to continue operation of the OSWI unit. You must submit the first status report 4 weeks after you notify the Administrator of the deviation under paragraph (c)(1) of this section. If EPA notifies you that your request to continue operation of the OSWI unit

is disapproved, the OSWI unit may continue operation for 90 days, then must cease operation. Operation of the unit may resume if you meet the two requirements in paragraphs (c)(2)(i) and (ii) of this section.

(i) A qualified operator is accessible as required under 60.3014(a).

(ii) You notify EPA that a qualified operator is accessible and that you are resuming operation.

MODEL RULE—EMISSION LIMITATIONS AND OPERATING LIMITS

§60.3022 What emission limitations must I meet and by when?

You must meet the emission limitations specified in table 2 of this subpart on the date the initial performance test is required or completed (whichever is earlier). Section 60.3031 specifies the date by which you are required to conduct your performance test.

§60.3023 What operating limits must I meet and by when?

(a) If you use a wet scrubber to comply with the emission limitations, you must establish operating limits for four operating parameters (as specified in table 3 of this subpart) as described in paragraphs (a)(1) through (4) of this section during the initial performance test.

(1) Maximum charge rate, calculated using one of the two different procedures in paragraphs (a)(1)(i) or (ii) of this section, as appropriate.

(i) For continuous and intermittent units, maximum charge rate is the average charge rate measured during the most recent performance test demonstrating compliance with all applicable emission limitations.

(ii) For batch units, maximum charge rate is the charge rate measured during the most recent performance test demonstrating compliance with all applicable emission limitations.

(2) Minimum pressure drop across the wet scrubber, which is calculated as the average pressure drop across the wet scrubber measured during the most recent performance test demonstrating compliance with the particulate matter emission limitations; or minimum amperage to the wet scrubber, which is calculated as the average amperage to the wet scrubber measured during the most recent performance test demonstrating compliance with the particulate matter emission limitations.

(3) Minimum scrubber liquor flow rate, which is calculated as the average liquor flow rate at the inlet to the wet scrubber measured during the most recent performance test demonstrating compliance with all applicable emission limitations.

(4) Minimum scrubber liquor pH, which is calculated as the average liquor pH at the inlet to the wet scrubber measured during the most recent performance test demonstrating compliance with the hydrogen chloride and sulfur dioxide emission limitations.

(b) You must meet the operating limits established during the initial performance test beginning on the date 180 days after your final compliance date in table 1 of this subpart.

§60.3024 What if I do not use a wet scrubber to comply with the emission limitations?

If you use an air pollution control device other than a wet scrubber or limit emissions in some other manner to comply with the emission limitations under §60.3022, you must petition EPA for specific operating limits, the values of which are to be established during the initial performance test and then continuously monitored thereafter. You must not conduct the initial performance test until after the petition has been approved by EPA. Your petition must include the five items listed in paragraphs (a) through (e) of this section.

(a) Identification of the specific parameters you propose to use as operating limits.

(b) A discussion of the relationship between these parameters and emissions of regulated pollutants, identifying how emissions of regulated pollutants change with changes in these parameters, and how limits on these parameters will serve to limit emissions of regulated pollutants.

(c) A discussion of how you will establish the upper and/or lower values for these parameters that will establish the operating limits on these parameters.

§60.3025

(d) A discussion identifying the methods you will use to measure and the instruments you will use to monitor these parameters, as well as the relative accuracy and precision of these methods and instruments.

(e) A discussion identifying the frequency and methods for recalibrating the instruments you will use for monitoring these parameters.

§60.3025 What happens during periods of startup, shutdown, and malfunction?

The emission limitations and operating limits apply at all times except during OSWI unit startups, shutdowns, or malfunctions.

MODEL RULE—PERFORMANCE TESTING

§60.3027 How do I conduct the initial and annual performance test?

(a) All performance tests must consist of a minimum of three test runs conducted under conditions representative of normal operations.

(b) All performance tests must be conducted using the methods in table 2 of this subpart.

(c) All performance tests must be conducted using the minimum run duration specified in table 2 of this subpart.

(d) Method 1 of appendix A of this part must be used to select the sampling location and number of traverse points.

(e) Method 3A or 3B of appendix A of this part must be used for gas composition analysis, including measurement of oxygen concentration. Method 3A or 3B of appendix A of this part must be used simultaneously with each method.

(f) All pollutant concentrations, except for opacity, must be adjusted to 7 percent oxygen using Equation 1 in §60.3076.

(g) Method 26A of appendix A of this part must be used for hydrogen chloride concentration analysis, with the additional requirements specified in paragraphs (g)(1) through (3) of this section.

(1) The probe and filter must be conditioned prior to sampling using the procedure described in paragraphs (g)(1)(i) through (iii) of this section.

(i) Assemble the sampling train(s) and conduct a conditioning run by col-

40 CFR Ch. I (7–1–23 Edition)

lecting between 14 liters per minute (0.5 cubic feet per minute) and 30 liters per minute (1.0 cubic feet per minute) of gas over a 1-hour period. Follow the sampling procedures outlined in section 8.1.5 of Method 26A of appendix A of this part. For the conditioning run, water can be used as the impinger solution.

(ii) Remove the impingers from the sampling train and replace with a fresh impinger train for the sampling run, leaving the probe and filter (and cyclone, if used) in position. Do not recover the filter or rinse the probe before the first run. Thoroughly rinse the impingers used in the preconditioning run with deionized water and discard these rinses.

(iii) The probe and filter assembly are conditioned by the stack gas and are not recovered or cleaned until the end of testing.

(2) For the duration of sampling, a temperature around the probe and filter (and cyclone, if used) between 120 $^{\circ}\mathrm{C}$ (248 $^{\circ}\mathrm{F})$ and 134 $^{\circ}\mathrm{C}$ (273 $^{\circ}\mathrm{F})$ must be maintained.

(3) If water droplets are present in the sample gas stream, the requirements specified in paragraphs (g)(3)(i) and (ii) of this section must be met.

(i) The cyclone described in section 6.1.4 of Method 26A of appendix A of this part must be used.

(ii) The post-test moisture removal procedure described in section 8.1.6 of Method 26A of appendix A of this part must be used.

§60.3028 How are the performance test data used?

You use results of performance tests to demonstrate compliance with the emission limitations in table 2 of this subpart.

MODEL RULE—INITIAL COMPLIANCE REQUIREMENTS

§60.3030 How do I demonstrate initial compliance with the emission limitations and establish the operating limits?

You must conduct an initial performance test, as required under §60.8, to determine compliance with the emission limitations in table 2 of this subpart and to establish operating limits using the procedure in §60.3023 or

§60.3024. The initial performance test must be conducted using the test methods listed in table 2 of this subpart and the procedures in §60.3027.

§60.3031 By what date must I conduct the initial performance test?

The initial performance test must be conducted no later than 180 days after your final compliance date. Your final compliance date is specified in table 1 of this subpart.

MODEL RULE—CONTINUOUS COMPLIANCE REQUIREMENTS

§60.3033 How do I demonstrate continuous compliance with the emission limitations and the operating limits?

(a) You must conduct an annual performance test for all of the pollutants in table 2 of this subpart for each OSWI unit to determine compliance with the emission limitations. The annual performance test must be conducted using the test methods listed in table 2 of this subpart and the procedures in §60.3027.

(b) You must continuously monitor carbon monoxide emissions to determine compliance with the carbon monoxide emissions limitation. Twelvehour rolling average values are used to determine compliance. A 12-hour rolling average value above the carbon monoxide emission limit in table 2 constitutes a deviation from the emission limitation.

(c) You must continuously monitor the operating parameters specified in §60.3023 or established under §60.3024. Three-hour rolling average values are used to determine compliance with the operating limits unless a different averaging period is established under §60.3024. A 3-hour rolling average value (unless a different averaging period is established under §60.3024) above the established maximum or below the established minimum operating limits constitutes a deviation from the established operating limits. Operating limits do not apply during performance tests

§60.3034 By what date must I conduct the annual performance test?

You must conduct annual performance tests within 12 months following the initial performance test. Conduct subsequent annual performance tests within 12 months following the previous one.

§60.3035 May I conduct performance testing less often?

(a) You can test less often for a given pollutant if you have test data for at least three consecutive annual tests, and all performance tests for the pollutant over that period show that you comply with the emission limitation. In this case, you do not have to conduct a performance test for that polluant for the next 2 years. You must conduct a performance test during the 3rd year and no more than 36 months following the previous performance test.

(b) If your OSWI unit continues to meet the emission limitation for the pollutant, you may choose to conduct performance tests for that pollutant every 3rd year, but each test must be within 36 months of the previous performance test.

(c) If a performance test shows a deviation from an emission limitation for any pollutant, you must conduct annual performance tests for that pollutant until three consecutive annual performance tests for that pollutant all show compliance.

§60.3036 May I conduct a repeat performance test to establish new operating limits?

Yes, you may conduct a repeat performance test at any time to establish new values for the operating limits. The Administrator may request a repeat performance test at any time.

MODEL RULE-MONITORING

§60.3038 What continuous emission monitoring systems must I install?

(a) You must install, calibrate, maintain, and operate continuous emission monitoring systems for carbon monoxide and for oxygen. You must monitor the oxygen concentration at each location where you monitor carbon monoxide.

(b) You must install, evaluate, and operate each continuous emission monitoring system according to the "Monitoring Requirements" in §60.13.

§60.3039 How do I make sure my continuous emission monitoring systems are operating correctly?

(a) Conduct initial, daily, quarterly, and annual evaluations of your continuous emission monitoring systems that measure carbon monoxide and oxygen.

(b) Complete your initial evaluation of the continuous emission monitoring systems within 180 days after your final compliance date in table 1 of this subpart.

(c) For initial and annual evaluations, collect data concurrently (or within 30 to 60 minutes) using your carbon monoxide and oxygen continuous emission monitoring systems. To validate carbon monoxide concentration levels, use EPA Method 10, 10A, or 10B of appendix A of this part. Use EPA Method 3 or 3A to measure oxygen. Collect the data during each initial and annual evaluation of your continuous emission monitoring systems following the applicable performance specifications in appendix B of this part. table 4 of this subpart shows the required span values and performance specifications that apply to each continuous emission monitoring system.

(d) Follow the quality assurance procedures in Procedure 1 of appendix F of this part for each continuous emission monitoring system. The procedures include daily calibration drift and quarterly accuracy determinations.

§60.3040 What is my schedule for evaluating continuous emission monitoring systems?

(a) Conduct annual evaluations of your continuous emission monitoring systems no more than 12 months after the previous evaluation was conducted.

(b) Evaluate your continuous emission monitoring systems daily and quarterly as specified in appendix F of this part.

§60.3041 What is the minimum amount of monitoring data I must collect with my continuous emission monitoring systems, and is the data collection requirement enforceable?

(a) Where continuous emission monitoring systems are required, obtain 1hour arithmetic averages. Make sure the averages for carbon monoxide are in parts per million by dry volume at 7 40 CFR Ch. I (7–1–23 Edition)

percent oxygen. Use the 1-hour averages of oxygen data from your continuous emission monitoring system to determine the actual oxygen level and to calculate emissions at 7 percent oxygen.

(b) Obtain at least two data points per hour in order to calculate a valid 1hour arithmetic average. Section 60.13(e)(2) requires your continuous emission monitoring systems to complete at least one cycle of operation (sampling, analyzing, and data recording) for each 15-minute period.

(c) Obtain valid 1-hour averages for at least 75 percent of the operating hours per day for at least 90 percent of the operating days per calendar quarter. An operating day is any day the unit combusts any municipal or institutional solid waste.

(d) If you do not obtain the minimum data required in paragraphs (a) through (c) of this section, you have deviated from the data collection requirement regardless of the emission level monitored.

(e) If you do not obtain the minimum data required in paragraphs (a) through (c) of this section, you must still use all valid data from the continuous emission monitoring systems in calculating emission concentrations.

(f) If continuous emission monitoring systems are temporarily unavailable to meet the data collection requirements, refer to table 4 of this subpart. It shows alternate methods for collecting data when systems malfunction or when repairs, calibration checks, or zero and span checks keep you from collecting the minimum amount of data.

§60.3042 How do I convert my 1-hour arithmetic averages into the appropriate averaging times and units?

(a) Use Equation 1 in §60.3076 to calculate emissions at 7 percent oxygen.

(b) Use Equation 2 in §60.3076 to calculate the 12-hour rolling averages for concentrations of carbon monoxide.

§60.3043 What operating parameter monitoring equipment must I install, and what operating parameters must I monitor?

(a) If you are using a wet scrubber to comply with the emission limitations

under §60.3022, you must install, calibrate (to manufacturers' specifications), maintain, and operate devices (or establish methods) for monitoring the value of the operating parameters used to determine compliance with the operating limits listed in table 3 of this subpart. These devices (or methods) must measure and record the values for these operating parameters at the frequencies indicated in table 3 of this subpart at all times.

(b) You must install, calibrate (to manufacturers' specifications), maintain, and operate a device or method for measuring the use of any stack that could be used to bypass the control device. The measurement must include the date, time, and duration of the use of the bypass stack.

(c) If you are using a method or air pollution control device other than a wet scrubber to comply with the emission limitations under §60.3022, you must install, calibrate (to the manufacturers' specifications), maintain, and operate the equipment necessary to monitor compliance with the sitespecific operating limits established using the procedures in §60.3024.

§60.3044 Is there a minimum amount of operating parameter monitoring data I must obtain?

(a) Except for monitor malfunctions, associated repairs, and required quality assurance or quality control activities (including, as applicable, calibration checks and required zero and span adjustments of the monitoring system), you must conduct all monitoring at all times the OSWI unit is operating.

(b) You must obtain valid monitoring data for at least 75 percent of the operating hours per day for at least 90 percent of the operating days per calendar quarter. An operating day is any day the unit combusts any municipal or institutional solid waste.

(c) If you do not obtain the minimum data required in paragraphs (a) and (b) of this section, you have deviated from the data collection requirement regardless of the operating parameter level monitored.

(d) Do not use data recorded during monitor malfunctions, associated repairs, and required quality assurance or quality control activities for meeting the requirements of this subpart, including data averages and calculations. You must use all the data collected during all other periods in assessing compliance with the operating limits.

MODEL RULE—RECORDKEEPING AND REPORTING

§60.3046 What records must I keep?

You must maintain the 14 items (as applicable) as specified in paragraphs (a) through (n) of this section for a period of at least 5 years.

(a) Calendar date of each record.

(b) Records of the data described in paragraphs (b)(1) through (8) of this section.

(1) The OSWI unit charge dates, times, weights, and hourly charge rates.

(2) Liquor flow rate to the wet scrubber inlet every 15 minutes of operation, as applicable.

(3) Pressure drop across the wet scrubber system every 15 minutes of operation or amperage to the wet scrubber every 15 minutes of operation, as applicable.

(4) Liquor pH as introduced to the wet scrubber every 15 minutes of operation, as applicable.

(5) For OSWI units that establish operating limits for controls other than wet scrubbers under §60.3024, you must maintain data collected for all operating parameters used to determine compliance with the operating limits.

(6) All 1-hour average concentrations of carbon monoxide emissions.

(7) All 12-hour rolling average values of carbon monoxide emissions and all 3hour rolling average values of continuously monitored operating parameters.

(8) Records of the dates, times, and durations of any bypass of the control device.

(c) Identification of calendar dates and times for which continuous emission monitoring systems or monitoring systems used to monitor operating limits were inoperative, inactive, malfunctioning, or out of control (except for downtime associated with zero and span and other routine calibration checks). Identify the pollutant emissions or operating parameters not measured, the duration, reasons for not

40 CFR Ch. I (7–1–23 Edition)

obtaining the data, and a description of corrective actions taken.

(d) Identification of calendar dates, times, and durations of malfunctions, and a description of the malfunction and the corrective action taken.

(e) Identification of calendar dates and times for which monitoring data show a deviation from the carbon monoxide emissions limit in table 2 of this subpart or a deviation from the operating limits in table 3 of this subpart or a deviation from other operating limits established under §60.3024 with a description of the deviations, reasons for such deviations, and a description of corrective actions taken.

(f) Calendar dates when continuous monitoring systems did not collect the minimum amount of data required under §§ 60.3041 and 60.3044.

(g) For carbon monoxide continuous emissions monitoring systems, document the results of your daily drift tests and quarterly accuracy determinations according to Procedure 1 of appendix F of this part.

(h) Records of the calibration of any monitoring devices required under §60.3043.

(i) The results of the initial, annual, and any subsequent performance tests conducted to determine compliance with the emission limits and/or to establish operating limits, as applicable. Retain a copy of the complete test report including calculations and a description of the types of waste burned during the test.

(j) Records showing the names of OSWI unit operators who have completed review of the information in §60.3019(a) as required by §60.3019(b), including the date of the initial review and all subsequent annual reviews.

(k) Records showing the names of the OSWI unit operators who have completed the operator training requirements under §60.3014, met the criteria for qualification under §60.3016, and maintained or renewed their qualification under §60.3017 or §60.3018. Records must include documentation of training, the dates of the initial and refresher training, and the dates of their qualification and all subsequent renewals of such qualifications.

(1) For each qualified operator, the phone and/or pager number at which

they can be reached during operating hours.

(m) Equipment vendor specifications and related operation and maintenance requirements for the incinerator, emission controls, and monitoring equipment.

(n) The information listed in §60.3019(a).

§60.3047 Where and in what format must I keep my records?

(a) You must keep each record on site for at least 2 years. You may keep the records off site for the remaining 3 years.

(b) All records must be available in either paper copy or computer-readable format that can be printed upon request, unless an alternative format is approved by the Administrator.

§60.3048 What reports must I submit?

See table 5 of this subpart for a summary of the reporting requirements.

§60.3049 What information must I submit following my initial performance test?

You must submit the information specified in paragraphs (a) through (c) of this section no later than 60 days following the initial performance test. All reports must be signed by the facilities manager.

(a) The complete test report for the initial performance test results obtained under §60.3030, as applicable.

(b) The values for the site-specific operating limits established in 60.3023 or 60.3024.

(c) The waste management plan, as specified in §§ 60.3010 through 60.3012.

§60.3050 When must I submit my annual report?

You must submit an annual report no later than 12 months following the submission of the information in §60.3049. You must submit subsequent reports no more than 12 months following the previous report.

§60.3051 What information must I include in my annual report?

The annual report required under §60.3050 must include the ten items listed in paragraphs (a) through (j) of this section. If you have a deviation

from the operating limits or the emission limitations, you must also submit deviation reports as specified in §§ 60.3052 through 60.3054.

(a) Company name and address.

(b) Statement by the owner or operator, with their name, title, and signature, certifying the truth, accuracy, and completeness of the report. Such certifications must also comply with the requirements of 40 CFR 70.5(d) or 40 CFR 71.5(d).

(c) Date of report and beginning and ending dates of the reporting period.

(d) The values for the operating limits established pursuant to 60.3023 or 60.3024.

(e) If no deviation from any emission limitation or operating limit that applies to you has been reported, a statement that there was no deviation from the emission limitations or operating limits during the reporting period, and that no monitoring system used to determine compliance with the emission limitations or operating limits was inoperative, inactive, malfunctioning or out of control.

(f) The highest recorded 12-hour average and the lowest recorded 12-hour average, as applicable, for carbon monoxide emissions and the highest recorded 3-hour average and the lowest recorded 3-hour average, as applicable, for each operating parameter recorded for the calendar year being reported.

(g) Information recorded under §60.3046(b)(6) and (c) through (e) for the calendar year being reported.

(h) If a performance test was conducted during the reporting period, the results of that test.

(i) If you met the requirements of (0, 0, 0, 0) (b), and did not conduct a performance test during the reporting period, you must state that you met the requirements of (0, 0, 0, 0) (b), and, therefore, you were not required to conduct a performance test during the reporting period.

(j) Documentation of periods when all qualified OSWI unit operators were unavailable for more than 12 hours, but less than 2 weeks.

§60.3052 What else must I report if I have a deviation from the operating limits or the emission limitations?

(a) You must submit a deviation report if any recorded 3-hour average parameter level is above the maximum operating limit or below the minimum operating limit established under this subpart, if any recorded 12-hour average carbon monoxide emission rate is above the emission limitation, if the control device was bypassed, or if a performance test was conducted that showed a deviation from any emission limitation.

(b) The deviation report must be submitted by August 1 of that year for data collected during the first half of the calendar year (January 1 to June 30), and by February 1 of the following year for data you collected during the second half of the calendar year (July 1 to December 31).

§60.3053 What must I include in the deviation report?

In each report required under §60.3052, for any pollutant or operating parameter that deviated from the emission limitations or operating limits specified in this subpart, include the seven items described in paragraphs (a) through (g) of this section.

(a) The calendar dates and times your unit deviated from the emission limitations or operating limit requirements.

(b) The averaged and recorded data for those dates.

(c) Durations and causes of each deviation from the emission limitations or operating limits and your corrective actions.

(d) A copy of the operating limit monitoring data during each deviation and any test report that documents the emission levels.

(e) The dates, times, number, duration, and causes for monitor downtime incidents (other than downtime associated with zero, span, and other routine calibration checks).

(f) Whether each deviation occurred during a period of startup, shutdown, or malfunction, or during another period.

(g) The dates, times, and durations of any bypass of the control device.

§60.3054 What else must I report if I have a deviation from the requirement to have a qualified operator accessible?

(a) If all qualified operators are not accessible for 2 weeks or more, you must take the two actions in paragraphs (a)(1) and (2) of this section.

(1) Submit a notification of the deviation within 10 days that includes the three items in paragraphs (a)(1)(i)through (iii) of this section.

(i) A statement of what caused the deviation.

(ii) A description of what you are doing to ensure that a qualified operator is accessible.

(iii) The date when you anticipate that a qualified operator will be available.

(2) Submit a status report to EPA every 4 weeks that includes the three items in paragraphs (a)(2)(i) through (iii) of this section.

(i) A description of what you are doing to ensure that a qualified operator is accessible.

(ii) The date when you anticipate that a qualified operator will be accessible.

(iii) Request approval from EPA to continue operation of the OSWI unit.

(b) If your unit was shut down by EPA, under the provisions of \$0.3020(c)(2), due to a failure to provide an accessible qualified operator, you must notify EPA that you are resuming operation once a qualified operator is accessible.

§60.3055 Are there any other notifications or reports that I must submit?

Yes, you must submit notifications as provided by §60.7.

§60.3056 In what form can I submit my reports?

Submit initial, annual, and deviation reports electronically or in paper format, postmarked on or before the submittal due dates.

§60.3057 Can reporting dates be changed?

If the Administrator agrees, you may change the semiannual or annual reporting dates. See §60.19(c) for procedures to seek approval to change your reporting date.

40 CFR Ch. I (7–1–23 Edition)

MODEL RULE—TITLE V OPERATING PERMITS

§60.3059 Am I required to apply for and obtain a title V operating permit for my unit?

Yes, if you are subject to an applicable EPA-approved and effective Clean Air Act section 111(d)/129 State or Tribal plan or an applicable and effective Federal plan, you are required to apply for and obtain a title V operating permit unless you meet the relevant requirements for an exemption specified in §60.2993.

§ 60.3060 When must I submit a title V permit application for my existing unit?

(a)(1) If your existing unit is not subject to an earlier permit application deadline, a complete title V permit application must be submitted on or before the earlier of the dates specified in paragraphs (a)(1)(i) through (iii) of this section. (See sections 129(e), 503(c), 503(d), and 502(a) of the Clean Air Act and 40 CFR 70.5(a)(1)(i) and 40 CFR 71.5(a)(1)(i).)

(i) 12 months after the effective date of any applicable EPA-approved Clean Air Act section 111(d)/129 State or Tribal plan.

(ii) 12 months after the effective date of any applicable Federal plan.

(iii) December 16, 2008.

(2) For any existing unit not subject to an earlier permit application deadline, the application deadline of 36 months after the promulgation of 40 CFR part 60, subpart FFFF, applies regardless of whether or when any applicable Federal plan is effective, or whether or when any applicable Clean Air Act section 111(d)/129 State or Tribal plan is approved by EPA and becomes effective.

(b) If your existing unit is subject to title V as a result of some triggering requirement(s) other than those specified in paragraph (a) of this section (for example, a unit may be a major source or part of a major source), then your unit may be required to apply for a title V permit prior to the deadlines specified in paragraph (a). If more than one requirement triggers a source's obligation to apply for a title V permit, the 12-month timeframe for filing a

title V permit application is triggered by the requirement which first causes the source to be subject to title V. (See section 503(c) of the Clean Air Act and 40 CFR 70.3(a) and (b), 40 CFR 70.5(a)(1)(1), 40 CFR 71.3(a) and (b), and 40 CFR 71.5(a)(1)(i).)

(c) A "complete" title V permit application is one that has been determined or deemed complete by the relevant permitting authority under section 503(d) of the Clean Air Act and 40 CFR 70.5(a)(2) or 40 CFR 71.5(a)(2). You must submit a complete permit application by the relevant application deadline in order to operate after this date in compliance with Federal law. (See sections 503(d) and 502(a) of the Clean Air Act and 40 CFR 70.7(b) and 40 CFR 71.7(b).)

MODEL RULE—TEMPORARY-USE INCINER-ATORS AND AIR CURTAIN INCINERATORS USED IN DISASTER RECOVERY

§60.3061 What are the requirements for temporary-use incinerators and air curtain incinerators used in disaster recovery?

Your incinerator or air curtain incinerator is excluded from the requirements of this subpart if it is used on a temporary basis to combust debris from a disaster or emergency such as a tornado, hurricane, flood, ice storm, high winds, or act of bioterrorism. To qualify for this exclusion, the incinerator or air curtain incinerator must be used to combust debris in an area declared a State of Emergency by a local or State government, or the President, under the authority of the Stafford Act, has declared that an emergency or a major disaster exists in the area, and you must follow the requirements specified in paragraphs (a) through (c) of this section.

(a) If the incinerator or air curtain incinerator is used during a period that begins on the date the unit started operation and lasts 8 weeks or less within the boundaries of the same emergency or disaster declaration area, then it is excluded from the requirements of this subpart. You do not need to notify the Administrator of its use or meet the emission limitations or other requirements of this subpart.

(b) If the incinerator or air curtain incinerator will be used during a period

that begins on the date the unit started operation and lasts more than 8 weeks within the boundaries of the same emergency or disaster declaration area, you must notify the Administrator that the temporary-use incinerator or air curtain incinerator will be used for more than 8 weeks and request permission to continue to operate the unit as specified in paragraphs (b)(1) and (2) of this section.

(1) The notification must be submitted in writing by the date 8 weeks after you start operation of the temporary-use incinerator or air curtain incinerator within the boundaries of the current emergency or disaster declaration area.

(2) The notification must contain the date the incinerator or air curtain incinerator started operation within the boundaries of the current emergency or disaster declaration area. identification of the disaster or emergency for which the incinerator or air curtain incinerator is being used, a description of the types of materials being burned in the incinerator or air curtain incinerator, a brief description of the size and design of the unit (for example, an air curtain incinerator or a modular starved-air incinerator). the reasons the incinerator or air curtain incinerator must be operated for more than 8 weeks, and the amount of time for which you request permission to operate including the date you expect to cease operation of the unit.

(c) If you submitted the notification containing the information in paragraph (b)(2) by the date specified in paragraph (b)(1), you may continue to operate the incinerator or air curtain incinerator for another 8 weeks, which is a total of 16 weeks from the date the unit started operation within the boundaries of the current emergency or disaster declaration area. You do not have to meet the emission limitations or other requirements of this subpart during this period.

(1) At the end of 16 weeks from the date the incinerator or air curtain incinerator started operation within the boundaries of the current emergency or disaster declaration area, you must cease operation of the unit or comply with all requirements of this subpart, unless the Administrator has approved

501

in writing your request to continue operation.

(2) If the Administrator has approved in writing your request to continue operation, then you may continue to operate the incinerator or air curtain incinerator within the boundaries of the current emergency or disaster declaration area until the date specified in the approval, and you do not need to comply with any other requirements of this subpart during the approved time period.

MODEL RULE—AIR CURTAIN INCINER-ATORS THAT BURN ONLY WOOD WASTE, CLEAN LUMBER, AND YARD WASTE

§60.3062 What is an air curtain incinerator?

(a) An air curtain incinerator operates by forcefully projecting a curtain of air across an open, integrated combustion chamber (fire box) or open pit or trench (trench burner) in which combustion occurs. For the purpose of this subpart and subpart EEEE of this part only, air curtain incinerators include both firebox and trench burner units.

(b) Air curtain incinerators that burn only the materials listed in paragraphs (b)(1) through (4) of this section are required to meet only the requirements in \S 60.3062 through 60.3069 and are exempt from all other requirements of this subpart.

(1) 100 percent wood waste.

(2) 100 percent clean lumber.

(3) 100 percent yard waste.

(4) 100 percent mixture of only wood waste, clean lumber, and/or yard waste.

§60.3063 When must I comply if my air curtain incinerator burns only wood waste, clean lumber, and yard waste?

Table 1 of this subpart specifies the final compliance date. You must submit a notification to the Administrator postmarked within 10 business days after the final compliance date in table 1 of this subpart.

§60.3064 What must I do if I close my air curtain incinerator that burns only wood waste, clean lumber, and yard waste and then restart it?

(a) If you close your incinerator but will reopen it prior to the final compli-

40 CFR Ch. I (7–1–23 Edition)

ance date in your State plan, you must meet the final compliance date specified in table 1 of this subpart.

(b) If you close your incinerator but will restart it after your final compliance date, you must meet the emission limitations on the date your incinerator restarts operation.

§60.3065 What must I do if I plan to permanently close my air curtain incinerator that burns only wood waste, clean lumber, and yard waste and not restart it?

You must close the unit before the final compliance date specified in table 1 of this subpart.

§60.3066 What are the emission limitations for air curtain incinerators that burn only wood waste, clean lumber, and yard waste?

(a) Within 180 days after your final compliance date in table 1 of this subpart, you must meet the two limitations specified in paragraphs (a)(1) and (2) of this section.

(1) The opacity limitation is 10 percent (6-minute average), except as described in paragraph (a)(2) of this section.

(2) The opacity limitation is 35 percent (6-minute average) during the startup period that is within the first 30 minutes of operation.

(b) The limitations in paragraph (a) of this section apply at all times except during malfunctions.

§60.3067 How must I monitor opacity for air curtain incinerators that burn only wood waste, clean lumber, and yard waste?

(a) Use Method 9 of appendix A of this part to determine compliance with the opacity limitation.

(b) Conduct an initial test for opacity as specified in §60.8 within 180 days after the final compliance date in table 1 of this subpart.

(c) After the initial test for opacity, conduct annual tests no more than 12 months following the date of your previous test.

(d) If the air curtain incinerator has been out of operation for more than 12 months following the date of your previous test, then you must conduct a test for opacity upon startup of the unit.

§60.3068 What are the recordkeeping and reporting requirements for air curtain incinerators that burn only wood waste, clean lumber, and yard waste?

(a) Keep records of results of all initial and annual opacity tests in either paper copy or computer-readable format that can be printed upon request, unless the Administrator approves another format, for at least 5 years. You must keep each record on site for at least 2 years. You may keep the records off site for the remaining 3 years.

(b) Make all records available for submittal to the Administrator or for an inspector's review.

(c) You must submit the results (each 6-minute average) of the initial opacity tests no later than 60 days following the initial test. Submit annual opacity test results within 12 months following the previous report.

(d) Submit initial and annual opacity test reports as electronic or paper copy on or before the applicable submittal date.

(e) Keep a copy of the initial and annual reports for a period of 5 years. You must keep each report on site for at least 2 years. You may keep the reports off site for the remaining 3 years.

§60.3069 Am I required to apply for and obtain a title V operating permit for my air curtain incinerator that burns only wood waste, clean lumber, and yard waste?

Yes, if your air curtain incinerator is subject to this subpart, you are required to apply for and obtain a title V operating permit as specified in §§ 60.3059 and 60.3060.

MODEL RULE-EQUATIONS

§60.3076 What equations must I use?

(a) *Percent oxygen*. Adjust all pollutant concentrations to 7 percent oxygen using Equation 1 of this section.

$$C_{adj} = C_{meas} * (20.9 - 7)/(20.9 - \%O_2)$$
 (Eq. 1)
Where:

 C_{adj} = pollutant concentration adjusted to 7 percent oxygen

 C_{meas} = pollutant concentration measured on a dry basis

(20.9–7) = 20.9 percent oxygen-7 percent oxygen (defined oxygen correction basis) 20.9 = oxygen concentration in air, percent %O₂ = oxygen concentration measured on a dry basis, percent

(b) Capacity of a very small municipal waste combustion unit. For very small municipal waste combustion units that can operate continuously for 24-hour periods, calculate the unit capacity based on 24 hours of operation at the maximum charge rate. To determine the maximum charge rate, use one of two methods:

(1) For very small municipal waste combustion units with a design based on heat input capacity, calculate the maximum charging rate based on the maximum heat input capacity and one of two heating values:

(i) If your very small municipal waste combustion unit combusts refuse-derived fuel, use a heating value of 12,800 kilojoules per kilogram (5,500 British thermal units per pound).

(ii) If your very small municipal waste combustion unit combusts municipal solid waste, use a heating value of 10,500 kilojoules per kilogram (4,500 British thermal units per pound).

(2) For very small municipal waste combustion units with a design not based on heat input capacity, use the maximum design charging rate.

(c) Capacity of a batch very small municipal waste combustion unit. Calculate the capacity of a batch OSWI unit as the maximum design amount of municipal solid waste it can charge per batch multiplied by the maximum number of batches it can process in 24 hours. Calculate the maximum number of batches by dividing 24 by the number of hours needed to process one batch. Retain fractional batches in the calculation. For example, if one batch requires 16 hours, the OSWI unit can combust 24/16, or 1.5 batches, in 24 hours.

(d) Carbon monoxide pollutant rate. When hourly average pollutant rates (E_h) are obtained (e.g., CEMS values), compute the rolling average carbon monoxide pollutant rate (E_a) for each 12-hour period using the following equation:

$$E_a = \frac{1}{12} \sum_{j=1}^{12} E_{hj}$$
 (Eq. 2)

Where:

- $$\begin{split} \mathbf{E}_a &= \mathrm{Average\ carbon\ monoxide\ pollutant\ rate} \\ & \text{for the\ 12-hour\ period,\ ppm\ corrected\ to} \\ & 7\ percent\ O_2. \end{split}$$
- $E_{hj} = Hourly \mbox{ arithmetic average pollutant} \\ \mbox{ rate for hour ``j,'' ppm corrected to 7 per$ $cent O_2. }$

MODEL RULE—DEFINITIONS

§60.3078 What definitions must I know?

Terms used but not defined in this subpart are defined in the Clean Air Act and subpart A (General Provisions) of this part.

Administrator means:

(1) For approved and effective State section 111(d)/129 plans, the Director of the State air pollution control agency, or his or her delegatee;

(2) For Federal section 111(d)/129 plans, the Administrator of the EPA, an employee of the EPA, the Director of the State air pollution control agency, or employee of the State air pollution control agency to whom the authority has been delegated by the Administrator of the EPA to perform the specified task; and

(3) For NSPS, the Administrator of the EPA, an employee of the EPA, the Director of the State air pollution control agency, or employee of the State air pollution control agency to whom the authority has been delegated by the Administrator of the EPA to perform the specified task.

Air curtain incinerator means an incineration unit operating by forcefully projecting a curtain of air across an open, integrated combustion chamber (fire box) or open pit or trench (trench burner) in which combustion occurs. For the purpose of this subpart and subpart EEEE only, air curtain incinerators include both firebox and trench burner units.

Auxiliary fuel means natural gas, liquified petroleum gas, fuel oil, or diesel fuel.

Batch OSWI unit means an OSWI unit that is designed such that neither waste charging nor ash removal can occur during combustion.

Calendar quarter means three consecutive months (nonoverlapping) beginning on: January 1, April 1, July 1, or October 1. 40 CFR Ch. I (7–1–23 Edition)

Calendar year means 365 consecutive days starting on January 1 and ending on December 31.

Chemotherapeutic waste means waste material resulting from the production or use of anti-neoplastic agents used for the purpose of stopping or reversing the growth of malignant cells.

Class II municipal solid waste landfill means a landfill that meets four criteria:

(1) Accepts, for incineration or disposal, less than 20 tons per day of municipal solid waste or other solid wastes based on an annual average;

(2) Is located on a site where there is no evidence of groundwater pollution caused or contributed to by the landfill;

(3) Is not connected by road to a Class I municipal solid waste landfill, as defined by Alaska regulatory code 18 AAC 60.300(c) or, if connected by road, is located more than 50 miles from a Class I municipal solid waste landfill; and

(4) Serves a community that meets one of two criteria:

(i) Experiences for at least three months each year, an interruption in access to surface transportation, preventing access to a Class I municipal solid waste landfill; or

(ii) Has no practicable waste management alternative, with a landfill located in an area that annually receives 25 inches or less of precipitation.

Class III municipal solid waste landfill is a landfill that is not connected by road to a Class I municipal solid waste landfill, as defined by Alaska regulatory code 18 AAC 60.300(c) or, if connected by road, is located more than 50 miles from a Class I municipal solid waste landfill, and that accepts, for disposal, either of the following two criteria:

(1) Ash from incinerated municipal waste in quantities less than one ton per day on an annual average, which ash must be free of food scraps that might attract animals; or

(2) Less than five tons per day of municipal solid waste, based on an annual average, and is not located in a place that meets either of the following criteria:

(i) Where public access is restricted, including restrictions on the right to move to the place and reside there; or

(ii) That is provided by an employer and that is populated totally by persons who are required to reside there as a condition of employment and who do not consider the place to be their permanent residence.

Clean lumber means wood or wood products that have been cut or shaped and include wet, air-dried, and kilndried wood products. Clean lumber does not include wood products that have been painted, pigment-stained, or pressure-treated by compounds such as chromate copper arsenate, pentachlorophenol, and creosote, or manufactured wood products that contain adhesives or resins (e.g., plywood, particle board, flake board, and oriented strand board).

Collected from means the transfer of material from the site at which the material is generated to a separate site where the material is burned.

Contained gaseous material means gases that are in a container when that container is combusted.

Continuous emission monitoring system or CEMS means a monitoring system for continuously measuring and recording the emissions of a pollutant from an OSWI unit.

Continuous OSWI unit means an OSWI unit that is designed to allow waste charging and ash removal during combustion.

Deviation means any instance in which a unit that meets the requirements in §60.2991, or an owner or operator of such a source:

(1) Fails to meet any requirement or obligation established by this subpart, including but not limited to any emission limitation, operating limit, or operator qualification and accessibility requirements;

(2) Fails to meet any term or condition that is adopted to implement an applicable requirement in this subpart and that is included in the operating permit for any unit that meets requirements in §60.2991 and is required to obtain such a permit; or

(3) Fails to meet any emission limitation, operating limit, or operator qualification and accessibility requirement in this subpart during startup, shutdown, or malfunction, regardless of whether or not such failure is allowed by this subpart.

Dioxins/furans means tetra-through octachlorinated dibenzo-p-dioxins and dibenzofurans.

Energy recovery means the process of recovering thermal energy from combustion for useful purposes such as steam generation or process heating.

EPA means the Administrator of the EPA or employee of the EPA that is delegated the authority to perform the specified task.

Institutional facility means a landbased facility owned and/or operated by an organization having a governmental, educational, civic, or religious purpose such as a school, hospital, prison, military installation, church, or other similar establishment or facility.

Institutional waste means solid waste (as defined in this subpart) that is combusted at any institutional facility using controlled flame combustion in an enclosed, distinct operating unit: Whose design does not provide for energy recovery (as defined in this subpart); operated without energy recovery (as defined in this subpart); or operated with only waste heat recovery (as defined in this subpart). Institutional waste also means solid waste (as defined in this subpart) combusted on site in an air curtain incinerator that is a distinct operating unit of any institutional facility.

Institutional waste incineration unit means any combustion unit that combusts institutional waste (as defined in this subpart) and is a distinct operating unit of the institutional facility that generated the waste. Institutional waste incineration units include fielderected, modular, cyclonic burn barrel, and custom built incineration units operating with starved or excess air, and any air curtain incinerator that is a distinct operating unit of the institutional facility that generated the institutional waste (except those air curtain incinerators listed in §60.2994(b)).

Intermittent OSWI unit means an OSWI unit that is designed to allow waste charging, but not ash removal, during combustion.

Low-level radioactive waste means waste material that contains radioactive nuclides emitting primarily beta or gamma radiation, or both, in concentrations or quantities that exceed applicable Federal or State standards for unrestricted release. Low-level radioactive waste is not high-level radioactive waste, spent nuclear fuel, or byproduct material as defined by the Atomic Energy Act of 1954 (42 U.S.C. 2014(e)(2)).

Malfunction means any sudden, infrequent, and not reasonably preventable failure of air pollution control equipment, process equipment, or a process to operate in a normal or usual manner. Failures that are caused, in part, by poor maintenance or careless operation are not malfunctions.

Metropolitan Statistical Area means any areas listed as metropolitan statistical areas in OMB Bulletin No. 05-02 entitled "Update of Statistical Area Definitions and Guidance on Their Uses" dated February 22, 2005 (available on the Web at http:// www.whitehouse.gov/omb/bulletins/).

Modification or modified unit means an incineration unit you have changed on or after June 16, 2006 and that meets one of two criteria:

(1) The cumulative cost of the changes over the life of the unit exceeds 50 percent of the original cost of building and installing the unit (not including the cost of land) updated to current costs (current dollars). For an OSWI unit, to determine what systems are within the boundary of the unit used to calculate these costs, see the definition of OSWI unit.

(2) Any physical change in the OSWI unit or change in the method of operating it that increases the amount of any air pollutant emitted for which section 129 or section 111 of the Clean Air Act has established standards.

Municipal solid waste means refuse (and refuse-derived fuel) collected from the general public and from residential, commercial, institutional, and industrial sources consisting of paper, wood, yard wastes, food wastes, plastics, leather, rubber, and other combustible materials and non-combustible materials such as metal, glass and rock, provided that: (1) The term does not include industrial process wastes or medical wastes that are segregated from such other wastes; and (2) an incineration unit shall not be considered 40 CFR Ch. I (7-1-23 Edition)

to be combusting municipal solid waste for purposes of this subpart if it combusts a fuel feed stream, 30 percent or less of the weight of which is comprised, in aggregate, of municipal solid waste, as determined by §60.2993(b).

Municipal waste combustion unit means, for the purpose of this subpart and subpart EEEE, any setting or equipment that combusts municipal solid waste (as defined in this subpart) including, but not limited to, fielderected, modular, cyclonic burn barrel, and custom built incineration units (with or without energy recovery) operating with starved or excess air, boilers, furnaces, pyrolysis/combustion units, and air curtain incinerators (except those air curtain incinerators listed in §60.2994(b)).

Other solid waste incineration (OSWI) *unit* means either a very small municipal waste combustion unit or an institutional waste incineration unit, as defined in this subpart. Unit types listed in §60.2993 as being excluded from the subpart are not OSWI units subject to this subpart. While not all OSWI units will include all of the following components, an OSWI unit includes, but is not limited to, the municipal or institutional solid waste feed system, grate system, flue gas system, waste heat recovery equipment, if any, and bottom ash system. The OSWI unit does not include air pollution control equipment or the stack. The OSWI unit boundary starts at the municipal or institutional waste hopper (if applicable) and extends through two areas:

(1) The combustion unit flue gas system, which ends immediately after the last combustion chamber or after the waste heat recovery equipment, if any; and

(2) The combustion unit bottom ash system, which ends at the truck loading station or similar equipment that transfers the ash to final disposal. The OSWI unit includes all ash handling systems connected to the bottom ash handling system.

Particulate matter means total particulate matter emitted from OSWI units as measured by Method 5 or Method 29 of appendix A of this part.

Pathological waste means waste material consisting of only human or animal remains, anatomical parts, and/or

tissue, the bags/containers used to collect and transport the waste material, and animal bedding (if applicable).

Reconstruction means rebuilding an incineration unit and meeting two criteria:

(1) The reconstruction begins on or after June 16, 2006.

(2) The cumulative cost of the construction over the life of the incineration unit exceeds 50 percent of the original cost of building and installing the unit (not including land) updated to current costs (current dollars). For an OSWI unit, to determine what systems are within the boundary of the unit used to calculate these costs, see the definition of OSWI unit.

Refuse-derived fuel means a type of municipal solid waste produced by processing municipal solid waste through shredding and size classification. This includes all classes of refusederived fuel including two fuels:

(1) Low-density fluff refuse-derived fuel through densified refuse-derived fuel.

(2) Pelletized refuse-derived fuel.

Shutdown means the period of time after all waste has been combusted in the primary chamber. For continuous OSWI, shutdown shall commence no less than 2 hours after the last charge to the incinerator. For intermittent OSWI, shutdown shall commence no less than 4 hours after the last charge to the incinerator. For batch OSWI, shutdown shall commence no less than 5 hours after the high-air phase of combustion has been completed.

Solid waste means any garbage, refuse, sludge from a waste treatment plant, water supply treatment plant, or air pollution control facility and other discarded material, including solid, liquid. semisolid, or contained gaseous material resulting from industrial, commercial, mining, agricultural operations, and from community activities, but does not include solid or dissolved material in domestic sewage, or solid or dissolved materials in irrigation return flows or industrial discharges that are point sources subject to permits under section 402 of the Federal Water Pollution Control Act, as amended (33 U.S.C. 1342), or source, special nuclear, or byproduct material as defined by the

Atomic Energy Act of 1954, as amended (42 U.S.C. 2014).

Standard conditions, when referring to units of measure, means a temperature of 68 $^{\circ}$ F (20 $^{\circ}$ C) and a pressure of 1 atmosphere (101.3 kilopascals).

Startup period means the period of time between the activation of the system and the first charge to the OSWI unit. For batch OSWI, startup means the period of time between activation of the system and ignition of the waste.

Very small municipal waste combustion unit means any municipal waste combustion unit that has the capacity to combust less than 35 tons per day of municipal solid waste or refuse-derived fuel, as determined by the calculations in §60.3076.

Waste heat recovery means the process of recovering heat from the combustion flue gases outside of the combustion firebox by convective heat transfer only.

Wet scrubber means an add-on air pollution control device that utilizes an aqueous or alkaline scrubbing liquor to collect particulate matter (including nonvaporous metals and condensed organics) and/or to absorb and neutralize acid gases.

Wood waste means untreated wood and untreated wood products, including tree stumps (whole or chipped), trees, tree limbs (whole or chipped), bark, sawdust, chips, scraps, slabs, millings, and shavings. Wood waste does not include:

(1) Grass, grass clippings, bushes, shrubs, and clippings from bushes and shrubs from residential, commercial/retail, institutional, or industrial sources as part of maintaining yards or other private or public lands.

(2) Construction, renovation, or demolition wastes.

(3) Clean lumber.

(4) Treated wood and treated wood products, including wood products that have been painted, pigment-stained, or pressure treated by compounds such as chromate copper arsenate, pentachlorophenol, and creosote, or manufactured wood products that contain adhesives or resins (*e.g.*, plywood, particle board, flake board, and oriented strand board).

Pt. 60, Subpt. FFFF, Table 1

40 CFR Ch. I (7-1-23 Edition)

Yard waste means grass, grass clippings, bushes, shrubs, and clippings from bushes and shrubs. Yard waste comes from residential, commercial/retail, institutional, or industrial sources as part of maintaining yards or other

private or public lands. Yard waste does not include two items:

(1) Construction, renovation, and demolition wastes.

(2) Clean lumber.

TABLE 1 TO SUBPART FFFF OF PART 60-MODEL RULE-COMPLIANCE SCHEDULE

As stated in §60.3000, you must comply with the following:

Complete this action	□y this date ^a	
□inal compliance ^b	(□ates to be specified in State plan) ^c .	

 $^{\rm a}\,\textsc{Site-specific schedules}$ can be used at the discretion of the State.

^b □inal compliance means that you complete all process changes and retrofit of control devices so that, when the incin-eration unit is brought on line, all process changes and air pollution control devices necessary to meet the emission limi-tations operate as designed. ° □he date can be no later than 3 years after the effective date of State plan approval or □ecember 16, 2010, whichever is earlier

is earlier.

TABLE 2 TO SUBPART FFFF OF PART 60-MODEL RULE-EMISSION LIMITATIONS

As stated in §60.3022, you must comply with the following:

□or the air pollutant	You must meet this emission limitation ^a	□sing this averaging time	And determining compliance using this method
1. Cadmium	1 □ micrograms per dry standard cubic meter.	3-run average (1 hour minimum sample time per run).	Method 29 of appendix A of this part.
2. Carbon monoxide	□0 parts per million by dry volume.	3-run average (1 hour minimum sample time per run during performance test), and 12-hour rolling averages meas- ured using CEMS ^b .	Method 10, 10A, or 10⊡ of appendix A of this part and CEMS.
 ∃ioxins furans (total basis). 	33 nanograms per dry standard cubic meter.	3-run average (1 hour minimum sample time per run).	Method 23 of appendix A of this part.
□ □ydrogen chloride	1□ parts per million by dry volume.	3-run average (1 hour minimum sample time per run).	Method 26A of appendix A of this part.
□ □ead	226 micrograms per dry standard cubic meter.	3-run average (1 hour minimum sample time per run).	Method 29 of appendix A of this part.
6. Mercury	7 micrograms per dry standard cubic meter.	3-run average (1 hour minimum sample time per run).	Method 29 of appendix A of this part.
7. Opacity	10 percent	6-minute average (observe over three 1-hour test runs□i.e., thirty 6-minute averages).	Method 9 of appendix A of this part.
□ Oxides of nitrogen	103 parts per mil- lion by dry vol- ume.	3-run average (1 hour minimum sample time per run).	Method 7, 7A, 7C, 7□, or 7E of appen- dix A of this part, or ANSTASME □C 19.10□19□1 (IIII), see §60.17(h)) in lieu of Methods 7 and 7C only.
9. □articulate matter	0.013 grains per dry standard cubic foot.	3-run average (1 hour minimum sample time per run).	Method or 29 of appendix A of this part.
10. Sulfur dioxide	3.1 parts per million by dry volume.	3-run average (1 hour minimum sample time per run).	Method 6 or 6C of appendix A of this part, or ANSTASME CC 19.10_19_1 (TC, see §60.17(h)) in lieu of Meth- od 6 only.

^a All emission limitations (except for opacity) are measured at 7 percent oxygen, dry basis at standard conditions. ^b Calculated each hour as the average of the previous 12 operating hours

TABLE 3 TO SUBPART FFFF OF PART 60-MODEL RULE-OPERATING LIMITS FOR INCINERATORS AND WET SCRUBBERS

As stated in §60.3023, you must comply with the following:

Pt. 60, Subpt. FFFF, Table 5

□or these operating pa-	You must establish op-	And monitoring using these minimum frequencies			
rameters	erating limits	□ata measurement	□ata recording	Averaging time	
1. Charge rate	Maximum charge rate	Continuous	Every hour	□aily for batch units. 3- hour rolling for con- tinuous and intermit- tent units. ^a	
 Pressure drop across the wet scrub- ber or amperage to wet scrubber. 	Minimum pressure drop or amperage.	Continuous	Every 1 minutes	3-hour rolling. ^a	
 Scrubber liquor flow rate. 	Minimum flow rate	Continuous	Every 1 minutes	3-hour rolling. ^a	
□. Scrubber liquor p□	Minimum p	Continuous	Every 1 minutes	3-hour rolling. a	

^aCalculated each hour as the average of the previous 3 operating hours.

TABLE 4 TO SUBPART FFFF OF PART 60—MODEL RULE—REQUIREMENTS FOR CONTINUOUS EMISSION MONITORING SYSTEMS (CEMS)

As stated in §60.3039, you must comply with the following:

□or the following pollutants	□se the following span values for your CEMS	□se the following perform- ance specifications (□.S.) in appendix □ of this part for your CEMS	If needed to meet minimum data requirements, use the following alternate methods in appendix A of this part to col- lect data
1. Carbon Monoxide	12□ percent of the maximum hourly potential carbon monoxide emissions of the waste combustion unit.		Method 10.
2. Oxygen	2 percent oxygen	D.S.3	Method 3A or 3□, or ANS ASME □□C 19.10□19□1 (□□□, see § 60.17(h)) in lieu of Method 3□ only.

TABLE 5 TO SUBPART FFFF OF PART 60—MODEL RULE—SUMMARY OF REPORTING REQUIREMENTS

As stated in §60.3048, you must comply with the following:

□eport	□ue date	Contents	□eference
1. Initial test report	a. No later than 60 days fol- lowing the initial perform- ance test	i. Complete test report for the initial performance test⊡and.	§ 60.30 ⊡ 9.
		ii. □he values for the site-spe- cific operating limits.	§ 60.30 □ 9.
2. □ aste management plan	a. No later than 60 days fol- lowing the initial perform- ance test	ieduction or separation of recyclable materials_and.	§§ 60.3010 through 60.3012.
		ii. Identification of additional waste management meas- ures and how they will be implemented.	§§ 60.3010 through 60.3012.
3. Annual ⊡eport	a. No later than 12 months following the submission of the initial test report. Sub- sequent reports are to be submitted no more than 12 months following the pre- vious report	i. Company Name and ad- dress⊡	§§ 60.30⊡0 and 60.30⊡1.
		ii. Statement and signature by the owner or operator	§§60.30⊡0 and 60.30⊡1.
		iii. ate of report	§§ 60.30 ⊡0 and 60.30 ⊡1.
		limits	38 00.30 L0 and 00.30 L1.
		 v. If no deviations or malfunc- tions were reported, a statement that no devi- ations occurred during the reporting period. 	§§ 60.30⊡0 and 60.30⊡1.

Pt. 60, Subpt. FFFF, Table 5

40 CFR Ch. I (7-1-23 Edition)

□eport	□ue date Contents		□eference
		vi. □ighest and lowest re- corded 12-hour averages, as applicable, for carbon monoxide emissions and highest and lowest re- corded 3-hour averages, as applicable, for each oper- ating parameter recorded for the calendar year being reported□	§§60.30⊡0 and 60.30⊡1.
		vii. Thermatic for deviations or malfunctions recorded under §60.29 (b)(6) and	§§ 60.30⊡0 and 60.30⊡1.
		viii. f a performance test was conducted during the re- porting period, the results	§§ 60.30⊡0 and 60.30⊡1.
		ix. If a performance test was not conducted during the reporting period, a state- ment that the requirements of § 60.293 (a) or (b) were met[and]	§§ 60.30⊐0 and 60.30⊐1.
		 x. Cocumentation of periods when all qualified OS unit operators were unavail- able for more than 12 hours but less than 2 wee s. 	§§ 60.30⊡0 and 60.30⊡1.
 Emission limitation or oper- ating limit deviation report. 	a. □y August 1 of that year for data collected during the first half of the calendar year. □y □ebruary 1 of the following year for data col- lected during the second half of the calendar year	i. ⊡ates and times of devi- ation⊡	§§ 60.30⊡2 and 60.30⊡3.
		ii. Averaged and recorded	§§60.30⊡2 and 60.30⊡3.
		iii. Uration and causes of each deviation and the cor- rective actions ta en	§§ 60.30⊡2 and 60.30⊡3.
		iv. Copy of operating limit monitoring data and any test reports	§§ 60.30⊡2 and 60.30⊡3.
		v. □ates, times, and causes for monitor downtime inci- dents□	§§ 60.30⊡2 and 60.30⊡3.
		 vi. □ hether each deviation occurred during a period of startup, shutdown, or mal- function□and. 	§§ 60.30⊡2 and 60.30⊡3.
		vii. □ates, times, and duration of any bypass of the control device.	§§ 60.30⊡2 and 60.30⊡3.
 ualified operator deviation notification. 	a. ithin 10 days of deviation	i. Statement of cause of devi- ation	§60.30□□(a)(1).
		ii. □escription of efforts to have an accessible quali- fied operator□and.	§ 60.30⊒(a)(1).
		iii. □he date a qualified oper- ator will be accessible.	§60.30□⊒(a)(1).
 □ualified operation deviation status report. 	a. Every □ wee⊡s following deviation	 i. □escription of efforts to have an accessible quali- fied operator □ 	§60.30⊒(a)(2).
		ii. □he date a qualified oper- ator will be accessible □and.	§60.30□□(a)(2).
		iii. □equest to continue oper- ation.	§60.30□□(a)(2).
 □ ualified operator deviation notification of resumed oper- ation. 	a. □rior to resuming operation	i. Notification that you are re- suming operation.	§60.30⊐⊒(b).

Note: \Box his table is only a summary, see the referenced sections of the rule for the complete requirements.

510

[70 FR 74907, Dec. 16, 2005, as amended at 71 FR 67806, Nov. 24, 2006]

Subparts GGGG-HHHH [Reserved]

Subpart IIII—Standards of Performance for Stationary Compression Ignition Internal Combustion Engines

Source: 71 FR 39172, July 11, 2006, unless otherwise noted.

WHAT THIS SUBPART COVERS

§ 60.4200 Am I subject to this subpart?

(a) The provisions of this subpart are applicable to manufacturers, owners, and operators of stationary compression ignition (CI) internal combustion engines (ICE) and other persons as specified in paragraphs (a)(1) through (4) of this section. For the purposes of this subpart, the date that construction commences is the date the engine is ordered by the owner or operator.

(1) Manufacturers of stationary CI ICE with a displacement of less than 30 liters per cylinder where the model year is:

(i) 2007 or later, for engines that are not fire pump engines;

(ii) The model year listed in Table 3 to this subpart or later model year, for fire pump engines.

(2) Owners and operators of stationary CI ICE that commence construction after July 11, 2005, where the stationary CI ICE are:

(i) Manufactured after April 1, 2006, and are not fire pump engines, or

(ii) Manufactured as a certified National Fire Protection Association (NFPA) fire pump engine after July 1, 2006.

(3) Owners and operators of any stationary CI ICE that are modified or reconstructed after July 11, 2005 and any person that modifies or reconstructs any stationary CI ICE after July 11, 2005.

(4) The provisions of §60.4208 of this subpart are applicable to all owners and operators of stationary CI ICE that commence construction after July 11, 2005.

(b) The provisions of this subpart are not applicable to stationary CI ICE

being tested at a stationary CI ICE test cell/stand.

(c) If you are an owner or operator of an area source subject to this subpart, you are exempt from the obligation to obtain a permit under 40 CFR part 70 or 40 CFR part 71, provided you are not required to obtain a permit under 40 CFR 70.3(a) or 40 CFR 71.3(a) for a reason other than your status as an area source under this subpart. Notwithstanding the previous sentence, you must continue to comply with the provisions of this subpart applicable to area sources.

(d) Stationary CI ICE may be eligible for exemption from the requirements of this subpart as described in 40 CFR part 1068, subpart C, except that owners and operators, as well as manufacturers, may be eligible to request an exemption for national security.

(e) Owners and operators of facilities with CI ICE that are acting as temporary replacement units and that are located at a stationary source for less than 1 year and that have been properly certified as meeting the standards that would be applicable to such engine under the appropriate nonroad engine provisions, are not required to meet any other provisions under this subpart with regard to such engines.

[71 FR 39172, July 11, 2006, as amended at 76 FR 37967, June 28, 2011; 86 FR 34357, June 29, 2021]

EMISSION STANDARDS FOR MANUFACTURERS

§ 60.4201 What emission standards must I meet for non-emergency engines if I am a stationary CI internal combustion engine manufacturer?

(a) Stationary CI internal combustion engine manufacturers must certify their 2007 model year and later nonemergency stationary CI ICE with a maximum engine power less than or equal to 2,237 kilowatt (KW) (3,000 horsepower (HP)) and a displacement of less than 10 liters per cylinder to the certification emission standards for new nonroad CI engines in 40 CFR 1039.101, 1039.102, 1039.104, 1039.105, 1039.107, and 1039.115 and 40 CFR part Appendix N

NOTIFICATION OF COMPLIANCE

Revision 0

Pursuant to

NESHAP: Standards for Hazardous Air Pollutants for Hazardous Waste Combustors (40 CFR 63 Subpart EEE, Phase II)

Prepared by

FUTUREFUEL CHEMICAL COMPANY P.O. Box 2357 Batesville, AR 72503 ARD089234884

and

RISK MANAGEMENT AND ENGINEERING, LTD. 9206 Briarcrest, Ste. 102 Rowlett, TX 75089

September 01, 2010

EXECUTIVE SUMMARY

FutureFuel Chemical Company (FFCC) owns and operates an organic chemical manufacturing plant located southeast of Batesville, Arkansas, that was formerly owned and operated by Eastman Chemical Company. As part of plant operations, FFCC (EPA 1D# ARD089234884) generates wastes regulated under the Resource Conservation and Recovery Act (RCRA) and burns some of this waste in its three coal-fired boilers for energy recovery. FFCC is currently operating these boilers under Arkansas Department of Environmental Quality (ADEQ) Hazardous Waste Permit Number 11H-RN1-M005. FFCC also destroys waste in its on-site incinerator currently operating under its Title V Permit (1085-AOP-R8).

The documentation of compliance (DOC) was developed in order to certify that the boilers are designed and will be operated in a manner that ensures compliance with the emission standards of 40 CFR 63.1216. This documentation of compliance was prepared in accordance with 40 CFR 63.1211(d) and was placed in the operating record on October 14, 2008. FFCC placed the DOC in its facility operating record as specified in the NESHAP: Final Standards for Hazardous Air Pollutants for Waste Combustors; Final Rule, effective October 12, 2005.

FFCC submitted the boiler Comprehensive Performance Test (CPT) plan for approval on April 11, 2008. The plan was conditionally approved by EPA Region 6 on April 13, 2010. FFCC began the CPT on April 14, 2010 and concluded the testing on June 4, 2010. The results of this test are being submitted as the Notice of Compliance (NOC) to the EPA and ADEQ. FFCC will begin complying with the operational parameters and limits established within this NOC immediately upon submittal.

CONCLUSIONS ON MEETING CPT OBJECTIVES

The overall objectives of the Comprehensive Performance Test were to demonstrate that the boilers are capable of meeting the solid fuel-fired emission standards established in the MACT Combustion Rule (40 CFR 63 Subpart EEE) and to establish operating parameter limits that are equivalent with that demonstration. Table ES-1 summarizes the specific objectives demonstrated during the CPT.

FFCC was successful in meeting the overall CPT objective, and this NOC establishes the operating conditions that reflect the results of this successful test.
NOTIFICATION OF COMPLIANCE

Revision 0

Pursuant to

NESHAP: Standards for Hazardous Air Pollutants for Hazardous Waste Combustors (40 CFR 63 Subpart EEE, Phase II)

Prepared by

FUTUREFUEL CHEMICAL COMPANY P.O. Box 2357 Batesville, AR 72503 ARD089234884

and

RISK MANAGEMENT AND ENGINEERING, LTD. 9206 Briarcrest, Ste. 102 Rowlett, TX 75089

September 01, 2010

EXECUTIVE SUMMARY

FutureFuel Chemical Company (FFCC) owns and operates an organic chemical manufacturing plant located southeast of Batesville, Arkansas, that was formerly owned and operated by Eastman Chemical Company. As part of plant operations, FFCC (EPA 1D# ARD089234884) generates wastes regulated under the Resource Conservation and Recovery Act (RCRA) and burns some of this waste in its three coal-fired boilers for energy recovery. FFCC is currently operating these boilers under Arkansas Department of Environmental Quality (ADEQ) Hazardous Waste Permit Number 11H-RN1-M005. FFCC also destroys waste in its on-site incinerator currently operating under its Title V Permit (1085-AOP-R8).

The documentation of compliance (DOC) was developed in order to certify that the boilers are designed and will be operated in a manner that ensures compliance with the emission standards of 40 CFR 63.1216. This documentation of compliance was prepared in accordance with 40 CFR 63.1211(d) and was placed in the operating record on October 14, 2008. FFCC placed the DOC in its facility operating record as specified in the NESHAP: Final Standards for Hazardous Air Pollutants for Waste Combustors; Final Rule, effective October 12, 2005.

FFCC submitted the boiler Comprehensive Performance Test (CPT) plan for approval on April 11, 2008. The plan was conditionally approved by EPA Region 6 on April 13, 2010. FFCC began the CPT on April 14, 2010 and concluded the testing on June 4, 2010. The results of this test are being submitted as the Notice of Compliance (NOC) to the EPA and ADEQ. FFCC will begin complying with the operational parameters and limits established within this NOC immediately upon submittal.

CONCLUSIONS ON MEETING CPT OBJECTIVES

The overall objectives of the Comprehensive Performance Test were to demonstrate that the boilers are capable of meeting the solid fuel-fired emission standards established in the MACT Combustion Rule (40 CFR 63 Subpart EEE) and to establish operating parameter limits that are equivalent with that demonstration. Table ES-1 summarizes the specific objectives demonstrated during the CPT.

FFCC was successful in meeting the overall CPT objective, and this NOC establishes the operating conditions that reflect the results of this successful test.

Boiler Notice of ComplianceRevision No.:0Revision:09/01/10Pageii of six

Table ES-1

Specific Objective	CPT Result
Demonstrate 99.99 % DRE of the designated POHC Chlorobenzene	≥ 99.9991 %
Demonstrate control of hydrocarbon emissions to less than 10 parts per million dry volume (ppmv), corrected 7 percent oxygen, on a hourly rolling average.	≤0.4 ppmv (a 7% O ₂
Conduct a one time emission test for Dioxin/Furan or submit adequate data in lieu of testing, and Demonstrate control of hydrocarbon emissions to less than 10 parts per million dry volume (ppmv), corrected 7 percent oxygen, on a hourly rolling average (as shown above)	Agency approved data in lieu of D/F test located in Appendix F of the CPT Plan. One time D/F test results were 0.092 ng TEQ/dscm (å: 7% O ₂
Demonstrate control of particulate emissions to less than 68 micrograms per dry standard cubic meter (ug/dscm) at to 7 percent oxygen.	26.2 ug/dscm (# 7% O ₂
Demonstrate that control of HCl and free chlorine emissions are equal to or less than limit established using the health-based alternative compliance demonstration described in Appendix D of the CPT Plan (< 1,886.8 lb/hr chloride feed rate)	494 lb/hr
Demonstrate that control of mercury emissions are equal to or less than 11 micrograms per dry standard cubic meter (ug/dsem) corrected to 7 percent oxygen. During Test 2.	3.9 ug/dsem (a 7% O 2
Demonstrate that control of Semi-Volatile metals (SVM) emissions are equal to or less than 180 micrograms per dry standard cubic meter (ug/dscm) corrected to 7 percent oxygen during Test 2.	158.8 ug/dsem (a 7% O ₂
Demonstrate that control of Low-Volatile metals (LVM) emissions are equal to or less than 380 micrograms per dry standard cubic meter (ug/dscm) corrected to 7 percent oxygen during Test 2.	117.3 ug/dsem (a) 7% O2
Gather data regarding waste feed characteristics and process operating conditions that will be used to develop operational permit limits that will ensure compliance with regulatory performance standards.	FFCC obtained all the necessary information to establish permit limits and demonstrate performance standards during the CPT.

Boiler Notice of ComplianceRevision No.:0Revision:09/01/10Pageiii of xix

DEVIATIONS FROM THE CPT PLAN

There were several operational related deviations from the approved CPT plan that were made prior to conducting the CPT. Table ES-2 details the changes made after formal EPA approval of the CPT plan.

Change to the CPT Plan	Explanation of Change
Coal Sampling Location	The original plan called for FFCC to sample the coal and diatomaceous earth as fed; however, the Agency directed us to modily the original sampling plan to obtain the coal samples prior to the addition of the diatomaceous earth. The coal samples were then taken from a sample point at the exit of the coal silo which is approximately 8 feet before the area where the ash (diatomaceous earth) was spiked into the coal.
VOST Sampling Rate	The original plan called for FFCC to obtain 4 VOST tube pairs over 120 minute sampling period. The Agency directed us to obtain samples at a rate of 0.5 liters per minute, which increased the sampling time to 160 minutes.
Chloride Analysis	The CPT plan called for the commercial lab to use SW-846 Method 9076 or other approved method. The Laboratory used SW-846 Method 9057, which is an EPA approved method for measuring chloride. The Independent QA Reviewer examined this issue in detail (see Section 4) and determined that the use of Method 9057 rather than 9076 did not compromise the data.
Maximum Hazardous Waste Feed Rate Limit, Maximum Combustion Air Flow Rate Limit, and Maximum Percent Stack Gas Oxygen Limit	FFCC will set the maximum hazardous waste feed rate, the maximum combustion air flow rate limit, and the maximum percent stack gas oxygen limit based on the average of the test run averages, instead the average of the maximum hourly rolling averages as specified in the CPT plan. This is a more conservative approach.

Table ES-2

Boiler Notice of ComplianceRevision No.:0Revision:09/01/10Pageiv of xix

Change to the CPT Plan	Explanation of Change
Maximum Mercury Feed Rate Limit	FFCC plans to use the Maximum Theoretical Emission Calculation (MTEC) to establish the maximum mercury feed rate limit. FFCC had planned to use the average of the test run averages to establish a maximum mercury feed rate limit: however the mercury present in the coal was uncharacteristically low. As seen in Table 2.0 of the CPT plan, the average mercury in FFCC's coal is 0.15 ppm. The mercury in the coal used during the CPT was 0.044 ppm. The MTEC assumes zero percent removal efficiency for mercury. To further alleviate concerns, FFCC will set the feed rate limit at just 90% of the MTEC Calculated limit to ensure it stays well below the MACT EEE standard for solid fuel-fired boilers.
Laboratory Analytical Methods	The CPT plan stated that certain analytical methods or other approved methods would be used by the laboratory. The laboratory used EPA approved scientific methods for all analytical results, but some of those methods were not specifically mentioned in the plan. The methods used, that were not specifically listed, were evaluated by the independent third party QA/QC officer and determined to be approved and viable EPA methods. This evaluation is discussed in section 4.0 of the NOC.

Table ES-2, continued

Boiler Notice of ComplianceRevision No.:0Revision:09/01/10Pagev of xix

CPT ISSUES AND SOLUTIONS

The boilers met all CPT objectives; however, the following issues were encountered during the CPT:

- 1) THC CEMS Malfunction
- 2) VOST Audit Process
- 3) Coal Feed Rate

Issue 1 – THC CEMS Malfunction

We began stabilizing the boiler for Test 2, Run 1 at 05:00 am. At 06:11 am we encountered a mechanical malfunction on the THC CEMS which initiated an AWFCO. The control system mechanic for the CEMS did not arrive at the plant site until 08:00 am. He repaired the malfunction and we were back on line by 09:00 am stabilizing again. Test 2, Run 1 began at 10:04 am instead of the planned 07:00 am start.

Issue 2 - VOST Audit Process

The Agency did not ship the VOST audit materials with fittings that would work with the stack sampling crew's equipment, and did not provide a means to measure the temperature of the audit stack. FFCC initiated an extensive search for fittings at its plant site, but none were found that would provide leak free flow from the audit cylinder. In the end, the stack sampling crew was forced to break one of its sampling probes so that the VOST audit process could be completed.

Issue 3 – Coal Feed Rate

During Test 2, Run 3, the process operators failed to select the correct analytical data for waste tank WB01. FFCC was hurning the test waste from WB01 but the computer was using analytical data entered for tank WB02. This did not impact the test results but it did cause the coal feed rate to read artificially low because the heating value for the waste in tank WB02 was higher than that of the test waste. They discovered this error at the end of Run 3 when they were switching the system back to burn from waste tank WB02. We informed agency oversight personnel of the mistake and demonstrated that that we were indeed burning from the CPT waste tank (WB01), during the test, by showing him the valve positions in the plant information system.

As shown in Section 1.2.1.1, the coal feed rate is not a direct measurement, but it is calculated indirectly based on various analytical and process data. It uses the heating value of the waste, the heating value of the coal, the feed rate of the waste, the steam production rate, various

Prepared by FutureFuel Chemical Company, Batesville, Arkansas, and Risk Management and Engineering, Ltd., Dallas, Texas

Boiler Notice of	Compliance
Revision No.:	0
Revision:	09/01/10
Page	vi of xix

temperature measurements, and the boiler efficiency to calculate the coal feed rate. Because the analytical for tank WB02 was selected, the computer used an artificially high heating value for the waste, which caused it to calculate an incorrect low coal feed rate. This explains the difference in the coal feed rate that was displayed on the computer screen between run 3 (Friday, June 4th) and runs 1 and 2 (Thursday, June 3rd).

In order to get a true and accurate coal feed rate, FFCC recalculated the coal feed rate based on the true Btu analysis of botb the coal and the waste obtained from the sampling completed during the CPT. Appendix E-6 provides the process data for the coal feed rate correction that was used along with the correct Btu to obtain the true and accurate coal feed rate during the CPT.

Boiler Notice of ComplianceRevision No.:0Revision:09/01/10Pagevii of xix

PERMIT RELATED OPERATIONAL DATA

Table ES-3 provides a summary of the permit related operational data derived from the CPT.

Table ES-3

Parameter	Basis	Result
Maximum total hazardous waste feed (lb/hr-12HRA)	Average of the test run averages recorded during each run of Test 2	2,518 lb/hr
Maximum Chloride Feed Rate (lb/hr-12HRA)	Average of the test run average feed rates recorded during each run of Test 2	494 lb/hr
Maximum Ash Feed Rate (lb/hr-12HRA)	Average of the test run average feed rates recorded during each run of Test 2	791 lb/hr
Maximum Mercury Feed Rate (lb/hr-12HRA)	Based on 90% of the MACT Standard using the maximum theoretical emissions calculation (MTEC)	0,00083 lb/hr
Maximum Semi-Volatile Metal Feed Rate (lb/hr-12HRA)	Average of the test run average feed rates recorded during each run of Test 2 with extrapolation	0.64 lb/hr
Maximum Low-Volatile Metal Feed Rate (lb/hr-12 HRA)	Average of the test run average feed rates recorded during each run of Test 2 with extrapolation	4.58 lb/hr

Boiler Notice of ComplianceRevision No.:0Revision:09/01/10Pageviii of xix

Table ES-3, continued

Parameter	Basis	Result
Minimum Combustion temperature (degrees F)	Average of the test run averages of Test 1	1,061 deg F
Maximum Combustion Air Flow Rate (scim)	Average of the test run averages recorded during each run of Test 2	21,130 scfm
Maximum Stack Gas Oxygen Content (%)	Average of the test run averages recorded during each run of Test 1	12.2 %
Maximum ESP Inlet Temperature	Average of the test run averages of Test 2	515 deg F
Minimum ESP Power (kW)	Average of the test run averages of Test 2	8 kW
Minimum Atomization Pressure (psig)	Manufacturer's Recommendation	30 psig
Maximum Combustion Chamber Pressure (in. w.c.)	Negative Pressure System	< ()

Boiler Notice of ComplianceRevision No.:0Revision:09/01/10Pageix of xix

FINAL PERMIT LIMITS

The final limits adopted pursuant to this NOC are described in Table ES-3. All parameters will be continuously monitored process parameters that will be tied to automatic waste feed cutoffs (AWFCO).

Boiler Notice of ComplianceRevision No.:0Revision:09/01/10Pagex of xix

TABLE OF CONTENTS

Page

Executive Summary	i
Table of Contents	х
1.0 – Notification of Compliance	l
1.1 – Facility Information and List of Key Project Personnel	1
1.1.1 – Facility Information	ł
1.1.2 – List of Key Project Personnel	1
1.2 – Detailed Engineering Description of the Boiler System	4
1.2.1 – Combustion and Fuel Feed Systems	6
1.2.1.1 – Primary Fuel Feed Systems	6
1.2.1.2 – Liquid Waste Feed Systems	7
1.2.1.3 – Sludge Feed Systems	8
1.2.1.4 – Alternative Solid Fuel Feed System	8
1.2.1.5 – Waste Vent Gases	9
I.2.2 – Prime Mover	10
1.2.3 – Waste Liquid Injection Nozzles	11
1.2.5 – Automatic Waste Feed Cutoff System	12
1.2.6 – Description of Air Pollution Control Equipment	13
1.2.7– Ash Handling System	15

Prepared by EutureFuel Chemical Company, Batesville, Arkansas, and Risk Management and Engineering, Ltd., Dallas, Texas

Page

1.2.8 – Boiler Stack	16
1.2.9 – Location and Description of Temperature, Pressure, And Flow Indicating and Control Devices	16
1.2.10 – Stack Gas Monitoring System	17
1.3 – Boiler Operator Training and Certification	19
2.0 – Sampling and Analysis Program	20
2.1 – Sampling Program	20
2.1.1 – Waste Feed Preparation	20
2.1.2 – Sampling and Monitoring Procedures	21
2.1.2.1 – Solid Fuel Feed	21
2.1.2.1.1 – Coal Samples	21
2.1.2.1.2 – Ash Spiking	21
2.1.2.2 – Organic Waste Feed	22
2.1.2.2.1 – Volatile Organic Analysis Samples	22
2.1.2.2.2 – Organic Waste Samples	22
2.1.2.3 – Metals Spiking Solution	22
2.1.3 – Combustion Gas	22
2.1.3.1 – Combustion Gas Temperature	22
2.1.3.2 – Combustion Gas Monitoring	23

	Page
2.1.4 – Stack Gas	25
2.1.4.1 – Hydrogen Chloride and Chlorine Train	25
2.1.4.2 – Multi-Mctals Train	26
2.1.4.3 – Volatile Organic Sampling Train	26
2.1.4.4 – Dioxin/Furan Sampling Train	26
2.2 – Analytical Program	27
2.2.1 – Analytical Methods	27
2.2.2 – Analytical Results	27
3.0 – Description of the Test Program	28
3.1 – CPT Objectives	28
3.2 – CPT Approach	29
3.2.1 – POHC Selection	29
3.3 – CPT Program	30
3.3.1 – Comprehensive Performance Test 1	30
3.3.2 – Comprehensive Performance Test 2	31

	Page
3.4 – CPT Sampling and Analytical Protocols	31
4.0 – Comparison of Actual Test Conditions versus Planned Conditions and Independent QA Review	33
4.1 – Combustion Device Parameters	33
4.2 – Feed Rate Parameters	34
4.3 – Constituent Feed Rate Parameters	34
4.4 – Stack Gas Parameters	34
4.5 – Independent QA Review	35
4.5.1 – Stack Testing Equipment	35
4.5.2 – Stack Gas-Particulate Analysis	35
4.5.3 – Stack Gas – Metals Analysis	35
4.5.4 – Stack Gas – Volatile Organics Analysis	36
4.5.5 - Stack Gas - Continuous Emissions Monitoring Data	36
4.5.6 – Waste-Volatile Organics	36
4.5.7 – Feed-Metals and Chloride	37
4.5.8 – Feed – General Chemistry	37
4.5.9 – Analytical Method Deviations	38
4.5.10 – Conclusion	40

	Page
5.0 - Comparison of Test Results with Regulatory Compliance Limits	4!
5.1 - Regulatory Requirements for the CPT	41
5.1.1 – Destruction Removal Efficiency	41
5.1.2 – Particulate Emissions	41
5.1.3 – HC1 / Cl ₂ Emissions	42
5.1.4 – Mercuty Emissions	42
5.1.5 – Semi-Volatile Metals Emissions	42
5.1.6 – Low-Volatile Metals Emissions	42
5.1.7 – Dioxin/Furan Emissions	43
5.1.8 – Total Hydrocarbon Emissions	43
6.0 – Procedures and Limitations for Operating Parameters	44
6.1 – Waste Feed Rate Limitations	44
6.1.1 – Maximum Hazardous Waste Feed Rate Limit	44
6.1.2 – Maximum Chloride Feed Rate Limit	44
6.1.3 – Maximum Ash Feed Rate Limit	44
6.1.4 – Maximum Metals Feed Rate Limits	44
6.1.4.1 – Maximum Mercury Feed Rate	45

	Page
6.1.4.2 – Maximum Semi-Volatile Metals Feed Rate	45
6.1.4.3 – Maximum Low-Volatile Metals Feed Rate	45
6.1.4.4 – Metals Feed Rate Extrapolation	45
6.2 – Combustion Chamber Limitations	46
6.2.1 – Minimum Combustion Chamber Temperature Limit	46
6.2.2 – Minimum Atomization Pressure Limit	46
6.2.3 – Maximum Furnace Pressure Limit	47
6.2.4 – Maximum Combustion Air Flow Rate Limit	47
6.3 – Air Pollution Control Limitations	47
6.3.1 – Minimum ESP Power Limit	47
6.3.2 – Maximum Inlet Temperature to the ESP	48
6.4 – Stack Gas Limitations	48
6.4.1 – Maximum Total Hydrocarbon Limit	48
6.4.2 – Maximum Percent Oxygen Limit	48
7.0 - Statement of Compliance	49
END OF TEXT.	49

Boiler Notice of ComplianceRevision No.:0Revision:09/01/10Pagexvi of xix

LIST OF TABLES

- Table 1.0 Feed Stream Analysis Data
- Table 2.0 Feed Stream Characteristics
- Table 3.0 Process Monitoring Instruments
- Table 4.0 Process Monitoring Instruments, Calibration, and Maintenance
- Table 5.0 CPT Process Feed Analytical Results
- Table 6.0 CPT Spiking Information
- Table 7.0 CPT Stack Gas Analytical Results
- Table 8.0 CPT Test 1 Averages
- Table 9.0 CPT Test 2 Averages
- Table 10.0 CPT Comparison of Planned Conditions to Actual Conditions -Test 1, Run 1
- Table 11.0 CPT Comparison of Planned Conditions to Actual Conditions -Test 1, Run 2
- Table 12.0 CPT Comparison of Planned Conditions to Actual Conditions -Test 1, Run 3
- Table 13.0 CPT Comparison of Planned Conditions to Actual Conditions -Test 2, Run 1
- Table 14.0- CPT Comparison of Planned Conditions to Actual Conditions -Test 2, Run 2
- Table 15.0 CPT Comparison of Planned Conditions to Actual Conditions -Test 2, Run 3

Boiler Notice of Compliance Revision No.: 0 Revision: 09/01/10 Page xvii of xix

LIST OF TABLES, continued

- Table 16.0 Demonstration of Compliance
- Table 17.0 Boiler Removal Efficiency
- Table 18.0 Summary of Operating Parameter Limits
- Table 19.0 Metal Extrapolation
- Table 20.0 Mercury MTEC Feed Rate Limit
- Table 21.0 Summary of Sampling and Analysis Program

LIST OF ATTACHMENTS

- Attachment A Alliance Source Testing CPT Report
- Attachment B Quality Assurance Project Plan
- Attachment C Instrument Calibration Records
- Attachment D CPT Field Logs
- Attachment E Other Information

Attachment E-I – Metals Spiking Certificate of Analysis

Attachment E-2 - Field Sampling and Spiking Logs

Attachment E-3 – Combustion Air Correlation Curve

Attachment E-4 – Relative Accuracy Test Audit

Attachment E-5 – Process Test Data

Attachment E-6 – Coal Feed Rate Data

LIST OF DRAWINGS

Process Flow Diagram Coal Fired Boiler	Figure 1.0
Boiler House Operating Floor Plan	6M01-1A-021
Boiler Sectional Side View	6M01-4V-014
APCD and Ductwork Arrangement	6M01-4V-016
Waste Liquid Injection Nozzle	6M01-4E-554
Boiler and Instrument Block Diagram Boiler 1	6M01-9T-024
Boiler and Instrument Block Diagram Boiler 2	6M01-9T-025
Boiler and Instrument Block Diagram Boiler 3	6M01-9T-026
I.D. Fan	6M01-5-048
F.D. Fan	6M01-5-049
Overfire Fan	6M01-5-093
Waste Feed Piping and Instrumentation Boiler 1	6M01-9T-031
Waste Feed Piping and Instrumentation Boiler 2	6M01-9T-032
Waste Feed Piping and Instrumentation Boiler 3	6M01-9T-089
CEMS Piping and Instrumentation	6M01-9T-005

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page1 of 49

1.0 Notice of Compliance

The following notification of compliance summarizes the activities associated with the comprehensive performance test (CPT), documents the results of the CPT, and serves as the basis for the development of permit conditions. The CPT was generally conducted in accordance with the CPT plan as approved April 11, 2008 as revised May 14, 2010.

1.1 Facility Information and List of Key Project Personnel

The following section gives hackground facility information and provides a list of key project personnel.

1.1.1 Facility Information

Background facility information is listed below:

Facility Name:	FutureFuel Chemical Company
Contact:	Thomas L Floyd
Address:	P.O. Box 2357
	2800 Gap Rd. Hwy 394-S
	Batesville, Arkansas 72503
Telephone Number:	(870) 698-1811
U.S. EPA Identification No.	ARD089234884
Permitting Agencies:	Environmental Protection Agency, Region 6
	Arkansas Department of Environmental Quality, Air Division
	Arkansas Department of Environmental Quality, Hazardous
	Waste Division

1.1.2 List of Key Project Personnel

A list of key project personnel is shown helow:

Project	Name	Steve Case, P.E.
Manager	Company	FutureFuel Chemical Company
	Title	Technical Associate
	Address	P.O. Box 2357
		Batesville, Arkansas 72503
	Telephone No.	(870) 698-1811
	Responsibility:	Responsible for all aspects of the CPT

СРТ	Name	Thomas L Floyd
Manager	Company	FutureFuel Chemical Company
	Title	Senior Environmental Biologist
	Address	P.O. Box 2357
		Batesville, Arkansas 72503
	Telephone No.	(870) 698-1811
	Responsibility:	Responsible for implementing and coordinating the CPT

Independent	Name	David Weeks
Third Party	Company	Risk Management & Engineering, LTD.
Quality	Title	Senior Environmental Engineer
Assurance	Address	Dallas, TX
Auditor	Telephone No.	(972) 412-6819
	Responsibility:	Independent third party quality assurance auditor for all
		aspects of the CPT

Stack	Name	Jeremy Hutchens
Sampling	Company	Alliance Source Testing
Project	Title	Environmental Scientist
Director	Address	214 Central Circle SW
		Decatur, Alabama 25603
	Telephone No.	(256)260-3974
	Responsibility:	QA/QC assurance of stack gas sampling to ensure integrity
		of emissions data.

Process	Name	Mike Gillihan
Spiking	Company	FutureFuel Chemical Company
Coordinator	Title	Process Assistant
	Address	P.O. Box 2357
		Batesville, Arkansas 72503
	Telephone No.	(870) 698-1811
	Responsibility:	Coordinating process spiking

Process	Name	Marshel Bray
Sampling	Company	FutureFuel Chemical Company
Coordinator	Title	Environmental Process Assistant
	Address	P.O. Box 2357
		Batesville, Arkansas 72503
	Telephone No.	(870) 698-1811
	Responsibility:	Coordinating process sampling

Stack	Name	Ryan O'Dea
Sampling	Company	Alliance Source Testing
Field	Title	
Supervisor	Address	
	Telephone No.	
	Responsibility:	Coordinating the collection of stack gas emissions

Laboratory	Name	Michael D. Challis
Coordinator	Company	Maxxam
& QA/QC	Title	Laboratory QA/QC Manager
Manager	Address	P.O. Box 598
		Addison, Texas 75001
	Telephone No.	(972) 931-7127
	Responsibility:	Coordinating the laboratory analysis of all samples and
		ensuring QA/QC procedures on sampling and analysis.

EPA	Name	Harry Shah
Oversight	Company	EPA, Region 6
Personnel	Title	Project Engineer
	Address	1445 Ross Ave,
		Dallas, TX
	Telephone No.	
	Responsibility:	EPA Oversight of the CPT

Agency	Name	Larissa Brown Arkansas Department of Environmental Quality	
Oversight	Agency		
Personnel	Address	1141 E. Main St, Suite #315	
		Batesville, AR 72501	
	Telephone No.	(870) 793-4762	
	Responsibility:	ADEQ oversight of the CPT	

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page4 of 49

1.2 Detailed Engineering Description of Boiler System

FFCC operates three hazardous-waste-burning coal-fired boilers (Nos. 1, 2 and 3) at its Batesville, Arkansas plant. These boilers generate steam for plant operations by burning coal, non-hazardous and hazardous liquid process wastes, biomass fuel, bio-sludge and non-hazardous solid waste for energy recovery.

The boilers and associated coal feeding systems, liquid waste feed systems, ash handling equipment, and air pollution control equipment are housed in FFCC's B-6M01 boiler area.

The three coal-fired boilers are Model MKB units built by E. Keeler Co. The boilers are identical in design, capacity, and operation. Preventive maintenance programs are also identical for each boiler. The boilers were installed at the FFCC site between October 1975 and December 1976. The practice of third party engineer construction certification was not prevalent at the time the boilers were installed. However, construction supervisors from E. Keeler were hired and present onsite as the boilers were installed and taken through start-up. At the time of construction, the E. Keeler construction supervisor ensured that the boilers were field erected within manufacturer design tolerances.

The boilers are balanced, draft, power generation, water-tube type units that have been fitted with atomizing nozzles to facilitate burning of liquid wastes. Each boiler consists of a combustion air fan, a traveling grate stoker, a steam drum, a mud drum, a super-heater, an economizer, an induced draft fan, an over-fire air fan, a soot blower system and a computerized boiler control system. The current design criteria for the boilers are as follows:

Design Criteria

- Pounds of Steam per Hour (continuous) 50,000
- Pounds of Steam per Hour (4-hour peak) 57,500
- Operating Pressure (psig) 610
- Design Pressure (psig) 675
- Steam Temperature (°F) 700
- Feedwater Temperature (°F) 270
- Total Heating Surface $(ft^2) 4,839$

- Radiant Heating Surface (ft²) 689
- Convection Heating Surface (ft²) 4,150
- Furnace Volume (ft³) 3,390
- Fuel coal, solid refuse, gas, liquid fuels and liquid waste fuels

Drawing 6M01-1A-021 and Drawing 6M01-4V-014 show equipment arrangements and a cutaway view of the Keeler boilers. The firebox dimensions are approximately 11 feet wide by 19 feet long by 45 feet tall. The boilers are rated for a maximum heat release of 76.6 million British thermal units (Btu) per hour each and a solid processing rate of 4.2 tons per hour. A simplified block flow diagram of the boiler system is provided on Figure-1.0.

The spreader-type stoker firebox combusts coal in both suspension and via mass burning. This combustion provides all of the heat needed to maintain steady flame characteristics in the boiler combustion chamber. Waste combustion is ancillary to this primary combustion. The suspension burning takes place as the coal exits the spreader. The coal ignites and burns as it travels to the grate, where the heavier coal particles continue to burn as a mass on the grate. Combustion air is routed to the firebox, both under the grates, (which supports the mass burning and overall combustion), and as overfire air, (which provides secondary air to complete the combustion process and to add turbulence in the firebox). The combustion of waste fuels and coal produces flame temperatures inside the firebox between 2000 and 2400 °F.

Cleaning and Maintenance

Each boiler is equipped with a soot-blowing system, which blows high-pressure steam through the tubes to remove soot.

The boilers are also washed out once a year before the annual preventative maintenance is performed on the boiler. The boilers are normally shutdown one at a time during the months of June, July and August for preventative maintenance. The firebox, super heater, economizer, ash handling system and precipitator sections of the boilers are washed out. High pressure water is used during the cleaning process to ensure that the debris is removed from the grates, walls and tubes through out the boiler.

In addition, if the operators determine that boiler performance has significantly deteriorated during typical process run cycles, the unit is taken out of service, inspected, and cleaned if necessary.

1.2.1 <u>Combustion and Fuel Feed Systems</u>

Currently, approximately 13,000 tons per year of fuel-quality liquid waste are burned as fuels in FFCC's three coal-fired industrial boilers in addition to coal and other types of waste fuels. The fuels and waste streams that can be fed are summarized as follows:

Feed Stream	Feed System
Coal	2 Spreader Stokers per Boiler
Nonhazardous/Hazardous Liquid Waste	1 Steam-Atomized Nozzle per Boiler
Bio-Sludge	2 Air-Atomized Nozzles per Boiler
Biomass Fuels (e.g., Wood chips)	2 Spreader Stokers per Boiler

FFCC processes the feed streams described above as follows:

- The feeder stokers feed coal and biomass fuels into the boiler.
- All liquid wastes are atomized through waste nozzles by high-pressure steam into the boiler.
- The bio-sludge wastes are atomized through waste nozzles by high-pressure air into the hoiler.

It is FFCC's understanding that successful performance of a CPT in accordance with this plan will constitute approval to hurn the types of waste described in this section.

1.2.1.1 Primary Fuel Feed Systems

Stoker coal is the primary fuel used to maintain the Boilers at a steady state. Coal is fed to the boilers on a continuous basis (i.e., 24 hours/day, 7 days/week) to maintain the desired steam demand. The coal used by the boiler system is a solid material with a maximum particle size of 2 inches. Heating value and other characteristics are described in Tables-1.0 and 2.0.

Coal is delivered in rail cars or trucks, which are unloaded into track-hoppers. From the hoppers, the coal is conveyed on a helt conveyor system up to three separate coal bunkers inside the B-6M01 boiler area. The coal is gravity fed from the bunkers into the boilers. The coal is mechanically spread by stokers which flip it to the opposite side of the boiler, (the boilers are equipped with two spreader stokers, identical in design and operation). The coal falls onto a traveling grate that slowly moves the burning bed of coal across the hoiler.

Boiler Notice of Compliance	
Revision No.:	0
Revision:	9/1/10
Page	7 of 49

The feed rate of the coal is determined by measuring the steam flow rate, economizer water temperature outlet, waste feed rate, and the heat contents of the waste and coal. Coal feed rate is determined by the following equation:

$$Q = \frac{\left\{ \frac{\left[A \times \left(1,308\frac{BTU}{lb} - 244\frac{BTU}{lb}\right)\right] - \left[A \times \left(B - 274^{\circ}F\right) \times \left(0.525\frac{BTU}{lb - T}\right)\right]}{\left(\frac{E}{100}\right)} - \left(M \times F\right) \right\}}{G}$$

Where,

Q = Coal Feed Rate (lb/hr)

A =Steam Flow Rate (lb/hr)

B = Economizer Water Outlet Temperature (° F)

E = Boiler Efficiency (typically 70 to 80%)

F = Heat Content of Waste (BTU/lb)

G = Heat Content of Coal (BTU/lb)

M = Waste Feed Rate (lb/hr)

1.2.1.2 Liquid Waste Feed Systems

The characteristics of the FFCC wastes are described in Tables-1.0 and 2.0. The waste characteristics provided in the tables are typical of as-fired conditions; althougb FFCC may occasionally blend hazardous waste by moving material between tanks to meet desired waste feed characteristics and regulatory requirements. The liquid wastes burned in the boilers are normally supplied from either the No. 1 Tank Farm or the Sludge Tanks. The No. 2 Tank Farm normally does not send waste directly to the boilers, but vent gases from the No. 2 Farm are routed to the boilers. The No. 1 Tank Farm consists of three storage tanks. These tanks are permitted hazardous waste tanks referred to as WB-01, WB-02, and WB-03. Each of these tanks has a 10,000-gallon capacity and is equipped to hold liquid wastes for treatment at the RCRA facility. The tanks are equipped with high- and low-level alarms and level indicators, located in the Incinerator and Boiler Control Rooms.

The sludge tanks, WDT-01 (North Tank) and WDT-02 (South Tank), are RCRA-permitted hazardous waste tanks. Both tanks are 2,500-gallon in capacity, insulated, and externally lined

Boiler Notice of Compliance		
Revision No.:	0	
Revision:	9/1/10	
Page	8 of 49	

with 30 psig heat tracing. Both tanks are equipped with high- and low-level alarms and level indicators, located in the Incinerator and Boiler Control Rooms.

Liquid waste can also be burned directly in the boilers from containers. FFCC uses several 750gallon portable containers to transfer wastes to the treatment facility. FFCC also has the ability to burn directly from a tank truck if necessary. The portable containers are used for special manufacturing situations and for waste that have the potential to be incompatible with the waste stored in FFCC storage tanks.

Centrifugal pumps are used to transfer waste from the storage tanks and containers to the boilers tbrough a variety of lines. The flow rate of the liquid wastes is continuously monitored by mass flow meters and can be adjusted with a control valve located at each boiler. Two automatic block valves in the liquid waste line ensure proper shut-off when not burning. The liquid hazardous waste feed system has the capability of supplying up to 5,100 lb/hr of liquid waste to the boilers; however, typical waste feedrates are approximately 2,000 - 2,500 lbs/hour. The liquid waste feed line is controlled by atomizing the waste feed with 150 psig steam.

The boiler waste feed piping and instrumentation can be seen in drawing 6M09-9T-089.

1.2.1.3 Sludge Feed Systems

In addition to liquid hazardous waste, the boilers are also equipped to burn various types of sludges. Currently, only non-hazardous sludge from FFCC's biological waste water treatment plant is burned in the boilers via nozzles (separate from those used to burn liquid waste) which use high-pressure air to atomize the waste as it is introduced into the boilers. The bio-sludge feed system has a capacity of 7,000 lb/hr.

1.2.1.4 Alternative Solid Fuel Feed System

Alternative fuel is conveyed on a belt conveyor system along with the coal up to three separate coal bunkers inside the B-6M01 boiler area. It is then gravity fed from the bunkers with the coal into the boilers. The alternative fuel and coal is mechanically spread by stokers which flip it to the opposite side of the boiler, (the boilers are equipped with two spreader stokers, identical in design and operation). The alternative fuel falls onto a traveling grate that slowly moves with the burning bed of coal across the boiler.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page9 of 49

1.2.1.5 Waste Vent Gases

Fugitive emissions from FFCC's hazardous waste storage tanks are routed to Boilers No. 1 and 2 for destruction. The P&ID diagram for the vent system is attached. Each tank vents through a detonation arrestor and pressure safety valve (PSV) into a common stainless steel header. The header is equipped with low point drains that are steam traced. The temperature of the steam tracing prevents the vapors from condensing in the header. The drains are checked periodically to ensure that vapor condensation is not occurring. The coal-fired boilers operated under negative pressure with an induced draft fan on the back end of the boiler system. This induced draft pulls the tank vapors into boiler combustion zone for destruction. The vapors pass through a flow meter, a double block and bleed valve arrangement, a manual valve, and a flame arrestor, before entering the liquid waste feed injection nozzle. The vent system is controlled by the boiler operators.

The vent system ductwork is constructed of 316L stainless steel from the tanks up to the point where the duct enters building 6M01. The ductwork construction material then changes to carbon steel. Carbon steel is used because the ductwork from the edge of building 6M01 to the boilers was present prior to installation of the vent gas feed system. This existing carbon steel ductwork was not being used for another purpose at the time the vent system was designed and constructed.

Detailed information about the vent system can be obtained from FFCC's Title V permit application. Emissions from each of the 11 tanks are included in the volatile organic compound emission estimates. EPA's TANKs program was used to derive the VOC emission rate estimates. These calculations resulted in an estimate of 3.9 lb/hr VOCs vented to the boilers (this amount of VOCs is negligible when compared to the trial burn liquid waste feed rates of 1,350 and 3,672 lb/hr). Although the waste vent gases are currently routed only to boilers 1 and 2, as noted above, the impact from these VOCs is negligible. Given this negligible impact, boiler 3 is representative of the operational conditions and performance of all units. Based on the 3.9 lb/hr emission rate, the worst case chlorine emission is 0.6 lb/hr chlorine assuming all the VOCs in the tanks is monochlorobenzene.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page10 of 49

1.2.2 Prime Mover

The induced draft fan will provide motive force to transport the combustion gas through the unit where it is expected to exit the boilers at temperatures between 375 and 515 °F. The fan is a Model 6,000 Radial Tip type assembly manufactured by Chicago Blower (see drawing 6M01-5-048). The flue gas flow rate is expected to range between 20,000 to 40,000 actual cubic feet per minute (acfin) when producing steam. The maximum flue gas flow rate capacity was set at 49,827 acfm during 1999 trial burn. We plan to target 42,000 acfin during the CPT. Specific design criteria for the prime mover are provided below and are based on a specific static pressure.

The forced draft fan will provide the majority of the combustion air to the boiler system (see drawing 6M01-5-049). The forced draft (FD) fan is the only variable combustion air input. Combustion air is also supplied by the overfire fan (shown in drawing 6M01-5-093) and waste chemical fan (shown in drawing 6M01-9T-089), but these two combustion air inputs are fixed at 3,050 scfin and 1,800 scfm respectively. These two fixed air supplies will be on when burning waste and their respective total comhustion air input (4,850 scfm) will be added to the combustion air provided by the FD fans. The FD fans provide 21,456 scfm combustion air at 100% of design. The amount of combustion air provided by these identical fans is measured the same on all three boilers. The measurement is the square root of a differential pressure across an annular ring. This measurement is recorded as 0 to 100% of measured flow. The % measured flow will be converted to scfm based on the design of the FD Fans.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page11 of 49

1.2.3 Waste Liquid Injection Nozzles

Each boiler has one waste liquid injection nozzle located above the coal fuel bed. Each nozzle is designed to atomize liquid waste using 150 pounds per square inch gauge (psig) high pressure steam for atomization which helps to prevent blockage. The waste liquid injection nozzle can be seen in drawing 6M01-4E-554. The steam atomization pressure is maintained at a minimum of 40 psig. The liquid waste pressure is maintained at a minimum of 20 psig when burning waste. The burner is also equipped with piping to facilitate burning of waste vent gases and a separate blower to provide combustion air that facilitates waste burning.

Even though the waste injection system is composed of traditional burner elements that combine fuel and air, the system does not function as a primary fuel burner. Rather, the liquid waste injection system is merely designed to facilitate the combustion of the waste by the primary coal fed combustion system. The temperature and turbulence needed to ensure waste combustion is provided by the coal fuel bed. The coal fuel bed supplies the flame stability for the system. For example, FFCC procedures establish a minimum steam production and minimum temperature before waste is introduced to the boiler. Hence, traditional burner design criteria are not totally applicable to the liquid waste injection system.

Design Criteria

- Size 100 million BTU/hour
- Viscosity The liquid waste injection nozzle does not function as a stand-alone burner that is solely responsible for waste combustion performance. Rather combustion inside the boiler firebox is controlled and effectuated by the combustion of the coal-fuel bed. The liquid waste injection nozzle serves only to introduce waste into the boiler firebox. Therefore, viscosity is not a design concern for this nozzle because the nozzle does not function as a burner and combustion stability is maintained by the coal fuel bed. For informational purposes, the nozzle as a burner is designed to maintain a stable flame with liquids ranging from No. 2 to No. 6 fuel oil. No. 6 fuel oil at 100°F has a viscosity of 310 centipoise. If the nozzle was used as a main waste burner, then a design viscosity to ensure stable flame conditions might be appropriate. However, the nozzle is not being used as a burner in this boiler design and thus, should not be a limiting parameter of the permit.
- Minimum Heating Value None.
- Excess Air The liquid waste injection nozzle and boiler forced draft fan jointly supply excess air to the boiler firebox.
- Burner Pressure The pressure range for the liquid waste injection nozzle is 10 to 125 psig. The pressure range for the steam atomization pressure is 3 50 psig.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page12 of 49

1.2.5 Automatic Waste Feed Cutoff System

The primary function of the AWFCO system is to prevent the feeding of hazardous waste if combustion conditions are outside notification of compliance (NOC) limits. The AWFCO system is implemented in the boiler distributive control system (DCS).

FFCC boilers are equipped with a central control system paired with a Data Acquisition System (DAS) to provide the necessary monitoring and control. The control system is comprised of a DCS (MOD 300), and the DAS is a separate, interfaced data historian (OSISoft PI System). The DCS is linked to the Continuous Emission Monitoring System (CEMS) and other process instrumentation necessary to properly control boiler combustion and the fuel feed systems and to monitor permit parameters.

The DCS is equipped with operator interfaces so that boiler operation can be monitored and modified, if necessary, by the operator when the boiler is in use. The DCS receives as inputs the operating variables from the CEMS and other system monitoring devices and transmits their values to the operator interfaces and the DAS system. The values are also stored within the DCS for short-term recordkeeping. The DCS receives new input readings of each required variable at least every 15 seconds and writes the value to the DAS. The DCS also computes a number of calculated values that are necessary to ensure permit compliance. Periodic and hourly rolling averages are also calculated in the DCS as required and where not already calculated in the CEMS. The DCS is capable of short-term recordkeeping and the PI system provides the long-term historical record. Industry standard data compression techniques are used to minimize the storage space required for the long-term historical data.

The 60-minute rolling averages for the CEMS (THC and O2) are computed by an ESC data logger. The ESC uses the data measured by the CEMS at least every 15 seconds to compute 1-minute periodic averages. The ESC then calculates hourly rolling averages once a minute based on the previous sixty 1-minute periodic averages for each parameter. The hourly rolling averages computed by the ESC data logger are sent directly to the DCS before being forwarded to the DAS. The DCS uses the hourly rolling average CEMS data to activate interlocks if necessary.

The DCS monitors the boiler operation including all the required permit operating parameters. If any parameter exceeds its regulatory limit, the DCS initiates an automatic waste feed cutoff by transmitting a digital signal which immediately closes the dual block valves that prohibit waste from entering the unit. Simultaneously, the DCS initiates an increase in the coal feed rate to maintain boiler load and temperature. The radiant heat in the boiler will not allow the chamber temperature to drop significantly during the cutoff of liquid hazardous waste. Therefore, any residual liquid hazardous waste fuel in the boiler will be burned. In accordance with regulatory requirements, FFCC will test the waste feed cutoff system, for each boiler, once each week during normal operation. This test includes all associated alarms.

Boiler Notice of Compliance		
Revision No.:	0	
Revision:	9/1/10	
Page	13 of 49	

In order to minimize the number of waste feed eutoffs, the DCS will annunciate the appropriate alarm when any of the monitored variables reach a high percentage of their limits. This early warning permits the boiler operator to take necessary corrective action before established limits are exceeded. The early alarm does not activate the AWFCO system. The AWFCO system is activated immediately whenever a permit related parameter reaches its limit. The DAS records all automatic waste feed cutoffs, either tests or actual.

The final AWFCO parameter values will be based on the operating limits established during the DPT and will be identified in the notification of compliance.

1.2.6 Description of Air Pollution Control Equipment

Combustion gases from the boilers are treated by individual electrostatic precipitators (ESPs) to remove the flyash and other contaminants. Drawing 6M01-4V-016 provides a cut out view of the boiler pollution control equipment arrangement. The design criteria for the ESP are as follows.

Design Criteria

- Manufacturer Name: Research-Cottrell, Inc.
- Model No: IP3280
- Removal Efficiency: 99.6% (based on 1999 Trial Burn)
- Performance Monitoring: Power Input (KW)
- Nucndorfer Precipitation Controls

Each section of the three ESPs has fifteen 9-inch wide ducts, formed by sixteen vertically aligned collecting plates, 15 feet high. The collecting plates are arranged in two banks of eight plates, and each bank of plates is supported at its upper leading and trailing edges by individual anvil beams. There are 48 collecting plates in each ESP. Suspended in the center of the ducts of each ESP section are 120 discharge electrode wires. These wires are formed by 0.1055" diameter, hard-drawn basic wire without coating, with a capped shroud at the top and bottom. Each wire supports a 15 or 25 pound cast iron weight, in plumb bob suspension, except four wire frames, which support the steadying bars. There are 360 discharge electrodes in each ESP. To provide for more uniform gas distribution, perforated distribution plates are located at the inlet end of each ESP. A 3/16" -thick, mild-steel shell encloses the collecting plates and discharge electrodes. Two pyramidal-type hoppers are located at the bottom of each ESP and three insulator compartments are located on the roof. Each insulator compartment serves one unit section in each ESP.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page14 of 49

Transformer-Rectifier Sets

High voltage, uni-directional power for energizing wire supporting frames in each ESP is supplied by three, 45 kv and 250 ma transformer-rectifier sets (silicon). The three transformers on each ESP are located on the ESP roof. Connections to the interior of each unit section are made with one run of pipe and guard. Each transformer-rectifier energizes one wire frame in one unit section with full-wave voltage. The high voltage transformer and silicon rectifier of each set are submerged in an oil-filled tank, which is equipped with two high-voltage output bushings.

An advanced microprocessor energization system, designed to maintain maximum usable ESP power by monitoring and automatically controlling electrical variables, is provided. It continuously and automatically adjusts to conditions such as changing boiler loads, variable fuel supplies, and upset situations without readjustment.

The control units for the transformer rectifier sets (silicon) are free standing, metal cabinets for indoor installation. Manual and automatic controls are provided. Each control unit contains all of the components required to control and protect one transformer-rectifier set except for a resistor board, which is located in the rectifier ground switch enclosure.

Collecting Plate Rappers

The collecting plates of each ESP are cleaned of collected material by the action of seven magnetic-impulse, gravity impact (MIGI) rappers. There are two rappers in each unit section and one in the inlet flues.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page15 of 49

Discharge Electrode Vibrator

The discharge electrode wires of each ESP are cleaned of collected material by the action of three vibrators. There is one vibrator in each unit section.

Collecting Plate Rapper Control

The control for the MIGI rappers is installed in a single cabinet for each ESP. The control for each ESP is a Neundorfer MicroRap microprocessor-based rapper control system. The system energizes a rapper on a custom frequency. The rappers can be managed controlled in groups or individually. This allows the rapping intensity of each group or rapper to be adjusted independently of the other groups.

Discharge Electrode Vibrator Controls

The controls for the vibrators are installed in a single cabinet for each ESP. Each control includes one vibration period controller and one cam timer. The cam timer has a synchronous gear-motor, cam shaft, cams and switches. The motor drives the camshaft at a constant speed and the rotating cams actuate eight load switches. A vibrator is energized each time a load switch closes. A multi-notch cam and switch on each cam timer is used to actuate the vibration period controller which limits the period of vibration of each vibrator to a maximum of six seconds. The cam timer is adjusted so that the vibrators are energized in consecutive order without overlapping. The cam timer has a time cycle of 30 minutes.

1.2.7 Ash Handling System

The ash handling system is composed of two subsystems; the bottom ash hopper sluicing system and the fly ash system. The bottom ash hopper sluicing system removes clinkers (bottom ash) from the three firebox bottom ash hoppers and flushes them out of the plant to an ash management area. Each bottom ash hopper is equipped with a clinker crusher and Jetpulsion pump. The crusher grinds the clinker material down to a size suitable for sluicing out of the plant by the Jetpulsion pump using high pressure ash service water. The Jetpulsion pumps are controlled by valves in the supply line to each pump. The three bottom ash hoppers are normally emptied one after the other in sequence manually or by an automatic control system. The control panel also contains controls to operate the sluice gate associated with the precipitator hopper washdown drain sump.

The fly ash system removes stored fly ash from six precipitator hoppers, three economizer hoppers and three siftings hoppers, mixes it with water and sluices it out of the plant to the ash management area. The hopper sections are equipped with self-feeding fly ash intakes, which control the flow of ash from the collecting hoppers to the ash transport lines. The ash is conveyed pneumatically through the transport lines to a Hydroveyor exhauster. The ash/air/water mixture then passes through an air separator where the air is bled off. The

Boiler Notice (of Compliance
Revision No.:	0
Revision:	9/1/10
Page	16 of 49

ash/water slurry continues down the sluicing line to the management area under pressure from the Hydroveyor exhauster. The fly ash intakes are operated as a programmed sequence by an automatic control system. This automatic sequence is initiated by push-button from the fly ash system mimic control panel and proceeds to completion.

1.2.8 Boiler Stack

The three boilers share a common 200-foot tall by 108-inch (at exit) diameter stack. The stack interior is constructed of gunnite and red brick. The external part of the stack is made from 10-inch thick reinforced concrete. The base of the stack is 232 inches in diameter. During the trial burn, combustion gases will bypass the main stack and be sampled from a "stub" stack especially designed to facilitate sampling of stack gases. The main stack cannot be used for sampling because the emissions from Boiler No. 3 cannot be differentiated from the non-sampled coal fired boilers. The agency has approved use of the "stub" stack in all previous testing. The "stub" stack is constructed such that cyclonic flow is controlled.

1.2.9 <u>Location and Description of Temperature, Pressure, and Flow Indicating</u> and Control Devices

FFCC uses a variety of devices to control the operation of the boiler system. A list of the instruments important to the combustion of hazardous waste including their description and location can be seen in Tables-3.0 and 4.0.
Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page17 of 49

1.2.10 Stack Gas Monitoring System

FFCC continuously monitors oxygen (O2) and total hydrocarbon (THC) levels in the combustion stack gases. The continuous emission monitoring system (CEMS) presently utilized by FFCC is an extractive system. The CEMS and other process instrumentation are summarized in Tables 3.0 and 4.0. The CEMS piping and instrumentation diagram can be seen in drawing 6M09-9T-005.

The CEMS is divided into three parts: a sample probe (installed directly on the dedicated breech to a common stack); a sampling gas conditioning and analysis cabinet (SGC&AC); and a heated sample line (HSL) connecting these two parts. The probe/primary filter assembly is a sample extractive device, which is placed in direct contact with the flue gas environment. It is used to draw in and filter a representative sample of the flue gas stream to be processed by the SGC&AC. A clam shell designed heating jacket, which surrounds the probe body, heats the sample probe.

A sample pump located in the SGC&AC, universal gas conditioning system, is used to draw in a continuous gas sample from the Extractive Probe Assembly via the HSL. The main function of the SGC&AC is to dry the sample gas by means of lowering the dew point of the gas sample, thus removing the moisture. Then, the dried gas sample is delivered to the O2 and THC analyzers. The SGC&AC conditions the sample for analysis hy the analyzers. This SGC&AC system can be divided into four basic sections: the gas sample drying section, dried sample distribution sections, the analysis section and auto-calibration/data collection. The first point of contact for the gas sample after being transported via the HSL is the pre-cooler. From here, the partially dried gas sample and the condensed moisture flow into the secondary Condensate Collector. Next, the gas sample flows on to the Sample Conditioner to have the remaining moisture removed. From the sample conditioner (Universal), the dried gas sample flows through the Instrument Plate, through a set of rotometers which control the flows and into the Analyzer Solenoid Plates (ASPs) via a distribution manifold. From the ASPs, the gas sample is valved again and then flows into the analyzers for measurement. Lastly, the gas sample is dumped into an exhaust manifold and vented.

The Heated Sample Line (HSL) is the device used to transport the gas sample from the Extraction Probe to the SGC&AC. The HSL will also maintain the gas sample temperature to ensure that the moisture in the gas sample does not condense within the HSL. The sample temperature is set between 250 °F to 300 °F. The HSL consists of the following items: one 240 VAC heating cahle, three 1/4" Teflon tubes, one twisted pair cable to be used for the RTD signal from the probe RTD, one RTD to sense the temperature of the sample line, and three 14 American Wire Gauge (AWG) wires for the probe power.

The CEMS system (sample conditioning equipment, analyzers, and data collection equipment) is located in a separate building just south of Boiler No. 3. When a malfunction occurs with this system, an AWFCO will occur, a local alarm light inside the building will be displayed, and an audible alarm will sound in the Boiler control room area. The Boiler Operator stationed in the area will acknowledge the alarm and determine the why a problem occurred. Maintenance personnel will then be dispatched to fix the problem. The following conditions will cause the alarms mentions above.

- Sample probe/sample line temperature low
- Moisture in sample
- Sample flow low
- Sample pump off
- Condensate trap level high
- Total Hydrocarbon analyzer flame out condition

The O2 analyzer is a stack gas analyzer. The O2 analyzer has the following characteristics:

Principle	Paramagnetic
Operation:	
Range:	0-25% O2
Response Time:	90% of full scale, 20 seconds
Repeatability:	0.01% O2 or \pm 1% of full scale, whichever is greater
Zero Drift:	± 1% full scale per day, ambient temperature does not change more
	than 20 degree F± 2% full scale per eek, ambient temperature change
	over entire range
Span Drift:	± 1% full scale per week

The THC analyzer is a standard stack gas analyzer and has the following characteristics:

Principle Operation:	Flame Ionization
Range:	Range switch has eight positions: 1, 2.5, 10, 25, 100, 250, 1000 and REMOTE. SPAN control provides continuously variable adjustment within a dynamic range of 1:4
Response Speed:	90% of full scale within 0.6 seconds with sample bypass flow at 3 liters/minutes
Repeatability:	1% of full scale for successive identical samples
Zero Drift:	± 1% of full scale per 24 hours
Span Drift:	\pm 1% of full scale per 24 hours

Raw data collected by the CEMS is delivered to the Odessa control unit and the DCS. The Odessa control unit also computes 60-minute rolling averages for each CEMS variable. The 60-minute rolling average is then delivered to the DCS and logged into the DAS. The Odessa control unit also initiates, through a timer, the daily calibration of the CEMS. This calibration is sequenced automatically.

1.3 Boiler Operator Training and Certification

All boiler operators receive initial training by completing the Utilities Operator Apprentice Program. This is a site-specific procedure based training program that has been reviewed and certified by the state Department of Labor. The training includes written materials, bands-onlabs, and on-the-job training. Once an apprentice operator has received training for a task or course and has completed that section of the Apprenticeship Program the operator is then allowed to do that task by their self.

A certified operator will instruct the apprentice until they bave successfully completed the program and passed a written examination administered by the instructor. The completions are recorded in an on-site training management system called TEDS.

An annual review is administered to the operators and critical staff that include both Title V/MACT and RCRA updates along with pertinent operational training, including but not limited to the SSM plan. The Utilities Operator Apprentice Program, including the Boiler Operator Training, is reviewed and updated whenever changes have been made that could improve the training or make the existing training obsolete.

A certified control room operator will be on duty at the site when the boiler is operating.

Boiler Notice of Compliance Revision No.: 0 Revision: 9/1/10 Page 20 of 49

2.0 Sampling and Analysis Program

This section summarizes the sampling and analysis programs conducted during the CPT.

2.1 <u>Sampling Program</u>

FFCC fed the following streams during the CPT:

- I. Coal
- 2. Ash Spiking
- 3. Organic Waste
- 4. Metals Spiking Solution

The characteristics of the hazardous waste feed during the CPT are summarized in Table 5.0, and CPT spiking information is summarized in Table 6.0.

2.1.1 Waste Feed Preparation

Coal – Coal is the primary fuel used to produce steam at the coal-fired hoilers. The coal is a high Btu fuel that contains no POHCs. Diatomaceous earth was spiked into the coal to maximize the ash feed rate during the CPT.

Organic Waste - The organic waste used in hoth Test 1 and Test 2 was prepared by using waste from FFCC's manufacturing units. Acetic Acid, chlorobenzene and perchloroethlyene were transferred from manufacturing to a permitted waste tank at Utilities. The stream used in Test 1 was prepared as a 96% Acetic Acid, and 4% chlorobenzene solution. The purpose of the stream was to provide a low Btu waste feed that contained a primary organic hazardous constituent (POHC) to demonstrate DRE at low temperatures. The stream used in Test 2 was prepared as an 87% Acetic Acid, 4% chlorobenzene, and 6% perchloroethlyene solution. The purpose of this stream was to provide a waste containing a POHC, as well as, chlorine to support the metals removal efficiency of the system with chlorine present.

Metals Spiking Solution – The metals spiking solution was prepared by Blue Ridge Chemicals, Inc. headquartered in Bala Cynwyd, Pennsylvania. This solution was prepared in order to establish MACT EEE feed limits for semi-volatile (cadmium and lead) and low-volatile (arsenic, beryllium, and chromium) metals. Lead was chosen to represent the semi-volatile metals and chromium was chosen to represent the low volatile metals. The solution was ordered to exact specifications and certified by Blue Ridge Chemicals as to its composition (certificate of analysis is located in Attachment E-1 The high volatile metal, mercury, was not spiked in to the feed

Boiler Notice o	f Compliance
Revision No.:	0
Revision:	9/1/10
Page	21 of 49

stream. Mercury is not expected to be in FFCC waste but is expected to be present in the coal feed.

2.1.2 <u>Sampling and Monitoring Procedures</u>

The CPT involved two different tests (Test 1 and Test 2) consisting of three replicate runs at the extreme range of normal conditions. Each test involved stack emission sampling for POHC destruction to ensure 99.99% DRE and to demonstrate compliance with THC emissions. Test 2 also involved stack sampling to ensure particulate matter, mercury, semi-volatile metal, and low-volatile metal emission standards. Waste feed and coal samples were obtained during each run of the CPT.

The procedures for collecting samples are summarized in Table 7.0. Sampling frequency and reference methods also are included in Table 7.0. Additional details regarding sampling frequencies and methods follow.

2.1.2.1 Solid Fuel Feed

2.1.2.1.1 <u>Coal Samples</u>

The coal feed was sampled at 30-minute intervals during each trial burn run under both test conditions. The original plan called for FFCC to sample the coal and diatomaceous earth as fed; however, the Agency directed us to modify the original sampling plan to obtain the coal samples prior to the addition of the diatomaceous earth. The coal samples were then taken from a sample point at the exit of the coal silo which is approximately 8 feet before the area where the ash (diatomaceous earth) was spiked into the coal.

2.1.2.1.2 <u>Ash Spiking</u>

The diatomaceous earth was added to the coal just before the bucket elevator takes the coal up the 100-ton coal bunker. The diatomaceous earth was added to the coal at a rate of 7.5 lbs of diatomaceous earth to every 100 lbs of coal. A summary of the ash spiking information can be found in Table 6.0 and the field spiking logs can be seen in Attachment E-3.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page22 of 49

2.1.2.2 Organic Waste Feed

2.1.2.2.1 Volatile Organic Analysis Samples

The volatile organic analysis (VOA) samples where collected from a tap in the waste feed line prior to the metals solution injection point. Grab samples for the VOA were taken every 30 minutes, packaged separately in 40ml vials, and then placed in ice and cooled to 4 degrees C. These samples were composited by the laboratory immediately before analysis.

2.1.2.2.2 Organic Waste Samples

The organic waste samples were also collected from a tap in the waste feed line for analysis other than volatile organics. These grab samples were taken every 30 minutes and were composited into 4oz jars at the sampling sight in an ice cooled environment. The composites were maintained in an ice cooled environment and transported to the designated analytical laboratory for analysis.

2.1.2.3 Metals Spiking Solution

The metals spiking solution was not sampled as a separate waste stream. It was prepared off-site by a third party who certified the composition. Attachment E-1 contains a copy of that certification. The solution was fed to the liquid waste feed line upstream of the combustion chamber injection point, and downstream of where the hazardous waste samples were taken. The feed rate of the metals solution was controlled using a chemical metering pump and a scale. The feed rate was maintained by calculating the weight loss of the container holding the metals solution. The scale calibration documentation can be found in attachment E -2.

2.1.3 <u>Combustion Gas</u>

2.1.3.1 Combustion Gas Temperature

The combustion gas temperature was measured in the combustion chamber by thermocouple with an instrument range of 0 - 2200 degrees Fahrenheit. The exact location of the thermocouple can be seen in drawing 6M01-4V-016, and is described in drawing 6M01-9T-026.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page23 of 49

2.1.3.2 Combustion Gas Monitoring

FFCC continuously monitors oxygen (O2) and total hydrocarbon (THC) levels in the combustion stack gases. The continuous emission monitoring system (CEMS) presently utilized by FFCC is an extractive system. (The CEMS and other process instrumentation are summarized in Tables-3.0 and 4.0). This system was calibrated and a RATA was performed on the system just prior to the CPT. The results of the RATA can be reviewed in Attachment E-4.

The CEMS is divided into three parts: a sample probe (installed directly on the dedicated breech to a common stack); a sampling gas conditioning and analysis cabinet (SGC&AC); and a heated sample line (HSL) connecting these two parts. The probe/primary filter assembly is a sample extractive device, which is placed in direct contact with the flue gas environment. It is used to draw in and filter a representative sample of the flue gas stream to be processed by the SGC&AC. A clam shell designed heating jacket, which surrounds the probe body, heats the sample probe.

A sample pump located in the SGC&AC, universal gas conditioning system, is used to draw in a continuous gas sample from the Extractive Probe Assembly via a heated sample line (HSL). The main function of the SGC&AC is to dry the sample gas by means of lowering the dew point of the gas sample, thus removing the moisture. Then, the dried gas sample is delivered to the O2 and THC analyzers. The SGC&AC conditions the sample for analysis by the analyzers. This SGC&AC system can be divided into four basic sections: the gas sample drying section, dried sample distribution sections, the analysis section and auto-calibration/data collection. The first point of contact for the gas sample after heing transported via the HSL is the pre-cooler. From here, the partially dried gas sample and the condensed moisture flow into the secondary Condensate Collector. Next, the gas sample flows on to the Sample Conditioner to have the remaining moisture removed. From the sample conditioner (Universal), the dried gas sample flows through the Instrument Plate, through a set of rotometers which control the flows and into the Analyzer Solenoid Plates (ASPs) via a distribution manifold. From the ASPs, the gas sample then flows into the analyzers for measurement. Lastly, the gas sample is dumped into an exhaust manifold and vented.

Boiler Notice o.	f Compliance
Revision No.:	0
Revision:	9/1/10
Page	24 of 49

The HSL transports the gas sample from the Extraction Probe to the SGC&AC. The HSL will also maintain the gas sample temperature to ensure that the moisture in the gas sample does not condense within the HSL. The sample temperature is set between 250 °F to 300 °F. The HSL consists of the following items: one 240 VAC heating cable, three 1/4" Teflon tubes, one twisted pair cable to be used for the RTD signal from the probe RTD, one RTD to sense the temperature of the sample line, and three 14 American Wire Gauge (AWG) wires for the probe power.

The CEMS system (sample conditioning equipment, analyzers, and data collection equipment) is located in a separate building just south of Boiler No. 3. When a malfunction occurs with this system, an AWFCO will occur, a local alarm light inside the building will be displayed, and an audible alarm will sound in the Boiler control room area. The Boiler Operator stationed in the area will acknowledge the alarm and determine the why a problem occurred. Maintenance personnel will then be dispatched to fix the problem. The following conditions will cause the alarms mentions above.

- Sample probe/sample line temperature low
- Moisture in sample
- Sample flow low
- Sample pump off
- Condensate trap level high
- Total Hydrocarbon analyzer flame out condition

The O2 analyzer is a stack gas analyzer. The O2 analyzer has the following characteristics:

Principle	Paramagnetic
Operation:	
Range:	0-25% O2
Response Time:	90% of full scale, 20 seconds
Repeatability:	0.01% O2 or $\pm 1\%$ of full scale, whichever is greater
Zero Drift:	\pm 1% full scale per 24 hours, ambient temperature does not change more than 20 degree F \pm 2.5% full scale per 24 hours, ambient
	temperature change over entire range
Span Drift:	\pm 1% full scale per 24 hours
	$\pm 2.5\%$ full scale per 24 hours

Boiler Notice of	Compliance
Revision No.:	0
Revision:	9/1/10
Page	25 of 49

The THC analyzer is a standard stack gas analyzer and has the following characteristics:

Principle Operation:	Flame Ionization
Range:	 Range switch has eight positions: 1, 2.5, 10, 25, 100, 250, 1000 and REMOTE. SPAN control provides continuously variable adjustment within a dynamic range of 1:4
Response Speed:	90% of full scale within 0.6 seconds with sample bypass flow at 3 liters/minutes
Repeatability:	1% of full scale for successive identical samples
Zero Drift:	\pm 1% of full scale per 24 hours
Span Drift:	\pm 1% of full scale per 24 hours

Raw data collected by the CEMS is delivered to the Odessa control unit and the DCS. The Odessa control unit also computes 60-minute rolling averages for each CEMS variable. The 60-minute rolling average is then delivered to the DCS and logged into the DAS. The Odessa control unit also initiates, through a timer, the daily calibration of the CEMS. This calibration is sequenced automatically.

2.1.4 Stack Gas

During the trial burn, combustion gases was diverted from the main stack and sampled from a "stub" stack especially designed to facilitate sampling of stack gases. The main stack cannot be used for sampling because the emissions from Boiler No. 3 cannot be differentiated from the other two identical coal fired boilers. The agency has approved use of the "stub" stack in all previous testing. The "stub" stack is constructed such that cyclonic flow is controlled.

2.1.4.1 Hydrogen Chloride and Chlorine Train (Method 26A)

FFCC demonstrated compliance with the alternative risk based standard for total chlorine by measuring the chlorine feed rate and using MTEC to determine stack emissions. The Alternative Risk Based Standard for Chlorine was provided in Appendix D of the FFCC CPT Plan.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page26 of 49

2.1.4.2 Multi-Metals Train (Method 29)

An MMT (multi-metals train) (EPA Method 29) was used for collection of metals and particulate from the stack gas during the CPT. The sampling train impingers were be charged with a solution of 5 percent nitric acid and 10 percent hydrogen peroxide to capture metals (i.e., arsenie, beryllium, cadmium, chromium, lead, and mercury) and acidified potassium permanganate (composed of 4 percent potassium permanganate and 10 percent sulfuric acid) to capture any mercury that will not be captured by the nitric acid and bydrogen peroxide solution. The total sampling time was about 2 hours during each replicate sampling run. The MMT was operated concurrently with the other sampling train(s) and sampled approximately 3 cubic meters of stack gas. The MMT procedure included measurement of the stack gas flow rate and temperature according to EPA Methods 1 and 2. Total hydrocarbon and oxygen determinations were made by the FFCC CEMS and Method 3A. Stack gas moisture was sampled in accordance with EPA Method 4.

2.1.4.3 Volatile Organic Sampling Train (Method 0030)

A volatile organic sampling train (VOST) was used during the CPT to collect the POHC chlorobenzene from the stack gas on sorbent resin. The VOST was configured in accordance with SW-846 Method 0030 with two Tenax® resin tubes and one Anasorb® tube in series. Because the VOST was a non-isokinetic sampling train, it could share a sampling port with any of the isokinetic sampling trains with no impact to the operation of either sampling train. The VOST was operated concurrently with the other sampling trains to collect a total of four sets of VOST cartridges for each test run. Three sets were targeted for analysis. The fourth set served as a backup in the event of tube hreakage or damage during sbipment and laboratory handling. About 20 liters of stack gas were sampled per set of VOST cartridges at 0.5 liters per minute for 40 minutes (slow-VOST conditions). The VOST cartridges were capped immediately upon removal from the train, wrapped in aluminum foil, placed in glass tubes, and sealed. The Method 26A stack parameter measurements were used for the VOST calculations.

2.1.4.4 Dioxin/Furan Sampling Train (Method 23A)

FFCC demonstrated compliance with the dioxin and furan standard through the use of data "in lieu of" testing. The data used by FFCC and documentation of its quality was provided in Appendix F of the FFCC CPT Plan.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page27 of 49

2,2 <u>Analytical Program</u>

Maxxam Analytics, Inc. is located in Mississauga, Ontario. This laboratory was used to perform the analytical testing on both the process feed samples and the stack gas samples collected during the CPT.

2.2.1 <u>Analytical Methods</u>

Analytical methods and procedures are summarized in Table 21.0. Detailed analytical methods and procedures can be found in Attachment A. Quality Assurance evaluation of methods used are discussed in section 4.0.

2.2.2 <u>Analytical Results</u>

A summary of analytical results are located in Table 5.0 and complete analytical reports can be seen in Attachment A.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page28 of 49

3.0 Description of the Test Program

This section summarizes the comprehensive performance test. The CPT plan contains the detailed plans of the test that was conducted June 2 - 4, 2010. The Executive Summary in this NOC discusses those deviations from the plan that were necessary to address unexpected conditions.

3.1 <u>CPT Objectives</u>

The overall objective of this Comprehensive Performance Test is to demonstrate that the boilers are capable of meeting the emission standards for solid fuel-fired boilers that burn hazardous waste as established in the MACT Combustion Rule (40 CFR 63 Subpart EEE) and to establish operating conditions and parameters that are equivalent with that demonstration.

The specific objectives for the FFCC comprehensive performance test are listed below. Each emissions parameter was reported with concentration corrected to 7% Oxygen.

- Demonstrate 99.99 percent DRE of the designated POHC (chlorohenzene).
- Demonstrate control of particulate emissions to less than or equal to 68 milligram/dry standard cubic meter (mg/dscm) at maximum ash feed rates.
- Demonstrate that hydrogen chloride/chlorine emissions are less than or equal to than the health-based compliance alternative established in FFCC's eligibility demonstration.
- Demonstrate that mercury emissions are less than or equal to 11 micrograms per dry standard cubic meter (ug/dscm).
- Demonstrate that semi-volatile metal (cadmium and lead) emissions are less than or equal to 180 micrograms per dry standard cubic meter (ug/dscm at maximum semi-volatile metal feed rates.
- Demonstrate that low-volatile metal (arsenic, beryllium, and chromium) emissions are less than or equal to 380 micrograms per dry standard cubic meter (ug/dscm) at maximum low-volatile metal feed rates.
- Demonstrate that total hydrocarbon emissions are less than 10 ppmv (dry) on an hourly rolling average basis.

- Meet THC limits to demonstrate that dioxin/furan emissions are less than or equal to 0.20 nanograms total equivalent quotient per dry standard cubic meter (ng TEQ/dscm).
- Gather data regarding waste feed characteristics and process operating conditions to be used to develop operational permit limits that will ensure compliance with regulatory performance standard.

3.2 <u>CPT Approach</u>

The proposed CPT is hased upon 40 CFR 63 Subpart EEE. 40 CFR 63.1207 establishes the requirements for conducting a comprehensive performance test. The universal approach establishes one set of permit conditions or limits applicable to all modes of operation. This approach, as proposed in the following section, will allow FFCC to treat the complete variety of hazardous wastes produced by the FFCC facility under one well-defined set of operating limits.

3.2.1 POHC Selection

FFCC has based its selection of POHC on two primary factors: (1) the University of Dayton thermal stability ranking, and (2) the composition of the actual wastes to be burned. Consistent with these factors, FFCC used the following criteria to select the POHC for this CPT:

- The POHC should be present as an Appendix VIII constituent in the actual wastes to the maximum extent practicable,
- The POHC should be considered an organic hazardous air pollutant under 42 U.S.C. 7412(b)(1), and
- The POHC should have a high ranking on the University of Dayton thermal stability-ranking list.

The constituents and hazardous air pollutants identified at 42 U.S.C. 7412(b)(1) that are likely to be found in FFCC's liquid waste based on testing and process knowledge are listed in Table-2.0 along with the historical ranges of these feed streams. Many of these compounds are included on the University of Dayton thermal stahility ranking including acetonitrile, henzene, chlorobenzene, and toluene. Chlorobenzene was selected as the POHC because it is a Class I compound, it is typically present in greater quantities in the waste than the other Class 1 compounds, and it is readily available.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page30 of 49

3.3 <u>CPT Program</u>

The CPT program consisted of one test of six runs. Three of the runs are identified as Test 1 and three of the runs are identified as Test 2. The target operating conditions from both tests were designed to establish operating limits for the boiler that would ensure all MACT combustion standards were met for one hazardous waste operating mode. Pursuant to Alternative Monitoring Request No. 1, all testing was done on Boiler No. 3 with data from this test to be used in lieu of testing Boilers Nos. 1 and 2 (as previously approved via letter from U.S. EPA Region 6 dated July 2, 1992). Boiler 3 was chosen because it affords better accessibility to the unit by the stack testing contractor. Alternative Monitoring Request No. 1 was provided as Appendix A of the FFCC CPT Plan.

3.3.1 Comprehensive Performance Test 1

The Test I operating conditions were designed to demonstrate worst case conditions for POHC destruction at the lower range of normal by testing the boiler's performance under minimum combustion chamber temperatures. The liquid hazardous waste feed was not maximized in order to keep operating temperatures less than typical operations. Specifically, liquid hazardous waste feed rates were not maximized during the low temperature test conditions (as opposed the approach typically used in incinerator performance tests) because the boilers can not operate at reduced temperatures when the waste feed rate is maximized. The operating temperature of the boiler is in direct relationship with the BTU input of the fuel. In addition, liquid hazardous waste feed rate cannot be maximized while the feed rate of coal is minimized hecause this condition would compromise the Bevill requirement to burn at least 50 percent coal when burning hazardous waste.

The Test 1 waste was a blend of acetic acid spiked with chlorobenzene (4 - 5 % of hazardous) waste feed.) The justification for this waste feed selection has already been presented.

The following operating conditions were established during Test 1:

- Minimum Furnace Temperature
- Maximum % O2

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page31 of 49

3.3.2 <u>Comprehensive Performance Test 2</u>

Test 2 is designed to establish permit conditions for minimum residence time which occurs at maximum feed and flue gas flow rates. DRE was measured to demonstrate that DRE is not affected at combustion gas residence times that are less than those demonstrated in Test 1 (i.e. the stack gas flowrate will be greater due to higher temperatures, thus leading to shorter residence times in the combustion chamber). Test 2 was also used to demonstrate compliance with the MACT EEE standards for solid-fuel fired boilers except low combustion chamber temperature.

The following operating conditions were demonstrated during Test 2:

- Maximum metals feed rates
- Maximum chlorine feed rates (as an operating parameter limit for SVM)
- Maximum liquid waste feed rates
- Maximum ash feed rate
- Maximum ESP inlet temperature
- Minimum Power to the ESP
- Maximum combustion air flow rate

Test 2 represented worst-case conditions for metals removal efficiency. FFCC will maximize the liquid hazardous waste feed rate during Test 2 in order to maximize steam production, which maximizes combustion chamber temperature and flue gas flow rate. The waste used during the minimum residence time test was a solution (i.e., acetic acid, methanol, or other solvent) spiked with chlorobenzene (4% of total waste feed) and perchloroethylene (6% of total waste feed). The justification for this waste feed selection has already been presented.

3.4 <u>CPT Sampling and Analytical Program</u>

The CPT sampling and analysis program is summarized in Table 21.0 and can be seen in the Alliance Source Testing Report found in Attachment A. The structure of this CPT was based on the previously stated objectives.

The stack was sampled during the CPT as follows:

- The volatile POHC chlorobenzene was sampled using the VOST train (SW-846 Method 0030).
- Metals using multi-metals train (40 CFR Part 60, EPA Method 29).
- Particulate matter using Method 5 procedures in conjunction with the multimetals train (40 CFR Part 60, Method 0029
- Total Hydrocarbon (THC) testing using in-house CEM (40 CFR Part 60, Method 25A).
- Total Oxygen (O2) testing using in-house CEM and Contractor Method 3A)

In addition to stack sampling, the feed materials (coal and liquid wastes) were sampled and analyzed for a number of physical and chemical parameters.

4.0 <u>Comparison of Actual Test Conditions versus Planned Conditions and</u> <u>Independent QA Review</u>

A comparison of the actual test conditions experienced during the CPT with the planned (target) test conditions documented in the CPT plan is provided below. Tables 8 through 15 provide a summary of the various operating parameter data with respect to the target conditions for each run. A detailed discussion of the operating parameter limits and the run-by-run data can be found in Section 6.0 of this report.

Comparisons were made on three sets of parameters:

- Combustion Device Parameters
- Feed Rate Parameters
- Constituent Feed Rate Parameters
- Stack Gas Parameters

4.1 <u>Combustion Device Parameters</u>

These parameters were used to establish limits for ensuring proper combustion and pollution control device operation. They include:

- Combustion chamber temperature In general, the average combustion chamber temperature exceeded the target by approximately 10 °F.
- Combustion Air Flow Rate The combustion air flow rate is a fairly new parameter. The flue gas flow rate was measured at the stack was consistent with the correlation equation discussed in Attachment E-5.
- ESP Power The average kW of the ESP was in line with the target value of 8 kW. Just prior to Test 2, the CPT manager revised the original target from 7 kW to 8 kW.
- ESP Inlet Temperature The average temperature to the inlet of the dry pollution control device was in line with the target temperature.
- Percent O2 The Percent O2 measured in the stack was in line with expectations.
- Total Hydrocarbons The THC concentration was in line with expectations.

• Atomization Pressure - The atomization pressure limit of 30 psig is based on manufacturer's recommendations.

4.2 Feed Rate Parameters

These parameters were used to establish constituent and POHC feed rates. They include:

- Total Waste Feed Rate The waste feed rate during the CPT was in line with the target value.
- Total Coal Feed Rate The coal feed rate is based on steam demand and waste feed characteristics. The process control logic determines how much coal is needed to make the steam production rate. The system did not need as much coal as estimated to meet the steam demand during the high temperature portion of the test.

4.3 <u>Constituent Feed Rate Parameters</u>

These parameters were calculated from the fuel feed rate parameters and the constituent concentrations in the fuel feed for the CPT Test. They include:

- Total Chloride Feed Rate In general, the total chloride feed rate was 17% lower than the target value.
- Total Ash Feed Rate In general, the total ash feed rate was 12% lower than the target value.
- Total Mercury Feed Rate Mercury was not detected in the waste feed. The mercury in the coal was well below the normal concentration seen in coal.
- Total Semi-Volatile Metal Feed Rate In general, the total SVM feed rate was in line with target value.
- Total Low-Volatile Metal Feed Rate In general, the LVM feed rate was in line with the target value.

4.4 <u>Stack Gas Parameters</u>

These parameters were monitored by a continuous emission monitor. They include:

- Total Hydrocarbon Concentration Emissions were well below the standard as expected.
- Oxygen Concentration Oxygen concentrations were consistent with expectations.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page35 of 49

4.5 Independent QA Review

Mr. David A. Weeks, P.E., BCEE, CIH (Risk Management and Engineering, Ltd.) provided independent oversight of the CPT and conducted an independent review of the quality assurance and quality control (QA/QC) performed by the stack testing company and laboratories during the test.

4.5.1 Stack Testing Equipment

Mr. Weeks was present during the entire CPT and inspected the stack sampling calibration records including the dry gas meter calibration records, pitot tube calibration records, and temperature indicator calibration records at the time of the test. Mr. Weeks observed the leak checks of the sampling equipment during the tests and found them to be in good order. The post calibration records were in order and showed that the stack gas sampling equipment functioned properly and provided data tbat were in accordance with the U.S. EPA methods used to collect tbe data.

4.5.2 Stack Gas - Particulate Analysis

Mr. Weeks conducted an independent review of the analytical data. The balance calibration data is summarized on page 339 of Attachment A. The results showed no deviation at a standard weight of 0.0000 grams, and deviations ranging from 0 to 0.0002 grams at a standard weight of 3,000 gram. The maximum deviation is less than 0.5 mg. Mr. Weeks' independent review of the data confirms the results reported by Alliance in their report.

4.5.3 <u>Stack Gas - Metals Analysis</u>

Mr. Weeks conducted an independent review of the analytical data. The stack sampling contractor's QA/QC summary is contained on pages 14 and 15 of Attachment A. The laboratory QC data is contained on pages 319 - 323 of Appendix C. The matrix spike and matrix spike duplicate recoveries were within the 70 - 130 % accuracy objective as defined in the QAPP. Matrix spikes were performed for mercury witb recoveries that ranged from 93 to 103%, showing that metals were not lost during the digestion process. The recoveries for semi- and low-volatile metals ranged from 95 to 106%. The RPD between the matrix spike and matrix spike duplicate analyses were less than the 25% objective established in the QAPP, with results ranging between 0.3 to 3% RPD for all metals. Thus, laboratory analyses were not biased high or low. Mr. Weeks' independent review of the data confirms the results reported by Alliance.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page36 of 49

4.5.4 Stack Gas - Volatile Organics Analysis

Mr. Weeks conducted an independent review of the analytical data. The laboratory's QC discussion is contained on pages 72 and 73 of Attachment A, the surrogate spike recoveries are reported with the analytical results on pages 63 - 65. Alliance reported, and Mr. Weeks confirmed by inspection of the laboratory data sheets, that surrogate recoveries ranged from 81 to 110 percent as compared to the target objective of 50 - 150%. The surrogate recovery for the spiked blanks and method blanks 99 to 102%. The RSD for the recoveries of the surrogate toluene-d8 ranged from 4 to 13 % as compared to the QAPP target of less than 35% RSD. Mr. Weeks' independent review of the data confirms the results reported by Alliance.

4.5.5 Stack Gas – Continuous Emissions Monitoring Data

Mr. Weeks conducted an independent review of the calibration data for continuous emissions monitoring (CEM) data contained in the Alliance report. The Alliance data was used to measure the oxygen and carbon monoxide concentration in the stack gas for the purpose of emission calculations. The calibration data is contained on pages 638 to 643. All of the calibration measurements for the oxygen analyzer were less than 2% of the span value, which is consistent with the calibration requirements contained in Section 13 of Method 3.

Mr. Weeks conducted an independent review of the calibration data for continuous emissions monitoring (CEM) data for the FFCC CEMs. The FFCC data was used to measure the oxygen and THC concentration in the stack gas for the purpose of THC emission calculations. For the FFCC THC analyzer, the mean difference between the CEMS and reference values for the daily calibrations ranged between 0.1 and 0.6 ppm for all runs, which is less than the 5 ppm requirement of Performance Specification 8A. The O2 daily calibrations reported a difference that ranged from 0.1 to 0.3 ppm for all runs, which is less than 2% of the span value, which is consistent with the calibration requirements contained in Section 13 of Method 3. The problem which resulted in a status of "out of control" on June 3 was corrected before the CPT runs were started.

The data from the April 14, 2010 RATA was also reviewed. The RATA data showed that the THC CEMs demonstrated a 3% difference between the reference method data and the FFCC CEMs, which is within the performance criterion of $\leq 10\%$. The O2 CEMs demonstrated a 0.1% difference, which is less than the performance criterion of $\pm 1\%$.

4.5.6 <u>Waste-Volatile Organics</u>

The QAPP does not identify specific QA/QC objectives for the wastes and spiking materials. However, the laboratory reported the results of the QC measurements collected during the analysis of these samples. Mr. Weeks reviewed this data in order to ensure the overall quality of the project. The laboratory's QC report for the waste analysis is contained on page 586 of Attachment A. The spiked hlank recovery for the organic waste was 108 %. This is consistent with standard QC limits of 70 - 130%.

Finally, the lahoratory reported that it did not analyze the waste feed samples for volatile organics within the 14-day holding time because of the need to dilute the samples. As a result, it might be argued that the volatile organic analytical results for the waste feed samples are biased low. If this is the case, the DRE calculations presented in this CPT report are more conservative than actually reported because a lower mass of POHC into the combustion system hiases the DRE calculation on the low side. Hence, the DRE values provided in the report may be greater than was reported.

4.5.7 <u>Feed-Metals and Chloride</u>

The QAPP does not identify specific QA/QC objectives for the wastes and spiking materials. However, the laboratory reported the results of the QC measurements collected during the analysis of these samples. Mr. Weeks reviewed this data in order to ensure the overall quality of the project data. The laboratory's QC discussion for semi- and low-volatile metals is contained on page 586; and page 593 provides the quality assurance discussion related to chloride and mercury.

The matrix spike recovery for semi- and low-volatile metals in the waste samples ranged between 102 to 108%; and the laboratory's QC standard recoveries ranged between 79 and 92%. Both QC measurements are consistent with the laboratory's recovery limits of 80 - 120. The mercury recoveries were 79 and 92% for the QC standard and the QC standard duplicate. The RPD between the standard and its duplicate was 15.4% compared to the laboratory criteria of 35%. The chloride QC standard recovery was 95% compared to the laboratory performance criterion of 90 - 100%, and the spiked blank recovery was 100% compared to the laboratory performance performance criterion of 80 - 120%.

Thus, the waste feed analytical results are not biased either high or low for metals and chloride.

4.5.8 <u>Feed – General Chemistry</u>

The analysis of certain process parameters such as heating value and ash do not have any specific QC requirements. Field duplicate samples were collected in order to evaluate the quality associated with this data. For higher heating value, the difference between the sample and its duplicate ranged between 0.02 and 0.3% for both the waste and coal. For ash content, the difference between the sample and its duplicate for coal was 4.9%. For the waste, the difference was found to be 50% (0.04% ash compared to 0.02% ash). However, the total ash content of the waste was very low (0.01 to 0.04% ash for all samples in all runs). Because the ash content of the waste is very low, the percent difference value of 50% does not materially affect the overall conclusions of this report.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page38 of 49

4.5.9 Analytical Method Deviations

A review of the analytical reports provided in Attachment A revealed that the laboratory used methods that were equivalent to those identified in the CPT Plan, but were different in name. A summary of the differences and impacts follows.

CPT PLAN METHOD	METHOD USED	DISCUSSION OF IMPACTS
S.G. of Waste: ASTM D-891 or other approved method	ASTM D-1298	None: the two methods provide similar data.
Ash Content of Coal: ASTM D- 482 or other approved method	ASTM D-3174	None; the method used by the laboratory is more appropriate for the matrix that was analyzed.
Chlorine Content of Waste and Coal: SW-846 9076 or other approved method	ASTM E776 / SW-846 9056	None: the two methods provide similar data.
Heat Content of Coal: ASTM-D- 240 or other approved method	ASTM D-5865	None; the method used by the laboratory is more appropriate for the matrix that was analyzed.
LVM and SVM Concentration in Waste: ICP-AES (Method 200.7 or 200.8)	SW-846 6020	None: the two methods provide similar data.
LVM and SVM Concentration in Coal: ICP-AES (Method 200.7 or 200.8)	SW-846 6010B	None: the two methods provide similar data.
Mercury Concentration in Waste and Coal: EPA Method SW- 7470A	ASTM D-3684-01	None: the two methods use the same technique and provide similar data.
Stack Gas Metals: SW-846 6010b and 7470a	SW-846 6020 and 7470a	None; the two methods provide similar data.

Specific Gravity of Waste

The two test methods are ASTM D891 - 09 Standard Test Methods for Specific Gravity, Apparent, of Liquid Industrial Chemicals; and ASTM D1298 - 99(2005) Standard Test Method for Density, Relative Density (Specific Gravity), or API Gravity of Crude Petroleum and Liquid Petroleum Products by Hydrometer Method. The two methods provide similar data for liquid substances in an organic matrix. No significant difference in results is expected between the two methods.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page39 of 49

Ash Content of Coal

The two test methods are ASTM D482 - 07 Standard Test Method for Ash from Petroleum Products; and ASTM D3174 - 04 Standard Test Method for Ash in the Analysis Sample of Coal and Coke from Coal. The method used by the laboratory is more appropriate for the sample matrix that was analyzed, which is coal.

Chlorine Content of Waste and Coal

The CPT Plan specified SW-846 Method 9076 - Test Method for Total Chlorine in New and Used Petroleum Products by Oxidative Combustion and Microcoulometry. The laboratory used ASTM E776 - 87(2009) Standard Test Method for Forms of Chlorine in Refuse-Derived Fuel, followed by SW-846 Method 9056A - Determination of Inorganic Anions by Ion Chromatography. The ASTM method is essentially an extraction method. The two analytical methods are essentially equivalent. This issue is not substantive because SW-846 Method 9057 is also an EPA approved method.

EPA SW-846 Method 9057 describes the use of liquid anion chromatography to separate the chloride anions from the sample matrix. The chloride ions are then detected by a conductivity detector that uses coulometry to measure the flow of electrons between two electrodes.

EPA SW-846 Method 9076 describes a process in which the material is heated, and the resulting gas is reacted with silver cations to form silver chloride. The chloride ions are determined quantitatively by the use of coulometry based on the stoichiometry of the reaction and the flow of silver cations that is detected by the electrodes. The detection method is the essentially the same between the two methods; it is the separation mechanism that is different.

SW-846 Method 9057 has a within-laboratory relative standard deviation of 6.2 percent and 3.2 percent at HCl concentrations of 3.9 and 15.3 ppm, respectively. Similar statistics are not provided for Method 9076, but the statistics that are provided in SW-846 state that the method is biased low by up to 9% for relatively low concentrations, reproducibility in the range of 1,400 ppm, and repeatability in the range of 400 ppm for higher concentrations. It is the conclusion of the independent QA reviewer that the two methods provide equal performance for chloride concentrations that are in the range of 50,000 ppm; such as this case.

Heat Content of Coal

The CPT Plan specified the use of ASTM D240 - 09 Standard Test Method for Heat of Combustion of Liquid Hydrocarbon Fuels by Bomb Calorimeter, whereas the laboratory used ASTM D5865 - 10a Standard Test Method for Gross Calorific Value of Coal and Coke. The method used by the laboratory is more appropriate for analysis of coal than the method specified in the Plan.

Boiler Notice of	Compliance
Revision No.:	0
Revision:	9/1/10
Page	40 of 49

LVM and SVM Concentration in Waste and Coal

The CPT Plan specified either EPA Method 200.7-Determination of Metals and Trace Elements in Water and Wastes by Inductively Coupled Plasma-Atomic Spectrometry, or EPA Method 200.8-Determination of Trace Elements in Waters and Wastes by ICP-MS.

The laboratory used SW-846 6010C-Inductively Coupled Plasma-Atomic Emission Spectrometry for the coal, and SW-846 6020A- Inductively Coupled Plasma-Mass Spectrometry for the waste. EPA Method 200.7 is essentially equivalent to SW-846 Method 6010 except that EPA Method 200.7 is more commonly used for aqueous materials (although it can also be used for waste materials.) EPA Method 200.8 is essentially equivalent to SW-846 Method 6020 although is more commonly used for aqueous materials. All of these methods provide equivalent results for metals in the range of concentrations that are indicative of the waste and coal.

Mercury Concentration in Waste and Coal

The CPT Plan specified EPA Method SW-846 Method 7470A - Mercury in Liquid Waste (Manual Cold-Vapor Atomic Absorption Technique) whereas the laboratory used ASTM D3684 - 01(2006) Standard Test Method for Total Mercury in Coal by the Oxygen Bomb Combustion/Atomic Absorption Method. Both methods are essentially equivalent because they both use cold vapor atomic absorption for the analysis.

Stack Gas Metals

The CPT Plan specified SW-846 6010C-Inductively Coupled Plasma-Atomic Emission Spectrometry whereas the laboratory used SW-846 6020A- Inductively Coupled Plasma-Mass Spectrometry. SW-846 Method 6020 has lower detection limits. However, detection limits were not an issue for this project and since Method 6020 has lower detection limits, it provides equivalent data to the specified method.

4.5.10 Conclusion

The Independent QA Review determined that the sampling and analysis was completed in accordance with the data quality objectives established for the project. Mr. Weeks opined that the data is of sufficient quality to demonstrate compliance with the standards and develop operating parameter limits.

5.0 Comparison of Test Results with Regulatory Compliance Limits

This section compares the test results of the CPT, conducted June 1 - 4, 2010, with regulatory compliance limits established by the MACT Combustion Rule for Solid Fuel-Fired boilers. Table 16.0 demonstrates that FFCC hazardous waste boiler is in full compliance with the emission standards of 40 CFR 63, Subpart EEE.

5.1 <u>Regulatory Requirements for the CPT</u>

The regulatory requirements to discuss from the CPT are:

- Destruction Removal Efficiency (DRE)
- Particulate Emissions
- HCl / Cl₂ Emissions
- Mercury Emissions
- Semi-Volatile Metals Emissions
- Low-Volatile Metals Emissions
- Dioxin/Furan Emissions
- Total Hydrocarbon Emissions

5.1.1 Destruction Removal Efficiency

40 CFR 63.1216(c) requires solid fuel boilers to achieve a destruction removal efficiency (DRE) of 99.99% during the CPT for each principal organic hazardous constituent (POHC) designated in the waste feed. FFCC was able to make this demonstration during the CPT. The average DRE measured during Test 1 was >99.9991%. The average DRE measured during Test 2 was >99.9995%.

5.1.2 Particulate Emissions

40 CFR 63.1216(a)(7) requires existing solid fuel boilers to not emit particulate matter in excess of 68 mg/dscm after correction to a stack gas concentration of 7% oxygen. FFCC's boiler emitted an average of 26.2 mg/dscm at 7% oxygen during Test 2.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page42 of 49

5.1.3 HCl / Cl₂ Emissions

The hydrogen chloride/chlorine feed rate during the CPT was less than the health-based compliance alternative feed rate limit for total chlorine developed under the procedures prescribed in 40 CFR 63.1215. Appendix F of the CPT Plan calculates the HCl/Cl2 alternative feed rate limit as 1886.8 lb/hr. An average feed rate equivalent to 508 lb/hr HCl was demonstrated during the CPT.

5.1.4 Mercury Emissions

40 CFR 63.1216 (a)(2) requires existing solid fuel boilers to not emit mercury in excess of 11 micrograms per dry standard cubic meter (ug/dscm) after correction to a stack gas concentration of 7% oxygen. The average mercury emission during the CPT was 3.9 ug/dscm at 7% oxygen.

5.1.5 Semi-Volatile Metals Emissions

40 CFR 63.1216(a)(3) requires existing solid fuel boilers to not emit semi-volatile metals in excess of 180 micrograms per dry standard cubic meter (ug/dscm) after correction to a stack gas concentration of 7% oxygen. FFCC's boiler emitted an average of 158.8 ug/dscm at 7% oxygen during the CPT.

5.1.6 Low-Volatile Metals Emissions

40 CFR 63.1203(a)(4) requires existing solid fuel boilers to not emit low-volatile metals in excess of 380 micrograms per dry standard cubic meter (ug/dscm) after correction to a stack gas concentration of 7% oxygen. FFCC's boiler emitted an average of 117.3 ug/dscm at 7% oxygen during the CPT.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page43 of 49

5.1.7 Dioxin/Furan Emissions

40 CFR 63.1216(a)(1)(ii) requires existing solid fuel boilers to not emit total hydrocarbons (THC) in excess of 10 parts per million by volume (ppmv) after correction to a stack gas concentration of 7% oxygen. FFCC's boiler emitted an average 0.4 ppmv at 7% oxygen during the Test 1 of the CPT, and emitted an average 0.3 ppmv at 7% oxygen during the Test 2 of the CPT. FFCC submitted data in lieu of Dioxin/Furan testing in Appendix F of the CPT Plan. This data in lieu of was obtained from a 1999 test were Dioxin/Furan emissions from FFCC boilers were 0.092 ng TEQ/dscm at 7% oxygen.

5.1.8 Total Hydrocarbon Emissions

40 CFR 63.1216(a)(5)(ii) requires existing solid fuel boilers to not emit total bydrocarbons (THC) in excess of 10 parts per million by volume (ppmv) after correction to a stack gas concentration of 7% oxygen. FFCC's boiler emitted an average 0.4 ppmv at 7% oxygen during the Test 1 of the CPT, and emitted an average 0.3 ppmv at 7% oxygen during the Test 2 of the CPT.

6.0 Procedures and Limitations for Operating Parameters

This section describes the operating parameter limits (OPL) and bow they were established. This section also describes waste, chloride, ash and metal feed rate limits and how they are determined, including extrapolation.

6.1 Waste Feed Rate Limitations

The waste feed rate limits are summarized in Table 18.0. All feed rate limits are 12-hour HRAs. Tables 8 - 15 summarize the waste feed rate data by run. The process data upon which Tables 8 - 15 are based can be found in Attachment E-5.

6.1.1 Maximum Hazardous Waste Feed Rate Limit

FFCC established a maximum hazardous waste feed rate limit based on the average of the test run averages from Test 2.

6.1.2 <u>Maximum Chloride Feed Rate Limit</u>

The maximum chloride feed rate limit for LVM and SVM control was determined as average of the test run averages from Test 2. The concentration of chloride in both the waste and coal were used to determine the limit.

6.1.3 <u>Maximum Ash Feed Rate Limit</u>

The maximum ash feed rate limit was determined as average of the test run averages from Test 2. The amount of ash spiked (diatomaceous earth) and the ash content of the coal and waste was used to determine the limit.

6.1.4 Metals Feed Rate Limits

Metals feed rate limits were also determined during the CPT. A combination of methods was used to develop the metal feed rate limits. The final feed rate limits determined from the CPT are summarized in Table 18.0.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page45 of 49

6.1.4.1 Maximum Mercury Feed Rate

The maximum mercury feed rate limit is hased on the maximum theoretical emission concentration (MTEC). MTEC was chosen to develop this feed rate limit because mercury was not spiked into the system, and the concentration of mercury in the coal was significantly lower than anticipated hased on historical analysis.

To develop the MTEC feed rate limit, FFCC used the combustion air correlation curve to determine the stack gas flow rate at the maximum comhustion air flow rate demonstrated during the CPT. The MACT EEE mercury emission standard was used to calculate the maximum mercury feed rate limit. FFCC used only 90% of the feed rate calculated using MTEC in order to establish the maximum feed rate limit. This calculation is shown in Table 20.0.

6.1.4.2 Maximum Semi-Volatile Metals Feed Rate

The maximum SVM feed rate limit was determined using the average of the test run averages from Test 2. The amount of SVM spiked and the SVM content of the coal and waste was used to determine the limit. The limit was further extrapolated as described in Section 6.2.3.4 and shown in Table 19.0.

6.1.4.3 Maximum Low Volatile Metals Feed Rate

The maximum LVM feed rate limit was determined using the average of the test run averages from Test 2. The amount of LVM spiked and the LVM content of the coal and waste was used to determine the limit. The limit was further extrapolated as described in Section 6.2.3.4 and shown in Table 19.0.

6.1.4.4 Metals Feed Rate Extrapolation

FFCC used the analytical results from the process feed analysis to extrapolate up to higher metal feed rate limits. The extrapolation was minimized by using only 90% of the MACT Standard.

The extrapolation was done by a simple algebraic calculation:

Metal feed – CPT		Metal feed – Permit
	=	
Emission level - CPT		MACT Standard

• Metal feed – CPT refers to the metals feed rate derived from Table 9.0.

- Emission level CPT: Refers to the average measured concentration, in ug/dsem corrected to 7% oxygen, of the metals from the three corresponding stack test runs of Test 2. This can be found in Table 16.0.
- Metal feed Permit: This will be the calculated feed limit for the metals.
- MACT Standard: These are 180 ug/dscm for LVM and 380 ug/dscm for SVM at 7% oxygen.

The extrapolated metal feed rate calculations are shown below:

SVM feed rate (lb/hr) = $[(0.633 \text{ lb/hr SVM}) \times (180 \text{ ug/dsem SVM}) \times (0.90)] / [158.8 \text{ ug/dsem}]$

Maximum SVM feed rate (lb/hr) = 0.65 lb/hr

LVM feed rate (lb/hr) = $[(1.58 \text{ lb/hr LVM}) \times (380 \text{ ug/dsem LVM}) \times (0.90)] / [117.3 \text{ ug/dsem LVM}]$

Maximum LVM feed rate (lb/hr) = 4.61 lb/hr

6.2 <u>Combustion Chamber Limitations</u>

The operating parameter limits are summarized in Table 18.0. Tables 8 - 15 summarizes the OPL data by run. The limits are calculated as the average of the test run average measurements collected from each of the process monitoring instruments. The limits were established as either maximums or minimums depending on the parameter. The minimum temperature limit and maximum percent O2 was established from the Test I data. All other combustion chamber limits were established from the Test 2 data. The process data from each run are provided in Attachment E-5.

6.2.1 Minimum Combustion Chamber Temperature Limit

FFCC established a minimum combustion chamber temperature of 1,061 deg F. during the CPT while demonstrating satisfactory DRE. The minimum combustion chamber temperature was established as the average of the test run averages from Test 1.

6.2.2 Minimum Atomization Pressure Limit

The minimum atomization pressure is based on the manufacturer's specification of 30 psig as an instantaneous limit for the liquid hazardous waste feed system.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page47 of 49

6.2.3 <u>Maximum Furnace Pressure Limit</u>

The boiler system is an induced draft system that operates under a constant negative pressure. A limit of 0 psig will be established and will be monitored on an instantaneous basis.

6.2.4 Maximum Combustion Air Flow Rate Limit

A maximum combustion air flow rate of 21,130 scfm was established during Test 2. This limit was an average of the test run average total combustion air flow measured during the three runs of Test 2. It will he established as an hourly rolling average limit.

6.3 <u>Air Pollution Control Limitations</u>

The operating parameter limits are summarized in Table 18.0. Tables 8 - 15 summarizes the OPL data by run. The limits are calculated as the average of the test run average measurements collected from each of the process monitoring instruments during Test 2. The limits were established as either maximums or minimums depending on the parameter. The process data from each run are provided in Attachment E-5.

6.3.1 Minimum ESP Power Limit

FFCC established a minimum electrostatic precipitator (ESP) power limit of 8 kilowatts (kW). This limit was established as the average of the test run averages recorded during Test 2. This limit will be monitored as an hourly rolling average limit.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page48 of 49

6.3.2 Maximum Inlet Temperature to the ESP

FFCC established a maximum inlet temperature to the ESP of 515 deg F. This limit was established as the average of the test run averages recorded during Test 2. This limit will be monitored as an hourly rolling average limit.

6.4 <u>Stack Gas Limitations</u>

The only stack gas measurement data from which an OPL is derived is oxygen. (The THC OPL is a regulatory standard.) The stack gas OPL limits are summarized in Table 18.0. Tables 8 - 15 summarizes the OPL data by run. For O2, the limits are calculated as the average of the test run average measurements during Test 1. The process data from each run are provided in Attachment E-5.

6.4.1 <u>Maximum Total Hydrocarbon Limit</u>

FFCC demonstrated compliance with the total hydrocarbon (THC) limit standard of 10 ppmv established in the MACT EEE regulations. This limit will be continuously monitored as an hourly rolling average.

6.4.2 <u>Maximum Percent Oxygen Limit</u>

EPA Region 6 requested FFCC to establish a maximum percent oxygen limit in conjunction with their request to use combustion air flow rate as a surrogate for flue gas flow rate. This limit was set as 12.2 % oxygen in the stack gas during the CPT. The maximum percent oxygen in the stack gas limit was established as an average of the test run average percent oxygen recorded during the three runs of Test 1. This limit will be continuously monitored as an hourly rolling average.

Boiler Notice of ComplianceRevision No.:0Revision:9/1/10Page49 of 49

7.0 <u>Statement of Compliance</u>

Based on my inquiry of the person or persons who manage the system, or those persons directly responsible for gathering the information, the information submitted is, to be the best of my knowledge and belief, true, accurate, and complete.

The information contained in the notification establishes the operating controls and limits that ensure that the solid fuel boilers located at the FutureFuel Chemical Company in Batesville, Arkansas is in compliance with all the applicable emission standards of 40 CFR 63, Subpart EEE for Phase Π units.

Signed this 2nd day of September 2010.

Sam Dortch Executive Vice President, General Manager, FutureFuel Chemical Company

Boiler Notice of ComplianceRevision No.:0Revision:09/01/10Tables1 - 21

Tables

- Table 1.0 Feed Stream Analysis Data
- Table 2.0 Feed Stream Characteristics
- Table 3.0 Process Monitoring Instruments
- Table 4.0 Process Monitoring Instruments, Calibration, and Maintenance
- Table 5.0 CPT Process Feed Analytical Results
- Table 6.0 CPT Spiking Information
- Table 7.0 CPT Stack Gas Analytical Results
- Table 8.0 CPT Test 1 Averages
- Table 9.0 CPT Test 2 Averages
- Table 10.0 CPT Comparison of Planned Conditions to Actual Conditions Test 1, Run 1
- Table 11.9 CPT Comparison of Planned Conditions to Actual Conditions Test 1, Run 2
- Table 12.0 CPT Comparison of Planned Conditions to Actual Conditions Test 1, Run 3
- Table 13.0 CPT Comparison of Planned Conditions to Actual Conditions Test 2, Run 1
- Table 14.0- CPT Comparison of Planned Conditions to Actual Conditions Test 2, Run 2
- Table 15.0 CPT Comparison of Planned Conditions to Actual Conditions Test 2, Run 3
- Table 16.0 Demonstration of Compliance
- Table 17.0 Boiler Removal Efficiency
- Table 18.9 Summary of Operating Parameter Limits
- Table 19.0 Metal Extrapolation
- Table 20.0 Mercury MTEC Feed Rate Limit
- Table 21.0 Summary of Sampling and Analysis Program

Notification of Compliance Revision No.: 09/01/10 Revision Date: 09/01/10 Table 1, Page: 1 of 1

TABLE 1.0

FEED STREAM ANALYSIS DATA

	Average Hant Value	JM	ow-Volati etals (Av	le 3)	Semi-V Metals	olatile (Avg)	Mercury (Avg)		Other (A)	Metals (g)		Avg	Avg	Physical
Feed Stream *	(Btu/lb)	As ppm	Beppm	Cr ppm	Cd	Pb ppm	Hg ppm	Sb ppm	Ba ppm	Ag ppm	uudd LL,	%	₽%6	Form
Process Intermediate Waste	12,000	BDL	BDL	5	BDL	BDL	BDL	BDL	BDL	BDL	BDL	4	<1	Liquid
Fatty Acid Waste	12,000	BDL	BDI,	106	BDL	BDL	BDL	BDI,	BDL,	BDL	BDL.	0	<2	Liquid
Organic Process Waste	9,000	BDL.	BDL	2	BDI,	BDL	< 0.1	BDL	< 0.5	BDL	BDL.	< 1	< 3	Liquid
Spent Solvent Waste	11,000	BDL	BDL	1	BDL	1	BDI.	BDL	BDL	BDL	BDL	<1	<2	Liquid
DIPB Auxiliary Fuel	18,600	BDI,	BDI.	BDL	BDL	BDL	BDL.	BDL	BDL	BDL.	BDL	0	< ! >	pinbr1
Biomass Fuel	11,800	BDL	BDL	BDL	BDL	BDL	BDL.	BDL	BDL	BDL	BDL	0	e	Solid
Vapor Recovery Vent Gas	< 1000	BDL	BDL	BDL	BDL	BDL	BDL	BDL	BDL	BDI.	BDL	<0.02	0	Gas
WWT Sludge	724	0.13	BDL	16	0.4	2.5	BDL	BDL	FI	BDL	BDL	0	5.6	Semi- Solid
Coal	12.300	9	1	13	1	8	0,15	3	43		BDL	0.1	10	Solid

BDI, = Below Detection Limit

Notification of Compliance Revision No.: 09/01/10 Revision Date: 09/01/10 Table 2; Page: 1 of 3

TABLE 2.0

FEED STREAM CHARACTERISTICS

Feed Stream	Constituents and Other Characteristics	Avg %	Max %	Organic Hazardous	Air Pollutants	Quantity Burned per Year (Ibs)	Waste Codes
	Toluene	15	30	Acetonitrile	< 1 %	500,000 to	F003 D001,
	Water	00	15	Chlorobenzene	0 - 3%	2,500,000	D021
	Methanol	15	50	Phenol	0-1-0		
	Heptane	10	40	Triethylamine	< 1%		
	TXIB	10	20	Toluene	0-30 %		
Urganic Process waste	IDMAP	2	2	Xylene	() - 40%		
	Acetone	10	40	Fornaldehyde	<1%		
	Isopropanol	n	30	Methanol	0 - 50%		
	Xylene	10	40	MIBK	0 - 5%		
	MIJK	2	S	Ethyl Benzene	0 - 2%		
	Other Organics	r: -	20				
	Chlorobenzene	10	40	Acetonitrile	<1%	15,000,000 to	F002, F003,
	Water	90	20	Chlorobenzene	0 - 40%	20,000,000	F005, D001.
	Toluene	20	50	Triethylamine	< 1%		D002, D021
	Methanol	10	40	Toluene	0 - 50 %		
	IFT Product	3	9	Formaldehyde	<1%		
open solven waste	Tars from organic process	14	30	MIBK	0 - 5%		
	Acetic Acid	25	50	Methanol	0 - 40%		
	Triethylene Glycol	7	2	Xylene	0 - 5 %		
	o-Dichlorobenzene	5	20	Ethyl Benzene	0 - 1 %		
	Other Organics	8	20				
Notification of Compliance Revision No.: 09/01/10 Revision: 2 of 3

TABLE 2.0 (Continued)

FEED STREAM CHARACTERISTICS

Feed Stream	Constituents and Other Characteristics	Avg %	Max.	Organic Hazardous Air Pollutants	Quantity Burned per Year (lbs)	Waste Codes
ss Intermediate Waste	Chlorobenzene Acetone Isopropanol Heptane Xyhene Toluene Other organics	10 15 15 15 10 15 15 10 10	30 30 30 30 30 30 30 30 30 30 30 30 30 3	Chlorobenzene $0-30\%$ Toluene $0-30\%$ Xylene $0-30\%$	1,500,000 to 2,500,000	F002, F003, F005, D001, LX021
Acıd Waste	Nonanoic Acid Octanoic Acid TXIB	86 0	100 	None	1,000,000 to 1,500,000	D007
tass Fuels	Woody Biomass	100	100	None	0 to 100,000	None

Prepared by FutureFuel Chemical Company Batesville Arkansas, and Risk Management & Engineering, Ltd., Dallas, Texas

3 01 3 0 01/10/60 Notification of Compliance Revision No.: 09/01/10 Revision Date: 09/01/10 Table 2, Page: 3 of 3

TABLE 2.0 (Continued)

FEED STREAM CHARACTERISTICS

Feed Stream	Constituents and Other Characteristics	Avg %	Max. %	Organic Hazardous A	vir Pollutants	Quantuy Burned per Year (lbs)	Waste Codes
Vapor Recovery Vent Gas	Nitrogen (Inert N2 Gas) Acetone Methanol Water Chlorobenzene Heptane Toluene Acetic Acid Isopropanol Xylene	99.5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	99.9 0.1 0.08 0.05 0.04 0.04 0.02 0.02 0.02 0.02	Methanol Chlorobenzene Heptane Toluene Xylene MilBK Ethyl Benzene	0.062 % 0.026 % 0.014 % 0.014 % 0.004 % 0.001 % 0.0002 %	1,500,000 to 2,000,000	None
WWT Sludge	Water Organic Microorganisms Inorganic Solids	70 15 00	90 55	None		0 to 500,000	None
Coal	Bituminous Coal	100	100	None		70,000,000 to 100,000,000	None

Notes:

- П
- British Thermal Unit British Thermal Unit per pound Ш
- Parts per million Weight production Btu Btu/Ib ppm wt%
 - 11 11
- Only those organic HAPs that may be potentially present in the waste are reported in Table D-5.3. All other organic HAPs are not present based on onsite material use and process chemistry evaluations. .

Notification of Compliance Revision No.: 0 Revision Date: 09/01/10 Table 3, Page: 1 of 1

TABLE 3.0

PROCESS MONITORING INSTRUMENTS

1.1	Instrument Number(s)	Type of Instrument	Instrument Range	Expected Operating Range	Accuracy	Drawing Location Reference
	FT-6M01-100 FT-6M01-200 FT-6M01-300	Micromotion Flow Meter	0 – 100 fb/min	0 – 65 lb/min	+/- 2 °o rate	6M01-9T-031 6M01-9T-032 6M01-9T-089
PSI, PSI,	6M01-WCB-112A 6M01-WCB-124A 6M01-WCB-1164	Low Pressure Switch	Set at 40 psig (Falling)	50 - 90 psig	Repeatability < 2.5% Full scale	6M01-9T-031 6M01-9T-032 6M01-9T-089
	TE-6M01-9195 TE-6M01-9196 TE-6M01-9197	Type K Thermocouple	0 – 2200 degrees F	900 - 1500 degrees F	+/« 1 % Temperature	6M01-9T-024 6M01-9T-025 6M01-9T-026
	Al-PPTR-1 Al-PPTR-2 Al-PPTR-3	Neundorfer Precipitator Controls	0 – 50 kW	10 – 24 kW	+/- 5 %	6M01-9T-024 6M01-9T-025 6M01-9T-026
自自自	6M01-FAS-608 6M01-FAS-616 6M01-FAS-624	Type J Thermocouple	0-1000 degrees F	420 - 510 degrees F	+/- 1% of Temperature	6M01-9T-024 6M01-9T-025 6M01-9T-026
EE	6M01-FAS-713 6M01-FAS-714 6M01-FAS-715	Pressure Transmitter	-6 to 6 in. w.c.	-3 to 3 in. w.c.	°₀ [~/+	6M01-9T-024 6M01-9T-025 6M01-9T-026
2.2.3	AT-6M09-116 AT-6M09-216 AT-6M09-316	Rosemount Mod. 400A	0 – 100 ppm	0 – 10 ppm	+/~ 30°	6M01-9T-005 6M01-9T-026
	10 - 1 - 10 - 10 - 10 - 10 - 10 - 10 -	osemount Mod.	0 - 25 °°	$0 = 25 \ ^{0}0$	+/- 2°o	6M01-9T-005 6M01-9T-026

Prepared by FutureFuel Chemical Company Batesville Arkansas, and Risk Management & Engineering, Ltd., Dallas, Texas Notification of Compliance Revision No.: 09/01/10 Revision Date: 09/01/10 Table 4, Page: 1 of 1

TABLE 4.0

PROCESS MONITORING INSTRUMENTS, CALIBRATION, and MAINTENANCE

Parameter	Location	Instrument Number(s)	Inspection/ Calibration Procedure	Calibration Frequency	Preventive Maintenance Procedure	Preventive Maintenance Frequency
Liquid Waste Feed Rate	Liquid Waste Feed Flow Monitor	FT-6M01-100 FT-6M01-200 FT-6M01-300	Check against ISO 9000 Standard	Annual	Check against ISO 9000 Standard	Annual
Steam Atomization Pressure	Steam Line	PSL-6M01-WCB-112A PSL-6M01-WCB-124A PSL-6M01-WCB-124A	Operator reviews for inconsistency with other process parameters.	As песеssагу	Calibrate or Replace	Annual
Combustion Temperature	Boiler Combustion Temperature	TE-6M01-9195 TE-6M01-9196 TE-6M01-9197	Visual Inspection with a density test against known material.	Amual	Cheek Accuracy or Replace	Annual
ESP Power Input	3rd floor precipitator panel	Al- PPTR-1 Al-PPTR-2 Al-PPTR-3	Primary Current. Secondary Current and Voltage are checked for accuracy	Amual	Calibrate	Annual
ESP Inlet Temperature	Precipitator Inlet	TE-6M01-FAS-608 TE-6M01-FAS-616 TE-6M01-FAS-624	Visual Inspection with a density test against known material.	Annual	Cheek Accuracy or Replace	Annual
Air Flow Control	Forced Draft Fan Outlet	FT-6M01-FAS-713 FT-6M01-FAS-714 FT-6M01-FAS-715	Calibrate against a known standard	Amual	Check and blow clear the circular annular ring	Annual
Total Hydrocarbon	L.D. Fan Outlet Duet	AT-6M09-116 AT-6M09-216 AT-6M09-316	Calibrate with known calibration gas +/- 2 % acuracy	Daily	PM is on call From daily check clean and calibrate	Annual
Oxygen	L.D. Fan Outlet Duct	AT-6M09-118 AT-6M09-218 AT-6M09-318	Calibrate with known calibration gas +/- 2 % accuracy	Daily	PM is on call From daily check clean and calibrate	Annual

Notification of Compliance Revision No.: 0 Revision Date: 09 01/10 Table 5, Page: 1 of 1

	Pro	ocess Waste	Analytical	Results		
Constituent	Test 1 Run 1	Test 1 Run 2	Test 1 Run 3	Test 2 Run 1	Test 2 Run 2	Test 2 Run 3
Chlorobenzene (%)	4.83	5.05	4.89	4.38	4.74	4.86
Heating Value (btu/lb)	6460	6482	6622	6254	6319	6333
Specific Gravity	1.036	1.032	1.044	1.052	1.056	1.052
Chloride (%)	NA	NA	NA	5.4	5.4	5.5
Ash (%)	0.03	0.03	0.02	0.04	0.03	0.01
Mercury (ug/g)	NA	NA	NA	0.004	0.002	0.003
SVM (ug/g)	0.034	0.034	0.034	0.033	0.033	0.033
Cadmium	0.005	0.005	0.005	0.005	0.005	0.005
Lead	0.029	0.029	0.029	0.029	0.028	0.029
LVM (ug/g)	18.450	18.521	18.308	17.217	17.152	16.265
Arsenic	0.048	0.049	0.048	0.048	0.047	0.048
Beryllium	0.029	0.029	0.029	0.029	0.028	0.029
Chromium	18.372	18.443	18.232	17.141	17.076	16.188
	P	rimary Fuel	Analytical F	Results		
Constituent	Test 1 Run 1	Test 1 Run 2	Test 1 Run 3	Test 2 Run 1	Test 2 Run 2	Test 2 Run 3
Chlorobenzene (%)	0	0	0	0	0	0
Heating Value (btu/lb)	10890	11090	11070	10630	10630	10600
Specific Gravity	NA	NA	NA	NA	NA	NA
Chloride (%)	NA	NA	NA	0.08	0.08	0.07
Ash (%)	10.4	8.45	8.44	9.26	9.26	9.79
Mercury (ug/g)	0.063	0.037	0.042	0.048	0.048	0.040
SVM (ug/g)	NA	NA	NA	4.98	4.98	5.09
Cadmium	NA	NA	NA	1.01	1.01	1.28
Lead	NA	NA	NA	3.97	3.97	3.81
LVM (ug/g)	NA	NA	NA	23.663	23.663	22.432
Arsenic	NA	NA	NA	1.33	1.33	1.4
Beryllium	NA	NA	NA	0.933	0.933	0.832
Chromium	NA	NA	NA	21.4	21.4	20.2

Table 5.0 - CPT Process Feed Analytical Results

Notification of Compliance Revision No.: 0 Revision Date: 09/01/10 Table 6, Page: 1 of 1

Table 6.0 - CPT Spiking Information

Ash Sp	iking -	Test 2	
Test 2 (Run #)	Coal Added (Ibs)	Diatomaceous Earth Added (Ibs)	Ash Spike
Run 1	100000	7500	7.5%
Run 2	100000	7500	7.5%
Run 3	50000	3750	7.5%

	Metals Spik	ing Inform	lation - Tesh	12			Metals S	piking Infe	irmation -]	Test 2			Metals 5	piking Inf	ormation			
Run 1 Run 2 Run 2 Run 3 Run 3 </th <th>6/3/2010</th> <th></th> <th></th> <th></th> <th></th> <th></th> <th>6/3/2010</th> <th></th> <th></th> <th></th> <th></th> <th></th> <th>6/4/2010</th> <th></th> <th></th> <th></th> <th></th> <th></th>	6/3/2010						6/3/2010						6/4/2010					
	Run 1						Run 2				Spik	e Rate	Run 3					
	Time	Scale Weight	Change	SolutionAvg lb/hr	Avg Pb lb/hr	Avg Cr Ib/hr	Time	Scale Weight	Change	Solution Avg lb/hr	Pb lb/hr	Cr Ib/hr	Thme	Scale Weight	Change	Solution Avg lb/br	Pb lb/hr	Cr Ib/hr
	10:00	312.30		64.8	0.60	1,41	14:12	466.0		62.1	0.58	1.35	7:00	335.7		68.5	0.64	1.49
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	10:19	283.90	28,40				14:27	444.6	21.4				7:15	322.9	12,8			
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	10;34	251.30	32.60				14:42	431,5	13.1				7:30	308.1	14.8			
11:04 226.00 14.60 15:12 404.3 12.2 404.3 12.2 8:00 267.6 2 11:19 211.30 14.70 14.70 15:27 391.5 12.8 8:15 255.2 1 11:149 183.50 15.10 15:42 373.9 17.6 8:35 241.8 1 11:149 183.50 12.70 15:42 373.9 17.6 8:30 241.8 1 12:04 109.90 13.60 14.90 8:30 241.0 14.5 9:30 241.8 1	10:49	240.60	10.70				14:57	416.5	15.0				7:45	288.8	19.3			
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	11:04	226.00	14.60				15:12	404.3	12.2				8:00	267.6	21.2			
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	11:19	211,30	14.70				15:27	391,5	12.8				8:15	255.2	12.4			
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	11:34	196.20	15.10				15:42	373.9	17.6				8:30	241.8	13.4			
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	11:49	183.50	12.70				15:57	355.5	18.4				8:45	228.2	13.6			
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	12:04	169,90	13.60				16:12	341.0	14.5				9:00	211.1	17.1			
12:34 142.70 12.30 12.30 18.14 1 12:49 128.40 14.30 9:45 163.9 1 12:04 113.60 14.80 147.2 1	12:19	155,00	14.90				16:27	326.0	15.0				9:15	197.2	13.9			
12:49 128.40 14.30 9.45 163.9 1 13:04 113.60 14.80 14.80 147.2 1	12:34	142.70	12.30				16:42	310.7	15.3				9:30	181.4	15.8			
13:04 113:60 14:80 147.2 1	12:49	128,40	14.30									1	9:45	163.9	17.5			
	13:04	113.60	14.80										10:00	147.2	16.7			

Spike Solution Concentration = 0.933% Pb and 2.18% Cr

Prepared by FutureFuel Chemical Company, Batesville, Arkansas, and Risk Management Engineering, Ltd., Dalfas, Texas

Nofication of Compliance Revision No.: 0 Revision Date: 09-01/10 Table 7, Page: 1 of 1

					Concentration	Emission
Test	Emission	Stack Flow	Concentration	02	@ 7% O2	Rate
Run	Constituent	DSCFM	ug/dscm	%	ug/dscm	lb/hr
Test 1 Run 1	POHC	19728	7.76	12.1	NA	0.000478
Test 1 Run 2	POHC	19786	6.33	12.2	NA	0.000391
Test 1 Run 3	POHC	19575	7.69	12.3	NA	0.000470
	POHC	22225	9.47	9.8	NA	0.000657
	Particulate	22222	14640	9.8	18300	1.015602
	Mercury	22222	4.08	9.8	5.1	0.000283
	SVM	22222	103.04	9.8	128.8	0.007148
Test 2 Run 1	Cd	22222	0.8	9.8	1	0.000055
Test 2 Ruit T	Pb	22222	102.24	9.8	127.8	0.007093
	LVM	22222	63.68	9.8	79.6	0.004418
	As	22222	15.36	9.8	19.2	0.001066
	Be	22222	0.48	9.8	0.6	0.000033
	Cr	22222	47.84	9.8	59.8	0.003319
	POHC	23078	7.83	9.8	NA	0.000564
	Particulate	23043	24160	9.8	30200	1.737942
	Mercury	23043	2.8	9.8	3.5	0.000201
	SVM	23043	141.36	9.8	176.7	0.010169
Test 2 Run 2	Cd	23043	1.04	9.8	1.3	0.000075
TOSTETRUTE	Pb	23043	140.32	9.8	175.4	0.010094
	LVM	23043	90.08	9.8	112.6	0.006480
	As	23043	16.32	9.8	20.4	0.001174
	Be	23043	0.48	9.8	0.6	0.000035
	Cr	23043	73.28	9.8	91.6	0.005271
	POHC	21781	8.64	9.9	NA	0.000587
	Particulate	21749	23865	9.9	30100	1.620317
	Mercury	21749	2.54	9.9	3.2	0.000172
	SVM	21749	135.50	9.9	170.9	0.009200
Test 2 Run 3	Cd	21749	1.98	9.9	2.5	0.000135
	Pb	21749	133.52	9,9	168.4	0.009065
	LVM	21749	126.70	9.9	159.8	0.008602
	As	21749	13.16	9.9	16.6	0.000894
	Be	21749	0.48	9.9	0.6	0.000032
	Cr	21749	113.06	9.9	142.6	0.007676

Table 7.0 - CPT Stack Gas Analytical Results

Notification of Compliance Revision No.: 0 Revision Date: 09/01/10 Table 8, Page: 1 of 1

Table 8.0 - CPT Test 1 Averages

		Planned				
Test Parameters		Conditions	Test 1	Test	1 Run Aver	ages
Description	Units	Target	Average	Run 1	Run 2	Run 3
Total Waste Feed Rate	lb/hr	1000	960	959	965	956
Total Coal Feed Rate	Ib/hr	2500	3221	3266	3249	3148
Total POHC Feed Rate	lb/hr	25	47	46	49	47
Combustion Chamber Temperature	Ц. Ф	1050	1061	1061	1059	1064
Combustion Chamber Pressure	psig	0 >	-0.3	-0.3	-0.3	-0.3
Combustion Air Flow Rate – FD Fan	scfm	NA	11270	11281	11253	11275
Combustion Air Flow Rate – Overfire	scfm	3050	3050	3050	3050	3050
Combustion Air Flow Rate – Atomizing	scfm	1800	1800	1800	1800	1800
Total Combustion Air Flow Rate	scfm	NA	16120	16131	16103	16125
THC Concentration	ppmv	< 10	0.4	0.5	0.5	0.4
O2 Concentration	%	13	12.2	12.2	12.3	12.0
Gun Atomization Pressure	psig	30	30	30	30	30

Prepared by FutureFuel Chemical Company, Batesville, Arkansas, and Risk Management Engineering, Ltd., Dallas, Texas Notification of Compliance Revision No.: 0 Revision Date: 09/01/10 Table 9, Page: 1 of 1

Table 9.0 - CPT Test 2 Averages

		Planned				
Test Parameters		Conditions	Test 2	Test	2 Run Aver	ages
Description	Units	Target	Average	Run 1	Run 2	Run 3
Total Waste Feed Rate	Ib/hr	2500	2518	2509	2509	2535
Total Coal Feed Rate	Ib/hr	6000	4664	4603	4603	4787
Total POHC Feed Rate	Ib/hr	75	117	109.9	119.0	123.1
Total Chloride Feed Rate	Ib/hr	150	494	504	504	474
Total Ash Feed Rate	Ib/hr	1050	791	772	772	828
Total Mercury Feed Rate	Ib/hr	0.0008	0.00019	0.00023	0.00013	0.00020
Total SVM Feed Rate	Ib/hr	0.61	0.63	0.63	0.60	0.66
Total LVM Feed Rate	Ib/hr	1.45	1.57	1.56	1.51	1.64
Combustion Chamber Temperature	Ц.	NA	1304	1304	1301	1308
Combustion Chamber Pressure	psig	0 >	-0.5	-0.5	-0.5	-0.4
Combustion Air Flow Rate - FD Fan	scfm	NA	16280	16190	16178	16471
Combustion Air Flow Rate – Overfire	scfm	3050	3050	3050	3050	3050
Combustion Air Flow Rate – Atomizing	scfm	1800	1800	1800	1800	1800
Total Combustion Air Flow Rate	scfm	NA	21130	21040	21028	21321
ESP Power	kW	7	ω	80	8	80
ESP Inlet Temperature	Ц.	510	515	514	518	514
THC Concentration	ppmv	< 10	0.3	0.3	0.4	0.4
O2 Concentration	%	NA	10.1	10.0	10.1	10.3
Gun Atomization Pressure	psig	30	30	30	30	30

Prepared by FutureFuel Chemical Company, Batesville, Arkansas, and Risk Management Engineering, Ltd., Dallas, Texas Notification of Compliance Revision No.: 0 Revision Date: 09/01/10 Table 10, Page: 1 of 1

Table 10.0 - CPT Comparison of Planned Conditions to Actual Conditions

Test 1 Run 1

		Flanned				
Test Parameters		Conditions	Ac	utal Conditi	ions	Comparison*
Description	Units	Target	Average	Minimum	Maximum	Comment
tal Waste Feed Rate	lb/hr	1000	959	359	1470	
tal Coal Feed Rate	lb/hr	2500	3266	2950	3933	Decided to target higher temp requiring more coal
tal POHC Feed Rate	lb/hr	25	46	17	71	Waste was 4.83% POHC rather than 4% target
tal Chloride Feed Rate	lb/hr	NA	NA	NA	NA	
tal Ash Feed Rate	lb/hr	NA	NA	NA	NA	
tal Mercury Feed Rate	lb/hr	NA	NA	NA	NA	
tal SVM Feed Rate	lb/hr	NA	NA	NA	NA	
tal LVM Feed Rate	lb/hr	NA	NA	NA	NA	
mbustion Chamber Temperature	٩	1050	1061	1048	1071	
Imbustion Chamber Pressure	psig	0 >	-0.3	-0.3	-0.3	
mbustion Air Flow Rate - FD Fan	scfm	NA	11281	10861	11803	
imbustion Air Flow Rate - Overfire	scfm	3050	3050	3050	3050	
mbustion Air Flow Rate – Atomizing	scfm	1800	1800	1800	1800	
tal Combustion Air Flow Rate	scfm	NA	16131	15711	16653	
SP Power	kW	NA	20.4	16.9	24.0	
SP Inlet Temperature	₽°	NA	463	462	464	
IC Concentration	ppmv	< 10	0.5	0.2	0.7	
2 Concentration	%	13	12.2	11.7	17.8	
in Atomization Pressure	osia	30	30	30	30	

* No comment means the difference between the planned and actual conditons were not significant (e.g., < 10%)

Prepared by FutureFuel Chemical Company, Batesville, Arkansas, and Risk Management Engineering, Ltd., Dallas, Texas Notification of Compliance Revision No.: 0 Revision Date: 09/01/10 Table 11, Page: 1 of 1

Table 11.0 - CPT Comparison of Planned Conditions to Actual Conditions

Test 1 Run 2

		Planned				
Test Parameters		Conditions	Ac	utal Conditi	ons	Comparison*
Description	Units	Target	Average	Minimum	Maximum	Comment
Total Waste Feed Rate	lb/hr	1000	965	723	1265	
Total Coal Feed Rate	lb/hr	2500	3249	3006	3881	Decided to target higher temp requiring more coal
Total POHC Feed Rate	lb/hr	25	49	37	64	Waste was 5.05% POHC rather than 4% target
Total Chloride Feed Rate	lb/hr	NA	NA	NA	NA	
Total Ash Feed Rate	lb/hr	NA	NA	NA	NA	
Total Mercury Feed Rate	lb/hr	NA	NA	NA	NA	
Total SVM Feed Rate	lb/hr	NA	NA	NA	NA	
Total LVM Feed Rate	lb/hr	NA	NA	NA	NA	
Combustion Chamber Temperature	ц. °	1050	1059	1054	1073	
Combustion Chamber Pressure	psig	< 0 >	-0.3	-0.3	-0.3	
Combustion Air Flow Rate - FD Fan	scfm	NA	11253	10795	11705	
Combustion Air Flow Rate - Overfire	scfm	3050	3050	3050	3050	
Combustion Air Flow Rate – Atomizing	scfm	1800	1800	1800	1800	
Total Combustion Air Flow Rate	scfm	NA	16103	15645	16555	
ESP Power	kW	NA	22.8	21.0	24.4	
ESP Inlet Temperature	H.	NA	460	459	461	
THC Concentration	ppmv	< 10	0.5	0,3	0.6	
O2 Concentration	%	13	12.3	11.7	17.9	
Gun Atomization Pressure	psig	30	30	30	30	

Notification of Compliance Revision No.: 0 Revision Date: 09/01/10 Table 12, Page: 1 of 1

Table 12.0 - CPT Comparison of Planned Conditions to Actual Conditions

Test 1 Run 3

		Planned				
Test Parameters		Conditions	Ac	utal Condit	ions	Comparison*
Description	Units	Target	Average	Minimum	Maximum	Comment
al Waste Feed Rate	Ib/hr	1000	956	563	1319	
al Coal Feed Rate	lb/hr	2500	3148	2849	3573	Decided to target higher temp requiring more coal
al POHC Feed Rate	lb/hr	25	47	28	65	Waste was 4.89% POHC rather than 4% target
al Chloride Feed Rate	lb/hr	NA	NA	NA	NA	
al Ash Feed Rate	lb/hr	NA	NA	NA	NA	
al Mercury Feed Rate	lb/hr	NA	NA	NA	NA	
al SVM Feed Rate	Ib/hr	NA	NA	NA	NA	
al LVM Feed Rate	lb/hr	NA	NA	NA	NA	
nbustion Chamber Temperature	ų.	1050	1064	1054	1075	
nbustion Chamber Pressure	psig	0 >	-0.3	-0.3	-0.3	
nbustion Air Flow Rate - FD Fan	scfm	NA	11275	10813	11788	
nbustion Air Flow Rate - Overfire	scfm	3050	3050	3050	3050	
hbustion Air Flow Rate – Atomizing	scfm	1800	1800	1800	1800	
I Combustion Air Flow Rate	scfm	NA	16125	15663	16638	
Power	kW	NA	22.7	20.7	24.8	
P Inlet Temperature	ų.	NA	458	458	459	
Concentration	ppmv	< 10	0.4	0,3	0.7	
Concentration	%	13	12.0	10.5	17.4	
Atomization Pressure	psig	30	30	30	30	

Notification of Compliance Revision No.: 0 Revision Date: 09/01/10 Table 13, Page: 1 of 1

Table 13.0 - CPT Comparison of Planned Conditions to Actual Conditions

Test 2 Run 1

		Planned				
Test Parameters		Conditions	Ac	utal Condit	ions	Comparison
Description	Units	Target	Average	Minimum	Maximum	Comment
Total Waste Feed Rate	lb/hr	2500	2509	261	3054	
Total Coal Feed Rate	lb/hr	6000	4603	4236	5677	Less coal was need to make Steam Production Target
Total POHC Feed Rate	lb/hr	75	110	11	134	Waste was 4.38% POHC rather than 4% target
Total Chloride Feed Rate	lb/hr	150	504	353	619	Chloride concentration was 5.4% and target was 6%
Total Ash Feed Rate	lb/hr	1050	772	710	953	Less coal was need to make Steam Production Target
Total Mercury Feed Rate	Ib/hr	0.0008	0.00023	NA	NA	Mercury content in coal was much less than normal
Total SVM Feed Rate	lb/hr	0.61	0.62753	NA	NA	
Total LVM Feed Rate	lb/hr	1.45	1.56461	NA	NA	
Combustion Chamber Temperature	ц.	NA	1304	1237	1318	
Combustion Chamber Pressure	psig	< 0 >	-0.5	-0.5	-0.4	
Combustion Air Flow Rate – FD Fan	scfm	NA	16190	15671	17349	
Combustion Air Flow Rate - Overfire	scfm	3050	3050	3050	3050	
Combustion Air Flow Rate – Atomizing	scfm	1800	1800	1800	1800	
Total Combustion Air Flow Rate	scfm	NA	21040	20521	22199	
ESP Power	kW	ŝ	8.4	7.3	8.6	
ESP Inlet Temperature	Ч°	510	514	511	516	
THC Concentration	ppmv	< 10	0.3	0.0	1.1	
O2 Concentration	%	NA	10.0	9.2	16.5	
Gun Atomization Pressure	psig	30				

Notification of Compliance Revision No. 0 Revision Date: 09/01/10 Table 14, Page: 1 of 1

Table 14.0 - CPT Comparison of Planned Conditions to Actual Conditions

Test 2 Run 2

		Planned				
Test Parameters		Conditions	Ac	utal Condit	ons	Comparison*
Description	Units	Target	Average	Minimum	Maximum	Comment
Total Waste Feed Rate	lb/hr	2500	2509	2502	2518	
Total Coal Feed Rate	lb/hr	6000	4603	4507	5619	Less coal was need to make Steam Production Target
Total POHC Feed Rate	lb/hr	75	119	119	119	Waste was 4.74% POHC rather than 4% target
Total Chloride Feed Rate	lb/hr	150	504	496	586	Chloride concentration was 5.4% and target was 6%
Total Ash Feed Rate	lb/hr	1050	772	756	943	Less coal was need to make Steam Production Target
Total Mercury Feed Rate	lb/hr	0.0008	0.00013	NA	NA	Mercury content in coal was much less than normal
Total SVM Feed Rate	lb/hr	0.61	0.60	NA	NA	
Total LVM Feed Rate	lb/hr	1.45	1.51	NA	NA	
Combustion Chamber Temperature	ų.	NA	1301	1284	1318	
Combustion Chamber Pressure	psig	0 >	-0.5	-0.5	-0.5	
Combustion Air Flow Rate – FD Fan	scfm	NA	16178	15793	16582	
Combustion Air Flow Rate - Overfire	scfm	3050	3050	3050	3050	
Combustion Air Flow Rate – Atomizing	scfm	1800	1800	1800	1800	
Total Combustion Air Flow Rate	scfm	NA	21028	20643	21432	
ESP Power	kW	80	8.2	7.5	8.3	
ESP Inlet Temperature	4.	510	518	516	519	
THC Concentration	ppmv	< 10	0.4	0.2	0.5	
O2 Concentration	%	NA	10.1	9.7	16.8	
Gun Atomization Pressure	psig	30	30	30	30	

Notification of Compliance Revision No.: 0 Revision Date: 09/01/10 Table 15, Page: 1 of 1

Table 15.0 - CPT Comparison of Planned Conditions to Actual Conditions

Test 2 Run 3

		Planned				
Test Parameters		Conditions	Aci	utal Condit	ions	Comparison
Description	Units	Target	Average	Minimum	Maximum	Comment
Total Waste Feed Rate	lb/hr	2500	2535	2521	2548	
Total Coal Feed Rate	lb/hr	6000	4787	4605	6047	Less coal was need to make Steam Production Target
Total POHC Feed Rate	lb/hr	75	123	122	124	Waste was 4.74% POHC rather than 4% target
Total Chloride Feed Rate	lb/hr	150	474	461	563	Chloride concentration was 5.4% and target was 6%
Total Ash Feed Rate	lb/hr	1050	828	796	1046	Less coal was need to make Steam Production Target
Total Mercury Feed Rate	lb/hr	0.0008	0.00020	NA	NA	Mercury content in coal was much less than normal
Total SVM Feed Rate	lb/hr	0.61	0.66	NA	NA	
Total LVM Feed Rate	lb/hr	1.45	1.64	NA	NA	Spike team fed a little more than planned
Combustion Chamber Temperature	Ļ	NA	1308	1291	1315	
Combustion Chamber Pressure	psig	0 >	-0.4	-0.4	-0.4	
Combustion Air Flow Rate – FD Fan	scfm	NA	16471	16127	16940	
Combustion Air Flow Rate - Overfire	scfm	3050	3050	3050	3050	
Combustion Air Flow Rate – Atomizing	scfm	1800	1800	1800	1800	
Total Combustion Air Flow Rate	scfm	NA	21321	20977	21790	
ESP Power	kW	ŝ	7.9	7.3	8.2	
ESP Inlet Temperature	Ч°	510	514	514	514	
THC Concentration	ppmv	< 10	0.4	0.3	1.0	
O2 Concentration	%	AN	10.3	9.8	17.2	
Gun Atomization Pressure	psig	30	30	30	30	

Notification of Compliance Revision No.: 0 Revision Date: 09/01/10 Table 16, Page: 1 of 1

Table 16.0 - Demonstration of Compliance

Test Unit Test 1 - Minimum Temperature Test 0 Destruction Removal Efficiency % Total Hydrocarbon Emissions ³ ppm Test 2 - Minimum Residence Time Test 1	Units % ppmv	Standard ≥ 99.99% ≤ 10 > 00 00%	CPT Result
Test 1 - Minimum Temperature Test % Destruction Removal Efficiency % Total Hydrocarbon Emissions ³ ppr Test 2 - Minimum Residence Time Test	wmqq	≥ 99.99% ≤ 10 > ao ao%	
Destruction Removal Efficiency Total Hydrocarbon Emissions ³ ppm Test 2 - Minimum Residence Time Test	%	≥ 99.99% ≤ 10 > ao ao%	
Total Hydrocarbon Emissions ³ ppm Test 2 - Minimum Residence Time Test	ppmv %	≤ 10 > ao ao%	99.9991%
Test 2 - Minimum Residence Time Test	70	> 00 00%	0.4
Test 2 - Minimum Residence Time Test	10	> 00 00%	
	10	> 00 00%	
Destruction Removal Efficiency	/0	1 00.001	99.9995%
Particulate Emission ³ mg/ds	mg/dscm	≤ 68	26.2
HCI/Chlorine Emissions ¹ Ib/h	Ib/hr	1886.8	508
Mercury Emissions ³ ug/ds	ug/dscm	≤11	3.9
Semi-Volatile Metal Emissions ³ ug/ds	ug/dscm	≤ 180	158.8
Low-Volatile Metal Emissions ³ ug/ds	ug/dscm	≤ 380	117.3
Dioxin/Furan ^{2 and 3} (as demonstrated by THC emissions) ppr	ppmv	≤ 10	0.3
Total Hydrocarbon Emissions ³ ppm	ppmv	≤ 10	0.3

¹ Limit established using Health-based Alternative Compliance Demonstration described in Appendix D of the CPT Plan ² Requirement to conduct one time D/F test fuliled by submittal of data in lieu of test. See Appendix F of CPT Plan.

³ Results corrected to 7% O ₂

Prepared by FutureFuel Chemical Company, Batesville. Arkansas, and Risk Management Engineering. Ltd., Dallas, Texas

Notification of ComplianceRevision No.:0Revision Date:09/01/10Table 17, Page:1 of 1

Test	DRE	SRE	SRE	SRE
Run	POHC	Particulate	SVM	LVM
Test 1 Run 1	99.9990%	NA	NA	NA
Test 1 Run 2	99.9992%	NA	NA	NA
Test 1 Run3	99.9990%	NA	NA	NA
Test 1 Average	99.9991%	NA	NA	NA
Test 2 Run 1	99.9994%	99.8685%	98.8609%	99.7177%
Test 2 Run 2	99.9995%	99.8685%	98.8138%	99.7067%
Test 2 Run 3	99.9995%	99.8773%	98.9235%	99.7311%
Test 2 Average	99.9995%	99.8714%	98.8660%	99.7185%

Table 17.0 - Boiler Removal Efficiency

Notification of Compliance Revision No.: 0 Revision Date: 09/01/10 Table 18. Page: 1 of 1

Table 18.0 - Summary of Operating Parameter Limits

		Final	
		Operating	
Operating Parameters		Limit	Comment
Waste Feed Limitations			
Maximum Hazardous Waste Feed Rate	lb/hr	2518	Obtained as the Average of the Test Run Average from Test 2
Maximum Chloride Feed Rate	lb/hr	494	Obtained as the Average of the Test Run Average from Test 2
Maximum Ash Feed Rate	lb/hr	791	Obtained as the Average of the Test Run Average from Test 2
Maximum Mercury Feed Rate	lb/hr	0.00083	Obtained as the Average of the Test Run Average from Test 2
Maximum Semi-Volatile Feed Rate	lb/hr	0.64	Obtained as the Average of the Test Run Average from Test 2
Maximum Low-Volatile Feed Rate	lb/hr	4.58	Obtained as the Average of the Test Run Average from Test 2
Combustion Chamber Limitations			
Minimum Combustion Chamber Temperature	Ļ	1061	Obtained as the Average of the Test Run Average from Test 1
Minimum Atomization Pressure	psig	30	Manufacturer's recommendation
Maximum Furnace Pressure	in.wc.	0 >	Induced draft system, must maintain negative furnace pressure
Maximum Combustion Air Flow Rate	scfm	21130	Obtained as the Average of the Test Run Average from Test 2
Air Pollution Control Limitaions			
Minimum ESP Power	kW	8	Obtained as the Average of the Test Run Average from Test 2
Maximum ESP Inlet Temperature	۴	515	Obtained as the Average of the Test Run Average from Test 2
Stack Gas Limitations			
Maximum THC Concentration	ppmv	10	MACT Emission Standard
Maximum O ₂ Concentration	%	12.2	Obtained as the Average of the Test Run Average from Test 1

Nottfication of Compliance Revision No.: 0 Revision Date: 09/01/10 Tables 19-20. Page: 1 of 1

trapolation
Metal Ex
19.0
Table

		CPT Emisson		Extrapolated
	CPT Feed	Concentration	90% of MACT	Operating
Metal	Rate	@ 7% 02	Standard	Limit
Group	(lb/hr)	(MDSCM)	(MDSCM)	(Ib/hr)
SVM	0.631	158.8	162	0.64
LVM	1.571	117.3	342	4.58

Table 20.0 - Mercury MTEC Feed Rate Limit

(MACT Limit ug/dscm) x (2.204586E-9 lb/ug) x (Stack Flow dscm/min) x (60 min/lr) / (1-SRE)	(11 ug/dscm) x (2.204586E-9 lb/ug) x (22499 dscf/min / 35.34 dscf/dscm) x (60 min/hr) / (1-0)),(K(09)3	0.08083
Hg Feed Rate Limit =	Hg Fccd Rate Limit =	Hg Fccd Rate Limit =	90% Hg Feed Rate Limit =

Notification of Compliance Revision No.: 09/01/10 Revision Date: 09/01/10 Table 21, Page: 1 of 3

TABLE 21.0

SUMMARY OF SAMPLING AND ANALYSIS PROGRAM

sters	Sampling Method	Sample Frequency	Field QA/QC Samples	Analytical Methods
ity			None	ASTM D-1298
	Grab, 4 oz. collected		None	ASTM D-482
	at each sample		None	ASTM E776 / SW-846 9056
	Interval	Initially and every 30 minutes; each run	None	ASTM D-240
	(Two composites		None	SW-846 6020
	prepared for each run)		None	SW-846 6020
			None	ASTM D-3684-01
	Grab, 40ml collected at each sample interval	Every 30 minutes, each run	None	SW-846, 8260
			None	ASTM D-3174
	Grab, 4 oz. collected		None	ASTM E776 / SW-846 9056
	interval	Initially and every 30	None	ASTM D-5865
	-	minutes, each run	None	SW-846 6010B
	(1 wo composites prepared for each run)		None	SW-846 6010B
			None	ASTM D-3684-01

Notification of Compliance Revision No.: 09/01/10 Revision Date: 09/01/10 Table 21, Page: 2 of 3

TABLE 21.0 (Continued)

SUMMARY OF SAMPLING AND ANALYSIS PROGRAM

Methods	Oa					
Analytica	SW-846 6020 and 747	sting will be performed.		SW-0030, 5041		
Field QA/QC Samples	Reagent Blanks	al efficiency. No stack te	One condensate trip blank	One pair VOST tube trip blank	One set field blank tubes (four pairs each) per test condition	ompleted in 1999.
Sample Frequency	120 minutes each run	on feed rate and zero remov		120 minutes each run		testing based on trial burn ee
Sampling Method	Method 5 and 29, MMT, isokmetic sample	mative Chloride Limit based		Method 0030, VOST train, 4 tube pairs		
Parameters	As, Be, Cd, Cr, Hg, and Pb	Compliance with Alter		Volatile Organics		Compliance will be der
Sample Type	 Stack gas metals and particulate 	4) Stack gas hydrogen chloride and Chlorine		6) Stack gas volatile	M Battes - + 00 I	7) Stack gas PCDDs/PCDFs – MM5

3 01 3 Notification of Compliance Revision No.: 09/01/10 Revision: 09/01/10 Table 21, Page: 3 of 3 0 01/10/60

TABLE 21.0

SUMMARY OF SAMPLING AND ANALYSIS PROGRAM

Sampling Method Sample Frequency Field QA/QC Analytical Methods	ORSAT (EPA Method 3) Each run None ORSAT Analyzer	EPA Method 25A Each Run None Continuous, Extractive, Flame Ionization Analyzer	Continuous, Each run None Continuous Extractive; 3B	EPA Method 4 Each run None Measure volume to 0.5 ml	EPA Method 2 Each run None None Measure temperature, pressure, and volume
Sampling Method Sample Fr	DRSAT (EPA Method 3) Each run	PA Method 25A Each Run	Continuous, Each run xtractive, 3B	2PA Method 4 Each run	SPA Method 2 Each run
Parameters	Carbon Dioxide	Total Hydrocarbon (from I: in-house CEMS)	Oxygen (from in-house CEMS & Stack Team) e	Stack Moisture E	Stack Velocity and Elowrate
Sample No./Type					

Notes to Table 5.0:

 Polynuclear aromatic hydrocarbon 	 American Society for Testing and Materials 	 Principal organic hazardous constituents 	 Cold vapor atomic absorption 	 Quality assurance and quality control 	 Gas chromatography and mass spectroscopy 	 Gas chromatography and flame ionization detector 	 Volatile organic sampling train 	 Inductively coupled argon plasma spectroscopy 	 Inductively coupled plasma-atomic emission spectron
PAH	ASTM	POHC	CVAA	QAQC	GC/MS	GC/FID	VOST	ICP	ICP-AES

Arsenic	Beryllium	Cadmium	Chromium	Mercury	Sulfurie acid	Potassium hydroxide	Gallons per minute	British thermal units	Lead
Ð	8	ü	11	11	ß	8	1	U	4
As.	Be	Cd	C	Hg	H2SO4	KOH	gpm	Bu	Ph
Low-Volatile Metals	Milliliter	EPA Method 0011	Modified Method 5	Multi-metals train	Sodium hydroxide	Nondispersive infrared	Semi-Volatile Metals	Dioxin	Furan
It	H	()	11	R.	Ð	31	30	в	ł.
INNI	ML	M0011	NIM5	TMM	NaOH	NDIR	NVS	PCDD	PCDF

NOTIFICATION OF COMPLIANCE

FutureFuel Chemical Company

Hazardous Waste Incinerator

Revision 0

Pursuant to

NESHAP: Standards for Hazardous Air Pollutants for Hazardous Waste Combustors (40 CFR 63 Subpart EEE)

Prepared by

FUTUREFUEL CHEMICAL COMPANY ARKANSAS OPERATIONS P.O. Box 2357 Batesville, AR 72503 ARD089234884

RISK MANAGEMENT & ENGINEERING, LTD. 705 W. Avenue B, Suite 400 Garland, TX 75040

January 12, 2011

EXECUTIVE SUMMARY

FutureFuel Chemical Company (FFCC) owns and operates an organic chemical manufacturing plant located southeast of Batesville, Arkansas. As part of plant operations, FFCC generates wastes regulated under the Resource Conservation and Recovery Act (RCRA) and destroys waste in a MACT EEE compliant phase I incinerator currently operating under Title V permit # 1088-AOP-R8. FFCC also burns waste for energy recovery in three coal-fired boilers. Those boilers are currently operating under an NOC submitted September 1, 2010, as well as, Arkansas Department of Environmental Quality (ADEQ) Waste Permit Number 11H-M005

FFCC submitted a Comprehensive Performance Test (CPT) plan for approval on August 1, 2008. The plan was not reviewed by EPA Region VI until August 2010. On October 5, 2010, EPA Region VI submitted a letter of conditional approval to FFCC. FFCC then conducted the Comprehensive Performance Test (CPT) the week of October 11, 2010. The results of this test are being submitted as the Notice of Compliance (NOC) to the EPA and ADEQ. FFCC will begin complying with the operational parameters and limits established within this NOC immediately upon submittal.

CONCLUSIONS ON MEETING CPT OBJECTIVES

The overall objective of the Comprehensive Performance Test was to demonstrate that the incinerator is capable of meeting the incinerator emission standards established in the MACT Combustion Rule (40 CFR 63 Subpart EEE) and to establish operating conditions and parameters that are equivalent with that demonstration. Table ES-1 summarizes the specific objectives demonstrated during the CPT.

FFCC was successful in meeting the overall CPT objective, and this NOC establishes the operating conditions that reflect the results of this successful test.

Incinerator Notification of C	օոր	ha	ice
Revision No			0
Revision Date	$\{0\}$	12	П
Executive Summary			ii

Table ES-1

Specific Objective	CPT Result
	Run I DRE = > 99.99996 %
Demonstrate 99.99 % DRE of the designated POHC Chlorobenzene	Run 2 DRE = > 99.99997 %
	Run 3 DRE = > 99,99995 %
Demonstrate control of carbon monoxide (CO) emissions to less than 100 parts per million dry volume (ppmv), corrected to 7 percent oxygen, on a 60-minute rolling average basis.	The average CO demonstrated during all three runs of the CPT was 1.2 ppmv
Demonstrate control of hydrocarbon emissions to less than 10 parts per million dry volume (ppmv), corrected to 7 percent oxygen, on a 60-minute rolling average basis.	The average THC demonstrated during all three runs of the CPT was 2.2 ppmy
Demonstrate control of particulate emissions to less than 0.013 grains per dry standard cubic foot (gr/dsef) corrected to 7 percent oxygen.	The average particulate emission demonstrated during three runs of the CPT was <u>0.0042 gr/dsef</u>
Demonstrate that control of HCl and free chlorine emissions are equal to or less than 32 ppmv corrected to 7 percent oxygen.	The average HCI/CI_2 emission demonstrated during the three runs of the CPT was <u>10.0 ppmv</u>
Demonstrate that control of mercury emissions are equal to or less than 130 micrograms per dry standard cubic meter (ug/dscm) corrected to 7 percent oxygen.	The average mercury emission demonstrated during the three runs of the CPT was <u>1.3 ug/dscm</u>
Demonstrate that control of Semi-Volatile metals (SVM) emissions are equal to or less than 230 micrograms per dry standard cubic meter (ug/dscm) corrected to 7 percent oxygen.	The average SVM emission, demonstrated during the three runs of the CPT was 224.8 ug/dscm
Demonstrate that control of Low-Volatile metals (LVM) emissions are equal to or less than 92 micrograms per dry standard cubic meter (ug/dsem) corrected to 7 percent oxygen.	The average LVM emission, demonstrated during the three runs of the CPT was <u>65.5 ug/dsem</u>
Demonstrate that control of Dioxin/Furan (D/F) emissions are equal to or less than 0.4 nanograms TEQ per dry standard cubic meter (ng TEQ/dscm) corrected to 7 percent oxygen.	The average D/F emission, demonstrated during the three runs of the CPT was 0.0032 ng/dscm

Incinerator Notification of C	omp	hance
Revision No		()
Revision Date	$\{0\}$	12 11
Executive Summary		iii

DEVIATIONS FROM THE CPT PLAN

There were a few deviations from the approved CPT plan. Table ES-2 details the changes made after formal EPA and ADEQ approval of the CPT plan.

Change to the CPT Plan	Explanation of Change
The heat content, POHC content, and chloride content of Run 2 was higher than planned.	When FFCC began pumping the low-btu waste to the incinerator for stabilization for Run 2, a valve was not closed all the way at a transfer station and approximately 2000 gallons of low-btu aqueous waste slipped into the high-btu waste tank. This caused phase separation in the high-btu waste tank. The heavy chlorinated waste migrated to the bottom taking most of the toluene with it, and the water phase migrated to the top taking most of the methanol with it. Our feed to the incinerator was coming from the lower portion of the tank, and this section had a higher concentration of perchloroethylene, chlorobenzene, and toluene than planned.
FFCC added small amounts of sulfuric acid to the scrubber recirculation tank during runs 1 and 3 to maintain the pH target submitted in the CPT Plan. Approximately 5 lbs an hour was introduced into the system.	FFCC set the target scrubber pH during the CPT for a pH of 4. In order to demonstrate the system was capable of removing HCl/Cl2 at low pH. FFCC found it necessary to add a small quantity of sulfuric acid into the scrubber recirculation tank during Run 1 and Run 3 in order to lower the scrubber pH into the desired target range. This issue is discussed in more detail in the following "Issues and Solutions" section of the executive summary.

Table	ES-2
-------	------

Incinerator Notification of Compliance Revision No 0 Revision Date 01/12/11 Executive Summary iv

CPT ISSUES AND SOLUTIONS

The incinerator performed well. Two issues were encountered during the CPT. One involved the high-btu waste feed composition, and the other involved the pH of the scrubber water.

Issue 1 - Waste Feed Composition

When FFCC began pumping the low-btu waste to the incinerator for stabilization for Run 2, a transfer line valve was not closed all the way and approximately 2000 gallons of low-btu aqueous waste slipped out of the feed line into the high-blu waste tank before it was discovered. When we were stabilizing we knew that something was different because the waste appeared to have higher-Btu than in Run 1 and the scrubbing system was behaving as though there were more chlorides present than in Run 1. However, the differences did not appear to be so significant that Run 2 could not be completed and we moved forward with Run 2. We continued to notice the differences throughout Run 2. We theorized that the water caused some phase separation in the tank. At the end of the run we pulled samples from different layers in the tank. and determined that phase separation did indeed occur. The heavy chlorinated organics (perchloroethylene and chlorobenzene) migrated to the bottom of the tank taking most of the toluene and a little methanol with it. The low-btu aqueous phase migrated to the top taking most of the methanol with it and a little toluene. This told us we probably fed way more chloride in Run 2 that we had planned, because we were burning from the lower portion of the tank. That night we decanted off the top water contaminated methanol layer and replaced the methanol. This worked very well and got us back close to our target composition for Run 3. Since the Run 2 chloride feed rate was greater than the feed rates where we had historically demonstrated compliance, we were concerned with HCl/Cl₂ emissions for Run 2. We believed the unit was capable of removing the HCl/Cl₂ but we did not have any historical data to support that hypothesis. For insurance, FFCC actually conducted a Run 4 for HCl/Cl₂ only in case Run 2 had a detrimental effect on the CPT. It turned out the unit performed well in Run 2 and the data from Run 4 was not needed, and thus not used.

<u>Issue 2 – Scrubber pH</u>

During stabilization prior to Run 1, the pH of the scrubber did not drop into the target range (pH = 4). In order to demonstrate the performance of the scrubber to remove HCl/Cl_2 at a low pH, it was decided to use a small pump to meter sulfuric acid into the scrubber water on the suction side of the scrubber water recirculation pump. This lowered the pH of the scrubber water well before the scrubber pH probes. We were able to keep the pH around 4 with a spike rate of approximately 2.4 lb every 15 minutes of sulfuric acid. This was done on both Run 1 and Run 3.

PERMIT RELATED OPERATIONAL DATA

Table ES-3 provides a summary of the permit related operational data derived from the CPT.

Table	ES-3
-------	------

Parameter Limit	Basis	Result
Maximum total hazardous waste feed (lb/hr-HRA)	Average of the test run averages	15.348
Maximum Chloride Feed Rate (lb/hr-12HRA)	Average of the test run averages	1,428
Maximum Ash Feed Rate (lb/hr-12HRA)	Average of the test run averages	1,453
Maximum Mercury Feed Rate (lb/hr-12HRA)	The mercury feed rate is established by 90% of the Maximum Theoretical Emission Calculation (MTEC). The MTEC is based on the maximum flue gas flow rate limit. MACT EEE mercury emission limit, and zero percent mercury removal efficiency.	0.01
Maximum Semi-Volatile Metal Feed Rate (lb/hr-12HRA)	Average of the test run averages	0,84
Maximum Low-Volatile Metal Feed Rate (lb/hr-12 HRA)	Average of the test run averages	1,62
Maximum Flue Gas Flow Rate (sefm - HRA)	Correlated from the Average of the Combustion Air Flow Rate Test Run Averages	23,602
Minimum Combustion temperature (degrees F - HRA)	Average of the test run averages	1,558

Incinerator Notification of Compliance					
Revision No	0				
Revision Date	01/12/11				
Executive Summary	vi				

Table ES-3, continued

Parameter	Basis	Result
Minimum Serabber dP (in. w.e HRA)	Average of the test run averages	71
Minimum Scrubber Water pH (HRA)	Average of the test run averages	4.4
Minimum Quench Tank pH (HRA)	Average of the test run averages	0,6
Minimum 1 ⁴ Stage Scrubber Water Flow Rate (gpm - HRA)	Average of the test run averages	477
Minimum 2 nd Stage Scrubber Water Flow Rate (gpm – HRA)	Average of the test run averages	3()]
Minimum Scrubber Water Blowdown Rate (gpm – HRA)	Average of the test run averages	100
Minimum Scrubber Tank Level (% - HRA)	Average of the test run averages	72
Minimum Atomization Pressure (psig – Instantaneous)	Manufacturer's Recommendation	30
Maximum Combustion Chamber Pressure (in. w.c Instantaneous)	90% of the Maximum System Design Pressure	200

Incinerator Notification of Compliance Revision No 0 Revision Date 01/12/11 Executive Summary VII

FINAL PERMIT LIMITS

The final limits adopted pursuant to this NOC are described in Table ES-3. All parameters will be continuously monitored process parameters that will be tied to automatic waste feed cutoffs (AWFCO).

TABLE OF CONTENTS

Page

Executive Summary	i
Table of Contents	viii
1.0 – Notification of Compliance	I
1.1 – Facility Information and List of Key Project Personnel	1
1.1.1 – Facility Information	
1.1.2 – List of Key Project Personnel	1
1.2 – Detailed Engineering Description of the Incinerator	5
1.2.1 – Combustion and Fuel Feed Systems	5
1.2.1.1 – Auxiliary Fuel Feed Systems	6
1.2.1.2 – Liquid Waste Feed Systems	6
1.2.2 – Prime Mover	7
1.2.3 – Waste Liquid Injection Nozzles	7
1.2.4 – Description of Rapid Quench System	7
1.2.5 – Automatic Waste Feed Cutoff System	8
1.2.6 – Description of Air Pollution Control Equipment	9
1.2.7– Incinerator Stack	9

	Page
1.2.8 – Location and Description of Temperature, Pressure, And Flow Indicating and Control Devices	9
1.2.9 – Stack Gas Monitoring System	10
1.3 – Incinerator Operator Training and Certification	10
2.0 – Sampling and Analysis Program	1 8
2.1 – Sampling Program	1 8
2.1.1 – Waste Feed Preparation	11
2.1.2 – Sampling and Monitoring Procedures	12
2.1.3 – Volatile Organic Analysis Samples	12
2.1.4 - Organic Waste Samples	12
2.1.5 – Aqueous Waste Samples	13
2.1.6 – Metals Spiking Solution	13
2.1.7 – Combustion Gas	13
2.1.7.1 – Combustion Gas Temperature	13
2.1.7.2 – Combustion Gas Monitoring	13
2.1.8 – Stack Gas	14
2.1.8.1 – Hydrogen Chloride, Chlorine, and Particulate Train	14
2.1.8.2 – Multi-Metals Train	14

	Page
2.1.8.3 – Volatile Organic Sampling Train	15
2.1.8.4 – Dioxin/Furan Sampling Train	15
2.2 – Analytical Program	16
2.2.1 – Analytical Methods	16
2.2.2 – Analytical Results	16
3.0 – Description of the Test Program	17
3.1 – CPT Objectives	17
3.2 – CPT Approach	18
3.2.1 – POHC Selection	18
3.3 – CPT Program	19
3.3.1 – Comprehensive Performance Test	19
3.4 – CPT Sampling and Analytical Protocols	19
4.0 – Comparison of Actual Test Conditions versus Planned Conditions and Independent QA Review	21
4.1 – Combustion Device Parameters	21

	Page
4.2 – Waste Feed Rate Parameters	22
4.3 – Constituent Feed Rate Parameters	22
4.4 – Stack Gas Parameters	23
4.5 – Independent QA Review	23
4.5.1 – Stack Testing Equipment	23
4.5.2 – Stack Gas-Particulate Analysis	23
4.5.3 – Stack Gas – HCl/Cl ₂ Analysis	24
4.5.4 – Stack Gas – Metal Analysis	24
4.5.5 – Stack Gas – Volatile Organics Analysis	25
4.5.6 – Stack Gas – PCDD/PCDF	25
4.5.7 - Stack Gas - Continuous Emissions Monitoring Data	26
4.5.8 – Waste - Volatile Organics	26
4.5.9 - Waste - Non-Organics	27
4.5.10 – Process Instruments	28
4.5.11 – Conclusion	29

	Page
5.0 - Comparison of Test Results with Regulatory Compliance Limits	30
5.1 - Regulatory Requirements for the CPT	30
5.1.1 – Destruction Removal Efficiency	30
5.1.2 – Particulate Emissions	31
5.1.3 – HCl / Cl ₂ Emissions	31
5.1.4 – Mercury Emissions	31
5.1.5 – Semi-Volatile Metals Emissions	31
5.1.6 – Low-Volatile Metals Emissions	32
5.1.7 – Dioxin/Furan Emissions	32
5.1.8 – Carbon Monoxide Emissions	32
5.1.9 - Total Hydrocarbon Emissions	32
6.0 – Procedures and Limits for Operating Parameters	33
6.1 – Waste Feed Rate Limitations	33
6.1.1 – Maximum Hazardous Waste Feed Rate Limit	33
6.1.2 – Maximum Chloride Feed Rate Limit	33
6.1.3 – Maximum Ash Feed Rate Limit	33
6.1.4 – Metals Feed Rate Limits	34

	Page
6.1.4.1 – Maximum Mercury Feed Rate Limit	34
6.1.4.2 – Maximum Semi-Volatile Metals Feed Rate	34
6.1.4.3 – Maximum Low Volatile Metals Feed Rate	34
6.2 – Combustion Chamber Limitations	35
6.2.1 – Minimum Combustion Chamber Temperature Limitation	35
6.2.2 – Minimum Atomization Pressure Limitation	35
6.2.3 – Maximum Furnace Pressure Limitation	35
6.2.4 – Maximum Flue Gas Flow Rate Limitation	35
6.3 – Air Pollution Control Limitations	36
6.3.1 – Minimum Pressure Drop Across the Scrubber	36
6.3.2 – Minimum Scrubber pH	36
6.3.3 – Minimum Quench Tank pH	36
6.3.4 – Minimum Water Flow Rate to 1 st Stage Scrubber	37
6.3.5 – Minimum Water Flow Rate to 2 nd Stage Scrubber	37
6.3.6 - Minimum Scrubber Blowdown Flow Rate	37
6.3.7 – Minimum Scrubber Recirculation Tank Volume	37
6.4 – Stack Gas Limitations	38
TABLE OF CONTENTS, continued

	Page
7.0 – CPT Conditional Approval Items	39
7.1 – Quality Assurance Project Plan (QAPP)	39
7.1.1 – Signature Page	39
7.1.2 – Hazardous Waste Combustion Table	39
7.2 – Combustion Zone Pressure	39
7.2.1 – Combustion Zone Design and Maintenance	39
7.2.1.1 – Manufacturer's Design	39
7.2.1.2 – Maintenance Process for Seals	40
7.2.1.3 – Maintenance / Replacement History for Seals	40
7.2.2 – Combustion Zone Inspection Program	40
7.2.3 – Recordkeeping	40
7.2.4 – Response to Leaks	40
7.3 – General Pending Comments	41
7.3.1 – Feed Stream Analysis Plan	41
7.3.2 – CO CEMS Range	41
7.3.4 – Combustion Air and Flue Gas Flow Correlation	41
8.0 – Statement of Compliance	42

LIST OF TABLES

- Table 1 Feed Stream Analysis Data
- Table 2 Feed Stream Characteristics
- Table 3 Summary of Sampling and Analysis Program
- Table 4 Process Monitoring Instruments
- Table 5 Process Monitoring Instruments, Calibration, and Maintenance
- Table 6 CPT Process Feed Analytical Results
- Table 7 CPT Spiking Information
- Table 8 CPT Stack Gas Analytical Results
- Table 9 CPT Test Averages
- Table 10 CPT Comparison of Planned Conditions to Actual Conditions Run 1
- Table 11 CPT Comparison of Planned Conditions to Actual Conditions Run 2
- Table 12 CPT Comparison of Planned Conditions to Actual Conditions Run 3
- Table 13 Demonstration of Compliance
- Table 14 Summary of Operating Parameter Limits
- Table 15 Mercury MTEC Feed Rate Limit
- Table 16 Destruction Removal Efficiency
- Table 17 CPT Operational Data Summary

LIST OF ATTACHMENTS

- Attachment A Alliance Source Testing Stack Test Report
- Attachment B Quality Assurance Project Plan
- Attachment C Instrument Calibration Records
- Attachment D Feed Stream Analytical Results
- Attachment E Other Information

Attachment E-1 - Metals Spiking Certificate of Analysis

Attachment E-2 - Field Sampling and Spiking Logs

Attachment E-3 – Combustion Air Correlation Curve

Attachment E-4 – Relative Accuracy Test Audit

Attachment E-5 – Process Test Data

Attachment F – Destructor Flue-Gas Scan

Attachment G – Description of Revisions

LIST OF DRAWINGS

P&I Diagram Burner and Controls CD-201	6M10-9T-001
P&I Diagram Burner and Controls CD-200R	6M10-9T-002
P&I Diagram Weir Tank CD-220	6M10-9T-003
P&I Diagram Quench Separator CD-236	6M10-9T-004
P&I Diagram Scrubber and Separator CD-223 & CD-227	6M10-9T-005
P&I Diagram Scrubber Recirculation Tank CD-224	6M10-9T-006
P&I Diagram Combustion Air Blower Item CD-207R	oM10-9T-007
P&I Diagram Fuel Guns and Waste Guns	6M10-9T-009
P&I Diagram Continuous Emission Monitoring System	6M10-9T-012
T-Thermal Destructor Steel Outline CD-200R1	6M10-4V-032
T-Thermal Destructor Steel Outline CD-200R1	6M10-4V-033
American Boa Non-Metallic Ducting Expansion Joints	6M10-4E-050

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 1-of 42

1.0 Notice of Compliance

The following notification of compliance summarizes the activities associated with the comprehensive performance test (CPT), documents the results of the CPT, and serves as the basis for the development of operating conditions. The CPT was generally conducted in accordance with the CPT plan as approved August 1, 2008 and adjusted based on EPA Region VI "conditional approval" letter received October 5, 2010.

1.1 Facility Information and List of Key Project Personnel

The following section gives background facility information and provides a list of key project personnel.

1.1.1 Facility Information

Background facility information is listed below:

Facility Name:	FutureFuel Chemical Company
Contact:	Thomas L Floyd
Address:	P.O. Box 2357
	2800 Gap Rd. Hwy 394-S
	Batesville, Arkansas 72503
Telephone Number:	(870) 698-1811
U.S. EPA Identification No.	ARD089234884
Regulatory Agencies:	Environmental Protection Agency, Region VI
	Arkansas Department of Environmental Quality, Air Division

1.1.2 List of Key Project Personnel

A list of key project personnel is shown below:

CPT	Name	Thomas L Floyd
Manager	Company	FutureFuel Chemical Company
	Title	Senior Environmental Biologist
	Address	P.O. Box 2357
		Batesville, Arkansas 72503
	Telephone No.	(870) 698-1811
	Responsibility:	Responsible for implementing and coordinating the CPT

Independent	Name	David Weeks
Third Party	Company	Risk Management & Engineering, LTD.
Quality	Title	Senior Environmental Engineer
Assurance	Address	Dallas, TX
Auditor	Telephone No.	(972) 412-6819
	Responsibility:	Independent third party quality assurance auditor for all
		aspects of the CPT

Stack	Name	Jeremy Hutchens
Sampling	Company	Alliance Source Testing
Project	Title	Environmental Scientist
Director	Address	214 Central Circle SW
		Decatur, Alabama 25603
	Telephone No.	(256)260-3974
	Responsibility:	QA/QC assurance of stack gas sampling to ensure integrity
		of emissions data, as well as reporting coordinator.

Stack	Name	Ryan O'Dea
Sampling	Company	Alliance Source Testing
Field	Title	Stack Sampling Supervisor
Supervisor	Address	8020 Counts Massie Road
		N. Little Rock, AR 72113
	Telephone No.	(501) 771-9969
	Responsibility:	Coordinating the collection of stack gas emissions

Laboratory	Name	Michael D. Challis
Coordinator	Company	Maxxam
& QA/QC	Title	Laboratory QA/QC Manager
Manager	Address	P.O. Box 598
		Addison, Texas 75001
	Telephone No.	(972) 931-7127
	Responsibility:	Coordinating the laboratory analysis of all stack samples
		and ensuring QA/QC procedures on sampling and analysis.

Process	Name	John Overbey
Sampling	Company	American Interplex
Laboratory	Title	Laboratory Director
Manager	Address	8600 Kanis Road
		Little Rock, AR 72204
	Telephone No.	(501) 224-5060
	Responsibility:	Coordinating the laboratory analysis of all process samples
		and ensuring QA/QC procedures on sampling and analysis.

Incinerator	Name	Eddie Brown
Operations	Company	FutureFuel Chemical Company
Supervisor	Title	Opertions Supervisor
	Address	P.O. Box 2357
		Batesville, Arkansas 72503
	Telephone No.	(870) 698-1811
	Responsibility:	Supervisor of Incinerator Operations

Process	Name	Mike Gillihan
Spiking	Company	FutureFuel Chemical Company
Coordinator	Title	Process Assistant
	Address	P.O. Box 2357
		Batesville, Arkansas 72503
	Telephone No.	(870) 698-1811
	Responsibility:	Coordinating process spiking

Process	Name	Marshel Bray
Sampling	Company	FutureFuel Chemical Company
Coordinator	Title	Environmental Process Assistant
	Address	P.O. Box 2357
		Batesville, Arkansas 72503
	Telephone No.	(870) 698-1811
	Responsibility:	Coordinating process sampling

Process	Name	Lynn Cornelius
Sample	Company	FutureFuel
Laboratory	Title	Senior Environmental Technologist
Coordinator	Address	P.O. Box 2357
		Batesville, Arkansas 72503
	Telephone No.	(870) 698-1811
	Responsibility:	Coordinating the sampling and analysis of all process
		samples.

Agency	Name	Larissa Brown
Oversight	Agency	Arkansas Department of Environmental Quality
Personnel	Address	1141 E. Main St, Suite #315
		Batesville, AR 72501
	Telephone No.	(870) 793-4762
	Responsibility:	ADEQ oversight of the CPT

1.2 Detailed Engineering Description of Incinerator System

FFCC operates one waste-burning incinerator at its Batesville, Arkansas plant. The incinerator is a John Zink HI-50, with a rapid quench and a Hydrosonic scrubbing system, which burns liquid process wastes for destruction.

1.2.1 Combustion and Fuel Feed Systems

The primary combustion chamber or burner is a small vertical refractory lined cylinder with five main guns mounted at the top in the burner. Natural gas can be fed through a gun located in the center of the burner at the upper level. There are four air-atomized guns on the burner as well. One of the air-atomized guns is used to feed liquid auxiliary fuels; the other three air-atomized guns are used to feed liquid waste.

The secondary level, just below the upper level, utilizes six additional air-atomized guns fed from one or two feed lines for combustion of low-Btu waste or injection of water.

The secondary combustion chamber is a vertical down-fired John Zink designed unit that measures 12'-0" outside diameter and 10'-5" inside diameter. It has a straight side height of 33'-5". The cross-sectional area of the primary chamber is 17 square feet, and the cross-sectional area of the secondary chamber is 86 square feet. The unit is constructed of refractory-lined carbon steel.

After leaving the combustion chamber, waste gases enter the rapid quench system. The system is described in section 1.2.4

Next, the waste gases pass through the air pollution control system (APCS) described in Section 1.2.6 before exiting out the stack.

Design Criteria

- Furnace Residence Time 2.2 3 seconds at 1900 degree F
- Design Firing Rate 50 MM BTU/hr
- Design Pressure 8 psig
- Fuel liquid waste, fuel oil, and natural gas

Incinerator Notice of Compliance			
Revision No	t F		
Revision Date:	01.12.11		
Page	6 of 42		

The fuel used for the incinerator is liquid waste, fuel oil, and natural gas with the liquid waste serving as the primary fuel.

1.2.1.1 Auxiliary Fuel Feed Systems

The auxiliary fuel feed system consists of fuel oil storage tanks, pumps, supply piping and valves, natural gas pilots, electric igniters, atomizing air, and flame detectors, as well as the burner itself. The fuel supply is interlocked with combustion control and automatic shutdown systems. The main auxiliary fuel burner is equipped to use fuel oil and/or natural gas and to also use high heating value non-regulated waste liquids. The auxiliary fuel system is used only for start-ups or when necessary to maintain the desired temperature in the combustion chamber. The capacity of the auxiliary fuel system is 1,500 lb/hr for fuel oil, and 36,000 SCFM for natural gas.

1.2.1.2 Liquid Waste Feed Systems

The characteristics of the liquid waste are described in Tables 1 and 2. The waste characteristics provided in the tables are typical of "as-fired" conditions; although FFCC may occasionally blend waste by moving material between tanks to meet desired waste feed characteristics and regulatory requirements. Wastes are generally fed from the tanks permitted under RCRA Permit 11H-M005.

Liquid waste can also be burned directly in the incinerator from containers. FFCC uses various portable containers to transfer wastes to the treatment facility (e.g., 750-gallon portable dumpsters, tankers, etc.). The portable containers are used for special manufacturing situations and for wastes that have the potential to be incompatible with the waste stored in FFCC feed tanks.

Centrifugal pumps are used to transfer waste from the storage tanks and containers to the incinerator through a variety of lines. The waste is segregated and fed based on Btu value (liquid aqueous waste and liquid organic waste). The flow rate of the liquid wastes is continuously monitored by mass flow meters and can be adjusted from the control room by control valves located at the incinerator.

The feed rate design capacity of the liquid aqueous waste feed is 13,500 lb/hr. The design capacity of the liquid organic waste feed is 4,500 lb/hr.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 7 of 42

1.2.2 Prime Mover

The combustion air blower is a Model 67103A centrifugal fan manufactured by Hoffman. The design conditions for the blower are 15,300 standard cubic feet per minute at 100 degree F, at 14.55 PSIA, 144 in. w.c., with a 500-horsepower motor. The incinerator is operated as a forced draft system. The system will be maintained sealed to prevent fugitive emissions.

1.2.3 Waste Liquid Injection Nozzles

The incinerator uses a total of 10 nozzles. The three nozzles used to burn liquid organic waste are Turbotak air-atomized nozzles. The six aqueous waste nozzles are John Zink type with Delavan tip air-atomized nozzles. The auxiliary fuel nozzle is a Turbotak air atomized nozzle. A minimum atomizing air pressure of 30 psig was specified by the manufacturer.

1.2.4 Description of Rapid Quench System

Combustion gases leave the combustion chamber and enter a T-Thermal SUB-X® quench system. The hot gases enter a weir tank through a water-washed downcomer. The weir tank is a carbon steel tank that is FRP (Derakane 510A Resin with 5% antimony trioxide) lined. The dimensions of the weir tank will be 7'-6" I.D. x 13'-9" overall height (OAH). Salts dripping down from the incinerator outlet drop into and are dissolved in the water bath. The downcomer directs the flow of hot gases approximately 24" below the surface of the water bath in the weir tank. The downcomer is constructed of alloy C-22 and has a 64" I.D. The gases are then forced into the water bath as they pass through openings near the bottom of the downcomer. As the gases mix with the water, they transfer heat to the water bath. The saturated gases then rise between the downcomer and the weir tank wall, carrying some of the quench water into the quench separator.

The quench separator is constructed of FRP. The dimensions of the quench separator are $10^{\circ}-6^{\circ}$ I.D. x 28'-2" OAH. It takes the liquid laden gases and reduces the velocity of those gases. As the velocity is dropped, the entrained liquid disengages and drops into the quench separator sump. The exit temperature of the combustion gases leaving the rapid quench system will be approximately 190 deg F. Makeup water to this system is primarily the blowdown from the Hydrosonics scrubber. Blowdown from the quench is pumped to the plant permitted wastewater treatment facility.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 8 of 42

1.2.5 Automatic Waste Feed Cutoff System

The primary function of the Automatic Waste Feed Cutoff (AWFCO) system interlocks is to prevent the feeding of waste if operating conditions are outside the permit limits. The AWFCO system is controlled by the incinerator facility Programmable Logic Controller (PLC).

The FFCC incinerator is equipped with a PLC-based control system and a central data acquisition system (DAS) which provide the necessary monitoring and control. The control system is composed of a Siemens Simatic 505 series controller and four operator interface terminals. The operator interfaces are personal computers running the Techomatix FactoryLink^C human/machine interface (HMI) software. The two primary operator interfaces are independently connected to the PLC. Two additional stations are located in the site powerhouse. These stations are connected to the primary operator interfaces in a client/server arrangement via the plant Ethernet network. The primary operator can use the operator interfaces to modify the incinerator operation within defined operating parameters if necessary. A thirty day historical dataset of one minute samples is maintained on the operator interfaces connected to the PLC. This data can be used to back-populate data into the central data acquisition system in the event that the central data acquisition system becomes unavailable for an extended period of time.

The central data acquisition system (DAS) is composed of a local data gathering node and a remote data archiving node. This system uses the OSISoft PI[°] software. The local data gathering node is implemented using the PI[°] OPC interface software on the two PCs that are connected directly to the PLC. Whichever of these two PCs is currently designated as the primary server will function as the local data gathering node. This node is connected to the remote data archiving node via the plant Ethernet network. The remote data archiving node is implemented using the PI[°] server software. The local data gathering node is referred to as a PI-API Node, and the remote data archiving node is referred to as the PI Home Node. The PI Home Node is the primary data repository. In the event that the PI Home Node is unavailable, data is buffered on the PI-API Node until the PI Home Node is again available. The PI-API Node polls the HMI software for the analog operating variables at least every 15 seconds. This data is transmitted to the PI Home Node on an exception basis and is stored in a compressed format.

The PLC monitors and controls the incinerator operations including all required permit operating parameters. If any parameter exceeds a shutoff limit that is set at a high percentage of its regulatory limit, the PLC initiates an automatic waste feed cutoff by de-energizing the signal that allows the waste feed dual block valves to open. This loss of signal immediately closes the waste feed dual block valves and prohibits waste from entering the incinerator. Radiant heat in the incinerator will not allow the combustion chamber temperature to drop significantly during the cutoff of liquid waste. Therefore, any residual liquid waste fuel in the incinerator will be burned. In accordance with regulatory requirements, FFCC will test the waste feed cutoff system once each week during normal operation.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 9 of 42

In order to minimize the number of waste feed cutoffs, the operator interface will also alarm when any of the variables reach a high percentage of their shutoff limits. This early warning will permit the incinerator operator to take necessary corrective action before the shutoff limits are exceeded. The early alarm does not activate the AWFCO system. The AWFCO system is activated immediately whenever a permit related parameter that is tied to the AWFCO system is exceeded. A first-out system implemented in the PLC software will indicate which parameter caused the automatic waste feed cutoff, either tests or actual. A numerical code indicating the trip condition is recorded in the DAS.

1.2.6 Description of Air Pollution Control Equipment

A Hydrosonics two-stage high-energy scrubber treats the combustion gases from the incinerator. The saturated gases are introduced into the scrubber first stage via a sub-sonic nozzle forming a free jet. Scrubbing liquor is sprayed into the periphery of the exhaust of the free jet. This forms a water curtain of fine high-speed water droplets through which the gases must pass, thereby removing the particulate from the gas. Acids in the gas stream are simultaneously removed by the scrubbing liquor. Addition of caustic to the scrubber liquor maintains its pH at a level high enough to maintain acid-scrubbing capability. Large droplets formed fall and are drained from the first stage, while gases pass through to the second stage. The second stage operates in the same manner as that of the first. Gases leaving the second stage then pass through a vane demister prior to exhausting through the Stack to the atmosphere. The design liquid flow rate of the scrubber is 500 gallons per minute in the first stage and 300 gallons per minute in the second stage. Plant process water is used as makeup to this recirculating system, which is blown down to the rapid quench weir tank.

1.2.7 Incinerator Stack

The new exhaust stack has a 48" I.D. and is 84 feet in height. The top of the stack will be 20 ft higher than the top of the combustion chamber. The stack will be constructed of FRP and will have 4 different sample ports located at three different levels on the stack, for total of 12 sample ports. At each level there will be four 6" sample ports at 90-degree angles in a horizontal plane.

1.2.8 Location and Description of Temperature, Pressure, and Flow Indicating and Control Devices

FFCC uses a variety of devices to control the operation of the incinerator system. A complete list of the instruments including their description and location can be seen in Tables 4 and 5.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 10 of 42

1.2.9 Stack Gas Monitoring System

FFCC continuously monitors carbon monoxide (CO) and oxygen (O₂) levels in the combustion stack gases. The gas analyzer will be a high performance multiple gas analyzer. The gas analyzer's carbon monoxide (CO) monitor will be an IR adsorption device capable of measuring CO at levels from 40 ppb to 3,000 ppm. The oxygen monitor operates using paramagnetic transduction and is capable of measuring oxygen concentrations ranging from 0% - 25% in the flue gas.

1.3 Incinerator Operator Training and Certification

All incinerator operators receive initial training by completing the Utilities Operator Apprentice Program. This is a site-specific procedure based training program that has been reviewed and certified by the state Department of Labor. The training includes written materials, hands-onlabs, and on-the-job training. Once an apprentice operator has received training for a task or course and has completed that section of the Apprenticeship Program the operator is then allowed to do that task by their self.

A certified operator will instruct the apprentice until they have successfully completed the program and passed a written examination administered by the instructor. The completions are recorded in an on-site training management system.

An annual review is administered to the operators and critical staff that include both Title V, MACT and RCRA updates along with pertinent operational training, including but not limited to the SSM plan. The Utilities Operator Apprentice Program, including the Incinerator Operator Training, is reviewed and updated whenever changes have been made that could improve the training or make the existing training obsolete.

A certified control room operator will be on duty at the site when the incinerator is operating.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 11-of 42

2.0 Sampling and Analysis Program

This section summarizes the sampling and analysis programs conducted during the CPT.

2.1 <u>Sampling Program</u>

FFCC fed the following streams during the CPT:

- I. Hazardous waste
 - a. High-Btu Organic Waste
 - b. Low-Btu Aqueous Waste
- 2. Metals Spiking Solution
- 3. Natural Gas

2.1.1 Waste Feed Preparation

Organic Waste - The organic waste was prepared by FFCC's manufacturing unit in a 2,000gallon reactor. The stream was prepared as a 40% toluene, 36% methanol, 2% chlorobenzene, and 22% perchloroethylene solution. The purpose of the stream was to provide a high Btu waste feed that contained a primary organic hazardous constituent (POHC) and as well as significant amount of chlorides.

Aqueous Waste – The aqueous waste was prepared in FFCC's waste feed tank. The stream was prepared to provide an 87% water and 13% sodium chloride solution. The purpose of the stream was to provide a low Btu waste feed that consisted of 13% total ash content.

Metals Spiking Solution – The metals spiking solution was prepared by Blue Ridge Chemicals, Inc. headquartered in Bala Cynwyd, Pennsylvania. This solution was prepared in order to establish MACT EEE feed limits for semi-volatile (cadmium and lead) and low-volatile (arsenic, beryllium, and chromium) metals. Lead was chosen to represent the semi-volatile metals and chromium was chosen to represent the low volatile metals. The solution was ordered to exact specifications and certified by Blue Ridge Chemicals as to its composition (certificate of analysis is located in Attachment E-1). The high volatile metal, mercury, was not spiked in to the feed stream. Mercury is not expected to be found in FFCC's waste and the feedrate limit will be based on the maximum theoretical emission concentration (MTEC).

Prepared by FutureFuel Chemical Company, Batesville, Arkansas, and Risk Management and Engineering, Ltd., Dallas, Texas

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 12-of 42

Natural Gas – The natural gas, used as auxiliary fuel support during the testing, and is supplied by CenterPoint Energy Arkla. The natural gas is a high Btu process stream that is absent of ash, chlorine and POHCs.

2.1.2 <u>Sampling and Monitoring Procedures</u>

The CPT consisted of three replicate runs at the extreme range of normal conditions. Each run involved stack emission sampling for POHC destruction to ensure 99.99% DRE and to demonstrate compliance with the THC, CO, particulate matter, hydrogen chloride/chlorine, mercury, semi-volatile metal, low-volatile metal and D/F emission standards. Waste feed and process samples were obtained during each run of the CPT.

The procedures for collecting samples are summarized in Table 3.0. Sampling frequency and reference methods also are included in Table 3.0. Additional details regarding sampling frequencies and methods follow.

2.1.3 Volatile Organic Analysis Samples

The volatile organic analysis (VOA) samples of the high-btu wastes were collected from a tap in the waste feed line prior to combustion. The VOA samples of the low-btu wastes were collected from a tank outlet upstream from the metals spiking point. Grab samples for the VOA were taken every 30 minutes, packaged separately in 40ml vials, and then placed in ice and cooled to 4 degrees C. These samples were composited by the laboratory before analysis.

2.1.4 Organic Waste Samples

The organic waste samples were also collected from a tap in the waste feed line for analysis of non-organic constituents. These grab samples were taken every 30 minutes in 8-oz jars and then placed in an ice cooled environment and transported to the designated analytical laboratory for analysis. These samples were composited by the laboratory prior to analysis.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 13 of 42

2.1.5 Aqueous Waste Samples

The aqueous waste samples were also collected from a tank outlet, upstream from the metals spiking point, for analysis organic and non-organic constituents. These grab samples were taken every 30 minutes in 8-oz jars and then placed in an ice cooled environment and transported to the designated analytical laboratory for analysis. These samples were composited by the laboratory prior to analysis.

2.1.6 <u>Metals Spiking Solution</u>

The metals spiking solution was not sampled as a separate waste stream. It was prepared off-site by a third party who certified the composition. Attachment E-1 contains a copy of that certification. The solution was fed to the liquid waste feed line upstream of the combustion chamber injection point, and down-stream of the point where the organic and aqueous waste samples were taken. The feed rate of the metals solution was controlled using a chemical metering pump and a scale. The feed rate was maintained by calculating the weight loss of the container holding the metals solution. The scale calibration documentation can be found in Attachment C.

2.1.7 <u>Combustion Gas</u>

2.1.7.1 Combustion Gas Temperature

The combustion gas temperature was measured in the combustion chamber by thermocouple with an instrument range of 0 - 2200 degrees Fahrenheit. This thermocouple is located about 18 inches above the bottom of the cylindrical section of the oxidizer.

2.1.7.2 Combustion Gas Monitoring

A stack gas monitoring system was used to monitor the carbon monoxide and oxygen content of the flue gas. The carbon monoxide (CO) monitor was an IR adsorption device capable of measuring CO at levels from 40 ppb to 3,000 ppm. The oxygen monitor operates using paramagnetic transduction and is capable of measuring oxygen concentrations ranging from 0% – 25% in the flue gas. A contractor-supplied total hydrocarbon (THC) monitor was utilized in parallel with these instruments.

Prepared by FutureFuel Chemical Company, Batesville, Arkansas, and Risk Management and Engineering, Ltd., Dallas, Texas

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 14 of 42

2.1.8 <u>Stack Gas</u>

Stack gas was collected from the incinerator stack. The stack's sampling ports were designed to meet all stack gas sampling needs (e.g., isokinetic sampling, etc.).

2.1.8.1 Hydrogen Chloride, Chlorine, and Particulate Train (Method 26A)

The Method 26A hydrogen Halide and Halogen isokinetic sampling train was used to collect hydrogen chloride and chlorine samples during the three runs of the CPT. The total sampling time was about 3 hours during each replicate sampling run. The Method 26A train was operated concurrently with the other sampling train(s) to sample stack gas. Gaseous and particulate pollutants were withdrawn isokinetically from the source and collected on a filter, and in absorbing solutions. The Method 26A procedure includes measurement of the stack gas flow rate and temperature according to EPA Methods 1 and 2. EPA Method 3Awas used to determine oxygen and carbon monoxide concentration in the stack. EPA Method 4 was used to determine the stack gas moisture content.

2.1.8.2 Multi-Metals Train (Method 29)

An MMT (multi-metals train) was used for collection of metals from the stack gas during the CPT. The sampling train impingers were charged with a solution of 5 percent nitric acid and 10 percent hydrogen peroxide to capture metals (i.e., arsenic, beryllium, cadmium, chromium, lead, and mercury) and acidified potassium permanganate (composed of 4 percent potassium permanganate and 10 percent sulfuric acid) to capture any mercury that was not captured by the nitric acid and hydrogen peroxide solution. The total sampling time was about 3 hours during each replicate sampling run. The MMT was operated concurrently with the other sampling train(s) to sample about 3 cubic meters of stack gas. The MMT procedure includes measurement of the stack gas flow rate and temperature according to EPA Methods 1 and 2. Carbon monoxide and oxygen determinations were made by Method 3A. Stack gas moisture was sampled in accordance with EPA Method 4. The Method 26A stack measurement of these parameters was used for the MMT calculations.

2.1.8.3 Volatile Organic Sampling Train (Method 0030)

A volatile organic sampling train (VOST) was used during the CPT to collect the POHC chlorobenzene from the stack gas on sorbent resin. The VOST was configured in accordance with SW-846 Method 0030 with two Tenax® resin tubes and one Anasorb® tube in series. The VOST was operated concurrently with the other sampling trains to collect a total of four sets of VOST cartridges for each test run. Three sets were targeted for analysis. The fourth set served as a backup in the event of tube breakage or damage during shipment and laboratory handling. About 20 liters of stack gas were sampled per set of VOST cartridges at 0.5 liters per minute for 40 minutes (slow-VOST conditions). The VOST cartridges were capped immediately upon removal from the train, wrapped in aluminum foil, placed in glass tubes, and sealed. The Method 23A stack parameter measurements were used for the VOST calculations.

2.1.8.4 Dioxin/Furan Sampling Train (Method 23A)

An MM5 sampling train was used for collecting PCDDs and PCDFs during the CPT. The sorbent trap contained XAD-2 resin to capture organics, and the impingers were filled with water. The total sampling time for this train was about 3 hours during each replicate sampling run. The sampling train was operated to sample approximately 4 cubic meters of stack gas. The sampling train was operated and recovered according to the procedures described in Method 0023A for PCDDs and PCDFs.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 16 of 42

2.2 <u>Analytical Program</u>

Maxxam Analytics laboratory headquartered in Mississauga, Ontario. This laboratory was used for all stack sampling analysis.

American Interplex is located in Little Rock, Arkansas and was used for all process feed stream analysis conducted during the CPT.

2.2.1 <u>Analytical Methods</u>

A summary of the analytical methods and procedures are summarized in Table 3.0 Detailed analytical methods and procedures for the stack samples are described in Attachment A. Analytical methods and procedures for the feed stream analysis can be found in Attachment D

2.2.2 <u>Analytical Results</u>

The stack sample analytical results are located in Attachment A. The process feed analytical results are summarized in Table 6.0. The actual feed stream analysis can be found in Attachment D.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 17 of 42

3.0 Description of the Test Program

This section summarizes the comprehensive performance test plan. The plan¹ contains the details of the test that was conducted October 12 - 14, 2010. The Executive Summary in this NOC discusses those deviations from the plan that were necessary to address unexpected or unplanned conditions.

3.1 <u>CPT Objectives</u>

The overall objective of this Comprehensive Performance Test is to demonstrate that the incinerator is capable of meeting the incinerator emission standards established in the MACT Combustion Rule (40 CFR 63 Subpart EEE) and to establish operating conditions and parameters that are equivalent with that demonstration.

The specific objectives for the FFCC comprehensive performance test are listed below. Each emissions parameter will be reported with concentration corrected to 7% Oxygen.

- Demonstrate 99.99 percent DRE of the designated POHC (chlorobenzene).
- Demonstrate control of particulate emissions to less than or equal to 0.013 grains per dry standard cubic foot (gr/dscf) at maximum ash feed rates.
- Demonstrate that hydrogen chloride/chlorine emissions are less than or equal to 32 ppmv at maximum chloride feed rates.
- Demonstrate that mercury emissions are less than or equal to 130 micrograms per dry standard cubic meter (ug/dscm).
- Demonstrate that semi-volatile metal (cadmium and lead) emissions are less than or equal to 230 micrograms per dry standard cubic meter (ug/dscm at maximum semi-volatile metal feed rates.
- Demonstrate that low-volatile metal (arsenic, beryllium, and chromium) emissions are less than or equal to 92 micrograms per dry standard cubic meter (ug/dscm) at maximum low-volatile metal feed rates.
- Demonstrate that carbon monoxide (CO) emissions are less than 100 ppmv (dry) on an hourly rolling average basis.

¹ Submitted August 1, 2008.

Prepared by FutureFuel Chemical Company, Batesville, Arkansas, and Risk Management and Engineering, Ltd., Dallas, Texas

- Demonstrate that total hydrocarbon emissions are less than 10 ppmv (dry) on an hourly rolling average basis.
- Demonstrate that dioxin/furan emissions are less than or equal to 0.40 nanograms total equivalent quotient per dry standard cubic meter (ng TEQ/dscm).
- Gather data regarding waste feed characteristics and process operating conditions that will be used to develop operational permit limits that will ensure compliance with regulatory performance standard.

3.2 <u>CPT Approach</u>

The CPT is based upon 40 CFR 63 Subpart EEE. 40 CFR 63.1207 establishes the requirements for conducting a comprehensive performance test. The universal approach establishes one set of permit conditions or limits applicable to all modes of operation. This approach allows FFCC to treat the complete variety of wastes produced by the FFCC facility under one well-defined set of operating limits. Operating limits have been derived from data assembled during the CPT.

3.2.1 POHC Selection

FFCC has based its selection of POHC on two primary factors: (1) the University of Dayton thermal stability ranking, and (2) the composition of the actual wastes to be burned. Consistent with these factors, FFCC used the following criteria to select the POHC for this CPT:

- The POHC should be present as an Appendix VIII constituent in the actual wastes to the maximum extent practicable.
- The POHC should be considered an organic hazardous air pollutant under 42 U.S.C. 7412(b)(1), and
- The POHC should have a high ranking on the University of Dayton thermal stability-ranking list.

The constituents that are likely to be found in FFCC's liquid waste are listed in Table 2. Many of these compounds are included on the University of Dayton thermal stability ranking including acetonitrile, benzene, chlorobenzene, and toluene. Chlorobenzene was selected as the POHC because it is a Class I compound, it is typically present in greater quantities in the waste than the other Class I compounds, and it is readily available.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 19 of 42

3.3 <u>CPT Program</u>

The proposed CPT program consists of one test of three replicate runs. The target operating conditions are designed to establish operating limits for the incinerator that will ensure all MACT combustion standards are met.

3.3.1 Comprehensive Performance Test

The CPT was conducted at the extreme range of normal conditions as prescribed in 40 CFR 63.1207. It was conducted at low combustion chamber temperature to demonstrate DRE of the designated POHC at extreme conditions for organic hazardous constituent destruction. In addition, the incinerator combustion gas velocity was maximized in order to decrease gas residence time. The liquid waste feed was maximized in order to show the DRE performance can be met at maximum allowable feed rates. Other operating conditions were set at extremes as above to assure that normal emissions are less than what is measured herein. This test condition was also used to demonstrate that Dioxin/Furan, carbon monoxide, total hydrocarbon, mercury, semi-volatile metal, low-volatile metal, hydrogen chloride/chlorine, and particulate emission standards are being met.

The aqueous waste feed was a solution of waste, metal salts, and sodium chloride designed to represent low Btu waste feed, as well as provide ash and chlorides during the CPT. The organic waste feed was a mixture of chlorobenzene, perchloroethylene, methanol, and toluene designed to represent high Btu waste feed, as well as provide POHC (chlorobenzene) and organic chlorides (chlorobenzene, perchloroethylene).

3.4 <u>CPT Sampling and Analytical Protocols</u>

The CPT sampling and analysis program is summarized in Table 3. The structure of this CPT is based on the previously stated objectives.

The stack was sampled during the CPT as follows:

- The volatile POHC chlorobenzene were sampled using the VOST train (SW-846 Method 0030).
- Hydrogen chloride, chlorine, and particulate emissions combined, were sampled using the EPA Reference Test Method 5/26A..
- PCDD/PCDFs were sampled using the SW-846 Method 0023A.
- Metals, including Mercury, were sampled using EPA Referenced Test Method 29.

- CO testing was done with the facility's permanent CEM system.
- Total Hydrocarbon (THC) testing was conducted using EPA Reference Test Method 25A.

In addition to stack sampling, the feed material (liquid wastes), and process water was sampled and analyzed for a number of physical and chemical parameters. These methods are summarized in Table 3.0. The actual waste analytical results and QA/QC can be found in Attachment D. A summary of the waste feed results can be seen in Table 6.0.

4.0 <u>Comparison of Actual Test Conditions versus Planned Conditions and</u> <u>Independent QA Review</u>

A comparison of the actual test conditions experienced during the CPT with the planned (target) test conditions documented in the CPT plan is provided below. Tables 10.0, 11.0, and 12.0 provide a summary of the various operating parameter data with respect to the target conditions for each run.

Comparisons were made on three sets of parameters:

- Combustion Device Parameters
- Feed Rate Parameters
- Constituent Feed Rate Parameters
- Stack Gas Parameters

4.1 <u>Combustion Device Parameters</u>

These parameters were used to establish limits for ensuring proper combustion and pollution control device operation. They include:

- Combustion chamber temperature In general, the average combustion chamber temperature exceeded the target by approximately 58 °F due to higher than anticipated waste heating values.
- Gun Atomization Pressure This parameter is based on manufacturer's recommendations. The atomization pressure during the CPT is consistent with those recommendations.
- Combustion Air Flow Rate The combustion air flow rate is consistent with the target operating conditions.
- Scrubber Pressure Drop The pressure drop across the scrubber is consistent with the target conditions.
- Scrubber Water pH The pH of the scrubber water is consistent with the target conditions.
- Quench Tank pH This is a new parameter that was established during the CPT based on FFCC's observation that chloride scrubbing efficiency is affected by this parameter. The values demonstrated during the CPT are consistent with FFCC's anticipated operations.

- 1st Stage Scrubber Flow The 1st Stage scrubber flow rate is consistent with the target conditions.
- 2nd Stage Scrubber Flow The 2nd Stage scrubber flow rate is less than target operating conditions by approximately 49 gpm.
- Scrubber Blowdown Rate The scrubber blowdown rate is consistent with the target conditions.
- Scrubber Recirculation Tank Level The scrubber recirculation tank level is consistent with the target conditions.

4.2 Feed Rate Parameters

These parameters were used to establish constituent and POHC feed rates. They include:

- Total Waste Feed Rate The feed rate demonstrated during the test, although less than described in the CPT Plan, should be sufficient for continued operation.
- Low BTU Aqueous Waste Feed Rate The feed rate demonstrated during the test, although less than described in the CPT Plan, should be sufficient for continued operation.
- High BTU Organic Waste Feed Rate The feed rate demonstrated during the test, although less than described in the CPT Plan, should be sufficient for continued operation.

4.3 Constituent Feed Rate Parameters

These parameters were calculated from the fuel feed rate parameters and the constituent concentrations in the fuel feed for the CPT Test. They include:

- Total Chloride Feed Rate In general, the overall average total chloride feed rate was 13% lower than the target value.
- Total Ash Feed Rate In general, the overall average total ash feed rate was 12% lower than the target value.
- Total Mercury Feed Rate Mercury was not detected in the waste feed, but the detection limits were slightly better than anticipated, resulting in a lower mercury feed rate than the target value.

- Total Semi-Volatile Metal Feed Rate The overall average SVM feed rate was higher than the target conditions due to increased spiking rates.
- Total Low-Volatile Metal Feed Rate The overall average LVM feed rate was higher than the target conditions due to increased spiking rates.

4.4 <u>Stack Gas Parameters</u>

These parameters were monitored by a continuous emission monitor. They include:

- Total Hydrocarbon Concentration Emissions were well below the standard as expected.
- Carbon Monoxide Concentration Emissions were well below the standard as expected.

4.5 Independent QA Review

Mr. David A. Weeks, P.E., BCEE, CIH (Risk Management and Engineering, Ltd.) provided independent oversight of the CPT and conducted an independent review of the quality assurance and quality control (QA/QC) performed by the stack testing company and laboratories during the test. A summary of the stack testing QC measurements is included in Section 4 of the stack test report (provided as Attachment A) in addition to the discussion contained in this section.

4.5.1 Stack Testing Equipment

Mr. Weeks was present during the entire CPT and inspected the stack sampling calibration records including the dry gas meter calibration records, pitot tube calibration records, and temperature indicator calibration records at the time of the test. Mr. Weeks observed the leak checks of the sampling equipment during the tests and found them to be in good order. The post calibration records were in order and showed that the stack gas sampling equipment functioned properly and provided data that were in accordance with the U.S. EPA methods used to collect the data. The calibration records are provided in Appendix D to Attachment A.

4.5.2 Stack Gas - Particulate Analysis

Mr. Weeks conducted an independent review of the analytical data. The balance calibration data is summarized on Appendix D of Attachment A. The results showed no deviation at a standard weight of 0.5000, 10, and 50 grams, and deviations ranging from 0.0001 to 0.0002 grams at standard weights of 100 and 1 grams, respectively. The maximum deviation (0.2 mg) is less than the 0.5 mg performance criterion. The particulate quantity stated on the laboratory certificate (see Appendix C of Attachment A) matches the particulate quantity used by Alliance to calculate

Incinerator Notice	Incinerator Notice of Compliance	
Revision No	Revision No 0	
Revision	014241	
Page	24 of 42	
_		

the particulate concentration. Mr. Weeks' independent review of the data confirms the results reported by Alliance in their report.

4.5.3 Stack Gas – HCl/Cl₂ Analysis

Mr. Weeks conducted an independent review of the analytical data. The available QA/QC data is summarized in Appendix C to Attachment A. The results show that the recovery of HCl from matrix spikes and spiked blanks was 99%. The recovery of Chlorine from matrix spikes and spiked blanks was 102 and 101% respectively. The QA/QC Plan does not provide specific performance criteria for this method because the method itself is very limited on QA/QC requirements. The Method states that the results of audit samples are the primary means of quality assurance/control other than the sampling checks that are prescribed by Method 5. The Method recommends that the difference between the known value of an audit sample and the measured value be less than 10%. As demonstrated above, recoveries of matrix and spiked blanks were within 1 to 2 percent of the known spiked quantity. Thus, the available QC data for this method demonstrate that the laboratory is generally performing within the QC limits specified by the method. Further, the laboratory did analyze a laboratory duplicate using the Run I impinger solutions. The analysis shows that the relative percent difference between the mass of HCl and Chlorine in the sample and the duplicate sample is 4.5% and -3.0%, respectively, which shows good precision in the analysis. HCl and chlorine were not detected in the laboratory blanks. The HCl and chlorine mass stated on the laboratory certificate matches the HCI and chlorine mass used by Alliance to calculate the total chloride concentration. Mr. Weeks' independent review of the data confirms the results reported by Alliance in their report.

4.5.4 Stack Gas - Metals Analysis

Mr. Weeks conducted an independent review of the analytical data. The stack sampling contractor's QA/QC summary is contained in Section 4 of Attachment A. The laboratory QC data is contained on pages in Appendix C to Attachment A. The matrix spike and matrix spike duplicate recoveries were within the 70 - 130 % accuracy objective as defined in the QAPP. Matrix spikes were performed for mercury on all the components of the blank train and portions of the Run 1 samples. (The QAPP required at least one matrix spike per test.) The mercury recoveries from the matrix spikes ranged from 87 to 112%, showing that metals were not lost during the digestion process. The recoveries for semi- and low-volatile metals ranged from 87 to 107%. The RPD between the matrix spike and matrix spike duplicate analyses were less than the 25% objective established in the QAPP, with the results ranging between 0 to 4% RPD for all metals. Thus, laboratory analyses were not biased high or low. The mass of metals stated on the laboratory certificate matches the mass of metals used by Alliance to calculate the concentration of metals in the stack gas. Mr. Weeks' independent review of the data confirms the results reported by Alliance.

Prepared by FutureFuel Chemical Company, Batesville, Arkansas, and Risk Management and Engineering, Ltd., Dallas, Texas

Incinerator Notice of Compliance Revision No 01 12 11 Page 25 of 42

4.5.5 Stack Gas - Volatile Organics Analysis

Mr. Weeks conducted an independent review of the analytical data. The laboratory's QC discussion and surrogate spike recovery data is contained in Appendix C to Attachment A. Alliance reports, and Mr. Weeks confirmed by inspection of the laboratory data sheets, that surrogate recoveries ranged from 86 to 106 percent as compared to the target objective of 50 - 150%. The surrogate recovery for the spiked blanks and method blanks was 94 to 116%. The RSD for the recoveries of the surrogate toluene-d8 calculated as 5.4% as compared to the QAPP target of less than 35% RSD. The mass of chlorobenzene stated on the laboratory certificate matches the mass of chlorobenzene used by Alliance to calculate the concentration of chlorobenzene in the stack gas. Mr. Weeks' independent review of the data confirms the results reported by Alliance.

4.5.6 Stack Gas – PCDD/PCDF

Mr. Weeks conducted an independent review of the analytical data. The laboratory's QC discussion and the surrogate spike recoveries are reported with the analytical results on pages in Appendix C to Attachment A. Alliance reports, and Mr. Weeks confirmed by inspection of the laboratory data sheets, that internal standard surrogate recoveries for tetra to hexa-substituted congeners ranged from 50 to 123 percent as compared to the target objective of 40 - 130%. The internal standard surrogate recoveries for hepta to octa-substituted congeners ranged from 88 to 116 percent as compared to the target objective of 25 - 130%.

Further, the sampling surrogate recoveries for tetra, penta, hexa, and hepta-substituted ranged from 101 to 141% as compared to the target QC criteria of 40 - 130%. The Run 2 backhalf sampling surrogate C13-2,3,4,7,8-Penta CDF was the sole sampling surrogate the exceeded the target QC criteria at 141%. The laboratory reported that the excessive recovery was due to a matrix interference. Thus, the mass of penta-CDF reported by the laboratory may be biased high by a minor quantity. This exception to the QC target criterion does not compromise the results because although the recovery of penta-CDF is biased high, the stack concentrations are still below the regulatory standard.

The precision QC criterion ($\leq 50\%$) was met for the internal standards with the precision of the front half (FH) internal standards demonstrating a 21% relative standard deviation (RSD) and the back half internal standards demonstrating a 26% RSD. Likewise, the precision of the sampling surrogates met the QC criterion of $\leq 50\%$ with a 10% RSD. (Recall that sampling surrogates are not added to the front half of the dioxin/furan sampling train.)

Finally, the mass of dioxin and furan stated on the laboratory certificate matches the mass of dioxin and furan used by Alliance to calculate the concentration of dioxin and furan in the stack gas. Mr. Weeks' independent review of the data confirms the results reported by Alliance.

Incinerator Notice of Compliance Revision No 014244 Page 26 of 42

4.5.7 <u>Stack Gas – Continuous Emissions Monitoring Data</u>

Mr. Weeks conducted an independent review of the calibration data for continuous emissions monitoring (CEM) data contained in the Alliance report and those records that demonstrate calibration of the Future Fuel CEMs. The Alliance data was used to measure the oxygen and carbon dioxide concentration in the stack gas for the purpose of emission calculations and THC concentration. The calibration data is contained Appendix D to Attachment A.

Mr. Weeks also conducted an independent review of the calibration data for continuous emissions monitoring (CEM) data for the FFCC CEMs. The FFCC CEMs analyzers were used to demonstrate compliance with the CO standard. The FFCC calibration records are contained in Attachment C to the CPT Report.

The Alliance O_2/CO_2 CEMs was calibrated daily and demonstrated values ranging from 0.0 to 0.5% of span. This difference is less than the 2% criterion required by the method. Likewise, the bias ranged from 0.0 to 0.8% of span (which is less than the 5% criterion required by the method) and the drift ranged from 0.0 to 0.3% of span (which is less than the 3% criterion required by the method.) The Alliance THC analyzer demonstrated daily calibration ranging from 0.0 to 1.3% of the Protocol gas concentration, which is less than 5% criterion; and the drift was measured at 0.0 to 0.1% which is less than the 3% of span required by the method.

The data from the October 15, 2010 RATA for the FFCC CEMs was also reviewed. The RATA data showed that the CO CEMs demonstrated a 0% difference between the reference method data and the FFCC CEMs, which is within the performance criterion of $\leq 5\%$. The O₂ CEMs demonstrated a 0.1% difference, which is less than the performance criterion of $\pm 1\%$. The CO daily calibration drift during the test ranged from 0 to 2.5% of span, which is less than the 5% criterion in the performance specification, and the O₂ monitor demonstrated no difference between the reference gas and the monitor response. Thus, the FFCC CEMs provided good data that can be used to evaluate compliance with the CO standard. (The daily calibration drift data for the FFCC CEMs is provided in Attachment C to the CPT report.)

4.5.8 <u>Waste-Volatile Organics</u>

The QAPP does not identify specific QA/QC objectives for the waste sampling and analytical procedures. However, the laboratory reported the results of the QC measurements collected during the analysis of these samples. Mr. Weeks reviewed this data in order to ensure the overall quality of the project data. The laboratory's analytical data and QC report for the waste analyses is contained in Attachment D. No anomalies were identified from the review of the data. One QA test that was implemented for the test was the collection of duplicate field samples during Run 2. The sample results for the high-BTU organic waste exhibited a chlorobenzene

Incinerator Notice	e of Compliance
Revision No	t F
Revision	014244
Page	27 of 42

concentration of 3.7% whereas the field duplicate exhibited a concentration of 3.9%. The relative percent difference (RPD) between the two values is -5.4%, which demonstrates good agreement. Likewise, the sample results for the low-BTU aqueous waste demonstrated a concentration of 1.0 mg/kg whereas the field duplicate exhibited a concentration of 1.1 mg/kg, which shows a RPD of -10%.

The laboratory also ran an internal batch duplicate analyses using the Run 1 sample material for the organic waste and Run 2 sample material for the aqueous waste. The laboratory RPD for the organic waste and aqueous waste was 6.04% and 8.05%, respectively. The toluene-d8 surrogate recoveries for these laboratory quality control samples ranged from 102 to 106%. Thus, RME concludes from this review that the concentration of chlorobenzene in the waste samples used to calculate DRE are accurate and precise.

4.5.9 <u>Waste Feed-Non-Organics</u>

Similar to the volatile organics, the QAPP does not identify specific QA/QC objectives for the metals, chloride, and other non-organic waste sampling and analytical procedures. However, the laboratory reported the results of the QC measurements collected during the analysis of these samples. Mr. Weeks reviewed this data in order to ensure the overall quality of the project data. The laboratory's analytical data and QC report for the waste analyses is contained in Attachment D. No anomalies were identified from the review of the data. One QA test that was implemented for the test was the collection of duplicate field samples during Run 2. The results of the field duplicate analysis are summarized in the following chart.

Parameter	Chromium	Lead	Mercury	Chloride	Specific	Heat	Ash
		Lotid		Cinonae	Gravity	Content	
			Organie	e Waste			
	mg/kg	mg/kg	mg/kg	0. - 0	NA	BTU/lb	0 a
R2 Sample	<0.7	<4	<0.1	25	1.02	13,000	0,006
Field Dup.	<0.7	<4	<0.1	24	1.01	13,000	<0.001
RPD	0%	0%	0%	4%	0,98%	() ⁰ 'u	83,3%
Matrix							
Spike	95,8%	87,9%	NA	NA	NA	NA	NA
Recovery							
Matrix							
Spike Dup.	95,5%	87,6%	NA	NA	NA	NA	NA
Recovery							
RPD	-0,256%	-0.479%	NA	NA	NA	NA	NA
Aqueous Waste							
	mg/L	mg/L	mg/L	mg/L		BTU/lb	0 a
R2 Sample	<0,007	<0.04	<0.1	78,000	1,09	<200	12
Field Dup.	<0,007	<0.04	<0.1	78,000	1,09	<200	12
RPD	0%a	0%	0%	0%	0%	() ⁰ '0	0%0
Lab Sample	NA	NA	<0,1	15	1,09	3,800	12
Lab Dup.	NA	NA	<(),]	14	1,09	3,800	12

Prepared by FutureFuel Chemical Company, Batesville, Arkansas, and Risk Management and Engineering, Ltd., Dallas, Texas

Incinerator Notice of Compliance Revision No 0 Revision: 01.12.11 Page 28 of 42

Parameter	Chromium	Lead	Mercury	Chloride	Specific Gravity	Heat Content	Ash
RPD	NA	NA	$\Theta_{\rm D}$	6,7%a	0.6	0^{0}	0%
Matrix Spike Recovery	101%	98,9%u	86,0%a	NA	NA	NA	NA
Matrix Spike Dup. Recovery	103%	100%	88,0%	NA	NA	NA	NA
RPD	1.84%	1,08%6	2,30%	NA	NA	NA	NA

The results show that there is little to no difference between the sample and the field duplicate sample results with the exception of the ash content of the organic waste. However, given the minor amounts of ash present in the organic waste, this difference is not significant as the preponderance of the ash in the waste feed was derived from the aqueous waste. The ash content in the aqueous waste (12%) is thousands of time greater than the ash content in the organic waste (<0.001 to 0.006%). Thus, the analytical difference does not affect the feed rate limits derived from the CPT.

The MS/MSD recoveries for the metals are good and consistent with the internal laboratory QC limits, and the RPD between the MS/MSD recoveries is very small. The laboratory's QC measurements demonstrate good accuracy and precision in the waste feed sampling. Thus, the waste feed analytical results are not biased either high or low for metals other non-organic waste feed constituents.

4.5.10 Process Instruments

The process instrumentation was calibrated or checked prior to the CPT. The result of the calibrations and checks is summarized in the chart below.

Instrument	Tag No.	Difference Between Reference Value and Measured Value	FFCC Performance Standard (±)
Organic Waste Feed Gun Flow Monitor	FT-20153	0.87565%	2 ⁴ %a
Organic Waste Feed Gun Flow Monitor	FT-20155	0.50175%	<u>2</u> %,0
Organic Waste Feed Gun Flow Monitor	FT-20156	1.95508%6	20,0
Aqueous Waste Feed Ring Flow Monitor	FT-20157	-1.28320ª%	<u>2</u> ⁿ . ₀
Aqueous Waste Feed Ring Flow Monitor	FT-20355	1.27455%	$29\%_{0}$
Combustion Air Flow Rate Pressure	FT-20180	$\theta = 0.3\%$	2 ⁰ %
Atomization Air Pressure	PSL-20270	0.67 - 1.0%	2,5%

Incinerator Notice of Compliance Revision No 0 Revision: 01.12.11 Page 29.01.42

Instrument	Tag No.	Difference Between Reference Value and Measured Value	FFCC Performance Standard (±)	
to Organic Waste Feed Gun				
Atomization Air Pressure to Organic Waste Feed Gun	PSL-20271	() ⁰ 'o	2,5%	
Atomization Air Pressure to Organic Waste Feed Gun	PSL-20272	0%	2.5%	
Atomization Air Pressure to Aqueous Waste Ring Nozzles	PSL-20137	0.34%	2.5%	
Combustion Chamber Temperature Thermocouple	TT-20182	0 - 0.06%	39/0	
Combustion Clamber Temperature Transmitter	TT-20183	0.02 - 0.05%	300	
Sembber Differential Pressure	PT-20036	-0.06 - 0.25%	0.5^{0} o	
Scrubber Differential Pressure	PT-20114	0%	0.5^{9} o	
Scrubber Water pH	AIT-20043	-0.4 - 1.25%	3 ⁰ .0	
Scrubber Water pH	AIT-20044	0 ⁰ 20	326	
Scrubber Water Blowdown Flow Transmitter	FT-20012	0 - 0,25%	3%a	
Sembber Tank Level Monitor	LT-20049	0.86 - 1.6%	2%6	
Sembber First Stage Flowrate	FT-20034	-0,6 - 1,8%	3%a	
Scrubber Second Stage Flow rate	FT-20032	0%	30/0	

The calibration of all the process monitoring instruments was consistent with the FFCC performance specifications and the data can be used to develop quality operating parameter limits.

4.5.11 Conclusion

The Independent QA Review determined that the sampling and analysis was completed in accordance with the data quality objectives established for the project. Mr. Weeks opined that the data is of sufficient quality to demonstrate compliance with the standards and develop operating parameter limits.

incinerator Notice of Compliance				
Revision No	t F			
Revision Date:	011211			
Page	30 of 42			

5.0 Comparison of Test Results with Regulatory Compliance Limits

This section compares the test results of the CPT, conducted October 12-14, 2010, with regulatory compliance limits established by the MACT Combustion Rule. Table 13.0 demonstrates that FFCC hazardous waste incinerator is in full compliance with the 40 CFR 63, Subpart EEE revised emission standards for hazardous waste incinerators.

5.1 **Regulatory Requirements for the CPT**

The regulatory requirements to discuss from the CPT are:

- Destruction Removal Efficiency (DRE)
- Particulate Emissions
- HCI / Cl₂ Emissions
- Mercury Emissions
- Semi-Volatile Metals Emissions
- Low-Volatile Metals Emissions
- Dioxin/Furan Emissions
- Carbon Monoxide Emissions
- Total Hydrocarbon Emissions

5.1.1 Destruction Removal Efficiency

40 CFR 63.1203(c) requires existing hazardous waste incinerators to achieve a destruction removal efficiency (DRE) of 99.99% during the CPT for each principal organic hazardous constituent (POHC) designated in the waste feed. FFCC was able to make this demonstration during the CPT. The average DRE measured e CPT was 99.99996%.

Table 16.0 summarizes the DRE during each run of the CPT. Additional quantitative and qualitative analysis can be found in Attachment A – Stack Test Report. The POHC emission rate and feed rates can bee seen in Table 8.0 and Table 9.0.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 31 of 42

5.1.2 Particulate Emissions

40 CFR 63.1203(a)(7) requires existing hazardous waste incinerators to not emit particulate matter in excess of 0.013 grains per dry standard cubic foot (gr/dscf)) after correction to a stack gas concentration of 7% oxygen. FFCC's incinerator emitted an average of 0.0042 gr/dscf at 7% oxygen during the compliance test. A summary of the particulate emission results can be found in Table 8.0 and in Attachment A.

5.1.3 HCl / Cl₂ Emissions

40 CFR 63.1203(a)(6) requires existing hazardous waste incinerators to not emit hydrogen chloride and chlorine gas (HCl/Cl₂) in excess of 32 parts per million by volume (ppmv) after correction to a stack gas concentration of 7% oxygen. FFCC's incinerator emitted an average 10.0 ppmv HCl/Cl₂ at 7% oxygen during the CPT. A summary of the HCl/Cl₂ emission results can be found in Table 8.0 and in Attachment A.

5.1.4 Mercury Emissions

40 CFR 63.1203 (a)(2) requires existing hazardous waste incinerators to not emit mercury in excess of 130 micrograms per dry standard cubic meter (ug/dscm) after correction to a stack gas concentration of 7% oxygen. FFCC's incinerator mercury emissions were nondetect. The average detection limit was 1.3 ug/dscm at 7% oxygen during the CPT. A summary of the mercury emission results can be found in Table 8.0 and in Attachment A.

5.1.5 Semi-Volatile Metals Emissions

40 CFR 63.1203(a)(3) requires existing hazardous waste incinerators to not emit semi-volatile metals in excess of 230 micrograms per dry standard cubic meter (ug/dscm) after correction to a stack gas concentration of 7% oxygen. FFCC's incinerator emitted an average of 224.8 ug/dscm at 7% oxygen during the CPT. A summary of the semi-volatile metal emission results can be found in Table 8.0 and in Attachment A.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 32 of 42

5.1.6 Low-Volatile Metals Emissions

40 CFR 63.1203(a)(4) requires existing hazardous waste incinerators to not emit low-volatile metals in excess of 92 micrograms per dry standard cubic meter (ug/dscm) after correction to a stack gas concentration of 7% oxygen. FFCC's incinerator low-volatile emissions were non-detect in the stack gas. The average detection limit was 65.5 ug/dscm at 7% oxygen during the CPT. A summary of the low-volatile metal emission results can be found in Table 8.0 and in Attachment A.

5.1.7 Dixoin/Furan Emissions

40 CFR 63.1203(a)(1)(ii) requires existing hazardous waste incinerators to not emit dioxin and furans in excess of 0.40 toxicity equivalents per dry standard cubic meter (TEQ/dscm) after correction to a stack gas concentration of 7% oxygen. FFCC's incinerator emitted an average of 0.0032 TEQ/dscm at 7% oxygen during the CPT. A summary of the dioxin/furan emission results can be found in Table 8.0 and in Attachment A.

5.1.8 Carbon Monoxide Emissions

40 CFR 63.1203(b)(5)(i) requires existing hazardous waste incinerators to not emit carbon monoxide (CO) in excess of 100 parts per million by volume (ppmv) after correction to a stack gas concentration of 7% oxygen. FFCC's incinerator emitted an average 1.2 ppmv at 7% oxygen during the CPT. A summary of the CO emission results can be found in Table 9.0.

5.1.9 Total Hydrocarbon Emissions

40 CFR 63.1203(b)(5)(i) requires existing hazardous waste incinerators to not emit total hydrocarbons (THC) in excess of 10 parts per million by volume (ppmv) after correction to a stack gas concentration of 7% oxygen. FFCC's incinerator emitted an average 2.2 ppmv at 7% oxygen during the CPT. A summary of the THC emission results can be found in Table 8.0 and the average THC emission result can be seen in Table 9.0.
6.0 Procedures and Limitations for Operating Parameters

This section describes the operating parameter limits (OPL) and how they were established. This section also describes waste, chloride, ash and metal feed rate limits and how they are determined. Table 14.0 summarizes the operating limits established by the CPT.

6.1 Waste Feed Rate Limitations

The maximum hazardous waste feed rate will be based on an hourly rolling average (HRA). All constituent specific feed rate limits (e.g. chloride, LVM) will be based on a 12-hour HRA. Table 17.0 summarizes the waste feed rate data by run. The process data upon which Table 17.0 is based can be found in Attachment E-5.

6.1.1 <u>Maximum Hazardous Waste Feed Rate Limit</u>

FFCC established a maximum hazardous waste feed rate limit based on the average of the test run averages from the CPT. This limit will be an hourly rolling average (HRA) limit of 15,348 lb/hr. The test run averages can be seen in Table 9.0.

6.1.2 <u>Maximum Chloride Feed Rate Limit</u>

FFCC established a maximum chloride feed rate limit based on the average of the test run averages from the CPT. The concentration of chloride in the waste, along with the waste feed rate, was used to determine the limit. This limit will be a 12-hour hourly rolling average (HRA) limit of 1,428 lb/hr. The test run averages can be seen in Table 9.0.

6.1.3 <u>Maximum Ash Feed Rate Limit</u>

FFCC established a maximum ash feed rate limit based on the average of the test run averages from the CPT. The concentration of ash in the waste, along with the waste feed rate, was used to determine the limit. This limit will be a 12-hour hourly rolling average (HRA) limit of 1,453 lb/hr. The test run averages can be seen in Table 9.0.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 34 of 42

6.1.4 <u>Metals Feed Rate Limits</u>

Metals feed rate limits were also determined during the CPT. A combination of methods was used to develop the metal feed rate limits. The final feed rate limits determined from the CPT are summarized in Table 14.0.

6.1.4.1 <u>Maximum Mercury Feed Rate</u>

The maximum mercury feed rate limit is based on the maximum theoretical emission concentration (MTEC). MTEC was chosen to develop this feed rate limit because mercury was not present in the waste.

To develop the MTEC feed rate limit, FFCC used the combustion air correlation curve to determine the stack gas flow rate at the maximum combustion air flow rate demonstrated during the CPT. The MACT EEE mercury emission standard was used to calculate the maximum mercury feed rate limit. FFCC used only 90% of the feed rate calculated using MTEC in order to establish the maximum feed rate limit. This calculation is shown in Table 15.0.

6.1.4.2 Maximum Semi-Volatile Metals Feed Rate

FFCC established a maximum semi-volatile metal (SVM) feed rate limit based on the average of the test run averages from the CPT. The amount of SVM spiked and the sVM content of the waste was used to determine the limit. This limit will be a 12-hour hourly rolling average (HRA) limit of 0.84 lb/hr. The test run averages can be seen in Table 9.0.

6.1.4.3 Maximum Low Volatile Metals Feed Rate

FFCC established a maximum low-volatile metal (LVM) feed rate limit based on the average of the test run averages from the CPT. The amount of LVM spiked and the LVM content of the waste was used to determine the limit. This limit will be a 12-hour hourly rolling average (HRA) limit of 1.62 lb/hr. The test run averages can be seen in Table 9.0.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 35 of 42

6.2 <u>Combustion Chamber Limitations</u>

The operating parameter limits are summarized in Table 14.0. Table 17.0 summarizes the OPL data by run. A combination of methods was used to develop the combustion chamber limits. The limits were established as either maximums or minimums depending on the parameter. Process data was used when required and manufacturing and design data was used when necessary. The process data from each run is provided in Attachment E-5.

6.2.1 <u>Minimum Combustion Chamber Temperature Limitation</u>

FFCC established a minimum combustion chamber temperature limit based on the average of the test run averages from the CPT. This limit will be an hourly rolling average (HRA) limit of 1,558 deg F. The test run averages can be seen in Table 9.0.

6.2.2 Minimum Atomization Pressure Limitation

FFCC established a minimum atomization pressure limit based on the manufacturer's design of the atomization nozzles. The nozzles for the high-btu waste are referred to as the "main guns", and the nozzles for the low-btu waste are referred to as the "ring nozzles". The limit for the main guns will be an instantaneous limit of 30 psig, and the limit for the ring nozzles will be an instantaneous limit of 30 psig. These are based on system design.

6.2.3 <u>Maximum Furnace Pressure Limitation</u>

FFCC established a maximum furnace pressure limit at 90% of the system's combustion zone design pressure. The design pressure of the oxidizer is 222 inches water column (in. w.c.). This can be seen in drawing 6M10-4V-033. Ninety percent of this design pressure is 200 in. w.c., so the limit for the furnace pressure will be an instantaneous limit of 200 in. w.c.

6.2.4 <u>Maximum Flue Gas Flow Rate Limitation</u>

FFCC established a maximum flue gas flow rate limit based on a correlation of flue gas flow rate to the average of the test run average combustion air flow rate recorded during the CPT. The correlation can be seen in Attachment E-3. This limit will be an hourly rolling average (HRA) limit of 21,106 scfm. The test run average combustion air flow rate can be seen in Table 9.0.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 36 of 42

6.3 <u>Air Pollution Control Limitations</u>

The operating parameter limits for the air pollution control devices are summarized in Table 14.0, and Table 17.0 summarizes the OPL data by run. The limits are calculated as the average of the test run average measurements collected from each of the process monitoring instruments during the CPT. The limits were established as either maximums or minimums depending on the parameter. The process data from each run are provided in Attachment E-5.

6.3.1 Minimum Pressure Drop Across the Scrubber

FFCC established a minimum pressure drop across the scrubbber limit based on the average of the test run averages from the CPT. The minimum differential pressure measurement was established due to its relationship to particulate, metals and HCl/Cl₂ removal efficiency. This parameter indirectly measures gas/liquid contact, which controls removal of sub-micron particulate and acid gases. This limit will be an HRA limit of 71 in, w.c. The test run averages can be seen in Table 9.0

6.3.2 <u>Minimum Scrubber pH</u>

FFCC established a minimum scrubber pH limit based on the average of the test run averages from the CPT. The minimum scrubber water pH was established to ensure the scrubbing system's ability to neutralize acid gases and meet the HCl/Cl₂ MACT EEE emission standard. This limit will be an HRA limit of 4.4 pH. The test run averages can be seen in Table 9.0

6.3.3 Minimum Quench Tank pH

Originally FFCC did not plan any quench tank limitations, but during the CPT FFCC decided that the pH of the Quench tank could have an impact on the systems ability to remove HCl/Cl_2 from the stack gas. Therefore FFCC decided to establish a minimum pH on the rapid quench water. The limit is established as the average of the test run average measurements collected during the CPT. This limit will be an HRA limit of 0.6 pH. The test run averages can be seen in Table 9.0

6.3.4 Minimum Water Flow Rate to 1st Stage Scrubber

FFCC established a minimum water flow rate to 1^{st} stage scrubber limit based on the average of the test run averages from the CPT. The minimum scrubber water flow rate measurement, to the 1^{st} stage scrubber, is established as an alternative to feed pressure, which is preferred in the MACT standard. Flow is more crucial to the performance of the specific scrubber to be operated at FFCC. This limit will be an HRA limit of 477 gpm. The test run averages can be seen in Table 9.0

6.3.5 <u>Minimum Water Flow Rate to 2nd Stage Scrubber</u>

FFCC established a minimum water flow rate to 2^{nd} stage scrubber limit based on the average of the test run averages from the CPT. The minimum scrubber water flow rate measurement, to the 2^{nd} stage scrubber, is established as an alternative to feed pressure, which is preferred in the MACT standard. Flow is more crucial to the performance of the specific scrubber to be operated at FFCC. This limit will be an HRA limit of 301 gpm. The test run averages can be seen in Table 9.0

6.3.6 Minimum Scrubber Blowdown Flow Rate

FFCC established a minimum scrubber blowdown rate limit based on the average of the test run averages from the CPT. The minimum scrubber blowdown rate measurement is established as the parameter to control solids build up in the scrubbing medium. This limit will be an HRA limit of 100 gpm. The test run averages can be seen in Table 9.0

6.3.7 Minimum Scrubber Recirculation Tank Volume

FFCC established a minimum scrubber recirculation tank volume limit based on the average of the test run averages from the CPT. The minimum scrubber tank volume will proposed in conjunction with the minimum scrubber blowdown rate to control solids build up in the scrubbing medium. This limit will be an HRA limit of 72%. The test run averages can be seen in Table 9.0

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 38 of 42

6.4 Stack Gas Limitations

The only stack gas OPL is carbon monoxide and that limit is set by regulation. The regulatory limit for CO in the stack gas is 100 ppmv corrected to 7% oxygen. The test run average CO can be seen in Table 9.0.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 39 of 42

7.0 <u>CPT Conditional Approval Items</u>

This section discusses the issues the agency requested be addressed as part of their approval to conduct the CPT.

7.1 Quality Assurance Project Plan (QAPP)

The agency requested two additions to the QAPP. A signature page and a HW Combustion table summary.

7.1.1 Signature Page

FFCC created a QAPP Signature page and distributed to all key personnel for signatures. It is located in the QAPP which can be found in Attachment B.

7.1.2 Hazardous Waste Combustion Table

FFCC created and added a table to the QAPP in the format requested by the EPA. This table is located in the QAPP, and the QAPP is located in Attachment B.

7.2 <u>Combustion Zone Pressure</u>

The agency requested several additional requirements for positive pressure units. Since the incinerator is a positive pressure unit, FFCC will address these requests.

7.2.1 Combustion Zone Design and Maintenance

7.2.1.1 Manufacturer's Design

The manufacturer's design combustion zone pressure is 222 in wc. or 8 psig. This is shown in Drawing 6M10-4V-033. The maximum design pressure of the seal between the combustion zone and the rapid quench is 276 in. w.c. or 10 psig. This is shown in Drawing 6M10-4E-050.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 40 of 42

7.2.1.2 Maintenance Process for Seals

The manufacturer provided no recommendation for maintenance or replacement of the seals in the combustion zone. FFCC maintains all of the equipment in its area. The incinerator is inspected daily for signs of corrosion, leaks and other concerns. The incinerator is also inspected monthly for potential flue gas leaks using an infrared camera. Most of the seals are not readily accessible when the unit is operating. Any time the incinerator is taken down and cooled, maintenance inspects all seals for functionality. If any seal or gasket appears to need repair, then FFCC will replace the seal.

7.2.1.3 Maintenance/Replacement History for Seals

FFCC has not maintained a historical documentation of seals that it has replaced in the past. FFCC will begin maintaining a history of seal repairs and maintain that documentation in the operating record.

7.2.2 Combustion Zone Inspection Program

FFCC currently visually inspects the incinerator daily to identify leaks, spills, fugitive emissions, and /or signs of deterioration that could lead to any of those events. FFCC has added a monthly inspection using an infrared camera to inspect the combustion zone for potential flue gas leaks. An infrared inspection record can be seen in Attachment F. The inspection in Attachment F was conducted during the CPT.

7.2.3 Recordkeeping

FFCC maintains records of its daily visual inspection, including leaks and corrective actions, in its operating record for a minimum of three years. FFCC will also maintain the results of the monthly infrared camera flue gas inspection as well.

7.2.4 Response to Leaks

If any inspection reveals a leak of combustion gases from the combustion zone, FFCC will cutoff the hazardous waste feed and repair the leak before burning hazardous waste again. A flue gas leak will be considered any leak of combustion gases from the combustion zone with a carbon monoxide (CO) concentration greater than 100 ppm, or volatile organic compound (VOC) emissions greater than 500 ppm above background.

Incinerator Notice of Compliance Revision No 0 Revision Date: 01-12-11 Page 41 of 42

7.3 General Pending Comments

The agency made several general comments on various issues involving the combustion facility.

7.3.1 Feedstream Analysis Plan

The agency was concerned with the frequency of analysis for the fatty acid stream due to its concentration of chromium. FFCC will not combust the Fatty Acid stream shown in Table 1.0 in its onsite incinerator, until FFCC demonstrates to its Title V permitting authority that its sampling frequency has sufficient confidence and power to ensure that emission standards are not exceeded.

7.3.2 CO CEMS Range

The facility CO CEMS has two ranges, a low range with a span of 200 ppmv and a high range with a span of 3000 ppmv at an oxygen correction factor of 1.

7.3.3 Combustion Air and Flue Gas Flow Correlation

See Attachment E-3 for the flue gas flow correlation and its discussion.

Incinerator Notice of Compliance Revision No.: 0 Revision Date: 01/12/11 Page 42 of 42

8.0 Statement of Compliance

Based on my inquiry of the person or persons who manage the system, or those persons directly responsible for gathering the information, the information submitted is, to be the best of my knowledge and belief, true, accurate, and complete.

The information contained in the notification establishes the operating controls and limits that ensure that the waste incinerator located at the FutureFuel Chemical Company located in Batesville, Arkansas is in compliance with all the applicable emission standards of 40 CFR 63, Subpart EEE.

Signed this 11th day of January 2011.

i

Sam Dortch Senior Vice President and General Manager FutureFuel Chemical Company

Tables

- Table 1 Feed Stream Analysis Data
- Table 2 Feed Stream Characteristics
- Table 3 Summary of Sampling and Analysis Program
- Table 4 Process Monitoring Instruments
- Table 5 Process Monitoring Instruments, Calibration, and Maintenance
- Table 6 CPT Process Feed Analytical Results
- Table 7 CPT Spiking Information
- Table 8 CPT Stack Gas Analytical Results
- Table 9 CPT Test Averages
- Table 10 CPT Comparison of Planned Conditions to Actual Conditions Run 1
- Table 11 CPT Comparison of Planned Conditions to Actual Conditions Run 2
- Table 12 CPT Comparison of Planned Conditions to Actual Conditions Run 3
- Table 13 Demonstration of Compliance
- Table 14 Summary of Operating Parameter Limits
- Table 15 Mercury MTEC Feed Rate Limit
- Table 16 Destruction Removal Efficiency
- Table 17 CPT Operational Data Summary

TABLE - 1.0

FEED STREAM ANALYSIS DATA

40 CFR 63.1207(f)(1)(i)

Food Stream #	Average Hoot Value	N	ow-Volati letals (Av	le 2)	Semi-V Metals	olatile (Avg)	Mercury (Avg)		Other (A	Metals vg)		Avg	Avg	Physical
Leeu ou cam	(Btu/lb)	As ppm	Be ppm	Cr ppm	Cd	Pb ppm	Hg ppm	Sh ppm	Ba ppm	Ag	ndq IT	0/0	%	Form
Process Intermediate Waste	12,000	BDL	BDL	5	BDL	BDL	BDL	BDL	BDL	BDL	BDL	4	<1	Liquid
Fatty Acid Waste	12.000	BDI.	BDL	106	BDL	BDL	BDL	BDI.	BDL	BDI.	BDL	0	<2	Liquid
Organic Process Waste	9,500	BDI.	BDL	2	BDL.	BDI.	< 0.1	BDI.	< 0.5	BDI,	BDL.	<1	< 3	I.iquid
Spent Solvent Waste	11,000	BDI.	BDL	1	BDI.	1	BDL	BDL.	BDI.	BDL	BDL.	< 1	< 2	Liquid
Aqueous Waste	1,500	BDL	BDL	BDL	BDI.	BDL	BDL	BDL.	0.26	BDI.	BDL	4	12	Liquid
Liquid Auxiliary Fuel	18.600	BDL	BDL	BDL	BDL	BDL	BDL	BDL	BDL	BDL	BDL	0	>	Liquid
River Water	0	BDL	BDL	BDL	BDL	BDL	BDL	BDL	BDL	BDL.	BDL	<00'0>	<1	Liquid
Combustion Air	0	0	0	0	0	0	0	0	0	0	0	0	0	Aŭ
Natural Gas Fuel	23,000	0	0	0	0	0	0	0	0	0	0	0	0	Gas

BDL = Below Detection Limit (for constituents that are BDL, detection limits are input into the process control system for purposes of compliance monitoring pursuant to 40 CFR 63.1209)

¹ Fatty Acid Waste will not be burned in the incinerator without Agency approval of the sampling frequency.

TABLE - 2.0

FEED STREAM CHARACTERISTICS

40 CFR 63.1207(f)(1)(ii)

Feed Stream	Constituents and Other Characteristics	Avg %	Max %	Organic Hazardous	Air Pollutants ¹	Quantity Burned per Year (lbs)	Waste Codes
Organic Process Waste	Toluenc	12	30	Acetonitrile	< 1 %	500,000 to	F003 D001.
	Water	90	15	Chlorobenzene	0 - 3%	2,500,000	D021
	Methanol	15	50	Phenol	0-1-0		
	Heptane	7	40	Triethylamine	< 1%		
	TXIB	7	20	Toluene	0-30 %		
	DMAP	2	5	Xylene	0 - 40%		
	Acdone	10	40	Formaldehyde	< 1 %		
	Isopropanol	5	30	Methanol	0 - 50%		
	Xylene	10	40	MIBK	0 - 5%		
	Glycerin	12	40	Ethyl Benzene	0 - 2%		
	Other Organics	10	20				
Spent Solvent Waste	Chlorobenzene	10	40	Acetonitrile	<1 %	15,000,000 to	F002, F003,
	Water	90	20	Chlorobenzene	0 - 40%	20,000,000	F005, D001,
	Toluene	20	50	Triethylamine	< 1%		D002, D021
	Methanol	10	40	Toluene	0 - 50%		
	IFT Product	2	9	Formaldehyde	< 1 %		
	Tars from organic process	14	30	MIBK	0 - 5%		
	Acetic Acid	25	50	Methanol	0 - 40 %		
	Triethylene Glycol	2	5	Xylene	0 - 5 %		
	o-Dichlorobenzene	1	10	Ethyl Benzene	0 - 1 %		
	Other Organics	~	20				

TABLE - 2.0 (Continued)

FEED STREAM CHARACTERISTICS

40 CFR 63.1207(f)(1)

Feed Stream	Constituents and Other Characteristics	Avg %	Max. %	Organic Hazardous Air Pollut	ants Burne Year	ntity ed per (Ibs)	Waste Codes
Process Intermediate Waste	Chlorobenzene Acetone Isopropanol Heptane Xylene Toluene Other organics	10 25 15 10 10 10	30 30 30 30 30 30 30 30 30 30 30 30 30 3	Chlorobenzene 0 – 30 Toluene 0 – 30 Xylene 0 – 30	% 1,500, % 2,500 %	,000 to ,000	F002, F003, F005, D001, D021
Fatty Acid Waste ²	Nonunoic Acid Octanoic Acid TXIB	98 1 1	100 2 2	None	1,000,1	000 to 0,000	D007
Natural Gas Auxiliary Fuel	Natural Gas	100	100	Methane 99.99	% 0 to 10	000'00	None
1,4 Dioxane Waste	Water NaOH 1,4-Dioxane Benzene	98.9 1 0.04	99.5 2 0.1 0.1	Benzene 0 – 0,1 1,4-Dixoane 0 – 0.	% 2,000 to	o 5,000	D018

of Compliance	0	1/12/11
or Nouffication (No.:	Date:
Incinerate	Revision	Revision

TABLE -2.0 (Continued)

FEED STREAM CHARACTERISTICS

40 CFR 63.1207(f)(1)

Feed Stream	Heat Value (Btu/lb)	Constituents and Other Characteristics	Avg %	Max %	Organic Hazardous A	vir Pollutants	Quantity Burned per Year (lbs)	Waste Codes
Aqueous Waste	1,000 to 4,000	Water Sodium Chloride Sodium Sulfate Methanol Acetone Propronic Acid Dimethyl Sulfoxide N.N-dimethylacetamide Glycerin Ethyl Acetate Isopropanol Toluene Chlorobenzene TXIB (CAS# 6846-50-0) Hydrogen Chloride Potassium Chloride Sulfuric Acid	$ 53 \pm 6 \pm 4 = 6 = 6 = 6 = 6 = 6 = 6 = 6 = 6 = 6 =$	95 26 v v v v v v v v v v v v 0 27 v v v v v v v v v v v 0 28 v v v v v v v v v v v v v v v v v v v	Acetonitrile Chlorobenzene Toluene Phenol Phenol	<pre><1% 0-1% 0-1% </pre> <pre>c 1% c 1% c 1% 0-1% 0-1% </pre>	35,000,000 lbs	D001,D002, D021
Liquid Auxiliary Fuel	13,000 to 16,000	Diesel Diisopropyl Benzene Derivatives Triisopropyl Benzene Derivatives	30 65 5	95 95 15	Benzene	0-0.01 %	50,000 to 700,000	None

of Compliance	0	1/12/11
Incinerator Notification	Revision No.:	Revision Date:

TABLE -2.0 (Continued)

FEED STREAM CHARACTERISTICS

40 CFR 63.1207(f)(1)

Waste Codes	None
Quantity Burned per Year (lbs)	1,000,000 to 10,000,000
Organic Hazardous Air Pollutants	None
Max %	100
Avg %	100
Constituents and Other Characteristics	Water
Heat Value (Btu/lb)	0
Feed Stream	River Water

Notes:

- British Thermal Unit British Thermal Unit per pound
- Btu Btu/Ib ppm wt%
- Parts per million Weight production H H H H

¹ Only those organic HAPs that may be potentially present in the waste are reported in Table 3.0. All other organic HAPs are not present based on on-site material use and process chemistry evaluations.

² Fatty Acid Waste will not be combusted in the incinerator without Agency approval of the sampling frequency.

TABLE-3.0

SUMMARY OF SAMPLING AND ANALYSIS PROGRAM

40 CFR 63.1207(f)(1)(iv)

Sample No./Type	Parameters	Sampling Method	Sample Frequency	Field QA/QC Samples	Analytical Methods
	Specific Gravity			None	ASTM D-891 or other approved method
	Ash			None	ASTM D-482 or other approved method
	СћІоппе	Grab, 4 oz. collected at each sample interval		None	SW-846 9076 or other approved method
1) Liquid Organic and	Heat Value		Every 30 mmutes, cach run	None	ASTM-D-240 or other approved method
Aqueous Waste	SVM - Lead	 (1 wo composites prepared for each run) 		None	ICP-AES (Method 200.7 or 200.8)
	LVM - Chromium			None	ICP-AES (Method 200.7 or 200.8)
	Mercury			None	EPA Method SW-7470A
	POHC	Grab, 40ml collected at each sample interval	Every 30 minutes; each run	None	Volatile POHC by SW-846 5030b, 8260b
2) Stack gas metals – MM I	As, Be, Cd, Cr, Hg, and Pb	Method 29, MMT. isokinetic sample	160 minutes each run	Reagent blanks (filter, probe rinse solution)	SW-846 6010b and 7470a
 Stack gas hydrogen chloride, chlorine, and particulate matter 	HCl and Chlorine	Method 26A,	160 minutes each run	Reagent blanks (H ₂ SO ₄ and NaOH impuger solutions	Ion Chromatography (SW-9056/9057)
 Method 26A sampling train 	Particulate matter	ISOMIKUC SUIDIC		None	Gravimetric

TABLE -3.0 (Continued)

SUMMARY OF SAMPLING AND ANALYSIS PROGRAM

40 CFR 63.1207(f)(1)(iv)

Field QA/QC Samples
One condensate trip blank
One pair VOST tı trip blank
One set field bla tubes (four pairs per test conditio
One field blank pe condition
One train blank pe condition
One trip blank res tube
One dejonized w

TABLE - 3.0 (Continued)

SUMMARY OF SAMPLING AND ANALYSIS PROGRAM

40 CFR 63.1207(f)(1)(iv)

Sample No./Type	Parameters	Sampling Method	Sample Frequency	Field QA/QC Samples	Analytical Methods
	Carbon Monoxide	Continuous, extractive, ORSAT (EPA Method 3)	Each run	None	Continuous Extractive-NDIR, ORSAT Analyzer
	Carbon Dioxide	ORSAT (EPA Method 3)	Each run	None	ORSAT Analyzer
6) Stack gas	Total Hydrocarbon	EPA Method 25A	Each Run	None	Continuous, Extractive, Flame Ionization Analyzer
	Oxygen	Continuous, extractive, ORSAT	Each run	None	Continuous Extractive-Paramagnetic, ORSAT Analyzer
	Stack Moisture	EPA Method 4	Each run	None	Measure volume to 0.5 ml
	Stack Velocity and Flowrate	EPA Method 2	Each run	None	Measure temperature, pressure, and volume

				Revision Date:	1/12/11
	TABLE -	- 3.0 (Continue	(pa		
	SUMMARY OF SAMPLI	NG AND ANA	TYS	IS PROGRAM	
	40 CFR	63.1207(f)(1)(i	()		
Notes to Tal	4e D-5.5:				
Is	= Arsenic	LVM		.ow-Volatile Metals	
WISE	 American Society for Testing and Materials 	ML		Ailliliter	
Sc	= Berylhum	1100W	11	PA Method 0011	
MA A	 British thermal units Pold neuror atomic absorbiout 	CIMIM		C Doubled Methods 2	
Prove Prove	- Cota rapor acoust acoust puos	NaOH		continue by developments	
	 Chromiun 	NDIR		Jondispersive infrared	
SC/MS	- Gas chromatorrashy and mass shortmenony	PAH	11	Advantelear aromatic hydrocarbon	
CIEID	Class chromatography and flame ionization detector	hh	0	agineren a onnan nyasoaron ead	
unda	= Gallons per minute	PCDD	Ш	Dioxin	
Ip	= Mercury	PCDF	1	uran	
12804	= Sulfuric acid	POHC	11	rincipal organic hazardous constituents	
CP	Inductively coupled argon plasma spectroscopy	0A/OC		Juality assurance and quality control	
CP-AES	= Inductively coupled plasma-atomic emission spectrometry	SVM	11	iemi-Volatile Metals	
HON	= Potassium hydroxide	VOST	11	Volatile organic sampling train	
HON		TODA	1	volatile organic sampung uant	

As ASTM Be Btu CVAA CC CC GC/MS GC/FID gpm Hg HDS CP ICP-AES KOH

TABLE - 4.0

PROCESS MONITORING INSTRUMENTS

40 CFR 62.1207(f)(1)(iv)

	Instrument Number(s)	Type of Instrument	Instrument Range	Expected Operating Range	Accuracy	Drawing Location Reference
o †	FT-6M10-20153 FT-6M10-20155 FT-6M10-20156	Coriolis Mass Flow Meter	0 - 100 lb/min	5 - 75 lb/min	+-2% of calibrated range	6M10-9T-001 6M10-9T-001 6M10-9T-001
	FT-6M10-20157 FT-6M10-20355	Coriolis Mass Flow Meter	0 - 300 lb/min	30 – 225 lb/min	+/-2% of calibrated range	6M10-9T-002 6M10-9T-002
14 ·	rT-6M10-20180	Diff. Pressure w/ Averaging Pitot Array	0 - 17,125 scfm	0 - 15,300 sefm	+/-2°o of calibrated range	6M10-9T-007
PS PS PS	L-6M10-20270 L-6M10-20271 L-6M10-20272	Welded Diaphragm Pressure Switch	2 – 50 psig	> 30 psig	Repeatability of +/- 2.5% of Adjustable Range	6M10-9T-009 6M10-9T-009 6M10-9T-009
ISd	-6M10-20137	Welded Diaphragm Pressure Switch	12 – 100 psig	> 30 psig	Repeatability of +/- 2.5% of Adjustable Range	6M10-9T-002
μĻ	6M10-20182 6M10-20183	Thermocouple and Transmitter	0-2200 Deg. F	1500 – 1900 deg. F	Transmitter : +/- 3% of calibrated range Thermocouple: +/- 3% of calibrated range	6M10-9T-002 6M10-9T-002
AT-	6M10-3269	CEMS Analyzer	0 - 500 ppm 0 - 3000 ppm	20 – 100 ppm	+/- 3 % of calibrated range	6M10-9T-012
AT-(SM10-3270	CEMS Analyzer	0 - 25 ° o	0-25 00	+/- 2% of calibrated range	6M10-9T-012

TABLE - 4.0 (continued)

PROCESS MONITORING INSTRUMENTS

40 CFR 62.1207(f)(1)(iv)

Drawing Location Reference	6M10-9T-005	6M10-9T-005	6M10-9T-004	6M10-9T-004	6M10-9T-006	6M10-9T-005	6MI10-9T-005
Accuracy	+- 0.5 % of calibrated range	+/+ 3% of calibrated range	+/- 3°° of calibrated range	+/- 3% of calibrated range	-/- 2°° of calibrated range	+/- 3° of calibrated range	-/- 30° of calibrated range
Expected Operating Range	70 - 90 ° WC	4 – 9 pH	Hq 8 – 0	60-150 gpm	4 – 10 feet	500-600 gpm	300 400 gpm
Instrument Range	0 - 100 ° WC	0 – 14 pH	0 – 14 pH	0 – 150 gpm	0 – 116 * WC	0 - 600 gpm	0 -450 gpm
Type of Instrument	Pressure Transmitter	pH Transmitter	pH Transmitter	Magnetic Flow Meter	Pressure Transmitter	Magnetic Flow Meter	Magnetic Flow Meter
Instrument Number(s)	PT-6M10-20036 PT-6M10-20114	AT-6M10-20043 AT-6M10-20044	AIT-6M10-20016 AIT-6M10-20017	FT-6M10-20012	LT-6M10-20049	FT-6M10-20034	FT-6M10-20032
Location	Top of Scribber	Serubber Re- circulation Pump Line to Serubbers	Rapid Quench Weir Tank	Destructor Pad Near Weir Tank Circulation Pumps	CD-224 Scrubber Re-circulation Tank	CD-223A Serubber	CD-223B Scrubber
Parameter	Scrubber Differential Pressure	Scrubber Water pH Measurement	Quench Water pH Measurement	Scrubber Water Blowdown Flow Monitor	Scrubber Tank Level Monitor	Scrubber 1ª Stage Flow Rate	Scrubber 2 nd Stage Flow Rate

TABLE - 5.0

PROCESS MONITORING INSTRUMENTS, CALIBRATION, and MAINTENANCE

40 CFR 63.1207(f)(1)(iv)

Parameter	Location	Instrument Number(s)	Inspection/ Calibration Procedure	Calibration Frequency	Preventive Maintenance Procedure	Preventive Maintenance Frequency
Liquid Waste Feed to Organic Waste Guns Flow Monitor	Liquid Waste Lines to Gun Numbers 2, 3, & 4	FT-6M10-20153 FT-6M10-20155 FT-6M10-20156	Calibrate using certified electronic scales	Amual	Inspect During Calibration	Annual. During Calibration
Liquid Waste Feed to Aqueous Waste Guns Flow Monitor	Liquid Waste Lines to Ring Nozzles	FT-6M10-20157 FT-6M10-20355	Calibrate using certified electronic scales	Amual	Inspect During Calibration	Annual, During Calibration
Combustion Air Flow Rate	Combustion Air Line	FT-6M10-20180	Calibrate against standard	Annual	Clean During Calibration	Annual
Atomization Air Pressure to Organic Waste Guns	Plant Air Lines to Organic Waste Guns	PSL-6M10-20270 PSL-6M10-20271 PSL-6M10-20272	Calibrated using a certified pressure standard	Amual	Inspect and Clean During Calibration	Annual
Atomization Air Pressure to Aqueous Waste Guns	Plant Air Lines to Aqueous Waste Guns	PSL-6M10-20137	Calibrated using a certified pressure standard	Annual	Inspect and Check Actuation on Loss of Air Pressure	Annual
Combustion Chamber Temperature	Combustion Chamber	TT-6M10-20182 TT-6M10-20183	Calibrated using certified temperature standard	Amual	Inspect During Calibration	Annual. During Calibration
Scrubber Differential Pressure	Top of Scribber	PT-6M10-20036 PT-6M10-20114	Calibrated using certified pressure standard	Amual	Inspect During Calibration	Annual, During Calibration
Scrubber Water pH Measurement	Scrubber Recirculation Pump Line to Scrubbers	AT-6M10-20043 AT-6M10-20044	Calibrated using standard pH buffer solution	Quarterly	Inspect During Calibration	Quarterly, During Calibration
Quench Water pH Measurement	Scrubber Recirculation Pump Line to Scrubbers	AIT-6M10-20016 AIT-6M10-20017	Calibrated using standard pH buffer solution	Quarterly	Inspect During Calibration	Quarterly, During Calibration

TABLE-5.0

PROCESS MONITORING INSTRUMENTS, CALIBRATION, and MAINTENANCE

40 CFR 63.1207(f)(1)(iv)

Parameter	Location	Instrument Number(s)	Inspection/ Calibration Procedure	Calibration Frequency	Preventive Maintenance Procedure	Preventive Maintenance Frequency
Serubber Water Blowdown Flow Monitor	Destructor Pad Near Weir Tank Circulation Pumps	FT-6M10-20012	Calibrate using certified standard	Annual	Inspect During Calibration	Annual during Calibration
Scrubber Tank Level Monitor	CD-224 Scrubber Recirculation Tank	LT-6M10-20049	Calibrate using certified pressure standard	Amual	Visual Inspection During Calibration	Annal
Scrubber 1ª Stage Flow Rate	CD-223A Scrubber	FT-6M10-20034	Calibrate using certified standard	Amual	Inspect During Calibration	Amual
Scrubber 2 nd Stage Flow Rate	CD-223B Scrubber	FT-6M10-20032	Calibrate using certified standard	Annual	Inspect During Calibration	Amual
Carbon Monoxide	CEMS Building	AT-6M10-3269	Calibrate with known Calibration Gas +/-2ºa Accuracy	Daily Calibration Check with Calibration when Calibration Check Indicates an Error of 15 PPM	Visual Inspection Daily, Clean and Calibrate as Necessary	Daily
Oxygen	CEMS Building	AT-6M10-3270	Calibrate with known Calibration Gas +/-2% Accuracy	Daily Calibration Check with Calibration when Calibration Check Indicates an Error of 0.5%	Visual Inspection Daily, Clean and Calibrate as Necessary	Daily

Low	-Btu Waste An	alytical Results	
Constituent	Run 1	Run 2	Run 3
Chlorobenzene (%)	0.00014	0.00010	0.00011
Heating Value (btu/lb)	<200	<200	<200
Specific Gravity	1.09	1.09	1.09
Chloride (mg/L)	80000	78000	79000
Ash (%)	12	12	12
Mercury (mg/L)	0.1	0.1	0.1
SVM (mg/L)	0.040	0.040	0.040
Lead	0.040	0.040	0.040
LVM (mg/L)	0.007	0.007	0.007
Chromium	0.007	0.007	0.007
High	-Btu Waste An	alytical Results	
Constituent	Run 1	Run 2	Run 3
Chlorobenzene (%)	2.1	3.7	1.7
Heating Value (btu/lb)	11000	13000	10000
Specific Gravity	0.917	1.02	0.905
Chloride (%)	15	25	11
Ash (%)	0.023	0.006	0.48
Mercury (mg/kg)	0.1	0.1	0.100
SVM (mg/kg)	4	4	4
Lead	4	4	4
LVM (mg/kg)	0.7	0.7	0.7
Chromium	0.7	0.7	0.7

Table 6.0 - CPT Process Feed Analytical Results

Table 7.0 - CPT Spiking Information

Metals Spik	ting Inform	ation				Metals S	oiking Info	rmation - 1	1 est 2			Metals !	Spiking Inf	ormation			
10/12/2010						10/13/201	0					10/14/20	110				
Run I				Spike	Rate	Run 2				Spike	Rate	Run 3				Spike	Rate
	Scale		Solution Avg	Avg	Avg		Scale		Solution	Pb	ċ		Scale		Solution	Ph	S.
Time	Weight	Change	lb/min	lb/hr	lb/hr	Time	Weight	Change	Avg lb/min	lb/hr	lb/hr	Time	Weight	Change	Avg lb/hr	lb/hr	lb/hr
8:32	516.82		1.06	0.58	1.13	8:10	509.22		1.73	0.95	1.85	8:30	467.29		1.71	0.94	1,83
8:47	504.32	12.50				8:15	500.35	8.87				8:35	459.15	8.14			
9:02	487.60	16.72				8:20	490.35	10.00				8:40	450.47	8.68			
9:17	474.47	13.13				8:25	481.72	8.63				8:45	441.92	8.55			
9:32	455.65	18.82				8:30	472.98	8.74				8:50	433.42	8.50			
9:47	439.19	16.46				8:45	446.69	26.29				8:55	425.01	8.41			
10:02	423.86	15.33				9:00	420.58	26.11				9:00	416.60	8.41			
10:17	407.53	16.33				9:15	394.56	26.02				9:15	390.79	25.81			
10:32	390.92	16.61				9:30	369.09	25.47				9:30	364.69	26.10			
10:47	373.93	16.99				9:45	343.52	25.57				9:45	338.59	26.10			
11:02	356.79	FI.71				10:00	318.06	25.46				10:00	312.53	26.06			
11:17	340.42	16.37				10:15	292.70	25.36				10:15	287.43	25.10			
11:32	324,90	15.52				10:30	267.27	25.43				10:30	262.25	25.18			
11:47	310,23	14.67				10:45	241.36	25.91				10:45	235.84	26.41			
12:02	297.26	12.97			Π	11:00	215.25	26.11				11:00	210.10	25.74			
12:17	284.30	12.96				11:15	189.23	26.02				11:15	185.70	24.40			
12:32	268.14	16.16				11:30	163.14	26.09				11:30	159.39	26.31			
12:47	253.90	14.24				11:45	137.23	25.91				11:45	133.25	26.14			
13:02	239.45	14.45				11:50	128.58	8.65				12:00	106.51	26.74			
13:17	223.28	16.17															
13:32	206.65	16.63															
13:47	190.30	16.35															

Spike Feedrate averages were calculated based on data collected during M29 metals sampling.

Spike Solution Concentration by Weight = 0.914% Pb and 1.78% Cr

		Ī	Concentration		Concentration	Corrected	
Test	Emission	Stack Flow	Converted to	02	Corrected to	Concentration	Emission Rate
Run	Constituent	DSCFM	ug/dscm	. ¹⁰ /6	(q. 7% O2	Units	lb/hr
	POLIC	12179	0.61	7.7	NA	ug dsem	0.000028
	HCLCL ₂	12018	8786 51	77	6.1	ppmvd	0.395575
	Particulate	12018	7181.22	7.7	0,0034	gr dsef	0.323304
	Mercury	12288	2.95	77	.3.1	ug dsem	0.00136
	SVM	12288	115.26	7.7	121.33	ug dsem	0.005306
	Cd	12288	0.31	77	0.33	ug dsem	(FEEDERF)
Run I	Pb	12288	114.95	77	121	ug dsem	0.005291
	LVM	12288	39.33	77	41 4037	ug dsem	0.001811
	Δ8	12288	(кти)	77	0	ug dsem	ET ELETERETENES
	Be	12288	411163	j 77	0	ug dsem	TETH ROUND
	Cr	12288	36.29	77	38.2	ug dsem	0.001671
	DE	12179	0.0000035	77	0.0037	ng dsem	1.60E-10
	THC	12179	NA	77	3.2	ppmvd	NA
	POLIC	12064	0.77	67	NΛ	ug dsem	03000035
	HCI CI ₂	12151	26173.26	67	16.9	ppmvd	1 191380
	Particulate	12151	1249011	6.7	0.0055	gr dsef	0.568537
	Mercury	12406	0.61	67	0.6	ug dsem	0.000028
	SVM	12406	297.24	67	291.0	ug dsem	0.013814
	Cd	12406	0.00	67	1141	ug dsem	О, ОКНАНИ
Run 2	Pb	12406	297.24	6.7	291	ug dsem	0.013814
	I VM	12406	77.53	67	75.9	ug dsem	0.003603
	As	12406	(1116)	67	0	ug dsem	A FA H HOM H HO
	Be	12406	11110	67	0	ug dsem	O CONTROLICE
	. Cr	12288	77.53	6.7	75.9	ug dsem	0.003569
	DE	12179	110000143	67	00042	ng dsem	96[-10
	THC	12179	NA	67	1.8	ppmvd	NΛ
	POIIC	12092	0.63	7.3	NA	ug dsem	0.000029
	HCLCI ₂	12229	10534 50	7.3	71	ppmvd	0.482598
	Particulate	12229	8049.89	7.3	0.0037	gr dsef	0.368775
	Mercury	12252	41.]0	73	01	ug dsem	а қан қан қад
	SVM	12252	256.39	7.3	262.0	ug dsem	0.011767
	. Cd	12252	()))	7.3	0.0	ug dsem	LETENDERMO
Run 3	Pb	12252	256.39		262	ug dsem	0.011767
	I VM	12252	77,60	7.3	79.3	ug dsem	0.003562
	As	12252	+++(H)	7.3	0	ug dsem	(F,CICHOR HONO
	Be	12252	()())	7.3	0	ug dsem	СЕСЕНИИ
	Ci	12288	77.60	73	79.3	ug dsem	0003572
	DF	12179	0.0000036	7.3	0,0016	ng dsem	7 148-11
	THC T	12179	NA] 73	6	ppmvd	NA

Table 8.0 - CPT Stack Gas Analytical Results

 $lb\ hr\ Emission\ Rate=(Stack\ Flow\ dscf\ mm)\ x\ (Conc.\ ug\ dscm)\ x\ (60\ mm\ hr)\ x\ (4\ m3\ 35.34\ ft3)\ x\ (4\ g\ -1.000,000)\ ug)\ x\ (4\ lb\ -453.6\ g)$

ug dsem = (11, -10x61) x (g=g-mole) x (1.0001, -m3) x (1.000.000 ug=g) x (1.g-mole = 24.041.) -

ug dsem = (gr=4t3) x (64.8 mg = gr) x (1.900 ug = mg) x (34.31 ft3 = m3) -

ug dsem (ng m3) x (Eug 1980 ng)

Table 9.0 - CPT Run Averages

		Planned				Γ
Test Parameters		Conditions	CPT	CPT	Run Avera	iges
Description	Units	Target	Average	Run 1	Run 2	Run 3
Total Waste Feed Rate	lb/hr	16500	15348	14704	15699	15642
Low-Btu Waste Feed Rate	lb/hr	12500	12059	11346	12527	12303
High-Btu Waste Feed Rate	lb/hr	4000	3289	3358	3171	3339
Total POHC Feed Rate	lb/hr	85	82	71	117	57
Total Chloride Feed Rate	lb/hr	1650	1428	1336	1689	1259
Total Ash Feed Rate	lb/hr	1650	1453	1362	1503	1492
Total Mercury Feed Rate	lb/hr	0.002	0.0014	0.0014	0.0015	0.0015
Total SVM Feed Rate	lb/hr	0.65	0.84	0.59	0.96	0.95
Total LVM Feed Rate	lb/hr	1.25	1.62	1.15	1.86	1.84
Combustion Air Flow Rate	scfm	10000	9994	9666	9989	9998
Combustion Temperature	deg F	1500	1558	1544	1588	1544
Scrubber Pressure Drop	in.wc.	70	71	72	72	71
Scrubber Water pH	Hđ	4	4.3	4.4	4.4	4.2
Quency Tank pH	Hd	¢ >	0.5	0.6	0.3	0.6
1st Stage Scrubber Flow	gpm	475	476	477	475	475
2nd Stage Scrubber Flow	gpm	350	301	301	300	301
Scrubber Blowdown	gpm	100	100	100	100	101
Recirc Tank Level	%	%02	72	72	71	71
CO Concentration	ppmv	≤ 100	1.2	0	<i>t</i>	3
THC Concentration	ppmv	10	2.2	3.2	1.8	1.6
Low-Btu Gun Atomization Pressure	psig	30	30	30	30	30
High-Btu Gun Atomization Pressure	psig	20	20	20	20	20

Table 10.0 - CPT Comparison of Planned Conditions to Actual Conditions

Run 1

		Planned				
Test Parameters		Conditions	Ac	utal Condit	ions	Comparison*
Description	Units	Target	Average	Minimum	Maximum	Comment
Total Waste Feed Rate	Ib/hr	16500	14704	14601	14807	Feed rates not as high as planned
Low-Btu Waste Feed Rate	Ib/hr	12500	11346	11243	11450	Was comfortable with low-btu feed rate during test
High-Btu Waste Feed Rate	Ib/hr	4000	3358	3358	3358	Heating value limited High-btu waste feed rate
Total POHC Feed Rate	Ib/hr	85	71	71	71	Feed rates not as high as planned
Total Chloride Feed Rate	Ib/hr	1650	1336	1329	1344	Feed rates not as high as planned
Total Ash Feed Rate	Ib/hr	1650	1362	1350	1375	Feed rates not as high as planned
Total Mercury Feed Rate	lb/hr	0.002	0.0014	NA	NA	Feed rates not as high as planned
Total SVM Feed Rate	lb/hr	0.65	0.59	NA	NA	
Total LVM Feed Rate	Ib/hr	1.25	1.15	NA	NA	
Combustion Air Flow Rate	scfm	10000	9666	9751	10223	
Combustion Temperature	deg F	1500	1544	1531	1564	Heating value of high-btu waste created a higher temp
Scrubber Pressure Drop	in.wc.	70	72	70	75	
Scrubber Water pH	Hd	4	4.4	1.7	5.9	
Quench Tank pH	Hd	< 1	0.6	0.6	0.6	
1st Stage Scrubber Flow	gpm	475	477	460	535	
2nd Stage Scrubber Flow	gpm	300	301	259	339	
Scrubber Blowdown	gpm	100	100	66	104	
Recirc Tank Level	%	70	72	70	74	
CO Concentration	ppmv	≤ 100	0	0	0	
THC Concentration	ppmv	≤ 10	3.2	NA	NA	
Low-Btu Gun Atomization Pressure	psig	30	30	30	30	
High-Btu Gun Atomization Pressure	psid	20	20	20	20	

* No comment means the difference between the planned and actual conditons were not significant (e.g., < 20%)

Table 11.0 - CPT Comparison of Planned Conditions to Actual Conditions

Run 2

					es				ased because of need to maximize	to maximize chromium feed rate.		ted heating value of waste											
Comparison*	Comment				See Significant Issue				Lead feed rate incre chromium feed rate.	Adjusted spike rate t		Higher than anticipal											
and	Maximum	15792	12629	3204	119	1705	1516	NA	NA	NA	10117	1601	76	5.8	0.4	553	307	110	73	1.7	NA	30	00
tal Condit	Minimum	15589	12426	3123	116	1670	1491	NA	NA	NA	9791	1546	71	2.4	0.2	469	291	92	69	0.0	NA	30	00
V	Averade	15699	12527	3171	117	1689	1503	0.0015	0.96	1.86	9989	1588	72	4.4	0.3	475	300	100	71	0.8	1.8	30	00
Planned	Target	16500	12500	4000	85	1650	1650	0.002	0.65	1.25	10500	1500	70	4	<1	475	300	100	%02	≤ 100	< 10	30	20
	Units	Ib/hr	lb/hr	lb/hr	lb/hr	lb/hr	lb/hr	lb/hr	lb/hr	lb/hr	scfm	deg F	in.wc.	Hd	Hd	gpm	gpm	gpm	%	ppmv	ppmv	psig	nsin
Tast Daramatare	Description	Total Waste Feed Rate	Low-Btu Waste Feed Rate	High-Btu Waste Feed Rate	Total POHC Feed Rate	Total Chloride Feed Rate	Total Ash Feed Rate	Total Mercury Feed Rate	Total SVM Feed Rate	Total LVM Feed Rate	Combustion Air Flow Rate	Combustion Temperature	Scrubber Pressure Drop	Scrubber Water pH	Quench Tank pH	1st Stage Scrubber Flow	2nd Stage Scrubber Flow	Scrubber Blowdown	Recirc Tank Level	CO Concentration	THC Concentration	Low-Btu Gun Atomization Pressure	High-Rtu Gun Atomization Pressure

* No comment means the difference between the planned and actual conditons were not significant (e.g., < 20%)

Table 12.0 - CPT Comparison of Planned Conditions to Actual Conditions

Run 3

		Planned				
Test Parameters		Conditions	Ac	utal Condit	ions	Comparison*
Description	Units	Target	Average	Minimum	Maximum	Comment
Total Waste Feed Rate	lb/hr	16500	15642	15642	15643	
Low-Btu Waste Feed Rate	lb/hr	12500	12303	12300	12306	
High-Btu Waste Feed Rate	lb/hr	4000	3339	3337	3342	
Total POHC Feed Rate	lb/hr	85	57	57	57	See significant issues
Total Chloride Feed Rate	lb/hr	1650	1259	1259	1259	See significant issues
Total Ash Feed Rate	lb/hr	1650	1492	1492	1493	
Total Mercury Feed Rate	lb/hr	0.002	0.0015	NA	NA	
Total SVM Feed Rate	lb/hr	0.65	0.95	NA	NA	Lead feed rate increased because of need to maximize chromium feed rate.
Total LVM Feed Rate	lb/hr	1.25	1.84	NA	NA	Adjusted spike rate to maximize chromium feed rate.
Combustion Air Flow Rate	scfm	10000	9998	9837	10264	
Combustion Temperature	deg F	1500	1544	1527	1555	
Scrubber Pressure Drop	in.wc.	20	17	69	74	
Scrubber Water pH	Hq	4	4.2	1.5	5.8	
Quench Tank pH	Hd	<1	0.6	0.5	0.6	
1st Stage Scrubber Flow	gpm	475	475	445	516	
2nd Stage Scrubber Flow	gpm	300	301	273	347	
Scrubber Blowdown	gpm	100	101	89	115	
Recirc Tank Level	%	%02	71	70	73	
CO Concentration	ppmv	≤ 100	2.7	1.2	4.2	
THC Concentration	ppmv	≤ 10	1.6	NA	NA	
Low-Btu Gun Atomization Pressure	psig	30	30	30	30	
High-Btu Gun Atomization Pressure	psig	20	20	20	20	

* No comment means the difference between the planned and actual conditons were not significant (e.g., < 20%)

Emission Results Compared to M/	VCT EEE Standards for Ha	izardous Wast	e Incinerators
Test	Units	Standard	CPT Result ²
Destruction Removal Efficiency	%	≥ 99.99%	990000000000000000000000000000000000000
Particulate Emission ¹	gr/dscf	≤ 0.013	0,0042
HCI/Chlorine Emissions ¹	http://www.	32	10
Mercury Emissions ¹	ng/dscm	≤ 130	1.3
Semi-Volatile Metal Emissions ¹	ug/dscm	≤ 230	224.8
Low-Volatile Metal Emissions ¹	ug/dscm	≤ 92	65.5
Dioxin/Furan ¹	ng/dscm TEQ	0.40	0.0032
Carbon Monoxide Emissions ¹	http://www.	≤ 100	1.2
Total Hydrocarbon Emissions ¹	http://www.	≤ 10	2.2

Table 13.0 - Demonstration of Compliance

¹ Results corrected to 7% O ₂

² CPT Results are shown as the average of the three runs

Table 14.0 - Summary of Operating Parameter Limits

		Final	
Operating Parameters		Operating	Comment
Waste Feed Limitations			
Maximum Hazardous Waste Feed Rate	lb/hr	15,348	Average of the Test Run Averages
Maximum Chloride Feed Rate	lb/hr	1,428	Average of the Test Run Averages
Maximum Ash Feed Rate	Ib/hr	1,453	Average of the Test Run Averages
Maximum Mercury Feed Rate	Ib/hr	0.011	90% of MTEC Feedrate based on Maximum Flue Gas Flow Rate
Maximum Semi-Volatile Feed Rate	lb/hr	0.84	Average of the Test Run Averages
Maximum Low-Volatile Feed Rate	lb/hr	1.62	Average of the Test Run Averages
Combustion Chamber Limitations			
Minimum Combustion Chamber Temperature	ц.	1,558	Average of the Test Run Averages
Minimum Atomization Pressure (High-Btu)	psig	30	Manufacturer's recommendation for main guns
Minimum Atomization Pressure (Low-Btu)	psig	30	Manufacturer's recommendation for ring nozzles
Maximum Furnace Pressure	in.wc.	≤200	Set at 90% of the system design pressure
Maximum Flue Gas Air Flow Rate	scfm	25,106	Correlated from the Average of the Combustion Air Flow Rate Test Run Averages (See Attachment E-3)
Air Pollution Control Limitaions			
Minimum Pressure Drop across Scrubber	in.wc.	71	Average of the Test Run Averages
Minimum Scrubber pH	Hd	4.4	Average of the Test Run Averages
Minimum Quench Tank pH	Hd	0.6	Average of the Test Run Averages
Minimum 1st Stage Scrubber Flow Rate	gpm	477	Average of the Test Run Averages
Minimum 2nd Stage Scrubber Flow Rate	gpm	301	Average of the Test Run Averages
Minimum Scrubber Blowdown Rate	gpm	100	Average of the Test Run Averages
Minimum Scrubber Recirculation Tank Level	%	72	Average of the Test Run Averages
Stack Gas Limitations			
Maximum Carbon Monoxide Concentration	ppmv	100	MACT Emission Standard

Table 15.0 - Mercury MTEC Feed Rate Limit

Hg Feed Rate Limit (Ib/hr) = $[M/$	ACT Limit ug/dscm) x (2.204586E-9 lb/ug) x (Stack Flow dscm/min) x (60 mm/lr) / (1-5RE)
Hg Feed Rate Limit (lb/hr) = (13)	0 ug/dscm) x (2.204586E-9 lb/ug) x (25106 dscf/min / 35.31 dscf/dscm) x (60 min/lr) / (1-0)
Hg MTEC Feed Rate Limit (lb/hr) = [0.0]	1223
90% Hg MTEC Feed Rate Limit (lb/lr) = 0.0	

Notification of Compliance Revision No 0 Revision Date 01/12/11

CPT Run	POHC IN Ib/hr	POHC OUT ib/hr	DRE POHC
Run 1	71	0.0000278	99.99996%
Run 2	117	0.0000348	99.99997%
Run 3	57	0.00002854	99.99995%
		-	
CPT Average			99.99996%

Table 16.0 - Destruction Removal Efficiency

					CPT Run	1 Sumn	nary						
	Total	Low Btu	High Btu			Scrubber			1st Stage	2nd Stage	Scrubber	Recirc	
	Waste	Waste	Waste	Combustion	Combustion	Pressure	Scrubher	Quench	Scrubber	Scrubber	Blewdown	Tank	00
	Feed	Feed	Feed	Air Flow	Temperature	Drop	Water pH	Tank pH	Flow	Flow	Flow	Level	Conc.
Calculation	lb/hr	lh/hr	lb/hr	sefin	deg F	in.wc.	Hd	pH	եթա	gpin	gpm [0%	hpmv
Run 1 Average	14704	11346	3358	9666	1544	72	4.4	0.6	477	301	100	72	0.0
Run 1 Minimum	14601	11243	3358	9751	1531	20	1.7	0.6	460	259	66	70	0.0
Run 1 Maximum	14807	11450	3358	10223	1564	75	9	-	535	339	104	74	0

>
5
g
5
ភ
~
g
Ъ,
Õ
a
0
÷.
ъ.
Ð.
ā.
0
Ē
2
0
N.
-
đ
ž
1
Ľ.
_

						2 6 1							Γ
						IIIIIne 7 I	liaiy						
Tot	ы	Low Btu	High Btu			Scrubber			1st Stage	2nd Stage	Scrubber	Recirc	
Wa	ste	Waste	Waste	Combustion	Combustion	Pressure	Scrubber	Quench	Scrubber	Scrubber	Blowdown	Tank	00
Fe	ed	Feed	Feed	Air Flow	Temperature	Drop	Water pH	Tank pH	Flow	Flow	Flow	Level	Conc.
â	/hr	lh/hr	lb/hr	sefin	deg F	in.w.c.	pII	pH	gal/min	gal/min	gal/min	a⁄,₀	hpmv
15(699	12527	3171	9989	1588	72	4.2	0.3	475	300	100	71	0.8
15	589	12426	3123	9791	1546	71	2.4	0.2	469	291	92	69	0.0
15	792	12629	3204	10117	1601	76	5.8	0.4	553	307	110	73	1.7

				CPT Run	3 Sumn	nary						
	ow Btu 1	High Btu			Scrubber			1st Stage	2nd Stage	Scruhher	Recirc	
Market	Waste	Waste (Combustion	Combustion	Pressure	Scrubber	Quench	Scrubber	Scrubber	Blowdown	Tank	00
	Feed	Feed	Air Flow	Temperature	Drop	Water pH	Tank pH	Flow	Flow	Flow	Level	Conc.
	lh/hr	lh/hr	sefin	deg F	in.w.c.	pH	PH	gal/min	gal/min	gal/min	9/0	ymidd
~	12303	3339	9998	1544	11	4.2	0.6	475	301	101	71	2.7
	12300	3337	9837	1527	69	1.5	0.5	445	273	89	70	1.2
	12306	3342	10264	1555	74	5.8	0.6	516	347	115	73	4.2
Appendix O

AUTHENTICATED U.S. GOVERNMENT INFORMATION

SOURCE: 57 FR 61992, Dec. 29, 1992, unless otherwise noted.

Subpart ZZZ—National Emissions Standards for Hazardous Air Pollutants for Stationary Reciprocating Internal Combustion Engines

SOURCE: 69 FR 33506, June 15, 2004, unless otherwise noted.

WHAT THIS SUBPART COVERS

§63.6580 What is the purpose of subpart ZZZZ?

Subpart ZZZZ establishes national emission limitations and operating limitations for hazardous air pollutants (HAP) emitted from stationary reciprocating internal combustion engines (RICE) located at major and area sources of HAP emissions. This subpart also establishes requirements to demonstrate initial and continuous compliance with the emission limitations and operating limitations.

[73 FR 3603, Jan. 18, 2008]

§63.6585 Am I subject to this subpart?

You are subject to this subpart if you own or operate a stationary RICE at a major or area source of HAP emissions, except if the stationary RICE is being tested at a stationary RICE test cell/ stand.

(a) A stationary RICE is any internal combustion engine which uses reciprocating motion to convert heat energy into mechanical work and which is not mobile. Stationary RICE differ from mobile RICE in that a stationary RICE is not a non-road engine as defined at 40 CFR 1068.30, and is not used to propel a motor vehicle or a vehicle used solely for competition.

(b) A major source of HAP emissions is a plant site that emits or has the potential to emit any single HAP at a rate of 10 tons (9.07 megagrams) or more per year or any combination of HAP at a rate of 25 tons (22.68 megagrams) or more per year, except that for oil and gas production facilities, a major source of HAP emissions is determined for each surface site.

(c) An area source of HAP emissions is a source that is not a major source.

(d) If you are an owner or operator of an area source subject to this subpart, your status as an entity subject to a standard or other requirements under this subpart does not subject you to the obligation to obtain a permit under 40 CFR part 70 or 71, provided you are not required to obtain a permit under 40 CFR 70.3(a) or 40 CFR 71.3(a) for a reason other than your status as an area source under this subpart. Notwithstanding the previous sentence, you must continue to comply with the provisions of this subpart as applicable.

(e) If you are an owner or operator of a stationary RICE used for national security purposes, you may be eligible to request an exemption from the requirements of this subpart as described in 40 CFR part 1068, subpart C.

(f) The emergency stationary RICE listed in paragraphs (f)(1) through (3) of this section are not subject to this subpart. The stationary RICE must meet the definition of an emergency stationary RICE in 63.6675, which includes operating according to the provisions specified in 63.6640(f).

(1) Existing residential emergency stationary RICE located at an area source of HAP emissions that do not operate for the purpose specified in $\S63.6640(f)(4)(ii)$.

(2) Existing commercial emergency stationary RICE located at an area source of HAP emissions that do not operate for the purpose specified in $\S63.6640(f)(4)(ii)$.

(3) Existing institutional emergency stationary RICE located at an area source of HAP emissions that do not operate for the purpose specified in $\S63.6640(f)(4)(ii)$.

[69 FR 33506, June 15, 2004, as amended at 73
FR 3603, Jan. 18, 2008; 78 FR 6700, Jan. 30, 2013; 87 FR 48607, Aug. 10, 2022]

§63.6590 What parts of my plant does this subpart cover?

This subpart applies to each affected source.

(a) Affected source. An affected source is any existing, new, or reconstructed stationary RICE located at a major or area source of HAP emissions, excluding stationary RICE being tested at a stationary RICE test cell/stand.

(1) Existing stationary RICE.

(i) For stationary RICE with a site rating of more than 500 brake horsepower (HP) located at a major source of HAP emissions, a stationary RICE is existing if you commenced construction or reconstruction of the stationary RICE before December 19, 2002.

(ii) For stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions, a stationary RICE is existing if you commenced construction or reconstruction of the stationary RICE before June 12, 2006.

(iii) For stationary RICE located at an area source of HAP emissions, a stationary RICE is existing if you commenced construction or reconstruction of the stationary RICE before June 12, 2006.

(iv) A change in ownership of an existing stationary RICE does not make that stationary RICE a new or reconstructed stationary RICE.

(2) New stationary RICE. (i) A stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions is new if you commenced construction of the stationary RICE on or after December 19, 2002.

(ii) A stationary RICE with a site rating of equal to or less than 500 brake HP located at a major source of HAP emissions is new if you commenced construction of the stationary RICE on or after June 12, 2006.

(iii) A stationary RICE located at an area source of HAP emissions is new if you commenced construction of the stationary RICE on or after June 12, 2006.

(3) Reconstructed stationary RICE. (i) A stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions is reconstructed if you meet the definition of reconstruction in $\S63.2$ and reconstruction is commenced on or after December 19, 2002.

(ii) A stationary RICE with a site rating of equal to or less than 500 brake HP located at a major source of HAP emissions is reconstructed if you meet the definition of reconstruction in §63.2 and reconstruction is commenced on or after June 12, 2006.

(iii) A stationary RICE located at an area source of HAP emissions is recon-

40 CFR Ch. I (7–1–23 Edition)

structed if you meet the definition of reconstruction in §63.2 and reconstruction is commenced on or after June 12, 2006.

(b) Stationary RICE subject to limited requirements. (1) An affected source which meets either of the criteria in paragraphs (b)(1)(i) through (ii) of this section does not have to meet the requirements of this subpart and of subpart A of this part except for the initial notification requirements of §63.6645(f).

(i) The stationary RICE is a new or reconstructed emergency stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions.

(ii) The stationary RICE is a new or reconstructed limited use stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions.

(2) A new or reconstructed stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions which combusts landfill or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis must meet the initial notification requirements of §63.6645(f) and the requirements of §§63.6625(c), 63.6650(g), and 63.6655(c). These stationary RICE do not have to meet the emission limitations and operating limitations of this subpart.

(3) The following stationary RICE do not have to meet the requirements of this subpart and of subpart A of this part, including initial notification requirements:

(i) Existing spark ignition 2 stroke lean burn (2SLB) stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions;

(ii) Existing spark ignition 4 stroke lean burn (4SLB) stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions;

(iii) Existing emergency stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions.

(iv) Existing limited use stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions;

(v) Existing stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions that combusts landfill gas or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis;

(c) Stationary RICE subject to Regulations under 40 CFR Part 60. An affected source that meets any of the criteria in paragraphs (c)(1) through (7) of this section must meet the requirements of this part by meeting the requirements of 40 CFR part 60 subpart IIII, for compression ignition engines or 40 CFR part 60 subpart JJJJ, for spark ignition engines. No further requirements apply for such engines under this part.

(1) A new or reconstructed stationary RICE located at an area source;

(2) A new or reconstructed 2SLB stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions;

(3) A new or reconstructed 4SLB stationary RICE with a site rating of less than 250 brake HP located at a major source of HAP emissions;

(4) A new or reconstructed spark ignition 4 stroke rich burn (4SRB) stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions;

(5) A new or reconstructed stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions which combusts landfill or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis;

(6) A new or reconstructed emergency or limited use stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions;

(7) A new or reconstructed compression ignition (CI) stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions.

[69 FR 33506, June 15, 2004, as amended at 73
FR 3604, Jan. 18, 2008; 75 FR 9674, Mar. 3, 2010;
75 FR 37733, June 30, 2010; 75 FR 51588, Aug.
20, 2010; 78 FR 6700, Jan. 30, 2013; 87 FR 48607,
Aug. 10, 2022]

§63.6595 When do I have to comply with this subpart?

(a) Affected sources. (1) If you have an existing stationary RICE, excluding existing non-emergency CI stationary RICE, with a site rating of more than 500 brake HP located at a major source of HAP emissions, you must comply with the applicable emission limitations, operating limitations and other requirements no later than June 15. 2007. If you have an existing non-emergency CI stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, an existing stationary CI RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions, or an existing stationary CI RICE located at an area source of HAP emissions, you must comply with the applicable emission limitations, operating limitations, and other requirements no later than May 3, 2013. If you have an existing stationary SI RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions, or an existing stationary SI RICE located at an area source of HAP emissions, you must comply with the applicable emission limitations, operating limitations, and other requirements no later than October 19, 2013.

(2) If you start up your new or reconstructed stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions before August 16, 2004, you must comply with the applicable emission limitations and operating limitations in this subpart no later than August 16, 2004.

(3) If you start up your new or reconstructed stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions after August 16, 2004, you must comply with the applicable emission limitations and operating limitations in this subpart upon startup of your affected source.

(4) If you start up your new or reconstructed stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions before January 18, 2008, you

§63.6600

must comply with the applicable emission limitations and operating limitations in this subpart no later than January 18, 2008.

(5) If you start up your new or reconstructed stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions after January 18, 2008, you must comply with the applicable emission limitations and operating limitations in this subpart upon startup of your affected source.

(6) If you start up your new or reconstructed stationary RICE located at an area source of HAP emissions before January 18, 2008, you must comply with the applicable emission limitations and operating limitations in this subpart no later than January 18, 2008.

(7) If you start up your new or reconstructed stationary RICE located at an area source of HAP emissions after January 18, 2008, you must comply with the applicable emission limitations and operating limitations in this subpart upon startup of your affected source.

(b) Area sources that become major sources. If you have an area source that increases its emissions or its potential to emit such that it becomes a major source of HAP, the compliance dates in paragraphs (b)(1) and (2) of this section apply to you.

(1) Any stationary RICE for which construction or reconstruction is commenced after the date when your area source becomes a major source of HAP must be in compliance with this subpart upon startup of your affected source.

(2) Any stationary RICE for which construction or reconstruction is commenced before your area source becomes a major source of HAP must be in compliance with the provisions of this subpart that are applicable to RICE located at major sources within 3 years after your area source becomes a major source of HAP.

(c) If you own or operate an affected source, you must meet the applicable notification requirements in §63.6645 and in 40 CFR part 63, subpart A.

[69 FR 33506, June 15, 2004, as amended at 73
FR 3604, Jan. 18, 2008; 75 FR 9675, Mar. 3, 2010;
75 FR 51589, Aug. 20, 2010; 78 FR 6701, Jan. 30, 2013]

40 CFR Ch. I (7–1–23 Edition)

EMISSION AND OPERATING LIMITATIONS

§ 63.6600 What emission limitations and operating limitations must I meet if I own or operate a stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions?

Compliance with the numerical emission limitations established in this subpart is based on the results of testing the average of three 1-hour runs using the testing requirements and procedures in §63.6620 and Table 4 to this subpart.

(a) If you own or operate an existing, new, or reconstructed spark ignition 4SRB stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you must comply with the emission limitations in Table 1a to this subpart and the operating limitations in Table 1b to this subpart which apply to you.

(b) If you own or operate a new or reconstructed 2SLB stationary RICE with a site rating of more than 500 brake HP located at major source of HAP emissions, a new or reconstructed 4SLB stationary RICE with a site rating of more than 500 brake HP located at major source of HAP emissions, or a new or reconstructed CI stationary RICE with a site rating of more than $500\ {\rm brake}\ {\rm HP}$ located at a major source of HAP emissions, you must comply with the emission limitations in Table 2a to this subpart and the operating limitations in Table 2b to this subpart which apply to you.

(c) If you own or operate any of the following stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you do not need to comply with the emission limitations in Tables 1a, 2a, 2c, and 2d to this subpart or operating limitations in Tables 1b and 2b to this subpart: an existing 2SLB stationary RICE; an existing 4SLB stationary RICE; a stationary RICE that combusts landfill gas or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis; an emergency stationary RICE: or a limited use stationary RICE.

(d) If you own or operate an existing non-emergency stationary CI RICE with a site rating of more than 500 brake HP located at a major source of

HAP emissions, you must comply with the emission limitations in Table 2c to this subpart and the operating limitations in Table 2b to this subpart which apply to you.

[73 FR 3605, Jan. 18, 2008, as amended at 75 FR 9675, Mar. 3, 2010]

§63.6601 What emission limitations must I meet if I own or operate a new or reconstructed 4SLB stationary RICE with a site rating of greater than or equal to 250 brake HP and less than or equal to 500 brake HP located at a major source of HAP emissions?

Compliance with the numerical emission limitations established in this subpart is based on the results of testing the average of three 1-hour runs using the testing requirements and procedures in §63.6620 and Table 4 to this subpart. If you own or operate a new or reconstructed 4SLB stationary RICE with a site rating of greater than or equal to 250 and less than or equal to 500 brake HP located at major source of HAP emissions manufactured on or after January 1, 2008, you must comply with the emission limitations in Table 2a to this subpart and the operating limitations in Table 2b to this subpart which apply to you.

[73 FR 3605, Jan. 18, 2008, as amended at 75 FR 9675, Mar. 3, 2010; 75 FR 51589, Aug. 20, 2010]

§63.6602 What emission limitations and other requirements must I meet if I own or operate an existing stationary RICE with a site rating of equal to or less than 500 brake HP located at a major source of HAP emissions?

If you own or operate an existing stationary RICE with a site rating of equal to or less than 500 brake HP located at a major source of HAP emissions, you must comply with the emission limitations and other requirements in Table 2c to this subpart which apply to you. Compliance with the numerical emission limitations established in this subpart is based on the results of testing the average of three 1-hour runs using the testing requirements and procedures in §63.6620 and Table 4 to this subpart.

[78 FR 6701, Jan. 30, 2013]

§63.6603 What emission limitations, operating limitations, and other requirements must I meet if I own or operate an existing stationary RICE located at an area source of HAP

Compliance with the numerical emission limitations established in this subpart is based on the results of testing the average of three 1-hour runs using the testing requirements and procedures in §63.6620 and Table 4 to this subpart.

emissions?

(a) If you own or operate an existing stationary RICE located at an area source of HAP emissions, you must comply with the requirements in Table 2d to this subpart and the operating limitations in Table 2b to this subpart that apply to you.

(b) If you own or operate an existing stationary non-emergency CI RICE with a site rating of more than 300 HP located at an area source of HAP that meets either paragraph (b)(1) or (2) of this section, you do not have to meet the numerical CO emission limitations specified in Table 2d of this subpart. Existing stationary non-emergency CI RICE with a site rating of more than 300 HP located at an area source of HAP that meet either paragraph (b)(1)or (2) of this section must meet the management practices that are shown for stationary non-emergency CI RICE with a site rating of less than or equal to 300 HP in Table 2d of this subpart.

(1) The area source is located in an area of Alaska that is not accessible by the Federal Aid Highway System (FAHS).

(2) The stationary RICE is located at an area source that meets paragraphs (b)(2)(i), (ii), and (iii) of this section.

(i) The only connection to the FAHS is through the Alaska Marine Highway System (AMHS), or the stationary RICE operation is within an isolated grid in Alaska that is not connected to the statewide electrical grid referred to as the Alaska Railbelt Grid.

(ii) At least 10 percent of the power generated by the stationary RICE on an annual basis is used for residential purposes.

(iii) The generating capacity of the area source is less than 12 megawatts,

or the stationary RICE is used exclusively for backup power for renewable energy.

(c) If you own or operate an existing stationary non-emergency CI RICE with a site rating of more than 300 HP located on an offshore vessel that is an area source of HAP and is a nonroad vehicle that is an Outer Continental Shelf (OCS) source as defined in 40 CFR 55.2, you do not have to meet the numerical CO emission limitations specified in Table 2d of this subpart. You must meet all of the following management practices:

(1) Change oil every 1,000 hours of operation or annually, whichever comes first. Sources have the option to utilize an oil analysis program as described in §63.6625(i) in order to extend the specified oil change requirement.

(2) Inspect and clean air filters every 750 hours of operation or annually, whichever comes first, and replace as necessary.

(3) Inspect fuel filters and belts, if installed, every 750 hours of operation or annually, whichever comes first, and replace as necessary.

(4) Inspect all flexible hoses every 1,000 hours of operation or annually, whichever comes first, and replace as necessary.

(d) If you own or operate an existing non-emergency CI RICE with a site rating of more than 300 HP located at an area source of HAP emissions that is certified to the Tier 1 or Tier 2 emission standards in Table 1 of 40 CFR 89.112 and that is subject to an enforceable state or local standard that requires the engine to be replaced no later than June 1, 2018, you may until January 1, 2015, or 12 years after the installation date of the engine (whichever is later), but not later than June 1, 2018, choose to comply with the management practices that are shown for stationary non-emergency CI RICE with a site rating of less than or equal to 300 HP in Table 2d of this subpart instead of the applicable emission limitations in Table 2d, operating limitations in Table 2b. and crankcase ventilation system requirements in §63.6625(g). You must comply with the emission limitations in Table 2d and operating limitations in Table 2b that apply for nonemergency CI RICE with a site rating

40 CFR Ch. I (7–1–23 Edition)

of more than 300 HP located at an area source of HAP emissions by January 1, 2015, or 12 years after the installation date of the engine (whichever is later), but not later than June 1, 2018. You must also comply with the crankcase ventilation system requirements in $\S63.6625(g)$ by January 1, 2015, or 12 years after the installation date of the engine (whichever is later), but not later than June 1, 2018.

(e) If you own or operate an existing non-emergency CI RICE with a site rating of more than 300 HP located at an area source of HAP emissions that is certified to the Tier 3 (Tier 2 for engines above 560 kilowatt (kW)) emission standards in Table 1 of 40 CFR 89.112, you may comply with the requirements under this part by meeting the requirements for Tier 3 engines (Tier 2 for engines above 560 kW) in 40 CFR part 60 subpart IIII instead of the emission limitations and other requirements that would otherwise apply under this part for existing non-emergency CI RICE with a site rating of more than 300 HP located at an area source of HAP emissions.

(f) An existing non-emergency SI 4SLB and 4SRB stationary RICE with a site rating of more than 500 HP located at area sources of HAP must meet the definition of remote stationary RICE in §63.6675 on the initial compliance date for the engine, October 19, 2013, in order to be considered a remote stationary RICE under this subpart. Owners and operators of existing non-emergency SI 4SLB and 4SRB stationary RICE with a site rating of more than 500 HP located at area sources of HAP that meet the definition of remote stationary RICE in §63.6675 of this subpart as of October 19, 2013 must evaluate the status of their stationary RICE every 12 months. Owners and operators must keep records of the initial and annual evaluation of the status of the engine. If the evaluation indicates that the stationary RICE no longer meets the definition of remote stationary RICE in §63.6675 of this subpart, the owner or operator must comply with all of the requirements for existing non-emergency SI 4SLB and 4SRB stationary RICE with a site rating of more than 500 HP located at area sources of HAP

that are not remote stationary RICE within 1 year of the evaluation.

[75 FR 9675, Mar. 3, 2010, as amended at 75 FR 51589, Aug. 20, 2010; 76 FR 12866, Mar. 9, 2011; 78 FR 6701, Jan. 30, 2013]

§63.6604 What fuel requirements must I meet if I own or operate a stationary CI RICE?

(a) If you own or operate an existing non-emergency, non-black start CI stationary RICE with a site rating of more than 300 brake HP with a displacement of less than 30 liters per cylinder that uses diesel fuel, you must use diesel fuel that meets the requirements in 40 CFR 1090.305 for nonroad diesel fuel.

(b) Beginning January 1, 2015, if you own or operate an existing emergency CI stationary RICE with a site rating of more than 100 brake HP and a displacement of less than 30 liters per cylinder that uses diesel fuel and operates for the purpose specified in 63.6640(f)(4)(ii), you must use diesel fuel that meets the requirements in 40 CFR 1090.305 for nonroad diesel fuel, except that any existing diesel fuel purchased (or otherwise obtained) prior to January 1, 2015, may be used until depleted.

(c) [Reserved]

(d) Existing CI stationary RICE located in Guam, American Samoa, the Commonwealth of the Northern Mariana Islands, at area sources in areas of Alaska that meet either $\S63.6603(b)(1)$ or $\S63.6603(b)(2)$, or are on offshore vessels that meet $\S63.6603(c)$ are exempt from the requirements of this section.

[78 FR 6702, Jan. 30, 2013, as amended at 85 FR 78463, Dec. 4, 2020; 87 FR 48607, Aug. 10, 2022]

GENERAL COMPLIANCE REQUIREMENTS

§ 63.6605 What are my general requirements for complying with this subpart?

(a) You must be in compliance with the emission limitations, operating limitations, and other requirements in this subpart that apply to you at all times.

(b) At all times you must operate and maintain any affected source, including associated air pollution control equipment and monitoring equipment, in a manner consistent with safety and good air pollution control practices for minimizing emissions. The general duty to minimize emissions does not require you to make any further efforts to reduce emissions if levels required by this standard have been achieved. Determination of whether such operation and maintenance procedures are being used will be based on information available to the Administrator which may include, but is not limited to, monitoring results, review of operation and maintenance procedures, review of operation and maintenance records, and inspection of the source.

[75 FR 9675, Mar. 3, 2010, as amended at 78 FR 6702, Jan. 30, 2013]

TESTING AND INITIAL COMPLIANCE REQUIREMENTS

§63.6610 By what date must I conduct the initial performance tests or other initial compliance demonstrations if I own or operate a stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions?

If you own or operate a stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions you are subject to the requirements of this section.

(a) You must conduct the initial performance test or other initial compliance demonstrations in Table 4 to this subpart that apply to you within 180 days after the compliance date that is specified for your stationary RICE in $\S63.6595$ and according to the provisions in \$63.7(a)(2).

(b) If you commenced construction or reconstruction between December 19, 2002 and June 15, 2004 and own or operate stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you must demonstrate initial compliance with either the proposed emission limitations or the promulgated emission limitations no later than February 10, 2005 or no later than 180 days after startup of the source, whichever is later, according to §63.7(a)(2)(ix).

(c) If you commenced construction or reconstruction between December 19, 2002 and June 15, 2004 and own or operate stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, and you chose to comply with the proposed emission limitations when demonstrating initial compliance, you must conduct a second performance test to demonstrate compliance with the promulgated emission limitations by December 13, 2007 or after startup of the source, whichever is later, according to $\S63.7(a)(2)(ix)$.

(d) An owner or operator is not required to conduct an initial performance test on units for which a performance test has been previously conducted, but the test must meet all of the conditions described in paragraphs (d)(1) through (5) of this section.

(1) The test must have been conducted using the same methods specified in this subpart, and these methods must have been followed correctly.

(2) The test must not be older than 2 years.

(3) The test must be reviewed and accepted by the Administrator.

(4) Either no process or equipment changes must have been made since the test was performed, or the owner or operator must be able to demonstrate that the results of the performance test, with or without adjustments, reliably demonstrate compliance despite process or equipment changes.

(5) The test must be conducted at any load condition within plus or minus 10 percent of 100 percent load.

[69 FR 33506, June 15, 2004, as amended at 73 FR 3605, Jan. 18, 2008]

§63.6611 By what date must I conduct the initial performance tests or other initial compliance demonstrations if I own or operate a new or reconstructed 4SLB SI stationary RICE with a site rating of greater than or equal to 250 and less than or equal to 500 brake HP located at a major source of HAP emissions?

If you own or operate a new or reconstructed 4SLB stationary RICE with a site rating of greater than or equal to 250 and less than or equal to 500 brake HP located at a major source of HAP emissions, you must conduct an initial performance test within 240 days after the compliance date that is specified for your stationary RICE in §63.6595 and according to the provisions speci40 CFR Ch. I (7–1–23 Edition)

fied in Table 4 to this subpart, as appropriate.

[73 FR 3605, Jan. 18, 2008, as amended at 75 FR 51589, Aug. 20, 2010]

§63.6612 By what date must I conduct the initial performance tests or other initial compliance demonstrations if I own or operate an existing stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions or an existing stationary RICE located at an area source of HAP emissions?

If you own or operate an existing stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions or an existing stationary RICE located at an area source of HAP emissions you are subject to the requirements of this section.

(a) You must conduct any initial performance test or other initial compliance demonstration according to Tables 4 and 5 to this subpart that apply to you within 180 days after the compliance date that is specified for your stationary RICE in §63.6595 and according to the provisions in §63.7(a)(2).

(b) An owner or operator is not required to conduct an initial performance test on a unit for which a performance test has been previously conducted, but the test must meet all of the conditions described in paragraphs (b)(1) through (4) of this section.

(1) The test must have been conducted using the same methods specified in this subpart, and these methods must have been followed correctly.

(2) The test must not be older than 2 years.

(3) The test must be reviewed and accepted by the Administrator.

(4) Either no process or equipment changes must have been made since the test was performed, or the owner or operator must be able to demonstrate that the results of the performance test, with or without adjustments, reliably demonstrate compliance despite process or equipment changes.

[75 FR 9676, Mar. 3, 2010, as amended at 75 FR 51589, Aug. 20, 2010]

§63.6615 When must I conduct subsequent performance tests?

If you must comply with the emission limitations and operating limitations, you must conduct subsequent performance tests as specified in Table 3 of this subpart.

§63.6620 What performance tests and other procedures must I use?

(a) You must conduct each performance test in Tables 3 and 4 of this subpart that applies to you.

(b) Each performance test must be conducted according to the requirements that this subpart specifies in Table 4 to this subpart. If you own or operate a non-operational stationary RICE that is subject to performance testing, you do not need to start up the engine solely to conduct the performance test. Owners and operators of a non-operational engine can conduct the performance test when the engine is started up again. The test must be conducted at any load condition within plus or minus 10 percent of 100 percent load for the stationary RICE listed in paragraphs (b)(1) through (4) of this section.

(1) Non-emergency 4SRB stationary RICE with a site rating of greater than 500 brake HP located at a major source of HAP emissions.

(2) New non-emergency 4SLB stationary RICE with a site rating of greater than or equal to 250 brake HP located at a major source of HAP emissions.

(3) New non-emergency 2SLB stationary RICE with a site rating of greater than 500 brake HP located at a major source of HAP emissions.

(4) New non-emergency CI stationary RICE with a site rating of greater than 500 brake HP located at a major source of HAP emissions.

(c) [Reserved]

(d) You must conduct three separate test runs for each performance test required in this section, as specified in §63.7(e)(3). Each test run must last at least 1 hour, unless otherwise specified in this subpart.

(e)(1) You must use Equation 1 of this section to determine compliance with the percent reduction requirement:

$$\frac{C_{i} - C_{o}}{C_{i}} \times 100 = R \quad (Eq. 1)$$

Where:

- C_i = concentration of carbon monoxide (CO), total hydrocarbons (THC), or formaldehyde at the control device inlet.
- C_{o} = concentration of CO, THC, or formaldehyde at the control device outlet, and
- R = percent reduction of CO, THC, or formaldehyde emissions.

(2) You must normalize the CO, THC, or formaldehyde concentrations at the inlet and outlet of the control device to a dry basis and to 15 percent oxygen, or an equivalent percent carbon dioxide (CO₂). If pollutant concentrations are to be corrected to 15 percent oxygen and CO₂ concentration is measured in lieu of oxygen concentration measurement, a CO₂ correction factor is needed. Calculate the CO₂ correction factor as described in paragraphs (e)(2)(i) through (iii) of this section.

(i) Calculate the fuel-specific F_o value for the fuel burned during the test using values obtained from Method 19, Section 5.2, and the following equation:

$$F_{O} = \frac{0.209 F_{d}}{F_{C}}$$
 (Eq. 2)

§63.6620

Where:

- $F_o = Fuel \mbox{ factor based on the ratio of oxygen } volume to the ultimate CO_2 volume produced by the fuel at zero percent excess air.$
- 0.209 = Fraction of air that is oxygen, percent/100.

40 CFR Ch. I (7–1–23 Edition)

- F_d = Ratio of the volume of dry effluent gas to the gross calorific value of the fuel from Method 19, dsm³/J (dscf/10⁶ Btu).
- $\label{eq:Fc} F_c = Ratio \mbox{ of the volume of } CO_2 \mbox{ produced to} \\ the gross calorific value of the fuel from Method 19, dsm^3/J \ (dscf/10^6 \mbox{ Btu})$

(ii) Calculate the CO_2 correction factor for correcting measurement data to 15 percent O_2 , as follows:

$$X_{CO2} = \frac{5.9}{F_{O}}$$
 (Eq. 3)

Where:

 X_{CO2} = CO₂ correction factor, percent.

5.9 = 20.9 percent O_2 —15 percent O_2 , the defined O_2 correction value, percent.

(iii) Calculate the CO, THC, and formaldehyde gas concentrations adjusted to 15 percent O_2 using CO_2 as follows:

$$C_{adj} = C_d \frac{X_{CO2}}{%CO_2} \quad (Eq.4)$$

Where:

- C_{adj} = Calculated concentration of CO, THC, or formaldehyde adjusted to 15 percent O_{2} .
- C_d = Measured concentration of CO, THC, or formaldehyde, uncorrected.

 $X_{CO2} = CO_2$ correction factor, percent.

 $%CO_2$ = Measured CO_2 concentration measured, dry basis, percent.

(f) If you comply with the emission limitation to reduce CO and you are not using an oxidation catalyst, if you comply with the emission limitation to reduce formaldehyde and you are not using NSCR, or if you comply with the emission limitation to limit the concentration of formaldehyde in the stationary RICE exhaust and you are not using an oxidation catalyst or NSCR, you must petition the Administrator for operating limitations to be established during the initial performance test and continuously monitored thereafter; or for approval of no operating limitations. You must not conduct the initial performance test until after the petition has been approved by the Administrator.

(g) If you petition the Administrator for approval of operating limitations, your petition must include the information described in paragraphs (g)(1) through (5) of this section.

(1) Identification of the specific parameters you propose to use as operating limitations;

(2) A discussion of the relationship between these parameters and HAP emissions, identifying how HAP emissions change with changes in these parameters, and how limitations on these parameters will serve to limit HAP emissions;

(3) A discussion of how you will establish the upper and/or lower values for these parameters which will establish the limits on these parameters in the operating limitations;

(4) A discussion identifying the methods you will use to measure and the instruments you will use to monitor these parameters, as well as the relative accuracy and precision of these methods and instruments; and

(5) A discussion identifying the frequency and methods for recalibrating the instruments you will use for monitoring these parameters.

(h) If you petition the Administrator for approval of no operating limitations, your petition must include the

information described in paragraphs (h)(1) through (7) of this section.

(1) Identification of the parameters associated with operation of the stationary RICE and any emission control device which could change intentionally (*e.g.*, operator adjustment, automatic controller adjustment, etc.) or unintentionally (*e.g.*, wear and tear, error, etc.) on a routine basis or over time;

(2) A discussion of the relationship, if any, between changes in the parameters and changes in HAP emissions;

(3) For the parameters which could change in such a way as to increase HAP emissions, a discussion of whether establishing limitations on the parameters would serve to limit HAP emissions;

(4) For the parameters which could change in such a way as to increase HAP emissions, a discussion of how you could establish upper and/or lower values for the parameters which would establish limits on the parameters in operating limitations;

(5) For the parameters, a discussion identifying the methods you could use to measure them and the instruments you could use to monitor them, as well as the relative accuracy and precision of the methods and instruments;

(6) For the parameters, a discussion identifying the frequency and methods for recalibrating the instruments you could use to monitor them; and

(7) A discussion of why, from your point of view, it is infeasible or unreasonable to adopt the parameters as operating limitations.

(i) The engine percent load during a performance test must be determined by documenting the calculations, assumptions, and measurement devices used to measure or estimate the percent load in a specific application. A written report of the average percent load determination must be included in the notification of compliance status. The following information must be included in the written report: the engine model number, the engine manufacturer, the year of purchase, the manufacturer's site-rated brake horsepower, the ambient temperature, pressure, and humidity during the performance test, and all assumptions that were made to estimate or calculate percent load during the performance test must be clearly explained. If measurement devices such as flow meters, kilowatt meters, beta analyzers, stain gauges, etc. are used, the model number of the measurement device, and an estimate of its accurate in percentage of true value must be provided.

 $[69\ {\rm FR}$ 33506, June 15, 2004, as amended at 75 FR 9676, Mar. 3, 2010; 78 FR 6702, Jan. 30, 2013]

§ 63.6625 What are my monitoring, installation, collection, operation, and maintenance requirements?

(a) If you elect to install a CEMS as specified in Table 5 of this subpart, you must install, operate, and maintain a CEMS to monitor CO and either O_2 or CO_2 according to the requirements in paragraphs (a)(1) through (4) of this section. If you are meeting a requirement to reduce CO emissions, the CEMS must be installed at both the inlet and outlet of the control device. If you are meeting a requirement to limit the concentration of CO, the CEMS must be installed at the outlet of the control device.

(1) Each CEMS must be installed, operated, and maintained according to the applicable performance specifications of 40 CFR part 60, appendix B.

(2) You must conduct an initial performance evaluation and an annual relative accuracy test audit (RATA) of each CEMS according to the requirements in §63.8 and according to the applicable performance specifications of 40 CFR part 60, appendix B as well as daily and periodic data quality checks in accordance with 40 CFR part 60, appendix F, procedure 1.

(3) As specified in §63.8(c)(4)(ii), each CEMS must complete a minimum of one cycle of operation (sampling, analyzing, and data recording) for each successive 15-minute period. You must have at least two data points, with each representing a different 15-minute period, to have a valid hour of data.

(4) The CEMS data must be reduced as specified in $\S63.8(g)(2)$ and recorded in parts per million or parts per billion (as appropriate for the applicable limitation) at 15 percent oxygen or the equivalent CO₂ concentration.

(b) If you are required to install a continuous parameter monitoring system (CPMS) as specified in Table 5 of

this subpart, you must install, operate, and maintain each CPMS according to the requirements in paragraphs (b)(1) through (6) of this section. For an affected source that is complying with the emission limitations and operating limitations on March 9, 2011, the requirements in paragraph (b) of this section are applicable September 6, 2011.

(1) You must prepare a site-specific monitoring plan that addresses the monitoring system design, data collection, and the quality assurance and quality control elements outlined in paragraphs (b)(1)(i) through (v) of this section and in §63.8(d). As specified in §63.8(f)(4), you may request approval of monitoring system quality assurance and quality control procedures alternative to those specified in paragraphs (b)(1) through (5) of this section in your site-specific monitoring plan.

(i) The performance criteria and design specifications for the monitoring system equipment, including the sample interface, detector signal analyzer, and data acquisition and calculations;

(ii) Sampling interface (*e.g.*, thermocouple) location such that the monitoring system will provide representative measurements;

(iii) Equipment performance evaluations, system accuracy audits, or other audit procedures;

(iv) Ongoing operation and maintenance procedures in accordance with provisions in 63.8(c)(1)(i) and (c)(3); and

(v) Ongoing reporting and recordkeeping procedures in accordance with provisions in 63.10(c), (e)(1), and (e)(2)(i).

(2) You must install, operate, and maintain each CPMS in continuous operation according to the procedures in your site-specific monitoring plan.

(3) The CPMS must collect data at least once every 15 minutes (see also §63.6635).

(4) For a CPMS for measuring temperature range, the temperature sensor must have a minimum tolerance of 2.8 degrees Celsius (5 degrees Fahrenheit) or 1 percent of the measurement range, whichever is larger.

(5) You must conduct the CPMS equipment performance evaluation, system accuracy audits, or other audit 40 CFR Ch. I (7–1–23 Edition)

procedures specified in your site-specific monitoring plan at least annually.

(6) You must conduct a performance evaluation of each CPMS in accordance with your site-specific monitoring plan.

(c) If you are operating a new or reconstructed stationary RICE which fires landfill gas or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis, you must monitor and record your fuel usage daily with separate fuel meters to measure the volumetric flow rate of each fuel. In addition, you must operate your stationary RICE in a manner which reasonably minimizes HAP emissions.

(d) If you are operating a new or reconstructed emergency 4SLB stationary RICE with a site rating of greater than or equal to 250 and less than or equal to 500 brake HP located at a major source of HAP emissions, you must install a non-resettable hour meter prior to the startup of the engine.

(e) If you own or operate any of the following stationary RICE, you must operate and maintain the stationary RICE and after-treatment control device (if any) according to the manufacturer's emission-related written instructions or develop your own maintenance plan which must provide to the extent practicable for the maintenance and operation of the engine in a manner consistent with good air pollution control practice for minimizing emissions:

(1) An existing stationary RICE with a site rating of less than 100 HP located at a major source of HAP emissions;

(2) An existing emergency or black start stationary RICE with a site rating of less than or equal to 500 HP located at a major source of HAP emissions;

(3) An existing emergency or black start stationary RICE located at an area source of HAP emissions;

(4) An existing non-emergency, nonblack start stationary CI RICE with a site rating less than or equal to 300 HP located at an area source of HAP emissions;

§63.6625

(5) An existing non-emergency, nonblack start 2SLB stationary RICE located at an area source of HAP emissions;

(6) An existing non-emergency, nonblack start stationary RICE located at an area source of HAP emissions which combusts landfill or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis.

(7) An existing non-emergency, nonblack start 4SLB stationary RICE with a site rating less than or equal to 500 HP located at an area source of HAP emissions;

(8) An existing non-emergency, nonblack start 4SRB stationary RICE with a site rating less than or equal to 500 HP located at an area source of HAP emissions;

(9) An existing, non-emergency, nonblack start 4SLB stationary RICE with a site rating greater than 500 HP located at an area source of HAP emissions that is operated 24 hours or less per calendar year; and

(10) An existing, non-emergency, nonblack start 4SRB stationary RICE with a site rating greater than 500 HP located at an area source of HAP emissions that is operated 24 hours or less per calendar year.

(f) If you own or operate an existing emergency stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions or an existing emergency stationary RICE located at an area source of HAP emissions, you must install a non-resettable hour meter if one is not already installed.

(g) If you own or operate an existing non-emergency, non-black start CI engine greater than or equal to 300 HP that is not equipped with a closed crankcase ventilation system, you must comply with either paragraph (g)(1) or paragraph (2) of this section. Owners and operators must follow the manufacturer's specified maintenance requirements for operating and maintaining the open or closed crankcase ventilation systems and replacing the crankcase filters, or can request the Administrator to approve different maintenance requirements that are as protective as manufacturer requirements. Existing CI engines located at area sources in areas of Alaska that meet either §63.6603(b)(1) or §63.6603(b)(2) do not have to meet the requirements of this paragraph (g). Existing CI engines located on offshore vessels that meet §63.6603(c) do not have to meet the requirements of this paragraph (g).

(1) Install a closed crankcase ventilation system that prevents crankcase emissions from being emitted to the atmosphere, or

(2) Install an open crankcase filtration emission control system that reduces emissions from the crankcase by filtering the exhaust stream to remove oil mist, particulates and metals.

(h) If you operate a new, reconstructed, or existing stationary engine, you must minimize the engine's time spent at idle during startup and minimize the engine's startup time to a period needed for appropriate and safe loading of the engine, not to exceed 30 minutes, after which time the emission standards applicable to all times other than startup in Tables 1a, 2a, 2c, and 2d to this subpart apply.

(i) If you own or operate a stationary CI engine that is subject to the work, operation or management practices in items 1 or 2 of Table 2c to this subpart or in items 1 or 4 of Table 2d to this subpart, you have the option of utilizing an oil analysis program in order to extend the specified oil change requirement in Tables 2c and 2d to this subpart. The oil analysis must be performed at the same frequency specified for changing the oil in Table 2c or 2d to this subpart. The analysis program must at a minimum analyze the following three parameters: Total Base Number, viscosity, and percent water content. The condemning limits for these parameters are as follows: Total Base Number is less than 30 percent of the Total Base Number of the oil when new; viscosity of the oil has changed by more than 20 percent from the viscosity of the oil when new; or percent water content (by volume) is greater than 0.5. If all of these condemning limits are not exceeded, the engine owner or operator is not required to change the oil. If any of the limits are exceeded, the engine owner or operator must change the oil within 2 business days of receiving the results of the

analysis; if the engine is not in operation when the results of the analysis are received, the engine owner or operator must change the oil within 2 business days or before commencing operation, whichever is later. The owner or operator must keep records of the parameters that are analyzed as part of the program, the results of the analysis, and the oil changes for the engine. The analysis program must be part of the maintenance plan for the engine.

(j) If you own or operate a stationary SI engine that is subject to the work, operation or management practices in items 6, 7, or 8 of Table 2c to this subpart or in items 5, 6, 7, 9, or 11 of Table 2d to this subpart, you have the option of utilizing an oil analysis program in order to extend the specified oil change requirement in Tables 2c and 2d to this subpart. The oil analysis must be performed at the same frequency specified for changing the oil in Table 2c or 2d to this subpart. The analysis program must at a minimum analyze the following three parameters: Total Acid Number, viscosity, and percent water content. The condemning limits for these parameters are as follows: Total Acid Number increases by more than 3.0 milligrams of potassium hydroxide (KOH) per gram from Total Acid Number of the oil when new; viscosity of the oil has changed by more than 20 percent from the viscosity of the oil when new; or percent water content (by volume) is greater than 0.5. If all of these condemning limits are not exceeded, the engine owner or operator is not required to change the oil. If any of the limits are exceeded, the engine owner or operator must change the oil within 2 business days of receiving the results of the analysis; if the engine is not in operation when the results of the analysis are received, the engine owner or operator must change the oil within 2 business days or before commencing operation, whichever is later. The owner or operator must keep records of the parameters that are analyzed as part of the program, the results of the analysis, and the oil changes for the engine. The analysis

40 CFR Ch. I (7–1–23 Edition)

program must be part of the maintenance plan for the engine.

[69 FR 33506, June 15, 2004, as amended at 73
FR 3606, Jan. 18, 2008; 75 FR 9676, Mar. 3, 2010;
75 FR 51589, Aug. 20, 2010; 76 FR 12866, Mar. 9, 2011; 78 FR 6703, Jan. 30, 2013]

§63.6630 How do I demonstrate initial compliance with the emission limitations, operating limitations, and other requirements?

(a) You must demonstrate initial compliance with each emission limitation, operating limitation, and other requirement that applies to you according to Table 5 of this subpart.

(b) During the initial performance test, you must establish each operating limitation in Tables 1b and 2b of this subpart that applies to you.

(c) You must submit the Notification of Compliance Status containing the results of the initial compliance demonstration according to the requirements in §63.6645.

(d) Non-emergency 4SRB stationary RICE complying with the requirement to reduce formaldehyde emissions by 76 percent or more can demonstrate initial compliance with the formaldehyde emission limit by testing for THC instead of formaldehyde. The testing must be conducted according to the requirements in Table 4 of this subpart. The average reduction of emissions of THC determined from the performance test must be equal to or greater than 30 percent.

(e) The initial compliance demonstration required for existing nonemergency 4SLB and 4SRB stationary RICE with a site rating of more than 500 HP located at an area source of HAP that are not remote stationary RICE and that are operated more than 24 hours per calendar year must be conducted according to the following requirements:

(1) The compliance demonstration must consist of at least three test runs.

(2) Each test run must be of at least 15 minute duration, except that each test conducted using the method in appendix A to this subpart must consist of at least one measurement cycle and include at least 2 minutes of test data phase measurement.

(3) If you are demonstrating compliance with the CO concentration or CO

percent reduction requirement, you must measure CO emissions using one of the CO measurement methods specified in Table 4 of this subpart, or using appendix A to this subpart.

(4) If you are demonstrating compliance with the THC percent reduction requirement, you must measure THC emissions using Method 25A, reported as propane, of 40 CFR part 60, appendix A.

(5) You must measure O_2 using one of the O_2 measurement methods specified in Table 4 of this subpart. Measurements to determine O_2 concentration must be made at the same time as the measurements for CO or THC concentration.

(6) If you are demonstrating compliance with the CO or THC percent reduction requirement, you must measure CO or THC emissions and O_2 emissions simultaneously at the inlet and outlet of the control device.

[69 FR 33506, June 15, 2004, as amended at 78 FR 6704, Jan. 30, 2013]

CONTINUOUS COMPLIANCE REQUIREMENTS

§63.6635 How do I monitor and collect data to demonstrate continuous compliance?

(a) If you must comply with emission and operating limitations, you must monitor and collect data according to this section.

(b) Except for monitor malfunctions, associated repairs, required performance evaluations, and required quality assurance or control activities, you must monitor continuously at all times that the stationary RICE is operating. A monitoring malfunction is any sudden, infrequent, not reasonably preventable failure of the monitoring to provide valid data. Monitoring failures that are caused in part by poor maintenance or careless operation are not malfunctions.

(c) You may not use data recorded during monitoring malfunctions, associated repairs, and required quality assurance or control activities in data averages and calculations used to report emission or operating levels. You must, however, use all the valid data collected during all other periods.

 $[69\ {\rm FR}$ 33506, June 15, 2004, as amended at 76 FR 12867, Mar. 9, 2011]

§63.6640 How do I demonstrate continuous compliance with the emission limitations, operating limitations, and other requirements?

(a) You must demonstrate continuous compliance with each emission limitation, operating limitation, and other requirements in Tables 1a and 1b, Tables 2a and 2b, Table 2c, and Table 2d to this subpart that apply to you according to methods specified in Table 6 to this subpart.

(b) You must report each instance in which you did not meet each emission limitation or operating limitation in Tables 1a and 1b, Tables 2a and 2b, Table 2c, and Table 2d to this subpart that apply to you. These instances are deviations from the emission and operating limitations in this subpart. These deviations must be reported according to the requirements in §63.6650. If you change your catalyst, you must reestablish the values of the operating parameters measured during the initial performance test. When you reestablish the values of your operating parameters, you must also conduct a performance test to demonstrate that you are meeting the required emission limitation applicable to your stationary RICE.

(c) The annual compliance demonstration required for existing nonemergency 4SLB and 4SRB stationary RICE with a site rating of more than 500 HP located at an area source of HAP that are not remote stationary RICE and that are operated more than 24 hours per calendar year must be conducted according to the following reouirements:

(1) The compliance demonstration must consist of at least one test run.

(2) Each test run must be of at least 15 minute duration, except that each test conducted using the method in appendix A to this subpart must consist of at least one measurement cycle and include at least 2 minutes of test data phase measurement.

(3) If you are demonstrating compliance with the CO concentration or CO percent reduction requirement, you must measure CO emissions using one of the CO measurement methods specified in Table 4 of this subpart, or using appendix A to this subpart. (4) If you are demonstrating compliance with the THC percent reduction requirement, you must measure THC emissions using Method 25A, reported as propane, of 40 CFR part 60, appendix A.

(5) You must measure O_2 using one of the O_2 measurement methods specified in Table 4 of this subpart. Measurements to determine O_2 concentration must be made at the same time as the measurements for CO or THC concentration.

(6) If you are demonstrating compliance with the CO or THC percent reduction requirement, you must measure CO or THC emissions and O_2 emissions simultaneously at the inlet and outlet of the control device.

(7) If the results of the annual compliance demonstration show that the emissions exceed the levels specified in Table 6 of this subpart, the stationary RICE must be shut down as soon as safely possible, and appropriate corrective action must be taken (e.g., repairs, catalyst cleaning, catalyst replacement). The stationary RICE must be retested within 7 days of being restarted and the emissions must meet the levels specified in Table 6 of this subpart. If the retest shows that the emissions continue to exceed the specified levels, the stationary RICE must again be shut down as soon as safely possible, and the stationary RICE may not operate, except for purposes of startup and testing, until the owner/operator demonstrates through testing that the emissions do not exceed the levels specified in Table 6 of this subpart.

(d) For new, reconstructed, and rebuilt stationary RICE, deviations from the emission or operating limitations that occur during the first 200 hours of operation from engine startup (engine burn-in period) are not violations. Rebuilt stationary RICE means a stationary RICE that has been rebuilt as that term is defined in 40 CFR 94.11(a).

(e) You must also report each instance in which you did not meet the requirements in Table 8 to this subpart that apply to you. If you own or operate a new or reconstructed stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions (except

40 CFR Ch. I (7–1–23 Edition)

new or reconstructed 4SLB engines greater than or equal to 250 and less than or equal to 500 brake HP), a new or reconstructed stationary RICE located at an area source of HAP emissions, or any of the following RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you do not need to comply with the requirements in Table 8 to this subpart: An existing 2SLB stationary RICE, an existing 4SLB stationary RICE, an existing emergency stationary RICE, an existing limited use stationary RICE, or an existing stationary RICE which fires landfill gas or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis. If you own or operate any of the following RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you do not need to comply with the requirements in Table 8 to this subpart, except for the initial notification requirements: a new or reconstructed stationary RICE that combusts landfill gas or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis, a new or reconstructed emergency stationary RICE, or a new or reconstructed limited use stationary BICE.

(f) If you own or operate an emergency stationary RICE, you must operate the emergency stationary RICE according to the requirements in paragraphs (f)(1) through (4) of this section. In order for the engine to be considered an emergency stationary RICE under this subpart, any operation other than emergency operation, maintenance and testing, and operation in non-emergency situations for 50 hours per year, as described in paragraphs (f)(1) through (4), is prohibited. If you do not operate the engine according to the requirements in paragraphs (f)(1) through (4), the engine will not be considered an emergency engine under this subpart and must meet all requirements for non-emergency engines.

(1) There is no time limit on the use of emergency stationary RICE in emergency situations.

(2) You may operate your emergency stationary RICE for the purpose specified in paragraph (f)(2)(i) of this section

for a maximum of 100 hours per calendar year. Any operation for nonemergency situations as allowed by paragraphs (f)(3) and (4) of this section counts as part of the 100 hours per calendar year allowed by this paragraph (f)(2).

(i) Emergency stationary RICE may be operated for maintenance checks and readiness testing, provided that the tests are recommended by federal, state or local government, the manufacturer, the vendor, the regional transmission organization or equivalent balancing authority and transmission operator, or the insurance company associated with the engine. The owner or operator may petition the Administrator for approval of additional hours to be used for maintenance checks and readiness testing, but a petition is not required if the owner or operator maintains records indicating that federal, state, or local standards require maintenance and testing of emergency RICE beyond 100 hours per calendar year.

(ii)–(iii) [Reserved]

(3) Emergency stationary RICE located at major sources of HAP may be operated for up to 50 hours per calendar year in non-emergency situations. The 50 hours of operation in non-emergency situations are counted as part of the 100 hours per calendar year for maintenance and testing provided in paragraph (f)(2) of this section. The 50 hours per year for non-emergency situations cannot be used for peak shaving or non-emergency demand response, or to generate income for a facility to supply power to an electric grid or otherwise supply power as part of a financial arrangement with another entity

(4) Emergency stationary RICE located at area sources of HAP may be operated for up to 50 hours per calendar year in non-emergency situations. The 50 hours of operation in non-emergency situations are counted as part of the 100 hours per calendar year for maintenance and testing provided in paragraph (f)(2) of this section. Except as provided in paragraphs (f)(4)(i) and (ii) of this section, the 50 hours per year for non-emergency situations cannot be used for peak shaving or non-emergency demand response, or to generate income for a facility to an electric grid or otherwise supply power as part of a financial arrangement with another entity.

(i) Prior to May 3, 2014, the 50 hours per year for non-emergency situations can be used for peak shaving or nonemergency demand response to generate income for a facility, or to otherwise supply power as part of a financial arrangement with another entity if the engine is operated as part of a peak shaving (load management program) with the local distribution system operator and the power is provided only to the facility itself or to support the local distribution system.

(ii) The 50 hours per year for nonemergency situations can be used to supply power as part of a financial arrangement with another entity if all of the following conditions are met:

(A) The engine is dispatched by the local balancing authority or local transmission and distribution system operator.

(B) The dispatch is intended to mitigate local transmission and/or distribution limitations so as to avert potential voltage collapse or line overloads that could lead to the interruption of power supply in a local area or region.

(C) The dispatch follows reliability, emergency operation or similar protocols that follow specific NERC, regional, state, public utility commission or local standards or guidelines.

(D) The power is provided only to the facility itself or to support the local transmission and distribution system.

(E) The owner or operator identifies and records the entity that dispatches the engine and the specific NERC, regional, state, public utility commission or local standards or guidelines that are being followed for dispatching the engine. The local balancing authority or local transmission and distribution system operator may keep these records on behalf of the engine owner or operator.

[69 FR 33506, June 15, 2004, as amended at 71
FR 20467, Apr. 20, 2006; 73 FR 3606, Jan. 18, 2008; 75 FR 9676, Mar. 3, 2010; 75 FR 51591, Aug. 20, 2010; 78 FR 6704, Jan. 30, 2013; 87 FR 48607, Aug. 10, 2022]

§63.6640

NOTIFICATIONS, REPORTS, AND RECORDS

§63.6645 What notifications must I submit and when?

(a) You must submit all of the notifications in \S 63.7(b) and (c), 63.8(e), (f)(4) and (f)(6), 63.9(b) through (e), and (g) and (h) that apply to you by the dates specified if you own or operate any of the following:

(1) An existing stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions.

(2) An existing stationary RICE located at an area source of HAP emissions.

(3) A stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions.

(4) A new or reconstructed 4SLB stationary RICE with a site rating of greater than or equal to 250 HP located at a major source of HAP emissions.

(5) This requirement does not apply if you own or operate an existing stationary RICE less than 100 HP, an existing stationary emergency RICE, or an existing stationary RICE that is not subject to any numerical emission standards.

(b) As specified in §63.9(b)(2), if you start up your stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions before the effective date of this subpart, you must submit an Initial Notification not later than December 13, 2004, or no later than 120 days after the source becomes subject to this subpart, whichever is later.

(c) If you start up your new or reconstructed stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions on or after August 16, 2004, you must submit an Initial Notification not later than 120 days after you become subject to this subpart.

(d) As specified in §63.9(b)(2), if you start up your stationary RICE with a site rating of equal to or less than 500 brake HP located at a major source of HAP emissions before the effective date of this subpart and you are required to submit an initial notification, you must submit an Initial Notification not later than July 16, 2008, or no later than 120 days after the source 40 CFR Ch. I (7–1–23 Edition)

becomes subject to this subpart, whichever is later.

(e) If you start up your new or reconstructed stationary RICE with a site rating of equal to or less than 500 brake HP located at a major source of HAP emissions on or after March 18, 2008 and you are required to submit an initial notification, you must submit an Initial Notification not later than 120 days after you become subject to this subpart.

(f) If you are required to submit an Initial Notification but are otherwise not affected by the requirements of this subpart, in accordance with $\S63.6590(b)$, your notification should include the information in $\S63.9(b)(2)(i)$ through (v), and a statement that your stationary RICE has no additional requirements and explain the basis of the exclusion (for example, that it operates exclusively as an emergency stationary RICE if it has a site rating of more than 500 brake HP located at a major source of HAP emissions).

(g) If you are required to conduct a performance test, you must submit a Notification of Intent to conduct a performance test at least 60 days before the performance test is scheduled to begin as required in $\S63.7(b)(1)$.

(h) If you are required to conduct a performance test or other initial compliance demonstration as specified in Tables 4 and 5 to this subpart, you must submit a Notification of Compliance Status according to §63.9(h)(2)(ii).

(1) For each initial compliance demonstration required in Table 5 to this subpart that does not include a performance test, you must submit the Notification of Compliance Status before the close of business on the 30th day following the completion of the initial compliance demonstration.

(2) For each initial compliance demonstration required in Table 5 to this subpart that includes a performance test conducted according to the requirements in Table 3 to this subpart, you must submit the Notification of Compliance Status, including the performance test results, before the close of business on the 60th day following the completion of the performance test according to §63.10(d)(2).

(i) If you own or operate an existing non-emergency CI RICE with a site rating of more than 300 HP located at an area source of HAP emissions that is certified to the Tier 1 or Tier 2 emission standards in Table 1 of 40 CFR 89.112 and subject to an enforceable state or local standard requiring engine replacement and you intend to meet management practices rather than emission limits, as specified in §63.6603(d), you must submit a notification by March 3, 2013, stating that you intend to use the provision in §63.6603(d) and identifying the state or local regulation that the engine is subiect to.

[73 FR 3606, Jan. 18, 2008, as amended at 75
FR 9677, Mar. 3, 2010; 75 FR 51591, Aug. 20, 2010; 78 FR 6705, Jan. 30, 2013; 85 FR 73912, Nov. 19, 2020]

§63.6650 What reports must I submit and when?

(a) You must submit each report in Table 7 of this subpart that applies to you.

(b) Unless the Administrator has approved a different schedule for submission of reports under §63.10(a), you must submit each report by the date in Table 7 of this subpart and according to the requirements in paragraphs (b)(1) through (b)(9) of this section.

(1) For semiannual Compliance reports, the first Compliance report must cover the period beginning on the compliance date that is specified for your affected source in §63.6595 and ending on June 30 or December 31, whichever date is the first date following the end of the first calendar half after the compliance date that is specified for your source in §63.6595.

(2) For semiannual Compliance reports, the first Compliance report must be postmarked or delivered no later than July 31 or January 31, whichever date follows the end of the first calendar half after the compliance date that is specified for your affected source in §63.6595.

(3) For semiannual Compliance reports, each subsequent Compliance report must cover the semiannual reporting period from January 1 through June 30 or the semiannual reporting period from July 1 through December 31.

(4) For semiannual Compliance reports, each subsequent Compliance report must be postmarked or delivered no later than July 31 or January 31, whichever date is the first date following the end of the semiannual reporting period.

(5) For each stationary RICE that is subject to permitting regulations pursuant to 40 CFR part 70 or 71, and if the permitting authority has established dates for submitting semiannual reports pursuant to 40 CFR 70.6(a)(3)(iii)(A) or 40 CFR 71.6(a)(3)(iii)(A), you may submit the first and subsequent Compliance reports according to the dates the permitting authority has established instead of according to the dates in paragraphs (b)(1) through (b)(4) of this section.

(6) For annual Compliance reports, the first Compliance report must cover the period beginning on the compliance date that is specified for your affected source in §63.6595 and ending on December 31.

(7) For annual Compliance reports, the first Compliance report must be postmarked or delivered no later than January 31 following the end of the first calendar year after the compliance date that is specified for your affected source in §63.6595.

(8) For annual Compliance reports, each subsequent Compliance report must cover the annual reporting period from January 1 through December 31.

(9) For annual Compliance reports, each subsequent Compliance report must be postmarked or delivered no later than January 31.

(c) The Compliance report must contain the information in paragraphs (c)(1) through (6) of this section.

(1) Company name and address.

(2) Statement by a responsible official, with that official's name, title, and signature, certifying the accuracy of the content of the report.

(3) Date of report and beginning and ending dates of the reporting period.

(4) If you had a malfunction during the reporting period, the compliance report must include the number, duration, and a brief description for each type of malfunction which occurred during the reporting period and which

40 CFR Ch. I (7–1–23 Edition)

caused or may have caused any applicable emission limitation to be exceeded. The report must also include a description of actions taken by an owner or operator during a malfunction of an affected source to minimize emissions in accordance with §63.6605(b), including actions taken to correct a malfunction.

(5) If there are no deviations from any emission or operating limitations that apply to you, a statement that there were no deviations from the emission or operating limitations during the reporting period.

(6) If there were no periods during which the continuous monitoring system (CMS), including CEMS and CPMS, was out-of-control, as specified in $\S63.8(c)(7)$, a statement that there were no periods during which the CMS was out-of-control during the reporting period.

(d) For each deviation from an emission or operating limitation that occurs for a stationary RICE where you are not using a CMS to comply with the emission or operating limitations in this subpart, the Compliance report must contain the information in paragraphs (c)(1) through (4) of this section and the information in paragraphs (d)(1) and (2) of this section.

(1) The total operating time of the stationary RICE at which the deviation occurred during the reporting period.

(2) Information on the number, duration, and cause of deviations (including unknown cause, if applicable), as applicable, and the corrective action taken.

(e) For each deviation from an emission or operating limitation occurring for a stationary RICE where you are using a CMS to comply with the emission and operating limitations in this subpart, you must include information in paragraphs (c)(1) through (4) and (e)(1) through (12) of this section.

(1) The date and time that each malfunction started and stopped.

(2) The date, time, and duration that each CMS was inoperative, except for zero (low-level) and high-level checks.

(3) The date, time, and duration that each CMS was out-of-control, including the information in §63.8(c)(8).

(4) The date and time that each deviation started and stopped, and whether each deviation occurred during a period of malfunction or during another period.

(5) A summary of the total duration of the deviation during the reporting period, and the total duration as a percent of the total source operating time during that reporting period.

(6) A breakdown of the total duration of the deviations during the reporting period into those that are due to control equipment problems, process problems, other known causes, and other unknown causes.

(7) A summary of the total duration of CMS downtime during the reporting period, and the total duration of CMS downtime as a percent of the total operating time of the stationary RICE at which the CMS downtime occurred during that reporting period.

(8) An identification of each parameter and pollutant (CO or formaldehyde) that was monitored at the stationary RICE.

(9) A brief description of the stationary RICE.

(10) A brief description of the CMS.

(11) The date of the latest CMS certification or audit.

(12) A description of any changes in CMS, processes, or controls since the last reporting period.

(f) Each affected source that has obtained a title V operating permit pursuant to 40 CFR part 70 or 71 must report all deviations as defined in this subpart in the semiannual monitoring report required by 40 CFR 70.6 (a)(3)(iii)(A) or 40 CFR 71.6(a)(3)(iii)(A). If an affected source submits a Compliance report pursuant to Table 7 of this subpart along with, or as part of, the semiannual monitoring report required by 40 CFR 70.6(a)(3)(iii)(A) or 40 CFR 71.6(a)(3)(iii)(A), and the Compliance report includes all required information concerning deviations from any emission or operating limitation in this subpart, submission of the Compliance report shall be deemed to satisfy any obligation to report the same deviations in the semiannual monitoring report. However, submission of a Compliance report shall not otherwise affect any obligation the affected source may have to report deviations from permit requirements to the permit authority.

§63.6650

(g) If you are operating as a new or reconstructed stationary RICE which fires landfill gas or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis, you must submit an annual report according to Table 7 of this subpart by the date specified unless the Administrator has approved a different schedule, according to the information described in paragraphs (b)(1) through (b)(5) of this section. You must report the data specified in (g)(1) through (g)(3) of this section.

(1) Fuel flow rate of each fuel and the heating values that were used in your calculations. You must also demonstrate that the percentage of heat input provided by landfill gas or digester gas is equivalent to 10 percent or more of the total fuel consumption on an annual basis.

(2) The operating limits provided in your federally enforceable permit, and any deviations from these limits.

(3) Any problems or errors suspected with the meters.

(h) If you own or operate an emergency stationary RICE with a site rating of more than 100 brake HP that operates for the purpose specified in $\S63.6640(f)(4)(ii)$, you must submit an annual report according to the requirements in paragraphs (h)(1) through (3) of this section.

(1) The report must contain the following information:

(i) Company name and address where the engine is located.

(ii) Date of the report and beginning and ending dates of the reporting period.

(iii) Engine site rating and model year.

(iv) Latitude and longitude of the engine in decimal degrees reported to the fifth decimal place.

(v)-(vi) [Reserved]

(vii) Hours spent for operation for the purpose specified in $\S63.6640(f)(4)(ii)$, including the date, start time, and end time for engine operation for the purposes specified in $\S63.6640(f)(4)(ii)$. The report must also identify the entity that dispatched the engine and the situation that necessitated the dispatch of the engine.

(viii) If there were no deviations from the fuel requirements in §63.6604 that apply to the engine (if any), a statement that there were no deviations from the fuel requirements during the reporting period.

(ix) If there were deviations from the fuel requirements in §63.6604 that apply to the engine (if any), information on the number, duration, and cause of deviations, and the corrective action taken.

(2) The first annual report must cover the calendar year 2015 and must be submitted no later than March 31, 2016. Subsequent annual reports for each calendar year must be submitted no later than March 31 of the following calendar year.

(3) The annual report must be submitted electronically using the subpart specific reporting form in the Compliance and Emissions Data Reporting Interface (CEDRI) that is accessed through EPA's Central Data Exchange (CDX) (www.epa.gov/cdx). However, if the reporting form specific to this subpart is not available in CEDRI at the time that the report is due, the written report must be submitted to the Administrator at the appropriate address listed in §63.13.

[69 FR 33506, June 15, 2004, as amended at 75 FR 9677, Mar. 3, 2010; 78 FR 6705, Jan. 30, 2013; 87 FR 48607, Aug. 10, 2022]

§63.6655 What records must I keep?

(a) If you must comply with the emission and operating limitations, you must keep the records described in paragraphs (a)(1) through (a)(5), (b)(1) through (b)(3) and (c) of this section.

(1) A copy of each notification and report that you submitted to comply with this subpart, including all documentation supporting any Initial Notification or Notification of Compliance Status that you submitted, according to the requirement in §63.10(b)(2)(xiv).

(2) Records of the occurrence and duration of each malfunction of operation (i.e., process equipment) or the air pollution control and monitoring equipment.

(3) Records of performance tests and performance evaluations as required in §63.10(b)(2)(viii).

(4) Records of all required maintenance performed on the air pollution control and monitoring equipment. (5) Records of actions taken during periods of malfunction to minimize emissions in accordance with §63.6605(b), including corrective actions to restore malfunctioning process and air pollution control and monitoring equipment to its normal or usual manner of operation.

(b) For each CEMS or CPMS, you must keep the records listed in paragraphs (b)(1) through (3) of this section.

(1) Records described in §63.10(b)(2)(vi) through (xi).

(2) Previous (*i.e.*, superseded) versions of the performance evaluation plan as required in 63.8(d)(3).

(3) Requests for alternatives to the relative accuracy test for CEMS or CPMS as required in 63.8(f)(6)(i), if applicable.

(c) If you are operating a new or reconstructed stationary RICE which fires landfill gas or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis, you must keep the records of your daily fuel usage monitors.

(d) You must keep the records required in Table 6 of this subpart to show continuous compliance with each emission or operating limitation that applies to you.

(e) You must keep records of the maintenance conducted on the stationary RICE in order to demonstrate that you operated and maintained the stationary RICE and after-treatment control device (if any) according to your own maintenance plan if you own or operate any of the following stationary RICE;

(1) An existing stationary RICE with a site rating of less than 100 brake HP located at a major source of HAP emissions.

(2) An existing stationary emergency RICE.

(3) An existing stationary RICE located at an area source of HAP emissions subject to management practices as shown in Table 2d to this subpart.

(f) If you own or operate any of the stationary RICE in paragraphs (f)(1) through (2) of this section, you must keep records of the hours of operation of the engine that is recorded through the non-resettable hour meter. The owner or operator must document how many hours are spent for emergency 40 CFR Ch. I (7–1–23 Edition)

operation, including what classified the operation as emergency and how many hours are spent for non-emergency operation. If the engine is used for the purpose specified in $\S63.6640(f)(4)(ii)$, the owner or operator must keep records of the notification of the emergency situation, and the date, start time, and end time of engine operation for these purposes.

(1) An existing emergency stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions that does not meet the standards applicable to non-emergency engines.

(2) An existing emergency stationary RICE located at an area source of HAP emissions that does not meet the standards applicable to non-emergency engines.

[69 FR 33506, June 15, 2004, as amended at 75
FR 9678, Mar. 3, 2010; 75 FR 51592, Aug. 20, 2010; 78 FR 6706, Jan. 30, 2013; 87 FR 48607, Aug. 10, 2022]

§63.6660 In what form and how long must I keep my records?

(a) Your records must be in a form suitable and readily available for expeditious review according to §63.10(b)(1).

(b) As specified in \$63.10(b)(1), you must keep each record for 5 years following the date of each occurrence, measurement, maintenance, corrective action, report, or record.

(c) You must keep each record readily accessible in hard copy or electronic form for at least 5 years after the date of each occurrence, measurement, maintenance, corrective action, report, or record, according to §63.10(b)(1).

[69 FR 33506, June 15, 2004, as amended at 75 FR 9678, Mar. 3, 2010]

OTHER REQUIREMENTS AND INFORMATION

§63.6665 What parts of the General Provisions apply to me?

Table 8 to this subpart shows which parts of the General Provisions in §§ 63.1 through 63.15 apply to you. If you own or operate a new or reconstructed stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions (except new or reconstructed 4SLB engines greater than or equal to

250 and less than or equal to 500 brake HP), a new or reconstructed stationary RICE located at an area source of HAP emissions, or any of the following RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you do not need to comply with any of the requirements of the General Provisions specified in Table 8: An existing 2SLB stationary RICE, an existing 4SLB stationary RICE, an existing stationary RICE that combusts landfill or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis, an existing emergency stationary RICE, or an existing limited use stationary RICE. If you own or operate any of the following RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you do not need to comply with the requirements in the General Provisions specified in Table 8 except for the initial notification requirements: A new stationary RICE that combusts landfill gas or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis, a new emergency stationary RICE, or a new limited use stationary RICE.

[75 FR 9678, Mar. 3, 2010]

§63.6670 Who implements and enforces this subpart?

(a) This subpart is implemented and enforced by the U.S. EPA, or a delegated authority such as your State, local, or tribal agency. If the U.S. EPA Administrator has delegated authority to your State, local, or tribal agency, then that agency (as well as the U.S. EPA) has the authority to implement and enforce this subpart. You should contact your U.S. EPA Regional Office to find out whether this subpart is delegated to your State, local, or tribal agency.

(b) In delegating implementation and enforcement authority of this subpart to a State, local, or tribal agency under 40 CFR part 63, subpart E, the authorities contained in paragraph (c) of this section are retained by the Administrator of the U.S. EPA and are not transferred to the State, local, or tribal agency. (c) The authorities that will not be delegated to State, local, or tribal agencies are:

(1) Approval of alternatives to the non-opacity emission limitations and operating limitations in 63.6600 under 63.6(g).

(2) Approval of major alternatives to test methods under $\S63.7(e)(2)(ii)$ and (f) and as defined in $\S63.90$.

(3) Approval of major alternatives to monitoring under 63.8(f) and as defined in 63.90.

(4) Approval of major alternatives to recordkeeping and reporting under §63.10(f) and as defined in §63.90.

(5) Approval of a performance test which was conducted prior to the effective date of the rule, as specified in $\S63.6610(b)$.

§63.6675 What definitions apply to this subpart?

Terms used in this subpart are defined in the Clean Air Act (CAA); in 40 CFR 63.2, the General Provisions of this part; and in this section as follows:

Alaska Railbelt Grid means the service areas of the six regulated public utilities that extend from Fairbanks to Anchorage and the Kenai Peninsula. These utilities are Golden Valley Electric Association; Chugach Electric Association; Matanuska Electric Association; Homer Electric Association; Anchorage Municipal Light & Power; and the City of Seward Electric System.

Area source means any stationary source of HAP that is not a major source as defined in part 63.

Associated equipment as used in this subpart and as referred to in section 112(n)(4) of the CAA, means equipment associated with an oil or natural gas exploration or production well, and includes all equipment from the well bore to the point of custody transfer, except glycol dehydration units, storage vessels with potential for flash emissions, combustion turbines, and stationary RICE.

Backup power for renewable energy means an engine that provides backup power to a facility that generates electricity from renewable energy resources, as that term is defined in Alaska Statute 42.45.045(1)(5) (incorporated by reference, see §63.14). Black start engine means an engine whose only purpose is to start up a combustion turbine.

CAA means the Clean Air Act (42 U.S.C. 7401 *et seq.*, as amended by Public Law 101-549, 104 Stat. 2399).

Commercial emergency stationary RICE means an emergency stationary RICE used in commercial establishments such as office buildings, hotels, stores, telecommunications facilities, restaurants, financial institutions such as banks, doctor's offices, and sports and performing arts facilities.

Compression ignition means relating to a type of stationary internal combustion engine that is not a spark ignition engine.

Custody transfer means the transfer of hydrocarbon liquids or natural gas: After processing and/or treatment in the producing operations, or from storage vessels or automatic transfer facilities or other such equipment, including product loading racks, to pipelines or any other forms of transportation. For the purposes of this subpart, the point at which such liquids or natural gas enters a natural gas processing plant is a point of custody transfer.

Deviation means any instance in which an affected source subject to this subpart, or an owner or operator of such a source:

(1) Fails to meet any requirement or obligation established by this subpart, including but not limited to any emission limitation or operating limitation;

(2) Fails to meet any term or condition that is adopted to implement an applicable requirement in this subpart and that is included in the operating permit for any affected source required to obtain such a permit; or

(3) Fails to meet any emission limitation or operating limitation in this subpart during malfunction, regardless or whether or not such failure is permitted by this subpart.

(4) Fails to satisfy the general duty to minimize emissions established by 63.6(e)(1)(i).

Diesel engine means any stationary RICE in which a high boiling point liquid fuel injected into the combustion chamber ignites when the air charge has been compressed to a temperature 40 CFR Ch. I (7–1–23 Edition)

sufficiently high for auto-ignition. This process is also known as compression ignition.

Diesel fuel means any liquid obtained from the distillation of petroleum with a boiling point of approximately 150 to 360 degrees Celsius. One commonly used form is fuel oil number 2. Diesel fuel also includes any non-distillate fuel with comparable physical and chemical properties (*e.g.* biodiesel) that is suitable for use in compression ignition engines.

Digester gas means any gaseous byproduct of wastewater treatment typically formed through the anaerobic decomposition of organic waste materials and composed principally of methane and CO_2 .

Dual-fuel engine means any stationary RICE in which a liquid fuel (typically diesel fuel) is used for compression ignition and gaseous fuel (typically natural gas) is used as the primary fuel.

Emergency stationary RICE means any stationary reciprocating internal combustion engine that meets all of the criteria in paragraphs (1) through (3) of this definition. All emergency stationary RICE must comply with the requirements specified in §63.6640(f) in order to be considered emergency stationary RICE. If the engine does not comply with the requirements specified in §63.6640(f), then it is not considered to be an emergency stationary RICE under this subpart.

(1) The stationary RICE is operated to provide electrical power or mechanical work during an emergency situation. Examples include stationary RICE used to produce power for critical networks or equipment (including power supplied to portions of a facility) when electric power from the local utility (or the normal power source, if the facility runs on its own power production) is interrupted, or stationary RICE used to pump water in the case of fire or flood, etc.

(2) The stationary RICE is operated under limited circumstances for situations not included in paragraph (1) of this definition, as specified in §63.6640(f).

(3) The stationary RICE operates as part of a financial arrangement with

another entity in situations not included in paragraph (1) of this definition only as allowed in 63.6640(f)(4)(i)or (ii).

Engine startup means the time from initial start until applied load and engine and associated equipment reaches steady state or normal operation. For stationary engine with catalytic controls, engine startup means the time from initial start until applied load and engine and associated equipment, including the catalyst, reaches steady state or normal operation.

Four-stroke engine means any type of engine which completes the power cycle in two crankshaft revolutions, with intake and compression strokes in the first revolution and power and exhaust strokes in the second revolution.

Gaseous fuel means a material used for combustion which is in the gaseous state at standard atmospheric temperature and pressure conditions.

Gasoline means any fuel sold in any State for use in motor vehicles and motor vehicle engines, or nonroad or stationary engines, and commonly or commercially known or sold as gasoline.

Glycol dehydration unit means a device in which a liquid glycol (including, but not limited to, ethylene glycol, diethylene glycol, or triethylene glycol) absorbent directly contacts a natural gas stream and absorbs water in a contact tower or absorption column (absorber). The glycol contacts and absorbs water vapor and other gas stream constituents from the natural gas and becomes "rich" glycol. This glycol is then regenerated in the glycol dehydration unit reboiler. The "lean" glycol is then recycled.

Hazardous air pollutants (HAP) means any air pollutants listed in or pursuant to section 112(b) of the CAA.

Institutional emergency stationary RICE means an emergency stationary RICE used in institutional establishments such as medical centers, nursing homes, research centers, institutions of higher education, correctional facilities, elementary and secondary schools, libraries, religious establishments, police stations, and fire stations.

ISO standard day conditions means 288 degrees Kelvin (15 degrees Celsius), 60

percent relative humidity and 101.3 kilopascals pressure.

Landfill gas means a gaseous by-product of the land application of municipal refuse typically formed through the anaerobic decomposition of waste materials and composed principally of methane and CO₂.

Lean burn engine means any twostroke or four-stroke spark ignited engine that does not meet the definition of a rich burn engine.

Limited use stationary RICE means any stationary RICE that operates less than 100 hours per year.

Liquefied petroleum gas means any liquefied hydrocarbon gas obtained as a by-product in petroleum refining of natural gas production.

Liquid fuel means any fuel in liquid form at standard temperature and pressure, including but not limited to diesel, residual/crude oil, kerosene/naphtha (jet fuel), and gasoline.

Major Source, as used in this subpart, shall have the same meaning as in §63.2, except that:

(1) Emissions from any oil or gas exploration or production well (with its associated equipment (as defined in this section)) and emissions from any pipeline compressor station or pump station shall not be aggregated with emissions from other similar units, to determine whether such emission points or stations are major sources, even when emission points are in a contiguous area or under common control;

(2) For oil and gas production facilities, emissions from processes, operations, or equipment that are not part of the same oil and gas production facility, as defined in §63.1271 of subpart HHH of this part, shall not be aggregated;

(3) For production field facilities, only HAP emissions from glycol dehydration units, storage vessel with the potential for flash emissions, combustion turbines and reciprocating internal combustion engines shall be aggregated for a major source determination; and

(4) Emissions from processes, operations, and equipment that are not part of the same natural gas transmission and storage facility, as defined in §63.1271 of subpart HHH of this part, shall not be aggregated. *Malfunction* means any sudden, infrequent, and not reasonably preventable failure of air pollution control equipment, process equipment, or a process to operate in a normal or usual manner which causes, or has the potential to cause, the emission limitations in an applicable standard to be exceeded. Failures that are caused in part by poor maintenance or careless operation are not malfunctions.

Natural gas means a naturally occurring mixture of hydrocarbon and nonhydrocarbon gases found in geologic formations beneath the Earth's surface, of which the principal constituent is methane. Natural gas may be field or pipeline quality.

Non-selective catalytic reduction (NSCR) means an add-on catalytic nitrogen oxides (NO_X) control device for rich burn engines that, in a two-step reaction, promotes the conversion of excess oxygen, NO_X, CO, and volatile organic compounds (VOC) into CO₂, nitrogen, and water.

Oil and gas production facility as used in this subpart means any grouping of equipment where hydrocarbon liquids are processed, upgraded (*i.e.*, remove impurities or other constituents to meet contract specifications), or stored prior to the point of custody transfer; or where natural gas is processed, upgraded, or stored prior to entering the natural gas transmission and storage source category. For purposes of a major source determination, facility (including a building, structure, or installation) means oil and natural gas production and processing equipment that is located within the boundaries of an individual surface site as defined in this section. Equipment that is part of a facility will typically be located within close proximity to other equipment located at the same facility. Pieces of production equipment or groupings of equipment located on different oil and gas leases, mineral fee tracts, lease tracts, subsurface or surface unit areas, surface fee tracts, surface lease tracts, or separate surface sites, whether or not connected by a road, waterway, power line or pipeline, shall not be considered part of the same facility. Examples of facilities in the oil and natural gas production source category include, but are not 40 CFR Ch. I (7–1–23 Edition)

limited to, well sites, satellite tank batteries, central tank batteries, a compressor station that transports natural gas to a natural gas processing plant, and natural gas processing plants.

Oxidation catalyst means an add-on catalytic control device that controls CO and VOC by oxidation.

Peaking unit or engine means any standby engine intended for use during periods of high demand that are not emergencies.

Percent load means the fractional power of an engine compared to its maximum manufacturer's design capacity at engine site conditions. Percent load may range between 0 percent to above 100 percent.

Potential to emit means the maximum capacity of a stationary source to emit a pollutant under its physical and operational design. Any physical or operational limitation on the capacity of the stationary source to emit a pollutant, including air pollution control equipment and restrictions on hours of operation or on the type or amount of material combusted, stored, or processed, shall be treated as part of its design if the limitation or the effect it would have on emissions is federally enforceable. For oil and natural gas production facilities subject to subpart HH of this part, the potential to emit provisions in §63.760(a) may be used. For natural gas transmission and storage facilities subject to subpart HHH of this part, the maximum annual facility gas throughput for storage facilities may be determined according to §63.1270(a)(1) and the maximum annual throughput for transmission facilities may be determined according to §63.1270(a)(2).

Production field facility means those oil and gas production facilities located prior to the point of custody transfer.

Production well means any hole drilled in the earth from which crude oil, condensate, or field natural gas is extracted.

Propane means a colorless gas derived from petroleum and natural gas, with the molecular structure C_3H_8 .

Remote stationary RICE means stationary RICE meeting any of the following criteria:

(1) Stationary RICE located in an offshore area that is beyond the line of ordinary low water along that portion of the coast of the United States that is in direct contact with the open seas and beyond the line marking the seaward limit of inland waters.

(2) Stationary RICE located on a pipeline segment that meets both of the criteria in paragraphs (2)(i) and (ii) of this definition.

(i) A pipeline segment with 10 or fewer buildings intended for human occupancy and no buildings with four or more stories within 220 yards (200 meters) on either side of the centerline of any continuous 1-mile (1.6 kilometers) length of pipeline. Each separate dwelling unit in a multiple dwelling unit building is counted as a separate building intended for human occupancy.

(ii) The pipeline segment does not lie within 100 yards (91 meters) of either a building or a small, well-defined outside area (such as a playground, recreation area, outdoor theater, or other place of public assembly) that is occupied by 20 or more persons on at least 5 days a week for 10 weeks in any 12month period. The days and weeks need not be consecutive. The building or area is considered occupied for a full day if it is occupied for any portion of the day.

(iii) For purposes of this paragraph (2), the term pipeline segment means all parts of those physical facilities through which gas moves in transportation, including but not limited to pipe, valves, and other appurtenance attached to pipe, compressor units, metering stations, regulator stations, delivery stations, holders, and fabricated assemblies. Stationary RICE located within 50 yards (46 meters) of the pipeline segment providing power for equipment on a pipeline segment are part of the pipeline segment. Transportation of gas means the gathering, transmission, or distribution of gas by pipeline, or the storage of gas. A building is intended for human occupancy if its primary use is for a purpose involving the presence of humans.

(3) Stationary RICE that are not located on gas pipelines and that have 5 or fewer buildings intended for human occupancy and no buildings with four or more stories within a 0.25 mile radius around the engine. A building is intended for human occupancy if its primary use is for a purpose involving the presence of humans.

Residential emergency stationary RICE means an emergency stationary RICE used in residential establishments such as homes or apartment buildings.

Responsible official means responsible official as defined in 40 CFR 70.2.

Rich burn engine means any fourstroke spark ignited engine where the manufacturer's recommended operating air/fuel ratio divided by the stoichiometric air/fuel ratio at full load conditions is less than or equal to 1.1. Engines originally manufactured as rich burn engines, but modified prior to December 19, 2002 with passive emission control technology for NO_X (such as pre-combustion chambers) will be considered lean burn engines. Also, existing engines where there are no manufacturer's recommendations regarding air/fuel ratio will be considered a rich burn engine if the excess oxygen content of the exhaust at full load conditions is less than or equal to 2 percent.

Site-rated HP means the maximum manufacturer's design capacity at engine site conditions.

Spark ignition means relating to either: A gasoline-fueled engine: or any other type of engine with a spark plug (or other sparking device) and with operating characteristics significantly similar to the theoretical Otto combustion cycle. Spark ignition engines usually use a throttle to regulate intake air flow to control power during normal operation. Dual-fuel engines in which a liquid fuel (typically diesel fuel) is used for CI and gaseous fuel (typically natural gas) is used as the primary fuel at an annual average ratio of less than 2 parts diesel fuel to 100 parts total fuel on an energy equivalent basis are spark ignition engines.

Stationary reciprocating internal combustion engine (RICE) means any reciprocating internal combustion engine which uses reciprocating motion to convert heat energy into mechanical work and which is not mobile. Stationary RICE differ from mobile RICE in that a stationary RICE is not a nonroad engine as defined at 40 CFR 1068.30, and is not used to propel a

Pt. 63, Subpt. ZZZZ, Table 1a

motor vehicle or a vehicle used solely for competition.

Stationary RICE test cell/stand means an engine test cell/stand, as defined in subpart PPPPP of this part, that tests stationary RICE.

Stoichiometric means the theoretical air-to-fuel ratio required for complete combustion.

Storage vessel with the potential for flash emissions means any storage vessel that contains a hydrocarbon liquid with a stock tank gas-to-oil ratio equal to or greater than 0.31 cubic meters per liter and an American Petroleum Institute gravity equal to or greater than 40 degrees and an actual annual average hydrocarbon liquid throughput equal to or greater than 79,500 liters per day. Flash emissions occur when dissolved hydrocarbons in the fluid evolve from solution when the fluid pressure is reduced.

40 CFR Ch. I (7–1–23 Edition)

Subpart means 40 CFR part 63, subpart ZZZZ.

Surface site means any combination of one or more graded pad sites, gravel pad sites, foundations, platforms, or the immediate physical location upon which equipment is physically affixed.

Two-stroke engine means a type of engine which completes the power cycle in single crankshaft revolution by combining the intake and compression operations into one stroke and the power and exhaust operations into a second stroke. This system requires auxiliary scavenging and inherently runs lean of stoichiometric.

[69 FR 33506, June 15, 2004, as amended at 71
FR 20467, Apr. 20, 2006; 73 FR 3607, Jan. 18, 2008; 75 FR 9679, Mar. 3, 2010; 75 FR 51592, Aug. 20, 2010; 76 FR 12867, Mar. 9, 2011; 78 FR 6706, Jan. 30, 2013; 87 FR 48608, Aug. 10, 2022]

TABLE 1A TO SUBPART ZZZZ OF PART 63—EMISSION LIMITATIONS FOR EXISTING, NEW, AND RECONSTRUCTED SPARK IGNITION, 4SRB STATIONARY RICE >500 HP LOCATED AT A MAJOR SOURCE OF HAP EMISSIONS

As stated in §§ 63.6600 and 63.6640, you must comply with the following emission limitations at 100 percent load plus or minus 10 percent for existing, new and reconstructed 4SRB stationary RICE >500 HP located at a major source of HAP emissions:

For each	Yo⊡ m⊡t meet the forowing emi⊡ion imitation, e⊡cept ⊡ring perio⊡ of ⊡tart⊡p	D⊡ring perio⊡ of ⊡tart⊡p ⊡o⊡ m⊡t
1. 4SR Itationar RC	a. Re⊡ce forma ⊡eh ⊡e emi⊡ion □ 6 per cent or more. If ⊡ commence con Tr⊂tion or recon Tr⊂tion Etween Decem ⊡er 19, 2002 an] Jne 15, 2004, ⊡ ma re⊡ce forma ⊡e h ⊡e emi⊡ion □ 5 percent or more inti J⊡ne 15, 200 or. Limit the concentration of forma ⊡eh ⊡e in the Etationar R:C e ⊡aa⊡t to 350 pp ⊡ or @⊡ at 15 percent O ₂ .	Minimice the engine time pent at ice and minimice the engine tart p time at lart p to a periodenee for appropriate and lafe load ing of the engine, not to ecceed 30 minited after which time the non tart p emicion finitiation app 1

¹ So rce can petition the aminiatrator parameter to the rearrement of 40 CFR 63.6 g for alternatice work practice

[75 FR 9679, Mar. 3, 2010, as amended at 75 FR 51592, Aug. 20, 2010]

TABLE 1b to SUBPART ZZZZ OF PART 63—OPERATING LIMITATIONS FOR EXISTING, NEW, AND RECONSTRUCTED SI 4SRB STATIONARY RICE >500 HP LOCATED AT A MAJOR SOURCE OF HAP EMISSIONS

As stated in §§ 63.6600, 63.6603, 63.6630 and 63.6640, you must comply with the following operating limitations for existing, new and reconstructed 4SRB stationary RICE >500 HP located at a major source of HAP emissions:

Pt. 63, Subpt. ZZZZ, Table 2a

For each	Yo⊡ m⊡t meet the fo®wing operating ®mitation, e⊡cept ⊡ring perio⊡ of ⊡tart⊡p
1. eliting, new and reconstructed 4SR dationar RiC >500 HP located at a major locree of HCP emillion compling with the redirement to redice formalehine emilion in 6 percent or more lor in 5 percent or more, if applicated and ling SCRC or eliting, new and reconstructed 4SR dationar RiC >500 HP located at a major locree of HCP emillion compling with the redirement to limit the concentration of formaled hime in the dationar RiC enhalito 350 ppill or lett at 15 percent 0 ₂ and ling SCR	a. maintain bor catallit to that the prelime propacrol the catallit boe not change to more than 2 inche of water at 100 percent to a pill or minim 10 percent from the prelime propacrol the catallit mealine initial performance tellant. I maintain the temperature of bor clationar RiC e chant to 50 °F and et than or ettain 1250 °F.1
2. editing, new and recondrate 4SR dationar RiC >500 HP locate at a maior correct of HP emittion company with the redirement to redice formate emittion in 6 percent or more for to 5 percent or more, if applicate and not ting ISCR or	Comp⊡ with an operating îmitation appro e the comin i⊡rator.
e ining, new an reconstructed 4SR intainant RiC >500 HP located at a matrix increase of HCP emission composing with the residence to limit the concentration of formalled here in the diationar RiC estability to 350 pp or less at 15 percent O_2 and not ling SCR.	

¹So rce can petition the control rate primate to the recirement of 40 CFR 63.8 for a fifterent temperative range.

[78 FR 6706, Jan. 30, 2013]

TABLE 2a to Subpart ZZZZ of Part 63—Emission Limitations for New and Reconstructed 2SLB and Compression Ignition Stationary RICE >500 HP and New and Reconstructed 4SLB Stationary RICE >250 HP Located at a Major Source of HAP Emissions

As stated in \$63.6600 and 63.6640, you must comply with the following emission limitations for new and reconstructed lean burn and new and reconstructed compression ignition stationary RICE at 100 percent load plus or minus 10 percent:

For each	Yo m m meet the for wing emi mon îmitation, e cept ⊡ring perio of tart p	During perio of tart pomet
1. 2SL □ tationar □ R C □	a. Re⊡ce CO emi⊡on □ 58 percent or more⊡or Limit concentration of forma ⊡eh ⊡e in the ⊡ationar □ Rt© e cha⊡t to 12 ppm ⊡ or ē⊡ at 15 percent O ₂ . If oo commence contriction or recon ⊡riction etween Decem er 19, 2002 an J ∩e 15, 2004, oo ma⊡imit con- centration of forma ⊡eh ⊡e to 10 ppm ⊡ or iē⊡ at 15 percent O ₂ _nti J ∩e 15, 2000	$\begin{array}{ccccc} \mbox{percent} & Minimize the engine limit important in an limit important impo$
2. 4SL □ dationar □ R IC □	a. Re⊡ce CO emi⊡on⊡ ⊡ 93 percent or more⊡or Limit concentration of forma⊡eh⊡e in the ⊡ationar⊡ RICe etha⊡t to 14	
3. C⊡tationar⊡ RtC□	a. Re⊡ce CO emiiion □ □ 0 percent or or more⊡or □ Limit concentration of forma ⊡eh ⊡e in the ⊡ationar⊡ RiCe e ha ⊡t to 580 pp⊡ or ie⊡ at 15 percent O ₂ .	

¹So rce can petition the minitator primat to the redirement of 40 CFR 63.6 g for alternative work practice

[75 FR 9680, Mar. 3, 2010]

Pt. 63, Subpt. ZZZZ, Table 2b

40 CFR Ch. I (7-1-23 Edition)

TABLE 2b TO SUBPART ZZZZ OF PART 63—OPERATING LIMITATIONS FOR NEW AND RECONSTRUCTED 2SLB AND CI STATIONARY RICE >500 HP LOCATED AT A MAJOR SOURCE OF HAP EMISSIONS, NEW AND RECONSTRUCTED 4SLB STATIONARY RICE \geq 250 HP LOCATED AT A MAJOR SOURCE OF HAP EMISSIONS, EXISTING CI STATIONARY RICE >500 HP

As stated in §§63.6600, 63.6601, 63.6603, 63.6630, and 63.6640, you must comply with the following operating limitations for new and reconstructed 2SLB and CI stationary RICE >500 HP located at a major source of HAP emissions; new and reconstructed 4SLB stationary RICE >250 HP located at a major source of HAP emissions; and existing CI stationary RICE >500 HP:

For each	Yo⊡ m⊡t meet the fo®wing operating ≋mitation, e⊡cept ⊡ring perio⊡ of ⊡tart⊡p
 lew an recontrictel 2SL an Contaionar RiC >500 HP locate at a maior borce of HP emission an new an recontrictel 4SL dationar RiC ≥250 HP locate at a maior borce of HP emission compling with the re- direment to retice CO emission and the main of ation cataintan ew an recontrictel 2SL an Contaionar RiC >500 HP locate at a maior locre of HP emission and new an recontrictel 4SL dationar RiC ≥250 HP locate at a maior locre of HP emission compling with the retire ment to limit the concentration of formate/lie in the taa tionar RiC e that an ing an of ation catain. Contain RiC Picto HP control at a maior locre of HP emission compling with the retire ment to limit the concentration of formate/lie in the taa tionar RiC e that an ing an of ation catain. Containar RiC >500 HP control at an ing an of ation catain. 	 a. maintain bor catallie to that the prelime propactor the catallie coent change more than 2 inche of water at 100 percent loap minor 10 percent from the prelime propactor the catallit that wall mealine maintain the temperature of bor clationar RIC echant to 450 °F and emitted to that the prelime propactor the catallit to that the prelime propactor the catallit coent change more than 2 inche of water at room the catallit coent change of the catallit that wall mealine the prelime roop acrows the catallit coent change more than 2 inche of water from the prelime roop acrows the catallit the temperature of coent the catallit the catallit the catallit the temperature of coent the catallit the catallit the temperature of coent the catallit the catallit the catallit the temperature of coent the catallit the catallit the catallit the temperature of coent the catallit the catallit the temperature of coent the catallit the catallit the catallit the temperature of coent the catallit the temperature of coent the catallit the catallit the temperature of coent the catallit the catallit the catallit the temperature of coent the catallit the temperature of coe
3. ew an recontricte 2SL an Contationar RiC >500 HP locate at a maior borce of HD emillion an new an recontricte 4SL dationar RiC ≥250 HP locate at a maior borce of HD emillion company with the re- direment to reice CO emillion an not ling an olia- tion catallan ew an recontricte 2SL an Contationar RiC >500 HP locate at a maior borce of HD emillion an new an recontricte 4SL dationar RiC ≥250 HP locate at a maior to reice company with the relian	Comp⊡ with an operating imitation appro⊡e □ □ the □□min□ i⊡trator.
maio Corce of HCP emilion compling with the recirce ment to imit the concentration of formal@hire in the ta- tionar RIC ethal an not cing an of ation catalin an editing Collationar RIC >500 HP compling with the re circement to limit or recirce the concentration of CO in the tationar RIC ethal an not cing an of ation catalin.	

¹So rce can petition the cminitrator promat to the recirement of 40 CFR 63.8 for a fifterent temperatore range.

[78 FR 6707, Jan. 30, 2013]

TABLE 2C TO SUBPART ZZZZ OF PART 63—REQUIREMENTS FOR EXISTING COMPRESSION IGNITION STATIONARY RICE LOCATED AT A MAJOR SOURCE OF HAP EMISSIONS AND EXISTING SPARK IGNITION STATIONARY RICE \leq 500 HP LOCATED AT A MAJOR SOURCE OF HAP EMISSIONS

As stated in §§ 63.6600, 63.6602, and 63.6640, you must comply with the following requirements for existing compression ignition stationary RICE located at a major source of HAP emissions and existing spark ignition stationary RICE ≤ 500 HP located at a major source of HAP emissions:

Pt. 63, Subpt. ZZZZ, Table 2c

For each	Yo⊡ m⊡t meet the fo⊡owing re⊡ire⊡ ment, e⊡cept ⊡ring perio⊡ of ⊡tart⊡p 	D⊡ring perio⊞ of ⊡tart⊡p ⊡o⊡ m⊡t	
I. omergenco dationaro CC RiCo ano ack dart dationaro CoRiCo 1.	a. Change oi an fitter eler 500 hor of operation or ann all, whicheler come fir 1.2 hepect air ceaner eler 1,000 hor of operation or ann all, whicheler come fir 1, an replace a nece for c. hepect all hole an eler 500 hor of operation or ann all, which eler come fir 1, an replace a nece	Minimic the engine time pent at ice an minimic the engine tart to time at tart to a perior necessfor ap propriate and the facing of the en gine, not to ecceed 30 minited after which time the nontart pemic or imitation app 3	
2. □on	a. Change oi an filter e er 1,000 hor of operation or ann a which e er come fir 1.2 in pect air cleaner e er 1,000 hor of operation or ann a whicher come fir 1, an replace a necetar c. In pect a mhore an ett e er 500 hor of operation or ann a which e Er come fir 1, an replace a nece		
3. □on IIImergenc□, non IIIack Itart C IIta I tionar I RIC I 100≤HP≤300 HP.	Limit concentration of CO in the ta tionar RIC = cha⊡t to 230 ppm □ or e □ at 15 percent O ₂ .		
4. □on□mergenc□, non□Tāck ttart C□ta□ tionar□ RIC□ 300 <hp≤500.< td=""><td>a. Limit concentration of CO in the tationar RtC = ethatt to 49 ppm or at 15 percent O2_or Retrice CO emittion 0 percent</td><td></td></hp≤500.<>	a. Limit concentration of CO in the tationar RtC = ethatt to 49 ppm or at 15 percent O2_or Retrice CO emittion 0 percent		
5. Oon III mergenci, non III ack Itart Ita I tionar III CIRICIII >500 HP.	a. Limit concentration of CO in the tationar RCC ethat to 23 ppm or ethat 15 percent O ₂ or Retrice CO emittion 0 percent or more		
6. mergenc: tationar S RC an ack tart tationar S RC.1	a. Change oilan filter eler 500 horr of operation or ann all, whicheler come firit? in pect park pige eler 1,000 horr of operation or ann all, whicheler come firit, an replace a necetart c. in pect all hole and etter 500 horr of operation or ann all, which eler come firit, an replace a nece		
□ □on⊞mergenc□, non⊞ack ⊡art ⊡a tionar□ S□ RIC□ <100 HP that are not 2SL□ ⊡tationar□ RIC□.	a. Change oil and filter eler 1,440 hobro of operation or annial which eler comed fir 1:3 in pect park piggeter 1,440 hobro of operation or annial whicheler comed fir 1, and replace all nece tard 1,440 hobro of operation or annial which hobbro of operation or annial which whicheler comed fir 1, and replace all nece tard		
8. Oon mergenc non ack tart 2SL tationar S RC <100 HP.	a. Change oil an filter eler 4,320 hor of operation or ann all which eler come firt? In pect park pigeer 4,320 hor of operation or ann all whicher come firt, an replace a nece art c. htpect all hole an lett eler 4,320 hor of operation or ann all whicheler come firt, an replace a nece art		
9. □on@mergenc□, non□lack tart 2SL□ tationar□ R.C□ 100≤HP≤500.	Limit concentration of CO in the tall tionar RC elhat to 225 ppm or e at 15 percent O ₂ .		
 tationar RiC 100≤HP≤500. 11. □on@mergenc□ non mack dart 4SR 	tinnar Concentration of CO in the Ital tionar RiC e cha⊡t to 4 ppm⊡ or e at 15 percent O ₂ . Limit concentration of forma⊡eh⊡e in		
⊡ationar⊡ R̃iC⊡ 100≤HP≤500.	the \Box tationar \square R \square e \square ha \square t to 10.3 ppm \square or \blacksquare at 15 percent O ₂ .		

Pt. 63, Subpt. ZZZZ, Table 2d

40 CFR Ch. I (7-1-23 Edition)

For each	Yo⊡ m⊡t meet the fo⊡owing re⊡ire⊡ ment, e⊡cept ⊡ring perio⊡ of ⊡tart⊡p 	D⊡ring perio⊡ of ⊡tart⊡p ⊡o⊡ m⊡t
12. ontemergenc_ non mack that that tionar R:C 100≤HP≤500 which com onto an fillion dgeter gale islatent to 10 percent or more of the gro—heat inpt on an annacat	Limit concentration of CO in the ⊡a tionar□ RIC□ e⊡ha⊡t to 1 □ ppm □ or e □ at 15 percent O ₂ .	

¹ If an emergence engine in operating there in a second and it is not possible to the form the engine in order to perform the work practice retirement on the table retire in a 2 c of this that, or if performing the work practice on the retire in a 2 c of this there for a condaw, the work practice can be retired in the emergence in the the emergence in the there is a condaw in the work practice on the retire in the emergence is an a second with the emergence is a condaw in the work practice on the retired at the emergence is a condaw in the work practice in the emergence is a condaw in the work practice is a condaw in the emergence is a condaw in the term of the condaw is a condaw in the emergence is a condaw in the term of the condaw is a condaw in the emergence is a condaw in the term of the condaw is a condaw in the emergence is a condaw in the term of the condaw is a condaw in the term of the emergence is a condaw in the term of the condaw is a condaw in the term of the term of the emergence is a condaw in the term of the term of the condaw is a condaw in the term of the term of the term of the term of the condaw is a condaw in the term of the term of the condaw is a condaw in the term of term of term of term of term of term of terms of terms of the term of terms of the terms of the terms of the terms of terms of terms of the terms of terms of terms of the terms of terms of terms of the terms of the terms of terms

[78 FR 6708, Jan. 30, 2013, as amended at 78 FR 14457, Mar. 6, 2013]

TABLE 2d TO SUBPART ZZZZ OF PART 63-REQUIREMENTS FOR EXISTING STATIONARY RICE LOCATED AT AREA SOURCES OF HAP EMISSIONS

As stated in \S 63.6603 and 63.6640, you must comply with the following requirements for existing stationary RICE located at area sources of HAP emissions:

For each	Yo⊡ m⊡t meet the fo⊡owing re⊡re⊡ ment, e⊡cept ⊡ring perio⊡ of ⊡tart⊡p	D⊡ring perio⊡ of ⊡tart⊡p ⊡o⊡ m⊡t	
1. ⊡on⊞mergenc□ non⊞äck ttart C⊡ta⊟ tionar⊡RiC ≤300 HP.	a. Change oi an filter e er 1,000 hor of operation or ann a m, which e er come fir 1 n pect air cleaner e er 1,000 hor of operation or ann a m, whiche er come fir 1, an replace a nece ar c. n pect a hor en an et er of 500 hor of operation or ann a m, which e er come fir 1, an replace a nece	Minimice the engine time pent at ice and minimice the engine tartcp time at dartcp to a period neeced for appropriate and cafe loading of the en- gine, not to eloced 30 minced after which time the non tartcp emition imitation app.	
 On Imergenc non Imack Lart Colla tionar RIC 300<hp≤500.< li=""> </hp≤500.<>	a. Limit concentration of CO in the ta tionar□ RtC□ ethatt to 49 ppm⊡ at 15 percent O₂_or Ret_ce CO emition□ □ 0 percent or more.		
3. □onⅢmergenc□ nonⅢāck 且art C□⊡a□ tionar□RIC□ >500 HP.	a. Limit concentration of CO in the ⊡a tionar□ RtC□ echa⊡t to 23 ppm⊡ at 15 percent O2_or □ Re⊡ce CO emi⊡on□ □ 0 percent or more		
4mergenc: _tationar: C R:C an Dack _tart _tationar: C R:C.2	a. Change oilani filter elleri 500 holini of operation or anniali, whicheller come finiti in pect air cleaner elleri 1,000 holini of operation or anniali, whicheller come finit, an replace all necellari ani c. in pect all hole and letter elleri 500 holini of operation or anniali, which eller come finit, and replace all necel		
5. mergenc: ationar: S: RIC: ack lart dationar: S: RIC: nonTemer genc: nonTack dart 45L: dationar RIC: 500 HP that operate 24 hortor e: per calentar tear: nonTemergenc; nonTack dart 45R: dationar: RIC: >500 HP that operate 24 hortor is: per calentar tear. ²	a. Change oilant filter etert 500 hothof of operation or anntat, whichefer come firtition in peet t park pigterer 1,000 hothof of operation or anntat, whichefer come firt, and replace at necetart and c. In peet atthofed and tetter 500 hothof of operation or anntat, which effer come firt, and replace at nece		

Pt. 63, Subpt. ZZZZ, Table 2d

For each	Yo□ m it meet the following re ire ment, elcept iring perio of itart p	D⊡ring perio⊡ of ⊡tart⊡p ⊑o⊡ m⊡t	
6. □ontemergenc□ nont⊡ack tart 2SL□ tationar□RtC□.	 a. Change oi an filer eler 4,320 hor of operation or ann all, which eler come fir 11 in pect park pig eler 4,320 hor of operation or ann all, which eler come fir 1, an replace a nece are ano c. In pect all hole an elet eler 4,320 hor of operation or ann all, which eler come fir 1, an replace a 		
□ □onīemergenc□ non⊡īāck [tart 4SL] □tationar□ RīC□ ≤500 HP.	a. Change oi an filter eler 1,440 hor of operation or ann all, which eler come firt. of operation or ann all, which eler come firt, an replace a nece art an c. htpect all hole an ele eler 1,440 hor of operation or ann all, whicheler come firt, an replace a		
8. ⊡oniemergenc□ non⊞iack ⊡art 4SL⊡ remote ⊡tationar⊡ RIC□ >500 HP.	a. Change oi an filter eler 2,160 hor of operation or ann all, which eler come fir 11 in pect park pig eler 2,160 hor of operation or ann all, which eler come fir 1, an replace a nece are and c. https://whicheler.org/line.com/l		
9. Contemergenco, non Tack tart 4SL tationaro RIC >500 HP that are not remote tationaro RIC and that operor ate more than 24 hotro per catendar tear.	nece Lar∟ In:Ta⊡ an of Cation cata⊡t to re⊡ce H⊡P emi⊡on⊡ from the Cationar⊡ R:C□.		
10. □oniemergenc□ noni⊡ack itart 4SR□ Itationar□ RIC□ ≤500 HP.	a. Change oil an filter eler 1,440 hord of operation or ann all, which eler come firt: in pect park pigeer 1,440 hord of operation or ann all, whicheler come firt, an replace all necestart and c. in pect all hole an eletter 1,440 hord of operation or ann all, whicheler come firt, and replace all pece and		
11. □on@mergenc□ non⊞ack ⊡art 4SR□ remote ⊡ationar□ R:C□ >500 HP.	a. Change oil and filter eler 2,160 hord of operation or annial which eler comed fir 11 in pect park piggle er 2,160 hord of operation or annial whicheler comed fir 1, and replace ad nece and c. In pect ad hole and elle eler 2,160 hord of operation or annial, whicheler comed fir 1, and replace ad nece ard		
12. OnTemergenc, nonmack tart 4SR tationar RIC >500 HP that are not remote tationar RIC an that oper ate more than 24 horr per calenar tear.	In⊡ta⊞ SCR to re⊡ce H P emi⊡ion from the ⊡ationar RC.		
13. On@mergenc, non@ack dart da tionar: RC: which com an filler igedler gale and to 10 percent or more of the grom heat input on an an n@ada.	a. Change oi an filter erer 1,440 hor of operation or ann all, which ere come fir 11 n pect park pigeer 1,440 hor of operation or ann all, which erer come fir 1, an replace a nece ar		

Pt. 63, Subpt. ZZZZ, Table 3

40 CFR Ch. I (7-1-23 Edition)

For each	Yo⊡ m⊡t meet the fo⊡owing re⊡ire⊡ ment, e⊡cept ⊡ring perio⊡ of ⊡tart⊡p 	D⊡ring perio⊡ of ⊡tart⊡p ⊡o⊡ m⊡t
	c. http://pect.all.hotel.anl.tettleter 1,440 hotel.of operation or anntatt, whicheter comel firt, anl replace at necettar	

¹ Sorce hale the option to title an oilanatio program a learne in 63.66250 or thin or er to elten the pecifie oil change retirement in the 2 of thit to part. ² If an emergenciengine is operating the index of the change of the change of the change retirement in the change of t

[78 FR 6709, Jan. 30, 2013]

TABLE 3 TO SUBPART ZZZZ OF PART 63—SUBSEQUENT PERFORMANCE TESTS

As stated in §§ 63.6615 and 63.6620, you must comply with the following subsequent performance test requirements:

For each	Compiling with the reliement to	Yo m I	
1. ew or recontricte 2SL tationar R(C) >500 HP ocate at maior borce new or recontricte 4SL tationar R(C) ≥250 HP ocate at tationar R(C) ≥250 HP ocate at maior or recontricte 4SL tricte C 1ationar R(C) ≥500 HP ocate at tricte C 1ationar R(C) >500 HP ocate at	Re⊡ce CO emi⊡on⊡ an⊡ not ⊡ing a C⊡MS.	Con t t cont performance te t cemiann cat 1	
2. 4SR □ tationar □ RIC □ ≥5,000 HP Io □ cate □ at maior □o □rce □	Re⊡ce forma≣eh ⊡e emi ⊡ion □	Con ct cent performance te c	
3. Stationar RIC >500 HP locate at malor locree an new or recon thread 4SL tationar RIC 2504HP5500 locate at malor locree	Limit the concentration of formallehle in the lationar RIC elhalt.	Con ct certe cent performance te to cemiann ca c 1	
4. □□ ting non emergenc, non □ ack □tart C □ tationar □ RIC □ >500 HP that are not imite □ c □ tationar □ RIC □.	Limit or re⊡ce CO emi⊡ion⊡ an⊡ not ⊡ing a C⊡MS.	Con ct cerement performance tell eler 8,60 hold or 3 lear which eler come fir t.	
5. If ting nontemergenci, nontrack that C tationar RC >500 HP that are timite the tationar RC.	Limit or re⊡ce CO emi⊡on⊡ an⊡ not ⊡ing a C⊡MS.	Con ct cere ent performance tell eer 8,60 hor or 5 ear, which eer come fir t.	

¹ Efter to hate temontarate compliance for two contectific tetting to man refice the frettence of the term performance tetting to man and the terminal of and the contract of the terminal performance tetting the terminal termi

[78 FR 6711, Jan. 30, 2013]

TABLE 4 TO SUBPART ZZZZ OF PART 63—REQUIREMENTS FOR PERFORMANCE TESTS

As stated in §§ 63.6610, 63.6611, 63.6620, and 63.6640, you must comply with the following requirements for performance tests for stationary RICE:

Pt. 63, Subpt. ZZZZ, Table 4

For each	Comp ing with the re irrement to	Yo⊡ m⊡t	□ □ ing	□ccor ing to the fo owing re irement
1. 2SL□, 4SL□, an□ C□:tationar□ R [®] C□.	a. Re⊡ce CO emi⊞ion⊡	i. Select the Tam pling port Toca tion and the ntmer Tocation of trater Te point at the infet and oftet of the cont tro Te Toce Tan		a For CO, O ₂ , an moiltre mealler ment, ict ≤6 inche in iameter maile ampielat a ingè point io cate at the ict centroi an ict >6 an ≤12 inche in iameter maile cate at 16., 50.0, an 83.3 of the meallerement line 3 point long ine if the ict i >12 inche in i ameter and the imping port loca to meet the two an inafiliameter criterion of lection 11.1.1 of metho 1 of 40 CFR part 60, appenier ing ine interwise, conict the irratification telling an leact amp ping point according to fection 8.1.2 of metho.
		ii. Mea⊡re the O ₂ at the in et an o tiet of the con tro TeliceCan	□ Metho 3 or 3 or 3 of 40 CFR part 60, appen □ □ 2, or S M D6522 100 Re appro ⊡ 2005 □ 3 Meate pro ⊡ not nec e □ ar □	Im Mea⊡rement□ to Eetermine O ₂ m⊡t ⊡e ma⊡e at the Eame time a⊡ the mea⊡rement□ for CO concentra⊡ tion.
		iii. Mea⊡re the CO at the in et an the o_tet of the contro_eice an	12 S M D6522 00 Reappro_e 2005 23 10 Reat 10 re 10 of 0 CFR part 60, ap pen 14.4	□c⊡he CO concentration m⊡t ⊡e at 15 percent O ₂ , □r□□a□□
		i Mea re moi tre content at the in et an ot et of the contro effice a nee e to etermine CO an O ₂ con centration on a refer	[3] Metho □ 4 of 40 CFR part 60, ap pen □ □ □ 3, or metho □ 320 of 40 CFR part 63, appen □ □, or □ S □ M D6348 03 1 ³ .	□ Mea rement to retermine moi tre content mit re mare at the ame time an ocation a the mea rement for CO concentration.
2. 4SR⊡ tationar⊡ RïC⊡.	a. Re⊡ce form⊡ a⊡eh⊡e or ⊡HC emi⊡ion⊡	i. Select the lam pling port location tion and the nlmeriocation of trailerie point at the inlet an oritet of the cont troileicelan		a For forma ■eh ■e, □HC, O ₂ , an moiltre meatrement, □ctl ≤6 inchein in ameter male mappia at a ingle point locate at the □ct centroil an □ctl >6 anl ≤12 inchein in ameter male a mappia at 3 tratere point locate at 16. 50.0, an 83.30 of the meatrement ine location gine if the □ct i >12 inchein in ameter and the □am pling port location meet the two an haffiameter criterion of lection 11.1.1 of methol 1 of 40 CFR part 60, appenIi, the □ct and □ct angle at 30point long ine □ther wite, con □ct the tratification tell ing an □elect Impling point ac- coring to lection 8.1.2 of methol □ of 40 CFR part 60, appenIi.
Pt. 63, Subpt. ZZZZ, Table 4

40 CFR Ch. I (7-1-23 Edition)

For each	Compiling with the religious remembers to	Yo = m = t	□ ⊑ing	□ccor ing to the fo owing re irement
		ii. Mea⊡re O₂ at the in≊t an o ott iet of the contro iet ceEan	Metho 3 or 3 or 3 or 3 or 3 or 40 CFR part 60, appen	Mea⊡rement⊡ to ⊡etermine O ₂ concentration m⊡t ⊑e ma⊡e at the ⊡ame time a⊡ the mea⊡rement⊡ for forma⊡eh⊡e or ⊡HC concentration.
		iii. Mea⊡re moi t⊡re content at the in at an o t at of the contro e⊡ce a nee e⊡to _etermine forma leh —e or HC an O₂ con centration _ on a	2Metho 4 of 40 CFR part 60, ap pent 320 of 40 CFR part 63, appen 0, or SIM D6348 0313.	CcIMea⊡rement⊡ to _etermine moi⊡ t⊡re content m⊡t ⊡e ma⊡e at the ⊡ame time an⊡ocation a⊡the mea⊡ ⊡rement⊡ for forma⊡eh⊡e or ⊡HC concentration.
		i d ⊡emontrating compliance with the forma⊡ehine percent re⊡ction re⊡rement, mea⊡re form a⊡ehine at the inet an the ot et of the contro re⊡ce.	3 Metho 320 or 323 of 40 CFR part 63, appen 1 □ or S M D6348:03,13 pro 1 e in S M D6348:03 nne 5 a te Spiking echni e the percent R m t e greater than or e a to 0 an e than or e a to 130.	Forma □eh □e concentration m □t e at 15 percent O ₂ , □ □ □a □ Re □1□ of thi□ e□ con □1 of the a □er age of the three 1□ho□r or longer r□n□
		☐ f emon traing comp fance with the HC percent re incline re in re ment, mea⊡re HC at the in at an the ottet of the contro incline	4 1 Metho 25., reporte a pro pane, of 40 CFR part 60, appen	■ HC concentration m te at 15 percent Q ₂ , the all a Recent of this test consist of the allerage of the three 1shost or longer rins
3. Stationar RC	a. Limit the con centration of forma the the or CO in the that tionar RC et hat.	i. Selict the Tam pling port Toca tion and the nomer Tocation of trailer ce point at the echact of the Clationar RICC an	· ·····	a For formalehte, CO, O ₂ , an moittre meatrement, tct ≤6 inchel in lameter malle lampêd at a lngê point locatel at the tct centroil and tct >6 and ≤12 inchel in lameter malle lampêd at 3 tratere point locatel at 16. 50.0, ans 83.30 of the meatrement line 13 point locatel at 16. >12 inchel in lameter malle lampêd at 3 tratere point locatel at 16. >12 inchel in lameter and the lamping point location meet the two and hall lameter or terion of lection 11.1.1 of methol 1 of 40 CFR part 60, appendiel, the lampîed at 3 point long line lothere wite, cond the lampîng point location 8.1.2 of methol cording to lection 8.1.2 of methol fing an leect lampîng intere dit 6. appendiel t lampîng te motole at at tere and the lampîng te motole locate at the cording to lection 8.1.2 of methol fing an leectere at the cite of to the lampîng te motole locatel at tere cite of tere and the lampîng te motole cording to lection 8.1.2 of methol fille and tere at the cite of tere and the lampîng te motole cite at the cite of tere and the lampîng te motole cite at the cite of tere and the lampîng tere motole cite at the cite of tere and the lampîng tere motole cite at the cite of tere at the cite of tere and the cite of tere and the lampîng tere motole cite at the cite of tere and the lampîng tere motole cite at the cite of tere and the lampîng tere motole cite at the cite of tere and the lampîng tere motole cite at the cite of tere and tere at the cite of tere and tere and the cite of tere and tere cite of tere and tere and ter

Pt. 63, Subpt. ZZZZ, Table 5

For each	Comp≣ing with the re⊒irement to	Yo⊡ m⊡t	□ [ing	□ccor ing to the fo to wing re irement
		ii. Determine the O2 concentration of the Itationar RIC = Ina It at the Iamp îng port Tocation Ian	Metho 3 or 3 or 3 or 3 or 3 or 3 of 40 CFR part 60, appen 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Mea⊡rement to ⊡etermine O ₂ concentration m ⊡ e ma⊡e at the □ame time an ⊡cation a the mea⊡ □rement for forma □eh □e or CO concentration.
		iii. Mea⊡re moi⊡ t⊡re content of the ⊡ationar⊡ R:CC e ha t at the ⊡amping port ocation a nee: e to ⊡etermine forma⊡eh⊡e or CO an O₂ con centration⊡ on a	ear [2]_Metho_ 4 of 40 CFR part 60, ap pen □ □ □ 3, or metho_ 320 of 40 CFR part 63, appen □ □ , or □SIM D6348 0313.	□ciMeairement to etermine moi tre content mit e male at the Eame time an ⊡ocation all the meai rement for formalehie or CO concentration.
		C C C C C C C C C C C C C C C C C C C	3 Metho 320 or 323 of 40 CFR part 63, appen □ or S M D6348:03,13 pro ene pro ene in SIM D6348:03,13 pro ene pro ene in SIM D6348:03,13 pro ene pro ene in SIM D6348:03 ene in pro ene in ene fs ene percent R milling the preater than on on end the	Forma □eh □e concentration m □t e at 15 percent O ₂ , □r □ □ □ □ Re □10 of thi□ te□ con □1 of the a □r □ age of the three 1 □ho□r or longer r □n□
		☐ Mea⊡re CO at the e⊡ha⊡t of the ⊡ationar⊡ RĭC⊡.	ani be that 0 ei⊡tto 130. 4 Metho 10 of 40 CFR part 60, appeni 2 74, SIM D6522:00 2005 1 3 metho 320 of 40 CFR part 63, appeni 0 , or SIM D6348:03 1 3.	CO concentration m⊡t ce at 15 percent O ₂ , cr caid Re⊡tt of thi telt con⊡t of the alerage of the three 1 hor or longer r n.

 I Yo mail a loo te methol 3: an 10 a option to ISIM_D6522_00 (2005)

 2Yo mail o tain a copil of ISIM_D6348.03 from at least one of the following all reliet imerican Societ for leating an IMateriation 100 carr Harlor Drice, etc. Conthohocken, Pi 19428/2959, or Initer it Microfilm Internationa, 300 orth Ieee Roa, Ion nr Cror, M48106.

 3 incorporate
 reference, Iee IG3.14.

[88 FR 18413, Mar. 29, 2023]

TABLE 5 TO SUBPART ZZZZ OF PART 63—INITIAL COMPLIANCE WITH EMISSION LIMITATIONS, OPERATING LIMITATIONS, AND OTHER REQUIREMENTS

As stated in §§ 63.6612, 63.6625 and 63.6630, you must initially comply with the emission and operating limitations as required by the following:

For each	Comp Ting with the redirement to	Yo⊟ ha⊡e ⊑emon⊡trate⊟ initia⊒compî⊡ ance if
1. ew or recontrected nontemergenc 2SL ationar RtC >500 HP locate at a maio force of HCP, new or re- contrected nontemergenc 4SL fai- tionar RtC >250 HP locate at a maio force of HCP, nontemergenc tationar CCRC >500 HP locate at a maio force of HCP, an effing nontemergenc fationar CCRC >500 HP locate at an area force of HCP.	a. Re⊡ce CO emi⊡on⊡ an⊡ ⊡ng o⊡ ⊡ation cata⊡t, an⊡ ⊡ing a CPMS.	i. □he a □erage re □ction of emi⊡on of CO □etermine from the initia pero formance tel achie et the re □re CO percent re□ction □an ii. Yoo ha⊡e in ta⊡e a CPMS to con- tin o □ monitor cata in inet temp peratire accoring to the re□re mento in 063.6625□□an iii. Yoo ha⊡e recore the cata it pre □re □rop an cata □t inet tempera tre □ring the initia_performance te 1.

Pt. 63, Subpt. ZZZZ, Table 5

40 CFR Ch. I (7-1-23 Edition)

For each	Comp≣ing with the re⊡irement to	Yo ha e emon trate initia compî ance if
2. Contemergenc: Cationar C RC >500 HP Tocate: at a mater correct of H:P, and citting nontemergenc: Cationar CRC >500 HP Tocate: at an area toorce of H:P.	a. Limit the concentration of CO, ⊡ing o⊡ation cata⊡t, an⊡ ⊡ing a CPMS.	i. □he a erage CO concentration □eter□ mine□ from the initia performance te □ i□ ֎□ than or e □a to the CO emi □ □on fimitation □an □
3. [ew or recon]r[cte] non[emergenc] 25L [ationar] R.C] >500 HP [ocate] at a maio [o]rce of H_P, new or re] con]r[cte] non]emergenc] 45L [at]	a. Re⊡ce CO emi⊡on⊡ an⊡ not ⊡ing oïication cata⊡t.	 ii. Yoo hale in table a CPMS to continue the init and the call in the temperature according to the retire mentain 63.6625 that iii. Yoo hale recorde the catality pretative according to init temperature according the initial performance tell. i. The alerage retiction of emittion of CO tetermine from the initial performance tell achieves the retire CO percent retiction and
tionar□ R:C□ ≥250 HP @cate□ at a maör ⊡rce of H□P, nonimergenc□ Itationar□ C□R:C□ >500 HP @cate□ at a maör ⊡rce of H□P, an = eliting nonimergenc□ Itationar□ C□ R:C□ >500 HP @cate□ at an area ⊡rce of H□P.		ii. Yoo hale in table a CPMS to condition the contract of the
4. Onlemergenc Dationar C RC >500 HP locate at a malor locroe of H_P, an editing nonlemergenc Ita- tionar CRC >500 HP locate at an area locroe of H_P.	a. Limit the concentration of CO, an⊟ not ⊡ing o⊡⊡ation cata⊡t.	i. The alerage CO concentration leter mine: from the initia_performance telt i let than or ella to the CO emili ion finitation and ii. Yo hale inalite a CPMS to con- tin or monitor operating param eter approle the ominitrator if an according to the retirement in
5. Dew or recontracted nontemergence 2SL dationar RC >500 HP locate at a mator barce of HP, new or re- contracted nontemergence 4SL dationar tionar RC >250 HP locate at a mator barce of HP, nontemergence dationar CCRCC >500 HP locate at a mator barce of HDP, and editing nontemergence dationar CCRCC >500 HD acote dationar CCRCC	a. Re⊡ce CO emi⊡ion⊒ an⊡ ⊡ing a C⊡MS.	iii. Yo hate recorte the approte op erating parameter if an interprote op initia_performance tell. i. Yoo hate initiate a CoMS to contine of monitor CO an either O₂ or CO₂ at oth the initiat an ofitet of the ofication cataint according to the re direment in 63.6625/allan ii. Yoo hate conticte a performance efallation of for CCMS ding PS 3 an 4 of 40 CFR part 60, appendi
H⊡P.		iii. he alerage reliction of CO call clate ling 63.6620 elal or el ceel the relified percent reliction. he initial tel comprise the first 4- hor perior after liccerficalization of the CIMS. Compliance i alel on the alerage percent reliction
6. Onlemergenc Itationar C RC >500 HP locate at a maior force of HDP, and efficiency for the second	a. Limit the concentration of CO, an⊡ ⊡ing a C⊡MS.	achiele Inng the 41hor periol i. Yo hare in target a CMS to contin of monitor CO an either 02 or CO2 at the offet of the offet of the offet in according to the redirement in 63.6625/a man ii. Yo hare condicted a performance efaultion of for CMS ing PS 3 and 4 of 40 CFR part 60, appendi- iii. The areage concentration of CO call called ing 63.6620 if and the initiation. The initiated comprises the first 4 hor perior after coefficient and of the CMS. Complance is all on the areage concentration meature of the data perior

Pt. 63, Subpt. ZZZZ, Table 5

For each	Comp≣ing with the re⊞irement to	Yo ha e emon trate initia comp in ance if
□ oniemergenc□ 4SR□ ttationar□ RIC□ >500 HP iocate□ at a maior io⊡rce of H□P.	a. Re⊡ce forma⊡eh⊡e emi⊡on⊡ an⊡ ⊡ing □SCR.	 i. □he alerage reliction of emilion of formallehile letermine from the initial performance tell i ella to or greater than the relice formallehile hie percent reliction, or the alerage reliction of emilion of HC elletermine from the initial performance tell i ella to or greater than 30 periorentano ii. Yoo hale initale a CPMS to continuo monitor catallit initiat temiperative according to the relifered mention 63.6625 man iii. Yoo hale recorde the catallit prelimeter of the catallity prelimeter of the catallit
8. Oon@mergencO 4SRO tationarO RCO >500 HP locateO at a malor loonce of HOP.	a. Re⊡ce forma⊡eh⊡e emi⊡on⊡ an⊡ not ⊡ing ⊡SCR.	tire cling the initial performance tell. i. The alerage reliction of emilion of formallehile cletermine from the ini- tial performance tell is cline at to or greater than the relifier formalle hile percent reliction or the alerage reliction of emilion of -HC cleter mine from the initial performance tell is cline at the action of greater than 30 percent and i. Yoo hale initiale a CPMS to con
		tin o monitor operating parameter appro c the comini trator if an accoring to the remaini trator in 63.6625 in an accoring the appro c operating parameter if an c ing the initia parameter to t
9. Dew or recontracted nontemergence trationart RIC >500 HP locate at a mator to the of HDP, new or recont tracted nontemergence 4SL tatter tionart RIC 250 <hp<500 a<br="" at="" locate="">mator to the HDP, and eliting non- emergence 4SR tationart RIC >500 HP locate at a mator to the of HDP.</hp<500>	a. Limit the concentration of forma e here in the tationar RC echa at an cling of ation cata to cSCR.	i. The alerage formallehtle concentral tion, correctel to 15 percent O ₂ , r ⁻ all from the three tel r∩n i en than or ella to the formallehtle emilion imitationan ii. Yol hale intale a CPMS to con tincoll monitor catall init tem peratire according to the retire
10. ew or recontracte nontemergenc tationar RC >500 HP focate at a mator force of HP, new or recontracted tracte nontemergenc 4SL tationar tionar RC 2504HP500 focate at a mator force of HP, an etiting non- emergenc 4SR tationar RC 5500 HP focate at a mator force of HP.	a. Limit the concentration of forma⊡e h⊡e in the ⊡ationar⊡ RiC⊡ e⊡ha⊡t an⊡ not ⊡ing o⊡ation cata⊡it or ⊡SCR.	iii. Yo hale recordent the catalination in the initial performance tell. i. The all erecordent initial performance tell. i. The all erage formallehille concentration, corrected to 15 percent O2, or all, from the three tell rholie initial performance tell. ii. Yoo hale initial a CPMS to condition initiation and iii. Yoo hale initial a CPMS to condition of the all erage of the all performance of the all pe
11. Difting nonTemergenc Itationar RTC 1005HPS500 Tocate at a maTor ToTree of HTP, and edifting nonTemer genc Itationar CRTC 300 <hps500 Tocate at an area ToTree of HTP.</hps500 	a. Re⊡ce CO emi⊡ion⊡	 i. The resolute in application opposed opposed opposed in the initial performance tell. i. The allerage relation of emillion of CO or formallehile, all application end termine from the initial performance tell in equato or greater than the relation or formallehile, all application opposed in the initial performance tell in encoder the opposed opposed in the initial performance tell in encoder the opposed oppo
12. Iding nonTemergenc Idationar RC 1005HP5500 Tocate at a mator Torce of HCP, an editing nonTemer genc Idationar CRC 3005HP5500 Tocate at an area Torce of HCP.	a. Limit the concentration of forma⊡e h⊡e or CO in the ⊡ationar⊡ RiC⊡ e⊡ ha⊡t.	i. The alreage forma the corrected to centration, all applicate, corrected to 15 percent 0 ₂ , the that or ellato there test rendie than or ellato the formattentie or CO emilion limit tation, all applicate.

Pt. 63, Subpt. ZZZZ, Table 6

40 CFR Ch. I (7-1-23 Edition)

For each	Comp≣ing with the re⊡irement to	Yo⊟ ha⊑e ⊑emon⊡trate⊟ initia⊡compi⊡ ance if
 13. Clifting nonTemergenc: 4SL: ta: tionar: R:C: >500 HP Tocate: at an area To:Toc of H:P that are not remote Dationar: R:C: and that are operate: more than 24 ho:T: per caTentar Tear. 14. Clifting nonTemergenc: 4SR: ta: tionar: R:C: >500 HP Totat: at an 	a. Initallian offication catallit	i. Yo: ha⊡e con tet= an initia compI ance emontration a pecifie in B3.6630 et to how that the alerage retrained of emition of CO 93 percent or more, or the alerage CO concentration is et that an or eta to d ppmi at 15 percent 0 ₂ . ii. Yo: hale in ta⊡e a CPMS to con tinotte accoring to the retrier ment(in B3.6625t) or to hale in Date: ettipment to alomaticat that compute if the catatt in at temperattre etceet 1350 °F. i. Yo: hale on tete an initiaCcompI
area ⊡Crce of H⊡P that are not remote ⊡ationar⊡ R:C⊡ an⊡ that are operate more than 24 ho⊡r⊡ per caïen⊡ar ⊡ear.		ance entormation appendent apernent entormation appendent approximation appendent approximation appendent approximation approximati approxima

[78 FR 6712, Jan. 30, 2013]

TABLE 6 TO SUBPART ZZZZ OF PART 63—CONTINUOUS COMPLIANCE WITH EMISSION LIMITATIONS, AND OTHER REQUIREMENTS

As stated in 63.6640, you must continuously comply with the emissions and operating limitations and work or management practices as required by the following:

For each	Comp \blacksquare ing with the re \Box irement to	Yo m t emon trate contin co com plance
1. ew or recontracted nontemergence 2SL tationar RtC >500 HP locate at a maior Ence of HD, new or re- contracted nontemergence 4SL taa tionar RtC >250 HP locate at a maior Ence of HD, an new or re- contracted nontemergence C taa tionar RtC >500 HP locate at a maior corce of HDP.	a. Re⊡ce CO emi⊡on⊡ an⊡ ⊡ing an o⊡ation cata⊡t, an⊡ ⊡ing a CPMS.	 i. Condicting demianned performance tells for CO to demondrate that the redired CO percent rediction is achied and the catally indet temperative for the catally indet temperative for the catally the formation of the catally indet temperative for the catally indet the performance tell.
2. ew or recontricte nontemergence 2SL dationar RCC >500 HP locate at a maior loice of HLP, new or re- contricte nontemergenc 4SL ta- tionar RCC >250 HP locate at a maior loice of HLP, an new or re- contricte nontemergenc C ta- tionar RCC >500 HP locate at a maior loice of HLP.	a. Re⊡ce CO emi⊡on⊡ an ⊡not ⊡ing an o⊡ation cata⊡t, an ⊡ing a CPMS.	 i. Conlicting lemiannial performance tell for CO to lemonIrate that the relifie CO percent reliction is achiele alani ii. Collecting the approle operating palarameter if an lata according to 63.6625 manilii. Relicting the lata to 4 thoir rolling alerage an

Pt. 63, Subpt. ZZZZ, Table 6

For each	Comp≣ing with the re⊡irement to	Yo m t emon trate contin co com plance
3. □ew or recon rcte non emergenc 2SL tationar RC >500 HP coate at a maior corce of H_P, new or recon contricte non emergenc 4SL ta tionar RC ≥250 HP coate at a maior corce of H_P, new or recon tricte non emergenc tationar C RC >500 HP coate at a maior corce of H_P, an elling non emergenc genc tationar C RC >500 HP.	a. Rece CO emiion □ or _imit the concentration of CO in thetationar R IC = e hat, aning a C _MS.	 Maintaining the 4.hor rolling alerage within the operating limitation for the operating parameter ella limbe criming the performance tell. Collecting the monitoring lata accoring to 63.6625 an relicing the mealmement to 1.hor alerage callading the percent reliciton or concentration of CO emillion accoring to 63.6620 and limit achiele the relime percent relicit to aleraging period or that the emillion remain at or lew whe CO concentral
4. □onTemergenc□ 4SR□ ttationar□ RIC□ >500 HP locate□ at a mator to□rce of H□P.	a. Reince formainehine eminioni and thing SCR.	tion imitCan iii. ConCting PS 3 an 4 of 40 CFR part 60, appentie , a wellar aim an period tata faith check in ac- corCance with 40 CFR part 60, appen- ifF, proceffre 1. i. Collecting the catalist in at tempera- tire fata according to 63.6625 and ii. Reficting the fata to 4 thort rolling
		 aerage and the thole of the second second
5. □oni≊mergenc□ 4SR□ tiationar□ RtC□ >500 HP iocate□ at a maior io⊡rce of H⊡P.	a. Re⊡ce forma⊡eh⊡e emi⊡ion⊡ an⊡ not ⊡ing ⊒SCR.	i. Collecting the approle⊡ operating pa⊡ rameter fif an □ tata according to action according to action action according to action action action action action iii. Maintaining the 4thor rolling action age within the operating fimitation for the operating parameter entation if the mind the performance tent.
6. □oniemergenc□ 4SR□ ttationar□ RIC□ with a trake HP ≥5,000 locate□ at a malor totrce of H□P.	a. Re iice forma iieh iie emi iion ii	Con cting emianna performance tell for formallehie to emonitrate that the remire formallehie per- cent reiction i achiele, or to emi onitrate that the alerage reliction of emillion of HC tetermine from the performance tell i emails or greater than 30 percent a
□ □ew or recon⊡r⊏cte□ non⊡emergenc□ □tationar□ RIC□ >500 HP locate□ at a ma or ⊡rec of H□P an new or recon□ □tricte□ non⊡emergenc□ 4SL□ tai tionar□ RIC□ 250≤HP≤500 locate□ at a ma or ⊡o⊡rce of H□P.	a. Limit the concentration of forma <u>■</u> e h <u></u> e in the <u>ationar</u> R <u>C</u> e <u>ha</u> <u>t</u> an <u></u> <u>n</u> g o <u></u> ation cata <u></u> or SCR.	 i. Contains to performance i. Contains to perform ance ii. Contains the cata and the performance iii. Contains the cata and the performance iii. Retricting the cata to the performance iii. Retricting the tamperative and iiii. Retricting the performance to perform and the performance iiii. Retricting the performance and the performance iiii. Retricting the tamperative and iiii. Retricting the performance and the performance of the cata and the performance and th
		ating imitation entailine entry the period ating imitation entailine entailine entailine entailine entry the performance tell.

Pt. 63, Subpt. ZZZZ, Table 6

40 CFR Ch. I (7-1-23 Edition)

For each	Comp≣ing with the re⊡irement to	Yo m t cemon trate contin co com plance
8. Dew or recondricted non@mergence Dationard R(C) >500 HP Docated at a maior bolice of HIP and new or recond Dricted non@mergence 4SL tad tionard R(C) 250≤HP≤500 Docated at a maior bolice of HIP.	a. Limit the concentration of forma h te in the tationar RtC = that an not ting offation cata to SCR.	i. Conficting remiannial performance tell for formallehile to remonitate that for mailehile to remonitate that for mailehile concentration finitalian ii. Collecting the approle operating pa rameter if and tata according to 63.66250000 iii. Reficing the leata to 4 thor rolling alerage fian i Maintaining the 4 thor rolling aler age within the operating fimitation for the operating parameter end.
9. ☐ ting emergenc an ☐ ack tart ta tonar R:C ≤500 HP ocate at a maior borce of HP, effing non emergenc 1ationar R:C <100 HP locate at a maior borce of HP, effin ing emergenc an 1ack tart ta tionar R:C locate at an area force of HP, effing nontemergenc 1a tionar C:RC ≤300 HP ocate at an area force of HP, effing nontemer genc 2SL 1ationar R:C locate at an area force of HP, effing non emergenc 1ationar S:RC locate at an area force of HP, which com in antificor ige ga efficient to 10 percent or more of the grow hat inp1 on a nanal efficient at an area loce of HP, effing nontemergenc 4SL an 4SR 1ationar R:C >500 HP locate at an area loce of HP, effing nontemergenc 4SL an 4SR 1ationar R:C >500 HP locate at an area loce of HP, effing nontemergenc 4SL an 4SR 1ationar R:C >500 HP locate at an area force of HP	a. □ ork or Management practice□	i. Operating and maintaining the liah i. Operating and maintaining the liah factirer according to the mand factirer for the maintenance in Deletop and follow for own mainten nance plan which mod profile to the effect practicate for the maintenance and operation of the engine in a mand ner conditient with good air polition controd practice for minimiding emidi
H⊡P that are remote ⊡ationar ⊂ RtC⊡. 10Cting ⊡ationar ⊂ C=RtC >500 HP that are not îmite⊡ ⊡e ⊡ationar ⊂ RtC⊡.	a. Re⊡ce CO emi⊡on or imit the concentration of CO in the ⊡ationar RIC elha⊡t, an ⊡ing ollation cata⊞t.	 i. Conflicting performance tell elerities, 60 hold or 3 fear, whichefer come first, for CO or formallehte, all appropriate, to femonitate that the retire CO or formallehte, all appropriate, percent retiction is achielee or that for emittion remain at or elow the CO or formallehte, all appropriate that the retire concentration limitant in the temperative for the concentration limitant in the temperative fat according to 63.6625111 and limitation the operating limitational for the catallities the presented and the presented and the presented and the concentration in the concentration for the catallities and the presented and the presented and the concentration in the presented and the concentrating the theory of the catallities and the presented and the concentration and the presented and the concentration and the concentration and the concentration and the presented and the presented and the concentration and the concentration and the concentration and the presented and the concentration and the concentration and the concentration and the presented and the concentration and t
11. ⊡⊡ting tationar⊡ C⊡RtC⊡ >500 HP that are not timite⊡ tationar⊡ RtC⊡.	a. Re⊡ce CO emi⊡ion or timit the concentration of CO in the ⊡ationar RIC enhant, an not ⊡ing of ation cata⊞1.	 performance te1. i. Con cting performance te1: eer a, E0 hor or 3 ear whichefer come fir1, for CO or formaleher, appropriate, to emon tate that the relifier CO or formaleher, al appropriate, percent reliction i achiefer or that for emilian reliminant

Pt. 63, Subpt. ZZZZ, Table 6

For each	Comp≣ing with the re⊡irement to	Yo m t cemon trate contin co com pîance
12. ⊡iling îmite⊡ ⊡e C⊡ itationar⊡ RîC□ >500 HP.	a. Re⊡ce CO emi⊡on⊡ or imit the concentration of CO in the ⊡ationar⊡ RIC□ e⊡na⊡t, an⊡ ⊡ing an o⊡ation cata⊡t.	 ii. Collecting the approle operating parameter if an ata accoring to 63.6625an iii. Relicing the le tata to 4 thoir roling alerage and iii. Maintaining the 4 thoir roling alerage within the operating limitation for the operating parameter ellation for the operating performance tell. ii. Conlicting performance telle elerent 8,60 hoir or 5 lear whicheler come firth, for CO or formallether, all appropriate, percent reliction i achiele or that loor emilion remain at or leave the CO or formallether. ii. Collecting the catal for formallether. iii. Collecting the catal in at temperation according to the concentration in the and the relifier according to 63.6625
13. □⊡ing īmite⊡ ⊡e C⊡ tationar⊡ RīC□ >500 HP.	a. Re⊡ce CO emi⊡on⊡ or imit the concentration of CO in the ⊡ationar⊡ RtC□ e⊡a⊡t, an⊡ not ⊡ing an o⊡ ⊡ation cata⊡t.	 and iii. Relicing the le tata to 4 thoir rolling alterage iii. Maintaining the 4 thoir rolling alterage iiii the operating limitation for the cataling the temperate lean Mealing the prelime loop acroin the cataling the temperatere and Mealing the prelime loop acroin the cataling the temperatere and on trating that the prelime loop acroin the cataling limitation etails in the temperative and the performance tet. i. Conliciting performance tet elerer 8,60 hoir or 5 lear whicheler come limit, for CO or formallehile, all appropriate, to remonitrate that the relime CO or formallehile, all appropriate, percent relicition is achiele or that loor emilion remain at or leow the CO or formalle hild concentration limitan ii. Collecting the approfe operating palarameter if and lata accoring to 63.6625
14. □□1ing nonTemergenc□ 4SL□ Ital tionar□ R.C□ >500 HP Tocate□ at an area To⊡ce of H□P that are not remote thationar□ R.C□ an□ that are operate□ more than 24 hor□per caten□ar Tear.	a. In⊡a⊡an o⊡⊡ation cata⊡it	 i Maintaining the 4 thon rolling aller age within the operating limitation for the operating parameter eltan intel complexity parameter eltan intel complexity annual complexity end on tration an pecifie in easily end to be all the allerage remeters of emillion of CO in 93 percent or more, or the allerage CO concentration in all then or eltantio 4 ppm at 15 percent O₂ an either ii. Collecting the cataling in et temperan there that according to elsafe52 remeters and maintaining the 4 thon rolling allerage within the limitation of greater than 450 °F an ent than or eltanto 1350 °F for the cataling in et temperatheror iii. Immediate in thing cown the engine if the cataling in the temperature ell ceeming 1550 °F

Pt. 63, Subpt. ZZZZ, Table 7

40 CFR Ch. I (7-1-23 Edition)

For each	Comp \blacksquare ing with the re \Box irement to	Yo m t emon trate contin co com pliance
15. ☐☐ing non@mergenc 4SR 1a tionar RC >500 HP locate at an area lolice of H_P that are not remote ⊡ationar RC an that are operate more than 24 hor per calen_ar lear.	a. Indamoscr	i. Concting ann a compliance emi onration apecifie in 63.6640;c_to how that the a erage rection of emiion of CO is percent or more, the a erage CO con centration i e than or e a to 2:0 ppm at 15 percent O ₂ , or the a er age rection of emiion of HC i 30 percent or more an either ii. Collecting the cata init temperal the tat a ccorfing to63.6625 recting the te ata to 4 thor rolling a erage an maintaining the 4 hor rolling a erage within the limitation of greater than or e a to 1250 °F an e that or e to 1250 °F an e that or e to 1250 °F or the iii. Immeiate htting own the engine if the cata init temperat_re cee if the cata init temperat_re cee

a fter to hate temontrate compliance for two contectife tell, to matter the trainer of the temperature of the term of term

[78 FR 6715, Jan. 30, 2013]

TABLE 7 TO SUBPART ZZZZ OF PART 63-REQUIREMENTS FOR REPORTS

As stated in 63.6650, you must comply with the following requirements for reports:

For each	Yo□ m□□t □□□mit a	□he report m□□t contain	Yo m t the report
I. □iting non@mergenc□, non□iack □art □ationar□ RiC□ 1005HP2500 iocate□ at a maior ⊡orce of H□P□ e□ting non@mergenc□, non□iack □art □ationar□CC RiC□ 5500 HP ocate□ at maior ⊡orce of H□P□e□1□ ing non@mergenc□ 4SR□ tationar□ RiC□ >500 HP ioc cate□ at a maior ⊡orce of H□P□e□1ing non@mer□ genc□, non□iack □art □at tionar□ C□RiC□ >300 HP io cate□ at an area ⊡orce of H□P□new or recon□tricte□ non@mergenc□ lationar□ RiC□ >500 HP iocate□ at a maior ⊡orce of H□P□an□ new or recon□tricte□ non emergenc□ 4SL□ lationar□ RiC□ 2505HP2500 iocate□ at a maior ⊡orce of H□P.	Comp îance report	a. If there are no cellation from an emilion limitation tion or operating limitation tion that applie to Co a tatement that there were no cellation from the emilion limitation or op- erating limitatio	i. Semiann a accor ing to the re irement in 03.6650 if its for en- gine that are not imite tation and its in the imite it. Inn a accor ing to the re irement in 03.6650 if its for en- gine that are imite is tationar RC if its to nomerica emiiton imita
		If ioc hall a left attorn from an emilion imitation or operating imitation cring the reporting periol, the information in 63.6650 If there were periol	 i. Semiann a □ accor ing to the re_irement in 63.6650 □ □ i. Semiann a □ accor ing to the re_irement in 63.6650 □ □

Pt. 63, Subpt. ZZZZ, Table 8

For each	Yo⊡ m⊡tt ⊡⊡mit a	□he report m□⊒t contain	Yo m t m the report
2. ew or recontrictel non emergenc: ationar RIC that commit an filing and ige ing a eminant to 10 percent or more of the grom heat inpit on an annia	Report	a. the flee flow rate of each flee and the heating larget that were the in the call clation and the method clation and the method clation and the method clation and the method clatic and fligged or the geter ga, i.e. the area to 10 percent or more of the growtheat inpl to an and national and the operating finit pro- flee in floer fleeratteen forceale permit, and and the fleeratteen forceale permit and and c. In profilem or error	 i. □nn□a□□, accor⊡ing to the re□⊡irement□ in □63.6650. i. See item 2.a.i. i. See item 2.a.i.
3ing nontemergenc_ nontack tart 4SL an 4SR tationar R:C >500 HP tocate at an area to cree of HP that are not remote tationar R:C an that operate more than 24	Compliance report	a. □hectel with the meter a. □he re □to of the ann □a compliance □emon Irration, if con □cte □ □ring the re □ porting perio □	i. Semiann⊡a i accor ing to the re⊡rement in 63.6650 III II50
4mergenc_ tationar_R:C_ that operate for the p_r_ po_e_ pecifie_ in _63.6640.ft_4_tit_	Report	a. ⊡he information in ⊡63.6650th⊡1⊡	i. ann a⊞ accor ing to the re⊡irement in ⊡63.6650 h 12 113 .

[87 FR 48608, Aug. 10, 2022]

TABLE 8 TO SUBPART ZZZZ OF PART 63—APPLICABILITY OF GENERAL PROVISIONS TO SUBPART ZZZZ.

As stated in 63.6665, you must comply with the following applicable general provisions.

Genera□pro⊡⊡on⊡citation	S === ect of citation	□ppîe□ to □□□ part	□□pānation
□63.1	Genera□appīca⊡iīt□ of the Genera□ Pro⊡⊡on□	Ye□	
□63.2	Definition	Ye 🗆	□□□itiona term □ □efine □ in □63.66 □5.
□63.3	□nit□an□a□re⊡ation□	Ye□	
□63.4	Prohi_ite acti_itie an circ_m_en tion	Ye□	
□63.5	Con tr ction an recon tr ction	Ye	
		Ye	
63 6 1 1 4	Compliance rate for new an recon	Ye	
63 6 5 5		Ye	
63 6 116	Referen		
63.6	Compliance rate for new an recon	Ye⊓	
	Tr ctell area for rcell that lecome		
□63.6 c □1 □ 2 □	Compliance Tate for efilting	Ye⊓	
63.6 c 3 4			
63.6 c 5	Compliance □ate□ for e□i□ting area	Ye⊓	
	orcenthat recome malor forcen		
□63.6□□□	Refere		
63.6 e	Operation an maintenance	□0.	
□63.6 f 1 □	□ppica⊡it□of □tan□ar□□	□ 0.	
□63.6 f □ 2 □	Metho II for I etermining compliance	Ye□	
□63.6 f 3 □	Finding of compliance	Ye□	
63.6 g 1 3	□ e of atternate tan ar	Ye□	
□63.6ҧ¯□	Opacit□ an□ ū́⊡ē emi⊡ion Itan⊡	□o	S part contain opac
	ar		it□or ⊡⊡⊡e emi⊡ion ⊡tan⊡ar⊡.
□63.6 🗊	Compliance entention procentre	Ye□	
	an 🗆 criteria.		
□63.6 Ⅲ	Pre i entia compiance e emption	Ye□	

Pt. 63, Subpt. ZZZZ, Table 8

40 CFR Ch. I (7-1-23 Edition)

Genera□pro⊡⊡on□citation	S === ect of citation	□ppîie□ to □□□□ part	□□panation
63.021120	Performance te t ate	Ye	S part accontain performance tet ate at 63.6610, 63.6611, an 63.6612.
63. a 3	C C cection 114 a thorit	Ye□	
63.00010	□otification of performance te It	Ye	□ cept that □63.□□□1 □ on □ appîe □ a□ cecifie □ in □63.6645.
63. 22	□otification of re⊡che □□îng	Ye 🗆	□ cept that □63.□□□12□ on □ appîe□
□63. □C □	□ laît la III rance te lt plan	Ye 🗆	□ cept that □63. □c□ on □ apple □ a □ □ cept in □63.6645
□63.□□□	□e⊡ting faciûtie□	Ye□	
063. De 11 0	Con ition for con cting perform ance te c	0.	S part conception for concepting performance tend at 63.6620.
63. e 2	Con ct of performance te an re	Ye 🗆	S part conception tent method at conception conception of the second sec
63. e 3	□e t r □n □□ration	Ye□	
63. e 4	□_mini_trator ma□ reire other te to ing _n_er _ection 114 of the C □	Ye□	
□63. □f □	□ternatice tect metho procicion	Ye□	
□63.□g□	Performance te t □ata ana □□□□, rec □ or □keeping, an □ reporting.	Ye□	
□63.□ħ□	□ ai⊡er of te⊡t □	Ye□	
□63.8 🖬 🔲	□ppitca⊡itt of monitoring re⊡ire□ ment□	Ye	S part contain pecific re frement for monitoring at 63.6625.
	Performance pecification	Ye□	
63.8 a 4	Monitoring for contro De Ce	□ 0.	
□63.8 □□1 □	Monitoring	Ye□	
63.8 2 3	M⊡tiple eff⊡ent⊡ an⊡ m⊡tiple moni⊡ toring ⊡_tem⊡	Ye□	
□63.8℃□1□	Monitoring <u>t</u> tem operation an maintenance.	Ye□	
□63.8 c 1 1	Ro⊡tine an□ pre⊡cta⊡e SSM	□o .	
□63.8 c □1 □ĭi □	SSM not in Start⊡p Sh⊡t⊡own Ma⊡ f⊡nction P⊡n.	Ye□	
063.8 c 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Comptiance with operation an□ main□ tenance re⊡irement□	□ 0.	
63.8 c 2 3	Monitoring inter in ta ation	Ye□	
□63.8℃□4□	Contin⊡o	Ye::	□ cept that □ part □ □ □ □ □ □ □ □ oe □ not re □ ire Contin □ □ □ Opacit□ Moni □ toring S □ tem □COMS □
□63.8℃ ¹⁵ □	COMS minim m proce re	□o	S⊡part □□□□ ⊡oe□ not re⊡ire COMS.
□63.8℃ □ 6□□8□	CMS re irement	Ye 🗆	□ cept that □ part □ □ □ ce□ not re □ re COMS.
□63.8 □□	CMS □aít□ contro □	Ye□	
□63.8 e	CMS performance e a ation	Ye 🗆	□ cept for □63.8 c □5 □ii □, which ap □ pie □ to COMS.
		□ cept that □63.8 e □	
		on III app îe I a II ⊡pecifie II	
		in □63.6645.	
63.8 f 1 5	□ternatice monitoring metho	Ye 🗆	□ cept that □63.8 f 4 on 1 app ie □ a □ pecifie in □63.6645.
□63.8 f □6 □	□ternatile to relatile acclracitett	Ye	□ cept that □63.8 f □6□ on □ app ie □ a □ pecifie □ in □63.6645.
□63.8 g □	Data re ⊡ction	Ye 🗆	□ cept that pro i i on □ for COMS are
			not applica e. eraging perio for emontrating compliance are pecifie at 63.6635 an 63.6640.
□63.9ª□	□ppica□it an State eegation of notification re	Ye□	
63.9 11 15	nitia □notification □	Ye	□ cept that □63.9 □ 3 □i □ re er e .
		□□c¢pt that	
		63.9	
		on app Ie 🗌	
		in □63.6645.	

Pt. 63, Subpt. ZZZZ, Table 8

Genera⊡pro⊡⊡on⊡ citation	Sect of citation	□ppîe□to □□□ part	□□panation
□63.9℃□	Relient for compliance entention	Ye:	□ cept that □63.9 c □ on □ appîe □ a □
□63.9Ⅲ□	□otification of □pecia□compilance re□ □□irement□ for new □o□rce□	Ye:	□ cept that □63.9 □ on □ app ie □ a □ □ cept in □63.6645.
□63.9℃	□otification of performance te t	Ye 🗆	□ cept that □63.9 c on □ appie a□ □ pecifie in □63.6645.
□63.9ff□	□otification of □□□e emi□ion Ⅳ□□ opacit□te□t.	□o	S part COL Coe not contain opac it or V tan ar
□63.9 g 🖽 🗆	□otification of performance e⊡a⊞ation	Ye	□ cept that □63.9 g □ on □ app te □ a □ □ pecifie □ in □63.6645.
□63.9 g □2 □	□otification of □ e of COMS □ata	□o	S part contain opac it or V tan ar
□63.9 'g □3□	□otification that criterion for after□ natice to R□□□ i□ e cee e□	Ye□	tîatîernati⊑e i⊡ in ⊡e.
063.91h111160	□otification of compliance ⊡at⊡	☐ cept that G3.9īg on □ appīe a □ pecifie in G3.6645. Ye □	Cept that notification for corree ing a C MS are corrected and a corrected and a corrected and corre
□63.9 i□	Change in production	Ye⊡	
□63.9 k□ □63.10 a□	Change in pre_or information Ectronic reporting proce re minitratile prolition for recor	Ye□ Ye□ Ye□	On⊡a⊡ pecifie ⊡in ⊡63.9 ⊡⊡
63.10	keeping⊡eporting. Recor⊡ retention	Ye:]	□ Ccept that the mo t recent 2 ⊡ear □ of □ata □o not ha⊡e to ⊡e retaine □ on □te.
63.10 2	Recor relate to SSM	□0.	
63 10 2 1		Ye∟ Vo⊡	
63.10 2 1ii	Recor⊡ when ⊡ing atternati⊡e to R□□.	Ye⊒	For CO ⊡tan⊡ar⊡ if ⊡ing R⊡⊡ after⊡ nati⊑e.
63.10 2 1	Recor of pporting oc mentation	Ye□	
□63.10 □□3 □	Recor II of app îca I ît I I etermination	Ye□	
□63.10℃□	CIMS.	Ye□	are referee.
		Yell	
63.10 3	Report of performance tell reliation.	Ye⊥ ⊡o	S part 000 oe not contain opac
63.10 4	Progre report	Ye□	
63.10 5	Start_p, ⊡h⊡t⊡own, an⊡ mat⊡nction report⊡	□0.	
63.10 e 1 an 2 1	COMS Ire ate □ report □	Ye□ □o	Soopart ooo ooo not reoor
□63.10 e □3 □	□ ce	Ye□	Cons. Ccept that C3.10 C i re cerce
□63.10 @ □ 4 □	Reporting COMS ata	□o	Sopart con coe not recire COMS.
□63.1011□	□ ai er for recor keeping proting	Ye□	
□63.11	Fare	□0.	
□63.12	State a⊡thorit an ceegation	Ye□	
□63.13		Ye□	
□63.14	Incorporation III reference	Ye□	
□63.15	□ aia iit of information	Ye□	

[75 FR 9688, Mar. 3, 2010, as amended at 78 FR 6720, Jan. 30, 2013; 85 FR 73912, Nov. 19, 2020]

Pt. 63, Subpt. ZZZZ, App. A

APPENDIX A TO SUBPART ZZZZ OF PART 63—PROTOCOL FOR USING AN ELEC-TROCHEMICAL ANALYZER TO DETER-MINE OXYGEN AND CARBON MON-OXIDE CONCENTRATIONS FROM CER-TAIN ENGINES

1.0 Scope and Application. What is this $$\operatorname{Protocol}$$

This protocol is a procedure for using portable electrochemical (EC) cells for meas-

40 CFR Ch. I (7-1-23 Edition)

uring carbon monoxide (CO) and oxygen (O_2) concentrations in controlled and uncontrolled emissions from existing stationary 4stroke lean burn and 4-stroke rich burn reciprocating internal combustion engines as specified in the applicable rule.

1.1 Analytes. What does this protocol determine?

This protocol measures the engine exhaust gas concentrations of carbon monoxide (CO) and oxygen (O_2) .

□na⊡te	C□S □o.	Sen⊡ti⊡t⊡
Car⊡on mono⊡⊡e ICO⊡	630=08=0	Minim _m _etecta _e fimit _ho e 2 percent of the nomina _range or 1 ppm, whiche _er i _ e re _tric _ ti _e.
$O \underline{\ } Gen \ \underline{\mathbb{O}}_2 \underline{\ }$		

1.2 Applicability. When is this protocol acceptable?

This protocol is applicable to 40 CFR part 63, subpart ZZZZ. Because of inherent cross sensitivities of EC cells, you must not apply this protocol to other emissions sources without specific instruction to that effect.

1.3 Data Quality Objectives. How good must my collected data be?

Refer to Section 13 to verify and document acceptable analyzer performance.

1.4 Range. What is the targeted analytical range for this protocol?

The measurement system and EC cell design(s) conforming to this protocol will determine the analytical range for each gas component. The nominal ranges are defined by choosing up-scale calibration gas concentrations near the maximum anticipated flue gas concentrations for CO and O_2 , or no more than twice the permitted CO level.

1.5 Sensitivity. What minimum detectable limit will this protocol yield for a particular gas component?

The minimum detectable limit depends on the nominal range and resolution of the specific EC cell used, and the signal to noise ratio of the measurement system. The minimum detectable limit should be 2 percent of the nominal range or 1 ppm, whichever is less restrictive.

2.0 SUMMARY OF PROTOCOL

In this protocol, a gas sample is extracted from an engine exhaust system and then conveyed to a portable EC analyzer for measurement of CO and O_2 gas concentrations. This method provides measurement system performance specifications and sampling protocols to ensure reliable data. You may use additions to, or modifications of vendor supplied measurement systems (e.g., heated or unheated sample lines, thermocouples, flow meters, selective gas scrubbers, etc.) to meet the design specifications of this protocol. Do not make changes to the measurement system from the as-verified configuration (Section 3.12).

3.0 DEFINITIONS

3.1 Measurement System. The total equipment required for the measurement of CO and O_2 concentrations. The measurement system consists of the following major subsystems:

3.1.1 Data Recorder. A strip chart recorder, computer or digital recorder for logging measurement data from the analyzer output. You may record measurement data from the digital data display manually or electronically.

3.1.2 Electrochemical (EC) Cell. A device, similar to a fuel cell, used to sense the presence of a specific analyte and generate an electrical current output proportional to the analyte concentration.

3.1.3 Interference Gas Scrubber. A device used to remove or neutralize chemical compounds that may interfere with the selective operation of an EC cell.

3.1.4 Moisture Removal System. Any device used to reduce the concentration of moisture in the sample stream so as to protect the EC cells from the damaging effects of condensation and to minimize errors in measurements caused by the scrubbing of soluble gases.

3.1.5 Sample Interface. The portion of the system used for one or more of the following: sample acquisition; sample transport; sample conditioning or protection of the EC cell from any degrading effects of the engine exhaust effluent; removal of particulate matter and condensed moisture.

3.2 Nominal Range. The range of analyte concentrations over which each EC cell is operated (normally 25 percent to 150 percent of up-scale calibration gas value). Several

nominal ranges can be used for any given cell so long as the calibration and repeatability checks for that range remain within specifications.

3.3 Calibration Gas. A vendor certified concentration of a specific analyte in an appropriate balance gas.

3.4 Zero Calibration Error. The analyte concentration output exhibited by the EC cell in response to zero-level calibration gas.

3.5 Up-Scale Calibration Error. The mean of the difference between the analyte concentration exhibited by the EC cell and the certified concentration of the up-scale calibration gas.

3.6 Interference Check. A procedure for quantifying analytical interference from components in the engine exhaust gas other than the targeted analytes.

3.7 Repeatability Check. A protocol for demonstrating that an EC cell operated over a given nominal analyte concentration range provides a stable and consistent response and is not significantly affected by repeated exposure to that gas.

3.8 Sample Flow Rate. The flow rate of the gas sample as it passes through the EC cell. In some situations, EC cells can experience drift with changes in flow rate. The flow rate must be monitored and documented during all phases of a sampling run.

3.9 Sampling Run. A timed three-phase event whereby an EC cell's response rises and plateaus in a sample conditioning phase, remains relatively constant during a measurement data phase, then declines during a refresh phase. The sample conditioning phase exposes the EC cell to the gas sample for a length of time sufficient to reach a constant response. The measurement data phase is the time interval during which gas sample measurements can be made that meet the acceptance criteria of this protocol. The refresh phase then purges the EC cells with CO-free air. The refresh phase replenishes requisite O_2 and moisture in the electrolyte reserve and provides a mechanism to de-gas or desorb any interference gas scrubbers or filters so as to enable a stable CO EC cell response. There are four primary types of sampling runs: pre- sampling calibrations; stack gas sampling; post-sampling calibration checks: and measurement system repeatability checks. Stack gas sampling runs can be chained together for extended evaluations, providing all other procedural specifications are met.

3.10 Sampling Day. A time not to exceed twelve hours from the time of the pre-sampling calibration to the post-sampling calibration check. During this time, stack gas sampling runs can be repeated without repeated recalibrations, providing all other sampling specifications have been met.

3.11 Pre-Sampling Calibration/Post-Sampling Calibration Check. The protocols executed at the beginning and end of each sampling day

Pt. 63, Subpt. ZZZZ, App. A

to bracket measurement readings with controlled performance checks.

3.12 Performance-Established Configuration. The EC cell and sampling system configuration that existed at the time that it initially met the performance requirements of this protocol.

4.0 INTERFERENCES.

When present in sufficient concentrations, NO and NO₂ are two gas species that have been reported to interfere with CO concentration measurements. In the likelihood of this occurrence, it is the protocol user's responsibility to employ and properly maintain an appropriate CO EC cell filter or scrubber for removal of these gases, as described in Section 6.2.12.

5.0 SAFETY. [RESERVED]

6.0 Equipment and Supplies.

6.1 What equipment do I need for the measurement system?

The system must maintain the gas sample at conditions that will prevent moisture condensation in the sample transport lines, both before and as the sample gas contacts the EC cells. The essential components of the measurement system are described below.

6.2 Measurement System Components.

6.2.1 Sample Probe. A single extractionpoint probe constructed of glass, stainless steel or other non-reactive material, and of length sufficient to reach any designated sampling point. The sample probe must be designed to prevent plugging due to condensation or particulate matter.

6.2.2 Sample Line. Non-reactive tubing to transport the effluent from the sample probe to the EC cell.

6.2.3 Calibration Assembly (optional). A three-way valve assembly or equivalent to introduce calibration gases at ambient pressure at the exit end of the sample probe during calibration checks. The assembly must be designed such that only stack gas or calibration gas flows in the sample line and all gases flow through any gas path filters.

6.2.4 Particulate Filter (optional). Filters before the inlet of the EC cell to prevent accumulation of particulate material in the measurement system and extend the useful life of the components. All filters must be fabricated of materials that are non-reactive to the gas mixtures being sampled.

6.2.5 Sample Pump. A leak-free pump to provide undiluted sample gas to the system at a flow rate sufficient to minimize the response time of the measurement system. If located upstream of the EC cells, the pump must be constructed of a material that is non-reactive to the gas mixtures being sampled.

6.2.8 Sample Flow Rate Monitoring. An adjustable rotameter or equivalent device used

Pt. 63, Subpt. ZZZZ, App. A

to adjust and maintain the sample flow rate through the analyzer as prescribed.

6.2.9 Sample Gas Manifold (optional). A manifold to divert a portion of the sample gas stream to the analyzer and the remainder to a by-pass discharge vent. The sample gas manifold may also include provisions for introducing calibration gases directly to the analyzer. The manifold must be constructed of a material that is non-reactive to the gas mixtures being sampled.

 $6.2.10 \ EC \ cell$. A device containing one or more EC cells to determine the CO and O₂ concentrations in the sample gas stream. The EC cell(s) must meet the applicable performance specifications of Section 13 of this protocol.

6.2.11 Data Recorder. A strip chart recorder, computer or digital recorder to make a record of analyzer output data. The data recorder resolution (i.e., readability) must be no greater than 1 ppm for CO; 0.1 percent for O₂; and one degree (either °C or °F) for temperature. Alternatively, you may use a digital or analog meter having the same resolution to observe and manually record the analyzer responses.

6.2.12 Interference Gas Filter or Scrubber. A device to remove interfering compounds upstream of the CO EC cell. Specific interference gas filters or scrubbers used in the performance-established configuration of the analyzer must continue to be used. Such a filter or scrubber must have a means to determine when the removal agent is exhausted. Periodically replace or replenish it in accordance with the manufacturer's recommendations.

7.0 REAGENTS AND STANDARDS. WHAT CALIBRATION GASES ARE NEEDED?

7.1 Calibration Gases. CO calibration gases for the EC cell must be CO in nitrogen or CO in a mixture of nitrogen and O_2 . Use CO calibration gases with labeled concentration values certified by the manufacturer to be within ± 5 percent of the label value. Dry ambient air (20.9 percent O_2) is acceptable for calibration of the O_2 cell. If needed, any lower percentage O_2 calibration gas must be a mixture of O_2 in nitrogen.

7.1.1 Up-Scale CO Calibration Gas Concentration. Choose one or more up-scale gas concentrations such that the average of the stack gas measurements for each stack gas sampling run are between 25 and 150 percent of those concentrations. Alternatively, choose an up-scale gas that does not exceed twice the concentration of the applicable outlet standard. If a measured gas value exceeds 150 percent of the up-scale CO calibration gas value at any time during the stack gas sampling run, the run must be discarded and repeated.

7.1.2 Up-Scale O_2 Calibration Gas Concentration.

40 CFR Ch. I (7–1–23 Edition)

Select an O_2 gas concentration such that the difference between the gas concentration and the average stack gas measurement or reading for each sample run is less than 15 percent O_2 . When the average exhaust gas O_2 readings are above 6 percent, you may use dry ambient air (20.9 percent O_2) for the upscale O_2 calibration gas.

7.1.3 Zero Gas. Use an inert gas that contains less than 0.25 percent of the up-scale CO calibration gas concentration. You may use dry air that is free from ambient CO and other combustion gas products (e.g., CO_2).

8.0 SAMPLE COLLECTION AND ANALYSIS

8.1 Selection of Sampling Sites.

8.1.1 Control Device Inlet. Select a sampling site sufficiently downstream of the engine so that the combustion gases should be well mixed. Use a single sampling extraction point near the center of the duct (e.g., within the 10 percent centroidal area), unless instructed otherwise.

8.1.2 Exhaust Gas Outlet. Select a sampling site located at least two stack diameters downstream of any disturbance (e.g., turbocharger exhaust, crossover junction or recirculation take-off) and at least one-half stack diameter upstream of the gas discharge to the atmosphere. Use a single sampling extraction point near the center of the duct (e.g., within the 10 percent centroidal area), unless instructed otherwise.

8.2 Stack Gas Collection and Analysis. Prior to the first stack gas sampling run, conduct that the pre-sampling calibration in accordance with Section 10.1. Use Figure 1 to record all data. Zero the analyzer with zero gas. Confirm and record that the scrubber media color is correct and not exhausted. Then position the probe at the sampling point and begin the sampling run at the same flow rate used during the up-scale calibration. Record the start time. Record all EC cell output responses and the flow rate during the "sample conditioning phase" once per minute until constant readings are obtained. Then begin the "measurement data phase" and record readings every 15 seconds for at least two minutes (or eight readings), or as otherwise required to achieve two continuous minutes of data that meet the specification given in Section 13.1. Finally, perform the "refresh phase" by introducing dry air, free from CO and other combustion gases, until several minute-to-minute readings of consistent value have been obtained. For each run use the "measurement data phase" readings to calculate the average stack gas CO and O₂ concentrations.

8.3 EC Cell Rate. Maintain the EC cell sample flow rate so that it does not vary by more than ± 10 percent throughout the pre-sampling calibration, stack gas sampling and post-sampling calibration check. Alternatively, the EC cell sample flow rate can be maintained within a tolerance range that

does not affect the gas concentration readings by more than ± 3 percent, as instructed by the EC cell manufacturer.

9.0 QUALITY CONTROL (RESERVED)

10.0 Calibration and Standardization

10.1 Pre-Sampling Calibration. Conduct the following protocol once for each nominal range to be used on each EC cell before performing a stack gas sampling run on each field sampling day. Repeat the calibration if you replace an EC cell before completing all of the sampling runs. There is no prescribed order for calibration of the EC cells: however, each cell must complete the measurement data phase during calibration. Assemble the measurement system by following the manufacturer's recommended protocols including for preparing and preconditioning the EC cell. Assure the measurement system has no leaks and verify the gas scrubbing agent is not depleted. Use Figure 1 to record all data.

10.1.1 Zero Calibration. For both the O_2 and CO cells, introduce zero gas to the measurement system (e.g., at the calibration assembly) and record the concentration reading every minute until readings are constant for at least two consecutive minutes. Include the time and sample flow rate. Repeat the steps in this section at least once to verify the zero calibration for each component gas.

10.1.2 Zero Calibration Tolerance. For each zero gas introduction, the zero level output must be less than or equal to ± 3 percent of the up-scale gas value or ± 1 ppm, whichever is less than or equal to ± 0.3 percent O₂ for the O₂ channel.

10.1.3 Up-Scale Calibration. Individually introduce each calibration gas to the measurement system (e.g., at the calibration assembly) and record the start time. Record all EC cell output responses and the flow rate during this "sample conditioning phase" once per minute until readings are constant for at least two minutes. Then begin the "measurement data phase" and record readings every 15 seconds for a total of two minutes, or as otherwise required. Finally, perform the "refresh phase" by introducing dry air, free from CO and other combustion gases, until readings are constant for at least two consecutive minutes. Then repeat the steps in this section at least once to verify the calibration for each component gas. Introduce all gases to flow through the entire sample handling system (i.e., at the exit end of the sampling probe or the calibration assembly).

10.1.4 Up-Scale Calibration Error. The mean of the difference of the "measurement data phase" readings from the reported standard gas value must be less than or equal to ± 5 percent or ± 1 ppm for CO or ± 0.5 percent O₂, whichever is less restrictive, respectively.

Pt. 63, Subpt. ZZZZ, App. A

The maximum allowable deviation from the mean measured value of any single "measurement data phase" reading must be less than or equal to ± 2 percent or ± 1 ppm for CO or ± 0.5 percent O₂, whichever is less restrictive, respectively.

10.2 Post-Sampling Calibration Check. Conduct a stack gas post-sampling calibration check after the stack gas sampling run or set of runs and within 12 hours of the initial calibration. Conduct up-scale and zero calibration checks using the protocol in Section 10.1. Make no changes to the sampling system or EC cell calibration until all post-sampling calibration checks have been recorded. If either the zero or up-scale calibration error exceeds the respective specification in Sections 10.1.2 and 10.1.4 then all measurement data collected since the previous successful calibrations are invalid and re-calibration and re-sampling are required. If the sampling system is disassembled or the EC cell calibration is adjusted, repeat the calibration check before conducting the next analyzer sampling run.

11.0 ANALYTICAL PROCEDURE

The analytical procedure is fully discussed in Section 8.

12.0 CALCULATIONS AND DATA ANALYSIS

Determine the CO and O_2 concentrations for each stack gas sampling run by calculating the mean gas concentrations of the data recorded during the "measurement data phase".

13.0 PROTOCOL PERFORMANCE

Use the following protocols to verify consistent analyzer performance during each field sampling day.

13.1 Measurement Data Phase Performance Check. Calculate the mean of the readings from the "measurement data phase". The maximum allowable deviation from the mean for each of the individual readings is ± 2 percent, or ± 1 ppm, whichever is less restrictive. Record the mean value and maximum deviation for each gas monitored. Data must conform to Section 10.1.4. The EC cell flow rate must conform to the specification in Section 8.3.

Example: A measurement data phase is invalid if the maximum deviation of any single reading comprising that mean is greater than ± 2 percent or ± 1 ppm (the default criteria). For example, if the mean = 30 ppm, single readings of below 29 ppm and above 31 ppm are disallowed).

13.2 Interference Check. Before the initial use of the EC cell and interference gas scrubber in the field, and semi-annually thereafter, challenge the interference gas scrubber with NO and NO_2 gas standards that are

Pt. 63, Subpt. ZZZZ, App. A

generally recognized as representative of diesel-fueled engine NO and NO_2 emission values. Record the responses displayed by the CO EC cell and other pertinent data on Figure 1 or a similar form.

13.2.1 Interference Response. The combined NO and NO₂ interference response should be less than or equal to ± 5 percent of the upscale CO calibration gas concentration.

13.3 Repeatability Check. Conduct the following check once for each nominal range that is to be used on the CO EC cell within 5 days prior to each field sampling program. If a field sampling program lasts longer than 5 days, repeat this check every 5 days. Immediately repeat the check if the EC cell is replaced or if the EC cell is exposed to gas concentrations greater than 150 percent of the highest up-scale gas concentration.

13.3.1 Repeatability Check Procedure. Perform a complete EC cell sampling run (all three phases) by introducing the CO calibration gas to the measurement system and record the response. Follow Section 10.1.3. Use Figure 1 to record all data. Repeat the run three times for a total of four complete runs. During the four repeatability check runs, do not adjust the system except where necessary to achieve the correct calibration gas flow rate at the analyzer.

13.3.2 Repeatability Check Calculations. Determine the highest and lowest average "measurement data phase" CO concentra-

40 CFR Ch. I (7–1–23 Edition)

tions from the four repeatability check runs and record the results on Figure 1 or a similar form. The absolute value of the difference between the maximum and minimum average values recorded must not vary more than ± 3 percent or ± 1 ppm of the up-scale gas value, whichever is less restrictive.

14.0 POLLUTION PREVENTION (RESERVED)

15.0 WASTE MANAGEMENT (RESERVED)

16.0 ALTERNATIVE PROCEDURES (RESERVED)

17.0 References

(1) "Development of an Electrochemical Cell Emission Analyzer Test Protocol", Topical Report, Phil Juneau, Emission Monitoring, Inc., July 1997.

(2) "Determination of Nitrogen Oxides, Carbon Monoxide, and Oxygen Emissions from Natural Gas-Fired Engines, Boilers, and Process Heaters Using Portable Analyzers", EMC Conditional Test Protocol 30 (CTM-30), Gas Research Institute Protocol GRI-96/0008, Revision 7, October 13, 1997.

(3) "ICAC Test Protocol for Periodic Monitoring", EMC Conditional Test Protocol 34 (CTM-034), The Institute of Clean Air Companies, September 8, 1999.

(4) "Code of Federal Regulations", Protection of Environment, 40 CFR, Part 60, Appendix A, Methods 1-4; 10.

				CD	S	MPL	R D					
	Faciût			□	ine D.			Date				
R Dn Ope:F	reSamp	 ē Caí	Tation	0	tack Ga	⊡ □ a⊡ Sam∣	þē	Pott	amp.e	_ Ca⊡Check	 Repeata⊡í	1 Check
R 🗅 🛛		- 0 ⁰	- 8	0 0	0 2	ς O	ωÖ	4 0 ₂	4 O	□ □	ScrOK	Fow⊡Rate
sampē Con⊡												
<i>"</i>												
" "												
Mea⊡rement Data Pha⊡e												
// 												
"												
"												
"												
" 												
,,												
"												
Mean												
Refre⊡h												
Phale												
<i>"</i>												
" "												
		-										

Pt. 63, Subpt. ZZZZ, App. A

§63.7080

[78 FR 6721, Jan. 30, 2013]

Subpart AAAAA—National Emission Standards for Hazardous Air Pollutants for Lime Manufacturing Plants

SOURCE: 69 FR 416, Jan. 5, 2004, unless otherwise noted.

WHAT THIS SUBPART COVERS

§63.7080 What is the purpose of this subpart?

This subpart establishes national emission standards for hazardous air pollutants (NESHAP) for lime manufacturing plants. This subpart also establishes requirements to demonstrate initial and continuous compliance with the emission limitations.

§63.7081 Am I subject to this subpart?

(a) You are subject to this subpart if you own or operate a lime manufacturing plant (LMP) that is a major source, or that is located at, or is part of, a major source of hazardous air pollutant (HAP) emissions, unless the LMP is located at a kraft pulp mill, soda pulp mill, sulfite pulp mill, beet sugar manufacturing plant, or only processes sludge containing calcium carbonate from water softening processes.

(1) An LMP is an establishment engaged in the manufacture of lime product (calcium oxide, calcium oxide with magnesium oxide, or dead burned dolomite) by calcination of limestone, dolomite, shells or other calcareous substances.

(2) A major source of HAP is a plant site that emits or has the potential to emit any single HAP at a rate of 9.07 megagrams (10 tons) or more per year or any combination of HAP at a rate of 22.68 megagrams (25 tons) or more per year from all emission sources at the plant site.

(b) [Reserved]

§ 63.7082 What parts of my plant does this subpart cover?

(a) This subpart applies to each existing or new lime kiln(s) and their associated cooler(s), and processed stone handling (PSH) operations system(s)

40 CFR Ch. I (7–1–23 Edition)

located at an LMP that is a major source.

(b) A new lime kiln is a lime kiln, and (if applicable) its associated lime cooler, for which construction or reconstruction began after December 20, 2002, if you met the applicability criteria in §63.7081 at the time you began construction or reconstruction.

(c) A new PSH operations system is the equipment in paragraph (g) of this section, for which construction or reconstruction began after December 20, 2002, if you met the applicability criteria in §63.7081 at the time you began construction or reconstruction.

(d) A lime kiln or PSH operations system is reconstructed if it meets the criteria for reconstruction defined in $\S63.2$.

(e) An existing lime kiln is any lime kiln, and (if applicable) its associated lime cooler, that does not meet the definition of a new kiln of paragraph (b) of this section.

(f) An existing PSH operations system is any PHS operations system that does not meet the definition of a new PSH operations system in paragraph (c) of this section.

(g) A PSH operations system includes all equipment associated with PSH operations beginning at the processed stone storage bin(s) or open storage pile(s) and ending where the processed stone is fed into the kiln. It includes man-made processed stone storage bins (but not open processed stone storage piles), conveying system transfer points, bulk loading or unloading systems, screening operations, surge bins, bucket elevators, and belt conveyors. No other materials processing operations are subject to this subpart.

(h) Nuisance dust collectors on lime coolers are part of the lime materials processing operations and are not covered by this subpart.

(i) Lime hydrators are not subject to this subpart.

(j) Open material storage piles are not subject to this subpart.

§63.7083 When do I have to comply with this subpart?

(a) If you have a new affected source, you must comply with this subpart according to paragraphs (a)(1) and (2) of this section.

Appendix P

AUTHENTICATED U.S. GOVERNMENT INFORMATION GPO

Environmental Protection Agency

§63.7480

Citation	S⊡lēct	□pp1e□to S□□part CCCCC□	□□panation
63.1 63.2 63.3 63.4	oppicalit Definition nitlan ordination Prohilite opticial opticial	Ye Ye Ye Ye	
63.6a, II, C, II, e, f, g, hI2II8 63.6hI9	Complance with Stanlar an Maintenance Re irement	Ye I	
63. a 3, ., c.h. 63. a 12	Performance celting Recirement opplicatific and Performance celt Date	Ye□ □o	S part CCCCC pecifie applica
63.8 a 1	Monitoring Reirement	Ye	CMS remirement in 63.8 c 4 imit c 5, an c 6 app on to COMS for catter tack
C63.8 Ca C4 C	for Contro De ice in 63.11.	□o	Fare□ are not a contro□⊡e⊡ce for S⊡part CCCCC affecte□ ro⊓rce□
63.8 c 4	Contin o Monitoring Stem	□o	S part CCCCC pecifie re ire
63.8 e 4 115	Performance	Ye	iccept COMS performance e a la tion m⊡t e con⊡cte efore
□63.8 1 16 □	R⊡⊡ ⊡ternati⊡e	□o	S part CCCCC oe not re re
□63.8頃□15□	Data Re ⊡ction	□o	S part CCCCC pecifie ata that canti e e in compiting a era
□63.9	□otification Re⊡irement□	Ye 🗆	□ itiona notification for CMS in □ 63.9 g app on to COMS for □ atter □ fack □
63.10°a,	Recor⊡keeping an⊡ Reporting Re⊡ ⊡irement⊡	Ye	□forforCMS in
063.10 02 01 01i 0	CMS Recor for R C C Iternatice	□o	S⊡part CCCCC ⊡oe nt re⊡ire C⊓MS.
63.10 c80	Recor of cce onicion an Parameter Monitoring	□o	S⊡part CCCCC ⊡pecifie⊡ recor⊡ re⊡irement⊡
□63.10 le □ 3 □		□o	S □ part CCCCC □ pecifie □ reporting re □ irement □
□63.11	Contro De ice Re ⊡irement□	□o	S □ part CCCCC □ oe □ not re □ ire fare □
□63.12 □□63.13□63.15	Statethorit_ an De egation re_e, incorporation Ref erence,aia_itt of information.	Ye□ Ye□	

Subpart DDDDD—National Emission Standards for Hazardous Air Pollutants for Major Sources: Industrial, Commercial, and Institutional Boilers and Process Heaters

SOURCE: 76 FR 15664, Mar. 21, 2011, unless otherwise noted.

WHAT THIS SUBPART COVERS

§63.7480 What is the purpose of this subpart?

This subpart establishes national emission limitations and work practice standards for hazardous air pollutants (HAP) emitted from industrial, commercial, and institutional boilers and process heaters located at major sources of HAP. This subpart also establishes requirements to demonstrate initial and continuous compliance with the emission limitations and work practice standards.

§63.7485 Am I subject to this subpart?

You are subject to this subpart if you own or operate an industrial, commercial, or institutional boiler or process heater as defined in §63.7575 that is located at, or is part of, a major source of HAP, except as specified in §63.7491. For purposes of this subpart, a major source of HAP is as defined in §63.2, except that for oil and natural gas production facilities, a major source of HAP is as defined in §63.7575.

[78 FR 7162, Jan. 31, 2013]

§63.7490 What is the affected source of this subpart?

(a) This subpart applies to new, reconstructed, and existing affected sources as described in paragraphs (a)(1) and (2) of this section.

(1) The affected source of this subpart is the collection at a major source of all existing industrial, commercial, and institutional boilers and process heaters within a subcategory as defined in §63.7575.

(2) The affected source of this subpart is each new or reconstructed industrial, commercial, or institutional boiler or process heater, as defined in §63.7575, located at a major source.

(b) A boiler or process heater is new if you commence construction of the boiler or process heater after June 4, 2010, and you meet the applicability criteria at the time you commence construction.

(c) A boiler or process heater is reconstructed if you meet the reconstruction criteria as defined in §63.2, you commence reconstruction after June 4, 2010, and you meet the applicability criteria at the time you commence reconstruction.

(d) A boiler or process heater is existing if it is not new or reconstructed.

(e) An existing electric utility steam generating unit (EGU) that meets the applicability requirements of this subpart after the effective date of this final rule due to a change (e.g., fuel switch) is considered to be an existing source under this subpart.

[76 FR 15664, Mar. 21, 2011, as amended at 78 FR 7162, Jan. 31, 2013]

40 CFR Ch. I (7–1–23 Edition)

§63.7491 Are any boilers or process heaters not subject to this subpart?

The types of boilers and process heaters listed in paragraphs (a) through (n) of this section are not subject to this subpart.

(a) An electric utility steam generating unit (EGU) covered by subpart UUUUU of this part or a natural gasfired EGU as defined in subpart UUUUU of this part firing at least 85 percent natural gas on an annual heat input basis.

(b) A recovery boiler or furnace covered by subpart MM of this part.

(c) A boiler or process heater that is used specifically for research and development, including test steam boilers used to provide steam for testing the propulsion systems on military vessels. This does not include units that provide heat or steam to a process at a research and development facility.

(d) A hot water heater as defined in this subpart.

(e) A refining kettle covered by subpart X of this part.

(f) An ethylene cracking furnace covered by subpart YY of this part.

(g) Blast furnace stoves as described in EPA-453/R-01-005 (incorporated by reference, see \S 63.14).

(h) Any boiler or process heater that is part of the affected source subject to another subpart of this part, such as boilers and process heaters used as control devices to comply with subparts JJJ, OOO, PPP, and U of this part.

(i) Any boiler or process heater that is used as a control device to comply with another subpart of this part, or part 60, part 61, or part 65 of this chapter provided that at least 50 percent of the average annual heat input during any 3 consecutive calendar years to the boiler or process heater is provided by regulated gas streams that are subject to another standard.

(j) Temporary boilers and process heaters as defined in this subpart.

(k) Blast furnace gas fuel-fired boilers and process heaters as defined in this subpart.

(1) Any boiler or process heater specifically listed as an affected source in any standard(s) established under section 129 of the Clean Air Act.

(m) A unit that burns hazardous waste covered by Subpart EEE of this

part. A unit that is exempt from Subpart EEE as specified in §63.1200(b) is not covered by Subpart EEE.

(n) Residential boilers as defined in this subpart.

[76 FR 15664, Mar. 21, 2011, as amended at 78 FR 7162, Jan. 31, 2013; 80 FR 72806, Nov. 20, 2015]

§63.7495 When do I have to comply with this subpart?

(a) If you have a new or reconstructed boiler or process heater, you must comply with this subpart by April 1, 2013, or upon startup of your boiler or process heater, whichever is later.

(b) If you have an existing boiler or process heater, you must comply with this subpart no later than January 31, 2016, except as provided in §63.6(i).

(c) If you have an area source that increases its emissions or its potential to emit such that it becomes a major source of HAP, paragraphs (c)(1) and (2) of this section apply to you.

(1) Any new or reconstructed boiler or process heater at the existing source must be in compliance with this subpart upon startup.

(2) Any existing boiler or process heater at the existing source must be in compliance with this subpart within 3 years after the source becomes a major source.

(d) You must meet the notification requirements in 63.7545 according to the schedule in 63.7545 and in subpart A of this part. Some of the notifications must be submitted before you are required to comply with the emission limits and work practice standards in this subpart.

(e) If you own or operate an industrial, commercial, or institutional boiler or process heater and would be subject to this subpart except for the exemption in §63.7491(1) for commercial and industrial solid waste incineration units covered by part 60, subpart CCCC or subpart DDDD, and you cease combusting solid waste, you must be in compliance with this subpart and are no longer subject to part 60, subparts CCCC or DDDD beginning on the effective date of the switch as identified under the provisions of §60.2145(a)(2) and (3) or §60.2710(a)(2) and (3). (f) If you own or operate an existing EGU that becomes subject to this subpart after January 31, 2016, you must be in compliance with the applicable existing source provisions of this subpart on the effective date such unit becomes subject to this subpart.

(g) If you own or operate an existing industrial, commercial, or institutional boiler or process heater and would be subject to this subpart except for a exemption in §63.7491(i) that becomes subject to this subpart after January 31, 2013, you must be in compliance with the applicable existing source provisions of this subpart within 3 years after such unit becomes subject to this subpart.

(h) If you own or operate an existing industrial, commercial, or institutional boiler or process heater and have switched fuels or made a physical change to the boiler or process heater that resulted in the applicability of a different subcategory after the compliance date of this subpart, you must be in compliance with the applicable existing source provisions of this subpart on the effective date of the fuel switch or physical change.

(i) If you own or operate a new industrial, commercial, or institutional boiler or process heater and have switched fuels or made a physical change to the boiler or process heater that resulted in the applicability of a different subcategory, you must be in compliance with the applicable new source provisions of this subpart on the effective date of the fuel switch or physical change.

[76 FR 15664, Mar. 21, 2011, as amended at 78 FR 7162, Jan. 31, 2013; 80 FR 72807, Nov. 20, 2015]

EMISSION LIMITATIONS AND WORK PRACTICE STANDARDS

§63.7499 What are the subcategories of boilers and process heaters?

The subcategories of boilers and process heaters, as defined in §63.7575 are:

(a) Pulverized coal/solid fossil fuel units.

(b) Stokers designed to burn coal/ solid fossil fuel.

(c) Fluidized bed units designed to burn coal/solid fossil fuel.

§63.7500

(d) Stokers/sloped grate/other units designed to burn kiln dried biomass/ bio-based solid.

(e) Fluidized bed units designed to burn biomass/bio-based solid.

(f) Suspension burners designed to burn biomass/bio-based solid.

(g) Fuel cells designed to burn biomass/bio-based solid.

(h) Hybrid suspension/grate burners designed to burn wet biomass/bio-based solid.

(i) Stokers/sloped grate/other units designed to burn wet biomass/bio-based solid.

(j) Dutch ovens/pile burners designed to burn biomass/bio-based solid.

(k) Units designed to burn liquid fuel that are non-continental units.

(1) Units designed to burn gas 1 fuels.(m) Units designed to burn gas 2 (other) gases.

(n) Metal process furnaces.

(o) Limited-use boilers and process heaters

(p) Units designed to burn solid fuel.

(q) Units designed to burn liquid fuel. (r) Units designed to burn coal/solid

fossil fuel.

(s) Fluidized bed units with an integrated fluidized bed heat exchanger designed to burn coal/solid fossil fuel.

(t) Units designed to burn heavy liquid fuel.

(u) Units designed to burn light liquid fuel.

[76 FR 15664, Mar. 21, 2011, as amended at 78 FR 7163, Jan. 31, 2013]

§63.7500 What emission limitations, work practice standards, and operating limits must I meet?

(a) You must meet the requirements in paragraphs (a)(1) through (3) of this section, except as provided in paragraphs (b) through (e) of this section. You must meet these requirements at all times the affected unit is operating, except as provided in paragraph (f) of this section.

(1) You must meet each emission limit and work practice standard in Tables 1 through 3 and 11 through 15 to this subpart that applies to your boiler or process heater, for each boiler or process heater at your source, except as provided under §63.7522. The outputbased emission limits, in units of pounds per million Btu of steam out-

40 CFR Ch. I (7–1–23 Edition)

put, in Table 1 or 2 to this subpart are an alternative applicable only to boilers and process heaters that generate either steam, cogenerate steam with electricity, or both. The output-based emission limits, in units of pounds per megawatt-hour, in Table 1 or 2 to this subpart are an alternative applicable only to boilers that generate only electricity. Boilers that perform multiple functions (cogeneration and electricity generation) or supply steam to common headers would calculate a total steam energy output using Equation 1 of §63.7575 to demonstrate compliance with the output-based emission limits, in units of pounds per million Btu of steam output, in Table 1 or 2 to this subpart. If you operate a new boiler or process heater, you can choose to comply with alternative limits as discussed in paragraphs (a)(1)(i) through (iv) of this section, but on or after October 6, 2025, you must comply with the emission limits in Table 1 to this subpart. If you operate an existing boiler or process heater, you can choose to comply with alternative limits as discussed in paragraph (a)(1)(v) of this section, but on or after October 6, 2025 you must comply with the emission limits in Table 2 to this subpart.

(i) If your boiler or process heater commenced construction or reconstruction after June 4, 2010, and before May 20, 2011, you may comply with the emission limits in Table 11 or 14 to this subpart until January 31, 2016.

(ii) If your boiler or process heater commenced construction or reconstruction on or after May 20, 2011, and before December 23, 2011, you may comply with the emission limits in Table 12 or 14 to this subpart until January 31, 2016.

(iii) If your boiler or process heater commenced construction or reconstruction on or after December 23, 2011, and before April 1, 2013, you may comply with the emission limits in Table 13 or 14 to this subpart until January 31, 2016.

(iv) If you operate a new boiler or process heater, you must comply with either the emission limits in Table 1 to this subpart or the emission limits in Table 14 to this subpart until you must comply with the emission limits in Table 1.

(v) If you operate an existing boiler or process heater, you must comply with either the emission limits in Table 2 to this subpart or the emission limits in Table 15 to this subpart until you must comply with the emission limits in Table 2.

(2) You must meet each operating limit in Table 4 to this subpart that applies to your boiler or process heater. If you use a control device or combination of control devices not covered in Table 4 to this subpart, or you wish to establish and monitor an alternative operating limit or an alternative monitoring parameter, you must apply to the EPA Administrator for approval of alternative monitoring under §63.8(f).

(3) At all times, you must operate and maintain any affected source (as defined in §63.7490), including associated air pollution control equipment and monitoring equipment, in a manner consistent with safety and good air pollution control practices for minimizing emissions. Determination of whether such operation and maintenance procedures are being used will be based on information available to the Administrator that may include, but is not limited to, monitoring results, review of operation and maintenance procedures, review of operation and maintenance records, and inspection of the source.

(b) As provided in §63.6(g), EPA may approve use of an alternative to the work practice standards in this section.

(c) Limited-use boilers and process heaters must complete a tune-up every 5 years as specified in §63.7540. They are not subject to the emission limits in Tables 1 and 2 or Tables 11 through 15 to this subpart, the annual tune-up, or the energy assessment requirements in Table 3 to this subpart, or the operating limits in Table 4 to this subpart.

(d) Boilers and process heaters with a heat input capacity of less than or equal to 5 million Btu per hour in the units designed to burn gas 2 (other) fuels subcategory or units designed to burn light liquid fuels subcategory must complete a tune-up every 5 years as specified in §63.7540.

(e) Boilers and process heaters in the units designed to burn gas 1 fuels subcategory with a heat input capacity of less than or equal to 5 million Btu per hour must complete a tune-up every 5 years as specified in §63.7540. Boilers and process heaters in the units designed to burn gas 1 fuels subcategory with a heat input capacity greater than 5 million Btu per hour and less than 10 million Btu per hour must complete a tune-up every 2 years as specified in §63.7540. Boilers and process heaters in the units designed to burn gas 1 fuels subcategory are not subject to the emission limits in Tables 1 and 2 or Tables 11 through 15 to this subpart, or the operating limits in Table 4 to this subpart.

(f) These standards apply at all times the affected unit is operating, except during periods of startup and shutdown during which time you must comply only with items 5 and 6 of Table 3 to this subpart.

[76 FR 15664, Mar. 21, 2011, as amended at 78
 FR 7163, Jan. 31, 2013; 80 FR 72807, Nov. 20, 2015; 87 FR 60840, Oct. 6, 2022]

§63.7501 [Reserved]

GENERAL COMPLIANCE REQUIREMENTS

§ 63.7505 What are my general requirements for complying with this subpart?

(a) You must be in compliance with the emission limits, work practice standards, and operating limits in this subpart. These emission and operating limits apply to you at all times the affected unit is operating except for the periods noted in §63.7500(f).

(b) [Reserved]

(c) You must demonstrate compliance with all applicable emission limits using performance stack testing, fuel analysis, or continuous monitoring systems (CMS), including a continuous emission monitoring system (CEMS), continuous opacity monitoring system (COMS), continuous parameter monitoring system (CPMS), or particulate matter continuous parameter monitoring system (PM CPMS), where applicable. You may demonstrate compliance with the applicable emission limit for hydrogen chloride (HCl), mercury, or total selected metals (TSM) using fuel analysis if the emission rate calculated according to §63.7530(c) is less than the applicable emission limit. For gaseous fuels, you may not use fuel

analyses to comply with the TSM alternative standard or the HCl standard. Otherwise, you must demonstrate compliance for HCl, mercury, or TSM using performance stack testing, if subject to an applicable emission limit listed in Table 1 or 2 or Tables 11 through 15 to this subpart.

(d) If you demonstrate compliance with any applicable emission limit through performance testing and subsequent compliance with operating limits through the use of CPMS, or with a CEMS or COMS, you must develop a site-specific monitoring plan according to the requirements in paragraphs (d)(1) through (4) of this section for the use of any CEMS, COMS, or CPMS. This requirement also applies to you if you petition the EPA Administrator for alternative monitoring parameters under §63.8(f).

(1) For each CMS required in this section (including CEMS, COMS, or CPMS), you must develop, and submit to the Administrator for approval upon request, a site-specific monitoring plan that addresses design, data collection, and the quality assurance and quality control elements outlined in §63.8(d) and the elements described in paragraphs (d)(1)(i) through (iii) of this section. You must submit this site-specific monitoring plan, if requested, at least 60 days before your initial performance evaluation of your CMS. This requirement to develop and submit a site specific monitoring plan does not apply to affected sources with existing CEMS or COMS operated according to the performance specifications under appendix B to part 60 of this chapter and that meet the requirements of §63.7525. Using the process described in §63.8(f)(4), you may request approval of alternative monitoring system quality assurance and quality control procedures in place of those specified in this paragraph and, if approved, include the alternatives in your site-specific monitoring plan.

(i) Installation of the CMS sampling probe or other interface at a measurement location relative to each affected process unit such that the measurement is representative of control of the exhaust emissions (e.g., on or downstream of the last control device); 40 CFR Ch. I (7–1–23 Edition)

(ii) Performance and equipment specifications for the sample interface, the pollutant concentration or parametric signal analyzer, and the data collection and reduction systems; and

(iii) Performance evaluation procedures and acceptance criteria (e.g., calibrations, accuracy audits, analytical drift).

(2) In your site-specific monitoring plan, you must also address paragraphs (d)(2)(i) through (iii) of this section.

(i) Ongoing operation and maintenance procedures in accordance with the general requirements of 63.8(c)(1)(ii), (c)(3), and (c)(4)(ii);

(ii) Ongoing data quality assurance procedures in accordance with the general requirements of §63.8(d); and

(iii) Ongoing recordkeeping and reporting procedures in accordance with the general requirements of 63.10(c) (as applicable in Table 10 to this subpart), (e)(1), and (e)(2)(i).

(3) You must conduct a performance evaluation of each CMS in accordance with your site-specific monitoring plan.

(4) You must operate and maintain the CMS in continuous operation according to the site-specific monitoring plan.

(e) If you have an applicable emission limit, and you choose to comply using definition (2) of "startup" in §63.7575, you must develop and implement a written startup and shutdown plan (SSP) according to the requirements in Table 3 to this subpart. The SSP must be maintained onsite and available upon request for public inspection.

[76 FR 15664, Mar. 21, 2011, as amended at 78
FR 7164, Jan. 31, 2013; 80 FR 72807, Nov. 20, 2015; 87 FR 60841, Oct. 6, 2022]

TESTING, FUEL ANALYSES, AND INITIAL COMPLIANCE REQUIREMENTS

§63.7510 What are my initial compliance requirements and by what date must I conduct them?

(a) For each boiler or process heater that is required or that you elect to demonstrate compliance with any of the applicable emission limits in Table 1 or 2 or Tables 11 through 15 to this subpart through performance (stack) testing, your initial compliance requirements include all the following:

(1) Conduct performance tests according to 63.7520 and Table 5 to this subpart.

(2) Conduct a fuel analysis for each type of fuel burned in your boiler or process heater according to $\S63.7521$ and Table 6 to this subpart, except as specified in paragraphs (a)(2)(i) through (iii) of this section.

(i) For each boiler or process heater that burns a single type of fuel, you are not required to conduct a fuel analysis for each type of fuel burned in your boiler or process heater according to § 63.7521 and Table 6 to this subpart. For purposes of this subpart, units that use a supplemental fuel only for startup, unit shutdown, and transient flame stability purposes still qualify as units that burn a single type of fuel, and the supplemental fuel is not subject to the fuel analysis requirements under § 63.7521 and Table 6 to this subpart.

(ii) When natural gas, refinery gas, or other gas 1 fuels are co-fired with other fuels, you are not required to conduct a fuel analysis of those Gas 1 fuels according to §63.7521 and Table 6 to this subpart. If gaseous fuels other than natural gas, refinery gas, or other gas 1 fuels are co-fired with other fuels and those non-Gas 1 gaseous fuels are subject to another subpart of this part, part 60, part 61, or part 65, you are not required to conduct a fuel analysis of those non-Gas 1 fuels according to §63.7521 and Table 6 to this subpart.

(iii) You are not required to conduct a chlorine fuel analysis for any gaseous fuels. You must conduct a fuel analysis for mercury on gaseous fuels unless the fuel is exempted in paragraphs (a)(2)(i)and (ii) of this section.

(3) Establish operating limits according to §63.7530 and Table 7 to this subpart.

(4) Conduct CMS performance evaluations according to §63.7525.

(b) For each boiler or process heater that you elect to demonstrate compliance with the applicable emission limits in Table 1 or 2 or Tables 11 through 15 to this subpart for HCl, mercury, or TSM through fuel analysis, your initial compliance requirement is to conduct a fuel analysis for each type of fuel burned in your boiler or process heater according to §63.7521 and Table 6 to this subpart and establish operating limits according to 63.7530 and Table 8 to this subpart. The fuels described in paragraphs (a)(2)(i) and (ii) of this section are exempt from these fuel analysis and operating limit requirements. The fuels described in paragraph (a)(2)(ii) of this section are exempt from the chloride fuel analysis and operating limit requirements. Boilers and process heaters that use a CEMS for mercury or HCl are exempt from the performance testing and operating limit requirements specified in paragraph (a) of this section for the HAP for which CEMS are used.

(c) If your boiler or process heater is subject to a carbon monoxide (CO) limit, your initial compliance demonstration for CO is to conduct a performance test for CO according to Table 5 to this subpart or conduct a performance evaluation of your continuous CO monitor, if applicable, according to §63.7525(a). Boilers and process heaters that use a CO CEMS to comply with the applicable alternative CO CEMS emission standard listed in Table 1 or 2 or Tables 11 through 15 to this subpart, as specified in §63.7525(a), are exempt from the initial CO performance testing and oxygen concentration operating limit requirements specified in paragraph (a) of this section.

(d) If your boiler or process heater is subject to a PM limit, your initial compliance demonstration for PM is to conduct a performance test in accordance with §63.7520 and Table 5 to this subpart.

(e) For existing affected sources (as defined in §63.7490), you must complete the initial compliance demonstrations, as specified in paragraphs (a) through (d) of this section, no later than 180 days after the compliance date that is specified for your source in §63.7495 and according to the applicable provisions in (63.7(a)(2)) as cited in Table 10 to this subpart, except as specified in paragraph (j) of this section. You must complete an initial tune-up by following the procedures described in §63.7540(a)(10)(i) through (vi) no later than the compliance date specified in §63.7495, except as specified in paragraph (j) of this section. You must

§63.7510

complete the one-time energy assessment specified in Table 3 to this subpart no later than the compliance date specified in §63.7495.

(f) For new or reconstructed affected sources (as defined in §63.7490), you must complete the initial compliance demonstration with the emission limits no later than July 30, 2013, or within 180 days after startup of the source, whichever is later.

(1) If you are demonstrating compliance with an emission limit in Tables 11 through 13 to this subpart that is less stringent than the applicable emission limit in Table 14 to this subpart, you must demonstrate compliance with the applicable emission limit in Table 14 no later than July 29, 2016.

(2) If you are demonstrating compliance with an emission limit in Table 14 to this subpart that is less stringent than the applicable emission limit in Table 1 to this subpart, you must demonstrate compliance with the applicable emission limit in Table 1 no later than October 6, 2025.

(g) For new or reconstructed affected sources (as defined in $\S63.7490$), you must demonstrate initial compliance with the applicable work practice standards in Table 3 to this subpart within the applicable annual, biennial, or 5-year schedule as specified in $\S63.7515(d)$ following the initial compliance date specified in $\S63.7495(a)$. Thereafter, you are required to complete the applicable annual, biennial, or 5-year tune-up as specified in $\S63.7515(d)$.

(h) For affected sources (as defined in §63.7490) that ceased burning solid waste consistent with §63.7495(e) and for which the initial compliance date has passed, you must demonstrate compliance within 60 days of the effective date of the waste-to-fuel switch. If you have not conducted your compliance demonstration for this subpart within the previous 12 months, you must complete all compliance demonstrations for this subpart before you commence or recommence combustion of solid waste.

(i) For an existing EGU that becomes subject after January 31, 2016, you must demonstrate compliance within 180 days after becoming an affected source.

40 CFR Ch. I (7–1–23 Edition)

(j) For existing affected sources (as defined in §63.7490) that have not operated between the effective date of the rule and the compliance date that is specified for your source in §63.7495, you must complete the initial compliance demonstration, if subject to the emission limits in Table 2 or 14 to this subpart, as applicable, as specified in paragraphs (a) through (d) of this section, no later than 180 days after the re-start of the affected source and according to the applicable provisions in (63.7(a)(2)) as cited in Table 10 to this subpart. You must complete an initial tune-up by following the procedures described in §63.7540(a)(10)(i) through (vi) no later than 30 days after the re-start of the affected source and, if applicable, complete the one-time energy assessment specified in Table 3 to this subpart, no later than the compliance date specified in §63.7495.

(k) For affected sources, as defined in §63.7490, that switch subcategories consistent with §63.7545(h) after the initial compliance date, you must demonstrate compliance within 60 days of the effective date of the switch, unless you had previously conducted your compliance demonstration for this subcategory within the previous 12 months.

[78 FR 7164, Jan. 31, 2013, as amended at 80 FR 72808, Nov. 20, 2015; 87 FR 60841, Oct. 6, 2022]

§63.7515 When must I conduct subsequent performance tests, fuel analyses, or tune-ups?

(a) You must conduct all applicable performance tests according to §63.7520 on an annual basis, except as specified in paragraphs (b) through (e), (g), and (h) of this section. Annual performance tests must be completed no more than 13 months after the previous performance test, except as specified in paragraphs (b) through (e), (g), and (h) of this section.

(b) If your performance tests for a given pollutant for at least 2 consecutive years show that your emissions are at or below 75 percent of the emission limit (or, in limited instances as specified in Tables 1 and 2 or 11 through 15 to this subpart, at or below the emission limit) for the pollutant,

and if there are no changes in the operation of the individual boiler or process heater or air pollution control equipment that could increase emissions, you may choose to conduct performance tests for the pollutant every third year. Each such performance test must be conducted no more than 37 months after the previous performance test. If you elect to demonstrate compliance using emission averaging under §63.7522, you must continue to conduct performance tests annually. The requirement to test at maximum chloride input level is waived unless the stack test is conducted for HCl. The requirement to test at maximum mercury input level is waived unless the stack test is conducted for mercury. The requirement to test at maximum TSM input level is waived unless the stack test is conducted for TSM.

(c) If a performance test shows emissions exceeded the emission limit or 75 percent of the emission limit (as specified in Tables 1 and 2 or 11 through 15 to this subpart) for a pollutant, you must conduct annual performance tests for that pollutant until all performance tests over a consecutive 2-year period meet the required level (at or below 75 percent of the emission limit, as specified in Tables 1 and 2 or 11 through 15).

(d) If you are required to meet an applicable tune-up work practice standard, you must conduct an annual, biennial, or 5-year performance tune-up according to §63.7540(a)(10), (11), or (12), respectively. Each annual tune-up specified in §63.7540(a)(10) must be no more than 13 months after the previous tune-up. Each biennial tune-up specified in §63.7540(a)(11) must be conducted no more than 25 months after the previous tune-up. Each 5-year tuneup specified in $\S63.7540(a)(12)$ must be conducted no more than 61 months after the previous tune-up. For a new or reconstructed affected source (as defined in §63.7490), the first annual, biennial, or 5-year tune-up must be no later than 13 months, 25 months, or 61 months, respectively, after April 1, 2013 or the initial startup of the new or reconstructed affected source, whichever is later.

(e) If you demonstrate compliance with the mercury, HCl, or TSM based

§63.7515

on fuel analysis, you must conduct a monthly fuel analysis according to §63.7521 for each type of fuel burned that is subject to an emission limit in Table 1 or 2 or Tables 11 through 15 to this subpart. You may comply with this monthly requirement by completing the fuel analysis any time within the calendar month as long as the analysis is separated from the previous analysis by at least 14 calendar days. If you burn a new type of fuel, you must conduct a fuel analysis before burning the new type of fuel in your boiler or process heater. You must still meet all applicable continuous compliance requirements in §63.7540. If each of 12 consecutive monthly fuel analyses demonstrates 75 percent or less of the compliance level, you may decrease the fuel analysis frequency to quarterly for that fuel. If any quarterly sample exceeds 75 percent of the compliance level or you begin burning a new type of fuel, you must return to monthly monitoring for that fuel, until 12 months of fuel analyses are again less than 75 percent of the compliance level. If sampling is conducted on 1 day per month, samples should be no less than 14 days apart, but if multiple samples are taken per month, the 14-day restriction does not apply

(f) You must report the results of performance tests and the associated fuel analyses within 60 days after the completion of the performance tests. This report must also verify that the operating limits for each boiler or process heater have not changed or provide documentation of revised operating limits established according to § 63.7530 and Table 7 to this subpart, as applicable. The reports for all subsequent performance tests must include all applicable information required in § 63.7550.

(g) For affected sources (as defined in §63.7490) that have not operated since the previous compliance demonstration and more than 1 year has passed since the previous compliance demonstration, you must complete the subsequent compliance demonstration, if subject to the emission limits in Table 1 or 2 or Tables 11 through 15 to this subpart, no later than 180 days after the re-start of the affected source and

40 CFR Ch. I (7–1–23 Edition)

according to the applicable provisions in §63.7(a)(2) as cited in Table 10 to this subpart. You must complete a subsequent tune-up by following the procedures described in §63.7540(a)(10)(i) through (vi) and the schedule described in §63.7540(a)(13) for units that are not operating at the time of their scheduled tune-up.

§63.7520

(h) If your affected boiler or process heater is in the unit designed to burn light liquid subcategory and you combust ultra-low sulfur liquid fuel, you do not need to conduct further performance tests (stack tests or fuel analyses) if the pollutants measured during the initial compliance performance tests meet the emission limits in Tables 1 or 2 of this subpart providing you demonstrate ongoing compliance with the emissions limits by monitoring and recording the type of fuel combusted on a monthly basis. If you intend to use a fuel other than ultra-low sulfur liquid fuel, natural gas, refinery gas, or other gas 1 fuel, you must conduct new performance tests within 60 days of burning the new fuel type.

(i) If you operate a CO CEMS that meets the Performance Specifications outlined in §63.7525(a)(3) to demonstrate compliance with the applicable alternative CO CEMS emission standard listed in Table 1 or 2 or Tables 11 through 15 to this subpart, you are not required to conduct CO performance tests and are not subject to the oxygen concentration operating limit requirement specified in §63.7510(a).

[78 FR 7165, Jan. 31, 2013, as amended at 80 FR 72808, Nov. 20, 2015; 87 FR 60842, Oct. 6, 2022]

§63.7520 What stack tests and procedures must I use?

(a) You must conduct all performance tests according to $\S63.7(c)$, (d), (f), and (h). You must also develop a sitespecific stack test plan according to the requirements in $\S63.7(c)$. You shall conduct all performance tests under such conditions as the Administrator specifies to you based on the representative performance of each boiler or process heater for the period being tested. Upon request, you shall make available to the Administrator such records as may be necessary to determine the conditions of the performance tests.

(b) You must conduct each performance test according to the requirements in Table 5 to this subpart.

(c) You must conduct each performance test under the specific conditions listed in Tables 5 and 7 to this subpart. You must conduct performance tests at representative operating load conditions while burning the type of fuel or mixture of fuels that has the highest content of chlorine and mercury, and TSM if you are opting to comply with the TSM alternative standard and you must demonstrate initial compliance and establish your operating limits based on these performance tests. These requirements could result in the need to conduct more than one performance test. Following each performance test and until the next performance test, you must comply with the operating limit for operating load conditions specified in Table 4 to this subpart.

(d) You must conduct a minimum of three separate test runs for each performance test required in this section, as specified in $\S63.7(e)(3)$. Each test run must comply with the minimum applicable sampling times or volumes specified in Tables 1 and 2 or 11 through 15 to this subpart.

(e) To determine compliance with the emission limits, you must use the F-Factor methodology and equations in sections 12.2 and 12.3 of EPA Method 19 at 40 CFR part 60, appendix A-7 of this chapter to convert the measured particulate matter (PM) concentrations, the measured HCl concentrations, the measured HCl concentrations, and the measured TSM concentrations that result from the performance test to pounds per million Btu heat input emission rates.

(f) Except for a 30-day rolling average based on CEMS (or sorbent trap monitoring system) data, if measurement results for any pollutant are reported as below the method detection level (e.g., laboratory analytical results for one or more sample components are below the method defined analytical detection level), you must use the method detection level as the measured emissions level for that pollutant

in calculating compliance. The measured result for a multiple component analysis (e.g., analytical values for multiple Method 29 fractions both for individual HAP metals and for total HAP metals) may include a combination of method detection level data and analytical data reported above the method detection level.

[76 FR 15664, Mar. 21, 2011, as amended at 78 FR 7166, Jan. 31, 2013; 87 FR 60842, Oct. 6, 2022]

§63.7521 What fuel analyses, fuel specification, and procedures must I use?

(a) For solid and liquid fuels, you must conduct fuel analyses for chloride and mercury according to the procedures in paragraphs (b) through (e) of this section and Table 6 to this subpart, as applicable. For solid fuels and liquid fuels, you must also conduct fuel analyses for TSM if you are opting to comply with the TSM alternative standard. For gas 2 (other) fuels, you must conduct fuel analyses for mercury according to the procedures in paragraphs (b) through (e) of this section and Table 6 to this subpart, as applicable. For gaseous fuels, you may not use fuel analyses to comply with the TSM alternative standard or the HCl standard. For purposes of complying with this section, a fuel gas system that consists of multiple gaseous fuels collected and mixed with each other is considered a single fuel type and sampling and analysis is only required on the combined fuel gas system that will feed the boiler or process heater. Sampling and analysis of the individual gaseous streams prior to combining is not required. You are not required to conduct fuel analyses for fuels used for only startup, unit shutdown, and transient flame stability purposes. You are required to conduct fuel analyses only for fuels and units that are subject to emission limits for mercury, HCl, or TSM in Tables 1 and 2 or 11 through 15 to this subpart. Gaseous and liquid fuels are exempt from the sampling requirements in paragraphs (c) and (d) of this section.

(b) You must develop a site-specific fuel monitoring plan according to the following procedures and requirements in paragraphs (b)(1) and (2) of this sec-

tion, if you are required to conduct fuel analyses as specified in §63.7510.

(1) If you intend to use an alternative analytical method other than those required by Table 6 to this subpart, you must submit the fuel analysis plan to the Administrator for review and approval no later than 60 days before the date that you intend to conduct the initial compliance demonstration described in §63.7510.

(2) You must include the information contained in paragraphs (b)(2)(i) through (vi) of this section in your fuel analysis plan.

(i) The identification of all fuel types anticipated to be burned in each boiler or process heater.

(ii) For each anticipated fuel type, the notification of whether you or a fuel supplier will be conducting the fuel analysis.

(iii) For each anticipated fuel type, a detailed description of the sample location and specific procedures to be used for collecting and preparing the composite samples if your procedures are different from paragraph (c) or (d) of this section. Samples should be collected at a location that most accurately represents the fuel type, where possible, at a point prior to mixing with other dissimilar fuel types.

(iv) For each anticipated fuel type, the analytical methods from Table 6, with the expected minimum detection levels, to be used for the measurement of chlorine or mercury.

(v) If you request to use an alternative analytical method other than those required by Table 6 to this subpart, you must also include a detailed description of the methods and procedures that you are proposing to use. Methods in Table 6 shall be used until the requested alternative is approved.

(vi) If you will be using fuel analysis from a fuel supplier in lieu of site-specific sampling and analysis, the fuel supplier must use the analytical methods required by Table 6 to this subpart.

(c) You must obtain composite fuel samples for each fuel type according to the procedures in paragraph (c)(1) or (2)of this section, or the methods listed in Table 6 to this subpart, or use an automated sampling mechanism that provides representative composite fuel samples for each fuel type that includes both coarse and fine material. At a minimum, for demonstrating initial compliance by fuel analysis, you must obtain three composite samples. For monthly fuel analyses, at a minimum, you must obtain a single composite sample. For fuel analyses as part of a performance stack test, as specified in §63.7510(a), you must obtain a composite fuel sample during each performance test run.

(1) If sampling from a belt (or screw) feeder, collect fuel samples according to paragraphs (c)(1)(i) and (ii) of this section.

(i) Stop the belt and withdraw a 6inch wide sample from the full crosssection of the stopped belt to obtain a minimum two pounds of sample. You must collect all the material (fines and coarse) in the full cross-section. You must transfer the sample to a clean plastic bag.

(ii) Each composite sample will consist of a minimum of three samples collected at approximately equal intervals during the testing period for sampling during performance stack testing.

(2) If sampling from a fuel pile or truck, you must collect fuel samples according to paragraphs (c)(2)(i) through (iii) of this section.

(i) For each composite sample, you must select a minimum of five sampling locations uniformly spaced over the surface of the pile.

(ii) At each sampling site, you must dig into the pile to a uniform depth of approximately 18 inches. You must insert a clean shovel into the hole and withdraw a sample, making sure that large pieces do not fall off during sampling; use the same shovel to collect all samples.

(iii) You must transfer all samples to a clean plastic bag for further processing.

(d) You must prepare each composite sample according to the procedures in paragraphs (d)(1) through (7) of this section.

(1) You must thoroughly mix and pour the entire composite sample over a clean plastic sheet.

(2) You must break large sample pieces (e.g., larger than 3 inches) into smaller sizes.

40 CFR Ch. I (7–1–23 Edition)

(3) You must make a pie shape with the entire composite sample and subdivide it into four equal parts.

(4) You must separate one of the quarter samples as the first subset.

(5) If this subset is too large for grinding, you must repeat the procedure in paragraph (d)(3) of this section with the quarter sample and obtain a one-quarter subset from this sample.

(6) You must grind the sample in a mill.

(7) You must use the procedure in paragraph (d)(3) of this section to obtain a one-quarter subsample for analysis. If the quarter sample is too large, subdivide it further using the same procedure.

(e) You must determine the concentration of pollutants in the fuel (mercury and/or chlorine and/or TSM) in units of pounds per million Btu of each composite sample for each fuel type according to the procedures in Table 6 to this subpart, for use in Equations 7, 8, and 9 of this subpart.

(f) To demonstrate that a gaseous fuel other than natural gas or refinery gas qualifies as an other gas 1 fuel, as defined in §63.7575, you must conduct a fuel specification analyses for mercury according to the procedures in paragraphs (g) through (i) of this section and Table 6 to this subpart, as applicable, except as specified in paragraph (f)(1) through (4) of this section, or as an alternative where fuel specification analysis is not practical, you must measure mercury concentration in the exhaust gas when firing only the gaseous fuel to be demonstrated as an other gas 1 fuel in the boiler or process heater according to the procedures in Table 6 to this subpart.

(1) You are not required to conduct the fuel specification analyses in paragraphs (g) through (i) of this section for natural gas or refinery gas.

(2) You are not required to conduct the fuel specification analyses in paragraphs (g) through (i) of this section for gaseous fuels that are subject to another subpart of this part, part 60, part 61, or part 65.

(3) You are not required to conduct the fuel specification analyses in paragraphs (g) through (i) of this section on

gaseous fuels for units that are complying with the limits for units designed to burn gas 2 (other) fuels.

(4) You are not required to conduct the fuel specification analyses in paragraphs (g) through (i) of this section for gas streams directly derived from natural gas at natural gas production sites or natural gas plants.

(g) You must develop a site-specific fuel analysis plan for other gas 1 fuels according to the following procedures and requirements in paragraphs (g)(1) and (2) of this section.

(1) If you intend to use an alternative analytical method other than those required by Table 6 to this subpart, you must submit the fuel analysis plan to the Administrator for review and approval no later than 60 days before the date that you intend to conduct the initial compliance demonstration described in §63.7510.

(2) You must include the information contained in paragraphs (g)(2)(i) through (vi) of this section in your fuel analysis plan.

(i) The identification of all gaseous fuel types other than those exempted from fuel specification analysis under (f)(1) through (3) of this section anticipated to be burned in each boiler or process heater.

(ii) For each anticipated fuel type, the identification of whether you or a fuel supplier will be conducting the fuel specification analysis.

(iii) For each anticipated fuel type, a detailed description of the sample location and specific procedures to be used for collecting and preparing the samples if your procedures are different from the sampling methods contained in Table 6 to this subpart. Samples should be collected at a location that most accurately represents the fuel type, where possible, at a point prior to mixing with other dissimilar fuel types. If multiple boilers or process heaters are fueled by a common fuel stream it is permissible to conduct a single gas specification at the common point of gas distribution.

(iv) For each anticipated fuel type, the analytical methods from Table 6 to this subpart, with the expected minimum detection levels, to be used for the measurement of mercury. (v) If you request to use an alternative analytical method other than those required by Table 6 to this subpart, you must also include a detailed description of the methods and procedures that you are proposing to use. Methods in Table 6 to this subpart shall be used until the requested alternative is approved.

(vi) If you will be using fuel analysis from a fuel supplier in lieu of site-specific sampling and analysis, the fuel supplier must use the analytical methods required by Table 6 to this subpart. When using a fuel supplier's fuel analysis, the owner or operator is not required to submit the information in §63.7521(g)(2)(iii).

(h) You must obtain a single fuel sample for each fuel type for fuel specification of gaseous fuels.

(i) You must determine the concentration in the fuel of mercury, in units of microgram per cubic meter, dry basis, of each sample for each other gas 1 fuel type according to the procedures in Table 6 to this subpart.

[78 FR 7167, Jan. 31, 2013, as amended at 80 FR 72808, Nov. 20, 2015; 87 FR 60842, Oct. 6, 2022]

§63.7522 Can I use emissions averaging to comply with this subpart?

(a) As an alternative to meeting the requirements of §63.7500 for PM (or TSM), HCl, or mercury on a boiler or process heater-specific basis, if you have more than one existing boiler or process heater in any subcategories located at your facility, you may demonstrate compliance by emissions averaging, if your averaged emissions are not more than 90 percent of the applicable emission limit, according to the procedures in this section. You may not include new boilers or process heaters in an emissions average.

(b) For a group of two or more existing boilers or process heaters in the same subcategory that each vent to a separate stack, you may average PM (or TSM), HCl, or mercury emissions among existing units to demonstrate compliance with the limits in Table 2 or 15 to this subpart as specified in paragraphs (b)(1) through (3) of this section, if you satisfy the requirements in paragraphs (c) through (g) of this section.

§63.7522

(1) You may average units using a CEMS or PM CPMS for demonstrating compliance.

(2) For mercury and HCl, averaging is allowed as follows:

(i) You may average among units in any of the solid fuel subcategories.

(ii) You may average among units in any of the liquid fuel subcategories.

(iii) You may average among units in a subcategory of units designed to burn gas 2 (other) fuels.

(iv) You may not average across the units designed to burn liquid, units designed to burn solid fuel, and units designed to burn gas 2 (other) subcategories.

(3) For PM (or TSM), averaging is only allowed between units within each of the following subcategories and you may not average across subcategories:

(i) Units designed to burn coal/solid fossil fuel.

(ii) Stokers/sloped grate/other units designed to burn kiln dried biomass/ bio-based solids.

(iii) Stokers/sloped grate/other units designed to burn wet biomass/bio-based solids.

(iv) Fluidized bed units designed to burn biomass/bio-based solid.

(v) Suspension burners designed to burn biomass/bio-based solid.

(vi) Dutch ovens/pile burners designed to burn biomass/bio-based solid.

(vii) Fuel Cells designed to burn biomass/bio-based solid.

(viii) Hybrid suspension/grate burners designed to burn wet biomass/bio-based solid.

(ix) Units designed to burn heavy liquid fuel.

 $\left(x\right)$ Units designed to burn light liquid fuel.

(xi) Units designed to burn liquid fuel that are non-continental units.

Equation 1a to paragraph (e)(1)

40 CFR Ch. I (7–1–23 Edition)

(xii) Units designed to burn gas 2 (other) gases.

(c) For each existing boiler or process heater in the averaging group, the emission rate achieved during the initial compliance test for the HAP being averaged must not exceed the emission level that was being achieved on April 1, 2013 or the control technology employed during the initial compliance test must not be less effective for the HAP being averaged than the control technology employed on April 1, 2013.

(d) The averaged emissions rate from the existing boilers and process heaters participating in the emissions averaging option must not exceed 90 percent of the limits in Table 2 or 15 to this subpart at all times the affected units are subject to numeric emission limits following the compliance date specified in §63.7495.

(e) You must demonstrate initial compliance according to paragraph (e)(1) or (2) of this section using the maximum rated heat input capacity or maximum steam generation capacity of each unit and the results of the initial performance tests or fuel analysis.

(1) You must use Equation 1a or 1b or 1c to this paragraph (e)(1) to demonstrate that the PM (or TSM), HCl, or mercury emissions from all existing units participating in the emissions averaging option for that pollutant do not exceed the emission limits in Table 2 or 15 to this subpart. Use Equation 1a if you are complying with the emission limits on a heat input basis, use Equation 1b if you are complying with the emission limits on a steam generation (output) basis, and use Equation 1c if you are complying with the emission limits on a electric generation (output) basis.

AveWeightedEmissions = $1.1 \times \sum_{r=1}^{n} (Er \times Hm) \div \sum_{r=1}^{n} Hm$ (Eq. 1a)

Where:

AveWeightedEmissions = Average weighted emissions for PM (or TSM), HCl, or mercury, in units of pounds per million Btu of heat input.

Er = Emission rate (as determined during the initial compliance demonstration) of PM

(or TSM), HCl, or mercury from unit, i, in units of pounds per million Btu of heat input. Determine the emission rate for PM (or TSM), HCl, or mercury by performance testing according to Table 5 to this subpart, or by fuel analysis for HCl §63.7522

or mercury or TSM using the applicable equation in §63.7530(c).

- Hm = Maximum rated heat input capacity of unit, i, in units of million Btu per hour.
- n = Number of units participating in the emissions averaging option.

1.1 =Required discount factor.

Equation 1b to paragraph (e)(1)

AveWeightedEmissions =
$$1.1 \times \sum_{r=1}^{n} (Er \times So) \div \sum_{r=1}^{n} So$$
 (Eq. 1b)

Where:

- AveWeightedEmissions = Average weighted emissions for PM (or TSM), HCl, or mercury, in units of pounds per million Btu of steam output.
- Er = Emission rate (as determined during the initial compliance demonstration) of PM (or TSM), HCl, or mercury from unit, i, in units of pounds per million Btu of steam output. Determine the emission rate for PM (or TSM), HCl, or mercury by performance testing according to Table 5 to this subpart, or by fuel anal-

ysis for HCl or mercury or TSM using the applicable equation in $\S63.7530(c)$. If you are taking credit for energy conservation measures from a unit according to $\S63.7533$, use the adjusted emission level for that unit, Eadj, determined according to $\S63.7533$ for that unit.

- So = Maximum steam output capacity of unit, i, in units of million Btu per hour, as defined in §63.7575.
- n = Number of units participating in the emissions averaging option.
- 1.1 = Required discount factor.

Equation 1c to paragraph (e)(1)

AveWeightedEmissions =
$$1.1 \times \sum_{r=1}^{\infty} (Er \times Eo) \div \sum_{r=1}^{\infty} Eo$$
 (Eq. 1c)

Where:

- AveWeightedEmissions = Average weighted emissions for PM (or TSM), HCl, or mercury, in units of pounds per megawatt hour.
- Er = Emission rate (as determined during the initial compliance demonstration) of PM (or TSM), HCl, or mercury from unit, i, in units of pounds per megawatt hour. Determine the emission rate for PM (or TSM), HCl, or mercury by performance testing according to Table 5 to this subpart, or by fuel analysis for HCl or mercury or TSM using the applicable equation in §63.7530(c). If you are taking credit for energy conservation measures from a unit according to §63.7533, use the adjusted emission level for that unit, Eadj, determined according to §63.7533 for that unit.
- Eo = Maximum electric generating output capacity of unit, i, in units of megawatt hour, as defined in §63.7575.
- n = Number of units participating in the emissions averaging option.

1.1 = Required discount factor.

(2) If you are not capable of determining the maximum rated heat input capacity of one or more boilers that generate steam, you may use Equation 2 to this paragraph (e)(2) as an alternative to using Equation 1a of paragraph (e)(1) of this section to demonstrate that the PM (or TSM), HCl, or mercury emissions from all existing units participating in the emissions averaging option do not exceed the emission limits for that pollutant in Table 2 or 15 to this subpart that are in pounds per million Btu of heat input.

§63.7522

40 CFR Ch. I (7-1-23 Edition)

Equation 2 to paragraph (e)(2)

Ave Weight additional scalar
$$1.1 \times \sum_{i=1}^{n} \{Er \times Sin \times C_i^n\} \times \sum_{i=1}^{n} \{Sin \times C_i^n\} = \{Eq. -2\}$$

Where:

- AveWeightedEmissions = Average weighted emission level for PM (or TSM), HCl, or mercury, in units of pounds per million Btu of heat input.
- Er = Emission rate (as determined during the most recent compliance demonstration) of PM (or TSM), HCl, or mercury from unit, i, in units of pounds per million Btu of heat input. Determine the emission rate for PM (or TSM), HCl, or mercury by performance testing according to Table 5 to this subpart, or by fuel analysis for HCl or mercury or TSM using the applicable equation in §63.7530(c).
- Sm = Maximum steam generation capacity by unit, i, in units of pounds per hour.
- *Cfi* = Conversion factor, calculated from the most recent compliance test, in units of million Btu of heat input per pounds of steam generated for unit, i.
- 1.1 =Required discount factor.

(f) After the initial compliance demonstration described in paragraph (e) of this section, you must demonstrate compliance on a monthly basis determined at the end of every month (12 times per year) according to paragraphs (f)(1) through (3) of this section. The first monthly period begins on the compliance date specified in §63.7495. If the affected source elects to collect monthly data for up the 11 months preceding the first monthly period, these additional data points can be used to compute the 12-month rolling average in paragraph (f)(3) of this section.

(1) For each calendar month, you must use Equation 3a or 3b or 3c of this section to calculate the average weighted emission rate for that month. Use Equation 3a and the actual heat input for the month for each existing unit participating in the emissions averaging option if you are complying with emission limits on a heat input basis. Use Equation 3b and the actual steam generation for the month if you are complying with the emission limits on a steam generation (output) basis. Use Equation 3c and the actual electrical generation for the month if you are complying with the emission limits on an electrical generation (output) basis.

AveWeightedEmissions =
$$1.1 \times \sum_{i=1}^{n} (Er \times Hb) \div \sum_{i=1}^{n} Hb$$
 (Eq. 3a)

Where:

- AveWeightedEmissions = Average weighted emission level for PM (or TSM), HCl, or mercury, in units of pounds per million Btu of heat input, for that calendar month.
- Er = Emission rate (as determined during the most recent compliance demonstration) of PM (or TSM), HCl, or mercury from unit, i, in units of pounds per million Btu of heat input. Determine the emission

rate for PM (or TSM), HCl, or mercury by performance testing according to Table 5 to this subpart, or by fuel analysis for HCl or mercury or TSM according to Table 6 to this subpart.

- Hb = The heat input for that calendar month to unit, i, in units of million Btu.
- n = Number of units participating in the emissions averaging option.

1.1 =Required discount factor.

AveWeightedEmissions =
$$1.1 \times \sum_{i=1}^{n} (Er \times So) \div \sum_{i=1}^{n} So$$
 (Eq. 3b)
Where:

- AveWeightedEmissions = Average weighted emission level for PM (or TSM), HCl, or mercury, in units of pounds per million Btu of steam output, for that calendar month.
- Er = Emission rate (as determined during the most recent compliance demonstration) of PM (or TSM), HCl, or mercury from unit, i, in units of pounds per million Btu of steam output. Determine the emission rate for PM (or TSM), HCl, or mercury by performance testing according to Table 5 to this subpart, or by fuel anal-

ysis for HCl or mercury or TSM according to Table 6 to this subpart. If you are taking credit for energy conservation measures from a unit according to $\S63.7533$, use the adjusted emission level for that unit, E_{adj} , determined according to $\S63.7533$ for that unit.

- So = The steam output for that calendar month from unit, i, in units of million Btu, as defined in 63.7575.
- n = Number of units participating in the emissions averaging option.
- 1.1 = Required discount factor.

AveWeightedEmissions =
$$1.1 \times \sum_{i=1}^{n} (Er \times Eo) \div \sum_{i=1}^{n} Eo$$
 (Eq. 3c)

Where:

- AveWeightedEmissions = Average weighted emission level for PM (or TSM), HCl, or mercury, in units of pounds per megawatt hour, for that calendar month.
- Er = Emission rate (as determined during the most recent compliance demonstration) of PM (or TSM), HCl, or mercury from unit, i, in units of pounds per megawatt hour. Determine the emission rate for PM (or TSM), HCl, or mercury by performance testing according to Table 5 to this subpart, or by fuel analysis for HCl or mercury or TSM according to Table 6 to this subpart. If you are taking credit for energy conservation measures from a unit according to §63.7533, use the adjusted emission level for that unit, E_{adi},

determined according to 63.7533 for that unit.

- Eo = The electric generating output for that calendar month from unit, i, in units of megawatt hour, as defined in §63.7575.
- n = Number of units participating in the emissions averaging option.
- 1.1 = Required discount factor.

(2) If you are not capable of monitoring heat input, you may use Equation 4 of this section as an alternative to using Equation 3a of this section to calculate the average weighted emission rate using the actual steam generation from the boilers participating in the emissions averaging option.

AveWeightedEmissions =
$$1.1 \times \sum_{i=1}^{n} (Er \times Sa \times Cfi) \div \sum_{i=1}^{n} (Sa \times Cfi)$$
 (Eq. 4)

Where:

- AveWeightedEmissions = average weighted emission level for PM (or TSM), HCl, or mercury, in units of pounds per million Btu of heat input for that calendar month.
- Er = Emission rate (as determined during the most recent compliance demonstration of PM (or TSM), HCl, or mercury from unit, i, in units of pounds per million Btu of heat input. Determine the emission rate for PM (or TSM), HCl, or mercury by performance testing according to Table 5 to this subpart, or by fuel analysis for HCl or mercury or TSM according to Table 6 to this subpart.
- Sa = Actual steam generation for that calendar month by boiler, i, in units of pounds.
- Cfi = Conversion factor, as calculated during the most recent compliance test, in units of million Btu of heat input per pounds of steam generated for boiler, i.

1.1 = Required discount factor.

(3) Until 12 monthly weighted average emission rates have been accumulated, calculate and report only the average weighted emission rate determined under paragraph (f)(1) or (2) of this section for each calendar month. After 12 monthly weighted average emission rates have been accumulated,

§63.7522

40 CFR Ch. I (7–1–23 Edition)

for each subsequent calendar month, use Equation 5 of this section to calculate the 12-month rolling average of the monthly weighted average emission rates for the current calendar month and the previous 11 calendar months.

$$Eavg = \sum_{i=1}^{n} ERi \div 12 \quad (Eq. 5)$$

Where:

Eavg = 12-month rolling average emission rate, (pounds per million Btu heat input)

ERi = Monthly weighted average, for calendar month "i" (pounds per million Btu heat input), as calculated by paragraph (f)(1) or (2) of this section.

(g) You must develop, and submit upon request to the applicable Administrator for review and approval, an implementation plan for emission averaging according to the following procedures and requirements in paragraphs (g)(1) through (4) of this section.

(1) If requested, you must submit the implementation plan no later than 180 days before the date that the facility intends to demonstrate compliance using the emission averaging option.

(2) You must include the information contained in paragraphs (g)(2)(i) through (vii) of this section in your implementation plan for all emission sources included in an emissions average:

(i) The identification of all existing boilers and process heaters in the averaging group, including for each either the applicable HAP emission level or the control technology installed as of January 31, 2013 and the date on which you are requesting emission averaging to commence;

(ii) The process parameter (heat input or steam generated) that will be monitored for each averaging group;

(iii) The specific control technology or pollution prevention measure to be used for each emission boiler or process heater in the averaging group and the date of its installation or application. If the pollution prevention measure reduces or eliminates emissions from multiple boilers or process heaters, the owner or operator must identify each boiler or process heater;

(iv) The test plan for the measurement of PM (or TSM), HCl, or mercury $\,$

emissions in accordance with the requirements in §63.7520;

(v) The operating parameters to be monitored for each control system or device consistent with §63.7500 and Table 4, and a description of how the operating limits will be determined;

(vi) If you request to monitor an alternative operating parameter pursuant to §63.7525, you must also include:

(A) A description of the parameter(s) to be monitored and an explanation of the criteria used to select the parameter(s); and

(B) A description of the methods and procedures that will be used to demonstrate that the parameter indicates proper operation of the control device; the frequency and content of monitoring, reporting, and recordkeeping requirements; and a demonstration, to the satisfaction of the Administrator, that the proposed monitoring frequency is sufficient to represent control device operating conditions; and

(vii) A demonstration that compliance with each of the applicable emission limit(s) will be achieved under representative operating load conditions. Following each compliance demonstration and until the next compliance demonstration, you must comply with the operating limit for operating load conditions specified in Table 4 to this subpart.

(3) If submitted upon request, the Administrator shall review and approve or disapprove the plan according to the following criteria:

(i) Whether the content of the plan includes all of the information specified in paragraph (g)(2) of this section; and

(ii) Whether the plan presents sufficient information to determine that compliance will be achieved and maintained.

(4) The applicable Administrator shall not approve an emission averaging implementation plan containing any of the following provisions:

(i) Any averaging between emissions of differing pollutants or between differing sources; or

(ii) The inclusion of any emission source other than an existing unit in the same subcategories.

(h) For a group of two or more existing affected units, each of which vents through a single common stack, you may average PM (or TSM), HCl, or mercury emissions to demonstrate compliance with the limits for that pollutant in Table 2 or 15 to this subpart if you satisfy the requirements in paragraph (i) or (j) of this section.

(i) For a group of two or more existing units in the same subcategory, each of which vents through a common emissions control system to a common

Equation 6 to paragraph (j)(1)

stack, that does not receive emissions from units in other subcategories or categories, you may treat such averaging group as a single existing unit for purposes of this subpart and comply with the requirements of this subpart as if the group were a single unit.

(j) For all other groups of units subject to the common stack requirements of paragraph (h) of this section, including situations where the exhaust of affected units are each individually controlled and then sent to a common stack, the owner or operator may elect to:

(1) Conduct performance tests according to procedures specified in $\S63.7520$ in the common stack if affected units from other subcategories vent to the common stack. The emission limits that the group must comply with are determined by the use of Equation 6 to this paragraph (j)(1).

$$En = \sum_{i=1}^{n} (ELi \times Hi) \div \sum_{i=1}^{n} Hi \qquad (Eq. -6)$$

Where:

- En = HAP emission limit, pounds per million British thermal units (lb/MMBtu) or parts per million (ppm).
- *ELi* = Appropriate emission limit from Table 2 or 15 to this subpart for unit i, in units of lb/MMBtu or ppm.

Hi = Heat input from unit i, MMBtu.

(2) Conduct performance tests according to procedures specified in §63.7520 in the common stack. If affected units and non-affected units vent to the common stack, the non-affected units must be shut down or vented to a different stack during the performance test unless the facility determines to demonstrate compliance with the non-affected units venting to the stack; and

(3) Meet the applicable operating limit specified in $\S63.7540$ and Table 8 to this subpart for each emissions control system (except that, if each unit venting to the common stack has an applicable opacity operating limit, then a single continuous opacity monitoring system may be located in the

common stack instead of in each duct to the common stack).

(k) The common stack of a group of two or more existing boilers or process heaters in the same subcategories subject to paragraph (h) of this section may be treated as a separate stack for purposes of paragraph (b) of this section and included in an emissions averaging group subject to paragraph (b) of this section.

[76 FR 15664, Mar. 21, 2011, as amended at 78
FR 7168, Jan. 31, 2013; 80 FR 72809, Nov. 20, 2015; 87 FR 60843, Oct. 6, 2022]

§63.7525 What are my monitoring, installation, operation, and maintenance requirements?

(a) If your boiler or process heater is subject to a CO emission limit in Table 1 or 2 or Tables 11 through 15 to this subpart, you must install, operate, and maintain an oxygen analyzer system, as defined in §63.7575, or install, certify, operate and maintain continuous emission monitoring systems for CO and oxygen (O_2) (or carbon dioxide (CO_2)) according to the procedures in paragraphs (a)(1) through (6) of this section.

(1) Install the CO CEMS including an O_2 (or CO_2) analyzer by the compliance date specified in §63.7495. The CO and O_2 (or CO_2) levels shall be monitored at the same location at the outlet of the boiler or process heater. An owner or operator may determine compliance with the CO emissions limit using a CO_2 analyzer as the diluent monitor. If a CO_2 analyzer is used as the diluent monitor, EPA Method 19 F-factors in 40 CFR part 60, appendix A-7, for the fuel type(s) being burned in the unit and EPA Method 19 equations in 40 CFR part 60, appendix A-7, must be used to calculate the emissions corrected to 3 percent O_2 using the measured CO_2 percentage, and must also take into account that the 3 percent oxygen correction is to be done on a dry basis. The equations used to calculate the emissions, must also account for any CO₂ being added to, or removed from, the emissions gas stream as a result of limestone injection, scrubber media, etc. The methodology used to calculate the CO emissions and the methodology used to account for any CO_2 being added to, or removed from the emissions gas stream shall be detailed and approved in the site-specific monitoring plan developed according to §63.7505(d).

(2) To demonstrate compliance with the applicable alternative CO CEMS emission standard listed in Table 1 or 2 or Tables 11 through 15 to this subpart, you must install, certify, operate, and maintain a CO CEMS and an oxygen analyzer according to the applicable procedures under Performance Specification 4, 4A, or 4B at 40 CFR part 60, appendix B; part 75 of this chapter (if an CO_2 analyzer is used); the site-specific monitoring plan developed according to §63.7505(d); and the requirements in (63.7540(a)(8)) and this paragraph (a). Any boiler or process heater that has a CO CEMS that is compliant with Performance Specification 4, 4A, or 4B at 40 CFR part 60, appendix B, a site-specific monitoring plan developed according to §63.7505(d), and the requirements in (63.7540(a)(8)) and this paragraph (a) must use the CO CEMS to comply with

40 CFR Ch. I (7–1–23 Edition)

the applicable alternative CO CEMS emission standard listed in Table 1 or 2 or Tables 11 through 15 to this subpart.

(i) You must conduct a performance evaluation of each CO CEMS according to the requirements in §63.8(e) and according to Performance Specification 4, 4A, or 4B at 40 CFR part 60, appendix B.

(ii) During each relative accuracy test run of the CO CEMS, you must collect emission data for CO concurrently using both the CO CEMS and Method 10, 10A, or 10B at 40 CFR part 60, appendix A-4. The relative accuracy testing must be conducted at representative operating conditions.

(iii) You must follow the quality assurance procedures (e.g., quarterly accuracy determinations and daily calibration drift tests) of Procedure 1 of appendix F to part 60. The measurement span value of the CO CEMS must be two times the applicable CO emission limit, expressed as a concentration.

(iv) Any CO CEMS that does not comply with this paragraph (a) cannot be used to meet any requirement in this subpart to demonstrate compliance with a CO emission limit listed in Table 1 or 2 or Tables 11 through 15 to this subpart.

(v) For a new unit, complete the initial performance evaluation no later than July 30, 2013, or 180 days after the date of initial startup, whichever is later. For an existing unit, complete the initial performance evaluation no later than July 29, 2016.

(vi) When CO_2 is used to correct COemissions and CO₂ is measured on a wet basis, if needed, correct for moisture as follows: Install, operate, maintain, and quality assure a continuous moisture monitoring system for measuring and recording the moisture content of the flue gases, in order to correct the measured hourly volumetric flow rates for moisture when calculating CO concentrations. The following continuous moisture monitoring systems are acceptable: a continuous moisture sensor; an oxygen analyzer (or analyzers) capable of measuring O_2 both on a wet basis and on a dry basis; or a stack temperature sensor and a moisture look-up table, i.e., a psychrometric chart (for saturated gas streams following wet

scrubbers or other demonstrably saturated gas streams, only). The moisture monitoring system shall include as a component the automated data acquisition and handling system (DAHS) for recording and reporting both the raw data (e.g., hourly average wet-and drybasis O_2 values) and the hourly average values of the stack gas moisture content derived from those data. When a moisture look-up table is used, the moisture monitoring system shall be represented as a single component, the certified DAHS, in the monitoring plan for the unit or common stack.

(3) Complete a minimum of one cycle of CO and oxygen (or CO_2) CEMS operation (sampling, analyzing, and data recording) for each successive 15minute period. Collect CO and oxygen (or CO_2) data concurrently. Collect at least four CO and oxygen (or CO_2) CEMS data values representing the four 15-minute periods in an hour, or at least two 15-minute data values during an hour when CEMS calibration, quality assurance, or maintenance activities are being performed.

(4) Reduce the CO CEMS data as specified in §63.8(g)(2).

(5) Calculate one-hour arithmetic averages, corrected to 3 percent oxygen (or corrected to an CO_2 percentage determined to be equivalent to 3 percent oxygen) from each hour of CO CEMS data in parts per million CO concentration. The one-hour arithmetic averages required shall be used to calculate the 30-day or 10-day rolling average emissions. Use Equation 19–19 in section 12.4.1 of Method 19 of 40 CFR part 60, appendix A-7 for calculating the average CO concentration from the hourly values.

(6) For purposes of collecting CO data, operate the CO CEMS as specified in 63.7535(b). You must use all the data collected during all periods in calculating data averages and assessing compliance, except that you must exclude certain data as specified in 63.7535(c). Periods when CO data are unavailable may constitute monitoring deviations as specified in 63.7535(d).

(7) Operate an oxygen trim system with the oxygen level set no lower than the lowest hourly average oxygen concentration measured during the most recent CO performance test as the operating limit for oxygen according to Table 7 to this subpart.

(b) If your boiler or process heater is in the unit designed to burn coal/solid fossil fuel subcategory or the unit designed to burn heavy liquid subcategory and has an average annual heat input rate greater than 250 MMBtu per hour from solid fossil fuel and/or heavy liquid, and you demonstrate compliance with the PM limit instead of the alternative TSM limit, you must install, maintain, and operate a PM CPMS monitoring emissions discharged to the atmosphere and record the output of the system as specified in paragraphs (b)(1) through (4) of this section. As an alternative to use of a PM CPMS to demonstrate compliance with the PM limit, you may choose to use a PM CEMS. If you choose to use a PM CEMS to demonstrate compliance with the PM limit instead of the alternative TSM limit, you must install, certify, maintain, and operate a PM CEMS monitoring emissions discharged to the atmosphere and record the output of the system as specified in paragraph (b)(5) through (8) of this section. For other boilers or process heaters, you may elect to use a PM CPMS or PM CEMS operated in accordance with this section in lieu of using other CMS for monitoring PM compliance (e.g., bag leak detectors, ESP secondary power, and PM scrubber pressure). Owners of boilers and process heaters who elect to comply with the alternative TSM limit are not required to install a PM CPMS.

(1) Install, operate, and maintain your PM CPMS according to the procedures in your approved site-specific monitoring plan developed in accordance with 63.7505(d), the requirements in 63.7540(a)(9), and paragraphs (b)(1)(i) through (iii) of this section.

(i) The operating principle of the PM CPMS must be based on in-stack or extractive light scatter, light scintillation, beta attenuation, or mass accumulation detection of PM in the exhaust gas or representative exhaust gas sample. The reportable measurement output from the PM CPMS must be expressed as milliamps.

(ii) The PM CPMS must have a cycle time (i.e., period required to complete sampling, measurement, and reporting for each measurement) no longer than 60 minutes.

(iii) The PM CPMS must have a documented detection limit of 0.5 milligram per actual cubic meter, or less.

(2) For a new unit, complete the initial performance evaluation no later than July 30, 2013, or 180 days after the date of initial startup, whichever is later. For an existing unit, complete the initial performance evaluation no later than July 29, 2016.

(3) Collect PM CPMS hourly average output data for all boiler or process heater operating hours except as indicated in §63.7535(a) through (d). Express the PM CPMS output as milliamps.

(4) Calculate the arithmetic 30-day rolling average of all of the hourly average PM CPMS output data collected during all boiler or process heater operating hours (milliamps).

(5) Install, certify, operate, and maintain your PM CEMS according to the procedures in your approved site-specific monitoring plan developed in accordance with §63.7505(d), the requirements in §63.7540(a)(9), and paragraphs (b)(5)(i) through (iv) of this section.

(i) You shall conduct a performance evaluation of the PM CEMS according to the applicable requirements of §60.8(e), and Performance Specification 11 at 40 CFR part 60, appendix B of this chapter.

(ii) During each PM correlation testing run of the CEMS required by Performance Specification 11 at 40 CFR part 60, appendix B of this chapter, you shall collect PM and oxygen (or carbon dioxide) data concurrently (or within a 30-to 60-minute period) by both the CEMS and conducting performance tests using Method 5 at 40 CFR part 60, appendix A-3 or Method 17 at 40 CFR part 60, appendix A-6 of this chapter.

(iii) You shall perform quarterly accuracy determinations and daily calibration drift tests in accordance with Procedure 2 at 40 CFR part 60, appendix F of this chapter. You must perform Relative Response Audits annually and perform Response Correlation Audits every 3 years.

(iv) Within 60 days after the date of completing each CEMS relative accuracy test audit or performance test conducted to demonstrate compliance 40 CFR Ch. I (7–1–23 Edition)

with this subpart, you must submit the relative accuracy test audit data and performance test data to the EPA by successfully submitting the data electronically into the EPA's Central Data Exchange by using the Electronic Reporting Tool (see http://www.epa.gov/ttn/chief/ert/erttool.html/).

(6) For a new unit, complete the initial performance evaluation no later than July 30, 2013, or 180 days after the date of initial startup, whichever is later. For an existing unit, complete the initial performance evaluation no later than July 29, 2016.

(7) Collect PM CEMS hourly average output data for all boiler or process heater operating hours except as indicated in §63.7535(a) through (d).

(8) Calculate the arithmetic 30-day rolling average of all of the hourly average PM CEMS output data collected during all boiler or process heater operating hours.

(c) If you have an applicable opacity operating limit in this rule, and are not otherwise required or elect to install and operate a PM CPMS, PM CEMS, or a bag leak detection system, you must install, operate, certify and maintain each COMS according to the procedures in paragraphs (c)(1) through (7) of this section by the compliance date specified in §63.7495.

(1) Each COMS must be installed, operated, and maintained according to Performance Specification 1 at appendix B to part 60 of this chapter.

(2) You must conduct a performance evaluation of each COMS according to the requirements in §63.8(e) and according to Performance Specification 1 at appendix B to part 60 of this chapter.

(3) As specified in 63.8(c)(4)(i), each COMS must complete a minimum of one cycle of sampling and analyzing for each successive 10-second period and one cycle of data recording for each successive 6-minute period.

(4) The COMS data must be reduced as specified in 63.8(g)(2).

(5) You must include in your site-specific monitoring plan procedures and acceptance criteria for operating and maintaining each COMS according to

the requirements in §63.8(d). At a minimum, the monitoring plan must include a daily calibration drift assessment, a quarterly performance audit, and an annual zero alignment audit of each COMS.

(6) You must operate and maintain each COMS according to the requirements in the monitoring plan and the requirements of §63.8(e). You must identify periods the COMS is out of control including any periods that the COMS fails to pass a daily calibration drift assessment, a quarterly performance audit, or an annual zero alignment audit. Any 6-minute period for which the monitoring system is out of control and data are not available for a required calculation constitutes a deviation from the monitoring requirements.

(7) You must determine and record all the 6-minute averages (and daily block averages as applicable) collected for periods during which the COMS is not out of control.

(d) If you have an operating limit that requires the use of a CMS other than a PM CPMS or COMS, you must install, operate, and maintain each CMS according to the procedures in paragraphs (d)(1) through (5) of this section by the compliance date specified in §63.7495.

(1) The CPMS must complete a minimum of one cycle of operation every 15-minutes. You must have a minimum of four successive cycles of operation, one representing each of the four 15minute periods in an hour, to have a valid hour of data.

(2) You must operate the monitoring system as specified in §63.7535(b), and comply with the data calculation requirements specified in §63.7535(c).

(3) Any 15-minute period for which the monitoring system is out-of-control and data are not available for a required calculation constitutes a deviation from the monitoring requirements. Other situations that constitute a monitoring deviation are specified in §63.7535(d).

(4) You must determine the 30-day rolling average of all recorded readings, except as provided in §63.7535(c).

(5) You must record the results of each inspection, calibration, and validation check.

(e) If you have an operating limit that requires the use of a flow monitoring system, you must meet the requirements in paragraphs (d) and (e)(1) through (4) of this section.

(1) You must install the flow sensor and other necessary equipment in a position that provides a representative flow.

(2) You must use a flow sensor with a measurement sensitivity of no greater than 2 percent of the design flow rate.

(3) You must minimize, consistent with good engineering practices, the effects of swirling flow or abnormal velocity distributions due to upstream and downstream disturbances.

(4) You must conduct a flow monitoring system performance evaluation in accordance with your monitoring plan at the time of each performance test but no less frequently than annually.

(f) If you have an operating limit that requires the use of a pressure monitoring system, you must meet the requirements in paragraphs (d) and (f)(1) through (6) of this section.

(1) Install the pressure sensor(s) in a position that provides a representative measurement of the pressure (*e.g.*, PM scrubber pressure drop).

(2) Minimize or eliminate pulsating pressure, vibration, and internal and external corrosion consistent with good engineering practices.

(3) Use a pressure sensor with a minimum tolerance of 1.27 centimeters of water or a minimum tolerance of 1 percent of the pressure monitoring system operating range, whichever is less.

(4) Perform checks at least once each process operating day to ensure pressure measurements are not obstructed (*e.g.*, check for pressure tap pluggage daily).

(5) Conduct a performance evaluation of the pressure monitoring system in accordance with your monitoring plan at the time of each performance test but no less frequently than annually.

(6) If at any time the measured pressure exceeds the manufacturer's specified maximum operating pressure range, conduct a performance evaluation of the pressure monitoring system in accordance with your monitoring plan and confirm that the pressure monitoring system continues to meet the performance requirements in you monitoring plan. Alternatively, install and verify the operation of a new pressure sensor.

(g) If you have an operating limit that requires a pH monitoring system, you must meet the requirements in paragraphs (d) and (g)(1) through (4) of this section.

(1) Install the pH sensor in a position that provides a representative measurement of scrubber effluent pH.

(2) Ensure the sample is properly mixed and representative of the fluid to be measured.

(3) Calibrate the pH monitoring system in accordance with your monitoring plan and according to the manufacturer's instructions. Clean the pH probe at least once each process operating day. Maintain on-site documentation that your calibration frequency is sufficient to maintain the specified accuracy of your device.

(4) Conduct a performance evaluation (including a two-point calibration with one of the two buffer solutions having a pH within 1 of the pH of the operating limit) of the pH monitoring system in accordance with your monitoring plan at the time of each performance test but no less frequently than annually.

(h) If you have an operating limit that requires a secondary electric power monitoring system for an electrostatic precipitator (ESP) operated with a wet scrubber, you must meet the requirements in paragraphs (h)(1) and (2) of this section.

(1) Install sensors to measure (secondary) voltage and current to the precipitator collection plates.

(2) Conduct a performance evaluation of the electric power monitoring system in accordance with your monitoring plan at the time of each performance test but no less frequently than annually.

(i) If you have an operating limit that requires the use of a monitoring system to measure sorbent injection rate (e.g., weigh belt, weigh hopper, or hopper flow measurement device), you must meet the requirements in paragraphs (d) and (i)(1) through (2) of this section.

(1) Install the system in a position(s) that provides a representative meas-

40 CFR Ch. I (7–1–23 Edition)

urement of the total sorbent injection rate.

(2) Conduct a performance evaluation of the sorbent injection rate monitoring system in accordance with your monitoring plan at the time of each performance test but no less frequently than annually.

(j) If you are not required to use a PM CPMS and elect to use a fabric filter bag leak detection system to comply with the requirements of this subpart, you must install, calibrate, maintain, and continuously operate the bag leak detection system as specified in paragraphs (j)(1) through (6) of this section.

(1) You must install a bag leak detection sensor(s) in a position(s) that will be representative of the relative or absolute PM loadings for each exhaust stack, roof vent, or compartment (e.g., for a positive pressure fabric filter) of the fabric filter.

(2) Conduct a performance evaluation of the bag leak detection system in accordance with your monitoring plan and consistent with the guidance provided in EPA-454/R-98-015 (incorporated by reference, see §63.14).

(3) Use a bag leak detection system certified by the manufacturer to be capable of detecting PM emissions at concentrations of 10 milligrams per actual cubic meter or less.

(4) Use a bag leak detection system equipped with a device to record continuously the output signal from the sensor.

(5) Use a bag leak detection system equipped with a system that will alert plant operating personnel when an increase in relative PM emissions over a preset level is detected. The alert must easily recognizable (e.g., heard or seen) by plant operating personnel.

(6) Where multiple bag leak detectors are required, the system's instrumentation and alert may be shared among detectors.

(k) For each unit that meets the definition of limited-use boiler or process heater, you must keep fuel use records for the days the boiler or process heater was operating.

(1) For each unit for which you decide to demonstrate compliance with the mercury or HCl emissions limits in Table 1 or 2 or Tables 11 through 15 to

this subpart by use of a CEMS for mercury or HCl, you must install, certify, maintain, and operate a CEMS measuring emissions discharged to the atmosphere and record the output of the system as specified in paragraphs (1)(1) through (8) of this section. For HCl, this option for an affected unit takes effect on the date of approval of a sitespecific monitoring plan.

(1) Notify the Administrator one month before starting use of the CEMS, and notify the Administrator one month before stopping use of the CEMS.

(2) Each CEMS shall be installed, certified, operated, and maintained according to the requirements in $\S63.7540(a)(14)$ for a mercury CEMS and $\S63.7540(a)(15)$ for a HCl CEMS.

(3) For a new unit, you must complete the initial performance evaluation of the CEMS by the latest of the dates specified in paragraph (1)(3)(i)through (iii) of this section.

(i) No later than July 30, 2013.

(ii) No later 180 days after the date of initial startup.

(iii) No later 180 days after notifying the Administrator before starting to use the CEMS in place of performance testing or fuel analysis to demonstrate compliance.

(4) For an existing unit, you must complete the initial performance evaluation by the latter of the two dates specified in paragraph (1)(4)(i) and (ii) of this section.

(i) No later than July 29, 2016.

(ii) No later 180 days after notifying the Administrator before starting to use the CEMS in place of performance testing or fuel analysis to demonstrate compliance.

(5) Compliance with the applicable emissions limit shall be determined based on the 30-day rolling average of the hourly arithmetic average emissions rates using the continuous monitoring system outlet data. The 30-day rolling arithmetic average emission rate (lb/MMBtu) shall be calculated using the equations in EPA Reference Method 19 at 40 CFR part 60, appendix A-7, but substituting the mercury or HCl concentration for the pollutant concentrations normally used in Method 19. (6) Collect CEMS hourly averages for all operating hours on a 30-day rolling average basis. Collect at least four CMS data values representing the four 15-minute periods in an hour, or at least two 15-minute data values during an hour when CMS calibration, quality assurance, or maintenance activities are being performed.

(7) The one-hour arithmetic averages required shall be expressed in lb/ MMBtu and shall be used to calculate the boiler 30-day and 10-day rolling average emissions.

(8) You are allowed to substitute the use of the PM, mercury or HCl CEMS for the applicable fuel analysis, annual performance test, and operating limits specified in Table 4 to this subpart to demonstrate compliance with the PM, mercury or HCl emissions limit, and if you are using an acid gas wet scrubber or dry sorbent injection control technology to comply with the HCl emission limit, you are allowed to substitute the use of a sulfur dioxide (SO_2) CEMS for the applicable fuel analysis, annual performance test, and operating limits specified in Table 4 to this subpart to demonstrate compliance with HCl emissions limit.

(m) If your unit is subject to a HCl emission limit in Table 1 or 2 or Tables 11 through 15 to this subpart and you have an acid gas wet scrubber or dry sorbent injection control technology and you elect to use an SO_2 CEMS to demonstrate continuous compliance with the HCl emission limit, you must install the monitor at the outlet of the boiler or process heater, downstream of all emission control devices, and you must install, certify, operate, and maintain the CEMS according to either part 60 or part 75 of this chapter.

(1) The SO₂ CEMS must be installed by the compliance date specified in $\S63.7495$.

(2) For on-going quality assurance (QA), the SO₂ CEMS must meet either the applicable daily and quarterly requirements in Procedure 1 of appendix F of part 60 or the applicable daily, quarterly, and semiannual or annual requirements in sections 2.1 through 2.3 of appendix B to part 75 of this chapter, with the following addition: You must perform the linearity checks required in section 2.2 of appendix B to part 75

of this chapter if the SO_2 CEMS has a span value of 30 ppm or less.

(3) For a new unit, the initial performance evaluation shall be completed no later than July 30, 2013, or 180 days after the date of initial startup, whichever is later. For an existing unit, the initial performance evaluation shall be completed no later than July 29, 2016.

(4) For purposes of collecting SO_2 data, you must operate the SO_2 CEMS as specified in §63.7535(b). You must use all the data collected during all periods in calculating data averages and assessing compliance, except that you must exclude certain data as specified in §63.7535(c). Periods when SO_2 data are unavailable may constitute monitoring deviations as specified in §63.7535(d).

(5) Collect CEMS hourly averages for all operating hours on a 30-day rolling average basis.

(6) Use only unadjusted, quality-assured SO₂ concentration values in the emissions calculations; do not apply bias adjustment factors to the part 75 SO₂ data and do not use part 75 substitute data values.

[76 FR 15664, Mar. 21, 2011, as amended at 78 FR 7171, Jan. 31, 2013; 80 FR 72810, Nov. 20, 2015; 87 FR 60844, Oct. 6, 2022]

§63.7530 How do I demonstrate initial compliance with the emission limitations, fuel specifications and work practice standards?

(a) You must demonstrate initial compliance with each emission limit that applies to you by conducting initial performance tests and fuel analyses and establishing operating limits, as applicable, according to §63.7520, paragraphs (b) and (c) of this section, and Tables 5 and 7 to this subpart. The requirement to conduct a fuel analysis is not applicable for units that burn a single type of fuel, as specified by

40 CFR Ch. I (7–1–23 Edition)

§63.7510(a)(2). If applicable, you must also install, operate, and maintain all applicable CMS (including CEMS, COMS, and CPMS) according to §63.7525.

(b) If you demonstrate compliance through performance stack testing, you must establish each site-specific operating limit in Table 4 to this subpart that applies to you according to the requirements in §63.7520, Table 7 to this subpart, and paragraph (b)(4) of this section, as applicable. You must also conduct fuel analyses according to §63.7521 and establish maximum fuel pollutant input levels according to paragraphs (b)(1) through (3) of this section, as applicable, and as specified in §63.7510(a)(2). (Note that §63.7510(a)(2) exempts certain fuels from the fuel analysis requirements.) However, if you switch fuel(s) and cannot show that the new fuel(s) does (do) not increase the chlorine, mercury, or TSM input into the unit through the results of fuel analysis, then you must repeat the performance test to demonstrate compliance while burning the new fuel(s).

(1) You must establish the maximum chlorine fuel input (Clinput) during the initial fuel analysis according to the procedures in paragraphs (b)(1)(i) through (iii) of this section.

(i) You must determine the fuel type or fuel mixture that you could burn in your boiler or process heater that has the highest content of chlorine.

(ii) During the fuel analysis for hydrogen chloride, you must determine the fraction of the total heat input for each fuel type burned (Qi) based on the fuel mixture that has the highest content of chlorine, and the average chlorine concentration of each fuel type burned (Ci).

(iii) You must establish a maximum chlorine input level using Equation 7 of this section.

$$Clinput = \sum_{i=1}^{n} (Ci \times Qi) \quad (Eq. 7)$$

Clinput = Maximum amount of chlorine entering the boiler or process heater

Where:

through fuels burned in units of pounds per million Btu.

- Ci = Arithmetic average concentration of chlorine in fuel type, i, analyzed according to §63.7521, in units of pounds per million Btu.
- Qi = Fraction of total heat input from fuel type, i, based on the fuel mixture that has the highest content of chlorine during the initial compliance test. If you do not burn multiple fuel types during the performance testing, it is not necessary to determine the value of this term. Insert a value of "1" for Qi. For continuous compliance demonstration, the actual fraction of the fuel burned during the month should be used.
- n = Number of different fuel types burned in your boiler or process heater for the mixture that has the highest content of chlorine.

(2) You must establish the maximum mercury fuel input level

(Mercuryinput) during the initial fuel analysis using the procedures in paragraphs (b)(2)(i) through (iii) of this section.

(i) You must determine the fuel type or fuel mixture that you could burn in your boiler or process heater that has the highest content of mercury.

(ii) During the compliance demonstration for mercury, you must determine the fraction of total heat input for each fuel burned (Qi) based on the fuel mixture that has the highest content of mercury, and the average mercury concentration of each fuel type burned (HGi).

(iii) You must establish a maximum mercury input level using Equation 8 of this section.

$$Mercuryinput = \sum_{i=1}^{n} (HGi \times Qi) \quad (Eq. 8)$$

Where:

- Mercuryinput = Maximum amount of mercury entering the boiler or process heater through fuels burned in units of pounds per million Btu.
- HGi = Arithmetic average concentration of mercury in fuel type, i, analyzed according to §63.7521, in units of pounds per million Btu.
- Qi = Fraction of total heat input from fuel type, i, based on the fuel mixture that has the highest mercury content during the initial compliance test. If you do not burn multiple fuel types during the performance test, it is not necessary to determine the value of this term. Insert a value of "1" for Qi. For continuous compliance demonstration, the actual fraction of the fuel burned during the month should be used.
- n = Number of different fuel types burned in your boiler or process heater for the mixture that has the highest content of mercury.

(3) If you opt to comply with the alternative TSM limit, you must establish the maximum TSM fuel input (TSMinput) for solid or liquid fuels during the initial fuel analysis according to the procedures in paragraphs (b)(3)(i) through (iii) of this section.

(i) You must determine the fuel type or fuel mixture that you could burn in your boiler or process heater that has the highest content of TSM.

(ii) During the fuel analysis for TSM, you must determine the fraction of the total heat input for each fuel type burned (Qi) based on the fuel mixture that has the highest content of TSM, and the average TSM concentration of each fuel type burned (TSMi).

(iii) You must establish a maximum TSM input level using Equation 9 of this section.

$$TSMinput = \sum_{i=1}^{n} (TSMi \times Qi) \quad (Eq. 9)$$

Where:

- TSMinput = Maximum amount of TSM entering the boiler or process heater through fuels burned in units of pounds per million Btu.
- TSMi = Arithmetic average concentration of TSM in fuel type, i, analyzed according to §63.7521, in units of pounds per million Btu.
- Qi = Fraction of total heat input from fuel type, i, based on the fuel mixture that has the highest content of TSM during the initial compliance test. If you do not burn multiple fuel types during the performance testing, it is not necessary to determine the value of this term. Insert a value of "1" for Qi. For continuous compliance demonstration, the actual fraction of the fuel burned during the month should be used.
- n = Number of different fuel types burned in your boiler or process heater for the mixture that has the highest content of TSM.

(4) You must establish parameter operating limits according to paragraphs (b)(4)(i) through (ix) of this section. As indicated in Table 4 to this subpart, you are not required to establish and comply with the operating parameter limits when you are using a CEMS to monitor and demonstrate compliance with the applicable emission limit for that control device parameter.

(i) For a wet acid gas scrubber, you must establish the minimum scrubber effluent pH and liquid flow rate as defined in §63.7575, as your operating limits during the performance test during which you demonstrate compliance with your applicable limit. If you use a wet scrubber and you conduct separate performance tests for HCl and mercury emissions, you must establish one set of minimum scrubber effluent pH, liquid flow rate, and pressure drop operating limits. The minimum scrubber effluent pH operating limit must be established during the HCl performance test. If you conduct multiple performance tests, you must set the minimum liquid flow rate operating limit at the higher of the minimum values established during the performance tests.

(ii) For any particulate control device (e.g., ESP, particulate wet scrubber, fabric filter) for which you use a PM CPMS, you must establish your PM CPMS operating limit and determine compliance with it according to para-

40 CFR Ch. I (7–1–23 Edition)

graphs (b)(4)(ii)(A) through (F) of this section.

(A) Determine your operating limit as the average PM CPMS output value recorded during the most recent performance test run demonstrating compliance with the filterable PM emission limit or at the PM CPMS output value corresponding to 75 percent of the emission limit if your PM performance test demonstrates compliance below 75 percent of the emission limit. You must verify an existing or establish a new operating limit after each repeated performance test. You must repeat the performance test annually and reassess and adjust the site-specific operating limit in accordance with the results of the performance test.

(1) Your PM CPMS must provide a 4-20 milliamp output and the establishment of its relationship to manual reference method measurements must be determined in units of milliamps.

(2) Your PM CPMS operating range must be capable of reading PM concentrations from zero to a level equivalent to at least two times your allowable emission limit. If your PM CPMS is an auto-ranging instrument capable of multiple scales, the primary range of the instrument must be capable of reading PM concentration from zero to a level equivalent to two times your allowable emission limit.

(3) During the initial performance test or any such subsequent performance test that demonstrates compliance with the PM limit, record and average all milliamp output values from the PM CPMS for the periods corresponding to the compliance test runs (e.g., average all your PM CPMS output values for three corresponding 2hour Method 5I test runs).

(B) If the average of your three PM performance test runs are below 75 percent of your PM emission limit, you must calculate an operating limit by establishing a relationship of PM CPMS signal to PM concentration using the PM CPMS instrument zero, the average PM CPMS values corresponding to the three compliance test runs, and the average PM concentration from the Method 5 or performance test with the procedures in paragraphs (b)(4)(ii)(B)(1) through (4) of this section.

§63.7530

(1) Determine your instrument zero output with one of the following procedures:

(i) Zero point data for *in-situ* instruments should be obtained by removing the instrument from the stack and monitoring ambient air on a test bench.

(*ii*) Zero point data for *extractive* instruments should be obtained by removing the extractive probe from the stack and drawing in clean ambient air.

(*iii*) The zero point may also be established by performing manual reference method measurements when the flue gas is free of PM emissions or contains very low PM concentrations (e.g., when your process is not operating, but the fans are operating or your source is combusting only natural gas) and plotting these with the compliance data to find the zero intercept.

(iv) If none of the steps in paragraphs (b)(4)(ii)(B)(1)(i) through (iii) of this section are possible, you must use a zero output value provided by the manufacturer.

(2) Determine your PM CPMS instrument average in milliamps, and the average of your corresponding three PM compliance test runs, using equation 10.

$$\overline{x} = \frac{1}{n} \sum_{i=1}^{n} X_{1,i} \overline{y} = \frac{1}{n} \sum_{i=1}^{n} Y_{1} \qquad (\text{Eq. 10})$$

Where:

- X_1 = the PM CPMS data points for the three runs constituting the performance test,
- \mathbf{Y}_1 = the PM concentration value for the three runs constituting the performance test, and
- n = the number of data points.

(3) With your instrument zero expressed in milliamps, your three run average PM CPMS milliamp value, and your three run average PM concentration from your three compliance tests, determine a relationship of lb/MMBtu per milliamp with equation 11.

$$\mathbf{R} = \frac{Y_1}{\left(X_1 - z\right)} \quad (Eq. 11)$$

Where:

- R = the relative lb/MMBtu per milliamp for your PM CPMS,
- Y_1 = the three run average lb/MMBtu PM concentration,
- \mathbf{X}_1 = the three run average milliamp output from you PM CPMS, and
- z = the milliamp equivalent of your instrument zero determined from (B)(i).

(4) Determine your source specific 30day rolling average operating limit using the lb/MMBtu per milliamp value from Equation 11 in equation 12, below. This sets your operating limit at the PM CPMS output value corresponding to 75 percent of your emission limit.

$$\mathcal{O}_i = z + \frac{0.75(L)}{R}$$

τ7

Where:

- O_1 = the operating limit for your PM CPMS on a 30-day rolling average, in milliamps.
- L = your source emission limit expressed in lb/MMBtu.

(Eq. 12)

z = your instrument zero in milliamps, determined from (B)(i), and

R = the relative lb/MMBtu per milliamp for your PM CPMS, from Equation 11.

(C) If the average of your three PM compliance test runs is at or above 75 percent of your PM emission limit you must determine your 30-day rolling average operating limit by averaging the PM CPMS milliamp output cor-

40 CFR Ch. I (7–1–23 Edition)

responding to your three PM performance test runs that demonstrate compliance with the emission limit using equation 13 and you must submit all compliance test and PM CPMS data according to the reporting requirements in paragraph (b)(4)(ii)(F) of this section.

$$O_{h} = \frac{1}{n} \sum_{i=1}^{n} X_{1}$$
 (Eq. 13)

Where:

 ${\rm X}_1$ = the PM CPMS data points for all runs i, n = the number of data points, and

 O_h = your site specific operating limit, in milliamps.

(D) To determine continuous compliance, you must record the PM CPMS output data for all periods when the process is operating and the PM CPMS is not out-of-control. You must demonstrate continuous compliance by using all quality-assured hourly average data collected by the PM CPMS for all operating hours to calculate the arithmetic average operating parameter in units of the operating limit (milliamps) on a 30-day rolling average basis, updated at the end of each new operating hour. Use Equation 14 to determine the 30-day rolling average.

$$30 - \text{day} = \frac{\sum_{i=1}^{n} Hpw}{n} \quad (\text{Eq. 14})$$

Where:

30-day = 30-day average.

Hpvi = is the hourly parameter value for hour i

n = is the number of valid hourly parameter values collected over the previous 30 operating days.

(E) Use EPA Method 5 of appendix A to part 60 of this chapter to determine PM emissions. For each performance test, conduct three separate runs under the conditions that exist when the affected source is operating at the highest load or capacity level reasonably expected to occur. Conduct each test run to collect a minimum sample volume specified in Table 1 or 2 or Tables 11 through 15 to this subpart, as applicable, for determining compliance with a new source limit or an existing source limit. Calculate the average of the results from three runs to determine the runs to determine the complexity of the results from three runs to determine the complexity of the results from three runs to determine the complexity of the results from three runs to determine the complexity of the results from three runs to determine the complexity of the results from three runs to determine the complexity of the results from three runs to determine the complexity of the results from three runs to determine the complexity of the results from three runs to determine the complexity of the results from three runs to determine the complexity of the results from three runs to determine the complexity of the results from three runs to determine the complexity of the results from three runs to determine the complexity of the results from three runs to determine the complexity of the results from three runs to determine the complexity of the results from three runs to determine the complexity of the runs to determine the complexity of the results from three runs to determine the complexity of the runs to det

mine compliance. You need not determine the PM collected in the impingers ("back half") of the Method 5 particulate sampling train to demonstrate compliance with the PM standards in this subpart. This shall not preclude the permitting authority from requiring a determination of the "back half" for other purposes.

(F) For PM performance test reports used to set a PM CPMS operating limit, the electronic submission of the test report must also include the make and model of the PM CPMS instrument, serial number of the instrument, analytical principle of the instrument (*e.g.* beta attenuation), span of the instruments primary analytical range, milliamp value equivalent to the instrument zero output, technique by which this zero value was determined,

and the average milliamp signals corresponding to each PM compliance test run.

(iii) For a particulate wet scrubber, you must establish the minimum pressure drop and liquid flow rate as defined in §63.7575, as your operating limits during the three-run performance test during which you demonstrate compliance with your applicable limit. If you use a wet scrubber and you conduct separate performance tests for PM and TSM emissions, you must establish one set of minimum scrubber liquid flow rate and pressure drop operating limits. If you conduct multiple performance tests, you must set the minimum liquid flow rate and pressure drop operating limits at the higher of the minimum values established during the performance tests.

(iv) For an electrostatic precipitator (ESP) operated with a wet scrubber, you must establish the minimum total secondary electric power input, as defined in §63.7575, as your operating limit during the three-run performance test during which you demonstrate compliance with your applicable limit. (These operating limits do not apply to ESP that are operated as dry controls without a wet scrubber.)

(v) For a dry scrubber, you must establish the minimum sorbent injection rate for each sorbent, as defined in $\S63.7575$, as your operating limit during the three-run performance test during which you demonstrate compliance with your applicable limit.

(vi) For activated carbon injection, you must establish the minimum activated carbon injection rate, as defined in §63.7575, as your operating limit during the three-run performance test during which you demonstrate compliance with your applicable limit.

(vii) The operating limit for boilers or process heaters with fabric filters that demonstrate continuous compliance through bag leak detection systems is that a bag leak detection system be installed according to the requirements in §63.7525, and that each fabric filter must be operated such that the bag leak detection system alert is not activated more than 5 percent of the operating time during a 6-month period.

(viii) For a minimum oxygen level, if you conduct multiple performance tests, you must set the minimum oxygen level at the lower of the minimum values established during the performance tests.

(ix) The operating limit for boilers or process heaters that demonstrate continuous compliance with the HCl emission limit using a SO₂ CEMS is to install and operate the SO₂ according to the requirements in $\S63.7525(m)$ establish a maximum SO₂ emission rate equal to the highest hourly average SO₂ measurement during the most recent three-run performance test for HCl.

(c) If you elect to demonstrate compliance with an applicable emission limit through fuel analysis, you must conduct fuel analyses according to $\S63.7521$ and follow the procedures in paragraphs (c)(1) through (5) of this section.

(1) If you burn more than one fuel type, you must determine the fuel mixture you could burn in your boiler or process heater that would result in the maximum emission rates of the pollutants that you elect to demonstrate compliance through fuel analysis.

(2) You must determine the 90th percentile confidence level fuel pollutant concentration of the composite samples analyzed for each fuel type using the one-sided t-statistic test described in Equation 15 of this section.

$$P90 = mean + (SD \times t) \quad (Eq. 15)$$

Where:

- P90 = 90th percentile confidence level pollutant concentration, in pounds per million Btu.
- Mean = Arithmetic average of the fuel pollutant concentration in the fuel samples analyzed according to §63.7521, in units of pounds per million Btu.
- SD = Standard deviation of the mean of pollutant concentration in the fuel samples

§63.7530

analyzed according to §63.7521, in units of pounds per million Btu. SD is calculated as the sample standard deviation divided by the square root of the number of samples.

t = t distribution critical value for 90th percentile $(t_{0,1})$ probability for the appropriate degrees of freedom (number of samples minus one) as obtained from a t-Distribution Critical Value Table. 40 CFR Ch. I (7–1–23 Edition)

(3) To demonstrate compliance with the applicable emission limit for HCl, the HCl emission rate that you calculate for your boiler or process heater using Equation 16 of this section must not exceed the applicable emission limit for HCl.

$$HCl = \sum_{i=1}^{n} \left(Ci90 \times Qi \times 1.028 \right)$$
 (Eq. 16)

Where:

- HCl = HCl emission rate from the boiler or process heater in units of pounds per million Btu.
- Ci90 = 90th percentile confidence level concentration of chlorine in fuel type, i, in units of pounds per million Btu as calculated according to Equation 15 of this section.
- Qi = Fraction of total heat input from fuel type, i, based on the fuel mixture that has the highest content of chlorine. If you do not burn multiple fuel types, it is not necessary to determine the value of this term. Insert a value of "1" for Qi. For continuous compliance demonstra-

tion, the actual fraction of the fuel burned during the month should be used.

- n = Number of different fuel types burned in your boiler or process heater for the mixture that has the highest content of chlorine.
- 1.028 = Molecular weight ratio of HCl to chlorine.

(4) To demonstrate compliance with the applicable emission limit for mercury, the mercury emission rate that you calculate for your boiler or process heater using Equation 17 of this section must not exceed the applicable emission limit for mercury.

$$Mercury = \sum_{i=1}^{n} (Hgi90 \times Qi) \quad (Eq. 17)$$

Where:

- Mercury = Mercury emission rate from the boiler or process heater in units of pounds per million Btu.
- Hgi90 = 90th percentile confidence level concentration of mercury in fuel, i, in units of pounds per million Btu as calculated according to Equation 15 of this section.
- Qi = Fraction of total heat input from fuel type, i, based on the fuel mixture that has the highest mercury content. If you do not burn multiple fuel types, it is not necessary to determine the value of this term. Insert a value of "1" for Qi. For continuous compliance demonstration,

the actual fraction of the fuel burned during the month should be used.

n = Number of different fuel types burned in your boiler or process heater for the mixture that has the highest mercury content.

(5) To demonstrate compliance with the applicable emission limit for TSM for solid or liquid fuels, the TSM emission rate that you calculate for your boiler or process heater from solid fuels using Equation 18 of this section must not exceed the applicable emission limit for TSM.

§63.7533

$$Metals = \sum_{i=1}^{n} (TSM90i \times Qi) \quad (Eq. 18)$$

Where:

- Metals = TSM emission rate from the boiler or process heater in units of pounds per million Btu.
- TSMi90 = 90th percentile confidence level concentration of TSM in fuel, i, in units of pounds per million Btu as calculated according to Equation 15 of this section.
- Qi = Fraction of total heat input from fuel type, i, based on the fuel mixture that has the highest TSM content. If you do not burn multiple fuel types, it is not necessary to determine the value of this term. Insert a value of "1" for Qi. For continuous compliance demonstration, the actual fraction of the fuel burned during the month should be used.
- $$\begin{split} n &= \text{Number of different fuel types burned in} \\ \text{your boiler or process heater for the mixture that has the highest TSM content.} \end{split}$$

(d)[Reserved]

(e) You must include with the Notification of Compliance Status a signed certification that either the energy assessment was completed according to Table 3 to this subpart, and that the assessment is an accurate depiction of your facility at the time of the assessment, or that the maximum number of on-site technical hours specified in the definition of energy assessment applicable to the facility has been expended.

(f) You must submit the Notification of Compliance Status containing the results of the initial compliance demonstration according to the requirements in §63.7545(e).

(g) If you elect to demonstrate that a gaseous fuel meets the specifications of another gas 1 fuel as defined in §63.7575, you must conduct an initial fuel specification analyses according to §63.7521(f) through (i) and according to the frequency listed in §63.7540(c) and maintain records of the results of the testing as outlined in §63.7555(g). For samples where the initial mercury specification has not been exceeded, you will include a signed certification with the Notification of Compliance Status that the initial fuel specification test meets the gas specification outlined in the definition of other gas 1 fuels.

(h) If you own or operate a unit subject to emission limits in Table 1 or 2 or Tables 11 through 15 to this subpart, you must meet the work practice standard according to Table 3 to this subpart. During startup and shutdown, you must only follow the work practice standards according to items 5 and 6 of Table 3 to this subpart.

(i) If you opt to comply with the alternative SO_2 CEMS operating limit in Tables 4 and 8 to this subpart, you may do so only if your affected boiler or process heater:

(1) Has a system using wet scrubber or dry sorbent injection and SO_2 CEMS installed on the unit; and

(2) At all times, you operate the wet scrubber or dry sorbent injection for acid gas control on the unit consistent with 63.7500(a)(3); and

(3) You establish a unit-specific maximum SO_2 operating limit by collecting the maximum hourly SO_2 emission rate on the SO_2 CEMS during the paired 3run test for HCl. The maximum SO_2 operating limit is equal to the highest hourly average SO_2 concentration measured during the HCl performance test.

[76 FR 15664, Mar. 21, 2011, as amended at 78
FR 7174, Jan. 31, 2013; 80 FR 72811, Nov. 20, 2015; 87 FR 60845, Oct. 6, 2022]

§63.7533 Can I use efficiency credits earned from implementation of energy conservation measures to comply with this subpart?

(a) If you elect to comply with the alternative equivalent output-based emission limits, instead of the heat input-based limits listed in Table 2 or 15 to this subpart, and you want to take credit for implementing energy conservation measures identified in an energy assessment, you may demonstrate compliance using efficiency credits according to the procedures in this section. You may use this compliance approach for an existing affected boiler for demonstrating initial compliance according to §63.7522(e) and for

demonstrating monthly compliance according to §63.7522(f). Owners or operators using this compliance approach must establish an emissions benchmark, calculate and document the efficiency credits, develop an Implementation Plan, comply with the general reporting requirements, and apply the efficiency credit according to the procedures in paragraphs (b) through (f) of this section. You cannot use this compliance approach for a new or reconstructed affected boiler. Additional guidance from the Department of Energy on efficiency credits is available at https://www.epa.gov/ttn/atw/boiler/ boilerpg.html.

(b) For each existing affected boiler for which you intend to apply emissions credits, establish a benchmark from which emission reduction credits may be generated by determining the actual annual fuel heat input to the affected boiler before initiation of an energy conservation activity to reduce energy demand (*i.e.*, fuel usage) according to paragraphs (b)(1) through (4) of this section. The benchmark shall be expressed in trillion Btu per year heat input.

(1) The benchmark from which efficiency credits may be generated shall be determined by using the most representative, accurate, and reliable process available for the source. The benchmark shall be established for a one-year period before the date that an energy demand reduction occurs, unless it can be demonstrated that a different time period is more representative of historical operations.

(2) Determine the starting point from which to measure progress. Inventory all fuel purchased and generated onsite (off-gases, residues) in physical units (MMBtu, million cubic feet, etc.).

(3) Document all uses of energy from the affected boiler. Use the most recent data available.

(4) Collect non-energy related facility and operational data to normalize, if necessary, the benchmark to current 40 CFR Ch. I (7–1–23 Edition)

operations, such as building size, operating hours, etc. If possible, use actual data that are current and timely rather than estimated data.

(c) Efficiency credits can be generated if the energy conservation measures were implemented after January 1, 2008 and if sufficient information is available to determine the appropriate value of credits.

(1) The following emission points cannot be used to generate efficiency credits:

(i) Energy conservation measures implemented on or before January 1, 2008, unless the level of energy demand reduction is increased after January 1, 2008, in which case credit will be allowed only for change in demand reduction achieved after January 1, 2008.

(ii) Efficiency credits on shut-down boilers. Boilers that are shut down cannot be used to generate credits unless the facility provides documentation linking the permanent shutdown to energy conservation measures identified in the energy assessment. In this case, the bench established for the affected boiler to which the credits from the shutdown will be applied must be revised to include the benchmark established for the shutdown boiler.

(2) For all points included in calculating emissions credits, the owner or operator shall:

(i) Calculate annual credits for all energy demand points. Use Equation 19 to calculate credits. Energy conservation measures that meet the criteria of paragraph (c)(1) of this section shall not be included, except as specified in paragraph (c)(1)(i) of this section.

(3) Credits are generated by the difference between the benchmark that is established for each affected boiler, and the actual energy demand reductions from energy conservation measures implemented after January 1, 2008. Credits shall be calculated using Equation 19 of this section as follows:

(i) The overall equation for calculating credits is:

$$ECredits = \left(\sum_{i=1}^{n} EIS_{iactual}\right) \div EI_{baseline} \quad (Eq. 19)$$

Where:

- ECredits = Energy Input Savings for all energy conservation measures implemented for an affected boiler, expressed as a decimal fraction of the baseline energy input.
- EIS_{iactual} = Energy Input Savings for each energy conservation measure, i, implemented for an affected boiler, million Btu per year.
- $\mathrm{EI}_{\mathrm{baseline}}$ = Energy Input baseline for the affected boiler, million Btu per year.
- n = Number of energy conservation measures
 included in the efficiency credit for the
 affected boiler.
 - (ii) [Reserved]

(d) The owner or operator shall develop, and submit for approval upon request by the Administrator, an Implementation Plan containing all of the information required in this paragraph for all boilers to be included in an efficiency credit approach. The Implementation Plan shall identify all existing affected boilers to be included in applying the efficiency credits. The Implementation Plan shall include a description of the energy conservation measures implemented and the energy sav-

Equation 20 to paragraph (f)

ings generated from each measure and an explanation of the criteria used for determining that savings. If requested, you must submit the implementation plan for efficiency credits to the Administrator for review and approval no later than 180 days before the date on which the facility intends to demonstrate compliance using the efficiency credit approach.

(e) The emissions rate as calculated using Equation 20 in paragraph (f) of this section from each existing boiler participating in the efficiency credit option must be in compliance with the limits in Table 2 or 15 to this subpart at all times the affected unit is subject to numeric emission limits, following the compliance date specified in §63.7495.

(f) You must use Equation 20 of this paragraph (f) to demonstrate initial compliance by demonstrating that the emissions from the affected boiler participating in the efficiency credit compliance approach do not exceed the emission limits in Table 2 or 15 to this subpart.

$E_{ac} = E_{a} \times (1 - ECredits) \quad (Eq. 20)$

Where:

- $$\begin{split} \mathbf{E}_{adj} &= \mathbf{Emission} \text{ level adjusted by applying} \\ \text{ the efficiency credits earned, lb per million Btu steam output (or lb per MWh)} \\ \text{ for the affected boiler.} \end{split}$$
- E_m = Emissions measured during the performance test, lb per million Btu steam output (or lb per MWh) for the affected boiler.
- ECredits = Efficiency credits from Equation 19 to paragraph (c)(3)(i) of this section for the affected boiler.

(g) As part of each compliance report submitted as required under §63.7550, you must include documentation that the energy conservation measures implemented continue to generate the credit for use in demonstrating compliance with the emission limits.

[76 FR 15664, Mar. 21, 2011, as amended at 78
 FR 7178, Jan. 31, 2013; 80 FR 72812, Nov. 20, 2015; 87 FR 60845, Oct. 6, 2022]

CONTINUOUS COMPLIANCE REQUIREMENTS

§63.7535 Is there a minimum amount of monitoring data I must obtain?

(a) You must monitor and collect data according to this section and the site-specific monitoring plan required by §63.7505(d).

(b) You must operate the monitoring system and collect data at all required intervals at all times that each boiler or process heater is operating and compliance is required, except for periods of monitoring system malfunctions or

out of control periods (see §63.8(c)(7) of this part), and required monitoring system quality assurance or control activities, including, as applicable, calibration checks, required zero and span adjustments, and scheduled CMS maintenance as defined in your site-specific monitoring plan. A monitoring system malfunction is any sudden, infrequent, not reasonably preventable failure of the monitoring system to provide valid data. Monitoring system failures that are caused in part by poor maintenance or careless operation are not malfunctions. You are required to complete monitoring system repairs in response to monitoring system malfunctions or out-of-control periods and to return the monitoring system to operation as expeditiously as practicable.

(c) You may not use data recorded during periods of startup and shutdown, monitoring system malfunctions or out-of-control periods, repairs associated with monitoring system malfunctions or out-of-control periods, or required monitoring system quality assurance or control activities in data averages and calculations used to report emissions or operating levels. You must record and make available upon request results of CMS performance audits and dates and duration of periods when the CMS is out of control to completion of the corrective actions necessary to return the CMS to operation consistent with your site-specific monitoring plan. You must use all the data collected during all other periods in assessing compliance and the operation of the control device and associated control system.

(d) Except for periods of monitoring system malfunctions, repairs associated with monitoring system malfunctions, and required monitoring system quality assurance or quality control activities (including, as applicable, system accuracy audits, calibration checks, and required zero and span adjustments), failure to collect required data is a deviation of the monitoring requirements. In calculating monitoring results, do not use any data collected during periods of startup and shutdown, when the monitoring system is out of control as specified in your site-specific monitoring plan, while conducting repairs associated with pe-

40 CFR Ch. I (7–1–23 Edition)

riods when the monitoring system is out of control, or while conducting required monitoring system quality assurance or quality control activities. You must calculate monitoring results using all other monitoring data collected while the process is operating. You must report all periods when the monitoring system is out of control in your semi-annual report.

[76 FR 15664, Mar. 21, 2011, as amended at 78 FR 7179, Jan. 31, 2013; 80 FR 72812, Nov. 20, 2015]

§63.7540 How do I demonstrate continuous compliance with the emission limitations, fuel specifications and work practice standards?

(a) You must demonstrate continuous compliance with each emission limit in Tables 1 and 2 or 11 through 15 to this subpart, the work practice standards in Table 3 to this subpart, and the operating limits in Table 4 to this subpart that applies to you according to the methods specified in Table 8 to this subpart and paragraphs (a)(1) through (19) of this section.

(1) Following the date on which the initial compliance demonstration is completed or is required to be completed under §§63.7 and 63.7510, whichever date comes first, operation above the established maximum or below the established minimum operating limits shall constitute a deviation of established operating limits listed in Table 4 of this subpart except during performance tests conducted to determine compliance with the emission limits or to establish new operating limits. Operating limits must be confirmed or reestablished during performance tests.

(2) As specified in §63.7555(d), you must keep records of the type and amount of all fuels burned in each boiler or process heater during the reporting period to demonstrate that all fuel types and mixtures of fuels burned would result in either of the following:

(i) Equal to or lower emissions of HCl, mercury, and TSM than the applicable emission limit for each pollutant, if you demonstrate compliance through fuel analysis.

(ii) Equal to or lower fuel input of chlorine, mercury, and TSM than the maximum values calculated during the

last performance test, if you demonstrate compliance through performance testing.

(3) If you demonstrate compliance with an applicable HCl emission limit through fuel analysis for a solid or liquid fuel and you plan to burn a new type of solid or liquid fuel, you must recalculate the HCl emission rate using Equation 16 of §63.7530 according to paragraphs (a)(3)(i) through (iii) of this section. You are not required to conduct fuel analyses for the fuels described in §63.7510(a)(2)(i) through (iii). You may exclude the fuels described in §63.7510(a)(2)(i) through (iii) when recalculating the HCl emission rate.

(i) You must determine the chlorine concentration for any new fuel type in units of pounds per million Btu, based on supplier data or your own fuel analysis, according to the provisions in your site-specific fuel analysis plan developed according to §63.7521(b).

(ii) You must determine the new mixture of fuels that will have the highest content of chlorine.

(iii) Recalculate the HCl emission rate from your boiler or process heater under these new conditions using Equation 16 of §63.7530. The recalculated HCl emission rate must be less than the applicable emission limit.

(4) If you demonstrate compliance with an applicable HCl emission limit through performance testing and you plan to burn a new type of fuel or a new mixture of fuels, you must recalculate the maximum chlorine input using Equation 7 of §63.7530. If the results of recalculating the maximum chlorine input using Equation 7 of §63.7530 are greater than the maximum chlorine input level established during the previous performance test, then you must conduct a new performance test within 60 days of burning the new fuel type or fuel mixture according to the procedures in §63.7520 to demonstrate that the HCl emissions do not exceed the emission limit. You must also establish new operating limits based on this performance test according to the procedures in §63.7530(b). In recalculating the maximum chlorine input and establishing the new operating limits, you are not required to conduct fuel analyses for and include the fuels described in §63.7510(a)(2)(i) through (iii).

(5) If you demonstrate compliance with an applicable mercury emission limit through fuel analysis, and you plan to burn a new type of fuel, you must recalculate the mercury emission rate using Equation 17 of 63.7530 according to the procedures specified in paragraphs (a)(5)(i) through (iii) of this section. You are not required to conduct fuel analyses for the fuels described in 63.7510(a)(2)(i) through (iii). You may exclude the fuels described in 63.7510(a)(2)(i) through (iii) when recalculating the mercury emission rate.

(i) You must determine the mercury concentration for any new fuel type in units of pounds per million Btu, based on supplier data or your own fuel analysis, according to the provisions in your site-specific fuel analysis plan developed according to §63.7521(b).

(ii) You must determine the new mixture of fuels that will have the highest content of mercury.

(iii) Recalculate the mercury emission rate from your boiler or process heater under these new conditions using Equation 17 of §63.7530. The recalculated mercury emission rate must be less than the applicable emission limit.

(6) If you demonstrate compliance with an applicable mercury emission limit through performance testing, and you plan to burn a new type of fuel or a new mixture of fuels, you must recalculate the maximum mercury input using Equation 8 of §63.7530. If the results of recalculating the maximum mercury input using Equation 8 of §63.7530 are higher than the maximum mercury input level established during the previous performance test, then you must conduct a new performance test within 60 days of burning the new fuel type or fuel mixture according to the procedures in §63.7520 to demonstrate that the mercury emissions do not exceed the emission limit. You must also establish new operating limits based on this performance test according to the procedures in §63.7530(b). You are not required to conduct fuel analyses for the fuels described in §63.7510(a)(2)(i) through (iii). You may exclude the fuels described in §63.7510(a)(2)(i) through (iii) when recalculating the mercury emission rate.

§63.7540

§63.7540

(7) If your unit is controlled with a fabric filter, and you demonstrate continuous compliance using a bag leak detection system, you must initiate corrective action within 1 hour of a bag leak detection system alert and complete corrective actions as soon as practical, and operate and maintain the fabric filter system such that the periods which would cause an alert are no more than 5 percent of the operating time during a 6-month period. You must also keep records of the date, time, and duration of each alert, the time corrective action was initiated and completed, and a brief description of the cause of the alert and the corrective action taken. You must also record the percent of the operating time during each 6-month period that the conditions exist for an alert. In calculating this operating time percentage, if inspection of the fabric filter demonstrates that no corrective action is required, no alert time is counted. If corrective action is required, each alert shall be counted as a minimum of 1 hour. If you take longer than 1 hour to initiate corrective action, the alert time shall be counted as the actual amount of time taken to initiate corrective action.

(8) To demonstrate compliance with the applicable alternative CO CEMS emission limit listed in Table 1 or 2 or Tables 11 through 15 to this subpart, you must meet the requirements in paragraphs (a)(8)(i) through (iv) of this section.

(i) Continuously monitor CO according to §§ 63.7525(a) and 63.7535.

(ii) Maintain a CO emission level below or at your applicable alternative CO CEMS-based standard in Table 1 or 2 or Tables 11 through 15 to this subpart at all times the affected unit is subject to numeric emission limits.

(iii) Keep records of CO levels according to §63.7555(b).

(iv) You must record and make available upon request results of CO CEMS performance audits, dates and duration of periods when the CO CEMS is out of control to completion of the corrective actions necessary to return the CO CEMS to operation consistent with your site-specific monitoring plan.

(9) The owner or operator of a boiler or process heater using a PM CPMS or

40 CFR Ch. I (7–1–23 Edition)

a PM CEMS to meet requirements of this subpart shall install, certify (PM CEMS only), operate, and maintain the PM CPMS or PM CEMS in accordance with your site-specific monitoring plan as required in §63.7505(d).

(10) If your boiler or process heater has a heat input capacity of 10 million Btu per hour or greater, you must conduct an annual tune-up of the boiler or process heater to demonstrate continuous compliance as specified in paragraphs (a)(10)(i) through (vi) of this section. You must conduct the tune-up while burning the type of fuel (or fuels in case of units that routinely burn a mixture) that provided the majority of the heat input to the boiler or process heater over the 12 months prior to the tune-up. This frequency does not apply to limited-use boilers and process heaters, as defined in §63.7575, or units with continuous oxygen trim systems that maintain an optimum air to fuel ratio.

(i) As applicable, inspect the burner, and clean or replace any components of the burner as necessary (you may perform the burner inspection any time prior to the tune-up or delay the burner inspection until the next scheduled unit shutdown). Units that produce electricity for sale may delay the burner inspection until the first outage, not to exceed 36 months from the previous inspection. At units where entry into a piece of process equipment or into a storage vessel is required to complete the tune-up inspections, inspections are required only during planned entries into the storage vessel or process equipment;

(ii) Inspect the flame pattern, as applicable, and adjust the burner as necessary to optimize the flame pattern. The adjustment should be consistent with the manufacturer's specifications, if available;

(iii) Inspect the system controlling the air-to-fuel ratio, as applicable, and ensure that it is correctly calibrated and functioning properly (you may delay the inspection until the next scheduled unit shutdown). Units that produce electricity for sale may delay the inspection until the first outage, not to exceed 36 months from the previous inspection;

(iv) Optimize total emissions of CO. This optimization should be consistent

with the manufacturer's specifications, if available, and with any NO_X requirement to which the unit is subject;

(v) Measure the concentrations in the effluent stream of CO in parts per million, by volume, and oxygen in volume percent, before and after the adjustments are made (measurements may be either on a dry or wet basis, as long as it is the same basis before and after the adjustments are made). Measurements may be taken using a portable CO analyzer; and

(vi) Maintain on-site and submit, if requested by the Administrator, a report containing the information in paragraphs (a)(10)(vi)(A) through (C) of this section,

(A) The concentrations of CO in the effluent stream in parts per million by volume, and oxygen in volume percent, measured at high fire or typical operating load, before and after the tune-up of the boiler or process heater;

(B) A description of any corrective actions taken as a part of the tune-up; and

(C) The type and amount of fuel used over the 12 months prior to the tuneup, but only if the unit was physically and legally capable of using more than one type of fuel during that period. Units sharing a fuel meter may estimate the fuel used by each unit.

(11) If your boiler or process heater has a heat input capacity of less than 10 million Btu per hour (except as specified in paragraph (a)(12) of this section), you must conduct a biennial tune-up of the boiler or process heater as specified in paragraphs (a)(10)(i) through (vi) of this section to demonstrate continuous compliance.

(12) If your boiler or process heater has a continuous oxygen trim system that maintains an optimum air to fuel ratio, or a heat input capacity of less than or equal to 5 million Btu per hour and the unit is in the units designed to burn gas 1; units designed to burn gas 2 (other); or units designed to burn light liquid subcategories, or meets the definition of limited-use boiler or process heater in §63.7575, you must conduct a tune-up of the boiler or process heater every 5 years as specified in paragraphs (a)(10)(i) through (vi) of this section to demonstrate continuous compliance. You may delay the burner

inspection specified in paragraph (a)(10)(i) of this section until the next scheduled or unscheduled unit shutdown, but you must inspect each burner at least once every 72 months. If an oxygen trim system is utilized on a unit without emission standards to reduce the tune-up frequency to once every 5 years, set the oxygen level no lower than the oxygen concentration measured during the most recent tune-up.

(13) If the unit is not operating on the required date for a tune-up, the tune-up must be conducted within 30 calendar days of startup.

(14) If you are using a CEMS measuring mercury emissions to meet requirements of this subpart you must install, certify, operate, and maintain the mercury CEMS as specified in paragraphs (a)(14)(i) and (ii) of this section.

(i) Operate the mercury CEMS in accordance with performance specification 12A of 40 CFR part 60, appendix B or operate a sorbent trap based integrated monitor in accordance with performance specification 12B of 40 CFR part 60, appendix B. The duration of the performance test must be 30 operating days if you specified a 30 operating day basis in §63.7545(e)(2)(iii) for mercury CEMS or it must be 720 hours if you specified a 720 hour basis in §63.7545(e)(2)(iii) for mercury CEMS. For each day in which the unit operates, you must obtain hourly mercury concentration data, and stack gas volumetric flow rate data.

(ii) If you are using a mercury CEMS, you must install, operate, calibrate, and maintain an instrument for continuously measuring and recording the mercury mass emissions rate to the atmosphere according to the requirements of performance specifications 6 and 12A of 40 CFR part 60, appendix B, and quality assurance procedure 6 of 40 CFR part 60, appendix F.

(15) If you are using a CEMS to measure HCl emissions to meet requirements of this subpart, you must install, certify, operate, and maintain the HCl CEMS as specified in paragraphs (a)(15)(i) and (ii) of this section. This option for an affected unit takes effect on the date of approval of a sitespecific monitoring plan.

(i) Operate the continuous emissions monitoring system in accordance with the applicable performance specification in 40 CFR part 60, appendix B. The duration of the performance test must be 30 operating days if you specified a operating dav basis in §63.7545(e)(2)(iii) for HCl CEMS or it must be 720 hours if you specified a 720 hour basis in §63.7545(e)(2)(iii) for HCl CEMS. For each day in which the unit operates, you must obtain hourly HCl concentration data, and stack gas volumetric flow rate data.

(ii) If you are using a HCl CEMS, you must install, operate, calibrate, and maintain an instrument for continuously measuring and recording the HCl mass emissions rate to the atmosphere according to the requirements of the applicable performance specification of 40 CFR part 60, appendix B, and the quality assurance procedures of 40 CFR part 60, appendix F.

(16) If you demonstrate compliance with an applicable TSM emission limit through performance testing, and you plan to burn a new type of fuel or a new mixture of fuels, you must recalculate the maximum TSM input using Equation 9 of §63.7530. If the results of recalculating the maximum TSM input using Equation 9 of §63.7530 are higher than the maximum total selected input level established during the previous performance test, then you must conduct a new performance test within 60 days of burning the new fuel type or fuel mixture according to the procedures in §63.7520 to demonstrate that the TSM emissions do not exceed the emission limit. You must also establish new operating limits based on this performance test according to the procedures in §63.7530(b). You are not required to conduct fuel analyses for the fuels described in §63.7510(a)(2)(i) through (iii). You may exclude the fuels described in §63.7510(a)(2)(i) through (iii) when recalculating the TSM emission rate.

(17) If you demonstrate compliance with an applicable TSM emission limit through fuel analysis for solid or liquid fuels, and you plan to burn a new type of fuel, you must recalculate the TSM emission rate using Equation 18 of §63.7530 according to the procedures specified in paragraphs (a)(5)(i) through 40 CFR Ch. I (7–1–23 Edition)

(iii) of this section. You are not required to conduct fuel analyses for the fuels described in $\S63.7510(a)(2)(i)$ through (iii). You may exclude the fuels described in $\S63.7510(a)(2)(i)$ through (iii) when recalculating the TSM emission rate.

(i) You must determine the TSM concentration for any new fuel type in units of pounds per million Btu, based on supplier data or your own fuel analysis, according to the provisions in your site-specific fuel analysis plan developed according to §63.7521(b).

(ii) You must determine the new mixture of fuels that will have the highest content of TSM.

(iii) Recalculate the TSM emission rate from your boiler or process heater under these new conditions using Equation 18 of §63.7530. The recalculated TSM emission rate must be less than the applicable emission limit.

(18) If you demonstrate continuous PM emissions compliance with a PM CPMS you will use a PM CPMS to establish a site-specific operating limit corresponding to the results of the performance test demonstrating compliance with the PM limit. You will conduct your performance test using the test method criteria in Table 5 of this subpart. You will use the PM CPMS to demonstrate continuous compliance with this operating limit. You must repeat the performance test annually and reassess and adjust the site-specific operating limit in accordance with the results of the performance test.

(i) To determine continuous compliance, you must record the PM CPMS output data for all periods when the process is operating and the PM CPMS is not out-of-control. You must demonstrate continuous compliance by using all quality-assured hourly average data collected by the PM CPMS for all operating hours to calculate the arithmetic average operating parameter in units of the operating limit (milliamps) on a 30-day rolling average basis.

(ii) For any deviation of the 30-day rolling PM CPMS average value from the established operating parameter limit, you must:

(A) Within 48 hours of the deviation, visually inspect the air pollution control device (APCD);

(B) If inspection of the APCD identifies the cause of the deviation, take corrective action as soon as possible and return the PM CPMS measurement to within the established value; and

(C) Within 30 days of the deviation or at the time of the annual compliance test, whichever comes first, conduct a PM emissions compliance test to determine compliance with the PM emissions limit and to verify or re-establish the CPMS operating limit. You are not required to conduct additional testing for any deviations that occur between the time of the original deviation and the PM emissions compliance test required under this paragraph.

(iii) PM CPMS deviations from the operating limit leading to more than four required performance tests in a 12-month operating period constitute a separate violation of this subpart.

(19) If you choose to comply with the PM filterable emissions limit by using PM CEMS you must install, certify, operate, and maintain a PM CEMS and record the output of the PM CEMS as specified in paragraphs (a)(19)(i) through (vii) of this section. The compliance limit will be expressed as a 30-day rolling average of the numerical emissions limit value applicable for your unit in Table 1 or 2 or Tables 11 through 15 to this subpart.

(i) Install and certify your PM CEMS according to the procedures and requirements in Performance Specification 11—Specifications and Test Procedures for Particulate Matter Continuous Emission Monitoring Systems at Stationary Sources in Appendix B to part 60 of this chapter, using test criteria outlined in Table V of this rule. The reportable measurement output from the PM CEMS must be expressed in units of the applicable emissions limit (e.g., lb/MMBtu, lb/MWh).

(ii) Operate and maintain your PM CEMS according to the procedures and requirements in Procedure 2— Quality Assurance Requirements for Particulate Matter Continuous Emission Monitoring Systems at Stationary Sources in Appendix F to part 60 of this chapter.

(A) You must conduct the relative response audit (RRA) for your PM CEMS at least once annually.

(B) You must conduct the relative correlation audit (RCA) for your PM CEMS at least once every 3 years.

(iii) Collect PM CEMS hourly average output data for all boiler operating hours except as indicated in paragraph(v) of this section.

(iv) Calculate the arithmetic 30-day rolling average of all of the hourly average PM CEMS output data collected during all nonexempt boiler or process heater operating hours.

(v) You must collect data using the PM CEMS at all times the unit is operating and at the intervals specified this paragraph (a), except for periods of monitoring system malfunctions, repairs associated with monitoring system malfunctions, and required monitoring system quality assurance or quality control activities.

(vi) You must use all the data collected during all boiler or process heater operating hours in assessing the compliance with your operating limit except:

(A) Any data collected during monitoring system malfunctions, repairs associated with monitoring system malfunctions, or required monitoring system quality assurance or control activities conducted during monitoring system malfunctions in calculations and report any such periods in your annual deviation report;

(B) Any data collected during periods when the monitoring system is out of control as specified in your site-specific monitoring plan, repairs associated with periods when the monitoring system is out of control, or required monitoring system quality assurance or control activities conducted during out of control periods in calculations used to report emissions or operating levels and report any such periods in your annual deviation report;

(C) Any data recorded during periods of startup or shutdown.

(vii) You must record and make available upon request results of PM CEMS system performance audits, dates and duration of periods when the PM CEMS is out of control to completion of the corrective actions necessary to return the PM CEMS to operation consistent with your site-specific monitoring plan. (b) You must report each instance in which you did not meet each emission limit and operating limit in Tables 1 through 4 or 11 through 15 to this subpart that apply to you. These instances are deviations from the emission limits or operating limits, respectively, in this subpart. These deviations must be reported according to the requirements in §63.7550.

(c) If you elected to demonstrate that the unit meets the specification for mercury for the unit designed to burn gas 1 subcategory, you must follow the sampling frequency specified in paragraphs (c)(1) through (4) of this section and conduct this sampling according to the procedures in 63.7521(f) through (i).

(1) If the initial mercury constituents in the gaseous fuels are measured to be equal to or less than half of the mercury specification as defined in §63.7575, you do not need to conduct further sampling.

(2) If the initial mercury constituents are greater than half but equal to or less than 75 percent of the mercury specification as defined in §63.7575, you will conduct semi-annual sampling. If 6 consecutive semi-annual fuel analyses demonstrate 50 percent or less of the mercury specification, you do not need to conduct further sampling. If any semi-annual sample exceeds 75 percent of the mercury specification, you must return to monthly sampling for that fuel, until 12 months of fuel analyses again are less than 75 percent of the compliance level.

(3) If the initial mercury constituents are greater than 75 percent of the mercury specification as defined in §63.7575, you will conduct monthly sampling. If 12 consecutive monthly fuel analyses demonstrate 75 percent or less of the mercury specification, you may decrease the fuel analysis frequency to semi-annual for that fuel.

(4) If the initial sample exceeds the mercury specification as defined in §63.7575, each affected boiler or process heater combusting this fuel is not part of the unit designed to burn gas 1 subcategory and must be in compliance with the emission and operating limits for the appropriate subcategory. You may elect to conduct additional monthly sampling while complying

40 CFR Ch. I (7–1–23 Edition)

with these emissions and operating limits to demonstrate that the fuel qualifies as another gas 1 fuel. If 12 consecutive monthly fuel analyses samples are at or below the mercury specification as defined in §63.7575, each affected boiler or process heater combusting the fuel can elect to switch back into the unit designed to burn gas 1 subcategory until the mercury specification is exceeded.

(d) For startup and shutdown, you must meet the work practice standards according to items 5 and 6 of Table 3 of this subpart.

[78 FR 7179, Jan. 31, 2013, as amended at 80 FR 72813, Nov. 20, 2015; 87 FR 60846, Oct. 6, 2022]

§63.7541 How do I demonstrate continuous compliance under the emissions averaging provision?

(a) Following the compliance date, the owner or operator must demonstrate compliance with this subpart on a continuous basis by meeting the requirements of paragraphs (a)(1)through (5) of this section.

(1) For each calendar month, demonstrate compliance with the average weighted emissions limit for the existing units participating in the emissions averaging option as determined in §63.7522(f) and (g).

(2) You must maintain the applicable opacity limit according to paragraphs (a)(2)(i) and (ii) of this section.

(i) For each existing unit participating in the emissions averaging option that is equipped with a dry control system and not vented to a common stack, maintain opacity at or below the applicable limit.

(ii) For each group of units participating in the emissions averaging option where each unit in the group is equipped with a dry control system and vented to a common stack that does not receive emissions from non-affected units, maintain opacity at or below the applicable limit at the common stack.

(3) For each existing unit participating in the emissions averaging option that is equipped with a wet scrubber, maintain the 30-day rolling average parameter values at or above the operating limits established during the most recent performance test.

(4) For each existing unit participating in the emissions averaging option that has an approved alternative operating parameter, maintain the 30day rolling average parameter values consistent with the approved monitoring plan.

(5) For each existing unit participating in the emissions averaging option venting to a common stack configuration containing affected units from other subcategories, maintain the appropriate operating limit for each unit as specified in Table 4 to this subpart that applies.

(b) Any instance where the owner or operator fails to comply with the continuous monitoring requirements in paragraphs (a)(1) through (5) of this section is a deviation.

[76 FR 15664, Mar. 21, 2011, as amended at 78 FR 7182, Jan. 31, 2013]

NOTIFICATION, REPORTS, AND RECORDS

§63.7545 What notifications must I submit and when?

(a) You must submit to the Administrator all of the notifications in \S 63.7(b) and (c), 63.8(e), (f)(4) and (6), and 63.9(b) through (h) that apply to you by the dates specified.

(b) As specified in §63.9(b)(2), if you startup your affected source before January 31, 2013, you must submit an Initial Notification not later than 120 days after January 31, 2013, or no later than 120 days after the source becomes subject to this subpart, whichever is later.

(c) As specified in §63.9(b)(4) and (5), if you startup your new or reconstructed affected source on or after January 31, 2013, you must submit an Initial Notification not later than 15 days after the actual date of startup of the affected source. For a new or reconstructed affected source that has reclassified to major source status, you must submit an Initial Notification not later 120 days after the source becomes subject to this subpart.

(d) If you are required to conduct a performance test you must submit a Notification of Intent to conduct a performance test at least 60 days before the performance test is scheduled to begin.

(e) If you are required to conduct an initial compliance demonstration as specified in §63.7530, you must submit a Notification of Compliance Status according to (63.9(h)(2)(ii)). For the initial compliance demonstration for each boiler or process heater, you must submit the Notification of Compliance Status, including all performance test results and fuel analyses, before the close of business on the 60th day following the completion of all performance test and/or other initial compliance demonstrations for all boiler or process heaters at the facility according to §63.10(d)(2). The Notification of Compliance Status report must contain all the information specified in paragraphs (e)(1) through (8) of this section, as applicable. If you are not required to conduct an initial compliance demonstration as specified in §63.7530(a), the Notification of Compliance Status must only contain the information specified in paragraphs (e)(1) and (8) of

specified at §63.7495(b). (1) A description of the affected unit(s) including identification of which subcategories the unit is in, the design heat input capacity of the unit, a description of the add-on controls used on the unit to comply with this subpart, description of the fuel(s) burned, including whether the fuel(s) were a secondary material determined by you or the EPA through a petition process to be a non-waste under §241.3 of this chapter, whether the fuel(s) were a secondary material processed from discarded non-hazardous secondary materials within the meaning of §241.3 of this chapter, and justification for the selection of fuel(s) burned during the compliance demonstration.

this section and must be submitted

within 60 days of the compliance date

(2) Summary of the results of all performance tests and fuel analyses, and calculations conducted to demonstrate initial compliance including all established operating limits, and including:

(i) Identification of whether you are complying with the PM emission limit or the alternative TSM emission limit.

(ii) Identification of whether you are complying with the output-based emission limits or the heat input-based (i.e., lb/MMBtu or ppm) emission limits, (iii) Identification of whether you are complying the arithmetic mean of all valid hours of data from the previous 30 operating days or of the previous 720 hours. This identification shall be specified separately for each operating parameter.

(3) A summary of the maximum CO emission levels recorded during the performance test to show that you have met any applicable emission standard in Table 1 or 2 or Tables 11 through 15 to this subpart, if you are not using a CO CEMS to demonstrate compliance.

(4) Identification of whether you plan to demonstrate compliance with each applicable emission limit through performance testing, a CEMS, or fuel analysis.

(5) Identification of whether you plan to demonstrate compliance by emissions averaging and identification of whether you plan to demonstrate compliance by using efficiency credits through energy conservation:

(i) If you plan to demonstrate compliance by emission averaging, report the emission level that was being achieved or the control technology employed on January 31, 2013.

(ii) [Reserved]

(6) A signed certification that you have met all applicable emission limits and work practice standards.

(7) If you had a deviation from any emission limit, work practice standard, or operating limit, you must also submit a description of the deviation, the duration of the deviation, and the corrective action taken in the Notification of Compliance Status report.

(8) In addition to the information required in $\S63.9(h)(2)$, your notification of compliance status must include the following certification(s) of compliance, as applicable, and signed by a responsible official:

(i) "This facility completed the required initial tune-up for all of the boilers and process heaters covered by 40 CFR part 63 subpart DDDDD at this site according to the procedures in $\S63.7540(a)(10)(i)$ through (vi)."

(ii) "This facility has had an energy assessment performed according to §63.7530(e)."

(iii) Except for units that burn only natural gas, refinery gas, or other gas 1

40 CFR Ch. I (7–1–23 Edition)

fuel, or units that qualify for a statutory exemption as provided in section 129(g)(1) of the Clean Air Act, include the following: "No secondary materials that are solid waste were combusted in any affected unit."

(f) If you operate a unit designed to burn natural gas, refinery gas, or other gas 1 fuels that is subject to this subpart, and you intend to use a fuel other than natural gas, refinery gas, gaseous fuel subject to another subpart of this part, part 60, 61, or 65, or other gas 1 fuel to fire the affected unit during a period of natural gas curtailment or supply interruption, as defined in §63.7575, you must submit a notification of alternative fuel use within 48 hours of the declaration of each period of natural gas curtailment or supply interruption, as defined in §63.7575. The notification must include the information specified in paragraphs (f)(1) through (5) of this section.

(1) Company name and address.

(2) Identification of the affected unit.

(3) Reason you are unable to use natural gas or equivalent fuel, including the date when the natural gas curtailment was declared or the natural gas supply interruption began.

(4) Type of alternative fuel that you intend to use.

(5) Dates when the alternative fuel use is expected to begin and end.

(g) If you intend to commence or recommence combustion of solid waste, you must provide 30 days prior notice of the date upon which you will commence or recommence combustion of solid waste. The notification must identify:

(1) The name of the owner or operator of the affected source, as defined in 63.7490, the location of the source, the boiler(s) or process heater(s) that will commence burning solid waste, and the date of the notice.

(2) The currently applicable subcategories under this subpart.

(3) The date on which you became subject to the currently applicable emission limits.

(4) The date upon which you will commence combusting solid waste.

(h) If you have switched fuels or made a physical change to the boiler or process heater and the fuel switch or

physical change resulted in the applicability of a different subcategory, you must provide notice of the date upon which you switched fuels or made the physical change within 30 days of the switch/change. The notification must identify:

(1) The name of the owner or operator of the affected source, as defined in 63.7490, the location of the source, the boiler(s) and process heater(s) that have switched fuels, were physically changed, and the date of the notice.

(2) The currently applicable subcategory under this subpart.

(3) The date upon which the fuel switch or physical change occurred.

[76 FR 15664, Mar. 21, 2011, as amended at 78
FR 7183, Jan. 31, 2013; 80 FR 72814, Nov. 20, 2015; 85 FR 73913, Nov. 19, 2020; 85 FR 84262, Dec. 28, 2020; 87 FR 60846, Oct. 6, 2022]

§63.7550 What reports must I submit and when?

(a) You must submit each report in Table 9 to this subpart that applies to you.

(b) Unless the EPA Administrator has approved a different schedule for submission of reports under §63.10(a), you must submit each report, according to paragraph (h) of this section, by the date in Table 9 to this subpart and according to the requirements in paragraphs (b)(1) through (4) of this section. For units that are subject only to a requirement to conduct subsequent annual, biennial, or 5-year tune-up according to §63.7540(a)(10), (11), or (12), respectively, and not subject to emission limits or Table 4 operating limits, you may submit only an annual, biennial, or 5-year compliance report, as applicable, as specified in paragraphs (b)(1) through (4) of this section, instead of a semi-annual compliance report.

(1) The first semi-annual compliance report must cover the period beginning on the compliance date that is specified for each boiler or process heater in §63.7495 and ending on June 30 or December 31, whichever date is the first date that occurs at least 180 days after the compliance date that is specified for your source in §63.7495. If submitting an annual, biennial, or 5-year compliance report, the first compliance report must cover the period beginning on the compliance date that is specified for each boiler or process heater in §63.7495 and ending on December 31 within 1, 2, or 5 years, as applicable, after the compliance date that is specified for your source in §63.7495.

(2) The first semi-annual compliance report must be postmarked or submitted no later than July 31 or January 31, whichever date is the first date following the end of the first calendar half after the compliance date that is specified for each boiler or process heater in §63.7495. The first annual, biennial, or 5-year compliance report must be postmarked or submitted no later than January 31.

(3) Each subsequent semi-annual compliance report must cover the semiannual reporting period from January 1 through June 30 or the semiannual reporting period from July 1 through December 31. Annual, biennial, and 5-year compliance reports must cover the applicable 1-, 2-, or 5-year periods from January 1 to December 31.

(4) Each subsequent semi-annual compliance report must be postmarked or submitted no later than July 31 or January 31, whichever date is the first date following the end of the semiannual reporting period. Annual, biennial, and 5-year compliance reports must be postmarked or submitted no later than January 31.

(5) For each affected source that is subject to permitting regulations pursuant to part 70 or part 71 of this chapter, and if the permitting authority has established dates for submitting semiannual reports pursuant to 70.6(a)(3)(iii)(A) or 71.6(a)(3)(iii)(A), you may submit the first and subsequent compliance reports according to the dates the permitting authority has established in the permit instead of according to the dates in paragraphs (b)(1) through (4) of this section.

(c) A compliance report must contain the following information depending on how the facility chooses to comply with the limits set in this rule.

(1) If the facility is subject to the requirements of a tune up you must submit a compliance report with the information in paragraphs (c)(5)(i) through (iii) of this section, (xiv) and (xvii) of this section, and paragraph (c)(5)(iv) of this section for limited-use boiler or process heater.

(2) If you are complying with the fuel analysis you must submit a compliance report with the information in paragraphs (c)(5)(i) through (iii), (vi), (x), (xi), (xiii), (xv), (xvii), (xviii) and paragraph (d) of this section.

(3) If you are complying with the applicable emissions limit with performance testing you must submit a compliance report with the information in (c)(5)(i) through (iii), (vi), (vii), (vii), (ix), (xi), (xii), (xv), (xvii), (xviii) and paragraph (d) of this section.

(4) If you are complying with an emissions limit using a CMS the compliance report must contain the information required in paragraphs (c)(5)(i) through (iii), (v), (vi), (xi) through (xiii), (xv) through (xviii), and paragraph (e) of this section.

(5)(i) Company and Facility name and address.

(ii) Process unit information, emissions limitations, and operating parameter limitations.

(iii) Date of report and beginning and ending dates of the reporting period.

(iv) The total operating time during the reporting period.

(v) If you use a CMS, including CEMS, COMS, or CPMS, you must include the monitoring equipment manufacturer(s) and model numbers and the date of the last CMS certification or audit.

(vi) The total fuel use by each individual boiler or process heater subject to an emission limit within the reporting period, including, but not limited to, a description of the fuel, whether the fuel has received a non-waste determination by the EPA or your basis for concluding that the fuel is not a waste, and the total fuel usage amount with units of measure.

(vii) If you are conducting performance tests once every 3 years consistent with §63.7515(b) or (c), the date of the last 2 performance tests and a statement as to whether there have been any operational changes since the last performance test that could increase emissions.

(viii) A statement indicating that you burned no new types of fuel in an individual boiler or process heater subject to an emission limit. Or, if you did 40 CFR Ch. I (7–1–23 Edition)

burn a new type of fuel and are subject to a HCl emission limit, you must submit the calculation of chlorine input, using Equation 7 of §63.7530, that demonstrates that your source is still within its maximum chlorine input level established during the previous performance testing (for sources that demonstrate compliance through performance testing) or you must submit the calculation of HCl emission rate using Equation 16 of §63.7530 that demonstrates that your source is still meeting the emission limit for HCl emissions (for boilers or process heaters that demonstrate compliance through fuel analysis). If you burned a new type of fuel and are subject to a mercury emission limit, you must submit the calculation of mercury input, using Equation 8 of §63.7530, that demonstrates that your source is still within its maximum mercury input level established during the previous performance testing (for sources that demonstrate compliance through performance testing), or you must submit the calculation of mercury emission rate using Equation 17 of §63.7530 that demonstrates that your source is still meeting the emission limit for mercury emissions (for boilers or process heaters that demonstrate compliance through fuel analysis). If you burned a new type of fuel and are subject to a TSM emission limit, you must submit the calculation of TSM input, using Equation 9 of §63.7530, that demonstrates that your source is still within its maximum TSM input level established during the previous performance testing (for sources that demonstrate compliance through performance testing), or you must submit the calculation of TSM emission rate, using Equation 18 of §63.7530, that demonstrates that your source is still meeting the emission limit for TSM emissions (for boilers or process heaters that demonstrate compliance through fuel analvsis).

(ix) If you wish to burn a new type of fuel in an individual boiler or process heater subject to an emission limit and you cannot demonstrate compliance with the maximum chlorine input operating limit using Equation 7 of §63.7530 or the maximum mercury input operating limit using Equation 8 of §63.7530,

or the maximum TSM input operating limit using Equation 9 of §63.7530 you must include in the compliance report a statement indicating the intent to conduct a new performance test within 60 days of starting to burn the new fuel.

(x) A summary of any monthly fuel analyses conducted to demonstrate compliance according to §§ 63.7521 and 63.7530 for individual boilers or process heaters subject to emission limits, and any fuel specification analyses conducted according to §§ 63.7521(f) and 63.7530(g).

(xi) If there are no deviations from any emission limits or operating limits in this subpart that apply to you, a statement that there were no deviations from the emission limits or operating limits during the reporting period.

(xii) If there were no deviations from the monitoring requirements including no periods during which the CMSs, including CEMS, COMS, and CPMS, were out of control as specified in §63.8(c)(7), a statement that there were no deviations and no periods during which the CMS were out of control during the reporting period.

(xiii) If a malfunction occurred during the reporting period, the report must include the number, duration, and a brief description for each type of malfunction which occurred during the reporting period and which caused or may have caused any applicable emission limitation to be exceeded. The report must also include a description of actions taken by you during a malfunction of a boiler, process heater, or associated air pollution control device or CMS to minimize emissions in accordance with §63.7500(a)(3), including actions taken to correct the malfunction.

(xiv) Include the date of the most recent tune-up for each unit subject to only the requirement to conduct an annual, biennial, or 5-year tune-up according to §63.7540(a)(10), (11), or (12) respectively. Include the date of the most recent burner inspection if it was not done annually, biennially, or on a 5-year period and was delayed until the next scheduled or unscheduled unit shutdown.

(xv) If you plan to demonstrate compliance by emission averaging, certify the emission level achieved or the control technology employed is no less stringent than the level or control technology contained in the notification of compliance status in $\S63.7545(e)(5)(i)$.

(xvi) For each reporting period, the compliance reports must include all of the calculated 30 day rolling average values for CEMS (CO, HCl, SO₂, and mercury), 10 day rolling average values for CO CEMS when the limit is expressed as a 10 day instead of 30 day rolling average, and the PM CPMS data.

(xvii) Statement by a responsible official with that official's name, title, and signature, certifying the truth, accuracy, and completeness of the content of the report.

(xviii) For each instance of startup or shutdown include the information required to be monitored, collected, or recorded according to the requirements of §63.7555(d).

(d) For each deviation from an emission limit or operating limit in this subpart that occurs at an individual boiler or process heater where you are not using a CMS to comply with that emission limit or operating limit, or from the work practice standards for periods if startup and shutdown, the compliance report must additionally contain the information required in paragraphs (d)(1) through (3) of this section.

(1) A description of the deviation and which emission limit, operating limit, or work practice standard from which you deviated.

(2) Information on the number, duration, and cause of deviations (including unknown cause), as applicable, and the corrective action taken.

(3) If the deviation occurred during an annual performance test, provide the date the annual performance test was completed.

(e) For each deviation from an emission limit, operating limit, and monitoring requirement in this subpart occurring at an individual boiler or process heater where you are using a CMS to comply with that emission limit or operating limit, the compliance report must additionally contain the information required in paragraphs (e)(1) through (9) of this section. This includes any deviations from your site-specific monitoring plan as required in §63.7505(d).

(1) The date and time that each deviation started and stopped and description of the nature of the deviation (i.e., what you deviated from).

(2) The date and time that each CMS was inoperative, except for zero (low-level) and high-level checks.

(3) The date, time, and duration that each CMS was out of control, including the information in 63.8(c)(8).

(4) The date and time that each deviation started and stopped.

(5) A summary of the total duration of the deviation during the reporting period and the total duration as a percent of the total source operating time during that reporting period.

(6) A characterization of the total duration of the deviations during the reporting period into those that are due to control equipment problems, process problems, other known causes, and other unknown causes.

(7) A summary of the total duration of CMS's downtime during the reporting period and the total duration of CMS downtime as a percent of the total source operating time during that reporting period.

(8) A brief description of the source for which there was a deviation.

(9) A description of any changes in CMSs, processes, or controls since the last reporting period for the source for which there was a deviation.

(f)-(g) [Reserved]

(h) You must submit the reports according to the procedures specified in paragraphs (h)(1) through (3) of this section.

(1) Within 60 days after the date of completing each performance test (as defined in §63.2) required by this subpart, you must submit the results of the performance tests, including any fuel analyses, following the procedure specified in either paragraph (h)(1)(i) or (ii) of this section.

(i) For data collected using test methods supported by the EPA's Electronic Reporting Tool (ERT) as listed on the EPA's ERT Web site (http:// www.epa.gov/ttn/chief/ert/index.html),

you must submit the results of the performance test to the EPA via the Com-

40 CFR Ch. I (7–1–23 Edition)

pliance and Emissions Data Reporting Interface (CEDRI). (CEDRI can be accessed through the EPA's Central (CDX) Data Exchange (https:// cdx.epa.gov/).) Performance test data must be submitted in a file format generated through use of the EPA's ERT or an electronic file format consistent with the extensible markup language (XML) schema listed on the EPA's ERT Web site. If you claim that some of the performance test information being submitted is confidential business information (CBI), you must submit a complete file generated through the use of the EPA's ERT or an alternate electronic file consistent with the XML schema listed on the EPA's ERT Web site, including information claimed to be CBI, on a compact disc, flash drive, or other commonly used electronic storage media to the EPA. The electronic media must be clearly marked as CBI and mailed to U.S. EPA/OAPQS/ CORE CBI Office, Attention: Group Leader, Measurement Policy Group, MD C404-02, 4930 Old Page Rd., Durham, NC 27703. The same ERT or alternate file with the CBI omitted must be submitted to the EPA via the EPA's CDX as described earlier in this paragraph.

(ii) For data collected using test methods that are not supported by the EPA'S ERT as listed on the EPA'S ERT Web site at the time of the test, you must submit the results of the performance test to the Administrator at the appropriate address listed in §63.13.

(2) Within 60 days after the date of completing each CEMS performance evaluation (as defined in 63.2), you must submit the results of the performance evaluation following the procedure specified in either paragraph (h)(2)(i) or (ii) of this section.

(i) For performance evaluations of continuous monitoring systems measuring relative accuracy test audit (RATA) pollutants that are supported by the EPA's ERT as listed on the EPA's ERT Web site at the time of the evaluation, you must submit the results of the performance evaluation to the EPA via the CEDRI. (CEDRI can be accessed through the EPA's CDX.) Performance evaluation data must be submitted in a file format generated through the use of the EPA's ERT or an alternate file format consistent

with the XML schema listed on the EPA's ERT Web site. If you claim that some of the performance evaluation information being transmitted is CBI, you must submit a complete file generated through the use of the EPA's ERT or an alternate electronic file consistent with the XML schema listed on the EPA's ERT Web site, including information claimed to be CBI, on a compact disc, flash drive, or other commonly used electronic storage media to the EPA. The electronic media must be clearly marked as CBI and mailed to U.S. EPA/OAPQS/CORE CBI Office, Attention: Group Leader, Measurement Policy Group, MD C404-02, 4930 Old Page Rd., Durham, NC 27703. The same ERT or alternate file with the CBI omitted must be submitted to the EPA via the EPA's CDX as described earlier in this paragraph.

(ii) For any performance evaluations of continuous monitoring systems measuring RATA pollutants that are not supported by the EPA's ERT as listed on the ERT Web site at the time of the evaluation, you must submit the results of the performance evaluation to the Administrator at the appropriate address listed in §63.13.

(3) You must submit all reports required by Table 9 of this subpart electronically to the EPA via the CEDRI. (CEDRI can be accessed through the EPA's CDX.) You must use the appropriate electronic report in CEDRI for this subpart. Instead of using the electronic report in CEDRI for this subpart, you may submit an alternate electronic file consistent with the XML schema listed on the CEDRI Web site (http://www.epa.gov/ttn/chief/cedri/

index.html), once the XML schema is available. If the reporting form specific to this subpart is not available in CEDRI at the time that the report is due, you must submit the report to the Administrator at the appropriate address listed in §63.13. You must begin submitting reports via CEDRI no later than 90 days after the form becomes available in CEDRI.

[78 FR 7183, Jan. 31, 2013, as amended at 80 FR 72814, Nov. 20, 2015]

§63.7555 What records must I keep?

(a) You must keep records according to paragraphs (a)(1) and (2) of this section.

(1) A copy of each notification and report that you submitted to comply with this subpart, including all documentation supporting any Initial Notification or Notification of Compliance Status or semiannual compliance report that you submitted, according to the requirements in §63.10(b)(2)(xiv).

(2) Records of performance tests, fuel analyses, or other compliance demonstrations and performance evaluations as required in §63.10(b)(2)(viii).

(3) For units in the limited use subcategory, you must keep a copy of the federally enforceable permit that limits the annual capacity factor to less than or equal to 10 percent and fuel use records for the days the boiler or process heater was operating.

(b) For each CEMS, COMS, and continuous monitoring system you must keep records according to paragraphs (b)(1) through (5) of this section.

(1) Records described in §63.10(b)(2)(vii) through (xi).

(2) Monitoring data for continuous opacity monitoring system during a performance evaluation as required in $\S63.6(h)(7)(i)$ and (ii).

(3) Previous (*i.e.*, superseded) versions of the performance evaluation plan as required in §63.8(d)(3).

(4) Request for alternatives to relative accuracy test for CEMS as required in 63.8(f)(6)(i).

(5) Records of the date and time that each deviation started and stopped.

(c) You must keep the records required in Table 8 to this subpart including records of all monitoring data and calculated averages for applicable operating limits, such as opacity, pressure drop, pH, and operating load, to show continuous compliance with each emission limit and operating limit that applies to you.

(d) For each boiler or process heater subject to an emission limit in Table 1 or 2 or Tables 11 through 15 to this subpart, you must also keep the applicable records in paragraphs (d)(1) through (11) of this section.

§63.7555

(1) You must keep records of monthly fuel use by each boiler or process heater, including the type(s) of fuel and amount(s) used.

(2) If you combust non-hazardous secondary materials that have been determined not to be solid waste pursuant to 241.3(b)(1) and (2) of this chapter, you must keep a record that documents how the secondary material meets each of the legitimacy criteria under §241.3(d)(1) of this chapter. If you combust a fuel that has been processed from a discarded non-hazardous secondary material pursuant to §241.3(b)(4) of this chapter, you must keep records as to how the operations that produced the fuel satisfy the definition of processing in §241.2 of this chapter. If the fuel received a nonwaste determination pursuant to the petition process submitted under §241.3(c) of this chapter, you must keep a record that documents how the fuel satisfies the requirements of the petition process. For operating units that combust non-hazardous secondary materials as fuel per §241.4 of this chapter, you must keep records documenting that the material is listed as a nonwaste under §241.4(a) of this chapter. Units exempt from the incinerator standards under section 129(g)(1) of the Clean Air Act because they are qualifying facilities burning a homogeneous waste stream do not need to maintain the records described in this paragraph (d)(2).

(3) A copy of all calculations and supporting documentation of maximum chlorine fuel input, using Equation 7 of §63.7530, that were done to demonstrate continuous compliance with the HCl emission limit, for sources that demonstrate compliance through performance testing. For sources that demonstrate compliance through fuel analysis, a copy of all calculations and supporting documentation of HCl emission rates, using Equation 16 of §63.7530, that were done to demonstrate compliance with the HCl emission limit. Supporting documentation should include results of any fuel analyses and basis for the estimates of maximum chlorine fuel input or HCl emission rates. You can use the results from one fuel analysis for multiple boilers and process heaters provided they are all burning

40 CFR Ch. I (7–1–23 Edition)

the same fuel type. However, you must calculate chlorine fuel input, or HCl emission rate, for each boiler and process heater.

(4) A copy of all calculations and supporting documentation of maximum mercury fuel input, using Equation 8 of §63.7530, that were done to demonstrate continuous compliance with the mercurv emission limit for sources that demonstrate compliance through performance testing. For sources that demonstrate compliance through fuel analysis, a copy of all calculations and supporting documentation of mercury emission rates, using Equation 17 of §63.7530, that were done to demonstrate compliance with the mercury emission limit. Supporting documentation should include results of any fuel analyses and basis for the estimates of maximum mercury fuel input or mercury emission rates. You can use the results from one fuel analysis for multiple boilers and process heaters provided they are all burning the same fuel type. However, you must calculate mercury fuel input, or mercury emission rates, for each boiler and process heater.

(5) If, consistent with §63.7515(b), you choose to stack test less frequently than annually, you must keep a record that documents that your emissions in the previous stack test(s) were less than 75 percent of the applicable emission limit (or, in specific instances noted in Tables 1 and 2 or 11 through 15 to this subpart, less than the applicable emission limit), and document that there was no change in source operations including fuel composition and operation of air pollution control equipment that would cause emissions of the relevant pollutant to increase within the past year.

(6) Records of the occurrence and duration of each malfunction of the boiler or process heater, or of the associated air pollution control and monitoring equipment.

(7) Records of actions taken during periods of malfunction to minimize emissions in accordance with the general duty to minimize emissions in $\S63.7500(a)(3)$, including corrective actions to restore the malfunctioning boiler or process heater, air pollution control, or monitoring equipment to

its normal or usual manner of operation.

(8) A copy of all calculations and supporting documentation of maximum TSM fuel input, using Equation 9 of §63.7530, that were done to demonstrate continuous compliance with the TSM emission limit for sources that demonstrate compliance through performance testing. For sources that demonstrate compliance through fuel analysis, a copy of all calculations and supporting documentation of TSM emission rates, using Equation 18 of §63.7530, that were done to demonstrate compliance with the TSM emission Supporting documentation limit. should include results of any fuel analyses and basis for the estimates of maximum TSM fuel input or TSM emission rates. You can use the results from one fuel analysis for multiple boilers and process heaters provided they are all burning the same fuel type. However, you must calculate TSM fuel input, or TSM emission rates, for each boiler and process heater.

(9) You must maintain records of the calendar date, time, occurrence and duration of each startup and shutdown.

(10) You must maintain records of the type(s) and amount(s) of fuels used during each startup and shutdown.

(11) For each startup period, for units selecting paragraph (2) of the definition of "startup" in §63.7575 you must maintain records of the time that clean fuel combustion begins; the time when you start feeding fuels that are not clean fuels; the time when useful thermal energy is first supplied; and the time when the PM controls are engaged.

(12) If you choose to rely on paragraph (2) of the definition of "startup" in §63.7575, for each startup period, you must maintain records of the hourly steam temperature, hourly steam pressure, hourly steam flow, hourly flue gas temperature, and all hourly average CMS data (e.g., CEMS, PM CPMS, COMS, ESP total secondary electric power input, scrubber pressure drop, scrubber liquid flow rate) collected during each startup period to confirm that the control devices are engaged. In addition, if compliance with the PM emission limit is demonstrated using a PM control device, you must maintain records as specified in paragraphs (d)(12)(i) through (iii) of this section.

(i) For a boiler or process heater with an electrostatic precipitator, record the number of fields in service, as well as each field's secondary voltage and secondary current during each hour of startup.

(ii) For a boiler or process heater with a fabric filter, record the number of compartments in service, as well as the differential pressure across the baghouse during each hour of startup.

(iii) For a boiler or process heater with a wet scrubber needed for filterable PM control, record the scrubber's liquid flow rate and the pressure drop during each hour of startup.

(13) If you choose to use paragraph (2) of the definition of "startup" in §63.7575 and you find that you are unable to safely engage and operate your PM control(s) within 1 hour of first firing of non-clean fuels, you may choose to rely on paragraph (1) of definition of "startup" in §63.7575 or you may submit to the delegated permitting authority a request for a variance with the PM controls requirement, as described below.

(i) The request shall provide evidence of a documented manufacturer-identified safety issue.

(ii) The request shall provide information to document that the PM control device is adequately designed and sized to meet the applicable PM emission limit.

(iii) In addition, the request shall contain documentation that:

(A) The unit is using clean fuels to the maximum extent possible to bring the unit and PM control device up to the temperature necessary to alleviate or prevent the identified safety issues prior to the combustion of primary fuel:

(B) The unit has explicitly followed the manufacturer's procedures to alleviate or prevent the identified safety issue; and

(C) Identifies with specificity the details of the manufacturer's statement of concern.

(iv) You must comply with all other work practice requirements, including but not limited to data collection, recordkeeping, and reporting requirements. (e) If you elect to average emissions consistent with §63.7522, you must additionally keep a copy of the emission averaging implementation plan required in §63.7522(g), all calculations required under §63.7522, including monthly records of heat input or steam generation, as applicable, and monitoring records consistent with §63.7541.

(f) If you elect to use efficiency credits from energy conservation measures to demonstrate compliance according to 63.7533, you must keep a copy of the Implementation Plan required in 63.7533(d) and copies of all data and calculations used to establish credits according to 63.7533(b), (c), and (f).

(g) If you elected to demonstrate that the unit meets the specification for mercury for the unit designed to burn gas 1 subcategory, you must maintain monthly records (or at the frequency required by $\S63.7540(c)$) of the calculations and results of the fuel specification for mercury in Table 6.

(h) If you operate a unit in the unit designed to burn gas 1 subcategory that is subject to this subpart, and you use an alternative fuel other than natural gas, refinery gas, gaseous fuel subject to another subpart under this part, other gas 1 fuel, or gaseous fuel subject to another subpart of this part or part 60, 61, or 65, you must keep records of the total hours per calendar year that alternative fuel is burned and the total hours per calendar year that the unit operated during periods of gas curtailment or gas supply emergencies.

[76 FR 15664, Mar. 21, 2011, as amended at 78 FR 7185, Jan. 31, 2013; 80 FR 72816, Nov. 20, 2015; 87 FR 60846, Oct. 6, 2022]

§63.7560 In what form and how long must I keep my records?

(a) Your records must be in a form suitable and readily available for expeditious review, according to §63.10(b)(1).

(b) As specified in §63.10(b)(1), you must keep each record for 5 years following the date of each occurrence, measurement, maintenance, corrective action, report, or record.

(c) You must keep each record on site, or they must be accessible from on site (for example, through a computer network), for at least 2 years after the date of each occurrence,

40 CFR Ch. I (7–1–23 Edition)

measurement, maintenance, corrective action, report, or record, according to §63.10(b)(1). You can keep the records off site for the remaining 3 years.

OTHER REQUIREMENTS AND INFORMATION

§63.7565 What parts of the General Provisions apply to me?

Table 10 to this subpart shows which parts of the General Provisions in §§ 63.1 through 63.15 apply to you.

§63.7570 Who implements and enforces this subpart?

(a) This subpart can be implemented and enforced by the EPA, or an Administrator such as your state, local, or tribal agency. If the EPA Administrator has delegated authority to your state, local, or tribal agency, then that agency (as well as the EPA) has the authority to implement and enforce this subpart. You should contact your EPA Regional Office to find out if this subpart is delegated to your state, local, or tribal agency.

(b) In delegating implementation and enforcement authority of this subpart to a state, local, or tribal agency under 40 CFR part 63, subpart E, the authorities listed in paragraphs (b)(1) through (4) of this section are retained by the EPA Administrator and are not transferred to the state, local, or tribal agency, however, the EPA retains oversight of this subpart and can take enforcement actions, as appropriate.

(1) Approval of alternatives to the emission limits and work practice standards in §63.7500(a) and (b) under §63.6(g), except as specified in §63.7555(d)(13).

(2) Approval of major change to test methods in Table 5 to this subpart under 63.7(e)(2)(i) and (f) and as defined in 63.90, and alternative analytical methods requested under 63.7521(b)(2).

(3) Approval of major change to monitoring under 63.8(f) and as defined in 63.90, and approval of alternative operating parameters under 63.7500(a)(2)and 63.7522(g)(2).
(4) Approval of major change to recordkeeping and reporting under §63.10(e) and as defined in §63.90.

[76 FR 15664, Mar. 21, 2011, as amended at 78 FR 7186, Jan. 31, 2013; 80 FR 72817, Nov. 20, 2015]

§63.7575 What definitions apply to this subpart?

Terms used in this subpart are defined in the Clean Air Act, in §63.2 (the General Provisions), and in this section as follows:

10-day rolling average means the arithmetic mean of the previous 240 hours of valid operating data. Valid data excludes hours during startup and shutdown, data collected during periods when the monitoring system is out of control as specified in your site-specific monitoring plan, while conducting repairs associated with periods when the monitoring system is out of control, or while conducting required monitoring system quality assurance or quality control activities, and periods when this unit is not operating. The 240 hours should be consecutive, but not necessarily continuous if operations were intermittent.

12-month rolling average means the arithmetic mean of the previous 12 months of valid fuel analysis data. The 12 months should be consecutive, but not necessarily continuous if operations were intermittent.

30-day rolling average means the arithmetic mean of the previous 720 hours of valid CO CEMS data. The 720 hours should be consecutive, but not necessarily continuous if operations were intermittent. For parameters other than CO, 30-day rolling average means either the arithmetic mean of all valid hours of data from 30 successive operating days or the arithmetic mean of the previous 720 hours of valid operating data. Valid data excludes hours during startup and shutdown, data collected during periods when the monitoring system is out of control as specified in your site-specific monitoring plan, while conducting repairs associated with periods when the monitoring system is out of control, or while conducting required monitoring system quality assurance or quality control activities, and periods when this unit is not operating.

Annual capacity factor means the ratio between the actual heat input to a boiler or process heater from the fuels burned during a calendar year and the potential heat input to the boiler or process heater had it been operated for 8,760 hours during a year at the maximum steady state design heat input capacity.

Annual heat input means the heat input for the 12 months preceding the compliance demonstration.

Average annual heat input rate means total heat input divided by the hours of operation for the 12 months preceding the compliance demonstration.

Bag leak detection system means a group of instruments that are capable of monitoring particulate matter loadings in the exhaust of a fabric filter (*i.e.*, baghouse) in order to detect bag failures. A bag leak detection system includes, but is not limited to, an instrument that operates on electrodynamic, triboelectric, light scattering, light transmittance, or other principle to monitor relative particulate matter loadings.

Benchmark means the fuel heat input for a boiler or process heater for the one-year period before the date that an energy demand reduction occurs, unless it can be demonstrated that a different time period is more representative of historical operations.

Biodiesel means a mono-alkyl ester derived from biomass and conforming to ASTM D6751-11b, Standard Specification for Biodiesel Fuel Blend Stock (B100) for Middle Distillate Fuels (incorporated by reference, see §63.14).

Biomass or bio-based solid fuel means any biomass-based solid fuel that is not a solid waste. This includes, but is not limited to, wood residue; wood products (e.g., trees, tree stumps, tree limbs, bark, lumber, sawdust, sander dust, chips, scraps, slabs, millings, and shavings); animal manure, including litter and other bedding materials; vegetative agricultural and silvicultural materials, such as logging residues (slash), nut and grain hulls and chaff (e.g., almond, walnut, peanut, rice, and wheat), bagasse, orchard prunings, corn stalks, coffee bean hulls and grounds. This definition of biomass is not intended to suggest that these materials are or are not solid waste.

§63.7575

Blast furnace gas fuel-fired boiler or process heater means an industrial/commercial/institutional boiler or process heater that receives 90 percent or more of its total annual gas volume from blast furnace gas.

Boiler means an enclosed device using controlled flame combustion and having the primary purpose of recovering thermal energy in the form of steam or hot water. Controlled flame combustion refers to a steady-state, or near steady-state, process wherein fuel and/ or oxidizer feed rates are controlled. A device combusting solid waste, as defined in §241.3 of this chapter, is not a boiler unless the device is exempt from the definition of a solid waste incineration unit as provided in section 129(g)(1) of the Clean Air Act. Waste heat boilers are excluded from this definition.

Boiler system means the boiler and associated components, such as, the feed water system, the combustion air system, the fuel system (including burners), blowdown system, combustion control systems, steam systems, and condensate return systems.

Calendar year means the period between January 1 and December 31, inclusive, for a given year.

Clean dry biomass means any biomassbased solid fuel that have not been painted, pigment-stained, or pressure treated, does not contain contaminants at concentrations not normally associated with virgin biomass materials and has a moisture content of less than 20 percent and is not a solid waste.

Coal means all solid fuels classifiable as anthracite, bituminous, sub-bituminous, or lignite by ASTM D388 (incorporated by reference, see §63.14), coal refuse, and petroleum coke. For the purposes of this subpart, this definition of "coal" includes synthetic fuels derived from coal, including but not limited to, solvent-refined coal, coal-oil mixtures, and coal-water mixtures. Coal derived gases are excluded from this definition.

Coal refuse means any by-product of coal mining or coal cleaning operations with an ash content greater than 50 percent (by weight) and a heating value less than 13,900 kilojoules per kilogram (6,000 Btu per pound) on a dry basis. 40 CFR Ch. I (7–1–23 Edition)

Commercial/institutional boiler means a boiler used in commercial establishments or institutional establishments such as medical centers, nursing homes, research centers, institutions of higher education, elementary and secondary schools, libraries, religious establishments, governmental buildings, hotels, restaurants, and laundries to provide electricity, steam, and/or hot water.

Common stack means the exhaust of emissions from two or more affected units through a single flue. Affected units with a common stack may each have separate air pollution control systems located before the common stack, or may have a single air pollution control system located after the exhausts come together in a single flue.

Cost-effective energy conservation measure means a measure that is implemented to improve the energy efficiency of the boiler or facility that has a payback (return of investment) period of 2 years or less.

Daily block average means the arithmetic mean of all valid emission concentrations or parameter levels recorded when a unit is operating measured over the 24-hour period from 12 a.m. (midnight) to 12 a.m. (midnight), except for periods of startup and shutdown or downtime.

Deviation. (1) Deviation means any instance in which an affected source subject to this subpart, or an owner or operator of such a source:

(i) Fails to meet any applicable requirement or obligation established by this subpart including, but not limited to, any emission limit, operating limit, or work practice standard; or

(ii) Fails to meet any term or condition that is adopted to implement an applicable requirement in this subpart and that is included in the operating permit for any affected source required to obtain such a permit.

(2) A deviation is not always a violation.

Dioxins/furans means tetra- through octa-chlorinated dibenzo-p-dioxins and dibenzofurans.

Distillate oil means fuel oils that contain 0.05 weight percent nitrogen or less and comply with the specifications for fuel oil numbers 1 and 2, as defined by the American Society of Testing

and Materials in ASTM D396 (incorporated by reference, see §63.14) or diesel fuel oil numbers 1 and 2, as defined by the American Society for Testing and Materials in ASTM D975 (incorporated by reference, see §63.14), kerosene, and biodiesel as defined by the American Society of Testing and Materials in ASTM D6751-11b (incorporated by reference, see §60.14).

Dry scrubber means an add-on air pollution control system that injects dry alkaline sorbent (dry injection) or sprays an alkaline sorbent (spray dryer) to react with and neutralize acid gas in the exhaust stream forming a dry powder material. Sorbent injection systems used as control devices in fluidized bed boilers and process heaters are included in this definition. A dry scrubber is a dry control system.

Dutch oven means a unit having a refractory-walled cell connected to a conventional boiler setting. Fuel materials are introduced through an opening in the roof of the dutch oven and burn in a pile on its floor. Fluidized bed boilers are not part of the dutch oven design category.

Efficiency credit means emission reductions above those required by this subpart. Efficiency credits generated may be used to comply with the emissions limits. Credits may come from pollution prevention projects that result in reduced fuel use by affected units. Boilers that are shut down cannot be used to generate credits unless the facility provides documentation linking the permanent shutdown to implementation of the energy conservation measures identified in the energy assessment.

Electric utility steam generating unit (EGU) means a fossil fuel-fired combustion unit of more than 25 megawatts electric (MWe) that serves a generator that produces electricity for sale. A fossil fuel-fired unit that cogenerates steam and electricity and supplies more than one-third of its potential electric output capacity and more than 25 MWe output to any utility power distribution system for sale is considered an electric utility steam generating unit. To be "capable of combusting" fossil fuels, an EGU would need to have these fuels allowed in their operating permits and have the

appropriate fuel handling facilities onsite or otherwise available (e.g., coal handling equipment, including coal storage area, belts and conveyers, pulverizers, etc.; oil storage facilities). In addition, fossil fuel-fired EGU means any EGU that fired fossil fuel for more than 10.0 percent of the average annual heat input in any 3 consecutive calendar years or for more than 15.0 percent of the annual heat input during any one calendar year after April 16, 2012.

Electrostatic precipitator (ESP) means an add-on air pollution control device used to capture particulate matter by charging the particles using an electrostatic field, collecting the particles using a grounded collecting surface, and transporting the particles into a hopper. An electrostatic precipitator is usually a dry control system.

Energy assessment means the following for the emission units covered by this subpart:

(1) The energy assessment for facilities with affected boilers and process heaters with a combined heat input capacity of less than 0.3 trillion Btu (TBtu) per year will be 8 on-site technical labor hours in length maximum, but may be longer at the discretion of the owner or operator of the affected source. The boiler system(s), process heater(s), and any on-site energy use system(s) accounting for at least 50 percent of the affected boiler(s) energy (e.g., steam, hot water, process heat, or electricity) production, as applicable, will be evaluated to identify energy savings opportunities, within the limit of performing an 8-hour on-site energy assessment.

(2) The energy assessment for facilities with affected boilers and process heaters with a combined heat input capacity of 0.3 to 1.0 TBtu/year will be 24 on-site technical labor hours in length maximum, but may be longer at the discretion of the owner or operator of the affected source. The boiler system(s), process heater(s), and any onsite energy use system(s) accounting for at least 33 percent of the energy (e.g., steam, hot water, process heat, or electricity) production, as applicable, will be evaluated to identify energy savings opportunities, within the limit of performing a 24-hour on-site energy assessment.

(3) The energy assessment for facilities with affected boilers and process heaters with a combined heat input capacity greater than 1.0 TBtu/year will be up to 24 on-site technical labor hours in length for the first TBtu/yr plus 8 on-site technical labor hours for every additional 1.0 TBtu/yr not to exceed 160 on-site technical hours, but may be longer at the discretion of the owner or operator of the affected source. The boiler system(s), process heater(s), and any on-site energy use system(s) accounting for at least 20 percent of the energy (e.g., steam, process heat, hot water, or electricity) production, as applicable, will be evaluated to identify energy savings opportunities.

(4) The on-site energy use systems serving as the basis for the percent of affected boiler(s) and process heater(s) energy production in paragraphs (1), (2), and (3) of this definition may be segmented by production area or energy use area as most logical and applicable to the specific facility being assessed (*e.g.*, product X manufacturing area; product Y drying area; Building Z).

Energy management practices means the set of practices and procedures designed to manage energy use that are demonstrated by the facility's energy policies, a facility energy manager and other staffing responsibilities, energy performance measurement and tracking methods, an energy saving goal, action plans, operating procedures, internal reporting requirements, and periodic review intervals used at the facility.

Energy management program means a program that includes a set of practices and procedures designed to manage energy use that are demonstrated by the facility's energy policies, a facility energy manager and other staffing responsibilities, energy performance measurement and tracking methods, an energy saving goal, action plans, operating procedures, internal reporting requirements, and periodic review intervals used at the facility. Facilities may establish their program through energy management systems compatible with ISO 50001.

40 CFR Ch. I (7–1–23 Edition)

Energy use system includes the following systems located on-site that use energy (steam, hot water, or electricity) provided by the affected boiler or process heater: process heating; compressed air systems; machine drive (motors, pumps, fans); process cooling; facility heating, ventilation, and airconditioning systems; hot water systems; building envelop; and lighting; or other systems that use steam, hot water, process heat, or electricity provided by the affected boiler or process heater. Energy use systems are only those systems using energy clearly produced by affected boilers and process heaters.

Equivalent means the following only as this term is used in Table 6 to this subpart:

(1) An equivalent sample collection procedure means a published voluntary consensus standard or practice (VCS) or EPA method that includes collection of a minimum of three composite fuel samples, with each composite consisting of a minimum of three increments collected at approximately equal intervals over the test period.

(2) An equivalent sample compositing procedure means a published VCS or EPA method to systematically mix and obtain a representative subsample (part) of the composite sample.

(3) An equivalent sample preparation procedure means a published VCS or EPA method that: Clearly states that the standard, practice or method is appropriate for the pollutant and the fuel matrix; or is cited as an appropriate sample preparation standard, practice or method for the pollutant in the chosen VCS or EPA determinative or analytical method.

(4) An equivalent procedure for determining heat content means a published VCS or EPA method to obtain gross calorific (or higher heating) value.

(5) An equivalent procedure for determining fuel moisture content means a published VCS or EPA method to obtain moisture content. If the sample analysis plan calls for determining metals (especially the mercury, selenium, or arsenic) using an aliquot of the dried sample, then the drying temperature must be modified to prevent vaporizing these metals. On the other hand, if metals analysis is done on an

"as received" basis, a separate aliquot can be dried to determine moisture content and the metals concentration mathematically adjusted to a dry basis.

(6) An equivalent pollutant (mercury, HCl) determinative or analytical procedure means a published VCS or EPA method that clearly states that the standard, practice, or method is appropriate for the pollutant and the fuel matrix and has a published detection limit equal or lower than the methods listed in Table 6 to this subpart for the same purpose.

Fabric filter means an add-on air pollution control device used to capture particulate matter by filtering gas streams through filter media, also known as a baghouse. A fabric filter is a dry control system.

Federally enforceable means all limitations and conditions that are enforceable by the EPA Administrator, including, but not limited to, the requirements of 40 CFR parts 60, 61, 63, and 65, requirements within any applicable state implementation plan, and any permit requirements established under 40 CFR 52.21 or under 40 CFR 51.18 and 40 CFR 51.24.

Fluidized bed boiler means a boiler utilizing a fluidized bed combustion process that is not a pulverized coal boiler.

Fluidized bed boiler with an integrated fluidized bed heat exchanger means a boiler utilizing a fluidized bed combustion where the entire tube surface area is located outside of the furnace section at the exit of the cyclone section and exposed to the flue gas stream for conductive heat transfer. This design applies only to boilers in the unit designed to burn coal/solid fossil fuel subcategory that fire coal refuse.

Fluidized bed combustion means a process where a fuel is burned in a bed of granulated particles, which are maintained in a mobile suspension by the forward flow of air and combustion products.

Fossil fuel means natural gas, oil, coal, and any form of solid, liquid, or gaseous fuel derived from such material.

Fuel cell means a boiler type in which the fuel is dropped onto suspended fixed grates and is fired in a pile. The refractory-lined fuel cell uses combustion air preheating and positioning of secondary and tertiary air injection ports to improve boiler efficiency. Fluidized bed, dutch oven, pile burner, hybrid suspension grate, and suspension burners are not part of the fuel cell subcategory.

Fuel type means each category of fuels that share a common name or classification. Examples include, but are not limited to, bituminous coal, sub-bituminous coal, lignite, anthracite, biomass, distillate oil, residual oil. Individual fuel types received from different suppliers are not considered new fuel types.

Gaseous fuel includes, but is not limited to, natural gas, process gas, landfill gas, coal derived gas, refinery gas, and biogas. Blast furnace gas and process gases that are regulated under another subpart of this part, or part 60, part 61, or part 65 of this chapter, are exempted from this definition.

Heat input means heat derived from combustion of fuel in a boiler or process heater and does not include the heat input from preheated combustion air, recirculated flue gases, returned condensate, or exhaust gases from other sources such as gas turbines, internal combustion engines, kilns, etc.

Heavy liquid includes residual oil and any other liquid fuel not classified as a light liquid.

Hourly average means the arithmetic average of at least four CMS data values representing the four 15-minute periods in an hour, or at least two 15minute data values during an hour when CMS calibration, quality assurance, or maintenance activities are being performed.

Hot water heater means a closed vessel with a capacity of no more than 120 U.S. gallons in which water is heated by combustion of gaseous, liquid, or biomass/bio-based solid fuel and is withdrawn for use external to the vessel. Hot water boilers (i.e., not generating steam) combusting gaseous, liquid, or biomass fuel with a heat input capacity of less than 1.6 million Btu per hour are included in this definition. The 120 U.S. gallon capacity threshold to be considered a hot water heater is independent of the 1.6 MMBtu/hr heat input capacity threshold for hot water boilers. Hot water heater also means a tankless unit that provides on demand hot water.

Hybrid suspension grate boiler means a boiler designed with air distributors to spread the fuel material over the entire width and depth of the boiler combustion zone. The biomass fuel combusted in these units exceeds a moisture content of 40 percent on an as-fired annual heat input basis as demonstrated by monthly fuel analysis. The drying and much of the combustion of the fuel takes place in suspension, and the combustion is completed on the grate or floor of the boiler. Fluidized bed, dutch oven, and pile burner designs are not part of the hybrid suspension grate boiler design category.

Industrial boiler means a boiler used in manufacturing, processing, mining, and refining or any other industry to provide steam, hot water, and/or electricity.

Light liquid includes distillate oil, biodiesel, or vegetable oil.

Limited-use boiler or process heater means any boiler or process heater that burns any amount of solid, liquid, or gaseous fuels and has a federally enforceable annual capacity factor of no more than 10 percent.

Liquid fuel includes, but is not limited to, light liquid, heavy liquid, any form of liquid fuel derived from petroleum, used oil, liquid biofuels, biodiesel, and vegetable oil.

Load fraction means the actual heat input of a boiler or process heater divided by heat input during the performance test that established the minimum sorbent injection rate or minimum activated carbon injection rate, expressed as a fraction (e.g., for 50percent load the load fraction is 0.5). For boilers and process heaters that cofire natural gas or refinery gas with a solid or liquid fuel, the load fraction is determined by the actual heat input of the solid or liquid fuel divided by heat input of the solid or liquid fuel fired during the performance test (e.g., if the performance test was conducted at 100 percent solid fuel firing, for 100 percent load firing 50 percent solid fuel and 50 percent natural gas the load fraction is 0.5).

Major source for oil and natural gas production facilities, as used in this sub40 CFR Ch. I (7–1–23 Edition)

part, shall have the same meaning as in §63.2, except that:

(1) Emissions from any oil or gas exploration or production well (with its associated equipment, as defined in this section), and emissions from any pipeline compressor station or pump station shall not be aggregated with emissions from other similar units to determine whether such emission points or stations are major sources, even when emission points are in a contiguous area or under common control;

(2) Emissions from processes, operations, or equipment that are not part of the same facility, as defined in this section, shall not be aggregated; and

(3) For facilities that are production field facilities, only HAP emissions from glycol dehydration units and storage vessels with the potential for flash emissions shall be aggregated for a major source determination. For facilities that are not production field facilities, HAP emissions from all HAP emission units shall be aggregated for a major source determination.

Metal process furnaces are a subcategory of process heaters, as defined in this subpart, which include natural gas-fired annealing furnaces, preheat furnaces, reheat furnaces, aging furnaces, heat treat furnaces, and homogenizing furnaces.

Million Btu (MMBtu) means one million British thermal units.

Minimum activated carbon injection rate means load fraction multiplied by the lowest hourly average activated carbon injection rate measured according to Table 7 to this subpart during the most recent performance test demonstrating compliance with the applicable emission limit.

Minimum oxygen level means the lowest hourly average oxygen level measured according to Table 7 to this subpart during the most recent performance test demonstrating compliance with the applicable emission limit.

Minimum pressure drop means the lowest hourly average pressure drop measured according to Table 7 to this subpart during the most recent performance test demonstrating compliance with the applicable emission limit.

Minimum scrubber effluent pH means the lowest hourly average sorbent liquid pH measured at the inlet to the wet

scrubber according to Table 7 to this subpart during the most recent performance test demonstrating compliance with the applicable hydrogen chloride emission limit.

Minimum scrubber liquid flow rate means the lowest hourly average liquid flow rate (e.g., to the PM scrubber or to the acid gas scrubber) measured according to Table 7 to this subpart during the most recent performance stack test demonstrating compliance with the applicable emission limit.

Minimum scrubber pressure drop means the lowest hourly average scrubber pressure drop measured according to Table 7 to this subpart during the most recent performance test demonstrating compliance with the applicable emission limit.

Minimum sorbent injection rate means:

(1) The load fraction multiplied by the lowest hourly average sorbent injection rate for each sorbent measured according to Table 7 to this subpart during the most recent performance test demonstrating compliance with the applicable emission limits; or

(2) For fluidized bed combustion not using an acid gas wet scrubber or dry sorbent injection control technology to comply with the HCl emission limit, the lowest average ratio of sorbent to sulfur measured during the most recent performance test.

Minimum total secondary electric power means the lowest hourly average total secondary electric power determined from the values of secondary voltage and secondary current to the electrostatic precipitator measured according to Table 7 to this subpart during the most recent performance test demonstrating compliance with the applicable emission limits.

Natural gas means:

(1) A naturally occurring mixture of hydrocarbon and nonhydrocarbon gases found in geologic formations beneath the earth's surface, of which the principal constituent is methane; or

(2) Liquefied petroleum gas, as defined in ASTM D1835 (incorporated by reference, see §63.14); or

(3) A mixture of hydrocarbons that maintains a gaseous state at ISO conditions. Additionally, natural gas must either be composed of at least 70 percent methane by volume or have a gross calorific value between 35 and 41 megajoules (MJ) per dry standard cubic meter (950 and 1,100 Btu per dry standard cubic foot); or

(4) Propane or propane derived synthetic natural gas. Propane means a colorless gas derived from petroleum and natural gas, with the molecular structure C_3H_8 .

Opacity means the degree to which emissions reduce the transmission of light and obscure the view of an object in the background.

Operating day means a 24-hour period between 12 midnight and the following midnight during which any fuel is combusted at any time in the boiler or process heater unit. It is not necessary for fuel to be combusted for the entire 24-hour period. For calculating rolling average emissions, an operating day does not include the hours of operation during startup or shutdown.

Other combustor means a unit designed to burn solid fuel that is not classified as a dutch oven, fluidized bed, fuel cell, hybrid suspension grate boiler, pulverized coal boiler, stoker, sloped grate, or suspension boiler as defined in this subpart.

Other gas 1 fuel means a gaseous fuel that is not natural gas or refinery gas and does not exceed a maximum mercury concentration of 40 micrograms/ cubic meters of gas.

Oxygen analyzer system means all equipment required to determine the oxygen content of a gas stream and used to monitor oxygen in the boiler or process heater flue gas, boiler or process heater, firebox, or other appropriate location. This definition includes oxygen trim systems. The source owner or operator must install, calibrate, maintain, and operate the oxygen analyzer system in accordance with the manufacturer's recommendations.

Oxygen trim system means a system of monitors that is used to maintain excess air at the desired level in a combustion device over its operating load range. A typical system consists of a flue gas oxygen and/or CO monitor that automatically provides a feedback signal to the combustion air controller or draft controller.

Particulate matter (PM) means any finely divided solid or liquid material,

other than uncombined water, as measured by the test methods specified under this subpart, or an approved alternative method.

Period of gas curtailment or supply interruption means a period of time during which the supply of gaseous fuel to an affected boiler or process heater is restricted or halted for reasons beyond the control of the facility. The act of entering into a contractual agreement with a supplier of natural gas established for curtailment purposes does not constitute a reason that is under the control of a facility for the purposes of this definition. An increase in the cost or unit price of natural gas due to normal market fluctuations not during periods of supplier delivery restriction does not constitute a period of natural gas curtailment or supply interruption. On-site gaseous fuel system emergencies or equipment failures qualify as periods of supply interruption when the emergency or failure is beyond the control of the facility.

Pile burner means a boiler design incorporating a design where the anticipated biomass fuel has a high relative moisture content. Grates serve to support the fuel, and underfire air flowing up through the grates provides oxygen for combustion, cools the grates, promotes turbulence in the fuel bed, and fires the fuel. The most common form of pile burning is the dutch oven.

Process heater means an enclosed device using controlled flame, and the unit's primary purpose is to transfer heat indirectly to a process material (liquid, gas, or solid) or to a heat transfer material (e.g., glycol or a mixture of glycol and water) for use in a process unit, instead of generating steam. Process heaters are devices in which the combustion gases do not come into direct contact with process materials. A device combusting solid waste, as defined in §241.3 of this chapter, is not a process heater unless the device is exempt from the definition of a solid waste incineration unit as provided in section 129(g)(1) of the Clean Air Act. Process heaters do not include units used for comfort heat or space heat, food preparation for on-site consumption, or autoclaves. Waste heat process heaters are excluded from this definition.

40 CFR Ch. I (7–1–23 Edition)

Pulverized coal boiler means a boiler in which pulverized coal or other solid fossil fuel is introduced into an air stream that carries the coal to the combustion chamber of the boiler where it is fired in suspension.

Qualified energy assessor means:

(1) Someone who has demonstrated capabilities to evaluate energy savings opportunities for steam generation and major energy using systems, including, but not limited to:

(i) Boiler combustion management.

(ii) Boiler thermal energy recovery, including

(A) Conventional feed water economizer,

(B) Conventional combustion air preheater, and

(C) Condensing economizer.

(iii) Boiler blowdown thermal energy recovery.

(iv) Primary energy resource selection, including

(A) Fuel (primary energy source) switching, and

(B) Applied steam energy versus direct-fired energy versus electricity.

(v) Insulation issues.

(vi) Steam trap and steam leak management.

(vi) Condensate recovery.

(viii) Steam end-use management.

(2) Capabilities and knowledge includes, but is not limited to:

(i) Background, experience, and recognized abilities to perform the assessment activities, data analysis, and report preparation.

(ii) Familiarity with operating and maintenance practices for steam or process heating systems.

(iii) Additional potential steam system improvement opportunities including improving steam turbine operations and reducing steam demand.

(iv) Additional process heating system opportunities including effective utilization of waste heat and use of proper process heating methods.

(v) Boiler-steam turbine cogeneration systems.

(vi) Industry specific steam end-use systems.

Refinery gas means any gas that is generated at a petroleum refinery and is combusted. Refinery gas includes natural gas when the natural gas is

combined and combusted in any proportion with a gas generated at a refinery. Refinery gas includes gases generated from other facilities when that gas is combined and combusted in any proportion with gas generated at a refinery.

Regulated gas stream means an offgas stream that is routed to a boiler or process heater for the purpose of achieving compliance with a standard under another subpart of this part or part 60, part 61, or part 65 of this chapter.

Residential boiler means a boiler used to provide heat and/or hot water and/or as part of a residential combined heat and power system. This definition includes boilers located at an institutional facility (e.g., university campus, military base, church grounds) or commercial/industrial facility (e.g., farm) used primarily to provide heat and/or hot water for:

(1) A dwelling containing four or fewer families; or

(2) A single unit residence dwelling that has since been converted or subdivided into condominiums or apartments.

Residual oil means crude oil, fuel oil that does not comply with the specifications under the definition of distillate oil, and all fuel oil numbers 4, 5, and 6, as defined by the American Society of Testing and Materials in ASTM D396-10 (incorporated by reference, see §63.14(b)).

Responsible official means responsible official as defined in §70.2.

Rolling average means the average of all data collected during the applicable averaging period. For demonstration of compliance with a CO CEMS-based emission limit based on CO concentration a 30-day (10-day) rolling average is comprised of the average of all the hourly average concentrations over the previous 720 (240) operating hours calculated each operating day. To demonstrate compliance on a 30-day rolling average basis for parameters other than CO, you must indicate the basis of the 30-day rolling average period you are using for compliance, as discussed in §63.7545(e)(2)(iii). If you indicate the 30 operating day basis, you must calculate a new average value each operating day and shall include the measured hourly values for the preceding 30 operating days. If you select the 720 operating hours basis, you must average of all the hourly average concentrations over the previous 720 operating hours calculated each operating day.

Secondary material means the material as defined in §241.2 of this chapter.

Shutdown means the period in which cessation of operation of a boiler or process heater is initiated for any purpose. Shutdown begins when the boiler or process heater no longer supplies useful thermal energy (such as heat or steam) for heating, cooling, or process purposes and/or generates electricity or when no fuel is being fed to the boiler or process heater, whichever is earlier. Shutdown ends when the boiler or process heater no longer supplies useful thermal energy (such as steam or heat) for heating, cooling, or process purposes and/or generates electricity, and no fuel is being combusted in the boiler or process heater.

Sloped grate means a unit where the solid fuel is fed to the top of the grate from where it slides downwards; while sliding the fuel first dries and then ignites and burns. The ash is deposited at the bottom of the grate. Fluidized bed, dutch oven, pile burner, hybrid suspension grate, suspension burners, and fuel cells are not considered to be a sloped grate design.

Solid fossil fuel includes, but is not limited to, coal, coke, petroleum coke, and tire derived fuel.

Solid fuel means any solid fossil fuel or biomass or bio-based solid fuel.

Startup means:

(1) Either the first-ever firing of fuel in a boiler or process heater for the purpose of supplying useful thermal energy for heating and/or producing electricity, or for any other purpose, or the firing of fuel in a boiler after a shutdown event for any purpose. Startup ends when any of the useful thermal energy from the boiler or process heater is supplied for heating, and/or producing electricity, or for any other purpose, or

(2) The period in which operation of a boiler or process heater is initiated for any purpose. Startup begins with either the first-ever firing of fuel in a boiler or process heater for the purpose of supplying useful thermal energy (such as steam or heat) for heating, cooling or process purposes, or producing electricity, or the firing of fuel in a boiler or process heater for any purpose after a shutdown event. Startup ends four hours after when the boiler or process heater supplies useful thermal energy (such as heat or steam) for heating, cooling, or process purposes, or generates electricity, whichever is earlier.

Steam output means:

(1) For a boiler that produces steam for process or heating only (no power generation), the energy content in terms of MMBtu of the boiler steam output,

(2) For a boiler that cogenerates process steam and electricity (also known as combined heat and power), the total energy output, which is the sum of the energy content of the steam exiting the turbine and sent to process in MMBtu and the energy of the electricity generated converted to MMBtu at a rate of 10,000 Btu per kilowatthour generated (10 MMBtu per megawatt-hour), and

(3) For a boiler that generates only electricity, the alternate output-based

40 CFR Ch. I (7–1–23 Edition)

emission limits would be the appropriate emission limit from Table 1, 2, 14, or 15 to this subpart in units of pounds per million Btu heat input (lb per MWh).

(4) For a boiler that performs multiple functions and produces steam to be used for any combination of paragraphs (1), (2), and (3) of this definition that includes electricity generation of paragraph (3) of this definition, the total energy output, in terms of MMBtu of steam output, is the sum of the energy content of steam sent directly to the process and/or used for heating (S_1) , the energy content of turbine steam sent to process plus energy in electricity according to paragraph (2) of this definition (S_2) , and the energy content of electricity generated by a electricity only turbine as paragraph (3) of this definition (MW_3) and would be calculated using Equation 1 to this definition. In the case of boilers supplying steam to one or more common headers, S_1 , S_2 , and $MW_{(3)}$ for each boiler would be calculated based on its (steam energy) contribution (fraction of total steam energy) to the common header.

Equation 1 to the definition *Steam Output*

$$SO_M = S_1 + S_2 + (MW_{(3)} \times CFn)$$
 (Eq. 1)

Where:

- SO_M = Total steam output for multi-function boiler, MMBtu.
- S₁ = Energy content of steam sent directly to the process and/or used for heating, MMBtu.
- S_2 = Energy content of turbine steam sent to the process plus energy in electricity according to paragraph (2) of this definition, MMBtu.
- $MW_{(3)}$ = Electricity generated according to paragraph (3) of this definition, MWh.
- CFn = Conversion factor for the appropriate subcategory for converting electricity generated according to paragraph (3) of this definition to equivalent steam energy, MMBtu/MWh.
- CFn for emission limits for boilers in the unit designed to burn solid fuel subcategory = 10.8.

- CFn PM and CO emission limits for boilers in one of the subcategories of units designed to burn coal = 11.7.
- CFn PM and CO emission limits for boilers in one of the subcategories of units designed to burn biomass = 12.1.
- CFn for emission limits for boilers in one of the subcategories of units designed to burn liquid fuel = 11.2.
- CFn for emission limits for boilers in the unit designed to burn gas 2 (other) subcategory = 6.2.

Stoker means a unit consisting of a mechanically operated fuel feeding mechanism, a stationary or moving grate to support the burning of fuel and admit under-grate air to the fuel, an overfire air system to complete combustion, and an ash discharge system. This definition of stoker includes

air swept stokers. There are two general types of stokers: Underfeed and overfeed. Overfeed stokers include mass feed and spreader stokers. Fluidized bed, dutch oven, pile burner, hybrid suspension grate, suspension burners, and fuel cells are not considered to be a stoker design.

Stoker/sloped grate/other unit designed to burn kiln dried biomass means the unit is in the units designed to burn biomass/bio-based solid subcategory that is either a stoker, sloped grate, or other combustor design and is not in the stoker/sloped grate/other units designed to burn wet biomass subcategory.

Stoker/sloped grate/other unit designed to burn wet biomass means the unit is in the units designed to burn biomass/biobased solid subcategory that is either a stoker, sloped grate, or other combustor design and any of the biomass/ bio-based solid fuel combusted in the unit exceeds 20 percent moisture on an annual heat input basis.

Suspension burner means a unit designed to fire dry biomass/biobased solid particles in suspension that are conveyed in an airstream to the furnace like pulverized coal. The combustion of the fuel material is completed on a grate or floor below. The biomass/ biobased fuel combusted in the unit shall not exceed 20 percent moisture on an annual heat input basis. Fluidized bed, dutch oven, pile burner, and hybrid suspension grate units are not part of the suspension burner subcategory.

Temporary boiler means any gaseous or liquid fuel boiler or process heater that is designed to, and is capable of, being carried or moved from one location to another by means of, for example, wheels, skids, carrying handles, dollies, trailers, or platforms. A boiler or process heater is not a temporary boiler or process heater if any one of the following conditions exists:

(1) The equipment is attached to a foundation.

(2) The boiler or process heater or a replacement remains at a location within the facility and performs the same or similar function for more than 12 consecutive months, unless the regulatory agency approves an extension. An extension may be granted by the

regulating agency upon petition by the owner or operator of a unit specifying the basis for such a request. Any temporary boiler or process heater that replaces a temporary boiler or process heater at a location and performs the same or similar function will be included in calculating the consecutive time period.

(3) The equipment is located at a seasonal facility and operates during the full annual operating period of the seasonal facility, remains at the facility for at least 2 years, and operates at that facility for at least 3 months each year.

(4) The equipment is moved from one location to another within the facility but continues to perform the same or similar function and serve the same electricity, process heat, steam, and/or hot water system in an attempt to circumvent the residence time requirements of this definition.

Total selected metals (TSM) means the sum of the following metallic hazardous air pollutants: arsenic, beryllium, cadmium, chromium, lead, manganese, nickel and selenium.

Traditional fuel means the fuel as defined in §241.2 of this chapter.

Tune-up means adjustments made to a boiler or process heater in accordance with the procedures outlined in $\S63.7540(a)(10)$.

Ultra low sulfur liquid fuel means a distillate oil that has less than or equal to 15 ppm sulfur.

Unit designed to burn biomass/bio-based solid subcategory includes any boiler or process heater that burns at least 10 percent biomass or bio-based solids on an annual heat input basis in combination with solid fossil fuels, liquid fuels, or gaseous fuels.

Unit designed to burn coal/solid fossil fuel subcategory includes any boiler or process heater that burns any coal or other solid fossil fuel alone or at least 10 percent coal or other solid fossil fuel on an annual heat input basis in combination with liquid fuels, gaseous fuels, or less than 10 percent biomass and bio-based solids on an annual heat input basis.

Unit designed to burn gas 1 subcategory includes any boiler or process heater that burns only natural gas, refinery gas, and/or other gas 1 fuels. Gaseous fuel boilers and process heaters that burn liquid fuel for periodic testing of liquid fuel, maintenance, or operator training, not to exceed a combined total of 48 hours during any calendar year, are included in this definition. Gaseous fuel boilers and process heaters that burn liquid fuel during periods of gas curtailment or gas supply interruptions of any duration are also included in this definition.

Unit designed to burn gas 2 (other) subcategory includes any boiler or process heater that is not in the unit designed to burn gas 1 subcategory and burns any gaseous fuels either alone or in combination with less than 10 percent coal/solid fossil fuel, and less than 10 percent biomass/bio-based solid fuel on an annual heat input basis, and no liquid fuels. Gaseous fuel boilers and process heaters that are not in the unit designed to burn gas 1 subcategory and that burn liquid fuel for periodic testing of liquid fuel, maintenance, or operator training, not to exceed a combined total of 48 hours during any calendar year, are included in this definition. Gaseous fuel boilers and process heaters that are not in the unit designed to burn gas 1 subcategory and that burn liquid fuel during periods of gas curtailment or gas supply interruption of any duration are also included in this definition.

Unit designed to burn heavy liquid subcategory means a unit in the unit designed to burn liquid subcategory where at least 10 percent of the heat input from liquid fuels on an annual heat input basis comes from heavy liquids.

Unit designed to burn light liquid subcategory means a unit in the unit designed to burn liquid subcategory that is not part of the unit designed to burn heavy liquid subcategory.

Unit designed to burn liquid subcategory includes any boiler or process heater that burns any liquid fuel, but less than 10 percent coal/solid fossil fuel and less than 10 percent biomass/ bio-based solid fuel on an annual heat input basis, either alone or in combination with gaseous fuels. Units in the unit design to burn gas 1 or unit designed to burn gas 2 (other) subcategories that burn liquid fuel for periodic testing of liquid fuel, mainte40 CFR Ch. I (7–1–23 Edition)

nance, or operator training, not to exceed a combined total of 48 hours during any calendar year are not included in this definition. Units in the unit design to burn gas 1 or unit designed to burn gas 2 (other) subcategories during periods of gas curtailment or gas supply interruption of any duration are also not included in this definition.

Unit designed to burn liquid fuel that is a non-continental unit means an industrial, commercial, or institutional boiler or process heater meeting the definition of the unit designed to burn liquid subcategory located in the State of Hawaii, the Virgin Islands, Guam, American Samoa, the Commonwealth of Puerto Rico, or the Northern Mariana Islands.

Unit designed to burn solid fuel subcategory means any boiler or process heater that burns only solid fuels or at least 10 percent solid fuel on an annual heat input basis in combination with liquid fuels or gaseous fuels.

Useful thermal energy means energy (*i.e.*, steam, hot water, or process heat) that meets the minimum operating temperature, flow, and/or pressure required by any energy use system that uses energy provided by the affected boiler or process heater.

Vegetable oil means oils extracted from vegetation.

Voluntary Consensus Standards or VCS mean technical standards (e.g., materials specifications, test methods, sampling procedures, business practices) developed or adopted by one or more voluntary consensus bodies. EPA/Office of Air Quality Planning and Standards, by precedent, has only used VCS that are written in English. Examples of VCS bodies are: American Society of Testing and Materials (ASTM 100 Barr Harbor Drive, P.O. Box CB700, West Conshohocken, Pennsylvania 19428-B2959 (800) 262 - 1373, http:// www.astm.org), American Society of Mechanical Engineers (ASME ASME, Three Park Avenue, New York, NY 10016–5990. (800) 843-2763. http:// www.asme.org), International Standards Organization (ISO 1, ch. de la Voie-Creuse, Case postale 56, CH-1211 Geneva 20, Switzerland, + 41 22 749 01 11,

http://www.iso.org/iso/home.htm), Standards Australia (AS Level 10, The Exchange Centre, 20 Bridge Street, Sydney, GPO Box 476, Sydney NSW 2001, + 61 2 9237 6171 http:// British Standwww.stadards.org.au), ards Institution (BSI, 389 Chiswick High Road, London, W4 4AL, United Kingdom, + 44 (0)20 8996 9001, http:// www.bsigroup.com), Canadian Standards Association (CSA 5060 Spectrum Way, Suite 100, Mississauga, Ontario L4W 5N6. Canada, 800-463-6727. http:// www.csa.ca), European Committee for Standardization (CEN CENELEC Management Centre Avenue Marnix 17 B-1000 Brussels, Belgium + 32 2 550 08 11, http://www.cen.eu/cen), and German Engineering Standards (VDI VDI Guidelines Department, P.O. Box 10 11 39 40002, Duesseldorf, Germany, + 49 211 6214-230, http://www.vdi.eu). The types of standards that are not considered VCS are standards developed by: The United States, e.g., California (CARB) and Texas (TCEQ); industry groups, such as American Petroleum Institute (API), Gas Processors Association (GPA), and Gas Research Institute (GRI); and other branches of the U.S. government, e.g., Department of Defense (DOD) and Department of Transportation (DOT). This does not preclude EPA from using standards developed by groups that are not VCS bodies within their rule. When this occurs, EPA has done searches and reviews for VCS equivalent to these non-EPA methods.

Pt. 63, Subpt. DDDDD, Table 1

Waste heat boiler means a device that recovers normally unused energy (i.e., hot exhaust gas) and converts it to usable heat. Waste heat boilers are also referred to as heat recovery steam generators. Waste heat boilers are heat exchangers generating steam from incoming hot exhaust gas from an industrial (e.g., thermal oxidizer, kiln, furnace) or power (e.g., combustion turbine, engine) equipment. Duct burners are sometimes used to increase the temperature of the incoming hot exhaust gas.

Waste heat process heater means an enclosed device that recovers normally unused energy (i.e., hot exhaust gas) and converts it to usable heat. Waste heat process heaters are also referred to as recuperative process heaters. This definition includes both fired and unfired waste heat process heaters.

Wet scrubber means any add-on air pollution control device that mixes an aqueous stream or slurry with the exhaust gases from a boiler or process heater to control emissions of particulate matter or to absorb and neutralize acid gases, such as hydrogen chloride. A wet scrubber creates an aqueous stream or slurry as a byproduct of the emissions control process.

Work practice standard means any design, equipment, work practice, or operational standard, or combination thereof, that is promulgated pursuant to section 112(h) of the Clean Air Act.

[78 FR 15664, Mar. 21, 2011, as amended at 78 FR 7163, Jan. 31, 2013; 80 FR 72817, Nov. 20, 2015; 87 FR 60846, Oct. 6, 2022]

TABLE 1 TO SUBPART DDDDD OF PART 63—EMISSION LIMITS FOR NEW OR RECONSTRUCTED BOILERS AND PROCESS HEATERS $^{\rm o}$

As stated in §63.7500, you must comply with the following applicable emission limits:

40 CFR Ch. I (7-1-23 Edition)

Cole 1 CO Sooper DDDDD of Para 63 Emission Limits for CE or Recorder CED Collers and Process Headers a

□ nit with heat inp t capacit of 10 mi to per hor or greater

If ⊡ r ⊡oier or proce heater i□ in thi□ incategor	For the fo⊡owing po⊡tant⊒	□he emi⊡ion □ m⊡t not eEcee the fo⊡owing emi⊡ion 1mit eCcept □ ring tartp an □ h t ⊡own	Or the emi in m t not creee the following all ternatice orphinate imit creept ring tartcp and httown	⊡îng thi⊡ ⊅ecifie⊡ ⊑ampîng
1. Init in a IIICat egorie III III III III IIIII IIIIIIIIIIIIIII	a. HC⊡	2.1E O4 ª	2.9E 04 a per MM to of team otpt or 2.EE03 a per M h.	For M26□, corect a minim⊡m of 1 ⊡cm per r⊡n⊡for M26 cor ect a minim⊡m of 120 îter⊡ per r⊡n.
	□ Merc□r□	8.0E⊡0⊒ª ⊞ per MM⊡t⊡ of heat inp⊡t.	8. E O P D P MM t O	For M29, collect a mining mon of 4 comper renfor M300 or M300, collect a mining mon cample a collect a mining mon collect a mining mon collect a mining mon of 4 com.
2. Init leigne to Im coa/Ioil foil f	a. Filtera 🗈 PM for ISM I	1.1E □03 □ per MM □t□ of heat inp□t□or 12.3E □05 □ per MM □t□ of heat inp □t□	1.1E_03 per MM_tt ofteam oftpft or 1.4E_02 per M h or 12.2E_05 per MM_tt ofteam oft pt or 2.9E_04 per M h	Co⊡ect a minim⊡m of 3 ⊡om per r⊡n.
3. Pierie coaloi erieigne to in coaloi foi fe	a. Car⊡on mono⊡⊡e ⊥CO⊡or CEMS⊡	130 ppm i olime on a i al orrecte to 3percent oligen, 31 n alerage or 320 ppm i bolme on a raicoror recte to 3 percent oligen, 30 la rolling ing alerage	0.11	1 hr minim⊡n ⊡amp1ing time.
4. Stoker://other:le: igne:ito ::::m coai/ coi:::fo::::fie::	a. CO lor CEMS	130 ppm i bilme on a r al correcte to 3 percent o gen, 3 r al erage or 340 ppm i bilme on a r al cor recte to 3 percent o gen , 30 a r o m ing a Perage	0.12	1 hr minim⊡m ⊡ampûng time.
5. Fuide conto Congreto com coalization fottofico	a. CO Tor CEMS	130 ppm b billime on a crocete to 3 percent o gen, 3 crocete 230 ppm b billime on a crocete rectel to 3 percent o gen, 30 crocete ing alerage	0.11	1 hr minim⊡m ⊡ampûng time.
6. Filline centit with an integrate heat enchanger centif ignent o finn coat	a. CO lor CEMS	140 ppm i collme on a collection a correcte to 3 percent o gen, 3 m a crage or 50 ppm i collme on a collection a coll recte to 3 percent o gen, 30 a coll ing a crage	1.2E:01 Ⅲ per MM⊡t of ⊡eam o⊡tp⊡t or 1.5 Ⅲ per M⊡ h⊡3⊡ r⊡n a⊡erage.	1 hr minim⊡m ⊡ampûng time.

Pt. 63, Subpt. DDDDD, Table 1

COLLE 1 DO SOLPARO DDDDD OF PARO 63 EMISSION LIMITS FOR DEC OR RECOLSARICOED
□oiters □□D Process He□ters °□ Continte
mater with best inn⊂t conseit of 10 mitten ⊐tr per ber or greater

 $\blacksquare nit \Box$ with heat inp $\blacksquare t$ capacit \Box of 10 mi $\blacksquare t \Box$ per ho $\boxdot r$ or greater \Box

If ⊡ r ⊡oier or proce heater i in thi □ categor	For the fo⊡owing po⊡tant □	□he emi⊡on m t not ecceo the fo⊡owing emi⊡on imit eccept ⊡ring tartp an ht own	Or the emilion mit not erceel the following all ternatice orpiliae imit_eccept fring lartp and thickness.	⊡ing thi⊡ pecifie⊡ ⊡amp1ing
□ Stoker//⊡ope□ grate/ other□ erigne□ to □rn wet Iroma⊡ f_e□	a. CO Tor CEMS	590 ppm : ollme on a r a: correcte to 3:percent ollgen, 3:r a areage or 390 ppm : collme on a r a: correcte to 3:percent ollgen, 30: coll or ing alerage.	6.1E::01 Ⅲ per MM⊡t of ⊡eam o⊡tp⊡t or 6.5 Ⅲ per M⊡ h⊡3⊡ r⊡n a⊡erage.	1 hr minim⊡m ⊡ampûng time.
	□ Filtera ⊡e PM lor □SM□	1.3E [®] 02 [™] per MM⊡t of heat inp⊡t⊡or 12.6E [®] 05 [™] per MM⊡t [®] of heat inp⊡t	1.4E 02	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
8. Stoker // ope grate/ other cigne to orn kin rie co ma fee	a. CO	460 ppm □	4.3E □01 □ per MM □t □ of □team o □tp □t or 5.1 □ per M □ h.	1 hr minim⊡m ⊡ampûng time.
	□ Filtera ⊡e PM ©r □SM□	3.0E ©2	3.5E 02	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
9. Filia e conta cagne to ann ao mai/dolla e coll in	a. CO lor CEMS	130 ppm : collme on a collection a corrected to 3 percent orgen, 3 rcm a crage or 310 ppm : collme on a crollection rected to 3 percent orgen, 30 cm coll ing a crage.	1.3E⊡01 III per MM⊡t of ⊡eam o⊡tp⊡ or 1.5 III per M⊡ h⊡3⊡ r⊡n a⊡erage.	1 hr minim⊡m ⊡ampûng time.
	□ Filtera ⊡e PM ©or □SM□	4.1E [®] 03 [™] per MM⊡t [®] of heat inp⊡t [®] or 18.4E [®] 06 ^a [™] per MM⊡t [®] of heat inp⊡t [®]	5.0E_03 m per MM_tt of _team o_tp_t or 5.8E_02 m per M_h or _1.1E_05* m per MM_tt_of _team o_tt p_t or 1.2E_04* m per M_h	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
10. Simpenfion inner Geigne to inn io mai/fiolia e com i	a. CO lor CEMS	220 ppm come on a corrected to 3 percent or gen, 3 mon a come come 2,000 ppm com corrected to 3 per- cent or gen, 10 ma roling alterage	0.18	1 hr minim⊡m ⊡ampûng time.
	Filtera e PM for SM	3.0E:02 III per MM:1 of heat inp:1 8.0E:03 III per MM:1 of heat inp:1	3.1E:02	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.

40 CFR Ch. I (7-1-23 Edition)

Collers od Process Heiders Continue init⊡ with heat inp⊡t capacit⊡ of 10 militon ⊡t⊡ per ho⊡r or greater⊡

f o r oier or proce heater i in thi □ categor	For the fo⊡owing po⊡tant□	□he emi⊡on□m⊡t not eīcee□ the fo⊡owing emi⊡on īmit□ eīcept ⊡ring ⊡art⊡p an□ h⊡t□ ⊡own	Or the emilion milt ot elceel the following all ternatice otplicate imit_elcept tring lartip and thitown	⊡ing thi⊡ pecifie⊡ ⊡amp1ing
11. Ditch Oieni/Pië irreri lefignet to irri fiomat/fioi laien coltin	a. CO lor CEMS	330 ppm Dollme on a conception of the service of the service of the service of the service of the service of the service of the service of the service of the service of the service of the service of the service of the service of the service of the service of the service of the service of th	3.5E:01 :::::::::::::::::::::::::::::::::::	1 hr minim⊡m ⊡ampûng time.
	L Filtera De PM Tor SMC	2.5E 03 m per MM to of heat inp tor (3.9E 05 m per MM to f heat inp to	3.4E_03 m per MM_tt of _team o_tp_t or 3.5E_02 m per M_h or _5.2E_05 m per MM_tt_of _team o_tt p_t or 5.5E_04 m per M_h	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
12. Feece init e ignet to in io mailionale of	a. CO	910 ppm	1.1	1 hr minim⊡m ⊡ampûng time.
	C Filtera Ce PM for	1.1E 02	2.0E_02	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
13. Hiri III penion grate loier leignet to Irn Ioma fio alei loit	a. CO lor CEMS	180 ppm : olime on a corrected to 3 percent oligen, 3 corrected to gen, 3 corrected to 3 percent oligen 3 0 corrected to 3 percent oligen 3 0 corrected to 3 percent oligen 3 corrected to 3 percent	0.22	1 hr minim⊡m ⊡amp1ing time.
	□ Fitera □e PM or □SM□	2.6E 02 per MM to of heat inp tor 3.4E 04 per MM to f heat inp to	3.3E_02 per MM_tt ofteam oftpft or 3.E_01 per M h or _5.5E_04 per MM_tt ofteam oft pft or 6.2E_03 per M h	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
14. Onito Defigned to Com 1995 files	a. HC⊡	1.5E⊡04 ª	1. E 04 a	For M26 :: Co ⊡ect a minim ☐m of 2 □ cm per r □n ☐for M26, co ⊡ect a minim ☐m of 240 îter ☐ per r □n.
	□ Merc□r□	4.8E⊡0⊒ª ⊞ per MM⊡t⊟ of heat inp⊡t.	5.3E 00 a m per MM to of 1 eam o tp1 or 6.0E 06 a m per M0 h.	For M29, collect a min imm of 4 cm per rin for M300 or M300, collect a min imm cample a pecifie in the meth offor SIM D6 B4 collect a minimm of 4 cm.
15. Init in in it in it is a constant of the init of the init of the iter in the init of the iter is a constant of the iter is constant of the iter is constant of the iter is	a. CO	130 ppm	0.13	1 hr minim⊡m ⊡ampûng time.

Pt. 63, Subpt. DDDDD, Table 1

DOLLE 1 DO SOLPERE DDDDD OF PERE 63 EMISSION LIMITS FOR DED OR RECORDERED Otters and Process Heaters Contin De

initial with heat inpat capacital of 10 mittion at a per hoar or greater

If ⊡o⊡r ⊡oi≣r or proce heater i⊡ in thi⊡ ⊡⊂categor⊡	For the fo⊡owing po⊡tant⊡	□he emi⊡on m⊡t not e[cee] the fo[@wing emi⊡on îmit] e[cept ⊡ring tart[p an] htt @wn	Or the emiion mit not eccee the following all ternatic orptimate limit, eccept fring tartip and hittown	□ ing thi □ ipecifie □ I ing îng
	□ Fitera⊡e PM tor □SM□	1.9E_03	2.1E:03	Cor≊ect a minim⊡m of 3 ⊡cm per r⊡n.
16. onito e igneo to	a. CO	130 ppm □ o me on a r a i correcte to 3 percent o gen.	0.13	1 hr minim⊡m ⊑ampûng time.
	L Filtera Ce PM for	1.1E⊡03°	1.2E ⁰ 3° □ per MM t of □team oltpt or 1.6E ⁰ 2° □ per M hor 3.2E ⁰ 5 □ per MM to of □team oltpt or 4.0E ⁰ 4 □ per M ho	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
1 onit e_igne to iiii f_e_that are non continenta nit	a. CO	130 ppm Comme on a Comme on to 3percent orgen, 3 Comme orgen, a comme or	0.13	1 hr minim⊡m ⊡ampûng time.
	L Filtera te PM for SMC	2.3E 02 per MM to of heat inp tor 18.6E 04 per MM to f heat inp to	2.5E 02 m per MM to of team otpt or 3.2E 01 m per M h or 19.4E 04 m per MM to of team ott pt or 1.2E 02 m per M h	Co⊡ect a minim⊡m of 4 ⊡cm per r⊡n.
18. onito celigneo to orn gao 2 tothero gaocol	a. CO	130 ppm □ o me on a □ a a correcte to 3 percent o gen.	0.16	1 hr minim⊡m ⊡ampûng time.
-	п нсп	1. E 03	2.9E 03	For M26 , co ect a minim m of 2 cm per rm for M26, co ect a minim m of 240 îter per rm.
	c. Merc⊡r⊡	□9E⊡06 III per MM⊡t⊡ of heat inp⊡t.	1.4E 05	For M29, collect a min imm of 3 icm per ronfor M300 or M300, collect a min imm cample a pecifie in the meth offor STM D6184 collect a minimm of 3 icm.
	□ Filtera⊡e PM tor □SM□	□3E□03 □ per MM⊡t□ of heat inp⊡t⊡or i2.1E□04 □ per MM⊡t□ of heat inp⊡t□	1.3E ⁰² ^(m) per MM ⁻ t ⁻ of ^{[team} o ⁻ tp ⁻ t or ^(a) 6E ⁰² ^(m) per M ⁻ h ⁻ or ⁽³⁾ 5E ⁰ 4 ^(m) per MM ⁻ t ⁻ of ^{(team} o ⁻ t ⁻) p ⁻ t or 2.2E ⁰³ ^(m) per M ⁻ h ⁻	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.

a fi foo are condicting tack tell to remonitrate compliance and food performance tells for this politant for at Ball 2 conditions are at or below that for emillions are at or below that for an intermediate that for a the other product of 63.1515 are met. For all other politant that for not contain a footnote failing performance tells for this politient for at Ball 2 conditions are at or below that for emillion are at or below that for emillion are at or below that for emillion are at or below that for a the other product of 63.1515 are met. For all other politant that for not contain a footnote failing performance tells for this politic ant for at Ball 2 conditions are at or below that for emillion are at or below to perform the other this politic for the other of the area of the other of the formation are at or below to be performed to the other the other the other and the other area of the other of t

40 CFR Ch. I (7-1-23 Edition)

[87 FR 60847, Oct. 6, 2022]

TABLE 2 TO SUBPART DDDDD OF PART 63—EMISSION LIMITS FOR EXISTING BOILERS AND PROCESS HEATERS $^{\rm d}$

As stated in §63.7500, you must com	ply with the following applicable emission limits:		
⊞nit⊟ with heat inp⊡t capacit⊟ of 10 mi⊞on ⊡t⊟ per ho⊡r or greater⊡			
	→ · · · · · · · · · · · · · · · · · · ·		

f or oier or proce heater i⊡in thi ⊡categor	For the fo⊡owing po⊡tant⊡	□he emi⊡on □ m □t not elceel the following emi⊡on fimit□ elcept □ring ttart□p an□ h t□ □own	e ceet the following a ternatice o the following a ternatice o the following a ternatice o the ternatice of	□ ing thi □ pecifie □ □ amp îng □ o □ me or te īt r n □ □ □ ration
1. □nit□ in a □□□□cat□ egorie □ □e □gne □ to □rn ⊡oî □ f e □	a. HC⊡	2.0E O2	2.3E 02 □ per MM to of team otpt or 0.26 □ per M h.	For M26⊟, corect a minim⊡m of 1 ⊡cm per r⊡n⊡for M26, co⊡ rect a minim⊡m of 120 îter⊡ per r⊡n.
	□ Merc⊡r⊡	5.4E⊡06 ⊞ per MM⊡t⊡ of heat inp⊡t.	6.2E 06	For M29, collect a minimum of 3 crom per ron-for M300 or M300, collect a mini- imum cample a DecifieD in the metho- o-for SCM D6184 collect a minimum of 3 crom.
2. Init le ign to Im coa/Io III fo II if e I	a. Filtera 🗈 PM for ISM I	3.9E 02 m per MM to of heat inp to 5.3E 05 m per MM to f heat inp to	4.1E_02 per MM_t ofteam o_tp_t or 4.8E_01 per M_ h or 15.6E_05 per MM_t ofteam o_t p_t or 6.5E_04 per M_ h_	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
3. Pollerice coalcoim er eigne to on coalcoim for fe	a. CO lor CEMS⊡	130 ppm □ o⊡me on a □r □ a□ correcte to 3;percent o⊡gen, 3;r □ a Terageor :320 ppm □ o⊡me on a □r □ a□ corr recte to 3;percent o⊡gen,° 30:□ a ro: ing a Terage □	0.11 Ⅲ per MM⊡1 of □team otp⊡t or 1.4 Ⅲ per M h ⊡3⊞n a⊡erage.	1 hr minim⊡m ⊡amping time.
4. Stoker:/other: e: igne: to ::::n coal coi:::fo::::f:e:	a. CO lor CEMS	150 ppm i olime on a concete to 3 percent oligen, 3 concete 340 ppm i olime on a concete to 3 percent oligen,° 30 concete ing a rerage	0.14	1 hr minim⊡m ⊡ampûng time.
5. Filidie ce nit ceigne to on coa/ibii fordfe	a. CO lor CEMS	130 ppm in image of the image o	0.12	1 hr minim⊡m ⊡ampûng time.

Pt. 63, Subpt. DDDDD, Table 2

I ⊡ coier or proce heater i⊡ in thi⊡ ⊡categor	For the fo⊡owing po⊞tantີ	he emi⊡on m⊡t not eccec the for@wing emi⊡on fimit_ eccept ⊡ring ctartcp an htt cown	□he emiiion m t not e cee the following a ternatile o tp t a c imit e cept ing tart p an thtown	□ Ing thi□ pecifie□ □ amp îng □ o Ime or te t r In □ □ □ ration
6. Filline ce nit with an integrate heat echanger ce igne to mn coat foil foil fee	a. CO lor CEMS	140 ppm Dollme on a Dollme on 3 Decrento Orgen, 3 Tha Derageor 150 ppm Dollme on a Dollme rectel to 3 Dercent orgen,° 30 Taromi ing Derage	1.3E ⊡1 III per MM⊡t of ⊡team o⊡tp⊡t or 1.5 III per M⊡ h⊡3 r⊡n a⊡erage.	1 hr minim⊡m ⊡ampûng time.
Stoker/Tope_grate/ other_ce_gne_to Tn wet Tomath fiel	a. CO lor CEMS	1,100 ppm i oime on a riaicori recte to 3ipercent oigen, 3in aeri agelor i20 ppm i oime on a ri ai correcte to 3 percent oigen,° 30 a roimg aerage	1.1	1 hr minim⊡m ⊡ampûng time.
	□ Fitera □e PM or □SM□	3.4E 02	4.0E ⁰² mer MM ¹ t of learn oltplt or 4.8E ⁰¹ mer M ¹ h or 12.4E ⁰⁴ mer MM ¹ t of learn olt plt or 2.8E ⁰³ mer M ¹ h	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
8. Stoker / Lope grate/ other eigne to In kin rie io main free	a. CO	460 ppm □	4.2E □01 □ per MM⊡t□ of □team o⊡tp⊡t or 5.1 □ per M⊡ h.	1 hr minim⊡m ⊡ampûng time.
	□ Filera⊡e PM lor □SM□	3.2E_01	3E_01per MM_tt of _team o_tp_t or 4.5per M_hor 5.9E_03per MM_tt of _team o_tt p_t or _0E_02per M_h	Co⊡ect a minim⊡m of 1 ⊡cm per r⊡n.
9. Filidice cel nito celigne to com do macificate com	a. CO lor CEMS	210 ppm in office on a in all corrected to 3 percent orgen, 3 in a lerage or 310 ppm in office on a in all corrected to 3 percent orgen,° 30 marorizing ing alerage	2.1E:01 III per MMIt of iteam oftpft or 2.3 III per Mith 3 rin afterage.	1 hr minim⊡m ⊡ampûng time.
	C Filtera Ce PM for	□4E □3 □ per MM □t□ of heat inp □t⊡or 16.4E □5 □ per MM □t□ of heat inp □t□	9.2E_03	Co⊡ect a minim⊡m of 1 ⊡cm per r⊡n.
10. Sipenion inner eigne to inn io mai/iolare oin	a. CO lor CEMS	2,400 ppm i oime on a criaticori recte to 3ipercent oigen, 3ich a eri agelor (2,000 ppm i oime on a cri al correcte to 3 percent oigen; 10 a roiling a erage	1.9 □ per MM to of □team o tpit or 2 □ □ per M h to 3 to a cero age.	1 hr minim⊡m ⊡ampûng time.

40 CFR Ch. I (7-1-23 Edition)

	\blacksquare nit \Box with heat inp \Box capacit \Box of 10 mittion $\Box t\Box$ per ho \Box r or greater \Box						
m			□he emi⊡ion □ m □ t not e cee □ the fo owing a □				

I ⊡ ⊡iër or proce heater i⊟ in thi ⊡categor	For the fo⊡owing po⊡tant□	eliceel the following emilion fimit, elicept ining tartp and hit own	e cee the following all ternatice o tp t a c fimit, e cept cring tart p an h t own	amp thi pecifie amp ing o tet ron and ration
	□ Filtera ⊡e PM lor SM⊡	4.1E_02 per MM_t of heat inp_tor 18.0E_03 per MM_t of heat inp_t	4.2E 02 □ per MM □t of □team o □tp□t or 5.8E 01 □ per M □ h or □8.1E 03 □ per MM □t of □team o □t□ p t or 0.12 □ per M □ b □	Corect a minim⊡m of 2 ⊡cm per r⊡n.
11. Ditch Olen JPie Dirner Cligne to The Toma Joo Cale Colin	a. CO Tor CEMS	O ppm Collme on a correcte to 3ipercent orgen, 3in a lerage lor f520 ppm forme on a correcte to 3ipercent orgen, to 3ipercent orgen, to 3ipercent ing alerage	8.4E [01] III per MM⊡t of ⊡team o⊡tp⊡t or 8.4 III per M⊡ h⊡3 r⊡n a⊡erage.	1 hr minim⊡m ⊡ampîhg time.
	L Filtera Le PM for	1.8E 01 per MM to of heat inp tor 12.0E 03 per MM to f heat inp to	2.5E_01	Co⊡ect a minim⊡m of 1 ⊡cm per r⊡n.
12. Felcellnit e ignet to om io matricolla e oit	a. CO	1,100 ppm C Come on a C Call cor recte to 3 percent o gen.	2.4 □ per MM to of	1 hr minim⊡m ⊡ampûng time.
	L Filtera Le PM for SMC	2.0E_02 _ per MM_tt of heat inp_ttor 5.8E_03 _ per MM_tt of heat inp_tt	5.5E_02 per MM_tt of _team o_tp_t or 2.8E_01 per M_h or1.6E_02 per MM_tt_of _team o_tt p_t or 8.1E_02 per M_h	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
13. H ini i i penion grate init i gine to inn ioma i io alei foit	a. CO lor CEMS	3,500 ppm i olime on a criaticori recte to 3ipercent oligen, 3ich aleri agelor 1900 ppm i olime on a criaticorrecte to 3 percent oligen,° 30 Calcoling alerage	3.5 m per MM to of learn oltpl or 39 m per M ho3rch aler age.	1 hr minim⊡m ⊡ampûng time.
	L Fiftera E PM for	4.4E ©1 □ per MM to of heat inp⊥or ⊈.5E ©4 □ per MM to f heat inp to	5.5E_01	Co⊡ect a minim⊡m of 1 ⊡cm per r⊡n.
14. □nit□ □eígne⊡ to □rn í⊡í□ f⊡e□	a. HC□	1.1E⊡03	1.4E 03	For M26⊟, corect a minim⊡m of 2 ⊡cm per r⊡n⊡for M26, cor ect a minim⊡m of 240 îter⊓per r⊡n.
	□ Merc⊡r□	□3E:00 □ per MM⊡t of heat inp⊡t.	8.8E:00	For M29, collect a min im_m of 3 incm per rbnfor M30 or M30 collect a min im_m Cample a pecifie in the meth of for DSTM D6:84 collect a minimm of 2 incm.

Pt. 63, Subpt. DDDDD, Table 2

□ nit with heat inp t capacit of 10 mi ton to per hor or greater

	For the fo⊡owing po⊡tantີ	he emi⊡on m⊡t not eccee the following emi⊡on 1mit, eccept ⊡ring tartcp an htt own	□he emi⊡on□m⊡t not e⊡cee□ the foïowing a□ ternatice o tp1⊡a e□ imit_ecept ⊡ring tart_p an□ th1town	□ ling thi□ lpecifie □ lampûng lome or telt rin □ □ □ ration
15. Init le igne to	a. CO	130 ppm	0.13 III per MMIt of Iteam oItpIt or 1.4 III per MIhII3IIIn	1 hr minim⊡m ⊡ampûng time.
	□ Filtera⊡e PM tor ⊡SM⊡	5.9ET02 @ per MM to of heat inp to 2.0ET04 @ per MM to of heat inp to	□2E:02 m per MM□t□ of iteam oltp□t or 8.2E:01 m per M□h□ or i2.5E:04 m per MM□t□ of iteam olt□ p□t or 2.8E:03 m per M□h□	Co⊡ect a minim⊡m of 1 ⊡cm per r⊡n.
16. onito e igneo to	a. CO	130 ppm □ o me on a □ a a correcte to 3 percent o gen.	0.13	1 hr minim⊡m ⊡ampûng time.
	L Fiftera le PM for SM	□9E 03 □ per MM to of heat inp tor 6.2E 05 □ per MM to f heat inp to	9.6E_03	Corect a minim⊡m of 3 ⊡om per r⊡n.
1 Onit Ceigne to Tr III feethat are non continenta Onit	a. CO	130 ppm □ co≣me on a □ a □ correcte to 3 percent o gen, 3 ⊡ n a erage a e on tack tet.	0.13	1 hr minim⊡m ⊡ampûng time.
	□ Filtera⊡e PM tor □SM□	2.2E 01	2. E 01 m per MM t of team otpt or 3.1 per M hor 1.1E 03 per MM t of team ot pt or 1.2E 02 per M ho	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
18. □nit□ ⊡igne□ to □rn ga□2 tother□ ga⊡e□	a. CO	130 ppm □ o me on a r aic correcte to 3 percent o gen.	0.16	1 hr minim⊡m ⊡ampûng time.
	L HC	1. E O3	2.9E 03 III per MM to of team otpot or 1.8E 02 III per M h.	For M26 , collect a minim m of 2 cm per r n for M26, coll lect a minim m of 240 liter per r n.
	c. Merc⊡r⊡	□9E⊡06 III per MM⊡t⊡ of heat inp⊡t.	1.4E:05	For M29, collect a minin imm of 3 comper rin for M300 or M30, collect a mini imm tample a pecifie in the meth offor SCM D6184 collect a minimm of 2 com.
	□ Fitera⊡e PM tor □SM□	□3E □03 □ per MM to of heat inp⊡t or □2.1E □04 □ per MM to of heat inp to	1.3E 02 m per MM t of deam oftpd or 06E 02 m per M h or 03.5E 04 m per MM of deam oft pd or 2.2E 03 m per M h	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.

a fi col are conjecting lack tell to remonizate compliance and four performance tell for this pointant for at real 2 conjective reard how that four emillions are at or reliave this limit, for an a four performance tell for this pointant for at real 2 conjective reard how that four emillions are at or reliave that for ot contain a footnote a, four performance tell for this pointant for at real 2 conjective reard much work that four emillions are at or reliave to reliave the performance tell for this pointant for at real 2 conjective reard much work that four emillions are at or reliave to perform for this for this for this point the tell for this point and the tell for this point and the tell for this for the tell for the tell for this for the tell for this for the tell for this for the tell for tell for

40 CFR Ch. I (7-1-23 Edition)

 $^{\circ}$ n owner or operator ma_cetermine compliance with the carlon monoline emilion limit cetermine ling CO₂ a.a. in control of the contro

[87 FR 60849, Oct. 6, 2022]

TABLE 3 TO SUBPART DDDDD OF PART 63-WORK PRACTICE STANDARDS

As stated in §63.7500, you must comply with the following applicable work practice standards:

₫ ¤o⊡r =nit i□	Yo⊡ m⊡t meet the fo⊡owing
1new or ell_tingoier or procell heater with a contin oll_gen trimem that maintain_an optim_m air to fee ratio, or a heat inplt capacito f en than or ellato 5 mill ion per hor in an of the forowingcategorieltint ellgne_ to mgal 10 the forowingcategorieltint or int ellgne_ to mgt, or a limite oier or procell heater.	Con⊡ct a t⊡ne⊞p of the ⊡oi@r or proce⊡ heater e⊡er⊡ 5 ⊡ear⊡ a⊡ ⊡pecifie⊡ in ⊡63.⊡540.
2. new or editing loier or proced heater without a contine local control of the second secon	Con ⊡ct a t⊡ne ⊡p of the ⊡oiær or proce⊡ heater ⊡ennia⊞ a⊟ ⊡pecifie⊟ in ⊡63.⊡540.
3. I new or editing Coller or procet heater without a contine Collect origen trim the and with heat input capacit of 10 million the per hour or greater.	Con ta t_nep of theoier or proce heater ann _a a pecifie in63540nit_ in either the Ga1 or Meta Proce France categorie will con this the a a work practice for a reg_ate entities nite in the _t
4. In eliting biler or procet heater boated at a mator borce facilit, not including timited the init.	 Mint hale a oneitime energi alleliment performe in a life energi allelior. In energi allelior complete on or after Janlard 1, 2008, that meet or in amene to meet the energi alleliment reinferent in this tale, tat infieithe energi alleliment reinferent. If facilit that operate in the energi alleliment reinferent in this tale, tat infieithe energi alleliment reinferent. If facilit that operate in the energi alleliment reinferent in this tale, tat infieithe energi alleliment reinferent. If facilit that operation or after Janlard 1, 2008 and the compliance lite the energi alleliment reinferent. If facilit that operating alleliment reinferent in the table of the energi alleliment reinferent. In the energi alleliment reinferent in the the energi alleliment reinferent. The energi alleliment reinferent is all one lite in the following with eltent of the elaliton for item a to e. appropriate for the onlite technicalhori the elaliton of operating contraint. c. In infentori of the lifer or proceil heater time. In elalitation of operating contraint. c. In infentori of male energi elimet conting energi from affecte lifer and proceil heater on maintenance proceil heater owner/operator. energiese. e. reilew of alaiale architectral and engineering plan, fall cittli operation and maintenance proceil enal for all fall of conteffectile energi contertation for improlement contilent with the facilitie energi contertation for improlement contilent with the facilitie energi contertation for improlement contilent with the facilitie energi contertation for improlement contert are within the facilitie energi contertation male energi endition for improlement contilent with the facilitie energi contertation mealment the are within the facilitie energi contertation mealment the are within the facilitie energi contertation mealment the are within the facilitie energi contertation mealment the area within the facilitie energi contertation mealment there is a conterte entertion mealment there is a

Pt. 63, Subpt. DDDDD, Table 3

 h. comprehencie report etailing the walched ciencies of the coll of pecific improcements time frame for recorping those in retirments and the ciencies of the coll of	
<pre>both the spin of the spin</pre>	to improle effi enefit, an the molt de one le matraga, 1 fed, ditate moake rag, ker
Image: Second	an free meeting miction tan ar
e:peliufold a poind. Startp end wh i	1 1 of the efini art firing fee that ion to the main xontro efice en committion fe Co actife cata fit or re- entone infection in SCR filtema a
Getart of the final with the final final and the final fina	1en leam or heat
Dan. a Decile in 105. Duble.	Yo mit end or of fir tee or of fir tee ing tart a applicate in an or tart in an an r tart an an an e other than thi thi ice other than thi thi
Yo mit compil with all applicate emittime time tecept minitime minitime monitoring tartip and hit own time time monitoring tartip and time time monitoring tartip and time time	i imit_ at a perio at which Yo mt conect p, apecifie in g perio oftart_ acti_itie anperi_
6. In eliting or new loier or procell heater field to emile Yo mill operate allCMS fing fillown. I hie fining fill that are not clean fill through 15 to this fining fill that are not clean fill fill fill fill fill fill fill fil	ing chittown, co an operate a ⊡ap e inæction in FC tit, in an cate, ar cate to ntrotecte. ation of chittown,
another file mill the import the file proce that a file mill the one or the following clean file that a file mill the one or the following clean file that file mill the oil fill the compared of the file pet You mill compared with a splicate emill time = cleapt for that por childrown period this work practice. You mill context month period of that the mill the splicate mill keep record fill the mill the fill the splicate mill the splicate of the fill the splicate mill the fill the splicate mill the splicate of the fill the splicate mill the splicate million of the splicate mil	httown a comination of thetic natiralga, thetic natiralga, thetic natiralga, thetic national thetic na

^{*} Cpecifie in 63.5551113; the force mained an alternative timeframe with the PM control relimement to the permitting althorit tate, focal or trial agenci that has been frequent althorit for this table. The force mained at the prole effect that is in the total engage and operate the PM control to meet the fileration PM control with the PM control of the table of table of table of table of the table of table

[87 FR 60852, Oct. 6, 2022]

40 CFR Ch. I (7-1-23 Edition)

TABLE 4 TO SUBPART DDDDD OF PART 63—OPERATING LIMITS FOR BOILERS AND PROCESS HEATERS

As stated in 63.7500, you must comply with the applicable operating limits:

COLE 4 CO SOCPORO DDDDD OF PORO 630 OPERCOMIG LIMITS FOR COLLERS COD PROCESS HECCERS

□ hen comp⊡ing with a n⊡merica⊡emi⊡ion îmit □n⊡er □a⊡e 1, 2, 11, 12, 13, 14, or 15 of thi□ □□□ part □ing	Yo \square m \blacksquare t meet the \blacksquare operating $\mbox{limit}\square$
 □ et PM ccrimer contro on a coier or proce heater not ming a PM CPMS. 	Maintain the 30 Tal rolling allerage prettire trop and the 30 Tal rolling allerage item from rate at or a long the loweld one hor allerage prettine trop and the loweld one hor allerage item from rate, respectively measure the performance test genonstrating compliance with the PM emission limitation according to 63.530 Talana label to this to the part.
 □ et aci□ ga□ IHC □□ cr □ cr □ contro □ on a coi □ er or proce □ heater not □ ing a HC □CEMS. 	Maintain the 30 a roiling allerage efficient pH at or alloce the lowed one hour allerage pH and the 30 a roiling allerage into flow rate at or alloce the lowed onethour allerage into flow rate measured thing the performance tell cemonitrating compliance with the HClemicion limital tion according to 63.530 and alle to this appart.
 Fa⊡ric fitter contro⊡on a ⊡oiær or proce — heater not —ing a PM CPMS. 	a. Maintain opacit to end than or end to 10 percent opacit or the high eff horm a frage opacit reading meatre thing the performance tet ron termoftrating compliance with the PM for SM emittion imitation that to compliance with the PM for the termination imitation
	□ Inla⊡an□ operate a Lag Bak Letectiontem accorling to L63525 an□ operate the fallic filterch that the Lag Bak Letectiontem alart i□ not actilate□ more than 5 percent of the operating timering each 6□ month perio□
 EĒctro⊡atic precipitator contro⊡on a ⊡oiêr or proce⊡heater not ⊡ing a PM CPMS. 	a. hi option i for i oier an proce heater that operate r contro . et al. a ESP without a wet control Editing an ew oier an proce heater mit maintain opacit to is than or emato 10 percent opacit or the higher horm alerage opacit realing meatre tring the performance tet r n emonitrating compliance with the PM or ISM emition imitation main mock alerage the option i on infor oier an proce heater not effect to PM CPMS or continuo compliance with an opacit imit file, r ESP Maintain the 30 a roling alerage total econtar effectic power inpl of the efformance tet a coring to 63.530 main ale to thi oppart.
5. Dr□ Cr□=er or car⊡on iniection contro⊡on a coiær or proce⊡ heater not ⊡ing a merc⊡r⊡ CEMS.	Maintain the minim⊡m ⊡or⊡ent or car⊡on iniection rate a⊡ ⊡efine⊡ in ⊡63.⊡5⊡5 of thi⊡ ⊡⊐part.
6. □n□ other a□on air po⊡tion contro⊡t_pe on a īoi@r or proce⊡ heater not ⊡ing a PM CPMS.	hi option i for oier an proce heater that operate r contro tem Elling an new boler an proce heater mit maintain opaci it to eit than or ellato 10 percent opacitor the higher hofma are age opacit realing meatre tring the performance tet r n em ontrating compliance with the PM for SM emilion finitation tait
□ Performance te⊡ing	For loi@r an proce heater that lemon trate compliance with a per- formance tell, maintain the 30 a rolling a lerage operating loa of each in that it loe not elcee 110 percent of the highet hor a ler age operating loa recore ing the performance tell.
8. O⊡gen ana⊡er ⊡tem	For loier an proce heater liet to a CO emilion limit that lem on trate compliance with an O ₂ analler lime all pecifie in c3.:525 an maintain the 30 to ling alrage orgen content at or a loie the lower hor alrage orgen concentration meatree iring the CO performance tell, all pecifie in late 8. this returnent loed not appli to init that in later orgen time there in the will be the trim line mote the let pecifie in c3.:525 all
9. SO ₂ CEMS	For <code>ioi@r</code> or proce heater ioi @ct to an HC emiiion fimit that <code>emi</code> on <code>irate</code> complance with an SO ₂ CEMS, maintain the 30 ⁱⁱⁱⁱ @ rolling an erage SO ₂ emiiiion rate at or <code>ieiow</code> the higher horm <code>arerage</code> SO ₂ concentration mealine irring the HCCperformance ter, <code>accpecifie</code> in <code>cace 8</code> .

a wet aci gal crimer is a controlletice that remoted aci galed contacting the committion gal with an a kaine mirror contacting the committion gal with an a kaine mirror contacting the committee committee committee contacting the committee co

[80 FR 72874, Nov. 20, 2015, as amended at 87 FR 60853, Oct. 6, 2022]

Pt. 63, Subpt. DDDDD, Table 5

TABLE 5 TO SUBPART DDDDD OF PART 63—PERFORMANCE TESTING REQUIREMENTS

As stated in §63.7520, you must comply with the following requirements for performance testing for existing, new or reconstructed affected sources:

\Box con \Box ct a performance te \Box for the fo \Box owing po \Box tant	Yo□m⊡t	□īng, a□appropriate
1. Fitera⊡e PM	a. Seïect camping port ocation and the nomer of tracerce point Determine ceïocit and commetric fiow rate of the Clack gad	Metho 1 at 40 CFR part 60, appen :
	c. Determine o⊡gen or car⊡on ⊡o⊡⊡e concentration of the ⊡tack ga⊡	cnapter. Metho 3 or 3 at 40 CFR part 60, ap pen 2 to part 60 of thi chapter, or SZSME P C 19.10 981.ª
	□ Mea⊡re the moi⊡t⊡re content of the □tack ga□	Metho□ 4 at 40 CFR part 60, appen ⊡ □ □ 3 of thi⊓chapter.
	e. Mea⊡re the PM emi⊡ion concentration	Metho 5 or 1 point premere fair fin ter mit pe Metho 5D at 40 CFR part 60, appen 3 3 or 6 of thi chapter.
	f. Con⊡ert emi⊡ion⊡ concentration to per MM⊡t⊡ emi⊡ion rate⊡	Metho 19 Fifactor metho og at 40 CFR part 60, appen 1 of thi chap ter.
2. □SM	a. Select ⊡amp1ng port⊡ location an⊡ the n⊡m⊡er of tra⊡er⊡e point⊡	Metho 1 at 40 CFR part 60, appen 1 = 1 of thi chapter.
	□ Determine □elocit□ an□ □o□metric flow□ rate of the □tack ga□	Metho 2, 2F, or 2G at 40 CFR part 60, appen 2 01 or 2 of thi chapter.
	c. Determine o⊡gen or car⊡on ⊡o⊡⊡e concentration of the ⊡tack ga□	Metho 3 or 3 at 40 CFR part 60, ap pen 3 of thi chapter, or ST SME P C 19.10 1981. ^a
	□ Mea⊡re the moi⊡t⊡re content of the □tack ga□	Metho 4 at 40 CFR part 60, appen 1 0 0 3 of thi chapter.
	e. Mea⊡re the □SM emi⊡ion concentra⊡ tion.	Metho□ 29 at 40 CFR part 60, appen⊡□ □□8 of thi□ chapter
	f. Con ert emi on concentration to per MM to emi on rate	Metho 19 Ffactor metho og at 40 CFR part 60, appen i of thi chap ter.
3. H⊡rogen chiori⊡e	a. Select ⊡ampling port⊡ location an⊡ the n⊡m⊡er of tra⊡er⊡e point⊡	Metho□ 1 at 40 CFR part 60, appen ⊡ □ □ 1 of thi□ chapter.
	 □ Determine le@cit□ an □ o□metric fow = rate of the lack ga□ c. Determine o□gen or car⊡on liolle concentration of the lack ga□ 	Metho□ 2, 2F, or 2G at 40 CFR part 60, appen□□□2 of thi□chapter. Metho□ 3 or 3 □ at 40 CFR part 60, ap□ pen□□ □2 of thi□chapter, or □S1
	□. Mea⊡re the moi⊡t⊡re content of the	□SME P□C 19.10□1981.ª Metho□ 4 at 40 CFR part 60, appen□i□ □□
	 dack ga□ e. Mea⊡re the h⊡rogen chôri⊡e emi⊡on concentration. 	3 of thi⊡ chapter. Metho⊡ 26 or 26⊡ 1M26 or M26⊡ at 40 CFR part 60, appen⊡ ⊡ 18 of thi⊡ chap
	f. Con⊡ert emi⊡ion⊡ concentration to per MM⊡t⊡ emi⊡ion rate⊡	ter. Metho 19 Ffactor metho o g at 40 CFR part 60, appen of thi chap
4. Merc□r□	a. Select _ampling port_ location an the n_m_er of tra_er_e point_	Metho 1 at 40 CFR part 60, appen 1 = 1 of thi chapter.
	Determine elocit an commetric flow rate of the tack gal	Metho 2, 2F, or 2G at 40 CFR part 60,
	c. Determine o⊡gen or car⊡on ⊡o⊡⊡e concentration of the ⊡tack ga⊡	Metho 3 or 3 at 40 CFR part 60, ap pen 0 01 of thio chapter, or 0.5 ₪ SME P C 19.10 1981.ª
	□ Mea⊡re the moi⊡t⊡re content of the □tack ga□	Metho 4 at 40 CFR part 60, appen 2 = 3 of thi chapter.
	e. Mea⊡re the merc⊡r⊡ emi⊡ion con⊡ centration.	Metho 29, 30, or 30 M29, M30, or M30 at 40 CFR part 60, appen 40 of thi chapter or Metho 101 at 40 CFR part 61, appen 6 thi chapter, or SM Metho D6:84. ^a
	f. Con⊡ert emi⊡on⊡ concentration to ⊡ per MM⊡t⊡ emi⊡on rate⊡	Metho 19 Flactor metho o og at 40 CFR part 60, appen of thi chap ter.
5. CO	a. Select the campling port location and the nomer of tracerce point	Metho 1 at 40 CFR part 60, appen i 0

40 CFR Ch. I (7-1-23 Edition)

\Box o con \Box ct a performance te \Box for the fo \Box wing po \Box tant	Yo⊡m⊡t	□⊡ng, a□appropriate
	$\hfill\square$ Determine o $\hfill\blacksquare$ gen concentration of the $\hfill\blacksquare$ fack ga $\hfill\blacksquare$	Metho 3 or 3 at 40 CFR part 60, ap pen 3 of thi chapter, or SM D652200 [Reapprote 2005] or S1 SME PC 19.10[1981.ª
	c. Mea⊡re the moi⊡t⊡re content of the ⊡tack ga□	Metho 4 at 40 CFR part 60, appen i 0
	□ Mea⊡re the CO emi⊞on concentration	Metho 10 at 40 CFR part 60, append 4 of thi chapter. The a meathreement pan table of 2 time the concentration of the applicate emittion fimit.

a incorporate in reference, ince incorporate.

[76 FR 15664, Mar. 21, 2011, as amended at 78 FR 7200, Jan. 31, 2013; 80 FR 72825, Nov. 20, 2015]

TABLE 6 TO SUBPART DDDDD OF PART 63—FUEL ANALYSIS REQUIREMENTS

As stated in $\S63.7521$, you must comply with the following requirements for fuel analysis testing for existing, new or reconstructed affected sources. However, equivalent methods (as defined in $\S63.7575$) may be used in lieu of the prescribed methods at the discretion of the source owner or operator:

Co con⊡ct a f e ana ⊡i for the fo⊡owing po⊡tant	Yo :: m :: t	□□ing
1. Merc⊒r⊡	a. Coïect f⊡e⊡ampïe⊡	Procence in 635217c0 or SCM D5192ª, or SCM D_430ª, or SCM D6883ª, or SCM D2234/D2234Mª for coall or EP0 1631 or EP0 1631E or SCM D6323ª for bill, or EP0 821CR 01:013 ffor into or bill, or SCM D410ª ffor into, or SCM D405ª ffor into, or ectal ient.
	Compo ite fe ampe	Proce re in 63.521 or e alent.
	c. Prepare compolite fle ample	EP: S: 846:3050; for oil ample: S: M D2013/ D2013M for coal S: M D5198 for for in coll or EP: 3050 for foil feet or EP: 821:R:01:013 for int or foil, or editatet.
	□ Determine heat content of the f□e □t□pe.	SIM D5865° for coall or SIM E11° for formation or SIM D5864° for finite and other coffic, or SIM D240° or efficient.
	e. Determine moi⊡tre content of the f⊡ettrpe.	SCM D3103°, CSCM E801°, or CSCM D5864°, or CSCM D240°, or CSCM D95° for inc. fleet, or CSCM D4006° for inc. fleet, or eccletent.
	f. Mea⊡re merc⊡r concentra tion in f⊡e ⊡amp.e.	SCM D6:22ª ifor coall EP: S: #46:41.2 or EP: 1631 or EP: 1631Eª ifor ioii: ampie: or EP: S: #46:46: C4:02ª or EP: S: #46:41.3 ifor ioi: ampie: or EP: 821 R:01:013ª ifor iii: or coii: or coii: or ecii: aent.
	g. Concert concentration into nit of poof merc_r per MM_t of heat content.	For fleelmiltered are Elecation 8 in a63.2530.
2. HC	a. Collect file Lampel	Proce_re in _63521c_ or _S_M D5192a, or _S_M D_430a, or _S_M D6883a, or _S_M D2234/D2234Ma for coall or _S_M D6323a for coall or _iomaS_M D41c_ for _ioi_ feetilor _S_M D405_ for _ioi_ feetilor or ecclatent.
	□ Compo ite f⊡e ⊡ampie	Proce re in 63.521 or e i alent.
	c. Prepare compolite fle ample	EP S 846 3050 a for coil ample SM D2013/ D2013Ma for coal or SM D5198a for ioma or EP 3050a or e i aent.
	□ Determine heat content of the f□e□t□pe.	□SIM D5865ª [for coall or □SIM EI11ª [for liomal] □SIM D5864ª, □SIM D240ª or ell alent.
	e. Determine moi tre content of the flette.	SIM D31 3ª or SIM E8 1ª, or D5864ª, or SIM D240ª, or SIM D95ª for interference or SIM D4006ª for inter filett, or enclatent.
	f. Mea⊡re chorine concentra⊡ tion in f⊡e⊡ampie.	EP: S::::::::::::::::::::::::::::::::::::
	g. Con ert concentration into nit of po n of HC per MM t of heat content.	For fleemittree re Eration in 63.530 and contert from chorine to HC remittipting 1.028.

Pt. 63, Subpt. DDDDD, Table 7

the fo@wing po@tant	Yo I m III	□ □ lng
3. Merc⊡r⊡ F⊡e⊡Specification for other ga⊡ 1 f⊡e⊡	a. Mea remerc con centration in the fer ample an concert to init of microgram per circ meter, or.	Metho 30 M30 at 40 CFR part 60, appen 1 8 of thi chapter or SM D5954ª, SM D6350ª, SO 6988 1:2003⊞9, or SO 6982:2003⊞9, or EP 1631ª or e ann.
	Mea⊡re merc⊡ con centration in the e_ha⊡t ga⊡when firing on □ the other ga 1 fee □ fire □ in the coiler or proce □ heater.	Metho 29, 30 , or 30 M29, M30 , or M30 at 40 CFR part 60, appen 2 B of thi chapter or Metho 101 or Metho 102 at 40 CFR part 61, appen 1 of thi chapter, or SM Metho D6 84ª or e 2 2 ant.
4. CSM	a. Co⊡ct f⊡e⊡ampē⊡	Procence in 63.521c or SM D5192ª, or SM D5430ª, or SM D6883ª, or SM D2234/D2234Mª for coal or SM D6823ª for coal or iomain or SM D410ª, for filled, for filled, or SM D405ª for filled, or SM D405ª for filled, or SM D405B for filled, or SM D405B
	Compo ite fe ampe	Proce re in 63.521 or e aent.
	c. Prepare compolite fle Lample	EP□ S□ [846][3050]] ^a (for [o][] [amp]ē, S□M D2013/ D2013M ^a (for coa, S□M D5198 ^a or □□PP□ [266 ^a] (for ioma, or EP□ 3050 ^a or e][a.ent.
	□ Determine heat content of the f⊡e⊡t⊡pe.	SCM D5865 ^a ffor coall or CSCM EC11 ^a ffor Ciomatti, or SCM D5864 ^a for finite and other coffin, or CSCM D240 ^a or ecclerate.
	e. Determine moi⊡t⊡re content of the f⊡e⊡t⊡pe.	SCM D31 3ª or SCM E81ª, or D5864ª, or SCM D240ª, or SCM D95ª for tool feett or SCM D406ª for tool feett or SCM D410ª for too feett or SCM D405ª for too feett or SCM D410ª
	f. Mea⊡re ⊡SM concentration in f⊡e⊡ampīe.	SIM D3683ª, or SIM D4606ª, or SIM D635Iª or EPI 200.8ª or EPI SI 846:6020aª, or EPI SI 846:6020aa, or EPI
	g. Con⊡ert concentration□ into _nit□ of po_n⊡ of _SM per MM_t□ of heat content.	For fletmittret the Ethation 9 in t63.530.

^a incorporate □ □ reference, □ee □63.14.

[83 FR 56725, Nov. 14, 2018]

TABLE 7 TO SUBPART DDDDD OF PART 63—ESTABLISHING OPERATING LIMITS $^{a\,b}$

As stated in §63.7520, you must comply with the following requirements for establishing operating limits:

io le 🗆 do Soopord DDDD!) of Pord 63	ESOLISHIG	OPER G	L _M S ^a
--------------------------	--------------	-----------	--------	-------------------------------

fi ⊡o ha e an appica e emi⊡ion imit for	□n□ ⊡o⊡r operating imit□ are □a⊡e□ on	Yo II m III t	□ ⊑ing	□ccor⊡ing to the fo⊡ owing re⊡irement□
1. PM, ⊡SM, or mer⊡ c⊡	a. □ et ⊡cr⊡⊡er oper⊡ ating parameter⊡	i. Eltalith a litelipe cific minimim licring rer prelite rop and minimim flow rate operating limit according to 063.0530000	□ □Data from the □crimer preime □cop an □imiof fow rate monitor□ an the PM, □SM, or mercim performance tell.	a Yo m t collect crimer preme roop an information of the performance tet Determine the low eit hor a crage crimer preme roop an information the hor a crage ing all of the 15 minite realing taken ring each performance tet.

40 CFR Ch. I (7-1-23 Edition)

	of Para 63 Escalshing Operating	L _M s ^a Contin e

ff ⊡o⊟ ha⊡e an appîca⊟ ⊡ē emi⊡ion îimit for	□n□ ⊡o⊡r operating imit□ are □a⊡e□ on	Yo II m III	□ ⊑ing	□ccor ing to the fo □ iowing re □ rement □
	□ Eectro⊡tatic precipi⊡ tator operating pa rameter⊡option on for Cnit that operate wet cr⊡ter	i. Eftailin a Itellipe cific minimim tota leconlare ectric power inpl accort ing to 63.1530	□ □Data from the □ot□ age an □ □econ□ar□ amperage monitor□ □rring the PM or merc□r□ performance te □.	a Yo m t collect econ ar o tage an econ ar am perage for each ESP cellan calculate tota econ ar erec tric power inpl tata eler 15 minte ming the entire pel rio of the perform ance telt Determine the arr age tota econ ar electric power inpl comp ting the hor areage taken ming each performance telt.
	c. Opacit	i. E⊡a⊡⊡h a ⊡te⊡pe⊟ cific ma⊡m⊡m opac⊟ it⊡ e⊡e⊡	I ⊡Data from the opac it⊡ monitoring ⊡tem ⊡ring the PM per⊟ formance te⊡.	all vol mail collect opacit: realing: eler: 15 minite: initiation of the performance ance tell: initiation of the performance and the performance initiation of the performance realing compliang the horizaterage initiation of the total minite realing taken oring each performance tell. colletermine the high eff horizaterage opacit: realing meather end initiation the tell rin emble on initiating complian ance with the PM for SMILemilion fimitation.
2. HC	a. □ et ⊡cr⊡er oper⊡ ating parameter⊡	i. Eta⊡th ⊡te⊡pecific minim⊡m eff⊡ent pH an_ föw rate oper ating imit⊡ accor⊡ng to _63.⊑530	□ Data from the pH an □ IIII fow/rate monitor an the HC performance te 1.	a Yo m corect pH an in i fow rate at a e er 15 min te i ng the entire perio of the per formance te 1 Determine the hor a erage pH an i fow rate comp ting the hor a erage ing a for the 15 min te rea ing taken i ng each performance te 1.

Pt. 63, Subpt. DDDDD, Table 7

	² □R□ 63 □	ESOLSHOG	OPER	L1MIIS ^a	Contin⊡e□
--	-----------------------	----------	------	---------------------	-----------

fi ⊡o ⊨ha e an appîca ⊡e emi⊡ion îmit for	□n□ ⊡o□r operating imit□ are □a⊡e□ on	Yo II m III	□ ⊑ing	□ccor ing to the fo □ owing re □ irement □
	□ Dr□ cr==er oper□ ating parameter□	i. E[la][] h a []tellpe cific minim] for lent in Betion rate operating limit ac cor[] ing to 63.530[] ff ff ferent aci ga forCent] are i.e limit he HC[per] formance tell, the allerage for each forCent i.e come the fitellpe cific operating limit for that forCent.	☐ □Data from the or ent in action rate monitor an HCor mercor performance telt.	a Yo m to collect or ent in action rate at a e er 15 min te or gene of the perio formance te to
				□□Determine the hor □ arrage for ent in action rate com pting the hor arrage ing afor the 15 min te rea ing taken tring each performance te 1.
				□ c:Determine the ōw□
	c. ⊡ternati⊡e Ma⊡m⊡m SO₂ emi⊡ion rate.	i. Etta tith a fite tpe cific ma m m So, emiifion rate oper ating finit according to 63.530	□ □Data from SO ₂ CEMS an□ the HC□ performance te īt.	a Yo m t collect the SO ₂ emilion ata accoring to accoring to accoring to accoring to accoring to accoring tend the molt recent HC performance tett accore the accore the accore the to accore the accore to accore to
3. Merc⊡r⊡	a. □cti⊡ate□ car⊡on in□ ection.	i. E[ta:]]Th a []te:]]pe: cific minim]m acti []ate:] car[]on in @c] tion rate operating]Imit accor[]ing to []63.[]530[]]]	□ □Data from the acti ate □ car_on rate monitor□ an □ mer□ c □ □ performance te □.	a No min collect ac tilate carlon initic tion rate lata eler 15 min te ining the entire perio of the performance telt initiate actilate carlon initiction rate carlon initiate carlon gallof the 15 min te realing taken tring each performance telt.

40 CFR Ch. I (7-1-23 Edition)

fi ⊡o ha⊡e an applica ⊡e emi⊡ion iîmit for	□n□ ⊡o□r operating imit□ are □a⊡e□ on	Yo II m III	□ īing	□ccor ing to the fo □ iowing re □irement □
 Car⊡on mono⊡⊡e for which compîance i□ ⊑emon∐rate□ □ a 	a. O⊡gen	i. E⊡ta⊡th a ⊡nit⊡pe⊡ cific îlmit for minim⊡m o⊡gen ie⊡e⊐accor⊡	□□Data from the o gen ana⊡rer □□tem □pecifie□ in	c:Determine the low ell hor alerage ella line ling the performance tell a lor operating limit. hen lor nit operate at lower loa milpior actinate caron in ection rate the loa fraction, a lef fine in 63.55, to the termine the re lire in ection rate. a Yo mil collect of gen lata eller 15 minile liring the
performance te 1.		ing to □63.⊏530⊞⊡	_63. ⊑525™ □	entire perio of the performance te1 Determine the hor arrage orgen concentration compling the hor arrage ing arrof the 15 min te real ing taken ing each performance te1. cDetermine the low eilthor arrage elasifie oring the performance te1 arrage finit
5. □n□ poⅢtant for which comp≋ance i□ _emonîtrateî □ a performance teīt.	a. ⊡oier or proce∷ heater operating oa⊥	i. Eſta⊡th a ⊡nit ⊡pe⊟ cific îlmit for ma⊟ im⊡m operating toa⊟ accorîng to ⊡63.⊡5201c⊡	□ Data from the oper ating ba monitor or from ⊡eam gen eration monitor	a Yo m c corection generating ball. a Yo m c corection generation team generation team generation team generation team generation team generating the entire period of the perdisor formance ted. Determine the alerd age operating the hord alerage taken ding each performance ted. Determine the highd ed hord alerage the three ted r ch alerage doing the performance ted. and m tipd thidd alerage 1.1 do percental Determine the disor alerage doing the the performance ted. and the three ted the the ted the the performance ted. and the three ted the the ted t

COLE O SOCPORO DDDDD OF PORO 63 ESCOLISHING OPERCOMIG LIMINS & C Contin Leo

^a Operating limit mit the confirmer or reellatified tring performance tella ^d to conflict mittipe performance tella to mit tet the minimum little flow rate and prettine from operating limit at the higher of the minimum latter established tring the performance tella For a minimum origen life if to conflict mittipe performance formance tella for mailment the minimum origen life at the lower of the minimum are established tring the performance tell For mailment operating ligal, if to conflict mittipe performance tella for mittine the mailment operating ligal at the lower of the mailment latter established tring the performance tella.

[87 FR 60853, Oct. 6, 2022]

Pt. 63, Subpt. DDDDD, Table 9

TABLE 8 TO SUBPART DDDDD OF PART 63—DEMONSTRATING CONTINUOUS COMPLIANCE

As stated in 63.7540, you must show continuous compliance with the emission limitations for each boiler or process heater according to the following:

Collecting the opacitic monitoring them that according to the construction and the construction according to the construction and the construction according to the construction according
Religing the opacitimonitoring lata to obminite aleragelian. Maintaining laterial lock alerage opacition is than or ellato 10 percent or he highest hore alerage opacitimeating meature initiation finitation.
Collecting the PM CPMS of pill fata according to 183.2550 Remains the fata to 30 Tail rolling allerage than Maintaining the 30 Tail rolling allerage than
perating init e_taline □ □ring the performance telt accoring to :63.□530□□14□
tating an operating a tag teak tetection to the according to 63.525 and operating the fabric filter that the retirement in 63.540 attract are met. Collecting the pretime to pan the fabric filter that the retirement of the fabric filter that according to the fabric filter fabric fabric filter that the fabric filter that that the fabric filter that that that the fabr
Rei Cing the Lata to 30 la rolling alerage lan Maintaining the 30 la rolling alerage pretter crop and to fow frate at or aloce the operating limit elatithe tring the performance tell according to 163 f530 ^m
Co⊡ecting the pH monitoring □□_tem □ata accor⊡ing to □=63.□525 an□ 63.□535□ an□
Reflecting the fata to 30 Tairoling alerage than a factor of the operating limit eltate Maintaining the 30 Tairoling alerage pH at or afole the operating limit eltate Theo thing the performance tell according to 683.5530 TT
Co⊡ecting the cortent or carton in≊ction rate monitoring tittem tata for the tre tortiter accorting to title3.t525 and 63.t535tand Retting the tata to 30tat rotting ateragettand
Maintaining the 30 a count of a carage concent or carcon intection rate at or aco the minimum concent or carcon intection rate a carcon in cost. 555.
Corecting the totaleconlar electric power inpl monitoring the late for he electrolatic precipitator according to 63.525 an 63.535an Relicing the late to 30.2 or colling aleragelan
Maintaining the 30⊡a rolling alerage tota⊡econ⊡ar⊟etectric power inp⊡t at or aloce the operating limit⊟e⊡alithe⊡ tring the performance tell according to 163.1530⊡
Con ct month fe ana for HC or mercir or SM accoring to a 6 o this copart and
Re⊡ce the Cata to 12.month ro⊡ng aCerage⊡an⊟ Maintain the 12.month ro⊡ng aCerage at or Ceröw the appfica⊡e emi⊡on imit or HCCor mercC⊡or CSM in Ca⊡e 1 an 2 or 11 thro⊑gh 15 to thi⊡ ⊡_part.
Cac⊡āte the HC_ merc⊡ an⊡or ⊡SM emi⊡ion rate from the ⊡oiær or proc e⊡ heater in ⊡nit⊡ of ⊡/MM⊡t⊒ ⊡ing E⊡ation 15 an⊡ E⊡ation⊡ 16, 1⊐, an⊡/or 8 in ⊡63.:530.
Contino and motion the origen content and an origen analiter and a continue to control that in talant control to an and a control that in talant and any and the time and the control that in talant and the time and the control the control that control the control that and the control the control that an
ie⊜ in ⊜63.5555a.⊞⊡ Re⊡cing the ⊡ata to 30.≣a⊜ ro≣ng a⊡erage⊡an⊡
Maintain the 30 a coing a crage of gen content at or a core the lower hore a crage of gen le concerned the concerned the concerned to the conc
collecting operating loal late of learn generation late elergins minitely Reflicing the late to 3011a1 rolling allerage tand Maintaining the 3011a1 rolling allerage operating load firch that it roed not efficient
he performance tell according to accord to ac
Co⊡ecting the SO₂ CEMS o⊡p⊡t ⊑ata accor⊡ing to _63.⊑525. Re⊡cing the ⊑ata to 30.≣a⊡ ro≣ng a⊡erage⊡an
Maintaining the 30 Ta ⁻ rolling a transformation of the second

[78 FR 7204, Jan. 31, 2013, as amended at 80 FR 72829, Nov. 20, 2015; 87 FR 60855, Oct. 6, 2022]

TABLE 9 TO SUBPART DDDDD OF PART 63-REPORTING REQUIREMENTS

As stated in 63.7550, you must comply with the following requirements for reports:

40 CFR Ch. I (7-1-23 Edition)

You mult comit a h	□he report m⊡t contain	Yo m t mut the report
1. Compilance report	a. Information re⊡ire⊟ in ⊡63.⊡5501è⊞1⊡thro⊡gh 15⊞an⊡	Semiann a III, ann a III, iennia III, or e er 5 ear accoring to the re irement in 63.550 III
	If there are no cellation from an emilion ilmitation iemilion ilmit an operating ilmit that apple to co an there are no cellation from the remement for work practice tan ar for period of tartp an httown in alle 3 to the sent that apple to co a tatement that there were no cellation from the emilion limitation an work practice tan ar fing the reporting period if there were no period tring which the CMS_ incluing continod emilion monitoring lifem, continod opacit monitoring filem, an operating parameter eter monitoring filement were of of control all pecifies in constant there were no teriod from an emilion from the emilion in the cMS were of of control there were no period file the cMS were of of control there were no period and comparing file of a constant there were no period and comparing file of the comparison of the comparison of the containt or operating file to an a constant the containt the information in 63.5500 man if there were period file monitoring filement, were of of control emilion file of the contain the information in constant in operating parameter monitoring filement were of for containt the information in containt the information in containt the information is don't monitoring filement contain the information in containt monitoring filement emilion file period file of the containt the information in containt monitoring filement emilion file period file of the containt the information in containt monitoring filement in contain the information in containt the information in containt the information in contain the information in containt the information in containt the information and the information and the containt the information in containt the information	

[76 FR 15664, Mar. 21, 2011, as amended at 78 FR 7205, Jan. 31, 2013; 80 FR 72830, Nov. 20, 2015]

TABLE 10 TO SUBPART DDDDD OF PART 63—APPLICABILITY OF GENERAL PROVISIONS TO SUBPART DDDDD

As stated in §63.7	565, you	must	comply	with	the	applicable	General	Provisions	according
to the following:									

Citation	S	□ppîe to □□part DDDDD
□63.1		Ye□
□63.2	Definition	Ye Continuaterm Cefine in Continuation 10
□63.3	□nit□an□ □□□re liation□	Ye□
□63.4	Prohi ite Cti itie an Circ mention	Ye□
□63.5	Precon⊡tr⊡ction Re⊡ew an⊡ □otification Re⊡irement□	Ye□
063.6 a, 111 1115, 111, 0c0.	Compliance with Stanlar and Maintenance Relifement	Ye□
-63.6 e 11 11	Genera□□t□ to minimi⊡e emi⊡lon□	o. See 63. 500 a 3 for the genera t re fre ment.
□63.6 e □1 □ìi □	Re ⊡irement to correct ma f⊡nction □ a □ □oon a □ practica □ē.	□o.
□63.6 e □3 □	Start⊡p, ⊡h⊡t⊡own, an □ malf⊡nction plan re⊡irement □	□ o.
□63.6 1 □1 □	Start_p, httown, an maffinction elemption for compili ance with non@paciteeniilion tantart.	□o.
63.6112 an 3	Compîance with non opacit emi ion tan ar	Ye□
□63.6 [°] g□	□ ce of atternatice ctancar contraction contractions and atternation contractions and atternations and atte	Ye ecept 63.555 113 pecifie the proce for application an appro a for an alternatile timeframe with the PM controll re former in the fartp work practice 12
□63.6 ⓑ □1 □	Start_p, httown, an mafinction elemption to opacit tan aro.	□o. See □63. □500 a □
□63.6 h □2 □to h □9 □	Determining compilance with opacit⊟emi⊡ion ⊡an⊡ar⊡	o. S⊡part DDDDD □pecifie opacit□a⊡an operating imit not an emi⊡on ⊡tan⊡ar⊡

Pt. 63, Subpt. DDDDD, Table 10

Citation	S	□ppîe□to □□part DDDDD
□63.61□	E⊡ten⊡on of comp1ance	Ye == ote: Facilitie = mails = mo re == eten=ion= of com pfance for the intation of com= heat an power, walte heat reco=er_ or ga pipefine or f== feeing infra tr=ct=re a = mean= of comp=ing with this == part.
63.6	Predientialeemption.	Ye Ye o. S part DDDDD pecifie contion for cont ting performance te at cont to cont cont to cont to cont to cont to cont to cont to con
	Performance □e⊡ting Re⊡irement□	Yel
63.8 a an 0 63.8 c 1 0	□ppî≿a⊡it⊡an□ Con □ct of Monitoring Operation an□ maintenance of CMS Genera□□□ to minimi⊡e emi⊡ion□ an□ CMS operation	Ye Ye Do See 063 500 3 3
63.8 c 1 1 1	Operation an maintenance of CMS	Ye□ □0.
63.8 C 2 to C 9	Operation an⊟ maintenance of CMS Monitoring Re⊡irement□ □ aît⊡ Contro⊡Program	Ye⊡ Ye⊡
G3.8 III 3 C	□ ritten proce⊡re□ for CMS	Ye_ elcept for the at ten tence, which refer to a tart_p, thtown, an main forction plan. Start_p, tht own, an mainction plan are not refire
63.8 @	Performance e a lation of a CMS	
63.8 0	Re⊓ction of monitoring rata	Ye⊓
□63.9	□otification Re⊡irement□	Ye
63.10 a., 11 63.10 2 ii	Recor⊡keeping an Reporting Re ⊡rement Recor⊡keeping of occ⊡rence an ⊡ration of ⊡tart⊡p or	Ye⊡ Ye⊡
□63.10 □□12 □ii□	□h □t ⊡own □ Recor □keeping of matf⊡nction □	□o. See □63. 555
		recorEkeeping of occErrence an Tration an 63.5555186 for action taken Tring matEnction
63.10 2 iii 63.10 2 iii	Maintenance recor⊡ □ction□ taken to minimi⊡e emi⊡ion□ □ring □tart□p, □h□t□ □own or maif□nction	Ye□ □o.
63.10 2 1	Recor⊑keeping for CMS mat⊡nction⊡ Other CMS re⊡rement	Ye⊡ Ye⊡
	Recor keeping re⊡irement for appîca îît ⊡etermination	□o. Xo□
□63.10ic 10 □an □ 11 □	Recording nature and cattle of mattinction, and corrective action	O. See 63. 555 for recor keeping of occ rrence an cration an 63. 555 for action taken cring matEnction
63.10 c 12 an 33	Recor keeping for corce with CMS	Ye
63.10 1 an 2	GeneraTreporting re∏irement⊓	⊔u. Ye⊓
□63.10 □□ 3□	Reporting opacit□ or □□□ e emi □ion o □er ation re □t□	□0.
63.10 4	Progre report ner an eten ion of compliance	Ye
⊔ʊᲙ.1ՍШШ5⊔	start∟p, ∟n⊥Lown, an⊔ ma⊥Lnction report⊡	Lo. See ∟63.∟550℃ 11 for maff_nction reporting re ⊡irement□
63.10 e	□□□itiona□reporting re□□irement□ for □o□rce□ with CMS	Ye□
□63.10ff□	□ ai⊡er of recor⊑keeping or reporting re⊡irement□	Ye
□63.12		_o. Ye⊡
□63.13□63.16	all	Yel
63.1a15;;a1111a19; 1112; c=3;q41; 63.61116; c=3;a44; 11; fe12; e1311; h13; h1511; 63.8a3; 63.9113; h14; 63.10;c12114; c19.	Relere	□0.

40 CFR Ch. I (7-1-23 Edition)

[76 FR 15664, Mar. 21, 2011, as amended at 78 FR 7205, Jan. 31, 2013; 80 FR 72830, Nov. 20, 2015]

TABLE 11 TO SUBPART DDDDD OF PART 63—ALTERNATIVE EMISSION LIMITS FOR NEW OR RECONSTRUCTED BOILERS AND PROCESS HEATERS THAT COMMENCED CON-STRUCTION OR RECONSTRUCTION AFTER JUNE 4, 2010, AND BEFORE MAY 20, 2011

f or ofer or proce heater i in thi com categor	For the fo⊡owing po⊡tant⊡	□he emi⊡on m⊡t not eccec the following emi⊡on imit, eccept ⊡ring perio⊡of tartcp an □httown	□ ing thi□ tpecifie□ tamp ing to the or test ron to the test ron
1. onito in a concato egorieo concato orn conto foeo	a. HC□	0.022	For M26□, co⊡ect a minim⊡m of 1 ⊡cm per r⊡n⊡for M26 co⊡ect a minim⊡m of 120 îter⊡ per r⊡n.
2. Init in a class of the second seco	a. Merc⊡r⊡	8.0E⊡0⊒ª ⊞ per MM⊡t⊡ of heat inp⊡t.	For M29, collect a minim⊡m of 4 lincm per r⊡n for M30 or M30, collect a minim⊡m liample a ⊡peolife in the metholifor _SSIM D6_84 o collect a minim⊡m of 4 lincm.
a control of the con	a. Merc⊡r⊡	2.0E⊡06 iii per MM⊡t⊡ of heat inp⊡t.	For M29, coilect a minimim of 4 incm per rini for M30 or M30, coilect a minimim lample a⊡pecifie in the methoinfor ISIM D6:84 o coilect a minimim of 4 incm.
4. □nit□ e ign to □rn coa/lo i□ fo □ fe □	a. Fittera⊡ë PM tor ⊡SM⊡	1.1E _03	Corect a minim⊡m of 3 ⊡cm per r⊡n.
5. Porerie coaloi er eigne to orn coaloi foi fe	a. Car⊡on mono⊡⊏e ICO ⊡or CEMS :	130 ppm □ o me on a □ a o recte to 3 percent o gen, 31 n a erage or 320 ppm □ b me on a r a corr recte to 3 percent o gen, c 30 a rom ing a erage	1 hr minim⊡m ⊡ampûng time.
6. Stoker eigne to in coa//oii foii fee	a. CO lor CEMS	130 ppm □ olme on a □ a □ a correcte to 3 percent olgen, 31 n alerage or 340 ppm □ bolme on a □ a corr recte to 3 percent olgen, ° 10 a roll ing alerage	1 hr minim⊡m ⊡amp1îng time.
□ F⊡ííce ce nit cígne to ⊡n coa/ coi⊡ fo⊡ífec	a. CO [or CEMS]	130 ppm	1 hr minim⊡m ⊡ampûng time.

Pt. 63, Subpt. DDDDD, Table 11

I ⊡ r ⊡oier or proce heater i⊡ in thi⊡ categor	For the fo⊡wing po⊡tant□	□he emi⊡on m t not e ceel the fo towing emi⊡on timt, e cept ⊡ring perio tof tart p an tht own	□ ũng thi□ tpecifie□ tamp îng to⊡me or tet r⊡n ⊡ration
8. Filiciae des chito	a. CO lor CEMS	140 ppm 🗆 💿 me on	1 hr minim⊡m ⊡ampûng time.
with an integrate⊡ heat e⊡changer ⊡e ⊡gne⊡ to ⊡m coa∄ co⊞ fo⊡ifice⊡		a in a correcte to 3 percent orgen, 310 n a erage or 50 ppm b bolme on a in a corr recte to 3 percent orgen,° 30 a roll ing a erage	
9. Stoker Cope grate/ other clgne to on wet coma fee	a. CO tor CEMS	620 ppm icine on a icine or a correcte to 3 percent orgen, 31 n alerage or 390 ppm icine or nectel to 3 percent orgen, corrected to 3 percent orgen, corrected to 3 percent	1 hr minim⊡m ⊡ampûng time.
	□ Fitera □ē PM tor □SM□	3.0E_02 per MM⊡t⊡ of heat inp⊡t⊡or I2.6E_05 per MM⊡t⊡ of heat inp Tt□	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
10. Stoker / cope grate/other cigne to cm kin rie co ma fee	a. CO	560 ppm	1 hr minim⊡m ⊡ampîng time.
	□ Fitera □ē PM tor □SM□	3.0E ⊡02	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n
11. Fuire e nit egge to on io mai/iomae oin	a. CO Ⴊr CEMS⊡	230 ppm i olime on a corrected to 3 percent oligen, 312m alerageor 310 ppm i bolime on a coloridation rected to 3 percent oligen,° 30 alerage	1 hr minim⊡m ⊡ampîng time.
	□ Filtera □ē PM tor □SM□	9.8E_03	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n
12. S peníon mere eigne to maio mai/iomate toim	a. CO tor CEMS	2,400 ppm ome on a o o recte_ to 3 percent o o a_er_ age_or _2,000 ppm ome on a a ome on a	1 hr minim⊡m ⊡ampîng time.
	□ Fitera □ē PM tor □SM□	3.0E ⊡02 per MM⊡t⊡ of heat inp⊡t⊡or 	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
13. Ditch Oleni/Pië	a. CO lor CEMS	1,010 pm i olime on a r-laicori rectel to 3 percent oligen, 3in aler agelor 1520 ppm i bolme on a r- alicorrectel to 3 percent oligen ≤ 100	1 hr minim⊡m ⊡ampûng time.
	□ Filtera⊡ē PM lor ISM□	8.0E_03	Cor≣ct a minim⊡m of 3 ⊡cm per r⊡n.

40 CFR Ch. I (7-1-23 Edition)

f or oier or proce heater i in thi con categor	For the fo⊡owing po⊞tant□	he emi⊡on m t not eccet the focwing emi⊡on imit, eccept ing perio of tartip an htown	□ îng thi □ pecifie □ amp îng ⊡o ⊡me or te t r⊡n □ ration
14. F e ce init e igne to in io ma iona e oim	a. CO	910 ppm □ o me on a □	1 hr minim⊡m ⊡ampûng time.
	SM	2.0E 02	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
15. Hirri Topenion grate Coller Engenen to Im Toma // for ale: coll	a. CO tor CEMS	1,100 ppm i o'ime on a r'aiori recte to 3 percent o'igen, 31'n a'er age'or 1900 ppm i o'ime on a r'aiori aiocorrecte to 3 percent o'igen, °30 a roiling a'erage	1 hr minim⊡m ⊡ampîng time.
	SMC	2.6E_02	Collect a minimum of 3 LLcm per run
16. Init Ie Igne to	a. HC□	4.4E 04	For M26 :: Collect a minim m of 2cm per r n for M26, collect a minim m of 240 liter per r n
	□ Merc□r□	4.8E [_] O□ ^a [_] per MM□t□ of heat inp□t.	For M29, collect a minimum of 4cm per r_n_ for M30_ or M30_, collect a minimum _amp@ apecifie_ in the methofor _S_M D6_84 _ collect a minimum of 4cm.
1 onito cerigneo to	a. CO	130 ppm	1 hr minim⊡m ⊡ampûng time.
	SM	1.3E 02	Corect a minim⊡m of 3 ⊡cm per r⊡n.
18. onito edigneo to	a. CO	130 ppm □ o me on a □ a □ correcte to 3 percent o gen.	1 hr minim⊡m ⊡ampûng time.
	C Filtera C PM for	2.0E 03 a per MM to of heat inp tor 2.9E 05 per MM to of heat inp to	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n
19. Init I I I I I I I I I I I I I I I I I I I	a. CO	130 ppm Collection on Collection o	1 hr minim⊡m ⊡ampûng time.
	C Filtera C PM for	2.3E 02	Co⊡ect a minim⊡m of 4 ⊡cm per r⊡n
20. onito congretations of the constraints of the c	a. CO	130 ppm	1 hr minim⊡m ⊑ampûng time.
-	□ HC□	1. □E □03 □ per MM □t □ of heat inp □t.	For M26_, Collect a minimlm of 2 licm per rlnfor M26, collect a minimlm of 240 liter per rln.
	c. Merc⊡r⊡	□9E□06	For M29, collect a minim m of 3 licem per rm for M30 or M30, collect a minim m tample a pecifie in the methor for SSM D6-84 or collect a minim m of 3 licem.
Pt. 63, Subpt. DDDDD, Table 12

I □ □ □ □i ēr or proce □ heater i □ in thi □ □ □ □ categor □	For the fo⊡owing po⊞tant⊡	□he emi iion m it not e cee the fo @wing emiiion imit e cept ⊡ring perio of tartp an t town	□ ling thi □ lpecifie □ lamping lo □me or telt r ln □ ration
	□ Filtera⊡ë PM lor ⊡SM□	6⊑03per MM⊡t⊡ of heat inp⊡t⊡or 2.1E04per MM⊡t⊡ of heat inp_tt	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.

a f to are context and the to the text of tex

[87 FR 60855, Oct. 6, 2022]

TABLE 12 TO SUBPART DDDDD OF PART 63—ALTERNATIVE EMISSION LIMITS FOR NEW OR RECONSTRUCTED BOILERS AND PROCESS HEATERS THAT COMMENCED CON-STRUCTION OR RECONSTRUCTION AFTER MAY 20, 2011, AND BEFORE DECEMBER 23, 2011

f or oier or proce heater i in thi⊡	For the fo⊡owing po⊞tant□	□he emi⊡ion□ m⊡t not eīcee□ the foïowing emi⊡ion îmit□ eīcept ⊡ring perio⊡ of ⊡art⊡p an□ hītūown	□ îng thi □ pecifie □ amp îng to □me or tet ron □ ration
1. onito in a cato egorieo cato to com coin fee	a. HC	0.022	For M26□, collect a minim⊡m of 1 ⊡cm per r⊡n⊡for M26 collect a minim⊡m of 120 fiter⊡per r⊡n.
	□ Merc□r□	3.5E⊡06 ª ⊞ per MM⊡t⊡ of heat inp⊡t	For M29, collect a minimum of 3 limm per rinfor M30 or M30, collect a minimum lample all pecified in the methomfor ISIM D6_84 collect a minimum of 3 limm.
2. Init I I I I I I I I I I I I I I I I I I I	a. Filtera ⊡e PM lor □SM⊡	1.1E_03 iii per MM_ti of heat inp_tior 2.3E_05 iii per MM_ti of heat inp_ti	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
3. P erie coa oier eigne to on coa/oii fo	a. Car⊡on mon⊡ o⊡e tCO⊡or CEMS⊡	130 ppm : come on a r: aic cor recte to 3 percent orgen, 31 n aic erage or 1320 ppm : come on a r: aic correcte to 3 percent or een e 30 a roima a rerage	1 hr minim⊡m ⊡ampûng time.
4. Stoker □ e igne □ to □m coa/⊥o î fo i f e □	a. CO lor CEMS⊡	130 ppm D D me on a D and cor rectel to 3 percent o gen, 37 n a erage or 1340 ppm D Dome on a real correctel to 3 percent o gen < 10 a rolling a Percent	1 hr minim⊡m ⊡ampûng time.
	a. CO lor CEMS⊡	130 ppm Dome on a Plad cor rectel to 3 percent orgen, 37 n an erage or 1230 ppm Dome on a relation corrected to 3 percent or gen < 30 main or her argen	1 hr minim⊡m ⊡ampûng time.
6. F i i e e nit with an inte grate heat e changer e gne to n coa//oi fo f fe	a. CO lor CEMS⊡	140 ppm in the first state of the state of t	1 hr minim⊡m ⊡ampûng time.
□ Stoker// ope grate/other □ e igne to orn wet ioma □ f e	a. CO lor CEMS⊡	620 ppm Dollme on a rolatic cor rected to 3 percent orgen, 31 n an erage or 1390 ppm Dollme on a rolatic corrected to 3 percent or gen, 6 30 mail rolling alterage	1 hr minim⊡m ⊡ampûng time.
	□ Fiftera ⊡e PM or SM□	3.0E_02	Co.≣ct a minim⊡m of 2 ⊡cm per r⊡n.

Pt. 63, Subpt. DDDDD, Table 12

40 CFR Ch. I (7-1-23 Edition)

I ⊡o⊡r ⊡oi≣r or proce	For the fo⊡owing po⊞tant□	□he emi⊡on m	□ îng thi □ pecifie □ ampîng □o □me or te lt r n □ □ ration
8. Stoker / @pe grate/other @e grate/other @e igne to @rn kin @rie doma fre	a. CO	460 ppm □	1 hr minim⊡m ⊡ampûng time.
	□ Fittera ⊡e PM tor □SM □	3.0E_02	Co≣ect a minim⊡m of 2 ⊡cm per r⊡n.
9. Filine e nit eigne to mioma filo a e oim	a. CO tor CEMS⊡	260 ppm Comme on a Comme on	1 hr minim⊡m ⊡ampîng time.
	□. Filtera □e PM or □SM□	9.8E_03	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
10. S pen ion	a. CO tor CEMS⊡	2,400 ppm i olime on a riai correcte to 3 percent oligen, 3rin alerage or 2,000 ppm i olime on a riai correcte to 3 percent oli Gen,e 10 aleroing alerage	1 hr minim⊡m <i>⊡</i> ampîng time.
	□ Fitera □e PM or □SM □	3.0E_02	Co≣ect a minim⊡m of 2 ⊡cm per r⊡n.
11. Dotch Olen //Pile Imero le Gigneo to om lioma // lio la leo lo îm	a. CO lor CEMS	4:0 ppm : comme on a creatic correcte: to 3 percent orgen, 3:cn a erage or :520 ppm : comme on a create correcte: to 3 percent or gen; 0:10:12: croiing a terage:	1 hr minim⊡m <i>⊡</i> ampîng time.
	□ Fitera⊡e PM or □SM□	3.2E 03	Co.≣ct a minim⊡m of 3 ⊡cm per r⊡n.
12. F e ce init	a. CO	910 ppm □ o me on a ro a ildo coro recte to 3 percent o me gen.	1 hr minim⊡m ⊑ampûng time.
	□ Fitera⊡e PM tor □SM□	2.0E 02	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
13. H i i pen ion grate coier eigne to m ioma i o ace o i	a. CO lor CEMS	1,500 ppm □ bolme on a r iai correctel to 3 percent oligen, 3 in alerageTor 1900 ppm □ tolme on a r all correctel to 3 percent oli gen,° 30 all rolling alerage	1 hr minim⊡m ⊡amping time.
		2.6E_02 iii per MM⊡ti of heat inp_tion ⊈.4E_04 iii per MM⊡ti of heat inp_ti	Collect a minimum of 3 LLcm per run.
to Corn inci fie.	a. HC	4.4EL04 III per MMLtL of neat inpLt	im m of 240 iter per r_n.
	⊔ Merc∟r⊔	4.8ELU⊔ ^a Ш per MM⊡t⊔ of neat inp⊥t	For M29, couect a minimum of 4cm per rbnfor M30 or M30 c. collect a minimum ramp'e a pecifie in the methor for SSM D6.84 collect a minimum of 4cm.
15. □nit□ □e፲igne□ to □rn hea □ î□□ □ □ f □e □	a. CO	130 ppm i collme on a critical cort recteto 3 percent of gen, 3 i n at erage.	1 hr minim⊡m ⊡ampûng time.
	□ Fitera	1.3E 02	Co≣ect a minim⊡m of 2 ⊡cm per r⊡n.
16. onito ce igne to orn ight into fee.	a. CO	130 ppm column on a r column or a column corrected to 3 percent or gen.	1 hr minim⊡m ⊑ampûng time.
	□ Fitera ⊡e PM tor □SM□	1.3E_03 a m per MM_to of heat inp_to or 2.9E_05 m per MM_to of heat inp_to	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
1 onit ceigne to orn in fe that are non conti nenta nit	a. CO	130 ppm 130 mme on a r 130 cor recte 10 3 percent o 1 gen, 3 r 1 a 1 erage 1 a 1 o 1 ack te 1.	1 hr minim⊡m ⊡ampûng time.
	□ Fitera □e PM or □SM□	2.3E ⊡02 □ per MM⊡t□ of heat inp⊡t⊡or 18.6E ⊡04 □ per MM⊡t□ of heat inp⊡t□	Co⊡ect a minim⊡m of 4 ⊡cm per r⊡n.
18. □nit□ ⊡e⊡gne□ to □rn ga□2 Iother⊡ga⊡e□	a. CO	130 ppm	1 hr minim⊡m ⊡ampûng time.

Pt. 63, Subpt. DDDDD, Table 13

for coi∉r or proce heater i in thi categor	For the fotowing pot≣tant□	□he emi⊡ion□ m □1 not eīcee□ the foïowing emi⊡ion îmit□ eīcept □ring perio⊡ of ⊡art⊡p an□ ch⊡town	□ îng thi □ pecifie □ amp îng □o îme or te ît r în □ ration
	□ HC□	1.⊡E⊡03 ⊞ per MM⊡t⊡ of heat inp⊡t	For M26_, Corect a minimom of 2 cm per rinofor M26, corect a mino imom of 240 iteroper ron.
	c. Merc□r□	□9EI06 III per MMIt□ of heat inpIt	For M29, collect a minimum of 3 cm per rincfor M300 or M300, collect a minimum cample a pecifie in the methocfor ISSM D6:84 collect a minimum of 3 ccm.
	□ Fifera ⊡e PM or □SM□	6. E 03 per MM t of heat inp t or 2.1E 04 per MM t of heat inp t.	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.

a fi to are contacting tlack tett to the monitrate compliance and to performance tett for this potitiant for at least 2 con-lectice card how that for emittions are at or provide this limit, to can keep the ling according to 63.515 if a month of the other provide of 63.515 are met. For all other potitiant, that to not contain a footnote tart to performance tett for this potition and for at least 2 concelling tear multiple with the limit of the other at or the limit is performance to the limit in order to the limit of the tetting.

Like telling. "Incorporate] = reference, the class that the common are at or between the percent of this limit in order to the limit of the class the common of the class the c

[87 FR 60857, Oct. 6, 2022]

TABLE 13 TO SUBPART DDDDD OF PART 63—ALTERNATIVE EMISSION LIMITS FOR NEW OR RECONSTRUCTED BOILERS AND PROCESS HEATERS THAT COMMENCED CON-STRUCTION OR RECONSTRUCTION AFTER DECEMBER 23, 2011, AND BEFORE APRIL 1, 2013

f o r oier or proce heater i in thi □categor	For the fo⊠wing po⊞tant⊡ 	he emi⊡on m t not e cee the forowing emi⊡on imit ecept oring perio of tart p an thatown	□ îng thi □ pecifie □ amp îng o îme or te î rîn □ ration
1. Initi in a IIII categorie Ieignei to IIIn IoiII fiel	a. HC□	0.022	For M26□, coïect a minim□m of 1 □cm per r□n□for M26 coïect a minim□m of 120 î⊡ ter□ per r□n.
	□ Merc□r□	8.6E ⊡ ^a	For M29, collect a minimum of 4 Comper Intfor M30 or M30, collect a minimum ample a pecifie in the metholofor CSIM D6 84 collect collect a minimum of 4 Collect minimum of 4
2. Pierie coaloiar e gne to in coaloit foi fe	a. Car⊡on mono⊡e ເCO⊡or CEMS⊡	130 ppm i olme on a r a correcte to 3 percent oligen, 31 a erage or 320 ppm i olme on a cent oligen,° 30 a roling a ferage	1 hr minim⊡m ⊡ampîng time.
	C Filtera De PM for CSMC	1.1E_03	Co≣ect a minim⊡m of 3 ⊡cm per r⊡n.
3. Stoker: eigne: to inn coa/ioii foiife:	a. CO lor CEMS	130 ppm lotter on a lotter lotter of a percent other of a lotter of a percent other of a lotter of a lotter is a lotter of a lotter cent other of a lotter of a lotter a lotter of a lotter of a lotter a lotter of a lotter o	1 hr minim⊡m ⊡ampîng time.
	C Fitera C PM for SMC	2.8E 02	Collect a minim m of 2 incm per r n.

Pt. 63, Subpt. DDDDD, Table 13

40 CFR Ch. I (7-1-23 Edition)

∬ ⊡ r ⊡oiêr or proce ⊡ heater i⊡ in thi⊡ ⊡⊂categor	For the fo⊡owing po⊞tant⊡ 	the emi⊡on m⊡t not e cee the fotowing emi⊡on ûmit, e cept ⊡ring perio⊡ of ⊡tartip an thatown .	□ ing thi□ pecifie□ amp ing o ime or te t r n □ ration
4. Finite en nitee gee to en coallo în forife	a. CO tor CEMS	130 ppm □ collme on a l □ a□ correcte to 3 percent o gen, 31 n a lerage or 1230 ppm □ collme on a □ a□ correcte to 3 per- cent o gen,° 30 a rolling a lerage 0	1 hr minim⊡m ⊡ampûng time.
	C Fitera C PM for CSMC	1.1E_03	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
5. Finitie: Ce: Chitawith an integrate heat echanger Edigned to Ern coa/Loin for fee	a. CO lor CEMS	140 ppm comme on a ro all corrected to 3 percent of gen, 310 a carage or d 50 ppm comme on a contracted to 3 per- cent of gen, ° 30 car rolling all carage	1 hr minim⊡m ⊡ampûng time.
	C Fitera C PM for SMC	1.1E_03	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
6. Stoker / ope grate/other efigne to trn wet foo matrifie	a. CO Tor CEMS	620 ppm lolme on a lo lol correcte to 3 percent oligen, 31 n alerage or 410 ppm lolme on a cent oligen,° 10 la roling alerage	1 hr minim⊡m ⊡ampûng time.
	C Filtera C PM for CSMC	3.0E_02	Co≣ect a minim⊡m of 2 ⊡cm per r⊡n.
Stoker Gpe grate/other e igne to mn kin in irie ioma fe	a. CO	460 ppm □ o □me on a □r ai correcte to 3 percent o □gen.	1 hr minim⊡m ⊡ampûng time.
	□ Fittera⊡ë PM tor □SM□	3.2E_01	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
8. Finite: Les Inits Les Ignes to Ern Ioma //Ios Lates Toits	a. CO for CEMS	230 ppm © collme on a r a correcte to 3 percent o gen, 31cn aterage or 310 ppm © collme on a control correcte to 3 per cent o gen, ° 30 a rolling aterage	1 hr minim⊡m ⊡ampûng time.
	C Filtera C PM for CSM	9.8E 03	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
9. Silpenion imeri e igne to im iomai/io ale oim	a. CO lor CEMS	2,400 ppm connected to 3 percent cent of gen, 3 cm a crage or (2,000 ppm connected to 3 percent of 3 percent	1 hr minim⊡m ⊡ampîng time.
	C Filtera C PM for CSM	5.1E_02 per MM_tt_ of heat inp_tt_or 16.5E_03 per MM_tt_ of heat inp_tt_	Co.≣ect a minim⊡m of 2 ⊡cm per r⊡n.
10. Dich Olen "Pie Liner Gignel to tim Iomatt Iottale Colt	a. CO lor CEMS	810 ppm i forme on a in all correcte to 3 percent origen, 31m alerage for 520 ppm i forme on a in alerage original cent origen, or 10 i a rolling alerage	1 hr minim⊡m ⊡ampîng time.
	□ Fitera⊡e PM tor tSM□	3.6E_02 per MM_t_ of heat inp_t_or .3.9E_05 per MM_t_ of heat inp_t_	Co.≣ect a minim⊡m of 2 ⊡cm per r⊡n.
11. Felcellnit eligne to	a. CO	910 ppm	1 hr minim⊡m ⊡amp1ng time.

Pt. 63, Subpt. DDDDD, Table 13

௴ ௦ஂ௺ ௦ஂௐ௭ஂ௭ஂ௺௭ஂ௺௭ஂௐஂஂ ௰ஂௗஂ௹ஂஂஂஂஂ	For the fo⊡owing po⊞tant⊡ · · ·	□he emi⊡on□ m⊡t not e□ cee□ the foïowing emi⊡on îmit□ e□cept □ring perio⊡ of ፲tart□p an□ h□t□own	□ing thi□ pecifie□ ampîng ⊡o≣me or tet r⊡n ⊡ration
	C Fifera PM for SMC	2.0E_02	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
12. Hiri III III and Angle III and Angle III and Angle III and Angle III angle IIII angle III an	a. CO Tor CEMS	1,500 ppm □ olime on a r a correcte to 3 per- cent oligen, 3 r a erage or 1900 ppm □ colime on a r a correcte to 3 percent oligen,° 30 a r olima aerage	1 hr minim⊡m ⊡ampûng time.
	□ Fitera⊡e PM for CSM□	2.6E 02	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
13. Inita leagned to ann îna în faet	a. HC□	1.2E ⊡03	For M26:: Corect a minimum of 2 cm per rin for M26, correct a minimum of 240 in teruper run.
	□ Merc□r□	4.9E ^{OD} ^a ^m per MM⊡to of heat inp⊡t.	For M29, collect a minimum of 4 Comper rincfor M30 or M30, collect a minimum collect a minimum of 4 Collect a minimum of 4
14nit_ reigne to _mn hea1 fre	a. CO lor CEMS	130 ppm income on a income on a income of a corrected to 3 percent on gen, 31 and a corrected to 3 percent on gen, 10 marching a corrected to 3 percent on gen, 10 marching a constant on gen.	1 hr minim⊡m ⊡ampîng time.
	C Fitera C PM for SMC	1.3E 03	Colect a minim⊡m of 3 ⊡cm per r⊡n.
15. Onito Cetigneo to Com Ight IIIIo ficeo	a. CO lor CEMS	130 a ppmorme on a acorrecte_to 3 percent oormeor 60 ppmorm me on aacor recte_to 3 percent ogen,° 1aock a erage	1 hr minim⊡m ⊡ampûng time.
	□ Fitera⊡e PM tor □SM□	1.1E ⊡03 ^a □ per MM to of heat inpotor 2.9E 05 □ per MM to of heat inpoto	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
16nit_ ce_gne_ tom im fce_that are non@onti_ nentanit_	a. CO	130 ppm i follme on a in all correcte to 3 percent oligen, 3 in a lerage all on tack tellior (91 ppm i follme on a in all correcte to 3 percent oligen, c 3 thoir rolling all or ane	1 hr minim⊡m <i>⊡</i> ampîng time.
	C Fifera PM for SMC	2.3E 02	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
1 onito colorn gao 2 otherogateo	a. CO	130 ppm Collection on a Cr Call correcte to 3 percent o gen.	1 hr minim⊡m ⊡ampûng time.
	□ HC□	1. E 03	For M26_, Collect a minimum of 2m per r_h_for M26, collect a minimum of 240 iii ter_per r_h.
	c. Merc⊡r⊡	□9E:06 III per MM⊡t⊡ of heat inp⊡t.	For M29, collect a minimum of 3 for M30_ or M30_, collect a minimum Lample a pecifiel in the metho_forS_M D6_B4_collect a minimum of 3

Pt. 63, Subpt. DDDDD, Table 14

40 CFR Ch. I (7-1-23 Edition)

II ⊡ r ⊡oiër or proce heater i⊡ in thi ccategor	For the fo⊠wing po⊞tant⊡ 	he emi⊡on m⊡ not e cee the fo™wing emi⊡on imit_ ecept ⊡ring perio⊡ of tartp an httown	□ ing thi □ pecifie □ amp îng □o □ me or te ा r n □ ration
	□ Fitera⊡e PM tor □SM□	6. E 03	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.

MMC of heat inp c
 MMC of heat inp c

[87 FR 60859, Oct. 6, 2022]

TABLE 14 TO SUBPART DDDDD OF PART 63-ALTERNATIVE EMISSION LIMITS FOR NEW OR RECONSTRUCTED BOILERS AND PROCESS HEATERS $^{\rm o}$

As stated in §63.7500, you may continue to comply with the following applicable emission limits until October 6, 2025:

fi ⊡or coi≅r or proce heater i⊡ in thi⊡ imcategor	For the fo⊡owing po⊞tant⊡	□he emiiion m⊡t not eīcee the folowing emiiion limit, eīcept ⊡ring tartīp an hīt īown	Or the emiiion mit not creee the following a ternatice otptimate imitice crept fring tartcp and hittown	□ing thi□ pecifie □amping □o⊡me or te⊡ ron □□□ ration
1nit_ in acat_ egoriee_igne_ to rnoii_ f_e_	a. HC⊡	2.2E 02	2.5E 02	For M26□, co⊡ect a minim⊡m of 1 □ccm per r⊡n⊡for M26 co⊡ ïect a minim⊡m of 120 îter⊡ per r⊡n.
	□ Merc□r□	8.0E⊡0⊡ª ⊞ per MM⊡t⊟ of heat inp⊡t.	8. E = 00 a m per MM to of _team o tp t or 1.1E = 05 a m per M h.	For M29, collect a miniti imitian of 4 to mer rontfor M30 or M300, collect a miniti imitian cample at the cifet in the methic ottor SIM D6/E84 collect a minitian of 4 to cm.
2. Init: eigne to Im coato in for f	a. Filtera 🗈 PM for SM	1.1E_03 □ per MM⊡t of heat inp⊡or 2.3E_05 □ per MM⊡t of heat inp⊡t	1.1E_03	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
3. Pilleri el coalición er l'eligne to inn coa//oil fottife.	a. Car⊡on mono⊡⊡e ICO⊡or CEMS⊡	130 ppm : colline on a colline correcte to 3:percent orgen, 3:cin alerage or 3:20 ppm : colline on a colline rectel to 3:percent orgen, 3:0:cin ing alerage	0.11	1 hr minim⊡m ⊡ampûng time.

init with heat inp t capacit of 10 mittion t per hor or greater

Pt. 63, Subpt. DDDDD, Table 14

□ nit with heat inp t capacit of 10 mi ton t per hor or greater

I or oier or proce heater i in thi □ categor	For the fo⊡owing po⊡tant□	he emi⊡on m⊡t not eccee the fo⊡owing emi⊡on imit_eccept ⊡ring tartp an htt ⊡own	Or the emi lon m t not ecceed the following all ternatice orpoticale imit_eccept tring tartip and thitown	□ ûng thi□ pecifie □ □ ampûng ⊡ me or te⊡t r⊡n □ □ ration
4. Stoker Jother Ce igne to Crn coal coll for if e	a. CO Tor CEMS	130 ppm : colline on a colline correcte to 3:percent orgen, 3:rim aleragefor :340 ppm : colline on a colline rectel to 3:percent orgen, 30:ra rotti ing alerage	0.12	1 hr minim⊡m ⊡ampûng time.
5. Filidie centit celigne to the coalloit for fe	a. CO lor CEMS	130 ppm i olime on a claim correcte to 3 percent oligen, 3 m a crage or 1230 ppm i olime on a claim corr rectel to 3 percent oligen, 30 maiorim ing a crage	0.11	1 hr minim⊡m ⊡ampûng time.
6. Filline en nit with an integrate heat enchanger en igner to min coall toth for free	a. CO tor CEMS	140 ppm i olime on a claric correcte to 3 percent oligen, 3 claric correcte on a claric correcte on a claric correcte oligen, 30 claric correct ing alerage	1.2E_01	1 hr minim⊡m ⊡ampûng time.
□ Stoker://⊡ope⊡ grate/ other: □eigne: to □:rn wet ioma: fie:	a. CO Tor CEMS	620 ppm : ollme on a road correcte to 3:percent oligen, 3:ron alerage or :390 ppm : ollme on a road correcte recte: to 3:percent oligen, 30:ra rotto ing alerage	5.8E::01 III per MM::1 of iteam oftpit or 6.8 III per M:: h::3 r:n afterage.	1 hr minim⊡m ⊡ampûng time.
	L Filtera Ce PM for SMC	3.0E ^{O2} Der MMoto of heat inpotor 12.6E ^{O5} per MMoto of heat inpoto	3.5E_02 m per MM_tt of _team o_tp_t or 4.2E_01 m per M_h or [2.E=05 m per MM_tt of _team o_tt p_t or 3.E=04 m per M_h	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
8. Stoker / ope grate/ other cigne to orn kin rie io ma fee	a. CO	460 ppm □ o	4.2E ⊡01	1 hr minim⊡m ⊡ampûng time.
	L Filtera Ce PM for	3.0E 02 per MM to of heat inp tor 4.0E 03 per MM to f heat inp to	3.5E 02 m per MM to of leam oltplt or 4.2E 01 m per M ho or 4.2E 03 m per MM to of leam olto plt or 5.6E 02 m per M ho	Co⊡cct a minim⊡m of 2 ⊡cm per r⊡n.
9. Filidie e e nit eigne to in io mai/ioliae com i	a. CO lor CEMS⊡	230 ppm : olme on a clarce or corrected to 3 percent oligen, 310 ppm : olime on a clarce or rected to 3 percent oligen, 30 a roll ing a crage	2.2E:01	1 hr minim⊡m ⊡ampûng time.

Pt. 63, Subpt. DDDDD, Table 14

40 CFR Ch. I (7-1-23 Edition)

□nit□ with heat inp⊡t capacit□ of 10 mi ⊡on □t□ per ho⊡r or greater□

If ⊡o⊡r ⊡oiër or proce heater i⊡ in thi⊡ ⊡⊂ategor⊡	For the fo⊡owing po⊡tantີ	□he emii⊡ion □ m⊡t not eEcee□ the fo⊡owing emi⊡ion îmit□, eEcept ⊡ring ⊡artEp an □ h⊡t□ Eown	Or the emiiion mit not creee the following all ternatice otputate imit crept tring tartcp and hotown	□ing thi□ pecifie□ ampling ⊡ome or tett rin □□ ration
	□ Filtera⊡e PM ⊡or ⊡SM⊡	9.8E ⊡03	1.2E [02]] per MM [1] of [team o [tp]t or 0.14]] per M [] h [] or [] .1E [04 a]] per MM [1] of [team o [1] p[] tor 1.2E [03 a]] per M [] h []	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
10. Sipenion innen eigne to inn io mai/iolae col i	a. CO lor CEMS	2,400 ppm :: o:me on a cr: a:: cori recte: to 3:percent o::	1.9 m per MM to of team otpt or 2 m per M h t3 m a tern age.	1 hr minim⊡m ⊡ampûhg time.
	□ Filtera ⊡e PM for □SM□	3.0E C2 m per MM to of heat inpCtor 16.5E C03 m per MM to of heat inpCto	3.1E ⁰² = per MM ¹ t ⁰ of [leam oftpft or 4.2E ⁰¹ = per M ¹ h ⁰ or 6.6E ⁰³ = per MM ¹ t ⁰ of [leam oft] p ¹ t or 9.1E ⁰² = per M ¹ h ⁰	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
11. Ditch Olen /Pie Trne: Cignel to Trn forma//io ale: coitt	a. CO tor CEMS	330 ppm : olime on a : a : a : correcte to 3:percent oligen, 3: a : a : a : a : a : a : a : a : a : a	3.5E::01 □ per MM::t: of ⊡eam o:Tp1 or 3.6 □ per M:: h::30 r:n a⊡erage.	1 hr minim⊡m ⊡amp ing time.
	□ Filtera⊡e PM ⊡or ⊡SM⊡	3.2E[03] [] per MM [1] of heat inp [1]or [3.9E[05] [] per MM [1] of heat inp [1]	4.3E_03	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
12. Felcennit e igne to orn io matricolate of	a. CO	910 ppm	1.1	1 hr minim⊡m ⊡ampûng time.
_	□ Filtera⊡e PM ⊡or ⊡SM⊡	2.0E ⁰ 2	3.0E_02 per MM_tt of _team o_tp_t or 2.8E_01 per M_h or _5.1E_05 per MM_tt of _team o_t p_t or 4.1E_04 per M_h	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
13. Hini IIIpeniion grate loier eiignei to IIm Iomaiiio alei loi	a. CO Tor CEMS	1,100 ppmor_me on aacor recte_ to 3;percent orgen, 3;r a er age_or :900 ppm or_me on a r acorrecte_ to 3 percent orgen, 30	1.4	1 hr minim⊡m ⊡ampûng time.
	□ Fitera⊡e PM tor □SM□	a rolling a reage 2.6E 02 per MM to of heat inp to r 4.4E 04 per MM to of heat inp to	3.3E ⁰² mer MM ¹ of leam olpt or 3.E ⁰¹ mer M h or 15.5E ⁰⁴ mer MM ¹ of leam olt pt or 6.2E ⁰³ mer M h	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.

Pt. 63, Subpt. DDDDD, Table 14

I ⊡⊡r ⊡oier or proce⊞ heater i⊟ in thi⊟ ⊡rcategor⊡	For the fo⊡owing po⊞tant⊟	□he emi ion mit not eccet the fo©wing emi ion mit eccet iring tart p an ht own	Or the emilion mit not creee the following all ternatice otptmale limit creet ming tartcp and hittown	□ ing thi □ pecifie □ amping rotime or tet r in □ □ ration
14. onito religned to	a. HC⊡	4.4E ⊡04	4.8E 04	For M26 :: Corect a minim m of 2 _ cm per r n for M26, cor lect a minim m of 240 liter per r n.
	□ Merc□r□	4.8E⊡0 =	5.3E:00 * :::::::::::::::::::::::::::::::::	For M29, collect a mining immore of 4 comper renefor M300 or M300, collect a mining immore and pecified in the methologication D6184 collect a minimore of 4 com.
15. □nit□ ceːigne□ to □rn hea⊡ i̇́⊡i□ f.e.	a. CO	130 ppm □ o me on a r ai correcte to 3 percent o gen, 3 r a erage.	0.13	1 hr minim⊡m ⊡ampûng time.
	□ Fitera ⊡e PM ⊡or □SM□	1.3E 02	1.5E_02 per MM_tt of _team o_tp_t or 1.8E_01 per M_h or _8.2E_05 a per MM_tt of _team o_tt p_t or 1.1E_03 a per M_h	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
16. onito ce igned to	a. CO	130 ppm □ o □me on a □r□ □a □ correcte to 3 percent o □gen.	0.13	1 hr minim⊡m ⊡ampûng time.
	□ Filtera⊡e PM for □SM□	1.1E□03 a □ per MM□t□ of heat inpt□or 12.9E□05 □ per MM□t□ of heat inpt□	1.2E:03 a □ per MM⊡ of ⊡eam oftpft or 1.6E:02 a □ per M□ h□or 3.2E:05 □ per MM⊡t of ⊡eam oftpft or 4.0E:04 □ per M□ h□	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡h.
1 eigne to fee_that are non@ontinenta 	a. CO	130 ppm 130 ppm 130 me on 131 correcte 153 percent orgen, 131 n a erage are on tack tert.	0.13	1 hr minim⊡m ⊡amp1îng time.
	□ Fifera⊡e PM or □SM□	2.3E 02 m per MM to of heat inp tor B.6E 04 m per MM to f heat inp to	2.5E 02	Co⊡ect a minim⊡m of 4 ⊡cm per r⊡n.
18. □nit□ □e igne □ to □rn ga □ 2 tother □ ga □e □	a. CO	130 ppm	0.16	1 hr minim⊡m ⊡ampûng time.
	□ HC□	1. E 03	2.9E 03	For M26□, Corect a minim⊡m of 2 ⊡cm per r⊡n⊡for M26, cor ect a minim⊡m of 240 îter: per r⊡n.
	c. Merc□r□	□9E:06 III per MM⊡t⊡ of heat inp⊡t.	1.4E:05	For M29, collect a minin imiliar of 3 litem per rinifor M300 or M30, collect a mini imiliar ample a litectified in the methi- olifor ISIM D6184 collect a minimiliar of 3 litem.

Pt. 63, Subpt. DDDDD, Table 15

40 CFR Ch. I (7-1-23 Edition)

init with heat inp t capacit of 10 mitton t per hor or greater

If ⊡o⊡r ⊡oiër or proce⊡ heater i⊡ in thi⊡ ⊡⊂categor⊡	For the fo⊡owing po⊡tant⊡	□he emiiion m⊡t not eCcee the fo⊡owing emiiion îmit eCcept ⊡ring tart p an htt Cown	Or the emilion mit not create the following all ternatice otput are imitice crept tring tartcp and that own	□ Ing thi□ tpecifie □ amp Ing Io Ime or te I r In □ □ □ ration
	□ Fitera⊡e PM for □SM□	6E03per MMt. of heat inp.tor 12.1E04per MMt. of heat inp.t.	1.2E ⁰² [] per MM ¹ .1 of []eam o ¹ .1p ¹ or 0E ⁰² [] per M h or 13.5E ⁰⁴ [] per MM ¹ .5 of Ieam o ¹ .1 p ¹ or 2.2E ⁰³ [] per M ¹ h	Co⊡ct a minim⊡m of 3 ⊡cm per r⊡n.

a fi ⊡ are con⊡cting tlack tell to cemonIrate compliance an ⊡ ⊡ r performance tell for thi poIIItant for at @aI 2 con ectice card how that ⊡ emiIIIon are at or performance time to the citing according to e63.515 if a to the other profilon 0 =63.515 are met. For a Tother poIIItant that to not contain a footnote IaII⊡or performance tell for thip poIII ant or at @aI 2 contective ceard mill how that ⊡or emiIIIon are at or performance to for thip III in the context of the time of the context of the time of the context of the cont □kp te ting.

ncorporate C reference, ce C63.14.

ⁱ mcorporatel □ reference, tee 163.14. ^c If too affecte to roe is a new or recontractel affecte to roe that commence contraction or recontraction after J = 4, 2010, an 1 efore origin 1, 2013, too mail complex with the emilion limit in alle 11, 12, or 13 to this top at thit Janar 31, 2016, on a later Janar 31, 2016, the fore October 6, 2025 to mail complex with the emilion limit in alle 11, 12, or 13 to this this alle 14. On an after October 6, 2025, to mail complex with the emilion limit in alle 11, 12, or 13 to this the alle 14. On an after October 6, 2025, to mail complex with the carton monofile emilion limit ing carton to de a a fillent correction in place of otigen a fibric 10 after 10 this to the otigen of the fibric 10 after 0, appending and the fibric 10 and 10 after 0, appending and the fibric 10 after 0, appending and the fibric 10 and 10 after 0, appending and the fibric 10 after 0, appending and the fibric 10 after 0, appending and the fibric 10 and 10 after 0, appending and 10 after 0, appending after 0, appending and the fibric 10 after 0, appending and 10 after 0, appending and 10 after 0, appending and 10 after 0, appending a fibric 0, and 0 after 0, appending a fibric 0, and 0 after 0, appending a fibric 0, or remote 10 from, the emilion 10 to 10 on a fibric 10 and 10 after 0 after 0 and 10 after 0 after 0 and 10 after 0 and 10 after 0 and 10 after 0 after 0 and 10 after 0 after 0 and 10 after 0 after 0 after 0 after

[87 FR 60860, Oct. 6, 2022]

TABLE 15 TO SUBPART DDDDD OF PART 63—ALTERNATIVE EMISSION LIMITS FOR EXISTING BOILERS AND PROCESS HEATERS D

As stated in §63.7500, you may continue to comply with following emission limits until October 6, 2025:

□nit□ with heat in	p⊡t capacit] of	10 mi≣on ⊡t⊡	per ho r or greater
			/

I ⊡o⊡r ⊡oi≣r or proce heater i⊡ in thi⊡ ⊡⊏categor⊡	For the fo⊡owing po⊡tant⊡	□he emi⊡ion □ m⊡t not eEcee□ the fo⊡owing emi⊡ion îmit□, eEcept ⊡ring ⊡art⊡p an □htEcown	□he emiiion□m⊡t not eīcee□ the fo⊠wing a⊡ ternatīce oīpiīna eo imit_ eīcept ⊡ring tartīp an□ thītīown	□ ing thi□ tpecifie □ iamp1ing io ime or tet rin □ ration
1. init: in a cat egorie: e gree to an coil fee	a. HC	2.2E 02 □ per MM t of heat inp t.	2.5E 02	For M26□, Co⊡ect a minim⊡m of 1 □ccm per r⊡n⊡for M26, co⊡ iect a minim⊡m of 120 ifter⊡ per r⊡n.
	□ Merc□r□	5.□E 06	6.4E 06	For M29, collect a min imm of 3 icm per rinfor M300 or M300, collect a min imm cample a pecifie in the meth offor ISIM D6184 collect a minimm of 3 icm.
2. Init: eign to Im coa/Ioin fo Iife	a. Filfera 🖻 PM for ISM	4.0E 02 per MM to of heat inp tor 5.3E 05 per MM to f heat inp to	4.2E_02 per MM_tt of _team o_tp_t or 4.9E_01 per M_h or 15.6E_05 per MM_tt_of _team o_tt p_t or 6.5E_04 per M_h	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.

Pt. 63, Subpt. DDDDD, Table 15

I ⊡o_r ⊡oier or proce heater i⊡ in thi⊡ ⊡categor⊡	For the fo⊡owing po⊞tantີ	□he emi⊡on□m⊡t not eEcee□ the fo®wing emi⊡on îmit□ eEcept ⊡ring ⊡art⊡p an□ □h∃t≣own	he emi ion m t not eccet the following a ternatice otptaac imit, eccept tring tartp an htown	□ Ing thi□ □pecifie□ □ amp1îng □ □ Ime or teît rîn □ □ □ ration
3. Pilerie coaloi er eigne to in coaloi foi fe	a. CO Tor CEMS	130 ppm comme on a comme comme on 3 concerted to 3	0.11 □ per MM⊡to of _team oitpit or 1.4 □ per M0 hi310h allerage.	1 hr minim⊡m ⊡ampûng time.
4. Stoker Tother e gne to mn coal to the form of the	a. CO Tor CEMS	160 ppm to Ime on a to Ime on a to Ime or orgen, Ime on a to Ime on a to Ime	0.14	1 hr minim⊡m ⊡ampûng time.
5. Fuide e nit eigne to m coa/ioin for fe	a. CO lor CEMS	130 ppm Dollme on a Dollme on a Dollme on a Dollme or a Dollme a Dollme on a Dollme on a Dollme on a Dollme on a Dollme on a Dollme on a Dollme a Doll	0.12 □ per MM⊡t⊡ of ☐eam o⊡p⊡t or 1.4 □ per M⊡ h⊡3⊡n a⊡erage.	1 hr minim⊡m ⊡ampûng time.
6. Filling control nite with an integrate heat echanger co igne to imn coall to imform free	a. CO lor CEMS	140 ppm in the image of the ima	1.3E:01 □ per MM:0 of iteam oftpit or 1.5 □ per M0 h:30 r0h allerage.	1 hr minim⊡m ⊡amp≣ng time.
Stoker/Tope_grate/ other_Ce_gne_to The wet formato fiel	a. CO lor CEMS	1,500 ppm i olime on a in all corr recte to 3ipercent oligen, 3iin aler age or ii20 ppm i olime on a in all correcte to 3 percent oligen,° 30 ial rolling alerage	1.4 □ per MM to of □team oopt or 1 □ □ per M h to Tron aler age.	1 hr minim⊡m ⊡ampûng time.
	□ Filera⊡e PM lor □SM□	3. E 02	4.3E_02 per MM_tt of _team o_tp_t or 5.2E_01 per M_h or _12.8E_04 per MM_tt_of team o_tt p_t or 3.4E_04 per M_h	Co⊡ect a minim⊡m of 2 ⊡cm per r⊡n.
8. Stoker / lope grate/ other eigne to in kinirie io mail fee	a. CO	460 ppm □	4.2E ⊡01	1 hr minim⊡m ⊡ampûng time.
	E Filera E PM for SM	3.2E_01	3E_01per MM_tt of _team o_tp_t or 4.5per M_hor /4.6E_03per MM_tt of _team o_tt p_t or 5.6E_02per M_h	Co⊡ect a minim⊡m of 1 ⊡cm per r⊡n.

Pt. 63, Subpt. DDDDD, Table 15

40 CFR Ch. I (7-1-23 Edition)

f ⊡o□r ⊡oiēr or proce heater i⊡ in thi⊡ ⊡categor⊡	For the fo⊡owing po⊡tant⊡	□he emi⊡on□m⊡t not eEcee□ the fo⊡owing emi⊡on îmit□ eEcept ⊡ring ⊡artEp an□ Ih=tEown	□he emi iion □ m □t not e cee □ the fo @wing a □ ternatice o cpt □ a e □ imit ecept oring tart p an □ h t cown	□ ing thi□ tpecifie □ □ amp îng □ o □ me or te ît r □n □ □ □ ration
9. Filice conto constant mail/iolace coll	a. CO lor CEMS	4:0 ppm : Dilme on a r al correcte to 3:percent orgen, 3:r n alerageor 3:10 ppm : Dilme on a r al corr rectel to 3:percent orgen, al al arotti ing alerage	4.6E ©1 □ per MM⊡t of ⊡eam o⊡tp⊡t or 5.2 □ per M □ h⊡3 r ⊡n a⊡erage.	1 hr minim⊡m ⊡ampûng time.
	□ Filtera⊡e PM tor □SM□	1.1E_01	1.4E_01 □ per MM⊡t of leam oltplt or 1.6 □ per M hor 0.5E_03 □ per MM⊡t of leam olt plt of 1.E_02 □ per M ho	Co⊡ect a minim⊡m of 1 ⊡cm per r⊡n.
10. Sipenion inner Deigne to inn io mai/ioliate coin	a. CO lor CEMS	2,400 ppm i oime on a r-laicor rectel to 3ipercent oigen, 3inn aer agelor (2,000 ppm i oime on a r ai correctel to 3 percent oigen,° 10 ta roimg aterage	1.9	1 hr minim⊡m ⊡ampîng time.
	□ Fitera □e PM for □SM□	5.1E ⁰² mer MM ^{to} of heat inp ^{to} 6.5E ⁰³ mer MM ^{to} of heat inp ^{to}	5.2E ⁰² m per MM ¹ t ⁰ of fleam of pft or 1E ⁰¹ per M ¹ h ⁰ or f6.6E ⁰³ m per MM ¹ t ⁰ of fleam of t ⁰ pft or 9.1E ⁰² m per M ¹ h ⁰	Corect a minimor of 2
11. Ditch Olen //Pie imer Defignen to im fomat//io ale toin	a. CO lor CEMS	O ppm Collectore on a collectore of the collectore of th	8.4E 01 m per MM t of deam of tp t or 8.4 m per M h 13 r n afterage.	1 hr minim⊡m ⊡ampîng time.
	□ Filera⊡e PM lor □SM□	2.8E 01 m per MM to of heat inp tor 12.0E 03 m per MM to f heat inp to	3.9E_01 per MM_tt of _team o_tp_t or 3.9 per M_hor 12.8E_03 per MM_tt_of _team o_tt p_t or 2.8E_02 per M_h0	Co⊡ect a minim⊡m of 1 ⊡cm per r⊡n.
12. Feecellnit e ignet to orn fo mat/follate coin	a. CO	1,100 ppm ome on a ome on a ome	2.4	1 hr minim⊡m ⊡ampûng time.
	L Filtera Ce PM Cor CSMC	2.0E_02	5.5E 02 m per MM to of leam oltplt or 2.8E 01 m per M h or d.6E 02 m per MM to of leam oltplt or 8.1E 02 m per M h	Cor≣ct a minim⊡m of 2 ⊡cm per r⊡n.

□ nit with heat inp t capacit of 10 mi ton t per hor or greater

Pt. 63, Subpt. DDDDD, Table 15

f or oier or proce heater i⊡ in thi ⊡categor	For the foïowing po⊞tant□	□he emi⊡ion□ m⊡t not e⊡cee□ the fo⊡owing emi⊡ion îimit□ e⊡cept ⊡ring ttart⊡p an□ thttown	he emi⊡on m t not eccee the fo⊡wing a ternatie o tpt tae imit, eccept oring tartop an htown	□ ing thi □ pecifie □ □ amp îng □ □ □ me or te □t r □ n □ □ □ ration
13. Haria appendion grate anita edignet to an doma affice are to in	a. CO tor CEMS	3,500 ppm on a read of a second rectain the second	3.5 mper MM_to of team otpt or 39 m per M_h_3rth alera age.	1 hr minim⊡m ⊡amp1ng time.
	□ Filtera⊡e PM tor □SM□	4.4E ⊡01	5.5E_01	Co⊡ect a minim⊡m of 1 ⊡cm per r⊡n.
14. Init: Deligne to	a. HC	1.1E ⊡03	1.4E 03	For M26 , co ect a minim of 2 ccm per ron for M26, co ect a minim of 240 fter per ron.
	□ Merc⊡r⊡	2.0E ⊡06 III per MM⊡t⊡ of heat inp⊡t.	2.5E 06	For M29, collect a min imim of 3 icm per rinifor M300 or M300 collect a min imim lample a ipecifie in the meth of, for ISIM D6IB4 collect a minimom of 2 icm.
15. ⊡nit⊡ e⊡gne⊟ to ⊡rn hea⊡ î⊞i⊡ f⊡e	a. CO	130 ppm	0.13	1 hr minim⊡m ⊡ampûng time.
	L Filtera E PM for SMC	6.2E_02 _ per MM_t of heat inp_tor 2.0E_04 _ per MM_t of heat inp_t	□.5E_02	Co⊡ct a minim⊡m of 1 ⊡cm per r⊡n.
16. onito ce igneo to	a. CO	130 ppm □	0.13	1 hr minim⊡m ⊡ampûng time.
	L Filtera E PM for SMC	□9E_03	9.6E_03	Co⊡ect a minim⊡m of 3 ⊡cm per r⊡n.
1 Onit Ceigne to Tr III feethat are non continenta Init	a. CO	130 ppm □ co≣me on a □r□ a □ correcte to 3 percent o gen, 3 □n a erage a e on tack tet.	0.13	1 hr minim⊡m ⊡ampûng time.
	□ Fitera⊡e PM tor ⊡SM⊡	2. E 01 per MM_t of heat inp_tor 18.6E 04 per MM_t of heat inp_t	3.3E_01 per MM_tt of _team o_tp_t or 3.8 per M_hor 1E_03 per MM_tt_of _team o_tt p_t or 1.2E_02 per M_h	Cor≊ct a minim⊡m of 2 ⊡cm per r⊡n.
18. onito congression of the constant of the c	a. CO	130 ppm	0.16	1 hr minim⊡m ⊡ampûng time.

§63.7680

40 CFR Ch. I (7-1-23 Edition)

init with heat inp t capacit of 10 mitton t per hor or greater

I ⊡o⊡r ⊡oi≣r or proce heater i⊡ in thi⊡ ⊡categor⊡	For the fo⊡owing po⊞tant⊡	he emi⊡on m⊡t not eccee the fo⊡owing emi⊡on imit_ eccept ⊡ring tartp an httown	he emi∷ion m t not ercee the for or most a m ternatice or potrae imit_ercept	□ ing thi□ □pecifie□ □ amp ing □ o ime or te it rin □ □ □ ration
	□ HC□	1. E O3 per MM t □ of heat inp t.	2.9E 03	For M26 □, corect a minim □m of 2 □ cm per r □n □for M26, cor ect a minim □m of 240 îter □ per r □n.
	c. Merc⊡r⊡	□9E 06	1.4E 05	For M29, collect a minum of 3 icm per runfor M30 or M30, collect a minum imm cample a lepecifie in the methodoffor SSM D6:84 oclect a minimum of 2 icm.
	□ Filtera □ e PM for □SM□	6. E 03 m per MM t of heat inp t or 2.1E 04 m per MM t of heat inp t	1.2E_02 per MM_t ofteam oftpt or 0E_02 per M_ h or _3.5E_04 per MM_t ofteam oft pt or 2.2E_03 per M_ h	Co⊡ect a minim⊡m of three ⊡cm per r⊡n.

aff to are contacting thack tetta to temontare compliance and to performance tetta for this pointant for at lead 2 contacting the tetta of tetta of the tetta of the tetta of tetta of the tetta of tetta of the tetta of tetta of tetta of the tetta of tetta te ting.

te ting. Incorporate reference, be 63.14. on owner or operator mail etermine compliance with the carlon monoile emilion limit ling carlon logice as a lient correction in place of oligen as learning in 63.1525 and EPS Methol 19 Flactor in 40 CFR part 60, appending and EPS Methol 19 estation in 40 CFR part 60, appending and the into account that the 3 oligen correction is to be one on a re-age for the first period of the init, and mit allo take into account that the 3 oligen correction is to be one on a re-all the methologic mit account for an CO₂ teng area to, or remote from, the emillion gain the area area in of fine to a correction in the list of the init and take into account that the 3 oligen correction is the area area of fine to a correction in the init and take into account the init of the in

[87 FR 60863, Oct. 6, 2022]

Subpart EEEEE—National Emission Standards for Hazardous Air Pollutants for Iron and Steel **Foundries**

SOURCE: 69 FR 21923, Apr. 22, 2004, unless otherwise noted.

WHAT THIS SUBPART COVERS

§63.7680 What is the purpose of this subpart?

This subpart establishes national emission standards for hazardous air pollutants (NESHAP) for iron and steel foundries. This subpart also establishes requirements to demonstrate initial and continuous compliance with the emissions limitations, work practice standards, and operation and maintenance requirements in this subpart.

§63.7681 Am I subject to this subpart?

You are subject to this subpart if you own or operate an iron and steel foundry that is (or is part of) a major source of hazardous air pollutant (HAP) emissions. Your iron and steel foundry is a major source of HAP for purposes of this subpart if it emits or has the potential to emit any single HAP at a rate of 10 tons or more per year or any combination of HAP at a rate of 25 tons or more per year or if it is located at a facility that emits or has the potential to emit any single HAP at a rate of 10 tons or more per year or any combination of HAP at a rate of 25 tons or more per year as defined in §63.2.

[69 FR 21923, Apr. 22, 2004, as amended at 73 FR 7218, Feb. 7, 2008]

Appendix Q

AUTHENTICATED U.S. GOVERNMENT INFORMATION GPO

Environmental Protection Agency

§60.4230

TABLE 8 TO SUBPART IIII OF PART 60—APPLICABILITY OF GENERAL PROVISIONS TO SUBPART IIII

□□ tate□ in □60.4218, □o□ m□t comp□ with the fo⊡owing apptica⊡e Genera⊡Protition ⊡

Genera⊡Pro⊡⊡on⊡ citation	Sect of citation	□ppîe□ to ⊡□part	E□panation
□60.1	Genera⊡appîca⊡ît⊡ of the Genera⊡Pro⊡⊡ ⊡on⊡	Ye□	
□60.2	Definition	Ye	□□itiona term □ efine in □60.4219.
□60.3	□nit□an□a□re iation□	Ye□	
□60.4	□	Ye□	
□60.5	Determination of con⊡tr⊡ction or mo⊡ifica⊡ tion.	Ye□	
□60.6	Re⊡ew of pan⊡	Ye□	
□60.□	□otification an□ Recor⊡keeping	Ye□	E cept that 60.0 on applie a pecifie in 60.4214 a
□60.8	Performance te to	Ye□	E cept that 60.8 on appie to ta tionar C CCE with a pacement of ≥30 iter per c in er an engine that are not certifie
□60.9	□ aia iit of information	Ye□	
□60.10	State □_thorit□	Ye□	
□60.11	Comptiance with tantart ant maintet nance rettirement	□o	Re⊡irement⊡are ⊡pecifie⊡ in ⊡⊃part
□60.12	Circ⊡m⊡ention	Ye□	
□60.13	Monitoring re irrement	Ye□	E cept that 060.13 on □ app1e to ta tionar C 0 CE with a 0 pacement of ≥30 fiter per c 0 her.
□60.14	Mo⊑ification	Ye□	
□60.15	Recon⊡tr⊡ction	Ye□	
□60.16	Priorit□ i⊞t	Ye□	
□60.1□	Incorporation	Ye□	
□60.18	Genera contro ce ice re irement	□o.	
□60.19	Genera⊡notification an⊡ reporting re⊡ire⊡ ment⊡	Ye□	

Subpart JJJJ—Standards of Performance for Stationary Spark Ignition Internal Combustion Engines

SOURCE: 73 FR 3591, Jan. 18, 2008, unless otherwise noted.

WHAT THIS SUBPART COVERS

§60.4230 Am I subject to this subpart?

(a) The provisions of this subpart are applicable to manufacturers, owners, and operators of stationary spark ignition (SI) internal combustion engines (ICE) as specified in paragraphs (a)(1) through (6) of this section. For the purposes of this subpart, the date that construction commences is the date the engine is ordered by the owner or operator.

(1) Manufacturers of stationary SI ICE with a maximum engine power less than or equal to 19 kilowatt (KW) (25 horsepower (HP)) that are manufactured on or after July 1, 2008.

(2) Manufacturers of stationary SI ICE with a maximum engine power greater than 19 KW (25 HP) that are gasoline fueled or that are rich burn engines fueled by liquefied petroleum gas (LPG), where the date of manufacture is:

(i) On or after July 1, 2008; or

(ii) On or after January 1, 2009, for emergency engines.

(3) Manufacturers of stationary SI ICE with a maximum engine power greater than 19 KW (25 HP) that are not gasoline fueled and are not rich burn engines fueled by LPG, where the manufacturer participates in the voluntary manufacturer certification program described in this subpart and where the date of manufacture is:

(i) On or after July 1, 2007, for engines with a maximum engine power greater than or equal to 500 HP (except lean burn engines with a maximum engine power greater than or equal to 500 HP and less than 1,350 HP);

§60.4231

(ii) On or after January 1, 2008, for lean burn engines with a maximum engine power greater than or equal to 500 HP and less than 1,350 HP;

(iii) On or after July 1, 2008, for engines with a maximum engine power less than 500 HP; or

(iv) On or after January 1, 2009, for emergency engines.

(4) Owners and operators of stationary SI ICE that commence construction after June 12, 2006, where the stationary SI ICE are manufactured:

(i) On or after July 1, 2007, for engines with a maximum engine power greater than or equal to 500 HP (except lean burn engines with a maximum engine power greater than or equal to 500 HP and less than 1,350 HP);

(ii) on or after January 1, 2008, for lean burn engines with a maximum engine power greater than or equal to 500 HP and less than 1,350 HP;

(iii) on or after July 1, 2008, for engines with a maximum engine power less than 500 HP; or

(iv) on or after January 1, 2009, for emergency engines with a maximum engine power greater than 19 KW (25 HP).

(5) Owners and operators of stationary SI ICE that are modified or reconstructed after June 12, 2006, and any person that modifies or reconstructs any stationary SI ICE after June 12, 2006.

(6) The provisions of §60.4236 of this subpart are applicable to all owners and operators of stationary SI ICE that commence construction after June 12, 2006.

(b) The provisions of this subpart are not applicable to stationary SI ICE being tested at an engine test cell/ stand.

(c) If you are an owner or operator of an area source subject to this subpart, you are exempt from the obligation to obtain a permit under 40 CFR part 70 or 40 CFR part 71, provided you are not required to obtain a permit under 40 CFR 70.3(a) or 40 CFR 71.3(a) for a reason

40 CFR Ch. I (7–1–23 Edition)

other than your status as an area source under this subpart. Notwithstanding the previous sentence, you must continue to comply with the provisions of this subpart as applicable.

(d) For the purposes of this subpart, stationary SI ICE using alcohol-based fuels are considered gasoline engines.

(e) Stationary SI ICE may be eligible for exemption from the requirements of this subpart as described in 40 CFR part 1068, subpart C (or the exemptions described in 40 CFR parts 1048 and 1054, for engines that would need to be certified to standards in those parts), except that owners and operators, as well as manufacturers, may be eligible to request an exemption for national security.

(f) Owners and operators of facilities with internal combustion engines that are acting as temporary replacement units and that are located at a stationary source for less than 1 year and that have been properly certified as meeting the standards that would be applicable to such engine under the appropriate nonroad engine provisions, are not required to meet any other provisions under this subpart with regard to such engines.

[73 FR 3591, Jan. 18, 2008, as amended at 76 FR 37972, June 28, 2011; 86 FR 34360, June 29, 2021]

Emission Standards for Manufacturers

§ 60.4231 What emission standards must I meet if I am a manufacturer of stationary SI internal combustion engines or equipment containing such engines?

(a) Stationary SI internal combustion engine manufacturers must certify their stationary SI ICE with a maximum engine power less than or equal to 19 KW (25 HP) manufactured on or after July 1, 2008 to the certification emission standards and other requirements for new nonroad SI engines in 40 CFR part 1054, as follows:

∄engine ⊡⊡pacement i⊡	an⊡ man⊡fact⊡ring ⊡ate⊡ are	the engine m⊡t meet the for⊠wing non:han:he⊡ emi⊡on ⊡tan⊡ar⊡ i⊡entifie⊡ in 40 CFR part 1054 an⊡ re™te⊡ re⊡irement⊡
1 0	J□□ 1, 2008 to Decem⊡er 31, 2011 Jan⊡ar□ 1, 2012 or later J□□ 1, 2008 to Decem⊡er 31, 2010 Jan⊡ar□ 1, 2011 or later	Pha⊡e 2. Pha⊡e 3. Pha⊡e 2. Pha⊡e 3.

(b) Stationary SI internal combustion engine manufacturers must certify their stationary SI ICE with a maximum engine power greater than 19 KW (25 HP) (except emergency stationary ICE with a maximum engine power greater than 25 HP and less than 130 HP) that use gasoline and that are manufactured on or after the applicable date in §60.4230(a)(2), or manufactured on or after the applicable date in §60.4230(a)(4) for emergency stationary ICE with a maximum engine power greater than or equal to 130 HP, to the certification emission standards and other requirements for new nonroad SI engines in 40 CFR part 1048. Stationary SI internal combustion engine manufacturers must certify their emergency stationary SI ICE with a maximum engine power greater than 25 HP and less than 130 HP that use gasoline and that are manufactured on or after the applicable date in (60.4230(a)(4)) to the Phase 1 emission standards in 40 CFR part 1054, appendix I, applicable to class II engines, and other requirements for new nonroad SI engines in 40 CFR part 1054. Stationary SI internal combustion engine manufacturers may certify their stationary SI ICE with a maximum engine power less than or equal to 30 KW (40 HP) with a total displacement less than or equal to 1,000 cubic centimeters (cc) that use gasoline to the certification emission standards and other requirements as appropriate for new nonroad SI engines in 40 CFR part 1054.

(c) Stationary SI internal combustion engine manufacturers must certify their stationary SI ICE with a maximum engine power greater than 19 KW (25 HP) (except emergency stationary ICE with a maximum engine power greater than 25 HP and less than 130 HP) that are rich burn engines that use LPG and that are manufactured on or after the applicable date in §60.4230(a)(2), or manufactured on or applicable date after the in §60.4230(a)(4) for emergency stationary ICE with a maximum engine power greater than or equal to 130 HP, to the certification emission standards and other requirements for new nonroad SI engines in 40 CFR part 1048. Stationary SI internal combustion engine manufacturers must certify their emergency §60.4231

stationary SI ICE greater than 25 HP and less than 130 HP that are rich burn engines that use LPG and that are manufactured on or after the applicable date in (60.4230(a)(4)) to the Phase 1 emission standards in 40 CFR part 1054, appendix I, applicable to class II engines, and other requirements for new nonroad SI engines in 40 CFR part 1054. Stationary SI internal combustion engine manufacturers may certify their stationary SI ICE with a maximum engine power less than or equal to 30 KW (40 HP) with a total displacement less than or equal to 1,000 cc that are rich burn engines that use LPG to the certification emission standards and other requirements as appropriate for new nonroad SI engines in 40 CFR part 1054.

(d) Stationary SI internal combustion engine manufacturers who choose to certify their stationary SI ICE with a maximum engine power greater than 19 KW (25 HP) and less than 75 KW (100 HP) (except gasoline and rich burn engines that use LPG and emergency stationary ICE with a maximum engine power greater than 25 HP and less than 130 HP) under the voluntary manufacturer certification program described in this subpart must certify those engines to the certification emission standards for new nonroad SI engines in 40 CFR part 1048. Stationary SI internal combustion engine manufacturers who choose to certify their emergency stationary SI ICE greater than 25 HP and less than 130 HP (except gasoline and rich burn engines that use LPG), must certify those engines to the Phase 1 emission standards in 40 CFR part 1054, appendix I, applicable to class II engines, for new nonroad SI engines in 40 CFR part 1054. Stationary SI internal combustion engine manufacturers may certify their stationary SI ICE with a maximum engine power less than or equal to 30 KW (40 HP) with a total displacement less than or equal to 1,000 cc (except gasoline and rich burn engines that use LPG) to the certification emission standards and other requirements as appropriate for new nonroad SI engines in 40 CFR part 1054. For stationary SI ICE with a maximum engine power greater than 19 KW (25 HP) and less than 75 KW (100 HP) (except gasoline and rich burn engines

or@on LcPcH6H6L3 with DcSctLER

that use LPG and emergency stationary ICE with a maximum engine power greater than 25 HP and less than 130 HP) manufactured prior to January 1, 2011, manufacturers may choose to certify these engines to the standards in Table 1 to this subpart applicable to engines with a maximum engine power greater than or equal to 100 HP and less than 500 HP.

(e) Stationary SI internal combustion engine manufacturers who choose to certify their stationary SI ICE with a maximum engine power greater than or equal to 75 KW (100 HP) (except gasoline and rich burn engines that use LPG) under the voluntary manufacturer certification program described in this subpart must certify those engines to the emission standards in Table 1 to this subpart. Stationary SI internal combustion engine manufacturers may certify their stationary SI ICE with a maximum engine power greater than or equal to 75 KW (100 HP)that are lean burn engines that use LPG to the certification emission standards for new nonroad SI engines in 40 CFR part 1048. For stationary SI ICE with a maximum engine power greater than or equal to 100 HP (75 KW) and less than 500 HP (373 KW) manufactured prior to January 1, 2011, and for stationary SI ICE with a maximum engine power greater than or equal to 500 HP (373 KW) manufactured prior to July 1, 2010, manufacturers may choose to certify these engines to the certification emission standards for new nonroad SI engines in 40 CFR part 1048 applicable to engines that are not severe duty engines.

(f) Manufacturers of equipment containing stationary SI internal combustion engines meeting the provisions of 40 CFR part 1054 must meet the provisions of 40 CFR part 1060, to the extent they apply to equipment manufacturers.

(g) Notwithstanding the requirements in paragraphs (a) through (c) of this section, stationary SI internal combustion engine manufacturers are not required to certify reconstructed engines; however manufacturers may elect to do so. The reconstructed engine must be certified to the emission standards specified in paragraphs (a) through (e) of this section that are ap40 CFR Ch. I (7–1–23 Edition)

plicable to the model year, maximum engine power and displacement of the reconstructed stationary SI ICE.

[73 FR 3591, Jan. 18, 2008, as amended at 73 FR 59175, Oct. 8, 2008; 76 FR 37973, June 28, 2011; 78 FR 6697, Jan. 30, 2013; 86 FR 34360, June 29, 2021]

§60.4232 How long must my engines meet the emission standards if I am a manufacturer of stationary SI internal combustion engines?

Engines manufactured by stationary SI internal combustion engine manufacturers must meet the emission standards as required in §60.4231 during the certified emissions life of the engines.

EMISSION STANDARDS FOR OWNERS AND OPERATORS

§60.4233 What emission standards must I meet if I am an owner or operator of a stationary SI internal combustion engine?

(a) Owners and operators of stationary SI ICE with a maximum engine power less than or equal to 19 KW (25 HP) manufactured on or after July 1, 2008, must comply with the emission standards in §60.4231(a) for their stationary SI ICE.

(b) Owners and operators of stationary SI ICE with a maximum engine power greater than 19 KW (25 HP) manufactured on or after the applicable date in 60.4230(a)(4) that use gasoline must comply with the emission standards in 60.4231(b) for their stationary SI ICE.

(c) Owners and operators of stationary SI ICE with a maximum engine power greater than 19 KW (25 HP) manufactured on or after the applicable date in $\S60.4230(a)(4)$ that are rich burn engines that use LPG must comply with the emission standards in $\S60.4231(c)$ for their stationary SI ICE.

(d) Owners and operators of stationary SI ICE with a maximum engine power greater than 19 KW (25 HP) and less than 75 KW (100 HP) (except gasoline and rich burn engines that use LPG) must comply with the emission standards for field testing in 40 CFR 1048.101(c) for their non-emergency stationary SI ICE and with the emission standards in Table 1 to this subpart for their emergency stationary SI ICE.

Owners and operators of stationary SI ICE with a maximum engine power greater than 19 KW (25 HP) and less than 75 KW (100 HP) manufactured prior to January 1, 2011, that were certified to the standards in Table 1 to this subpart applicable to engines with a maximum engine power greater than or equal to 100 HP and less than 500 HP, may optionally choose to meet those standards.

(e) Owners and operators of stationary SI ICE with a maximum engine power greater than or equal to 75 KW (100 HP) (except gasoline and rich burn engines that use LPG) must comply with the emission standards in Table 1 to this subpart for their stationary SI ICE. For owners and operators of stationary SI ICE with a maximum engine power greater than or equal to 100 HP (except gasoline and rich burn engines that use LPG) manufactured prior to January 1, 2011 that were certified to the certification emission standards in 40 CFR part 1048 applicable to engines that are not severe duty engines, if such stationary SI ICE was certified to a carbon monoxide (CO) standard above the standard in Table 1 to this subpart, then the owners and operators may meet the CO certification (not field testing) standard for which the engine was certified.

(f) Owners and operators of any modified or reconstructed stationary SI ICE subject to this subpart must meet the requirements as specified in paragraphs (f)(1) through (5) of this section.

(1) Owners and operators of stationary SI ICE with a maximum engine power less than or equal to 19 KW (25 HP), that are modified or reconstructed after June 12, 2006, must comply with emission standards in §60.4231(a) for their stationary SI ICE. Engines with a date of manufacture prior to July 1, 2008 must comply with the emission standards specified in §60.4231(a) applicable to engines manufactured on July 1, 2008.

(2) Owners and operators of stationary SI ICE with a maximum engine power greater than 19 KW (25 HP) that are gasoline engines and are modified or reconstructed after June 12, 2006, must comply with the emission standards in §60.4231(b) for their stationary SI ICE. Engines with a date of manufacture prior to July 1, 2008 (or January 1, 2009 for emergency engines) must comply with the emission standards specified in §60.4231(b) applicable to engines manufactured on July 1, 2008 (or January 1, 2009 for emergency engines).

(3) Owners and operators of stationary SI ICE with a maximum engine power greater than 19 KW (25 HP) that are rich burn engines that use LPG, that are modified or reconstructed after June 12, 2006, must comply with the same emission standards as those specified in §60.4231(c). Engines with a date of manufacture prior to July 1, 2008 (or January 1, 2009 for emergency engines) must comply with the emission standards specified in §60.4231(c) applicable to engines manufactured on July 1, 2008 (or January 1, 2009 for emergency engines).

(4) Owners and operators of stationary SI natural gas and lean burn LPG engines with a maximum engine power greater than 19 KW (25 HP), that are modified or reconstructed after June 12, 2006, must comply with the same emission standards as those specified in paragraph (d) or (e) of this section, except that such owners and operators of non-emergency engines and emergency engines greater than or equal to 130 HP must meet a nitrogen oxides (NO_X) emission standard of 3.0 grams per HP-hour (g/HP-hr), a CO emission standard of 4.0 g/HP-hr (5.0 g/ HP-hr for non-emergency engines less than 100 HP), and a volatile organic compounds (VOC) emission standard of 1.0 g/HP-hr, or a NO_X emission standard of 250 ppmvd at 15 percent oxygen (O_2) , a CO emission standard 540 ppmvd at 15 percent O₂ (675 ppmvd at 15 percent O₂ for non-emergency engines less than 100 HP), and a VOC emission standard of 86 ppmvd at 15 percent O_2 , where the date of manufacture of the engine is:

(i) Prior to July 1, 2007, for non-emergency engines with a maximum engine power greater than or equal to 500 HP (except lean burn natural gas engines and LPG engines with a maximum engine power greater than or equal to 500 HP and less than 1,350 HP);

(ii) Prior to July 1, 2008, for nonemergency engines with a maximum engine power less than 500 HP;

(iii) Prior to January 1, 2009, for emergency engines;

(iv) Prior to January 1, 2008, for nonemergency lean burn natural gas engines and LPG engines with a maximum engine power greater than or equal to 500 HP and less than 1,350 HP.

(5) Owners and operators of stationary SI landfill/digester gas ICE engines with a maximum engine power greater than 19 KW (25 HP), that are modified or reconstructed after June 12, 2006, must comply with the same emission standards as those specified in paragraph (e) of this section for stationary landfill/digester gas engines. Engines with maximum engine power less than 500 HP and a date of manufacture prior to July 1, 2008 must comply with the emission standards specified in paragraph (e) of this section for stationary landfill/digester gas ICE with a maximum engine power less than 500 HP manufactured on July 1, 2008. Engines with a maximum engine power greater than or equal to 500 HP (except lean burn engines greater than or equal to 500 HP and less than 1,350 HP) and a date of manufacture prior to July 1, 2007 must comply with the emission standards specified in paragraph (e) of this section for stationary landfill/digester gas ICE with a maximum engine power greater than or equal to 500 HP (except lean burn engines greater than or equal to 500 HP and less than 1,350 HP) manufactured on July 1, 2007. Lean burn engines greater than or equal to 500 HP and less than 1,350 HP with a date of manufacture prior to January 1, 2008 must comply with the emission standards specified in paragraph (e) of this section for stationary landfill/digester gas ICE that are lean burn engines greater than or equal to 500 HP and less than 1.350 HP and manufactured on January 1, 2008.

(g) Owners and operators of stationary SI wellhead gas ICE engines may petition the Administrator for approval on a case-by-case basis to meet emission standards no less stringent than the emission standards that apply to stationary emergency SI engines greater than 25 HP and less than 130 HP due to the presence of high sulfur levels in the fuel, as specified in Table 1 to this subpart. The request must, at a minimum, demonstrate that the fuel has high sulfur levels that prevent the use of aftertreatment controls and also

40 CFR Ch. I (7–1–23 Edition)

that the owner has reasonably made all attempts possible to obtain an engine that will meet the standards without the use of aftertreatment controls. The petition must request the most stringent standards reasonably applicable to the engine using the fuel.

(h) Owners and operators of stationary SI ICE that are required to meet standards that reference 40 CFR 1048.101 must, if testing their engines in use, meet the standards in that section applicable to field testing, except as indicated in paragraph (e) of this section.

[73 FR 3591, Jan. 18, 2008, as amended at 76 FR 37973, June 28, 2011]

§60.4234 How long must I meet the emission standards if I am an owner or operator of a stationary SI internal combustion engine?

Owners and operators of stationary SI ICE must operate and maintain stationary SI ICE that achieve the emission standards as required in §60.4233 over the entire life of the engine.

OTHER REQUIREMENTS FOR OWNERS AND OPERATORS

§ 60.4235 What fuel requirements must I meet if I am an owner or operator of a stationary SI gasoline fired internal combustion engine subject to this subpart?

Owners and operators of stationary SI ICE subject to this subpart that use gasoline must use gasoline that meets the per gallon sulfur limit in 40 CFR 1090.205.

 $[73\ {\rm FR}\ 3591,\ {\rm Jan.}\ 18,\ 2008,\ {\rm as}\ {\rm amended}\ {\rm at}\ 85\ {\rm FR}\ 78463,\ {\rm Dec.}\ 4,\ 2020]$

§60.4236 What is the deadline for importing or installing stationary SI ICE produced in previous model years?

(a) After July 1, 2010, owners and operators may not install stationary SI ICE with a maximum engine power of less than 500 HP that do not meet the applicable requirements in §60.4233.

(b) After July 1, 2009, owners and operators may not install stationary SI ICE with a maximum engine power of greater than or equal to 500 HP that do not meet the applicable requirements in §60.4233, except that lean burn engines with a maximum engine power

greater than or equal to 500 HP and less than 1,350 HP that do not meet the applicable requirements in §60.4233 may not be installed after January 1, 2010.

(c) For emergency stationary SI ICE with a maximum engine power of greater than 19 KW (25 HP), owners and operators may not install engines that do not meet the applicable requirements in §60.4233 after January 1, 2011.

(d) In addition to the requirements specified in §§ 60.4231 and 60.4233, it is prohibited to import stationary SI ICE less than or equal to 19 KW (25 HP), stationary rich burn LPG SI ICE, and stationary gasoline SI ICE that do not meet the applicable requirements specified in paragraphs (a), (b), and (c) of this section.

(e) The requirements of this section do not apply to owners and operators of stationary SI ICE that have been modified or reconstructed, and they do not apply to engines that were removed from one existing location and reinstalled at a new location.

§ 60.4237 What are the monitoring requirements if I am an owner or operator of an emergency stationary SI internal combustion engine?

(a) Starting on July 1, 2010, if the emergency stationary SI internal combustion engine that is greater than or equal to 500 HP that was built on or after July 1, 2010, does not meet the standards applicable to non-emergency engines, the owner or operator must install a non-resettable hour meter.

(b) Starting on January 1, 2011, if the emergency stationary SI internal combustion engine that is greater than or equal to 130 HP and less than 500 HP that was built on or after January 1, 2011, does not meet the standards applicable to non-emergency engines, the owner or operator must install a nonresettable hour meter.

(c) If you are an owner or operator of an emergency stationary SI internal combustion engine that is less than 130 HP, was built on or after July 1, 2008, and does not meet the standards applicable to non-emergency engines, you must install a non-resettable hour meter upon startup of your emergency engine.

§60.4239

COMPLIANCE REQUIREMENTS FOR MANUFACTURERS

§60.4238 What are my compliance requirements if I am a manufacturer of stationary SI internal combustion engines ≤19 KW (25 HP) or a manufacturer of equipment containing such engines?

Stationary SI internal combustion engine manufacturers who are subject to the emission standards specified in §60.4231(a) must certify their stationary SI ICE using the certification and testing procedures required in 40 CFR part 1054, subparts C and F. Manufacturers of equipment containing stationary SI internal combustion engines meeting the provisions of 40 CFR part 1054 must meet the provisions of 40 CFR part 1060, subpart C, to the extent they apply to equipment manufacturers.

[86 FR 34361, June 29, 2021]

§60.4239 What are my compliance requirements if I am a manufacturer of stationary SI internal combustion engines >19 KW (25 HP) that use gasoline or a manufacturer of equipment containing such engines?

Stationary SI internal combustion engine manufacturers who are subject to the emission standards specified in §60.4231(b) must certify their stationary SI ICE using the certification procedures required in 40 CFR part 1048, subpart C, and must test their engines as specified in that part. Stationary SI internal combustion engine manufacturers who certify their stationary SI ICE with a maximum engine power less than or equal to 30 KW (40 HP) with a total displacement less than or equal to 1,000 cc to the certification emission standards and other requirements for new nonroad SI engines in 40 CFR part 1054, and manufacturers of stationary SI emergency engines that are greater than 25 HP and less than 130 HP who meet the Phase 1 emission standards in 40 CFR part 1054, appendix I, applicable to class II engines, must certify their stationary SI ICE using the certification and testing procedures required in 40 CFR part 1054, subparts C and F. Manufacturers of equipment containing stationary SI internal combustion engines meeting

the provisions of 40 CFR part 1054 must meet the provisions of 40 CFR part 1060, subpart C, to the extent they apply to equipment manufacturers.

[86 FR 34361, June 29, 2021]

§60.4240 What are my compliance requirements if I am a manufacturer of stationary SI internal combustion engines >19 KW (25 HP) that are rich burn engines that use LPG or a manufacturer of equipment containing such engines?

Stationary SI internal combustion engine manufacturers who are subject to the emission standards specified in §60.4231(c) must certify their stationary SI ICE using the certification procedures required in 40 CFR part 1048, subpart C, and must test their engines as specified in that part. Stationary SI internal combustion engine manufacturers who certify their stationary SI ICE with a maximum engine power less than or equal to 30 KW (40 HP) with a total displacement less than or equal to $1,000 \ cc$ to the certification emission standards and other requirements for new nonroad SI engines in 40 CFR part 1054, and manufacturers of stationary SI emergency engines that are greater than 25 HP and less than 130 HP who meet the Phase 1 emission standards in 40 CFR part 1054, appendix I, applicable to class II engines, must certify their stationary SI ICE using the certification and testing procedures required in 40 CFR part 1054, subparts C and F. Manufacturers of equipment containing stationary SI internal combustion engines meeting the provisions of 40 CFR part 1054 must meet the provisions of 40 CFR part 1060, subpart C, to the extent they apply to equipment manufacturers.

[86 FR 34361, June 29, 2021]

§60.4241 What are my compliance requirements if I am a manufacturer of stationary SI internal combustion engines participating in the voluntary certification program or a manufacturer of equipment containing such engines?

(a) Manufacturers of stationary SI internal combustion engines with a maximum engine power greater than 19 KW (25 HP) that do not use gasoline and are not rich burn engines that use

40 CFR Ch. I (7-1-23 Edition)

LPG can choose to certify their engines to the emission standards in §60.4231(d) or (e), as applicable, under the voluntary certification program described in this subpart. Manufacturers who certify their engines under the voluntary certification program must meet the requirements as specified in paragraphs (b) through (g) of this section. In addition, manufacturers of stationary SI internal combustion engines who choose to certify their engines under the voluntary certification program, must also meet the requirements as specified in §60.4247. Manufacturers of stationary SI internal combustion engines who choose not to certify their engines under this section must notify the ultimate purchaser that testing requirements apply as described in §60.4243(b)(2); manufacturers must keep a copy of this notification for five years after shipping each engine and make those documents available to EPA upon request.

(b) Manufacturers of engines other than those certified to standards in 40 CFR part 1054 must certify their stationary SI ICE using the certification procedures required in 40 CFR part 1048, subpart C, and must follow the same test procedures that apply to Large SI nonroad engines under 40 CFR part 1048, but must use the D-1 cycle of International Organization for Standardization 8178-4: 1996(E) (incorporated by reference, see §60.17) or the test cycle requirements specified in Table 3 to 40 CFR 1048.505, except that Table 3 of 40 CFR 1048.505 applies to high load engines only. Manufacturers of any size may certify their stationary emergency engines at or above 130 hp using assigned deterioration factors established by EPA, consistent with 40 CFR 1048.240. Stationary SI internal combustion engine manufacturers who certify their stationary SI ICE with a maximum engine power less than or equal to 30 KW (40 HP) with a total displacement less than or equal to 1,000 cc to the certification emission standards and other requirements for new nonroad SI engines in 40 CFR part 1054, and manufacturers of emergency engines that are greater than 25 HP and less than 130 HP who meet the Phase 1 standards in 40 CFR part 1054, appendix I, applicable to class II engines, must

certify their stationary SI ICE using the certification and testing procedures required in 40 CFR part 1054, subparts C and F. Manufacturers of equipment containing stationary SI internal combustion engines meeting the provisions of 40 CFR part 1054 must meet the provisions of 40 CFR part 1060, subpart C, to the extent they apply to equipment manufacturers.

(c) Certification of stationary SI ICE to the emission standards specified in §60.4231(d) or (e), as applicable, is voluntary, but manufacturers who decide to certify are subject to all of the requirements indicated in this subpart with regard to the engines included in their certification. Manufacturers must clearly label their stationary SI engines as certified or non-certified engines.

(d) Manufacturers of natural gas fired stationary SI ICE who conduct voluntary certification of stationary SI ICE to the emission standards specified in §60.4231(d) or (e), as applicable, must certify their engines for operation using fuel that meets the definition of pipeline-quality natural gas. The fuel used for certifying stationary SI natural gas engines must meet the definition of pipeline-quality natural gas as described in §60.4248. In addition, the manufacturer must provide information to the owner and operator of the certified stationary SI engine including the specifications of the pipelinequality natural gas to which the engine is certified and what adjustments the owner or operator must make to the engine when installed in the field to ensure compliance with the emission standards.

(e) Manufacturers of stationary SI ICE that are lean burn engines fueled by LPG who conduct voluntary certification of stationary SI ICE to the emission standards specified in $\S 60.4231(d)$ or (e), as applicable, must certify their engines for operation using fuel that meets the specifications in 40 CFR 1065.720.

(f) Manufacturers may certify their engines for operation using gaseous fuels in addition to pipeline-quality natural gas; however, the manufacturer must specify the properties of that fuel and provide testing information showing that the engine will meet

the emission standards specified in §60.4231(d) or (e), as applicable, when operating on that fuel. The manufacturer must also provide instructions for configuring the stationary engine to meet the emission standards on fuels that do not meet the pipelinequality natural gas definition. The manufacturer must also provide information to the owner and operator of the certified stationary SI engine regarding the configuration that is most conducive to reduced emissions where the engine will be operated on gaseous fuels with different quality than the fuel that it was certified to.

(g) A stationary SI engine manufacturer may certify an engine family solely to the standards applicable to landfill/digester gas engines as specified in §60.4231(d) or (e), as applicable, but must certify their engines for operation using landfill/digester gas and must add a permanent label stating that the engine is for use only in landfill/digester gas applications. The label must be added according to the labeling requirements specified in 40 CFR 1048.135(b).

(h) For purposes of this subpart, when calculating emissions of volatile organic compounds, emissions of formaldehyde should not be included.

(i) For engines being certified to the voluntary certification standards in Table 1 of this subpart, the VOC measurement shall be made by following the procedures in 40 CFR part 1065, subpart C, to determine the total NMHC emissions. As an alternative, manufacturers may measure ethane, as well as methane, for excluding such levels from the total VOC measurement.

[73 FR 3591, Jan. 18, 2008, as amended at 73 FR 59176, Oct. 8, 2008; 76 FR 37974, June 28, 2011; 86 FR 34361, June 29, 2021]

§60.4242 What other requirements must I meet if I am a manufacturer of stationary SI internal combustion engines or equipment containing stationary SI internal combustion engines or a manufacturer of equipment containing such engines?

(a) Stationary SI internal combustion engine manufacturers must meet the provisions of 40 CFR parts 1048, 1054, and 1068, as applicable, except that engines certified pursuant to the

voluntary certification procedures in §60.4241 are subject only to the provisions indicated in §60.4247 and are permitted to provide instructions to owners and operators allowing for deviations from certified configurations, if such deviations are consistent with the provisions of §60.4241(c) through (f). Manufacturers of equipment containing stationary SI internal combustion engines meeting the provisions of 40 CFR part 1054 must meet the provisions of 40 CFR part 1060, as applicable. Labels on engines certified to 40 CFR part 1048 must refer to stationary engines, rather than or in addition to nonroad engines, as appropriate.

(b) An engine manufacturer certifying an engine family or families to standards under this subpart that are identical to standards identified in 40 CFR part 1048 or 1054 for that model year may certify any such family that contains both nonroad and stationary engines as a single engine family and/ or may include any such family containing stationary engines in the averaging, banking and trading provisions applicable for such engines under those parts. This paragraph (b) also applies to equipment or component manufacturers certifying to standards under 40 CFR part 1060.

(c) Manufacturers of engine families certified to 40 CFR part 1048 may meet the labeling requirements referred to in paragraph (a) of this section for stationary SI ICE by either adding a separate label containing the information required in paragraph (a) of this section or by adding the words "and stationary" after the word "nonroad" to the label.

(d) For all engines manufactured on or after January 1, 2011, and for all engines with a maximum engine power greater than 25 HP and less than 130 HP manufactured on or after July 1, 2008, a stationary SI engine manufacturer that certifies an engine family solely to the standards applicable to emergency engines must add a permanent label stating that the engines in that family are for emergency use only. The label must be added according to the labeling requirements specified in 40 CFR 1048.135(b). 40 CFR Ch. I (7–1–23 Edition)

(e) All stationary SI engines subject to mandatory certification that do not meet the requirements of this subpart must be labeled and exported according to 40 CFR 1068.230. Manufacturers of stationary engines with a maximum engine power greater than 25 HP that are not certified to standards and other requirements under 40 CFR part 1048 are subject to the labeling provisions of 40 CFR 1048.20 pertaining to excluded stationary engines.

(f) For manufacturers of gaseousfueled stationary engines required to meet the warranty provisions in 40 CFR 1054.120, we may establish an hour-based warranty period equal to at least the certified emissions life of the engines (in engine operating hours) if we determine that these engines are likely to operate for a number of hours greater than the applicable useful life within 24 months. We will not approve an alternate warranty under this paragraph (f) for nonroad engines. An alternate warranty period approved under this paragraph (f) will be the specified number of engine operating hours or two years, whichever comes first. The engine manufacturer shall request this alternate warranty period in its application for certification or in an earlier submission. We may approve an alternate warranty period for an engine family subject to the following conditions:

(1) The engines must be equipped with non-resettable hour meters.

(2) The engines must be designed to operate for a number of hours substantially greater than the applicable certified emissions life.

(3) The emission-related warranty for the engines may not be shorter than any published warranty offered by the manufacturer without charge for the engines. Similarly, the emission-related warranty for any component shall not be shorter than any published warranty offered by the manufacturer without charge for that component.

[86 FR 34362, June 29, 2021]

COMPLIANCE REQUIREMENTS FOR OWNERS AND OPERATORS

§60.4243 What are my compliance requirements if I am an owner or operator of a stationary SI internal combustion engine?

(a) If you are an owner or operator of a stationary SI internal combustion engine that is manufactured after July 1, 2008, and must comply with the emission standards specified in 60.4233(a)through (c), you must comply by purchasing an engine certified to the emission standards in 60.4231(a) through (c), as applicable, for the same engine class and maximum engine power. In addition, you must meet one of the requirements specified in (a)(1) and (2) of this section.

(1) If you operate and maintain the certified stationary SI internal combustion engine and control device according to the manufacturer's emission-related written instructions, you must keep records of conducted maintenance to demonstrate compliance, but no performance testing is required if you are an owner or operator. You must also meet the requirements as specified in 40 CFR part 1068, subparts A through D, as they apply to you. If you adjust engine settings according to and consistent with the manufacturer's instructions, your stationary SI internal combustion engine will not be considered out of compliance.

(2) If you do not operate and maintain the certified stationary SI internal combustion engine and control device according to the manufacturer's emission-related written instructions, your engine will be considered a noncertified engine, and you must demonstrate compliance according to (a)(2)(i) through (iii) of this section, as appropriate.

(i) If you are an owner or operator of a stationary SI internal combustion engine less than 100 HP, you must keep a maintenance plan and records of conducted maintenance to demonstrate compliance and must, to the extent practicable, maintain and operate the engine in a manner consistent with good air pollution control practice for minimizing emissions, but no performance testing is required if you are an owner or operator. (ii) If you are an owner or operator of a stationary SI internal combustion engine greater than or equal to 100 HP and less than or equal to 500 HP, you must keep a maintenance plan and records of conducted maintenance and must, to the extent practicable, maintain and operate the engine in a manner consistent with good air pollution control practice for minimizing emissions. In addition, you must conduct an initial performance test within 1 year of engine startup to demonstrate compliance.

(iii) If you are an owner or operator of a stationary SI internal combustion engine greater than 500 HP, you must keep a maintenance plan and records of conducted maintenance and must, to the extent practicable, maintain and operate the engine in a manner consistent with good air pollution control practice for minimizing emissions. In addition, you must conduct an initial performance test within 1 year of engine startup and conduct subsequent performance testing every 8,760 hours or 3 years, whichever comes first, thereafter to demonstrate compliance.

(b) If you are an owner or operator of a stationary SI internal combustion engine and must comply with the emission standards specified in 60.4233(d)or (e), you must demonstrate compliance according to one of the methods specified in paragraphs (b)(1) and (2) of this section.

(1) Purchasing an engine certified according to procedures specified in this subpart, for the same model year and demonstrating compliance according to one of the methods specified in paragraph (a) of this section.

(2) Purchasing a non-certified engine and demonstrating compliance with the emission standards specified in 60.4233(d) or (e) and according to the requirements specified in 60.4244, as applicable, and according to paragraphs (b)(2)(i) and (ii) of this section.

(i) If you are an owner or operator of a stationary SI internal combustion engine greater than 25 HP and less than or equal to 500 HP, you must keep a maintenance plan and records of conducted maintenance and must, to the extent practicable, maintain and operate the engine in a manner consistent with good air pollution control practice for minimizing emissions. In addition, you must conduct an initial performance test to demonstrate compliance.

(ii) If you are an owner or operator of a stationary SI internal combustion engine greater than 500 HP, you must keep a maintenance plan and records of conducted maintenance and must, to the extent practicable, maintain and operate the engine in a manner consistent with good air pollution control practice for minimizing emissions. In addition, you must conduct an initial performance test and conduct subsequent performance testing every 8,760 hours or 3 years, whichever comes first, thereafter to demonstrate compliance.

(c) If you are an owner or operator of a stationary SI internal combustion engine that must comply with the emission standards specified in §60.4233(f), you must demonstrate compliance according paragraph (b)(2)(i) or (ii) of this section, except that if you comply according to paragraph (b)(2)(i) of this section, you demonstrate that your non-certified engine complies with the emission standards specified in §60.4233(f).

(d) If you own or operate an emergency stationary ICE, you must operate the emergency stationary ICE according to the requirements in paragraphs (d)(1) through (3) of this section. In order for the engine to be considered an emergency stationary ICE under this subpart, any operation other than emergency operation, maintenance and testing, and operation in non-emergency situations for 50 hours per year, as described in paragraphs (d)(1) through (3), is prohibited. If you do not operate the engine according to the requirements in paragraphs (d)(1)through (3), the engine will not be considered an emergency engine under this subpart and must meet all requirements for non-emergency engines.

(1) There is no time limit on the use of emergency stationary ICE in emergency situations.

(2) You may operate your emergency stationary ICE for the purpose specified in paragraph (d)(2)(i) of this section for a maximum of 100 hours per calendar year. Any operation for non-emergency situations as allowed by

40 CFR Ch. I (7–1–23 Edition)

paragraph (d)(3) of this section counts as part of the 100 hours per calendar year allowed by this paragraph (d)(2).

(i) Emergency stationary ICE may be operated for maintenance checks and readiness testing, provided that the tests are recommended by federal, state or local government, the manufacturer, the vendor, the regional transmission organization or equivalent balancing authority and transmission operator, or the insurance company associated with the engine. The owner or operator may petition the Administrator for approval of additional hours to be used for maintenance checks and readiness testing, but a petition is not required if the owner or operator maintains records indicating that federal, state, or local standards require maintenance and testing of emergency ICE beyond 100 hours per calendar year.

(ii)–(iii) [Reserved]

(3) Emergency stationary ICE may be operated for up to 50 hours per calendar year in non-emergency situations. The 50 hours of operation in non-emergency situations are counted as part of the 100 hours per calendar year for maintenance and testing provided in paragraph (d)(2) of this section. Except as provided in paragraph (d)(3)(i) of this section, the 50 hours per year for nonemergency situations cannot be used for peak shaving or non-emergency demand response, or to generate income for a facility to an electric grid or otherwise supply power as part of a financial arrangement with another entity.

(i) The 50 hours per year for nonemergency situations can be used to supply power as part of a financial arrangement with another entity if all of the following conditions are met:

(A) The engine is dispatched by the local balancing authority or local transmission and distribution system operator;

(B) The dispatch is intended to mitigate local transmission and/or distribution limitations so as to avert potential voltage collapse or line overloads that could lead to the interruption of power supply in a local area or region.

(C) The dispatch follows reliability, emergency operation or similar protocols that follow specific NERC, regional, state, public utility commission or local standards or guidelines.

(D) The power is provided only to the facility itself or to support the local transmission and distribution system.

(E) The owner or operator identifies and records the entity that dispatches the engine and the specific NERC, regional, state, public utility commission or local standards or guidelines that are being followed for dispatching the engine. The local balancing authority or local transmission and distribution system operator may keep these records on behalf of the engine owner or operator.

(ii) [Reserved]

(e) Owners and operators of stationary SI natural gas fired engines may operate their engines using propane for a maximum of 100 hours per year as an alternative fuel solely during emergency operations, but must keep records of such use. If propane is used for more than 100 hours per year in an engine that is not certified to the emission standards when using propane, the owners and operators are required to conduct a performance test to demonstrate compliance with the emission standards of §60.4233.

(f) If you are an owner or operator of a stationary SI internal combustion engine that is less than or equal to 500 HP and you purchase a non-certified engine or you do not operate and maintain your certified stationary SI internal combustion engine and control device according to the manufacturer's written emission-related instructions, you are required to perform initial performance testing as indicated in this section, but you are not required to conduct subsequent performance testing unless the stationary engine undergoes rebuild, major repair or maintenance. Engine rebuilding means to overhaul an engine or to otherwise perform extensive service on the engine (or on a portion of the engine or engine system). For the purpose of this paragraph (f), perform extensive service means to disassemble the engine (or portion of the engine or engine system), inspect and/or replace many of the parts, and reassemble the engine

(or portion of the engine or engine system) in such a manner that significantly increases the service life of the resultant engine.

(g) It is expected that air-to-fuel ratio controllers will be used with the operation of three-way catalysts/nonselective catalytic reduction. The AFR controller must be maintained and operated appropriately in order to ensure proper operation of the engine and control device to minimize emissions at all times.

(h) If you are an owner/operator of an stationary SI internal combustion engine with maximum engine power greater than or equal to 500 HP that is manufactured after July 1, 2007 and before July 1, 2008, and must comply with the emission standards specified in sections 60.4233(b) or (c), you must comply by one of the methods specified in paragraphs (h)(1) through (h)(4) of this section.

(1) Purchasing an engine certified according to 40 CFR part 1048. The engine must be installed and configured according to the manufacturer's specifications.

(2) Keeping records of performance test results for each pollutant for a test conducted on a similar engine. The test must have been conducted using the same methods specified in this subpart and these methods must have been followed correctly.

(3) Keeping records of engine manufacturer data indicating compliance with the standards.

(4) Keeping records of control device vendor data indicating compliance with the standards.

(i) If you are an owner or operator of a modified or reconstructed stationary SI internal combustion engine and must comply with the emission standards specified in $\S 60.4233(f)$, you must demonstrate compliance according to one of the methods specified in paragraphs (i)(1) or (2) of this section.

(1) Purchasing, or otherwise owning or operating, an engine certified to the emission standards in §60.4233(f), as applicable.

(2) Conducting a performance test to demonstrate initial compliance with the emission standards according to the requirements specified in §60.4244. The test must be conducted within 60 days after the engine commences operation after the modification or reconstruction.

[73 FR 3591, Jan. 18, 2008, as amended at 76
FR 37974, June 28, 2011; 78 FR 6697, Jan. 30, 2013; 86 FR 34362, June 29, 2021; 87 FR 48606, Aug. 10, 2022]

TESTING REQUIREMENTS FOR OWNERS AND OPERATORS

§60.4244 What test methods and other procedures must I use if I am an owner or operator of a stationary SI internal combustion engine?

Owners and operators of stationary SI ICE who conduct performance tests must follow the procedures in paragraphs (a) through (f) of this section.

(a) Each performance test must be conducted within 10 percent of 100 percent peak (or the highest achievable) load and according to the requirements in §60.8 and under the specific condi40 CFR Ch. I (7–1–23 Edition)

tions that are specified by Table 2 to this subpart.

(b) You may not conduct performance tests during periods of startup, shutdown, or malfunction, as specified in §60.8(c). If your stationary SI internal combustion engine is non-operational, you do not need to startup the engine solely to conduct a performance test; however, you must conduct the performance test immediately upon startup of the engine.

(c) You must conduct three separate test runs for each performance test required in this section, as specified in $\S60.8(f)$. Each test run must be conducted within 10 percent of 100 percent peak (or the highest achievable) load and last at least 1 hour.

(d) To determine compliance with the NO_X mass per unit output emission limitation, convert the concentration of NO_X in the engine exhaust using Equation 1 of this section:

$$ER = \frac{C_d \times 1.912 \times 10^{-3} \times Q \times T}{HP - hr} \qquad (Eq. 1)$$

Where:

- $ER = Emission rate of NO_x in g/HP-hr.$
- C_d = Measured NO_x concentration in parts per million by volume (ppmv). 1.912 × 10⁻³ = Conversion constant for ppm
- NO_X to grams per standard cubic meter at 20 degrees Celsius.
- Q = Stack gas volumetric flow rate, in standard cubic meter per hour, dry basis.

T = Time of test run, in hours.

HP-hr = Brake work of the engine, horsepower-hour (HP-hr).

(e) To determine compliance with the CO mass per unit output emission limitation, convert the concentration of CO in the engine exhaust using Equation 2 of this section:

$$ER = \frac{C_{d} \times 1.164 \times 10^{-3} \times Q \times T}{HP - hr} \qquad (Eq. 2)$$

Where:

ER = Emission rate of CO in g/HP-hr.

- C_d = Measured CO concentration in ppmv.
- 1.164×10^{-3} = Conversion constant for ppm CO to grams per standard cubic meter at 20 degrees Celsius.
- Q = Stack gas volumetric flow rate, in standard cubic meters per hour, dry basis.
- T = Time of test run, in hours. HP-hr = Brake work of the engine, in HP-hr.

(f) For purposes of this subpart, when calculating emissions of VOC, emis-

sions of formaldehyde should not be included. To determine compliance with the VOC mass per unit output emission limitation, convert the concentration of VOC in the engine exhaust using Equation 3 of this section:

§60.4245

$$ER = \frac{C_d \times 1.833 \times 10^{-3} \times Q \times T}{HP - hr}$$
 (Eq. 3)

Where:

ER = Emission rate of VOC in g/HP-hr.

- $C_d = VOC$ concentration measured as propane in ppmv.
- 1.833×10^{-3} = Conversion constant for ppm VOC measured as propane, to grams per standard cubic meter at 20 degrees Celsius.

Q = Stack gas volumetric flow rate, in standard cubic meters per hour, dry basis.

T = Time of test run, in hours.

HP-hr = Brake work of the engine, in HP-hr.

(g) If the owner/operator chooses to measure VOC emissions using either Method 18 of 40 CFR part 60, appendix A, or Method 320 of 40 CFR part 63, appendix A, then it has the option of correcting the measured VOC emissions to account for the potential differences in measured values between these methods and Method 25A. The results from Method 18 and Method 320 can be corrected for response factor differences using Equations 4 and 5 of this section. The corrected VOC concentration can then be placed on a propane basis using Equation 6 of this section.

$$RF_{i} = \frac{C_{Mi}}{C_{Ai}} \qquad (Eq. 4)$$

Where:

 RF_i = Response factor of compound i when measured with EPA Method 25A.

- C_{Mi} = Measured concentration of compound i in ppmv as carbon.
- $C_{A\,i}$ = True concentration of compound i in ppmv as carbon.

$$C_{icorr} = RF_i \times C_{imeas}$$
 (Eq. 5)

Where:

- $C_{\rm icorr}$ = Concentration of compound i corrected to the value that would have been measured by EPA Method 25A, ppmv as carbon.
- $C_{\rm imeas}$ = Concentration of compound i measured by EPA Method 320, ppmv as carbon.

$$C_{Peq} = 0.6098 \times C_{icorr}$$
 (Eq. 6)

Where:

 C_{Peq} = Concentration of compound i in mg of propane equivalent per DSCM.

NOTIFICATION, REPORTS, AND RECORDS FOR OWNERS AND OPERATORS

§60.4245 What are my notification, reporting, and recordkeeping requirements if I am an owner or operator of a stationary SI internal combustion engine?

Owners or operators of stationary SI ICE must meet the following notification, reporting and recordkeeping requirements.

(a) Owners and operators of all stationary SI ICE must keep records of the information in paragraphs (a)(1) through (4) of this section.

(1) All notifications submitted to comply with this subpart and all documentation supporting any notification.

(2) Maintenance conducted on the engine.

(3) If the stationary SI internal combustion engine is a certified engine, documentation from the manufacturer that the engine is certified to meet the emission standards and information as required in 40 CFR parts 1048, 1054, and 1060, as applicable.

(4) If the stationary SI internal combustion engine is not a certified engine or is a certified engine operating in a non-certified manner and subject to \$60.4243(a)(2), documentation that the engine meets the emission standards.

(b) For all stationary SI emergency ICE greater than or equal to 500 HP manufactured on or after July 1, 2010, that do not meet the standards applicable to non-emergency engines, the owner or operator of must keep records of the hours of operation of the engine that is recorded through the non-resettable hour meter. For all stationary SI emergency ICE greater than or equal to 130 HP and less than 500 HP manufactured on or after July 1, 2011 that do not meet the standards applicable to non-emergency engines, the owner or operator of must keep records of the hours of operation of the engine that is recorded through the non-resettable hour meter. For all stationary SI emergency ICE greater than 25 HP and less than 130 HP manufactured on or after

H6H6L3 with DISOLLER

July 1, 2008, that do not meet the standards applicable to non-emergency engines, the owner or operator of must keep records of the hours of operation of the engine that is recorded through the non-resettable hour meter. The owner or operator must document how many hours are spent for emergency operation, including what classified the operation as emergency and how many hours are spent for non-emergency operation.

(c) Owners and operators of stationary SI ICE greater than or equal to 500 HP that have not been certified by an engine manufacturer to meet the emission standards in 60.4231 must submit an initial notification as required in 60.7(a)(1). The notification must include the information in paragraphs (c)(1) through (5) of this section.

(1) Name and address of the owner or operator;

 $\left(2\right)$ The address of the affected source;

(3) Engine information including make, model, engine family, serial number, model year, maximum engine power, and engine displacement;

(4) Emission control equipment; and

(5) Fuel used.

(d) Owners and operators of stationary SI ICE that are subject to performance testing must submit a copy of each performance test as conducted in §60.4244 within 60 days after the test has been completed. Performance test reports using EPA Method 18, EPA Method 320. or ASTM D6348-03 (incorporated by reference—see 40 CFR 60.17) to measure VOC require reporting of all QA/QC data. For Method 18, report results from sections 8.4 and 11.1.1.4; for Method 320, report results from sections 8.6.2, 9.0, and 13.0; and for ASTM D6348-03 report results of all QA/QC procedures in Annexes 1-7.

(e) If you own or operate an emergency stationary SI ICE with a maximum engine power more than 100 HP that operates for the purpose specified in 60.4243(d)(3)(i), you must submit an annual report according to the requirements in paragraphs (e)(1) through (3) of this section.

(1) The report must contain the following information:

(i) Company name and address where the engine is located.

40 CFR Ch. I (7–1–23 Edition)

(ii) Date of the report and beginning and ending dates of the reporting period.

(iii) Engine site rating and model year.

(iv) Latitude and longitude of the engine in decimal degrees reported to the fifth decimal place.

(v)-(vi) [Reserved]

(vii) Hours spent for operation for the purposes specified in 60.4243(d)(3)(i), including the date, start time, and end time for engine operation for the purposes specified in 60.4243(d)(3)(i). The report must also identify the entity that dispatched the engine and the situation that necessitated the dispatch of the engine.

(2) The first annual report must cover the calendar year 2015 and must be submitted no later than March 31, 2016. Subsequent annual reports for each calendar year must be submitted no later than March 31 of the following calendar year.

(3) The annual report must be submitted electronically using the subpart specific reporting form in the Compliance and Emissions Data Reporting Interface (CEDRI) that is accessed through EPA's Central Data Exchange (CDX) (www.epa.gov/cdx). However, if the reporting form specific to this subpart is not available in CEDRI at the time that the report is due, the written report must be submitted to the Administrator at the appropriate address listed in §60.4.

[73 FR 3591, Jan. 18, 2008, as amended at 73
FR 59177, Oct. 8, 2008; 78 FR 6697, Jan. 30, 2013;
81 FR 59809, Aug. 30, 2016; 86 FR 34362, June 29, 2021; 87 FR 48606, Aug. 10, 2022]

GENERAL PROVISIONS

§60.4246 What General Provisions and confidential information provisions apply to me?

(a) Table 3 to this subpart shows which parts of the General Provisions in §§ 60.1 through 60.19 apply to you.

(b) The provisions of 40 CFR 1068.10 and 1068.11 apply for engine manufacturers. For others, the general confidential business information (CBI) provisions apply as described in 40 CFR part 2.

[88 FR 4471, Jan. 24, 2023]

MOBILE SOURCE PROVISIONS

§60.4247 What parts of the mobile source provisions apply to me if I am a manufacturer of stationary SI internal combustion engines or a manufacturer of equipment containing such engines?

(a) Manufacturers certifying to emission standards in 40 CFR part 1054 must meet the provisions of 40 CFR part 1054. Note that 40 CFR part 1054, appendix I, describes various provisions that do not apply for engines meeting Phase 1 standards in 40 CFR part 1054. Manufacturers of equipment containing stationary SI internal combustion engines meeting the provisions of 40 CFR part 1054 must meet the provisions of 40 CFR part 1060 to the extent they apply to equipment manufacturers.

(b) Manufacturers required to certify to emission standards in 40 CFR part 1048 must meet the provisions of 40 CFR part 1048. Manufacturers certifying to emission standards in 40 CFR part 1048 pursuant to the voluntary certification program must meet the requirements in Table 4 to this subpart as well as the standards in 40 CFR 1048.101.

(c) For manufacturers of stationary SI internal combustion engines participating in the voluntary certification program and certifying engines to Table 1 to this subpart, Table 4 to this subpart shows which parts of the mobile source provisions in 40 CFR parts 1048, 1065, and 1068 apply to you. Compliance with the deterioration factor provisions under 40 CFR 1048.205(n) and 1048.240 will be required for engines built new on and after January 1, 2010. Prior to January 1, 2010, manufacturers of stationary internal combustion engines participating in the voluntary certification program have the option to develop their own deterioration factors based on an engineering analysis.

[73 FR 3591, Jan. 18, 2008, as amended at 73 FR 59177, Oct. 8, 2008; 86 FR 34362, June 29, 2021]

DEFINITIONS

§60.4248 What definitions apply to this subpart?

As used in this subpart, all terms not defined herein shall have the meaning

given them in the CAA and in subpart A of this part.

Certified emissions life means the period during which the engine is designed to properly function in terms of reliability and fuel consumption, without being remanufactured, specified as a number of hours of operation or calendar years, whichever comes first. The values for certified emissions life for stationary SI ICE with a maximum engine power less than or equal to 19 KW (25 HP) are given in 40 CFR 1054.107 and 1060.101, as appropriate. The values for certified emissions life for stationary SI ICE with a maximum engine power greater than 19 KW (25 HP) certified to 40 CFR part 1048 are given in 40 CFR 1048.101(g). The certified emissions life for stationary SI ICE with a maximum engine power greater than 75 KW (100 HP) certified under the voluntary manufacturer certification program of this subpart is 5,000 hours or 7 years, whichever comes first. You may request in your application for certification that we approve a shorter certified emissions life for an engine family. We may approve a shorter certified emissions life, in hours of engine operation but not in years, if we determine that these engines will rarely operate longer than the shorter certified emissions life. If engines identical to those in the engine family have already been produced and are in use, your demonstration must include documentation from such in-use engines. In other cases, your demonstration must include an engineering analysis of information equivalent to such in-use data, such as data from research engines or similar engine models that are already in production. Your demonstration must also include any overhaul interval that you recommend, any mechanical warranty that you offer for the engine or its components, and any relevant customer design specifications. Your demonstration may include any other relevant information. The certified emissions life value may not be shorter than any of the following:

(1) 1,000 hours of operation.

(2) Your recommended overhaul interval.

(3) Your mechanical warranty for the engine.

§60.4248

Certified stationary internal combustion engine means an engine that belongs to an engine family that has a certificate of conformity that complies with the emission standards and requirements in this part, or of 40 CFR part 1048 or 1054, as appropriate.

Combustion turbine means all equipment, including but not limited to the turbine, the fuel, air, lubrication and exhaust gas systems, control systems (except emissions control equipment), and any ancillary components and subcomponents comprising any simple cycle combustion turbine, any regenerative/recuperative cycle combustion turbine, the combustion turbine portion of any cogeneration cycle combustion system, or the combustion turbine portion of any combined cycle steam/ electric generating system.

Compression ignition means relating to a type of stationary internal combustion engine that is not a spark ignition engine.

Date of manufacture means one of the following things:

(1) For freshly manufactured engines and modified engines, date of manufacture means the date the engine is originally produced.

(2) For reconstructed engines, date of manufacture means the date the engine was originally produced, except as specified in paragraph (3) of this definition.

(3) Reconstructed engines are assigned a new date of manufacture if the fixed capital cost of the new and refurbished components exceeds 75 percent of the fixed capital cost of a comparable entirely new facility. An engine that is produced from a previously used engine block does not retain the date of manufacture of the engine in which the engine block was previously used if the engine is produced using all new components except for the engine block. In these cases, the date of manufacture is the date of reconstruction or the date the new engine is produced.

Diesel fuel means any liquid obtained from the distillation of petroleum with a boiling point of approximately 150 to 360 degrees Celsius. One commonly used form is number 2 distillate oil.

Digester gas means any gaseous byproduct of wastewater treatment typically formed through the anaerobic de40 CFR Ch. I (7–1–23 Edition)

composition of organic waste materials and composed principally of methane and carbon dioxide (CO_2) .

Emergency stationary internal combustion engine means any stationary reciprocating internal combustion engine that meets all of the criteria in paragraphs (1) through (3) of this definition. All emergency stationary ICE must comply with the requirements specified in 60.4243(d) in order to be considered emergency stationary ICE. If the engine does not comply with the requirements specified in 60.4243(d), then it is not considered to be an emergency stationary ICE under this subpart.

(1) The stationary ICE is operated to provide electrical power or mechanical work during an emergency situation. Examples include stationary ICE used to produce power for critical networks or equipment (including power supplied to portions of a facility) when electric power from the local utility (or the normal power source, if the facility runs on its own power production) is interrupted, or stationary ICE used to pump water in the case of fire or flood, etc.

(2) The stationary ICE is operated under limited circumstances for situations not included in paragraph (1) of this definition, as specified in §60.4243(d).

(3) The stationary ICE operates as part of a financial arrangement with another entity in situations not included in paragraph (1) of this definition only as allowed in 60.4243(d)(3)(i).

Engine manufacturer means the manufacturer of the engine. See the definition of "manufacturer" in this section.

Four-stroke engine means any type of engine which completes the power cycle in two crankshaft revolutions, with intake and compression strokes in the first revolution and power and exhaust strokes in the second revolution.

Freshly manufactured engine means an engine that has not been placed into service. An engine becomes freshly manufactured when it is originally produced.

Gasoline means any fuel sold in any State for use in motor vehicles and motor vehicle engines, or nonroad or stationary engines, and commonly or commercially known or sold as gasoline.

Installed means the engine is placed and secured at the location where it is intended to be operated.

Landfill gas means a gaseous by-product of the land application of municipal refuse typically formed through the anaerobic decomposition of waste materials and composed principally of methane and CO_2 .

Lean burn engine means any twostroke or four-stroke spark ignited engine that does not meet the definition of a rich burn engine.

Liquefied petroleum gas means any liquefied hydrocarbon gas obtained as a by-product in petroleum refining or natural gas production.

Manufacturer has the meaning given in section 216(1) of the Clean Air Act. In general, this term includes any person who manufactures a stationary engine for sale in the United States or otherwise introduces a new stationary engine into commerce in the United States. This includes importers who import stationary engines for resale.

Maximum engine power means maximum engine power as defined in 40 CFR 1048.801.

Model year means the calendar year in which an engine is manufactured (see "date of manufacture"), except as follows:

(1) Model year means the annual new model production period of the engine manufacturer in which an engine is manufactured (see "date of manufacture"), if the annual new model production period is different than the calendar year and includes January 1 of the calendar year for which the model year is named. It may not begin before January 2 of the previous calendar year and it must end by December 31 of the named calendar year.

(2) For an engine that is converted to a stationary engine after being placed into service as a nonroad or other nonstationary engine, model year means the calendar year or new model production period in which the engine was manufactured (see "date of manufacture").

Natural gas means a naturally occurring mixture of hydrocarbon and nonhydrocarbon gases found in geologic formations beneath the Earth's surface, of which the principal constituent is methane. Natural gas may be field or pipeline quality.

Other internal combustion engine means any internal combustion engine, except combustion turbines, which is not a reciprocating internal combustion engine or rotary internal combustion engine.

Pipeline-quality natural gas means a naturally occurring fluid mixture of hydrocarbons (e.g., methane, ethane, or propane) produced in geological formations beneath the Earth's surface that maintains a gaseous state at standard atmospheric temperature and pressure under ordinary conditions, and which is provided by a supplier through a pipeline. Pipeline-quality natural gas must either be composed of at least 70 percent methane by volume or have a gross calorific value between 950 and 1,100 British thermal units per standard cubic foot.

Rich burn engine means any fourstroke spark ignited engine where the manufacturer's recommended operating air/fuel ratio divided by the stoichiometric air/fuel ratio at full load conditions is less than or equal to 1.1. Engines originally manufactured as rich burn engines, but modified prior to June 12, 2006, with passive emission control technology for NO_X (such as pre-combustion chambers) will be considered lean burn engines. Also, existing engines where there are no manufacturer's recommendations regarding air/fuel ratio will be considered a rich burn engine if the excess oxygen content of the exhaust at full load conditions is less than or equal to 2 percent.

Rotary internal combustion engine means any internal combustion engine which uses rotary motion to convert heat energy into mechanical work.

Spark ignition means relating to either: a gasoline-fueled engine; or any other type of engine with a spark plug (or other sparking device) and with operating characteristics significantly similar to the theoretical Otto combustion cycle. Spark ignition engines usually use a throttle to regulate intake air flow to control power during normal operation. Dual-fuel engines in which a liquid fuel (typically diesel fuel) is used for compression ignition and gaseous fuel (typically natural gas)

Pt. 60, Subpt. JJJJ, Table 1

is used as the primary fuel at an annual average ratio of less than 2 parts diesel fuel to 100 parts total fuel on an energy equivalent basis are spark ignition engines.

Stationary internal combustion engine means any internal combustion engine, except combustion turbines, that converts heat energy into mechanical work and is not mobile. Stationary ICE differ from mobile ICE in that a stationary internal combustion engine is not a nonroad engine as defined at 40 CFR 1068.30 (excluding paragraph (2)(ii) of that definition), and is not used to propel a motor vehicle, aircraft, or a vehicle used solely for competition. Stationary ICE include reciprocating ICE, rotary ICE, and other ICE, except combustion turbines.

Stationary internal combustion engine test cell/stand means an engine test cell/ stand, as defined in 40 CFR part 63, subpart PPPPP, that tests stationary ICE.

Stoichiometric means the theoretical air-to-fuel ratio required for complete combustion.

Subpart means 40 CFR part 60, subpart JJJJ.

40 CFR Ch. I (7-1-23 Edition)

Two-stroke engine means a type of engine which completes the power cycle in single crankshaft revolution by combining the intake and compression operations into one stroke and the power and exhaust operations into a second stroke. This system requires auxiliary scavenging and inherently runs lean of stoichiometric.

Volatile organic compounds means volatile organic compounds as defined in 40 CFR 51.100(s).

Voluntary certification program means an optional engine certification program that manufacturers of stationary SI internal combustion engines with a maximum engine power greater than 19 KW (25 HP) that do not use gasoline and are not rich burn engines that use LPG can choose to participate in to certify their engines to the emission standards in §60.4231(d) or (e), as applicable.

[73 FR 3591, Jan. 18, 2008, as amended at 73 FR 59177, Oct. 8, 2008; 76 FR 37974, June 28, 2011; 78 FR 6698, Jan. 30, 2013; 86 FR 34363, June 29, 2021; 87 FR 48606, Aug. 10, 2022]

TABLE 1 TO SUBPART JJJJ OF PART 60-NO_X, CO, AND VOC EMISSION STANDARDS FOR STATIONARY NON-EMERGENCY SI ENGINES ≥100 HP (EXCEPT GASOLINE AND RICH BURN LPG), STATIONARY SI LANDFILL/DIGESTER GAS ENGINES, AND STA-TIONARY EMERGENCY ENGINES >25 HP

			Emission standards a					
Engine type and fuel	Maximum engine power	Manufacture date	g/HP-hr ppmvd at 15				% O ₂	
			NO _x	со	VOC^{d}	$NO_{\rm X}$	со	VOC ^d
Non-Emergency SI Natural Gas ^b and Non-Emergency SI Lean Burn LPG ^b .	100≤HP<500	7/1/2008	2.0	4.0	1.0	160	540	86
• •		1/1/2011	1.0	2.0	0.7	82	270	60
Non-Emergency SI Lean Burn Natural Gas and LPG.	500≤HP<1,350	1/1/2008	2.0	4.0	1.0	160	540	86
		7/1/2010	1.0	2.0	0.7	82	270	60
Non-Emergency SI Natural Gas and Non-Emergency SI Lean Burn LPG (except lean burn 500≤HP<1,350).	HP≥500	7/1/2007	2.0	4.0	1.0	160	540	86
	HP≥500	7/1/2010	1.0	2.0	0.7	82	270	60
Landfill/Digester Gas (except lean burn 500 <hp<1.350).< td=""><td>HP<500</td><td>7/1/2008</td><td>3.0</td><td>5.0</td><td>1.0</td><td>220</td><td>610</td><td>80</td></hp<1.350).<>	HP<500	7/1/2008	3.0	5.0	1.0	220	610	80
,		1/1/2011	2.0	5.0	1.0	150	610	80
	HP≥500	7/1/2007	3.0	5.0	1.0	220	610	80
		7/1/2010	2.0	5.0	1.0	150	610	80
Landfill/Digester Gas Lean Burn	500≤HP<1,350	1/1/2008	3.0	5.0	1.0	220	610	80
		7/1/2010	2.0	5.0	1.0	150	610	80
Emergency	25 <hp<130< td=""><td>1/1/2009</td><td>° 10</td><td>387</td><td>N/A</td><td>N/A</td><td>N/A</td><td>N/A</td></hp<130<>	1/1/2009	° 10	387	N/A	N/A	N/A	N/A
	HP≥130		2.0	4.0	1.0	160	540	86

^a Owners and operators of stationary non-certified SI engines may choose to comply with the emission standards in units of ei-

^a Owners and operators of stationary indirecting of engines may choose is example, intermediate there g/HP-hr or ppmvd at 15 percent O₂. ^b Owners and operators of new or reconstructed non-emergency lean burn SI stationary engines with a site rating of greater than or equal to 250 brake HP located at a major source that are meeting the requirements of 40 CFR part 63, subpart ZZZZ, Table 2a do not have to comply with the CO emission standards of Table 1 of this subpart. ^c The emission standards applicable to emergency engines between 25 HP and 130 HP are in terms of NO_X + HC.

Pt. 60, Subpt. JJJJ, Table 2

^dFor purposes of this subpart, when calculating emissions of volatile organic compounds, emissions of formaldehyde should not be included.

[76 FR 37975, June 28, 2011]

TABLE 2 TO SUBPART JJJJ OF PART $60\mbox{--}Requirements$ for Performance Tests

As stated in 60.4244, you must comply with the following requirements for performance tests within 10 percent of 100 percent peak (or the highest achievable) load].

For each	Complying with the requirement to	□ou must	□sing	According to the following requirements
 Stationary SI in- ternal combustion engine dem- onstrating compli- ance according to □60.4244. 	a. Limit the con- centration of NO _X in the stationary SI internal com- bustion engine exhaust.	i. Select the sam- pling port loca- tion and the number/location of traverse points at the exhaust of the stationary in- ternal combus- tion engine□	(1) Method 1 or 1A of 40 CFR part 60, appendix A⊡ 1, if measuring flow rate.	(a) Alternatively, for NO _X , O ₂ , and moisture measurement, ducts ≤6 inches in diameter may be sampled at a single point located at the duct centroid and ducts >6 and ≤12 inches in diameter may be sampled at 3 traverse points located at 16.7, 50.0, and 83.3% of the measurement line (:3-point long line). If the duct is >12 inches in diameter and the sam- pling port location meets the two and half-diameter criterion of Section 11.1.1 of Method 1 of 40 CFR part 60, Appendix A, the duct may be sampled at :3-point long line: otherwise, conduct the stratification test- ing and select sampling points ac- cording to Section 8.1.2 of Method 7E of 40 CFR part 60, Appendix A.
		Determine the O₂ concentration of the stationary internal combus- tion engine ex- haust at the sam- pling port loca- tion□	(2) Method 3, 3A, or 3B ^b of 40 CFR part 60, ap- pendix A ⁻ Z or ASTM Method D6522 ⁻ D0 (Re- approved 2005) ^a d.	(b) Measurements to determine O ₂ concentration must be made at the same time as the measurements for NO _X concentration.
		iii. If necessary, de- termine the ex- haust flowrate of the stationary in- ternal combus- tion engine ex- haust□	(3) Method 2 or 2C of 40 CFR part 60, appendix A⊡1 or Method 19 of 40 CFR part 60, appendix A⊡7.	(c) Measurements to determine the exhaust flowrate must be made (1) at the same time as the measurement for NO _X concentration or, alternatively (2) according to the option in Section 11.1.2 of Method 1A of 40 CFR part 60, Appendix A \Box 1, if applicable.
		iv. If necessary, measure mois- ture content of the stationary in- ternal combus- tion engine ex- haust at the sam- pling port loca- tion_and	(4) Method 4 of 40 CFR part 60, ap- pendix A:3, Method 320 of 40 CFR part 63, appendix A, ^e or ASTM Method D6348:03 d.e.	(d) Measurements to determine mois- ture must be made at the same time as the measurement for NO _x con- centration.
		v. Measure NO _x at the exhaust of the stationary in- ternal combus- tion engine⊡if using a control device, the sam- pling site must be located at the outlet of the con- trol device	(5) Method 7E of 40 CFR part 60, appendix $A \Box 4$, ASTM Method D6522 \Box 00 (Re- approved 2005), a ^{.d} Method 320 of 40 CFR part 63, appendix A, ^e or ASTM Method D6348 \Box 03 d ^{.e} .	(e) Results of this test consist of the average of the three 1-hour or longer runs.
Pt. 60, Subpt. JJJJ, Table 2

40 CFR Ch. I (7-1-23 Edition)

For each	Complying with the requirement to	□ou must	□sing	According to the following requirements
	b. Limit the con- centration of CO in the stationary SI internal com- bustion engine exhaust.	i. Select the sam- pling port loca- tion and the number/location of traverse points at the exhaust of the stationary in- ternal combus- tion engine⊡	(1) Method 1 or 1A of 40 CFR part 60, appendix A□ 1, if measuring flow rate.	(a) Alternatively, for CO, O ₂ , and moisture measurement, ducts ≤6 inches in diameter may be sampled at a single point located at the duct centroid and ducts >6 and ≤12 inches in diameter may be sampled at 3 traverse points located at 16.7, 50.0, and 83.3% of the measurement line (3-point long line). If the duct is >12 inches in diameter and the sampling port location meets the two and half-diameter criterion of Section 11.1.1 of Method 1 of 40 CFR part 60, Appendix A, the duct may be sampled at 3-point long line⊡otherwise, conduct the stratification testing and select sampling points according to Section 8.1.2 of Method 7E of 40 CFR part 60. Appendix A.
		 Determine the O₂ concentration of the stationary internal combus- tion engine ex- haust at the sam- pling port loca- tion⁻ 	(2) Method 3, 3A, or 3B ^b of 40 CFR part 60, ap- pendix A⊡2 or ASTM Method D6522⊡00 (Re- approved 2005:ad	(b) Measurements to determine O ₂ concentration must be made at the same time as the measurements for CO concentration.
		 iii. If necessary, determine the exhaust flowrate of the stationary internal combustion engine exhaust⁻ iv. If necessary, measure moist ture content of the stationary internal combustion engine exhaust at the sampling port location_and 	 (3) Method 2 or 2C of 40 CFR 60, appendix A⊡1 or Method 19 of 40 CFR part 60, ap- pendix A⊡7. (4) Method 4 of 40 CFR part 60, ap- pendix A⊒3, Method 320 of 40 CFR part 63, appendix A,ª or ASTM Method D6348⊡03 ^de. 	 (c) Measurements to determine the exhaust flowrate must be made (1) at the same time as the measurement for CO concentration or, alternatively (2) according to the option in Section 11.1.2 of Method 1A of 40 CFR part 60, Appendix A⊡1, if applicable. (d) Measurements to determine moisture must be made at the same time as the measurement for CO concentration.
		v. Measure CO at the exhaust of the stationary in- ternal combus- tion engine⊡if using a control device, the sam- pling site must be located at the outlet of the con- trol device	(5) Method 10 of 40 CFR part 60, appendix A4, ASTM Method D6522⊡00 (Re- approved 2005),ªde Meth- od 320 of 40 CFR part 63, ap- pendix A,e or ASTM Method D6348⊡03 de.	(e) Results of this test consist of the average of the three 1-hour or longer runs.

Pt. 60, Subpt. JJJJ, Table 2

For each	Complying with the requirement to	□ou must	□sing	According to the following requirements
	c. Limit the con- centration of VOC in the sta- tionary SI internal combustion en- gine exhaust.	i. Select the sam- pling port loca- tion and the number/location of traverse points at the exhaust of the stationary in- ternal combus- tion engine□	(1) Method 1 or 1A of 40 CFR part 60, appendix A □ 1, if measuring flow rate.	(a) Alternatively, for VOC, O ₂ , and moisture measurement, ducts ≤6 inches in diameter may be sampled at a single point located at the duct centroid and ducts >6 and ≤12 inches in diameter may be sampled at 3 traverse points located at 16.7, 50.0, and 83.3% of the measurement line (3-point long line). If the duct is >12 inches in diameter and the sam- pling port location meets the two and half-diameter criterion of Section 11.1.1 of Method 1 of 40 CFR part 60, Appendix A, the duct may be sampled at 13-point long line⊡other- wise, conduct the startification test- ing and select sampling points ac- cording to Section 8.1.2 of Method 7E of 40 CFR part 60, Appendix A.
		ii. Determine the O₂ concentration of the stationary internal combus- tion engine ex- haust at the sam- pling port loca- tion □	(2) Method 3, 3A, or 3B ^b of 40 CFR part 60, ap- pendix A⊡2 or ASTM Method D6522:D0 (Re- approved 2005) ^{ad} .	(b) Measurements to determine O ₂ concentration must be made at the same time as the measurements for VOC concentration.
		iii. If necessary, de- termine the ex- haust flowrate of the stationary in- ternal combus- tion engine ex- haust□	(3) Method 2 or 2C of 40 CFR 60, appendix A□ or Method 19 of 40 CFR part 60, ap- pendix A□7.	(c) Measurements to determine the exhaust flowrate must be made (1) at the same time as the measurement for VOC concentration or, alter- natively (2) according to the option in Section 11.1.2 of Method 1A of 40 CFR part 60, Appendix A□1, if appli- cable.
		iv. If necessary, measure mois- ture content of the stationary in- ternal combus- tion engine ex- haust at the sam- pling port loca- tion_and	(4) Method 4 of 40 CFR part 60, ap- pendix A::3, Method 320 of 40 CFR part 63, appendix A, ^e or ASTM Method D6348::03 d ^{.e} .	(d) Measurements to determine mois- ture must be made at the same time as the measurement for VOC con- centration.
		v. Measure VOC at the exhaust of the stationary in- ternal combus- tion engine if using a control device, the sam- pling site must be located at the outlet of the con- trol device	(5) Methods 25A and 18 of 40 CFR part 60, ap- pendices A⊡6 and A⊡7, Method 25A with the use of a hydrocarbon cutter as de- scribed in 40 CFR 1065.265, Method 18 of 40 CFR part 60, ap- pendix A⊡6,°° Method 320 of 40 CFR part 63, appendix A.° or ASTM Method DE348.733.4°	(e) Results of this test consist of the average of the three 1-hour or longer runs.

^a Also, you may petition the Administrator for approval to use alternative methods for portable analy⊡er. ^b ⊡ou may use ASME PTC 19.10⊡1981, Flue and Exhaust Gas Analyses, for measuring the O₂ content of the exhaust gas as an alternative to EPA Method 3B. AMSE PTC 19.10⊡1981 incorporated by reference, see 40 CFR 60.17 ^c ⊡ou may use EPA Method 18 of 40 CFR part 60, appendix A⊡6, provided that you conduct an adequate pre-survey test prior to the emissions test, such as the one described in OTM 11 on EPA's website (^a Incorporated by reference⊡see 40 CFR 60.17. ^e ⊡ou must meet the requirements in □60.4245(d).

[85 FR 63408, Oct. 7, 2020]

Pt. 60, Subpt. JJJJ, Table 3

40 CFR Ch. I (7-1-23 Edition)

TABLE 3 TO SUBPART JJJJ OF PART 60—Applicability of General Provisions to Subpart JJJJ

As stated in \Box 60.4246, you must comply with the following applicable General Provisions \Box

General provisions citation	Subject of citation	Applies to subpart	Explanation
□60.1	General applicability of the General Provisions.	□es.	
□60.2	Definitions	□es	Additional terms defined in □60.4248.
□60.3	□nits and abbreviations	□es.	
□60.4	Address	□es.	
□60.5	Determination of construction or modification.	□es.	
□60.6	Review of plans	□es.	
□60.7	Notification and Record- keeping.	□es	Except that □60.7 only ap- plies as specified in □60.4245.
□60.8	Performance tests	_es	Except that G0.8 only applies to owners and operators who are subject to performance testing in subpart JJJJ.
□60.9	Availability of information	□es.	
□60.10	State Authority	□es.	
□60.11	Compliance with standards and maintenance require- ments.	□es	Requirements are specified in subpart JJJJ.
□60.12	Circumvention	□es.	
□60.13	Monitoring requirements	No.	
□60.14	Modification	□es.	
□60.15	Reconstruction	□es.	
□60.16	Priority list	□es.	
□60.17	Incorporations by reference	□es.	
□60.18	General control device re- quirements.	No.	
□60.19	General notification and re- porting requirements.	□es.	

TABLE 4 TO SUBPART JJJJ OF PART 60—APPLICABILITY OF MOBILE SOURCE PROVI-SIONS FOR MANUFACTURERS PARTICIPATING IN THE VOLUNTARY CERTIFICATION PROGRAM AND CERTIFYING STATIONARY SI ICE TO EMISSION STANDARDS IN TABLE 1 OF SUBPART JJJJ

As stated in □60.4247, you must comply with the following applicable mobile source provisions if you are a manufacturer participating in the voluntary certification program and certifying stationary SI ICE to emission standards in Table 1 of subpart JJJJ□

Mobile source provisions cita- tion	Subject of citation	Applies to subpart	Explanation	
1048 subpart A 1048 subpart B	Overview and Applicability Emission Standards and Re- lated Requirements.	□es. □es	Except for the specific sec- tions below.	
1048.101 1048.105	Exhaust Emission Standards Evaporative Emission Stand- ards.	No. No.		
1048.110 1048.140	Diagnosing Malfunctions Certifying Blue Sky Series Engines.	No. No.		
1048.145 1048 subpart C	Interim Provisions Certifying Engine Families	No. □es	Except for the specific sec- tions below.	
1048.205(b) 1048.205(c)	AECD reporting OBD Requirements	□es. No.		
1048.205(n)	Deterioration Factors	Les	60.4247(c).	
1048.205(p)(1)	sion.	∟es.		
1048.205(p)(2)	Liquid Fuels as they require	No.		
1048.240(D)(C)(d)	Deterioration Factors	⊔es.		
1048 subpart D	gines.	∟es.		
1048 subpart E	Testing In-□se Engines	No.		
1048 subpart F	Test Procedures	l □es.		

§60.4310

Mobile source provisions cita- tion	Subject of citation	Applies to subpart	Explanation
1065.5(a)(4)	Raw sampling (refers reader back to the specific emis- sions regulation for guid- ance).	Des.	
1048 subpart G	Compliance Provisions	□es.	
1048 subpart I	Definitions and Other Ref- erence Information.	□es.	
1048 appendix I and II	□es.		
1065 (all subparts)	Engine Testing Procedures	□es	Except for the specific section below.
1065.715	Test Fuel Specifications for Natural Gas.	No.	
1068 (all subparts)	General Compliance Provi- sions for Nonroad Pro- grams.	□es	Except for the specific sec- tions below.
1068.245	Hardship Provisions for usual Circumstances.	No.	
1068.250	Hardship Provisions for Small-Volume Manufactur-	No.	
1068.255	Hardship Provisions for Equipment Manufacturers and Secondary Engine Manufacturers.	No.	

As stated in □60.4247, you must comply with the following applicable mobile source provisions if you are a manufacturer participating in the voluntary certification program and certifying stationary SI ICE to emission standards in Table 1 of subpart JJJJ□

Subpart KKKK—Standards of Performance for Stationary Combustion Turbines

SOURCE: 71 FR 38497, July 6, 2006, unless otherwise noted.

INTRODUCTION

§60.4300 What is the purpose of this subpart?

This subpart establishes emission standards and compliance schedules for the control of emissions from stationary combustion turbines that commenced construction, modification or reconstruction after February 18, 2005.

APPLICABILITY

§60.4305 Does this subpart apply to my stationary combustion turbine?

(a) If you are the owner or operator of a stationary combustion turbine with a heat input at peak load equal to or greater than 10.7 gigajoules (10 MMBtu) per hour, based on the higher heating value of the fuel, which commenced construction, modification, or reconstruction after February 18, 2005, your turbine is subject to this subpart. Only heat input to the combustion turbine should be included when determining whether or not this subpart is applicable to your turbine. Any additional heat input to associated heat recovery steam generators (HRSG) or duct burners should not be included when determining your peak heat input. However, this subpart does apply to emissions from any associated HRSG and duct burners.

(b) Stationary combustion turbines regulated under this subpart are exempt from the requirements of subpart GG of this part. Heat recovery steam generators and duct burners regulated under this subpart are exempted from the requirements of subparts Da, Db, and Dc of this part.

§60.4310 What types of operations are exempt from these standards of performance?

(a) Emergency combustion turbines, as defined in 60.4420(i), are exempt from the nitrogen oxides (NO_X) emission limits in 60.4320.

(b) Stationary combustion turbines engaged by manufacturers in research and development of equipment for both combustion turbine emission control techniques and combustion turbine efficiency improvements are exempt from the NO_X emission limits in Appendix R

AUTHENTICATED U.S. GOVERNMENT INFORMATION

> (c) The reports required under paragraph (b) shall be postmarked within 30 days following the end of the second and fourth calendar quarters.

> (d) The requirements of this subsection remain in force until and unless the Agency, in delegating enforcement authority to a State under section 111(c) of the Act, approves reporting requirements or an alternative means of compliance surveillance adopted by such States. In that event, affected sources within the State will be relieved of the obligation to comply with this subsection, provided that they comply with requirements established by the State.

Subpart GGG—Standards of Performance for Equipment Leaks of VOC in Petroleum Refineries for which Construction, Reconstruction, or Modification Commenced After January 4, 1983, and on or Before November 7, 2006

SOURCE: 49 FR 22606, May 30, 1984, unless otherwise noted.

§60.590 Applicability and designation of affected facility.

(a)(1) The provisions of this subpart apply to affected facilities in petroleum refineries.

(2) A compressor is an affected facility.

(3) The group of all the equipment (defined in §60.591) within a process unit is an affected facility.

(b) Any affected facility under paragraph (a) of this section that commences construction, reconstruction, or modification after January 4, 1983, and on or before November 7, 2006, is subject to the requirements of this subpart.

(c) Addition or replacement of equipment (defined in §60.591) for the purpose of process improvement which is accomplished without a capital expenditure shall not by itself be considered a modification under this subpart.

(d) Facilities subject to subpart VV, subpart VVa, or subpart KKK of this part are excluded from this subpart.

(e) Stay of standards. Owners or operators are not required to comply with the definition of "process unit" in §60.590 of this subpart until the EPA takes final action to require compliance and publishes a document in the FEDERAL REGISTER. While the definition of "process unit" is stayed, owners or operators should use the following definition:

Process unit means components assembled to produce intermediate or final products from petroleum, unfinished petroleum derivatives, or other intermediates; a process unit can operate independently if supplied with sufficient feed or raw materials and sufficient storage facilities for the product.

[49 FR 22606, May 30, 1984, as amended at 72 FR 64895, Nov. 16, 2007; 73 FR 31376, June 2, 2008]

§ 60.591 Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the act, in subpart A of part 60, or in subpart VV of part 60, and the following terms shall have the specific meanings given them.

Alaskan North Slope means the approximately 69,000 square mile area extending from the Brooks Range to the Arctic Ocean.

Asphalt (also known as Bitumen) is a black or dark brown solid or semi-solid thermo-plastic material possessing waterproofing and adhesive properties. It is a complex combination of higher molecular weight organic compounds containing a relatively high proportion of hydrocarbons having carbon numbers greater than C25 with a high carbon to hydrogen ratio. It is essentially nonvolatile at ambient temperatures with closed cup flash point of 445 °F (230 °C) or greater.

Equipment means each valve, pump, pressure relief device, sampling connection system, open-ended valve or line, and flange or other connector in VOC service. For the purposes of recordkeeping and reporting only, compressors are considered equipment.

In hydrogen service means that a compressor contains a process fluid that meets the conditions specified in §60.593(b).

In light liquid service means that the piece of equipment contains a liquid that meets the conditions specified in §60.593(c).

Petroleum means the crude oil removed from the earth and the oils derived from tar sands, shale, and coal.

Petroleum refinery means any facility engaged in producing gasoline, kerosene, distillate fuel oils, residual fuel oils, lubricants, or other products through the distillation of petroleum, or through the redistillation, cracking, or reforming of unfinished petroleum derivatives.

Process unit means the components assembled and connected by pipes or ducts to process raw materials and to produce intermediate or final products from petroleum, unfinished petroleum derivatives, or other intermediates. A process unit can operate independently if supplied with sufficient feed or raw materials and sufficient storage facilities for the product. For the purpose of this subpart, process unit includes any feed, intermediate and final product storage vessels (except as specified in §60.482-1(g)), product transfer racks, and connected ducts and piping. A process unit includes all equipment as defined in this subpart.

[49 FR 22606, May 30, 1984, as amended at 72 FR 64895, Nov. 16, 2007]

EFFECTIVE DATE NOTE: At 73 FR 31376, June 2, 2008, §60.591, the definition of "process unit" was stayed until further notice.

§60.592 Standards.

(a) Each owner or operator subject to the provisions of this subpart shall comply with the requirements of §§ 60.482-1 to 60.482-10 as soon as practicable, but no later than 180 days after initial startup.

(b) For a given process unit, an owner or operator may elect to comply with the requirements of paragraphs (b)(1), (2), or (3) of this section as an alternative to the requirements in $\S60.482-7$.

(1) Comply with §60.483–1.

(2) Comply with §60.483–2.

(3) Comply with the Phase III provisions in 40 CFR 63.168, except an owner or operator may elect to follow the provisions in §60.482–7(f) instead of 40 CFR 63.168 for any valve that is designated as being leakless.

(c) An owner or operator may apply to the Administrator for a determination of equivalency for any means of emission limitation that achieves a reduction in emissions of VOC at least 40 CFR Ch. I (7–1–23 Edition)

equivalent to the reduction in emissions of VOC achieved by the controls required in this subpart. In doing so, the owner or operator shall comply with requirements of §60.484.

(d) Each owner or operator subject to the provisions of this subpart shall comply with the provisions of §60.485 except as provided in §60.593.

(e) Each owner or operator subject to the provisions of this subpart shall comply with the provisions of \$ 60.486 and 60.487.

[49 FR 22606, May 30, 1984, as amended at 72 FR 64896, Nov. 16, 2007]

§60.593 Exceptions.

(a) Each owner or operator subject to the provisions of this subpart may comply with the following exceptions to the provisions of subpart VV.

(b)(1) Compressors in hydrogen service are exempt from the requirements of 60.592 if an owner or operator demonstrates that a compressor is in hydrogen service.

(2) Each compressor is presumed not to be in hydrogen service unless an owner or operator demonstrates that the piece of equipment is in hydrogen service. For a piece of equipment to be considered in hydrogen service, it must be determined that the percent hydrogen content can be reasonably expected always to exceed 50 percent by volume. For purposes of determining the percent hydrogen content in the process fluid that is contained in or contacts a compressor, procedures that conform to the general method described in ASTM E260-73, 91, or 96, E168-67, 77, or 92, or E169-63, 77, or 93 (incorporated by reference as specified in §60.17) shall be used.

(3)(i) An owner or operator may use engineering judgment rather than procedures in paragraph (b)(2) of this section to demonstrate that the percent content exceeds 50 percent by volume, provided the engineering judgment demonstrates that the content clearly exceeds 50 percent by volume. When an owner or operator and the Administrator do not agree on whether a piece of equipment is in hydrogen service, however, the procedures in paragraph (b)(2) shall be used to resolve the disagreement.

(ii) If an owner or operator determines that a piece of equipment is in hydrogen service, the determination can be revised only after following the procedures in paragraph (b)(2).

(c) Any existing reciprocating compressor that becomes an affected facility under provisions of 60.14 or 60.15is exempt from 60.482-3(a), (b), (c), (d), (e), and (h) provided the owner or operator demonstrates that recasting the distance piece or replacing the compressor are the only options available to bring the compressor into compliance with the provisions of 60.482-3(a), (b), (c), (d), (e), and (h).

(d) An owner or operator may use the following provision in addition to §60.485(e): Equipment is in light liquid service if the percent evaporated is greater than 10 percent at 150 °C as determined by ASTM Method D86-78, 82, 90, 95, or 96 (incorporated by reference as specified in §60.17).

(e) Pumps in light liquid service and valves in gas/vapor and light liquid service within a process unit that is located in the Alaskan North Slope are exempt from the requirements of §§ 60.482–2 and 60.482–7.

(f) Open-ended values or lines containing asphalt as defined in 60.591 are exempt from the requirements of 60.482-6(a) through (c).

[49 FR 22606, May 30, 1984, as amended at 65 FR 61768, Oct. 17, 2000; 72 FR 64896, Nov. 16, 2007]

Subpart GGGa—Standards of Performance for Equipment Leaks of VOC in Petroleum Refineries for Which Construction, Reconstruction, or Modification Commenced After November 7, 2006

SOURCE: $72\,$ FR 64896, Nov. 16, 2007, unless otherwise noted.

§60.590a Applicability and designation of affected facility.

(a)(1) The provisions of this subpart apply to affected facilities in petro-leum refineries.

(2) A compressor is an affected facility.

(3) The group of all the equipment (defined in 60.591a) within a process unit is an affected facility.

(b) Any affected facility under paragraph (a) of this section that commences construction, reconstruction, or modification after November 7, 2006, is subject to the requirements of this subpart.

(c) Addition or replacement of equipment (defined in §60.591a) for the purpose of process improvement which is accomplished without a capital expenditure shall not by itself be considered a modification under this subpart.

(d) Facilities subject to subpart VV, subpart VVa, subpart GGG, or subpart KKK of this part are excluded from this subpart.

(e) Stay of standards. Owners or operators are not required to comply with the definition of "process unit" in §60.590 of this subpart until the EPA takes final action to require compliance and publishes a document in the FEDERAL REGISTER. While the definition of "process unit" is stayed, owners or operators should use the following definition:

Process unit means components assembled to produce intermediate or final products from petroleum, unfinished petroleum derivatives, or other intermediates; a process unit can operate independently if supplied with sufficient feed or raw materials and sufficient storage facilities for the product.

[49 FR 22606, May 30, 1984, as amended at 73 FR 31376, June 2, 2008]

§60.591a Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Clean Air Act, in subpart A of part 60, or in subpart VVa of this part, and the following terms shall have the specific meanings given them.

Alaskan North Slope means the approximately 69,000 square mile area extending from the Brooks Range to the Arctic Ocean.

Asphalt (also known as Bitumen) is a black or dark brown solid or semi-solid thermo-plastic material possessing waterproofing and adhesive properties. It is a complex combination of higher molecular weight organic compounds containing a relatively high proportion of Appendix S

§60.40b

AUTHENTICATED U.S. GOVERNMENT INFORMATION

> (iii) Copies of all visible emission observer opacity field data sheets;

(2) For each performance test conducted using Method 22 of appendix A-4 of this part, the owner or operator shall keep the records including the information specified in paragraphs (b)(2)(i) through (iv) of this section.

(i) Dates and time intervals of all visible emissions observation periods;

(ii) Name and affiliation for each visible emission observer participating in the performance test;

(iii) Copies of all visible emission observer opacity field data sheets; and

(iv) Documentation of any adjustments made and the time the adjustments were completed to the affected facility operation by the owner or operator to demonstrate compliance with the applicable monitoring requirements.

(3) For each digital opacity compliance system, the owner or operator shall maintain records and submit reports according to the requirements specified in the site-specific monitoring plan approved by the Administrator.

[74 FR 5083, Jan. 28, 2009, as amended at 77 FR 9459, Feb. 16, 2012]

Subpart Db—Standards of Performance for Industrial-Commercial-Institutional Steam Generating Units

SOURCE: 72 FR 32742, June 13, 2007, unless otherwise noted.

§60.40b Applicability and delegation of authority.

(a) The affected facility to which this subpart applies is each steam generating unit that commences construction, modification, or reconstruction after June 19, 1984, and that has a heat input capacity from fuels combusted in the steam generating unit of greater than 29 megawatts (MW) (100 million British thermal units per hour (MMBtu/hr)).

(b) Any affected facility meeting the applicability requirements under paragraph (a) of this section and commencing construction, modification, or reconstruction after June 19, 1984, but

40 CFR Ch. I (7–1–23 Edition)

on or before June 19, 1986, is subject to the following standards:

(1) Coal-fired affected facilities having a heat input capacity between 29 and 73 MW (100 and 250 MMBtu/hr), inclusive, are subject to the particulate matter (PM) and nitrogen oxides (NO_X) standards under this subpart.

(2) Coal-fired affected facilities having a heat input capacity greater than 73 MW (250 MMBtu/hr) and meeting the applicability requirements under subpart D (Standards of performance for fossil-fuel-fired steam generators; $\S60.40$) are subject to the PM and NO_X standards under this subpart and to the sulfur dioxide (SO₂) standards under subpart D ($\S60.43$).

(3) Oil-fired affected facilities having a heat input capacity between 29 and 73 MW (100 and 250 MMBtu/hr), inclusive, are subject to the NO_X standards under this subpart.

(4) Oil-fired affected facilities having a heat input capacity greater than 73 MW (250 MMBtu/hr) and meeting the applicability requirements under subpart D (Standards of performance for fossil-fuel-fired steam generators; $\S60.40$) are also subject to the NO_X standards under this subpart and the PM and SO₂ standards under subpart D (\S 60.42 and 60.43).

(c) Affected facilities that also meet the applicability requirements under subpart J or subpart Ja of this part are subject to the PM and NO_X standards under this subpart and the SO_2 standards under subpart J or subpart Ja of this part, as applicable.

(d) Affected facilities that also meet the applicability requirements under subpart E (Standards of performance for incinerators; $\S60.50$) are subject to the NO_X and PM standards under this subpart.

(e) Steam generating units meeting the applicability requirements under subpart Da (Standards of performance for electric utility steam generating units; \S 60.40Da) are not subject to this subpart.

(f) Any change to an existing steam generating unit for the sole purpose of combusting gases containing total reduced sulfur (TRS) as defined under 60.281 is not considered a modification under 60.14 and the steam generating unit is not subject to this subpart.

(g) In delegating implementation and enforcement authority to a State under section 111(c) of the Clean Air Act, the following authorities shall be retained by the Administrator and not transferred to a State.

(1) Section 60.44b(f).

(2) Section 60.44b(g).

(3) Section 60.49b(a)(4).

(h) Any affected facility that meets the applicability requirements and is subject to subpart Ea, subpart Eb, subpart AAAA, or subpart CCCC of this

part is not subject to this subpart. (i) Affected facilities (i.e., heat recovery steam generators) that are associated with stationary combustion turbines and that meet the applicability requirements of subpart KKKK of this part are not subject to this subpart. This subpart will continue to apply to all other affected facilities (i.e. heat recovery steam generators with duct burners) that are capable of combusting more than 29 MW (100 MMBtu/ h) heat input of fossil fuel. If the affected facility (*i.e.* heat recovery steam generator) is subject to this subpart, only emissions resulting from combustion of fuels in the steam generating unit are subject to this subpart. (The stationary combustion turbine emissions are subject to subpart GG or KKKK, as applicable, of this part.)

(j) Any affected facility meeting the applicability requirements under paragraph (a) of this section and commencing construction, modification, or reconstruction after June 19, 1986 is not subject to subpart D (Standards of Performance for Fossil-Fuel-Fired Steam Generators, §60.40).

(k) Any affected facility that meets the applicability requirements and is subject to an EPA approved State or Federal section 111(d)/129 plan implementing subpart Cb or subpart BBBB of this part is not covered by this subpart.

(1) Affected facilities that also meet the applicability requirements under subpart BB of this part (Standards of Performance for Kraft Pulp Mills) are subject to the SO_2 and NO_X standards under this subpart and the PM standards under subpart BB. (m) Temporary boilers are not subject to this subpart.

[72 FR 32742, June 13, 2007, as amended at 74 FR 5084, Jan. 28, 2009; 77 FR 9459, Feb. 16, 2012]

§60.41b Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Clean Air Act and in subpart A of this part.

Annual capacity factor means the ratio between the actual heat input to a steam generating unit from the fuels listed in 60.42b(a), 60.43b(a), or 60.44b(a), as applicable, during a calendar year and the potential heat input to the steam generating unit had it been operated for 8,760 hours during a calendar year at the maximum steady state design heat input capacity. In the case of steam generating units that are rented or leased, the actual heat input shall be determined based on the combined heat input from all operations of the affected facility in a calendar year.

Byproduct/waste means any liquid or gaseous substance produced at chemical manufacturing plants, petroleum refineries, or pulp and paper mills (except natural gas, distillate oil, or residual oil) and combusted in a steam generating unit for heat recovery or for disposal. Gaseous substances with carbon dioxide (CO_2) levels greater than 50 percent or carbon monoxide levels greater than 10 percent are not byproduct/waste for the purpose of this subpart.

Chemical manufacturing plants mean industrial plants that are classified by the Department of Commerce under Standard Industrial Classification (SIC) Code 28.

Coal means all solid fuels classified as anthracite, bituminous, subbituminous, or lignite by the American Society of Testing and Materials in ASTM D388 (incorporated by reference, see § 60.17), coal refuse, and petroleum coke. Coal-derived synthetic fuels, including but not limited to solvent refined coal, gasified coal not meeting the definition of natural gas, coal-oil mixtures, coke oven gas, and coalwater mixtures, are also included in this definition for the purposes of this subpart. *Coal refuse* means any byproduct of coal mining or coal cleaning operations with an ash content greater than 50 percent, by weight, and a heating value less than 13,900 kJ/kg (6,000 Btu/lb) on a dry basis.

Cogeneration, also known as combined heat and power, means a facility that simultaneously produces both electric (or mechanical) and useful thermal energy from the same primary energy source.

Coke oven gas means the volatile constituents generated in the gaseous exhaust during the carbonization of bituminous coal to form coke.

Combined cycle system means a system in which a separate source, such as a gas turbine, internal combustion engine, kiln, etc., provides exhaust gas to a steam generating unit.

Conventional technology means wet flue gas desulfurization (FGD) technology, dry FGD technology, atmospheric fluidized bed combustion technology, and oil hydrodesulfurization technology.

Distillate oil means fuel oils that contain 0.05 weight percent nitrogen or less and comply with the specifications for fuel oil numbers 1 and 2, as defined by the American Society of Testing and Materials in ASTM D396 (incorporated by reference, see §60.17), diesel fuel oil numbers 1 and 2, as defined by the American Society for Testing and Materials in ASTM D975 (incorporated by reference, see §60.17), kerosine, as defined by the American Society of Testing and Materials in ASTM D3699 (incorporated by reference, see §60.17), biodiesel as defined by the American Society of Testing and Materials in ASTM D6751 (incorporated by reference, see §60.17), or biodiesel blends as defined by the American Society of Testing and Materials in ASTM D7467 (incorporated by reference, see §60.17).

Dry flue gas desulfurization technology means a SO_2 control system that is located downstream of the steam generating unit and removes sulfur oxides from the combustion gases of the steam generating unit by contacting the combustion gases with an alkaline reagent and water, whether introduced separately or as a premixed slurry or solution and forming a dry powder material. This definition includes devices 40 CFR Ch. I (7–1–23 Edition)

where the dry powder material is subsequently converted to another form. Alkaline slurries or solutions used in dry flue gas desulfurization technology include but are not limited to lime and sodium.

Duct burner means a device that combusts fuel and that is placed in the exhaust duct from another source, such as a stationary gas turbine, internal combustion engine, kiln, etc., to allow the firing of additional fuel to heat the exhaust gases before the exhaust gases enter a steam generating unit.

Emerging technology means any SO_2 control system that is not defined as a conventional technology under this section, and for which the owner or operator of the facility has applied to the Administrator and received approval to operate as an emerging technology under $\S 60.49b(a)(4)$.

Federally enforceable means all limitations and conditions that are enforceable by the Administrator, including the requirements of 40 CFR parts 60 and 61, requirements within any applicable State Implementation Plan, and any permit requirements established under 40 CFR 52.21 or under 40 CFR 51.18 and 51.24.

Fluidized bed combustion technology means combustion of fuel in a bed or series of beds (including but not limited to bubbling bed units and circulating bed units) of limestone aggregate (or other sorbent materials) in which these materials are forced upward by the flow of combustion air and the gaseous products of combustion.

Fuel pretreatment means a process that removes a portion of the sulfur in a fuel before combustion of the fuel in a steam generating unit.

Full capacity means operation of the steam generating unit at 90 percent or more of the maximum steady-state design heat input capacity.

Gaseous fuel means any fuel that is a gas at ISO conditions. This includes, but is not limited to, natural gas and gasified coal (including coke oven gas).

Gross output means the gross useful work performed by the steam generated. For units generating only electricity, the gross useful work performed is the gross electrical output from the turbine/generator set. For cogeneration units, the gross useful work

performed is the gross electrical or mechanical output plus 75 percent of the useful thermal output measured relative to ISO conditions that is not used to generate additional electrical or mechanical output or to enhance the performance of the unit (*i.e.*, steam delivered to an industrial process).

Heat input means heat derived from combustion of fuel in a steam generating unit and does not include the heat derived from preheated combustion air, recirculated flue gases, or exhaust gases from other sources, such as gas turbines, internal combustion engines, kilns, etc.

Heat release rate means the steam generating unit design heat input capacity (in MW or Btu/hr) divided by the furnace volume (in cubic meters or cubic feet); the furnace volume is that volume bounded by the front furnace wall where the burner is located, the furnace side waterwall, and extending to the level just below or in front of the first row of convection pass tubes.

Heat transfer medium means any material that is used to transfer heat from one point to another point.

High heat release rate means a heat release rate greater than 730,000 J/sec-m³ (70,000 Btu/hr-ft³).

ISO Conditions means a temperature of 288 Kelvin, a relative humidity of 60 percent, and a pressure of 101.3 kilopascals.

Lignite means a type of coal classified as lignite A or lignite B by the American Society of Testing and Materials in ASTM D388 (incorporated by reference, see §60.17).

Low heat release rate means a heat release rate of 730,000 J/sec-m³ (70,000 Btu/ hr-ft³) or less.

Mass-feed stoker steam generating unit means a steam generating unit where solid fuel is introduced directly into a retort or is fed directly onto a grate where it is combusted.

Maximum heat input capacity means the ability of a steam generating unit to combust a stated maximum amount of fuel on a steady state basis, as determined by the physical design and characteristics of the steam generating unit.

Municipal-type solid waste means refuse, more than 50 percent of which is waste consisting of a mixture of paper, wood, yard wastes, food wastes, plastics, leather, rubber, and other combustible materials, and noncombustible materials such as glass and rock.

Natural gas means:

(1) A naturally occurring mixture of hydrocarbon and nonhydrocarbon gases found in geologic formations beneath the earth's surface, of which the principal constituent is methane; or

(2) Liquefied petroleum gas, as defined by the American Society for Testing and Materials in ASTM D1835 (incorporated by reference, see §60.17); or

(3) A mixture of hydrocarbons that maintains a gaseous state at ISO conditions. Additionally, natural gas must either be composed of at least 70 percent methane by volume or have a gross calorific value between 34 and 43 megajoules (MJ) per dry standard cubic meter (910 and 1,150 Btu per dry standard cubic foot).

Noncontinental area means the State of Hawaii, the Virgin Islands, Guam, American Samoa, the Commonwealth of Puerto Rico, or the Northern Mariana Islands.

Oil means crude oil or petroleum or a liquid fuel derived from crude oil or petroleum, including distillate and residual oil.

Petroleum refinery means industrial plants as classified by the Department of Commerce under Standard Industrial Classification (SIC) Code 29.

Potential sulfur dioxide emission rate means the theoretical SO_2 emissions (nanograms per joule (ng/J) or lb/ MMBtu heat input) that would result from combusting fuel in an uncleaned state and without using emission control systems. For gasified coal or oil that is desulfurized prior to combustion, the Potential sulfur dioxide emission rate is the theoretical SO_2 emissions (ng/J) or lb/MMBtu heat input) that would result from combusting fuel in a cleaned state without using any post combustion emission control systems.

Process heater means a device that is primarily used to heat a material to initiate or promote a chemical reaction in which the material participates as a reactant or catalyst.

40 CFR Ch. I (7-1-23 Edition)

Pulp and paper mills means industrial plants that are classified by the Department of Commerce under North American Industry Classification System (NAICS) Code 322 or Standard Industrial Classification (SIC) Code 26.

Pulverized coal-fired steam generating *unit* means a steam generating unit in which pulverized coal is introduced into an air stream that carries the coal to the combustion chamber of the steam generating unit where it is fired in suspension. This includes both conventional pulverized coal-fired and micropulverized coal-fired steam generating units. Residual oil means crude oil, fuel oil numbers 1 and 2 that have a nitrogen content greater than $0.05\,$ weight percent, and all fuel oil numbers 4, 5 and 6, as defined by the American Society of Testing and Materials in ASTM D396 (incorporated by reference, see §60.17).

Spreader stoker steam generating unit means a steam generating unit in which solid fuel is introduced to the combustion zone by a mechanism that throws the fuel onto a grate from above. Combustion takes place both in suspension and on the grate.

Steam generating unit means a device that combusts any fuel or byproduct/ waste and produces steam or heats water or heats any heat transfer medium. This term includes any municipal-type solid waste incinerator with a heat recovery steam generating unit or any steam generating unit that combusts fuel and is part of a cogeneration system or a combined cycle system. This term does not include process heaters as they are defined in this subpart.

Steam generating unit operating day means a 24-hour period between 12:00 midnight and the following midnight during which any fuel is combusted at any time in the steam generating unit. It is not necessary for fuel to be combusted continuously for the entire 24hour period.

Temporary boiler means any gaseous or liquid fuel-fired steam generating unit that is designed to, and is capable of, being carried or moved from one location to another by means of, for example, wheels, skids, carrying handles, dollies, trailers, or platforms. A steam generating unit is not a temporary boiler if any one of the following conditions exists:

(1) The equipment is attached to a foundation.

(2) The steam generating unit or a replacement remains at a location for more than 180 consecutive days. Any temporary boiler that replaces a temporary boiler at a location and performs the same or similar function will be included in calculating the consecutive time period.

(3) The equipment is located at a seasonal facility and operates during the full annual operating period of the seasonal facility, remains at the facility for at least 2 years, and operates at that facility for at least 3 months each year.

(4) The equipment is moved from one location to another in an attempt to circumvent the residence time requirements of this definition.

Very low sulfur oil means for units constructed, reconstructed, or modified on or before February 28, 2005, oil that contains no more than 0.5 weight percent sulfur or that, when combusted without SO₂ emission control, has a SO_2 emission rate equal to or less than 215 ng/J (0.5 lb/MMBtu) heat input. For units constructed, reconstructed, or modified after February 28, 2005 and not located in a noncontinental area, very low sulfur oil means oil that contains no more than 0.30 weight percent sulfur or that, when combusted without SO₂ emission control, has a SO₂ emission rate equal to or less than 140 ng/J (0.32 lb/MMBtu) heat input. For units constructed, reconstructed, or modified after February 28, 2005 and located in a noncontinental area. very low sulfur oil means oil that contains no more than 0.5 weight percent sulfur or that, when combusted without SO₂ emission control, has a SO_2 emission rate equal to or less than 215 ng/J (0.50 lb/MMBtu) heat input.

Wet flue gas desulfurization technology means a SO_2 control system that is located downstream of the steam generating unit and removes sulfur oxides from the combustion gases of the steam generating unit by contacting the combustion gas with an alkaline slurry or solution and forming a liquid

material. This definition applies to devices where the aqueous liquid material product of this contact is subsequently converted to other forms. Alkaline reagents used in wet flue gas desulfurization technology include, but are not limited to, lime, limestone, and sodium.

Wet scrubber system means any emission control device that mixes an aqueous stream or slurry with the exhaust gases from a steam generating unit to control emissions of PM or SO₂.

Wood means wood, wood residue, bark, or any derivative fuel or residue thereof, in any form, including, but not limited to, sawdust, sanderdust, wood chips, scraps, slabs, millings, shavings, and processed pellets made from wood or other forest residues.

[72 FR 32742, June 13, 2007, as amended at 74 FR 5084, Jan. 28, 2009; 77 FR 9459, Feb. 16, 2012]

60.42b Standard for sulfur dioxide (SO₂).

(a) Except as provided in paragraphs (b), (c), (d), or (j) of this section, on and after the date on which the performance test is completed or required to be completed under §60.8, whichever comes first, no owner or operator of an affected facility that commenced construction, reconstruction, or modification on or before February 28, 2005, that combusts coal or oil shall cause to be discharged into the atmosphere any gases that contain SO_2 in excess of 87 ng/J (0.20 lb/MMBtu) or 10 percent (0.10) of the potential SO_2 emission rate (90) percent reduction) and the emission limit determined according to the following formula:

$$\mathbf{E}_{s} = \frac{\left(\mathbf{K}_{a}\mathbf{H}_{a} + \mathbf{K}_{b}\mathbf{H}_{b}\right)}{\left(\mathbf{H}_{a} + \mathbf{H}_{b}\right)}$$

Where:

 $E_{\rm s}$ = SO_2 emission limit, in ng/J or lb/MMBtu heat input;

 $K_a = 520 \text{ ng/J}$ (or 1.2 lb/MMBtu);

 $K_b = 340 \text{ ng/J} \text{ (or } 0.80 \text{ lb/MMBtu)};$

$$\label{eq:hamiltonian} \begin{split} H_a &= Heat \text{ input from the combustion of coal,} \\ &\text{ in } J \ (MMBtu); \text{ and} \end{split}$$

 H_{b} = Heat input from the combustion of oil, in J (MMBtu).

For facilities complying with the percent reduction standard, only the heat input supplied to the affected facility from the combustion of coal and oil is counted in this paragraph. No credit is provided for the heat input to the affected facility from the combustion of natural gas, wood, municipaltype solid waste, or other fuels or heat derived from exhaust gases from other sources, such as gas turbines, internal combustion engines, kilns, etc.

(b) On and after the date on which the performance test is completed or required to be completed under §60.8, whichever date comes first, no owner or operator of an affected facility that commenced construction, reconstruction, or modification on or before February 28, 2005, that combusts coal refuse alone in a fluidized bed combustion steam generating unit shall cause to be discharged into the atmosphere any gases that contain SO_2 in excess of 87 ng/J (0.20 lb/MMBtu) or 20 percent (0.20) of the potential SO₂ emission rate (80 percent reduction) and 520 ng/J (1.2 lb/MMBtu) heat input. If coal or oil is fired with coal refuse, the affected facility is subject to paragraph (a) or (d) of this section, as applicable. For facilities complying with the percent reduction standard, only the heat input supplied to the affected facility from the combustion of coal and oil is counted in this paragraph. No credit is provided for the heat input to the affected facility from the combustion of natural gas, wood, municipal-type solid waste, or other fuels or heat derived from exhaust gases from other sources, such as gas turbines, internal combustion engines, kilns, etc.

(c) On and after the date on which the performance test is completed or is required to be completed under §60.8, whichever comes first, no owner or operator of an affected facility that combusts coal or oil, either alone or in combination with any other fuel, and that uses an emerging technology for the control of SO₂ emissions, shall cause to be discharged into the atmosphere any gases that contain SO_2 in excess of 50 percent of the potential SO_2 emission rate (50 percent reduction) and that contain SO_2 in excess of the emission limit determined according to the following formula:

211

§60.42b

$$\mathbf{E}_{s} = \frac{\left(\mathbf{K}_{c}\mathbf{H}_{c} + \mathbf{K}_{d}\mathbf{H}_{d}\right)}{\left(\mathbf{H}_{c} + \mathbf{H}_{d}\right)}$$

Where:

 $E_{\rm s}$ = SO2 emission limit, in ng/J or lb/MM Btu heat input;

 $K_c = 260 \text{ ng/J} \text{ (or } 0.60 \text{ lb/MMBtu)};$

 $K_d = 170 \text{ ng/J} \text{ (or } 0.40 \text{ lb/MMBtu});$

 $H_{\rm c}$ = Heat input from the combustion of coal, in J (MMBtu); and

 H_d = Heat input from the combustion of oil, in J (MMBtu).

For facilities complying with the percent reduction standard, only the heat input supplied to the affected facility from the combustion of coal and oil is counted in this paragraph. No credit is provided for the heat input to the affected facility from the combustion of natural gas, wood, municipaltype solid waste, or other fuels, or from the heat input derived from exhaust gases from other sources, such as gas turbines, internal combustion engines, kilns, etc.

(d) On and after the date on which the performance test is completed or required to be completed under §60.8, whichever comes first, no owner or operator of an affected facility that commenced construction, reconstruction, or modification on or before February 28, 2005 and listed in paragraphs (d)(1), (2), (3), or (4) of this section shall cause to be discharged into the atmosphere any gases that contain SO_2 in excess of 520 ng/J (1.2 lb/MMBtu) heat input if the affected facility combusts coal, or 215 ng/J (0.5 lb/MMBtu) heat input if the affected facility combusts oil other than very low sulfur oil. Percent reduction requirements are not applicable to affected facilities under paragraphs (d)(1), (2), (3) or (4) of this section. For facilities complying with paragraphs (d)(1), (2), or (3) of this section, only the heat input supplied to the affected facility from the combustion of coal and oil is counted in this paragraph. No credit is provided for the heat input to the affected facility from the combustion of natural gas, wood, municipaltype solid waste, or other fuels or heat derived from exhaust gases from other sources, such as gas turbines, internal combustion engines, kilns, etc.

(1) Affected facilities that have an annual capacity factor for coal and oil

40 CFR Ch. I (7–1–23 Edition)

of 30 percent (0.30) or less and are subject to a federally enforceable permit limiting the operation of the affected facility to an annual capacity factor for coal and oil of 30 percent (0.30) or less;

(2) Affected facilities located in a noncontinental area; or

(3) Affected facilities combusting coal or oil, alone or in combination with any fuel, in a duct burner as part of a combined cycle system where 30 percent (0.30) or less of the heat entering the steam generating unit is from combustion of coal and oil in the duct burner and 70 percent (0.70) or more of the heat entering the steam generating unit is from the exhaust gases entering the duct burner; or

(4) The affected facility burns coke oven gas alone or in combination with natural gas or very low sulfur distillate oil.

(e) Except as provided in paragraph (f) of this section, compliance with the emission limits, fuel oil sulfur limits, and/or percent reduction requirements under this section are determined on a 30-day rolling average basis.

(f) Except as provided in paragraph (j)(2) of this section, compliance with the emission limits or fuel oil sulfur limits under this section is determined on a 24-hour average basis for affected facilities that (1) have a federally enforceable permit limiting the annual capacity factor for oil to 10 percent or less, (2) combust only very low sulfur oil, and (3) do not combust any other fuel.

(g) Except as provided in paragraph (i) of this section and 60.45b(a), the SO_2 emission limits and percent reduction requirements under this section apply at all times, including periods of startup, shutdown, and malfunction.

(h) Reductions in the potential SO_2 emission rate through fuel pretreatment are not credited toward the percent reduction requirement under paragraph (c) of this section unless:

(1) Fuel pretreatment results in a 50 percent or greater reduction in potential SO_2 emissions and

(2) Emissions from the pretreated fuel (without combustion or post-combustion SO_2 control) are equal to or

less than the emission limits specified in paragraph (c) of this section.

(i) An affected facility subject to paragraph (a), (b), or (c) of this section may combust very low sulfur oil or natural gas when the SO_2 control system is not being operated because of malfunction or maintenance of the SO_2 control system.

(j) Percent reduction requirements are not applicable to affected facilities combusting only very low sulfur oil. The owner or operator of an affected facility combusting very low sulfur oil shall demonstrate that the oil meets the definition of very low sulfur oil by: (1) Following the performance testing procedures as described in $\S60.45b(c)$ or $\S60.45b(d)$, and following the monitoring procedures as described in $\S60.47b(a)$ or $\S60.47b(b)$ to determine SO₂ emission rate or fuel oil sulfur content; or (2) maintaining fuel records as described in $\S60.49b(r)$.

(k)(1) Except as provided in paragraphs (k)(2), (k)(3), and (k)(4) of this section, on and after the date on which the initial performance test is completed or is required to be completed under §60.8, whichever date comes first, no owner or operator of an affected facility that commences construction, reconstruction, or modification after February 28, 2005, and that combusts coal, oil, natural gas, a mixture of these fuels, or a mixture of these fuels with any other fuels shall cause to be discharged into the atmosphere any gases that contain SO_2 in excess of 87 ng/J (0.20 lb/MMBtu) heat input or 8 percent (0.08) of the potential SO₂ emission rate (92 percent reduction) and $520\,$ ng/J (1.2 lb/MMBtu) heat input. For facilities complying with the percent reduction standard and paragraph (k)(3)of this section, only the heat input supplied to the affected facility from the combustion of coal and oil is counted in paragraph (k) of this section. No credit is provided for the heat input to the affected facility from the combustion of natural gas, wood, municipaltype solid waste, or other fuels or heat derived from exhaust gases from other sources, such as gas turbines, internal combustion engines, kilns, etc.

(2) Units firing only very low sulfur oil, gaseous fuel, a mixture of these fuels, or a mixture of these fuels with any other fuels with a potential SO_2 emission rate of 140 ng/J (0.32 lb/ MMBtu) heat input or less are exempt from the SO_2 emissions limit in paragraph (k)(1) of this section.

(3) Units that are located in a noncontinental area and that combust coal, oil, or natural gas shall not discharge any gases that contain SO₂ in excess of 520 ng/J (1.2 lb/MMBtu) heat input if the affected facility combusts coal, or 215 ng/J (0.50 lb/MMBtu) heat input if the affected facility combusts oil or natural gas.

(4) As an alternative to meeting the requirements under paragraph (k)(1) of this section, modified facilities that combust coal or a mixture of coal with other fuels shall not cause to be discharged into the atmosphere any gases that contain SO₂ in excess of 87 ng/J (0.20 lb/MMBtu) heat input or 10 percent (0.10) of the potential SO₂ emission rate (90 percent reduction) and 520 ng/J (1.2 lb/MMBtu) heat input.

[72 FR 32742, June 13, 2007, as amended at 74 FR 5084, Jan. 28, 2009; 76 FR 3523, Jan. 20, 2011]

§60.43b Standard for particulate matter (PM).

(a) On and after the date on which the initial performance test is completed or is required to be completed under §60.8, whichever comes first, no owner or operator of an affected facility that commenced construction, reconstruction, or modification on or before February 28, 2005 that combusts coal or combusts mixtures of coal with other fuels, shall cause to be discharged into the atmosphere from that affected facility any gases that contain PM in excess of the following emission limits:

(1) 22 ng/J (0.051 lb/MMBtu) heat input, (i) If the affected facility combusts only coal, or

(ii) If the affected facility combusts coal and other fuels and has an annual capacity factor for the other fuels of 10 percent (0.10) or less.

(2) 43 ng/J (0.10 lb/MMBtu) heat input if the affected facility combusts coal and other fuels and has an annual capacity factor for the other fuels greater than 10 percent (0.10) and is subject to a federally enforceable requirement limiting operation of the affected facility to an annual capacity factor greater than 10 percent (0.10) for fuels other than coal.

(3) 86 ng/J (0.20 lb/MMBtu) heat input if the affected facility combusts coal or coal and other fuels and

(i) Has an annual capacity factor for coal or coal and other fuels of 30 percent (0.30) or less,

(ii) Has a maximum heat input capacity of 73 MW (250 MMBtu/hr) or less,

(iii) Has a federally enforceable requirement limiting operation of the affected facility to an annual capacity factor of 30 percent (0.30) or less for coal or coal and other solid fuels, and

(iv) Construction of the affected facility commenced after June 19, 1984, and before November 25, 1986.

(4) An affected facility burning coke oven gas alone or in combination with other fuels not subject to a PM standard under 60.43b and not using a postcombustion technology (except a wet scrubber) for reducing PM or SO₂ emissions is not subject to the PM limits under 60.43b(a).

(b) On and after the date on which the performance test is completed or required to be completed under §60.8, whichever comes first, no owner or operator of an affected facility that commenced construction, reconstruction, or modification on or before February 28, 2005, and that combusts oil (or mixtures of oil with other fuels) and uses a conventional or emerging technology to reduce SO₂ emissions shall cause to be discharged into the atmosphere from that affected facility any gases that contain PM in excess of 43 ng/J (0.10 lb/MMBtu) heat input.

(c) On and after the date on which the initial performance test is completed or is required to be completed under §60.8, whichever comes first, no owner or operator of an affected facility that commenced construction, reconstruction, or modification on or before February 28, 2005, and that combusts wood, or wood with other fuels, except coal, shall cause to be discharged from that affected facility any gases that contain PM in excess of the following emission limits:

(1) 43 ng/J (0.10 lb/MMBtu) heat input if the affected facility has an annual

40 CFR Ch. I (7–1–23 Edition)

capacity factor greater than 30 percent (0.30) for wood.

(2) 86 ng/J (0.20 lb/MMBtu) heat input if (i) The affected facility has an annual capacity factor of 30 percent (0.30) or less for wood;

(ii) Is subject to a federally enforceable requirement limiting operation of the affected facility to an annual capacity factor of 30 percent (0.30) or less for wood; and

(iii) Has a maximum heat input capacity of 73 MW (250 MMBtu/hr) or less.

(d) On and after the date on which the initial performance test is completed or is required to be completed under §60.8, whichever date comes first, no owner or operator of an affected facility that combusts municipal-type solid waste or mixtures of municipaltype solid waste with other fuels, shall cause to be discharged into the atmosphere from that affected facility any gases that contain PM in excess of the following emission limits:

(1) 43 ng/J (0.10 lb/MMBtu) heat input;
(i) If the affected facility combusts only municipal-type solid waste; or

(ii) If the affected facility combusts municipal-type solid waste and other fuels and has an annual capacity factor for the other fuels of 10 percent (0.10) or less.

(2) 86 ng/J (0.20 lb/MMBtu) heat input if the affected facility combusts municipal-type solid waste or municipal-type solid waste and other fuels; and

(i) Has an annual capacity factor for municipal-type solid waste and other fuels of 30 percent (0.30) or less;

(ii) Has a maximum heat input capacity of 73 MW (250 MMBtu/hr) or less;

(iii) Has a federally enforceable requirement limiting operation of the affected facility to an annual capacity factor of 30 percent (0.30) or less for municipal-type solid waste, or municipal-type solid waste and other fuels; and

(iv) Construction of the affected facility commenced after June 19, 1984, but on or before November 25, 1986.

(e) For the purposes of this section, the annual capacity factor is determined by dividing the actual heat input to the steam generating unit during the calendar year from the combustion of coal, wood, or municipal-type

solid waste, and other fuels, as applicable, by the potential heat input to the steam generating unit if the steam generating unit had been operated for 8,760 hours at the maximum heat input capacity.

(f) On and after the date on which the initial performance test is completed or is required to be completed under §60.8, whichever date comes first, no owner or operator of an affected facility that combusts coal, oil, wood, or mixtures of these fuels with any other fuels shall cause to be discharged into the atmosphere any gases that exhibit greater than 20 percent opacity (6minute average), except for one 6minute period per hour of not more than 27 percent opacity. An owner or operator of an affected facility that elects to install, calibrate, maintain, and operate a continuous emissions monitoring system (CEMS) for measuring PM emissions according to the requirements of this subpart and is subject to a federally enforceable PM limit of 0.030 lb/MMBtu or less is exempt from the opacity standard specified in this paragraph.

(g) The PM and opacity standards apply at all times, except during periods of startup, shutdown, or malfunction.

(h)(1) Except as provided in paragraphs (h)(2), (h)(3), (h)(4), (h)(5), and (h)(6) of this section, on and after the date on which the initial performance test is completed or is required to be completed under §60.8, whichever date comes first, no owner or operator of an affected facility that commenced construction, reconstruction, or modification after February 28, 2005, and that combusts coal, oil, wood, a mixture of these fuels, or a mixture of these fuels with any other fuels shall cause to be discharged into the atmosphere from that affected facility any gases that contain PM in excess of 13 ng/J (0.030 lb/MMBtu) heat input,

(2) As an alternative to meeting the requirements of paragraph (h)(1) of this section, the owner or operator of an affected facility for which modification commenced after February 28, 2005, may elect to meet the requirements of this paragraph. On and after the date on which the initial performance test is completed or required to be com-

pleted under §60.8, no owner or operator of an affected facility that commences modification after February 28, 2005 shall cause to be discharged into the atmosphere from that affected facility any gases that contain PM in excess of both:

(i) 22 ng/J (0.051 lb/MMBtu) heat input derived from the combustion of coal, oil, wood, a mixture of these fuels, or a mixture of these fuels with any other fuels; and

(ii) 0.2 percent of the combustion concentration (99.8 percent reduction) when combusting coal, oil, wood, a mixture of these fuels, or a mixture of these fuels with any other fuels.

(3) On and after the date on which the initial performance test is completed or is required to be completed under §60.8, whichever date comes first, no owner or operator of an affected facility that commences modification after February 28, 2005, and that combusts over 30 percent wood (by heat input) on an annual basis and has a maximum heat input capacity of 73 MW (250 MMBtu/h) or less shall cause to be discharged into the atmosphere from that affected facility any gases that contain PM in excess of 43 ng/J (0.10 lb/MMBtu) heat input.

(4) On and after the date on which the initial performance test is completed or is required to be completed under §60.8, whichever date comes first, no owner or operator of an affected facility that commences modification after February 28, 2005, and that combusts over 30 percent wood (by heat input) on an annual basis and has a maximum heat input capacity greater than 73 MW (250 MMBtu/h) shall cause to be discharged into the atmosphere from that affected facility any gases that contain PM in excess of 37 ng/J (0.085 lb/MMBtu) heat input.

(5) On and after the date on which the initial performance test is completed or is required to be completed under §60.8, whichever date comes first, an owner or operator of an affected facility not located in a noncontinental area that commences construction, reconstruction, or modification after February 28, 2005, and that combusts only oil that contains no more than 0.30 weight percent sulfur, coke oven gas, a mixture of these fuels, or either fuel (or a mixture of these fuels) in combination with other fuels not subject to a PM standard in §60.43b and not using a post-combustion technology (except a wet scrubber) to reduce SO_2 or PM emissions is not subject to the PM limits in (h)(1) of this section.

(6) On and after the date on which the initial performance test is completed or is required to be completed under §60.8, whichever date comes first, an owner or operator of an affected facility located in a noncontinental area that commences construction, reconstruction, or modification after February 28, 2005, and that combusts only oil that contains no more than 0.5 weight percent sulfur, coke oven gas, a mixture of these fuels, or either fuel (or a mixture of these fuels) in combination with other fuels not subject to a PM standard in §60.43b and not using a post-combustion technology (except

40 CFR Ch. I (7–1–23 Edition)

a wet scrubber) to reduce SO_2 or PM emissions is not subject to the PM limits in (h)(1) of this section.

[72 FR 32742, June 13, 2007, as amended at 74 FR 5084, Jan. 28, 2009; 77 FR 9459, Feb. 16, 2012]

§60.44b Standard for nitrogen oxides (NO₂).

(a) Except as provided under paragraphs (k) and (l) of this section, on and after the date on which the initial performance test is completed or is required to be completed under §60.8, whichever date comes first, no owner or operator of an affected facility that is subject to the provisions of this section and that combusts only coal, oil, or natural gas shall cause to be discharged into the atmosphere from that affected facility any gases that contain NO_X (expressed as NO_2) in excess of the following emission limits:

Fuel/steam generating unit type		Nitrogen oxide emission limits (expressed as NO ₂) heat input	
	ng/J	lb/MMBTu	
(1) Natural gas and distillate oil, except (4):			
(i) Low heat release rate	43	0.10	
(ii) High heat release rate	86	0.20	
(2) Residual oil:			
(i) Low heat release rate	130	0.30	
(ii) High heat release rate	170	0.40	
(3) Coal:			
(i) Mass-feed stoker	210	0.50	
(ii) Spreader stoker and fluidi⊡ed bed combustion	260	0.60	
(iii) Pulveri⊡ed coal	300	0.70	
(iv) Lignite, except (v)	260	0.60	
(v) Lignite mined in North Dakota, South Dakota, or Montana and combusted in a slag tap			
furnace	340	0.80	
(vi) Coal-derived synthetic fuels	210	0.50	
(4) Duct burner used in a combined cycle system:			
(i) Natural gas and distillate oil	86	0.20	
(ii) Residual oil	170	0.40	

(b) Except as provided under paragraphs (k) and (l) of this section, on and after the date on which the initial performance test is completed or is required to be completed under §60.8, whichever date comes first, no owner or operator of an affected facility that simultaneously combusts mixtures of only coal, oil, or natural gas shall cause to be discharged into the atmosphere from that affected facility any gases that contain NO_X in excess of a limit determined by the use of the following formula:

$$\mathbf{E}_{n} = \frac{\left(\mathbf{EL}_{go}\mathbf{H}_{go}\right) + \left(\mathbf{EL}_{ro}\mathbf{H}_{ro}\right) + \left(\mathbf{EL}_{c}\mathbf{H}_{c}\right)}{\left(\mathbf{H}_{go} + \mathbf{H}_{ro} + \mathbf{H}_{c}\right)}$$

Where:

- $E_n = NO_X$ emission limit (expressed as NO₂), ng/J (lb/MMBtu);
- EL_{go} = Appropriate emission limit from paragraph (a)(1) for combustion of natural gas or distillate oil, ng/J (lb/MMBtu);
- H_{go} = Heat input from combustion of natural gas or distillate oil, J (MMBtu);

- EL_{ro} = Appropriate emission limit from paragraph (a)(2) for combustion of residual oil, ng/J (lb/MMBtu);
- H_{ro} = Heat input from combustion of residual oil, J (MMBtu);
- EL_c = Appropriate emission limit from paragraph (a)(3) for combustion of coal, ng/J (lb/MMBtu); and
- H_c = Heat input from combustion of coal, J (MMBtu).

(c) Except as provided under paragraph (d) and (l) of this section, on and after the date on which the initial performance test is completed or is required to be completed under §60.8, whichever date comes first, no owner or operator of an affected facility that simultaneously combusts coal or oil, natural gas (or any combination of the three), and wood, or any other fuel shall cause to be discharged into the atmosphere any gases that contain NO_X in excess of the emission limit for the coal, oil, natural gas (or any combination of the three), combusted in the affected facility, as determined pursuant to paragraph (a) or (b) of this section. This standard does not apply to an affected facility that is subject to and in compliance with a federally enforceable requirement that limits operation of the affected facility to an annual capacity factor of 10 percent (0.10)or less for coal, oil, natural gas (or any combination of the three).

(d) On and after the date on which the initial performance test is completed or is required to be completed under §60.8, whichever date comes first, no owner or operator of an affected facility that simultaneously combusts natural gas and/or distillate oil with a potential SO₂ emissions rate of 26 ng/J (0.060 lb/MMBtu) or less with wood, municipal-type solid waste, or other solid fuel, except coal, shall cause to be discharged into the atmosphere from that affected facility any gases that contain $NO_{\rm X}$ in excess of 130 ng/J (0.30 lb/ MMBtu) heat input unless the affected facility has an annual capacity factor for natural gas, distillate oil, or a mixture of these fuels of 10 percent (0.10) or less and is subject to a federally enforceable requirement that limits operation of the affected facility to an annual capacity factor of 10 percent (0.10) or less for natural gas, distillate oil, or a mixture of these fuels.

(e) Except as provided under paragraph (1) of this section, on and after the date on which the initial performance test is completed or is required to be completed under §60.8, whichever date comes first, no owner or operator of an affected facility that simultaneously combusts only coal, oil, or natural gas with byproduct/waste shall cause to be discharged into the atmosphere any gases that contain NO_x in excess of the emission limit determined by the following formula unless the affected facility has an annual capacity factor for coal, oil, and natural gas of 10 percent (0.10) or less and is subject to a federally enforceable requirement that limits operation of the affected facility to an annual capacity factor of 10 percent (0.10) or less:

(f) Any owner or operator of an affected facility that combusts byproduct/waste with either natural gas or oil may petition the Administrator within 180 days of the initial startup of the affected facility to establish a NO_X emission limit that shall apply specifically to that affected facility when the byproduct/waste is combusted. The petition shall include sufficient and appropriate data, as determined by the Administrator, such as NO_X emissions from the affected facility, waste composition (including nitrogen content), and combustion conditions to allow the Administrator to confirm that the affected facility is unable to comply with the emission limits in paragraph (e) of this section and to determine the appropriate emission limit for the affected facility.

(1) Any owner or operator of an affected facility petitioning for a facility-specific NO_X emission limit under this section shall:

(i) Demonstrate compliance with the emission limits for natural gas and distillate oil in paragraph (a)(1) of this section or for residual oil in paragraph (a)(2) or (1)(1) of this section, as appropriate, by conducting a 30-day performance test as provided in 60.46b(e). During the performance test only natural gas, distillate oil, or residual oil shall be combusted in the affected facility; and

(ii) Demonstrate that the affected facility is unable to comply with the emission limits for natural gas and distillate oil in paragraph (a)(1) of this section or for residual oil in paragraph (a)(2) or (1)(1) of this section, as appropriate, when gaseous or liquid byproduct/waste is combusted in the affected facility under the same conditions and using the same technological system of emission reduction applied when demonstrating compliance under paragraph (f)(1)(i) of this section.

(2) The NO_X emission limits for natural gas or distillate oil in paragraph (a)(1) of this section or for residual oil in paragraph (a)(2) or (1)(1) of this section, as appropriate, shall be applicable to the affected facility until and unless the petition is approved by the Administrator. If the petition is approved by the Administrator, a facility-specific NO_X emission limit will be established at the NO_X emission level achievable when the affected facility is combusting oil or natural gas and byproduct/waste in a manner that the Administrator determines to be consistent with minimizing NO_X emissions. In lieu of amending this subpart, a letter will be sent to the facility describing the facility-specific NO_X limit. The facility shall use the compliance procedures detailed in the letter and make the letter available to the public. If the Administrator determines it is appropriate, the conditions and requirements of the letter can be reviewed and changed at any point.

(g) Any owner or operator of an affected facility that combusts hazardous waste (as defined by 40 CFR part 261 or 40 CFR part 761) with natural gas or oil may petition the Administrator within 180 days of the initial startup of the affected facility for a waiver from compliance with the NO_X emission limit that applies specifically to that affected facility. The petition must include sufficient and appropriate data, as determined by the Administrator, on NO_X emissions from the affected facility, waste destruction efficiencies, waste composition (including nitrogen content), the quantity of specific wastes to be combusted and combustion conditions to allow the Administrator to determine if the affected facility is able to comply with the NO_X emission limits required by this section. The owner or operator of the af40 CFR Ch. I (7-1-23 Edition)

fected facility shall demonstrate that when hazardous waste is combusted in the affected facility, thermal destruction efficiency requirements for hazardous waste specified in an applicable federally enforceable requirement preclude compliance with the NO_X emission limits of this section. The NO_X emission limits for natural gas or distillate oil in paragraph (a)(1) of this section or for residual oil in paragraph (a)(2) or (l)(1) of this section, as appropriate, are applicable to the affected facility until and unless the petition is approved by the Administrator. (See 40 CFR 761.70 for regulations applicable to the incineration of materials containing polychlorinated biphenyls (PCB's).) In lieu of amending this subpart, a letter will be sent to the facility describing the facility-specific NO_X limit. The facility shall use the compliance procedures detailed in the letter and make the letter available to the public. If the Administrator determines it is appropriate, the conditions and requirements of the letter can be reviewed and changed at any point.

(h) For purposes of paragraph (i) of this section, the NO_X standards under this section apply at all times including periods of startup, shutdown, or malfunction.

(i) Except as provided under paragraph (j) of this section, compliance with the emission limits under this section is determined on a 30-day rolling average basis.

(j) Compliance with the emission limits under this section is determined on a 24-hour average basis for the initial performance test and on a 3-hour average basis for subsequent performance tests for any affected facilities that:

(1) Combust, alone or in combination, only natural gas, distillate oil, or residual oil with a nitrogen content of 0.30 weight percent or less;

(2) Have a combined annual capacity factor of 10 percent or less for natural gas, distillate oil, and residual oil with a nitrogen content of 0.30 weight percent or less; and

(3) Are subject to a federally enforceable requirement limiting operation of the affected facility to the firing of natural gas, distillate oil, and/or residual oil with a nitrogen content of 0.30

weight percent or less and limiting operation of the affected facility to a combined annual capacity factor of 10 percent or less for natural gas, distillate oil, and residual oil with a nitrogen content of 0.30 weight percent or less.

(k) Affected facilities that meet the criteria described in paragraphs (j)(1), (2), and (3) of this section, and that have a heat input capacity of 73 MW (250 MMBtu/hr) or less, are not subject to the NO_X emission limits under this section.

(1) On and after the date on which the initial performance test is completed or is required to be completed under 60.8, whichever date is first, no owner or operator of an affected facility that commenced construction after July 9, 1997 shall cause to be discharged into the atmosphere from that affected facility any gases that contain NOx (expressed as NO2) in excess of the following limits:

(1) 86 ng/J (0.20 lb/MMBtu) heat input if the affected facility combusts coal, oil, or natural gas (or any combination of the three), alone or with any other fuels. The affected facility is not subject to this limit if it is subject to and in compliance with a federally enforceable requirement that limits operation of the facility to an annual capacity factor of 10 percent (0.10) or less for coal, oil, and natural gas (or any combination of the three); or

(2) If the affected facility has a low heat release rate and combusts natural gas or distillate oil in excess of 30 percent of the heat input on a 30-day rolling average from the combustion of all fuels, a limit determined by use of the following formula:

$$E_{n} = \frac{(0.10 \times H_{go}) + (0.20 \times H_{r})}{(H_{go} + H_{r})}$$

Where:

 $E_n = NO_X$ emission limit, (lb/MMBtu);

- H_{go} = 30-day heat input from combustion of natural gas or distillate oil; and
- $H_{\rm r}$ = 30-day heat input from combustion of any other fuel.

(3) After February 27, 2006, units where more than 10 percent of total annual output is electrical or mechanical may comply with an optional limit of 270 ng/J (2.1 lb/MWh) gross energy output, based on a 30-day rolling average. Units complying with this outputbased limit must demonstrate compliance according to the procedures of §60.48Da(i) of subpart Da of this part, and must monitor emissions according to §60.49Da(c), (k), through (n) of subpart Da of this part.

[72 FR 32742, June 13, 2007, as amended at 74 FR 5086, Jan. 28, 2009; 77 FR 9459, Feb. 16, 2012]

§60.45b Compliance and performance test methods and procedures for sulfur dioxide.

(a) The SO_2 emission standards in §60.42b apply at all times. Facilities burning coke oven gas alone or in combination with any other gaseous fuels or distillate oil are allowed to exceed the limit 30 operating days per calendar year for SO_2 control system maintenance.

(b) In conducting the performance tests required under 60.8, the owner or operator shall use the methods and procedures in appendix A (including fuel certification and sampling) of this part or the methods and procedures as specified in this section, except as provided in 60.8(b). Section 60.8(f) does not apply to this section. The 30-4ay notice required in 60.8(d) applies only to the initial performance test unless otherwise specified by the Administrator.

(c) The owner or operator of an affected facility shall conduct performance tests to determine compliance with the percent of potential SO_2 emission rate (% P_s) and the SO_2 emission rate (E_s) pursuant to §60.42b following the procedures listed below, except as provided under paragraph (d) and (k) of this section.

(1) The initial performance test shall be conducted over 30 consecutive operating days of the steam generating unit. Compliance with the SO_2 standards shall be determined using a 30-day average. The first operating day included in the initial performance test shall be scheduled within 30 days after achieving the maximum production rate at which the affected facility will be operated, but not later than 180 days after initial startup of the facility.

§60.45b

(2) If only coal, only oil, or a mixture of coal and oil is combusted, the following procedures are used:

(i) The procedures in Method 19 of appendix A-7 of this part are used to determine the hourly SO₂ emission rate ($E_{\rm ho}$) and the 30-day average emission rate ($E_{\rm ao}$). The hourly averages used to compute the 30-day averages are obtained from the CEMS of §60.47b(a) or (b).

(ii) The percent of potential SO_2 emission rate (%P_s) emitted to the atmosphere is computed using the following formula:

$$\%P_{s} = 100 \left(1 - \frac{\%R_{g}}{100}\right) \left(1 - \frac{\%R_{f}}{100}\right)$$

Where:

%Ps = Potential SO₂ emission rate, percent; %Rg = SO₂ removal efficiency of the control device as determined by Method 19 of appendix A of this part, in percent; and

bendra A of this part, in percent, and %R_r = SO₂ removal efficiency of fuel pretreatment as determined by Method 19 of appendix A of this part, in percent.

(3) If coal or oil is combusted with other fuels, the same procedures required in paragraph (c)(2) of this section are used, except as provided in the following:

(i) An adjusted hourly SO_2 emission rate (E_{ho}°) is used in Equation 19–19 of Method 19 of appendix A of this part to compute an adjusted 30-day average emission rate (E_{ao}°) . The Eho° is computed using the following formula:

$$E_{ho}^{o} = \frac{E_{ho} - E_{w} (1 - X_{k})}{X_{k}}$$

Where:

 $E_{ho}{}^{\rm o}$ = Adjusted hourly SO_2 emission rate, ng/ J (lb/MMBtu);

- E_{ho} = Hourly SO_2 emission rate, ng/J (lb/ MMBtu);
- E_w = SO_2 concentration in fuels other than coal and oil combusted in the affected facility, as determined by the fuel sampling and analysis procedures in Method 19 of appendix A of this part, ng/J (lb/MMBtu). The value E_w for each fuel lot is used for each hourly average during the time that the lot is being combusted; and
- X_k = Fraction of total heat input from fuel combustion derived from coal, oil, or coal and oil, as determined by applicable procedures in Method 19 of appendix A of this part.

40 CFR Ch. I (7–1–23 Edition)

(ii) To compute the percent of potential SO₂ emission rate (%P_s), an adjusted %R_g (%R_g°) is computed from the adjusted E_{ao}° from paragraph (b)(3)(i) of this section and an adjusted average SO₂ inlet rate (E_{ai}°) using the following formula:

$$R_{g}^{o} = 100 \left(1.0 - \frac{E_{ao}^{o}}{E_{ai}^{o}} \right)$$

To compute E_{ai}^{o} , an adjusted hourly SO_2 inlet rate (E_{hi}^{o}) is used. The E_{hi}^{o} is computed using the following formula:

$$E_{hi}^{o} = \frac{E_{hi} - E_{w} (1 - X_{k})}{X_{k}}$$

Where:

 $E_{\rm hi^0}$ = Adjusted hourly SO_2 inlet rate, ng/J (lb/MMBtu); and

 E_{hi} = Hourly SO₂ inlet rate, ng/J (lb/MMBtu).

(4) The owner or operator of an affected facility subject to paragraph (c)(3) of this section does not have to measure parameters E_w or X_k if the owner or operator elects to assume that $X_k = 1.0$. Owners or operators of affected facilities who assume $X_k = 1.0$ shall:

(i) Determine $\ensuremath{\%P_s}$ following the procedures in paragraph (c)(2) of this section; and

(ii) Sulfur dioxide emissions (E_s) are considered to be in compliance with SO_2 emission limits under §60.42b.

(5) The owner or operator of an affected facility that qualifies under the provisions of §60.42b(d) does not have to measure parameters E_w or X_k in paragraph (c)(3) of this section if the owner or operator of the affected facility elects to measure SO_2 emission rates of the coal or oil following the fuel sampling and analysis procedures in Method 19 of appendix A-7 of this part.

(d) Except as provided in paragraph (j) of this section, the owner or operator of an affected facility that combusts only very low sulfur oil, natural gas, or a mixture of these fuels, has an annual capacity factor for oil of 10 percent (0.10) or less, and is subject to a federally enforceable requirement limiting operation of the affected facility to an annual capacity factor for oil of 10 percent (0.10) or less shall: ER13JN07.030</MATH>

(1) Conduct the initial performance test over 24 consecutive steam generating unit operating hours at full load;

(2) Determine compliance with the standards after the initial performance test based on the arithmetic average of the hourly emissions data during each steam generating unit operating day if a CEMS is used, or based on a daily average if Method 6B of appendix A of this part or fuel sampling and analysis procedures under Method 19 of appendix A of this part are used.

(e) The owner or operator of an affected facility subject to §60.42b(d)(1) shall demonstrate the maximum design capacity of the steam generating unit by operating the facility at maximum capacity for 24 hours. This demonstration will be made during the initial performance test and a subsequent demonstration may be requested at any other time. If the 24-hour average firing rate for the affected facility is less than the maximum design capacity provided by the manufacturer of the affected facility, the 24-hour average firing rate shall be used to determine the capacity utilization rate for the affected facility, otherwise the maximum design capacity provided by the manufacturer is used.

(f) For the initial performance test required under §60.8, compliance with the SO₂ emission limits and percent reduction requirements under §60.42b is based on the average emission rates and the average percent reduction for SO_2 for the first 30 consecutive steam generating unit operating days, except as provided under paragraph (d) of this section. The initial performance test is the only test for which at least 30 days prior notice is required unless otherwise specified by the Administrator. The initial performance test is to be scheduled so that the first steam generating unit operating day of the 30 successive steam generating unit operating days is completed within 30 days after achieving the maximum production rate at which the affected facility will be operated, but not later than 180 days after initial startup of the facility. The boiler load during the 30-day period does not have to be the maximum design load, but must be representative of future operating conditions and include at least one 24-hour period at full load.

(g) After the initial performance test required under §60.8, compliance with the SO₂ emission limits and percent reduction requirements under §60.42b is based on the average emission rates and the average percent reduction for SO_2 for 30 successive steam generating unit operating days, except as provided under paragraph (d). A separate performance test is completed at the end of each steam generating unit operating day after the initial performance test, and a new 30-day average emission rate and percent reduction for SO₂ are calculated to show compliance with the standard.

(h) Except as provided under paragraph (i) of this section, the owner or operator of an affected facility shall use all valid SO_2 emissions data in calculating %P_s and E_{ho} under paragraph (c), of this section whether or not the minimum emissions data requirements under §60.46b are achieved. All valid emissions data, including valid SO_2 emission data collected during periods of startup, shutdown and malfunction, shall be used in calculating %P_s and E_{ho} pursuant to paragraph (c) of this section.

(i) During periods of malfunction or maintenance of the SO_2 control systems when oil is combusted as provided under §60.42b(i), emission data are not used to calculate %P_s or E_s under §60.42b(a), (b) or (c), however, the emissions data are used to determine compliance with the emission limit under §60.42b(i).

(j) The owner or operator of an affected facility that only combusts very low sulfur oil, natural gas, or a mixture of these fuels with any other fuels not subject to an SO_2 standard is not subject to the compliance and performance testing requirements of this section if the owner or operator obtains fuel receipts as described in §60.49b(r).

(k) The owner or operator of an affected facility seeking to demonstrate compliance in $\S 60.42b(d)(4)$, 60.42b(j), 60.42b(k)(2), and 60.42b(k)(3) (when not burning coal) shall follow the applicable procedures in $\S 60.49b(r)$.

 $[72\ {\rm FR}\ 32742,\ {\rm June}\ 13,\ 2007,\ {\rm as}\ {\rm amended}\ {\rm at}\ 74$ ${\rm FR}\ 5086,\ {\rm Jan}.\ 28,\ 2009]$

§60.46b Compliance and performance test methods and procedures for particulate matter and nitrogen oxides.

(a) The PM emission standards and opacity limits under 60.43b apply at all times except during periods of startup, shutdown, or malfunction. The NO_x emission standards under 60.44b apply at all times.

(b) Compliance with the PM emission standards under §60.43b shall be determined through performance testing as described in paragraph (d) of this section, except as provided in paragraph (i) of this section.

(c) Compliance with the NO_x emission standards under §60.44b shall be determined through performance testing under paragraph (e) or (f), or under paragraphs (g) and (h) of this section, as applicable.

(d) To determine compliance with the PM emission limits and opacity limits under §60.43b, the owner or operator of an affected facility shall conduct an initial performance test as required under §60.8, and shall conduct subsequent performance tests as requested by the Administrator, using the following procedures and reference methods:

(1) Method 3A or 3B of appendix A-2 of this part is used for gas analysis when applying Method 5 of appendix A-3 of this part or Method 17 of appendix A-6 of this part.

(2) Method 5, 5B, or 17 of appendix A of this part shall be used to measure the concentration of PM as follows:

(i) Method 5 of appendix A of this part shall be used at affected facilities without wet flue gas desulfurization (FGD) systems; and

(ii) Method 17 of appendix A-6 of this part may be used at facilities with or without wet scrubber systems provided the stack gas temperature does not exceed a temperature of 160 °C (320 °F). The procedures of sections 8.1 and 11.1 of Method 5B of appendix A-3 of this part may be used in Method 17 of appendix A-6 of this part only if it is used after a wet FGD system. Do not use Method 17 of appendix A-6 of this part after wet FGD systems if the effluent is saturated or laden with water droplets. 40 CFR Ch. I (7–1–23 Edition)

(iii) Method 5B of appendix A of this part is to be used only after wet FGD systems.

(3) Method 1 of appendix A of this part is used to select the sampling site and the number of traverse sampling points. The sampling time for each run is at least 120 minutes and the minimum sampling volume is 1.7 dscm (60 dscf) except that smaller sampling times or volumes may be approved by the Administrator when necessitated by process variables or other factors.

(4) For Method 5 of appendix A of this part, the temperature of the sample gas in the probe and filter holder is monitored and is maintained at 160 ± 14 °C (320 ± 25 °F).

(5) For determination of PM emissions, the oxygen (O_2) or CO_2 sample is obtained simultaneously with each run of Method 5, 5B, or 17 of appendix A of this part by traversing the duct at the same sampling location.

(6) For each run using Method 5, 5B, or 17 of appendix A of this part, the emission rate expressed in ng/J heat input is determined using:

(i) The O_2 or CO_2 measurements and PM measurements obtained under this section;

(ii) The dry basis F factor; and

(iii) The dry basis emission rate calculation procedure contained in Method 19 of appendix A of this part.

(7) Method 9 of appendix A of this part is used for determining the opacity of stack emissions.

(e) To determine compliance with the emission limits for NO_X required under §60.44b, the owner or operator of an affected facility shall conduct the performance test as required under §60.8 using the continuous system for monitoring NO_X under §60.48(b).

(1) For the initial compliance test, NO_x from the steam generating unit are monitored for 30 successive steam generating unit operating days and the 30-day average emission rate is used to determine compliance with the NO_x emission standards under §60.44b. The 30-day average emission rate is calculated as the average of all hourly emissions data recorded by the monitoring system during the 30-day test period.

(2) Following the date on which the initial performance test is completed

or is required to be completed in §60.8, whichever date comes first, the owner or operator of an affected facility which combusts coal (except as specified under §60.46b(e)(4)) or which combusts residual oil having a nitrogen content greater than 0.30 weight percent shall determine compliance with the NO_X emission standards in §60.44b on a continuous basis through the use of a 30-day rolling average emission rate. A new 30-day rolling average emission rate is calculated for each steam generating unit operating day as the average of all of the hourly NO_X emission data for the preceding 30 steam generating unit operating days.

(3) Following the date on which the initial performance test is completed or is required to be completed under §60.8, whichever date comes first, the owner or operator of an affected facility that has a heat input capacity greater than 73 MW (250 MMBtu/hr) and that combusts natural gas, distillate oil, or residual oil having a nitrogen content of 0.30 weight percent or less shall determine compliance with the NO_X standards under §60.44b on a continuous basis through the use of a 30day rolling average emission rate. A new 30-day rolling average emission rate is calculated each steam generating unit operating day as the average of all of the hourly NO_X emission data for the preceding 30 steam generating unit operating days.

(4) Following the date on which the initial performance test is completed or required to be completed under §60.8, whichever date comes first, the owner or operator of an affected facility that has a heat input capacity of 73 MW (250 MMBtu/hr) or less and that combusts natural gas, distillate oil, gasified coal, or residual oil having a nitrogen content of 0.30 weight percent or less shall upon request determine compliance with the NO_X standards in §60.44b through the use of a 30-day performance test. During periods when performance tests are not requested, NO_X emissions data collected pursuant to §60.48b(g)(1) or §60.48b(g)(2) are used to calculate a 30-day rolling average emission rate on a daily basis and used to prepare excess emission reports, but will not be used to determine compliance with the $NO_{\rm X}$ emission standards.

A new 30-day rolling average emission rate is calculated each steam generating unit operating day as the average of all of the hourly NO_X emission data for the preceding 30 steam generating unit operating days.

(5) If the owner or operator of an affected facility that combusts residual oil does not sample and analyze the residual oil for nitrogen content, as specified in (60.49b(e)), the requirements of (60.48b(g)(1)) apply and the provisions of (60.48b(g)(2)) are inapplicable.

(f) To determine compliance with the emissions limits for NO_X required by \$60.44b(a)(4) or \$60.44b(1) for duct burners used in combined cycle systems, either of the procedures described in paragraph (f)(1) or (2) of this section may be used:

(1) The owner or operator of an affected facility shall conduct the performance test required under §60.8 as follows:

(i) The emissions rate (E) of NO_X shall be computed using Equation 1 in this section:

$$\mathbf{E} = \mathbf{E}_{sg} + \left(\frac{\mathbf{H}_{g}}{\mathbf{H}_{b}}\right) \left(\mathbf{E}_{sg} - \mathbf{E}_{g}\right) \qquad (\mathrm{Eq.1})$$

Where:

- E = Emissions rate of NO_x from the duct burner, ng/J (lb/MMBtu) heat input;
- E_{sg} = Combined effluent emissions rate, in ng/J (lb/MMBtu) heat input using appropriate F factor as described in Method 19 of appendix A of this part;
- H_g = Heat input rate to the combustion turbine, in J/hr (MMBtu/hr);
- $H_{\rm b}$ = Heat input rate to the duct burner, in J/ hr (MMBtu/hr); and
- $E_g = Emissions \ rate \ from \ the \ combustion \ turbine, \ in \ ng/J \ (lb/MMBtu) \ heat \ input \ calculated \ using \ appropriate \ F \ factor \ as \ described \ in \ Method \ 19 \ of \ appendix \ A \ of \ this \ part.$

(ii) Method 7E of appendix A of this part or Method 320 of appendix A of part 63 shall be used to determine the NO_X concentrations. Method 3A or 3B of appendix A of this part shall be used to determine O_2 concentration.

(iii) The owner or operator shall identify and demonstrate to the Administrator's satisfaction suitable methods to determine the average hourly heat input rate to the combustion turbine and the average hourly heat input rate to the affected duct burner.

LAPBH6H6L3 with DISTILLER

§60.46b

(iv) Compliance with the emissions limits under 60.44b(a)(4) or 60.44b(1) is determined by the three-run average (nominal 1-hour runs) for the initial and subsequent performance tests; or

(2) The owner or operator of an affected facility may elect to determine compliance on a 30-day rolling average basis by using the CEMS specified under §60.48b for measuring NO_X and O₂ and meet the requirements of §60.48b. The sampling site shall be located at the outlet from the steam generating unit. The NO_X emissions rate at the outlet from the steam generating unit shall constitute the NO_X emissions rate from the duct burner of the combined cycle system.

(g) The owner or operator of an affected facility described in §60.44b(j) or 60.44b(k) shall demonstrate the maximum heat input capacity of the steam generating unit by operating the facility at maximum capacity for 24 hours. The owner or operator of an affected facility shall determine the maximum heat input capacity using the heat loss method or the heat input method described in sections 5 and 7.3 of the ASME Power Test Codes 4.1 (incorporated by reference, see §60.17). This demonstration of maximum heat input capacity shall be made during the initial performance test for affected facilities that meet the criteria of §60.44b(j). It shall be made within 60 days after achieving the maximum production rate at which the affected facility will be operated, but not later than 180 days after initial start-up of each facility, for affected facilities meeting the criteria of §60.44b(k). Subsequent demonstrations may be required by the Administrator at any other time. If this demonstration indicates that the maximum heat input capacity of the affected facility is less than that stated by the manufacturer of the affected facility, the maximum heat input capacity determined during this demonstration shall be used to determine the capacity utilization rate for the affected facility. Otherwise, the maximum heat input capacity provided by the manufacturer is used.

(h) The owner or operator of an affected facility described in §60.44b(j) that has a heat input capacity greater than 73 MW (250 MMBtu/hr) shall: 40 CFR Ch. I (7–1–23 Edition)

(1) Conduct an initial performance test as required under 60.8 over a minimum of 24 consecutive steam generating unit operating hours at maximum heat input capacity to demonstrate compliance with the NO_X emission standards under 60.44b using Method 7, 7A, or 7E of appendix A of this part, Method 320 of appendix A of part 63 of this chapter, or other approved reference methods; and

(2) Conduct subsequent performance tests once per calendar year or every 400 hours of operation (whichever comes first) to demonstrate compliance with the NO_x emission standards under 60.44b over a minimum of 3 consecutive steam generating unit operating hours at maximum heat input capacity using Method 7, 7A, or 7E of appendix A of this part, Method 320 of appendix A of part 63, or other approved reference methods.

(i) The owner or operator of an affected facility seeking to demonstrate compliance with the PM limit in paragraphs 60.43b(a)(4) or 60.43b(h)(5)shall follow the applicable procedures in 60.49b(r).

(j) In place of PM testing with Method 5 or 5B of appendix A-3 of this part, or Method 17 of appendix A-6 of this part, an owner or operator may elect to install, calibrate, maintain, and operate a CEMS for monitoring PM emissions discharged to the atmosphere and record the output of the system. The owner or operator of an affected facility who elects to continuously monitor PM emissions instead of conducting performance testing using Method 5 or 5B of appendix A-3 of this part or Method 17 of appendix A-6 of this part shall comply with the requirements specified in paragraphs (j)(1) through (j)(14) of this section.

(1) Notify the Administrator one month before starting use of the system.

(2) Notify the Administrator one month before stopping use of the system.

(3) The monitor shall be installed, evaluated, and operated in accordance with §60.13 of subpart A of this part.

(4) The initial performance evaluation shall be completed no later than 180 days after the date of initial startup of the affected facility, as specified

under §60.8 of subpart A of this part or within 180 days of notification to the Administrator of use of the CEMS if the owner or operator was previously determining compliance by Method 5, 5B, or 17 of appendix A of this part performance tests, whichever is later.

(5) The owner or operator of an affected facility shall conduct an initial performance test for PM emissions as required under §60.8 of subpart A of this part. Compliance with the PM emission limit shall be determined by using the CEMS specified in paragraph (j) of this section to measure PM and calculating a 24-hour block arithmetic average emission concentration using EPA Reference Method 19 of appendix A of this part, section 4.1.

(6) Compliance with the PM emission limit shall be determined based on the 24-hour daily (block) average of the hourly arithmetic average emission concentrations using CEMS outlet data.

(7) At a minimum, valid CEMS hourly averages shall be obtained as specified in paragraphs (j)(7)(i) of this section for 75 percent of the total operating hours per 30-day rolling average.

(i) At least two data points per hour shall be used to calculate each 1-hour arithmetic average.

(ii) [Reserved]

(8) The 1-hour arithmetic averages required under paragraph (j)(7) of this section shall be expressed in ng/J or lb/ MMBtu heat input and shall be used to calculate the boiler operating day daily arithmetic average emission concentrations. The 1-hour arithmetic averages shall be calculated using the data points required under 60.13(e)(2) of subpart A of this part.

(9) All valid CEMS data shall be used in calculating average emission concentrations even if the minimum CEMS data requirements of paragraph (j)(7) of this section are not met.

(10) The CEMS shall be operated according to Performance Specification 11 in appendix B of this part.

(11) During the correlation testing runs of the CEMS required by Performance Specification 11 in appendix B of this part, PM and O_2 (or CO_2) data shall be collected concurrently (or within a 30-to 60-minute period) by both the continuous emission monitors and performance tests conducted using the following test methods.

(i) For PM, Method 5 or 5B of appendix A-3 of this part or Method 17 of appendix A-6 of this part shall be used; and

(ii) For O_2 (or CO_2), Method 3A or 3B of appendix A-2 of this part, as applicable shall be used.

(12) Quarterly accuracy determinations and daily calibration drift tests shall be performed in accordance with procedure 2 in appendix F of this part. Relative Response Audit's must be performed annually and Response Correlation Audits must be performed every 3 years.

(13) When PM emissions data are not obtained because of CEMS breakdowns, repairs, calibration checks, and zero and span adjustments, emissions data shall be obtained by using other monitoring systems as approved by the Administrator or EPA Reference Method 19 of appendix A of this part to provide, as necessary, valid emissions data for a minimum of 75 percent of total operating hours per 30-day rolling average.

(14) As of January 1, 2012, and within 90 days after the date of completing each performance test, as defined in §60.8, conducted to demonstrate compliance with this subpart, you must submit relative accuracy test audit (i.e., reference method) data and performance test (*i.e.*, compliance test) data, except opacity data, electronically to EPA's Central Data Exchange (CDX) by using the Electronic Reporting Tool (ERT) (see http://www.epa.gov/ *ttn/chief/ert/ert tool.html/*) or other compatible electronic spreadsheet. Only data collected using test methods compatible with ERT are subject to this requirement to be submitted electronically into EPA's WebFIRE database.

[72 FR 32742, June 13, 2007, as amended at 74 FR 5086, Jan. 28, 2009; 76 FR 3523, Jan. 20, 2011; 77 FR 9460, Feb. 16, 2012; 79 FR 11249, Feb. 27, 2014]

§60.47b Emission monitoring for sulfur dioxide.

(a) Except as provided in paragraphs (b) and (f) of this section, the owner or operator of an affected facility subject to the SO_2 standards in §60.42b shall install, calibrate, maintain, and operate

CEMS for measuring SO₂ concentrations and either O_2 or CO_2 concentrations and shall record the output of the systems. For units complying with the percent reduction standard, the SO₂ and either O_2 or CO_2 concentrations shall both be monitored at the inlet and outlet of the SO_2 control device. If the owner or operator has installed and certified SO_2 and O_2 or CO_2 CEMS according to the requirements of §75.20(c)(1) of this chapter and appendix A to part 75 of this chapter, and is continuing to meet the ongoing quality assurance requirements of §75.21 of this chapter and appendix B to part 75 of this chapter, those CEMS may be used to meet the requirements of this section, provided that:

(1) When relative accuracy testing is conducted, SO_2 concentration data and CO_2 (or O_2) data are collected simultaneously; and

(2) In addition to meeting the applicable SO_2 and CO_2 (or O_2) relative accuracy specifications in Figure 2 of appendix B to part 75 of this chapter, the relative accuracy (RA) standard in section 13.2 of Performance Specification 2 in appendix B to this part is met when the RA is calculated on a lb/MMBtu basis; and

(3) The reporting requirements of $\S60.49b$ are met. SO₂ and CO₂ (or O₂) data used to meet the requirements of $\S60.49b$ shall not include substitute data values derived from the missing data procedures in subpart D of part 75 of this chapter, nor shall the SO₂ data have been bias adjusted according to the procedures of part 75 of this chapter.

(b) As an alternative to operating CEMS as required under paragraph (a) of this section, an owner or operator may elect to determine the average SO_2 emissions and percent reduction by:

(1) Collecting coal or oil samples in an as-fired condition at the inlet to the steam generating unit and analyzing them for sulfur and heat content according to Method 19 of appendix A of this part. Method 19 of appendix A of this part provides procedures for converting these measurements into the format to be used in calculating the average SO₂ input rate, or

(2) Measuring SO₂ according to Method 6B of appendix A of this part at the inlet or outlet to the SO_2 control system. An initial stratification test is required to verify the adequacy of the sampling location for Method 6B of appendix A of this part. The stratification test shall consist of three paired runs of a suitable SO_2 and CO_2 measurement train operated at the candidate location and a second similar train operated according to the procedures in Section 3.2 and the applicable procedures in Section 7 of Performance Specification 2. Method 6B of appendix A of this part, Method 6A of appendix A of this part, or a combination of Methods 6 and 3 or 3B of appendix A of this part or Methods 6C or Method 320 of appendix A of part 63 of this chapter and 3A of appendix A of this part are suitable measurement techniques. If Method 6B of appendix A of this part is used for the second train, sampling time and timer operation may be adjusted for the stratification test as long as an adequate sample volume is collected; however, both sampling trains are to be operated similarly. For the location to be adequate for Method 6B of appendix A of this part, 24-hour tests, the mean of the absolute difference between the three paired runs must be less than 10 percent.

(3) A daily SO₂ emission rate, E_D , shall be determined using the procedure described in Method 6A of appendix A of this part, section 7.6.2 (Equation 6A-8) and stated in ng/J (lb/ MMBtu) heat input.

(4) The mean 30-day emission rate is calculated using the daily measured values in ng/J (lb/MMBtu) for 30 successive steam generating unit operating days using equation 19–20 of Method 19 of appendix A of this part.

(c) The owner or operator of an affected facility shall obtain emission data for at least 75 percent of the operating hours in at least 22 out of 30 successive boiler operating days. If this minimum data requirement is not met with a single monitoring system, the owner or operator of the affected facility shall supplement the emission data with data collected with other monitoring systems as approved by the Administrator or the reference methods

and procedures as described in paragraph (b) of this section.

(d) The 1-hour average SO_2 emission rates measured by the CEMS required by paragraph (a) of this section and required under §60.13(h) is expressed in ng/J or lb/MMBtu heat input and is used to calculate the average emission rates under §60.42(b). Each 1-hour average SO_2 emission rate must be based on 30 or more minutes of steam generating unit operation. The hourly averages shall be calculated according to 60.13(h)(2). Hourly SO₂ emission rates are not calculated if the affected facility is operated less than 30 minutes in a given clock hour and are not counted toward determination of a steam generating unit operating day.

(e) The procedures under §60.13 shall be followed for installation, evaluation, and operation of the CEMS.

(1) Except as provided for in paragraph (e)(4) of this section, all CEMS shall be operated in accordance with the applicable procedures under Performance Specifications 1, 2, and 3 of appendix B of this part.

(2) Except as provided for in paragraph (e)(4) of this section, quarterly accuracy determinations and daily calibration drift tests shall be performed in accordance with Procedure 1 of appendix F of this part.

(3) For affected facilities combusting coal or oil, alone or in combination with other fuels, the span value of the SO₂ CEMS at the inlet to the SO₂ control device is 125 percent of the maximum estimated hourly potential SO₂ emissions of the fuel combusted, and the span value of the CEMS at the outlet to the SO₂ control device is 50 percent of the maximum estimated hourly potential SO₂ emissions of the fuel combusted. Alternatively, SO₂ span values determined according to section 2.1.1 in appendix A to part 75 of this chapter may be used.

(4) As an alternative to meeting the requirements of requirements of paragraphs (e)(1) and (e)(2) of this section, the owner or operator may elect to implement the following alternative data accuracy assessment procedures:

(i) For all required CO_2 and O_2 monitors and for SO_2 and NO_X monitors with span values greater than or equal to 100 ppm, the daily calibration error

test and calibration adjustment procedures described in sections 2.1.1 and 2.1.3 of appendix B to part 75 of this chapter may be followed instead of the CD assessment procedures in Procedure 1, section 4.1 of appendix F to this part.

(ii) For all required CO_2 and O_2 monitors and for SO_2 and NO_X monitors with span values greater than 30 ppm, quarterly linearity checks may be performed in accordance with section 2.2.1 of appendix B to part 75 of this chapter, instead of performing the cylinder gas audits (CGAs) described in Procedure 1, section 5.1.2 of appendix F to this part. If this option is selected: The frequency of the linearity checks shall be as specified in section 2.2.1 of appendix B to part 75 of this chapter; the applicable linearity specifications in section 3.2 of appendix A to part 75 of this chapter shall be met; the data validation and out-of-control criteria in section 2.2.3 of appendix B to part 75 of this chapter shall be followed instead of the excessive audit inaccuracy and out-of-control criteria in Procedure 1, section 5.2 of appendix F to this part; and the grace period provisions in section 2.2.4 of appendix B to part 75 of this chapter shall apply. For the purposes of data validation under this subpart, the cylinder gas audits described in Procedure 1, section 5.1.2 of appendix F to this part shall be performed for SO₂ and NO_X span values less than or equal to 30 ppm; and

(iii) For SO_2 , CO_2 , and O_2 monitoring systems and for NO_X emission rate monitoring systems, RATAs may be performed in accordance with section 2.3 of appendix B to part 75 of this chapter instead of following the procedures described in Procedure 1, section 5.1.1 of appendix F to this part. If this option is selected: The frequency of each RATA shall be as specified in section 2.3.1 of appendix B to part 75 of this chapter; the applicable relative accuracy specifications shown in Figure 2 in appendix B to part 75 of this chapter shall be met; the data validation and out-of-control criteria in section 2.3.2 of appendix B to part 75 of this chapter shall be followed instead of the excessive audit inaccuracy and out-of-control criteria in Procedure 1, section 5.2 of appendix F to this part; and the grace period provisions in section 2.3.3 of appendix B to part 75 of this chapter shall apply. For the purposes of data validation under this subpart, the relative accuracy specification in section 13.2 of Performance Specification 2 in appendix B to this part shall be met on a lb/MMBtu basis for SO₂ (regardless of the SO₂ emission level during the RATA), and for NO_X when the average NO_X emission rate measured by the reference method during the RATA is less than 0.100 lb/MMBtu.

(f) The owner or operator of an affected facility that combusts very low sulfur oil or is demonstrating compliance under (60.45b(k)) is not subject to the emission monitoring requirements under paragraph (a) of this section if the owner or operator maintains fuel records as described in (60.49b(r)).

[72 FR 32742, June 13, 2007, as amended at 74 FR 5087, Jan. 28, 2009; 79 FR 11249, Feb. 27, 2014]

§60.48b Emission monitoring for particulate matter and nitrogen oxides.

(a) Except as provided in paragraph (j) of this section, the owner or operator of an affected facility subject to the opacity standard under §60.43b shall install, calibrate, maintain, and operate a continuous opacity monitoring systems (COMS) for measuring the opacity of emissions discharged to the atmosphere and record the output of the system. The owner or operator of an affected facility subject to an opacity standard under §60.43b and meeting the conditions under paragraphs (j)(1), (2), (3), (4), (5), or (6) of this section who elects not to use a COMS shall conduct a performance test using Method 9 of appendix A-4 of this part and the procedures in §60.11 to demonstrate compliance with the applicable limit in §60.43b by April 29, 2011, within 45 days of stopping use of an existing COMS, or within 180 days after initial startup of the facility, whichever is later, and shall comply with either paragraphs (a)(1), (a)(2), or (a)(3) of this section. The observation period for Method 9 of appendix A-4 of this part performance tests may be reduced from 3 hours to 60 minutes if all 6-minute averages are less than 10 percent and all individual 15-second observations are less than or equal to 20 percent during the initial 60 minutes of observation.

40 CFR Ch. I (7–1–23 Edition)

(1) Except as provided in paragraph (a)(2) and (a)(3) of this section, the owner or operator shall conduct subsequent Method 9 of appendix A-4 of this part performance tests using the procedures in paragraph (a) of this section according to the applicable schedule in paragraphs (a)(1)(i) through (a)(1)(iv) of this section, as determined by the most recent Method 9 of appendix A-4 of this part performance test results.

(i) If no visible emissions are observed, a subsequent Method 9 of appendix A-4 of this part performance test must be completed within 12 calendar months from the date that the most recent performance test was conducted or within 45 days of the next day that fuel with an opacity standard is combusted, whichever is later;

(ii) If visible emissions are observed but the maximum 6-minute average opacity is less than or equal to 5 percent, a subsequent Method 9 of appendix A-4 of this part performance test must be completed within 6 calendar months from the date that the most recent performance test was conducted or within 45 days of the next day that fuel with an opacity standard is combusted, whichever is later;

(iii) If the maximum 6-minute average opacity is greater than 5 percent but less than or equal to 10 percent, a subsequent Method 9 of appendix A-4 of this part performance test must be completed within 3 calendar months from the date that the most recent performance test was conducted or within 45 days of the next day that fuel with an opacity standard is combusted, whichever is later; or

(iv) If the maximum 6-minute average opacity is greater than 10 percent, a subsequent Method 9 of appendix A-4 of this part performance test must be completed within 45 calendar days from the date that the most recent performance test was conducted.

(2) If the maximum 6-minute opacity is less than 10 percent during the most recent Method 9 of appendix A-4 of this part performance test, the owner or operator may, as an alternative to performing subsequent Method 9 of appendix A-4 of this part performance tests, elect to perform subsequent monitoring using Method 22 of appendix A-

7 of this part according to the procedures specified in paragraphs (a)(2)(i) and (ii) of this section.

(i) The owner or operator shall conduct 10 minute observations (during normal operation) each operating day the affected facility fires fuel for which an opacity standard is applicable using Method 22 of appendix A-7 of this part and demonstrate that the sum of the occurrences of any visible emissions is not in excess of 5 percent of the observation period (i.e., 30 seconds per 10 minute period). If the sum of the occurrence of any visible emissions is greater than 30 seconds during the initial 10 minute observation, immediately conduct a 30 minute observation. If the sum of the occurrence of visible emissions is greater than 5 percent of the observation period (i.e., 90 seconds per 30 minute period), the owner or operator shall either document and adjust the operation of the facility and demonstrate within 24 hours that the sum of the occurrence of visible emissions is equal to or less than 5 percent during a 30 minute observation (i.e., 90 seconds) or conduct a new Method 9 of appendix A-4 of this part performance test using the procedures in paragraph (a) of this section within 45 calendar days according to the requirements in §60.46d(d)(7).

(ii) If no visible emissions are observed for 10 operating days during which an opacity standard is applicable, observations can be reduced to once every 7 operating days during which an opacity standard is applicable. If any visible emissions are observed, daily observations shall be resumed.

(3) If the maximum 6-minute opacity is less than 10 percent during the most recent Method 9 of appendix A-4 of this part performance test, the owner or operator may, as an alternative to performing subsequent Method 9 of appendix A-4 performance tests, elect to perform subsequent monitoring using a digital opacity compliance system according to a site-specific monitoring plan approved by the Administrator. The observations shall be similar, but not necessarily identical, to the requirements in paragraph (a)(2) of this section. For reference purposes in preparing the monitoring plan, see OAQPS "Determination of Visible Emission Opacity from Stationary Sources Using Computer-Based Photographic Analysis Systems." This document is available from the U.S. Environmental Protection Agency (U.S. EPA); Office of Air Quality and Planning Standards; Sector Policies and Programs Division; Measurement Policy Group (D243-02), Research Triangle Park, NC 27711. This document is also available on the Technology Transfer Network (TTN) under Emission Measurement Center Preliminary Methods.

(b) Except as provided under paragraphs (g), (h), and (i) of this section, the owner or operator of an affected facility subject to a NO_X standard under §60.44b shall comply with either paragraphs (b)(1) or (b)(2) of this section.

(1) Install, calibrate, maintain, and operate CEMS for measuring NO_X and O_2 (or CO_2) emissions discharged to the atmosphere, and shall record the output of the system; or

(2) If the owner or operator has installed a NO_X emission rate CEMS to meet the requirements of part 75 of this chapter and is continuing to meet the ongoing requirements of part 75 of this chapter, that CEMS may be used to meet the requirements of this section, except that the owner or operator shall also meet the requirements of §60.49b. Data reported to meet the requirements of §60.49b shall not include data substituted using the missing data procedures in subpart D of part 75 of this chapter, nor shall the data have been bias adjusted according to the procedures of part 75 of this chapter.

(c) The CEMS required under paragraph (b) of this section shall be operated and data recorded during all periods of operation of the affected facility except for CEMS breakdowns and repairs. Data is recorded during calibration checks, and zero and span adjustments.

(d) The 1-hour average NO_X emission rates measured by the continuous NO_X monitor required by paragraph (b) of this section and required under §60.13(h) shall be expressed in ng/J or lb/MMBtu heat input and shall be used to calculate the average emission rates under §60.44b. The 1-hour averages shall be calculated using the data points required under §60.13(h)(2).

§60.48b

(e) The procedures under §60.13 shall be followed for installation, evaluation, and operation of the continuous monitoring systems.

(1) For affected facilities combusting coal, wood or municipal-type solid waste, the span value for a COMS shall be between 60 and 80 percent.

(2) For affected facilities combusting coal, oil, or natural gas, the span value for NO_X is determined using one of the following procedures:

(i) Except as provided under paragraph (e)(2)(ii) of this section, NO_X span values shall be determined as follows:

Fuel	Span values for NO _X (ppm)
Natural gas	500.
Oil	500.
Coal	1,000.
Mixtures	500 (x + y) + 1,000□

Where:

x = Fraction of total heat input derived from natural gas;

 \mathbf{y} = Fraction of total heat input derived from oil; and

z = Fraction of total heat input derived from coal.

(ii) As an alternative to meeting the requirements of paragraph (e)(2)(i) of this section, the owner or operator of an affected facility may elect to use the NO_X span values determined according to section 2.1.2 in appendix A to part 75 of this chapter.

(3) All span values computed under paragraph (e)(2)(i) of this section for combusting mixtures of regulated fuels are rounded to the nearest 500 ppm. Span values computed under paragraph (e)(2)(ii) of this section shall be rounded off according to section 2.1.2 in appendix A to part 75 of this chapter.

(f) When NO_X emission data are not obtained because of CEMS breakdowns, repairs, calibration checks and zero and span adjustments, emission data will be obtained by using standby monitoring systems, Method 7 of appendix A of this part, Method 7A of appendix A of this part, or other approved reference methods to provide emission data for a minimum of 75 percent of the operating hours in each steam generating unit operating day, in at least 22 out of 30 successive steam generating unit operating days.

40 CFR Ch. I (7–1–23 Edition)

(g) The owner or operator of an affected facility that has a heat input capacity of 73 MW (250 MMBtu/hr) or less, and that has an annual capacity factor for residual oil having a nitrogen content of 0.30 weight percent or less, natural gas, distillate oil, gasified coal, or any mixture of these fuels, greater than 10 percent (0.10) shall:

(1) Comply with the provisions of paragraphs (b), (c), (d), (e)(2), (e)(3), and (f) of this section; or

(2) Monitor steam generating unit operating conditions and predict NO_X emission rates as specified in a plan submitted pursuant to §60.49b(c).

(h) The owner or operator of a duct burner, as described in 60.41b, that is subject to the NO_X standards in 60.44b(a)(4), 60.44b(e), or 60.44b(1) is not required to install or operate a continuous emissions monitoring system to measure NO_X emissions.

(i) The owner or operator of an affected facility described in $\S60.44b(j)$ or $\S60.44b(k)$ is not required to install or operate a CEMS for measuring NO_X emissions.

(j) The owner or operator of an affected facility that meets the conditions in either paragraph (j)(1), (2), (3), (4), (5), (6), or (7) of this section is not required to install or operate a COMS if:

(1) The affected facility uses a PM CEMS to monitor PM emissions; or

(2) The affected facility burns only liquid (excluding residual oil) or gaseous fuels with potential SO₂ emissions rates of 26 ng/J (0.060 lb/MMBtu) or less and does not use a post-combustion technology to reduce SO₂ or PM emissions. The owner or operator must maintain fuel records of the sulfur content of the fuels burned, as described under 60.49b(r); or

(3) The affected facility burns coke oven gas alone or in combination with fuels meeting the criteria in paragraph (j)(2) of this section and does not use a post-combustion technology to reduce SO_2 or PM emissions; or

(4) The affected facility does not use post-combustion technology (except a wet scrubber) for reducing PM, SO_2 , or carbon monoxide (CO) emissions, burns only gaseous fuels or fuel oils that contain less than or equal to 0.30 weight percent sulfur, and is operated such

that emissions of CO to the atmosphere from the affected facility are maintained at levels less than or equal to 0.15 lb/MMBtu on a steam generating unit operating day average basis. Owners and operators of affected facilities electing to comply with this paragraph must demonstrate compliance according to the procedures specified in paragraphs (j)(4)(i) through (iv) of this section; or

(i) You must monitor CO emissions using a CEMS according to the procedures specified in paragraphs (j)(4)(i)(A) through (D) of this section.

(A) The CO CEMS must be installed, certified, maintained, and operated according to the provisions in §60.58b(i)(3) of subpart Eb of this part.

(B) Each 1-hour CO emissions average is calculated using the data points generated by the CO CEMS expressed in parts per million by volume corrected to 3 percent oxygen (dry basis).

(C) At a minimum, valid 1-hour CO emissions averages must be obtained for at least 90 percent of the operating hours on a 30-day rolling average basis. The 1-hour averages are calculated using the data points required in $\S60.13(h)(2)$.

(D) Quarterly accuracy determinations and daily calibration drift tests for the CO CEMS must be performed in accordance with procedure 1 in appendix F of this part.

(ii) You must calculate the 1-hour average CO emissions levels for each steam generating unit operating day by multiplying the average hourly CO output concentration measured by the CO CEMS times the corresponding average hourly flue gas flow rate and divided by the corresponding average hourly heat input to the affected source. The 24-hour average CO emission level is determined by calculating the arithmetic average of the hourly CO emission levels computed for each steam generating unit operating day.

(iii) You must evaluate the preceding 24-hour average CO emission level each steam generating unit operating day excluding periods of affected source startup, shutdown, or malfunction. If the 24-hour average CO emission level is greater than 0.15 lb/MMBtu, you must initiate investigation of the relevant equipment and control systems within 24 hours of the first discovery of the high emission incident and, take the appropriate corrective action as soon as practicable to adjust control settings or repair equipment to reduce the 24-hour average CO emission level to 0.15 lb/MMBtu or less.

(iv) You must record the CO measurements and calculations performed according to paragraph (j)(4) of this section and any corrective actions taken. The record of corrective action taken must include the date and time during which the 24-hour average CO emission level was greater than 0.15 lb/MMBtu, and the date, time, and description of the corrective action.

(5) The affected facility uses a bag leak detection system to monitor the performance of a fabric filter (baghouse) according to the most current requirements in section §60.48Da of this part; or

(6) The affected facility uses an ESP as the primary PM control device and uses an ESP predictive model to monitor the performance of the ESP developed in accordance and operated according to the most current requirements in section §60.48Da of this part; or

(7) The affected facility burns only gaseous fuels or fuel oils that contain less than or equal to 0.30 weight percent sulfur and operates according to a written site-specific monitoring plan approved by the permitting authority. This monitoring plan must include procedures and criteria for establishing and monitoring specific parameters for the affected facility indicative of compliance with the opacity standard.

(k) Owners or operators complying with the PM emission limit by using a PM CEMS must calibrate, maintain, operate, and record the output of the system for PM emissions discharged to the atmosphere as specified in §60.46b(j). The CEMS specified in paragraph §60.46b(j) shall be operated and data recorded during all periods of operation of the affected facility except for CEMS breakdowns and repairs. Data is recorded during calibration checks, and zero and span adjustments.

(1) An owner or operator of an affected facility that is subject to an opacity standard under §60.43b(f) is not required to operate a COMS provided

that the unit burns only gaseous fuels and/or liquid fuels (excluding residue oil) with a potential SO_2 emissions rate no greater than 26 ng/J (0.060 lb/MMBtu), and the unit operates according to a written site-specific monitoring plan approved by the permitting authority is not required to operate a COMS. This monitoring plan must include procedures and criteria for establishing and monitoring specific parameters for the affected facility indicative of compliance with the opacity standard. For testing performed as part of this site-specific monitoring plan, the permitting authority may require as an alternative to the notification and reporting requirements specified in §§ 60.8 and 60.11 that the owner or operator submit any deviations with the excess emissions report required under §60.49b(h).

[72 FR 32742, June 13, 2007, as amended at 74
FR 5087, Jan. 28, 2009; 76 FR 3523, Jan. 20, 2011; 77 FR 9460, Feb. 16, 2012]

§60.49b Reporting and recordkeeping requirements.

(a) The owner or operator of each affected facility shall submit notification of the date of initial startup, as provided by §60.7. This notification shall include:

(1) The design heat input capacity of the affected facility and identification of the fuels to be combusted in the affected facility;

(2) If applicable, a copy of any federally enforceable requirement that limits the annual capacity factor for any fuel or mixture of fuels under $\S60.42b(d)(1)$, $\S60.43b(a)(2)$, (a)(3)(iii), (c)(2)(ii), (d)(2)(iii), $\S60.44b(c)$, (d), (e), (i), (j), (k), $\S60.45b(d)$, (g), $\S60.46b(h)$, or $\S60.48b(i)$;

(3) The annual capacity factor at which the owner or operator anticipates operating the facility based on all fuels fired and based on each individual fuel fired; and

(4) Notification that an emerging technology will be used for controlling emissions of SO₂. The Administrator will examine the description of the emerging technology and will determine whether the technology qualifies as an emerging technology. In making this determination, the Administrator may require the owner or operator of the affected facility to submit additional information concerning the control device. The affected facility is subject to the provisions of §60.42b(a) unless and until this determination is made by the Administrator.

(b) The owner or operator of each affected facility subject to the SO₂, PM, and/or NO_X emission limits under §§60.42b, 60.43b, and 60.44b shall submit to the Administrator the performance test data from the initial performance test and the performance evaluation of the CEMS using the applicable performance specifications in appendix B of this part. The owner or operator of each affected facility described in §60.44b(j) or §60.44b(k) shall submit to the Administrator the maximum heat input capacity data from the demonstration of the maximum heat input capacity of the affected facility.

(c) The owner or operator of each affected facility subject to the NO_X standard in §60.44b who seeks to demonstrate compliance with those standards through the monitoring of steam generating unit operating conditions in the provisions of §60.48b(g)(2) shall submit to the Administrator for approval a plan that identifies the operating conditions to be monitored in (60.48b(g)(2)) and the records to be maintained in §60.49b(g). This plan shall be submitted to the Administrator for approval within 360 days of the initial startup of the affected facility. An affected facility burning coke oven gas alone or in combination with other gaseous fuels or distillate oil shall submit this plan to the Administrator for approval within 360 days of the initial startup of the affected facility or by November 30, 2009, whichever date comes later. If the plan is approved, the owner or operator shall maintain records of predicted nitrogen oxide emission rates and the monitored operating conditions, including steam generating unit load, identified in the plan. The plan shall:

(1) Identify the specific operating conditions to be monitored and the relationship between these operating conditions and NO_x emission rates (*i.e.*, ng/J or lbs/MMBtu heat input). Steam generating unit operating conditions include, but are not limited to, the degree of staged combustion (*i.e.*, the
ratio of primary air to secondary and/ or tertiary air) and the level of excess air (*i.e.*, flue gas O_2 level);

(2) Include the data and information that the owner or operator used to identify the relationship between NO_X emission rates and these operating conditions; and

(3) Identify how these operating conditions, including steam generating unit load, will be monitored under §60.48b(g) on an hourly basis by the owner or operator during the period of operation of the affected facility; the quality assurance procedures or practices that will be employed to ensure that the data generated by monitoring these operating conditions will be representative and accurate; and the type and format of the records of these operating conditions, including steam generating unit load, that will be maintained by the owner or operator under §60.49b(g).

(d) Except as provided in paragraph (d)(2) of this section, the owner or operator of an affected facility shall record and maintain records as specified in paragraph (d)(1) of this section.

(1) The owner or operator of an affected facility shall record and maintain records of the amounts of each fuel combusted during each day and calculate the annual capacity factor individually for coal, distillate oil, residual oil, natural gas, wood, and municipal-type solid waste for the reporting period. The annual capacity factor is determined on a 12-month rolling average basis with a new annual capacity factor calculated at the end of each calendar month.

(2) As an alternative to meeting the requirements of paragraph (d)(1) of this section, the owner or operator of an affected facility that is subject to a federally enforceable permit restricting fuel use to a single fuel such that the facility is not required to continuously monitor any emissions (excluding opacity) or parameters indicative of emissions may elect to record and maintain records of the amount of each fuel combusted during each calendar month.

(e) For an affected facility that combusts residual oil and meets the criteria under 60.46b(e)(4), 60.44b(j), or (k), the owner or operator shall maintain records of the nitrogen content of the residual oil combusted in the affected facility and calculate the average fuel nitrogen content for the reporting period. The nitrogen content shall be determined using ASTM Method D4629 (incorporated by reference, see §60.17), or fuel suppliers. If residual oil blends are being combusted, fuel nitrogen specifications may be prorated based on the ratio of residual oils of different nitrogen content in the fuel blend.

(f) For an affected facility subject to the opacity standard in §60.43b, the owner or operator shall maintain records of opacity. In addition, an owner or operator that elects to monitor emissions according to the requirements in §60.48b(a) shall maintain records according to the requirements specified in paragraphs (f)(1) through (3) of this section, as applicable to the visible emissions monitoring method used.

(1) For each performance test conducted using Method 9 of appendix A-4 of this part, the owner or operator shall keep the records including the information specified in paragraphs (f)(1)(i) through (iii) of this section.

(i) Dates and time intervals of all opacity observation periods;

(ii) Name, affiliation, and copy of current visible emission reading certification for each visible emission observer participating in the performance test; and

(iii) Copies of all visible emission observer opacity field data sheets;

(2) For each performance test conducted using Method 22 of appendix A– 4 of this part, the owner or operator shall keep the records including the information specified in paragraphs (f)(2)(i) through (iv) of this section.

(i) Dates and time intervals of all visible emissions observation periods;

(ii) Name and affiliation for each visible emission observer participating in the performance test;

(iii) Copies of all visible emission observer opacity field data sheets; and

(iv) Documentation of any adjustments made and the time the adjustments were completed to the affected facility operation by the owner or operator to demonstrate compliance with the applicable monitoring requirements.

(3) For each digital opacity compliance system, the owner or operator shall maintain records and submit reports according to the requirements specified in the site-specific monitoring plan approved by the Administrator.

(g) Except as provided under paragraph (p) of this section, the owner or operator of an affected facility subject to the NO_X standards under 60.44bshall maintain records of the following information for each steam generating unit operating day:

(1) Calendar date;

(2) The average hourly NO_X emission rates (expressed as NO_2) (ng/J or lb/ MMBtu heat input) measured or predicted;

(3) The 30-day average NO_X emission rates (ng/J or lb/MMBtu heat input) calculated at the end of each steam generating unit operating day from the measured or predicted hourly nitrogen oxide emission rates for the preceding 30 steam generating unit operating days;

(4) Identification of the steam generating unit operating days when the calculated 30-day average NO_x emission rates are in excess of the NO_x emissions standards under §60.44b, with the reasons for such excess emissions as well as a description of corrective actions taken;

(5) Identification of the steam generating unit operating days for which pollutant data have not been obtained, including reasons for not obtaining sufficient data and a description of corrective actions taken;

(6) Identification of the times when emission data have been excluded from the calculation of average emission rates and the reasons for excluding data;

(7) Identification of "F" factor used for calculations, method of determination, and type of fuel combusted;

(8) Identification of the times when the pollutant concentration exceeded full span of the CEMS;

(9) Description of any modifications to the CEMS that could affect the ability of the CEMS to comply with Performance Specification 2 or 3; and 40 CFR Ch. I (7–1–23 Edition)

(10) Results of daily CEMS drift tests and quarterly accuracy assessments as required under appendix F, Procedure 1 of this part.

(h) The owner or operator of any affected facility in any category listed in paragraphs (h)(1) or (2) of this section is required to submit excess emission reports for any excess emissions that occurred during the reporting period.

(1) Any affected facility subject to the opacity standards in 60.43b(f) or to the operating parameter monitoring requirements in 60.13(i)(1).

(2) Any affected facility that is subject to the NO_X standard of §60.44b, and that:

(i) Combusts natural gas, distillate oil, gasified coal, or residual oil with a nitrogen content of 0.3 weight percent or less; or

(ii) Has a heat input capacity of 73 MW (250 MMBtu/hr) or less and is required to monitor NO_X emissions on a continuous basis under §60.48b(g)(1) or steam generating unit operating conditions under §60.48b(g)(2).

(3) For the purpose of §60.43b, excess emissions are defined as all 6-minute periods during which the average opacity exceeds the opacity standards under §60.43b(f).

(4) For purposes of (0.48b(g))(1), excess emissions are defined as any calculated 30-day rolling average NO_X emission rate, as determined under (0.46b(e)), that exceeds the applicable emission limits in (0.44b).

(i) The owner or operator of any affected facility subject to the continuous monitoring requirements for NO_X under §60.48(b) shall submit reports containing the information recorded under paragraph (g) of this section.

(j) The owner or operator of any affected facility subject to the SO_2 standards under §60.42b shall submit reports.

(k) For each affected facility subject to the compliance and performance testing requirements of 60.45b and the reporting requirement in paragraph (j) of this section, the following information shall be reported to the Administrator:

(1) Calendar dates covered in the reporting period;

(2) Each 30-day average SO_2 emission rate (ng/J or lb/MMBtu heat input) measured during the reporting period,

ending with the last 30-day period; reasons for noncompliance with the emission standards; and a description of corrective actions taken; For an exceedance due to maintenance of the SO_2 control system covered in paragraph 60.45b(a), the report shall identify the days on which the maintenance was performed and a description of the maintenance;

(3) Each 30-day average percent reduction in SO_2 emissions calculated during the reporting period, ending with the last 30-day period; reasons for noncompliance with the emission standards; and a description of corrective actions taken;

(4) Identification of the steam generating unit operating days that coal or oil was combusted and for which SO_2 or diluent (O_2 or CO_2) data have not been obtained by an approved method for at least 75 percent of the operating hours in the steam generating unit operating day; justification for not obtaining sufficient data; and description of corrective action taken;

(5) Identification of the times when emissions data have been excluded from the calculation of average emission rates; justification for excluding data; and description of corrective action taken if data have been excluded for periods other than those during which coal or oil were not combusted in the steam generating unit;

(6) Identification of "F" factor used for calculations, method of determination, and type of fuel combusted;

(7) Identification of times when hourly averages have been obtained based on manual sampling methods;

(8) Identification of the times when the pollutant concentration exceeded full span of the CEMS;

(9) Description of any modifications to the CEMS that could affect the ability of the CEMS to comply with Performance Specification 2 or 3;

(10) Results of daily CEMS drift tests and quarterly accuracy assessments as required under appendix F, Procedure 1 of this part; and

(11) The annual capacity factor of each fired as provided under paragraph (d) of this section.

(1) For each affected facility subject to the compliance and performance testing requirements of 60.45b(d) and

the reporting requirements of paragraph (j) of this section, the following information shall be reported to the Administrator:

(1) Calendar dates when the facility was in operation during the reporting period;

(2) The 24-hour average SO_2 emission rate measured for each steam generating unit operating day during the reporting period that coal or oil was combusted, ending in the last 24-hour period in the quarter; reasons for noncompliance with the emission standards; and a description of corrective actions taken;

(3) Identification of the steam generating unit operating days that coal or oil was combusted for which S0₂ or diluent (O₂ or CO₂) data have not been obtained by an approved method for at least 75 percent of the operating hours; justification for not obtaining sufficient data; and description of corrective action taken;

(4) Identification of the times when emissions data have been excluded from the calculation of average emission rates; justification for excluding data; and description of corrective action taken if data have been excluded for periods other than those during which coal or oil were not combusted in the steam generating unit;

(5) Identification of "F" factor used for calculations, method of determination, and type of fuel combusted;

(6) Identification of times when hourly averages have been obtained based on manual sampling methods;

(7) Identification of the times when the pollutant concentration exceeded full span of the CEMS;

(8) Description of any modifications to the CEMS that could affect the ability of the CEMS to comply with Performance Specification 2 or 3; and

(9) Results of daily CEMS drift tests and quarterly accuracy assessments as required under Procedure 1 of appendix F 1 of this part. If the owner or operator elects to implement the alternative data assessment procedures described in $\$\ 60.47b(e)(4)(i)$ through (e)(4)(iii), each data assessment report shall include a summary of the results of all of the RATAs, linearity checks, CGAs, and calibration error or drift assessments required by $\S 60.47b(e)(4)(i)$ through (e)(4)(ii).

(m) For each affected facility subject to the SO_2 standards in §60.42(b) for which the minimum amount of data required in §60.47b(c) were not obtained during the reporting period, the following information is reported to the Administrator in addition to that required under paragraph (k) of this section:

(1) The number of hourly averages available for outlet emission rates and inlet emission rates;

(2) The standard deviation of hourly averages for outlet emission rates and inlet emission rates, as determined in Method 19 of appendix A of this part, section 7;

(3) The lower confidence limit for the mean outlet emission rate and the upper confidence limit for the mean inlet emission rate, as calculated in Method 19 of appendix A of this part, section 7; and

(4) The ratio of the lower confidence limit for the mean outlet emission rate and the allowable emission rate, as determined in Method 19 of appendix A of this part, section 7.

(n) If a percent removal efficiency by fuel pretreatment (*i.e.*, $\ensuremath{\%R_{\rm f}}$) is used to determine the overall percent reduction (*i.e.*, $\ensuremath{\%R_{\rm o}}$) under §60.45b, the owner or operator of the affected facility shall submit a signed statement with the report.

(1) Indicating what removal efficiency by fuel pretreatment (*i.e.*, $\Re R_f$) was credited during the reporting period;

(2) Listing the quantity, heat content, and date each pre-treated fuel shipment was received during the reporting period, the name and location of the fuel pretreatment facility; and the total quantity and total heat content of all fuels received at the affected facility during the reporting period;

(3) Documenting the transport of the fuel from the fuel pretreatment facility to the steam generating unit; and

(4) Including a signed statement from the owner or operator of the fuel pretreatment facility certifying that the percent removal efficiency achieved by fuel pretreatment was determined in accordance with the provi40 CFR Ch. I (7–1–23 Edition)

sions of Method 19 of appendix A of this part and listing the heat content and sulfur content of each fuel before and after fuel pretreatment.

(o) All records required under this section shall be maintained by the owner or operator of the affected facility for a period of 2 years following the date of such record.

(p) The owner or operator of an affected facility described in §60.44b(j) or (k) shall maintain records of the following information for each steam generating unit operating day:

(1) Calendar date;

(2) The number of hours of operation; and

(3) A record of the hourly steam load. (q) The owner or operator of an affected facility described in §60.44b(j) or §60.44b(k) shall submit to the Administrator a report containing:

(1) The annual capacity factor over the previous 12 months;

(2) The average fuel nitrogen content during the reporting period, if residual oil was fired; and

(3) If the affected facility meets the criteria described in $\S60.44b(j)$, the results of any NO_x emission tests required during the reporting period, the hours of operation during the reporting period, and the hours of operation since the last NO_x emission test.

(r) The owner or operator of an affected facility who elects to use the fuel based compliance alternatives in $\S60.42b$ or $\S60.43b$ shall either:

(1) The owner or operator of an affected facility who elects to demonstrate that the affected facility combusts only very low sulfur oil, natural gas, wood, a mixture of these fuels, or any of these fuels (or a mixture of these fuels) in combination with other fuels that are known to contain an insignificant amount of sulfur in §60.42b(j) or §60.42b(k) shall obtain and maintain at the affected facility fuel receipts (such as a current, valid purchase contract, tariff sheet, or transportation contract) from the fuel supplier that certify that the oil meets the definition of distillate oil and gaseous fuel meets the definition of natural gas as defined in §60.41b and the applicable sulfur limit. For the purposes of this section, the distillate oil need not meet the fuel nitrogen content specification

in the definition of distillate oil. Reports shall be submitted to the Administrator certifying that only very low sulfur oil meeting this definition, natural gas, wood, and/or other fuels that are known to contain insignificant amounts of sulfur were combusted in the affected facility during the reporting period; or

(2) The owner or operator of an affected facility who elects to demonstrate compliance based on fuel analysis in §60.42b or §60.43b shall develop and submit a site-specific fuel analysis plan to the Administrator for review and approval no later than 60 days before the date you intend to demonstrate compliance. Each fuel analysis plan shall include a minimum initial requirement of weekly testing and each analysis report shall contain, at a minimum, the following information:

(i) The potential sulfur emissions rate of the representative fuel mixture in ng/J heat input;

(ii) The method used to determine the potential sulfur emissions rate of each constituent of the mixture. For distillate oil and natural gas a fuel receipt or tariff sheet is acceptable;

(iii) The ratio of different fuels in the mixture; and

(iv) The owner or operator can petition the Administrator to approve monthly or quarterly sampling in place of weekly sampling.

(s) Facility specific NO_X standard for Cytec Industries Fortier Plant's C.AOG incinerator located in Westwego, Louisiana:

(1) Definitions.

Oxidation zone is defined as the portion of the C.AOG incinerator that extends from the inlet of the oxidizing zone combustion air to the outlet gas stack.

Reducing zone is defined as the portion of the C.AOG incinerator that extends from the burner section to the inlet of the oxidizing zone combustion air.

Total inlet air is defined as the total amount of air introduced into the C.AOG incinerator for combustion of natural gas and chemical by-product waste and is equal to the sum of the air flow into the reducing zone and the air flow into the oxidation zone. (2) Standard for nitrogen oxides. (i) When fossil fuel alone is combusted, the NO_X emission limit for fossil fuel in §60.44b(a) applies.

(ii) When natural gas and chemical by-product waste are simultaneously combusted, the NO_X emission limit is 289 ng/J (0.67 lb/MMBtu) and a maximum of 81 percent of the total inlet air provided for combustion shall be provided to the reducing zone of the C.AOG incinerator.

(3) Emission monitoring. (i) The percent of total inlet air provided to the reducing zone shall be determined at least every 15 minutes by measuring the air flow of all the air entering the reducing zone and the air flow of all the air entering the oxidation zone, and compliance with the percentage of total inlet air that is provided to the reducing zone shall be determined on a 3-hour average basis.

(ii) The NO_X emission limit shall be determined by the compliance and performance test methods and procedures for NO_X in §60.46b(i).

(iii) The monitoring of the NO_X emission limit shall be performed in accordance with 60.48b.

(4) Reporting and recordkeeping requirements. (i) The owner or operator of the C.AOG incinerator shall submit a report on any excursions from the limits required by paragraph (a)(2) of this section to the Administrator with the quarterly report required by paragraph (i) of this section.

(ii) The owner or operator of the C.AOG incinerator shall keep records of the monitoring required by paragraph (a)(3) of this section for a period of 2 years following the date of such record.

(iii) The owner of operator of the C.AOG incinerator shall perform all the applicable reporting and record-keeping requirements of this section.

(t) Facility-specific NO_X standard for Rohm and Haas Kentucky Incorporated's Boiler No. 100 located in Louisville, Kentucky:

(1) *Definitions*.

Air ratio control damper is defined as the part of the low NO_X burner that is adjusted to control the split of total combustion air delivered to the reducing and oxidation portions of the combustion flame. *Flue gas recirculation line* is defined as the part of Boiler No. 100 that recirculates a portion of the boiler flue gas back into the combustion air.

(2) Standard for nitrogen oxides. (i) When fossil fuel alone is combusted, the NO_X emission limit for fossil fuel in §60.44b(a) applies.

(ii) When fossil fuel and chemical byproduct waste are simultaneously combusted, the NO_x emission limit is 473 ng/J (1.1 lb/MMBtu), and the air ratio control damper tee handle shall be at a minimum of 5 inches (12.7 centimeters) out of the boiler, and the flue gas recirculation line shall be operated at a minimum of 10 percent open as indicated by its valve opening position indicator.

(3) Emission monitoring for nitrogen oxides. (i) The air ratio control damper tee handle setting and the flue gas recirculation line valve opening position indicator setting shall be recorded during each 8-hour operating shift.

(ii) The NO_X emission limit shall be determined by the compliance and performance test methods and procedures for NO_X in §60.46b.

(iii) The monitoring of the NO_X emission limit shall be performed in accordance with §60.48b.

(4) Reporting and recordkeeping requirements. (i) The owner or operator of Boiler No. 100 shall submit a report on any excursions from the limits required by paragraph (b)(2) of this section to the Administrator with the quarterly report required by §60.49b(i).

(ii) The owner or operator of Boiler No. 100 shall keep records of the monitoring required by paragraph (b)(3) of this section for a period of 2 years following the date of such record.

(iii) The owner of operator of Boiler No. 100 shall perform all the applicable reporting and recordkeeping requirements of §60.49b.

(u) Site-specific standard for Merck & Co., Inc.'s Stonewall Plant in Elkton, Virginia. (1) This paragraph (u) applies only to the pharmaceutical manufacturing facility, commonly referred to as the Stonewall Plant, located at Route 340 South, in Elkton, Virginia ("site") and only to the natural gasfired boilers installed as part of the powerhouse conversion required pursuant to 40 CFR 52.2454(g). The require40 CFR Ch. I (7–1–23 Edition)

ments of this paragraph shall apply, and the requirements of §§60.40b through 60.49b(t) shall not apply, to the natural gas-fired boilers installed pursuant to 40 CFR 52.2454(g).

(i) The site shall equip the natural gas-fired boilers with low NO_X technology.

(ii) The site shall install, calibrate, maintain, and operate a continuous monitoring and recording system for measuring NO_X emissions discharged to the atmosphere and opacity using a continuous emissions monitoring system or a predictive emissions monitoring system.

(iii) Within 180 days of the completion of the powerhouse conversion, as required by 40 CFR 52.2454, the site shall perform a performance test to quantify criteria pollutant emissions.

(2) [Reserved]

(v) The owner or operator of an affected facility may submit electronic quarterly reports for SO₂ and/or NO_X and/or opacity in lieu of submitting the written reports required under paragraphs (h), (i), (j), (k) or (l) of this section. The format of each quarterly electronic report shall be coordinated with the permitting authority. The electronic report(s) shall be submitted no later than 30 days after the end of the calendar quarter and shall be accompanied by a certification statement from the owner or operator, indicating whether compliance with the applicable emission standards and minimum data requirements of this subpart was achieved during the reporting period. Before submitting reports in the electronic format, the owner or operator shall coordinate with the permitting authority to obtain their agreement to submit reports in this alternative format.

(w) The reporting period for the reports required under this subpart is each 6 month period. All reports shall be submitted to the Administrator and shall be postmarked by the 30th day following the end of the reporting period.

(x) Facility-specific NO_X standard for Weyerhaeuser Company's No. 2 Power Boiler located in New Bern, North Carolina:

(1) Standard for nitrogen oxides. (i) When fossil fuel alone is combusted,

the $NO_{\rm X}$ emission limit for fossil fuel in §60.44b(a) applies.

(ii) When fossil fuel and chemical byproduct waste are simultaneously combusted, the NO_X emission limit is 215 ng/J (0.5 lb/MMBtu).

(2) Emission monitoring for nitrogen oxides. (i) The NO_X emissions shall be determined by the compliance and performance test methods and procedures for NO_X in 60.46b.

(ii) The monitoring of the NO_X emissions shall be performed in accordance with §60.48b.

(3) Reporting and recordkeeping requirements. (i) The owner or operator of the No. 2 Power Boiler shall submit a report on any excursions from the limits required by paragraph (x)(2) of this section to the Administrator with the quarterly report required by §60.49b(i).

(ii) The owner or operator of the No. 2 Power Boiler shall keep records of the monitoring required by paragraph (x)(3) of this section for a period of 2 years following the date of such record.

(iii) The owner or operator of the No. 2 Power Boiler shall perform all the applicable reporting and recordkeeping requirements of §60.49b.

(y) Facility-specific NO_X standard for INEOS USA's AOGI located in Lima, Ohio:

(1) Standard for NO_X . (i) When fossil fuel alone is combusted, the NO_X emission limit for fossil fuel in §60.44b(a) applies.

(ii) When fossil fuel and chemical byproduct/waste are simultaneously combusted, the NO_X emission limit is 645 ng/J (1.5 lb/MMBtu).

(2) Emission monitoring for NO_X . (i) The NO_X emissions shall be determined by the compliance and performance test methods and procedures for NO_X in §60.46b.

(ii) The monitoring of the NO_X emissions shall be performed in accordance with §60.48b.

(3) Reporting and recordkeeping requirements. (i) The owner or operator of the AOGI shall submit a report on any excursions from the limits required by paragraph (y)(2) of this section to the Administrator with the quarterly report required by paragraph (i) of this section.

(ii) The owner or operator of the AOGI shall keep records of the moni-

toring required by paragraph (y)(3) of this section for a period of 2 years following the date of such record.

(iii) The owner or operator of the AOGI shall perform all the applicable reporting and recordkeeping requirements of this section.

[72 FR 32742, June 13, 2007, as amended at 74 FR 5089, Jan. 28, 2009; 77 FR 9461, Feb. 16, 2012]

Subpart Dc—Standards of Performance for Small Industrial-Commercial-Institutional Steam Generating Units

SOURCE: 72 FR 32759, June 13, 2007, unless otherwise noted.

§60.40c Applicability and delegation of authority.

(a) Except as provided in paragraphs (d), (e), (f), and (g) of this section, the affected facility to which this subpart applies is each steam generating unit for which construction, modification, or reconstruction is commenced after June 9, 1989 and that has a maximum design heat input capacity of 29 megawatts (MW) (100 million British thermal units per hour (MMBtu/h)) or less, but greater than or equal to 2.9 MW (10 MMBtu/h).

(b) In delegating implementation and enforcement authority to a State under section 111(c) of the Clean Air Act, 60.48c(a)(4) shall be retained by the Administrator and not transferred to a State.

(c) Steam generating units that meet the applicability requirements in paragraph (a) of this section are not subject to the sulfur dioxide (SO_2) or particulate matter (PM) emission limits, performance testing requirements, or monitoring requirements under this subpart (§60.42c, §60.43c, §60.44c, §60.45c, §60.46c, or §60.47c) during periods of combustion research, as defined in §60.41c.

(d) Any temporary change to an existing steam generating unit for the purpose of conducting combustion research is not considered a modification under §60.14.

(e) Affected facilities (*i.e.* heat recovery steam generators and fuel heaters) that are associated with stationary

Appendix T

AUTHENTICATED U.S. GOVERNMENT INFORMATION

(µg-min/m³)(4.68 \times 10⁻⁹ pound minutes per cubic foot (lb-min/ft³)) of air within the limits of 10 to 60 minutes, accumulated during any 2 consecutive weeks, in any area in which an adverse effect to public health could occur.

(b) If combustion products from the firing of beryllium propellant are collected in a closed tank, emissions from such tank shall not exceed 2.0 g/hr (0.0044 lb/hr) and a maximum of 10 g/ day (0.022 lb/day).

[38 FR 8826, Apr. 6, 1973, as amended at 65 FR 62151, Oct. 17, 2000; 79 FR 11275, Feb. 27, 2014]

§61.43 Emission testing—rocket firing or propellant disposal.

(a) Ambient air concentrations shall be measured during and after firing of a rocket motor or propellant disposal and in such a manner that the effect of these emissions can be compared with the standard. Such sampling techniques shall be approved by the Administrator.

(b) All samples shall be analyzed and results shall be calculated within 30 days after samples are taken and before any subsequent rocket motor firing or propellant disposal at the given site. All results shall be reported to the Administrator by a registered letter dispatched before the close of the next business day following determination of such results.

(c) Records of air sampling test results and other data needed to determine integrated intermittent concentrations shall be retained at the source and made available, for inspection by the Administrator, for a minimum of 2 years.

(d) The Administrator shall be notified at least 30 days prior to an air sampling test, so that he may at his option observe the test.

§61.44 Stack sampling.

(a) Sources subject to §61.42(b) shall be continuously sampled, during release of combustion products from the tank, according to Method 104 of appendix B to this part. Method 103 of appendix B to this part is approved by the Administrator as an alternative method for sources subject to §61.42(b).

(b) All samples shall be analyzed, and beryllium emissions shall be determined within 30 days after samples are

40 CFR Ch. I (7–1–23 Edition)

taken and before any subsequent rocket motor firing or propellant disposal at the given site. All determinations shall be reported to the Administrator by a registered letter dispatched before the close of the next business day following such determinations.

(c) Records of emission test results and other data needed to determine total emissions shall be retained at the source and made available, for inspection by the Administrator, for a minimum of 2 years.

(d) The Administrator shall be notified at least 30 days prior to an emission test, so that he may at his option observe the test.

[38 FR 8826, Apr. 6, 1973, as amended at 50 FR 46294, Nov. 7, 1985]

Subpart E—National Emission Standard for Mercury

§61.50 Applicability.

The provisions of this subpart are applicable to those stationary sources which process mercury ore to recover mercury, use mercury chlor-alkali cells to produce chlorine gas and alkali metal hydroxide, and incinerate or dry wastewater treatment plant sludge.

[40 FR 48302, Oct. 14, 1975]

§61.51 Definitions.

Terms used in this subpart are defined in the act, in subpart A of this part, or in this section as follows:

(a) *Mercury* means the element mercury, excluding any associated elements, and includes mercury in particulates, vapors, aerosols, and compounds.

(b) *Mercury ore* means a mineral mined specifically for its mercury content.

(c) *Mercury ore processing facility* means a facility processing mercury ore to obtain mercury.

(d) *Condenser stack gases* mean the gaseous effluent evolved from the stack of processes utilizing heat to extract mercury metal from mercury ore.

(e) *Mercury chlor-alkali cell* means a device which is basically composed of an electrolyzer section and a denuder (decomposer) section and utilizes mercury to produce chlorine gas, hydrogen gas, and alkali metal hydroxide.

(f) Mercury chlor-alkali electrolyzer means an electrolytic device which is part of a mercury chlor-alkali cell and utilizes a flowing mercury cathode to produce chlorine gas and alkali metal amalgam.

(g) *Denuder* means a horizontal or vertical container which is part of a mercury chlor-alkali cell and in which water and alkali metal amalgam are converted to alkali metal hydroxide, mercury, and hydrogen gas in a shortcircuited, electrolytic reaction.

(h) *Hydrogen gas stream* means a hydrogen stream formed in the chlor-al-kali cell denuder.

(i) *End box* means a container(s) located on one or both ends of a mercury chlor-alkali electrolyzer which serves as a connection between the electrolyzer and denuder for rich and stripped amalgam.

(j) *End box ventilation system* means a ventilation system which collects mercury emissions from the end-boxes, the mercury pump sumps, and their water collection systems.

(k) *Cell room* means a structure(s) housing one or more mercury electrolytic chlor-alkali cells.

(1) *Sludge* means sludge produced by a treatment plant that processes municipal or industrial waste waters.

(m) Sludge dryer means a device used to reduce the moisture content of sludge by heating to temperatures above 65 °C (ca. 150 °F) directly with combustion gases.

[38 FR 8826, Apr. 6, 1973, as amended at 40 FR 48302, Oct. 14, 1975]

§61.52 Emission standard.

(a) Emissions to the atmosphere from mercury ore processing facilities and mercury cell chlor-alkali plants shall not exceed 2.3 kg (5.1 lb) of mercury per 24-hour period.

(b) Emissions to the atmosphere from sludge incineration plants, sludge drying plants, or a combination of these that process wastewater treatment plant sludges shall not exceed 3.2 kg (7.1 lb) of mercury per 24-hour period.

[40 FR 48302, Oct. 14, 1975, as amended at 65 FR 62151, Oct. 17, 2000]

§61.53 Stack sampling.

(a) Mercury ore processing facility. (1) Unless a waiver of emission testing is obtained under §61.13, each owner or operator processing mercury ore shall test emissions from the source according to Method 101 of appendix B to this part. The emission test shall be performed—

(i) Within 90 days of the effective date in the case of an existing source or a new source which has an initial start-up date preceding the effective date; or

(ii) Within 90 days of startup in the case of a new source which did not have an initial startup date preceding the effective date.

(2) The Administrator shall be notified at least 30 days prior to an emission test, so that he may at his option observe the test.

(3) Samples shall be taken over such a period or periods as are necessary to accurately determine the maximum emissions which will occur in a 24-hour period. No changes in the operation shall be made, which would potentially increase emissions above that determined by the most recent source test, until the new emission level has been estimated by calculation and the results reported to the Administrator.

(4) All samples shall be analyzed and mercury emissions shall be determined within 30 days after the stack test. Each determination shall be reported to the Administrator by a registered letter dispatched within 15 calendar days following the date such determination is completed.

(5) Records of emission test results and other data needed to determine total emissions shall be retained at the source and made available, for inspection by the Administrator, for a minimum of 2 years.

(b) Mercury chlor-alkali plant—hydrogen and end-box ventilation gas streams. (1) Unless a waiver of emission testing is obtained under §61.13, each owner or operator employing mercury chlor-alkali cell(s) shall test emissions from hydrogen streams according to Method 102 and from end-box ventilation gas streams according to Method 101 of appendix B to this part. The emission test shall be performed—

40 CFR Ch. I (7–1–23 Edition)

(i) Within 90 days of the effective date in the case of an existing source or a new source which has an initial startup date preceding the effective date; or

(ii) Within 90 days of startup in the case of a new source which did not have an initial startup date preceding the effective date.

(2) The Administrator shall be notified at least 30 days prior to an emission test, so that he may at his option observe the test.

(3) Samples shall be taken over such a period or periods as are necessary to accurately determine the maximum emissions which will occur in a 24-hour period. No changes in the operation shall be made, which would potentially increase emissions above that determined by the most recent source test, until the new emission has been estimated by calculation and the results reported to the Administrator.

(4) All samples shall be analyzed and mercury emissions shall be determined within 30 days after the stack test. Each determination shall be reported to the Administrator by a registered letter dispatched within 15 calendar days following the date such determination is completed.

(5) Records of emission test results and other data needed to determine total emissions shall be retained at the source and made available, for inspection by the Administrator, for a minimum of 2 years.

(c) Mercury chlor-alkali plants—cell room ventilation system. (1) Stationary sources using mercury chlor-alkali cells may test cell room emissions in accordance with paragraph (c)(2) of this section or demonstrate compliance with paragraph (c)(4) of this section and assume ventilation emissions of 1.3 kg/day (2.9 lb/day) of mercury.

(2) Unless a waiver of emission testing is obtained under §61.13, each owner or operator shall pass all cell room air in force gas streams through stacks suitable for testing and shall test emissions from the source according to Method 101 in appendix B to this part. The emission test shall be performed—

(i) Within 90 days of the effective date in the case of an existing source or a new source which has an initial startup date preceding the effective date; or

(ii) Within 90 days of startup in the case of a new source which did not have an initial startup date preceding the effective date.

(3) The Administrator shall be notified at least 30 days prior to an emission test, so that he may at his option observe the test.

(4) An owner or operator may carry out approved design, maintenance, and housekeeping practices. A list of approved practices is provided in appendix A of "Review of National Emission Standards for Mercury," EPA-450/3-84-014a, December 1984. Copies are available from EPA's Central Docket Section, Docket item number A-84-41, III-B-1.

(d) Sludge incineration and drying plants. (1) Unless a waiver of emission testing is obtained under 61.13, each owner or operator of a source subject to the standard in 61.52(b) shall test emissions from that source. Such tests shall be conducted in accordance with the procedures set forth either in paragraph (d) of this section or in 61.54.

(2) Method 101A in appendix B or Method 29 in appendix A to part 60 shall be used to test emissions as follows:

(i) The test shall be performed by May 28, 2014 in the case of an existing source or a new source which has an initial startup date preceding February 27, 2014.

(ii) The test shall be performed within 90 days of startup in the case of a new source which did not have an initial startup date preceding February 27, 2014.

(3) The Administrator shall be notified at least 30 days prior to an emission test, so that he may at his option observe the test.

(4) Samples shall be taken over such a period or periods as are necessary to determine accurately the maximum emissions which will occur in a 24-hour period. No changes shall be made in the operation which would potentially increase emissions above the level determined by the most recent stack test, until the new emission level has been estimated by calculation and the results reported to the Administrator.

(5) All samples shall be analyzed and mercury emissions shall be determined within 30 days after the stack test. Each determination shall be reported to the Administrator by a registered letter dispatched within 15 calendar days following the date such determination is completed.

(6) Records of emission test results and other data needed to determine total emissions shall be retained at the source and shall be made available, for inspection by the Administrator, for a minimum of 2 years.

[38 FR 8826, Apr. 6, 1973, as amended at 40 FR 48302, Oct. 14, 1975; 47 FR 24704, June 8, 1982; 50 FR 46294, Nov. 7, 1985; 52 FR 8726, Mar. 19, 1987; 65 FR 62151, Oct. 17, 2000; 79 FR 11275, Feb. 27, 2014]

§61.54 Sludge sampling.

(a) As an alternative means for demonstrating compliance with §61.52(b), an owner or operator may use Method 105 of appendix B and the procedures specified in this section.

(1) A sludge test shall be conducted within 90 days of the effective date of these regulations in the case of an existing source or a new source which has an initial startup date preceding the effective date; or

(2) A sludge test shall be conducted within 90 days of startup in the case of a new source which did not have an initial startup date preceding the effective date.

(b) The Administrator shall be notified at least 30 days prior to a sludge sampling test, so that he may at his option observe the test.

(c) Sludge shall be sampled according to paragraph (c)(1) of this section, sludge charging rate for the plant shall be determined according to paragraph (c)(2) of this section, and the sludge analysis shall be performed according to paragraph (c)(3) of this section.

(1) The sludge shall be sampled according to Method 105—Determination of Mercury in Wastewater Treatment Plant Sewage Sludges. A total of three composite samples shall be obtained within an operating period of 24 hours. When the 24-hour operating period is not continuous, the total sampling period shall not exceed 72 hours after the first grab sample is obtained. Samples shall not be exposed to any condition that may result in mercury contamination or loss.

(2) The maximum 24-hour period sludge incineration or drying rate shall be determined by use of a flow rate measurement device that can measure the mass rate of sludge charged to the incinerator or dryer with an accuracy of ± 5 percent over its operating range. Other methods of measuring sludge mass charging rates may be used if they have received prior approval by the Administrator.

(3) The sampling, handling, preparation, and analysis of sludge samples shall be accomplished according to Method 105 in appendix B of this part.

(d) The mercury emissions shall be determined by use of the following equation.

$$E_{Hg} = \frac{MQ F_{sm(avg)}}{1000}$$

where:

 E_{Hg} = Mercury emissions, g/day.

M = Mercury concentration of sludge on a dry solids basis, $\mu g/g$.

Q = Sludge changing rate, kg/day.

 F_{sm} = Weight fraction of solids in the collected sludge after mixing.

1000 = Conversion factor, kg μ g/g.²

(e) No changes in the operation of a plant shall be made after a sludge test has been conducted which would potentially increase emissions above the level determined by the most recent sludge test, until the new emission level has been estimated by calculation and the results reported to the Administrator.

(f) All sludge samples shall be analyzed for mercury content within 30 days after the sludge sample is collected. Each determination shall be reported to the Administrator by a registered letter dispatched within 15 calendar days following the date such determination is completed.

(g) Records of sludge sampling, charging rate determination and other data needed to determine mercury content of wastewater treatment plant sludges shall be retained at the source and made available, for inspection by §61.55

the Administrator, for a minimum of 2 years.

[40 FR 48303, Oct. 14, 1975, as amended at 49
FR 35770, Sept. 12, 1984; 52 FR 8727, Mar. 19, 1987; 53 FR 36972, Sept. 23, 1988]

§61.55 Monitoring of emissions and operations.

(a) Wastewater treatment plant sludge incineration and drying plants. All the sources for which mercury emissions exceed 1.6 kg (3.5 lb) per 24-hour period, demonstrated either by stack sampling according to $\S61.53$ or sludge sampling according to $\S61.54$, shall monitor mercury emissions at intervals of at least once per year by use of Method 105 of appendix B or the procedures specified in $\S61.53$ (d) (2) and (4). The results of monitoring shall be reported and retained according to $\S61.53(d)$ (5) and (6) or $\S61.54$ (f) and (g).

(b) Mercury cell chlor-alkali plants hydrogen and end-box ventilation gas streams. (1) The owner or operator of each mercury cell chlor-alkali plant shall, within 1 year of the date of publication of these amendments or within 1 year of startup for a plant with initial startup after the date of publication, perform a mercury emission test that demonstrates compliance with the emission limits in §61.52, on the hydrogen stream by Method 102 and on the end-box stream by Method 101 for the purpose of establishing limits for parameters to be monitored.

(2) During tests specified in paragraph (b)(1) of this section, the following control device parameters shall be monitored, except as provided in paragraph (c) of this section, and recorded manually or automatically at least once every 15 minutes:

(i) The exit gas temperature from uncontrolled streams;

(ii) The outlet temperature of the gas stream for the final (i.e., the farthest downstream) cooling system when no control devices other than coolers and demisters are used;

(iii) The outlet temperature of the gas stream from the final cooling system when the cooling system is followed by a molecular sieve or carbon adsorber;

(iv) Outlet concentration of available chlorine, pH, liquid flow rate, and inlet

gas temperature of chlorinated brine scrubbers and hypochlorite scrubbers;

(v) The liquid flow rate and exit gas temperature for water scrubbers;

(vi) The inlet gas temperature of carbon adsorption systems; and

(vii) The temperature during the heating phase of the regeneration cycle for carbon adsorbers or molecular sieves.

(3) The recorded parameters in paragraphs (b)(2)(i) through (b)(2)(vi) of this section shall be averaged over the test period (a minimum of 6 hours) to provide an average number. The highest temperature reading that is measured in paragraph (b)(2)(vii) of this section is to be identified as the reference temperature for use in paragraph (b)(6)(ii) of this section.

(4)(i) Immediately following completion of the emission tests specified in paragraph (b)(1) of this section, the owner or operator of a mercury cell chlor-alkali plant shall monitor and record manually or automatically at least once per hour the same parameters specified in paragraphs (b)(2)(i) through (b)(2)(vi) of this section.

(ii) Immediately following completion of the emission tests specified in paragraph (b)(1) of this section, the owner or operator shall monitor and record manually or automatically, during each heating phase of the regeneration cycle, the temperature specified in paragraph (b)(2)(vii) of this section.

(5) Monitoring devices used in accordance with paragraphs (b)(2) and (b)(4) of this section shall be certified by their manufacturer to be accurate to within 10 percent, and shall be operated, maintained, and calibrated according to the manufacturer's instructions. Records of the certifications and calibrations shall be retained at the chlor-alkali plant and made available for inspection by the Administrator as follows: Certification, for as long as the device is used for this purpose; calibration for a minimum of 2 years.

(6)(i) When the hourly value of a parameter monitored in accordance with paragraph (b)(4)(i) of this section exceeds, or in the case of liquid flow rate and available chlorine falls below the

value of that same parameter determined in paragraph (b)(2) of this section for 24 consecutive hours, the Administrator is to be notified within the next 10 days.

(ii) When the maximum hourly value of the temperature measured in accordance with paragraph (b)(4)(ii) of this section is below the reference temperature recorded according to paragraph (b)(3) of this section for three consecutive regeneration cycles, the Administrator is to be notified within the next 10 days.

(7) Semiannual reports shall be submitted to the Administrator indicating the time and date on which the hourly value of each parameter monitored according to paragraphs (b)(4)(i) and (b)(4)(ii) of this section fell outside the value of that same parameter determined under paragraph (b)(3) of this section; and corrective action taken, and the time and date of the corrective action. Parameter excursions will be considered unacceptable operation and maintenance of the emission control system. In addition, while compliance with the emission limits is determined primarily by conducting a performance test according to the procedures in §61.53(b), reports of parameter excursions may be used as evidence in judging the duration of a violation that is determined by a performance test.

(8) Semiannual reports required in paragraph (b)(7) of this section shall be submitted to the Administrator on September 15 and March 15 of each year. The first semiannual report is to be submitted following the first full 6 month reporting period. The semiannual report due on September 15 (March 15) shall include all excursions monitored through August 31 (February 28) of the same calendar year.

(c) As an alternative to the monitoring, recordkeeping, and reporting requirements in paragraphs (b)(2) through (8) of this section, an owner or operator may develop and submit for the Administrator's review and approval a plant-specific monitoring plan. To be approved, such a plan must ensure not only compliance with the emission limits of $\S61.52(a)$ but also proper operation and maintenance of emissions control systems. Any sitespecific monitoring plan submitted must, at a minimum, include the following:

(1) Identification of the critical parameter or parameters for the hydrogen stream and for the end-box ventilation stream that are to be monitored and an explanation of why the critical parameter(s) selected is the best indicator of proper control system performance and of mercury emission rates.

(2) Identification of the maximum or minimum value of each parameter (e.g., degrees temperature, concentration of mercury) that is not to be exceeded. The level(s) is to be directly correlated to the results of a performance test, conducted no more than 180 days prior to submittal of the plan, when the facility was in compliance with the emission limits of $\S61.52(a)$.

(3) Designation of the frequency for recording the parameter measurements, with justification if the frequency is less than hourly. A longer recording frequency must be justified on the basis of the amount of time that could elapse during periods of process or control system upsets before the emission limits would be exceeded, and consideration is to be given to the time that would be necessary to repair the failure.

(4) Designation of the immediate actions to be taken in the event of an excursion beyond the value of the parameter established in paragraph (c)(2) of this section.

(5) Provisions for reporting, semiannually, parameter excursions and the corrective actions taken, and provisions for reporting within 10 days any significant excursion.

(6) Identification of the accuracy of the monitoring device(s) or of the readings obtained.

(7) Recordkeeping requirements for certifications and calibrations.

(d) Mercury cell chlor-alkali plants cell room ventilation system. (1) Stationary sources determining cell room emissions in accordance with §61.53(c)(4) shall maintain daily records of all leaks or spills of mercury. The records shall indicate the amount, location, time, and date the leaks or spills occurred, identify the cause of the leak or spill, state the immediate steps taken to minimize mercury emissions and steps taken to prevent future occurrences, and provide the time and date on which corrective steps were taken.

(2) The results of monitoring shall be recorded, retained at the source, and made available for inspection by the Administrator for a minimum of 2 years.

[52 FR 8727, Mar. 19, 1987, as amended at 65 FR 62151, Oct. 17, 2000]

§61.56 Delegation of authority.

(a) In delegating implementation and enforcement authority to a State under section 112(d) of the Act, the authorities contained in paragraph (b) of this section shall be retained by the Administrator and not transferred to a State.

(b) Authorities which will not be delegated to States: Sections 61.53(c)(4)and 61.55(d). The authorities not delegated to States listed are in addition to the authorities in the General Provisions, subpart A of 40 CFR part 61, that will not be delegated to States (§§ 61.04(b), 61.12(d)(1), and 61.13(h)(1)(ii)).

[52 FR 8728, Mar. 19, 1987]

Subpart F—National Emission Standard for Vinyl Chloride

SOURCE: 41 FR 46564, Oct. 21, 1976, unless otherwise noted.

§61.60 Applicability.

(a) This subpart applies to plants which produce:

(1) Ethylene dichloride by reaction of oxygen and hydrogen chloride with ethylene,

(2) Vinyl chloride by any process, and/or

(3) One or more polymers containing any fraction of polymerized vinyl chloride.

(b) This subpart does not apply to equipment used in research and development if the reactor used to polymerize the vinyl chloride processed in the equipment has a capacity of no more than 0.19 m^3 (50 gal).

(c) Sections of this subpart other than \$\$61.61; 61.64 (a)(1), (b), (c), and (d); 61.67; 61.68; 61.69; 61.70; and 61.71 do not

40 CFR Ch. I (7–1–23 Edition)

apply to equipment used in research and development if the reactor used to polymerize the vinyl chloride processed in the equipment has a capacity of greater than $0.19 \text{ m}^3(50 \text{ gal})$ and no more than $4.17 \text{ m}^3(1100 \text{ gal})$.

[41 FR 46564, Oct. 21, 1976, as amended at 42 FR 29006, June 7, 1977; 53 FR 36972, Sept. 23, 1988; 57 FR 60999, Dec. 23, 1992]

§61.61 Definitions.

Terms used in this subpart are defined in the Act, in subpart A of this part, or in this section as follows:

(a) *Ethylene dichloride plant* includes any plant which produces ethylene dichloride by reaction of oxygen and hydrogen chloride with ethylene.

(b) *Vinyl chloride plant* includes any plant which produces vinyl chloride by any process.

(c) *Polyvinyl chloride (PVC) plant* includes any plant where vinyl chloride alone or in combination with other materials is polymerized.

(d) Slip gauge means a gauge which has a probe that moves through the gas/liquid interface in a storage or transfer vessel and indicates the level of vinyl chloride in the vessel by the physical state of the material the gauge discharges.

(e) *Type of resin* means the broad classification of resin referring to the basic manufacturing process for producing that resin, including, but not limited to, the suspension, dispersion, latex, bulk, and solution processes.

(f) *Grade of resin* means the subdivision of resin classification which describes it as a unique resin, i.e., the most exact description of a resin with no further subdivision.

(g) *Dispersion resin* means a resin manufactured in such a way as to form fluid dispersions when dispersed in a plasticizer or plasticizer/diluent mixtures.

(h) *Latex resin* means a resin which is produced by a polymerization process which initiates from free radical catalyst sites and is sold undried.

(i) *Bulk resin* means a resin which is produced by a polymerization process in which no water is used.

(j) *Inprocess wastewater* means any water which, during manufacturing or processing, comes into direct contact